ETSI TS 124 501 V17.16.0 (2024-10)



5G; Non-Access-Stratum (NAS) protocol for 5G System (5GS); Stage 3 (3GPP TS 24.501 version 17.16.0 Release 17)



Reference RTS/TSGC-0124501vha0

Keywords

5G

ETSI

650 Route des Lucioles F-06921 Sophia Antipolis Cedex - FRANCE

Tel.: +33 4 92 94 42 00 Fax: +33 4 93 65 47 16

Siret N° 348 623 562 00017 - APE 7112B Association à but non lucratif enregistrée à la Sous-Préfecture de Grasse (06) N° w061004871

Important notice

The present document can be downloaded from the ETSI <u>Search & Browse Standards application.</u>

The present document may be made available in electronic versions and/or in print. The content of any electronic and/or print versions of the present document shall not be modified without the prior written authorization of ETSI. In case of any existing or perceived difference in contents between such versions and/or in print, the prevailing version of an ETSI deliverable is the one made publicly available in PDF format on ETSI deliver.

Users should be aware that the present document may be revised or have its status changed, this information is available in the <u>Milestones listing</u>.

If you find errors in the present document, please send your comments to the relevant service listed under <u>Committee Support Staff</u>.

If you find a security vulnerability in the present document, please report it through our <u>Coordinated Vulnerability Disclosure (CVD)</u> program.

Notice of disclaimer & limitation of liability

The information provided in the present deliverable is directed solely to professionals who have the appropriate degree of experience to understand and interpret its content in accordance with generally accepted engineering or other professional standard and applicable regulations.

No recommendation as to products and services or vendors is made or should be implied.

No representation or warranty is made that this deliverable is technically accurate or sufficient or conforms to any law and/or governmental rule and/or regulation and further, no representation or warranty is made of merchantability or fitness for any particular purpose or against infringement of intellectual property rights.

In no event shall ETSI be held liable for loss of profits or any other incidental or consequential damages.

Any software contained in this deliverable is provided "AS IS" with no warranties, express or implied, including but not limited to, the warranties of merchantability, fitness for a particular purpose and non-infringement of intellectual property rights and ETSI shall not be held liable in any event for any damages whatsoever (including, without limitation, damages for loss of profits, business interruption, loss of information, or any other pecuniary loss) arising out of or related to the use of or inability to use the software.

Copyright Notification

No part may be reproduced or utilized in any form or by any means, electronic or mechanical, including photocopying and microfilm except as authorized by written permission of ETSI. The content of the PDF version shall not be modified without the written authorization of ETSI.

The copyright and the foregoing restriction extend to reproduction in all media.

© ETSI 2024. All rights reserved.

Intellectual Property Rights

Essential patents

IPRs essential or potentially essential to normative deliverables may have been declared to ETSI. The declarations pertaining to these essential IPRs, if any, are publicly available for **ETSI members and non-members**, and can be found in ETSI SR 000 314: "Intellectual Property Rights (IPRs); Essential, or potentially Essential, IPRs notified to ETSI in respect of ETSI standards", which is available from the ETSI Secretariat. Latest updates are available on the ETSI Web server (https://ipr.etsi.org/).

Pursuant to the ETSI Directives including the ETSI IPR Policy, no investigation regarding the essentiality of IPRs, including IPR searches, has been carried out by ETSI. No guarantee can be given as to the existence of other IPRs not referenced in ETSI SR 000 314 (or the updates on the ETSI Web server) which are, or may be, or may become, essential to the present document.

Trademarks

The present document may include trademarks and/or tradenames which are asserted and/or registered by their owners. ETSI claims no ownership of these except for any which are indicated as being the property of ETSI, and conveys no right to use or reproduce any trademark and/or tradename. Mention of those trademarks in the present document does not constitute an endorsement by ETSI of products, services or organizations associated with those trademarks.

DECTTM, **PLUGTESTSTM**, **UMTSTM** and the ETSI logo are trademarks of ETSI registered for the benefit of its Members. **3GPPTM** and **LTETM** are trademarks of ETSI registered for the benefit of its Members and of the 3GPP Organizational Partners. **oneM2MTM** logo is a trademark of ETSI registered for the benefit of its Members and of the oneM2M Partners. **GSM**[®] and the GSM logo are trademarks registered and owned by the GSM Association.

Legal Notice

This Technical Specification (TS) has been produced by ETSI 3rd Generation Partnership Project (3GPP).

The present document may refer to technical specifications or reports using their 3GPP identities. These shall be interpreted as being references to the corresponding ETSI deliverables.

The cross reference between 3GPP and ETSI identities can be found under https://webapp.etsi.org/key/queryform.asp.

Modal verbs terminology

In the present document "shall", "shall not", "should", "should not", "may", "need not", "will", "will not", "can" and "cannot" are to be interpreted as described in clause 3.2 of the <u>ETSI Drafting Rules</u> (Verbal forms for the expression of provisions).

"must" and "must not" are NOT allowed in ETSI deliverables except when used in direct citation.

Contents

Intelle	ectual Property Rights	2
Legal	Notice	2
Moda	l verbs terminology	2
Forew	vord	24
1	Scope	25
2	References	25
3	Definitions and abbreviations	29
3.1	Definitions	29
3.2	Abbreviations	40
4	General	43
4.1	Overview	43
4.2	Coordination between the protocols for 5GS mobility management and 5GS session management	43
4.3	UE domain selection	44
4.3.1	UE's usage setting	44
4.3.2	Domain selection for UE originating sessions / calls	44
4.3.3	Change of UE's usage setting	46
4.3.4	Change or determination of IMS voice availability	46
4.4	NAS security	47
4.4.1	General	47
4.4.2	Handling of 5G NAS security contexts	47
4.4.2.1	General	47
4.4.2.2	Establishment of a mapped 5G NAS security context during inter-system change from 51 mode	50
4 4 9 9	to NI mode in SGMM-CONNECTED mode	50
4.4.2.3	Establishment of a SG NAS security context during inter system change from N1 mode to S1 mode.	
4.4.2.4	in 5GMM CONNECTED mode	52
1125	Fstablishment of secure exchange of NAS messages	52
4426	Change of security keys	<i>52</i> 54
4.4.3	Handling of NAS COUNT and NAS sequence number	
4.4.3.1	General	
4.4.3.2	Replay protection	56
4.4.3.3	Integrity protection and verification	56
4.4.3.4	Ciphering and deciphering	56
4.4.3.5	NAS COUNT wrap around	57
4.4.4	Integrity protection of NAS signalling messages	57
4.4.4.1	General	57
4.4.4.2	Integrity checking of NAS signalling messages in the UE	58
4.4.4.3	Integrity checking of NAS signalling messages in the AMF	58
4.4.5	Ciphering of NAS signalling messages	60
4.4.6	Protection of initial NAS signalling messages	60
4.4./	Protection of NAS IEs	62
4.5	Unified access control	63
4.5.1	Determination of the access identities and access category associated with a request for access for	05
4.J.2	UEs not operating in SNPN access mode	65
4.5 2A	Determination of the access identities and access category associated with a request for access for	05
	UEs operating in SNPN access mode	70
4.5.3	Operator-defined access categories	75
4.5.4	Access control and checking.	77
4.5.4.1	Access control and checking in 5GMM-IDLE mode and in 5GMM-IDLE mode with suspend	
	indication	77
4.5.4.2	Access control and checking in 5GMM-CONNECTED mode and in 5GMM-CONNECTED mode with RRC inactive indication	79
4.5.5	Exception handling and avoiding double barring	81

4.5.6	Mapping between access categories/access identities and RRC establishment cause	
4.6	Network slicing	86
4.6.1	General	86
4.6.2	Mobility management aspects	
4.6.2.1	General	
4.6.2.2	NSSAI storage	
4.6.2.3	Provision of NSSAI to lower layers in 5GMM-IDLE mode	94
4.6.2.4	Network slice-specific authentication and authorization	96
4.6.2.5	Mobility management based network slice admission control	97
4.6.2.6	Provision of NSAG information to lower layers	98
4.6.3	Session management aspects	99
4.6.3.0	General	99
4.6.3.1	Session management based network slice admission control	100
4.6.3.2	Support of network slice admission control and interworking with EPC	100
4.6.3.3	Session management based network slice data rate limitation control	100
4.7	NAS over non-3GPP access	101
4.7.1	General	101
4.7.2	5GS mobility management aspects	101
4.7.2.1	General	101
4.7.2.2	Establishment cause for non-3GPP access	102
4.7.3	5GS session management aspects	103
4.7.4	Limited service state over non-3GPP access	103
4.7.5	NAS signalling using trusted WLAN access network	104
4.8	Interworking with E-UTRAN connected to EPC	104
4.8.1	General	104
4.8.2	Single-registration mode	105
4.8.2.1	General	105
4.8.2.2	Single-registration mode with N26 interface	105
4.8.2.3	Single-registration mode without N26 interface	105
4.8.2.3.1	Interworking between NG-RAN and E-UTRAN	105
4.8.2.3.2	Interworking between TNGF or N3IWF connected to 5GCN and E-UTRAN	107
4.8.3	Dual-registration mode	108
4.8.4	Core Network selection for UEs not using CIoT 5GS optimizations	109
4.8.4A	Core Network selection and redirection for UEs using CIoT optimizations	109
4.8.4A.1	Core network selection	109
4.8.4A.2	Redirection of the UE by the core network	110
4.9	Disabling and re-enabling of UE's N1 mode capability	110
4.9.1	General	110
4.9.2	Disabling and re-enabling of UE's N1 mode capability for 3GPP access	110
4.9.3	Disabling and re-enabling of UE's N1 mode capability for non-3GPP access	113
4.10	Interworking with ePDG connected to EPC	113
4.11	UE configuration parameter updates	113
4.12	Access traffic steering, switching and splitting (ATSSS)	114
4.13	Support of NAS signalling using wireline access network	114
4.14	Non-public network	115
4.14.1	General	
4.14.2	Stand-alone non-public network	115
4.14.3	Public network integrated non-public network (PNI-NPN)	117
4.15	Time synchronization and time sensitive communication	
4.15.1	General	
4.15.2	Void	
4.15.2.1	Void	
4.15.2.2	Void	
4.15.2.3	Void	
4.15.3	Time synchronization	
4.15.4	User plane node management	
4.16	UE radio capability signalling optimisation	
4.17	5GS mobility management in NB-N1 mode	
4.18	5GS session management in NB-N1 mode	120
4.19	5GS mobility management in WB-N1 mode for IoT	
4.20	5GS session management in WB-N1 mode for IoT	
4.21	Authentication and Key Management for Applications (AKMA)	

4.22	Uncrewed aerial vehicle identification, authentication, and authorization	
4.22.1	General	
4.22.2	Authentication and authorization of UAV	
4.22.3	Authorization of C2 communication	
4.22.4	Void	
4.23	NAS OVer Non-Terrestrial Network	124 124
4.23.1	List of "PI MNs not allowed to operate at the present UE location"	124 124
4.23.2	5GS mobility management via a satellite NG-RAN cell	124 125
4 23 4	5GS session management via a satellite NG-RAN cell	125
4.23.5	Handling multiple tracking area codes from the lower layers	
4.24	Minimization of service interruption	
4.25	Support of MUSIM features	
5 Ele	ementary procedures for 5GS mobility management	128
5.1	Overview	128
5.1.1	General	
5.1.2	Types of 5GMM procedures	
5.1.3	5GMM sublayer states	
5.1.3.1	General	
5.1.3.2	5GMM sublayer states	
5.1.3.2.1	5GMM sublayer states in the UE	
5.1.3.2.1.1	General	
5.1.3.2.1.2	Main states	130
5.1.3.2.1.2	.1 5GMM-NULL	130
5.1.3.2.1.2	.2 5GMM-DEREGISTERED	130
5.1.3.2.1.2	5GMM-REGISTERED-INITIATED	130
5.1.3.2.1.2	.4 5GMM-REGISTERED	130
5.1.3.2.1.2	5GMM-DEREGISTERED-INITIATED	130
5.1.3.2.1.2	.6 5GMM-SERVICE-REQUEST-INITIATED	131
5.1.3.2.1.3	Substates of state 5GMM-DEREGISTERED	
5.1.3.2.1.3	.1 General	
5.1.3.2.1.3	.2 5GMM-DEREGISTERED.NORMAL-SERVICE	
5.1.3.2.1.3	.3 5GMM-DEREGISTERED.LIMITED-SERVICE	
5.1.3.2.1.3	5.4 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION	
5.1.3.2.1.3	5 SUMM-DEREGISTERED NO SUDI	
5.1.3.2.1.3	50 SUMM-DEREGISTERED NO CELL AVAILADLE	
5.1.3.2.1.3 5.1.2.2.1.3	5.7 JUMINI-DEREGISTERED.NU-CELL-AVAILABLE	
5.1.3.2.1.3	.0 SCMM DEDECISTEDED INITIAL DECISTDATION NEEDED	132
5.1.3.2.1.3	Substates of state 5CMM REGISTERED	132
5.1.3.2.1.4	1 General	132
5.1.3.2.1.7	2 5GMM_REGISTERED NORMAL_SERVICE	132
513214	.3 5GMM-REGISTERED NON-ALLOWED-SERVICE	132
513214	4 5GMM-REGISTERED ATTEMPTING-REGISTRATION-UPDATE	132
5.1.3.2.1.4	5 5GMM-REGISTERED LIMITED-SERVICE	
5.1.3.2.1.4	.6 5GMM-REGISTERED.PLMN-SEARCH	
5.1.3.2.1.4	.7 5GMM-REGISTERED.NO-CELL-AVAILABLE	
5.1.3.2.1.4	.8 5GMM-REGISTERED.UPDATE-NEEDED	
5.1.3.2.2	5GS update status in the UE	
5.1.3.2.3	5GMM sublayer states in the network side	134
5.1.3.2.3.1	General	134
5.1.3.2.3.2	5GMM-DEREGISTERED	135
5.1.3.2.3.3	5GMM-COMMON-PROCEDURE-INITIATED	135
5.1.3.2.3.4	5GMM-REGISTERED	135
5.1.3.2.3.5	5GMM-DEREGISTERED-INITIATED	
5.1.4	Coordination between 5GMM and EMM	135
5.1.4.1	General	
5.1.4.2	Coordination between 5GMM for 3GPP access and EMM with N26 interface	
5.1.4.3	Coordination between 5GMM for 3GPP access and EMM without N26 interface	136
5.1.5	Coordination between 5GMM and GMM	136
5.2	Behaviour of the UE in state 5GMM-DEREGISTERED and state 5GMM-REGISTERED	136

5.2.1	General	
5.2.2	UE behaviour in state 5GMM-DEREGISTERED	
5.2.2.1	General	136
5.2.2.2	Primary substate selection	137
5.2.2.2.1	Selection of the substate after power on	
5.2.2.3	Detailed description of UE behaviour in state 5GMM-DEREGISTERED	
5.2.2.3.1	NORMAL-SERVICE	
5.2.2.3.2	LIMITED-SERVICE	
5.2.2.3.3	ATTEMPTING-REGISTRATION	
5.2.2.3.4	PLMN-SEARCH	139
5.2.2.3.5	NO-SUPI	139
5.2.2.3.6	NO-CELL-AVAILABLE	139
5.2.2.3.7	eCALL-INACTIVE	139
5.2.2.3.8	INITIAL-REGISTRATION-NEEDED	139
5.2.2.4	Substate when back to state 5GMM-DEREGISTERED from another 5GMM state	139
5.2.3	UE behaviour in state 5GMM-REGISTERED	140
5.2.3.1	General	140
5.2.3.2	Detailed description of UE behaviour in state 5GMM-REGISTERED	140
5.2.3.2.1	NORMAL-SERVICE	140
5.2.3.2.2	NON-ALLOWED-SERVICE	140
5.2.3.2.3	ATTEMPTING-REGISTRATION-UPDATE	141
5.2.3.2.4	LIMITED-SERVICE	142
5.2.3.2.5	PLMN-SEARCH	142
5.2.3.2.6	NO-CELL-AVAILABLE	142
5.2.3.2.7	UPDATE-NEEDED	
5.3	General on elementary 5GMM procedures	
5.3.1	5GMM modes and N1 NAS signalling connection	
5.3.1.1	Establishment of the N1 NAS signalling connection	
5.3.1.2	Re-establishment of the N1 NAS signalling connection	
5.3.1.3	Release of the N1 NAS signalling connection	
5.3.1.4	5GMM-CONNECTED mode with RRC inactive indication	
5.3.1.5	Suspend and resume of the N1 NAS signalling connection	
5.3.2	Permanent identifiers	
5.5.5	I emporary identities	
5.3.4	Registration areas	
5.5.5	Service area restrictions	
5.3.5.1	General	
5.5.5.2	SGPP access service area restrictions.	
5.5.5.5	Whether access service area restrictions	
5.3.0	Handling of the periodic registration undete timer and mobile reachable timer	105
5.2.9	Handling of timer T2502	104
5.2.0	Handling of NAS level menogement congestion control	100
5 3 10	Handling of DNN based congestion control	100
5 3 11	Handling of S NSSAI based congestion control	107
5312	Handling of local emergency numbers	107
5 3 12 A	Handling of local emergency numbers received via 3GPP access and non-3GPP access	169
5 3 12A 1	General	169
5 3 12A 2	Receiving a REGISTRATION ACCEPT message via non-3GPP access	169
5 3 13	Lists of 5GS forbidden tracking areas	170
5 3 13A	Forbidden PLMN lists	171
5.3.14	List of equivalent PLMNs	
5.3.15	Transmission failure abnormal case in the UE	
5.3.16	Extended DRX cycle for UEs in 5GMM-IDLE and 5GMM-CONNECTED mode with RRC	inactive
-	indication	
5.3.17	Service Gap Control	
5.3.18	Restriction on use of enhanced coverage	
5.3.19	Handling of congestion control for transport of user data via the control plane	
5.3.19a	Specific requirements for UE configured to use timer T3245	
5.3.19a.1	UE not operating in SNPN access operation mode	175
5.3.19a.2	UE operating in SNPN access operation mode	176
5.3.20	Specific requirements for UE when receiving non-integrity protected reject messages	176

5.3.20.1	General	
5.3.20.2	Requirements for UE in a PLMN.	
5.3.20.3	Requirements for UE in an SNPN	
5.3.21	CIoT 5GS optimizations	
5.3.22	Interaction between MICO mode with active time and extended idle mode DRX cycle	
5.3.23	Forbidden wireline access area	
5.3.24	WUS assistance	
5.3.25	Paging Early Indication with Paging Subgrouping Assistance	
5.4	5GMM common procedures	
5.4.1	Primary authentication and key agreement procedure	
5.4.1.1	General	
5.4.1.2	EAP based primary authentication and key agreement procedure	
5.4.1.2.1	General	
5.4.1.2.2	EAP-AKA' related procedures	192
5.4.1.2.3	EAP-TLS related procedures	198
5.4.1.2.3A	Procedures related to EAP methods other than EAP-AKA' and EAP-TLS	203
5.4.1.2.3B	Procedures related to EAP methods used for primary authentication of an N5GC device.	207
5.4.1.2.4	EAP message reliable transport procedure	208
5.4.1.2.4.3	EAP message reliable transport procedure accepted by the UE	209
5.4.1.2.5	EAP result message transport procedure	211
5.4.1.3	5G AKA based primary authentication and key agreement procedure	212
5.4.1.3.1	General	212
5.4.1.3.2	Authentication initiation by the network	
5.4.1.3.3	Authentication response by the UE	213
5.4.1.3.4	Authentication completion by the network	214
5.4.1.3.5	Authentication not accepted by the network	
5.4.1.3.6	Authentication not accepted by the UE	217
5.4.1.3.7	Abnormal cases	
5.4.2	Security mode control procedure	
5.4.2.1	General	
5.4.2.2	NAS security mode control initiation by the network	
5.4.2.3	NAS security mode command accepted by the UE	
5.4.2.4	NAS security mode control completion by the network	
5.4.2.5	NAS security mode command not accepted by the UE	
5.4.2.6	Abnormal cases in the UE	
5.4.2.7	Abnormal cases on the network side	
5.4.5	Identification procedure	
5.4.5.1	General	
5.4.3.2	Identification initiation by the network	
5.4.5.5 5.4.2.4	Identification completion by the network	230
5 4 2 5	Abnormal agess in the UE	230
5.4.3.5	Abnormal cases on the network side	230
5.4.5.0	Conoria UE configuration undata procedure	230
5 4 4 1	General	231
5442	Generic LIE configuration update procedure initiated by the network	231
5443	Generic UE configuration update accented by the UE	235
5444	Generic UE configuration update completion by the network	245
5445	Abnormal cases in the UE	245
5.4.4.6	Abnormal cases on the network side	
5.4.5	NAS transport procedure(s)	
5.4.5.1	General	
5.4.5.2	UE-initiated NAS transport procedure	
5.4.5.2.1	General	
5.4.5.2.2	UE-initiated NAS transport procedure initiation	
5.4.5.2.3	UE-initiated NAS transport of messages accepted by the network	
5.4.5.2.4	UE-initiated NAS transport of messages not accepted by the network	255
5.4.5.2.5	Abnormal cases on the network side	
5.4.5.2.6	Abnormal cases in the UE	
5.4.5.3	Network-initiated NAS transport procedure	
5.4.5.3.1	General	
5.4.5.3.2	Network-initiated NAS transport procedure initiation	

5.4.5.3.3	Network-initiated NAS transport of messages	269
5.4.6	5GMM status procedure	274
5.4.6.1	General	274
5.4.6.2	5GMM status received in the UE	275
5.4.6.3	5GMM status received in the network	275
5.4.7	Network slice-specific authentication and authorization procedure	275
5.4.7.1	General	275
5.4.7.2	Network slice-specific EAP message reliable transport procedure	276
5.4.7.2.1	Network slice-specific EAP message reliable transport procedure initiation	276
5.4.7.2.2	Network slice-specific EAP message reliable transport procedure accepted by the UE	277
5.4.7.2.3	Abnormal cases on the network side	277
5.4.7.2.4	Abnormal cases in the UE	277
5.4.7.3	Network slice-specific EAP result message transport procedure	278
5.4.7.3.1	Network slice-specific EAP result message transport procedure initiation	278
5.5	5GMM specific procedures	278
5.5.1	Registration procedure	278
5.5.1.1	General	278
5.5.1.2	Registration procedure for initial registration	280
5.5.1.2.1	General	280
5.5.1.2.2	Initial registration initiation	280
5.5.1.2.3	5GMM common procedure initiation	288
5.5.1.2.4	Initial registration accepted by the network	288
5.5.1.2.5	Initial registration not accepted by the network	309
5.5.1.2.6	Initial registration for emergency services not accepted by the network	322
5.5.1.2.6A	Initial registration for initiating an emergency PDU session not accepted by the network	323
5.5.1.2.7	Abnormal cases in the UE	324
5.5.1.2.8	Abnormal cases on the network side	326
5.5.1.3	Registration procedure for mobility and periodic registration update	328
5.5.1.3.1	General	328
5.5.1.3.2	Mobility and periodic registration update initiation	328
5.5.1.3.3	5GMM common procedure initiation	341
5.5.1.3.4	Mobility and periodic registration update accepted by the network	342
5.5.1.3.5	Mobility and periodic registration update not accepted by the network	368
5.5.1.3.6	Mobility and periodic registration update for initiating an emergency PDU session not	
	accepted by the network	382
5.5.1.3.6A	Mobility and periodic registration update for an emergency services fallback not accepted by	202
<i>5 5</i> 1 2 7	the network	383
5.5.1.3.7	Abnormal cases in the UE	383
5.5.1.5.8	Adnormal cases on the network side	
5.5.2	Conorol	200
5.5.2.1	UE initiated de registration procedure	202
5.5.2.2	UE-initiated de registration procedure initiation	202
55222	UE-initiated de registration procedure initiation	
55222	UE-initiated de registration procedure completion for 5CS services over 3CDD access	202
55224	UE initiated de registration procedure completion for 5GS services over 5GFT access	202
5522.2.4	UE initiated de registration procedure completion for 5GS services over hoth 3GPP access	
5.5.2.2.5	and non 3GPP access	303
55226	Abnormal cases in the LIF	30/
55227	Abnormal cases in the network side	396
5523	Network-initiated de-registration procedure	396
55231	Network-initiated de-registration procedure initiation	396
55232	Network-initiated de-registration procedure completion by the UE	398
5.5.2.3 3	Network-initiated de-registration procedure completion by the network	409
5.5.2.3.4	Abnormal cases in the UE	
5.5.2.3.5	Abnormal cases in the network side	
5.5.3	eCall inactivity procedure	
5.5.4	Authentication and key agreement procedure for 5G ProSe UE-to-network relay	411
5.5.4.1	General	411
5.5.4.2	ProSe relay transaction identity (PRTI)	412
5.5.4.3	UE-initiated authentication and key agreement procedure initiation	412
5.5.4.4	UE-initiated authentication and key agreement procedure accepted by the network	413

5.5.4.5	UE-initiated authentication and key agreement procedure not accepted by the network	414
5.5.4.6	Abnormal cases in the UE	414
5.5.4.7	Abnormal cases on the network side	414
5.6	5GMM connection management procedures	415
5.6.1	Service request procedure	415
5.6.1.1	General	415
5.6.1.2	Service request procedure initiation	419
5.6.1.2.1	UE is not using 5GS services with control plane CIoT 5GS optimization	419
5.6.1.2.2	UE is using 5GS services with control plane CIoT 5GS optimization	421
5.6.1.3	Common procedure initiation	424
5.6.1.4	Service request procedure accepted by the network	424
5.6.1.4.1	UE is not using 5GS services with control plane CIoT 5GS optimization	424
5.6.1.4.2	UE is using 5GS services with control plane CIoT 5GS optimization	428
5.6.1.5	Service request procedure not accepted by the network	432
5.6.1.6	Service request procedure for initiating an emergency PDU session not accepted by the network .	444
5.6.1.6A	Service request procedure for an emergency services fallback not accepted by the network	445
5.6.1.7	Abnormal cases in the UE	445
5.6.1.8	Abnormal cases on the network side	449
5.6.2	Paging procedure	450
5.6.2.1	General	450
5.6.2.2	Paging for 5GS services	451
5.6.2.2.1	General	451
5.6.2.2.2	Abnormal cases on the network side	453
5.6.2.2.3	Abnormal cases in the UE	453
5.6.3	Notification procedure	454
5.6.3.1	General	454
5.6.3.2	Notification procedure initiation	454
5.6.3.3	Notification procedure completion	456
5.6.3.4	Abnormal cases on the network side	457
5.6.3.5	Abnormal cases on the UE side	457
6 F1	lamentary another for FCC according management	157
0 EI	lementary procedures for 5GS session management	457
C 1		457
6.1	Overview	457
6.1 6.1.1	Overview	457
6.1 6.1.1 6.1.2	Overview General Types of 5GSM procedures	457 457 458
6.1 6.1.1 6.1.2 6.1.3	Overview General Types of 5GSM procedures 5GSM sublayer states	457 457 458 459
6.1 6.1.1 6.1.2 6.1.3 6.1.3.1 6.1.3.1	Overview General Types of 5GSM procedures 5GSM sublayer states General	457 457 457 458 459 459
6.1 6.1.1 6.1.2 6.1.3 6.1.3.1 6.1.3.2 (1.2.2.1	Overview	457 457 457 458 459 459 459
6.1 6.1.1 6.1.2 6.1.3 6.1.3.1 6.1.3.2 6.1.3.2.1 (1.2.2.2)	Overview	457 457 457 458 459 459 459 459 459
6.1 6.1.2 6.1.3 6.1.3.1 6.1.3.2 6.1.3.2.1 6.1.3.2.2	Overview	457 457 457 458 459 459 459 459 459 459
6.1 6.1.2 6.1.3 6.1.3.1 6.1.3.2 6.1.3.2.1 6.1.3.2.2 6.1.3.2.3	Overview	457 457 458 459 459 459 459 459 459 459
$\begin{array}{c} 6.1 \\ 6.1.1 \\ 6.1.2 \\ 6.1.3 \\ 6.1.3.1 \\ 6.1.3.2 \\ 6.1.3.2.1 \\ 6.1.3.2.2 \\ 6.1.3.2.3 \\ 6.1.3.2.4 \\ 6.1.3.2.4 \end{array}$	Overview	457 457 457 459 459 459 459 459 459 459 459
$\begin{array}{c} 6.1 \\ 6.1.1 \\ 6.1.2 \\ 6.1.3 \\ 6.1.3.1 \\ 6.1.3.2 \\ 6.1.3.2.1 \\ 6.1.3.2.2 \\ 6.1.3.2.3 \\ 6.1.3.2.4 \\ 6.1.3.2.5 \\ 6.1.3.2.5 \end{array}$	Overview	457 457 457 459 459 459 459 459 459 459 459 459 459
$\begin{array}{c} 6.1 \\ 6.1.1 \\ 6.1.2 \\ 6.1.3 \\ 6.1.3.1 \\ 6.1.3.2 \\ 6.1.3.2.1 \\ 6.1.3.2.2 \\ 6.1.3.2.3 \\ 6.1.3.2.4 \\ 6.1.3.2.5 \\ 6.1.3.2.6 \end{array}$	Overview General Types of 5GSM procedures 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING	457 457 457 458 459 459 459 459 459 459 459 459 460 460
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.7\\ \end{array}$	Overview	457 457 457 458 459 459 459 459 459 459 459 459 460 460
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.2.8\end{array}$	Overview General Types of 5GSM procedures 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION MODIFICATION PENDING. PDU SESSION MODIFICATION PENDING. PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING	457 457 457 458 459 459 459 459 459 459 459 459 460 460 460
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.2\\ 6.1.3.2\\ \end{array}$	Overview General Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PDU SESSION ACTIVE PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION MOLIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING SGSM sublayer states in the network side Overview	457 457 457 458 459 459 459 459 459 459 459 459 460 460 460 460
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.3\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3.1\end{array}$	Overview	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.3\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3\\ 6.1.3.3.2\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.$	Overview General Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states in the UE General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION INACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING 5GSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION INACTIVE	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.2\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3\\ 6.$	Overview General Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states in the UE General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3\\ 6.1.3.3.2\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3.3\\ 6.1.3\\ 6.$	Overview General Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING 5GSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING 5GSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION ACTIVE PDU SESSION ACTIVE PDU SESSION INACTIVE PENDING PDU SESSION ACTIVE	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 460 460 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.2\\ 6.1.3.3.4\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.3.5\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5\\ 6.1.5$	Overview General Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING 5GSM sublayer states in the network side Overview PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING 5GSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION ACTIVE PDU SESSION ACTIVE	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 460 460 461 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.2.6\\ 6.1.3.2\\ 6.1.2.6\\ 6.1.2.6\\ 6.1.2.6\\ 6.1.2.6\\ 6.1.2.6\\ 6.1.2.2$	Overview General. Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION PENDING. FROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION MODIFICATION PENDING. PDU SESSION INACTIVE PDU SESSION INACTIVE	457 457 457 458 459 459 459 459 459 459 459 459 460 460 460 460 460 460 461 461 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.8\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.3.8\\ 6.1.3.3.6\\ 6.1.3.6\\ 6.1$	Overview General. Types of 5GSM procedures 5GSM sublayer states 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING. SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING	457 457 457 458 459 459 459 459 459 459 459 459 460 460 460 460 460 460 461 461 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4\\ 6.1.4\\ 1.4\\ 1.4\\ 1.4\\ 1.4\\ 1.4\\ 1.4\\ 1.4\\$	Overview General Types of 5GSM procedures 5GSM sublayer states General 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION NACTIVE PENDING PDU SESSION INACTIVE PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION MODIFICATION PENDING POCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM Coordination between 5GSM and ESM	457 457 457 458 459 459 459 459 459 459 459 459 460 460 460 460 460 460 461 461 461 461 461 461
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4.1\\ 6.1.4.1\\ 6.1.4.2\\ \end{array}$	Overview General Types of 5GSM procedures 5GSM sublayer states General SGSM sublayer states in the UE Overview Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PENDING PDU SESSION ACTIVE PDU SESSION NACTIVE PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING SGSM sublayer states in the network side Overview Overview PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM with N26 interface Coordination between 5GSM and ESM with N26 interface	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 462 462
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4.1\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4.1\\ 6.1.4.2\\$	Overview General. Types of 5GSM procedures 5GSM sublayer states General. 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION PENDING. PROCEDURE TRANSACTION PENDING. SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PROCEDURE TRANSACTION PENDING. SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PENDING. PDU SESSION MODIFICATION PENDING POU SESSION RACTIVE PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM with N26 interface Coordination between 5GSM and ESM without N26 interface Coordination between 5GSM and ESM without N26 interface	457 457 457 458 459 459 459 459 459 459 459 459 459 459 460 460 460 461 461 461 461 461 461 461 461 461 462 462 462
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4a\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4a\\ 6.1.5a\\ 6.1.4a\\ 6.1.$	Overview General. Types of 5GSM procedures 5GSM sublayer states General. 5GSM sublayer states General. 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING. PDU SESSION MODIFICATION PENDING. PROCEDURE TRANSACTION INACTIVE. PROCEDURE TRANSACTION PENDING. SGSM sublayer states in the network side Overview PUU SESSION INACTIVE PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. SGSM sublayer states in the network side Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PDU SESSION ACTIVE PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION INACTIVE PDU SESSION MODIFICATION PENDING. PDU SESSION MODIFICATION PENDING POCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING. PROCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM with N26 interface Coordination between 5GSM and ESM without N26 interface Coordination between 5GSM and ESM without N26 interface Coordination between 5GSM and SM <td>457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 461 461 462 462 475 477</td>	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 461 461 462 462 475 477
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.3\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4\\ 6.1.4.1\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4a\\ 6.1.5\\ 6.2\end{array}$	Overview General. Types of 5GSM procedures 5GSM sublayer states General. 5GSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING. PDU SESSION MODIFICATION PENDING. POUSESSION INACTIVE PENDING. PDU SESSION ACTIVE PENDING. PDU SESSION MODIFICATION PENDING. PDU SESSION ACTIVE PENDING. PROCEDURE TRANSACTION INACTIVE. PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION MODIFICATION PENDING. PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING POCEDURE TRANSACTION INACTIVE PROCEDURE TRANSACTION PENDING POCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM Coordination between 5GSM and ESM Coordination between 5GSM and ESM with N26 i	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 461 462 462 477 477 477
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4a\\ 6.1.5\\ 6.2\\ 6.2 \\ 1\end{array}$	Overview General. Types of 5GSM procedures SGSM sublayer states General. General. SGSM sublayer states in the UE Overview PDU SESSION INACTIVE PDU SESSION ACTIVE PENDING. PDU SESSION ACTIVE PENDING PDU SESSION NACTIVE PENDING. PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE. PROCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PROCEDURE TRANSACTION PENDING. PDU SESSION INACTIVE PDU SESSION INACTIVE. PDU SESSION INACTIVE PDU SESSION INACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION NACTIVE PDU SESSION INACTIVE PENDING. PDU SESSION NODIFICATION PENDING PDU SESSION MODIFICATION PENDING PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION INACTIVE PDU SESSION MODIFICATION PENDING PROCEDURE TRANSACTION PENDING POCEDURE TRANSACTION PENDING PROCEDURE TRANSACTION PENDING Coordination between 5GSM and ESM Coordination between 5GSM and ESM Coordination between 5GSM and ESM Coordination between 5GSM and SM Coordination between 5GSM and SM Coordination for interworking with ePDG connected to EPC <td>457 457 457 458 459 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 462 475 477 477</td>	457 457 457 458 459 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 462 475 477 477
$\begin{array}{c} 6.1\\ 6.1.1\\ 6.1.2\\ 6.1.3\\ 6.1.3.1\\ 6.1.3.2\\ 6.1.3.2.1\\ 6.1.3.2.2\\ 6.1.3.2.2\\ 6.1.3.2.3\\ 6.1.3.2.4\\ 6.1.3.2.5\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.6\\ 6.1.3.2.7\\ 6.1.3.2.8\\ 6.1.3.3\\ 6.1.3.3.1\\ 6.1.3.3.2\\ 6.1.3.3.3\\ 6.1.3.3.4\\ 6.1.3.3.5\\ 6.1.3.3.6\\ 6.1.3.3.7\\ 6.1.4\\ 6.1.4.1\\ 6.1.4.2\\ 6.1.4a\\ 6.1.5\\ 6.2\\ 6.2.1\\ 6.2.1\\ 6.2.1\\ 6.2.2\\ 6.2.1\\ 6.2.2\\ 6.2.1\\ 6.2.2\\ $	Overview	457 457 457 458 459 459 459 459 459 459 459 459 459 460 460 460 460 461 461 461 461 461 461 461 461 462 475 477 478 478

6.2.3	PDU session management	480
6.2.4	IP address allocation	480
6.2.4.1	General	480
6.2.4.2	IP address allocation via NAS signalling	481
6.2.4.3	Additional RG related requirements for IP address allocation	481
6.2.4.4	Additional requirements of the UE acting as 5G ProSe layer-3 UE-to-network relay UE for IP	
	address allocation	482
6.2.5	Quality of service	482
6.2.5.1	General	482
6.2.5.1.1	OoS rules	
6.2.5.1.1.1	General	482
6.2.5.1.1.2	Signalled QoS rules	482
6.2.5.1.1.3	Derived QoS rules	484
6.2.5.1.1.4	OoS flow descriptions	484
6.2.5.1.2	Session-AMBR	485
6.2.5.1.2A	Void	485
6.2.5.1.3	UL user data packet matching	485
6.2.5.1.4	Reflective QoS	485
6.2.5.1.4.1	General	485
6.2.5.1.4.2	Derivation of packet filter for UL direction from DL user data packet	486
6.2.5.1.4.3	Creating a derived OoS rule by reflective OoS in the UE	
6.2.5.1.4.4	Updating a derived QoS rule by reflective QoS in the UE	488
6.2.5.1.4.5	Deleting a derived QoS rule in the UE	488
6.2.5.1.4.6	Ignoring ROI in the UE	489
6252	OoS in MA PDU session	489
626	Local area data network (LADN)	489
627	Handling of DNN based congestion control	490
628	Handling of S-NSSAI based congestion control	491
629	Interaction with upper layers	494
6291	General	494
6292	URSP	494
6293	ProSeP	494
6 2 10	Handling of 3GPP PS data off	494
6211	Multi-homed IPv6 PDU session	496
6 2 12	Handling of network rejection not due to congestion control	496
6 2 13	Handling of Small data rate control	499
6 2 14	Handling of Serving PI MN rate control	499
6215	Handling of Reliable Data Service	500
6 2 16	Handling of header compression for control plane CIoT optimizations	500
6217	Handling of edge computing enhancements	500
6 2 18	Support of redundant PDU sessions	501
63	Network-requested 5GSM procedures	502
631	PDU session authentication and authorization procedure	502
6311	General	502
6312	PDU FAP message reliable transport procedure	504
63121	PDU FAP message reliable transport procedure initiation	504
63122	PDU FAP message reliable transport procedure accented by the UF	505
63123	Abnormal cases on the network side	506
63124	Abnormal cases in the LIF	506
6313	PDU FAP result message transport procedure	506
63131	PDU FAP result message transport procedure initiation	506
63132	Abnormal cases in the UF	507
631A	Service-level authentication and authorization procedure	507
63141	General	507
63147	Service-level authentication and authorization procedure initiation	508
631A3	Service-level authentication and authorization procedure accented by the UF	500
6314 <i>/</i>	Abnormal cases on the network side	509
63145	Abnormal cases in the IIF	509
637	Network-requested PDU session modification procedure	509
6321	General	509
6322	Network-requested PDU session modification procedure initiation	510
6323	Network-requested PDU session modification procedure accepted by the UF	515
0.5.2.5	network-requested i DO session mouncation procedure accepted by the OE	

6.3.2.4	Network-requested PDU session modification procedure not accepted by the UE	523
6.3.2.5	Abnormal cases on the network side	528
6.3.2.6	Abnormal cases in the UE	529
6.3.3	Network-requested PDU session release procedure	529
6.3.3.1	General	529
6.3.3.2	Network-requested PDU session release procedure initiation	529
6.3.3.3	Network-requested PDU session release procedure accepted by the UE	532
6.3.3.4	N1 SM delivery skipped	541
6.3.3.5	Abnormal cases on the network side	541
6.3.3.6	Abnormal cases in the UE	541
6.4	UE-requested 5GSM procedures	542
6.4.1	UE-requested PDU session establishment procedure	542
6.4.1.1	General	542
6.4.1.2	UE-requested PDU session establishment procedure initiation	542
6.4.1.3	UE-requested PDU session establishment procedure accepted by the network	552
6.4.1.4	UE-requested PDU session establishment procedure not accepted by the network	566
6.4.1.4.	1 General	566
6.4.1.4.	2 Handling of network rejection due to congestion control	569
6.4.1.4.	3 Handling of network rejection not due to congestion control	576
6.4.1.5	Handling the maximum number of established PDU sessions	583
6.4.1.5	A Handling the maximum number of allowed active user-plane resources for PDU sessions of UEs	
	in NB-N1 mode	585
6.4.1.6	Abnormal cases in the UE	585
6.4.1.7	Abnormal cases on the network side	587
6.4.2	UE-requested PDU session modification procedure	588
6.4.2.1	General	588
6.4.2.2	UE-requested PDU session modification procedure initiation	589
6.4.2.3	UE-requested PDU session modification procedure accepted by the network	594
6.4.2.4	UE-requested PDU session modification procedure not accepted by the network	594
6.4.2.4.	1 General	594
6.4.2.4.	2 Handling of network rejection due to congestion control	595
6.4.2.4.	3 Handling of network rejection not due to congestion control	602
6.4.2.5	Abnormal cases in the UE	605
6.4.2.6	Abnormal cases on the network side	606
6.4.3	UE-requested PDU session release procedure	607
6.4.3.1	General	607
6.4.3.2	UE-requested PDU session release procedure initiation	607
6.4.3.3	UE-requested PDU session release procedure accepted by the network	608
6.4.3.4	UE-requested PDU session release procedure not accepted by the network	608
6.4.3.5	Abnormal cases in the UE	609
6.4.3.6	Abnormal cases on the network side	610
6.5	5GSM status procedure	610
6.5.1	General	610
6.5.2	5GSM status received in the UE	611
6.5.3	5GSM status received in the SMF	611
6.6	Miscellaneous procedures	611
6.6.1	Exchange of extended protocol configuration options	611
6.6.2	Remote UE report procedure	612
6.6.2.1	General	612
6.6.2.2	Remote UE report procedure initiation	612
6.6.2.3	Al second by the LE	
0.0.2.4	Abnormal cases in the petwork side	013
0.0.2.3	Autoritian cases on the network side	013
7 1	Handling of unknown, unforeseen, and erroneous protocol data	.614
7.1	General	614
7.2	Message too short or too long	614
7.2.1	Message too short	614
7.2.2	Message too long	614
7.3	Unknown or unforeseen procedure transaction identity or PDU Session identity	614
7.3.1	Procedure transaction identity	614
7.3.2	PDU Session identity	615

7.4	Unknown or unforeseen message type	616
7.5	Non-semantical mandatory information element errors	
7.5.1	Common procedures	
7.5.2	5GS mobility management	
7.5.3	5GS session management	
7.6	Unknown and unforeseen IFs in the non-imperative message part	618
761	IEIs unknown in the message	618
762	Out of sequence IFs	618
7.6.2	Reneated IEs	
7.0.5	Non-imperative message part errors	
771	Syntactically incorrect ontional IFs	618
777	Conditional IE errors	
7.8	Messages with semantically incorrect contents	
7.0	wessages with semanticarry medirect contents	
8	Message functional definitions and contents	619
8.1	Overview	619
8.2	5GS mobility management messages	619
8.2.1	Authentication request	619
8.2.1.1	Message definition	619
8.2.1.2	Authentication parameter RAND	619
8.2.1.3	Authentication parameter AUTN	
8.2.1.4	Void	
8.2.1.5	EAP message	
8.2.2	Authentication response	
8.2.2.1	Message definition	
8.2.2.2	Authentication response parameter	620
8.2.2.3	EAP message	
823	Authentication result	620
8231	Message definition	
8232	ABBA	621
824	Authentication failure	621
8241	Message definition	621
8242	Authentication failure parameter	
825	Authentication reject	
8251	Message definition	
8252	FAD massage	
826	Pagistration request	
0.2.0 8.2.6.1	Message definition	
8262	Non current notive NAS key set identifier	
0.2.0.2	5GMM comphility	
0.2.0.3	JOIVINI Capadility	
0.2.0.4	Degreeted NSS AL	
0.2.0.3	L act visited resistand TAL	
0.2.0.0	C1 LIE notwork constiller	
0.2.0.1	SI UE network capability	
8.2.0.8		
8.2.0.9	PDU session status	
8.2.6.1	0 MICO indication	
8.2.6.1		
8.2.6.1	2 Additional GUIT	
8.2.6.1	3 Allowed PDU session status	
8.2.6.1	4 UE's usage setting	
8.2.6.1	5 Requested DRX parameters	
8.2.6.1	6 EPS NAS message container	
8.2.6.1	/ LADN indication	
8.2.6.1	/A Payload container type	
8.2.6.1	8 Payload container	
8.2.6.1	9 Network slicing indication	
8.2.6.2	0 5GS update type	
8.2.6.2	1 NAS message container	
8.2.6.2	2 Requested extended DRX parameters	
8.2.6.2	3 EPS bearer context status	
8.2.6.2	4 T3324 value	

8.2.6.25	Mobile station classmark 2	
8.2.6.26	Supported codecs	
8.2.6.27	UE radio capability ID	
8.2.6.28	Requested mapped NSSAI	
8.2.6.29	Additional information requested	
8.2.6.30	Requested WUS assistance information	
8.2.6.31	Void	
8.2.6.32	N5GC indication	
8.2.6.33	Requested NB-N1 mode DRX parameters	
8.2.6.34	UE request type	
8.2.6.35	Paging restriction	
8.2.6.35	Service-level-AA container	
8.2.6.36	NID	
8.2.6.37	MS determined PLMN with disaster condition	
8.2.6.38	Requested PEIPS assistance information	
8.2.6.39	Requested T3512 value	
8.2.7	Registration accept	
8.2.7.1	Message definition	
8.2.7.2	5G-GUTI	
8.2.7.3	Equivalent PLMNs	
8.2.7.4	TÂI list	
8.2.7.5	Allowed NSSAI	
8.2.7.6	Rejected NSSAI	
8.2.7.7	Configured NSSAI	
8.2.7.8	5GS network feature support	
8.2.7.9	PDU session status	
8.2.7.10	PDU session reactivation result	
8.2.7.11	PDU session reactivation result error cause	
8.2.7.12	LADN information	
8.2.7.13	MICO indication	
8.2.7.14	Network slicing indication	
8.2.7.15	Service area list	
8.2.7.16	T3512 value	
8.2.7.17	Non-3GPP de-registration timer value	
8.2.7.18	T3502 value	
8.2.7.19	Emergency number list	
8.2.7.20	Extended emergency number list	
8.2.7.21	SOR transparent container	
8.2.7.22	EAP message	
8.2.7.23	NSSAI inclusion mode	
8.2.7.24	Operator-defined access category definitions	
8.2.7.25	Negotiated DRX parameters	
8.2.7.26	Non-3GPP NW policies	
8.2.7.27	Negotiated extended DRX parameters	
8.2.7.28	T3447 value	
8.2.7.29	T3448 value	
8.2.7.30	T3324 value	
8.2.7.31	EPS bearer context status	
8.2.7.32	UE radio capability ID	
8.2.7.33	UE radio capability ID deletion indication	636
8.2.7.34	Pending NSSAI	
8.2.7.35	Ciphering key data	
8.2.7.36	CAG information list	636
8.2.7.37	Truncated 5G-S-TMSI configuration	
8.2.7.38	Negotiated NB-N1 mode DRX parameters	
8.2.7.39	Negotiated WUS assistance information	
8.2.7.40	Extended rejected NSSAI	
8.2.7.41	Service-level-AA container	
8.2.7.42	Negotiated PEIPS assistance information	
8.2.7.43	5GS additional request result	
8.2.7.44	NSSRG information	
8.2.7.45	Disaster roaming wait range	

82746	Disaster return wait range	637
0.2.7.40	Disaster ferun war fange	
8.2.7.47	List of PLMINs to be used in disaster condition	037
8.2.7.48	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	637
8.2.7.49	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	637
8.2.7.50	Extended CAG information list	637
8.2.7.51	NSAG information	.637
828	Registration complete	638
8281	Message definition	638
0.2.0.1	Nossage definition	.050
0.2.0.2	SOR transparent container	030
8.2.9	Registration reject.	638
8.2.9.1	Message definition	638
8.2.9.2	T3346 value	639
8.2.9.3	T3502 value	639
8.2.9.4	EAP message	.639
8.2.9.5	Rejected NSSAI	.639
8296	CAG information list	640
8207	Extended rejected NSSAI	640
0.2.9.7	Disease with an extension	.040
8.2.9.8	Disaster feturin wait range	040
8.2.9.9	Extended CAG information list	
8.2.9.10	Lower bound timer value	640
8.2.9.11	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	640
8.2.9.12	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	640
8.2.10	UL NAS transport	.640
8.2.10.1	Message definition	.640
8 2 10 2	PDU ession ID	6/1
8 2 10 3	Old PDU session ID	6/1
0.2.10.3		041
8.2.10.4	Request type	
8.2.10.5	S-NSSAI	641
8.2.10.6	DNN	641
8.2.10.7	Additional information	642
8.2.10.8	MA PDU session information	642
8.2.10.9	Release assistance indication	642
8 2 11	DL NAS transport	642
8 2 11 1	Message definition	642
8 2 11 2	PDU assign ID	6/3
0.2.11.2	Additional information	045
8.2.11.5	Additional information	043
8.2.11.4	SGMM cause	
8.2.11.5	Back-off timer value	643
8.2.11.6	Lower bound timer value	643
8.2.12	De-registration request (UE originating de-registration)	643
8.2.12.1	Message definition	
8 2 13	De-registration accent (UE originating de-registration)	644
8 2 13 1	Message definition	644
8 2 14	De registration request (IE terminated de registration)	644
0.2.14	De-registration request (OE terminated de-registration).	
8.2.14.1	Message definition	
8.2.14.2	SGMM cause	645
8.2.14.3	T3346 value	645
8.2.14.4	Rejected NSSAI	645
8.2.14.5	CAG information list	645
8.2.14.6	Extended rejected NSSAI	.646
8.2.14.7	Disaster return wait range	646
8 2 14 7 4	Extended CAG information list	646
8 2 14 8	Lower bound timer value	646
8 2 1 4 0	Earbiddon TAI(a) for the list of "5GS forhiddon treating areas for reamine"	614
0.2.14.7	For trade in TAI(s) for the list of JOS for bit the test list of $IOI for the list of JOS for the list of$	
0.2.14.10	Forbidden 1 AI(s) for the list of 5GS forbidden tracking areas for regional provision of service"	046
8.2.15	De-registration accept (UE terminated de-registration)	
8.2.15.1	Message definition	646
8.2.16	Service request	647
8.2.16.1	Message definition	647
8.2.16.2	Uplink data status	647
8.2.16.3	PDU session status	647
8 2 16 4	Allowed PDU session status	647

8.2.16.5	NAS message container	648
8.2.16.6	UE request type	648
8.2.16.7	Paging restriction	648
8.2.17	Service accept	648
8.2.17.1	Message definition	648
8.2.17.2	PDU session status	649
8.2.17.3	PDU session reactivation result	649
8.2.17.4	PDU session reactivation result error cause	649
8.2.17.5	EAP message	649
8.2.17.6	T3448 value	649
8.2.17.7	5GS additional request result	649
8.2.17.8	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	649
8.2.17.9	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	649
8.2.18	Service reject	649
8.2.18.1	Message definition	649
8.2.18.2	PDU session status	650
8.2.18.3	T3346 value	650
8.2.18.4	EAP message	650
8.2.18.5	T3448 value	650
8.2.18.6	CAG information list	651
8.2.18.7	Disaster return wait range	651
8.2.18.8	Extended CAG information list	651
8.2.18.9	Lower bound timer value	651
8.2.18.10	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	651
8.2.18.11	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	651
8.2.19	Configuration update command	651
8.2.19.1	Message definition	651
8.2.19.2	Configuration update indication	654
8.2.19.3	5G-GUTI	654
8.2.19.4	TAI list	654
8.2.19.5	Allowed NSSAI	654
8.2.19.6	Service area list	654
8.2.19.7	Full name for network	654
8.2.19.8	Short name for network	654
8.2.19.9	Local time zone	654
8.2.19.10	Universal time and local time zone	654
8.2.19.11	Network daylight saving time	654
8.2.19.12	LADN information	654
8.2.19.13	MICO indication	655
8.2.19.14	Network slicing indication	655
8.2.19.15	Configured NSSAI	655
8.2.19.16	Rejected NSSAI	655
8.2.19.17	Operator-defined access category definitions	655
8.2.19.18	SMS indication	655
8.2.19.19	T3447 value	655
8.2.19.20	CAG information list	655
8.2.19.21	UE radio capability ID	655
8.2.19.22	UE radio capability ID deletion indication	655
8.2.19.23	5GS registration result	655
8.2.19.24	Truncated 5G-S-TMSI configuration	655
8.2.19.25	Additional configuration indication	656
8.2.19.26	Extended rejected NSSAI	656
8.2.19.27	Service-level-AA container	656
8.2.19.28	NSSRG information	656
8.2.19.29	Disaster roaming wait range	656
8.2.19.30	Disaster return wait range	656
8.2.19.31	List of PLMNs to be used in disaster condition	656
8.2.19.32	Extended CAG information list	656
8.2.19.33	Updated PEIPS assistance information	656
8.2.19.34	NSAG information	656
8.2.19.35	Priority indicator	656
8.2.20	Configuration update complete	657

8.2.20.1	Message definition	
8.2.20.2	Void	
8.2.21	Identity request	
8.2.21.1	Message definition	
8.2.22	Identity response	
8.2.22.1	Message definition	
8.2.23	Notification	
8.2.23.1	Message definition	
8.2.24	Notification response	
8.2.24.1	Message definition	
8.2.24.2	PDU session status	
8.2.25	Security mode command	
8.2.25.1	Message definition	
8.2.25.2	IMEISV request	
8.2.25.3	Void	
8.2.25.4	Selected EPS NAS security algorithms	
8.2.25.5	Additional 5G security information	
8.2.25.6	EAP message	
8.2.25.7	ABBA	
8.2.25.8	Replayed S1 UE security capabilities	661
8 2 26	Security mode complete	661
8 2 26 1	Message definition	661
8 2 26 2	IMFISV	661
8 2 26 3	NAS message container	661
8 2 26 4	non IMEISV DEI	661
8.2.20.4	Security mode reject	
86271	Massage definition	
8.0.27.1	Security protected 5GS NAS massage	
0.2.20	Magazza definition	
0.2.20.1	SCMM status	
8.2.29	Marca de finition	
8.2.29.1	Message definition	
8.2.30	Control Plane Service request	
8.2.30.1	Message definition	
8.2.30.2	Clo1 small data container	
8.2.30.3	Payload container type	
8.2.30.4	Payload container	
8.2.30.5	PDU session ID	
8.2.30.6	PDU session status	
8.2.30.7	Release assistance indication	
8.2.30.8	Uplink data status	
8.2.30.9	NAS message container	
8.2.30.10	Additional information	
8.2.30.11	Allowed PDU session status	
8.2.30.12	UE request type	
8.2.30.13	Paging restriction	
8.2.31	Network slice-specific authentication command	
8.2.31.1	Message definition	
8.2.32	Network slice-specific authentication complete	
8.2.32.1	Message definition	
8.2.33	Network slice-specific authentication result	
8.2.33.1	Message definition	
8.2.34	Relay key request	
8.2.34.1	Message definition	
8.2.35	Relay key accept	
8.2.35.1	Message definition	
8.2.35.2	EAP message	
8.2.36	Relay key reject	
8.2.36.1	Message definition	
8.2.36.2	EAP message	
8.2.37	Relay authentication request	668
8.2.37.1	Message definition	668
8.2.38	Relay authentication response	669

8.2.38.1	Message definition	669
8.3	5GS session management messages	
8.3.1	PDU session establishment request	
8.3.1.1	Message definition	
8.3.1.2	PDU session type	
8.3.1.3	SSC mode	
8.3.1.4	Maximum number of supported packet filters	
8.3.1.5	5GSM capability	
8.3.1.6	Void	
8.3.1.7	Always-on PDU session requested	
8.3.1.8	SM PDU DN request container	
8.3.1.9	Extended protocol configuration options	
8.3.1.10	IP header compression configuration	
8.3.1.11	DS-TT Ethernet port MAC address	
8.3.1.12	UE-DS-11 residence time	
8.3.1.13	Port management information container	
8.3.1.14	Ethernet neader compression configuration	072
0.3.1.13 8 3 1 16	Suggested Interface Identifier	072
8.3.1.10 8.3.1.17	Dequested MRS container	
0.3.1.17	PDU session pair ID	
8 3 1 10	P DO SESSION PAIL ID	
832	PDU session establishment accent	
8321	Message definition	672
8.3.2.2	5GSM cause	
8.3.2.3	PDU address	
8.3.2.4	RQ timer value	
8.3.2.5	S-NSSAI	
8.3.2.6	Always-on PDU session indication	
8.3.2.7	Mapped EPS bearer contexts	
8.3.2.8	EAP message	
8.3.2.9	Authorized QoS flow descriptions	
8.3.2.10	Extended protocol configuration options	
8.3.2.11	DNN	
8.3.2.12	SGSM network feature support	
8.3.2.13		
8.3.2.14	ATSSS container	0/4
0.3.2.13 8 3 2 16	Control plane only indication	074
8.3.2.10	IP header compression configuration	
83218	Ethernet header compression configuration	
83219	Service-level-A A container	
8.3.2.20	Received MBS container.	
8.3.3	PDU session establishment reject	
8.3.3.1	Message definition	
8.3.3.2	Back-off timer value	
8.3.3.3	Allowed SSC mode	
8.3.3.4	EAP message	
8.3.3.4A	5GSM congestion re-attempt indicator	
8.3.3.5	Extended protocol configuration options	676
8.3.3.6	Re-attempt indicator	677
8.3.3.7	Service-level-AA container	
8.3.4	PDU session authentication command	
8.3.4.1	Message definition	
8.3.4.2	Extended protocol configuration options	
8.5.4.3 9.2.5	V010 DDU session authentication complete	
0.3.3	r المن session aumentication complete	0/8 270
8352	Extended protocol configuration options	078 678
8353	Void	078 678
8.3.6	PDU session authentication result	
8.3.6.1	Message definition	
	-	

8.3.6.2	EAP message	679
8.3.6.3	Extended protocol configuration options	679
8.3.7	PDU session modification request	679
8.3.7.1	Message definition	679
8.3.7.2	5GSM capability	
8.3.7.3	5GSM cause	681
8.3.7.4	Maximum number of supported packet filters	681
8.3.7.5	Always-on PDU session requested	
8.3.7.6	Integrity protection maximum data rate	681
8.3.7.7	Requested QoS rules	
8.3.7.8	Requested QoS flow descriptions	681
8.3.7.9	Extended protocol configuration options	681
8.3.7.10	Mapped EPS bearer contexts	
8.3.7.11	Port management information container	
8.3.7.12	IP header compression configuration	681
8.3.7.13	Ethernet header compression configuration	
8.3.7.14	Requested MBS container	
8.3.7.15	Service-level-AA container	
8.3.8	PDU session modification reject	
8.3.8.1	Message definition	
8.3.8.2	Back-off timer value	
8.3.8.2A	5GSM congestion re-attempt indicator	
8.3.8.3	Extended protocol configuration options	
8.3.8.4	Re-attempt indicator	
8.3.9	PDU session modification command	
8.3.9.1	Message definition	
8.3.9.2	5GSM cause	
8.3.9.3	Session-AMBR	
8.3.9.4	RQ timer value	
8.3.9.5	Always-on PDU session indication	
8.3.9.6	Authorized QoS rules	
8.3.9.7	Mapped EPS bearer contexts	
8.3.9.8	Authorized QoS flow descriptions	
8.3.9.9	Extended protocol configuration options	
8.3.9.10	Void	
8.3.9.11	ATSSS container	
8.3.9.12	IP header compression configuration	
8.3.9.13	Port management information container	
8.3.9.14	Serving PLMN rate control	
8.3.9.15	Ethernet header compression configuration	
8.3.9.16	Received MBS container	
8.3.9.17	Service-level-AA container	686
8.3.10	PDU session modification complete	686
8.3.10.1	Message definition	686
8.3.10.2	Extended protocol configuration options	686
8.3.10.3	Port management information container	
8.3.11	PDU session modification command reject	686
8.3.11.1	Message definition	686
8.3.11.2	Extended protocol configuration options	
8.3.12	PDU session release request	687
8.3.12.1	Message definition	
8.3.12.2	5GSM cause	
8.3.12.3	Extended protocol configuration options	
8.3.13	PDU session release reject	
8.3.13.1	Message definition	
8.3.13.2	Extended protocol configuration options	
8.3.14	PDU session release command	
8.3.14.1	Message definition	
8.3.14.2	Back-off timer value	
8.3.14.3	EAP message	
8.3.14.4	Extended protocol configuration options	689
8.3.14.5	5GSM congestion re-attempt indicator	

8.3.14.6	Access type	689
8.3.14.7	Service-level-AA container	689
8.3.15	PDU session release complete	690
8.3.15.1	Message definition	690
8.3.15.2	5GSM cause	690
8.3.15.3	Extended protocol configuration options	690
8.3.16	5GSM status	690
8.3.16.1	Message definition	690
8.3.17	Service-level authentication command	691
8.3.17.1	Message definition	691
8.3.18	Service-level authentication complete	691
8.3.18.1	Message definition	
8.3.19	Remote UE report	
8 3 19 1	Message definition	692
8 3 19 2	Remote LIF context connected	692
8 3 10 3	Remote UE context disconnected	
8 3 20	Remote UE context disconnected	
8 3 20 1	Massage definition	
0.3.20.1 9 2 20 2	Wessage definition	
0.3.20.2	Volu	
8.3.20.3	V 010	
8.3.20.4	V 01d	
9 Ge	neral message format and information elements coding.	
91	Overview	693
9.11	$N\Delta S$ message format	693
012	Field format and manning	
9.1.2	Evtended protocol discriminator	
9.2	Security based type	
9.5	DDU session identita	
9.4	PDU session identity	
9.5	Spare half octet	
9.6	Procedure transaction identity	
9.7	Message type	
9.8	Message authentication code	697
9.9	Plain 5GS NAS message	697
9.10	Sequence number	697
9.11	Other information elements	698
9.11.1	General	698
9.11.2	Common information elements	698
9.11.2.1	Additional information	698
9.11.2.1A	Access type	698
9.11.2.1B	DNN	699
9.11.2.2	EAP message	699
9.11.2.3	GPRS timer	699
9.11.2.4	GPRS timer 2	700
9.11.2.5	GPRS timer 3	700
9.11.2.6	Intra N1 mode NAS transparent container	
9.11.2.7	N1 mode to S1 mode NAS transparent container	
9.11.2.8	S-NSSAI	
9.11.2.9	S1 mode to N1 mode NAS transparent container	
9 11 2 10	Service-level-A A container	704
9 11 2 11	Service-level device ID	707
9 11 2 12	Service-level-A A server address	707
9.11.2.12	Service-level- $\Delta \Delta$ payload	708
0 11 0 1/	Service level $\Delta \Delta$ response	
0 11 0 15	Service-level-A A pavload type	
0 11 0 14	Void	
<i>7</i> .11.2.10	Volu	
0.11.2.1/	Service level A A service status indication	
9.11.2.18	Service-level-AA service status indication	
9.11.5	SONDA and like	
9.11.3.1		
9.11.3.2	JUMIM Cause	
9.11.3.2A	SGS DKX parameters	

9.11.3.3	5GS identity type	.718
9.11.3.4	5GS mobile identity	.719
9.11.3.5	5GS network feature support	.725
9.11.3.6	5GS registration result.	.730
9.11.3.7	5GS registration type	.731
9.11.3.8	5GS tracking area identity	.732
9.11.3.9	5GS tracking area identity list	.733
9.11.3.9A	5GS update type	.738
9.11.3.10	ABBA	.739
9.11.3.11	Void	.739
9.11.3.12	Additional 5G security information	.739
9.11.3.12A	Additional information requested	.740
9.11.3.13	Allowed PDU session status	.740
9.11.3.14	Authentication failure parameter	.741
9.11.3.15	Authentication parameter AUTN	.741
9.11.3.16	Authentication parameter RAND	.741
9.11.3.17	Authentication response parameter	.741
9.11.3.18	Configuration update indication	.741
9.11.3.18A	CAG information list	.742
9.11.3.18B	CIoT small data container	.743
9.11.3.18C	Ciphering key data	.747
9.11.3.18D	Control plane service type	.756
9.11.3.19	Daylight saving time	.756
9.11.3.20	De-registration type	.756
9.11.3.21	Void	.757
9.11.3.22	Void	.757
9.11.3.23	Emergency number list	.757
9.11.3.23A	EPS bearer context status	.757
9.11.3.24	EPS NAS message container	.757
9.11.3.25	EPS NAS security algorithms	.758
9.11.3.26	Extended emergency number list	.758
9.11.3.26A	Extended DRX parameters	.758
9.11.3.27	V 01d	.758
9.11.3.28	IMEIS V request	./58
9.11.3.29	LADN indication.	./58
9.11.3.30	LADN Information	./39
9.11.3.31	MA DDU accient information	./60
9.11.3.31A	Marrad NSCAL	./01
9.11.3.31B	Mapped NSSAI	762
9.11.3.310	NAS kay sat identifier	762
9.11.3.32	NAS Key set identifier	763
9.11.3.33	NAS message container	763
9.11.3.34	Network name	764
9 11 3 36	Network slicing indication	764
9 11 3 36A	Non-3GPP NW provided policies	765
9 11 3 37	NSSAI	765
9 11 3 37A	NSSAI inclusion mode	766
9 11 3 38	Operator-defined access category definitions	766
9.11.3.39	Pavload container	.770
9.11.3.40	Pavload container type	.775
9.11.3.41	PDU session identity 2	.776
9.11.3.42	PDU session reactivation result	.776
9.11.3.43	PDU session reactivation result error cause	.777
9.11.3.44	PDU session status	.778
9.11.3.45	PLMN list	.778
9.11.3.46	Rejected NSSAI	.778
9.11.3.46A	Release assistance indication	.780
9.11.3.47	Request type	.780
9.11.3.48	S1 UE network capability	.780
9.11.3.48A	S1 UE security capability	.780
9.11.3.49	Service area list	.780

9.11.3.50	Service type	
9.11.3.50A	SMS indication	
9.11.3.51	SOR transparent container	
9.11.3.51A	Supported codec list	
9.11.3.52	Time zone	
9.11.3.53	Time zone and time	
9.11.3.53A	UE parameters update transparent container	
9.11.3.54	UE security capability	
9.11.3.55	UE's usage setting	
9.11.3.56	UE status	
9.11.3.57	Uplink data status	
9.11.3.58	Void	
9.11.3.59	Void	
9.11.3.60	Void	
9.11.3.61	Void	805
9 11 3 62	Void	805
9.11.3.63	Void	
9.11.3.64	Void	805
9.11.3.65	Void	
9 11 3 66	Void	805
9 11 3 67	Void	805
9 11 3 68	UE radio canability ID	805
9 11 3 69	UE radio capability ID deletion indication	805
9 11 3 70	Truncated 5G-S-TMSI configuration	806
9 11 3 71	WIIS assistance information	807
9 11 3 72	NSGC indication	807
9 11 3 73	NB-N1 mode DRX parameters	808
9 11 3 74	Additional configuration indication	808
9 11 3 75	Extended rejected NSSAI	809
9 11 3 76	LIE request type	813
9 11 3 77	Paging restriction	813
9 11 3 78	Void	814
9 11 3 79	NID	81/
9 11 3 80	PEIPS assistance information	81/
9 11 3 81	5GS additional request result	816
9 11 3 82	NSSRG information	817
9 11 3 83	List of PLMNs to be used in disaster condition	818
9 11 3 8/	Registration wait range	810
9 11 3 85	PI MN identity	819
9.11.3.85	Extended CAG information list	820
9 11 3 87	NSAG information	822
0 11 3 88	ProSe relay transaction identity	823
0 11 3 80	Palay kay request parameters	823
9.11.3.09	Relay key response parameters	824
9 11 3 91	Priority indicator	825
9.11.3.91	5GS session management (5GSM) information elements	825
9.11.4	5GSM canability	825
9 11 / 2	5GSM capacity	826
9.11.4.2	Always on PDU session indication	
9.11.4.5	Always-on PDU session requested	828
0 11 / 5	Allowed SSC mode	828
0 11 4 6	Extended protocol configuration options	820
9.11.4.0	Integrity protection maximum data rate	829
9 11 4 8	Manned FPS bearer contexts	
9 11 / 0	Maximum number of supported packet filters	
9.11. 4 .2	PDU address	
9 11 / 11	PDU session type	
9.11.7.11 9.11/1.17	Oos flow descriptions	
9 11 4 13	Oos miles	
9 11 4 14	Session-AMRR	۰۰۰۰۰،۰۰۰،۰۰۰،۰۰۰،۰۰۰،۰۰۰،۰۰۰،۰۰۰،۰۰۰،
9 11 / 15	SM PDU DN request container	
9 11 / 16	SSC mode	
Z.II.T.IU		

9.11.4	.17 Re-attempt indicator	853
9.11.4	.18 5GSM network feature support	
9.11.4	.19 Void	
9.11.4	20 Serving PLMN rate control	854
9.11.4	22 ATSSS container	
9 11 4	22 Arissis container	
9.11.4	.24 IP header compression configuration	
9.11.4	.25 DS-TT Ethernet port MAC address	
9.11.4	.26 UE-DS-TT residence time	859
9.11.4	.27 Port management information container	
9.11.4	.28 Ethernet header compression configuration	
9.11.4	.29 Remote UE context list	861
9.11.4	.30 Requested MBS container	
9.11.4	.31 Received MBS container	
9.11.4	.32 PDU session pair ID	
9.11.4	24 ECS address	8/3
9.11.4	35 Void	
9.11.4	3GPP specific coding information defined within present document	
9.12.1	Serving network name (SNN)	
10	List of system parameters	
10.1	General	
10.2	Timers of 5GS mobility management	8/8
10.3	limers of 5GS session management	889
Anne	x A (informative): Cause values for 5GS mobility management	
A 1	Courses related to LIE identification	905
A.1	Causes related to UE identification	
A.2	Cause related to subscription options	
12	Courses related to DI MNI on SNDN specific naturally failures and connection/outhentication	
A.3	foilures	807
A.4	Causes related to invalid messages	
		000
Anne	x B (informative): Cause values for 5GS session management	900
B.1	Causes related to nature of request	
DЭ	Protocol amore (o o university massage)	002
В.2	Protocol errors (e.g., unknown message)	
Anne	x C (normative): Storage of 5GMM information	
~ .		
C.1	Storage of 5GMM information for UEs not operating in SNPN access operation mode	904
C.2	Storage of 5GMM information for UEs operating in SNPN access operation mode	
Anne	x D (normative): UE policy delivery service	909
D 1	General	000
D.1		000 000
D.1.2	Principles of PTI handling for UE policy delivery service procedures	
5		0.4.4
D.2	Procedures	
D.2.1	Network-requested UE policy management procedure	
D.2.1.	Otheral Network requested UE policy management precedure initiation	
D.2.1.	 Network-requested UE policy management procedure accorted by the UE Network-requested UE policy management procedure accorted by the UE 	
D.2.1.	4 Network-requested UE policy management procedure not accepted by the UE	
D.2.1	5 Abnormal cases on the network side	
<u>_</u>		
D.2.1.	6 Abnormal cases in the UE	
D.2.1. D.2.2	6 Abnormal cases in the UE UE-initiated UE state indication procedure	913 914

D 2 2 2 IF initiated IF state indication proceedure accented by the network 015
D.2.2.5 DE-initiated DE state indication procedure accepted by the network
D.2.2.4 Abnormal cases on the network side
D.3 UE policy re-assembly at the UE
D.4 Void
D.5 Message functional definition and contents
D.5.1 Manage UE policy command
D.5.1.1 Message definition
D.5.1.2 UE policy network classmark
D.5.2 Manage UE policy complete
D.5.2.1 Message definition
D.5.3 Manage UE policy command reject
D.5.3.1 Message definition
D.5.4 UE state indication
D.5.4.1 Message definition
D.6 Information elements coding 918
D 6 1 UE policy delivery service message type 918
D 6 2 UE policy section management list 918
D 6 3 UE policy section management result 924
D 6 4 UPSL list 926
D.6.5 UE policy classmark
D.6.6 UE OS Id
D.6.7 UE policy network classmark
D.7 Timers of UE policy delivery service
D.8 Handling of unknown unforeseen and erroneous UPDS data 929
D.8.1 General 929
D.8.2 Message too short or too long
D.8.2.1 Message too short
D.8.2.2 Message too long
D.8.3 Unknown or unforeseen procedure transaction identity
D.8.3.1 Procedure transaction identity
D.8.4 Unknown or unforeseen message type
D.8.5 Non-semantical mandatory information element errors
D.8.5.1 Common procedures
D.8.6 Unknown and unforeseen IEs in the non-imperative message part
D.8.6.1 IEIs unknown in the message
D.8.6.2 Out of sequence IEs
D.8.6.3 Repeated IEs
D.8.7 Non-imperative message part errors
D.8.7.1 Syntactically incorrect optional IEs
D.8.7.2 Conditional IE errors
D.8.8 Messages with semantically incorrect contents
Annex E (informative): Void
Annex F (informative): Change history934
History

Foreword

This Technical Specification has been produced by the 3rd Generation Partnership Project (3GPP).

The contents of the present document are subject to continuing work within the TSG and may change following formal TSG approval. Should the TSG modify the contents of the present document, it will be re-released by the TSG with an identifying change of release date and an increase in version number as follows:

Version x.y.z

where:

- x the first digit:
 - 1 presented to TSG for information;
 - 2 presented to TSG for approval;
 - 3 or greater indicates TSG approved document under change control.
- y the second digit is incremented for all changes of substance, i.e. technical enhancements, corrections, updates, etc.
- z the third digit is incremented when editorial only changes have been incorporated in the document.

1 Scope

The present document specifies the non-access stratum (NAS) procedures in the 5G system (5GS) used by the protocols for:

- mobility management between the user equipment (UE) and the access and mobility management function (AMF) for both 3GPP access and non-3GPP access; and
- session management between the user equipment (UE) and the session management function (SMF) for both 3GPP access and non-3GPP access.

The 5GS mobility management (5GMM) protocol defined in the present document provides procedures for the control of mobility when the user equipment (UE) is using the NG radio access network (NG-RAN) and/or non-3GPP access network. The 5GMM protocol also provides control of security for the NAS protocols.

The 5GS session management (5GSM) protocol defined in the present document provides procedures for the handling of 5GS PDU sessions. Together with the bearer control provided by the access stratum, this protocol is used for the control of user-plane resources.

For both NAS protocols the present document specifies procedures for the support of inter-system mobility between the NG-RAN and the evolved universal terrestrial radio access (E-UTRAN), between the NG-RAN and the non-3GPP access network connected to the EPC, and between the non-3GPP access network connected to the 5G core network (5GCN) and the E-UTRAN.

For both NAS protocols the present document specifies procedures for the support of mobility between the NG-RAN and the non-3GPP access network connected to the 5GCN.

In addition, the present document specifies the procedures in the 5GS for UE policy delivery service between the UE and the policy control function (PCF) for both 3GPP access and non-3GPP access.

The present document is applicable to the UE, the access and mobility management function (AMF), the session management function (SMF), and the PCF in the 5GS.

The clauses and subclauses in the present document are common for both 3GPP access and non-3GPP access unless it is explicitly stated that they apply to 3GPP access only or non-3GPP access only.

2 References

The following documents contain provisions which, through reference in this text, constitute provisions of the present document.

- References are either specific (identified by date of publication, edition number, version number, etc.) or non-specific.
- For a specific reference, subsequent revisions do not apply.
- For a non-specific reference, the latest version applies. In the case of a reference to a 3GPP document (including a GSM document), a non-specific reference implicitly refers to the latest version of that document in the same Release as the present document.
- [1] 3GPP TR 21.905: "Vocabulary for 3GPP Specifications".
- [1A] 3GPP TS 22.011: "Service accessibility".
- [2] 3GPP TS 22.101: "Service aspects; Service principles".
- [3] 3GPP TS 22.261: "Service requirements for the 5G system; Stage 1".
- [4] 3GPP TS 23.003: "Numbering, addressing and identification".
- [4A] 3GPP TS 23.040: "Technical realization of Short Message Service (SMS)".
- [4B] 3GPP TS 23.032: "Universal Geographical Area Description (GAD)".

- [5] 3GPP TS 23.122: "Non-Access-Stratum functions related to Mobile Station (MS) in idle mode". [6] 3GPP TS 23.167: "IP Multimedia Subsystem (IMS) emergency sessions". [6A] 3GPP TS 23.216: "Single Radio Voice Call Continuity (SRVCC); Stage 2". [6AB] 3GPP TS 23.256: "Support of Uncrewed Aerial Systems (UAS) connectivity, identification and tracking; Stage 2". [6B] 3GPP TS 23.273: "5G System (5GS) Location Services (LCS); Stage 2". [6C] 3GPP TS 23.287: "Architecture enhancements for 5G System (5GS) to support Vehicle-to-Everything (V2X) services". 3GPP TS 23.316: "Wireless and wireline convergence access support for the 5G System (5GS)". [6D] [6E] 3GPP TS 23.304: "Proximity based Services (ProSe) in the 5G System (5GS)". [7] 3GPP TS 23.401: "GPRS enhancements for E-UTRAN access". [8] 3GPP TS 23.501: "System Architecture for the 5G System; Stage 2". [9] 3GPP TS 23.502: "Procedures for the 5G System; Stage 2". [10] 3GPP TS 23.503: "Policy and Charging Control Framework for the 5G System; Stage 2". [10A] 3GPP TS 23.548: "5G System Enhancements for Edge Computing; Stage 2". 3GPP TS 24.007: "Mobile radio interface signalling layer 3; General aspects". [11] [12] 3GPP TS 24.008: "Mobile Radio Interface Layer 3 specification; Core Network Protocols; Stage 3". 3GPP TS 24.011: "Point-to-Point Short Message Service (SMS) support on mobile radio [13] interface". 3GPP TS 24.080: "Mobile radio interface layer 3 Supplementary services specification; Formats [13A] and coding". [13B] 3GPP TS 24.193: "Access Traffic Steering, Switching and Splitting; Stage 3". 3GPP TS 24.173: "IMS Multimedia telephony communication service and supplementary [13C] services; Stage 3". [13D] 3GPP TS 24.174: "Support of multi-device and multi-identity in the IP Multimedia Subsystem (IMS); Stage 3". 3GPP TS 24.229: "IP multimedia call control protocol based on Session Initiation Protocol (SIP) [14] and Session Description Protocol (SDP); Stage 3". [14AA] 3GPP TS 24.237: "IP Multimedia (IM) Core Network (CN) subsystem IP Multimedia Subsystem (IMS) service continuity; Stage 3". 3GPP TS 24.250: "Protocol for Reliable Data Service; Stage 3". [14A] 3GPP TS 24.301: "Non-Access-Stratum (NAS) protocol for Evolved Packet System (EPS); [15] Stage 3". 3GPP TS 24.302: "Access to the 3GPP Evolved Packet Core (EPC) via non-3GPP access [16] networks; Stage 3" [17] 3GPP TS 24.368: "Non-Access Stratum (NAS) configuration Management Object (MO)". 3GPP TS 24.502: "Access to the 3GPP 5G System (5GS) via non-3GPP access [18] networks; Stage 3".
- [19] 3GPP TS 24.526: "UE policies for 5G System (5GS); Stage 3".

[19BA]	3GPP TS 24.539: "5G System (5GS); Network to TSN translator (TT) protocol aspects; Stage 3".
[19A]	3GPP TS 24.535: "Device-Side Time-Sensitive Networking (TSN) Translator (DS-TT) to Network-Side TSN Translator (NW-TT) protocol aspects; Stage 3".
[19B]	3GPP TS 24.587: "Vehicle-to-Everything (V2X) services in 5G System (5GS); Protocol aspects; Stage 3"
[19C]	3GPP TS 24.588: "Vehicle-to-Everything (V2X) services in 5G System (5GS); User Equipment (UE) policies; Stage 3"
[19D]	Void.
[19E]	3GPP TS 24.554: "Proximity-service (ProSe) in 5G System (5GS) protocol aspects; Stage 3".
[19F]	3GPP TS 24.555: "Proximity-services (ProSe) in 5G System (5GS); User Equipment (UE) policies; Stage 3".
[20]	3GPP TS 24.623: "Extensive Markup Language (XML) Configuration Access Protocol (XCAP) over the Ut interface for Manipulating Supplementary Services".
[20AA]	3GPP TS 29.500: "5G System; Technical Realization of Service Based Architecture; Stage 3".
[20A]	3GPP TS 29.502: "5G System; Session Management Services; Stage 3".
[20AB]	3GPP TS 29.503: "5G System; Unified Data Management Services; Stage 3".
[20B]	3GPP TS 29.518: "5G System; Access and Mobility Management Services; Stage 3".
[21]	3GPP TS 29.525: "5G System; UE Policy Control Service; Stage 3".
[21A]	3GPP TS 29.526: "5G System; Network Slice-Specific Authentication and Authorization (NSSAA) services; Stage 3".
[21B]	3GPP TS 29.256: "5G System; Uncrewed Aerial Systems Network Function (UAS-NF); Aerial Management Services; Stage 3.
[22]	3GPP TS 31.102: "Characteristics of the Universal Subscriber Identity Module (USIM) application".
[22A]	3GPP TS 31.111: "USIM Application Toolkit (USAT)".
[22B]	3GPP TS 31.115: "Secured packet structure for (Universal) Subscriber Identity Module (U)SIM Toolkit applications".
[23]	3GPP TS 33.102: "3G security; Security architecture".
[23A]	3GPP TS 33.401: "3GPP System Architecture Evolution; Security architecture".
[24]	3GPP TS 33.501: "Security architecture and procedures for 5G System".
[24A]	3GPP TS 33.535: "Authentication and Key Management for Applications (AKMA) based on 3GPP credentials in the 5G System (5GS)".
[24B]	3GPP TS 33.256: "Security aspects of Uncrewed Aerial Systems (UAS)".
[25]	3GPP TS 36.323: "NR; Packet Data Convergence Protocol (PDCP) specification".
[25A]	3GPP TS 36.331: "Evolved Universal Terrestrial Radio Access (E-UTRA); Radio Resource Control (RRC) protocol specification".
[25B]	3GPP TS 36.300: "Evolved Universal Terrestrial Radio Access (E-UTRA) and Evolved Universal Terrestrial Radio Access Network (E-UTRAN); Overall description".
[050]	

[25C] 3GPP TS 36.304: "Evolved Universal Terrestrial Radio Access (E-UTRA); User Equipment (UE) procedures in idle mode".

- [25D] 3GPP TS 36.306: "Evolved Universal Terrestrial Radio Access (E-UTRA); User Equipment (UE) radio access capabilities".
- [25E] 3GPP TS 36.321: "Evolved Universal Terrestrial Radio Access (E-UTRA); Medium Access Control (MAC) protocol specification".
- [26] 3GPP TS 37.355: "LTE Positioning Protocol (LPP)".
- [27] 3GPP TS 38.300: "NR; NR and NG-RAN Overall Description; Stage 2".
- [28] 3GPP TS 38.304: "New Generation Radio Access Network; User Equipment (UE) procedures in Idle mode".
- [29] 3GPP TS 38.323: "Evolved Universal Terrestrial Radio Access (E-UTRA); Packet Data Convergence Protocol (PDCP) specification".
- [30] 3GPP TS 38.331: "NR; Radio Resource Control (RRC); Protocol Specification".
- [31] 3GPP TS 38.413: "NG Radio Access Network (NG-RAN); NG Application Protocol (NGAP)".
- [31A] IEEE Std 802.3TM-2022: "Ethernet".
- [31AA] 3GPP TS 38.509: "Special conformance testing functions for User Equipment (UE)".
- [32] IETF RFC 768: "User Datagram Protocol".
- [33] IETF RFC 793: "Transmission Control Protocol."
- [33A] IETF RFC 3095: "RObust Header Compression (ROHC): Framework and four profiles: RTP, UDP, ESP and uncompressed".
- [33B] Void.
- [33C] Void.
- [33D] IETF RFC 8415: "Dynamic Host Configuration Protocol for IPv6 (DHCPv6)".
- [34] IETF RFC 3748: "Extensible Authentication Protocol (EAP)".
- [34A] IETF RFC 3843: "RObust Header Compression (ROHC): A Compression Profile for IP".
- [35] Void.
- [35A] IETF RFC 4122: "A Universally Unique IDentifier (UUID) URN Namespace".
- [36] IETF RFC 4191: "Default Router Preferences and More-Specific Routes".
- [37] IETF RFC 7542: "The Network Access Identifier".
- [38] IETF RFC 4303: "IP Encapsulating Security Payload (ESP)".
- [38A] IETF RFC 4815: "RObust Header Compression (ROHC): Corrections and Clarifications to RFC 3095".
- [38B] IETF RFC 4861: "Neighbor Discovery for IP version 6 (IPv6)".
- [39] IETF RFC 4862: "IPv6 Stateless Address Autoconfiguration".
- [39A] IETF RFC 5225: "RObust Header Compression (ROHC) Version 2: Profiles for RTP, UDP, IP, ESP and UDP Lite".
- [39B] IETF RFC 5795: "The RObust Header Compression (ROHC) Framework".
- [40] IETF RFC 5448: "Improved Extensible Authentication Protocol Method for 3rd Generation Authentication and Key Agreement (EAP-AKA')".
- [40A] IETF RFC 6603: "Prefix Exclude Option for DHCPv6-based Prefix Delegation".

- [40B] IETF RFC 6846: "RObust Header Compression (ROHC): A Profile for TCP/IP (ROHC-TCP)". [41] IETF RFC 7296: "Internet Key Exchange Protocol Version 2 (IKEv2)". [42] ITU-T Recommendation E.212: "The international identification plan for public networks and subscriptions", 2016-09-23. [43] IEEE Std 802-2014: "IEEE Standard for Local and Metropolitan Area Networks: Overview and Architecture" (30 June 2014). [43A] IEEE Std 802.1AS-2020: "IEEE Standard for Local and metropolitan area networks--Timing and Synchronization for Time-Sensitive Applications". [43B] IEEE Std 1588TM-2019: "IEEE Standard for a Precision Clock Synchronization Protocol for Networked Measurement and Control Systems". Void. [43C] [43D] Void. Void. [43E] [44] Void. Void. [45] [46] Void. Void. [47] [48] IEEE "Guidelines for Use of Extended Unique Identifier (EUI), Organizationally Unique Identifier (OUI), and Company ID (CID)". [49] BBF TR-069: "CPE WAN Management Protocol". [50] BBF TR-369: "User Services Platform (USP)". 3GPP TS 37.340: "Evolved Universal Terrestrial Radio Access (E-UTRA) and NR; Multi-[51] connectivity; Stage 2". [52] IETF RFC 8106:"IPv6 Router Advertisement Options for DNS Configuration". 3GPP TS 23.247: "Architectural enhancements for 5G multicast-broadcast services; Stage 2". [53] [54] 3GPP TS 23.380: "IMS Restoration Procedures". IETF RFC 3948: "UDP Encapsulation of IPsec ESP Packets". [55] 3GPP TS 33.503: "Security Aspects of Proximity based Services (ProSe) in the 5G System [56] (5GS)". 3GPP TS 33.246: "Security of Multimedia Broadcast/Multicast Service (MBMS)". [57] 3GPP TS 38.321: "NR; Medium Access Control (MAC); Protocol specification". [58]
- [59] IETF RFC 4291:" IP Version 6 Addressing Architecture".

3 Definitions and abbreviations

3.1 Definitions

For the purposes of the present document, the terms and definitions given in 3GPP TR 21.905 [1] and the following apply. A term defined in the present document takes precedence over the definition of the same term, if any, in 3GPP TR 21.905 [1].

5GMM-IDLE mode: In this specification, if the term is used standalone, a UE in 5GMM-IDLE mode means the UE can be either in 5GMM-IDLE mode over 3GPP access or in 5GMM-IDLE mode over non-3GPP access.

5GMM-CONNECTED mode: In this specification, if the term is used standalone, a UE in 5GMM-CONNECTED mode means the UE can be either in 5GMM-CONNECTED mode over 3GPP access or in 5GMM-CONNECTED mode over non-3GPP access.

5GMM-IDLE mode over 3GPP access: A UE is in 5GMM-IDLE mode over 3GPP access when no N1 NAS signalling connection between the UE and network over 3GPP access exists. The term 5GMM-IDLE mode over 3GPP access used in the present document corresponds to the term CM-IDLE state for 3GPP access used in 3GPP TS 23.501 [8].

5GMM-CONNECTED mode over 3GPP access: A UE is in 5GMM-CONNECTED mode over 3GPP access when an N1 NAS signalling connection between the UE and network over 3GPP access exists. The term 5GMM-CONNECTED mode over 3GPP access used in the present document corresponds to the term CM-CONNECTED state for 3GPP access used in 3GPP TS 23.501 [8].

5GMM-IDLE mode over non-3GPP access: A UE is in 5GMM-IDLE mode over non-3GPP access when no N1 NAS signalling connection between the UE and network over non-3GPP access exists. The term 5GMM-IDLE mode over non-3GPP access used in the present document corresponds to the term CM-IDLE state for non-3GPP access used in 3GPP TS 23.501 [8].

5GMM-CONNECTED mode over non-3GPP access: A UE is in 5GMM-CONNECTED mode over non-3GPP access when an N1 NAS signalling connection between the UE and network over non-3GPP access exists. The term 5GMM-CONNECTED mode over non-3GPP access used in the present document corresponds to the term CM-CONNECTED state for non-3GPP access used in 3GPP TS 23.501 [8].

5GS services: Services provided by PS domain. Within the context of this specification, 5GS services is used as a synonym for EPS services.

5G-EA: 5GS encryption algorithms. The term 5G-EA, 5G-EA0, 128-5G-EA1, 128-5G-EA2, 128-5G-EA3, 5G-EA4, 5G-EA5, 5G-EA6 and 5G-EA7 used in the present document corresponds to the term NEA, NEA0, 128-NEA1, 128-NEA2, 128-NEA3, NEA4, NEA5, NEA6 and NEA7 defined in 3GPP TS 33.501 [24].

5G-IA: 5GS integrity algorithms. The term 5G-IA, 5G-IA0, 128-5G-IA1, 128-5G-IA2, 128-5G-IA3, 5G-IA4, 5G-IA5, 5G-IA6 and 5G-IA7 used in the present document corresponds to the term NIA, NIA0, 128-NIA1, 128-NIA2, 128-NIA3, NIA4, NIA5, NIA6 and NIA7 defined in 3GPP TS 33.501 [24].

Access stratum connection: A peer to peer access stratum connection:

- between the UE and the NG-RAN for 3GPP access;
- between the UE and the N3IWF for untrusted non-3GPP access;
- between the UE and the TNGF for trusted non-3GPP access used by the UE;
- within the TWIF acting on behalf of the N5CW device for trusted non-3GPP access used by the N5CW device;
- between the 5G-RG and the W-AGF for wireline access used by the 5G-RG;
- within the W-AGF acting on behalf of the FN-RG for wireline access used by the FN-RG; or
- within the W-AGF acting on behalf of the N5GC device for wireline access used by the N5GC device.

The access stratum connection for 3GPP access corresponds to an RRC connection via the Uu reference point. The creation of the access stratum connection for untrusted non-3GPP access corresponds to the completion of the IKE_SA_INIT exchange (see IETF RFC 7296 [41]) via the NWu reference point. The creation of the access stratum connection for trusted non-3GPP access used by the UE corresponds to the UE reception of an EAP-request/5G-start via NWt reference point (see 3GPP TS 23.502 [9]). The creation of the access stratum connection for trusted non-3GPP access used by the VIF's start of acting on behalf of the N5CW device. The creation of the access stratum connection for wireline access used by the 5G-RG corresponds to the 5G-RG reception of an EAP-request/5G-packet over the W-CP EAP connection via the Y4 reference point (see 3GPP TS 23.316 [6D]). The creation of the access stratum connection for wireline access used by the FN-RG corresponds to the W-AGF's start of acting on behalf of the FN-RG. The creation of the access stratum connection for wireline access stratum connection for wireline access used by the FN-RG corresponds to the W-AGF's start of acting on behalf of the FN-RG. The creation of the access stratum connection for wireline access used by the FN-RG corresponds to the W-AGF's start of acting on behalf of the FN-RG. The creation of the access stratum connection for wireline access used by the N5GC device.

Access to SNPN services via a PLMN/To access SNPN services via a PLMN: A UE is accessing SNPN services via a PLMN when the UE is connecting to the 5GCN of the SNPN using the 3GPP access of the PLMN.

Aggregate maximum bit rate: The maximum bit rate that limits the aggregate bit rate of a set of non-GBR bearers of a UE. Definition derived from 3GPP TS 23.501 [8].

Always-on PDU session: A PDU session for which user-plane resources have to be established during every transition from 5GMM-IDLE mode to 5GMM-CONNECTED mode. A UE requests a PDU session to be established as an always-on PDU session based on indication from upper layers and the network decides whether a PDU session is established as an always-on PDU session.

NOTE 1: How the upper layers in the UE are configured to provide an indication is outside the scope of the present document.

Applicable UE radio capability ID for the current UE radio configuration in the selected network: The UE has an applicable UE radio capability ID for the current UE radio configuration in the selected network if:

- a) the UE supports RACS; and
- b) the UE has:
 - 1) a stored network-assigned UE radio capability ID which is associated with the PLMN ID or SNPN identity of the serving network and which maps to the set of radio capabilities currently enabled at the UE; or
 - 2) a manufacturer-assigned UE radio capability ID which maps to the set of radio capabilities currently enabled at the UE.

CAG cell: A cell in which only members of the CAG can get normal service. Depending on local regulation, the CAG cell can provide emergency services also to subscribers who are not members of the CAG.

CAG-ID: A CAG-ID is a unique identifier within the scope of one PLMN defined in 3GPP TS 23.003 [4] which identifies a Closed Access Group (CAG) in the PLMN associated with a cell or group of cells to which access is restricted to members of the CAG.

CAG restrictions: Restrictions applied to a UE in accessing a PLMN's 5GCN via:

- a) a non-CAG cell if the entry for the PLMN in the UE's "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells"; or
- b) a CAG cell if none of the CAG-ID(s) supported by the CAG cell is included in the "allowed CAG list" for the PLMN in the UE's "CAG information list".

The CAG restrictions are not applied in a PLMN when a UE accesses the PLMN due to emergency services.

Cleartext IEs: Information elements that can be sent without confidentiality protection in initial NAS messages as specified in subclause 4.4.6.

Configuration of SNPN subscription parameters in PLMN via the user plane: Configuration of a UE in a PLMN with one or more entries of the "list of subscriber data" via the user plane.

Control plane CIoT 5GS optimization: Signalling optimizations to enable efficient transport of user data (IP, Ethernet, Unstructured or SMS) over control plane via the AMF including optional header compression of IP data and Ethernet data.

Current TAI: A TAI of a selected PLMN broadcast in the cell on which the UE is camping. If the cell is a satellite NG-RAN cell broadcasting multiple TACs of the selected PLMN, the UE NAS layer selects the current TAI from these multiple TACs of the selected PLMN as specified in subclause 4.23.5.

NOTE 2: For the purpose of this definition, the selected PLMN can either be the registered PLMN or a PLMN selected according to PLMN selection rules as specified in 3GPP TS 23.122 [5].

DNN determined by the AMF: If no DNN requested by the UE is provided, a DNN determined by the AMF based subscription information or local policy. Otherwise DNN determined by the AMF is the DNN requested by the UE.

DNN requested by the UE: A DNN explicitly requested by the UE and included in a NAS request message.

DNN selected by the network: If DNN replacement applies, a DNN selected and indicated to the AMF by PCF. Otherwise DNN selected by the network is the DNN determined by the AMF.

Default S-NSSAI: An S-NSSAI in the subscribed S-NSSAIs marked as default.

Globally-unique SNPN identity: An SNPN identity with an NID whose assignment mode is not set to 1 (see 3GPP TS 23.003 [4]).

HPLMN S-NSSAI: An S-NSSAI applicable in the HPLMN without any further mapping by the network. If the UE has an EHPLMN list which is not empty, then the HPLMN S-NSSAIs are applicable without any further mapping in the PLMN derived from the IMSI, regardless of whether or not this PLMN is included in the EHPLMN list.

The UE considers as HPLMN S-NSSAIs at least the following S-NSSAIs:

- a) any S-NSSAI included in the configured NSSAI or allowed NSSAI for a PLMN or SNPN if it is provided by
 - 1) the HPLMN, if the EHPLMN list is not present or is empty;
 - 2) the PLMN whose PLMN code is derived from the IMSI, if the EHPLMN list is available and not empty, regardless of whether or not the PLMN code derived from the IMSI is included in the EHPLMN list; or
 - 3) the subscribed SNPN;
- b) any S-NSSAI provided as mapped S-NSSAI for the configured NSSAI or allowed NSSAI for a PLMN or SNPN;
- c) any S-NSSAI associated with a PDU session if there is no mapped S-NSSAI associated with the PDU session and the UE is
 - 1) in the HPLMN, if the EHPLMN list is not present or is empty;
 - 2) the PLMN whose PLMN code is derived from the IMSI, if the EHPLMN list is available and not empty, regardless of whether or not the PLMN code derived from the IMSI is included in the EHPLMN list; or
 - 3) in the subscribed SNPN; and
- d) any mapped S-NSSAI associated with a PDU session.
- NOTE 3: The above list is not intended to be complete. E.g., also in case of PLMN the S-NSSAIs included in URSP rules or in the signalling messages for network slice-specific authentication and authorization are HPLMN S-NSSAIs.

User plane CIoT 5GS optimization: Signalling optimizations to enable efficient transport of user data (IP, Ethernet or Unstructured) over the user plane.

UE supporting CIoT 5GS optimizations: A UE that supports control plane CIoT 5GS optimization or user plane CIoT 5GS optimization and one or more other CIoT 5GS optimizations when the UE is in N1 mode.

Registered for 5GS services with control plane CIOT 5GS optimization: A UE supporting CIOT 5GS optimizations is registered for 5GS services, and control plane CIOT 5GS optimization along with one or more other CIOT 5GS optimizations have been accepted by the network.

Registered for 5GS services with user plane CIoT 5GS optimization: A UE supporting CIoT 5GS optimizations is registered for 5GS services, and user plane CIoT 5GS optimization along with one or more other CIoT 5GS optimizations have been accepted by the network.

Registered for 5GS services with CIoT 5GS optimization: A UE is registered for 5GS services with control plane CIoT 5GS optimization or registered for 5GS services with user plane CIoT 5GS optimization.

DNN based congestion control: Type of congestion control at session management level that is applied to reject session management requests from UEs or release PDU sessions when the associated DNN is congested. DNN based congestion control can be activated at the SMF over session management level and also activated at the AMF over mobility management level.

Emergency PDU session: A PDU session established with the request type "initial emergency request" or "existing emergency PDU session".

General NAS level congestion control: Type of congestion control at mobility management level that is applied at a general overload or congestion situation in the network, e.g. lack of processing resources.

Initial NAS message: A NAS message is considered as an initial NAS message, if this NAS message can trigger the establishment of an N1 NAS signalling connection. For instance, the REGISTRATION REQUEST message is an initial NAS message.

Initial registration for emergency services: A registration performed with 5GS registration type "emergency registration" in the REGISTRATION REQUEST message.

Initial registration for onboarding services in SNPN: A registration performed with 5GS registration type "SNPN onboarding registration" in the REGISTRATION REQUEST message.

Initial registration for disaster roaming services: A registration performed with 5GS registration type "disaster roaming initial registration" in the REGISTRATION REQUEST message.

Last visited registered TAI: A TAI which is contained in the registration area that the UE registered to the network and which identifies the tracking area last visited by the UE. If the cell is a satellite NG-RAN cell broadcasting multiple TAIs, a TAI which is contained in the registration area that the UE registered to the network and last selected by the UE as the current TAI.

Mapped 5G-GUTI: A 5G-GUTI which is mapped from a 4G-GUTI previously allocated by an MME. Mapping rules are defined in 3GPP TS 23.003 [4].

Mapped S-NSSAI: An S-NSSAI in the subscribed S-NSSAIs for the HPLMN or the subscribed SNPN, to which an S-NSSAI of the registered PLMN (in case of a roaming scenario) or the registered non-subscribed SNPN is mapped.

Mobility registration for disaster roaming services: A registration performed with 5GS registration type "disaster roaming mobility registration updating" in the REGISTRATION REQUEST message.

MUSIM UE: A UE with multiple valid USIMs, capable of initiating and maintaining simultaneous separate registration states over 3GPP access with PLMN(s) using identities and credentials associated with those USIMs and supporting one or more of the N1 NAS signalling connection release, the paging indication for voice services, the reject paging request, the paging restriction and the paging timing collision control (see 3GPP TS 23.501 [8]).

N1 mode: A mode of a UE allowing access to the 5G core network via the 5G access network.

Native 5G-GUTI: A 5G-GUTI previously allocated by an AMF.

Non 5G capable over WLAN (N5CW) device: A device that is not capable to operate as a UE supporting NAS signalling with the 5GCN over a WLAN access network. However, this device may be capable to operate as a UE supporting NAS signalling with 5GCN using the N1 reference point as specified in this specification over 3GPP access. An N5CW device may be allowed to access the 5GCN via trusted WLAN access network (TWAN) that supports a trusted WLAN interworking function (TWIF) as specified in 3GPP TS 24.502 [18].

Non-CAG Cell: An NR cell which does not broadcast any Closed Access Group identity or an E-UTRA cell connected to 5GCN.

Non-globally-unique SNPN identity: An SNPN identity with an NID whose assignment mode is set to 1 (see 3GPP TS 23.003 [4]).

In NB-N1 mode: Indicates this paragraph applies only to a system which operates in NB-N1 mode. For a multi-access system this case applies if the current serving radio access network provides access to network services via E-UTRA connected to 5GCN by NB-IoT (see 3GPP TS 36.300 [25B], 3GPP TS 36.331 [25A], 3GPP TS 36.306 [25D]).

In WB-N1 mode: Indicates this paragraph applies only to a system which operates in WB-N1 mode. For a multi-access system this case applies if the system operates in N1 mode with E-UTRA connected to 5GCN, but not in NB-N1 mode.

In WB-N1/CE mode: Indicates this paragraph applies only when a UE, which is a CE mode B capable UE (see 3GPP TS 36.306 [25D]), is operating in CE mode A or B in WB-N1 mode.

Initial small data rate control parameters: Parameters that, if received by the UE during the establishment of a PDU session, are used as initial parameters to limit the allowed data for the PDU session according to small data rate control after establishment of a PDU session as described in subclause 6.2.13. At expiry of the associated validity period, the initial small data rate control parameters are no longer valid and the small data rate control parameters apply.

Initial small data rate control parameters for exception data: Parameters corresponding to initial small data rate control parameters for small data rate control of exception data.

N1 NAS signalling connection: A peer to peer N1 mode connection between UE and AMF. An N1 NAS signalling connection is either the concatenation of an RRC connection via the Uu reference point and an NG connection via the N2 reference point for 3GPP access, or the concatenation of an IPsec tunnel via the NWu reference point and an NG connection via the N2 reference point for non-3GPP access.

N5CW device supporting 3GPP access: An N5CW device which supports acting as a UE in 3GPP access (i.e. which supports NAS over 3GPP access).

N6 PDU session: A PDU session established between the UE and the User Plane Function (UPF) for transmitting the UE's IP data, Ethernet data or Unstructured data related to a specific application.

NEF PDU session: A PDU session established between the UE and the Network Exposure Function (NEF) for transmitting the UE's Unstructured data related to a specific application.

Network slicing information: information stored at the UE consisting of one or more of the following:

- a) default configured NSSAI for PLMN or SNPN;
- b) configured NSSAI for a PLMN or an SNPN;
- b1)NSSRG information for the configured NSSAI for a PLMN or an SNPN;
- c) mapped S-NSSAI(s) for the configured NSSAI for a PLMN or an SNPN;
- d) pending NSSAI for a PLMN or an SNPN;
- e) mapped S-NSSAI(s) for the pending NSSAI for a PLMN or an SNPN;
- f) rejected NSSAI for the current PLMN or SNPN;
- g) mapped S-NSSAI(s) for the rejected NSSAI for the current PLMN or an SNPN;
- h) rejected NSSAI for the failed or revoked NSSAA;
- i) for each access type:
 - 1) allowed NSSAI for a PLMN or an SNPN;
 - 2) mapped S-NSSAI(s) for the allowed NSSAI for a PLMN;
 - 3) rejected NSSAI for the current registration area;
 - 4) mapped S-NSSAI(s) for the rejected NSSAI for the current registration area;
 - 5) rejected NSSAI for the maximum number of UEs reached; and
 - 6) mapped S-NSSAI(s) for the rejected NSSAI for the maximum number of UEs reached; and
- j) for 3GPP access type:
 - 1) NSAG information for the configured NSSAI for a PLMN or an SNPN.

Non-cleartext IEs: Information elements that are not cleartext IEs.

Non-emergency PDU session: Any PDU session which is not an emergency PDU session.

Onboarding SUCI: SUCI derived from onboarding SUPI.

Onboarding SUPI: SUPI with the SUPI format "network specific identifier" containing a network specific identifier or with the SUPI format "IMSI" containing an IMSI, derived by a UE in SNPN access mode, from default UE credentials for primary authentication and used to identify the UE during initial registration for onboarding services in SNPN and while registered for onboarding services in SNPN.

PDU address: An IP address assigned to the UE by the packet data network.

PDU session for LADN: A PDU session with a DNN associated with a LADN.

PDU session with suspended user-plane resources: A PDU session for which user-plane resources were established or re-established, and for which data radio bearers were suspended when transition to 5GMM-CONNECTED mode with RRC inactive indication.

Persistent PDU session: either a non-emergency PDU session contains a GBR QoS flow with QoS equivalent to QoS of teleservice 11 and where there is a radio bearer associated with that PDU session over 3GPP access, or an emergency PDU session where there is a radio bearer associated with that PDU session over 3GPP access.

NOTE 4: An example of a persistent PDU session is a non-emergency PDU session with 5QI = 1 where there is a radio bearer associated with that context.

Procedure transaction identity: An identity which is dynamically allocated by the UE for the UE-requested 5GSM procedures or allocated by the UE or the PCF for the UE policy delivery procedures. The procedure transaction identity is released when the procedure is completed but it should not be released immediately.

RAT frequency selection priority index: A parameter provided by the AMF to the NG-RAN via the N2 reference point. The AMF selects an RFSP index for a particular UE based on the subscribed RFSP index, the locally configured operator's policies, the allowed NSSAI and the UE context information, including the UE's usage setting, if received during the registration procedure. Definition derived from 3GPP TS 23.501 [8].

Registered for disaster roaming services: A UE is considered as "registered for disaster roaming services" when it has successfully completed initial registration or mobility registration for disaster roaming services.

Registered for emergency services: A UE is considered as "registered for emergency services" when it has successfully completed initial registration for emergency services.

Registered for onboarding services in SNPN: A UE is considered as "registered for onboarding services in SNPN" when it has successfully completed initial registration for onboarding services in SNPN. While registered for onboarding services in SNPN, services other than the onboarding services are not available.

Registered PLMN: The PLMN on which the UE performed the last successful registration. The identity of the registered PLMN (MCC and MNC) is provided to the UE within the GUAMI field of the 5G-GUTI.

Rejected NSSAI: Rejected NSSAI for the current PLMN or SNPN, rejected NSSAI for the current registration area, rejected NSSAI for the failed or revoked NSSAA or rejected NSSAI for the maximum number of UEs reached.

NOTE 5: Rejected NSSAI for the current PLMN or SNPN, rejected NSSAI for the current registration area or rejected NSSAI for the maximum number of UEs reached contains a set of S-NSSAI(s) associated with a PLMN identity or SNPN identity for the current PLMN or SNPN and in roaming scenarios also contains a set of mapped S-NSSAI(s) if available. Rejected NSSAI for the failed or revoked NSSAA only contains a set of S-NSSAI(s) associated with a PLMN identity or SNPN identity or SNPN.

Rejected NSSAI for the current PLMN or SNPN: A set of S-NSSAI(s) which was included in the requested NSSAI by the UE and is sent by the AMF with the rejection cause "S-NSSAI not available in the current PLMN or SNPN".

Rejected NSSAI for the current registration area: A set of S-NSSAI(s) which was included in the requested NSSAI by the UE and is sent by the AMF with the rejection cause "S-NSSAI not available in the current registration area".

Rejected NSSAI for the failed or revoked NSSAA: A set of S-NSSAI(s) which is sent by the AMF with the rejection cause "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization".

Rejected NSSAI for the maximum number of UEs reached: A set of S-NSSAI(s) which was included in the requested NSSAI by the UE and is sent by the AMF with the rejection cause "S-NSSAI not available due to maximum number of UEs reached".

Local release: Release of a PDU session without peer-to-peer signalling between the network and the UE.

NOTE 6: Local release can include communication among network entities.

Removal of eCall only mode restriction: All the limitations as described in 3GPP TS 22.101 [2] for the eCall only mode do not apply any more.

SNPN access operation mode: SNPN access mode or access to SNPN over non-3GPP access.
NOTE 7: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.

S-NSSAI based congestion control: Type of congestion control at session management level that is applied to reject session management requests from UEs or release PDU sessions when the associated S-NSSAI and optionally the associated DNN are congested. S-NSSAI based congestion control can be activated at the SMF over session management level and also activated at the AMF over mobility management level.

Satellite NG-RAN RAT type: In case of satellite NG-RAN access, RAT types are used to distinguish different types of satellite NG-RAN access, as defined in 3GPP TS 38.413 [31]. In this version of the specification, the defined satellite NG-RAN RAT types are "NR(LEO)", "NR(MEO)" and "NR(GEO)".

Selected core network type information: A type of core network (EPC or 5GCN) selected by the UE NAS layer in case of an E-UTRA cell connected to both EPC and 5GCN.

UE supporting UAS services: A UE which supports an aerial vehicle, such as a drone, with an onboard or built-in USIM and is able to perform UE NAS functionalities specified in this specification. Upper layers of the UE supporting UAS services are responsible for UAS related procedures, such as UUAA and C2 authorization, for which the NAS layer of the UE supporting UAS services performs the necessary NAS procedures.

UE configured for high priority access in selected PLMN: A UE configured with one or more access identities equal to 1, 2, or 11-15 applicable in the selected PLMN as specified in subclause 4.5.2. Definition derived from 3GPP TS 22.261 [3].

UE operating in single-registration mode in a network supporting N26 interface: A UE, supporting both N1 mode and S1 mode. During the last attach, tracking area update (see 3GPP TS 24.301 [15]) or registration procedures, the UE has received either a 5GS network feature support IE with IWK N26 bit set to "interworking without N26 interface not supported" or an EPS network feature support IE with IWK N26 bit set to "interworking without N26 interface not supported".

UE using 5GS services with control plane CIOT 5GS optimization: A UE that is registered for 5GS services with the control plane CIOT 5GS optimization accepted by the network.

User-plane resources: Resources established between the UE and the UPF. The user-plane resources consist of one of the following:

- user plane radio bearers via the Uu reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for 3GPP access;
- IPsec tunnels via the NWu reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for untrusted non-3GPP access;
- IPsec tunnels via the NWt reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for trusted non-3GPP access used by the UE;
- a layer-2 connection via the Yt reference point, a layer-2 or layer-3 connection via the Yw reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for trusted non-3GPP access used by the N5CW device;
- W-UP resources via Y4 reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for wireline access used by the 5G-RG; and
- L-W-UP resources via Y5 reference point, a tunnel via the N3 reference point and a tunnel via the N9 reference point (if any) for wireline access used by the FN-RG.

W-AGF acting on behalf of the N5GC device: A W-AGF that enables an N5GC device behind a 5G-CRG or an FN-CRG to connect to the 5G Core.

For the purposes of the present document, the following terms and definitions given in 3GPP TS 22.261 [3] apply:

Non-public network Disaster Roaming satellite NG-RAN

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.003 [4] apply:

5G-GUTI 5G-S-TMSI 5G-TMSI Global Line Identifier (GLI) Global Cable Identifier (GCI) GUAMI IMEI IMEISV IMSI PEI SUPI SUCI

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.122 [5] apply:

37

CAG selection Country EHPLMN HPLMN **Onboarding services in SNPN Registered SNPN** Selected PLMN Selected SNPN **Shared network SNPN** identity Steering of Roaming (SOR) Steering of roaming connected mode control information (SOR-CMCI) **Steering of Roaming information** Subscribed SNPN Suitable cell **VPLMN**

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.167 [6] apply:

eCall over IMS

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.216 [6A] apply:

SRVCC

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.401 [7] apply:

eCall only mode

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.501 [8] apply:

5G access network 5G core network 5G QoS flow **5G QoS identifier** 5G-RG 5G-BRG 5G-CRG **5G System** Allowed area Allowed NSSAI **AMF** region AMF set **Closed access group Configured NSSAI Credentials Holder (CH) Default Credentials Server (DCS)** Group ID for Network Selection (GIN) IAB-node

Local area data network Network identifier (NID) **Network slice** NG-RAN Non-allowed area **Onboarding Standalone Non-Public Network** PDU connectivity service PDU session PDU session type Pending NSSAI **Requested NSSAI Routing Indicator** Service data flow **Service Gap Control** Serving PLMN rate control Small data rate control status **SNPN** access mode **SNPN enabled UE** Stand-alone Non-Public Network **Time Sensitive Communication Time Sensitive Communication and Time Synchronization Function UE-DS-TT residence time UE-Slice-MBR** UE presence in LADN service area

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.503 [10] apply:

UE local configuration

For the purposes of the present document, the following terms and definitions given in 3GPP TS 24.008 [12] apply:

GMM MM A/Gb mode Iu mode GPRS Non-GPRS

For the purposes of the present document, the following terms and definitions given in 3GPP TS 24.301 [15] apply:

CIoT EPS optimization Control plane CIoT EPS optimization EENLV EMM **EMM-DEREGISTERED EMM-DEREGISTERED-INITIATED EMM-IDLE mode EMM-NULL EMM-REGISTERED EMM-REGISTERED-INITIATED EMM-SERVICE-REQUEST-INITIATED EMM-TRACKING-AREA-UPDATING-INITIATED** EPS **EPS** security context **EPS** services Lower layer failure Megabit Message header NAS signalling connection recovery Native GUTI NB-S1 mode **Non-EPS** services S1 mode

User plane CIoT EPS optimization WB-S1 mode

For the purposes of the present document, the following terms and definitions given in 3GPP TS 33.501 [24] apply:

5G security context 5G NAS security context ABBA Current 5G NAS security context Default UE credentials for primary authentication Default UE credentials for secondary authentication Full native 5G NAS security context K'_{AME} **K**AMF **K**_{ASME} Mapped 5G NAS security context Mapped security context Native 5G NAS security context NCC Non-current 5G NAS security context Partial native 5G NAS security context **RES***

For the purposes of the present document, the following terms and definitions given in 3GPP TS 38.413 [31] apply:

NG connection User Location Information

For the purposes of the present document, the following terms and definitions given in 3GPP TS 24.587 [19B] apply:

E-UTRA-PC5 NR-PC5 V2X

For the purposes of the present document, the following terms and its definitions given in 3GPP TS 23.256 [6AB] apply:

3GPP UAV ID CAA (Civil Aviation Administration)-Level UAV Identity Command and Control (C2) Communication UAV controller (UAV-C) UAS Services UAS Service Supplier (USS) Uncrewed Aerial System (UAS) USS communication UUAA UUAA-MM UUAA-SM

For the purposes of the present document, the following terms and definitions given in 3GPP TS 24.554 [19E] apply:

5G ProSe

For the purposes of the present document, the following terms and definitions given in 3GPP TS 23.548 [10A] apply:

Edge Application Server Edge DNS Client

For the purposes of the present document, the following terms and definitions given in 3GPP TS 24.526 [19] apply:

Non-subscribed SNPN signalled URSP

3.2 Abbreviations

For the purposes of the present document, the abbreviations given in 3GPP TR 21.905 [1] and the following apply. An abbreviation defined in the present document takes precedence over the definition of the same abbreviation, if any, in 3GPP TR 21.905 [1].

4G-GUTI	4G-Globally Unique Temporary Identifier
5GCN	5G Core Network
5G-GUTI	5G-Globally Unique Temporary Identifier
5GMM	5GS Mobility Management
5G-RG	5G Residential Gateway
5G-BRG	5G Broadband Residential Gateway
5G-CRG	5G Cable Residential Gateway
5GS	5G System
5GSM	5GS Session Management
5G-S-TMSI	5G S-Temporary Mobile Subscription Identifier
5G-TMSI	5G Temporary Mobile Subscription Identifier
50I	5G OoS Identifier
ACS	Auto-Configuration Server
AKA	Authentication and Key Agreement
AKMA	Authentication and Key Management for Applications
A-KID	AKMA Key Identifier
A-TID	AKMA Temporary Identifier
AMBR	Aggregate Maximum Bit Rate
AMF	Access and Mobility Management Function
APN	Access Point Name
AS	Access stratum
ATSSS	Access Traffic Steering Switching and Splitting
AUSE	Authentication Server Function
CAG	Closed access group
CGI	Cell Global Identity
CHAP	Challenge Handshake Authentication Protocol
CP-PRUK	Control Plane ProSe Remote User Key
DDX	Downlink Data Expected
DI	Downlink Data Expected
DN	Data Network
DNN	Data Network Name
DNS	Domain Name System
eDRX	Extended DRX cycle
DS-TT	Device-Side TSN Translator
FII	Extended Unique Identifier
E-UTRAN	Evolved Universal Terrestrial Radio Access Network
FAC	Farly Admission Control
FAP-AKA'	Improved Extensible Authentication Protocol method for 3rd generation Authentication and Key
	A greement
FAS	Edge Application Server
FASDE	Edge Application Server Discovery Function
FCIES	Elliptic Curve Integrated Encryption Scheme
FCS	Edge Configuration Server
FCSP	Edge Computing Service Provider
FDC	Edge DNS Client
FFC	Edge Enabler Client
FPD	Extended Protocol Discriminator
ENM	EPS Mobility Management
FPC	Evolved Packet Core Network
FPS	Evolved Packet System
FPS_UPIP	User-plane integrity protection in FPS
FSM	FPS Session Management
FN-RG	Fixed Network RG
FN_RRG	Fixed Network Broadband RG
FN_CRG	Fixed Network Cable RG

Gbps	Gigabits per second
GEO	Geostationary Orbit
GFBR	Guaranteed Flow Bit Rate
GUAMI	Globally Unique AMF Identifier
IAB	Integrated access and backhaul
IMEI	International Mobile station Equipment Identity
IMEISV	International Mobile station Equipment Identity and Software Version number
IMSI	International Mobile Subscriber Identity
IP-CAN	IP-Connectivity Access Network
KSI	Key Set Identifier
LADN	Local Area Data Network
LCS	LoCation Services
LEO	Low Earth Orbit
LMF	Location Management Function
LPP	LTE Positioning Protocol
MAC	Message Authentication Code
MA PDU	Multi-Access PDU
MRS	Multicast/Broadcast Services
Mbps	Megabits per second
MCS	Mission Critical Service
MEO	Madium Earth Orbit
MEDD	Movimum Flow Bit Poto
MICO	Mahila Initiated Connection Only
MINT	Minimization of Somilas Intermution
MDS	Multimedia Driarity Comica
MPS MSV	MDS Service Key
MSK	MBS Service Key
	MBS Ifainc Key
MUSIM	Multi-USIM
N3IWF	Non-3GPP Inter-Working Function
N5CW	Non-5G-Capable over WLAN
N5GC	Non-5G Capable
NAI	Network Access Identifier
NITZ	Network Identity and Time Zone
ngKSI	Key Set Identifier for Next Generation Radio Access Network
NPN	Non-public network
NR	New Radio
NSAC	Network Slice Admission Control
NSACF	Network Slice Admission Control Function
NSAG	Network slice AS group
NSSAA	Network slice-specific authentication and authorization
NSSAAF	Network Slice-Specific and SNPN authentication and authorization Function
NSSAI	Network Slice Selection Assistance Information
NSSRG	Network Slice Simultaneous Registration Group
NSWO	Non-Seamless WLAN Offload
ON-SNPN	Onboarding Standalone Non-Public Network
OS	Operating System
OS Id	OS Identity
PAP	Password Authentication Protocol
PCO	Protocol Configuration Option
PEI	Permanent Equipment Identifier
PEIPS	Paging Early Indication with Paging Subgrouping
PNI-NPN	Public Network Integrated Non-Public Network
ProSe	Proximity based Services
ProSeP	5G ProSe policy
PTI	Procedure Transaction Identity
PVS	Provisioning Server
OFI	OoS Flow Identifier
008	Quality of Service
CBI Kon	Oos Pula Identifier
2M DACS	Que ruie lucililier Dadia Canability Signalling Optimisation
(D)AN	Radio) Access Network
(K)AIN DESD	(Raulo) Access Incluork
кгэг	KAT Frequency Selection Priority

RG	Residential Gateway
RPLMN	Registered PLMN
RQA	Reflective QoS Attribute
RQI	Reflective QoS Indication
RSC	Relay Service Code
RSN	Redundancy Sequence Number
RSNPN	Registered SNPN
S-NSSAI	Single NSSAI
SA	Security Association
SDF	Service Data Flow
SDT	Small Data Transmission
SMF	Session Management Function
SGC	Service Gap Control
SNN	Serving Network Name
SNPN	Stand-alone Non-Public Network
SOR	Steering of Roaming
SOR-CMCI	Steering of Roaming Connected Mode Control Information
SUCI	Subscription Concealed Identifier
SUPI	Subscription Permanent Identifier
TA	Tracking Area
TAC	Tracking Area Code
TAI	Tracking Area Identity
Tbps	Terabits per second
TMGI	Temporary Mobile Group Identity
TNGF	Trusted Non-3GPP Gateway Function
TSC	Time Sensitive Communication
TSCTSF	Time Sensitive Communication and Time Synchronization Function
TWIF	Trusted WLAN Interworking Function
TSN	Time-Sensitive Networking
UAS	Uncrewed Aerial System
UAV	Uncrewed Aerial Vehicle
UAV-C	Uncrewed Aerial Vehicle-Controller
UDM	Unified Data Management
UL	Unlink
UPDS	UE policy delivery service
UPF	User Plane Function
UP-PRUK	User Plane ProSe Remote User Key
UPSC	UE Policy Section Code
UPSI	UE Policy Section Identifier
URN	Uniform Resource Name
URSP	UE Route Selection Policy
USS	UAS Service Supplier
UUAA	USS UAV Authorization/Authentication
V2X	Vehicle-to-Everything
V2XP	V2X policy
W-AGF	Wireline Access Gateway Function
WLAN	Wireless Local Area Network
WUS	Wake-un signal
	There of Biblion

4 General

4.1 Overview

The non-access stratum (NAS) described in the present document forms the highest stratum of the control plane between UE and AMF (reference point "N1" see 3GPP TS 23.501 [8]) for both 3GPP and non-3GPP access.

Main functions of the protocols that are part of the NAS are:

- support of mobility of the user equipment (UE) including also common procedures such as authentication, identification, generic UE configuration update and security mode control procedures;
- support of session management procedures to establish and maintain data connectivity between the UE and the data network; and
- NAS transport procedure to provide a transport of SMS, LPP, LCS, UE policy container, SOR transparent container and UE parameters update information payload.

Principles for the handing of 5GS security contexts and for the activation of ciphering and integrity protection, when a NAS signalling connection is established, are provided in subclause 4.4.

For the support of the above functions, the following procedures are supplied within this specification:

- elementary procedures for 5GS mobility management in clause 5; and
- elementary procedures for 5GS session management in clause 6.

Signalling procedures for the control of NAS security are described as part of the 5GMM common procedures in subclause 5.4.

Complete NAS transactions consist of specific sequences of elementary procedures. Examples of such specific sequences can be found in 3GPP TS 23.502 [9].

The NAS for 5GS follows the protocol architecture model for layer 3 as described in 3GPP TS 24.007 [11].

4.2 Coordination between the protocols for 5GS mobility management and 5GS session management

A 5GS session management (5GSM) message is piggybacked in specific 5GS mobility management (5GMM) transport messages. To this purpose, the 5GSM messages can be transmitted in an information element in the 5GMM transport messages. In this case, the UE, the AMF and the SMF execute the 5GMM procedure and the 5GSM procedure in parallel. The success of the 5GMM procedure is not dependent on the success of the piggybacked 5GSM procedure.

The UE can only initiate the 5GSM procedure when there is a 5GMM context established at the UE.

During 5GMM procedures, the UE and the AMF shall suspend the transmission of 5GSM messages, except when:

- a) the 5GMM procedure is piggybacking 5GSM messages; or
- b) the UE is in 5GMM-CONNECTED mode and a service request procedure for re-establishing user-plane resources of PDU session(s) is initiated without including PDU session status IE or Allowed PDU session status IE. In this case, the UE and the AMF need not suspend the transmission of 5GSM messages related to other PDU session(s) than the one(s) for which the user- plane resources re-establishment is requested.

If the UE determines to locally release the N1 NAS signalling connection upon receiving an SOR transparent container during a registration procedure as specified in 3GPP TS 23.122 [5] Annex C.2, the UE shall suspend the transmission of 5GSM messages after sending the REGISTRATION COMPLETE message and until the N1 NAS signalling connection is released to obtain service on a higher priority PLMN, with the exception of the case when the UE has an emergency PDU session.

A 5GMM message piggybacking a 5GSM message for a PDU session shall be delivered via the access associated with the PDU session, if any, with the following exceptions:

- a) the AMF shall send, via 3GPP access, a DL NAS TRANSPORT message piggybacking a downlink 5GSM message of a network-requested 5GSM procedure for a PDU session associated with non-3GPP access if the conditions specified in subclause 5.5.1.3.4 or subclause 5.6.1.4 are met;
- b) the UE shall send an UL NAS TRANSPORT message piggybacking a response message to the 5GSM message described in a) via either:
 - 1) 3GPP access; or
 - 2) non-3GPP access if the UE is in 5GMM-CONNECTED mode over non-3GPP access; and
- NOTE: The interaction between the 5GMM sublayer and the 5GSM sublayer to enable the UE to send the UL NAS TRANSPORT message containing the response message via 3GPP access is required. This is achieved via UE implementation.
- c) the UE shall send, via the target access, an UL NAS TRANSPORT message piggybacking a 5GSM message associated with a request type set to "existing PDU session" or "existing emergency PDU session" for handover of an existing PDU session between 3GPP access and non-3GPP access.

A 5GMM message piggybacking a 5GSM message as a response message to a request message associated with an MA PDU session, shall be delivered via the same access that the initial message was received.

4.3 UE domain selection

4.3.1 UE's usage setting

The UE's usage setting defined in 3GPP TS 24.301 [15] applies to voice capable UEs in 5GS and indicates whether the UE has preference for voice services over data services or vice-versa, where:

- a) voice services include IMS voice; and
- b) data services include any kind of user data transfer without a voice media component.

The UE's usage setting can be set to:

- a) "voice centric"; or
- b) "data centric".

If the UE is capable of S1 mode, there is a single UE's usage setting at the UE which applies to both 5GS and EPS.

4.3.2 Domain selection for UE originating sessions / calls

The behaviour of the UE for domain selection is determined by:

- a) the UE usage setting;
- b) the availability of IMS voice; and
- c) whether the UE operates in single-registration mode or dual-registration mode (see 3GPP TS 23.501 [8]).

In the present document the condition "the UE supports IMS voice over 3GPP access" evaluates to "true" if at least one of the following is fulfilled:

- 1) the UE supports IMS voice over NR connected to 5GCN;
- 2) the UE supports IMS voice over E-UTRA connected to 5GCN; or
- 3) the UE supports IMS voice in EPS.

In the present document the condition "the UE does not support IMS voice over 3GPP access" evaluates to "true" if the condition "the UE supports IMS voice over 3GPP access" evaluates to "false".

In the present document the condition "the UE supports IMS voice over non-3GPP access" evaluates to "true" if the UE supports IMS voice over non-3GPP access connected to 5GCN.

In the present document the condition "the UE does not support IMS voice over non-3GPP access" evaluates to "true" if the condition "the UE supports IMS voice over non-3GPP access" evaluates to "false".

In the present document, "IMS voice not available" is determined per access type independently, i.e. 3GPP access or non-3GPP access.

In the present document, "IMS voice not available" refers to one of the following conditions:

- a) the UE does not support IMS voice;
- b) the UE supports IMS voice, but the network indicates in the REGISTRATION ACCEPT message that IMS voice over PS sessions are not supported; or
- c) the UE supports IMS voice, the network indicates in the REGISTRATION ACCEPT message that IMS voice over PS sessions are supported, but the upper layers:
 - 1) provide no indication that the UE is available for voice call in the IMS within a manufacturer determined period of time; or
 - 2) indicate that the UE is not available for voice calls in the IMS.
- NOTE 1: If conditions a and b evaluate to false, the upper layers need time to attempt IMS registration. In the event an indication from the upper layers that the UE is available for voice calls in the IMS takes longer than the manufacturer determined period of time (e.g. due to delay when attempting IMS registration or due to delay in obtaining a QoS flow for SIP signalling), the NAS layer assumes the UE is not available for voice calls in the IMS.

Other conditions may exist but these are implementation specific.

In the present document, "IMS voice available" applies when "IMS voice not available" does not apply.

When IMS voice is not available over 3GPP access, if the UE's usage setting is "voice centric", the UE operates in single-registration mode, and the UE:

- a) does not have a persistent PDU session, and:
 - if the UE is only registered over 3GPP access, or if the UE is registered over both 3GPP access and non-3GPP access and IMS voice is not available over non-3GPP access, the UE shall disable the N1 mode capability for 3GPP access and proceed as specified in subclause 4.9.2 with modifications described below;. or
 - if the UE is registered over both 3GPP access and non-3GPP access and IMS voice is available over non-3GPP access, the UE may disable the N1 mode capability for 3GPP access and proceed as specified in subclause 4.9.2 with modifications described below; or
- b) has a persistent PDU session, then the UE waits until the radio bearer associated with the persistent PDU session has been released. When the radio bearer associated with the persistent PDU session has been released, then:
 - if the UE is only registered over 3GPP access, or if the UE is registered over both 3GPP access and non-3GPP access and IMS voice is not available over non-3GPP access, the UE shall disable the N1 mode capability for 3GPP access and proceed as specified in subclause 4.9.2 with modifications described below; or
 - 2) If the UE is registered over both 3GPP access and non-3GPP access and IMS voice is available over non-3GPP access, the UE may disable the N1 mode capability for 3GPP access and proceed as specified in subclause 4.9.2 with modifications described below.

The following modifications are applied to the procedure in subclause 4.9.2 for disabling the N1 mode capability for 3GPP access, if the UE's usage setting is "voice centric" and the UE operates in single-registration mode:

a) in item a) of subclause 4.9.2, the UE shall attempt to select an E-UTRA cell connected to EPC. If such a cell is found, the UE shall then perform voice domain selection procedures as defined in 3GPP TS 24.301 [15]; and

b) in item b) of subclause 4.9.2, if an E-UTRA cell connected to EPC cannot be found, the UE shall attempt to select another supported radio access technology which supports voice services.

When IMS voice is not available over non-3GPP access, if the UE's usage setting is "voice centric" and the UE operates in single-registration mode, then:

- a) if the UE is only registered over non-3GPP access, the UE shall disable the N1 mode capability for non-3GPP access (see subclause 4.9.3); or
- b) if the UE is registered over both 3GPP access and non-3GPP access and IMS voice is not available also over 3GPP access, the UE may disable the N1 mode capability for non-3GPP access (see subclause 4.9.3).
- NOTE 2: The UE can register over 3GPP access in another mode, e.g., S1 mode, for voice service, and in this case the UE can keep the N1 mode capability for non-3GPP access enabled.

4.3.3 Change of UE's usage setting

If the UE operates in single-registration mode, whenever the UE's usage setting changes, the UE shall execute procedures according to table 4.3.3.1:

UE's usage setting	Procedure to execute	
change		
From "data centric" to "voice centric" and "IMS voice not available" over 3GPP access only	Disable the N1 mode capability for 3GPP access (see subclause 4.9.2), if the UE is only registered over 3GPP access (NOTE)	
From "data centric" to "voice centric", and "IMS voice not available" over both 3GPP access and non-3GPP access	Disable the N1 mode capability for 3GPP access (see subclause 4.9.2) and non-3GPP access (see subclause 4.9.3), if the UE is registered over both 3GPP access and non-3GPP access Disable the N1 mode capability for 3GPP access (see subclause 4.9.2), if the UE is only registered over 3GPP access. Disable the N1 mode capability for non-3GPP access (see subclause 4.9.3), if the UE is only registered over non-3GPP access. (NOTE)	
From "voice centric" to "data centric" and the N1 mode capability for 3GPP access is disabled at the UE due to "IMS voice not available"	Re-enable the N1 mode capability for 3GPP access (see subclause 4.9.2)	
From "data centric" to "voice centric" and "IMS voice not available" over non-3GPP access only	Disable the N1 mode capability for non-3GPP access (see subclause 4.9.3), if the UE is only registered over non-3GPP access	
From "voice centric" to "data centric", and the N1 mode capability for non- 3GPP access is disabled at the UE due to "IMS voice not available"	Re-enable the N1 mode capability for non-3GPP access (see subclause 4.9.3)	
NOTE: If the UE is registered over 3GPP access and has a persistent PDU session, then the UE waits until the radio bearer associated with the persistent PDU session has been released		

Table 4.3.3.1: Change of UE's usage setting for a UE in single-registration mode

4.3.4 Change or determination of IMS voice availability

If the UE operates in single-registration mode, whenever the IMS voice availability is determined or changes, the UE shall execute procedures according to table 4.3.4.1:

Change of IMS voice	Procedure to execute
available condition	
"IMS voice not available"	Disable the N1 mode capability for 3GPP access, if the UE is only registered
over 3GPP access only	over 3GPP access (see subclause 4.9.2). (NOTE 2)
and the UE's usage	
setting is "voice centric"	
"IMS voice not available"	Disable the N1 mode capability for non-3GPP access (see subclause 4.9.3),
over non-3GPP access	if the UE is only registered over non-3GPP access. (NOTE 2)
only and the UE's usage	
setting is "voice centric"	
"IMS voice not available"	Disable the N1 mode capability for 3GPP access (see subclause 4.9.2) and
over both 3GPP access	non-3GPP access (see subclause 4.9.3), if the UE is registered over both
and non-3GPP access,	3GPP access and non-3GPP access.
and the UE's usage	Disable the N1 mode capability for 3GPP access (see subclause 4.9.2), if
setting is "voice centric"	the UE is only registered over 3GPP access
	Disable the N1 mode capability for non-3GPP access (see subclause 4.9.3),
	if the UE is only registered over non-3GPP access.
	(NOTE 1, NOTE 2)
NOTE 1: If the UE is regis	tered over 3GPP access and has a persistent PDU session, then the UE
waits until the ra	dio bearer associated with the persistent PDU session has been released.
NOTE 2: If the UE determines "IMS voice not available" upon receipt of a 5GS session management	
reject message i	ncluding a back-off timer value, and the re-attempt indicator indicates that the
UE is not allowe	d to re-attempt the procedure in S1 mode then, upon inter-system change
from N1 mode to	S1 mode, the UE proceeds as specified in 3GPP TS 24.301 [15],
subclause 4.3.2.	4, Change or determination of IMS registration status.

 Table 4.3.4.1: Change of IMS voice availability for a UE in single-registration mode

4.4 NAS security

4.4.1 General

This clause describes the principles for the handling of 5G NAS security contexts in the UE and in the AMF, the procedures used for the security protection of NAS messages between the UE and the AMF, and the procedures used for the protection of NAS IEs between the UE and the UDM. Security protection involves integrity protection and ciphering of the 5GMM messages. 5GSM messages are security protected indirectly by being piggybacked by the security protected 5GMM messages (i.e. UL NAS TRANSPORT message and the DL NAS TRANSPORT message).

The signalling procedures for the control of NAS security are part of the 5GMM protocol and are described in detail in clause 5.

NOTE: The use of ciphering in a network is an operator option. In this subclause, for the ease of description, it is assumed that ciphering is used, unless explicitly indicated otherwise. Operation of a network without ciphering is achieved by configuring the AMF so that it always selects the "null ciphering algorithm", 5G-EA0.

4.4.2 Handling of 5G NAS security contexts

4.4.2.1 General

The security parameters for authentication, integrity protection and ciphering are tied together in a 5G NAS security context and identified by a key set identifier (ngKSI). The relationship between the security parameters is defined in 3GPP TS 33.501 [24].

Before security can be activated, the AMF and the UE need to establish a 5G NAS security context. Usually, the 5G NAS security context is created as the result of a primary authentication and key agreement procedure between the AMF and the UE. A new 5G NAS security context may also be created during an N1 mode to N1 mode handover. Alternatively, during inter-system change from S1 mode to N1 mode, the AMF not supporting interworking without N26 and the UE operating in single-registration mode may derive a mapped 5G NAS security context from an EPS security context that has been established while the UE was in S1 mode.

The 5G NAS security context is taken into use by the UE and the AMF, when the AMF initiates a security mode control procedure, during an N1 mode to N1 mode handover, or during the inter-system change procedure from S1 mode to N1 mode. The 5G NAS security context which has been taken into use by the network most recently is called current 5G NAS security context. This current 5G NAS security context can be of type native or mapped, i.e. originating from a native 5G NAS security context or mapped 5G NAS security context.

The key set identifier ngKSI is assigned by the AMF either during the primary authentication and key agreement procedure or, for the mapped 5G NAS security context, during the inter-system change. The ngKSI consists of a value and a type of security context parameter indicating whether a 5G NAS security context is a native 5G NAS security context or a mapped 5G NAS security context. When the 5G NAS security context is a native 5G NAS security context, the ngKSI has the value of KSI_{AMF}, and when the current 5G NAS security context is of type mapped, the ngKSI has the value of KSI_{ASME}.

The 5G NAS security context which is indicated by an ngKSI can be taken into use to establish the secure exchange of NAS messages when a new N1 NAS signalling connection is established without executing a new primary authentication and key agreement procedure (see subclause 5.4.1) or when the AMF initiates a security mode control procedure. For this purpose, the initial NAS messages (i.e. REGISTRATION REQUEST, DEREGISTRATION REQUEST, SERVICE REQUEST and CONTROL PLANE SERVICE REQUEST) and the SECURITY MODE COMMAND message contain an ngKSI in the ngKSI IE indicating the current 5G NAS security context used to integrity protect the NAS message.

In the present document, when the UE is required to delete an ngKSI, the UE shall set the ngKSI to the value "no key is available" and consider also the associated keys K_{AMF} or K'_{AMF} , 5G NAS ciphering key and 5G NAS integrity key invalid (i.e. the 5G NAS security context associated with the ngKSI as no longer valid). In the initial registration procedure, when the key K_{AUSF} , is invalid, the UE shall delete the ngKSI.

NOTE: In some specifications the term ciphering key sequence number might be used instead of the term key set identifier (KSI).

As described in subclause 4.8 in order to interwork with E-UTRAN connected to EPC, the UE supporting both S1 mode and N1 mode can operate in either single-registration mode or dual-registration mode. A UE operating in dual-registration mode shall independently maintain and use both EPS security context (see 3GPP TS 24.301 [15]) and 5G NAS security context. When the UE operating in dual-registration mode performs an EPS attach procedure, it shall take into use an EPS security context and follow the handling of this security context as specified in 3GPP TS 24.301 [15]. However, when the UE operating in dual-registration mode performs an initial registration procedure, it shall take into use a 5G NAS security context and follow the handling of this security context as described in the present specification.

The UE and the AMF need to be able to maintain two 5G NAS security contexts simultaneously, i.e. a current 5G NAS security context and a non-current 5G NAS security context, since:

- a) after a 5G re-authentication, the UE and the AMF can have both a current 5G NAS security context and a noncurrent 5G NAS security context which has not yet been taken into use (i.e. a partial native 5G NAS security context); and
- b) after an inter-system change from S1 mode to N1 mode, the UE and the AMF can have both a mapped 5G NAS security context, which is the current 5G NAS security context, and a non-current native 5G NAS security context that was created during a previous access in N1 mode.

The number of 5G NAS security contexts that need to be maintained simultaneously by the UE and the AMF is limited by the following requirements:

- a) after a successful 5G (re-)authentication, which creates a new partial native 5G NAS security context, the AMF and the UE shall delete the non-current 5G NAS security context, if any;
- b) when a partial native 5G NAS security context is taken into use through a security mode control procedure, the AMF shall delete the previously current 5G NAS security context. If the UE does not support multiple records of NAS security context storage for multiple registration (see 3GPP TS 31.102 [22]), the UE shall delete the previously current 5G NAS security context. If the UE supports multiple records of NAS security context storage for multiple registration, the UE supports multiple records of NAS security context storage for multiple registration.
 - replace the previously current 5G NAS security context stored in the first 5G security context of that access (see 3GPP TS 31.102 [22]) with the new 5G security context (taken into use through a security mode control procedure), when the UE activates the new 5G security context for the same PLMN and access; or

- 2) store the previously current 5G NAS security context in the second 5G security context of that access (see 3GPP TS 31.102 [22]) and store the new 5G security context (taken into use through a security mode control procedure) in the first 5G security context, when the UE activates the new 5G security context for a different PLMN over that access but the previously current 5G NAS security context is associated with the 5G-GUTI of the other access;
- c) when the AMF and the UE create a 5G NAS security context using "null integrity protection algorithm" and "null ciphering algorithm" during an initial registration procedure for emergency services, or a registration procedure for mobility and periodic registration update for a UE that has an emergency PDU session (see subclause 5.4.2.2), the AMF and the UE shall delete the previous current 5G NAS security context;
- d) when a new mapped 5G NAS security context or 5G NAS security context created using "null integrity protection algorithm" and "null ciphering algorithm" is taken into use during the inter-system change from S1 mode to N1 mode, the AMF and the UE shall not delete the previously current native 5G NAS security context, if any. Instead, the previously current native 5G NAS security context shall become a non-current native 5G NAS security context, and the AMF and the UE shall delete any partial native 5G NAS security context;

If no previously current native 5G NAS security context exists, the AMF and the UE shall not delete the partial native 5G NAS security context, if any;

- e) when the AMF and the UE derive a new mapped 5G NAS security context during inter-system change from S1 mode to N1 mode, the AMF and the UE shall delete any existing current mapped 5G NAS security context;
- f) when a non-current full native 5G NAS security context is taken into use by a security mode control procedure, then the AMF and the UE shall delete the previously current mapped 5G NAS security context;
- g) when the UE or the AMF moves from 5GMM-REGISTERED to 5GMM-DEREGISTERED state, if the current 5G NAS security context is a mapped 5G NAS security context and a non-current full native 5G NAS security context exists, then the non-current 5G NAS security context shall become the current 5G NAS security context. Furthermore, the UE and the AMF shall delete any mapped 5G NAS security context or partial native 5G NAS security context.
- h) when the UE operating in single-registration mode in a network supporting N26 interface performs an intersystem change from N1 mode to S1 mode:
 - 1) if the UE has a mapped 5G NAS security context and the inter-system change is performed in:
 - i) 5GMM-IDLE mode, the UE shall delete the mapped 5G NAS security context after the successful completion of the tracking area update procedure or attach procedure (see 3GPP TS 24.301 [15]); or
 - ii) 5GMM-CONNECTED mode, the UE shall delete the mapped 5G NAS security context after the completion of the inter-system change.

After deletion of the mapped 5G NAS security context, if the UE has a non-current full native 5G NAS security context, then the non-current full native 5G NAS security context shall become the current full native 5G NAS security context; and

 i) when the UE operating in single-registration mode in a network supporting N26 interface performs an intersystem change from S1 mode to N1 mode in 5GMM-IDLE mode, if the UE has a non-current full native 5G NAS security context, then the UE shall make the non-current full native 5G NAS security context as the current native 5G NAS security context. The UE shall delete the mapped 5G NAS security context, if any.

If the UE is capable of registration over both 3GPP access and non-3GPP access, the UE in the state 5GMM-DEREGISTERED over both 3GPP access and non-3GPP access shall mark the 5G NAS security contexts of the 3GPP access and the non-3GPP access on the USIM or in the non-volatile memory as invalid when the UE initiates an initial registration procedure over either 3GPP access or non-3GPP access as described in subclause 5.5.1.2 or when the UE leaves state 5GMM-DEREGISTERED for any other state except 5GMM-NULL over either 3GPP access or non-3GPP access. Otherwise, the UE shall mark the 5G NAS security context on the USIM or in the non-volatile memory as invalid when the UE initiates an initial registration procedure as described in subclause 5.5.1.2 or when the UE leaves state 5GMM-DEREGISTERED for any other state except 5GMM-NULL.

If the UE is capable of registration over both 3GPP access and non-3GPP access, the UE shall store the current native 5G NAS security contexts of the 3GPP access and the non-3GPP access as specified in annex C and mark them as valid only when the UE enters state 5GMM-DEREGISTERED from any other state except 5GMM-NULL over both the 3GPP access and non-3GPP access or only when the UE aborts the initial registration procedure without having left

5GMM-DEREGISTERED over both the 3GPP access and non-3GPP access. Otherwise, the UE shall store the current native 5G NAS security context as specified in annex C and mark it as valid only when the UE enters state 5GMM-DEREGISTERED from any other state except 5GMM-NULL or when the UE aborts the initial registration procedure without having left 5GMM-DEREGISTERED.

4.4.2.2 Establishment of a mapped 5G NAS security context during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode

In order for the UE operating in single-registration mode in a network supporting N26 interface to derive a mapped 5G NAS security context for an inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, the AMF shall construct a mapped 5G NAS security context from the EPS security context received from the source MME as indicated in 3GPP TS 33.501 [24]. The AMF shall select the 5G NAS security algorithms and derive the 5G NAS keys (i.e. K_{NASenc} and K_{NASint}). The AMF shall define an ngKSI for the newly derived K'_{AMF} key such that the value field is taken from the eKSI of the K_{ASME} key and the type field is set to indicate a mapped security context and associate this ngKSI with the newly created mapped 5G NAS security context. The AMF shall then include the message authentication code, selected NAS algorithms, NCC and generated ngKSI in the S1 mode to N1 mode NAS transparent container IE (see subclause 9.11.2.9).

When the UE operating in single-registration mode in a network supporting N26 interface receives the command to perform inter-system change to N1 mode in 5GMM-CONNECTED mode, the UE shall derive a mapped K'_{AMF}, as indicated in 3GPP TS 33.501 [24], using the K_{ASME} from the EPS security context. Furthermore, the UE shall also derive the 5G NAS keys from the mapped K'_{AMF} using the selected NAS algorithm identifiers included in the S1 mode to N1 mode NAS transparent container IE and associate this mapped 5G NAS security context with the ngKSI value received. The UE shall then verify the received NAS MAC. In case the received NAS MAC is not verified successfully (see subclause 4.4.3.3) the UE shall discard the content of the received S1 mode to N1 mode NAS transparent container IE and inform the lower layers that the received S1 mode to N1 mode NAS transparent container is invalid.

When the UE operating in single-registration mode in a network supporting N26 interface has a PDN connection for emergency bearer services and has no current EPS security context, the AMF shall set 5G-IA0 and 5G-EA0 as the selected 5G NAS security algorithms in the S1 mode to N1 mode NAS transparent container IE. The AMF shall create a locally generated K'_{AMF}. The AMF shall set the ngKSI value of the associated security context to "000" and the type of security context flag to "mapped security context" in the S1 mode to N1 mode NAS transparent container IE.

When the UE operating in single-registration mode in a network supporting N26 interface receives the command to perform inter-system change to N1 mode in 5GMM-CONNECTED mode (see 3GPP TS 38.331 [30]) and has a PDN connection for emergency bearer services, if 5G-IA0 and 5G-EA0 as the selected 5G NAS security algorithms are included in the S1 mode to N1 mode NAS transparent container IE, the UE shall create a locally generated K'_{AMF}. Furthermore, the UE shall set the ngKSI value of the associated security context to the KSI value received.

After the new mapped 5G NAS security context is taken into use for the 3GPP access following a successful inter system change from S1 mode to N1 mode in 5GMM-CONNECTED mode and the UE is registered with the same PLMN over the 3GPP access and non-3GPP access:

- a) if a native 5G NAS security context is used on the non-3GPP access and:
 - the UE is in 5GMM-IDLE mode over non-3GPP access, then the AMF and the UE shall activate and take into use the new mapped 5G NAS security context on the 3GPP access for the non-3GPP access as described in 3GPP TS 33.501 [24] after the AMF sends or the UE receives the REGISTRATION ACCEPT message respectively. The UE and AMF shall keep the native 5G NAS security context which was used on the non-3GPP access and make it a non-current native 5G NAS security context. The non-current native 5G NAS security context may be re-activated later using the security mode control procedure; or
 - 2) the UE is in 5GMM-CONNECTED mode over non-3GPP access, in order to activate the native 5G NAS security context over the 3GPP access that is active on the non-3GPP access the AMF shall send the SECURITY MODE COMMAND message over the 3GPP access as described in 3GPP TS 33.501 [24]. The SECURITY MODE COMMAND message shall include the same ngKSI to identify the native 5G NAS security context that is used on the non-3GPP access; or
- b) if a mapped 5G NAS security context is used on the non-3GPP access and:
 - 1) the UE is in 5GMM-IDLE mode over non-3GPP access, the AMF and the UE shall activate and take into use the new mapped 5G NAS security context active on the 3GPP access for the non-3GPP access as described in

3GPP TS 33.501 [24] after the AMF sends or the UE receives the REGISTRATION ACCEPT message respectively; or

2) the UE is in 5GMM-CONNECTED mode over non-3GPP access, in order to activate the same mapped 5G NAS security context over one access that is used on the other access the AMF shall send the SECURITY MODE COMMAND message over one-access as described in 3GPP TS 33.501 [24]. The SECURITY MODE COMMAND message shall include the same ngKSI to identify the mapped 5G NAS security context that is used over the other access.

If the inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode is not completed successfully, the AMF and the UE operating in single-registration mode in a network supporting N26 interface shall delete the new mapped 5G NAS security context.

4.4.2.3 Establishment of a 5G NAS security context during N1 mode to N1 mode handover

During an N1 mode to N1 mode handover, the target AMF may derive a new 5G NAS security context for which the target AMF creates a new 5G NAS security context as indicated in 3GPP TS 33.501 [24].

When a new 5G NAS security context is derived using the same K_{AMF} , the target AMF includes the 8 least significant bits of the downlink NAS COUNT in the Intra N1 mode NAS transparent container IE, and indicates that a new K_{AMF} shall not be derived (see subclause 9.11.2.6). The AMF shall increment the downlink NAS COUNT by one after creating the Intra N1 mode NAS transparent container IE.

When a new 5G NAS security context is created from a new K_{AMF} , the target AMF includes the 8 least significant bits of the downlink NAS COUNT in the Intra N1 mode NAS transparent container IE and indicates that a new K_{AMF} shall be derived (see subclause 9.11.2.6). The AMF shall then set both the uplink and downlink NAS COUNT counters of this 5G NAS security context to zero. The AMF shall increment the downlink NAS COUNT by one after creating the Intra N1 mode NAS transparent container IE.

The target AMF also includes the ngKSI with the same value as the ngKSI currently being used with the UE, the message authentication code, and the selected NAS algorithms in the Intra N1 mode NAS transparent container IE.

When the UE receives a command to perform handover to NG-RAN including an Intra N1 mode NAS transparent container IE (see subclause 9.11.2.6), the UE derives a new 5G NAS security context as described in 3GPP TS 33.501 [24]. When the Intra N1 mode NAS transparent container IE indicates that a new K_{AMF} needs to be derived, the UE shall set both the downlink NAS COUNT and uplink NAS COUNT to zero after creating the new 5G NAS security context.

If the received Intra N1 mode NAS transparent container IE does not have a valid NAS COUNT (see subclause 4.4.3.2) or the received NAS MAC is not verified successfully (see subclause 4.4.3.3) the UE shall discard the content of the received Intra N1 mode NAS transparent container IE, continue to use the current 5G NAS security context, and inform the lower layers that the received Intra N1 mode NAS transparent container is invalid.

NOTE 1: During N1 mode to N1 mode handover, the Intra N1 mode NAS transparent container IE (see subclause 9.11.2.6) is equivalent to sending a SECURITY MODE COMMAND message to the UE in order to derive and use a new 5G NAS security context, optionally created with a new K_{AMF}. The UE maintains the Selected EPS NAS security algorithms until the UE receives a new Selected EPS NAS security algorithms.

After the new 5G NAS security context is taken into use for 3GPP access following a successful N1 mode to N1 mode handover and the UE is registered with the same PLMN over the 3GPP access and non-3GPP access:

- a) the UE is in 5GMM-IDLE mode over non-3GPP access, the AMF and the UE shall activate and take into use the new 5G NAS security context over the non-3GPP access as described in 3GPP TS 33.501 [24] after the AMF sends or the UE receives the REGISTRATION ACCEPT message respectively. If the new 5G NAS security context is created from a new K_{AMF}, the AMF and the UE shall set the downlink NAS COUNT and uplink NAS COUNT to zero also for the non-3GPP access, otherwise the downlink NAS COUNT and uplink NAS COUNT for the non-3GPP access are not changed; or
- b) the UE is in 5GMM-CONNECTED mode over non-3GPP access, in order to activate the new 5G NAS security context over the non-3GPP access that has been activated for the 3GPP access the AMF shall send the SECURITY MODE COMMAND message over the non-3GPP access as described in 3GPP TS 33.501 [24]. The SECURITY MODE COMMAND message shall include the same ngKSI to identify the new 5G NAS security

context that was activated over the 3GPP access and shall include the horizontal derivation parameter indicating " K_{AMF} derivation is not required". Otherwise, if the new 5G NAS security context is created from a new K_{AMF} , the AMF and the UE shall set the downlink NAS COUNT and uplink NAS COUNT to zero for the non-3GPP access.

NOTE 2: Explicit indication "K_{AMF} derivation is not required" for the non-3GPP access is to align security contexts within the UE without a subsequent derivation of a new K_{AMF} in the non-3GPP access.

4.4.2.4 Establishment of an EPS security context during inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode

In order for the UE operating in single-registration mode in a network supporting N26 interface to derive a mapped EPS security context for an inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode, the AMF shall prepare a mapped EPS security context for the target MME as indicated in 3GPP TS 33.501 [24].

The AMF shall derive a K'_{ASME} using the K_{AMF} key and the downlink NAS COUNT of the current 5G NAS security context, include the corresponding NAS sequence number in the N1 mode to S1 mode NAS transparent container IE (see subclause 9.11.2.7) and then increments its stored downlink NAS COUNT value by one.

NOTE: The creation of the N1 mode to S1 mode NAS transparent container and the increment of the stored downlink NAS COUNT value by one are performed in prior to transferring the mapped EPS security context to the MME.

The AMF shall select the NAS algorithms identifiers to be used in the target MME after the inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode, for encryption and integrity protection. The uplink and downlink NAS COUNT associated with the newly derived K'_{ASME} key are set to the uplink and downlink NAS COUNT value of the current 5G NAS security context, respectively. The eKSI for the newly derived K'_{ASME} key shall be defined such as the value field is taken from the ngKSI and the type field is set to indicate a mapped security context.

When the UE operating in single-registration mode in a network supporting N26 interface receives a command to perform inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode, the UE shall derive the mapped EPS security context, i.e. derive K'_{ASME} from K_{AMF} using a downlink NAS COUNT based on the NAS sequence number received in the N1 mode to S1 mode NAS transparent container IE (see subclause 9.11.2.7) as described in 3GPP TS 33.501 [24]. The UE shall set the uplink and downlink NAS COUNT values associated with the newly derived K'_{ASME} key to the uplink and downlink NAS COUNT values of the current 5G NAS security context respectively. The eKSI for the newly derived K'_{ASME} key is defined such that the value field is taken from the ngKSI and the type field is set to indicate a mapped security context. The UE shall also derive the NAS keys as specified in 3GPP TS 33.401 [23A] using the EPS NAS security algorithms identifiers that are stored in the UE's 5G NAS security context.

If the received N1 mode to S1 mode NAS transparent container IE does not have a valid NAS COUNT (see subclause 4.4.3.2) the UE shall discard the content of the received N1 mode to S1 mode NAS transparent container IE and inform the lower layers that the received N1 mode to S1 mode NAS transparent container is invalid.

If the inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode is not completed successfully, the AMF and the UE shall delete the new mapped EPS security context.

4.4.2.5 Establishment of secure exchange of NAS messages

Secure exchange of NAS messages via a NAS signalling connection is usually established by the AMF during the registration procedure by initiating a security mode control procedure. After successful completion of the security mode control procedure, all NAS messages exchanged between the UE and the AMF are sent integrity protected using the current 5G security algorithms, and except for the messages specified in subclause 4.4.5, all NAS messages exchanged between the UE and the AMF are sent ciphered using the current 5G security algorithms.

During inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, secure exchange of NAS messages is established between the AMF and the UE by:

a) the transmission of NAS security related parameters encapsulated in the AS signalling from the AMF to the UE triggering the inter-system change in 5GMM-CONNECTED mode (see 3GPP TS 33.501 [24]). The UE uses these parameters to generate the mapped 5G NAS security context (see subclause 8.6.2 of 3GPP TS 33.501 [24]); and

- b) after the inter-system change in 5GMM-CONNECTED mode, the transmission of a REGISTRATION REQUEST message from the UE to the AMF. The UE shall send this message integrity protected using the mapped 5G NAS security context and further protect this message as specified in subclause 4.4.6 and subclause 5.5.1.3.2. After the AMF receives the REGISTRATION REQUEST message:
 - if the AMF decides to take the native 5G NAS security context into use, the security mode control procedure is performed. From this time onward, all NAS messages exchanged between the UE and the AMF are sent integrity protected using the native 5G NAS security context, and except for the messages specified in subclause 4.4.5, all NAS messages exchanged between the UE and the AMF are sent ciphered using the native 5G NAS security context; or
 - 2) if the AMF decides to take the mapped 5G NAS security context into use, from this time onward, all NAS messages exchanged between the UE and the AMF are sent integrity protected using the mapped 5G NAS security context, and except for the messages specified in subclause 4.4.5, all NAS messages exchanged between the UE and the AMF are sent ciphered using the mapped 5G NAS security context.

During inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, if the UE is operating in single-registration mode and:

a) if the UE has a valid native 5G NAS security context, the UE shall transmit a REGISTRATION REQUEST message integrity protected with the native 5G NAS security context. The UE shall include the ngKSI indicating the native 5G NAS security context value in the REGISTRATION REQUEST message.

After receiving the REGISTRATION REQUEST message including the ngKSI indicating a native 5G NAS security context value, the AMF shall check whether the ngKSI included in the REGISTRATION REQUEST message belongs to a 5G NAS security context available in the AMF, and shall verify the MAC of the REGISTRATION REQUEST message. If the verification is successful, the AMF deletes the EPS security context received from the source MME if any, and the AMF re-establishes the secure exchange of NAS messages by either:

- replying with a REGISTRATION ACCEPT message that is integrity protected and ciphered using the native 5G NAS security context. From this time onward, all NAS messages exchanged between the UE and the AMF are sent integrity protected and except for the messages specified in subclause 4.4.5, all NAS messages exchanged between the UE and the AMF are sent ciphered; or
- initiating a security mode control procedure. This can be used by the AMF to take a non-current 5G NAS security context into use or to modify the current 5G NAS security context by selecting new NAS security algorithms.
- b) if the UE has no valid native 5G NAS security context, the UE shall send the REGISTRATION REQUEST message without integrity protection and encryption.

After receiving the REGISTRATION REQUEST message without integrity protection and encryption:

- 1) if N26 interface is supported:
 - if an EPS security context received from the source MME does not include the NAS security algorithms set to EIA0 and EEA0, the AMF shall either create a fresh mapped 5G NAS security context (see subclause 8.6.2 of 3GPP TS 33.501 [24]) or trigger a primary authentication and key agreement procedure to create a fresh native 5G NAS security context; or
 - ii) if an EPS security context received from the source MME includes the NAS security algorithms set to EIA0 and EEA0, the AMF shall trigger a primary authentication and key agreement procedure to create a fresh native 5G NAS security context; or
- 2) if N26 interface is not supported, the AMF shall trigger a primary authentication and key agreement procedure.

The newly created 5G NAS security context is taken into use by initiating a security mode control procedure and this context becomes the current 5G NAS security context in both the UE and the AMF. This re-establishes the secure exchange of NAS messages.

During an N1 mode to N1 mode handover, secure exchange of NAS messages is established between the AMF and the UE by:

- the transmission of NAS security related parameters encapsulated in the AS signalling from the target AMF to the UE triggering the N1 mode to N1 mode handover (see 3GPP TS 33.501 [24]). The UE uses these parameters to create a new 5G NAS security context.

The secure exchange of NAS messages shall be continued after N1 mode to N1 mode handover. It is terminated after inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode or when the NAS signalling connection is released.

When a UE in 5GMM-IDLE mode establishes a new NAS signalling connection and has a valid current 5G NAS security context, the UE shall transmit the initial NAS message integrity protected with the current 5G NAS security context and further protect this message as specified in subclause 4.4.6. The UE shall include the ngKSI indicating the current 5G NAS security context value in the initial NAS message. The AMF shall check whether the ngKSI included in the initial NAS message belongs to a 5G NAS security context available in the AMF, and shall verify the MAC of the NAS message. If the verification is successful, the AMF may re-establish the secure exchange of NAS messages:

- a) by replying with a NAS message that is integrity protected and ciphered using the current 5G NAS security context. From this time onward, all NAS messages exchanged between the UE and the AMF are sent integrity protected and except for the messages specified in subclause 4.4.5, all NAS messages exchanged between the UE and the AMF are sent ciphered; or
- b) by initiating a security mode control procedure. This can be used by the AMF to take a non-current 5G NAS security context into use or to modify the current 5G NAS security context by selecting new NAS security algorithms.

When a UE attempts multiple registrations in the same or different serving network, both the AMF and the UE shall follow the behavior specified in subclause 6.3.2 of 3GPP TS 33.501 [24]. The UE may support multiple records of NAS security context storage for multiple registration (see 3GPP TS 31.102 [22]). If the UE supports multiple records of NAS security context storage for multiple registration, the UE can select the appropriate one among the stored 5G security contexts to protect the initial NAS message (see 3GPP TS 33.501 [24]).

NOTE: For the case when the UE has two records of NAS security context stored and is attempting registration to the PLMN associated with the 5G-GUTI (or an equivalent PLMN) for that access, the UE uses the first NAS security context of that access to protect the initial NAS message. For the case when the UE has two records of NAS security context stored and is attempting registration to the PLMN associated with the second record (or an equivalent PLMN) of that access, the UE uses the second NAS security context of that access to protect the initial NAS message. For other cases when the UE has two records of NAS security context stored and is attempting registration to the PLMN associated with the second record (or an equivalent PLMN) of that access, the UE uses the second NAS security context of that access to protect the initial NAS message. For other cases when the UE has two records of NAS security context stored and is attempting registration to a PLMN which is not associated with any NAS security context record, the UE uses either record of the NAS security context of that access to protect the initial NAS message.

4.4.2.6 Change of security keys

When the AMF initiates a re-authentication to create a new 5G NAS security context, the messages exchanged during the authentication procedure are integrity protected and ciphered using the current 5G NAS security context, if any.

Both UE and AMF shall continue to use the current 5G NAS security context, until the AMF initiates a security mode control procedure. The SECURITY MODE COMMAND message sent by the AMF includes the ngKSI of the new 5G NAS security context to be used. The AMF shall send the SECURITY MODE COMMAND message integrity protected with the new 5G NAS security context, but unciphered. When the UE responds with a SECURITY MODE COMPLETE message, it shall send the message integrity protected and ciphered with the new 5G NAS security context.

The AMF can also modify the current 5G NAS security context or take the non-current native 5G NAS security context, if any, into use, by sending a SECURITY MODE COMMAND message including the ngKSI of the 5G NAS security context to be modified and including a new set of selected NAS security algorithms. In this case the AMF shall send the SECURITY MODE COMMAND message integrity protected with the modified 5G NAS security context, but unciphered. When the UE replies with a SECURITY MODE COMPLETE message, it shall send the message integrity protected and ciphered with the modified 5G NAS security context.

4.4.3 Handling of NAS COUNT and NAS sequence number

4.4.3.1 General

Each 5G NAS security context shall be associated with two separate counters NAS COUNT per access type in the same PLMN: one related to uplink NAS messages and one related to downlink NAS messages. If the 5G NAS security context is used for access via both 3GPP and non-3GPP access in the same PLMN, there are two NAS COUNT counter pairs associated with the 5G NAS security context. The NAS COUNT counters use 24-bit internal representation and are independently maintained by UE and AMF. The NAS COUNT shall be constructed as a NAS sequence number (8 least significant bits) concatenated with a NAS overflow counter (16 most significant bits).

When NAS COUNT is input to NAS ciphering or NAS integrity algorithms it shall be considered to be a 32-bit entity which shall be constructed by padding the 24-bit internal representation with 8 zeros in the most significant bits.

The value of the uplink NAS COUNT that is stored or read out of the USIM or non-volatile memory as described in annex C, is the value that shall be used in the next NAS message.

The value of the downlink NAS COUNT that is stored or read out of the USIM or non-volatile memory as described in annex C, is the largest downlink NAS COUNT used in a successfully integrity checked NAS message.

The value of the uplink NAS COUNT stored in the AMF is the largest uplink NAS COUNT used in a successfully integrity checked NAS message.

The value of the downlink NAS COUNT stored in the AMF is the value that shall be used in the next NAS message.

The NAS sequence number part of the NAS COUNT shall be exchanged between the UE and the AMF as part of the NAS signalling. After each new or retransmitted outbound SECURITY PROTECTED 5GS NAS MESSAGE message, the sender shall increase the NAS COUNT number by one, except for the initial NAS messages if the lower layers indicated the failure to establish the RRC connection (see 3GPP TS 38.331 [30]). Specifically, on the sender side, the NAS sequence number shall be increased by one, and if the result is zero (due to wrap around), the stored NAS overflow counter shall also be incremented by one (see subclause 4.4.3.5). If, through implementation-dependent means, the receiver determines that the NAS message is a replay of an earlier NAS message, then the receiver handles the received NAS message as described in subclause 4.4.3.2. Otherwise, in order to determine the estimated NAS COUNT value to be used for integrity verification of a received NAS message:

- The sequence number part of the estimated NAS COUNT value shall be equal to the sequence number in the received NAS message; and
- If the receiver can guarantee that this NAS message was not previously accepted, then the receiver may select the estimated NAS overflow counter so that the estimated NAS COUNT value is lower than the stored NAS COUNT value; otherwise, the receiver selects the estimated NAS overflow counter so that the estimated NAS COUNT value is higher than the stored NAS COUNT value.

During the inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, when a mapped 5G NAS security context is derived and taken into use, the AMF shall set both the uplink and downlink NAS COUNT counters of this 5G NAS security context to zero. The UE shall set both the uplink and downlink NAS COUNT counters of this 5G NAS security context to zero.

During the inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, the AMF shall increment the downlink NAS COUNT by one after it has created an S1 mode to N1 mode NAS transparent container (see subclause 9.11.2.9).

During the inter-system change from N1 mode to S1 mode in 5GMM-CONNECTED mode, the AMF shall increment the downlink NAS COUNT by one after it has created an N1 mode to S1 mode NAS transparent container (see subclause 9.11.2.7).

During N1 mode to N1 mode handover:

- a) if the new 5G NAS security context is created with the same K_{AMF}, the AMF shall signal the 8 least significant bits of the current downlink NAS COUNT value in an Intra N1 mode NAS transparent container (see subclause 9.11.2.6). The AMF shall then increment the downlink NAS COUNT by one; or
- b) if the new 5G NAS security context is created with a new K_{AMF}, the AMF shall signal the 8 least significant bits of the current downlink NAS COUNT value in an Intra N1 mode NAS transparent container (see

subclause 9.11.2.6) and shall then set both the uplink and downlink NAS COUNT counters of this 5G NAS security context to zero. The AMF shall then increment the downlink NAS COUNT by one. The UE shall also set both the uplink and downlink NAS COUNT counters to zero.

NOTE: During the inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, the S1 mode to N1 mode NAS transparent container (see subclause 9.11.2.9) is treated as an implicit SECURITY MODE COMMAND message for the UE and the AMF, and therefore the AMF regards the sending of the S1 mode to N1 mode NAS transparent container as the sending of an initial SECURITY MODE COMMAND message in order to derive and take into use a mapped 5G NAS security context for the purpose of the NAS COUNT handling.

4.4.3.2 Replay protection

Replay protection shall be supported for received NAS messages both in the AMF and the UE. However, since the realization of replay protection does not affect the interoperability between nodes, no specific mechanism is required for implementation.

Replay protection assures that one and the same NAS message is not accepted twice by the receiver. Specifically, for a given 5G NAS security context, a given NAS COUNT value shall be accepted at most one time and only if message integrity verifies correctly.

Replay protection is not applicable when 5G-IA0 is used.

4.4.3.3 Integrity protection and verification

The sender shall use its locally stored NAS COUNT as input to the integrity protection algorithm.

The receiver shall use the NAS sequence number included in the received message and an estimate for the NAS overflow counter as defined in subclause 4.4.3.1 to form the NAS COUNT input to the integrity verification algorithm.

The algorithm to calculate the integrity protection information is specified in 3GPP TS 33.501 [24], and in case of the:

- a) SECURITY PROTECTED 5GS NAS MESSAGE message, the integrity protection shall include octet 7 to n, i.e. the Sequence number IE and the NAS message IE.
- b) Intra N1 mode NAS transparent container IE and S1 mode to N1 mode NAS transparent container IE, the integrity protection shall include all octets of the value part of the IE starting from octet 7.
- NOTE: To ensure backward compatibility, the UE uses all octets starting from octet 7 in the received NAS transparent container for the purpose of integrity check of the NAS transparent container irrespective of the release/version it supports. After a successful integrity check, the UE can ignore the octets which are not specified in the release/version which the UE supports.

In addition to the data that is to be integrity protected, the BEARER ID, DIRECTION bit, NAS COUNT and 5G NAS integrity key are input to the integrity protection algorithm. These parameters are described in 3GPP TS 33.501 [24].

After successful integrity protection validation, the receiver shall update its corresponding locally stored NAS COUNT with the value of the estimated NAS COUNT for this NAS message.

Integrity verification is not applicable when 5G-IA0 is used.

4.4.3.4 Ciphering and deciphering

The sender shall use its locally stored NAS COUNT as input to the ciphering algorithm.

The receiver shall use the NAS sequence number included in the received message and an estimate for the NAS overflow counter as defined in subclause 4.4.3.1 to form the NAS COUNT input to the deciphering algorithm.

The input parameters to the NAS ciphering algorithm are the BEARER ID, DIRECTION bit, NAS COUNT, NAS encryption key and the length of the key stream to be generated by the encryption algorithm.

When applying initial NAS message protection to the REGISTRATION REQUEST or SERVICE REQUEST message as described in subclause 4.4.6, the length of the key stream is set to the length of the entire plain NAS message that is included in the NAS message container IE, i.e. the value part of the NAS message container IE, that is to be ciphered.

When applying initial NAS message protection to the CONTROL PLANE SERVICE REQUEST message as described in subclause 4.4.6, the length of the key stream is set to the length of:

- a) the value part of the CIoT small data container IE that is to be ciphered; or
- b) the value part of the NAS message container IE that is to be ciphered.

4.4.3.5 NAS COUNT wrap around

If, when increasing the NAS COUNT as specified above, the AMF detects that either its downlink NAS COUNT or the UE's uplink NAS COUNT is "close" to wrap around, (close to 2^{24}), the AMF shall take the following actions:

- If there is no non-current native 5G NAS security context with sufficiently low NAS COUNT values, the AMF shall initiate a new primary authentication and key agreement procedure with the UE, leading to a new established 5G NAS security context and the NAS COUNT being reset to 0 in both the UE and the AMF when the new 5G NAS security context is activated;
- Otherwise, the AMF can activate a non-current native 5G NAS security context with sufficiently low NAS COUNT values or initiate a new primary authentication and key agreement procedure as specified above.

If for some reason a new K_{AMF} has not been established using primary authentication and key agreement procedure before the NAS COUNT wraps around, the node (AMF or UE) in need of sending a NAS message shall instead release the NAS signalling connection. Prior to sending the next uplink NAS message, the UE shall delete the ngKSI indicating the current 5G NAS security context.

When the 5G-IA0 is used as the NAS integrity algorithm, the UE and the AMF shall allow NAS COUNT wrap around. If NAS COUNT wrap around occurs, the following requirements apply:

- a) the UE and the AMF shall continue to use the current 5G NAS security context;
- b) the AMF shall not initiate the primary authentication and key agreement procedure;
- c) the AMF shall not release the NAS signalling connection; and
- d) the UE shall not perform a local release of the NAS signalling connection.

4.4.4 Integrity protection of NAS signalling messages

4.4.4.1 General

For the UE, integrity protected signalling is mandatory for the 5GMM NAS messages once a valid 5G NAS security context exists and has been taken into use. For the network, integrity protected signalling is mandatory for the 5GMM NAS messages once a secure exchange of 5GS NAS messages has been established for the NAS signalling connection. Integrity protection of all NAS signalling messages is the responsibility of the NAS. It is the network which activates integrity protection.

The use of "null integrity protection algorithm" 5G-IA0 (see subclause 9.11.3.34) in the current 5G NAS security context is only allowed:

- a) for an unauthenticated UE for which establishment of emergency services is allowed;
- b) for an W-AGF acting on behalf of an FN-RG; and
- c) for a W-AGF acting on behalf of an N5GC device.

For setting the security header type in outbound NAS messages, the UE and the AMF shall apply the same rules irrespective of whether the "null integrity protection algorithm" or any other integrity protection algorithm is indicated in the 5G NAS security context.

If the "null integrity protection algorithm"5G-IA0 has been selected as an integrity protection algorithm, the receiver shall regard the NAS messages with the security header indicating integrity protection as integrity protected.

Details of the integrity protection and verification of NAS signalling messages are specified in 3GPP TS 33.501 [24].

When a NAS message needs to be sent both ciphered and integrity protected, the NAS message is first ciphered and then the ciphered NAS message and the NAS sequence number are integrity protected by calculating the MAC.

NOTE: NAS messages that are ciphered with the "null ciphering algorithm" 5G-EA0 are regarded as ciphered (see subclause 4.4.5).

When a NAS message needs to be sent only integrity protected and unciphered, the unciphered NAS message and the NAS sequence number are integrity protected by calculating the MAC.

When a 5GSM message is piggybacked in a 5GMM message, there is only one Sequence number IE and one Message authentication code IE for the 5GMM message piggybacking the 5GSM message.

4.4.4.2 Integrity checking of NAS signalling messages in the UE

Except the messages listed below, no NAS signalling messages shall be processed by the receiving 5GMM entity in the UE or forwarded to the 5GSM entity, unless the network has established secure exchange of 5GS NAS messages for the NAS signalling connection:

- a) IDENTITY REQUEST (if requested identification parameter is SUCI);
- b) AUTHENTICATION REQUEST;
- c) AUTHENTICATION RESULT;
- d) AUTHENTICATION REJECT;
- e) REGISTRATION REJECT (if the 5GMM cause is not #76 or #78);
- f) DEREGISTRATION ACCEPT (for non switch off); and
- g) SERVICE REJECT (if the 5GMM cause is not #76 or #78).
- NOTE: These messages are accepted by the UE without integrity protection, as in certain situations they are sent by the network before security can be activated.

Integrity protection is never applied directly to 5GSM messages, but to the 5GMM message in which the 5GSM message is included.

Once the secure exchange of NAS messages has been established, the receiving 5GMM entity in the UE shall not process any NAS signalling messages unless they have been successfully integrity checked by the NAS. If NAS signalling messages, having not successfully passed the integrity check, are received, then the NAS in the UE shall discard that message. The processing of the SECURITY MODE COMMAND message that has not successfully passed the integrity check is specified in subclause 5.4.2.5. If any NAS signalling message is received as not integrity protected even though the secure exchange of NAS messages has been established by the network, then the NAS shall discard this message.

4.4.4.3 Integrity checking of NAS signalling messages in the AMF

Except the messages listed below, no NAS signalling messages shall be processed by the receiving 5GMM entity in the AMF or forwarded to the 5GSM entity, unless the secure exchange of NAS messages has been established for the NAS signalling connection:

- a) REGISTRATION REQUEST;
- b) IDENTITY RESPONSE (if requested identification parameter is SUCI);
- c) AUTHENTICATION RESPONSE;
- d) AUTHENTICATION FAILURE;
- e) SECURITY MODE REJECT;
- f) DEREGISTRATION REQUEST; and
- g) DEREGISTRATION ACCEPT;

- NOTE 1: The REGISTRATION REQUEST message is sent by the UE without integrity protection, if the registration procedure is initiated due to an inter-system change in 5GMM-IDLE mode and no current 5G NAS security context is available in the UE. The other messages are accepted by the AMF without integrity protection, as in certain situations they are sent by the UE before security can be activated.
- NOTE 2: The DEREGISTRATION REQUEST message can be sent by the UE without integrity protection, e.g. if the UE is registered for emergency services and there is no valid 5G NAS security context available, or if due to user interaction a registration procedure is cancelled before the secure exchange of NAS messages has been established. For these cases the network can attempt to use additional criteria (e.g. whether the UE is subsequently still performing periodic registration update or still responding to paging) before marking the UE as 5GMM-DEREGISTERED.

Integrity protection is never applied directly to 5GSM messages, but to the 5GMM message in which the 5GSM message is included.

Once a current 5G NAS security context exists, until the secure exchange of NAS messages has been established for the NAS signalling connection, the receiving 5GMM entity in the AMF shall process the following NAS signalling messages, even if the MAC included in the message fails the integrity check or cannot be verified, as the 5G NAS security context is not available in the network:

- a) REGISTRATION REQUEST;
- b) IDENTITY RESPONSE (if requested identification parameter is SUCI);
- c) AUTHENTICATION RESPONSE;
- d) AUTHENTICATION FAILURE;
- e) SECURITY MODE REJECT;
- f) DEREGISTRATION REQUEST;
- g) DEREGISTRATION ACCEPT;
- h) SERVICE REQUEST; and
- i) CONTROL PLANE SERVICE REQUEST;
- NOTE 3: These messages are processed by the AMF even when the MAC that fails the integrity check or cannot be verified, as in certain situations they can be sent by the UE protected with a 5G NAS security context that is no longer available in the network.

If a REGISTRATION REQUEST message for initial registration fails the integrity check and it is not a registration request for emergency services, the AMF shall authenticate the subscriber before processing the registration request any further. Additionally, the AMF shall initiate a security mode control procedure, and include the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message as specified in subclause 5.4.2.2. For the case when the registration procedure is for emergency services see subclause 5.5.1.2.3 and subclause 5.4.1.3.5.

If a REGISTRATION REQUEST message for mobility and periodic registration update fails the integrity check and the UE provided EPS NAS message container IE which was successfully verified by the source MME, the AMF may create a mapped 5G NAS security context and initiate a security mode control procedure to take the new mapped 5G NAS security context into use; otherwise if the UE has only a non-emergency PDU session established, the AMF shall initiate a primary authentication and key agreement procedure to create a new native 5G NAS security context. Additionally, the AMF shall initiate a security mode control procedure, and include the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message as specified in subclause 5.4.2.2. For the case when the UE has an emergency PDU session see subclause 5.5.1.3.3 and subclause 5.4.1.3.5.

If a DEREGISTRATION REQUEST message fails the integrity check, the AMF shall proceed as follows:

- If it is not a deregistration request due to switch off, and the AMF can initiate an authentication procedure, the AMF should authenticate the subscriber before processing the deregistration request any further.
- If it is a deregistration request due to switch off, or the AMF does not initiate an authentication procedure for any other reason, the AMF may ignore the deregistration request and remain in state 5GMM-REGISTERED.

NOTE 4: The network can attempt to use additional criteria (e.g. whether the UE is subsequently still performing periodic registration update or still responding to paging) before marking the UE as 5GMM-DEREGISTERED.

If a SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message fails the integrity check and the UE has only non-emergency PDU sessions established, the AMF shall send the SERVICE REJECT message with 5GMM cause #9 "UE identity cannot be derived by the network" and keep the 5GMM-context and 5G NAS security context unchanged. For the case when the UE has an emergency PDU session and integrity check fails, the AMF may skip the authentication procedure even if no 5G NAS security context is available and proceed directly to the execution of the security mode control procedure as specified in subclause 5.4.2. Additionally, the AMF shall include the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message as specified in subclause 5.4.2.2. After successful completion of the service request procedure, the network shall perform a local release of all non-emergency PDU sessions. The emergency PDU sessions shall not be released.

Once the secure exchange of NAS messages has been established for the NAS signalling connection, the receiving 5GMM entity in the AMF shall not process any NAS signalling messages unless they have been successfully integrity checked by the NAS. If any NAS signalling message, having not successfully passed the integrity check, is received, then the NAS in the AMF shall discard that message. If any NAS signalling message is received, as not integrity protected even though the secure exchange of NAS messages has been established, then the NAS shall discard this message.

4.4.5 Ciphering of NAS signalling messages

The use of ciphering in a network is an operator option subject to AMF configuration. When operation of the network without ciphering is configured, the AMF shall indicate the use of "null ciphering algorithm" 5G-EA0 (see subclause 9.11.3.34) in the current 5G NAS security context for all UEs. For setting the security header type in outbound NAS messages, the UE and the AMF shall apply the same rules irrespective of whether the "null ciphering algorithm" or any other ciphering algorithm is indicated in the 5G NAS security context.

When the UE establishes a new N1 NAS signalling connection, it shall apply security protection to the initial NAS message as described in subclause 4.4.6.

The UE shall start the ciphering and deciphering of NAS messages when the secure exchange of NAS messages has been established for an N1 NAS signalling connection. From this time onward, unless explicitly defined, the UE shall send all NAS messages ciphered until the N1 NAS signalling connection is released, or the UE performs inter-system change to S1 mode.

The AMF shall start ciphering and deciphering of NAS messages as described in subclause 4.4.2.5. From this time onward, except for the SECURITY MODE COMMAND message, the AMF shall send all NAS messages ciphered until the N1 NAS signalling connection is released, or the UE performs inter-system change to S1 mode.

Ciphering is never applied directly to 5GSM messages, but to the 5GMM message in which the 5GSM message is included.

Once the encryption of NAS messages has been started between the AMF and the UE, the receiver shall discard the unciphered NAS messages which shall have been ciphered according to the rules described in this specification.

If the "null ciphering algorithm" 5G-EA0 has been selected as a ciphering algorithm, the NAS messages with the security header indicating ciphering are regarded as ciphered.

Details of ciphering and deciphering of NAS signalling messages are specified in 3GPP TS 33.501 [24].

4.4.6 Protection of initial NAS signalling messages

The 5GS supports protection of initial NAS messages as specified in 3GPP TS 33.501 [24]. The protection of initial NAS messages applies to the REGISTRATION REQUEST, SERVICE REQUEST and CONTROL PLANE SERVICE REQUEST message, and is achieved as follows:

 a) If the UE does not have a valid 5G NAS security context, the UE sends a REGISTRATION REQUEST message including cleartext IEs only. After activating a 5G NAS security context resulting from a security mode control procedure:

- if the UE needs to send non-cleartext IEs, the UE shall include the entire REGISTRATION REQUEST message (i.e. containing both cleartext IEs and non-cleartext IEs) in the NAS message container IE and shall include the NAS message container IE in the SECURITY MODE COMPLETE message; or
- 2) if the UE does not need to send non-cleartext IEs, the UE shall include the entire REGISTRATION REQUEST message (i.e. containing cleartext IEs only) in the NAS message container IE and shall include the NAS message container IE in the SECURITY MODE COMPLETE message.
- b) If the UE has a valid 5G NAS security context and:
 - the UE needs to send non-cleartext IEs in a REGISTRATION REQUEST or SERVICE REQUEST message, the UE includes the entire REGISTRATION REQUEST or SERVICE REQUEST message (i.e. containing both cleartext IEs and non-cleartext IEs) in the NAS message container IE and shall cipher the value part of the NAS message container IE. The UE shall then send a REGISTRATION REQUEST or SERVICE REQUEST message containing the cleartext IEs and the NAS message container IE;
 - 2) the UE needs to send non-cleartext IEs in a CONTROL PLANE SERVICE REQUEST message:
 - i) if CIoT small data container IE is the only non-cleartext IE to be sent, the UE shall cipher the value part of the CIoT small data container IE. The UE shall then send a CONTROL PLANE SERVICE REQUEST message containing the cleartext IEs and the CIoT small data container IE;
 - ii) otherwise, the UE includes non-cleartext IEs in the NAS message container IE and shall cipher the value part of the NAS message container IE. The UE shall then send a CONTROL PLANE SERVICE REQUEST message containing the cleartext IEs and the NAS message container IE;
 - 3) the UE does not need to send non-cleartext IEs in a REGISTRATION REQUEST or SERVICE REQUEST message, the UE sends the REGISTRATION REQUEST or SERVICE REQUEST message without including the NAS message container IE; or
 - 4) the UE does not need to send non-cleartext IEs in a CONTROL PLANE SERVICE REQUEST message, the UE sends the CONTROL PLANE SERVICE REQUEST message without including the NAS message container IE and the CIoT small data container IE.

When the initial NAS message is a REGISTRATION REQUEST message, the cleartext IEs are:

- Extended protocol discriminator;
- Security header type;
- Spare half octet;
- Registration request message identity;
- 5GS registration type;
- ngKSI;
- 5GS mobile identity;
- UE security capability;
- Additional GUTI;
- UE status;
- EPS NAS message container;
- NID; and
- PLMN with disaster condition.

When the initial NAS message is a SERVICE REQUEST message, the cleartext IEs are:

- Extended protocol discriminator;
- Security header type;

- Spare half octet;
- ngKSI;
- Service request message identity;
- Service type; and
- 5G-S-TMSI.

When the initial NAS message is a CONTROL PLANE SERVICE REQUEST message, the cleartext IEs are:

- Extended protocol discriminator;
- Security header type;
- Spare half octet;
- ngKSI;
- Control plane service request message identity; and
- Control plane service type.

When the UE sends a REGISTRATION REQUEST or SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message that includes a NAS message container IE, the UE shall set the security header type of the initial NAS message to "integrity protected".

When the AMF receives an integrity protected initial NAS message which includes a NAS message container IE, the AMF shall decipher the value part of the NAS message container IE. If the received initial NAS message is a REGISTRATION REQUEST message or a SERVICE REQUEST message, the AMF shall consider the NAS message that is obtained from the NAS message container IE as the initial NAS message that triggered the procedure.

When the AMF receives a CONTROL PLANE SERVICE REQUEST message which includes a CIoT small data container IE, the AMF shall decipher the value part of the CIoT small data container IE and handle the message as specified in subclause 5.6.1.4.2.

When the initial NAS message is a DEREGISTRATION REQUEST message, the UE always sends the NAS message unciphered.

If the UE:

- a) has 5G-EA0 as a selected 5G NAS security algorithm; and
- b) selects a PLMN other than Registered PLMN and EPLMN over one access;

the UE shall send an initial NAS message including cleartext IEs only via the access type associated with the newly selected PLMN as described in this subclause for the case when the UE does not have a valid 5G NAS security context.

If the UE:

- a) has 5G-EA0 as a selected 5G NAS security algorithm; and
- b) selects a PLMN other than Registered PLMN and EPLMN over one access, and the Registered PLMN or EPLMN is not registering or registered over other access;

the UE shall delete the 5G NAS security context.

NOTE: UE deletes the 5G NAS security context only if the UE is not in the connected mode.

4.4.7 Protection of NAS IEs

The network can provide the SOR transparent container IE during the registration procedure to the UE in the REGISTRATION ACCEPT message. The SOR transparent container IE is integrity protected by the HPLMN or subscribed SNPN as specified in 3GPP TS 33.501 [24].

The UE can provide the SOR transparent container IE during the registration procedure to the network in the REGISTRATION COMPLETE message. The SoR-MAC- I_{UE} in the SOR transparent container IE is generated by the UE as specified in 3GPP TS 33.501 [24].

The network can provide the Payload container IE during the Network-initiated NAS transport procedure to the UE in DL NAS TRANSPORT message. If the Payload container type IE is set to "SOR transparent container" or "UE parameters update transparent container", the Payload container IE is integrity protected by the HPLMN or subscribed SNPN as specified in 3GPP TS 33.501 [24]. If the Payload container type IE is set to "Multiple payloads" and the payload container type field of the payload container entry is set to "SOR transparent container" or "UE parameters update transparent container", the payload container entry is set to "SOR transparent container" or "UE parameters update transparent container", the payload container entry is set to "SOR transparent container" or "UE parameters update transparent container", the payload container entry contents field of the payload container entry is integrity protected correspondingly.

The UE can provide the Payload container IE during the UE-initiated NAS transport procedure to the network in UL NAS TRANSPORT message. If the Payload container type IE is set to "SOR transparent container" or "UE parameters update transparent container", the SoR-MAC-I_{UE} or UPU-MAC-I_{UE} in the Payload container IE is generated by the UE as specified in 3GPP TS 33.501 [24]. If the Payload container type IE is set to "Multiple payloads" and the payload container type field of the payload container entry is set to "SOR transparent container" or "UE parameters update transparent container", the SoR-MAC-I_{UE} or UPU-MAC-I_{UE} in the payload container or "UE parameters update transparent container", the SoR-MAC-I_{UE} or UPU-MAC-I_{UE} in the payload container entry is generated by the UE correspondingly.

4.5 Unified access control

4.5.1 General

When the UE needs to access the 5GS, the UE not operating as an IAB-node (see 3GPP TS 23.501 [8]) and not acting as a 5G ProSe layer-2 UE-to-network relay UE (see 3GPP TS 23.304 [6E]) whose access attempt is triggered by a 5G ProSe layer-2 remote UE, first performs access control checks to determine if the access is allowed. Access control checks shall be performed for the access attempts defined by the following list of events:

- NOTE 1: Although the UE operating as an IAB-node skips the access control checks, the UE operating as an IABnode determines an access category and one or more access identities for each access attempt in order to derive an RRC establishment cause. In this case the NAS provides the RRC establishment cause but does not provide the access category and the one or more access identities to the lower layers.
- a) the UE is in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication over 3GPP access and an event that requires a transition to 5GMM-CONNECTED mode occurs; and
- b) the UE is in 5GMM-CONNECTED mode over 3GPP access or 5GMM-CONNECTED mode with RRC inactive indication and one of the following events occurs:
 - 5GMM receives an MO-IMS-registration-related-signalling-started indication, an MO-MMTEL-voice-callstarted indication, an MO-MMTEL-video-call-started indication or an MO-SMSoIP-attempt-started indication from upper layers;
 - 5GMM receives a request from upper layers to send a mobile originated SMS over NAS unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode;
 - 3) 5GMM receives a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session establishment unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode;
 - 4) 5GMM receives a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session modification unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode;
 - 5) 5GMM receives a request to re-establish the user-plane resources for an existing PDU session;
 - 6) 5GMM is notified that an uplink user data packet is to be sent for a PDU session with suspended user-plane resources;

- 7) 5GMM receives a request from upper layers to send a mobile originated location request unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode; and
- 8) 5GMM receives a request from upper layers to send a mobile originated signalling transaction towards the PCF by sending an UL NAS TRANSPORT message including a UE policy container (see 3GPP TS 24.587 [19B] and 3GPP TS 24.554 [19E]) unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode to 5GMM-CONNECTED mode.
- NOTE 2: 5GMM specific procedures initiated by NAS in 5GMM-CONNECTED mode or 5GMM-CONNECTED mode with RRC inactive indication are not subject to access control, e.g. a registration procedure after PS handover will not be prevented by access control (see subclause 5.5).
- NOTE 3: LPP messages or location event report messages transported in the UL NAS TRANSPORT message sent in response to a mobile terminating or network induced location request, and the corresponding access attempts are handled as MT access.
- NOTE 4: Initiating a mobile originated signalling transaction towards the UDM by sending an UL NAS TRANSPORT message including an SOR transparent container is not supported. Therefore, access control for these cases has not been specified.

When the NAS detects one of the above events, the NAS needs to perform the mapping of the kind of request to one or more access identities and one access category and lower layers will perform access barring checks for that request based on the determined access identities and access category.

NOTE 5: The NAS is aware of the above events through indications provided by upper layers or through determining the need to start 5GMM procedures through normal NAS behaviour, or both.

To determine the access identities and the access category for a request, the NAS checks the reason for access, types of service requested and profile of the UE including UE configurations, against a set of access identities and access categories defined in 3GPP TS 22.261 [3], namely:

- a) a set of standardized access identities;
- b) a set of standardized access categories; and
- c) a set of operator-defined access categories, if available.

For the purpose of determining the applicable access identities from the set of standardized access identities defined in 3GPP TS 22.261 [3], the NAS shall follow the requirements set out in:

- a) subclause 4.5.2 and the rules and actions defined in table 4.5.2.1, if the UE is not operating in SNPN access mode; or
- b) subclause 4.5.2A and the rules and actions defined in table 4.5.2A.1, if the UE is operating in SNPN access mode.

In order to enable access barring checks for access attempts identified by lower layers in 5GMM-CONNECTED mode with RRC inactive indication, the UE provides the applicable access identities to lower layers.

- NOTE 6: When and how the NAS provides the applicable access identities to lower layers is UE implementation specific.
- NOTE 7: Although the UE operating as an IAB-node skips the access control checks, the UE provides the applicable access identities to lower layers for access attempts identified by lower layers in 5GMM-CONNECTED mode with RRC inactive indication.

For the purpose of determining the applicable access category from the set of standardized access categories and operator-defined access categories defined in 3GPP TS 22.261 [3], the NAS shall follow the requirements set out in:

- a) subclause 4.5.2 and the rules and actions defined in table 4.5.2.2, if the UE is not operating in SNPN access mode; or
- b) subclause 4.5.2A and the rules and actions defined in table 4.5.2A.2, if the UE is operating in SNPN access mode.

4.5.2 Determination of the access identities and access category associated with a request for access for UEs not operating in SNPN access mode

When the UE needs to initiate an access attempt in one of the events listed in subclause 4.5.1, the UE shall determine one or more access identities from the set of standardized access identities, and one access category from the set of standardized access categories, to be associated with that access attempt.

The set of the access identities applicable for the request is determined by the UE in the following way:

- a) for each of the access identities 1, 2, 3, 11, 12, 13, 14 and 15 in table 4.5.2.1, the UE shall check whether the access identity is applicable in the selected PLMN, if a new PLMN is selected, or otherwise if it is applicable in the RPLMN or equivalent PLMN; and
- b) if none of the above access identities is applicable, then access identity 0 is applicable.

Access Identity		UE configuration
number		UE is not configured with any parameters from this table
1 (1	NOTE 1)	UE is configured for multimedia priority service (MPS).
2 (1	NOTE 2)	UE is configured for mission critical service (MCS).
1) 8	NOTE 4)	UE for which a disaster condition applies
	4-10	Reserved for future use
11 (NOTE 3)	Access Class 11 is configured in the UE.
12 (NOTE 3)	Access Class 12 is configured in the UE.
13 (NOTE 3)	Access Class 13 is configured in the UE.
14 (NOTE 3)	Access Class 14 is configured in the UE.
15 (NOTE 3)	Access Class 15 is configured in the UE.
 NOTE 1: Access identity 1 is valid when: the USIM file EF_{UAC_AIC} indicates the UE is configured for access identity 1 and the selected PLMN, if a new PLMN is selected, or RPLMN is the HPLMN (if the EHPLMN list is not present or is empty) or EHPLMN (if the EHPLMN list is present), or a visited PLMN of the home country (see the definition of home country in 3GPP TS 24.301 [15]); the UE receives the 5GS network feature support IE with the MPS indicator bit set to "Access identity 1 valid" from the RPLMN as described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4; or the UE receives the Priority indicator IE with the MPS indicator bit set to "Access identity 1 valid" from the RPLMN as described in subclause 5.4.4.3. NOTE 2: Access identity 2 is used by UEs configured for MCS and is valid when: the USIM file EF_{UAC_AIC} indicates the UE is configured for access identity 2 and the selected PLMN, if a new PLMN is selected, or RPLMN is the HPLMN (if the EHPLMN list is not present or is empty) or EHPLMN (if the EHPLMN list is present), or a visited PLMN of the home country (see 3GPP TS 23.122 [5]); or the UE receives the 5GS network feature support IE with the MCS indicator bit set to "Access identity 2 valid" from the RPLMN as described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4; or 		
NOTE 3: NOTE 4:	Access identitie EHPLMN (if the visited PLMNs Access Identity (see 3GPP TS	es 11 and 15 are valid in HPLMN (if the EHPLMN list is not present or is empty) or e EHPLMN list is present). Access Identities 12, 13 and 14 are valid in HPLMN and of home country only (see the definition of home country in 3GPP TS 24.301 [15]). / 3 is valid when the UE is registering or registered for disaster roaming services /23.122 [5])

Table 4.5.2.1: Access identities

The UE uses the MPS indicator bit of the 5GS network feature support IE or the Priority indicator IE to determine if access identity 1 is valid. Processing of the MPS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message is described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4. Processing of the MPS indicator bit of the Priority indicator IE in the CONFIGURATION UPDATE COMMAND message is described in subclause 5.4.4.3. The UE shall not consider access identity 1 to be valid when the UE is not in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present) prior to receiving the MPS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message or of the Priority indicator IE in the CONFIGURATION UPDATE COMMAND message being set to "Access identity 1 valid".

When the UE is in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), the contents of the USIM file EF_{UAC_AIC} as specified in 3GPP TS 31.102 [22] and the rules specified in table 4.5.2.1 are used to determine

the applicability of access identity 1. When the UE is in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), and the USIM file EF_{UAC_AIC} does not indicate the UE is configured for access identity 1, the UE uses the MPS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message or of the Priority indicator IE in the CONFIGURATION UPDATE COMMAND message to determine if access identity 1 is valid. When the UE is in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), and the USIM file EF_{UAC_AIC} indicates the UE is configured for access identity 1, the MPS indicator bit of the 5GS network feature support IE and the Priority indicator IE are not applicable. When the UE is not in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), the contents of the USIM file EF_{UAC_AIC} are not applicable.

The UE uses the MCS indicator bit of the 5GS network feature support IE to determine if access identity 2 is valid. Processing of the MCS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message is described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4. The UE shall not consider access identity 2 to be valid when the UE is not in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present) prior to receiving the MCS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message being set to "Access identity 2 valid".

When the UE is in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), the contents of the USIM file EF_{UAC_AIC} as specified in 3GPP TS 31.102 [22] and the rules specified in table 4.5.2.1 are used to determine the applicability of access identity 2. When the UE is in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), and the USIM file EF_{UAC_AIC} does not indicate the UE is configured for access identity 2, the UE uses the MCS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message to determine if access identity 2 is valid. When the UE is in the country of its HPLMN or in an EHPLMN list is present), and the USIM file EF_{UAC_AIC} indicates the UE is configured for access identity 2, the MCS indicator bit of the 5GS network feature support IE in the country of its HPLMN or in an EHPLMN list is present), and the USIM file EF_{UAC_AIC} indicates the UE is configured for access identity 2, the MCS indicator bit of the 5GS network feature support IE is not applicable. When the UE is not in the country of its HPLMN or in an EHPLMN or in an EHPLMN (if the EHPLMN list is feature support IE is not applicable. When the UE is not in the country of its HPLMN or in an EHPLMN (if the EHPLMN list is present), the contents of the USIM file EF_{UAC_AIC} are not applicable.

The UE checks the conditions specified in subclause 4.4.3.1.1 of 3GPP TS 23.122 [5] to determine if access identity 3 is valid, and the applicability of access identity 3.

When the UE is in its HPLMN (if the EHPLMN list is not present or is empty) or in an EHPLMN (if the EHPLMN list is present), the contents of the USIM file EF_{ACC} as specified in 3GPP TS 31.102 [22] and the rules specified in table 4.5.2.1 are used to determine the applicability of access classes 11 and 15. When the UE is not in its HPLMN (if the EHPLMN list is not present or is empty) or in an EHPLMN (if the EHPLMN list is present), access classes 11 and 15 are not applicable.

When the UE is in the country of its HPLMN, the contents of the USIM file EF_{ACC} as specified in 3GPP TS 31.102 [22] and the rules specified in table 4.5.2.1 are used to determine the applicability of access classes 12 - 14. When the UE is not in the country of its HPLMN, access classes 12-14 are not applicable.

In order to determine the access category applicable for the access attempt, the NAS shall check the rules in table 4.5.2.2, and use the access category for which there is a match for barring check. If the access attempt matches more than one rule, the access category of the lowest rule number shall be selected. If the access attempt matches more than one operator-defined access category definition, the UE shall select the access category from the operator-defined access category definition with the lowest precedence value (see subclause 4.5.3).

NOTE: The case when an access attempt matches more than one rule includes the case when multiple events trigger an access attempt at the same time.

Table 4.5.2.2: Mapping table for access categories

Rule #	Type of access attempt	Requirements to be met	Access Category
1	Response to paging or	Access attempt is for MT access or	0 (-MT acc)
•		handover of ongoing MMTEL voice call	
		MMTEL video call or SMSoIP from pop	
	5GMM connection	3GPP access: or	
	management procedure	Access attempt is made upon receipt of	
	initiated for the purpose of	"coll pull initiated" (2CPP TS 24 174 [12D])	
	transporting on LPP		
	manage without on		
	procedure:		
	Access attempt to		
	handover of ongoing		
	MMTEL voice call		
	MMTEL video call or		
	SMSoIP from pop-3GPP		
	receipt of "call-pull-		
	initiated" indication from		
	the upper layers (see		
	3GPP TS 24 174 [13D])		
2	Emergency	UE is attempting access for an emergency	2 (= emergency)
		session (NOTE 1, NOTE 2)	00.00
3	Access attempt for	UE stores operator-defined access category	32-63
	operator-defined access	apposited in subclause 4.5.2, and appose	
	category	attempt is matching criteria of an operator	
		defined access category definition	classification
3.1	Access attempt for MO	LIF is in NB-N1 mode and allowed to use	10 (– MO
0.1	exception data	exception data reporting (see the	exception data)
		ExceptionDataReportingAllowed leaf of the	chooption data)
		NAS configuration MO in	
		3GPP TS 24.368 [17] or the USIM file	
		EF _{NASCONFIG} in 3GPP TS 31.102 [22]), and	
		access attempt is for MO data or for MO	
		signalling initiated upon receiving a request	
		from upper layers to transmit user data	
		related to an exceptional event.	
4	Access attempt for delay	(a) UE is configured for NAS signalling low	1 (= delay
	tolerant service	priority or UE supporting S1 mode is	tolerant)
		configured for EAB (see the	
		"ExtendedAccessBarring" leaf of NAS	
		configuration MO in 3GPP TS 24.368 [17] or	
		3GPP TS 31.102 [22]) where "EAB override"	
		does not apply, and	
		(b): the UE received one of the categories a,	
		b of c as part of the parameters for unified	
		information and the LIE is a member of the	
		broadcasted category in the selected PLMN	
		or RPI MN/equivalent PI MN	
		(NOTE 3. NOTE 5. NOTE 6. NOTE 7.	
		NOTE 8)	
5	MO MMTel voice call	Access attempt is for MO MMTel voice call	4 (= MO MMTel
		or for NAS signalling connection recovery	voice)
		during ongoing MO MMTel voice call	
		(NOTE 2)	
6	MO MMTel video call	Access attempt is for MO MMTel video call	5 (= MO MMTel
		or tor NAS signalling connection recovery	video)
7		(1001 ± 2)	6 (- MO SMS
'	SMSolP	(NOTE 4) or MO SMS over SMSoIP transfer	and SMSoIP)
		or for NAS signalling connection recovery	2
		during ongoing MO SMS or SMSoIP transfer	
		(NOTE 2)	

7.1	MO IMS registration related signalling	Access attempt is for MO IMS registration related signalling (e.g. IMS initial registration, re-registration, subscription refresh) or for NAS signalling connection recovery during ongoing procedure for MO IMS registration related signalling (NOTE 2a)	9 (= MO IMS registration related signalling)
8	UE NAS initiated 5GMM specific procedures	Access attempt is for MO signalling	3 (= MO_sig)
8.1	Mobile originated location request	Access attempt is for mobile originated location request (NOTE 9)	3 (= MO_sig)
8.2	Mobile originated signalling transaction towards the PCF	Access attempt is for mobile originated signalling transaction towards the PCF (NOTE 10)	3 (= MO_sig)
9	UE NAS initiated 5GMM connection management procedure or 5GMM NAS transport procedure	Access attempt is for MO data	7 (= MO_data)
10	An uplink user data packet is to be sent for a PDU session with suspended user-plane resources	No further requirement is to be met	7 (= MO_data)

NOTE 1:	This includes 5GMM specific procedures while the service is ongoing and 5GMM connection
	management procedures required to establish a PDU session with request type = "initial
	emergency request" or "existing emergency PDU session", or to re-establish user-plane
	resources for such a PDU session. This further includes the service request procedure initiated
	with a SERVICE REQUEST message with the Service type IE set to "emergency services
	fallback".
NOTE 2:	Access for the purpose of NAS signalling connection recovery during an ongoing service as
	defined in subclause 4.5.5, or for the purpose of NAS signalling connection establishment
	following fallback indication from lower layers during an ongoing service as defined in
	subclause 4.5.5, is mapped to the access category of the ongoing service in order to derive an
	RRC establishment cause, but barring checks will be skipped for this access attempt.
NOTE 2a:	Access for the purpose of NAS signalling connection recovery during an ongoing procedure for
	MO IMS registration related signalling as defined in subclause 4.5.5, or for the purpose of NAS
	signalling connection establishment following fallback indication from lower layers during an
	ongoing procedure for MO IMS registration related signalling as defined in subclause 4.5.5, is
	mapped to the access category of the MO IMS registration related signalling in order to derive an
	RRC establishment cause, but barring checks will be skipped for this access attempt.
NOTE 3:	If the UE selects a new PLMN, then the selected PLMN is used to check the membership;
	otherwise the UE uses the RPLMNor a PLMN equivalent to the RPLMN.
NOTE 4:	This includes the 5GMM connection management procedures triggered by the UE-initiated NAS
	transport procedure for transporting the MO SMS.
NOTE 5:	The UE configured for NAS signalling low priority is not supported in this release of specification.
	If a UE supporting both S1 mode and N1 mode is configured for NAS signalling low priority in S1
	mode as specified in 3GPP TS 24.368 [17] or 3GPP TS 31.102 [22], the UE shall ignore the
	configuration for NAS signalling low priority when in N1 mode.
NOTE 6:	If the access category applicable for the access attempt is 1, then the UE shall additionally
	determine a second access category from the range 3 to 7. If more than one access category
	matches, the access category of the lowest rule number shall be chosen. The UE shall use the
	second access category only to derive an RRC establishment cause for the access attempt.
NOTE 7:	"EAB override" does not apply, if the UE is not configured to allow overriding EAB (see the
	"Override_ExtendedAccessBarring" leaf of NAS configuration MO in 3GPP TS 24.368 [17] or
	3GPP TS 31.102 [22]), or if NAS has not received an indication from the upper layers to override
	EAB and the UE does not have a PDU session that was established with EAB override.
NOTE 8:	For the definition of categories a, b and c associated with access category 1, see
	3GPP TS 22.261 [3]. The categories associated with access category 1 are distinct from the
	categories a, b and c associated with EAB (see 3GPP TS 22.011 [1A]).
NOTE 9:	This includes:
	a) the UE-initiated NAS transport procedure for transporting a mobile originated location
	request;
	b) the 5GMM connection management procedure triggered by a) above; and
	c) NAS signalling connection recovery during an ongoing 5GC-MO-LR procedure.
NOTE 10:	This includes:
	a) the UE-initiated NAS transport procedure for transporting a mobile originated signalling
	transaction towards the PCF;
	b) the 5GMM connection management procedure triggered by a) above; and
	c) NAS signalling connection recovery during an ongoing UE-requested policy provisioning
	procedure for V2XP, ProSeP or both (see 3GPP TS 24.587 [19B] and see
	3GPP TS 24.554 [19E]).

4.5.2A Determination of the access identities and access category associated with a request for access for UEs operating in SNPN access mode

When the UE needs to initiate an access attempt in one of the events listed in subclause 4.5.1, the UE shall determine one or more access identities from the set of standardized access identities, and one access category from the set of standardized access categories, to be associated with that access attempt.

The set of the access identities applicable for the request is determined by the UE in the following way:

- a) for each of the access identities 1, 2, 11, 12, 13, 14 and 15 in table 4.5.2A.1, the UE shall check whether the access identity is applicable in the selected SNPN, if a new SNPN is selected, or otherwise if it is applicable in the RSNPN; and
- b) if none of the above access identities is applicable, then access identity 0 is applicable.

Access Identi number	ity	UE configuration
0		UE is not configured with any parameters from this table
1 (NOTE 1)		UE is configured for multimedia priority service (MPS).
2 (NOTE 2)		UE is configured for mission critical service (MCS).
3-10		Reserved for future use
11 (NOTE 3))	Access Class 11 is configured in the UE.
12 (NOTE 3))	Access Class 12 is configured in the UE.
13 (NOTE 3))	Access Class 13 is configured in the UE.
14 (NOTE 3))	Access Class 14 is configured in the UE.
15 (NOTE 3))	Access Class 15 is configured in the UE.
 NOTE 1: Access identity 1 is valid when: the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, indicates the UE is configured for access identity 1 in the selected SNPN, if a new SNPN is selected, or RSNPN; the UE receives the 5GS network feature support IE with the MPS indicator bit set to "Access identity 1 valid" from the RSNPN as described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4; or the UE receives the Priority indicator IE with the MPS indicator bit set to "Access identity 1 valid" from the RPLMN as described in subclause 5.4.4.3. NOTE 2: Access identity 2 is used by UEs configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the UE (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected. 		
3GPP 1 access - the UE identity NOTE 3: Access control an entry PLMN s	 3GPP TS 31.102 [22]), if the PLMN subscription is selected, indicates the UE is configured for access identity 2 in the selected SNPN, if a new SNPN is selected, or RSNPN; or the UE receives the 5GS network feature support IE with the MCS indicator bit set to "Access identity 2 valid" from the RSNPN as described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4. DTE 3: Access identities 11 to 15 are valid if indicated as configured for the UE in the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, in the selected SNPN, if a new SNPN is selected, or RSNPN. 	

The contents of the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, and the rules specified in table 4.5.2A.1 are used to determine the applicability of access identity 1 in the SNPN. When the contents of the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, do not indicate the UE is configured for access identity 1 for the SNPN, the UE uses the MPS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message and the MPS indicator bit of the Priority indicator IE in the CONFIGURATION UPDATE COMMAND message to determine if access identity 1 is valid.

The contents of the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, and the rules specified in table 4.5.2A.1 are used to determine the applicability of access identity 2 in the SNPN. When the contents of the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, do not indicate the UE is configured for access identity 2 for the SNPN, the UE uses the MCS indicator bit of the 5GS network feature support IE in the REGISTRATION ACCEPT message to determine if access identity 2 is valid.

The contents of the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]), if an entry of "list of subscriber data" is selected, or in the USIM (see 3GPP TS 31.102 [22]), if the PLMN subscription is selected, and the rules specified in table 4.5.2A.1 are used to determine the applicability of access classes 11 to 15 in the SNPN.

In order to determine the access category applicable for the access attempt, the NAS shall check the rules in table 4.5.2A.2, and use the access category for which there is a match for barring check. If the access attempt matches more than one rule, the access category of the lowest rule number shall be selected. If the access attempt matches more than one operator-defined access category definition, the UE shall select the access category from the operator-defined access category definition with the lowest precedence value (see subclause 4.5.3).
NOTE: The case when an access attempt matches more than one rule includes the case when multiple events trigger an access attempt at the same time.

Table 4.5.2A.2: Mapping table for access categories

Rule #	Type of access attempt	Requirements to be met	Access Category
1	Response to paging or NOTIFICATION over non- 3GPP access (NOTE 11); 5GMM connection management procedure initiated for the purpose of transporting an LPP message without an ongoing 5GC-MO-LR procedure; Access attempt to handover of MMTEL voice call, MMTEL video call or SMSoIP from non-3GPP access; Access attempt upon receipt of "call-pull- initiated" indication from the upper layers (see 3GPP TS 24.174 [13D])	Access attempt is for MT access, handover of ongoing MMTEL voice call, MMTEL video call or SMSoIP from non-3GPP access; or Access attempt is made upon receipt of "call-pull-initiated" indication (3GPP TS 24.174 [13D])	0 (= MT_acc)
2	Emergency	UE is attempting access for an emergency session (NOTE 1, NOTE 2)	2 (= emergency)
3	Access attempt for operator-defined access category	UE stores operator-defined access category definitions valid in the SNPN as specified in subclause 4.5.3, and access attempt is matching criteria of an operator-defined access category definition	32-63 (= based on operator classification)
4	Access attempt for delay tolerant service	 (a) UE is configured for NAS signalling low priority, and (b) the UE received one of the categories a, b or c as part of the parameters for unified access control in the broadcast system information, and the UE is a member of the broadcasted category in the selected SNPN or RSNPN (NOTE 3, NOTE 5, NOTE 6, NOTE 7, NOTE 8) 	1 (= delay tolerant)
5	MO MMTel voice call	Access attempt is for MO MMTel voice call or for NAS signalling connection recovery during ongoing MO MMTel voice call (NOTE 2)	4 (= MO MMTel voice)
6	MO MMTel video call	Access attempt is for MO MMTel video call or for NAS signalling connection recovery during ongoing MO MMTel video call (NOTE 2)	5 (= MO MMTel video)
7	MO SMS over NAS or MO SMSoIP	Access attempt is for MO SMS over NAS (NOTE 4) or MO SMS over SMSoIP transfer or for NAS signalling connection recovery during ongoing MO SMS or SMSoIP transfer (NOTE 2)	6 (= MO SMS and SMSoIP)
7.1	MO IMS registration related signalling	Access attempt is for MO IMS registration related signalling (e.g. IMS initial registration, re-registration, subscription refresh) or for NAS signalling connection recovery during ongoing procedure for MO IMS registration related signalling (NOTE 2a)	9 (= MO IMS registration related signalling)
8	UE NAS initiated 5GMM specific procedures	Access attempt is for MO signalling	3 (= MO_sig)
8.1	Mobile originated location request	Access attempt is for mobile originated location request (NOTE 9)	3 (= MO_sig)
8.2	Mobile originated signalling transaction towards the PCF	Access attempt is for mobile originated signalling transaction towards the PCF (NOTE 10)	3 (= MO_sig)

9	UE NAS initiated 5GMM	Access attempt is for MO data	7 (= MO_data)			
	connection management					
	procedure or 5GMM NAS					
	transport procedure					
10	An uplink user data packet	No further requirement is to be met	7 (= MO_data)			
	is to be sent for a PDU					
	session with suspended					
	user-plane resources					
NOTE 1:	Void					
NOTE 2	Access for the purpose of NAS si	analling connection recovery during an ongoin	a service as			
	defined in subclause 4.5.5 or for	the purpose of NAS signalling connection esta	blishment			
	following fallback indication from	lower lavers during an ongoing service as defin	ned in			
	subclause 4.5.5 is mapped to the	access category of the ongoing service in or	ter to derive an			
	BBC establishment cause, but be	arring checks will be skipped for this access att	en lo denve an			
NOTE 20	Access for the purpose of NAS si	anning checks will be skipped for this access att				
NOTE Za	Access for the purpose of NAS si	defined in subslauge 4.5.5 or for the purpage a	y NO INS			
	registration related signaling as t	a follback indication from lower lowers during a				
	Connection establishment following	ig failback indication from lower layers during a				
	INS registration related signalling	as defined in subclause 4.5.5, is mapped to the	1e access			
	category of the MO IMS registrati	ion related signalling in order to derive an RRC	establishment			
NOTE	cause, but barring checks will be	skipped for this access attempt.				
NOTE 3:	If the UE selects a new SNPN, then the selected SNPN is used to check the membership;					
	otherwise the UE uses the RSNPN.					
NOTE 4:	E This includes the 5GMM connection management procedures triggered by the UE-initiated NAS					
	transport procedure for transporting the MO SMS.					
NOTE 5:	5: The UE configured for NAS signalling low priority is not supported in this release of specification.					
NOTE 6:	6: If the access category applicable for the access attempt is 1, then the UE shall additionally					
	determine a second access categories	gory from the range 3 to 7. If more than one ac	cess category			
	matches, the access category of	the lowest rule number shall be chosen. The U	E shall use the			
	second access category only to derive an RRC establishment cause for the access attempt.					
NOTE 7:	': Void.					
NOTE 8:	: For the definition of categories a, b and c associated with access category 1, see					
	3GPP TS 22.261 [3]. The categories associated with access category 1 are distinct from the					
	categories a, b and c associated	with EAB (see 3GPP TS 22.011 [1A]).				
NOTE 9:	This includes:	· · · ·				
	a) the UE-initiated NAS transpor	t procedure for transporting a mobile originated	d location			
	request;					
	b) the 5GMM connection manage	ement procedure triggered by a) above: and				
	c) NAS signalling connection rec	covery during an ongoing 5GC-MO-LR procedu	ure.			
NOTE 10	This includes:					
	a) the UE-initiated NAS transpor	t procedure for transporting a mobile originated	d signalling			
	transaction towards the PCF					
	b) the 5GMM connection manage	ement procedure triggered by a) above: and				
	c) NAS signaling connection management procedure triggered by a) above, and					
	procedure for V/2XP or both (see	3GPP TS 24 587 [198])	provisioning			
NOTE 11	The term "non-3GPP access" ref	ers to the case when the LIF is accessing SND	N services via a			
	PI MNI	ors to the case when the OL is accessilly SINF	IN SCIVICES VIA A			
	I LIVIA.					

4.5.3 Operator-defined access categories

Operator-defined access category definitions can be signalled to the UE using NAS signalling. Each operator-defined access category definition consists of the following parameters:

- a) a precedence value which indicates in which order the UE shall evaluate the operator-defined category definition for a match;
- b) an operator-defined access category number, i.e. access category number in the 32-63 range that uniquely identifies the access category in the PLMN or SNPN in which the access categories are being sent to the UE;
- c) criteria consisting of one or more access category criteria type and associated access category criteria type values. The access category criteria type can be set to one of the following:
 - 1) DNN;
 - 2) Void;
 - 3) OS Id + OS App Id of application triggering the access attempt; or

4) S-NSSAI; and

NOTE 1: An access category criteria type can be associated with more than one access category criteria values.

d) optionally, a standardized access category. This standardized access category is used in combination with the access identities of the UE to determine the RRC establishment cause as specified in subclause 4.5.6.

If the access attempt is to establish a new PDU session i.e. it is triggered by:

- a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session establishment unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode; or
- a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode triggered by a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session establishment,

then:

- the access attempt matches access category criteria type DNN if the DNN requested by the UE during the PDU session establishment procedure matches any of the access criteria type values associated with the access criteria type DNN; and
- the access attempt matches access category criteria type S-NSSAI if the S-NSSAI requested by the UE during the PDU session establishment procedure matches any of the access criteria type values associated with the access criteria type S-NSSAI.

If the access attempt is for an existing PDU session i.e. it is triggered by:

- a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session modification unless the request triggered a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode;
- a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode triggered by a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session modification;
- a service request procedure to transition the UE from 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication to 5GMM-CONNECTED mode triggered by a request from upper layers to send an UL NAS TRANSPORT message for the purpose of PDU session release;
- a service request procedure requesting user-plane resources for a PDU session; or
- an uplink user data packet is to be sent for a PDU session with suspended user-plane resources,

then:

- the access attempt matches access category criteria type DNN if the DNN provided by the network in the PDU SESSION ESTABLISHMENT ACCEPT message matches any of the access criteria type values associated with the access criteria type DNN; and
- the access attempt matches access category criteria type S-NSSAI if the S-NSSAI associated with the PDU session matches any of the access criteria type values associated with the access criteria type S-NSSAI.

An access attempt matches the criteria of an operator-defined access category definition, if the access attempt matches all access category criteria types included in the criteria with any of the associated access criteria type values.

Each operator-defined access category definition has a different precedence value.

Several operator-defined access category definitions can have the same operator-defined access category number.

If:

- an access category in bullet d) is not provided;
- an access category in bullet d) is provided and is not a standardized access category; or

- an access category in bullet d) is provided, is a standardized access category and is not recognized by the UE;

the UE shall use instead access category 7 (MO_data) in combination with the access identities of the UE to determine the RRC establishment cause as specified in subclause 4.5.6.

The operator-defined access category definitions are valid in the PLMN which provided them and in a PLMN equivalent to the PLMN which provided them, or in the SNPN which provided them, as specified in annex C.

If the UE stores operator-defined access category definitions valid in the selected PLMN or the RPLMN, or valid in the selected SNPN or RSNPN, then access control in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication will only be performed for the event a) defined in subclause 4.5.1. If the transition from 5GMM-IDLE mode or 3GPP access to 5GMM-CONNECTED mode is due to a UE NAS initiated 5GMM specific procedure, then this access attempt shall be mapped to one of the standardized access categories in the range < 32, see subclause 4.5.2. I.e. for this case the UE shall skip the checking of operator-defined access category definitions.

If the UE stores operator-defined access category definitions valid in the selected PLMN or the RPLMN, or valid in the selected SNPN or RSNPN, then access control in 5GMM-CONNECTED mode and in 5GMM-CONNECTED mode with RRC inactive indication will only be performed for the events 1) to 8) defined in subclause 4.5.1.

The UE shall handle the operator-defined access category definitions stored for the RPLMN or RSNPN as specified in subclause 5.4.4.3, subclause 5.5.1.2.4, and subclause 5.5.1.3.4.

When the UE is switched off, the UE shall keep the operator-defined access category definitions so that the operatordefined access category definitions can be used after switch on.

When the UE selects a new PLMN which is not equivalent to the previously selected PLMN, or selects a new SNPN, the UE shall stop using the operator-defined access category definitions stored for the previously selected PLMN or SNPN and should keep the operator-defined access category definitions stored for the previously selected PLMN or SNPN.

NOTE 2: When the UE selects a new PLMN which is not equivalent to the previously selected PLMN, or selects a new SNPN, the UE can delete the operator-defined access category definitions stored for the previously selected PLMN or SNPN e.g. if there is no storage space in the UE.

4.5.4 Access control and checking

4.5.4.1 Access control and checking in 5GMM-IDLE mode and in 5GMM-IDLE mode with suspend indication

When the UE is in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication, upon receiving a request from the upper layers for an access attempt, the NAS shall categorize the access attempt into access identities and an access category following:

- a) subclause 4.5.2, table 4.5.2.1 and table 4.5.2.2, and subclause 4.5.3, if the UE is not operating in SNPN access mode; or
- b) subclause 4.5.2A, table 4.5.2A.1 and table 4.5.2A.2, and subclause 4.5.3, if the UE is operating in SNPN access mode,

and provide the applicable access identities and the access category to the lower layers for the purpose of access control checking. In this request to the lower layer the NAS can also provide to the lower layer the RRC establishment cause determined as specified in subclause 4.5.6 of this specification.

- NOTE 1: The access barring check is performed by the lower layers.
- NOTE 2: As an implementation option, the NAS can provide the RRC establishment cause to the lower layers after being informed by the lower layers that the access attempt is allowed.

If the UE has uplink user data pending for one or more PDU sessions when it builds a REGISTRATION REQUEST or SERVICE REQUEST message as initial NAS message, the UE shall indicate the respective PDU sessions in the Uplink data status IE as specified in subclause 5.5.1.3.2 and 5.6.1.2.1, regardless of the access category for which the access barring check is performed.

If the UE is registered for 5GS services with control plane CIoT 5GS optimization has uplink user data pending for one or more PDU sessions when it builds a CONTROL PLANE SERVICE REQUEST message as initial NAS message, the UE shall indicate the respective PDU sessions as specified in subclause 5.6.1.2.2, regardless of the access category for which the access barring check is performed.

NOTE 3: The UE indicates pending user data for all the respective PDU sessions, even if barring timers are running for some of the corresponding access categories.

If the lower layers indicate that the access attempt is allowed, the NAS shall initiate the procedure to send the initial NAS message for the access attempt.

If the lower layers indicate that the access attempt is barred, the NAS shall not initiate the procedure to send the initial NAS message for the access attempt. Additionally:

- a) if the event which triggered the access attempt was an MO-MMTEL-voice-call-started indication or an MO-MMTEL-video-call-started indication:
 - if the UE is operating in the single-registration mode, the UE's usage setting is "voice centric" and the UE has not disabled its E-UTRA capability as specified in 3GPP TS 24.301 [15], the UE may attempt to select an E-UTRA cell connected to EPC. If the UE finds a suitable E-UTRA cell connected to EPC, it then proceeds with the appropriate EMM specific procedures and, if necessary, ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.2 and 3GPP TS 24.301 [15];
 - if the UE is operating in the dual-registration mode, the UE may proceed in S1 mode with the appropriate EMM specific procedures and ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.3 and 3GPP TS 24.301 [15]; or
 - 3) otherwise, the NAS shall notify the upper layers that the access attempt is barred. In this case, upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS shall notify the upper layers that the barring is alleviated for the access category and may initiate the procedure to send the initial NAS message, if still needed;
- b) if the event which triggered the access attempt was an MO-SMSoIP-attempt-started indication or an MO-IMS-registration-related-signalling-started indication:
 - if the UE is operating in the single-registration mode, the UE may attempt to select an E-UTRA cell connected to EPC. If the UE finds a suitable E-UTRA cell connected to EPC, it then proceeds with the appropriate EMM specific procedures and, if necessary, ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.2 and 3GPP TS 24.301 [15];
 - if the UE is operating in the dual-registration mode, the UE may proceed in S1 mode with the appropriate EMM specific procedures and ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.3 and 3GPP TS 24.301 [15]; or
 - 3) otherwise, the NAS shall notify the upper layers that the access attempt is barred. In this case, upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS shall notify the upper layers that the barring is alleviated for the access category and may initiate the procedure to send the initial NAS message, if still needed; and
- c) if the access attempt is for emergency:
 - the NAS shall notify the upper layers that the access attempt is barred. In this case, upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS shall notify the upper layers of that the barring is alleviated for the access category and may initiate the procedure to send the initial NAS message, if still needed.
- NOTE 4: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6] and 3GPP TS 24.229 [14].
- NOTE 5: Barring timers, on a per access category basis, are run by the lower layers. At expiry of barring timers, the indication of alleviation of access barring is indicated to the NAS on a per access category basis.

4.5.4.2 Access control and checking in 5GMM-CONNECTED mode and in 5GMM-CONNECTED mode with RRC inactive indication

When the UE is in 5GMM-CONNECTED mode or 5GMM-CONNECTED mode with RRC inactive indication, upon detecting one of events 1) through 8) listed in subclause 4.5.1, the NAS shall categorize the corresponding access attempt into access identities and an access category following:

- a) subclause 4.5.2, table 4.5.2.1 and table 4.5.2.2, and subclause 4.5.3, if the UE is not operating in SNPN access mode; or
- b) subclause 4.5.2A, table 4.5.2A.1 and table 4.5.2A.2, and subclause 4.5.3, if the UE is operating in SNPN access mode,

and provide the access identities and the access category to the lower layers for the purpose of access control checking. In this request to the lower layer the NAS can also provide to the lower layer the RRC establishment cause determined as specified in subclause 4.5.6 of this specification.

NOTE 1: As an implementation option, the NAS can provide the RRC establishment cause to the lower layers after being informed by the lower layers that the access attempt is allowed.

If the UE has uplink user data pending for one or more PDU sessions when it builds a REGISTRATION REQUEST or SERVICE REQUEST message for the access attempt, the UE shall indicate the respective PDU sessions in the Uplink data status IE as specified in subclause 5.5.1.3.2 and 5.6.1.2, regardless of the access category for which the access barring check is performed.

NOTE 2: The UE indicates pending user data for all the respective PDU sessions, even if barring timers are running for some of the corresponding access categories.

If the lower layers indicate that the access attempt is allowed, the NAS shall take the following action depending on the event which triggered the access attempt:

- a) if the event which triggered the access attempt was an MO-MMTEL-voice-call-started indication, an MO-MMTEL-video-call-started indication, an MO-SMSoIP-attempt-started indication, or an MO-IMS-registration-related-signalling-started indication, the NAS shall notify the upper layers that the access attempt is allowed;
- b) if the event which triggered the access attempt was a request from upper layers to send a mobile originated SMS over NAS, 5GMM shall initiate the NAS transport procedure as specified in subclause 5.4.5 to send the SMS in an UL NAS TRANSPORT message;
- c) if the event which triggered the access attempt was a request from upper layers to establish a new PDU session, 5GMM shall initiate the NAS transport procedure as specified in subclause 5.4.5 to send the PDU SESSION ESTABLISHMENT REQUEST message;
- d) if the event which triggered the access attempt was a request from upper layers to modify an existing PDU session, 5GMM shall initiate the NAS transport procedure as specified in subclause 5.4.5 to send the PDU SESSION MODIFICATION REQUEST message;
- e) if the event which triggered the access attempt was a request to re-establish the user-plane resources for an existing PDU session, 5GMM shall initiate the service request procedure as specified in subclause 5.6.1;
- f) if the event which triggered the access attempt was an uplink user data packet to be sent for a PDU session with suspended user-plane resources, 5GMM shall consider that the uplink user data packet is allowed to be sent;
- g) if the event which triggered the access attempt was a request from upper layers to send a mobile originated location request, 5GMM shall initiate the NAS transport procedure as specified in clause 5.4.5 to send an LCS message in an UL NAS TRANSPORT message; and
- h) if the event which triggered the access attempt was a request from upper layers to send a mobile originated signalling transaction towards the PCF by sending an UL NAS TRANSPORT message including a UE policy container (see 3GPP TS 24.587 [19B] and 3GPP TS 24.554 [19E]), 5GMM shall initiate the NAS transport procedure as specified in subclause 5.4.5 to send the signalling transaction via an UL NAS TRANSPORT message.

If the lower layers indicate that the access attempt is barred, the NAS shall take the following action depending on the event which triggered the access attempt:

- a) if the event which triggered the access attempt was an MO-MMTEL-voice-call-started indication, an MO-MMTEL-video-call-started indication or an MO-SMSoIP-attempt-started indication, or an MO-IMS-registration-related-signalling-started indication:
 - if the UE is operating in the dual-registration mode, the UE may proceed in S1 mode with the appropriate EMM specific procedures and ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.3 and 3GPP TS 24.301 [15];
 - otherwise, the NAS shall notify the upper layers that the access attempt is barred. In this case, upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS shall notify the upper layers that the barring is alleviated for the access category;
- NOTE 3: In this case prohibiting the initiation of the MMTEL voice session, MMTEL video session or prohibiting sending of the SMS over IP or the IMS registration related signalling is performed by the upper layers.
- b) if the event which triggered the access attempt was a request from upper layers to send a mobile originated SMS over NAS, 5GMM shall not initiate the NAS transport procedure as specified in subclause 5.4.5 to send the SMS in an UL NAS TRANSPORT message. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, 5GMM may initiate the NAS transport procedure as specified in subclause 5.4.5 to send the SMS in an UL NAS TRANSPORT message, if still needed;
- c) if the event which triggered the access attempt was a request from upper layers to establish a new PDU session, 5GMM shall not initiate the NAS transport procedure to send the PDU SESSION ESTABLISHMENT REQUEST message. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS may initiate the NAS transport procedure as specified in subclause 5.4.5, if still needed;
- d) if the event which triggered the access attempt was a request from upper layers to modify an existing PDU session modification, 5GMM shall not initiate the NAS transport procedure to send the PDU SESSION MODIFICATION REQUEST message. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS may initiate the NAS transport procedure as specified in subclause 5.4.5, if still needed;
- e) if the event which triggered the access attempt was a request to re-establish the user-plane resources for an existing PDU session, the NAS shall not initiate the service request procedure as specified in subclause 5.6.1. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS may initiate the service request procedure as specified in subclause 5.6.1, if still needed;
- f) if the event which triggered the access attempt was an uplink user data packet to be sent for a PDU session with suspended user-plane resources, 5GMM shall consider that the uplink user data packet is not allowed to be sent. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, the NAS shall consider that the barring is alleviated for the access category;
- g) if the event which triggered the access attempt was a request from upper layers to send a mobile originated location request, 5GMM shall not initiate the NAS transport procedure as specified in clause 5.4.5 to send an LCS message in an UL NAS TRANSPORT message. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, 5GMM may initiate the NAS transport procedure as specified in clause 5.4.5 to send the LCS message in an UL NAS TRANSPORT message, if still needed; and
- h) if the event which triggered the access attempt was a request from upper layers to send a mobile originated signalling transaction towards the PCF by sending an UL NAS TRANSPORT message including a UE policy container (see 3GPP TS 24.587 [19B]), 5GMM shall not initiate the NAS transport procedure as specified in subclause 5.4.5 to send the mobile originated signalling transaction via an UL NAS TRANSPORT message. Upon receiving an indication from the lower layers that the barring is alleviated for the access category with which the access attempt was associated, 5GMM may initiate the NAS transport procedure as specified in subclause 5.4.5 to send the mobile originated signalling transaction via an UL NAS TRANSPORT message, if still needed.

4.5.5 Exception handling and avoiding double barring

Access attempts are allowed to proceed without further access control checking in order to avoid double barring for any service request or registration procedure initiated for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclauses 5.3.1.2 and 5.3.1.4).

NOTE 1: The case of NAS signalling connection recovery also includes the cases where the UE was in S1 mode when the RRC connection failure occurred.

For any service request or registration procedure of this kind the UE determines an access category as specified in subclause 4.5.1 and 4.5.2 or 4.5.2A, unless a different access category is specified in the rest of the present subclause.

NOTE 2: Although the access control checking is skipped, the access category is determined for the specific access attempt in order to derive an RRC establishment cause.

There are several services or an MO IMS registration related signalling for which the NAS needs to be informed when the service starts and stops,

- because, while the service is ongoing or the MO IMS registration related signalling is ongoing, the mapping of other access attempts to a specific access category can be affected; and
- in order to avoid double barring at the start of these services or at the start of the MO IMS registration related signalling.

These services are:

- a) emergency service;
- b) MMTEL voice;
- c) MMTEL video;
- d) SMSoIP;
- e) SMS over NAS;
- f) 5GC-MO-LR procedure;
- g) UE-requested policy provisioning procedure for V2XP, ProSeP or both; and
- h) CIoT user data transfer over the control plane.

The UE considers an emergency service a) as started when 5GMM receives a request from upper layers to register for emergency services or to establish a PDU session with request type = "initial emergency request" or "existing emergency PDU session". It considers the emergency service as stopped when this PDU session is released.

In addition, the UE considers an emergency service a) as started when the 5GMM receives a request from the upper layers to perform emergency services fallback and performs emergency services fallback as specified in subclause 4.13.4.2 of 3GPP TS 23.502 [9]. In this case, the UE considers the emergency service as stopped when:

- the emergency PDU session established during the emergency services fallback is released if the UE has moved to an E-UTRA cell connected to 5GCN; or
- the service request procedure involved in the emergency services fallback is completed otherwise.

While an emergency service a) is ongoing, any access attempt triggered by the initiation of a registration, de-registration or service request procedure or by an uplink user data packet to be sent for a PDU session with suspended user-plane resources is mapped to access category 2 = emergency.

Once the emergency service has successfully passed access control, then as long as the service is ongoing, the following access attempts are allowed to proceed without further access control checking in order to avoid double barring:

- any service request procedure related to the PDU session associated with request type = "initial emergency request" or "existing emergency PDU session"; and

- any uplink user data packet to be sent for a PDU session with suspended user-plane resources associated with request type = "initial emergency request" or "existing emergency PDU session".
- NOTE 3: Although the access control checking is skipped, the mapping is performed in order to derive an RRC establishment cause.

For services b) to h) the 5GMM receives explicit start and stop indications from the upper layers.

For the case of handover of ongoing services b) to d) from non-3GPP access, the 5GMM receives an additional explicit handover of ongoing service from non-3GPP access indication from the upper layers.

The 5GMM may receive an additional explicit "call-pull-initiated" indication from the upper layers (see 3GPP TS 24.174 [13D]).

Once the service has successfully passed access control, then as long as the service is ongoing, the following access attempts are allowed to proceed without further access control checking in order to avoid double barring:

- for services b), c) and d):
 - any service request procedure related to the PDU session established for DNN = "IMS" except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0, and receiving from the lower layers an indication that the barring is alleviated for the access category determined for the access attempt; and
 - 2) any uplink user data packet to be sent for a PDU session with suspended user-plane resources established for DNN = "IMS" except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0, and receiving from the lower layers an indication that the barring is alleviated for the access category determined for the access attempt;
- for service d), if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS":
 - any service request procedure related to the PDU session established for the DNN used for SMSoIP except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0, and receiving from the lower layers an indication that the barring is alleviated for access category 6; and
 - 2) any uplink user data packet to be sent for a PDU session with suspended user-plane resources established for the DNN used for SMSoIP except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access category except category 0, and receiving from the lower layers an indication that the barring is alleviated for access category 6.

For the MO IMS registration related signalling, the 5GMM receives explicit start and stop indications from the upper layers.

Once the MO IMS registration related signalling has successfully passed access control, then as long as the MO IMS registration related signalling is ongoing, the following access attempts are allowed to proceed without further access control checking in order to avoid double barring:

- any service request procedure related to the PDU session established for DNN = "IMS" and for the DNN used for SMSoIP, if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS", except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0 and receiving from the lower layers an indication that the barring is alleviated for the access category determined for the access attempt; and
- 2) any uplink user data packet to be sent for a PDU session with suspended user-plane resources established for DNN = "IMS" and for the DNN used for SMSoIP except between receiving from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0 and receiving from the lower layers an indication that the barring is alleviated for the access category determined for the access attempt;

While an MMTEL voice call is ongoing:

- any service request procedure related to the PDU session established for DNN = "IMS" is mapped to access category 4;
- any uplink user data packet to be sent for a PDU session with suspended user-plane resources established for DNN = "IMS" is mapped to access category 4; and

- any:

- 1) service request procedure; or
- 2) registration procedure;

initiated in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclause 5.3.1.2 and 5.3.1.4) is mapped to access category 4.

While an MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any service request procedure related to the PDU session established for DNN = "IMS" is mapped to access category 5;
- any uplink user data packet to be sent for a PDU session with suspended user-plane resources established for DNN = "IMS" is mapped to access category 5; and
- any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclause 5.3.1.2 and 5.3.1.4) is mapped to access category 5.

While an SMSoIP is ongoing, no MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any service request procedure related to the PDU session established:
 - 1) for DNN = "IMS"; or
 - for the DNN used for SMSoIP, if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS";

is mapped to access category 6; and

- any uplink user data packet to be sent for a PDU session with suspended user-plane resources established:
 - 1) for DNN = "IMS"; or
 - 2) for the DNN used for SMSoIP, if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS";

is mapped to access category 6; and

- any:

- 1) service request procedure; or
- 2) registration procedure;

initiated in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclause 5.3.1.2 and 5.3.1.4) is mapped to access category 6.

While an SMS over NAS is ongoing, no SMSoIP is ongoing, no MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any:

- 1) service request procedure; or
- 2) registration procedure;

initiated in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclause 5.3.1.2 and 5.3.1.4) is mapped to access category 6.

While an MO IMS registration related signalling is ongoing, no SMSoIP is ongoing, no MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any service request procedure related to the PDU session established:
 - 1) for DNN = "IMS"; and
 - for the DNN used for SMSoIP, if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS";

is mapped to access category 9; and

- any uplink user data packet to be sent for a PDU session with suspended user-plane resources established:
 - 1) for DNN = "IMS"; and
 - 2) for the DNN used for SMSoIP, if the upper layers have indicated a DNN used for SMSoIP and the indicated DNN used for SMSoIP is different from "IMS";

is mapped to access category 9; and

- if no SMS over NAS is ongoing, any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclause 5.3.1.2 and 5.3.1.4) is mapped to access category 9.

While a 5GC-MO-LR procedure is ongoing, no SMS over NAS is ongoing, no SMSoIP is ongoing, no MO IMS registration related signalling is ongoing, no MMTEL video call is ongoing, and no MMTEL voice call is ongoing:

- any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclauses 5.3.1.2 and 5.3.1.4) is mapped to access category 3.

While a UE-requested policy provisioning procedure for V2XP, ProSeP or both (see 3GPP TS 24.587 [19B] and 3GPP TS 24.554 [19E]), no 5GC-MO-LR procedure is ongoing, no SMS over NAS is ongoing, no SMSoIP is ongoing, no MMTEL video call is ongoing, and no MMTEL voice call is ongoing:

- any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode for the purpose of NAS signalling connection recovery or following a fallback indication from the lower layers (see subclauses 5.3.1.2 and 5.3.1.4) is mapped to access category 3.

While CIoT user data transfer over the control plane is ongoing, no 5GC-MO-LR procedure is ongoing, no SMS over NAS is ongoing, no SMSoIP is ongoing, no MMTEL video call is ongoing, and no MMTEL voice call is ongoing, any service request procedure initiated in 5GMM-IDLE mode following a fallback indication from the lower layers (see subclause 5.3.1.4) is mapped to access category 7.

NOTE 3: Although the access control checking is skipped, the mapping is performed in order to derive an RRC establishment cause.

If an access category is determined and the access control checking is skipped, the NAS shall determine the RRC establishment cause from one or more determined access identities and the access category as specified in subclause 4.5.6, the NAS shall initiate the procedure to send the initial NAS message for the access attempt and shall provide the RRC establishment cause to lower layers.

If the UE receives from the lower layers an indication that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0:

- a) if an MMTEL voice call or MMTEL video call is ongoing:
 - if the UE is operating in the single-registration mode and the UE's usage setting is "voice centric", the UE may attempt to select an E-UTRA cell connected to EPC. If the UE finds a suitable E-UTRA cell connected to EPC, it then proceeds with the appropriate EMM specific procedures and, if necessary, ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.2 and 3GPP TS 24.301 [15]; and
 - if the UE is operating in the dual-registration mode, the UE may proceed in S1 mode with the appropriate EMM specific procedures and ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.3 and 3GPP TS 24.301 [15]; and
- b) if SMSoIP is ongoing or an MO IMS registration related signalling is ongoing:
 - 1) if the UE is operating in the single-registration mode, the UE may attempt to select an E-UTRA cell connected to EPC. If the UE finds a suitable E-UTRA cell connected to EPC, it then proceeds with the appropriate EMM specific procedures and, if necessary, ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.2 and 3GPP TS 24.301 [15]; and
 - if the UE is operating in the dual-registration mode, the UE may proceed in S1 mode with the appropriate EMM specific procedures and ESM procedures to make a PDN connection providing access to IMS available; see subclause 4.8.3 and 3GPP TS 24.301 [15].

4.5.6 Mapping between access categories/access identities and RRC establishment cause

When 5GMM requests the establishment of a NAS-signalling connection, the RRC establishment cause used by the UE shall be selected according to one or more access identities (see subclauses 4.5.2 and 4.5.2A) and the determined access category by checking the rules specified in table 4.5.6.1 and table 4.5.6.2. If the access attempt matches more than one rule, the RRC establishment cause of the lowest rule number shall be used. If the determined access category is an operator-defined access category, then the RRC establishment cause used by the UE shall be selected according to table 4.5.6.1 and table 4.5.6.2 based on one or more access identities (see subclauses 4.5.2 and 4.5.2A) and the standardized access category determined for the operator-defined access category as described in subclause 4.5.3.

- NOTE 1: Following an RRC release with redirection, the lower layers can set the RRC establishment cause to "mps-PriorityAccess" in the case of redirection to an NR cell connected to 5GCN (see 3GPP TS 38.331 [30]) or to "highPriorityAccess" in the case of redirection to an E-UTRA cell connected to 5GCN (see 3GPP TS 36.331 [25A]), if the network indicates to the UE during RRC connection release with redirection that the UE has an active MPS session.
- NOTE 2: In case of the UE is acting as a 5G ProSe layer-2 UE-to-network relay UE, it is possible for the lower layer to decide an applicable RRC establishment cause according to the request from the 5G ProSe layer-2 remote UE as specified in 3GPP TS 38.331 [30].

Rule #	Access identities	Access categories	RRC establishment cause is set to	
1	1	Any category	mps-PriorityAccess	
2	2	Any category	mcs-PriorityAccess	
3	11, 15	Any category	highPriorityAccess	
4	12,13,14,	Any category	highPriorityAccess	
5	0	0 (= MT_acc)	mt-Access	
		1 (= delay tolerant)	Not applicable (NOTE 1)	
		2 (= emergency)	emergency	
		3 (= MO_sig)	mo-Signalling	
		4 (= MO MMTel voice)	mo-VoiceCall	
		5 (= MO MMTel video)	mo-VideoCall	
		6 (= MO SMS and SMSoIP)	mo-SMS	
		7 (= MO_data)	mo-Data	
		9 (= MO IMS registration related signalling)	mo-Data	
NOTE 1: A UE using access category 1 for the access barring check will determine a second access category in the				
range 3 to 7 that is to be used for determination of the RRC establishment cause. See subclause 4.5.2, table 4.5.2.2, NOTE 6.				
NOTE 2 See subclaus	2. See subclause 4.5.2 table 4.5.2.1 for use of the access identities of 0.1.2 and 11-15			

Table 4.5.6.1: Mapping table for access identities/access categories and RRC establishment cause when establishing N1 NAS signalling connection via NR connected to 5GCN

Table 4.5.6.2: Mapping table for access identities/access categories and RRC establishment cause when establishing N1 NAS signalling connection via E-UTRA connected to 5GCN

Rule #	Access identities	Access categories	RRC establishment cause is set to	
1	1	Any category	highPriorityAccess	
2	2	Any category	highPriorityAccess	
3	11, 15	Any category	highPriorityAccess	
4	12,13,14,	Any category	highPriorityAccess	
5	0	0 (= MT_acc)	mt-Access	
		1 (= delay tolerant)	Not applicable (NOTE 1)	
		2 (= emergency)	emergency	
		3 (= MO_sig)	mo-Signalling	
		4 (= MO MMTel voice)	mo-VoiceCall	
		5 (= MO MMTel video)	mo-VoiceCall	
		6 (= MO SMS and SMSoIP) mo-Da		
		7 (= MO_data) mo-Dat		
		9 (= MO IMS registration related signalling)	mo-Data	
		10 (= MO exception data)	mo-ExceptionData (NOTE 3)	
NOTE 1: A UE using access category 1 for the access barring check will determine a second access category in the				
range 3 to 7 that is to be used for determination of the RRC establishment cause. See subclause 4.5.2,				
table 4.5.2.2, NOTE 6.				
NOTE 2: See subclause 4.5.2, table 4.5.2.1 for use of the access identities of 0, 1, 2, and 11-15.				

NOTE 3: This applies to the UE in NB-N1 mode.

4.6 Network slicing

4.6.1 General

The 5GS supports network slicing as described in 3GPP TS 23.501 [8]. Within a PLMN or SNPN, a network slice is identified by an S-NSSAI, which is comprised of a slice/service type (SST) and a slice differentiator (SD). Inclusion of an SD in an S-NSSAI is optional. A set of one or more S-NSSAIs is called the NSSAI. The following NSSAIs are defined in 3GPP TS 23.501 [8]:

a) configured NSSAI;

- b) requested NSSAI;
- c) allowed NSSAI;
- d) subscribed S-NSSAIs; and
- e) pending NSSAI.

The following NSSAIs are defined in the present document:

- a) rejected NSSAI for the current PLMN or SNPN;
- b) rejected NSSAI for the current registration area;
- c) rejected NSSAI for the failed or revoked NSSAA; and
- d) rejected NSSAI for the maximum number of UEs reached.

In roaming scenarios, rejected NSSAI for the current PLMN or SNPN, or rejected NSSAI for the current registration area, or rejected NSSAI for the maximum number of UEs reached includes one or more S-NSSAI for the current PLMN and also contains a set of mapped S-NSSAI(s). An S-NSSAI included in the rejected NSSAI for the failed or revoked NSSAA is an HPLMN S-NSSAI.

In case of a PLMN, a serving PLMN may configure a UE with the configured NSSAI per PLMN, and NSSRG information if the UE has indicated it support the subscription-based restrictions to simultaneous registration of network slices feature. In addition, the HPLMN may configure a UE with a single default configured NSSAI and consider the default configured NSSAI as valid in a PLMN for which the UE has neither a configured NSSAI nor an allowed NSSAI.

NOTE 1: The value(s) used in the default configured NSSAI are expected to be commonly decided by all roaming partners, e.g., values standardized by 3GPP or other bodies.

In case of an SNPN, the SNPN may configure a UE with a configured NSSAI applicable to the SNPN, and NSSRG information if the UE has indicated it support the subscription-based restrictions to simultaneous registration of network slices feature, if the UE is neither registering nor registered for onboarding services in SNPN. In addition, the credential holder may configure a single default configured NSSAI associated with the selected entry of the "list of subscriber data" or the PLMN subscription and consider the default configured NSSAI as valid in a SNPN for which the UE has neither a configured NSSAI nor an allowed NSSAI. If the UE is registering or registered for onboarding services in SNPN, the serving SNPN shall not provide a configured NSSAI to the UE.

The allowed NSSAI and the rejected NSSAI for the current registration area are managed per access type independently, i.e. 3GPP access or non-3GPP access, and is applicable for the registration area. If the UE does not have a valid registration area, the rejected NSSAI for the current registration area is applicable to the tracking area on which it was received. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the allowed NSSAI and the rejected NSSAI for the current registration area are applicable to these PLMNs in this registration area.

The allowed NSSAI that is associated with a registration area containing TAIs belonging to different PLMNs, which are equivalent PLMNs, can be used to form the requested NSSAI for any of the equivalent PLMNs when the UE is outside of the registration area where the allowed NSSAI was received.

When the network slice-specific authentication and authorization procedure is to be initiated for one or more S-NSSAIs in the requested NSSAI or the network slice-specific authentication and authorization procedure is ongoing for one or more S-NSSAIs, these S-NSSAI(s) will be included in the pending NSSAI. When the network slice-specific authentication and authorization procedure is completed for an S-NSSAI that has been in the pending NSSAI, the S-NSSAI will be moved to the allowed NSSAI or rejected NSSAI depending on the outcome of the procedure. The AMF sends the updated allowed NSSAI to the UE over the same access of the requested S-NSSAI. The AMF sends the updated rejected NSSAI over either 3GPP access or non-3GPP access. The pending NSSAI is managed regardless of access type i.e. the pending NSSAI is applicable to both 3GPP access and non-3GPP access for the current PLMN even if sent over only one of the accesses. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the pending NSSAI is applicable to these PLMNs in this registration area.

The rejected NSSAI for the current PLMN or SNPN is applicable for the whole registered PLMN or SNPN. The AMF shall only send a rejected NSSAI for the current PLMN when the registration area consists of TAIs that only belong to the registered PLMN. If the UE receives a rejected NSSAI for the current PLMN, and the registration area also contains

TAIs belonging to different PLMNs, the UE shall treat the received rejected NSSAI for the current PLMN as applicable to the whole registered PLMN.

The rejected NSSAI for the failed or revoked NSSAA includes one or more S-NSSAIs that have failed the network slice-specific authentication and authorization or for which the authorization have been revoked, and are applicable for the whole registered PLMN or SNPN.

The rejected NSSAI for the maximum number of UEs reached is applicable for the whole registered PLMN or SNPN, and the access type over which the rejected NSSAI was sent. The AMF shall send a rejected NSSAI including S-NSSAI(s) with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", when one or more S-NSSAIs are indicated that the maximum number of UEs has been reached. If the timer T3526 associated with the S-NSSAI(s) was started upon reception of the rejected NSSAI for the maximum number of UEs reached, the UE may remove the S-NSSAI(s) from the rejected NSSAI including S-NSSAI(s) with the rejection cause "S-NSSAI not available due to maximum number of UEs reached, the UE may remove the S-NSSAI(s) from the rejected NSSAI including S-NSSAI(s) with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", if the timer T3526 associated with the S-NSSAI(s) expires. If one or more S-NSSAIs are removed from the rejected NSSAI for the maximum number of UEs reached, the timer T3526 associated with the removed S-NSSAI(s) shall be stopped, if running. The UE shall not stop the timer T3526 if the UE selects an E-UTRA cell connected to EPC.

If the UE receives a rejected NSSAI for the maximum number of UEs reached, the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the UE shall treat the received rejected NSSAI for the maximum number of UEs reached as applicable to these equivalent PLMNs when the UE is in this registration area.

- NOTE 2: Based on local policies, the UE can remove an S-NSSAI from the rejected NSSAI for the failed or revoked NSSAA when the UE wants to register to the slice identified by this S-NSSAI.
- NOTE 3: Based on network local policy, network slice-specific authentication and authorization procedure can be initiated by the AMF for an S-NSSAI in rejected NSSAI for the failed or revoked NSSAA when the S-NSSAI is requested by the UE based on its local policy.
- NOTE 4: At least one S-NSSAI in the default configured NSSAI or at least one default S-NSSAI is recommended as not subject to network slice-specific authentication and authorization, in order to ensure that at least one PDU session can be established to access service, even when Network Slice-specific Authentication and Authorization fails.
- NOTE 5: At least one S-NSSAI in the default configured NSSAI or in the subscribed S-NSSAIs marked as default S-NSSAI is recommended as not subject to network slice admission control, in order to ensure that at least one PDU session can be established to access service.
- NOTE 6: The rejected NSSAI can be provided by the network via either Rejected NSSAI IE or the Extended rejected NSSAI IE.

4.6.2 Mobility management aspects

4.6.2.1 General

Upon registration to a PLMN or SNPN (except for the registration procedure for periodic registration update, the initial registration for onboarding services in SNPN, and the registration procedure for mobility registration update when registered for onboarding services in SNPN), the UE shall send to the AMF the requested NSSAI which includes one or more S-NSSAIs of the allowed NSSAI for the PLMN or SNPN or the configured NSSAI for the PLMN or SNPN and corresponds to the network slice(s) to which the UE intends to register with, if:

- a) the UE has a configured NSSAI for the current PLMN or SNPN;
- b) the UE has an allowed NSSAI for the current PLMN or SNPN; or
- c) the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and has a default configured NSSAI. In this case the UE indicates to the AMF that the requested NSSAI is created from the default configured NSSAI.

In roaming scenarios, if the mapped S-NSSAI(s) associated to the allowed NSSAI or the configured NSSAI are missing, the UE shall locally set the mapped S-NSSAI to the same value as the received S-NSSAI.

NOTE 1: The above occurs only when the UE is roaming and the AMF compliant with earlier versions of the specification omits providing to the UE a mapped S-NSSAI for one or more S-NSSAIs in, e.g., the allowed NSSAI or configured NSSAI.

Other than S-NSSAIs contained in the NSSAIs described above, the requested NSSAI can be formed based on the S-NSSAI(s) available in the UE (see subclause 5.5.1.3.2 for further details). In roaming scenarios, the UE shall also provide the mapped S-NSSAI(s) for the requested NSSAI. The AMF verifies if the requested NSSAI is permitted based on the subscribed S-NSSAIs in the UE subscription and, in roaming scenarios the mapped S-NSSAI(s) provided by the UE, and if so then the AMF shall provide the UE with the allowed NSSAI for the PLMN or SNPN, and shall also provide the UE with the mapped S-NSSAI(s) for the allowed NSSAI for the PLMN or SNPN. Additionally, if the AMF allows one or more subscribed S-NSSAIs for the UE, the AMF may include the allowed subscribed S-NSSAI(s) in the allowed NSSAI in the REGISTRATION ACCEPT message. The AMF shall ensure that there are not two or more S-NSSAIs of the allowed NSSAI of the HPLMN or SNPN. If

- a) all the S-NSSAIs included in the requested NSSAI are rejected, or the requested NSSAI was not included by the UE;
- b) there is no default S-NSSAI(s) or all subscribed S-NSSAIs marked as default are not allowed; and
- c) the UE is neither registering nor registered for onboarding services in SNPN and the UE is neither registering nor registered for emergency services;

then the AMF may reject the registration request (see subclauses 5.5.1.2.5 and 5.5.1.3.5 for further details).

In roaming scenarios, if the mapped S-NSSAI(s) associated to requested NSSAI are missing, the AMF shall locally set the mapped S-NSSAI to the same value as the received S-NSSAI.

NOTE 2: When the UE compliant with earlier versions of the specification is roaming, the UE can omit providing to the AMF a mapped S-NSSAI for one or more S-NSSAIs in requested NSSAI.

The set of network slice(s) for a UE can be changed at any time while the UE is registered to a PLMN or SNPN, and the change may be initiated by the network or the UE. In this case, the allowed NSSAI and associated registration area may be changed during the registration procedure or the generic UE configuration update procedure. The configured NSSAI and the rejected NSSAI may be changed during the registration procedure or the generic UE configuration update procedure. The default configured NSSAI may be changed by sending a UE parameters update transparent container to the UE during the NAS transport procedure.

The UE in NB-N1 mode does not include the requested NSSAI during the registration procedure if the 5GS registration type IE indicates "mobility registration updating" for the UE in NB-N1 mode, procedure is not initiated to change the slice(s) that the UE is currently registered to, and the UE is still in the current registration area.

The UE does not include the requested NSSAI during the registration procedure if the 5GS registration type IE indicates "SNPN onboarding registration" or the UE is registered for onboarding services in SNPN. The AMF does not include the allowed NSSAI during a registration procedure with the 5GS registration type IE indicating "mobility registration updating" except if the allowed NSSAI has changed for the UE. The UE considers the last received allowed NSSAI as valid until the UE receives a new allowed NSSAI. The AMF does not include the allowed NSSAI during a registration procedure with the 5GS registration" or during a registration procedure with the 5GS negistration of the UE receives a new allowed NSSAI. The AMF does not include the allowed NSSAI during a registration procedure with the 5GS registration type IE indicating "SNPN onboarding registration" or during a registration procedure when the UE is registered for onboarding services in SNPN.

The UE considers the last received allowed NSSAI as valid until the UE receives a new allowed NSSAI.

4.6.2.2 NSSAI storage

If available, the configured NSSAI(s) shall be stored in a non-volatile memory in the ME as specified in annex C. For a configured NSSAI, if there is associated NSSRG information, the NSSRG information shall also be stored in a non-volatile memory in the ME as specified in annex C. For a configured NSSAI, if there is associated NSAG information, the NSAG information shall be stored in the ME. The support for NSSRG information and NSAG information by a UE or an AMF is optional.

The allowed NSSAI(s) should be stored in a non-volatile memory in the ME as specified in annex C.

Each of the configured NSSAI stored in the UE is a set composed of at most 16 S-NSSAIs. Each of the allowed NSSAI stored in the UE is a set composed of at most 8 S-NSSAIs and is associated with a PLMN identity or SNPN identity, an access type and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of

the "list of subscriber data" or the selected PLMN subscription. Each of the configured NSSAI except the default configured NSSAI, and the rejected NSSAI is associated with a PLMN identity or SNPN identity and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. Each of the pending NSSAI stored in the UE is a set composed of at most 16 S-NSSAIs and is associated with a PLMN identity or SNPN identity and, if the UE supports access to an SNPN using credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN identity or SNPN identity and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. The S-NSSAI(s) in the rejected NSSAI for the current registration area are further associated with one or more tracking areas where the rejected S-NSSAI(s) is not available. The S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN regardless of the access type. The S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA shall be considered rejected for the current PLMN or SNPN regardless of the access type. The S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached are further associated with the access type over which the rejected NSSAI was received. There shall be no duplicated PLMN identities or SNPN identities associated with each of the list of configured NSSAI(s), pending NSSAI(s) for the failed or revoked NSSAI for the maximum number of UEs reached NSSAI(s) for the failed or revoked NSSAI for the maximum number of UEs reached.

The UE stores NSSAIs as follows:

- a) The configured NSSAI shall be stored until a new configured NSSAI is received for a given PLMN or SNPN. The network may provide to the UE the mapped S-NSSAI(s) for the new configured NSSAI which shall also be stored in the UE. When the UE is provisioned with a new configured NSSAI for a PLMN or SNPN, the UE shall:
 - 1) replace any stored configured NSSAI for this PLMN or SNPN with the new configured NSSAI for this PLMN or SNPN;
 - 2) delete any stored mapped S-NSSAI(s) for the configured NSSAI and, if available, store the mapped S-NSSAI(s) for the new configured NSSAI;
 - 3) delete any stored allowed NSSAI for this PLMN or SNPN and, if available, the stored mapped S-NSSAI(s) for the allowed NSSAI, if the UE received the new configured NSSAI for this PLMN or SNPN and the Configuration update indication IE with the Registration requested bit set to "registration requested", in the same CONFIGURATION UPDATE COMMAND message but without any new allowed NSSAI for this PLMN or SNPN included;
 - 4) delete any stored rejected NSSAI, and stop the timer T3526 associated with the deleted rejected S-NSSAI for the maximum number of UEs reached if running;
 - 4A) remove from the stored mapped S-NSSAI(s) for the rejected NSSAI for the current PLMN or SNPN and the stored mapped S-NSSAI(s) for the rejected NSSAI for the current registration area and the stored rejected NSSAI for the maximum number of UEs reached, the S-NSSAI(s), if any, included in the mapped S-NSSAI(s) for the new configured NSSAI for the current PLMN or SNPN (if the UE is roaming), and stop the timer T3526 associated with the deleted rejected S-NSSAI for the maximum number of UEs reached if running; and
 - 5) delete any S-NSSAI(s) stored in the pending NSSAI that are not included in the new configured NSSAI for the current PLMN or SNPN or any mapped S-NSSAI(s), if any, stored in the pending NSSAI that are not included in the mapped S-NSSAI(s) for the configured NSSAI (if the UE is roaming);

If the UE receives an S-NSSAI associated with a PLMN ID from the network during the PDN connection establishment procedure in EPS as specified in 3GPP TS 24.301 [15] or via ePDG as specified in 3GPP TS 24.302 [16], the UE may store the received S-NSSAI in the configured NSSAI for the PLMN identified by the PLMN ID associated with the S-NSSAI, if not already included in the configured NSSAI;

The UE may continue storing a received configured NSSAI for a PLMN and associated mapped S-NSSAI(s), if available, when the UE registers in another PLMN.

- NOTE 1: The maximum number of configured NSSAIs and associated mapped S-NSSAIs for PLMNs other than the HPLMN that need to be stored in the UE, and how to handle the stored entries, are up to UE implementation.
- ab) The NSAG information shall be stored until:

- 1) a new NSAG information is received for the registered PLMN or the registered SNPN is received over 3GPP access; or
- 2) a new configured NSSAI without any associated NSAG information is received for the registered PLMN or the registered SNPN is received over 3GPP access.

When a new NSAG information for the registered PLMN or the registered SNPN is received over 3GPP access, the UE shall replace any stored NSAG information for the registered PLMN and its equivalent PLMN(s) or the registered SNPN with the new NSAG information for the registered PLMN or the registered SNPN.

When a new configured NSSAI without any associated NSAG information for the registered PLMN or the registered SNPN is received over 3GPP access, the UE shall delete any stored NSAG information for the registered PLMN and its equivalent PLMN(s) or the registered SNPN.

The UE shall be able to store 32 NSAG entries in the NSAG information stored for the registered PLMN or the registered SNPN.

The UE shall be able to store TAI lists for up to 4 NSAG entries in the NSAG information stored for the registered PLMN or the registered SNPN.

The UE needs not to store the NSAG information when the UE is switched off or when the UE is deregistered from the registered PLMN or the registered SNPN.

- b) The allowed NSSAI shall be stored until:
 - 1) a new allowed NSSAI for the same access type (i.e. 3GPP access or non-3GPP access) is received for a given PLMN or SNPN;
 - 2) the CONFIGURATION UPDATE COMMAND message with the Registration requested bit of the Configuration update indication IE set to "registration requested" is received and contains no other parameters (see subclauses 5.4.4.2 and 5.4.4.3); or
 - 3) the REGISTRATION ACCEPT message is received with the "NSSAA to be performed" indicator of the 5GS registration result IE set to "Network slice-specific authentication and authorization is to be performed", and the REGISTRATION ACCEPT message contains a pending NSSAI and no new allowed NSSAI as described in subclause 5.5.1.2.4 and subclause 5.5.1.3.4.

The network may provide to the UE the mapped S-NSSAI(s) for the new allowed NSSAI (see subclauses 5.5.1.2 and 5.5.1.3) which shall also be stored in the UE. When a new allowed NSSAI for a PLMN or SNPN is received, the UE shall:

- 1) replace any stored allowed NSSAI for this PLMN or SNPN and its equivalent PLMN(s) for the same access type with the new allowed NSSAI for this PLMN or SNPN;
- delete any stored mapped S-NSSAI(s) for the allowed NSSAI for this PLMN or SNPN and its equivalent PLMN(s) for the same access type and, if available, store the mapped S-NSSAI(s) for the new allowed NSSAI;
- 3) remove from the stored rejected NSSAI for the current PLMN or SNPN, the rejected NSSAI for the current registration area and rejected NSSAI for the maximum number of UEs reached, the S-NSSAI(s), if any, included in the new allowed NSSAI for the current PLMN or SNPN, unless the S-NSSAI(s), if the rejected NSSAI is associated with one or more S-NSSAI(s) in the stored mapped rejected NSSAI and these mapped S-NSSAI(s) are not included in the mapped S-NSSAI(s) for the new allowed NSSAI, and stop the timer T3526 associated with the deleted rejected S-NSSAI for the maximum number of UEs reached if running;
- remove from the stored rejected NSSAI for the failed or revoked NSSAA, the S-NSSAI(s), if any, included in the new allowed NSSAI for the current PLMN or SNPN (if the UE is not roaming) or the mapped S-NSSAI(s) for the new allowed NSSAI for the current PLMN or SNPN (if the UE is roaming);
- 5) remove from the stored mapped S-NSSAI(s) for the rejected NSSAI for the current PLMN or SNPN, the stored mapped S-NSSAI(s) for the rejected NSSAI for the current registration area and rejected NSSAI for the maximum number of UEs reached, the S-NSSAI(s), if any, included in the mapped S-NSSAI(s) for the new allowed NSSAI for the current PLMN or SNPN (if the UE is roaming), and stop the timer T3526 associated with the deleted rejected S-NSSAI for the maximum number of UEs reached if running; and

6) remove from the stored pending NSSAI for this PLMN or SNPN and its equivalent PLMN(s), one or more S-NSSAIs, if any, included in the new allowed NSSAI for the current PLMN or SNPN and its equivalent PLMN(s) (if the UE is not roaming) or the mapped S-NSSAI(s) for the new allowed NSSAI for the current PLMN or SNPN and its equivalent PLMN(s) (if the UE is roaming).

If the UE receives the CONFIGURATION UPDATE COMMAND message with the Registration requested bit of the Configuration update indication IE set to "registration requested" and contains no other parameters (see subclauses 5.4.4.2 and 5.4.4.3), the UE shall delete any stored allowed NSSAI for this PLMN or SNPN, and delete any stored mapped S-NSSAI(s) for the allowed NSSAI, if available;

- NOTE 2: Whether the UE stores the allowed NSSAI and the mapped S-NSSAI(s) for the allowed NSSAI also when the UE is switched off is implementation specific.
- c) When the UE receives the S-NSSAI(s) included in the rejected NSSAI in the REGISTRATION ACCEPT message, the REGISTRATION REJECT message, the DEREGISTRATION REQUEST message or in the CONFIGURATION UPDATE COMMAND message, the UE shall:
 - store the S-NSSAI(s) into the rejected NSSAI and the mapped S-NSSAI(s) for the rejected NSSAI based on the associated rejection cause(s);
 - 2) if the UE receives the S-NSSAI(s) included in the Rejected NSSAI IE, or if the UE receives the S-NSSAI(s) included in the Extended rejected NSSAI IE in non-roaming case, remove from the stored allowed NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), the S-NSSAI(s), if any, included in the:
 - i) rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - ii) rejected NSSAI for the current registration area, associated with the same access type; or
 - iii) rejected NSSAI for the maximum number of UEs reached, associated with the same access type;
 - 3) if the UE receives the S-NSSAI(s) included in the Extended rejected NSSAI IE in roaming case, remove from the stored allowed NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), the S-NSSAI(s), if any, included in the:
 - i) rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - ii) rejected NSSAI for the current registration area, associated with the same access type; or
 - iii) rejected NSSAI for the maximum number of UEs reached, associated with the same access type;

if the mapped S-NSSAI(s) for the S-NSSAI in the stored allowed NSSAI for the current PLMN or SNPN are stored in the UE, and the all of the mapped S-NSSAI are included in the Extended rejected NSSAI IE;

- 4) remove from the stored allowed NSSAI for the current PLMN or SNPN and its equivalent PLMN(s) (if the UE is not roaming) or the stored mapped S-NSSAI(s) for the allowed NSSAI (if available and if the UE is roaming), the S-NSSAI(s), if any, included in the:
 - i) rejected NSSAI for the failed or revoked NSSAA, for each and every access type;
 - ii) mapped S-NSSAI(s) for the rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - iii) mapped S-NSSAI(s) for the rejected NSSAI for the current registration area, associated with the same access type; or
 - iv) mapped S-NSSAI(s) for the rejected NSSAI for the maximum number of UEs reached, associated with the same access type;
- 5) if the UE receives the S-NSSAI(s) included in the Rejected NSSAI IE, or if the UE receives the S-NSSAI(s) included in the Extended rejected NSSAI IE in non-roaming case, remove from the stored pending NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), the S-NSSAI(s), if any, included in the:
 - i) rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - ii) rejected NSSAI for the current registration area, associated with the same access type; or

iii) rejected NSSAI for the maximum number of UEs reached, associated with the same access type;

- 6) if the UE receives the S-NSSAI(s) included in the Extended rejected NSSAI IE in roaming case, remove from the stored pending NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), the S-NSSAI(s), if any, included in the:
 - i) rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - ii) rejected NSSAI for the current registration area, associated with the same access type; or
 - iii) rejected NSSAI for the maximum number of UEs reached, associated with the same access type,

if the mapped S-NSSAI(s) for the S-NSSAI in the stored pending NSSAI are stored in the UE, and all of the mapped S-NSSAI(s) are included in the Extended rejected NSSAI IE; and

- 7) remove from the stored pending NSSAI for the current PLMN and its equivalent PLMN(s) or SNPN (if the UE is not roaming) or the stored mapped S-NSSAI(s) for the pending NSSAI (if available and if the UE is roaming), the S-NSSAI(s) included in the:
 - i) rejected NSSAI for the failed or revoked NSSAA, for each and every access type;
 - ii) mapped S-NSSAI(s) for the rejected NSSAI for the current PLMN or SNPN, for each and every access type;
 - iii) mapped S-NSSAI(s) for the rejected NSSAI for the current registration area, associated with the same access type; or
 - iv) mapped S-NSSAI(s) for the rejected NSSAI for the maximum number of UEs reached, associated with the same access type;
- 8) If the UE receives the CONFIGURATION UPDATE COMMAND message with the Registration requested bit of the Configuration update indication IE set to "registration requested" and contains no other parameters (see subclauses 5.4.4.2 and 5.4.4.3), the UE shall delete any stored rejected NSSAI.

When the UE:

- 1) enters state 5GMM-DEREGISTERED following an unsuccessful registration for 5GMM causes other than #62 "No network slices available" for the current PLMN or SNPN;
- 2) successfully registers with a new PLMN or SNPN;
- 3) enters state 5GMM-DEREGISTERED following an unsuccessful registration with a new PLMN; or
- 4) performs inter-system change from N1 mode to S1 mode and the UE successfully completes tracking area update procedure;

and the UE is not registered with the current PLMN or SNPN over another access, the rejected NSSAI for the current PLMN or SNPN and the rejected NSSAI for the failed or revoked NSSAA shall be deleted.

When the UE receive ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST message provided with S-NSSAI and the PLMN ID in the protocol configuration options IE or extended protocol configuration options IE (see subclause 6.2.2 of 3GPP TS 24.301 [15]), the UE shall remove the S-NSSAI from the rejected NSSAI for the current PLMN. When the UE receive ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST message provided with S-NSSAI and the PLMN ID in the protocol configuration options IE or extended protocol configuration options IE (see subclause 6.2.2 of 3GPP TS 24.301 [15]), the UE may remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached for each and every access type, if any, and stop the timer T3526 associated with the S-NSSAI if running.

When the UE:

- 1) deregisters over an access type;
- 2) successfully registers in a new registration area over an access type;
- 3) enters state 5GMM-DEREGISTERED or 5GMM-REGISTERED following an unsuccessful registration in a new registration area over an access type; or

4) performs inter-system change from N1 mode to S1 mode and the UE successfully completes tracking area update procedure;

the rejected NSSAI for the current registration area corresponding to the access type shall be deleted;

d) When the UE receives the pending NSSAI in the REGISTRATION ACCEPT message, the UE shall replace any stored pending NSSAI for this PLMN or SNPN with the new pending NSSAI received in the REGISTRATION ACCEPT message for this PLMN or SNPN. If the UE does not receive the pending NSSAI in the REGISTRATION ACCEPT message and the "NSSAA to be performed" indicator is not set to "Network slice-specific authentication and authorization is to be performed" in the 5GS registration result IE of the REGISTRATION ACCEPT message, the UE shall delete the stored pending NSSAI, if any, for this PLMN or SNPN and its equivalent PLMN(s).

If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, then for each of the equivalent PLMNs, the UE shall replace any stored pending NSSAI with the pending NSSAI received in the registered PLMN.

When the UE:

- 1) deregisters with the current PLMN or SNPN using explicit signalling or enters state 5GMM-DEREGISTERED for the current PLMN or SNPN;
- 2) successfully registers with a new PLMN or SNPN not in the list of equivalent PLMNs;
- 3) enters state 5GMM-DEREGISTERED following an unsuccessful registration with a new PLMN or SNPN; or
- 4) successfully initiates an attach or tracking area update procedure in S1 mode and the UE is operating in single-registration mode;

and the UE is not registered with the current PLMN or SNPN over another access, the pending NSSAI for the current PLMN or SNPN and its equivalent PLMN(s) shall be deleted;

- e) When the UE receives the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed" in the REGISTRATION ACCEPT message or in the CONFIGURATION UPDATE COMMAND message, the UE shall delete the network slicing information for each of the PLMNs or SNPNs that the UE has slicing information stored for (excluding the current PLMN or SNPN). The UE shall delete any stored rejected NSSAI and stop the timer T3526 associated with the deleted rejected S-NSSAI for the maximum number of UEs reached if running. The UE shall not delete the default configured NSSAI. Additionally, the UE shall update the network slicing information for the current PLMN or SNPN (if received) as specified above in bullets a), b), c) and d); and
- f) When the UE receives the new default configured NSSAI included in the default configured NSSAI update data in the Payload container IE of DL NAS TRANSPORT message, the UE shall replace any stored default configured NSSAI with the new default configured NSSAI. In case of SNPN, the UE shall replace the stored default configured NSSAI associated with the selected entry of the "list of subscriber data" or the PLMN subscription with the new default configured NSSAI.

4.6.2.3 Provision of NSSAI to lower layers in 5GMM-IDLE mode

The UE NAS layer may provide the lower layers with an NSSAI (either requested NSSAI or allowed NSSAI) when the UE in 5GMM-IDLE mode sends an initial NAS message.

The AMF may indicate, via the NSSAI inclusion mode IE of a REGISTRATION ACCEPT message, an NSSAI inclusion mode in which the UE shall operate over the current access within the current PLMN or SNPN, if any (see subclauses 5.5.1.2.4 and 5.5.1.3.4), where the NSSAI inclusion mode is chosen among the following NSSAI inclusion modes described in table 4.6.2.3.1.

Initial NAS message	NSSAI	NSSAI	NSSAI	NSSAI
	inclusion	inclusion	inclusion	inclusion
	Requested	Requested	Requested	
i) including the 5GS registration	NSSAL if	NSSAL if	NSSAL if	NUNSOAI
type IE set to "initial registration"	any	any	any	
REGISTRATION REQUEST message:	Requested	Requested	Requested	No NSSAI
including the 5GS registration	NSSAI, if	NSSAI, if	NSSAI, if	
type IE set to "mobility registration	any	any	any	
updating"; and				
a) or n) in subclause 5.5.1.3.2				
REGISTRATION REQUEST message:	Allowed	Allowed	No NSSAL if	No NSSAI
i) including the 5GS registration	NSSAI, if	NSSAI, if	any	
type IE set to "mobility registration	any	any	-	
updating"; and				
ii) initiated by case g) or n) in				
	Allowed	Allowed		
i) including the 5GS registration	NSSAL if	NSSAL if	NU NOSAI	NU NOSAI
type IE set to "periodic registration	any	any		
updating"	2	,		
SERVICE REQUEST message	Allowed	See	No NSSAI	No NSSAI
	NSSAI, if	NOTE 1		
	any any			
NOTE 1: All the S-NSSAIS of the PDU sessio	ns that have the	user-plane reso	ontrol plane inte	to be re-
triggering the service request is related to (see 3GPP TS 23 501 [8])				
NOTE 2: For a REGISTRATION REQUEST message which is triggered by emergency services, a				
DEREGISTRATION REQUEST message and a SERVICE REQUEST message which includes				
the service type IE set to "emergency services" or "emergency services fallback", no NSSAI is				
provided to the lower layers. If the UE performs initial registration for onboarding services in				
SINFIN OF IS REGISTERED FOR ONDOARDING SERVICES IN SINFIN, THE UE NAS layer shall not provide the				provide the
NOTE 3. The mapped configured S-NSSAI(s) from the S-NSSAI(s) of the HPI MN are not included as part				uded as part
of the S-NSSAIs in the requested NSSAI or the allowed NSSAI when it is provided to the lower				the lower
layers.			•	

Table 4.6.2.3.1: NSSAI inclusion modes and NSSAI which shall be provided to the lower layers

The UE shall store the NSSAI inclusion mode:

- a) indicated by the AMF, if the AMF included the NSSAI inclusion mode IE in the REGISTRATION ACCEPT message; or
- b) decided by the UE, if the AMF did not include the NSSAI inclusion mode IE in the REGISTRATION ACCEPT message;

together with the identity of the current PLMN or SNPN and access type in a non-volatile memory in the ME as specified in annex C.

The UE shall apply the NSSAI inclusion mode received in the REGISTRATION ACCEPT message over the current access within the current PLMN and its equivalent PLMN(s) or the current SNPN, if any, in the current registration area.

When a UE performs a registration procedure to a PLMN which is not a PLMN in the current registration area or an SNPN, if the UE has no NSSAI inclusion mode for the PLMN or the SNPN stored in a non-volatile memory in the ME, the UE shall provide the lower layers with:

- a) no NSSAI if the UE is performing the registration procedure over 3GPP access; or
- b) requested NSSAI if the UE is performing the registration procedure over non-3GPP access.

When a UE performs a registration procedure after an inter-system change from S1 mode to N1 mode, if the UE has no NSSAI inclusion mode for the PLMN stored in a non-volatile memory in the ME and the registration procedure is performed over 3GPP access, the UE shall not provide the lower layers with any NSSAI over the 3GPP access.

4.6.2.4 Network slice-specific authentication and authorization

The UE and network may support network slice-specific authentication and authorization.

A serving PLMN or SNPN shall perform network slice-specific authentication and authorization for the S-NSSAI(s) of the HPLMN or SNPN which are subject to it based on subscription information. The UE shall indicate whether it supports network slice-specific authentication and authorization in the 5GMM Capability IE in the REGISTRATION REQUEST message as specified in subclauses 5.5.1.2.2 and 5.5.1.3.2.

The upper layer stores an association between each S-NSSAI and its corresponding credentials for the network slice-specific authentication and authorization.

NOTE 1: The credentials for network slice-specific authentication and authorization and how to provision them in the upper layer are out of the scope of 3GPP.

The network slice-specific authentication and authorization procedure shall not be performed unless the primary authentication and key agreement procedure as specified in the subclause 5.4.1 has successfully been completed.

The AMF informs the UE about S-NSSAI(s) for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing in the pending NSSAI. The AMF informs the UE about S-NSSAI(s) for which NSSAA procedure is completed as success in the allowed NSSAI. The AMF informs the UE about S-NSSAI(s) for which NSSAA procedure is completed as failure in the rejected NSSAI for the failed or revoked NSSAA. The AMF stores and handles allowed NSSAI, pending NSSAI, rejected NSSAI, and 5GS registration result in the REGISTRATION ACCEPT message according to subclauses 5.5.1.2.4 and 5.5.1.3.4.

- NOTE 2: The AMF maintains the NSSAA procedure status for each S-NSSAI, as specified in 3GPP TS 29.518 [20B] and the NSSAA procedure status for each S-NSSAI is not impacted by NSAC as specified in subclauses 4.6.2.5 and 4.6.3.1.
- NOTE 3: Upon completion of NSSAA procedures, it can happen that the total number of S-NSSAIs which need to be included in the allowed NSSAI exceeds eight. In this case, it is up to the AMF implementation on how to pick up the S-NSSAIs included in the allowed NSSAI.
- NOTE 4: It can happen that one or more S-NSSAIs included in the received allowed NSSAI, are not the S-NSSAIs that the UE intends to register to. In this case, it is up to the UE implementation on how to use these S-NSSAIs.

To perform network slice-specific authentication and authorization for an S-NSSAI, the AMF invokes an EAP-based network slice-specific authentication and authorization procedure for the S-NSSAI, see subclause 5.4.7 and 3GPP TS 23.502 [9] using the EAP framework as described in 3GPP TS 33.501 [24].

The AMF updates the allowed NSSAI and the rejected NSSAI using the generic UE configuration update procedure as specified in the subclause 5.4.4 after the network slice-specific authentication and authorization procedure is completed.

The AMF shall send the pending NSSAI containing all S-NSSAIs for which the network slice-specific authentication and authorization procedure (except for re-NSSAA) will be performed or is ongoing in the REGISTRATION ACCEPT message. The AMF shall also include in the REGISTRATION ACCEPT message the allowed NSSAI containing one or more S-NSSAIs from the requested NSSAI which are allowed by the AMF and for which network slice-specific authentication procedure or the network slice-specific authorization procedure or the network slice-specific authorization procedure can be invoked by the network for a UE supporting NSSAA at any time. After the network performs the network slice-specific re-authentication and re-authorization procedure or network slice-specific authorization revocation procedure:

- a) if network slice-specific authentication and authorization fails or network slice-specific authorization is revoked for some but not all S-NSSAIs in the allowed NSSAI, the AMF updates the allowed NSSAI and the rejected NSSAI accordingly using the generic UE configuration update procedure as specified in the subclause 5.4.4 and inform the SMF to release all PDU sessions associated with the S-NSSAI for which network slice-specific reauthentication and re-authorization fails or network slice-specific authorization is revoked;
- b) if network slice-specific authentication and authorization fails or network slice-specific authorization is revoked for all S-NSSAIs in the allowed NSSAI but there are one or more default S-NSSAIs which are not subject to network slice-specific authentication and authorization or for which the network slice-specific authentication and authorization has been successfully performed, the AMF updates the allowed NSSAI containing these default S-NSSAIs and the rejected NSSAI accordingly using the generic UE configuration update procedure as specified

in the subclause 5.4.4. The AMF shall also inform the SMF to release all PDU sessions associated with the S-NSSAI for which network slice-specific re-authentication and re-authorization fails or network slice-specific authorization is revoked; or

c) if network slice-specific authentication and authorization fails or network slice-specific authorization is revoked for all S-NSSAIs in the allowed NSSAI and all default S-NSSAIs are subject to network slice-specific authentication and authorization, then AMF performs the network-initiated de-registration procedure and includes the rejected NSSAI in the DEREGISTRATION REQUEST message as specified in the subclause 5.5.2.3 except when the UE has an emergency PDU session established or the UE is establishing an emergency PDU session. In this case the AMF shall send the CONFIGURATION UPDATE COMMAND message containing rejected NSSAI and inform the SMF to release all PDU sessions associated with the S-NSSAI for which network slice-specific re-authentication and re-authorization fails or network slice-specific authorization is revoked. After the emergency PDU session is released, the AMF performs the network-initiated de-registration procedure as specified in the subclause 5.5.2.3.

The UE does not include in the requested NSSAI any of the S-NSSAIs from the pending NSSAI that the UE stores, regardless of the access type. When the UE storing a pending NSSAI intends to register to one or more additional S-NSSAIs not included in the pending NSSAI, the UE initiates the registration procedure with a requested NSSAI containing these S-NSSAIs as described in subclause 5.5.1.3.2. In this case, the requested NSSAI shall also include one or more S-NSSAIs from the allowed NSSAI, if the UE still wants to use the S-NSSAI(s) from the allowed NSSAI.

During the registration procedure, when the AMF receives a requested NSSAI from a UE over an access type, for which there is a pending NSSAI including one or more S-NSSAIs that were previously requested over the same access type, the AMF considers S-NSSAIs included in the requested NSSAI and S-NSSAIs in the pending NSSAI that were previously requested over the same access type as requested S-NSSAIs by the UE. The AMF handles the requested S-NSSAIs as described in subclause 5.5.1.3.4.

When performing the network slice-specific re-authentication and re-authorization procedure if the S-NSSAI is included in the allowed NSSAI for both 3GPP and non-3GPP accesses, and the UE is registered to both 3GPP and non-3GPP accesses in the same PLMN, then the AMF selects an access type to perform network slice-specific authentication and authorization based upon operator policy.

If network slice-specific authorization is revoked for an S-NSSAI that is in the current allowed NSSAI for an access type, the AMF shall:

- a) provide a new allowed NSSAI, excluding the S-NSSAI for which the network slice-specific authorization is revoked; and
- b) provide a new rejected NSSAI for the failed or revoked NSSAA, including the S-NSSAI for which the network slice-specific authorization is revoked, with the rejection cause "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization",

to the UE using the generic UE configuration update procedure as specified in the subclause 5.4.4 and inform the SMF to release all PDU sessions associated with the S-NSSAI for which the network slice-specific authorization is revoked for this access type.

If the UE requests the establishment of a new PDU session or the modification of a PDU session for an S-NSSAI for which the AMF is performing network slice-specific re-authentication and re-authorization procedure, the AMF may determine to not forward the 5GSM message to the SMF as described in subclause 5.4.5.2.4.

NOTE 5: If the AMF receives the HTTP code set to "4xx" or "5xx" as specified in 3GPP TS 29.500 [20AA] or the AMF detects that the NSSAAF failure as specified in 3GPP TS 29.526 [21A] during the NSSAA procedure for an S-NSSAI, then the AMF considers the NSSAA procedure has failed for this S-NSSAI.

4.6.2.5 Mobility management based network slice admission control

A serving PLMN or SNPN can perform network slice admission control for the S-NSSAI(s) subject to NSAC to monitor and control the number of registered UEs per network slice. The timing of the network slice admission control is managed by the EAC mode per network slice, which can be either activated or deactivated for the network performing network slice admission control. The EAC mode is activated when the number of UEs associated with the S-NSSAI reaches a certain threshold (see 3GPP TS 23.502 [9])

If the EAC mode is activated for an S-NSSAI, the AMF performs network slice admission control before the S-NSSAI subject to NSAC is included in the allowed NSSAI sent to the UE. During a registration procedure (including initial

registration or mobility registration updating from another AMF), if the AMF determines that the maximum number of UEs has been reached for:

- a) one or more S-NSSAIs but not all S-NSSAIs in the requested NSSAI, then the AMF includes the allowed NSSAI and the rejected NSSAI accordingly in the REGISTRATION ACCEPT message as specified in the subclauses 5.5.1.2.4 and 5.5.1.3.4;
- b) all S-NSSAIs in the requested NSSAI but there are one or more default S-NSSAIs which can be allowed to the UE, then the AMF includes the allowed NSSAI containing these default S-NSSAIs and the rejected NSSAI accordingly in the REGISTRATION ACCEPT message as specified in the subclauses 5.5.1.2.4 and 5.5.1.3.4; or
- c) all S-NSSAIs in the requested NSSAI and there are no default S-NSSAIs which can be allowed to the UE, then the AMF includes the rejected NSSAI accordingly in the REGISTRATION REJECT message as specified in the subclauses 5.5.1.2.5 and 5.5.1.3.5.

If the EAC mode is deactivated for an S-NSSAI, the AMF performs network slice admission control after the S-NSSAI subject to NSAC is included in the allowed NSSAI sent to the UE. While the AMF is waiting for response from the NSCAF for the S-NSSAI, the AMF processes the NAS signalling message related to the S-NSSAI as usual i.e. like S-NSSAI in the allowed NSSAI. After the network performs the network slice admission control, if the AMF determines that the maximum number of UEs has been reached for:

- a) one or more S-NSSAIs but not all S-NSSAIs in the allowed NSSAI, then the AMF updates the allowed NSSAI and the rejected NSSAI accordingly using the generic UE configuration update procedure as specified in the subclause 5.4.4;
- b) for all S-NSSAIs in the allowed NSSAI but there are one or more default S-NSSAIs which can be allowed to the UE, then the AMF updates the allowed NSSAI containing these default S-NSSAIs and the rejected NSSAI accordingly using the generic UE configuration update procedure as specified in the subclause 5.4.4; or
- c) for all S-NSSAIs in the allowed NSSAI and there are no default S-NSSAIs which can be allowed to the UE, then the AMF performs the network-initiated de-registration procedure and includes the rejected NSSAI in the DEREGISTRATION REQUEST message as specified in the subclause 5.5.2.3 except when the UE has an emergency PDU session established or the UE is establishing an emergency PDU session.

When the UE has an emergency PDU session established or the UE is establishing an emergency PDU session, the AMF updates the rejected NSSAI using the generic UE configuration update procedure as specified in the subclause 5.4.4 and informs the SMF to release all PDU sessions associated with the S-NSSAI. During the generic UE configuration update procedure, the AMF includes the 5GS registration result IE in the CONFIGURATION UPDATE COMMAND message and sets the Emergency registered bit of the 5GS registration result IE to "Registered for emergency services". After the emergency PDU session is released, the AMF performs the network-initiated de-registration procedure as specified in the subclause 5.5.2.3.

Based on operator policy, the mobility management based network slice admission control is not applicable for the S-NSSAI used for emergency services, or the mobility management based network slice admission control result is ignored for the S-NSSAI used for emergency services.

Based on operator policy, the mobility management based network slice admission control is not applicable for the UEs configured for priority services, or the mobility management based network slice admission control result is ignored for the UEs configured for priority services.

NOTE: A UE configured for priority services can be identified based on the RRC establishment cause received from the NG-RAN or based on the MPS priority information in the user's subscription context obtained from the UDM.

The mobility management based network slice admission control is not applicable to a UE that is registering or registered for onboarding services in SNPN.

4.6.2.6 Provision of NSAG information to lower layers

The NSAG information provided by the network and stored in the UE includes a list of NSAGs each of which contains:

- a) an NSAG ID;
- b) a list of S-NSSAI(s), which are associated with the NSAG and shall be part of the configured NSSAI;

- c) a priority value that is associated with the NSAG; and
- NOTE 1: The AMF can take local configuration, UE 5GMM capabilities, subscribed S-NSSAIs, HPLMN, etc. to determine the NSAG priority information for the associated NSAG to a UE.
- d) optionally a list of TAIs in which the NSAG is valid. If it is not provided by the network, the NSAG is valid in the PLMN or SNPN which has sent the NSAG information and its equivalent PLMN(s)..
- NOTE 2: If the NSAG for the PLMN and its equivalent PLMN(s) have different associations with S-NSSAIs, then the AMF includes a list of TAIs in the NSAG information.

The UE NAS layer shall provide the lower layers with:

- a) the most recent NSAG information stored in the UE (see subclause 4.6.2.2);
- b) the allowed NSSAI or the requested NSSAI for the purpose of network slice-based cell reselection (see 3GPP TS 23.501 [8]); and
- c) zero or more S-NSSAIs related to an access attempt for the purpose of network slice-based random access, when the access attempt is made by the UE in 5GMM-IDLE mode or 5GMM-CONNECTED mode with RRC inactive indication, determined as follows:
 - S-NSSAI(s) triggering the initial NAS message (requested NSSAI (if any) in case of the REGISTRATION REQUEST message and S-NSSAI(s) associated with all the PDU sessions included in the Uplink data status IE (if any), PDU session status IE (if any), or Allowed PDU session status IE (if any) in case of the (CONTROL PLANE) SERVICE REQUEST message), if an access attempt occurred due to an initial NAS message;
 - ii) the S-NSSAI associated with the PDU session, if an access attempt occurred due to:
 - an uplink user data packet to be sent for a PDU session with suspended user-plane resources;
 - an UL NAS TRANSPORT which carries a 5GSM message for a PDU session associated with an S-NSSAI (if any); or
 - CIOT user data to be sent in a CONTROL PLANE SERVICE REQUEST message or an UL NAS TRANSPORT message;
 - iii) no S-NSSAI, if an access attempt occurred due to:
 - the deregistration procedure;
 - the UE-initiated NAS transport procedure for sending SMS, LPP message, SOR transparent container, UE policy container, UE parameters update transparent container, or a location services message; or
 - emergency services; or
 - iv) the allowed NSSAI (if any), if an access attempt occurred for other reason than those specified in bullets i) iii).
- NOTE: The AMF can take local configuration, UE 5GMM capabilities, subscribed S-NSSAIs, HPLMN, etc. to determine the NSAG priority information for the associated NSAG to a UE.

4.6.3 Session management aspects

4.6.3.0 General

In order to enable PDU transmission in a network slice, the UE may request establishment of a PDU session in a network slice towards a data network (DN) which is associated with an S-NSSAI and a data network name (DNN) if there is no established PDU session adequate for the PDU transmission. The S-NSSAI included is part of allowed NSSAI of the serving PLMN or SNPN, which is an S-NSSAI value valid in the serving PLMN or SNPN, and in roaming scenarios the mapped S-NSSAI is also included for the PDU session if available. See subclause 6.4.1 for further details. The UE determines whether to establish a new PDU session or use one of the established PDU session(s) based on the URSP rules which include S-NSSAIs, if any (see subclause 6.2.9), or based on UE local configuration, as described in subclause 4.2.2 of 3GPP TS 24.526 [19].

4.6.3.1 Session management based network slice admission control

A serving PLMN or the HPLMN, or SNPN can perform network slice admission control for the S-NSSAI(s) subject to NSAC to monitor and control the total number of established PDU sessions per network slice. The SMF performs network slice admission control on the S-NSSAI during the PDU session establishment procedure. If the maximum number of PDU session on a network slice associated with an S-NSSAI has been already reached, the SMF rejects the PDU session establishment request using S-NSSAI based congestion control as specified in subclause 6.2.8 and 6.4.1.4.2.

The SMF performs network slice admission control on the S-NSSAI for a PDU session that is associated with the non-3GPP access, when the UE requests to transfer a session from the non-3GPP access to the 3GPP access with the Allowed PDU session status IE as described in subclause 5.6.1.4. If the maximum number of PDU sessions on a network slice associated with an S-NSSAI has been already reached, the SMF rejects the request to establish the user-plane resources (see 3GPP TS 29.502 [20A]).

Based on operator policy, the session management based network slice admission control is not applicable for the PDU session for emergency services, or the session management based network slice admission control result is ignored for the PDU session for emergency services.

Based on operator policy, the session management based network slice admission control is not applicable for the PDU session for priority services, or the session management based network slice admission control result is ignored for the PDU session for priority services.

NOTE: How the SMF determines that the PDU session is used for priority services is outside the scope of this release of the present document.

The session management based network slice admission control is not applicable to PDU session established for onboarding services in SNPN.

- NOTE 1: For the MA PDU session during the PDU session establishment procedure, the SMF performs network slice admission control only when it is newly established over the associated access type.
- NOTE 2: For a set of redundant PDU sessions, the SMF performs network slice admission control for each PDU session independently.

4.6.3.2 Support of network slice admission control and interworking with EPC

If EPS counting is required for a network slice, the network performs network slice admission control for the S-NSSAI(s) subject to NSAC to monitor and control the number of UEs per network slice and number of PDU sessions per network slice during the PDN connection establishment procedure. If the maximum number of UEs on a network slice associated with an S-NSSAI or the maximum number of PDU sessions on a network slice associated with an S-NSSAI have already been reached, the network rejects the PDN connectivity request using ESM cause #26 "insufficient resources" as specified in 3GPP TS 24.301 [15].

NOTE: If there are more than one S-NSSAI associated with the APN used in the PDN connectivity request and some of but not all associated S-NSSAIs are not available due to either maximum number of UEs reached or maximum number of PDU sessions reached, the network can use the associated S-NSSAI for which maximum number of UEs and maximum number of PDU sessions have not reached to avoid PDN connectivity request rejection.

4.6.3.3 Session management based network slice data rate limitation control

A serving PLMN or the HPLMN can perform network slice data rate limitation control for the S-NSSAI(s) subject to network slice data rate limitation control to monitor and control the total data rate of established PDU sessions per network slice as specified in 3GPP TS 23.503 [10]. If the maximum data rate of PDU sessions on a network slice associated with an S-NSSAI has been exceeded during the PDU session establishment procedure, the SMF may reject the PDU session establishment request using S-NSSAI based congestion control as specified in clause 6.2.8 and 6.4.1.4.2.

A serving PLMN or the HPLMN can perform management of Slice-Maximum Bit Rate per UE (UE-Slice-MBR) as specified in 3GPP TS 23.503 [10]. When the UE-Slice-MBR for the UE and S-NSSAI to which the PDU session is allocated is exceeded during the PDU session establishment procedure, the SMF may reject the PDU session establishment request using S-NSSAI based congestion control as specified in clause 6.2.8 and 6.4.1.4.2.

NOTE 1: Based on operator policy, the network slice data rate limitation control can be not applicable for the S-NSSAI(s) used for emergency services or priority services.

NOTE 2: The network slice data rate limitation control and UE-Slice-MBR management are performed by the PCF.

4.7 NAS over non-3GPP access

4.7.1 General

From the UE's NAS perspective, in general the procedures and messages defined for 5GMM and 5GSM are used over non-3GPP access as over 3GPP access. However, a number of aspects are different as described in the following subclauses.

4.7.2 5GS mobility management aspects

4.7.2.1 General

The mobility management procedures defined over 3GPP access are re-used over non-3GPP access with the following exceptions:

- a) the registration status, and the 5GMM parameters of the UE's 3GPP access and non-3GPP access 5GMM state machine instances are independent in each of these accesses and can be different;
- b) single-registration mode and dual-registration mode do not apply for 5GMM over non-3GPP access;
- c) the RPLMN over non-3GPP access can be different from the RPLMN over 3GPP access. The MCC of the RPLMN over 3GPP access and the MCC of the RPLMN over the non-3GPP access can also be different;
- d) the registration for 3GPP access and for non-3GPP access are performed separately. Like for 3GPP access, an access stratum connection exists before the UE can perform the registration procedure for non-3GPP access. As at registration over non-3GPP access the UE is allocated a registration area, which is associated with a single TAI, list management of registration areas is not required, and registration updating due to registration area change with the registered PLMN is not performed. Furthermore, the periodic registration update procedure is also not performed. New registration at change of PLMN is required;
- e) the 5GMM over non-3GPP access in the UE considers that the N1 NAS signalling connection is established when the lower layers indicate that the access stratum connection is established successfully;
- f) the UE-initiated service request procedure via non-3GPP access is supported. Upon indication from the lower layers of non-3GPP access, that the access stratum connection is established between the UE and the network, the UE in 5GMM-REGISTERED state and in 5GMM-IDLE mode over non-3GPP access shall initiate the service request procedure via non-3GPP access. The UE may indicate with the service request message the PDU session(s) associated with non-3GPP access to re-establish user-plane resources for which the UE has pending user data to be sent;
- g) paging procedure is not performed via non-3GPP access;
- h) service area restrictions do not apply for non-3GPP access other than the wireline access;
- i) the establishment cause for non-3GPP access is determined according to subclause 4.7.2.2;
- j) eCall inactivity procedure is not performed via non-3GPP access;
- k) local area data network (LADN) does not apply for non-3GPP access;
- the Allowed PDU session IE shall not be included in the REGISTRATION REQUEST message or the SERVICE REQUEST message sent over non-3GPP access;
- m) DRX parameters do not apply for non-3GPP access;
- n) Mobile initiated connection only mode (MICO) does not apply for non-3GPP access;

- o) CIoT 5GS optimizations do not apply for non-3GPP access;
- p) unified access control does not apply for non-3GPP access;
- q) UE radio capability signalling optimisation (RACS) does not apply for non-3GPP access;
- r) Closed access group (CAG) does not apply for non-3GPP access; and
- s) the N1 NAS signalling connection release, the paging indication for voice services and reject the paging request do not apply for non-3GPP access. The Paging restriction IE shall not be included in the REGISTRATION REQUEST message, the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message sent over non-3GPP access. The AMF shall not delete any stored paging restriction preferences for the UE and shall not stop restricting paging when receiving REGISTRATION REQUEST message, SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message over non-3GPP access.

4.7.2.2 Establishment cause for non-3GPP access

When establishment of an N1 NAS signalling connection over non-3GPP access is initiated, the UE shall:

- a) determine one or more access identities to be associated with the establishment of the N1 NAS signalling connection as specified in subclause 4.5.2 and table 4.5.2.1;
- b) select the establishment cause for non-3GPP access from the determined one or more access identities and the event which triggered initiation of the N1 NAS signalling connection over non-3GPP access by checking the rules specified in table 4.7.2.2.1; and
- c) provide the selected establishment cause for non-3GPP access to the lower layers.

While an MMTEL voice call is ongoing:

- any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode is mapped to "MO MMTel voice call" type access attempt.

While an MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any:
 - 1) service request procedure; or
 - 2) registration procedure;
 - initiated in 5GMM-IDLE mode is mapped to "MO MMTel video call" type access attempt.

While an SMSoIP is ongoing, no MMTEL video call is ongoing and no MMTEL voice call is ongoing:

- any:
 - 1) service request procedure; or
 - 2) registration procedure;

initiated in 5GMM-IDLE mode is mapped to "MO SMS over IP" type access attempt.

If the access attempt matches more than one rule, the establishment cause for non-3GPP access of the lowest rule number shall be used.

Rule #	Access identities	Type of access	Requirements to be	Establishment cause
		attempt	met	for non-3GPP access
1	1	Any	Any	mps-PriorityAccess
2	2	Any	Any	mcs-PriorityAccess
3	11, 15	Any	Any	highPriorityAccess
4	12,13,14,	Any	Any	highPriorityAccess
5	0	Emergency	UE is attempting	emergency
			access for an	
			emergency session	
			(NOTE 1)	
		UE NAS initiated	Access attempt is for	mo-Signalling
		5GMM specific	MO signalling	
		procedures		
		UE NAS initiated	Access attempt is for	mo-Data
		5GMM connection	MO data	
		management		
		procedures or 5GMM		
		NAS transport		
		procedure		
		MO SMS over NAS or	Access attempt is for	mo-SMS
		MO SMS over IP	MO SMS over NAS or	
			MO SMS over IP	
		MO MMTel voice call	Access attempt is for	mo-VoiceCall
			MO MMTel voice call	
		MO MMTel video call	Access attempt is for	mo-Videocall
			MO MMTel video call	
NOTE 1: This includes 5GMM specific procedures while the service is ongoing and 5GMM connection management				
procedures required to establish a PDU session with request type = "initial emergency request" or "existing				
emergency PDU session", or to re-establish user-plane resources for such a PDU session.				
NOTE 2: See subclause 4.5.2, table 4.5.2.1 for use of the access identities of 0, 1, 2, and 11-15.				

Table 4.7.2.2.1: Mapping table for determination of establishment cause for non-3GPP access

4.7.3 5GS session management aspects

The session management procedures defined over 3GPP access are re-used over non-3GPP access with the following exceptions:

The session management procedures defined over 3GPP access are re-used over non-3GPP access with the following exceptions:

- a) Serving PLMN rate control does not apply for non-3GPP access;
- b) Small data rate control does not apply for non-3GPP access;
- c) Handling of 5GSM cause value #82 "maximum data rate per UE for user-plane integrity protection is too low" does not apply for non-3GPP access;
- d) MBS does not apply for non-3GPP access; and
- e) Support of redundant PDU sessions does not apply for non-3GPP access.

4.7.4 Limited service state over non-3GPP access

There are a number of situations in which the UE is unable to obtain normal service from a PLMN over non-3GPP access and the UE enters the limited service state over non-3GPP access. These include:

- a) no USIM in the ME;
- b) an "illegal UE" or "illegal ME" is received when registration, network-initiated de-registration or service request is performed (any USIM in the ME is then considered "invalid");
- c) a "5GS services not allowed" is received when a registration, network-initiated de-registration or service request is performed;

- a "PLMN not allowed" is received when registration, network-initiated de-registration or service request is performed;
- e) a "Tracking area not allowed" is received when a registration, network-initiated de-registration or service request is performed;
- f) a "Roaming not allowed in this tracking area" is received when a registration, network-initiated de-registration or service request is performed;
- g) void; or
- h) a "Serving network not authorized" is received when a registration or service request is performed.

In limited service state with a valid USIM in the UE, the network selection is performed as defined in 3GPP TS 24.502 [18].

With the exception of performing initial registration for emergency services, no registration requests are made until a valid USIM is present. For registration for emergency services, the PLMN of the current N3IWF or TNGF is considered as the selected PLMN for the duration the UE is registered for emergency services.

4.7.5 NAS signalling using trusted WLAN access network

A trusted WLAN interworking function (TWIF) provides functionalities for a non-5G capable over WLAN (N5CW) device to access 5GCN, including:

- a) NAS signalling over N1 NAS signalling connection with AMF; and
- b) PDU session establishment, modification and release on behalf of the N5CW device, over N2 connection with the AMF.

The TWIF registers on behalf of the N5CW device to an AMF according to subclause 5.5.1.3 by populating the parameters for the registration by using implementation specific default values which are the same for N5CW devices.

The TWIF may request to establish a PDU session as specified in subclause 6.4.1.2 on behalf of the N5CW device upon receipt of an IP configuration request from the N5CW device by populating either all the required parameters or part of the required parameters for the PDU session establishment by using implementation specific default values from the TWIF's configuration. Only one PDU session is supported when N5CW device accessing 5GC via the TWIF.

NOTE 1: If part of the required parameters for the PDU session establishment is provided by the TWIF, the remaining of the required parameters are determined by the AMF or the SMF based on the N5CW device's subscription information.

Upon loss of the IP address of the N5CW device, the TWIF acting on behalf of the N5CW device shall initiate the UE-requested PDU session release procedure as defined in subclause 6.4.3.

NOTE 2: The established PDU session on behalf of the N5CW device can be modified by the TWIF or the network.

4.8 Interworking with E-UTRAN connected to EPC

4.8.1 General

In order to interwork with E-UTRAN connected to EPC, the UE supporting both S1 mode and N1 mode can operate in single-registration mode or dual-registration mode (see 3GPP TS 23.501 [8]). Support of single-registration mode is mandatory for UEs supporting both S1 mode and N1 mode.

During the EPS attach procedure (see 3GPP TS 24.301 [15]) or initial registration procedure (see subclause 5.5.1.2), the mode for interworking is selected if the UE supports both S1 mode and N1 mode, and the network supports interworking. The mode for interworking may also be selected during the EPS tracking area updating procedure (see 3GPP TS 24.301 [15]) or registration procedure for mobility and periodic registration update (see subclause 5.5.1.3).

For interworking between E-UTRAN connected to EPC and TNGF or N3IWF connected to 5GCN, the UE shall operate as specified in either subclause 4.8.2.3 or subclause 4.8.3. Which subclause the UE follows is chosen by the UE irrespective of the interworking without N26 interface indicator.

4.8.2 Single-registration mode

4.8.2.1 General

If the UE receives the indication that "interworking without N26 interface not supported" (see 3GPP TS 24.301 [15]), the UE operates as described in subclause 4.8.2.2.

If the UE receives the indication that "interworking without N26 interface supported" and

- a) the UE does not support dual-registration mode; or
- b) the UE supporting dual-registration mode determines to operate in single-registration mode,

the UE operates as described in subclause 4.8.2.3.

4.8.2.2 Single-registration mode with N26 interface

See subclause 5.1.4.2 for coordination between 5GMM and EMM and subclause 6.1.4.1 for coordination between 5GSM and ESM.

4.8.2.3 Single-registration mode without N26 interface

4.8.2.3.1 Interworking between NG-RAN and E-UTRAN

At inter-system change from N1 mode to S1 mode in EMM-IDLE mode when:

- a) the UE supports non-IP PDN type and at least one PDU session of Unstructured PDU session type is active;
- b) the UE supports IPv4 PDN type and at least one PDU session of IPv4 PDU session type is active;
- c) the UE supports IPv6 PDN type and at least one PDU session of IPv6 PDU session type is active;
- d) the UE supports IPv4v6 PDN type and at least one PDU session of IPv4v6 PDU session type is active; or
- e) at least one PDU session of Ethernet PDU session type is active and:
 - 1) the UE supports non-IP PDN type; or
 - 2) the UE and the network support Ethernet PDN type in S1 mode;

the UE shall proceed as follows:

- a) if the UE supports sending an ATTACH REQUEST message containing a PDN CONNECTIVITY REQUEST message with request type set to "handover" or "handover of emergency bearer services" to transfer a PDU session from N1 mode to S1 mode and the UE has received an "interworking without N26 interface supported" indication from the network, the UE shall:
 - 1) enter substates EMM-DEREGISTERED.NORMAL-SERVICE and 5GMM- DEREGISTERED.NO-CELL-AVAILABLE for 3GPP access;
 - 2) map the PDU session(s) which the UE intends to transfer to EPS to the default EPS bearer context of the corresponding PDN connection(s) as specified in subclause 6.1.4.2; and
 - 3) initiate an EPS attach procedure and include in the ATTACH REQUEST message a PDN CONNECTIVITY REQUEST message with:
 - the request type set to "handover of emergency bearer services" to activate a default EPS bearer context for an active emergency PDU session, if the session to be transferred is an emergency PDU session; or

- the request type set to "handover" message to activate a default EPS bearer context for an active nonemergency PDU session, if the session to be transferred is a non-emergency PDU session. If the selected PDU session is an MA PDU session established over 3GPP access, the UE shall include the ATSSS request parameter in the Protocol configuration options IE or the Extended protocol configuration options IE of the ESM INFORMATION RESPONSE message.

After successful completion of the EPS attach procedure, the UE shall reset the registration attempt counter for 3GPP access and the attach attempt counter (see 3GPP TS 24.301 [15]) and attempt to activate each of the other default EPS bearer contexts, if any, by initiating a stand-alone PDN connectivity procedure with request type set to "handover" for non-emergency PDU session or "handover of emergency bearer services" for emergency PDU session in the PDN CONNECTIVITY REQUEST message; and

b) otherwise, enter substates EMM-REGISTERED.NORMAL-SERVICE and 5GMM-REGISTERED.NO-CELL-AVAILABLE for 3GPP access and initiate a tracking area update procedure (see 3GPP TS 24.301 [15]).

At inter-system change from N1 mode to S1 mode in EMM-IDLE mode when:

- a) the UE does not support non-IP PDN type or no PDU session of Unstructured PDU session type is active;
- b) the UE does not support IPv4 PDN type or no PDU session of IPv4 PDU session type is active;
- c) the UE does not support IPv6 PDN type or no PDU session of IPv6 PDU session type is active;
- d) the UE does not support IPv4v6 PDN type or no PDU session of IPv4v6 PDU session type is active; and
- e) no PDU session of Ethernet PDU session type is active or:
 - 1) the UE does not support non-IP PDN type; and
 - 2) the UE, the network or both do not support Ethernet PDN type in S1 mode;

the UE shall enter substates EMM-DEREGISTERED.NORMAL-SERVICE and 5GMM-DEREGISTERED.NO-CELL-AVAILABLE for 3GPP access, and initiate an attach procedure.

At inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, the UE shall:

- a) enter substate 5GMM-REGISTERED.NORMAL-SERVICE for 3GPP access and substate EMM-REGISTERED.NO-CELL-AVAILABLE;
- b) map the default EPS bearer context(s) of the PDN connection(s) which the UE intends to transfer to 5GS, if any, to the corresponding PDU session(s) as specified in subclause 6.1.4.2; and
- c) initiate the registration procedure for mobility and periodic registration update over 3GPP access indicating "mobility registration updating" in the 5GS registration type IE of the REGISTRATION REQUEST message (see subclause 5.5.1.3).

After having successfully registered in N1 mode over 3GPP access, the UE shall reset the registration attempt counter for 3GPP access, and the attach attempt counter or tracking area updating attempt counter (see 3GPP TS 24.301 [15]) and:

- a) if the UE supports the PDU session establishment procedure with request type set to "existing PDU session" or "existing emergency PDU session" to transfer a PDN connection from S1 mode to N1 mode and the UE has received an "interworking without N26 interface supported" indication from the network, attempt to transfer the PDN connection(s) which the UE intends to transfer to 5GS, if any, from S1 mode to N1 mode by:
 - if the PDN connection which the UE intends to transfer is a PDN connection for emergency bearer services, initiating the PDU session establishment procedure with request type set to "existing emergency PDU session" to transfer the PDN connection for emergency bearer services; and
 - if the PDN connection which the UE intends to transfer is a non-emergency PDN connection, initiating the PDU session establishment procedure with request type set to:
 - 1) "MA PDU request", if the PDN connection to be transferred is a user-plane resource of an MA PDU session; or
 - 2) "existing PDU session" to transfer the non-emergency PDN connection; and

3GPP TS 24.501 version 17.16.0 Release 17

107

b) otherwise, establish PDU session(s) corresponding to the PDN connection(s) which the UE intends to transfer to 5GS, if any, by initiating the PDU session establishment procedure with request type set to "initial request".

See subclause 5.1.4.3 for coordination between 5GMM and EMM and subclause 6.1.4.2 for coordination between 5GSM and ESM.

4.8.2.3.2 Interworking between TNGF or N3IWF connected to 5GCN and E-UTRAN

If:

- a) the UE has registered in neither N1 mode over 3GPP access nor S1 mode yet; and
- b) the UE has at least one active PDU session associated with non-3GPP access which the UE intends to transfer to EPS,

the UE shall initiate an EPS attach procedure and include a PDN CONNECTIVITY REQUEST message with a request type in the ATTACH REQUEST message to activate a default EPS bearer context for one of the active PDU sessions which the UE intends to transfer to EPS (see 3GPP TS 24.301 [15]). The request type is set as follows:

- if the PDU session which the UE intends to transfer is a non-emergency PDU session, the request type is set to "handover"; and
- if the PDU session which the UE intends to transfer is an emergency PDU session, the request type is set to "handover of emergency bearer services" and the default bearer to be activated is the default EPS bearer context for the emergency PDU session.
- NOTE 1: It is necessary for the UE to support sending an ATTACH REQUEST message containing a PDN CONNECTIVITY REQUEST message with request type set to "handover" or "handover of emergency bearer services" to transfer a PDU session from N1 mode to S1 mode for interworking between TNGF or N3IWF connected to 5GCN and E-UTRAN.

NOTE 2: The order of PDU sessions to be tranferred to EPS is up to UE implementation.

After successful completion of the EPS attach procedure where the activated default EPS bearer context is not for emergency service, the UE shall initiate a UE requested PDN connectivity procedure with request type set to "handover" for non-emergency PDU session or "handover of emergency bearer services" for emergency PDU session in the PDN CONNECTIVITY REQUEST message to transfer each of the other PDU sessions which the UE intends to transfer to EPS, if any.

If:

- a) the UE has not registered in N1 mode over non-3GPP access yet; and
- b) the UE has at least one active PDN connection which the UE intends to transfer to TNGF or N3IWF connected to 5GCN,

the UE shall initiate an initial registration procedure over non-3GPP access (see subclause 5.5.1.2).

After successful completion of the 5GS initial registration in N1 mode over non-3GPP access, the UE shall initiate a UE-requested PDU session establishment procedure with a request type to transfer each of the PDN connections which the UE intends to transfer to TNGF or N3IWF connected to 5GCN, if any. The request type is set as follows:

- if the PDN connection which the UE intends to transfer is a PDN connection for emergency bearer services, the request type is set to "existing emergency PDU session" to transfer the PDN connection for emergency bearer services; and
- if the PDN connection which the UE intends to transfer is a non-emergency PDN connection, the request type is set to "existing PDU session" to transfer the non-emergency PDN connection.
- NOTE 3: If the UE has no active PDU session associated with non-3GPP access which the UE in N1 mode intends to transfer to EPS or no active PDN connection associated with 3GPP access which the UE in S1 mode intends to transfer to TNGF or N3IWF connected to 5GCN, the interworking between TNGF or N3IWF connected to 5GCN and E-UTRAN is not supported.

See subclause 6.1.4.2 for coordination between 5GSM and ESM.
4.8.3 Dual-registration mode

If both 5GMM and EMM are enabled, a UE, operating in the dual-registration mode shall maintain independent contexts for 5GMM and EMM and this includes independent lists of equivalent PLMNs. Coordination between 5GMM and EMM is not needed, except as specified in the present subclause, subclause 5.1.5 and 5.3.13A.

For dual-registration mode the following applies:

- a) a UE operating in the dual-registration mode may register to N1 mode only, S1 mode only, or to both N1 mode and S1 mode;
- b) when the UE decides to operate in dual-registration mode (see subclause 5.5.1.2.4), NAS informs the lower layers about this;
- c) if a UE is registered in N1 mode only, then for registration in S1 mode the UE shall use:
 - 1) the same PLMN to which it is registered in N1 mode; or
 - 2) an equivalent PLMN; and
- d) if a UE is registered in S1 mode only, then for registration in N1 mode the UE shall use:
 - 1) the same PLMN to which it is registered in S1 mode; or
 - 2) an equivalent PLMN.
- NOTE 1: It is up to UE implementation how to handle the case when the UE is registered in both N1 mode and S1 mode and the PLMNs to which the UE is registered, are not equivalent, e.g. search for a PLMN which is the same or equivalent to any of the registered ones.

When no PDU session is active and the UE has not registered to S1 mode yet, the UE may initiate the EPS attach procedure with PDN connection establishment if EMM-REGISTERED without PDN connection is not supported by the MME. If EMM-REGISTERED without PDN connection is supported by the MME, the UE may initiate either the EPS attach procedure without PDN connection establishment or the attach procedure with PDN connection establishment.

When at least one PDU session is active and the UE has not registered to S1 mode yet, the UE may initiate the EPS attach procedure. If necessary, the UE may transfer an active PDU session from N1 mode to S1 mode by initiating the EPS attach procedure with request type set to "handover" in the PDN CONNECTIVITY REQUEST message. After successfully attached in S1 mode, if necessary, the UE may transfer other active PDU sessions from N1 mode to S1 mode to S1 mode by initiating the PDN connectivity procedure with request type set to "handover" in the PDN CONNECTIVITY REQUEST message. Request type set to "handover" in the PDN connectivity procedure with request type set to "handover" in the PDN CONNECTIVITY REQUEST message.

NOTE 2: It is up to UE implementation to determine which active PDU session is transferred from N1 mode to S1 mode.

When the UE has not registered to N1 mode, the UE may initiate the initial registration procedure. After successfully registered in N1 mode, if necessary, the UE may transfer one or more active PDN connections from S1 mode to N1 mode by initiating the PDU session establishment procedure with request type set to "existing PDU session".

NOTE 3: It is up to UE implementation to determine which active PDN connection is transferred from S1 mode to N1 mode.

If the MME supports EMM-REGISTERED without PDN connection, the UE that transferred all PDN connections to the 5GS, may stay in state EMM-REGISTERED. Otherwise, the UE shall enter state EMM-DEREGISTERED upon transferring all PDN connection to the 5GS.

NOTE 4: When the UE has registered in both N1 mode and S1 mode, it is up to UE implementation to maintain the registration update to date in both N1 mode and S1 mode.

See subclause 6.1.4 for coordination between 5GSM and ESM.

See subclause 4.8.2.3.2 for interworking between TNGF or N3IWF connected to 5GCN and E-UTRAN.

4.8.4 Core Network selection for UEs not using CIoT 5GS optimizations

If the UE is capable of both N1 mode and S1 mode, when the UE needs to use one or more functionalities not supported in 5GS but supported in EPS and the UE is in 5GMM-IDLE mode, the UE may disable the N1 mode capability for 3GPP access (see subclause 4.9.2).

If the UE is capable of both N1 mode and S1 mode and lower layers provide an indication that the current E-UTRA cell is connected to both EPC and 5GCN without also providing an indication that a target core network type was received from the NG-RAN, the UE shall select a core network type (EPC or 5GCN) based on the PLMN selection procedures as specified in 3GPP TS 23.122 [5] and provide the selected core network type information to the lower layer during the initial registration procedure.

If the UE is capable of both N1 mode and S1 mode and the lower layers have provided an indication that the current E-UTRA cell is connected to both EPC and 5GCN and an indication of whether the network supports IMS emergency services via either EPC or 5GCN or both (see 3GPP TS 36.331 [25A]), the UE selects a core network type (EPC or 5GCN) as specified in 3GPP TS 23.167 [6] annex H.2 for initiating emergency calls when in the state 5GMM-DEREGISTERED.LIMITED-SERVICE or EMM-DEREGISTERED.LIMITED-SERVICE.

NOTE 1: If the PLMN selection information provisioned in the USIM does not contain any prioritization between E-UTRAN and NG-RAN for a PLMN, which core network type to select for that PLMN is up to UE implementation.

If the UE is capable of both N1 mode and S1 mode and lower layers provide an indication that the current E-UTRA cell is connected to both EPC and 5GCN with:

- 1) an indication that target core network type EPC was received from the NG-RAN, the UE shall select the EPC and proceed with the appropriate EMM procedure as specified in 3GPP TS 24.301 [15]; or
- 2) an indication that target core network type 5GCN was received from the NG-RAN, the UE shall select the 5GCN and proceed with the appropriate 5GMM procedure.
- NOTE 2: The NG-RAN can provide a target core network type to the UE during RRC connection release with redirection (see 3GPP TS 36.331 [25A] and 3GPP TS 38.331 [30]).

4.8.4A Core Network selection and redirection for UEs using CIoT optimizations

4.8.4A.1 Core network selection

A UE that supports CIoT optimizations performs core network selection (i.e. it selects EPC or 5GCN) if the lower layers have provided an indication that the current E-UTRA cell is connected to both EPC and 5GCN as specified in 3GPP TS 23.501 [8].

When selecting a PLMN as described in 3GPP TS 23.122 [5], the UE shall select a core network type (EPC or 5GCN) based on:

- a) indication from the lower layers about the CIoT EPS optimizations supported in EPC;
- b) indication from the lower layers about the CIoT 5GS optimizations supported in 5GCN;
- c) the CIoT EPS optimizations supported by the UE;
- d) the CIoT 5GS optimizations supported by the UE;
- e) the UE's preferred CIoT network behaviour for EPC; and
- f) the UE's preferred CIoT network behaviour for 5GCN.

The UE shall provide the selected core network type information to the lower layer during the initial registration procedure.

4.8.4A.2 Redirection of the UE by the core network

The network that supports CIoT optimizations can redirect a UE between EPC and 5GCN as specified in subclause 5.31.3 of 3GPP TS 23.501 [8]. The network can take into account the UE's N1 mode capability or S1 mode capability, the CIoT network behaviour supported and preferred by the UE or the CIoT network behaviour supported by the network to determine the redirection.

NOTE: It is assumed that the network would avoid redirecting the UE back and forth between EPC and 5GCN.

The network redirects the UE to EPC by rejecting the registration request or service request with the 5GMM cause #31 "Redirection to EPC required" as specified in subclause 5.5.1.2.5, 5.5.1.3.5 and 5.6.1.5. Upon receipt of reject message, the UE disables the N1 mode capability for 3GPP access as specified in subclause 4.9.2 and enables the E-UTRA capability if it was disabled in order to move to EPC.

When there is no ongoing registration procedure or service request procedure for a UE in 5GMM-CONNECTED mode, if the AMF determines to redirect the UE to EPC, the AMF shall initiate the generic UE configuration update procedure to indicate registration requested and release of the N1 NAS signalling connection not requested as described in subclause 5.4.4. The network then redirects the UE to EPC by rejecting the registration request as specified in subclause 5.5.1.3.5.

The network that supports CIoT optimizations can also redirect a UE from EPC to 5GCN as specified in subclause 5.3.19.2 of 3GPP TS 24.301 [15].

4.9 Disabling and re-enabling of UE's N1 mode capability

4.9.1 General

The UE shall re-enable the N1 mode capability when the UE powers off and powers on again, the USIM is removed or an entry of the "list of subscriber data" with the SNPN identity of the SNPN is updated.

4.9.2 Disabling and re-enabling of UE's N1 mode capability for 3GPP access

The UE shall only disable the N1 mode capability for 3GPP access when in 5GMM-IDLE mode.

When the UE is disabling the N1 mode capability for 3GPP access for a PLMN not due to redirection to EPC, it should proceed as follows:

- a) select an E-UTRA cell connected to EPC of the registered PLMN or a PLMN from the list of equivalent PLMNs, if the UE supports S1 mode and the UE has not disabled its E-UTRA capability as specified in 3GPP TS 24.301 [15];
- b) if an E-UTRA cell connected to EPC of the registered PLMN or a PLMN from the list of equivalent PLMNs cannot be found, the UE does not support S1 mode or the UE has disabled its E-UTRA capability as specified in 3GPP TS 24.301 [15], the UE may select another RAT of the registered PLMN or a PLMN from the list of equivalent PLMNs that the UE supports;
- c) if another RAT of the registered PLMN or a PLMN from the list of equivalent PLMNs cannot be found, then enter the state 5GMM-REGISTERED.PLMN-SEARCH or 5GMM-DEREGISTERED.PLMN-SEARCH, or the UE does not have a registered PLMN, then enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform PLMN selection as specified in 3GPP TS 23.122 [5]. If disabling of the N1 mode capability for 3GPP access was not due to a UE-initiated de-registration procedure for 5GS services over 3GPP access not due to switch-off, the UE may re-enable the N1 capability for this PLMN selection. As an implementation option, if the UE does not have a registered PLMN, instead of performing PLMN selection, the UE may select another RAT of the selected PLMN if the UE has chosen a PLMN and the RAT is supported by the UE; or
- d) if no other allowed PLMN and RAT combinations are available, then the UE may re-enable the N1 mode capability for 3GPP access and indicate to lower layers to remain camped in NG-RAN of the registered PLMN, and may periodically scan for another PLMN and RAT combination which can provide EPS services or non-EPS services (if the UE supports EPS services or non-EPS services). How this periodic scanning is done, is UE implementation dependent.

When the UE is disabling the N1 mode capability for 3GPP access for an SNPN, it should proceed as follows:

- a) enter the state 5GMM-REGISTERED.PLMN-SEARCH or 5GMM-DEREGISTERED.PLMN-SEARCH and perform SNPN selection as specified in 3GPP TS 23.122 [5]. If disabling of the N1 mode capability for 3GPP access was not due to a UE-initiated de-registration procedure for 5GS services over 3GPP access not due to switch-off, the UE may re-enable the N1 capability for this SNPN selection; or
- b) if no other SNPN is available, then the UE may re-enable the N1 mode capability for 3GPP access and indicate to lower layers to remain camped in NG-RAN of the registered SNPN.

When the UE is disabling the N1 mode capability upon receiving cause value #31 "Redirection to EPC required" as specified in subclauses 5.5.1.2.5, 5.5.1.3.5 and 5.6.1.5, it should proceed as follows:

- a) If the UE is in NB-N1 mode:
 - if lower layers do not provide an indication that the current E-UTRA cell is connected to EPC or lower layers do not provide an indication that the current E-UTRA cell supports CIoT EPS optimizations that are supported by the UE, search for a suitable NB-IoT cell connected to EPC according to 3GPP TS 36.304 [25C];
 - if lower layers provide an indication that the current E-UTRA cell is connected to EPC and the current E-UTRA cell supports CIoT EPS optimizations that are supported by the UE, perform a core network selection to select EPC as specified in subclause 4.8.4A.1; or
 - 3) if lower layers cannot find a suitable NB-IoT cell connected to EPC or there is no suitable NB-IoT cell connected to EPC which supports CIoT EPS optimizations that are supported by the UE, the UE, as an implementation option, may indicate to lower layers to remain camped in E-UTRA cell connected to 5GCN, may then start an implementation-specific timer and enter the state 5GMM-REGISTERED.LIMITED-SERVICE. The UE may may re-enable the N1 mode capability for 3GPP access at expiry of the implementation-specific timer, if the timer had been started, and may then proceed with the appropriate 5GMM procedure.
- b) If the UE is in WB-N1 mode:
 - if lower layers do not provide an indication that the current E-UTRA cell is connected to EPC or lower layers do not provide an indication that the current E-UTRA cell supports CIoT EPS optimizations that are supported by the UE, search for a suitable E-UTRA cell connected to EPC according to 3GPP TS 36.304 [25C];
 - if lower layers provide an indication that the current E-UTRA cell is connected to EPC and the current E-UTRA cell supports CIoT EPS optimizations that are supported by the UE, then perform a core network selection to select EPC as specified in subclause 4.8.4A.1; or
 - 3) if lower layers cannot find a suitable E-UTRA cell connected to EPC or there is no suitable E-UTRA cell connected to EPC which supports CIoT EPS optimizations that are supported by the UE, the UE, as an implementation option, may indicate to lower layers to remain camped in E-UTRA cell connected to 5GCN, may then start an implementation-specific timer and enter the state 5GMM-REGISTERED.LIMITED-SERVICE. The UE may re-enable the N1 mode capability for 3GPP access at expiry of the implementation-specific timer, if the timer had been started, and may then proceed with the appropriate 5GMM procedure.

When the UE supporting both N1 mode and S1 mode needs to stay in E-UTRA connected to EPC (e.g. due to the domain selection for UE originating sessions as specified in subclause 4.3.2), in order to prevent unintentional handover or cell reselection from E-UTRA connected to EPC to NG-RAN connected to 5GCN, the UE operating in single-registration mode shall disable the N1 mode capability for 3GPP access and:

- a) shall set the N1mode bit to "N1 mode for 3GPP access not supported" in the UE network capability IE (see 3GPP TS 24.301 [15]) of the ATTACH REQUEST message and the TRACKING AREA UPDATE REQUEST message in EPC; and
- b) the UE NAS layer shall indicate the access stratum layer(s) of disabling of the N1 mode capability for 3GPP access.

If the UE is required to disable the N1 mode capability for 3GPP access and select E-UTRA or another RAT, and the UE is in the 5GMM-CONNECTED mode,

- if the UE has a persistent PDU session, then the UE waits until the radio bearer associated with the persistent PDU session has been released;
- otherwise the UE shall locally release the established NAS signalling connection;

and enter the 5GMM-IDLE mode before selecting E-UTRA or another RAT.

If the UE is disabling its N1 mode capability for 3GPP access before selecting E-UTRA or another RAT, the UE shall not perform the UE-initiated de-registration procedure of subclause 5.5.2.2.

The UE shall re-enable the N1 mode capability for 3GPP access when the UE performs PLMN or SNPN selection over 3GPP access, unless

- disabling of the N1 mode capability for 3GPP access was due to a UE-initiated de-registration procedure for 5GS services over 3GPP access not due to switch-off; or
- the UE has already re-enabled the N1 mode capability for 3GPP access when performing items c) or d) above.

If the disabling of N1 mode capability for 3GPP access was due to IMS voice is not available over 3GPP access and the UE's usage setting is "voice centric", the UE shall re-enable the N1 mode capability for 3GPP access when the UE's usage setting is changed from "voice centric" to "data centric", as specified in subclauses 4.3.3.

The UE should memorize the identity of the PLMN or SNPN where N1 mode capability for 3GPP access was disabled and should use that stored information in subsequent PLMN or SNPN selections as specified in 3GPP TS 23.122 [5].

If the disabling of N1 mode capability for 3GPP access was due to successful completion of an emergency services fallback, the criteria to enable the N1 mode capability again are UE implementation specific.

The UE shall disable the N1 mode capability for 3GPP access if requested by the upper layers (e.g. see subclause U.2.2.6.4 in 3GPP TS 24.229 [14]). If the UE disabled the N1 mode capability for 3GPP access based on the request from the upper layers (e.g. see subclause U.2.2.6.4 in 3GPP TS 24.229 [14]), the criteria to re-enable the N1 mode capability for 3GPP access after the completion of an emergency service are UE implementation specific.

If the N1 mode capability for 3GPP access was disabled due to the UE initiated de-registration procedure for 3GPP access or for 3GPP access and non-3GPP access and the UE is operating in single-registration mode (see subclause 5.5.2.2.3), upon request of the upper layers to re-register for 5GS services over 3GPP access the UE shall enable the N1 mode capability for 3GPP access again.

As an implementation option, the UE may start a timer for enabling the N1 mode capability for 3GPP access when the UE's registration attempt counter reaches 5 and the UE disables the N1 mode capability for 3GPP access for cases described in subclauses 5.5.1.2.7 and 5.5.1.3.7. The UE should memorize the identity of the PLMNs where N1 mode capability for 3GPP access was disabled. On expiry of this timer:

- if the UE is in Iu mode or A/Gb mode and is in idle mode as specified in 3GPP TS 24.008 [13] on expiry of the timer, the UE should enable the N1 mode capability for 3GPP access;
- if the UE is in Iu mode and a PS signalling connection exists, but no RR connection exists, the UE may abort the PS signalling connection before enabling the N1 mode capability for 3GPP access; and
- if the UE is in S1 mode and is in EMM-IDLE mode as specified in 3GPP TS 24.301 [15], on expiry of the timer, the UE should enable the N1 mode capability for 3GPP access.

If the UE is in Iu mode or A/Gb mode and an RR connection exists, the UE should delay enabling the N1 mode capability for 3GPP access until the RR connection is released. If the UE is in S1 mode and is in EMM-CONNECTED mode as specified in 3GPP TS 24.301 [15], the UE should delay enabling the N1 mode capability for 3GPP access until the NAS signalling connection in S1 mode is released.

The UE may disable the N1 mode capability for currently camped PLMN or SNPN over 3GPP access (see 3GPP TS 23.122 [5]) if no network slice is available for the camped PLMN or SNPN.

If the UE attempts to establish an emergency PDU session in a PLMN where N1 mode capability was disabled due to the UE's registration attempt counter have reached 5, the UE may enable N1 mode capability for that PLMN memorized by the UE.

NOTE: If N1 mode capability is disabled due to the UE's registration attempt counter reaches 5, the value of the timer for re-enabling N1 mode capability is recommended to be the same as the value of T3502 which follows the handling specified in subclause 5.3.8.

4.9.3 Disabling and re-enabling of UE's N1 mode capability for non-3GPP access

When the UE disables the N1 mode capability for non-3GPP access, the UE NAS layer shall not initiate any 5GS NAS procedures towards the network over non-3GPP access.

When the UE supporting both N1 mode and S1 mode needs to stay in non-3GPP access connected to EPC (e.g. due to the domain selection for UE originating sessions as specified in subclause 4.3.2), in order to prevent unintentional selection of a non-3GPP access network connected to 5GCN, the UE operating in single-registration mode shall not transfer any PDN connection to a non-3GPP access network connected to the 5GCN.

If the disabling of N1 mode capability for non-3GPP access was due to IMS voice is not available over non-3GPP access in 5GS and the UE's usage setting is "voice centric", the UE shall re-enable the N1 mode capability for non-3GPP access when the UE's usage setting is changed from "voice centric" to "data centric" as specified in subclauses 4.3.3.

The UE shall re-enable the N1 mode capability for non-3GPP access when a new PLMN or SNPN is selected over non-3GPP access.

NOTE: In SNPN, the term "UE's N1 mode capability for non-3GPP access" in this subclause refers to the UE's N1 mode capability to access SNPN services via a PLMN.

The UE may disable the N1 mode capability for the currently camped PLMN or SNPN over non-3GPP access if no network slice is available for the camped PLMN or SNPN.

As an implementation option, the UE may start a timer for re-enabling the N1 mode capability for non-3GPP access, after the N1 mode capability for non-3GPP access was disabled. On the expiry of this timer, the UE should re-enable the N1 mode capability for non-3GPP access.

4.10 Interworking with ePDG connected to EPC

In order to interwork with ePDG connected to EPC, the UE shall operate as specified in either subclause 4.8.2.3 or subclause 4.8.3. Which subclause the UE follows is chosen by the UE irrespective of the interworking without N26 interface indicator.

The UE shall not attempt to transfer PDU sessions with PDU session type "Ethernet" or "Unstructured" to an ePDG connected to EPC.

NOTE: PDU sessions with PDU session type "Ethernet" or "Unstructured" cannot be transferred to an ePDG connected to EPC because PDN connections with PDN type "non-IP" or PDN type "Ethernet" are not supported over ePDG connected to EPC.

4.11 UE configuration parameter updates

The 5GS in a PLMN supports updating UE parameters via NAS signalling. The feature enables the HPLMN to securely and dynamically re-configure the UE configuration parameters stored on the USIM and the ME.

- In this release of the specification, updates of the following USIM configuration parameters are supported:
 - routing indicator.
- In this release of specification, updates of the following ME configuration parameters are supported:
 - default configured NSSAI.
 - disaster roaming information.

The 5GS in an SNPN supports updating UE parameters via NAS signalling. The feature enables the SNPN to securely and dynamically re-configure the UE configuration parameter stored on the USIM if the UE used the USIM for registration to the SNPN.

- In this release of the specification, updates of the following USIM configuration parameters are supported:
 - routing indicator.
- In this release of specification, updates of the following ME configuration parameters are supported:
 - routing indicator.
 - default configured NSSAI.

The update of UE configuration parameters is initiated by the network using the network-initiated downlink NAS transport procedure as described in subclause 5.4.5.3. The ME acknowledgement of successful reception of the updated UE configuration parameter information is sent back to the network using the UE-initiated uplink NAS transport procedure as described in subclause 5.4.5.2.

4.12 Access traffic steering, switching and splitting (ATSSS)

The ATSSS feature is an optional feature that may be supported by the UE and the 5GCN.

The ATSSS feature enables a multi-access PDU connectivity service, which can exchange PDUs between the UE and a data network by simultaneously using one 3GPP access network and one non-3GPP access network. The multi-access PDU connectivity service is realized by establishing a multi-access PDU session, i.e. a PDU session that can have user-plane resources on two access networks.

NOTE: MA PDU session is not applicable for CIoT 5GS optimization in this release of specification.

The UE can request an MA PDU session when the UE is registered via both 3GPP and non-3GPP accesses, or when the UE is registered via one access only. The MA PDU session management is performed based on the PDU session management procedures.

The detailed description of the procedures for ATSSS between the UE and the network across one 3GPP access network and one non-3GPP access network are specified in 3GPP TS 24.193 [13B].

4.13 Support of NAS signalling using wireline access network

A 5G-RG, a W-AGF acting on behalf of an FN-RG or a W-AGF acting on behalf of an N5GC device can use wireline access network to access the 5GCN by using NAS signalling procedures as described in 3GPP TS 23.501 [8], 3GPP TS 23.502 [9] and 3GPP TS 23.316 [6D].

Wireline access is a type of non-3GPP access.

A 5G-RG simultaneously connected to the same 5GCN of a PLMN over a 3GPP access and a wireline access is connected to a single AMF.

5G-RG maintains the N1 NAS signalling connection with the AMF over the wireline access network after all the PDU sessions for the 5G-RG over that access have been released or handed over to 3GPP access.

The 5G-RG connected to 5GCN via NG-RAN is specified in 3GPP TS 23.316 [6D].

When accessing the 5GCN over 3GPP access, in addition to requirements specified for the 5G-RG in the present document, the 5G-RG shall also perform requirements specified in the present document for a UE accessing 5GCN over 3GPP access. When accessing the 5GCN over wireline access, in addition to requirements specified for the 5G-RG in the present document, the 5G-RG shall also perform requirements specified in the present document for a UE accessing 5GCN over non-3GPP access. If a requirement specified for the 5G-RG in the present document contradicts a requirement specified for the UE in the present document, the 5G-RG shall perform the requirement specified in the present document for the 5G-RG.

For the scenario of FN-RG, which does not support N1 mode, the W-AGF acting on behalf of the FN-RG exchanges NAS signalling messages with an AMF.

For the scenario of N5GC device, which does not support N1 mode, the W-AGF acting on behalf of the N5GC device exchanges NAS signalling messages with an AMF.

In addition to requirements specified for the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) in the present document, the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) shall also perform requirements specified in the present document for a UE accessing 5GCN over non-3GPP access. If a requirement specified for the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) in the present document contradicts a requirement specified for the UE in the present document, the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) shall perform requirement specified in the present document for the W-AGF acting on behalf of the SGC device) shall perform requirement specified in the present document for the W-AGF acting on behalf of the N5GC device) shall perform requirement specified in the present document for the W-AGF acting on behalf of the N5GC device) shall perform requirement specified in the present document for the W-AGF acting on behalf of the N5GC device).

The PDU session authentication and authorization procedure is not supported in a PDU session established by the W-AGF acting on behalf of the FN-RG or on behalf of the N5GC device.

The W-AGF acting on behalf of the N5GC device requests the establishment of a PDU Session on behalf of the N5GC device upon registration. Only one PDU session per N5GC device is supported.

A 5G-RG or an FN-RG provide a non-3GPP access network to UEs. A UE connected to a non-3GPP access network provided by the 5G-RG or the FN-RG can access to the 5GCN via the N3IWF or via the TNGF as described in 3GPP TS 23.316 [6D].

The 5G-RG or the W-AGF acting on behalf of the FN-RG shall indicate "ANDSP not supported by the UE" in the UE policy classmark IE during the UE-initiated UE state indication procedure as specified in subclause D.2.2.

4.14 Non-public network

4.14.1 General

Two types of NPN can be deployed using 5GS: SNPN (see subclause 4.14.2) and PNI-NPN (see subclause 4.14.3).

4.14.2 Stand-alone non-public network

If the UE is not SNPN enabled, the UE is always considered to be not operating in SNPN access operation mode. If the UE is SNPN enabled, the UE can operate in SNPN access operation mode. Details of activation and deactivation of SNPN access operation mode at the SNPN enabled UE are up to UE implementation.

The functions and procedures of NAS described in the present document are applicable to an SNPN and an SNPN enabled UE unless indicated otherwise. The key differences brought by the SNPN to the NAS layer are as follows:

- a) instead of the PLMN selection process, the SNPN selection process is performed by a UE operating in SNPN access operation mode (see 3GPP TS 23.122 [5] for further details on the SNPN selection);
- b) a "permanently forbidden SNPNs" list and a "temporarily forbidden SNPNs" list are managed per access type independently (i.e. 3GPP access or non-3GPP access) and, if the UE supports access to an SNPN using credentials from a credentials holder, per entry of the "list of subscriber data" or the PLMN subscription, by a UE operating in SNPN access operation mode instead of forbidden PLMN lists. If the UE supports onboarding services in SNPN, an additional "permanently forbidden SNPNs" list for onboarding services and an additional "temporarily forbidden SNPNs" list for onboarding services are managed;
- c) inter-system change to and from S1 mode is not supported;
- d) void;
- e) CAG is not supported in SNPN access operation mode;
- f) with respect to the 5GMM cause values:
 - 1) 5GMM cause values #74 "Temporarily not authorized for this SNPN" and #75 "Permanently not authorized for this SNPN" are supported whereas these 5GMM cause values cannot be used in a PLMN; and
 - 5GMM cause values #11 "PLMN not allowed", #31 "Redirection to EPC required", #73 "Serving network not authorized", and #76 "Not authorized for this CAG or authorized for CAG cells only" are not supported whereas these 5GMM cause values can be used in a PLMN;

- NOTE 1: The network does not send 5GMM cause value #13 to the UE operating in SNPN access operation mode in this release of specification.
- g) a list of "5GS forbidden tracking areas for roaming" and a list of "5GS forbidden tracking areas for regional provision of service" are managed per SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, entry of the "list of subscriber data" or PLMN subscription (see 3GPP TS 23.122 [5]);
- h) when accessing SNPN services via a PLMN using 3GPP access, access to 5GCN of the SNPN is performed using 5GMM procedures for non-3GPP access, 5GMM parameters for non-3GPP access, the UE is performing access to SNPN over non-3GPP access and the UE is not operating in SNPN access mode over 3GPP access. When accessing PLMN services via a SNPN using 3GPP access, access to 5GCN of the PLMN is performed using 5GMM procedures for non-3GPP access, 5GMM parameters for non-3GPP access, the UE is not performing access to SNPN over non-3GPP access, 5GMM parameters for non-3GPP access, the UE is not performing access to SNPN over non-3GPP access, and the UE is operating in SNPN access mode over 3GPP access. From the UE's NAS perspective, accessing PLMN services via an SNPN and accessing SNPN services via a PLMN are treated as untrusted non-3GPP access. If the UE is accessing the PLMN using non-3GPP access, the access to 5GCN of the SNPN via PLMN is not specified in this release of the specification.

Emergency services are not supported in an SNPN when a UE accesses SNPN services via a PLMN;

- NOTE 2: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- i) when registered to an SNPN, the UE shall use only the UE policies provided by the registered SNPN;
- j) equivalent SNPN is not supported;
- k) void;
- l) void;
- m) UE mobility between SNPNs in 5GMM-CONNECTED mode is not supported, UE mobility between SNPNs in 5GMM-IDLE mode is supported when the UE supports access to an SNPN using credentials from a credentials holder, and UE mobility between an SNPN and a PLMN is not supported;
- n) CIoT 5GS optimizations are not supported;
- accessing SNPN services using non-3GPP access is not supported, except when accessing SNPN services via a PLMN using 3GPP access as specified in item h;
- p) when registering or registered to an SNPN, the UE shall handle the 5GS mobile identity as described in subclause 5.5.1.2.2;
- q) when registering or registered to an SNPN, the UE shall only consider:
 - 1) a last visited registered TAI visited in the same SNPN as an available last visited registered TAI; or
 - a last visited registered TAI visited using the same entry of the "list of subscriber data" or the same PLMN subscription as an available last visited registered TAI, if the UE supports access to an SNPN using credentials from a credentials holder;
- NOTE 3: If the last visited registered TAI is assigned by an SNPN other than the current SNPN, the serving AMF can determine the SNPN assigning the last visited registered TAI using the NID provided by the UE.
- r) emergency service fallback is not supported;
- s) when registering or registered for onboarding services in SNPN, the UE shall not provide the requested NSSAI to the network;
- s1) when performing initial registration for onboarding services in SNPN, the UE shall set the 5GS registration type value to "SNPN onboarding registration";
- t) when registering or registered for onboarding services in SNPN, the AMF shall not provide the configured NSSAI, the allowed NSSAI or the rejected NSSAI to the UE, shall use the S-NSSAI included in the AMF onboarding configuration data for onboarding services in SNPN and shall not perform NSSAA procedure for S-NSSAI used for onboarding services in SNPN;

- u) the UE can access an SNPN indicating that onboarding is allowed using default UE credentials for primary authentication in order for the UE to be configured with one or more entries of the "list of subscriber data";
- x) eCall over IMS is not supported in SNPN access operation mode and the UE ignores any USIM configuration for eCall only mode;
- y) when registering or registered for onboarding services in SNPN, the AMF shall store in the 5GMM context of the UE an indication that the UE is registered for onboarding services in SNPN;
- z) a UE with multiple valid entries of "list of subscriber data", or one or more valid USIMs and one or more valid entries of "list of subscriber data", capable of initiating and maintaining simultaneous separate registration states over 3GPP access with PLMN(s) or SNPN(s), using identities and credentials associated with those entries of "list of subscriber data", or USIMs and entries of "list of subscriber data", and supporting one or more of the N1 NAS signalling connection release, the paging indication for voice services, the reject paging request, the paging restriction and the paging timing collision control may use procedures defined for MUSIM UE, even if the UE does not include multiple valid USIMs;
- za) when the UE is registering or registered for onboarding services in SNPN, the network slice admission control is not performed; and
- NOTE 4: If the network determines that the UE cannot register to the onboarding SNPN due to lack of resources for the network slice used for onboarding, the AMF can reject the UE with 5GMM cause #22 "congestion".
- v) proximity based services (5G ProSe as specified in 3GPP TS 24.554 [19E]) are not supported.

4.14.3 Public network integrated non-public network (PNI-NPN)

A PNI-NPN is made available by means of e.g. dedicated DNNs or by one or more S-NSSAIs allocated for it. A CAG can be optionally used in order to prevent UEs not allowed to access a PNI-NPN from accessing the PNI-NPN. The key enablers for the CAG in the NAS layer are as follows:

- a) CAG selection (see 3GPP TS 23.122 [5]); and
- b) provisioning of a "CAG information list" as specified in 3GPP TS 23.122 [5], from network to UE via the generic UE configuration update procedure, the registration procedure, the service request procedure, and the network-initiated de-registration procedure.

The "CAG information list" provisioned by the network, if available, is stored in the non-volatile memory in the ME as specified in annex C. The "CAG information list" stored in the ME is kept when the UE enters 5GMM-DEREGISTERED state. Annex C specifies condition under which the "CAG information list" stored in the ME is deleted. Additionally, when a USIM is inserted, if:

- no "CAG information list" is stored in the non-volatile memory of the ME; or
- the SUPI from the USIM does not match the SUPI stored together with the "CAG information list" in the non-volatile memory of the ME;

and the UE has a "CAG information list" stored in the USIM (see 3GPP TS 31.102 [22]), the UE shall store the "CAG information list" from the USIM into the ME, as specified in annex C. The "Allowed CAG list" included in the entry for the HPLMN or EHPLMN in "CAG information list" stored in the USIM can contain a range of CAG-IDs.

The UE supporting CAG may perform the initial registration for emergency services via a non-CAG cell in a PLMN for which the UE has an "indication that the UE is only allowed to access 5GS via CAG cells" or via a CAG cell that is not included in the "Allowed CAG list" (see 3GPP TS 23.122 [5]) for the selected PLMN. If a UE supporting CAG having an emergency PDU session is camping on:

- a) a CAG cell and none of the CAG-IDs of the CAG cell are included in the "Allowed CAG list" for the current PLMN in the UE's subscription; or
- b) a non-CAG cell in a PLMN for which the UE's subscription contains an "indication that the UE is only allowed to access 5GS via CAG cells";

the AMF shall behave as specified in subclause 5.4.4.2, 5.5.1.3.4 or 5.6.1.4.1.

NOTE: The emergency services in a PLMN for which the UE's subscription contains an "indication that the UE is only allowed to access 5GS via CAG cells" can be subject to local regulation.

Proximity based services (5G ProSe as specified in 3GPP TS 24.554 [19E]) are not supported in this release of the specification when a UE is camping on a CAG cell.

4.15 Time synchronization and time sensitive communication

4.15.1 General

A 5GS can support time synchronization and TSC (see 3GPP TS 23.501 [8], 3GPP TS 23.502 [9], and 3GPP TS 23.503 [10]). The clause describes NAS-specific aspects of the 5GS features to support time synchronization and TSC. Interworking with EPS is not supported for a PDU session for time synchronization or TSC.

- 4.15.2 Void
- 4.15.2.1 Void
- 4.15.2.2 Void
- 4.15.2.3 Void

4.15.3 Time synchronization

Two types of synchronization processes are supported by the 5GS: 5GS synchronization and (g)PTP domain synchronization (see 3GPP TS 23.501 [8]).

For 5GS synchronization, the lower layers provide the 5G internal system clock signalled via the NG-RAN (see 3GPP TS 38.331 [30]) and the UE forwards the 5G internal system clock to the DS-TT(s).

For (g)PTP domain synchronization, the UE supports forwarding (g)PTP messages (see 3GPP TS 23.501 [8], 3GPP TS 23.502 [9], and 3GPP TS 24.535 [19A]). For all (g)PTP domains associated with a PDU session:

- a) if the UE receives (g)PTP message via the PDU session, the UE forwards the (g)PTP messages to the DS-TT associated with the PDU session; or
- b) if the UE receives (g)PTP messages from the DS-TT associated with the PDU session, the UE forwards the (g)PTP messages via the PDU session.

4.15.4 User plane node management

A 5G system (5GS) can act as a user plane node of an external network (e.g. IEEE TSN bridge) or a 5GS can be independently used to enable TSC. For these purposes, information available at a UE is provided to the network and port management information containers are exchanged between a DS-TT and a TSN AF or a TSCTSF (see 3GPP TS 24.539 [19BA]).

During a UE-requested PDU session establishment procedure, if the UE supports transfer of port management information containers, then the UE indicates that transfer of port management information container is supported and the UE provides a DS-TT Ethernet port MAC address (if the PDU session type is Ethernet), port management information container, and a UE-DS-TT residence time (if available) to the network (see subclause 6.4.1.2).

Once the UE has successfully established a PDU session and the UE has indicated that transfer of port management information container is supported during the UE-requested PDU session establishment procedure (see subclause 6.4.1.2), then port management information containers are exchanged via a UE-requested PDU session modification procedure and a network-requested PDU session modification procedure (see subclauses 6.3.2 and 6.4.2). The UE receiving a port management information container from the network shall forward the port management

information container to the DS-TT. The SMF receiving a port management information container from the UE shall operate as described in 3GPP TS 23.502 [9].

4.16 UE radio capability signalling optimisation

UE radio capability signalling optimisation (RACS) is a feature that is optional at both the UE and the network and which aims to optimise the transmission of UE radio capability over the radio interface (see 3GPP TS 23.501 [8]). RACS works by assigning an identifier to represent a set of UE radio capabilities. This identifier is called the UE radio capability ID. A UE radio capability ID can be either manufacturer-assigned or network-assigned. The UE radio capability ID is an alternative to the signalling of the radio capabilities container over the radio interface.

In this release of the specification, RACS is applicable to neither NB-N1 mode nor non-3GPP access.

If the UE supports RACS:

- a) the UE shall indicate support for RACS by setting the RACS bit to "RACS supported" in the 5GMM capability IE of the REGISTRATION REQUEST message;
- b) if the UE performs a registration procedure for initial registration and the UE has an applicable UE radio capability ID for the current UE radio configuration in the selected network, the UE shall include the UE radio capability ID in the UE radio capability ID IE as a non-cleartext IE in the REGISTRATION REQUEST message. If both a network-assigned UE radio capability ID and a manufacturer-assigned UE Radio Capability ID are applicable, the UE shall include the network-assigned UE radio capability ID in the REGISTRATION REGISTRATION REQUEST message;
- c) if the radio configuration at the UE changes (for instance because the UE has disabled a specific radio capability) then:
 - if the UE has an applicable UE radio capability ID for the new UE radio configuration, the UE shall initiate a registration procedure for mobility and periodic registration update. The UE shall include the applicable UE radio capability ID in the UE radio capability ID IE of the REGISTRATION REQUEST message and shall include the 5GS update type IE in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to "UE radio capability update needed". If both a network-assigned UE radio capability ID and a manufacturer-assigned UE Radio Capability ID are applicable, the UE shall include the network-assigned UE radio capability ID in the REGISTRATION REQUEST message; and
 - 2) if the UE does not have an applicable UE radio capability ID for the new UE radio configuration, the UE shall initiate a registration procedure for mobility and periodic registration update and include the 5GS update type IE in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to "UE radio capability update needed";
- NOTE: Performing the registration procedure for mobility and periodic registration update and including the 5GS update type IE in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to "UE radio capability update needed" without a UE radio capability ID included in the REGISTRATION REQUEST message can trigger the network to assign a new UE radio capability ID to the UE.
- d) upon receiving a network-assigned UE radio capability ID in the REGISTRATION ACCEPT message or the CONFIGURATION UPDATE COMMAND message, the UE shall store the network-assigned UE radio capability ID and the PLMN ID or SNPN identity of the serving network and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription along with a mapping to the current UE radio configuration in its non-volatile memory as specified in annex C. The UE shall be able to store at least the last 16 received network-assigned UE radio capability IDs with the associated PLMN ID or SNPN identity and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription and the mapping to the corresponding UE radio configuration;
- e) the UE shall not use a network-assigned UE radio capability ID assigned by a PLMN in PLMNs equivalent to the PLMN which assigned it;
- f) upon receiving a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested" in the REGISTRATION ACCEPT message or the CONFIGURATION UPDATE COMMAND message, the UE shall delete all network-assigned UE radio capability IDs stored at the UE for the serving network, initiate a registration procedure for mobility and periodic registration update and include an

applicable manufacturer-assigned UE radio capability ID for the current UE radio configuration, if available at the UE, in the UE radio capability ID IE of the REGISTRATION REQUEST message; and

g) if the UE performs a registration procedure for mobility and periodic registration update due to entering a tracking area that is not in the list of tracking areas that the UE previously registered in the AMF and the UE has an applicable UE radio capability ID for the current UE radio configuration in the selected network, the UE shall include the UE radio capability ID in the UE radio capability ID IE as a non-cleartext IE in the REGISTRATION REQUEST message. If both a network-assigned UE radio capability ID are applicable, the UE shall include the network-assigned UE radio capability ID are applicable, the UE shall include the network-assigned UE radio capability ID are applicable.

If the network supports RACS:

- a) the network may assign a network-assigned UE radio capability ID to a UE which supports RACS by including a UE radio capability ID IE in the REGISTRATION ACCEPT message or in the CONFIGURATION UPDATE COMMAND message;
- b) the network may trigger the UE to delete all network-assigned UE radio capability IDs stored at the UE for the serving network by including a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested" in the REGISTRATION ACCEPT message or in the CONFIGURATION UPDATE COMMAND message; and
- c) the network may send an IDENTITY REQUEST message to the UE that supports RACS to retrieve the PEI, if not available in the network.

4.17 5GS mobility management in NB-N1 mode

A UE in NB-N1 mode (see 3GPP TS 36.331 [25A]) shall calculate the value of the applicable NAS timer indicated in table 10.2.1 plus 240s.

The timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not re-calculate the use of the NAS timer value until the NAS procedure is completed, restarted or aborted.

When an AMF that supports NB-N1 mode performs NAS signalling with a UE, which is using NB-N1 mode, the AMF shall calculate the value of the applicable NAS timer indicated in table 10.2.2 plus 240s.

The timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not re-calculate the use of the NAS timer value until the NAS procedure is completed, restarted or aborted.

4.18 5GS session management in NB-N1 mode

A UE in NB-N1 mode (see 3GPP TS 36.331 [25A]) shall calculate the value of the applicable NAS timer indicated in table 10.3.1 plus 180s.

The timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not re-calculate the use of the NAS timer value until the NAS procedure is completed, restarted or aborted.

When an SMF that supports NB-N1 mode performs NAS signalling with a UE, which is using NB-N1 mode, the SMF shall calculate the value of the applicable NAS timer indicated in table 10.3.2 plus 180s.

The timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not re-calculate the use of the NAS timer value until the NAS procedure is completed, restarted or aborted.

4.19 5GS mobility management in WB-N1 mode for IoT

In WB-N1 mode, a UE operating in category CE can operate in either CE mode A or CE mode B (see 3GPP TS 36.306 [25D]). If a UE that supports CE mode B and operates in WB-N1 mode, the UE's usage setting is not set to "voice centric" (see 3GPP TS 23.501 [8]), and:

- a) the use of enhanced coverage is not restricted by the network; or
- b) CE mode B is not restricted by the network (see 3GPP TS 23.501 [8]);

the UE shall apply the value of the applicable NAS timer indicated in table 10.2.1 for WB-N1/CE mode.

A UE that supports CE mode B and operates in WB-N1 mode shall not apply the value of the applicable NAS timer indicated in table 10.2.1 for WB-N1/CE mode before receiving an indication from the network that the use of enhanced coverage is not restricted, or CE mode B is not restricted, as described in this subclause.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure, and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

The support of CE mode B by a UE is indicated to the AMF by lower layers and shall be stored by the AMF. When an AMF that supports WB-N1 mode performs NAS signalling with a UE, which supports CE mode B and operates in WB-N1 mode, the UE's usage setting is not set to "voice centric" (see 3GPP TS 23.501 [8]) and the AMF determines that:

- a) the use of enhanced coverage is not restricted for the UE; or
- b) CE mode B is not restricted for the UE (see 3GPP TS 23.501 [8]);

the AMF shall calculate the value of the applicable NAS timer indicated in table 10.2.2 for WB-N1/CE mode.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

4.20 5GS session management in WB-N1 mode for IoT

In WB-N1 mode, a UE operating in category CE can operate in either CE mode A or CE mode B (see 3GPP TS 36.306 [25D]). If a UE that supports CE mode B and operates in WB-N1 mode and the UE's usage setting is not set to "voice centric" (see 3GPP TS 23.501 [8]), and:

- a) the use of enhanced coverage is not restricted by the network; or
- b) CE mode B is not restricted by the network (see 3GPP TS 23.501 [8]);

the UE shall apply the value of the applicable NAS timer indicated in table 10.3.1 for WB-N1/CE mode.

A UE that supports CE mode B and operates in WB-N1 mode shall not apply the value of the applicable NAS timer indicated in table 10.3.1 for WB-N1/CE mode before receiving an indication from the network that the use of enhanced coverage is not restricted, or CE mode B is not restricted, as described in this subclause.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure, and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

If the use of extended NAS timer is indicated by the AMF (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]), the SMF shall calculate the value of the applicable NAS timer indicated in table 10.3.2 for WB-N1/CE mode.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

4.21 Authentication and Key Management for Applications (AKMA)

The UE may support AKMA.

The purpose of AKMA is to provide authentication and key management to applications based on 3GPP credentials used for 5GS access as specified in 3GPP TS 33.535 [24A], which allows the UE to securely exchange data with an AKMA application function.

Upon receiving a request from the upper layers to obtain AKMA Anchor Key (K_{AKMA}) and AKMA Key Identifier (A-KID), the UE supporting AKMA shall derive the K_{AKMA} and the AKMA Temporary Identifier (A-TID) from the valid K_{AUSF} if available as specified in 3GPP TS 33.535 [24A], shall further derive the A-KID from the A-TID as specified in 3GPP TS 33.535 [24A] and shall provide K_{AKMA} and A-KID to the upper layers.

The UE supporting AKMA shall notify the upper layers whenever there is a change of the K_{AUSF} upon reception of an EAP-success message in subclauses 5.4.1.2.2.8, 5.4.1.2.3.1 and 5.4.1.2.3A.1 or upon reception of SECURITY MODE COMMAND message in subclauses 5.4.2.3.

During an ongoing primary authentication and key agreement procedure (see subclause 5.4.1), if the UE receives a request from upper layers to obtain K_{AKMA} and A-KID, the UE shall derive the K_{AKMA} and A-TID after the completion of the ongoing primary authentication and key agreement procedure, shall further derive the A-KID from the A-TID as specified in 3GPP TS 33.535 [24A] and shall provide K_{AKMA} and A-KID to the upper layers.

- NOTE 1: The upper layers derive the AKMA Application Key (K_{AF}) from K_{AKMA} as specified in 3GPP TS 33.535 [24A].
- NOTE 2: The knowledge of whether a certain application needs to use AKMA or not is application specific and is out of the scope of 3GPP.
- NOTE 3: The exact method of securing the data exchange at the upper layers using K_{AF} is application specific and is out of the scope of 3GPP.
- NOTE 4: The upper layers request the UE NAS layer to provide K_{AKMA} and A-KID before the upper layers initiate communication with an AKMA application function.
- NOTE 5: Upon receiving a request from the upper layers to obtain K_{AKMA} and A-KID, if there is no K_{AUSF} available, the UE NAS layer cannot derive the K_{AKMA} and A-KID and provides an indication to the upper layers that K_{AKMA} and A-KID cannot be generated.

4.22 Uncrewed aerial vehicle identification, authentication, and authorization

4.22.1 General

A 5GS can support UAV identification, authentication, and authorization (see 3GPP TS 23.256 [6AB]). This subclause describes NAS-specific aspects of the 5GS features to support UAV identification, authentication, authorization and C2 communication authorization.

Before accessing 5GS for UAS services, the UE supporting UAS services must have an assigned CAA-level UAV ID. The UE can be registered to 5GS for UAS services if there is a valid aerial subscription in the UE's subscription.

4.22.2 Authentication and authorization of UAV

The 5GS supports the USS UAV Authorization and Authentication (UUAA) procedure for a UE supporting UAS services. Depending on operator policy or regulatory requirements, the UUAA-MM procedure can be performed by the UE and the AMF at a registration procedure as specified in subclause 5.5.1.2 or the UUAA-SM procedure can be performed by the UE and the SMF at a PDU session establishment procedure as specified in subclause 6.4.1.2. The UE shall support UUAA-MM and UUAA-SM, and the network shall support UUAA-SM and may optionally support UUAA-MM. The UUAA procedure needs to be performed by 5GS with USS successfully before the connectivity for UAS services is established.

During the registration procedure as described in subclause 5.5.1.2, the UE supporting UAS services provides CAAlevel UAV ID to the AMF, and the AMF may trigger the UUAA-MM procedure. If the UE supporting UAS services does not provide CAA-level UAV ID to the AMF and the network is configured to perform UUAA-MM at registration procedure, the AMF may accept the registration request and shall mark in the UE's 5GMM context that the UE is not allowed to request UAS services. If the UE wants to use the UAS services by providing the CAA-Level UAV ID later on, the UE shall perform the registration procedure for mobility and periodic registration update.

When a UE supporting UAS services requests to establish a PDU session as described in subclause 6.4.1.2 for USS communication, the UE provides CAA-level UAV ID to the network, and the SMF may trigger the UUAA-SM procedure. If the UE does not provide CAA-level UAV ID and the SM subscription data for the UE requires the UUAA-SM, the network rejects the UE-requested PDU session establishment procedure for the UAS services.

The UE supporting UAS services shall not provide CAA-level UAV ID to the network over non-3GPP access, and the network shall not perform UUAA procedure for non-3GPP access and shall ensure that the UE is not allowed to access any aerial services in non-3GPP access.

If provided by the upper layers, the UE supporting UAS services provides to the network the USS address during the registration procedure or PDU session establishment procedure so that the network uses the information to discover the USS.

NOTE: The parameters (e.g., CAA-level UAV ID or USS address) sent by a UE supporting UAS services to the network for UAS services are included in the Service-level-AA container IE which is a non-cleartext IE.

After successful UUAA procedure, either the AMF or the SMF may initiate re-authentication of the UAV when required by the USS. If UUAA-MM fails during a re-authentication and there are PDU sessions established using UAS services, the AMF shall request the SMF to perform the release of these PDU sessions and may trigger a network-initiated de-registration procedure based on operator policy. If UUAA-SM fails during a re-authentication, the SMF shall release the PDU session related to re-authentication.

If the UUAA is revoked, the PDU session related to the UAS services shall be released by the SMF. Based on operator policy, the AMF may decide to keep the UE registered or trigger a de-registration procedure.

4.22.3 Authorization of C2 communication

The 5GS supports USS authorization of C2 communication for pairing of UAV and UAV-C. The pairing of UAV and UAV-C needs to be authorized by USS successfully before the user plane connectivity for C2 communication is enabled. For C2 authorization procedure, the UE supporting UAS services provides to the network with CAA-level UAV ID.

The USS authorization of UAV flight can also be performed during the C2 authorization procedure. The UE supporting UAS services provides the UAV flight authorization information to the network if provided by upper layers.

NOTE 1: The C2 authorization payload in the service-level-AA payload can include the pairing information for C2 communication and the UAV flight authorization information (see subclauses 6.4.1.2 and 6.4.2.2).

If a UE supporting UAS services uses a common PDU session for both USS communication and C2 communication, the C2 communication can be authorized using UUAA-SM procedure during the PDU session establishment procedure or during the PDU session modification procedure. If the pairing of UAV and UAV-C is revoked, the network shall disable C2 communication for the PDU session.

NOTE 2: The network can disable C2 communication for the PDU session e.g., by removing the QoS flow for C2 communication during PDU session modification procedure as decribed in subclauses 6.3.2.2.

If a UE supporting UAS services uses separate PDU sessions for, respectively, USS communication and C2 communication, the C2 communication is authorized using UUAA-SM during the PDU session establishment procedure. If the pairing of UAV and UAV-C is revoked, the PDU session for C2 communication shall be released by the SMF.

4.22.4 Void

4.23 NAS over Non-Terrestrial Network

4.23.1 General

A 5GS can support 3GPP satellite NG-RAN access technology (see 3GPP TS 23.501 [8]). This clause describes NAS-specific aspects of the 5GS features to support 3GPP satellite NG-RAN access technology.

4.23.2 List of "PLMNs not allowed to operate at the present UE location"

For 3GPP satellite NG-RAN the UE shall store a list of "PLMNs not allowed to operate at the present UE location". Each entry consists of:

- a) the PLMN identity of the PLMN which sent a message including 5GMM cause value #78 "PLMN not allowed to operate at the present UE location" via satellite NG-RAN access technology; and
- b) the geographical location, if known by the UE, where 5GMM cause value #78 was received on satellite NG-RAN access technology; and
- c) if the geographical location exists, a UE implementation specific distance value.

Before storing a new entry in the list, the UE shall delete any existing entry with the same PLMN identity. Upon storing a new entry, the UE starts a timer instance associated with the entry with an implementation specific value that shall not be set to a value smaller than the timer value indicated by the network in the Lower bound timer value IE, if any. If the Lower bound timer value IE was not provided by the network, the value of the timer shall be set based on the UE implementation.

The UE is allowed to attempt to access a PLMN via satellite NG-RAN access technology which is part of the list of "PLMNs not allowed to operate at the present UE location" only if:

- a) the current UE location is known, a geographical location is stored for the entry of this PLMN, and the distance to the current UE location is larger than a UE implementation specific value.
- b) the timer associated with the entry of this PLMN has expired; or
- c) the access is for emergency services (see 3GPP TS 23.122 [5] for further details).
- NOTE: When the UE is accessing network for emergency services, it is up to operator and regulatory policies whether the network needs to determine if the UE is in a location where network is not allowed to operate.

The list shall accommodate three or more entries. The maximum number of entries is an implementation decision. When the list is full and a new entry has to be inserted, the oldest entry shall be deleted.

Each entry shall be removed if for the entry:

- a) the UE successfully registers via satellite NG-RAN access technology to the PLMN stored in the entry except when the UE registers for emergency services; or
- b) the timer instance associated with the entry expires.

The UE may delete the entry in the list, if the current UE location is known, a geographical location is stored for the entry of this PLMN, and the distance to the current UE location is larger than a UE implementation specific value.

If the UE is in 5GMM-DEREGISTERED.LIMITED-SERVICE state and an entry from the list of "PLMNs not allowed to operate at the present UE location" is removed, the UE shall perform PLMN selection according to 3GPP TS 23.122 [5].

When the UE is switched off, the UE shall keep the list of "PLMNs not allowed to operate at the present UE location" in its non-volatile memory. The UE shall delete the list of "PLMNs not allowed to operate at the present UE location" if the USIM is removed.

If the UE is switched off when the timer instance associated with the entry in the list is running, the UE shall behave as follows when the UE is switched on and the USIM in the UE remains the same:

let t1 be the time remaining for timer instance associated with the entry in the list to timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 - t. If t1 is equal to or less than t, then the timer need not be restarted and considered expired. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

4.23.3 5GS mobility management via a satellite NG-RAN cell

For 5GS mobility management via a satellite NG-RAN cell the UE shall apply the value of the applicable NAS timer indicated in table 10.2.1 for access via a satellite NG-RAN cell.

NOTE 1: The applied NAS timer values are based on the current satellite NG-RAN access RAT type determined based on information from lower layers.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure, and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

The access via a satellite NG-RAN cell by a UE is indicated to the AMF by lower layers and shall be stored by the AMF. When an AMF that supports access via satellite NG-RAN cells performs NAS signalling with a UE via satellite NG-RAN cells, the AMF shall calculate the value of the applicable NAS timer indicated in table 10.2.2 for access via a satellite NG-RAN cell.

NOTE 2: The applied NAS timer values are based on the current satellite NG-RAN access RAT type determined based on information from lower layers.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

4.23.4 5GS session management via a satellite NG-RAN cell

For 5GS session management via a satellite NG-RAN cell the UE shall apply the value of the applicable NAS timer indicated in table 10.2.1 for access via a satellite NG-RAN cell.

NOTE 1: The applied NAS timer values are based on the current satellite NG-RAN access RAT type determined based on information from lower layers.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure, and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

If the use of extended NAS timer for access via a satellite NG-RAN cell is indicated by the AMF (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]), the SMF shall calculate the value of the applicable NAS timer indicated in table 10.3.2 for access via a satellite NG-RAN cell.

The NAS timer value obtained is used as described in the appropriate procedure subclause of this specification. The NAS timer value shall be calculated at start of a NAS procedure and shall not be re-calculated until the NAS procedure is completed, restarted or aborted.

4.23.5 Handling multiple tracking area codes from the lower layers

When a UE camps on a satellite NG-RAN cell, the UE may receive multiple TACs from the lower layers. The UE shall construct TAIs from the multiple TACs (i.e. concatenate the identity of the current PLMN and each of the TACs) and select a TAI as follows:

a) if at least one TAI belongs to the current registration area of the UE, the UE shall select a TAI which belongs to the current registration area of the UE according to the followings.

If there are multiple TAIs which belong to the current registration area of the UE, the UE shall select a TAI as follows:

- if there is a TAI which belongs to the list of "allowed tracking area" (if any) and does not belong to the list of "non-allowed tracking areas" (if any), the UE shall select a TAI which belongs to the list of "allowed tracking area" (if any) and does not belong to the list of "non-allowed tracking areas" (if any). In this case, if there are multiple TAIs which belong to the list of "allowed tracking area" (if any) and does not belong to the list of "non-allowed tracking areas" (if any), then the UE shall consider each of these TAIs equal and select a TAI in an implementation-specific way (e.g. taking into account LADN service area information). If these multiple TAIs contain the previous current TAI, the current TAI can be left unchanged.
- 2) if there is no TAI which belongs to the list of "allowed tracking area" (if any) and does not belong to the list of "non-allowed tracking areas" (if any) or neither the list of "allowed tracking area" nor the list of "non-allowed tracking areas" is available, then the UE shall consider each of these TAIs equal and select a TAI in an implementation-specific way (e.g. taking into account LADN service area information). If these multiple TAIs contain the previous current TAI, the curren TAI can be left unchanged.
- b) if no TAI belongs to the current registration area of the UE and:
 - there is a TAI which belongs to neither the list of "5GS forbidden tracking areas for roaming" nor the list of "5GS forbidden tracking areas for regional provision of service", the UE shall select a TAI which belongs to neither the list of "5GS forbidden tracking areas for roaming" nor the list of "5GS forbidden tracking areas for regional provision of service". In this case, if there are multiple TAIs which belong to neither the list of "5GS forbidden tracking areas for roaming" nor the list of "5GS forbidden tracking areas for regional provision of service". In this case, if there are multiple TAIs which belong to neither the list of "5GS forbidden tracking areas for roaming" nor the list of "5GS forbidden tracking areas for regional provision of service", then the UE shall consider each of these TAIs equal and select a TAI in an implementation-specific way.
 - 2) all TAIs belong to the list of "5GS forbidden tracking areas for roaming" or the list of "5GS forbidden tracking areas for regional provision of service", then the UE shall consider each of these TAIs equal and select a TAI in an implementation-specific way.

The UE shall consider the selected TAI as the current TAI. The UE shall select a TAI when:

- a) the UE receives multiple TACs from the lower layers; or
- b) the UE has received multiple TACs from the lower layers upon starting to camping on the current cell and the registration area, the list of "allowed tracking areas", the list of "non-allowed tracking areas", the list of "5GS forbidden tracking areas for roaming", or the list of "5GS forbidden tracking areas for regional provision of service" is updated.

Handling of the list of "5GS forbidden tracking areas for roaming" and the list of "5GS forbidden tracking areas for regional provision of service" is specified in clause 5.3.13.

4.24 Minimization of service interruption

The UE and the network may support Minimization of service interruption (MINT). MINT aims to enable a UE to obtain service from a PLMN offering disaster roaming services when a disaster condition applies to the UE's determined PLMN with disaster condition.

If the UE supports MINT, the indication of whether disaster roaming is enabled in the UE, the indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN', the one or more "list of PLMN(s) to be used in disaster condition", disaster roaming wait range and disaster return wait range provisioned by the network, if available, are stored in the non-volatile memory in the ME as specified in annex C and are kept when the UE enters 5GMM-DEREGISTERED state. Annex C specifies condition under which the indication of whether disaster roaming is enabled in the UE, the indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN', the one or more "lists of PLMN(s) to be used in disaster roaming wait range and disaster return wait range stored in the ME are deleted.

Upon selecting a PLMN for disaster roaming as specified in 3GPP TS 23.122 [6]:

- a) if the UE does not have a stored disaster roaming wait range, the UE shall perform a registration procedure for disaster roaming services on the selected PLMN as described in clause 5.5.1; and
- b) if the UE has a stored disaster roaming wait range, the UE shall generate a random number within the disaster roaming wait range and start a timer with the generated random number. While the timer is running, the UE shall not initiate registration on the selected PLMN except if the UE needs to request an emergency PDU session, in

which case the UE shall initiate the registration procedure, set the 5GS registration type IE to "emergency registration" in the REGISTRATION REQUEST message and keep the timer running. Upon expiration of the timer, if the UE does not have an emergency PDU session, the UE shall perform a registration procedure for disaster roaming services as described in clause 5.5.1 if still camped on the selected PLMN. If the UE has an emergency PDU session when the timer expires, the registration procedure for disaster roaming services as described in clause 5.5.1 shall be performed after the release of the emergency PDU session, if the UE is still camped on the selected PLMN.

The timer started with a generated random number within the disaster roaming wait range is stopped and the UE shall perform a PLMN selection as described in 3GPP TS 23.122 [5], if:

- a) the UE has successfully registered over non-3GPP access on another PLMN;
- b) the UE has successfully registered with an allowable PLMN; or
- c) an NG-RAN cell selected for camping of the selected PLMN broadcasts neither the disaster related indication nor a "list of one or more PLMN(s) with disaster condition for which disaster roaming services is offered by the available PLMN" including the determined PLMN with Disaster Condition (see 3GPP TS 23.122 [5]).

Upon determining that a disaster condition has ended and that the UE shall perform PLMN selection as specified in 3GPP TS 23.122 [6]:

- a) if the UE does not have a stored disaster return wait range, the UE shall perform a registration procedure on the selected PLMN; and
- b) if the UE has a stored disaster return wait range, the UE shall generate a random number within the disaster return wait range and start a timer with the generated random number value. While the timer is running, the UE shall not initiate registration on the selected PLMN except if the UE needs to request an emergency PDU session, in which case the UE shall initiate the registration procedure, set the 5GS registration type IE to "emergency registration" in the REGISTRATION REQUEST message and keep the timer running. Upon expiration of the timer, if the UE does not have an emergency PDU session, the UE shall perform a registration procedure if still camped on the selected PLMN. If the UE has an emergency PDU session when the timer expires, the registration procedure as described in clause 5.5.1 shall be performed after the release of the emergency PDU session, if the UE is still camped on the selected PLMN.

When the AMF assigns a registration area to the UE registered for disaster roaming services, the AMF shall only include TAIs covering the area with the disaster condition.

When the AMF determines that the disaster condition has ended and the UE which is registered for disaster roaming services has an emergency PDU session, the AMF shall initiate the generic UE configuration update procedure to indicate that the UE is registered for emergency services as described in subclause 5.4.4.2.

Interworking with EPS is not supported for UEs that are registered for disaster roaming services. When registering for disaster roaming services, the UE indicates to the network that S1 mode is not supported as described in subclause 5.5.1.2.2. While registered for disaster roaming services and upon a need to establish an emergency PDU session or perform emergency services fallback, the UE initiates the registration procedure for mobility and periodic registration update and indicates that S1 mode is supported as described in subclause 5.5.1.3.2.

4.25 Support of MUSIM features

A network and a MUSIM UE may support one or more of the MUSIM features (i.e. the N1 NAS signalling connection release, the paging indication for voice services, the reject paging request, the paging restriction and the paging timing collision control).

If MUSIM UE supports one or more MUSIM features, the UE indicates support of one or more MUSIM features (except for the paging timing collision control) during the registration procedure. If the UE has indicated support of the N1 NAS signalling connection release or the reject paging request or both and the UE supports the paging restriction, the UE indicates support of the paging restriction.

If the UE indicates support of one or more MUSIM features and the network decides to accept one or more MUSIM features, the network indicates the support of one or more MUSIM features during the registration procedure. The network only indicates the support of the paging restriction together with the support of either N1 NAS signalling connection release or the reject paging request.

The network does not indicate support for any MUSIM feature to the UE during the registration for emergency services.

If a UE stops fulfilling the condition to be considered a MUSIM UE as defined in subclause 3.1, and the UE has negotiated support of one or more MUSIM features, then the UE shall initiate a registration procedure for mobility and periodic registration update to indicate that all the MUSIM features are not supported (except for the paging timing collision control) as specified in subclause 5.5.1.3.

A MUSIM UE operating in NB-N1 mode or in WB-N1 mode CE mode B does not indicate the support for paging indication for voice services during the registration procedure towards the network.

5 Elementary procedures for 5GS mobility management

5.1 Overview

5.1.1 General

The main function of the 5GS mobility management (5GMM) sublayer is to support the identification, security, mobility of a UE as well as generic message transport.

A further function of the 5GMM sublayer is to provide connection management services to the other sublayer(s).

NOTE: In a satellite NG-RAN access, a GNSS fix time in lower layers can delay transmission of an initial UL NAS message by up to 100 seconds (GNSS cold state).

5.1.2 Types of 5GMM procedures

Depending on how they can be initiated, three types of 5GMM procedures can be distinguished:

a) 5GMM common procedures:

5GMM common procedure can always be initiated when the UE is in 5GMM-CONNECTED mode. The procedures belonging to this type are:

- 1) Initiated by the network:
 - i) network-initiated NAS transport;
 - ii) primary authentication and key agreement;
 - iii) security mode control;
 - iv) generic UE configuration update;
 - v) identification; and
 - vi) network slice-specific authentication and authorization;
- 2) Initiated by the UE:

UE-initiated NAS transport.

 Initiated by the UE or the network and used to report certain error conditions detected upon receipt of 5GMM protocol data:

5GMM status.

b) 5GMM specific procedures:

At any time only one UE initiated 5GMM specific procedure can be running for each of the access network(s) that the UE is camping in. The procedures belonging to this type are:

1) Initiated by the UE and used e.g. to register to the network for 5GS services and establish a 5GMM context, to update the location/parameter(s) of the UE:

registration.

2) Initiated by the UE or the network and used to deregister from the network for 5GS services and to release a 5GMM context:

de-registration.

3) Initiated by the UE and used to deregister from the network for 5GS services and to release a 5GMM context:

eCall inactivity procedure.

- c) 5GMM connection management procedures:
 - 1) Initiated by the UE and used to establish a secure connection to the network or to request the resource reservation for sending data, or both:

service request.

The service request procedure can only be initiated if no UE initiated 5GMM specific procedure is ongoing for each of the access network(s) that the UE is camping in.

2) Initiated by the network and used to request the establishment of an N1 NAS signalling connection or to request re-establishment of user-plane resources for the PDU session(s) associated with 3GPP access or to request re-establishment of user-plane resources of the PDU session(s) associated with non-3GPP access over 3GPP access; not applicable for the non-3GPP access network:

paging.

3) Initiated by the network and used to request re-establishment of user-plane resources of the PDU session(s) associated with non-3GPP access over 3GPP access or to deliver 5GSM downlink signalling messages associated with non-3GPP access over 3GPP access, when the UE is in 5GMM-CONNECTED mode over 3GPP access and in 5GMM-IDLE mode over non-3GPP access; or

Initiated by the network and used to request re-establishment of user-plane resources of the PDU session(s) associated with 3GPP access over 3GPP access or to deliver downlink signalling associated with 3GPP access, when the UE is in 5GMM-CONNECTED mode over non-3GPP access, and when the UE is in 5GMM-IDLE mode over 3GPP access and not in MICO mode:

notification.

- NOTE 1: In NB-N1 mode, the UE NAS using 5GS services with control plane CIoT 5GS optimization can wait for the lower layers to complete the transmission of the previous UL NAS TRANSPORT messages carrying control plane user data before providing subsequent NAS messages. Other implementations are possible.
- NOTE 2: When providing NAS messages to the lower layers for transmission, the UE NAS using 5GS services with control plane CIoT 5GS optimization can prioritize sending NAS signalling messages over the UL NAS TRANSPORT messages carrying control plane user data. How the UE performs this prioritization is implementation specific.

5.1.3 5GMM sublayer states

5.1.3.1 General

In the following subclauses, the 5GS mobility management (5GMM) sublayer of the UE and the network is described by means of different state machines. The 5GMM sublayer states is managed per access type independently, i.e. 3GPP access or non-3GPP access. In subclause 5.1.3.2, the states of the 5GMM sublayer are introduced.

5.1.3.2 5GMM sublayer states

5.1.3.2.1 5GMM sublayer states in the UE

5.1.3.2.1.1 General

In the following subclauses, the possible 5GMM sublayer states of the UE are described and shown in Figure 5.1.3.2.1.1.1.



NOTE: Not all possible transitions are shown in this figure. **Figure 5.1.3.2.1.1.1: 5GMM main states in the UE**

5.1.3.2.1.2 Main states

5.1.3.2.1.2.1 5GMM-NULL

5GS services are disabled in the UE. No 5GS mobility management function shall be performed in this state.

5.1.3.2.1.2.2 5GMM-DEREGISTERED

In the state 5GMM-DEREGISTERED, no 5GMM context has been established and the UE location is unknown to the network and hence it is unreachable by a network. In order to establish a 5GMM context, the UE shall start the initial registration procedure.

5.1.3.2.1.2.3 5GMM-REGISTERED-INITIATED

A UE enters the state 5GMM-REGISTERED-INITIATED after it has started the initial registration procedure or the non-initial registration procedure, and is waiting for a response from the network.

5.1.3.2.1.2.4 5GMM-REGISTERED

In the state 5GMM-REGISTERED, a 5GMM context has been established. Additionally, one or more PDU session(s) may be established at the UE. The UE may initiate the non-initial registration procedure (including the normal registration update and periodic registration update) and the service request procedure. The UE in the state 5GMM-REGISTERED over non-3GPP access shall not initiate the periodic registration update procedure.

5.1.3.2.1.2.5 5GMM-DEREGISTERED-INITIATED

A UE enters the state 5GMM-DEREGISTERED-INITIATED after it has requested release of the 5GMM context by starting the de-registration procedure and is waiting for a response from the network.

5.1.3.2.1.2.6 5GMM-SERVICE-REQUEST-INITIATED

A UE enters the state 5GMM-SERVICE-REQUEST-INITIATED after it has started the service request procedure and is waiting for a response from the network.

5.1.3.2.1.3 Substates of state 5GMM-DEREGISTERED

5.1.3.2.1.3.1 General

The state 5GMM-DEREGISTERED is subdivided into a number of substates as described in this subclause. The following substates are not applicable to non-3GPP access:

- a) 5GMM-DEREGISTERED.PLMN-SEARCH:
- b) 5GMM-DEREGISTERED.NO-CELL-AVAILABLE; and
- c) 5GMM-DEREGISTERED.eCALL-INACTIVE.

Valid subscriber data are available for the UE before it enters the substates, except for the substate 5GMM-DEREGISTERED.NO-SUPI.

5.1.3.2.1.3.2 5GMM-DEREGISTERED.NORMAL-SERVICE

The substate 5GMM-DEREGISTERED.NORMAL-SERVICE is chosen in the UE when a suitable cell has been found and the PLMN, SNPN, or tracking area is not in the forbidden list.

5.1.3.2.1.3.3 5GMM-DEREGISTERED.LIMITED-SERVICE

The substate 5GMM-DEREGISTERED.LIMITED-SERVICE is chosen in the UE, when it is known that a selected cell for 3GPP access or TA for non-3GPP access is unable to provide normal service (e.g. the selected cell over 3GPP access is in a forbidden PLMN or SNPN or is in a forbidden tracking area or TA for non-3GPP access is forbidden) or the selected cell is a CAG cell whose CAG ID is not included in the "Allowed CAG list" in the entry of the "CAG information list" for the PLMN, or the selected cell is a non-CAG cell in a PLMN for which there exists an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the PLMN.

5.1.3.2.1.3.4 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION

The substate 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION is chosen in the UE if the initial registration procedure failed due to a missing response from the network or due to the circumstances described in subclauses 5.5.1.2.4, 5.5.1.2.5, 5.5.1.2.7 and 5.5.1.3.4.

5.1.3.2.1.3.5 5GMM-DEREGISTERED.PLMN-SEARCH

The substate 5GMM-DEREGISTERED.PLMN-SEARCH is chosen in the UE, if the UE is searching for PLMNs or SNPNs. This substate is left either when a cell has been selected (the new substate is NORMAL-SERVICE or LIMITED-SERVICE) or when it has been concluded that no cell is available at the moment (the new substate is NO-CELL-AVAILABLE).

This substate is not applicable to non-3GPP access.

5.1.3.2.1.3.6 5GMM-DEREGISTERED.NO-SUPI

The substate 5GMM-DEREGISTERED.NO-SUPI is chosen in the UE, if the N1 mode is enabled and the UE has no valid subscriber data available (USIM not available, the USIM is considered invalid by the UE or an entry of the "list of subscriber data" with the SNPN identity of the SNPN is considered invalid by the UE).

5.1.3.2.1.3.7 5GMM-DEREGISTERED.NO-CELL-AVAILABLE

No 5G cell can be selected. This substate is entered after a first intensive search failed when in substate 5GMM-DEREGISTERED.PLMN-SEARCH. Cells are searched for at a low rhythm. No 5GS services are offered.

This substate is not applicable to non-3GPP access.

5.1.3.2.1.3.8 5GMM-DEREGISTERED.eCALL-INACTIVE

The substate 5GMM-DEREGISTERED.eCALL-INACTIVE is chosen in the UE when:

- a) the UE is configured for eCall only mode as specified in 3GPP TS 31.102 [22];
- b) timer T3444 and timer T3445 have expired or are not running;
- c) a PLMN has been selected as specified in 3GPP TS 23.122 [5];
- d) the UE does not need to perform an eCall over IMS; and
- e) the UE does not need to perform a call to a non-emergency MSISDN or URI for test or terminal reconfiguration service.

In this substate, the UE shall not initiate any signalling towards the network, except to originate an eCall over IMS, or a call to a non-emergency MSISDN or URI for test or terminal reconfiguration service.

This substate is not applicable to non-3GPP access.

5.1.3.2.1.3.9 5GMM-DEREGISTERED.INITIAL-REGISTRATION-NEEDED

Valid subscriber data are available for the UE and for some reason a registration procedure for initial registration has to be performed as soon as possible. This substate can be entered if the access is barred due to unified access control, (see subclause 4.5.4) or if the network rejects the N1 NAS signalling connection establishment.

5.1.3.2.1.4 Substates of state 5GMM-REGISTERED

5.1.3.2.1.4.1 General

The state 5GMM-REGISTERED is subdivided into a number of substates as described in this subclause. The following substates are not applicable to non-3GPP access:

- a) 5GMM-REGISTERED.PLMN-SEARCH:
- b) 5GMM-REGISTERED.NON-ALLOWED-SERVICE; and
- c) 5GMM-REGISTERED.NO-CELL-AVAILABLE.

5.1.3.2.1.4.2 5GMM-REGISTERED.NORMAL-SERVICE

The substate 5GMM-REGISTERED.NORMAL-SERVICE is chosen by the UE as the primary substate when the UE enters the state 5GMM-REGISTERED, and:

- for 3GPP access, the cell the UE selected is known to be in an allowed area (see subclause 5.3.5.2); or
- for wireline access, the wireline access service area restrictions are not enforced.

5.1.3.2.1.4.3 5GMM-REGISTERED.NON-ALLOWED-SERVICE

The substate 5GMM-REGISTERED.NON-ALLOWED-SERVICE is chosen in the UE, if:

- for 3GPP access, the cell the UE selected is known to be in a non-allowed area (see subclause 5.3.5.2); or
- for wireline access, the wireline access service area restrictions are enforced.

This substate is applicable only to 3GPP access and to wireline access.

5.1.3.2.1.4.4 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE

The substate 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE is chosen by the UE if the registration procedure for mobility and periodic registration update failed due to a missing response from the network, or due to the circumstances described in subclauses 5.3.9, 5.5.1.3.5 and 5.5.1.3.7. No 5GMM procedure except registration procedure for mobility and periodic registration update (i.e. the 5GS registration type IE set to "mobility").

registration updating" or "periodic registration updating" in the REGISTRATION REQUEST message) and deregistration procedure, shall be initiated by the UE in this substate. No data shall be sent or received.

- NOTE 1: The registration procedure for mobility and periodic registration update over non-3GPP access can be triggered by, e.g. the change of S1 UE network capability or renegotiating some parameters.
- NOTE 2: This substate is entered irrespective whether:
 - the UE is camped on a cell which is in the registered PLMN or a PLMN from the list of equivalent PLMNs and the current TAI is not in the list of "allowed tracking areas"; or
 - the current TAI is in the list of "non-allowed tracking areas".

5.1.3.2.1.4.5 5GMM-REGISTERED.LIMITED-SERVICE

The substate 5GMM-REGISTERED.LIMITED-SERVICE is chosen in the UE, if the cell the UE selected is known not to be able to provide normal service.

5.1.3.2.1.4.6 5GMM-REGISTERED.PLMN-SEARCH

The substate 5GMM-REGISTERED.PLMN-SEARCH is chosen in the UE, while the UE is searching for PLMNs or SNPNs.

This substate is not applicable to non-3GPP access.

5.1.3.2.1.4.7 5GMM-REGISTERED.NO-CELL-AVAILABLE

5G coverage has been lost or MICO mode is active in the UE. If MICO mode is active, the UE can deactivate MICO mode at any time by activating the AS layer when the UE needs to send mobile originated signalling or user data. Otherwise, the UE shall not initiate any 5GMM procedure except for cell and PLMN reselection.

This substate is not applicable to non-3GPP access.

5.1.3.2.1.4.8 5GMM-REGISTERED.UPDATE-NEEDED

This state can be entered if the UE has to perform a registration procedure for mobility and periodic registration update but:

- a) the access is barred due to unified access control when in 3GPP access;
- b) the network rejects the N1 NAS signalling connection establishment when in 3GPP access or in non-3GPP access; or
- c) the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers that the resumption of the RRC connection has failed and for access is barred for all categories except categories '0' and '2' as specified in subclause 5.3.1.4.

No 5GMM procedure except:

- a) registration procedure for mobility and periodic registration update;
- b) service request procedure as a response to paging or notification; and
- c) deregistration procedure

shall be initiated by the UE in this substate.

- NOTE: This substate is entered irrespective whether:
 - the UE is camped on a cell which is in the registered PLMN or a PLMN from the list of equivalent PLMNs and the current TAI is not in the list of "allowed tracking areas"; or
 - the current TAI is in the list of "non-allowed tracking areas".

5.1.3.2.2 5GS update status in the UE

In order to describe the detailed UE behaviour, the 5GS update (5U) status pertaining to a specific subscriber is defined.

If the UE is not operating in SNPN access operation mode (see 3GPP TS 23.501 [8]), the 5GS update status is stored in a non-volatile memory in the USIM if the corresponding file is present in the USIM, else in the non-volatile memory in the ME, as described in annex C.

If the UE is operating in SNPN access operation mode, the 5GS update status for each SNPN whose SNPN identity is included in the "list of subscriber data" configured in the ME (see 3GPP TS 23.122 [5]) is stored in the non-volatile memory in the ME as described in annex C.

The 5GS update status value is changed only after the execution of a registration, network-initiated de-registration, 5GS based primary authentication and key agreement, service request, paging procedure or due to change in the current TAI which does not belong to the current registration area while T3346 is running.

5U1: UPDATED

The last registration attempt was successful.

5U2: NOT UPDATED

The last registration or service request attempt failed procedurally, e.g. no response or reject message was received from the AMF.

5U3: ROAMING NOT ALLOWED

The last registration, service request, or registration for mobility or periodic registration update attempt was correctly performed, but the answer from the AMF was negative (because of roaming or subscription restrictions).

5.1.3.2.3 5GMM sublayer states in the network side

5.1.3.2.3.1 General

In the following subclauses, the possible 5GMM sublayer states of the network are described and shown in Figure 5.1.3.2.3.1.1.



NOTE: Not all possible transitions are shown in this figure. Figure 5.1.3.2.3.1.1: 5GMM main states in the network

5.1.3.2.3.2 5GMM-DEREGISTERED

In the state 5GMM-DEREGISTERED, no 5GMM context has been established or the 5GMM context is marked as deregistered. The UE is deregistered. The network may answer to an initial registration procedure initiated by the UE. The network may also answer to a de-registration procedure initiated by the UE.

5.1.3.2.3.3 5GMM-COMMON-PROCEDURE-INITIATED

The network enters the state 5GMM-COMMON-PROCEDURE-INITIATED, after it has started a common 5GMM procedure and is waiting for a response from the UE.

5.1.3.2.3.4 5GMM-REGISTERED

In the state 5GMM-REGISTERED, a 5GMM context has been established. Additionally, one or more PDU session(s) may be established at the network.

5.1.3.2.3.5 5GMM-DEREGISTERED-INITIATED

The network enters the state 5GMM-DEREGISTERED-INITIATED after it has started a de-registration procedure and is waiting for a response from the UE.

5.1.4 Coordination between 5GMM and EMM

5.1.4.1 General

If both 5GMM and EMM are enabled, a UE, operating in single-registration mode, shall maintain one common registration for 5GMM for 3GPP access and EMM.

Coordination between 5GMM for 3GPP access and EMM for a UE, which is capable of N1 mode and S1 mode and operates in dual-registration mode, is not needed, except as specified in subclause 4.8.3.

The coordination between 5GMM for 3GPP access and EMM in subclauses 5.1.4.2 and 5.1.4.3 only applies to the UEs operating in single-registration mode.

Regarding the coordination of "SIM/USIM considered invalid" and "USIM considered invalid for 5GS services" between the various mobility management entities see subclause 5.1.5.

5.1.4.2 Coordination between 5GMM for 3GPP access and EMM with N26 interface

A UE that is not registered shall be in state EMM-DEREGISTERED and state 5GMM-DEREGISTERED for 3GPP access.

In N1 mode, upon successful completion of a registration procedure over 3GPP access, the UE operating in singleregistration mode shall enter substates 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE as described in subclause 5.3.5.2 for 3GPP access and EMM-REGISTERED.NO-CELL-AVAILABLE. The UE shall reset the registration attempt counter for 3GPP access and the attach attempt counter (see 3GPP TS 24.301 [15]).

At inter-system change from S1 mode to N1 mode, the UE shall enter substates 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE as described in subclause 5.3.5.2 for 3GPP accessand EMM-REGISTERED.NO-CELL-AVAILABLE and initiate a registration procedure for mobility and periodic registration update over 3GPP access indicating "mobility registration updating" in the 5GS registration type IE of the REGISTRATION REQUEST message (see subclause 5.5.1.3).

In S1 mode, upon successful completion of an attach or tracking area updating procedure, the UE operating in singleregistration mode shall enter substates 5GMM-REGISTERED.NO-CELL-AVAILABLE for 3GPP access and EMM-REGISTERED.NORMAL-SERVICE. The UE shall reset the registration attempt counter for 3GPP access and the attach attempt counter or tracking area updating attempt counter (see 3GPP TS 24.301 [15]).

At inter-system change from N1 mode to S1 mode when there is no active PDU session for which interworking with EPS is supported as specified in subclause 6.1.4.1, and EMM-REGISTERED without PDN connection is not supported by the UE or the MME, the UE shall enter state 5GMM-DEREGISTERED for 3GPP access and state EMM-

DEREGISTERED and then initiate the EPS attach procedure. If EMM-REGISTERED without PDN connection is supported by the UE and the MME, the UE shall enter substates EMM-REGISTERED.NORMAL-SERVICE and 5GMM-REGISTERED.NO-CELL-AVAILABLE for 3GPP access and initiate a tracking area updating procedure.

At inter-system change from N1 mode to S1 mode when there is at least one active PDU session for which interworking with EPS is supported as specified in subclause 6.1.4.1, the UE shall enter substates EMM-REGISTERED.NORMAL-SERVICE and 5GMM-REGISTERED.NO-CELL-AVAILABLE for 3GPP access and initiate a tracking area updating procedure (see 3GPP TS 24.301 [15]).

5.1.4.3 Coordination between 5GMM for 3GPP access and EMM without N26 interface

A UE operating in the single-registration mode that is not registered over 3GPP access shall be in state EMM-DEREGISTERED and in state 5GMM-DEREGISTERED for 3GPP access.

In N1 mode, upon successful completion of a registration procedure over 3GPP access, the UE operating in the singleregistration mode shall enter substates 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE as described in subclause 5.3.5.2 for 3GPP access and EMM-REGISTERED.NO-CELL-AVAILABLE.

At inter-system change from N1 mode to S1 mode in 5GMM-IDLE mode, the UE shall behave as specified in subclause 4.8.2.3.

In S1 mode, upon successful completion of an attach or tracking area updating procedure, the UE operating in the single-registration mode shall enter substates 5GMM-REGISTERED.NO-CELL-AVAILABLE for 3GPP access and EMM-REGISTERED.NORMAL-SERVICE.

At inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, the UE operating in the single-registration mode shall enter substates EMM-REGISTERED.NO-CELL-AVAILABLE and 5GMM- REGISTERED.NORMAL-SERVICE for 3GPP access and then initiate the registration procedure for mobility and periodic registration update over 3GPP access indicating "mobility registration updating" in the 5GS registration type IE of the REGISTRATION REQUEST message (see subclause 5.5.1.3).

5.1.5 Coordination between 5GMM and GMM

Coordination between 5GMM and GMM states is not required.

Regardless whether the UE is operating in single-registration mode or dual-registration mode,

- a) if the UE considers the SIM/USIM invalid for any of: 3GPP access in N1 mode, S1 mode, A/Gb mode or Iu mode, then it considers the SIM/USIM invalid for all of them; and
- b) if the UE considers the USIM invalid for 5GS services for any of: 3GPP access in N1 mode, S1 mode, A/Gb mode or Iu mode, then it considers the USIM invalid for 5GS services for all of them.

5.2 Behaviour of the UE in state 5GMM-DEREGISTERED and state 5GMM-REGISTERED

5.2.1 General

In this subclause, the detailed behaviour of the UE in the states 5GMM-DEREGISTERED and 5GMM-REGISTERED is described.

5.2.2 UE behaviour in state 5GMM-DEREGISTERED

5.2.2.1 General

The state 5GMM-DEREGISTERED is entered in the UE, when:

- a) the de-registration is performed either by the UE or by the network (see subclause 5.5.2);
- b) the registration request is rejected by the AMF (see subclause 5.5.1.2.5 and 5.5.1.3.5);
- c) the service request is rejected by the AMF (see subclause 5.6.1);
- d) the UE is switched on; or
- e) the UE registered for emergency services is in 5GMM-IDLE mode and its periodic registration update timer expires (see subclause 5.3.7).

In state 5GMM-DEREGISTERED, the UE shall behave according to the substate as explained in subclause 5.2.2.3.

5.2.2.2 Primary substate selection

5.2.2.2.1 Selection of the substate after power on

For a UE configured for eCall only mode as specified in 3GPP TS 31.102 [22], timers T3444 and T3445 are considered to have expired at power on. When the UE is switched on, the substate shall be PLMN-SEARCH if the USIM is available and valid or there are valid entries in the "list of subscriber data". See 3GPP TS 23.122 [5] for further details.

The substate chosen after PLMN-SEARCH, following power on is:

- a) if no cell can be selected, the substate shall be NO-CELL-AVAILABLE;
- b) if the UE is not operating in SNPN access operation mode, and no USIM is present, or the USIM is considered invalid by the UE, the substate shall be NO-SUPI;
- c) if the UE is operating in SNPN access operation mode, and:
 - 1) the selected entry in the "list of subscriber data" does not contain subscription identifier, and no USIM is present, or the USIM is considered invalid by the UE; or
 - 2) no valid entry in the "list of subscriber data" exists;

the substate shall be NO-SUPI;

- d) if a suitable cell has been found, the PLMN or SNPN identity of the cell is not in one of the forbidden PLMN lists, the "permanently forbidden SNPNs" list or the "temporarily forbidden SNPNs" list which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, and the tracking area is not in one of the lists of 5GS forbidden tracking areas, then the substate shall be NORMAL-SERVICE;
- e) if the selected cell is known not to be able to provide normal service, then the UE shall enter the substate LIMITED-SERVICE;
- f) if the UE is in manual network selection mode and no cell of the selected PLMN or SNPN has been found, the UE shall enter the substate NO-CELL-AVAILABLE; and
- g) if the UE is configured for eCall only mode as specified in 3GPP TS 31.102 [22], the substate shall be eCALL-INACTIVE.

5.2.2.3 Detailed description of UE behaviour in state 5GMM-DEREGISTERED

5.2.2.3.1 NORMAL-SERVICE

The UE shall initiate an initial registration procedure if the timer T3346 is not running. If timer T3346 is running, the UE shall initiate an initial registration procedure on the expiry of timer T3346.

The UE may initiate an initial registration procedure for emergency services or for a UE configured for high priority access in the selected PLMN, even if timer T3346 is running.

5.2.2.3.2 LIMITED-SERVICE

The UE shall initiate an initial registration procedure when entering a cell which provides normal service.

The UE may initiate initial registration for emergency services.

5.2.2.3.3 ATTEMPTING-REGISTRATION

The UE in 3GPP access:

- a) shall initiate an initial registration procedure on the expiry of timers T3502, T3511 or T3346;
- b) may initiate an initial registration procedure for emergency services even if timers T3502, T3511 or T3346 are running;
- b1)may initiate an initial registration procedure even if timer T3502, T3346 or T3447 is running, if the UE is a UE configured for high priority access in selected PLMN;
- c) shall initiate an initial registration procedure when entering a new PLMN or SNPN, except
 - i) if timer T3346 is running and the new PLMN is equivalent to the PLMN where the UE started timer T3346;
 - ii) if the PLMN identity of the new cell is in the forbidden PLMN lists or the SNPN identity of the new cell is in the "permanently forbidden SNPNs" list or the "temporarily forbidden SNPNs" list which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription; or
 - iii) if the current TAI is in one of the lists of 5GS forbidden tracking areas;
- d) shall initiate an initial registration procedure when the current TAI has changed, if timer T3346 is not running, the PLMN identity of the new cell is not in one of the forbidden PLMN lists or the SNPN identity of the new cell is in neither the "permanently forbidden SNPNs" list nor the "temporarily forbidden SNPNs" list which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, and the current TAI is not in one of the lists of 5GS forbidden tracking areas;
- e) shall initiate an initial registration procedure if the 5GS update status is set to 5U2 NOT UPDATED, and timers T3511, T3502 and T3346 are not running;
- f) may initiate an initial registration procedure for UE in NB-N1 mode upon receiving a request from upper layers to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17]) if timer T3346 is not already running for "MO exception data" and even if timer T3502 or timer T3511 is running; and
- g) may initiate an initial registration procedure with 5GS registration type IE set to "initial registration" for initiating of an emergency PDU session, upon request of the upper layers to establish the emergency PDU session.

The UE in non-3GPP access:

- a) shall initiate an initial registration procedure on the expiry of timers T3502, T3511 or T3346;
- b) may initiate an initial registration procedure for emergency services even if timers T3502, T3511 or T3346 are running;
- b1)may initiate an initial registration procedure even if timer T3502 or T3346 is running if the UE is a UE configured for high priority access in selected PLMN;
- c) shall initiate an initial registration procedure when entering a new PLMN, except if timer T3346 is running and the new PLMN is equivalent to the PLMN where the UE started timer T3346;
- shall initiate an initial registration procedure if the 5GS update status is set to 5U2 NOT UPDATED, and timers T3511, T3502 and T3346 are not running; and

3GPP TS 24.501 version 17.16.0 Release 17

139

e) may initiate an initial registration procedure with 5GS registration type IE set to "initial registration" for initiating of an emergency PDU session, upon request of the upper layers to establish the emergency PDU session.

5.2.2.3.4 PLMN-SEARCH

The UE shall perform PLMN selection or SNPN selection. If a new PLMN or SNPN is selected, the UE shall reset the registration attempt counter and initiate the registration procedure for initial registration (see subclause 5.5.1.2.2).

If the selected cell in the new PLMN is known not to be able to provide normal service, the UE may initiate the registration procedure for initial registration for emergency services.

5.2.2.3.5 NO-SUPI

The UE may initiate the registration procedure for initial registration for emergency services.

5.2.2.3.6 NO-CELL-AVAILABLE

The UE shall perform cell selection and choose an appropriate substate when a cell is found.

5.2.2.3.7 eCALL-INACTIVE

The UE camps on a suitable cell or an acceptable cell in a PLMN selected as specified in 3GPP TS 23.122 [5] but initiates no 5GMM signalling with the network and ignores any paging requests.

The UE shall leave substate 5GMM-DEREGISTERED.eCALL-INACTIVE state only when one of the following events occur:

- a) if the USIM is removed, the UE enters substate 5GMM-DEREGISTERED.NO-SUPI;
- b) if coverage is lost, the UE enters substate 5GMM-DEREGISTERED.PLMN-SEARCH;
- c) if the UE is deactivated (e.g. powered off) by the user, the UE enters state 5GMM-NULL;
- d) if the UE receives a request from upper layers to establish an eCall over IMS, the UE enters state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION. The UE then uses the relevant 5GMM and 5GSM procedures to establish the eCall over IMS at the earliest opportunity; or
- e) if the UE receives a request from upper layers to establish a call to an HPLMN designated non-emergency MSISDN or URI for test or terminal reconfiguration service, the UE enters state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION. Once the registration procedure is completed, the UE uses the relevant 5GMM and 5GSM procedures to establish the non-emergency call.

5.2.2.3.8 INITIAL-REGISTRATION-NEEDED

The UE shall initiate the initial registration procedure, if still needed, as soon as the access is allowed in the selected cell for the UE.

The UE may initiate registration procedure for emergency services.

5.2.2.4 Substate when back to state 5GMM-DEREGISTERED from another 5GMM state

When returning to state 5GMM-DEREGISTERED, the UE shall select a cell as specified in 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

The substate depends on the result of the cell selection procedure, the outcome of the previously performed 5GMM specific procedures, on the 5GS update status of the UE, on the tracking area data stored in the UE, on the presence of the USIM, on the UE configuration and on the reason for moving to 5GMM-DEREGISTERED:

- a) If no cell has been found, the substate is NO-CELL-AVAILABLE, until a cell is found;
- b) If no USIM is present or if the inserted USIM is considered invalid by the UE, the substate shall be NO-SUPI;

- c) If a suitable cell has been found and the PLMN or tracking area is not in one of the forbidden lists, the substate shall be NORMAL-SERVICE;
- d) If an initial registration shall be performed (e.g. network-requested re-registration), the substate shall be ATTEMPTING-REGISTRATION;
- e) If a PLMN reselection or SNPN reselection (according to 3GPP TS 23.122 [5]) is needed, the substate shall be PLMN-SEARCH;
- f) If the selected cell is known not to be able to provide normal service, the substate shall be LIMITED-SERVICE; and
- g) If the UE is configured for eCall only mode as specified in 3GPP TS 31.102 [22], T3444 and T3445 have expired or are not running, and substate PLMN-SEARCH is not required, the substate shall be eCALL-INACTIVE.

5.2.3 UE behaviour in state 5GMM-REGISTERED

5.2.3.1 General

The state 5GMM-REGISTERED is entered at the UE, when the initial registration procedure is performed by the UE (see subclause 5.5.1.2.2).

In state 5GMM-REGISTERED, the UE shall behave according to the substate as explained in subclause 5.2.3.2.

5.2.3.2 Detailed description of UE behaviour in state 5GMM-REGISTERED

5.2.3.2.1 NORMAL-SERVICE

The UE:

- a) shall initiate the mobility or the periodic registration update procedure (according to conditions given in subclause 5.5.1.3.2), except that the periodic registration update procedure shall not be initiated over non-3GPP access;
- b) shall initiate the service request procedure (according to conditions given in subclause 5.6.1);
- c) shall respond to paging;
- NOTE 1: Paging is not supported over non-3GPP access.
- NOTE 2: As an implementation option, the MUSIM UE is allowed to not respond to paging based on the information available in the paging message, e.g. voice service indication.
- d) if configured for eCall only mode as specified in 3GPP TS 31.102 [22], shall perform the eCall inactivity procedure at expiry of timer T3444 or timer T3445 (see subclause 5.5.3);
- e) shall initiate a registration procedure for mobility and periodic registration update on the expiry of timer T3511; and
- f) if acting as a 5G ProSe UE-to-network relay UE as specified in 3GPP 24.554 [19E], shall initiate the authentication and key agreement procedure (according to the conditions given in subclause 5.5.4).

5.2.3.2.2 NON-ALLOWED-SERVICE

The UE shall behave as specified in subclause 5.3.5.

The UE in 5GMM-REGISTERED.NON-ALLOWED-SERVICE substate, if configured for eCall only mode as specified in 3GPP TS 31.102 [22], shall perform the eCall inactivity procedure at expiry of timer T3444 or timer T3445 (see subclause 5.5.3).

5.2.3.2.3 ATTEMPTING-REGISTRATION-UPDATE

The UE in 3GPP access:

- a) shall not send any user data;
- b) shall initiate a registration procedure for mobility and periodic registration update on the expiry of timers T3502, T3511 or T3346;
- c) shall initiate a registration procedure for mobility and periodic registration update when entering a new PLMN, if timer T3346 is running and the new PLMN is not equivalent to the PLMN where the UE started timer T3346, the PLMN identity of the new cell is not in the forbidden PLMN lists, and the current TAI is not in one of the lists of 5GS forbidden tracking areas;
- d) shall initiate a registration procedure for mobility and periodic registration update when the current TAI has changed, if timer T3346 is not running, the PLMN identity of the new cell is not in one of the forbidden PLMN lists or the SNPN identity of the new cell is in neither the "permanently forbidden SNPNs" list nor the "temporarily forbidden SNPNs" list which are, if the UE supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, and the current TAI is not in one of the lists of 5GS forbidden tracking areas;
- e) may initiate a registration procedure for mobility and periodic registration update upon request of the upper layers to establish an emergency PDU session;
- e1) may initiate a registration procedure for mobility and periodic registration update upon request of the upper layers to establish a PDU session, even if timer T3502, T3346 or T3447 is running, if the UE is a UE configured for high priority access in the selected PLMN;
- f) may perform de-registration locally and initiate a registration procedure for initial registration for emergency services even if timer T3346 is running;
- g) shall initiate registration procedure for mobility and periodic registration update upon reception of paging, or upon reception of NOTIFICATION message with access type indicating 3GPP access;
- NOTE: As an implementation option, the MUSIM UE is allowed to not respond to paging based on the information available in the paging message, e.g. voice service indication.
- h) may initiate a registration procedure for mobility and periodic registration update upon request for an MMTEL voice call, MMTEL video call, or an MO IMS registration related signalling from the upper layers, and none of the following conditions is met:
 - 1) timer T3346 is running;
 - 2) the UE has stored a list of "non-allowed tracking areas" and the current TAI is in the list of "non-allowed tracking areas"; or
 - the UE has stored a list of "allowed tracking areas", the UE is camped on a cell which is in the registered PLMN or a PLMN from the list of equivalent PLMNs, and the current TAI is not in the list of "allowed tracking areas";
- i) shall initiate a registration procedure for mobility and periodic registration update if the 5GS update status is set to 5U2 NOT UPDATED, and timers T3511, T3502 and T3346 are not running;
- j) if configured for eCall only mode as specified in 3GPP TS 31.102 [22], shall perform the eCall inactivity procedure at expiry of timer T3444 or timer T3445 (see subclause 5.5.3);
- k) may initiate a registration procedure for mobility and periodic registration update for UE in NB-N1 mode upon receiving a request from upper layers to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17]) if timer T3346 is not already running for "MO exception data" and even if timer T3502 or timer T3511 is running; and
- l) shall not initiate the de-registration signalling procedure unless the current TAI is part of the TAI list.

The UE in non-3GPP access:

- a) shall not send any user data;
- b) shall initiate the registration procedure for mobility and periodic registration update on the expiry of timers T3502, T3511 or T3346;
- c) may initiate a registration procedure for mobility registration update upon request of the upper layers to establish an emergency PDU session;
- c1) may initiate a registration procedure for mobility and periodic registration update upon request of the upper layers to establish a PDU session, if the UE is a UE configured for high priority access in the selected PLMN even if timer T3346 or T3502 is running;
- d) may perform de-registration locally and initiate a registration procedure for initial registration for emergency services even if timer T3346 is running;
- e) may initiate a registration procedure for mobility and periodic registration update upon request for an MMTEL voice call, MMTEL video call, or an MO IMS registration related signalling from the upper layers, if timer T3346 is not running;
- f) shall initiate a registration procedure for mobility and periodic registration update if the 5GS update status is set to 5U2 NOT UPDATED, and timers T3511, T3502 and T3346 are not running; and
- g) shall not initiate the de-registration signalling procedure unless timer T3346 is running.

5.2.3.2.4 LIMITED-SERVICE

The UE:

- a) shall perform cell selection/reselection;
- b) may perform de-registration locally and initiate an initial registration for emergency services; and
- c) if configured for eCall only mode as specified in 3GPP TS 31.102 [22], shall perform the eCall inactivity procedure at expiry of timer T3444 or timer T3445 (see subclause 5.5.3).

5.2.3.2.5 PLMN-SEARCH

The UE shall perform PLMN selection or SNPN selection. If a new PLMN is selected, the UE shall reset the registration attempt counter and initiate a registration procedure for mobility and periodic registration update (see subclause 5.5.1.3). If a new SNPN is selected, the UE shall reset the registration attempt counter, may perform deregistration locally and shall initiate a registration procedure for initial registration (see subclause 5.5.1.2.2).

If the selected cell in the new PLMN is known not to be able to provide normal service, the UE may perform deregistration locally and initiate an initial registration for emergency services.

5.2.3.2.6 NO-CELL-AVAILABLE

The UE shall perform cell selection and choose an appropriate substate when a cell is found.

5.2.3.2.7 UPDATE-NEEDED

The UE:

- a) shall not send any user data;
- b) shall not send signalling information, unless it is a service request as a response to paging or to initiate signalling for emergency services or emergency services fallback;
- c) shall perform cell selection/reselection;
- d) shall enter the appropriate new substate as soon as the lower layers indicate that the barring is alleviated for the access category with which the access attempt for the registration procedure for mobility and periodic registration update was associated; and

e) if configured for eCall only mode as specified in 3GPP TS 31.102 [22], shall perform the eCall inactivity procedure at expiry of timer T3444 or T3445 (see subclause 5.5.3).

5.3 General on elementary 5GMM procedures

5.3.1 5GMM modes and N1 NAS signalling connection

5.3.1.1 Establishment of the N1 NAS signalling connection

When the UE is in 5GMM-IDLE mode over 3GPP access and needs to transmit an initial NAS message, the UE shall request the lower layer to establish an RRC connection. Upon indication from the lower layers that the RRC connection has been established, the UE shall consider that the N1 NAS signalling connection over 3GPP access is established and enter 5GMM-CONNECTED mode over 3GPP access.

When the UE is in 5GMM-IDLE mode over non-3GPP access, and the UE receives an indication from the lower layers of non-3GPP access, that the access stratum connection is established between the UE and the network, the UE shall send an initial NAS message, consider the N1 NAS signalling connection is established and enter 5GMM-CONNECTED mode over non-3GPP access.

Initial NAS messages are:

- a) REGISTRATION REQUEST message;
- b) DEREGISTRATION REQUEST message;
- c) SERVICE REQUEST message; and
- d) CONTROL PLANE SERVICE REQUEST.

If the UE is capable of both N1 mode and S1 mode and lower layers provide an indication that the current E-UTRA cell is connected to both EPC and 5GCN, for the routing of the REGISTRATION REQUEST message during the initial registration procedure to the appropriate core network (EPC or 5GCN), the UE NAS provides the lower layers with the selected core network type information.

For the routing of the initial NAS message to the appropriate AMF, if the UE holds a 5G-GUTI or 4G-GUTI, the UE NAS provides the lower layers with the UE identity according to the following rules:

- a) if the registration procedure for mobility and periodic update was triggered due to the last CONFIGURATION UPDATE COMMAND message containing the Configuration update indication IE with the Registration bit set to "registration requested" and including:
 - 1) no other parameters;
 - 2) one or both of the Allowed NSSAI IE and the Configured NSSAI IE; or
 - 3) the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed";

the UE NAS shall not provide the lower layers with the 5G-S-TMSI or the registered GUAMI;

- b) if the service request procedure was initiated over non-3GPP access, the UE NAS shall provide the lower layers with the registered GUAMI, but shall not provide the lower layers with the 5G-S-TMSI;
- c) if the initial NAS message other than the SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message was initiated over untrusted non-3GPP access, the UE NAS shall provide the lower layers with the GUAMI of the 5G-GUTI that the UE NAS has selected as specified in the subclause 5.5.1.2.2 and 5.5.1.3.2, but shall not provide the lower layers with the 5G-S-TMSI;

if the initial NAS message other than the SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message was initiated over trusted non-3GPP access, the UE NAS shall provide the lower layers with the 5G-GUTI, if available, otherwise shall provide the lower layers with the SUCI;
if the UE is the 5G-RG and the initial NAS message other than the SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message was initiated over wireline access, the UE NAS shall provide the lower layers with the GUAMI of the 5G-GUTI that the UE NAS has selected as specified in the subclause 5.5.1.2.2 and 5.5.1.3.2, if available, otherwise shall not provide the lower layers with any UE identity;

- d) if the UE does not hold a 5G-GUTI that was previously assigned by the same PLMN with which the UE is performing the registration procedure and if:
 - the UE operating in the single-registration mode performs a registration procedure for mobility and periodic update indicating "mobility registration updating" following an inter-system change from S1 mode to N1 mode; or
 - 2) the UE which was previously registered in S1 mode before entering state EMM-DEREGISTERED, performs an initial registration procedure, the UE has received the interworking without N26 interface indicator set to "interworking without N26 interface not supported" from the network, and the UE holds a 4G-GUTI;

then the UE NAS provides the lower layers with a GUAMI part of the 5G-GUTI mapped from 4G-GUTI as specified in 3GPP TS 23.003 [4] with an indication that the GUAMI is mapped from EPS; or

- e) otherwise:
 - 1) if the tracking area of the current cell is in the registration area, the UE NAS shall provide the lower layers with the 5G-S-TMSI, but shall not provide the registered GUAMI to the lower layers; or
 - 2) if the tracking area of the current cell is not in the registration area, the UE NAS shall provide the lower layers with the GUAMI of the 5G-GUTI that the UE NAS has selected as specified in the subclauses 5.5.1.2.2 and 5.5.1.3.2, but shall not provide the lower layers with the 5G-S-TMSI.

For 3GPP access and untrusted non-3GPP access, if the UE does not hold a 5G-GUTI and the UE does not hold a 4G-GUTI, the UE NAS does not provide the lower layers with the 5G-S-TMSI or the registered GUAMI. For trusted non-3GPP access, if the UE does not hold a 5G-GUTI and the UE does not hold a 4G-GUTI, the UE NAS provides the lower layers with the SUCI.

For 3GPP access, if a UE operating as an IAB-node performs a registration procedure or service request procedure (see 3GPP TS 23.501 [8]), the UE NAS shall indicate to the lower layers that the establishment of the NAS signalling connection is for a UE operating as an IAB-node.

The UE NAS also provides the lower layers with the identity of the selected PLMN (see 3GPP TS 38.331 [30]) if the UE is not operating in SNPN access operation mode. If the UE is operating in SNPN access operation mode, the UE NAS provides the lower layers with the SNPN identity of the selected SNPN. In a shared network, the UE shall choose one of the PLMN identity(ies) or SNPN identity(ies) as specified in 3GPP TS 23.122 [5].

The UE NAS layer may provide the lower layers with an NSSAI as specified in subclause 4.6.2.3.

If the UE performs initial registration for onboarding services in SNPN or is registered for onboarding services in SNPN, the UE NAS layer shall provide the lower layers with an indication that the connection is for onboarding.

5.3.1.2 Re-establishment of the N1 NAS signalling connection

When the UE in 5GMM-CONNECTED mode over 3GPP access receives a fallback indication from lower layers, and the UE has no pending NAS procedure and no pending uplink user data for PDU session(s) with user-plane resources already established, the UE shall:

- a) enter 5GMM-IDLE mode; and
- b) initiate the registration procedure for mobility and periodic registration update and include the Uplink data status IE in the REGISTRATION REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclause 5.5.1.3 for further details).

When the UE in 5GMM-CONNECTED mode over 3GPP access receives a fallback indication from lower layers, and the UE has pending uplink user data for PDU session(s) with user-plane resources already established but no pending NAS procedure, the UE shall:

a) enter 5GMM-IDLE mode; and

b) initiate the service request procedure and include the Uplink data status IE in the SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication (see subclause 5.6.1 for further details).

When the UE in 5GMM-CONNECTED mode over 3GPP access receives a fallback indication from lower layers, and the UE has a pending registration procedure, a service request procedure, or a de-registration procedure, the UE shall:

- a) enter 5GMM-IDLE mode;
- b) proceed with the pending procedure; and
- c) if the pending procedure is a service request or registration procedure and the SERVICE REQUEST message, the CONTROL PLANE SERVICE REQUEST message or the REGISTRATION REQUEST message does not include UE request type IE with Request type value set to "NAS signalling connection release", the UE shall include the Uplink data status IE in the SERVICE REQUEST message, or in the REGISTRATION REQUEST message, indicating the PDU session(s) for which user-plane resources were not active prior to receiving a fallback indication from the lower layers and the UE has pending user data to be sent over 3GPP access, if any, and the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclauses 5.5.1.3 and 5.6.1 for further details).

When the UE in 5GMM-CONNECTED mode over 3GPP access receives a fallback indication from lower layers, and the UE has a pending NAS procedure other than a registration procedure, a service request procedure, or a deregistration procedure, the UE shall:

- a) enter 5GMM-IDLE mode;
- b) initiate the service request procedure and include the Uplink data status IE in the SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclause 5.6.1 for further details); and
- c) upon successful service request procedure completion, proceed with any pending procedure.

When the UE in 5GMM-CONNECTED mode over 3GPP access receives a fallback indication from lower layers, and the UE has no pending NAS procedure and no pending uplink user data for PDU session(s) with user-plane resources already established, and the UE was using network resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5 (see 3GPP TS 23.304 [6E]), the UE shall:

- a) enter 5GMM-IDLE mode; and
- b) initiate the service request procedure and include the Uplink data status IE in the SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclause 5.6.1 for further details).

The cases above apply when the UE is in an allowed area or when the UE is not in a non-allowed area.

When the UE:

- a) is in a non-allowed area or is not in an allowed area;
- b) is in 5GMM-CONNECTED mode over 3GPP access;
- c) receives a fallback indication from lower layers; and
- d) does not have signalling pending,

the UE shall:

- a) enter 5GMM-IDLE mode; and
- b) initiate the registration procedure for mobility and periodic registration update. The UE shall not include the Uplink data status IE in the REGISTRATION REQUEST message except if the PDU session for which userplane resources were active is an emergency PDU session, or if the UE is configured for high priority access in the selected PLMN.

In the above cases when the UE receives a fallback indication from lower layers, if the UE is in non-allowed area or not in allowed area, the UE shall behave as specified in subclause 5.3.5.

5.3.1.3 Release of the N1 NAS signalling connection

The signalling procedure for the release of the N1 NAS signalling connection is initiated by the network.

In N1 mode, upon indication from lower layers that the access stratum connection has been released, the UE shall enter 5GMM-IDLE mode and consider the N1 NAS signalling connection released.

If the UE in 3GPP access is configured for eCall only mode as specified in 3GPP TS 31.102 [22] then:

- if the N1 NAS signalling connection that was released had been established for eCall over IMS, the UE shall start timer T3444; and
- if the N1 NAS signalling connection that was released had been established for a call to an HPLMN designated non-emergency MSISDN or URI for test or terminal reconfiguration service, the UE shall start timer T3445.

The UE shall start the timer T3447 if not already running when the N1 NAS signalling connection is released as specified in subclause 5.3.17.

To allow the network to release the N1 NAS signalling connection, the UE:

- a) shall start the timer T3540 if the UE receives any of the 5GMM cause values #7, #11, #12, #13, #15, #27, #31, #62, #72, #73, #74, #75, #76 and the UE does not consider the received 5GMM cause value as abnormal case as specified in subclauses 5.5.1.2.7, 5.5.1.3.7 and 5.5.2.3.4;
- b) shall start the timer T3540 for a UE in 3GPP access if:
 - 1) the UE receives a REGISTRATION ACCEPT message which does not include a Pending NSSAI IE or UE radio capability ID deletion indication IE;
 - 2) the UE has set the Follow-on request indicator to "No follow-on request pending" in the REGISTRATION REQUEST message;
 - the UE has not included the Uplink data status IE in the REGISTRATION REQUEST message, or the UE has included the Uplink data status IE in the REGISTRATION REQUEST message but the REGISTRATION ACCEPT message indicates that no user-plane resources of any PDU sessions are to be reestablished;
 - 4) the UE has not included the Allowed PDU session status IE or has included the Allowed PDU session status IE indicating there is no PDU session(s) for which the UE allowed the user-plane resource to be re-established over 3GPP access in the REGISTRATION REQUEST message, or the UE has included the Allowed PDU session status IE in the REGISTRATION REQUEST message but the REGISTRATION ACCEPT message does not indicate that any user-plane resources of any PDU sessions are to be re-established;
 - 5) the registration procedure has been initiated in 5GMM-IDLE mode, or the UE has set Request type to "NAS signalling connection release" in the UE request type IE in the REGISTRATION REQUEST message and the N1 NAS signalling connection release bit is set to "N1 NAS signalling connection release supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message;
 - 6) the user-plane resources for PDU sessions have not been set up, except for the case the UE has set Request type to "NAS signalling connection release" in the UE request type IE in the REGISTRATION REQUEST message and the N1 NAS signalling connection release bit is set to "N1 NAS signalling connection release supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message;
 - the UE need not request resources for V2X communication over PC5 reference point (see 3GPP TS 23.287 [6C]); and
 - the UE need not request resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5 (see 3GPP TS 23.304 [6E]);
- NOTE 1: The lower layers indicate when the user-plane resources for PDU sessions are successfully established or released.
- c) shall start the timer T3540 if the UE receives a REGISTRATION REJECT message indicating:

the 5GMM cause value #9 or #10;

d) shall start the timer T3540 if the UE receives a SERVICE REJECT message indicating:

the 5GMM cause value #9, #10; or

the 5GMM cause value #28 and with no emergency PDU session has been established;

- e) shall start the timer T3540 if:
 - 1) the UE receives a CONFIGURATION UPDATE COMMAND message containing the Configuration update indication IE with the Registration bit set to "registration requested" and with:
 - i) either new allowed NSSAI information or new configured NSSAI information or both included;
 - ii) the network slicing subscription change indication; or
 - iii) no other parameters;
 - 2) the user-plane resources for PDU sessions have not been set up; and
- f) shall start the timer T3540 for a UE in 3GPP access if:
 - 1) the UE receives a SERVICE ACCEPT message;
 - 2) for the case that the UE sent the:
 - SERVICE REQUEST message, the UE did not set the Service type IE to "signalling" or "high priority access", the UE has not included the Uplink data status IE in the SERVICE REQUEST message, or the UE has included the Uplink data status IE in the SERVICE REQUEST message but the SERVICE ACCEPT message indicates that no user-plane resources of any PDU sessions are to be re-established; or
 - ii) CONTROL PLANE SERVICE REQUEST message, the UE did not set the Control plane service type IE to "emergency services fallback", the UE has not included the Uplink data status IE in the CONTROL PLANE SERVICE REQUEST message, or the UE has included the Uplink data status IE in the CONTROL PLANE SERVICE REQUEST message but the SERVICE ACCEPT message indicates that no user-plane resources of any PDU sessions are to be re-established;
 - 3) the UE has not included the Allowed PDU session status IE or has included the Allowed PDU session status IE indicating there is no PDU session(s) for which the UE allowed the user-plane resource to be re-established over 3GPP access in the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message, or the UE has included the Allowed PDU session status IE in the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message but the SERVICE ACCEPT message does not indicate that any user-plane resources of any PDU sessions are to be re-established;
 - 4) the service request procedure has been initiated in 5GMM-IDLE mode;
 - 5) the user-plane resources for PDU sessions have not been set up;
 - 6) the UE need not request resources for V2X communication over PC5 reference point (see 3GPP TS 23.287 [6C]); and
 - 7) the UE need not request resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5 (see 3GPP TS 23.304 [6E]);
- NOTE 2: The lower layers indicate when the user-plane resources for PDU sessions are successfully established or released.
- g) may start the timer T3540 if the UE receives any of the 5GMM cause values #3 or #6 or if it receives an AUTHENTICATION REJECT message;
- h) shall start the timer T3540 upon completion of the configuration update procedure or the registration procedure if the UE does not have an emergency PDU session and:
 - the UE received a CONFIGURATION UPDATE COMMAND message or a REGISTRATION ACCEPT message while camping on a CAG cell and the entry for the current PLMN in the received "CAG information list" does not include any of the CAG-ID(s) supported by the current CAG cell;

- 2) the UE received a CONFIGURATION UPDATE COMMAND message or a REGISTRATION ACCEPT message while camping on a non-CAG cell and the entry for the current PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells"; or
- 3) the UE received a CONFIGURATION UPDATE COMMAND message while camping on a CAG cell and the entry for the current PLMN in not included in the received "CAG information list"; or
- i) shall start the timer T3540 for a UE in 3GPP access if:
 - 1) the UE receives a SERVICE ACCEPT message; and
 - 2) the UE:
 - has set Request type to "NAS signalling connection release" in the UE request type IE in the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message; or
 - has set Request type to "Rejection of paging" in the UE request type IE in the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message and the UE receives a CONFIGURATION UPDATE COMMAND message;
- j) shall start the timer T3540 if the UE receives the 5GMM cause value #22 along with a T3346 value, and the value indicates that the timer T3346 is neither zero nor deactivated and the NAS procedure has been initiated in 5GMM-IDLE mode; or
- k) shall start the timer T3540 if the UE receives a DEREGISTRATION ACCEPT message.

Upon expiry of T3540,

- in cases a), b), f), g), h), i), j) and k) the UE shall locally release the established N1 NAS signalling connection;
- in cases c) and d) the UE shall locally release the established N1 NAS signalling connection and the UE shall initiate the registration procedure as described in subclause 5.5.1.3 or 5.6.1.5; or
- in case e), the UE shall locally release the established N1 NAS signalling connection and perform a new registration procedure as specified in subclause 5.5.1.3.2.

In case a),

- upon receiving a request from the upper layers to perform emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall stop timer T3540 and shall locally release the N1 NAS signalling connection, before proceeding as specified in subclause 5.5.1.

In case b) and f),

- upon an indication from the lower layers that the user-plane resources for PDU sessions are set up, the UE shall stop timer T3540 and may send uplink signalling via the existing N1 NAS signalling connection or user data via user plane. If the uplink signalling is associated with emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall stop timer T3540 and send the uplink signalling via the existing N1 NAS signalling connection;

In case b), f) and i),

- upon receipt of a DEREGISTRATION REQUEST message, the UE shall stop timer T3540 and respond to the network-initiated de-registration request via the existing N1 NAS signalling connection as specified in subclause 5.5.2.3;
- upon receipt of a message of a network-initiated 5GMM common procedure, the UE shall stop timer T3540 and respond to the network-initiated 5GMM common procedure via the existing N1 NAS signalling connection as specified in subclause 5.4;
- if there is no user-plane resources established for PDU sessions, upon receiving a request from the upper layers to perform emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall stop timer T3540 and shall locally release the N1 NAS signalling connection, before proceeding as specified in subclause 5.6.1;

- if there is no user-plane resources established for PDU sessions, upon receiving a request from the upper layers to perform services other than emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall wait for the local release of the established N1 NAS signalling connection upon expiry of timer T3540 or wait for timer T3540 being stopped, before initiating NAS signalling;
- upon receipt of a DL NAS TRANSPORT message, the UE shall stop timer T3540 and may send uplink signalling via the existing N1 NAS signalling connection; or
- upon initiation of registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.2.7 for cases h), i), j), subclause 5.5.1.3.7 for cases j), k) or subclause 5.5.1.3.2 for case a), the UE shall stop timer T3540.

In case c) and d),

- upon an indication from the lower layers that the access stratum connection has been released, the UE shall stop timer T3540 and perform a new registration procedure as specified in subclause 5.5.1.3.5 or 5.6.1.5.
- upon receiving a request from the upper layers to perform emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall stop timer T3540 and shall locally release the N1 NAS signalling connection, before proceeding as specified in subclause 5.5.1.

In case e),

- upon an indication from the lower layers that the access stratum connection has been released, the UE shall stop timer T3540 and perform a new registration procedure as specified in subclause 5.5.1.3.2.
- upon an indication from the lower layers that the user-plane resources for PDU sessions are set up, the UE shall stop timer T3540 and may send user data via user plane.

NOTE 3: In this case, the new registration procedure is performed when the UE moves to the 5GMM-IDLE mode.

- upon receiving a request from the upper layers to perform emergency services fallback only for a UE in 3GPP access or establishing an emergency PDU session, the UE shall stop timer T3540 and shall locally release the N1 NAS signalling connection, before proceeding as specified in subclause 5.5.1.

If the UE had set the Follow-on request indicator to "Follow-on request pending" in the REGISTRATION REQUEST message due to pending uplink signalling but cannot send the pending signalling due to new service area restrictions received or due to network not supporting the feature as indicated in the REGISTRATION ACCEPT message (for example UE set the "Follow-on request pending" to send SMS over NAS but AMF notified "SMS over NAS not allowed") and if there is no further pending data or signalling and user plane resources have not been set up, the UE may locally release the established N1 NAS signalling connection upon completion of the registration procedure.

NOTE 4: the UE is allowed to inform the lower layers that there is no 5GMM or 5GSM messages need to be sent over non-3GPP access if the UE receives a REGISTRATION REJECT message over non-3GPP access or a SERVICE REJECT message over non-3GPP access.

If the timer T3540 is not running when the UE enters state 5GMM-DEREGISTERED.PLMN-SEARCH or 5GMM-REGISTERED.PLMN-SEARCH, the UE may locally release the N1 NAS signalling connection.

5.3.1.4 5GMM-CONNECTED mode with RRC inactive indication

This subclause is only applicable for UE's 5GMM mode over 3GPP access. The 5GMM-CONNECTED mode with RRC inactive indication is not supported when the UE is in NB-N1 mode.

The UE is in 5GMM-CONNECTED mode with RRC inactive indication when the UE is in:

- a) 5GMM-CONNECTED mode over 3GPP access at the NAS layer; and
- b) RRC_INACTIVE state at the AS layer (see 3GPP TS 38.300 [27]).

Unless stated otherwise, the UE behaviour in 5GMM-CONNECTED mode with RRC inactive indication follows the UE behaviour in 5GMM-CONNECTED over 3GPP access, except that:

a) the UE shall apply the mobility restrictions; and

b) the UE shall perform the PLMN selection procedures

as in 5GMM-IDLE mode over 3GPP access.

The UE shall transition from 5GMM-CONNECTED mode over 3GPP access to 5GMM-CONNECTED mode with RRC inactive indication upon receiving an indication from the lower layers that the RRC connection has been suspended.

NOTE 1: Any pending procedure or uplink data packet when receiving an indication from the lower layers that the RRC connection has been suspended, triggers a request to the lower layers to transition to RRC_CONNECTED state. This is also the case when the pending procedure or uplink data packet triggered a previous request to the lower layers to transition to RRC_CONNECTED state.

Upon:

- a trigger of a procedure which requires sending of a NAS message different from a REGISTRATION REQUEST message with the NG-RAN-RCU bit of the 5GS update type IE set to "UE radio capability update needed";
- b) an uplink user data packet to be sent for a PDU session with suspended user-plane resources;
- c) a trigger to request resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5; or
- d) a trigger to request resources for V2X communication over PC5 (see 3GPP TS 23.287 [6C]);

the UE in 5GMM-CONNECTED mode with RRC inactive indication over 3GPP access shall request the lower layers to transition to RRC_CONNECTED state (see 3GPP TS 38.300 [27]).

NOTE 2: If the UE supports Small Data Transmission (SDT) (see 3GPP TS 38.300 [27]), the following applies:

- a) if the UE due to pending uplink NAS messages or user data packets is requesting the lower layers to transition to RRC_CONNECTED state, but has not received a response from the lower layers, the UE can send the pending NAS messages or user data packets to the lower layers, and can receive multiple downlink NAS messages or multiple downlink user data packets from the lower layers while the UE remains in 5GMM-CONNECTED mode with RRC inactive indication over 3GPP access (i.e., without transitioning to 5GMM-CONNECTED mode). When the NAS layer triggers the transmission of pending uplink NAS messages or user data packets, and if the SDT is ongoing, the NAS layer will receive the response from the lower layers only after the SDT session has completed or failed;
- b) the NAS layer is not aware of the classification of NAS messages or the user data packets as belonging to the SDT session at the lower layers; and
- c) the setting of access category and the RRC establishment cause indicated to the lower layers when sending the pending uplink user data packets while the UE remains in 5GMM-CONNECTED mode with RRC inactive indication, is left to implementation.

Upon a trigger to send a REGISTRATION REQUEST message with the NG-RAN-RCU bit of the 5GS update type IE set to "UE radio capability update needed", the UE in 5GMM-CONNECTED mode with RRC inactive indication shall move to 5GMM-IDLE mode over 3GPP access and proceed with the registration procedure for mobility and periodic registration as specified in subclause 5.5.1.3.2.

The UE shall transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-CONNECTED mode over 3GPP access upon receiving an indication from the lower layers that the UE has transitioned to RRC_CONNECTED state (see 3GPP TS 38.300 [27]).

NOTE 3: The AMF can be aware of the transition between 5GMM-CONNECTED mode and 5GMM-CONNECTED mode with RRC inactive indication for a UE (see 3GPP TS 23.502 [9]).

The UE shall trigger a transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode upon selection of a PLMN that is not an equivalent PLMN to the registered PLMN. The UE shall not trigger a transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode upon entering a new PLMN which is in the list of equivalent PLMNs.

The UE shall trigger a transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode upon receiving REFRESH command from the UICC as specified in subclause 5.4.5.3.3.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers that the RRC connection has been suspended, the UE shall stay in 5GMM-CONNECTED mode with RRC inactive indication. The UE shall re-initiate any pending procedure that had triggered the request to the lower layers to transition to RRC_CONNECTED state, if still needed.

When the UE in 5GMM-CONNECTED mode with RRC inactive indication receives a fallback indication from lower layers, and the UE has no pending NAS procedure and no pending uplink user data for PDU session(s) with user-plane resources already established, the UE shall:

- a) enter 5GMM-IDLE mode; and
- b) initiate the registration procedure for mobility and periodic registration update as specified for case o) in subclause 5.5.1.3.2.

If the UE requests the lower layers to transition to RRC_CONNECTED state at initiation of a registration procedure, a service request procedure or a de-registration procedure, upon fallback indication from lower layers, the UE shall:

- enter 5GMM-IDLE mode;
- proceed with the pending procedure; and
- if the pending procedure is a service request or registration request procedure and the SERVICE REQUEST message, the CONTROL PLANE SERVICE REQUEST message or the REGISTRATION REQUEST message does not include UE request type IE with Request type value set to "NAS signalling connection release", the UE shall include the Uplink data status IE in the SERVICE REQUEST message, the CONTROL PLANE SERVICE REQUEST message or in the REGISTRATION REQUEST message, indicating the PDU session(s) without active user-plane resources for which the UE has pending user data to be sent, if any, and the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclauses 5.5.1.3 and 5.6.1 for further details).

If the UE requests the lower layers to transition to RRC_CONNECTED state for other reason than initiation of a registration procedure, or for other reason than a service request procedure, or for other reason than a de-registration procedure, upon fallback indication from lower layers, the UE shall:

- 1) enter 5GMM-IDLE mode;
- 2) initiate the service request procedure and include the Uplink data status IE in the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any (see subclause 5.6.1 for further details). If the procedure that triggered the request to the lower layers to transition to RRC_CONNECTED state is the UE-initiated NAS transport procedure and the UE had SMS, location services message, or CIoT user data to send, the UE shall also include the SMS, location services message, or CIoT user data in the CONTROL PLANE SERVICE REQUEST message as described in subclause 5.6.1.2.2; and
- 3) upon successful service request procedure completion, proceed with any pending procedure.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives a fallback indication from lower layers, and the UE has pending uplink user data for PDU session(s) with user-plane resources already established but no pending NAS procedure, the UE shall:

- 1) enter 5GMM-IDLE mode; and
- 2) initiate the service request procedure and include the Uplink data status IE in the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication (see subclause 5.6.1 for further details).

In the above cases when the UE receives a fallback indication from lower layers, if the UE is in non-allowed area or not in allowed area, the UE shall behave as specified in subclause 5.3.5.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers that the resumption of the RRC connection has failed, and:

- a) if the lower layers indicate that access barring is applicable for all access categories except categories 0 and 2, or access barring is applicable for all access categories except category 0, the UE shall:
 - 1) stay in 5GMM-CONNECTED mode with RRC inactive indication;
- b) else, the UE shall:
 - 1) enter 5GMM-IDLE mode; and
 - 2) initiate the registration procedure for mobility and periodic registration update used for mobility (i.e. the 5GS registration type IE set to "mobility registration updating" in the REGISTRATION REQUEST message) for N1 NAS signalling connection recovery as specified for case f) in subclause 5.5.1.3.2.
- NOTE 4: An indication from the lower layer that the RRC connection has been released with cause "RRC resume failure" can be considered as an indication that the resumption of the RRC connection has failed.

The UE shall transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode over 3GPP access upon receiving from the lower layers:

- a) indication of transition from RRC_INACTIVE state to RRC_IDLE state; or
- b) indication of cell selection to E-UTRAN or another RAT that the UE supports.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers about the cell (re-)selection to different RAT that the UE supports, the UE shall initiate the registration procedure for mobility or periodic registration update used for mobility (i.e. the 5GS registration type IE set to "mobility registration updating" in the REGISTRATION REQUEST message) as specified in subclause 5.5.1.3.2.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers of a transition from RRC_INACTIVE state to RRC_IDLE state and 5GMM-REGISTERED.LIMITED-SERVICE is entered, the UE shall subsequently upon entering state 5GMM-REGISTERED.NORMAL-SERVICE and if there is no uplink user data or signalling pending, initiate the registration procedure for mobility and periodic registration update used for mobility (i.e. the 5GS registration type IE set to "mobility registration updating" in the REGISTRATION REQUEST message) for N1 NAS signalling connection recovery as specified in subclause 5.5.1.3.2.

If the UE in 5GMM-CONNECTED mode with RRC inactive indication receives an indication from the lower layers about RAN paging and the MUSIM UE decides not to initiate the service request procedure with service type set to "mobile terminated services" or control plane service type set to "mobile terminating request" to respond to the RAN paging, the UE may initiate the service request procedure and set request type to "NAS signalling connection release" in the UE request type IE and service type to "signalling" in the SERVICE REQUEST message or set request type to "NAS signalling connection release" in the UE request type IE and control plane service type set to "mobile originating request" in the CONTROL PLANE SERVICE REQUEST message to reject the RAN paging as specified in subclause 5.6.1.2 for case o of subclause 5.6.1.1. The UE may include its paging restriction preferences in the Paging restriction IE in the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message as specified in subclause 5.6.1.2 for case o of subclause 5.6.1.1.

- NOTE 5: The interworking between the NAS layer and the AS layer triggered by RAN paging is up to UE implementation.
- NOTE 6: As an implementation option, the MUSIM UE is allowed to not respond to RAN paging based on the information available in the paging message, e.g. voice service indication.

Upon receiving AMF paging indication from the lower layers, the UE shall transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode over 3GPP access and handle the AMF paging same as the paging request received in the 5GMM-IDLE mode over 3GPP access as specified in subclause 5.6.1.

5.3.1.5 Suspend and resume of the N1 NAS signalling connection

Suspend of the N1 NAS signalling connection can be initiated by the network in 5GMM-CONNECTED mode when user plane CIoT 5GS optimization is used. Resume of the suspended N1 NAS signalling connection is initiated by the UE.

In the UE, when user plane CIoT 5GS optimization is used:

- Upon idle suspend indication from the lower layers, the UE shall enter 5GMM-IDLE mode with suspend indication, shall not consider the N1 NAS signalling connection released and shall not consider the secure exchange of NAS messages terminated (see subclause 4.4.2.5 and 4.4.5).
- Upon trigger of a procedure using an initial NAS message when in 5GMM-IDLE mode with suspend indication, the UE shall:
 - i) if the initial NAS message is a REGISTRATION REQUEST message with the NG-RAN-RCU bit of the 5GS update type IE set to "UE radio capability update needed", enter 5GMM-IDLE mode without suspend indication and proceed with the registration procedure; or
 - ii) otherwise, request the lower layer to resume the RRC connection.
- NOTE 1: In NB-N1 mode, in the request to the lower layer the data volume information of the initial NAS message is provided to the lower layers. Interactions between the NAS and the lower layers in order to obtain the data volume information of the initial NAS message (see 3GPP TS 36.321 [25E], 3GPP TS 36.331 [25A]) is left to implementations.
- Upon indication from the lower layers that the RRC connection has been resumed when in 5GMM-IDLE mode with suspend indication, the UE shall enter 5GMM-CONNECTED mode. If the pending NAS message is:
 - i) a SERVICE REQUEST message, the service type IE is not set to "emergency services fallback", and the UE did not include the NAS message container IE in the SERVICE REQUEST message; or
 - ii) a CONTROL PLANE SERVICE REQUEST message, and the UE did not include the CIoT small data container IE or the NAS message container IE in the CONTROL PLANE SERVICE REQUEST message,
- the message shall not be sent. Otherwise the UE shall cipher the message as specified in subclause 4.4.5 and send the pending initial NAS message upon entering 5GMM-CONNECTED mode;
- NOTE 2: If a NAS message is discarded and not sent to the network, the uplink NAS COUNT value corresponding to that message is reused for the next uplink NAS message to be sent.
- Upon fallback indication from the lower layers at RRC connection resume when in 5GMM-IDLE mode with suspend indication, the UE shall enter 5GMM-IDLE mode without suspend indication, send any pending initial NAS message and proceed as if RRC connection establishment had been requested;
- Upon indication from the lower layers that the RRC connection resume has failed and indication from the lower layers that the RRC connection is suspended, the UE shall enter 5GMM-IDLE mode with suspend indication and restart the ongoing NAS procedure if required; and
- Upon indication from the lower layers that the RRC connection resume has failed and indication from the lower layers that the RRC connection is not idle suspended, the UE shall enter 5GMM-IDLE mode without suspend indication and restart the ongoing NAS procedure if required.

In the network, when user plane CIoT 5GS optimization is used:

- Upon idle suspend indication from the lower layers, the network shall enter 5GMM-IDLE mode with suspend indication, shall not consider the N1 NAS signalling connection released and shall not consider the secure exchange of NAS messages terminated; and
- Upon indication from the lower layers that it has received the NGAP UE context resume request message as specified in 3GPP TS 38.413 [31] when in 5GMM-IDLE mode with suspend indication, the network shall enter 5GMM-CONNECTED mode.

5.3.2 Permanent identifiers

A globally unique permanent identity, the 5G subscription permanent identifier (SUPI), is allocated to each subscriber for 5GS-based services. The IMSI, the network specific identifier, the GCI and the GLI are valid SUPI types. When the SUPI contains a network specific identifier, a GCI or a GLI, it shall take the form of a network access identifier (NAI). When the UE performs initial registration for onboarding services in SNPN or is registered for onboarding services in SNPN, the SUPI contains the onboarding SUPI derived from the default UE credentials for primary authentication. The UE derives the onboarding SUPI before or during the initial registration for onboarding services in SNPN and uses the

derived onboarding SUPI in the initial registration for onboarding services in SNPN and while registered for onboarding services in SNPN.

The structure of the SUPI and its derivatives are specified in 3GPP TS 23.003 [4].

The UE provides the SUPI to the network in concealed form. The SUCI is a privacy preserving identifier containing the concealed SUPI. When the SUPI contains a network specific identifier, a GCI or a GLI, the SUCI shall take the form of a NAI as specified in 3GPP TS 23.003 [4].

A UE supporting N1 mode includes a SUCI:

- a) in the REGISTRATION REQUEST message when the UE is attempting initial registration procedure and a valid 5G-GUTI is not available;
- b) in the IDENTITY RESPONSE message, if the SUCI is requested by the network during the identification procedure; and
- c) in the DEREGISTRATION REQUEST message when the UE initiates a de-registration procedure and a valid 5G-GUTI is not available.

If the UE uses the "null-scheme" as specified in 3GPP TS 33.501 [24] to generate a SUCI, the SUCI contains the unconcealed SUPI.

When:

- not operating in SNPN access operation mode; or
- operating in SNPN access operation mode but not performing initial registration for onboarding services and not registered for onboarding services;

the UE shall use the "null-scheme" if:

- a) the home network has not provisioned the public key needed to generate a SUCI;
- b) the home network has configured "null-scheme" to be used for the UE;
- c) the UE needs to perform a registration procedure for emergency services after the failure of authentication procedure or after reception of a REGISTRATION REJECT message with the 5GMM cause #3 "Illegal UE", or to initiate a de-registration procedure before the registration procedure for emergency services was completed successfully, and the UE does not have a valid 5G-GUTI for the selected PLMN; or
- d) the UE receives an identity request for SUCI during a registration procedure for emergency services or during a de-registration procedure that was initiated before the registration procedure for emergency services was completed successfully.

When operating in SNPN access operation mode and:

- performing initial registration for onboarding services; or
- registered for onboarding services;

the UE shall use the "null-scheme" if:

- a) the public key needed to generate a SUCI is not configured as part of the default UE credentials for primary authentication; or
- b) "null-scheme" usage is configured as part of the default UE credentials for primary authentication.

If:

- a) the UE uses the "null-scheme" as specified in 3GPP TS 33.501 [24] to generate a SUCI;
- b) the UE operates in SNPN access operation mode and:
 - 1) an indication to use anonymous SUCI which is associated with the selected entry of the "list of subscriber data", is configured in the ME, if the UE is not registering or registered for onboarding services in SNPN; or

- an indication to use anonymous SUCI which is associated with the default UE credentials for primary authentication, is configured in the ME, if the UE is registering or registered for onboarding services in SNPN;
- NOTE 1: The ME can be configured with an indication to use anonymous SUCI associated with an entry of "list of subscriber data" when the EAP method associated with the credentials of the entry supports SUPI privacy at the EAP layer, or can be configured with an indication to use anonymous SUCI associated with the default UE credentials for primary authentication when the EAP method associated with the default UE credentials for primary authentication supports SUPI privacy at the EAP layer, or both.
- c) the UE does not need to perform a registration procedure for emergency services, or to initiate a de-registration procedure before the registration procedure for emergency services was completed successfully; and
- d) the UE does not receive an identity request for SUCI during a registration procedure for emergency services or during a de-registration procedure that was initiated before the registration procedure for emergency services was completed successfully;

then the UE shall use anonymous SUCI as specified in 3GPP TS 23.003 [4].

A W-AGF acting on behalf of an FN-RG shall use the "null-scheme" as specified in 3GPP TS 33.501 [24] to generate a SUCI.

A W-AGF acting on behalf of an N5GC device shall use the "null-scheme" as specified in 3GPP TS 33.501 [24] to generate a SUCI.

If a UE is a MUSIM UE, the UE shall use a separate permanent equipment identifier (PEI) for each USIM, if any, and each entry of "list of subscriber data", if any, the UE operates for accessing 5GS-based services; otherwise, a UE contains and uses a permanent equipment identifier (PEI) for accessing 5GS-based services. When the UE is registered with a network by using a PEI, the UE shall not use that PEI to register with another network until the UE is deregistered from the network.

In this release of the specification, the IMEI, the IMEISV, the MAC address together with the MAC address usage restriction indication and the EUI-64 are the only PEI formats supported by 5GS. The structure of the PEI and its formats are specified in 3GPP TS 23.003 [4].

Each UE supporting at least one 3GPP access technology (i.e. satellite NG-RAN, NG-RAN, satellite E-UTRAN, E-UTRAN, UTRAN or GERAN) contains a PEI in the IMEI format and shall be able to provide an IMEI and an IMEISV upon request from the network.

Each UE not supporting any 3GPP access technologies and supporting NAS over untrusted or trusted non-3GPP access shall have a PEI in the form of the Extended Unique Identifier EUI-64 [48] of the access technology the UE uses to connect to the 5GC.

A UE supporting N1 mode includes a PEI:

- a) when neither SUPI nor valid 5G-GUTI is available to use for emergency services in the REGISTRATION REQUEST message with 5GS registration type IE set to "emergency registration";
- b) when the network requests the PEI by using the identification procedure, in the IDENTITY RESPONSE message; and
- c) when the network requests the IMEISV by using the security mode control procedure, in the SECURITY MODE COMPLETE message.

Each 5G-RG supporting only wireline access and each FN-RG shall have a permanent MAC address configured by the manufacturer. For 5G-CRG, the permanent MAC address configured by the manufacturer shall be a cable modem MAC address.

When the 5G-RG contains neither an IMEI nor an IMEISV, the 5G-RG shall use as a PEI the 5G-RG's permanent MAC address configured by the manufacturer and the MAC address usage restriction indication set to "no restrictions".

The W-AGF acting on behalf of the FN-RG shall use as a PEI the MAC address provided by the FN-RG and if the MAC address provided by the FN-RG is not unique or does not correspond to the FN-RG's permanent MAC address according to W-AGF's configuration, the MAC address usage restriction indication set to "MAC address is not usable as an equipment identifier" otherwise the MAC address usage restriction indication set to "no restrictions".

The 5G-RG containing neither an IMEI nor an IMEISV shall include the PEI containing the MAC address together with the MAC address usage restriction indication:

- a) when neither SUPI nor valid 5G-GUTI is available to use for emergency services in the REGISTRATION REQUEST message with 5GS registration type IE set to "emergency registration";
- b) when the network requests the PEI by using the identification procedure, in the IDENTIFICATION RESPONSE message; and
- c) when the network requests the IMEISV by using the security mode control procedure, in the SECURITY MODE COMPLETE message.
- NOTE 2: In case c) above, the MAC address is provided even though AMF requests the IMEISV.

The W-AGF acting on behalf of the FN-RG shall include the PEI containing the MAC address together with the MAC address usage restriction indication:

- a) when the network requests the PEI by using the identification procedure, in the IDENTIFICATION RESPONSE message; and
- b) when the network requests the IMEISV by using the security mode control procedure, in the SECURITY MODE COMPLETE message.

NOTE 3: In case b) above, the MAC address is provided even though AMF requests the IMEISV.

The W-AGF acting on behalf of the N5GC device shall use as a PEI the MAC address provided by the N5GC device and the MAC address usage restriction indication set to "no restrictions". Based on operator policy, the W-AGF acting on behalf of the N5GC device may encode the MAC address of the N5GC device using the EUI-64 format as specified in [48] and use as a PEI the derived EUI-64.

NOTE 4: The MAC address of an N5GC device is universally/globally unique.

The AMF can request the PEI at any time by using the identification procedure.

5.3.3 Temporary identities

A temporary user identity for 5GS-based services, the 5G globally unique temporary identity (5G-GUTI), is used for identification within the signalling procedures. In case of PLMN the 5G-GUTI is globally unique and in case of SNPN the 5G-GUTI is unique within an SNPN. When the UE is registered to the same PLMN or SNPN over 3GPP and non-3GPP access, the UE and the AMF maintain one 5G-GUTI that is common to both 3GPP and non-3GPP access. When the UE is required to delete the 5G-GUTI according to a NAS procedure, the UE shall delete the 5G-GUTI only if it is not registered to the same PLMN or SNPN through other access. When the UE is registered to different PLMNs or SNPNs over 3GPP access and non-3GPP access, the UE maintains two 5G-GUTIs, a 5G-GUTI for the registration with a PLMN or SNPN over the 3GPP access and another 5G-GUTI for the registration with another PLMN or SNPN over the agent and service request procedures, a shortened form of the 5G-GUTI, the 5G S-temporary mobile subscriber identity (5G-S-TMSI), is used to enable more efficient radio signalling. The purpose of the 5G-GUTI and 5G-S-TMSI is to provide identity confidentiality, i.e., to protect a user from being identified and located by an intruder. The structure of the 5G-GUTI and its derivatives are specified in 3GPP TS 23.003 [4]. The 5G-GUTI has two main components (see 3GPP TS 23.501 [8]):

- a) the GUAMI; and
- b) the 5G-TMSI that provides an unambiguous identity of the UE within the AMF(s) identified by the GUAMI.
- NOTE: The UE registered with an SNPN over non-3GPP access refers to the UE accessing SNPN services via a PLMN.

The 5G-S-TMSI has three main components:

- a) the AMF set ID that uniquely identifies the AMF set within the AMF region;
- b) the AMF pointer that identifies one or more AMFs within the AMF set; and
- c) the 5G-TMSI.

A UE supporting N1 mode includes a valid 5G-GUTI, if any is available, in the REGISTRATION REQUEST and DEREGISTRATION REQUEST messages. In the SERVICE REQUEST message, the UE includes a valid 5G-S-TMSI as user identity. The AMF shall assign a new 5G-GUTI for a particular UE:

- a) during a successful initial registration procedure;
- b) during a successful registration procedure for mobility registration update;
- c) after a successful service request procedure invoked as a response to a paging request from the network and before the:
 - 1) release of the N1 NAS signalling connection; or
 - 2) suspension of the N1 NAS signalling connection due to user plane CIoT 5GS optimization i.e. before the UE and the AMF enter 5GMM-IDLE mode with suspend indication;

as specified in subclause 5.4.4.1; and

- d) after the AMF receives an indication from the lower layers that it has received the NGAP UE context resume request message as specified in 3GPP TS 38.413 [31] for a UE in 5GMM-IDLE mode with suspend indication and this resumption is a response to a paging request from the network, and before the:
 - 1) release of the N1 NAS signalling connection; or
 - suspension of the N1 NAS signalling connection due to user plane CIoT 5GS optimization i.e. before the UE and the AMF enter 5GMM-IDLE mode with suspend indication;

as specified in subclause 5.4.4.1.

The AMF should assign a new 5G-GUTI for a particular UE during a successful registration procedure for periodic registration update. The AMF may assign a new 5G-GUTI at any time for a particular UE by performing the generic UE configuration update procedure.

If a new 5G-GUTI is assigned by the AMF, the UE and the AMF handle the 5G-GUTI as follows:

- a) Upon receipt of a 5GMM message containing a new 5G-GUTI, the UE considers the new 5G-GUTI as valid and the old 5G-GUTI as invalid, stops timer T3519 if running, and deletes any stored SUCI. The new 5G-GUTI is stored in a non-volatile memory in the USIM if the corresponding file is present in the USIM, else in the non-volatile memory in the ME, as described in annex C.
- b) The AMF considers the old 5G-GUTI as invalid as soon as an acknowledgement for a registration or generic UE configuration update procedure is received.

5.3.4 Registration areas

Within the 5GS, the registration area is managed independently per access type, i.e., 3GPP access or non-3GPP access. The AMF assigns a registration area to the UE during the registration procedure. A registration area is defined as a set of tracking areas and each of these tracking areas consists of one or more cells that cover a geographical area. Within the 5GS, the concept of "registration to multiple tracking areas" applies:

- a) A tracking area is identified by a TAI which is broadcast in the cells of the tracking area. The TAI is constructed from a TAC and a PLMN identity. In case of a shared network:
 - 1) one or more TACs; and
 - 2) any of the following:
 - i) multiple PLMN identities;
 - ii) multiple SNPN identities; or
 - iii) one or more PLMN identities and one or more SNPN identities;

are broadcast.

- b) In order to reduce the tracking area update signalling within the 5GS, the AMF can assign several tracking areas to the UE. These tracking areas construct a list of tracking areas which is identified by a TAI list. When generating the TAI list, the AMF shall include only TAIs that are applicable on the access where the TAI list is sent. The AMF shall be able to allocate a TAI list over different NG-RAN access technologies. The AMF shall not allocate a TAI list containing both tracking areas in NB-N1 mode and tracking areas not in NB-N1 mode.
- c) The UE considers itself registered to a list of tracking areas and does not need to trigger the registration procedure for mobility and periodic registration update used for mobility (i.e. the 5GS registration type IE set to "mobility registration updating" in the REGISTRATION REQUEST message) as long as the UE stays in one of the tracking areas of the list of tracking areas received from the AMF.
- d) The UE will consider the TAI list as valid, until it receives a new TAI list in the next registration procedure for mobility and periodic registration update or generic UE configuration update procedure, or the UE is commanded by the network to delete the TAI list by a reject message or it is deregistered from the 5GS. If the registration request is accepted or the TAI list is reallocated by the AMF, the AMF shall provide at least one entry in the TAI list. If the new and the old TAI list are identical, the AMF does not need to provide the new TAI list to the UE during mobility registration update or periodic registration update.
- e) The TAI list can be reallocated by the AMF.
- f)- When the UE is deregistered from the 5GS, the TAI list in the UE is invalid.
- g) The UE includes the last visited registered TAI, if available, to the AMF. The last visited registered TAI is stored in a non-volatile memory in the USIM if the corresponding file is present in the USIM, else in the non-volatile memory in the ME, as described in annex C.

5.3.5 Service area restrictions

5.3.5.1 General

Service area restrictions are applicable only to 3GPP access and to wireline access.

Subclause 5.3.5.2 applies when the UE accesses 5GCN over 3GPP access.

Subclause 5.3.5.3 applies when the 5G-RG or the W-AGF acting on behalf of an FN-CRG (or on behalf of the N5GC device) access 5GCN over wireline access.

NOTE: Service area restrictions are not applicable for the W-AGF acting on behalf of the FN-BRG.

5.3.5.2 3GPP access service area restrictions

The service area restrictions consist of tracking areas forming either an allowed area, or a non-allowed area. The tracking areas belong to either the registered PLMN or its equivalent PLMNs in the registration area. The allowed area can contain up to 16 tracking areas or include all tracking areas in the registered PLMN and its equivalent PLMN(s) in the registration area. The non-allowed area can contain up to 16 tracking areas. The network conveys the service area restrictions to the UE by including either an allowed area, or a non-allowed area, but not both, in the Service area list IE of a REGISTRATION ACCEPT message or a CONFIGURATION UPDATE COMMAND message.

If the network does not convey the service area restrictions to the UE in the Service area list IE of a REGISTRATION ACCEPT message, the UE shall treat all tracking areas in the registered PLMN and its equivalent PLMN(s) in the registration area as allowed area and delete the stored list of "allowed tracking areas" or the stored list of "non-allowed tracking areas".

When the UE receives a Service area list IE with an allowed area indication during a registration procedure or a generic UE configuration update procedure:

- a) if the "Type of list" included in the Service area list IE does not indicate "all TAIs belonging to the PLMNs in the registration area are allowed area", the UE shall delete the old list of "allowed tracking areas" and store the tracking areas in the allowed area as the list of "allowed tracking areas". If the UE has a stored list of "nonallowed tracking areas", the UE shall delete that list; or
- b) if the "Type of list" included in the Service area list IE indicates "all TAIs belonging to the PLMNs in the registration area are allowed area", the UE shall treat all tracking areas in the registered PLMN and its equivalent

PLMN(s) as allowed area and delete the stored list of "allowed tracking areas" or the stored list of "non-allowed tracking areas".

When the UE receives a Service area list IE with a non-allowed area indication during a registration procedure or a generic UE configuration update procedure, the UE shall delete the old list of "non-allowed tracking areas" and store the tracking areas in the non-allowed area as the list of "non-allowed tracking areas". If the UE has a stored list of "allowed tracking areas", the UE shall delete that list.

If the UE is successfully registered to a PLMN and has a stored list of "allowed tracking areas":

- a) while the current TAI is in the list of "allowed tracking areas", the UE shall stay in, or enter, the state 5GMM-REGISTERED.NORMAL-SERVICE and is allowed to initiate any 5GMM and 5GSM procedures; and
- b) while the UE is camped on a cell which is in the registered PLMN or a PLMN from the list of equivalent PLMNs and the current TAI is not in the list of "allowed tracking areas", the UE shall enter the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE, and:
 - 1) if the UE is in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication over 3GPP access, the UE:
 - i) shall not include the Uplink data status IE in the registration procedure for mobility and periodic registration update except for emergency services or for high priority access;
 - ii) shall not perform the registration procedure for mobility and periodic registration update with Follow-on request indicator set to "Follow-on request pending", except for:
 - emergency services;
 - high priority access;
 - indicating a change of 3GPP PS data off UE status;
 - sending an SOR transparent container;
 - sending a UE policy container; or
 - sending a UE parameters update transparent container;
 - iii) shall not initiate a service request procedure or request the lower layers to resume a suspended connection, except for:
 - emergency services;
 - emergency services fallback;
 - high priority access;
 - responding to paging;
 - responding to notification received over non-3GPP access;
 - indicating a change of 3GPP PS data off UE status;
 - sending an SOR transparent container;
 - sending a UE policy container; or
 - sending a UE parameters update transparent container; and
 - 2) if the UE is in 5GMM-CONNECTED mode or 5GMM-CONNECTED mode with RRC inactive indication over 3GPP access, the UE:
 - i) shall not perform the registration procedure for mobility and periodic registration update with Uplink data status IE except for emergency services or for high priority access;
 - ii) shall not initiate a service request procedure except for:

- emergency services;
- emergency services fallback;
- high priority access;
- responding to paging or responding to notification received over non-3GPP access;
- iii) shall not initiate a 5GSM procedure except for:
 - emergency services;
 - high priority access; or
 - indicating a change of 3GPP PS data off UE status; and
- iv) shall not perform the NAS transport procedure except for the sending:
 - SMS;
 - an LPP message;
 - a location services message;
 - an SOR transparent container;
 - a UE policy container;
 - a UE parameters update transparent container; or
 - a CIoT user data container.
- NOTE 1: The contents of CIoT user data container can be data that is not for exception reports, or data that is for exception reports if allowed for the UE (see subclause 6.2.13).
- If the UE is successfully registered to a PLMN and has a stored list of "non-allowed tracking areas":
 - a) while the UE is camped on a cell which is in the registered PLMN or a PLMN from the list of equivalent PLMNs and the current TAI is not in the list of "non-allowed tracking areas", the UE shall stay in, or enter, the state 5GMM-REGISTERED.NORMAL-SERVICE and is allowed to initiate any 5GMM and 5GSM procedures; and
 - b) while the current TAI is in the list of "non-allowed tracking areas", the UE shall enter the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE, and:
 - if the UE is in 5GMM-IDLE mode or 5GMM-IDLE mode with suspend indication over 3GPP access, the UE:
 - i) shall not include the Uplink data status IE in the registration procedure for mobility and periodic registration update except for emergency services or for high priority access;
 - ii) shall not perform the registration procedure for mobility and periodic registration update with Follow-on request indicator set to "Follow-on request pending", except for:
 - emergency services;
 - high priority access;
 - indicating a change of 3GPP PS data off UE status;
 - sending an SOR transparent container;
 - sending a UE policy container; or
 - sending a UE parameters update transparent container; and
 - iii) shall not initiate a service request procedure or request the lower layers to resume a suspended connection, except for:

- emergency services;
- emergency services fallback;
- high priority access;
- responding to paging;
- responding to notification received over non-3GPP access;
- indicating a change of 3GPP PS data off UE status;
- sending an SOR transparent container;
- sending a UE policy container; or
- sending a UE parameters update transparent container; and
- 2) if the UE is in 5GMM-CONNECTED mode or 5GMM-CONNECTED mode with RRC inactive indication over 3GPP access, the UE:
 - i) shall not perform the registration procedure for mobility and registration update with the Uplink data status IE except for emergency services or for high priority access;
 - ii) shall not initiate a service request procedure or request the lower layers to resume a suspended connection, except for:
 - emergency services;
 - emergency services fallback;
 - high priority access; or
 - responding to paging or responding to notification received over non-3GPP access;
 - iii) shall not initiate a 5GSM procedure except for:
 - emergency services;
 - high priority access; or
 - indicating a change of 3GPP PS data off UE status; and
 - iv) shall not perform the NAS transport procedure except for the sending:
 - SMS;
 - an LPP message;
 - a location services message;
 - an SOR transparent container;
 - a UE policy container;
 - a UE parameters update transparent container; or
 - a CIoT user data container.
- NOTE 2: The contents of CIoT user data container can be data that is not for exception reports, or data that is for exception reports if allowed for the UE (see subclause 6.2.13).

The list of "allowed tracking areas", as well as the list of "non-allowed tracking areas" shall be erased when:

- a) the UE is switched off; and
- b) the UICC containing the USIM is removed or an entry of the "list of subscriber data" with the SNPN identity of the SNPN is updated.

When a tracking area is added to the list of "5GS forbidden tracking areas for roaming" or to the list of "5GS forbidden tracking areas for regional provision of service" as specified in the subclauses 5.5.1.2.5 or 5.5.1.3.5, the tracking area shall be removed from the list of "allowed tracking areas" if the tracking area is already present in the list of "allowed tracking areas" if the tracking area is already present in the list of "non-allowed tracking areas" if the tracking area is already present in the list of "non-allowed tracking areas".

5.3.5.3 Wireline access service area restrictions

If:

- a) a SERVICE REJECT message with the 5GMM cause #28 "Restricted service area";
- b) a DL NAS TRANSPORT message with the Payload container type IE set to "N1 SM information" and the 5GMM cause #28 "Restricted service area"; or
- c) a REGISTRATION ACCEPT message includes the PDU session reactivation result error cause IE with the 5GMM cause #28 "Restricted service area";

is received over wireline access then the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) shall start enforcing the wireline access service area restrictions and shall enter the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE.

While in the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE, the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) shall:

- a) if in 5GMM-IDLE mode over wireline access:
 - 1) shall not perform the registration procedure for mobility and periodic registration update with Uplink data status IE except for emergency services or for high priority access; and
 - 2) shall not initiate a service request procedure except for:
 - emergency services,
 - emergency services fallback; or
 - high priority access; and
- b) if in 5GMM-CONNECTED mode over wireline access:
 - 1) shall not perform the registration procedure for mobility and periodic registration update with Uplink data status IE except for:
 - emergency services;
 - emergency services fallback; or
 - high priority access;
 - 2) shall not initiate a service request procedure except for emergency services, or high priority access; and
 - 3) shall not initiate a 5GSM procedure except for emergency services or high priority access;

over the wireline access.

When the 5G-RG is switched off, the UICC containing the USIM is removed or the 5G-RG starts using another wireline access network, the 5G-RG shall stop enforcing the wireline access service area restrictions, if enforced.

When the W-AGF acting on behalf of the FN-CRG determines that the FN-CRG is switched off, the W-AGF acting on behalf of the FN-CRG stops enforcing the wireline access service area restrictions, if enforced.

When the W-AGF acting on behalf of the N5GC device determines that the FN-CRG serving the N5GC device is switched off, the W-AGF acting on behalf of the N5GC device stops enforcing the wireline access service area restrictions, if enforced.

5.3.6 Mobile initiated connection only mode

The UE can request the use of mobile initiated connection only (MICO) mode during the registration procedure (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]). The UE shall not request use of MICO mode over non-3GPP access. Furthermore, the UE in 3GPP access shall not request the use of MICO mode during:

- a) a registration procedure for initial registration for emergency services (see subclause 5.5.1.2);
- b) a registration procedure for initial registration for initiating an emergency PDU session (see subclause 5.5.1.2);
- c) a registration procedure for mobility and periodic registration update (see subclause 5.5.1.3) for initiating an emergency PDU session if the UE is in the state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE; or
- d) a registration procedure for mobility and periodic registration update (see subclause 5.5.1.3) when the UE has an emergency PDU session established.

If the UE requests the use of MICO mode, the network can accept the use of MICO mode by providing a MICO indication when accepting the registration procedure. The UE may use MICO mode only if the network has provided the MICO indication IE during the last registration procedure. The UE may also request an active time value together with the MICO mode indication during the registration procedure. If the UE requests an active time by including an active time value, the UE may also include the Requested T3512 value IE to request a particular T3512 value to be allocated.

If the network accepts the use of MICO mode and does not include an active time value in T3324 IE to the UE, the AMF may include an "all PLMN registration area allocated" indication in the MICO indication IE to the UE. If the UE indicated the support for strictly periodic registration timer in the MICO indication IE to the network, the network may include a "strictly periodic registration timer supported" indication in the MICO indication IE to the UE.

If the UE requested the use of active time by including an active time value and the network accepts the use of MICO mode and the use of active time, the AMF shall include an active time value in the T3324 IE to the UE. If the AMF indicates active time value to the UE, AMF should not indicate "all PLMN registration area allocated" indication in the MICO indication IE to the UE. Upon entering 5GMM-IDLE mode, AMF shall start the active timer with the active time value indicated to the UE and shall consider the UE is reachable for paging as long as the timer is running. If the UE enters 5GMM-CONNECTED mode over 3GPP access when the active timer is running, the AMF shall stop the active timer.

NOTE 1: The active time value assigned by AMF can be different from the active time value requested by the UE. AMF assigns the active time value based on several factors, e.g. local configuration, expected UE behaviour, UE requested active time value, UE subscription information, network policies etc.

If the UE requested an active time and a requested T3512 value and the network accepts the use of MICO mode, the AMF shall take the UE requested T3512 value into consideration when assigning a value of timer T3512 to the UE.

If the network accepts the use of MICO mode, the UE may deactivate the AS layer and activate MICO mode by entering the state 5GMM-REGISTERED.NO-CELL-AVAILABLE if:

- a) the UE is in 5GMM-IDLE mode over3GPP access;
- b) the UE is in the 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) state for 3GPP access; and
- c) no T3324 value is received from the network.

If the network accepts the use of MICO mode and indicates an active time value to the UE in a successful registration procedure, the UE shall start the timer T3324 with the value received from the network after entering 5GMM-IDLE mode over 3GPP access. At the expiry of the timer T3324, the UE may activate MICO mode by entering the state 5GMM-REGISTERED.NO-CELL-AVAILABLE if the UE is in the 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) state for 3GPP access. If the UE enters 5GMM-CONNECTED mode over 3GPP access when the timer T3324 is running, the UE shall stop the timer T3324.

When MICO mode is activated, all NAS timers are stopped and associated procedures aborted except for timers T3512, T3346, T3447, T3396, T3584, T3585, any back-off timers, T3247, and the timer T controlling the periodic search for HPLMN or EHPLMN or higher prioritized PLMNs (see 3GPP TS 23.122 [5]).

NOTE 2: When MICO mode is activated and if the UE is also registered over the non-3GPP access, the AMF will not send a NOTIFICATION message with access type indicating 3GPP access over the non-3GPP access for PDU sessions associated with 3GPP access.

The UE may deactivate MICO mode and activate the AS layer at any time. Upon deactivating MICO mode, the UE may initiate 5GMM procedures (e.g. for the transfer of mobile originated signalling or user data).

When an emergency PDU session is successfully established after the MICO mode was enabled, the UE and the AMF shall locally disable MICO mode. The UE and the AMF shall not enable MICO mode until the AMF accepts the use of MICO mode in the next registration procedure. To enable an emergency call back, the UE should wait for a UE implementation-specific duration of time before requesting the use of MICO mode after the completion of the emergency services.

If the AMF accepts the use of MICO mode and does not indicate "strictly periodic registration timer supported" in the MICO indication IE to the UE, the AMF starts the implicit de-registration timer for 3GPP access when entering 5GMM-IDLE mode for 3GPP access. If AMF accepts the use of MICO mode and indicates "strictly periodic registration timer supported" in the MICO indication IE to the UE, AMF shall start the strictly periodic monitoring timer with T3512 value indicated in the T3512 value IE after the registration procedure is completed. The AMF shall neither stop nor reset the strictly periodic monitoring timer when the NAS signalling connection is established or released for the UE. If the strictly periodic monitoring timer expires when NAS signalling connection is established for the UE, AMF shall restart the strictly periodic monitoring timer with the T3512 value, otherwise AMF shall start the implicit de-registration timer.

When an emergency PDU session is successfully established and the MICO mode is disabled, the UE shall stop timer T3512 if running and the AMF shall stop strictly periodic monitoring timer if running. The UE and the AMF shall behave as if no "strictly periodic registration timer supported" indication was given to the UE in the last registration attempt.

Upon successful completion of an attach procedure or tracking area updating procedure after inter-system change from N1 mode to S1 mode (see 3GPP TS 24.301 [15]), the UE operating in single-registration mode shall locally disable MICO mode. After inter-system change from S1 mode to N1 mode, the UE operating in single-registration mode may re-negotiate MICO mode with the network during the registration procedure for mobility and periodic registration update.

5.3.7 Handling of the periodic registration update timer and mobile reachable timer

The periodic registration update procedure is used over 3GPP access to periodically notify the availability of the UE to the network. The procedure is controlled in the UE by the periodic registration update timer, T3512.

If the UE is registered over the 3GPP access, the AMF maintains an implicit de-registration timer to control when the UE is considered implicitly de-registered over the 3GPP access. If the UE is registered over the non-3GPP access, the AMF also maintains a non-3GPP implicit de-registration timer to control when the UE is considered implicitly de-registered over the non-3GPP access. The UE registered over the non-3GPP access maintains a non-3GPP de-registration timer to control when the UE is considered implicitly de-registered over the non-3GPP access.

The AMF shall start a non-3GPP implicit de-registration timer for the UE registered over non-3GPP access when the N1 NAS signalling connection over non-3GPP access is released.

The UE registered over non-3GPP access shall reset and start a non-3GPP de-registration timer when the N1 NAS signalling connection over non-3GPP access is released. The non-3GPP de-registration timer is stopped when the UE enters 5GMM-CONNECTED mode over non-3GPP access or the 5GMM-DEREGISTERED state over non-3GPP access.

The non-3GPP implicit de-registration timer shall be longer than the non-3GPP de-registration timer.

The value of timer T3512 is sent by the network to the UE in the REGISTRATION ACCEPT message. The UE shall apply this value in all tracking areas of the list of tracking areas assigned to the UE until a new value is received. The periodic registration update timer only applies to the UE registered to the 5GS services over 3GPP access.

If timer T3512 received by the UE in a REGISTRATION ACCEPT message contains an indication that the timer is deactivated or the timer value is zero, then timer T3512 is deactivated and the UE shall not perform the periodic registration update procedure.

NOTE 1: The UE does not perform the periodic registration update procedure for non-3GPP access.

If during the registration procedure, the AMF does not indicate "strictly periodic registration timer supported" in the MICO indication IE to the UE, timer T3512 is reset and started with its initial value, when the UE changes from 5GMM-CONNECTED over 3GPP access to 5GMM-IDLE mode over 3GPP access. Timer T3512 is stopped when the UE enters 5GMM-CONNECTED mode over 3GPP access or the 5GMM-DEREGISTERED state over 3GPP access.

If during the registration procedure, the AMF indicates "strictly periodic registration timer supported" in the MICO indication IE to the UE, timer T3512 is started with its initial value after the completion of the registration procedure. The UE shall neither stop nor reset the timer T3512 when the UE enters 5GMM-CONNECTED or when changing from 5GMM-CONNECTED mode to 5GMM-IDLE mode. If the timer T3512 expires,

- a) the UE in 5GMM-CONNECTED mode over 3GPP access shall reset and start the timer T3512 with its initial value; or
- b) the UE in 5GMM-IDLE mode over 3GPP access shall perform the periodic registration procedure.

If the UE is registered for emergency services, and timer T3512 expires, the UE shall not initiate a periodic registration update procedure, but shall locally de-register from the network. When the UE is camping on a suitable cell, it may reregister to regain normal service.

When a UE is not registered for emergency services, and timer T3512 expires when the UE is in 5GMM-IDLE mode, the periodic registration update procedure shall be started.

If the UE is not registered for emergency services, and is in a state other than 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE over 3GPP access when timer T3512 expires, the periodic registration update procedure is delayed until the UE returns to 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE over 3GPP access.

NOTE 2: When the UE returns to 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE and it needs to initiate other 5GMM procedure than the periodic registration update procedure then, based on UE implementation, the 5GMM procedure can take precedence.

The network supervises the periodic registration update procedure of the UE by means of the mobile reachable timer.

If the UE is not registered for emergency services, the mobile reachable timer shall be longer than the value of timer T3512. In this case, by default, the mobile reachable timer is 4 minutes greater than the value of timer T3512.

The network behaviour upon expiry of the mobile reachable timer is network dependent, but typically the network stops sending paging messages to the UE on the first expiry, and may take other appropriate actions.

If the UE is registered for emergency services, the AMF shall set the mobile reachable timer with a value equal to timer T3512. When the mobile reachable timer expires, the AMF shall locally de-register the UE.

The mobile reachable timer shall be reset and started with the value as indicated above, when the AMF releases the NAS signalling connection for the UE. The mobile reachable timer shall be stopped when a NAS signalling connection is established for the UE.

Upon expiry of the mobile reachable timer the network shall start the implicit de-registration timer over 3GPP access. The value of the implicit de-registration timer over 3GPP access is network dependent. If MICO mode is activated, the network shall start the implicit de-registration timer over 3GPP access when the UE enters 5GMM-IDLE mode at the AMF over 3GPP access. The default value of the implicit de-registration timer over 3GPP access is 4 minutes greater than the value of timer T3512.

If the implicit de-registration timer expires before the UE contacts the network, the network shall implicitly de-register the UE. The implicit de-registration timer shall be stopped when a NAS signalling connection is established for the UE.

If the non-3GPP implicit de-registration timer expires before the UE contacts the network over the non-3GPP access, the network shall implicitly de-register the UE and enter the state 5GMM-DEREGISTERED over non-3GPP access for the UE. The non-3GPP implicit de-registration timer shall be stopped when a NAS signalling connection over non-3GPP access is established for the UE.

If the non-3GPP de-registration timer expires before the UE contacts the network over the non-3GPP access, the UE shall enter the state 5GMM-DEREGISTERED over non-3GPP access. The non-3GPP de-registration timer shall be stopped when a NAS signalling connection over non-3GPP access is established for the UE.

If the AMF provides T3346 value IE in the DEREGISTRATION REQUEST message with Access type set to "Non-3GPP access" in Deregistration type IE, REGISTRATION REJECT message during a registration procedure for mobility and periodic registration update or SERVICE REJECT message and the value of timer T3346 is greater than the value of timer T3512, the AMF sets the mobile reachable timer and the implicit de-registration timer such that the sum of the timer values is greater than the value of timer T3346.

If the AMF provides T3346 value IE in the DEREGISTRATION REQUEST message with Access type set to "3GPP access" in Deregistration type IE, REGISTRATION REJECT message during a registration procedure for mobility and periodic registration update or SERVICE REJECT message and the value of timer T3346 is greater than the value of the non-3GPP de-registration timer, the AMF sets the non-3GPP implicit de-registration timer value to be 8 minutes greater than the value of timer T3346.

If the UE receives T3346 value IE in the DEREGISTRATION REQUEST message with Access type set to "3GPP access" in Deregistration type IE, REGISTRATION REJECT message during a registration procedure for mobility and periodic registration update or SERVICE REJECT message and the value of timer T3346 is greater than the value of the non-3GPP de-registration timer, the UE sets the non-3GPP de-registration timer value to be 4 minutes greater than the value of timer T3346.

5.3.8 Handling of timer T3502

The value of timer T3502 can be sent by the network to the UE in the REGISTRATION ACCEPT message. The UE shall apply this value in all tracking areas of the registration area assigned to the UE, until a new value is received.

The value of timer T3502 can be sent by the network to the UE in the REGISTRATION REJECT message during the initial registration. If a REGISTRATION REJECT message including timer T3502 value was received integrity protected, the UE shall apply this value until a new value is received with integrity protection or a new PLMN is selected. Otherwise, the default value of this timer is used.

The default value of this timer is also used by the UE in the following cases:

- a) REGISTRATION ACCEPT message is received without a value specified;
- b) the UE does not have a stored value for this timer;
- c) a new PLMN which is not in the list of equivalent PLMNs has been entered, the initial registration procedure fails, the registration attempt counter is equal to 5 and no REGISTRATION REJECT message was received from the new PLMN;
- d) the network indicates that the timer is "deactivated"; or
- e) a new PLMN which is not in the list of equivalent PLMNs has been entered, the mobility and periodic update registration procedure fails and the registration attempt counter is equal to 5.

5.3.9 Handling of NAS level mobility management congestion control

The AMF may detect 5GMM signalling congestion and perform general NAS level congestion control. Under the 5GMM signalling congestion conditions the AMF may reject 5GMM signalling requests from UEs as specified in 3GPP TS 23.501 [8]. The AMF should not reject the following:

- a) requests for emergency services;
- b) requests for emergency services fallback;
- c) requests from UEs configured for high priority access in selected PLMN;
- d) DEREGISTRATION REQUEST message;
- e) requests for mobile terminated services, triggered by paging or a notification procedure; and
- f) requests for initial registration or mobility and periodic registration update, when emergency is indicated by lower layers.

When general NAS level congestion control is active, the AMF may include a value for the mobility management backoff timer T3346 in the reject messages. The UE starts the timer T3346 with the value received in the 5GMM reject messages. To avoid that large numbers of UEs simultaneously initiate deferred requests, the AMF should select the value for the timer T3346 for the rejected UEs so that timeouts are not synchronised.

If the UE is registered in the same PLMN over the 3GPP access and non-3GPP access, and the UE receives the timer T3346 from the AMF, the timer T3346 shall apply to both 3GPP access and non-3GPP access.

If the UE receives the paging message or NOTIFICATION message when timer T3346 is running and the UE is registered to the same PLMN over 3GPP access and non-3GPP access, the UE shall stop the timer T3346 for both accesses and respond to the paging message or NOTIFICATION message as specified in subclause 5.6.2 and subclause 5.6.3.

NOTE 1: As an implementation option, MUSIM UE is allowed to not respond to paging based on the information available in the paging message, e.g. voice service indication.

If the timer T3346 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE remains the same, then the timer T3346 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3346 is running, the UE shall behave as follows when the UE is switched on and the USIM in the UE remains the same:

let t1 be the time remaining for T3346 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 - t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the UE enters a new PLMN while timer T3346 is running, and the new PLMN is not equivalent to the PLMN where the UE started timer T3346, the UE shall stop timer T3346 when initiating 5GMM procedures in the new PLMN.

After a change in registration area, if the timer T3346 is running and 5GS update status is 5U1 UPDATED then the UE shall set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE.

If timer T3346 is running or is deactivated, and the UE is a UE configured for high priority access in selected PLMN, or the UE needs to initiate signalling for emergency services or emergency services fallback, then the UE is allowed to initiate 5GMM procedures.

NOTE 2: UE can, based on implementation, restrict lower layers of non-3GPP access from establishing access stratum connection on a registered PLMN when timer T3346 is running for the same PLMN.

5.3.10 Handling of DNN based congestion control

The AMF may detect and start performing DNN based congestion control when one or more DNN congestion criteria as specified in 3GPP TS 23.501 [8] are met. If the UE does not provide a DNN for a non-emergency PDU session, then the AMF uses the selected DNN or the DNN associated with the PDU session corresponding to the 5GSM procedure.

When DNN based congestion control is activated at the AMF, the AMF performs the congestion control as specified in subclause 5.4.5 and the UE performs the congestion control as specified in subclause 5.4.5 and subclause 6.2.7.

5.3.11 Handling of S-NSSAI based congestion control

The AMF may detect and start performing S-NSSAI based congestion control when one or more S-NSSAI congestion criteria as specified in 3GPP TS 23.501 [8] are met. If the UE does not provide a DNN for a non-emergency PDU session, then the AMF uses the selected DNN or the DNN associated with the PDU session corresponding to the 5GSM procedure. If the UE does not provide an S-NSSAI for a non-emergency PDU session, then the AMF uses the selected S-NSSAI or the S-NSSAI associated with the PDU session corresponding to the 5GSM procedure.

When S-NSSAI based congestion control is activated at the AMF, the AMF performs the congestion control as specified in subclause 5.4.5 and the UE performs the congestion control as specified in subclause 5.4.5 and subclause 6.2.8.

5.3.12 Handling of local emergency numbers

The additional requirements in subclause 5.3.12A apply to a UE supporting registration or attach procedures via 3GPP access and registration procedures via non-3GPP access.

In case of PLMN, the network may send a local emergency numbers list or an extended local emergency numbers list or both, in the REGISTRATION ACCEPT message, by including the Emergency number list IE and the Extended emergency number list IE, respectively. The Local emergency numbers list can be updated as described in 3GPP TS 24.301 [15], subclause 5.3.7.

In case of SNPN, the network may send an extended local emergency numbers list, in the REGISTRATION ACCEPT message, by including the Extended emergency number list IE. The network shall set the Extended Emergency Number List Validity (EENLV) field within the Extended emergency number list IE to "Extended Local Emergency Numbers List is valid only in the PLMN from which this IE is received". The UE shall consider the emergency number(s) received in the Extended emergency number list IE valid only in the SNPN from which this IE is received regardless of the received value of the EENLV field within the Extended emergency number list IE.

The UE shall store the local emergency numbers list and the extended local emergency numbers list, as provided by the network. The local emergency numbers list stored in the UE shall be replaced on each receipt of the Emergency number list IE. The extended local emergency numbers list stored in the UE shall be replaced on each receipt of the Extended emergency number list IE. The received local emergency numbers list or the received extended local emergency numbers list or both shall be provided to the upper layers.

If a REGISTRATION ACCEPT message is received via non-3GPP access from a PLMN in a country different from the current country of the UE, the UE shall keep the stored local emergency numbers list and the extended local emergency numbers list, if available.

The emergency number(s) received in the Emergency number list IE are valid only in PLMNs in the same country as the PLMN from which this IE is received. If no Emergency number list IE is contained in the REGISTRATION ACCEPT message, then the stored local emergency numbers list in the UE shall be kept, except if the UE has successfully registered to a PLMN in a country different from that of the PLMN that sent the list.

The emergency number(s) received in the Extended emergency number list IE are valid only in:

- PLMNs in the same country as the PLMN from which this IE is received, if the Extended Emergency Number List Validity (EENLV) field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid in the country of the PLMN from which this IE is received"; and
- the PLMN from which this IE is received, if the EENLV field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid only in the PLMN from which this IE is received"; and
- the SNPN from which this IE is received, regardless of the value of the EENLV field within the Extended emergency number list IE.

If no Extended Local Emergency Numbers List is contained in the REGISTRATION ACCEPT message, and the registered PLMN or the registered SNPN has not changed, then the stored Extended Local Emergency Numbers List in the UE shall be kept. If no Extended Local Emergency Numbers List is contained in the REGISTRATION ACCEPT message, but the registered PLMN or the registered SNPN has changed, then:

- if the last received indication in the EENLV field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid only in the PLMN from which this IE is received", the stored Extended Local Emergency Numbers List in the UE shall be deleted; and
- if the last received indication in the EENLV field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid in the country of the PLMN from which this IE is received" the list shall be kept except if the UE has successfully registered to a PLMN in a country different from that of the PLMN that sent the stored list.
- NOTE: To prevent the misrouting of emergency calls, all operators within a country need to follow the regulation or agree on the setting of the Extended emergency number list IE in accordance to national agreement either to indicate validity within a country or to indicate validity only within the PLMN.

The local emergency numbers list and the extended local emergency numbers list shall be deleted at switch off or removal of the USIM. The UE shall be able to store up to ten entries in the local emergency numbers list and up to twenty entries in the Extended local emergency numbers list, received from the network.

For the use of the local emergency numbers list and the extended local emergency numbers list by the UE see 3GPP TS 24.301 [15], subclause 5.3.7.

5.3.12A Handling of local emergency numbers received via 3GPP access and non-3GPP access

5.3.12A.1 General

The requirements in subclause 5.3.12 with the clarifications and additional conditions in subclause 5.3.12A apply to a UE supporting:

- attach procedures (see 3GPP TS 24.301 [15]) or registration procedures via 3GPP access; and
- registration procedures via non-3GPP access.

The UE shall ignore the presence or absence of local emergency numbers list, extended local emergency numbers list or both, in a REGISTRATION ACCEPT message received via non-3GPP access, unless conditions in subclause 5.3.12A.2 are met.

For the purposes of subclause 5.3.12A, the UE is considered neither registered nor attached over 3GPP access if:

- 1) the UE supports 3GPP access to EPC, the UE does not support 3GPP access to 5GC, and:
 - a) the EMM sublayer is in the EMM-NULL state, EMM-DEREGISTERED state or EMM-DEREGISTERED-INITIATED state; or
- 2) the UE supports 3GPP access to 5GC, the UE does not support 3GPP access to EPC, and:
 - a) the 5GMM sublayer is in the 5GMM-NULL state, 5GMM-DEREGISTERED state or 5GMM-DEREGISTERED-INITIATED state; or
- 3) supports both 3GPP access to EPC and 3GPP access to 5GC, and:
 - a) the EMM sublayer is in the EMM-NULL state, EMM-DEREGISTERED state or EMM-DEREGISTERED-INITIATED state; and
 - a) the 5GMM sublayer is in the 5GMM-NULL state, 5GMM-DEREGISTERED state or 5GMM-DEREGISTERED-INITIATED state.

5.3.12A.2 Receiving a REGISTRATION ACCEPT message via non-3GPP access

If the UE can determine the current country and after switch on or after removal of the USIM, has not been registered or has not been attached via 3GPP access in the current country, then the UE shall store the local emergency numbers list or the extended local emergency numbers list or both, as provided by the network with an MCC matching the current country via non-3GPP access.

NOTE: The UE determines, as the current country, the country in which it is located in accordance with 3GPP TS 24.502 [18].

The UE shall replace a previously stored local emergency numbers list or a previously stored extended local emergency numbers list or both with a local emergency numbers list or an extended local emergency numbers list or both received in a REGISTRATION ACCEPT message via non-3GPP access, if the previously stored local emergency numbers list was also received via non-3GPP access or the previously stored extended local emergency numbers list was also received via non-3GPP access.

The UE shall replace a previously stored extended local emergency numbers list with an extended local emergency numbers list received in a REGISTRATION ACCEPT message via non-3GPP access, if:

- the UE is neither registered nor attached over 3GPP access;

- the REGISTRATION ACCEPT message is received from a PLMN different from which the stored list was received; and
- the stored indication in the EENLV field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid only in the PLMN from which this IE is received".

If no extended local emergency numbers list is contained in a REGISTRATION ACCEPT message received via non-3GPP access and the UE is neither registered nor attached over 3GPP access, the stored extended local emergency numbers list in the UE shall be discarded if:

- the UE can determine the current country and the UE has successfully registered to a PLMN in the country and that country is different from that of the PLMN that sent the stored list; or
- the REGISTRATION ACCEPT message is received from a PLMN different from which the stored list was received, and the stored indication in the EENLV field within the Extended emergency number list IE indicates "Extended Local Emergency Numbers List is valid only in the PLMN from which this IE is received".

5.3.13 Lists of 5GS forbidden tracking areas

If the UE is not operating in SNPN access operation mode, the UE shall store a list of "5GS forbidden tracking areas for roaming", as well as a list of "5GS forbidden tracking areas for regional provision of service". Otherwise the UE shall store a list of "5GS forbidden tracking areas for roaming":

- per SNPN; and
- if the UE supports access to an SNPN using credentials from a credentials holder, per entry of the "list of subscriber data" or PLMN subscription;

and store a list of "5GS forbidden tracking areas for regional provision of service":

- per SNPN; and
- if the UE supports access to an SNPN using credentials from a credentials holder, per entry of the "list of subscriber data" or PLMN subscription.

Within the 5GS, these lists are managed independently per access type, i.e., 3GPP access or wireline access. These lists shall be erased when:

- a) the UE is switched off, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the subscribed SNPN identity identifying the current SNPN is updated or, if the UE supports access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" associated with the lists is updated; and
- b) periodically (with a period in the range 12 to 24 hours).

Over 3GPP access, when the lists are erased, the UE performs cell selection according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]. A tracking area shall be removed from the list of "5GS forbidden tracking areas for roaming", as well as the list of "5GS forbidden tracking areas for regional provision of service", if the UE receives the tracking area in the TAI list or the Service area list of "allowed tracking areas" in REGISTRATION ACCEPT message or a CONFIGURATION UPDATE COMMAND message. The UE shall not remove the tracking area from "5GS forbidden tracking areas for regional provision of service" if the UE is registered for emergency services.

In N1 mode over 3GPP access, the UE shall update the suitable list whenever a REGISTRATION REJECT, SERVICE REJECT or DEREGISTRATION REQUEST message is received with the 5GMM cause #12 "tracking area not allowed", #13 "roaming not allowed in this tracking area", or #15 "no suitable cells in tracking area", or a REGISTRATION ACCEPT or SERVICE ACCEPT message is received with the forbidden TAI(s).

Over wireline access, the 5G-RG, the W-AGF acting on behalf of an FN-RG or the W-AGF acting on behalf of an N5GC device shall update the suitable list whenever a REGISTRATION REJECT, SERVICE REJECT or DEREGISTRATION REQUEST message is received with the 5GMM cause #12 "tracking area not allowed" or #13 "roaming not allowed in this tracking area".

NOTE: In this release of the specification, for untrusted non-3GPP access and trusted non-3GPP access, neither the list of "5GS forbidden tracking areas for roaming" nor the list of "5GS forbidden tracking areas for regional provision of service" is maintained by the UE since the UE is not able to determine the corresponding TAI.

Each list shall accommodate 40 or more TAIs. When the list is full and a new entry has to be inserted, the oldest entry shall be deleted.

5.3.13A Forbidden PLMN lists

In N1 mode, two lists of forbidden PLMN are managed independently per access type, i.e., 3GPP access or non-3GPP access:

- the list of "forbidden PLMNs" as defined in 3GPP TS 23.122 [5] is applicable for 3GPP access in N1 mode. The same list is used by 5GMM for 3GPP access, EMM, GMM and MM (see 3GPP TS 24.301 [15] and 3GPP TS 24.008 [12]), regardless whether the UE is operating in single-registration mode or dual-registration mode.
- the list of "forbidden PLMNs for non-3GPP access to 5GCN" as defined in 3GPP TS 24.502 [18] is applicable for 5GMM for non-3GPP access.

The list of "forbidden PLMNs for GPRS service" as defined in 3GPP TS 23.122 [5] and 3GPP TS 24.008 [12] is applicable for 3GPP access in N1 mode. The same list is used by 5GMM for 3GPP access, EMM and GMM (see 3GPP TS 24.301 [15] and 3GPP TS 24.008 [12]), regardless whether the UE is operating in single-registration mode or dual-registration mode.

5.3.14 List of equivalent PLMNs

The UE shall store a list of equivalent PLMNs. These PLMNs shall be regarded by the UE as equivalent to each other for PLMN selection and cell selection/re-selection. The same list is used by 5GMM, EMM, GMM and MM (see 3GPP TS 24.301 [15] and 3GPP TS 24.008 [12]) except for the case when the UE operates in dual-registration mode (see subclause 4.8.3).

The UE shall update or delete this list at the end of each registration procedure. The stored list consists of a list of equivalent PLMNs as downloaded by the network plus the PLMN code of the registered PLMN that downloaded the list. When the UE is switched off, the UE shall keep the stored list so that it can be used for PLMN selection after switch on. The UE shall delete the stored list if the USIM is removed or when the UE registered for emergency services enters the state 5GMM-DEREGISTERED. The maximum number of possible entries in the stored list is 16.

5.3.15 Transmission failure abnormal case in the UE

The abnormal case 5GMM uplink message transmission failure indication by lower layers can be identified for 5GMM procedures:

When it is specified in the relevant procedure that it is up to the UE implementation to re-run the ongoing procedure that triggered that procedure, the procedure can typically be re-initiated using a retransmission mechanism of the uplink message (i.e. the one that has previously failed to be transmitted) with new sequence number and message authentication code information thus avoiding to re-start the whole procedure.

NOTE: The transmission failure can happen due to TAI change. The lower layer might take some time to read the system information and determine if the current TAI is changed. Therefore, the information of TAI change can be sent to the NAS layer a little after receiving the transmission failure indication from the lower layer. How to handle the retransmission procedure caused by the possible delayed TAI change information is up to UE implementation.

5.3.16 Extended DRX cycle for UEs in 5GMM-IDLE and 5GMM-CONNECTED mode with RRC inactive indication

Extended DRX (eDRX) cycle is supported for a UE in N1 mode. When eDRX is requested by the UE and accepted by the network:

- if the UE is not in NB-N1 mode, eDRX is used when the UE is in 5GMM-IDLE mode or in 5GMM-CONNECTED mode with RRC inactive indication; or
- if the UE is in NB-N1 mode, eDRX is used when the UE is in 5GMM-IDLE mode.

The UE may request the use of eDRX cycle during a registration procedure by including the Requested extended DRX parameters IE (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]). The UE shall not request the use of eDRX during a registration procedure for emergency services. The UE may use the extended idle mode DRX cycle length stored in the USIM (see 3GPP TS 31.102 [22]) when requesting the use of eDRX.

The UE and the network may negotiate eDRX parameters during a registration procedure when the UE has an emergency PDU session.

The network accepts the request to use the eDRX by providing the Negotiated extended DRX parameters IE when accepting the registration procedure. The UE shall use eDRX only if it received the Negotiated extended DRX parameters IE during the last registration procedure and the UE does not have an emergency PDU session.

NOTE: If the UE wants to keep using eDRX, the UE includes the Extended DRX parameters IE in each registration procedure.

If the UE received the Negotiated extended DRX parameters IE during the last registration procedure, upon successful completion of the PDU session release procedure of the emergency PDU session, the UE shall resume eDRX.

If the network has provided the Negotiated extended DRX parameters IE during the last registration procedure, upon successful completion of the PDU session release procedure of the emergency PDU session, the network shall resume eDRX.

If the UE or the network locally releases an emergency PDU session, the UE or the network shall not use eDRX until the UE receives eDRX parameters during a registration procedure with PDU session context synchronization or upon successful completion of a service request procedure with PDU session context synchronization.

If the UE did not receive the Negotiated extended DRX parameters IE, or if the UE has an emergency PDU session, the UE shall use the stored UE specific DRX parameter, if available.

If the network did not accept the request to use eDRX, or if the UE has an emergency PDU session, the network shall use the stored UE specific DRX parameter, if available.

If the network provided the Negotiated extended DRX parameters IE and also assigned a new 5G-GUTI for the UE as described in subclause 5.5.1.3.4 during the last registration procedure, the network shall use the stored UE specific DRX parameter, if available, with the old 5G-GUTI and use the eDRX provided by the network with the new 5G-GUTI until the old 5G-GUTI can be considered as invalid by the network (see subclauses 5.4.4.4 and 5.5.1.3.4).

5.3.17 Service Gap Control

Service gap control (SGC) only applies to 3GPP access.

The network may control the frequency with which UEs can transition from 5GMM-IDLE mode to 5GMM-CONNECTED mode via the SGC as specified in 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]. If the network supports SGC and the service gap time value i.e. T3447 value is available in the 5GMM context of the UE, the AMF shall consider SGC as active for the UE.

The UE and the network negotiate usage of SGC during the registration procedure for initial registration and registration procedure for mobility and periodic registration update:

- the UE supporting SGC indicates its support in the REGISTRATION REQUEST message. If the UE supports SGC and the SGC is active for the UE, the AMF includes T3447 value IE in the REGISTRATION ACCEPT message (see subclause 5.5.1.2 and subclause 5.5.1.3). The UE stores the T3447 value; and
- for UEs that do not support SGC when the network rejects mobility management signalling requests because SGC is active in the network, the mechanism for general NAS level mobility management congestion control as specified in subclause 5.3.9 applies.

The network can provide a new T3447 value to the UE to be used next time it is started or stop timer T3447 in the UE if running with the Generic UE configuration update procedure as specified in subclause 5.4.4.

The UE shall start timer T3447 when the N1 NAS signalling connection is released and if:

- the UE supports SGC, and the T3447 value is available in the UE and does not indicate zero; and
- the N1 NAS signalling connection released was not established for:
 - paging;
 - registration procedure for initial registration with Follow-on request indicator set to "No follow-on request pending";
 - registration procedure for mobility and periodic registration update with Follow-on request indicator set to "No follow-on request pending" and without Uplink data status IE included;
 - requests for emergency services; or
 - requests for exception data reporting.

If the SGC is active in the network, after the UE transitions from 5GMM-CONNECTED mode to 5GMM-IDLE mode except when the UE was in 5GMM-CONNECTED mode due to:

- paging;
- registration procedure for initial registration with Follow-on request indicator set to "No follow-on request pending";
- registration procedure for mobility and periodic registration update with Follow-on request indicator set to "No follow-on request pending" and without Uplink data status IE included,
- requests for emergency services; or
- requests for exception data reporting,

the network shall start timer T3447 if not already running:

- with the T3447 value available in the 5GMM context minus 4 minutes, if the UE supports SGC and the T3447 value has been sent to the UE with a non-zero value; or
- with the T3447 value available in the 5GMM context if the UE does not support SGC.

When timer T3447 is running, the network allows:

- requests for emergency service;
- requests for emergency services fallback;
- requests for high priority access;
- requests for exception data reporting;
- registration procedure for initial registration with Follow-on request indicator set to "No follow-on request pending";
- registration procedure for mobility and periodic registration update without Uplink data status IE included and with Follow-on request indicator set to "No follow-on request pending"; or
- service request procedure or registration procedure for mobility and periodic registration update triggered by paging and subsequent MO signalling or MO data, if any, until the UE enters 5GMM-IDLE mode.

The UE or the network with a running T3447 timer keeps the timer running when the UE transits from 5GMM-IDLE mode to 5GMM-CONNECTED mode.

NOTE: If the UE transitions from 5GMM-IDLE mode to 5GMM-CONNECTED mode due to registration procedure for initial registration with Follow-on request indicator set to "No follow-on request pending" or mobility and periodic registration update request without Uplink data status IE and with Follow-on request indicator set to "No follow-on request pending", the UE initiates no further MO signalling except for mobility and periodic registration update requests without Uplink data status and with Follow-on request indicator set to "No follow-on request pending" until the UE receives mobile terminated signalling (e.g. DL NAS TRANSPORT message for MT SMS) or MT data over user plane, or after the UE has moved to 5GMM-IDLE state and the service gap timer is not running.

If timer T3447 is running when the UE changes PLMN or enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE remains the same, then timer T3447 is kept running until it expires.

If the AMF determines that the UE operating in single-registration mode has performed an inter-system change from N1 mode to S1 mode and the timer T3447 is running in the AMF, the AMF stops the T3447.

Upon inter-system change from S1 mode to N1 mode, if the UE supports service gap control, T3447 is running in the UE, and the T3447 value IE is included in the REGISTRATION ACCEPT message received from the AMF (see subclause 5.5.1.2 and subclause 5.5.1.3), the UE shall keep T3447 running. Additionally, the UE shall store and replace the currently stored service gap time value with the received T3447 value. Upon expiry of the running T3447 timer, the UE shall use the new value when starting T3447 again.

If the UE is switched off when the timer T3447 is running, the UE shall behave as follows when the UE is switched on and the USIM in the UE remains the same:

let t1 be the time remaining for timer T3447 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

5.3.18 Restriction on use of enhanced coverage

In order to deal with use of extensive resources from the network, the operator may prevent specific subscribers from using enhanced coverage (see 3GPP TS 23.501 [8]). When in NB-N1 mode, the UE shall indicate support for restriction on use of enhanced coverage. When in WB-N1 mode, the UE supporting either CE mode A or CE mode B shall indicate support for restriction on use of enhanced coverage. The UE supporting restriction on use of enhanced coverage indicates its support for restriction on use of enhanced coverage in the REGISTRATION REQUEST message. If the UE supports restriction on use of enhanced coverage, the AMF indicates in the REGISTRATION ACCEPT message (see subclause 5.5.1.2 and subclause 5.5.1.3) that:

- a) when in WB-N1 mode, whether CE mode B is restricted for the UE, or both CE mode A and CE mode B are restricted for the UE, or both CE mode A and CE mode B are not restricted for the UE; or
- b) when in NB-N1 mode, whether the use of enhanced coverage is restricted or not for the UE.

If:

- a) the use of enhanced coverage is restricted;
- b) the use of CE mode B is restricted; or
- c) the use of CE mode A and CE mode B is restricted,

the UE shall not use enhanced coverage in the registered PLMN and in any PLMN which is in the list of equivalent PLMNs.

If the UE supports CE mode B and the network determines that

- a) the use of enhanced coverage is not restricted for the UE; or
- b) CE mode B is not restricted for the UE,

the applicable NAS timer values shall be calculated by the network as described in subclause 4.19 and subclause 4.20.

For a UE that supports restriction on use of enhanced coverage or CE mode B, if:

- a) the AMF determines to enforce a change in restriction on the use of enhanced coverage or a change in the restriction on the use of CE mode B as described in 3GPP TS 23.501 [8]; and
- b) the UE is in 5GMM-CONNECTED mode and there is no ongoing registration procedure,

the AMF shall initiate the generic UE configuration update procedure to indicate registration requested and release of the N1 NAS signalling connection not requested as described in subclause 5.4.4. After the successful completion of the registration procedure for mobility registration update including change of the restriction on the use of enhanced coverage, for any SMF with which the UE has an established PDU session, the AMF updates the SMF with the indication on the use of extended NAS timer setting as described in 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9].

5.3.19 Handling of congestion control for transport of user data via the control plane

The network may activate congestion control for transport of user data via the control plane, as specified in 3GPP TS 23.501 [8].

If the UE has indicated support for the control plane CIoT 5GS optimizations and the network decides to activate the congestion control for transport of user data via the control plane, the network may include a value for the control plane data back-off timer T3448 in REGISTRATION ACCEPT, SERVICE ACCEPT or SERVICE REJECT message, and shall store an control plane data back-off time on a per UE basis. The UE starts the timer T3448 with the value informed in the message. To avoid that large numbers of UEs simultaneously initiate deferred requests, the network should select the value for the timer T3448 for the informed UEs so that timeouts are not synchronised.

The network sends REGISTRATION ACCEPT message or SERVICE ACCEPT message without T3448 value IE to stop the timer T3448 running in the UE as specified in subclause 5.5.1.3.4 and subclause 5.6.1.4.

Based on the stored control plane data back-off time for the UE, the network may reject the transfer of user data via the control plane initiated by the UE.

While the timer T3448 is running, the UE in 5GMM-IDLE mode does not initiate the transport of user data via the control plane procedure, except if the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]) and the user data is related to an exceptional event.

The UE is allowed:

- a) to respond to paging with CONTROL PLANE SERVICE REQUEST message without uplink data; or
- b) to send a CONTROL PLANE SERVICE REQUEST message for emergency services or for emergency services fallback;

even if the timer T3448 is running.

Upon entering the state 5GMM-DEREGISTERED or a new PLMN which is not equivalent to the PLMN where the UE started the timer T3448, or upon being switched off while the timer T3448 is running, the UE stops the timer T3448.

5.3.19a Specific requirements for UE configured to use timer T3245

5.3.19a.1 UE not operating in SNPN access operation mode

The following requirements apply for a UE that is configured to use timer T3245 (see 3GPP TS 24.368 [17] or 3GPP TS 31.102 [22]).

When the UE adds a PLMN identity to the "forbidden PLMN list" or sets the USIM as invalid for 5GS services for 3GPP access or non-3GPP access, and timer T3245 (see 3GPP TS 24.008 [12]) is not running, the UE shall start timer T3245 as specified in 3GPP TS 24.008 [12], subclause 4.1.1.6.

Upon expiry of the timer T3245, the UE shall erase the "forbidden PLMN list" and "forbidden PLMNs for GPRS service" list and set the USIM to valid for 5GS services for 3GPP access and non-3GPP access. When the lists are erased, the UE performs cell selection according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

If the UE is switched off when the timer T3245 is running, the UE shall behave as follows when the UE is switched on and the USIM in the UE remains the same:

let t1 be the time remaining for T3245 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the UE will follow the behaviour as defined in the paragraph above upon expiry of the timer T3245. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

5.3.19a.2 UE operating in SNPN access operation mode

The following requirements apply for a UE that is configured to use timer T3245 (see 3GPP TS 24.368 [17]).

When the UE adds an SNPN to the "permanently forbidden SNPNs" list or "temporarily forbidden SNPNs" list which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, or sets the selected entry of the "list of subscriber data" or the selected PLMN subscription as invalid for 3GPP access or non-3GPP access, and timer T3245 (see 3GPP TS 24.008 [12]) is not running, the UE shall start timer T3245 with a random value, uniformly drawn from the range between 12h and 24h.

Upon expiry of the timer T3245, the UE shall erase the "permanently forbidden SNPNs" list(s) and "temporarily forbidden SNPNs" list(s) and set the selected entry of the "list of subscriber data" or the selected PLMN subscription to valid for 3GPP access and non-3GPP access. When the lists are erased, the UE performs cell selection according to 3GPP TS 38.304 [28].

If the UE is switched off when the timer T3245 is running, the UE shall behave as follows when the UE is switched on and the selected entry of the "list of subscriber data" or the selected PLMN subscription remain the same:

let t1 be the time remaining for T3245 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the UE will follow the behaviour as defined in the paragraph above upon expiry of the timer T3245. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

5.3.20 Specific requirements for UE when receiving non-integrity protected reject messages

5.3.20.1 General

This subclause specifies the requirements for a UE that is not configured to use timer T3245 (see 3GPP TS 24.368 [17] or 3GPP TS 31.102 [22]) and receives a REGISTRATION REJECT or SERVICE REJECT message without integrity protection with specific 5GMM causes.

NOTE: Additional UE requirements for this case, requirements for other 5GMM causes, and requirements for the case when the UE receives an integrity protected reject message are specified in subclauses 5.5.1 and 5.6.1.

5.3.20.2 Requirements for UE in a PLMN

The UE shall maintain:

- a list of PLMN-specific attempt counters (see 3GPP TS 24.301 [15]). The maximum number of possible entries in the list is implementation dependent. This list is applicable to access attempts via 3GPP access only;
- a list of PLMN-specific attempt counters for non-3GPP access, if the UE supports non-3GPP access. The maximum number of possible entries in the list is implementation dependent. This list is applicable to access attempts via non-3GPP access only;
- a list of PLMN-specific N1 mode attempt counters for 3GPP access. The maximum number of possible entries in the list is implementation dependent. This list is applicable to access attempts via 3GPP access only;
- a list of PLMN-specific N1 mode attempt counters for non-3GPP access, if the UE supports non-3GPP access. The maximum number of possible entries in the list is implementation dependent. This list is applicable to access attempts via non-3GPP access only;

3GPP TS 24.501 version 17.16.0 Release 17

177

- one counter for "SIM/USIM considered invalid for GPRS services" events (see 3GPP TS 24. 008 [12]); and
- one counter for "USIM considered invalid for 5GS services over non-3GPP access" events, if the UE supports non-3GPP access.

A UE supporting non-EPS services shall maintain one counter for "SIM/USIM considered invalid for non-GPRS services" events (see 3GPP TS 24.008 [12]).

The UE shall store the above lists of attempt counters and the event counters in its non-volatile memory. The UE shall erase the lists and reset the event counters to zero when the UICC containing the USIM is removed. The counter values shall not be affected by the activation or deactivation of MICO mode or power saving mode (see 3GPP TS 24.301 [15]).

The UE implementation-specific maximum value for any of the above counters shall not be greater than 10.

NOTE 1: Different counters can use different UE implementation-specific maximum values.

If the UE receives a REGISTRATION REJECT or SERVICE REJECT message without integrity protection with 5GMM cause value #3, #6, #7, #11, #12, #13, #15, #27, #31, #62, #72 or #73 before the network has established secure exchange of NAS messages for the N1 NAS signalling connection, the UE shall stop timer T3510 or T3517 if running, and start timer T3247 (see 3GPP TS 24.008 [12]) with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running, and take the following actions:

- 1) if the 5GMM cause value received is #3, #6 or #7, and:
 - a) if the 5GMM cause value is received over 3GPP access, the UE shall:
 - i) if the UE is already registered over another access:
 - store the current TAI in the list of "5GS forbidden tracking areas for roaming", memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
 - search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]; or
 - ii) otherwise, if the counter for "SIM/USIM considered invalid for GPRS services" events has a value less than a UE implementation-specific maximum value,
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI for 3GPP access;
 - if the 5GMM cause value received is #3 or #6, delete the list of equivalent PLMNs if any;
 - increment the counter for "SIM/USIM considered invalid for GPRS services" events;
 - if the 5GMM cause value received is #3 or #6, and if the counter for "SIM/USIM considered invalid for non-GPRS services" events has a value less than a UE implementation-specific maximum value, increment the counter;
 - if a registration procedure was performed, reset the registration attempt counter and if a service request procedure was performed, reset the service request attempt counter;
 - if the UE is operating in single-registration mode, handle the EMM parameters EMM state, EPS update status, EPS attach attempt counter, tracking area updating attempt counter or service request attempt counter, 4G-GUTI, TAI list, eKSI as specified in 3GPP TS 24.301 [15] for the case when the EPS attach, tracking area updating procedure or service request procedure is rejected with the EMM cause of the same value in a NAS message without integrity protection;
 - store the current TAI in the list of "5GS forbidden tracking areas for roaming", memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
 - search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]; and as a UE implementation option, the UE may perform registration attempt

over the non-3GPP access, if non-3GPP access is available, and the USIM is not considered invalid for 5GS services over non-3GPP access; and

iii) otherwise proceed as specified in subclauses 5.5.1 and 5.6.1;

- b) if the 5GMM cause value is received over non-3GPP access, the UE shall:
 - i) if the UE is already registered over another access:
 - enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
 - may perform registration attempt over the non-3GPP access if another access point for non-3GPP access is available; or
 - ii) otherwise, if the counter for "USIM considered invalid for 5GS services over non-3GPP access" events has a value less than a UE implementation-specific maximum value,
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete the 5G-GUTI, last visited registered TAI, TAI list and ngKSI for non-3GPP access;
 - enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE;
 - increment the counter for "USIM considered invalid for 5GS services over non-3GPP access" events; and as a UE implementation option, the UE may either perform registration attempt over the non-3GPP access if another access point for non-3GPP access is available, or if 3GPP access is available, and the SIM/USIM is not considered invalid for 5GS services over 3GPP access, perform registration attempt over the 3GPP access; and

NOTE 2: How to select another access point for non-3GPP access is implementation specific.

iii) otherwise proceed as specified in subclauses 5.5.1 and 5.6.1;

- 2) if the 5GMM cause value received is #12, #13 or #15, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1. Additionally, the UE may:
 - a) if the 5GMM cause value is received over 3GPP access, non-3GPP access is available, the UE is not registered over non-3GPP access yet, and the USIM is not considered invalid for 5GS services over non-3GPP access, perform registration attempt over the non-3GPP access; or
 - b) if the 5GMM cause value is received over non-3GPP access, 3GPP access is available, the UE is not registered over 3GPP access yet, and the USIM is not considered invalid for 5GS services over 3GPP access, perform registration attempt over the 3GPP access;
- 3) if the 5GMM cause value received is #11 or #73 and the UE is in its HPLMN or EHPLMN:
 - a) if the 5GMM cause value is received over 3GPP access, the UE shall:
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete, the 5G-GUTI, last visited registered TAI, TAI list, ngKSI for 3GPP access and the list of equivalent PLMNs. Additionally, if a registration procedure was performed, the UE shall reset the registration attempt counter and if a service request procedure was performed, reset the service request attempt counter;
 - if the 5GMM cause value received is #11 and the UE is operating in single-registration mode, handle the EMM parameters EMM state, EPS update status, EPS attach attempt counter, tracking area updating attempt counter or service request attempt counter, 4G-GUTI, TAI list, eKSI as specified in 3GPP TS 24.301 [15] for the case when the EPS attach, tracking area updating procedure or service request procedure is rejected with the EMM cause of the same value in a NAS message without integrity protection;
 - if the 5GMM cause value received is #73 and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI. Additionally, the UE shall reset the attach attempt counter or tracking area updating attempt counter, and enter the state EMM-DEREGISTERED;

- store the current TAI in the list of "5GS forbidden tracking areas for roaming", memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
- search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]; and as a UE implementation option, the UE may perform registration attempt over the non-3GPP access, if non-3GPP access is available, the UE is not registered over non-3GPP access yet, and the USIM is not considered invalid for 5GS services over non-3GPP access;
- b) if the 5GMM cause value is received over non-3GPP access, the UE shall:
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete the 5G-GUTI, last visited registered TAI, TAI list and ngKSI for non-3GPP access. Additionally, if a registration procedure was performed, the UE shall reset the registration attempt counter and if a service request procedure was performed, reset the service request attempt counter; and
 - enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. As a UE implementation option, the UE may perform registration attempt over the non-3GPP access if another access point for non-3GPP access is available, or if 3GPP access is available, the UE is not registered over 3GPP access yet, and the USIM is not considered invalid for 5GS services over 3GPP access, perform registration attempt over the 3GPP access;
- 4) if the 5GMM cause value received is #11 or #73 and the UE is not in its HPLMN or EHPLMN, in addition to the UE requirements specified in subclause 5.5.1 and 5.6.1:
 - if the message was received via 3GPP access and if the PLMN-specific attempt counter for the PLMN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment the PLMN-specific attempt counter for the PLMN; or
 - if the message was received via non-3GPP access and if the PLMN-specific attempt counter for non-3GPP access for the PLMN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment the PLMN-specific attempt counter for non-3GPP access for the PLMN;
- 5) if the 5GMM cause value received is #27, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1. Additionally, if the PLMN-specific N1 mode attempt counter for the respective access type and for the PLMN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment this counter for the PLMN;
- 6) if the 5GMM cause value received is #72, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1. Additionally, if the PLMN-specific N1 mode attempt counter for non-3GPP access for the PLMN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment this counter for the PLMN;
- 7) if the 5GMM cause value received is #31 for a UE that has indicated support for CIoT optimizations, the UE may discard the message or alternatively the UE should:
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2);
 - store the current TAI in the list of "5GS forbidden tracking areas for roaming", memorize the current TAI
 was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject
 message; and
 - search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]; and
- 8) if the 5GMM cause value received is #62, the UE may discard the message or alternatively the UE should:
 - set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2);
- store the current TAI in the list of "5GS forbidden tracking areas for roaming", memorize the current TAI
 was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject
 message; and
- search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

Upon expiry of timer T3247, the UE shall:

- remove all tracking areas from the list of "5GS forbidden tracking areas for regional provision of service" and the list of "5GS forbidden tracking areas for roaming", which were stored in these lists for non-integrity protected NAS reject message;
- set the USIM to valid for 5GS services for 3GPP access, if:
 - the counter for "SIM/USIM considered invalid for GPRS services" events has a value less than a UE implementation-specific maximum value;
- set the USIM to valid for 5GS services for non-3GPP access, if:
 - the counter for "USIM considered invalid for 5GS services over non-3GPP access" events has a value less than a UE implementation-specific maximum value;
- set the USIM to valid for non-EPS services, if:
 - the counter for "SIM/USIM considered invalid for non-GPRS services" events has a value less than a UE implementation-specific maximum value;
- for each PLMN-specific attempt counter that has a value greater than zero and less than a UE implementationspecific maximum value, remove the respective PLMN from the extension of the "forbidden PLMNs" list;
- for each PLMN-specific attempt counter for non-3GPP access that has a value greater than zero and less than a UE implementation-specific maximum value, remove the respective PLMN from the list of "forbidden PLMNs for non-3GPP access to 5GCN";
- re-enable the N1 mode capability for 3GPP access and, for each PLMN-specific N1 mode attempt counter for 3GPP access that has a value greater than zero and less than a UE implementation-specific maximum value, remove the respective PLMN from the list of PLMNs where N1 mode is not allowed for 3GPP access (see 3GPP TS 23.122 [5]);
- re-enable the N1 mode capability for non-3GPP access and, for each PLMN-specific N1 mode attempt counter for non-3GPP access that has a value greater than zero and less than a UE implementation-specific maximum value, remove the respective PLMN from the list of PLMNs where N1 mode is not allowed for non-3GPP access;
- if the UE is supporting A/Gb mode or Iu mode, perform the actions as specified in 3GPP TS 24.008 [12] for the case when timer T3247 expires;
- if the UE is supporting S1 mode, perform the actions as specified in 3GPP TS 24.301 [15] for the case when timer T3247 expires; and
- initiate a registration procedure, if still needed, dependent on 5GMM state and 5GS update status, or perform PLMN selection according to 3GPP TS 23.122 [5].

When the UE is switched off, the UE shall, for each PLMN-specific attempt counter that has a value greater than zero and less than the UE implementation-specific maximum value, remove the respective PLMN from the list of "forbidden PLMNs". When the USIM is removed, the UE should perform this action.

When the UE is switched off, the UE shall, for each PLMN-specific attempt counter for non-3GPP access that has a value greater than zero and less than the UE implementation-specific maximum value, remove the respective PLMN from the list of "forbidden PLMNs for non-3GPP access to 5GCN". When the USIM is removed, the UE should perform this action.

NOTE 3: If the respective PLMN was stored in the extension of the "forbidden PLMNs" list, then according to 3GPP TS 23.122 [5] the UE will delete the contents of this extension when the UE is switched off or the USIM is removed.

5.3.20.3 Requirements for UE in an SNPN

If the UE is operating in SNPN access operation mode, the UE shall maintain, for each of the entries in the "list of subscriber data":

- one SNPN-specific attempt counter for 3GPP access. The counter is applicable to access attempts via 3GPP access only;
- one SNPN-specific attempt counter for non-3GPP access, if the UE supports accessing SNPN services via a PLMN. The counter is applicable in case of accessing SNPN services via a PLMN only;
- one counter for "the entry for the current SNPN considered invalid for 3GPP access" events; and
- one counter for "the entry for the current SNPN considered invalid for non-3GPP access" events, if the UE supports accessing SNPN services via a PLMN. The counter is applicable in case of accessing SNPN services via a PLMN only.
- NOTE 1: The term "non-3GPP access" used in the counter for "SNPN-specific attempt counter for non-3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events, is used to express access to SNPN services via a PLMN.

The UE shall store the above counters in its non-volatile memory. The UE shall erase the attempt counters and reset the event counters to zero when the selected entry of the "list of subscriber data" is updated or USIM is removed for the selected PLMN subscription. The counter values shall not be affected by the activation or deactivation of MICO mode or power saving mode (see 3GPP TS 24.301 [15]).

The UE implementation-specific maximum value for any of the above counters shall not be greater than 10.

NOTE 2: Different counters can use different UE implementation-specific maximum values.

If the UE receives a REGISTRATION REJECT or SERVICE REJECT message without integrity protection with 5GMM cause value #3, #6, #7, #12, #13, #15, #27, #72, #74, or #75 before the network has established secure exchange of NAS messages for the N1 NAS signalling connection, the UE shall stop timer T3510 or T3517, if running, and start timer T3247 (see 3GPP TS 24.008 [12]) with a random value uniformly drawn from the range between:

- a) 15 minutes and 30 minutes for 5GMM cause value #74; or
- b) 30 minutes and 60 minutes for other 5GMM cause values;

if the timer is not running, and take the following actions:

- a) if the 5GMM cause value received is #3, #6, or #7 and the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN:
 - 1) if the 5GMM cause value is received over 3GPP access:
 - i) if the UE is already registered over another access, the UE shall:
 - A) store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
 - B) search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28]; or
 - ii) otherwise if the counter for "the entry for the current SNPN considered invalid for 3GPP access" events has a value less than a UE implementation-specific maximum value, the UE shall:
 - A) set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list, and ngKSI for 3GPP access;
 - B) increment the counter for "the entry for the current SNPN considered invalid for 3GPP access" events;

- C) reset the registration attempt counter in case of a REGISTRATION REJECT message or reset the service request attempt counter in case of a SERVICE REJECT message;
- D) store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
- E) search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28]. As a UE implementation option, if accessing SNPN services via a PLMN is available and the selected entry of the "list of subscriber data" or the selected PLMN subscription is not considered invalid for non-3GPP access, then the UE may attempt to access SNPN services via a PLMN; or
- iii) otherwise, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1;
- 2) if the 5GMM cause value is received over non-3GPP access:
- NOTE 3: A 5GMM cause value "received over non-3GPP access" in this subclause refers to a 5GMM cause value received via a PLMN when the UE attempts to access SNPN services via a PLMN.
 - i) if the UE is already registered over another access, the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; or
 - ii) otherwise if the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events has a value less than a UE implementation-specific maximum value, the UE shall:
 - A) set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete the 5G-GUTI, last visited registered TAI, TAI list, and ngKSI for non-3GPP access;
 - B) enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; and
 - C) increment the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events. As a UE implementation option, if 3GPP access is available and the selected entry of the "list of subscriber data" or the selected PLMN subscription is not considered invalid for 3GPP access, then the UE may make a registration attempt over 3GPP access; or
 - iii) otherwise, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1;
- b) if the 5GMM cause value received is #12, #13, or #15, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1. Additionally:
 - 1) if the 5GMM cause value is received over 3GPP access, accessing SNPN services via a PLMN is available, the UE has not accessed SNPN services via a PLMN yet, and the selected entry of the "list of subscriber data" or the selected PLMN subscription is not considered invalid for non-3GPP access, the UE may attempt to access SNPN services via a PLMN; or
 - if the 5GMM cause value is received over non-3GPP access, 3GPP access is available, the UE is not registered to the current SNPN over 3GPP access yet, and the selected entry of the "list of subscriber data" or the selected PLMN subscription is not considered invalid for 3GPP access, the UE may make a registration attempt over 3GPP access;
- NOTE 4: The network does not send 5GMM cause value #13 to the UE operating in SNPN access operation mode in this release of specification.
- c) if the 5GMM cause value received is #27, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1.
 Additionally, if the SNPN-specific attempt counter for the respective access type and for the current SNPN has a value less than a UE implementation-specific maximum value, the UE shall increment this counter for the SNPN;

c1) if the 5GMM cause value received is #72, the UE shall proceed as specified in subclauses 5.5.1 and 5.6.1. Additionally, if the SNPN-specific attempt counter for non-3GPP access for the current SNPN has a value less than a UE implementation-specific maximum value, the UE shall increment this counter for the SNPN; and

d) if:

- 1) the 5GMM cause value received is #74 or #75; or
- 2) the 5GMM cause value received is #3, #6, or #7 and the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN;

in addition to the UE requirements specified in subclauses 5.5.1 and 5.6.1:

- 1) if the message was received via 3GPP access and if the SNPN-specific attempt counter for 3GPP access for the SNPN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment the SNPN-specific attempt counter for 3GPP access for the SNPN; or
- 2) if the message was received via non-3GPP access and if the SNPN-specific attempt counter for non-3GPP access for the SNPN sending the reject message has a value less than a UE implementation-specific maximum value, the UE shall increment the SNPN-specific attempt counter for non-3GPP access for the SNPN.
- NOTE 5: The message "received via non-3GPP access" in this subclause refers to a message received via a PLMN when the UE attempts to access SNPN services via a PLMN.

Upon expiry of timer T3247, the UE shall:

- remove, for each SNPN, all tracking areas from the list of "5GS forbidden tracking areas for regional provision of service" and the list of "5GS forbidden tracking areas for roaming" for the SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, which were stored in these lists for non-integrity protected NAS reject message;
- set each entry of the "list of subscriber data" or the PLMN subscription to valid for 3GPP access, if the corresponding counter for "the entry for the current SNPN considered invalid for 3GPP access" events has a value less than a UE implementation-specific maximum value;
- set each entry of the "list of subscriber data" or the PLMN subscription to valid for non-3GPP access, if the corresponding counter for "the entry for the current SNPN considered invalid for non-3GPP access" events has a value less than a UE implementation-specific maximum value;
- remove each SNPN identity from the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, if the corresponding SNPN-specific attempt counter for 3GPP access has a value greater than zero and less than a UE implementation-specific maximum value and the SNPN identity is included in any of the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access;
- remove each SNPN identity from the "permanently forbidden SNPNs" list for non-3GPP access or "temporarily forbidden SNPNs" list for non-3GPP access which are, if the MS supports access to an SNPN using credentials from a credentials holder, associated with the selected entry of the "list of subscriber data" or the selected PLMN subscription, if the corresponding SNPN-specific attempt counter for non-3GPP access has a value greater than zero and less than a UE implementation-specific maximum value and the SNPN identity is included in any of the "permanently forbidden SNPNs" list for non-3GPP access or "temporarily forbidden SNPNs" list for non-3GPP access;
- re-enable the N1 mode capability for 3GPP access and, for each SNPN-specific attempt counter for 3GPP access that has a value greater than zero and less than a UE implementation-specific maximum value, remove the respective SNPN from the list of SNPNs for which the N1 mode capability was disabled due to receipt of a reject from the network with 5GMM cause #27 "N1 mode not allowed" (see 3GPP TS 23.122 [5]);
- re-enable the N1 mode capability for non-3GPP access and, for each SNPN-specific attempt counter for non-3GPP access that has a value greater than zero and less than a UE implementation-specific maximum value, remove the respective SNPN from the list of SNPNs for which N1 mode capability was disabled for non-3GPP

access due to receipt of a reject from the network with 5GMM cause #27 "N1 mode not allowed" or 5GMM cause #72 "non-3GPP access to 5GCN not allowed"; and

- initiate a registration procedure, if still needed, dependent on 5GMM state and 5GS update status, or perform SNPN selection according to 3GPP TS 23.122 [5].

When the UE is switched off or a UICC containing the USIM is removed:

- for each SNPN-specific attempt counter for 3GPP access having a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the respective SNPN identity from the "permanently forbidden SNPNs" list(s) for 3GPP access or "temporarily forbidden SNPNs" list(s) for 3GPP access, if available; and
- for each SNPN-specific attempt counter for non-3GPP access having a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the respective SNPN identity from the "permanently forbidden SNPNs" list(s) for non-3GPP access or "temporarily forbidden SNPNs" list(s) for non-3GPP access, if available.

When an entry of the "list of subscriber data" is updated:

- if the UE does not support access to an SNPN using credentials from a credentials holder and the SNPN-specific attempt counter for 3GPP access for the SNPN corresponding to the entry has a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the SNPN identity corresponding to the entry from the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access, if available;
- if the UE does not support access to an SNPN using credentials from a credentials holder and the SNPN-specific attempt counter for non-3GPP access for the SNPN corresponding to the entry has a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the SNPN identity corresponding to the entry from the "permanently forbidden SNPNs" list for non-3GPP access or "temporarily forbidden SNPNs" list for non-3GPP access, if available;
- if the UE supports access to an SNPN using credentials from a credentials holder and the SNPN-specific attempt counter for 3GPP access for an SNPN in the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access, associated with the entry, has a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the SNPN identity corresponding to the SNPN from the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access, associated with the entry, has a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the SNPN identity corresponding to the SNPN from the "permanently forbidden SNPNs" list for 3GPP access or "temporarily forbidden SNPNs" list for 3GPP access, associated with the entry; and
- if the UE supports access to an SNPN using credentials from a credentials holder and the SNPN-specific attempt counter for non-3GPP access for an SNPN in the "permanently forbidden SNPNs" list for non-3GPP access or "temporarily forbidden SNPNs" list for non-3GPP access associated with the entry, has a value greater than zero and less than the UE implementation-specific maximum value, the UE shall remove the SNPN identity corresponding to the SNPN from the "permanently forbidden SNPNs" list for non-3GPP access, associated with the entry.

5.3.21 CIoT 5GS optimizations

CIOT 5GS optimizations provide improved support of small data and SMS transfer. A UE supporting CIOT 5GS optimizations can indicate the 5GS CIoT network behaviour the UE can support and prefers to use during the registration procedure (see 3GPP TS 23.502 [9]). The UE may indicate the support for control plane CIoT 5GS optimization, user plane CIoT 5GS optimization, N3 data transfer and header compression (see subclause 9.11.3.1). Furthermore, the UE may, separately from the indication of support, indicate preference for control plane CIoT 5GS optimization or user plane CIoT 5GS optimization (see subclause 9.11.3.9A). The indication of preference is also considered as the request to use. A UE supporting CIoT EPS optimizations can also indicate the EPS CIoT network behaviour the UE can support during the registration procedure. Furthermore, the UE may, separately from the indication of support, indicate preference for control plane CIoT EPS optimization or user plane CIoT EPS optimization or user plane CIoT EPS optimization procedure. Furthermore, the UE may, separately from the indication of support, indicate preference for control plane CIoT EPS optimization or user plane CIoT EPS optimization.

NOTE 1: CIoT 5GS optimizations are not supported by NR connected to 5GCN.

NOTE 2: If the UE does not support user plane CIoT 5GS optimization, it does not indicate preference for user plane CIoT 5GS optimization.

The UE can be in NB-N1 mode or WB-N1 mode when requesting the use of CIoT 5GS optimizations during the registration procedure. A UE in NB-N1 mode always indicates support for control plane CIoT 5GS optimization.

In NB-N1 mode, the UE, when requesting the use of CIoT 5GS optimizations, does not:

- request an initial registration for emergency services;
- request a PDU session establishment for emergency PDU session; or
- indicate UE's usage setting during the registration procedure.

The network does not indicate to the UE support of emergency services when the UE is in NB-N1 mode (see subclause 5.5.1.2.4 and 5.5.1.3.4).

The control plane CIoT 5GS optimization enables support of efficient transport of user data (IP, Ethernet and Unstructured) or SMS messages over control plane via the AMF without triggering user-plane resources establishment. The support of control plane CIoT 5GS optimization is mandatory for the network in NB-N1 mode and optional in WB-N1 mode. Optional header compression of IP data and Ethernet data can be applied to PDU sessions with IP PDU session type and Ethernet PDU session type that are configured to support header compression. For IP header compression, Robust Header Compression (ROHC) protocol specified in IETF RFC 5795 [39B] is used. For Ethernet header compression, Ethernet Header Compression (EHC) protocol specified in 3GPP TS 38.323 [29] is used.

For a UE that supports Location Services (LCS) notification mechanisms in N1 mode, the control plane CIoT 5GS optimization also enables the transport of location services messages from 5GMM-IDLE mode using the CONTROL PLANE SERVICE REQUEST message when location services are requested (see subclause 6.7.1 in 3GPP TS 23.273 [6B]).

The user plane CIoT 5GS optimization enables support for change from 5GMM-IDLE mode over 3GPP access to 5GMM-CONNECTED mode over 3GPP access without the need for using the service request procedure (see subclause 5.3.1.5).

If the UE supports user plane CIoT 5GS optimization, it shall also support N3 data transfer.

If the UE indicates support of one or more CIoT 5GS optimizations and the network supports one or more CIoT 5GS optimizations and decides to accept the registration request, the network indicates the supported CIoT 5GS optimizations to the UE per registration area when accepting the UE request. Network indication of support is interpreted by the UE as the acceptance to use the respective feature. After completion of the registration procedure, the UE and the network can then use the accepted CIoT 5GS optimizations for the transfer of user data (IP, Ethernet, Unstructured and SMS).

A UE in NB-N1 mode or WB-N1 mode can request the use of SMS over NAS by setting the SMS requested bit of the 5GS update type IE in the REGISTRATION REQUEST message as specified in subclauses 5.5.1.2.2 and 5.5.1.3.2.

The AMF indicates whether it allows the use of SMS over NAS for a UE in NB-N1 mode or WB-N1 mode by setting the SMS allowed bit of the 5GS registration result IE in the REGISTRATION ACCEPT message as specified in subclauses 5.5.1.2.4 and 5.5.1.3.4.

If the UE and the network support both the control plane CIoT 5GS optimization and N3 data transfer, then when receiving the UE's request for a PDU session establishment, the AMF decides whether the PDU session should be NEF PDU session or N6 PDU session as specified in 3GPP TS 23.501 [8] and then:

- a) if NEF PDU session is to be established for unstructured data type, the AMF includes control plane only indication for the requested PDU session to the SMF;
- b) if N6 PDU session is to be established and the DNN or S-NSSAI of the newly requested N6 PDU session supports interworking with EPS as specified in TS 23.502 [9]:
 - 1) if there are existing N6 PDU sessions supporting interworking with EPS for this UE that were established with the control plane only indication, the AMF includes the control plane only indication for the newly requested N6 PDU session to the SMF; or
 - 2) if there are existing N6 PDU sessions supporting interworking with EPS for this UE that were established without the control plane only indication, the AMF does not include the control plane only indication for the newly requested N6 PDU session to the SMF;

- 3) if there is no existing N6 PDU session supporting interworking with EPS for this UE, the AMF determines whether to include the control plane only indication for the newly requested N6 PDU session to the SMF based on local policies, the UE's preferred CIoT network behaviour and the supported CIoT network behaviour; and
- c) if N6 PDU session is to be established and the DNN or S-NSSAI of the N6 PDU session does not support interworking with EPS as specified in TS 23.502 [9], the AMF determines whether to include the control plane only indication for the newly requested N6 PDU session to the SMF based on local policies, the UE's preferred CIoT network behaviour and the supported CIoT network behaviour.

In NB-N1 mode, if the UE or the network does not support N3 data transfer, then when receiving the UE's request for a PDU session establishment, the AMF decides whether the PDU session should be NEF PDU session or N6 PDU session as specified in 3GPP TS 23.501 [8] and then includes the control plane only indication for the requested PDU session to the SMF.

If the network supports user plane CIoT 5GS optimization, it shall also support N3 data transfer.

Broadcast system information may provide information about support of CIoT 5GS optimizations (see 3GPP TS 36.331 [25A]). At reception of new broadcast system information, the lower layers deliver it to the 5GMM layer in the UE. The information provided by lower layers is per PLMN and used by the UE to determine whether certain CIoT 5GS optimizations are supported in the cell.

The UE shall not attempt to use CIoT 5GS optimizations which are indicated as not supported.

In NB-N1 mode, at any given time, there cannot be user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources. The UE in NB-N1 mode shall not:

- a) request the establishment of user-plane resources for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources; or
- b) initiate the establishment of a new PDU session, or request the transfer of a PDU session from the non-3GPP access to the 3GPP access, if:
 - 1) the UE has indicated preference for user plane CIoT 5GS optimization;
 - 2) the network accepted the use of user plane CIoT 5GS optimization; and
 - 3) the UE currently has user-plane resources established for anumber of PDU sessions that is equal to the UE's maximum number of supported user-plane resources.

The AMF enforces a limit on the number of PDU sessions with active user-plane resources for a UE in NB-N1 mode based on the UE's maximum number of supported user-plane resources as follows:

- a) there can be a maximum of one PDU session with active user-plane resources when the Multiple user-plane resources support bit is set to "Multiple user-plane resources not supported", or
- b) there can be a maximum of two PDU sessions with active user-plane resources when the Multiple user-plane resources support bit is set to "Multiple user-plane resources supported".

A PDU session for a UE in NB-N1 mode shall only have one QoS rule and that is the default QoS rule. Reflective QoS is not supported in NB-N1 mode. Reflective QoS is not applicable for a PDU session with control plane only indication.

In NB-N1 mode, when the UE requests the lower layer to establish a RRC connection and the UE requests the use of user plane CIoT 5GS optimization, the UE shall pass an indication of the requested CIoT 5GS optimizations to the lower layers. If the UE requests the use of N3 data transfer without user plane CIoT 5GS optimization, then the UE shall also pass an indication of user plane CIoT 5GS optimization to lower layers.

In WB-N1 mode, when the UE requests the lower layer to establish a RRC connection and the UE requests the use of control plane CIoT 5GS optimization or user plane CIoT 5GS optimization, the UE shall pass an indication of the requested CIoT 5GS optimizations to the lower layers.

5.3.22 Interaction between MICO mode with active time and extended idle mode DRX cycle

The UE can request the use of both MICO mode with active time and eDRX during a registration procedure but it is up to the network to decide to enable none, one of them or both (see 3GPP TS 23.501 [8]).

If the network accepts the use of both MICO mode with active time (see subclause 5.3.6) and eDRX (see subclause 5.3.16), the extended DRX parameters provided to the UE should allow for multiple paging occasions before the active timer expires.

5.3.23 Forbidden wireline access area

The AMF shall send a REGISTRATION REJECT, SERVICE REJECT or DEREGISTRATION REQUEST message over a wireline access network with the 5GMM cause #77 "wireline access area not allowed", if conditions specified in 3GPP TS 23.316 [6D] for AMF's enforcement of forbidden area are fulfilled.

If a REGISTRATION REJECT, SERVICE REJECT or DEREGISTRATION REQUEST message is received over a wireline access network with the 5GMM cause #77 "wireline access area not allowed":

- a) the 5G-RG shall not access 5GCN over the wireline access network until the 5G-RG is switched off, the UICC containing the USIM is removed, or the 5G-RG starts using another wireline access network;
- b) the W-AGF acting on behalf of the FN-CRG shall not access 5GCN until the W-AGF acting on behalf of the FN-CRG determines that the FN-CRG is switched off; and
- c) the W-AGF acting on behalf of the N5GC device shall not access 5GCN until the W-AGF acting on behalf of the FN-CRG determines that the FN-CRG serving the N5GC device is switched off.

5.3.24 WUS assistance

A UE supporting reception of WUS assistance information indicates its capability for reception of WUS assistance information during registration procedure (see 3GPP TS 23.501 [8]). The UE supporting WUS assistance information may include its UE paging probability information in the Requested WUS assistance information IE in the REGISTRATION REQUEST message (see 3GPP TS 23.501 [8]). The UE shall not include its UE paging probability information the UE has an active emergency PDU session.

The UE and the network may negotiate the UE paging probability information during a registration procedure when the UE does not have an emergency PDU session. The UE paging probability information is an assistance information used to determine the WUS group for paging UE (see 3GPP TS 23.501 [8], 3GPP TS 36.300 [25B]).

NOTE: The determination of UE paging probability information is up to UE implementation.

If a UE supporting WUS assistance information did not receive the Negotiated WUS assistance information IE during the last registration procedure due to an active emergency PDU session over 3GPP access, the UE shall initiate a registration procedure for mobility and periodic registration update to request WUS assistance information after the emergency PDU session is released over 3GPP access.

If the UE does not have an emergency PDU session and the network accepts the use of the WUS assistance information for the UE, the network determines the negotiated UE paging probability information for the UE based on the requested UE paging probability information, if any, local configuration or previous statistical information for the UE, and then indicates the negotiated UE paging probability information in the Negotiated WUS assistance information IE to the UE in the REGISTRATION ACCEPT message. The network shall store the negotiated UE paging probability information in the 5GMM context of the UE for paging.

The UE shall use WUS assistance information only if the UE received the Negotiated WUS assistance information IE during the last registration procedure. If the UE did not receive the Negotiated WUS assistance information IE during the last registration procedure, the UE shall not use WUS assistance.

If the network did not accept the request to use WUS assistance information, the network shall delete the stored negotiated UE paging probability information for the UE, if available.

When an emergency PDU session is successfully established after the UE received the Negotiated WUS assistance information IE during the last registration procedure, the UE and the AMF shall not use WUS assistance information until:

- the successful completion of the PDU session release procedure of the emergency PDU session;
- the UE receives WUS assistance information during a registration procedure with PDU session status IE or upon successful completion of a service request procedure, if the UE or the network locally releases the emergency PDU session; or
- the successful completion of the handover of the emergency PDU session to non-3GPP access.

5.3.25 Paging Early Indication with Paging Subgrouping Assistance

A UE may indicate its capability to support NR paging subgrouping during registration procedure when the UE:

- initiates a registration procedure with 5GS registration type IE not set to "emergency registration"; and
- does not have an active emergency PDU session.

If a UE supporting NR paging subgrouping did not indicate its capability to support NR paging subgrouping during the last registration procedure due to an active emergency PDU session over 3GPP access, the UE shall initiate a registration procedure for mobility and periodic registration update procedure to indicate its capability to support NR paging subgrouping after the emergency PDU session is released over 3GPP access.

If the UE indicates support of NR paging subgrouping the UE may include its paging probability information in the Requested PEIPS assistance information IE in the REGISTRATION REQUEST message. If the UE indicates support of NR paging subgrouping and the network supports and accepts the use of the PEIPS assistance information for the UE, the network provides to the UE the Negotiated PEIPS assistance information, including the Paging subgroup ID, in the REGISTRATION ACCEPT message or the CONFIGURATION UPDATE COMMAND message. The Paging subgroup ID is used to determine the NR paging subgroup for paging the UE. The network shall store the Paging subgroup ID in the 5GMM context of the UE.

The UE shall use PEIPS assistance information only if the UE received the Negotiated PEIPS assistance information IE during the last registration procedure. If the UE did not receive the Negotiated PEIPS assistance information IE during the last registration procedure, the UE shall delete any existing PEIPS assistance information received from the network.

If the network did not accept the request to use PEIPS assistance information during the registration procedure, the network shall delete the stored PEIPS assistance information for the UE, if available.

If the UE supports the use of the PEIPS assistance information and the network supports and accepts the use of the PEIPS assistance information, the network may provide the PEIPS assistance information to the UE by including the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message.

When an emergency PDU session is successfully established over 3GPP access after the UE received the Negotiated PEIPS assistance information IE during the last registration procedure, the UE and the AMF shall not use PEIPS assistance information until:

- the successful completion of the PDU session release procedure of the emergency PDU;
- the UE receives PEIPS assistance information during a registration procedure with PDU session status IE or upon successful completion of a service request procedure, if the UE or the network locally releases the emergency PDU session;
- the successful completion of handover of emergency PDU session to non-3GPP access; or
- the successful transfer of the emergency PDU session in 5GS to the EPS or ePDG connected to EPC.

5.4 5GMM common procedures

5.4.1 Primary authentication and key agreement procedure

5.4.1.1 General

The purpose of the primary authentication and key agreement procedure is to enable mutual authentication between the UE and the network and to provide keying material that can be used between the UE and network in subsequent security procedures, as specified in 3GPP TS 33.501 [24].

Two methods are defined:

- a) EAP based primary authentication and key agreement procedure.
- b) 5G AKA based primary authentication and key agreement procedure.

The UE and the AMF shall support the EAP based primary authentication and key agreement procedure and the 5G AKA based primary authentication and key agreement procedure.

5.4.1.2 EAP based primary authentication and key agreement procedure

5.4.1.2.1 General

The purpose of the EAP based primary authentication and key agreement procedure is to provide mutual authentication between the UE and the network and to agree on the keys K_{AUSF} , K_{SEAF} and K_{AMF} (see 3GPP TS 33.501 [24]).

Extensible authentication protocol (EAP) as specified in IETF RFC 3748 [34] enables authentication using various EAP methods.

EAP defines four types of EAP messages:

- a) an EAP-request message;
- b) an EAP-response message;
- c) an EAP-success message; and
- d) an EAP-failure message.

Several rounds of exchanges of an EAP-request message and a related EAP-response message can be required to achieve the authentication (see example in figure 5.4.1.2.1.1).

The EAP based primary authentication and key agreement procedure is always initiated and controlled by the network.

The EAP-request message, the ngKSI and the ABBA are transported from the network to the UE using the AUTHENTICATION REQUEST message of the EAP message reliable transport procedure.

The EAP-response message is transported from the UE to the network using the AUTHENTICATION RESPONSE message of the EAP message reliable transport procedure.

If the authentication of the UE completes successfully, the serving AMF intends to initiate a security mode control procedure after the EAP based primary authentication and key agreement procedure and the security mode control procedure intends to bring into use the partial native 5G NAS security context created by the EAP based primary authentication and key agreement procedure, then the EAP-success message and the ngKSI are transported from the network to the UE using the SECURITY MODE COMMAND message of the security mode control procedure (see subclause 5.4.2).

If the authentication of the UE completes successfully and the serving AMF does not intend to initiate a security mode control procedure bringing into use the partial native 5G NAS security context created by the EAP based primary authentication and key agreement procedure, then the EAP-success message, and the ngKSI are transported from the network to the UE using the AUTHENTICATION RESULT message of the EAP result message transport procedure.

NOTE 1: The serving AMF will not initiate a security mode control procedure after the EAP based primary authentication and key agreement procedure e.g. in case of AMF relocation during registration procedure.

If the authentication of the UE completes unsuccessfully, the EAP-failure message is transported from the network to the UE using the AUTHENTICATION RESULT message or the AUTHENTICATION REJECT message of the EAP result message transport procedure or in a response of the initial 5GMM procedure as part of which the EAP based primary authentication and key agreement procedure is performed.

The AMF shall set the authenticator retransmission timer specified in IETF RFC 3748 [34] subclause 4.3 to infinite value.

NOTE 2: The EAP message reliable transport procedure provides a reliable transport of EAP messages and therefore retransmissions at the EAP layer do not occur.

The AUSF and the AMF support exchange of EAP messages using N12.

The UE shall detect and handle any duplication of EAP message as specified in IETF RFC 3748 [34].

UE		AMF	
•	AUTHENTICATION REQUEST <eap-request a="" message=""></eap-request>		Start T3560
	AUTHENTICATION RESPONSE <eap-response a="" eap-request="" message="" to=""></eap-response>	→	Stop T3560
•	AUTHENTICATION REQUEST <eap-request b="" message=""></eap-request>		Start T3560
	AUTHENTICATION RESPONSE <eap-response b="" eap-request="" message="" to=""> </eap-response>		Stop T3560
←	I SECURITY MODE COMMAND <eap-success message=""></eap-success>		
←	OR <u>REGISTRATION REJECT</u> <eap-failure message=""></eap-failure>		
4	OR SERVICE REJECT <eap-failure message=""></eap-failure>		
•	OR REGISTRATION ACCEPT <eap-failure message=""></eap-failure>		
•	OR <u>SERVICE ACCEPT</u> <eap-failure message=""></eap-failure>		
•	OR <u>AUTHENTICATION RESULT</u> <eap-success eap-failure="" message="" or=""></eap-success>		
•	OR AUTHENTICATION REJECT <eap-failure message=""></eap-failure>		

Figure 5.4.1.2.1.1: EAP based primary authentication and key agreement procedure

5.4.1.2.2 EAP-AKA' related procedures

5.4.1.2.2.1 General

The UE shall support acting as EAP-AKA' peer as specified in IETF RFC 5448 [40]. The AUSF may support acting as EAP-AKA' server as specified in IETF RFC 5448 [40]. The AAA server of the Credentials Holder (CH) or the Default Credentials Server (DCS) may support acting as EAP-AKA' server as specified in IETF RFC 5448 [40].

The EAP-AKA' enables mutual authentication of the UE and the network.

The UE can reject the EAP-request/AKA'-challenge message sent by the network. The UE shall proceed with an EAP-request/AKA'-challenge message only if a USIM is present.

During a successful EAP based primary authentication and key agreement procedure, the CK and IK are computed by the USIM. CK and IK are then used by the ME as key material to generate an EMSK or MSK.

5.4.1.2.2.2 Initiation

In order to initiate the EAP based primary authentication and key agreement procedure using EAP-AKA', the AUSF or the AAA server of the CH or the DCS shall send an EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40]. The AUSF or the AAA server of the CH or the DCS shall set the AT_KDF_INPUT attribute of the EAP-request/AKA'-challenge message to the SNN. The SNN is in format described in subclause 9.12.1. The AUSF or the AAA server of the CH or the DCS may include AT_RESULT_IND attribute in the EAP-request/AKA'-challenge message.

The network shall select an ngKSI value. If an ngKSI is contained in an initial NAS message during a 5GMM procedure, the network shall select a different ngKSI value. The network shall send the selected ngKSI value to the UE along with each EAP message. The network shall send the ABBA value as described in subclause 9.11.3.10 to the UE along with the EAP request message and EAP-success message.

Upon receiving an EAP-request/AKA'-challenge message, the UE shall check whether the UE has a USIM, shall check the key derivation function indicated in AT_KDF attributes as specified in IETF RFC 5448 [40], and if the value of the Key derivation function field within the received AT_KDF attribute, is of value 1, shall check:

- a) whether the network name field of the AT_KDF_INPUT attribute is the SNN constructed according to subclause 9.12.1; and
- b) whether the network name field of the AT_KDF_INPUT attribute matches the PLMN identity or the SNPN identity of the selected SNPN saved in the UE.

When not operating in SNPN access operation mode, the PLMN identity the UE uses for the above network name check is as follows:

- a) when the UE moves from 5GMM-IDLE mode to 5GMM-CONNECTED mode, until the first handover, the UE shall use the PLMN identity of the selected PLMN; and
- b) after handover or inter-system change to N1 mode in 5GMM-CONNECTED mode:
 - 1) if the target cell is not a shared network cell, the UE shall use the PLMN identity received as part of the broadcast system information;
 - 2) if the target cell is a shared network cell and the UE has a valid 5G-GUTI, the UE shall use the PLMN identity that is part of the 5G-GUTI; and
 - 3) if the target cell is a shared network cell and the UE has a valid 4G-GUTI, but not a valid 5G-GUTI, the UE shall use the PLMN identity that is part of the 4G-GUTI.

When operating in SNPN access operation mode, the SNPN identity the UE uses for the above network name check is the SNPN identity of the selected SNPN.

5.4.1.2.2.3 UE successfully authenticates network

If a USIM is present and the SNN check is successful, the UE shall handle the EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40]. The USIM shall derive CK and IK and compute the authentication response (RES)

using the 5G authentication challenge data received from the ME, and pass RES to the ME. The ME shall derive CK' and IK' from CK and IK, and if the UE operates in SNPN access operation mode and the credentials in the USIM contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then derive MSK from CK' and IK' otherwise derive EMSK from CK' and IK'.

NOTE 1: When the UE is registering or registered for onboarding services in SNPN, credentials in the USIM do not contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure.

Furthermore, if the UE operates in SNPN access operation mode and the credentials in the USIM

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then the ME may generate a new K_{AUSF} from the MSK otherwise the ME may generate a new K_{AUSF} from the EMSK.

If the ME generates a new K_{AUSF} , the ME shall generate a new K_{SEAF} from the new K_{AUSF} , and the K_{AMF} from the ABBA received together with the EAP-request/AKA'-challenge message, and the new K_{SEAF} as described in 3GPP TS 33.501 [24], and create a partial native 5G NAS security context identified by the ngKSI value received together with the EAP-request/AKA'-challenge message in subclause 5.4.1.2.4.2 in the volatile memory of the ME. If the K_{AMF} and the partial native 5G NAS security context are created, the ME shall store the K_{AMF} in the created partial native 5G NAS security context, and shall send an EAP-response/AKA'-challenge message as specified in IETF RFC 5448 [40].

NOTE 2: Generation of the new K_{AUSF} and the new K_{SEAF} does not result into deletion of the valid K_{AUSF} and the valid K_{SEAF}, if any.

The ME shall not use the new K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received, until receipt of an EAP-success message.

If the EAP-request/AKA'-challenge message contains AT_RESULT_IND attribute, the UE may include AT_RESULT_IND attribute in the EAP-response/AKA'-challenge message as specified in IETF RFC 5448 [40].

5.4.1.2.2.4 Errors when handling EAP-request/AKA'-challenge message

If a USIM is present, the SNN check fails or the UE does not accept AUTN during handling of the EAP-request/AKA'challenge message as specified in IETF RFC 5448 [40], the UE shall send an EAP-response/AKA'-authentication-reject message as specified in IETF RFC 5448 [40].

If a USIM is present, the SNN check is successful but the UE detects that the sequence number in AUTN is not correct during handling of the EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40], the UE shall send an EAP-response/AKA'-synchronization-failure message as specified in IETF RFC 5448 [40].

If a USIM is present, the SNN check is successful, the sequence number in AUTN is correct and the UE detects another error during handling of the EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40], the UE shall send an EAP-response/AKA'-client-error message as specified in IETF RFC 5448 [40].

If a USIM is not present, the UE shall send an EAP-response/AKA'-client-error message as specified in IETF RFC 5448 [40].

For any of the above, the UE shall start timer T3520 when the AUTHENTICATION RESPONSE message containing the EAP-response message is sent. Furthermore, the UE shall stop any of the timers T3510, T3517 or T3521 (if they were running). Upon receiving an AUTHENTICATION REQUEST message with the EAP message IE containing an EAP-request/AKA'-challenge from the network, the UE shall stop timer T3520, if running, and then process the EAP-request/AKA'-challenge information as normal.

5.4.1.2.2.5 Network successfully authenticates UE

Upon reception of the EAP-response/AKA'-challenge message, if procedures for handling an EAP-response/AKA'- challenge message as specified in IETF RFC 5448 [40] are successful and:

a) the AUSF acts as the EAP-AKA' server, the AUSF shall generate EMSK, the K_{AUSF} from the EMSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24]; or

b) the AAA server of the CH or the DCS acts as the EAP-AKA' server, the AAA server of the CH or the DCS shall generate MSK as described in 3GPP TS 33.501 [24];

and:

- a) if the AUSF or the AAA server of the CH or the DCS included the AT_RESULT_IND attribute in the EAPrequest/AKA'-challenge message and the AT_RESULT_IND attribute is included in the corresponding EAPresponse/AKA'-challenge message, the AUSF or the AAA server of the CH or the DCS shall send an EAPrequest/AKA'-notification message as specified in IETF RFC 5448 [40]; or
- b) if the AUSF or the AAA server of the CH or the DCS:
 - 1) included the AT_RESULT_IND attribute in the EAP-request/AKA'-challenge message and the AT_RESULT_IND attribute is not included in the EAP-response/AKA'-challenge message; or
 - 2) did not include the AT_RESULT_IND attribute in the EAP-request/AKA'-challenge message;

then the AUSF or the AAA server of the CH or the DCS shall send an EAP-success message as specified in IETF RFC 5448 [40] and shall consider the procedure complete.

- NOTE 1: When the AAA server of the CH or the DCS acts as the EAP-AKA' server, the AAA server of the CH or the DCS provides (via the NSSAAF) the MSK and the SUPI to the AUSF. Upon reception of the MSK, the AUSF generates the K_{AUSF} from the MSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24].
- NOTE 2: The AUSF provides the K_{SEAF} and optionally the SUPI (unless the SEAF provided the AUSF with the SUPI before) to the SEAF as described in 3GPP TS 33.501 [24]. Upon reception of the K_{SEAF} and optionally the SUPI, the SEAF generates the K_{AMF} based on the ABBA, the K_{SEAF} and the SUPI as described in 3GPP TS 33.501 [24] and provides ngKSI and the K_{AMF} to the AMF. Upon reception of the ngKSI and the K_{AMF}, the AMF creates a partial native 5G NAS security context identified by the ngKSI and stores the K_{AMF} in the created partial native 5G NAS security context.

5.4.1.2.2.6 UE handling EAP-AKA' notification message

Upon receiving an EAP-request/AKA'-notification message, the UE shall send an EAP-response/AKA'-notification message as specified in IETF RFC 5448 [40].

5.4.1.2.2.6A EAP based Identification initiation by the network

If the AUSF or the AAA server of the CH or the DCS decides to initiate the EAP based identification procedure, the AUSF or the AAA server of the CH or the DCS shall send an EAP-Request/Identity or EAP-Request/AKA'-Identity message as specified in IETF RFC 5448 [40].

The AMF shall encapsulate the EAP-Request/Identity or EAP-Request/AKA'-Identity message in the AUTHENTICATION REQUEST message and send it to the UE.

5.4.1.2.2.6B EAP based Identification response by the UE

Upon receipt of the AUTHENTICATION REQUEST message with EAP-Request/Identity message the UE shall send an AUTHENTICATION RESPONSE message with EAP-Response/Identity to the network. In the EAP-Response/Identity message, the UE shall provide the requested identity according to 3GPP TS 33.501 [24] annex F.2, in the UE identity in the EAP-Response/Identity message as specified in IETF RFC 5448 [40].

Upon receipt of the AUTHENTICATION REQUEST message with EAP-Request/AKA'-Identity message the UE shall send an AUTHENTICATION RESPONSE message with EAP-Response/AKA'-Identity to the network. Based on the attribute received in the EAP-Request/AKA'-Identity, the UE shall provide the requested identity according to 3GPP TS 33.501 [24] annex F.2, in the EAP-Response/AKA'-Identity message, as specified in IETF RFC 5448 [40].

If the EAP-Request/AKA'-Identity carries the AT_PERMANENT_REQ, the UE shall respond with EAP-Response/AKA'-Client-Error with the error code "unable to process packet".

5.4.1.2.2.7 Network sending EAP-success message

Upon reception of the EAP-response/AKA'-notification message, if earlier procedures for handling an EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40] were successful, the AUSF or the AAA server of the CH or the DCS shall send an EAP-success message as specified in IETF RFC 5448 [40] and shall consider the procedure complete.

NOTE: The AUSF provides the K_{SEAF} to the SEAF. Upon reception of the K_{SEAF} , the SEAF generates the K_{AMF} based on the ABBA and the K_{SEAF} as described in 3GPP TS 33.501 [24], and provides ngKSI and the K_{AMF} to the AMF. Upon reception of the ngKSI and the K_{AMF} , the AMF creates a partial native 5G NAS security context identified by the ngKSI, and stores the K_{AMF} in the created partial native 5G NAS security context.

5.4.1.2.2.8 UE handling EAP-success message

Upon receiving an EAP-success message, the ME shall:

- a) delete the valid K_{AUSF} and the valid K_{SEAF}, if any;
- b) if the ME has not generated a new K_{AUSF} and a new K_{SEAF} and has not created a partial native 5G NAS security context as described in subclause 5.4.1.2.2.3:
 - if the UE operates in SNPN access operation mode and the credentials in the USIM contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then generate a new K_{AUSF} from the MSK otherwise generate a new K_{AUSF} from the EMSK;
- NOTE: When the UE is registering or registered for onboarding services in SNPN, credentials in the USIM do not contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure.
 - 2) generate a new K_{SEAF} from the new K_{AUSF}, and the K_{AMF} from the ABBA that was received with the EAPsuccess message, and the new K_{SEAF} as described in 3GPP TS 33.501 [24];
 - 3) create a partial native 5G NAS security context identified by the ngKSI value in the volatile memory of the ME; and
 - 4) store the K_{AMF} in the created partial native 5G NAS security context; and
- c) consider the new K_{AUSF} to be the valid K_{AUSF}, and the new K_{SEAF} to be the valid K_{SEAF}, reset the SOR counter and the UE parameter update counter to zero, and store the valid K_{AUSF}, the valid K_{SEAF}, the SOR counter and the UE parameter update counter as specified in annex C, and use the valid K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received.

The UE shall consider the procedure complete.

5.4.1.2.2.9 Network not successfully authenticates UE

Upon reception of the EAP-response/AKA'-challenge message, if procedures for handling an EAP-response/AKA'challenge message as specified in IETF RFC 5448 [40] are not successful, the AUSF or the AAA server of the CH or the DCS shall send an EAP-request/AKA'-notification message that implies failure as specified in IETF RFC 5448 [40].

5.4.1.2.2.10 Network sending EAP-failure message

Upon reception of the EAP-response/AKA'-notification message, if earlier procedures for handling an EAP-request/AKA'-challenge message as specified in IETF RFC 5448 [40] were not successful, the AUSF or the AAA server of the CH or the DCS shall send an EAP-failure message as specified in IETF RFC 5448 [40] and shall consider the procedure complete.

If the authentication response (RES) returned by the UE in the AT_RES attribute of the EAP-response/AKA'-challenge message is not valid, the network handling depends upon the type of identity used by the UE in the initial NAS message, that is:

- if the 5G-GUTI was used; or

- if the SUCI was used.

If the 5G-GUTI was used, the network should transport the EAP-failure message in the AUTHENTICATION RESULT message of the EAP result message transport procedure, initiate an identification procedure to retrieve SUCI from the UE and restart the EAP based primary authentication and key agreement procedure with the received SUCI.

If the SUCI was used for identification in the initial NAS message or in a restarted EAP based primary authentication and key agreement procedure, or the network decides not to initiate the identification procedure to retrieve SUCI from the UE after an unsuccessful EAP based primary authentication and key agreement procedure, the network should transport the EAP-failure message in an AUTHENTICATION REJECT message of the EAP result message transport procedure.

Depending on local requirements or operator preference for emergency services, if the UE initiates a registration procedure with 5GS registration type IE set to "emergency registration" and the AMF is configured to allow emergency registration without user identity, the AMF needs not follow the procedures specified for transporting the EAP-failure message in the AUTHENTICATION REJECT message of the EAP result message transport procedure in the present subclause. The AMF may include the EAP-failure message in a response of the current 5GMM specific procedure or in the AUTHENTICATION RESULT of the EAP result message transport procedure.

5.4.1.2.2.11 UE handling EAP-failure message

Upon receiving an EAP-failure message, the UE shall delete the partial native 5G NAS security context and shall delete the new K_{AUSF} and the new K_{SEAF} , if any were created as described in subclause 5.4.1.2.2.3.

The UE shall consider the procedure complete.

If the EAP-failure message is received in an AUTHENTICATION REJECT message:

- 1) if the AUTHENTICATION REJECT message has been successfully integrity checked by the NAS:
 - the UE shall set the update status to 5U3 ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI;

In case of PLMN, the USIM shall be considered invalid until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN until switching off or the UICC containing the USIM is removed.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the entry until switching off or the UICC containing the USIM is removed.

If the UE is registered for onboarding services in SNPN or is performing initial registration for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5];

- if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall set:
 - i) the counter for "SIM/USIM considered invalid for GPRS services" events, the counter for "USIM considered invalid for 5GS services over non-3GPP access" events, and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
 - ii) the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN;

to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value; and

- if the UE is operating in single-registration mode, the UE shall handle EMM parameters, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and
- 2) if the AUTHENTICATION REJECT message is received without integrity protection, the UE shall start timer T3247 with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running (see subclause 5.3.20).

Additionally, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall:

- a) if the AUTHENTICATION REJECT message is received over 3GPP access, and the counter for "SIM/USIM considered invalid for GPRS services" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-a) of subclause 5.3.20.2 (if the UE is not operating in SNPN access operation mode) or list item a)-1) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3;
- b) if the AUTHENTICATION REJECT message is received over non-3GPP access, and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-b) of subclause 5.3.20.2 (if the UE is not operating in SNPN access operation mode) or list item a)-2) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3;
- c) otherwise:
 - i) if the AUTHENTICATION REJECT message is received over 3GPP access:
 - The UE shall set the update status for 3GPP access to 5U3 ROAMING NOT ALLOWED, delete for 3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.
 - In case of PLMN, the UE shall consider the USIM as invalid for 5GS services via 3GPP access and invalid for non-EPS service until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN via 3GPP access until switching off or the UICC containing the USIM is removed.

In case of SNPN, if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the entry via 3GPP access until switching off or the UICC containing the USIM is removed.

- The UE shall set:
 - the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
 - the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN;

to UE implementation-specific maximum value.

- If the UE is operating in single-registration mode, the UE shall handle 4G-GUTI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and
- ii) if the AUTHENTICATION REJECT message is received over non-3GPP access:
 - the UE shall set the update status for non-3GPP access to 5U3 ROAMING NOT ALLOWED, delete for non-3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI;
 - in case of PLMN, the UE shall consider the USIM as invalid for 5GS services via non-3GPP access until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for non-3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN and for non-3GPP access until switching off or the UICC containing the USIM is removed; and

- the UE shall set:
 - the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN; or
 - the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall:

- a) if the SNPN-specific attempt counter for the SNPN sending the AUTHENTICATION REJECT message has a value less than a UE implementation-specific maximum value, increment the SNPN-specific attempt counter for the SNPN; or
- b) otherwise, the UE shall set the update status to 5U3.ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI, and ngKSI, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the AUTHENTICATION REJECT message is received by the UE, the UE shall abort any 5GMM signalling procedure, stop any of the timers T3510, T3517, T3519 or T3521 (if they were running), enter state 5GMM-DEREGISTERED and delete any stored SUCI.

5.4.1.2.2.12 Abnormal cases in the UE

The following abnormal cases can be identified:

a) EAP-request/AKA'-challenge message with the key derivation function indicated in AT_KDF attributes set to a value other than 1.

The UE shall act as specified in IETF RFC 5448 [40] subclause 3.2 for the case when the AUTN had been incorrect.

5.4.1.2.3 EAP-TLS related procedures

5.4.1.2.3.1 General

The UE may support acting as EAP-TLS peer as specified in 3GPP TS 33.501 [24]. The AUSF may support acting as EAP-TLS server as specified in 3GPP TS 33.501 [24]. The AAA server of the CH or the DCS may support acting as EAP server of such EAP method as specified in 3GPP TS 23.501 [8].

The EAP-TLS enables mutual authentication of the UE and the network.

When initiating an EAP based primary authentication and key agreement procedure using EAP-TLS, the network shall select an ngKSI value. If an ngKSI is contained in an initial NAS message during a 5GMM procedure, the network shall

select a different ngKSI value. The network shall send the selected ngKSI value to the UE along with each EAP message. The network shall send the ABBA value as described in subclause 9.11.3.10 to the UE along with the EAP-request message and EAP-success message.

When the EAP based primary authentication and key agreement procedure uses EAP-TLS:

- a) if the UE operates in SNPN access operation mode and:
 - 1) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
 - 2) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then the ME shall generate MSK as described in 3GPP TS 33.501 [24] otherwise the ME shall generate EMSK as described in 3GPP TS 33.501 [24];

- b) if the AUSF acts as the EAP-TLS server, the AUSF shall generate EMSK as described in 3GPP TS 33.501 [24]; and
- c) if the AAA server of the CH or the DCS acts as the EAP-TLS server, the AAA server of the CH or the DCS shall generate MSK as described in 3GPP TS 33.501 [24].

When handling of an EAP-request message results into generation of MSK or EMSK, if the UE operates in SNPN access operation mode and:

- a) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
- b) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure, then the ME may generate a new K_{AUSF} from the MSK otherwise the ME may generate a new K_{AUSF} from the EMSK.

If the ME generates a new K_{AUSF} , the ME shall generate a new K_{SEAF} from the new K_{AUSF} , and the K_{AMF} from the ABBA received together with the EAP-request message, and the new K_{SEAF} as described in 3GPP TS 33.501 [24], and create a partial native 5G NAS security context identified by the ngKSI value received together with the EAP-request message in subclause 5.4.1.2.4.2, in the volatile memory of the ME. If the K_{AMF} and the partial native 5G NAS security context are created partial native 5G NAS security context.

NOTE 1: Generation of the new K_{AUSF} and the new K_{SEAF} does not result into deletion of the valid K_{AUSF} and the valid K_{SEAF} , if any.

The ME shall not use the new K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received, until receipt of an EAP-success message.

When the AUSF acts as the EAP-TLS server and handling of an EAP response message results into generation of EMSK, the AUSF shall generate the K_{AUSF} from the EMSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24].

- NOTE 2: When the AAA server of the CH or the DCS acts as the EAP-TLS server, the AAA server of the CH or the DCS provides (via the NSSAAF) the MSK and the SUPI to the AUSF. Upon reception of the MSK, the AUSF generates the K_{AUSF} from the MSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24].
- NOTE 3: The AUSF provides the K_{SEAF} and optionally the SUPI (unless the SEAF provided the AUSF with the SUPI before) to the SEAF as described in 3GPP TS 33.501 [24]. Upon reception of the K_{SEAF} and optionally the SUPI, the SEAF generates the K_{AMF} based on the ABBA, the K_{SEAF} and the SUPI as described in 3GPP TS 33.501 [24], and provides ngKSI and the K_{AMF} to the AMF. Upon reception of the ngKSI and the K_{AMF} , the AMF creates a partial native 5G NAS security context identified by the ngKSI, and stores the K_{AMF} in the created partial native 5G NAS security context.

If the UE does not accept the server certificate of the network, the UE shall start timer T3520 when the AUTHENTICATION RESPONSE message containing the EAP-response message is sent. Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon receiving an AUTHENTICATION REQUEST message with the EAP message IE containing an EAP-request message from the network, the UE shall stop timer T3520, if running, and then process the EAP-request message as normally.

If the network does not accept the client certificate of the UE, the network handling depends upon the type of identity used by the UE in the initial NAS message, that is:

- if the 5G-GUTI was used; or
- if the SUCI was used.

If the 5G-GUTI was used, the network should transport the EAP-failure message in the AUTHENTICATION RESULT message of the EAP result message transport procedure, initiate an identification procedure to retrieve SUCI from the UE and restart the EAP based primary authentication and key agreement procedure with the received SUCI.

If the SUCI was used for identification in the initial NAS message or in a restarted EAP based primary authentication and key agreement procedure, or the network decides not to initiate the identification procedure to retrieve SUCI from the UE after an unsuccessful the EAP based primary authentication and key agreement procedure, the network should transport the EAP-failure message in an AUTHENTICATION REJECT message of the EAP result message transport procedure.

Depending on local requirements or operator preference for emergency services, if the UE initiates a registration procedure with 5GS registration type IE set to "emergency registration" and the AMF is configured to allow emergency registration without user identity, the AMF needs not follow the procedures specified for transporting the EAP-failure message in the AUTHENTICATION REJECT message of the EAP result message transport procedure in the present subclause. The AMF may include the EAP-failure message in a response of the current 5GMM specific procedure or in the AUTHENTICATION RESULT of the EAP result message transport procedure.

If the EAP-failure message is received in an AUTHENTICATION REJECT message:

- a) if the AUTHENTICATION REJECT message has been successfully integrity checked by the NAS:
 - 1) the UE shall set the update status to 5U3 ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.

In case of PLMN, the USIM shall be considered invalid until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid until the UE is switched off or the entry is updated;

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid until the UE is switched off or the entry is updated.

If the UE is registered for onboarding services in SNPN or is performing initial registration for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5];

- 2) if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall set:
 - the counter for "SIM/USIM considered invalid for GPRS services" events, the counter for "USIM considered invalid for 5GS services over non-3GPP access" events, and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
 - ii) the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN;

- NOTE 4: The term "non-3GPP access" used in the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events, is used to express access to SNPN services via a PLMN.
 - to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value; and

- 3) if the UE is operating in single-registration mode, the UE shall handle EMM parameters, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and
- b) if the AUTHENTICATION REJECT message is received without integrity protection, the UE shall start timer T3247 with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running (see subclause 5.3.20).

Additionally, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall:

- if the AUTHENTICATION REJECT message is received over 3GPP access, and the counter for "SIM/USIM considered invalid for GPRS services" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-a) of subclause 5.3.20.2 (if the UE is not SNPN enabled or is not operating in SNPN access operation mode) or list item a) 1) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3;
- 2) if the AUTHENTICATION REJECT message is received over non-3GPP access, and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-b) of subclause 5.3.20.2 (if the UE is not operating in SNPN access operation mode) or list item a)-2) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3; or
- 3) otherwise:
 - i) if the AUTHENTICATION REJECT message is received over 3GPP access:
 - A) the UE shall set the update status for 3GPP access to 5U3 ROAMING NOT ALLOWED, delete for 3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services via 3GPP access and invalid for non-EPS service until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 3GPP access until the UE is switched off or the entry is updated;

In case of SNPN, if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off or the entry is updated;

- B) the UE shall set:
 - the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
 - the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN;

to UE implementation-specific maximum value; and

- C) If the UE is operating in single-registration mode, the UE shall handle 4G-GUTI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and
- ii) if the AUTHENTICATION REJECT message is received over non-3GPP access:
 - A) the UE shall set the update status for non-3GPP access to 5U3 ROAMING NOT ALLOWED, delete for non-3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI. In case of PLMN, the USIM shall be considered invalid for 5GS services via non-3GPP access until switching off the UE or the UICC containing the USIM is removed. In case of SNPN, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid for the entry is updated; and
 - B) the UE shall set the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall:

- if the SNPN-specific attempt counter for the SNPN sending the AUTHENTICATION REJECT message has a value less than a UE implementation-specific maximum value, increment the SNPN-specific attempt counter for the SNPN; or
- 2) otherwise, the UE shall set the update status to 5U3.ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI, and ngKSI, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the AUTHENTICATION REJECT message is received by the UE, the UE shall abort any 5GMM signalling procedure, stop any of the timers T3510, T3517, T3519 or T3521 (if they were running), enter state 5GMM-DEREGISTERED and delete any stored SUCI.

Upon receiving an EAP-success message, the ME shall:

- a) delete the valid K_{AUSF} and the valid K_{SEAF}, if any;
- b) if the ME has not generated a new K_{AUSF} and a new K_{SEAF} and has not created a partial native 5G NAS security context when handling the EAP-request message which resulted into generation of EMSK or MSK as described above:
 - 1) if the UE operates in SNPN access operation mode and:
 - i) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
 - ii) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then generate a new K_{AUSF} from the MSK otherwise generate a new K_{AUSF} from the EMSK;

- 2) generate a new K_{SEAF} from the new K_{AUSF}, and the K_{AMF} from the ABBA that was received with the EAPsuccess message, and the new K_{SEAF} as described in 3GPP TS 33.501 [24];
- 3) create a partial native 5G NAS security context identified by the ngKSI value in the volatile memory of the ME; and
- 4) store the K_{AMF} in the created partial native 5G NAS security context; and
- c) consider the new K_{AUSF} to be the valid K_{AUSF}, and the new K_{SEAF} to be the valid K_{SEAF}, reset the SOR counter and the UE parameter update counter to zero, store the valid K_{AUSF}, the valid K_{SEAF}, the SOR counter and the UE parameter update counter as specified in annex C, and use the valid K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received.

The UE shall consider the procedure complete.

Upon receiving an EAP-failure message, the UE shall delete the partial native 5G NAS security context and shall delete the new K_{AUSF} and the new K_{SEAF} , if any were created when handling the EAP-request message which resulted into generation of EMSK or MSK as described above.

The UE shall consider the procedure complete.

5.4.1.2.3A Procedures related to EAP methods other than EAP-AKA' and EAP-TLS

5.4.1.2.3A.1 General

This subclause applies when an EAP method:

- a) supporting mutual authentication;
- b) supporting EMSK or MSK generation; and
- c) other than EAP-AKA' and EAP-TLS;

is used for primary authentication and key agreement in an SNPN.

The UE may support acting as EAP peer of such EAP method as specified in 3GPP TS 33.501 [24]. The AUSF may support acting as EAP server of such EAP method as specified in 3GPP TS 33.501 [24]. The AAA server of the CH or the DCS may support acting as EAP server of such EAP method as specified in 3GPP TS 23.501 [8].

When initiating an EAP based primary authentication and key agreement procedure using such EAP method, the network shall select an ngKSI value. If an ngKSI is contained in an initial NAS message during a 5GMM procedure, the network shall select a different ngKSI value. The network shall send the selected ngKSI value to the UE along with each EAP message. The network shall send the ABBA value as described in subclause 9.11.3.10 to the UE along with the EAP-request message and EAP-success message.

When the EAP based primary authentication and key agreement procedure uses such EAP method:

a) if:

- 1) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
- 2) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then the ME shall generate MSK as described in 3GPP TS 33.501 [24] otherwise the ME shall generate EMSK as described in 3GPP TS 33.501 [24];

- b) if the AUSF acts as the EAP server, the AUSF shall generate EMSK as described in 3GPP TS 33.501 [24]; and
- c) if the AAA server of the CH or the DCS acts as the EAP server, the AAA server of the CH or the DCS shall generate MSK as described in 3GPP TS 33.501 [24].

When handling of an EAP-request message results into generation of MSK or EMSK, if:

- a) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
- b) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then the ME may generate a new K_{AUSF} from the MSK otherwise the ME may generate a new K_{AUSF} from the EMSK.

If the ME generates a new K_{AUSF} , the ME shall generate a new K_{SEAF} from the new K_{AUSF} , and the K_{AMF} from the ABBA received together with the EAP-request message, and the new K_{SEAF} as described in 3GPP TS 33.501 [24], and

create a partial native 5G NAS security context identified by the ngKSI value received together with the EAP-request message in subclause 5.4.1.2.4.2, in the volatile memory of the ME. If the K_{AMF} and the partial native 5G NAS security context are created, the ME shall store the K_{AMF} in the created partial native 5G NAS security context.

NOTE 1: Generation of the new K_{AUSF} and the new K_{SEAF} does not result into deletion of the valid K_{AUSF} and the valid K_{SEAF} , if any.

The ME shall not use the new K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received, until receipt of an EAP-success message.

When the AUSF acts as the EAP server and handling of an EAP response message results into generation of EMSK, the AUSF shall generate the K_{AUSF} from the EMSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24].

- NOTE 2: When the AAA server of the CH or the DCS acts as the EAP server and handling of an EAP response message results into generation of MSK, the AAA server of the CH or the DCS provides (via the NSSAAF) the MSK and the SUPI to the AUSF. Upon reception of the MSK, the AUSF generates the K_{AUSF} from the MSK, and the K_{SEAF} from the K_{AUSF} as described in 3GPP TS 33.501 [24].
- NOTE 3: The AUSF provides the K_{SEAF} and optionally the SUPI (unless the SEAF provided the AUSF with the SUPI before) to the SEAF as described in 3GPP TS 33.501 [24]. Upon reception of the K_{SEAF} and optionally the SUPI, the SEAF generates the K_{AMF} based on the ABBA, the K_{SEAF} and the SUPI as described in 3GPP TS 33.501 [24], and provides ngKSI and the K_{AMF} to the AMF. Upon reception of the ngKSI and the K_{AMF} , the AMF creates a partial native 5G NAS security context identified by the ngKSI, and stores the K_{AMF} in the created partial native 5G NAS security context.

If the UE fails to authenticate the network, the UE shall start timer T3520 when the AUTHENTICATION RESPONSE message containing the EAP-response message is sent. Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon receiving an AUTHENTICATION REQUEST message with the EAP message IE containing an EAP-request message from the network, the UE shall stop timer T3520, if running, and then process the EAP-request message as normally.

If the network fails to authenticate the UE, the network handling depends upon the type of identity used by the UE in the initial NAS message, that is:

- if the 5G-GUTI was used; or
- if the SUCI was used.

If the 5G-GUTI was used, the network should transport the EAP-failure message in the AUTHENTICATION RESULT message of the EAP result message transport procedure, initiate an identification procedure to retrieve SUCI from the UE and restart the EAP based primary authentication and key agreement procedure with the received SUCI.

If the SUCI was used for identification in the initial NAS message or in a restarted EAP based primary authentication and key agreement procedure, or the network decides not to initiate the identification procedure to retrieve SUCI from the UE after an unsuccessful the EAP based primary authentication and key agreement procedure, the network should transport the EAP-failure message in an AUTHENTICATION REJECT message of the EAP result message transport procedure.

If the EAP-failure message is received in an AUTHENTICATION REJECT message:

- a) if the AUTHENTICATION REJECT message has been successfully integrity checked by the NAS:
 - 1) the UE shall set the update status to 5U3 ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid until the UE is switched off or the entry is updated;

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid until the UE is switched off or the entry is updated.

In case of SNPN, if the UE is registered for onboarding services in SNPN or is performing initial registration for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]; and

- 2) if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN to UE implementation-specific maximum value.
- NOTE 4: The term "non-3GPP access" used in the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events, is used to express access to SNPN services via a PLMN.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value; and

b) if the AUTHENTICATION REJECT message is received without integrity protection, the UE shall start timer T3247 with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running (see subclause 5.3.20).

Additionally, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall:

- if the AUTHENTICATION REJECT message is received over 3GPP access, and the counter for "the entry for the current SNPN considered invalid for 3GPP access" events has a value less than a UE implementationspecific maximum value, proceed as specified in list item a) 1) of subclause 5.3.20.3 for the case that the 5GMM cause value received is #3;
- 2) if the AUTHENTICATION REJECT message is received over non-3GPP access, and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events has a value less than a UE implementation-specific maximum value, proceed as specified in list item a)-2) of subclause 5.3.20.3 for the case that the 5GMM cause value received is #3; or
- 3) otherwise:
 - i) if the AUTHENTICATION REJECT message is received over 3GPP access:
 - the UE shall set the update status for 3GPP access to 5U3 ROAMING NOT ALLOWED, delete for 3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI;

In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid for 3GPP access until the UE is switched off or the entry is updated;

In case of SNPN, if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid until the UE is switched off or the entry is updated; and

- the UE shall set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events to UE implementation-specific maximum value; and
- ii) if the AUTHENTICATION REJECT message is received over non-3GPP access:
 - the UE shall set the update status for non-3GPP access to 5U3 ROAMING NOT ALLOWED, delete for non-3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI. The entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid for non-3GPP access until the UE is switched off or the entry is updated; and
 - the UE shall set the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value.
- NOTE 5: The AUTHENTICATION REJECT message "received over non-3GPP access" in this subclause refers to an AUTHENTICATION REJECT message received via a PLMN when the UE attempts to access SNPN services via a PLMN.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall:

- 1) if the SNPN-specific attempt counter for the SNPN sending the AUTHENTICATION REJECT message has a value less than a UE implementation-specific maximum value, increment the SNPN-specific attempt counter for the SNPN; or
- 2) otherwise, the UE shall set the update status to 5U3.ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI, and ngKSI, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the AUTHENTICATION REJECT message is received by the UE, the UE shall abort any 5GMM signalling procedure, stop any of the timers T3510, T3517, T3519 or T3521 (if they were running), enter state 5GMM-DEREGISTERED and delete any stored SUCI.

Upon receiving an EAP-success message, the ME shall:

- a) delete the valid K_{AUSF} and the valid K_{SEAF} , if any;
- b) if the ME has not generated a new K_{AUSF} and a new K_{SEAF} and has not created a partial native 5G NAS security context when handling the EAP-request message which resulted into generation of EMSK as described above:

1) if:

- i) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
- ii) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN;

contain an indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure then generate a new K_{AUSF} from the MSK otherwise generate a new K_{AUSF} from the EMSK;

- 2) generate a new K_{SEAF} from the new K_{AUSF} , and the K_{AMF} from the ABBA that was received with the EAPsuccess message, and the K_{SEAF} as described in 3GPP TS 33.501 [24];
- 3) create a partial native 5G NAS security context identified by the ngKSI value in the volatile memory of the ME; and
- 4) store the K_{AMF} in the created partial native 5G NAS security context; and
- c) consider the new K_{AUSF} to be the valid K_{AUSF}, and the new K_{SEAF} to be the valid K_{SEAF}, reset the SOR counter and the UE parameter update counter to zero, store the valid K_{AUSF}, the valid K_{SEAF}, the SOR counter and the UE parameter update counter as specified in annex C, and use the valid K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received.

The UE shall consider the procedure complete.

Upon receiving an EAP-failure message, the UE shall delete the partial native 5G NAS security context and shall delete the new K_{AUSF} and the new K_{SEAF} , if any were created when handling the EAP-request message which resulted into generation of EMSK or MSK as described above.

The UE shall consider the procedure complete.

5.4.1.2.3A.2 EAP-TTLS with two phases of authentication

The UE may support acting as EAP peer of EAP-TTLS with two phases of authentication as specified in 3GPP TS 33.501 [24] and acting as peer of a legacy authentication protocol as specified in 3GPP TS 33.501 [24]. The AUSF may support acting as EAP server of EAP-TTLS with two phases of authentication as specified in 3GPP TS 33.501 [24]. The AAA server of CH may support acting a server of a legacy authentication protocol as specified in 3GPP TS 33.501 [24].

When EAP-TTLS with two phases of authentication as specified in 3GPP TS 33.501 [24] is used for primary authentication and key agreement in an SNPN:

- a) requirements in subclause 5.4.1.2.3A.1 shall apply in addition to requirements specified in 3GPP TS 33.501 [24] annex U.
- b) indication to use MSK for derivation of K_{AUSF} after success of primary authentication and key agreement procedure is not included in:
 - 1) the default UE credentials for primary authentication, if the UE is registering or registered for onboarding services in SNPN; or
 - 2) credentials in the selected entry of the "list of subscriber data", if the UE is not registering or registered for onboarding services in SNPN.
- c) the SUPI of the UE is in the form of a SUPI with the SUPI format "network specific identifier" containing a network-specific identifier.
- NOTE: Support of EAP-TTLS with two phases of authentication is based on the informative requirements as specified in 3GPP TS 33.501 [24].

5.4.1.2.3B Procedures related to EAP methods used for primary authentication of an N5GC device

5.4.1.2.3B.1 General

This subclause applies when an EAP method:

- a) supporting mutual authentication; and
- b) other than EAP-AKA',

is used for primary authentication of an N5GC device, when an W-AGF supports acting on behalf of the N5GC device, the AMF supports serving the W-AGF acting on behalf of the N5GC device and the AUSF supports authentication of the N5GC device. EAP-TLS is an example of such EAP method.

NOTE 1: Neither the N5GC device nor the AUSF derive any 5G related keys during or after the primary authentication.

The AUSF supporting authentication of the N5GC device shall support acting as EAP server of at least one such EAP method as specified in annex O of 3GPP TS 33.501 [24].

The N5GC device shall support acting as EAP peer of at least one such EAP method as specified in annex O of 3GPP TS 33.501 [24], which is also supported by the AUSF.

The W-AGF acting on behalf of the N5GC device provides to the N5GC device an EAP-request message, an EAP-success message or an EAP-failure message received from the network according to subclause 5.4.1.2.1 and sends to the network according to subclause 5.4.1.2.1 an EAP-response provided by the N5GC device. The N5GC device can inform the W-AGF acting on behalf of the N5GC device that the N5GC device fails to authenticate the network. Details of communication between the N5GC device and the W-AGF acting on behalf of the N5GC device are out of scope of this specification.

When initiating an EAP based primary authentication and key agreement procedure using such EAP method, the network shall select an ngKSI value. The network shall send the selected ngKSI value to the W-AGF acting on behalf of the N5GC device along with each EAP message. The network shall send the ABBA value as described in subclause 9.11.3.10 to the W-AGF acting on behalf of the N5GC device along with the EAP-request message and EAP-success message. The W-AGF acting on behalf of the N5GC device shall not forward the ngKSI value or the ABBA value to the N5GC device.

NOTE 2: The network provides the ngKSI value and the ABBA value since the ngKSI IE and the ABBA IE are mandatory IEs in AUTHENTICATION REQUEST message. The W-AGF acting on behalf of the N5GC device does not use the ngKSI value or the ABBA value provided by the network.

If the N5GC device fails to authenticate the network, the W-AGF acting on behalf of the N5GC device shall start timer T3520 when the AUTHENTICATION RESPONSE message containing the EAP-response message is sent.

Furthermore, the W-AGF acting on behalf of the N5GC device shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon receiving an AUTHENTICATION REQUEST message with the EAP message IE containing an EAP-request message from the network, the W-AGF acting on behalf of the N5GC device shall stop timer T3520, if running, and then provides the EAP-request message to the N5GC device as normally.

If the network fails to authenticate the N5GC device, the network handling depends upon the type of identity used by the W-AGF acting on behalf of the N5GC device in the initial NAS message, that is:

- a) if the 5G-GUTI was used; or
- b) if the SUCI was used.

If the 5G-GUTI was used, the network should transport the EAP-failure message in the AUTHENTICATION RESULT message of the EAP result message transport procedure, initiate an identification procedure to retrieve SUCI from the W-AGF acting on behalf of the N5GC device and restart the EAP based primary authentication and key agreement procedure with the received SUCI.

If the SUCI was used for identification in the initial NAS message or in a restarted EAP based primary authentication and key agreement procedure, or the network decides not to initiate the identification procedure to retrieve SUCI from the W-AGF acting on behalf of the N5GC device after an unsuccessful EAP based primary authentication and key agreement procedure, the network should transport the EAP-failure message in an AUTHENTICATION REJECT message of the EAP result message transport procedure.

If the EAP-failure message is received in an AUTHENTICATION REJECT message, the W-AGF acting on behalf of the N5GC device shall start timer T3247 with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running (see subclause 5.3.20). Additionally, the W-AGF acting on behalf of the N5GC device shall:

- a) if the counter for "USIM considered invalid for 5GS services over non-3GPP access" events has a value less than a W-AGF implementation-specific maximum value, proceed as specified in list item 1)-b) of subclause 5.3.20.2 for the case that the 5GMM cause value received is #3; or
- b) otherwise, set the update status to 5U3 ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI. The USIM shall be considered invalid for 5GS services via non-3GPP access until switching off or the UICC containing the USIM is removed.

If the AUTHENTICATION REJECT message is received by the W-AGF acting on behalf of the N5GC device, the W-AGF acting on behalf of the N5GC device shall abort any 5GMM signalling procedure, stop any of the timers T3510, T3517, T3519 or T3521 (if they were running), enter state 5GMM-DEREGISTERED and delete any stored SUCI.

Upon receiving an EAP-success message, the W-AGF acting on behalf of the N5GC device shall consider the procedure complete.

Upon receiving an EAP-failure message, the W-AGF acting on behalf of the N5GC device shall consider the procedure complete.

5.4.1.2.4 EAP message reliable transport procedure

5.4.1.2.4.1 General

The purpose of the EAP message reliable transport procedure is to provide a reliable transport of an EAP-request message, the ngKSI and the ABBA from the network to the UE and of an EAP-response message from the UE to the network.

The EAP message reliable transport procedure is initiated by an AUTHENTICATION REQUEST message with the EAP message IE.

5.4.1.2.4.2 EAP message reliable transport procedure initiation by the network

In order to initiate the EAP message reliable transport procedure, the AMF shall create an AUTHENTICATION REQUEST message.

The AMF shall set the EAP message IE of the AUTHENTICATION REQUEST message to the EAP-request message to be sent to the UE. The AMF shall set the ngKSI IE of the AUTHENTICATION REQUEST message to the ngKSI

value selected in subclause 5.4.1.2.2.2, subclause 5.4.1.2.3.1 or subclause 5.4.1.2.3A.1. In this release of specification, the AMF shall set the ABBA IE of the AUTHENTICATION REQUEST message with the length of ABBA IE to 2 and the ABBA contents to be 2 octets in length with value 0000H as described in subclause 9.11.3.10.

The AMF shall send the AUTHENTICATION REQUEST message to the UE, and the AMF shall start timer T3560 (see example in figure 5.4.1.2.4.2.1).

UE

AUTHENTICATION REQUEST

Start T3560

Stop T3560

AUTHENTICATION RESPONSE

Figure 5.4.1.2.4.2.1: EAP message reliable transport procedure

Upon receipt of an AUTHENTICATION REQUEST message with the EAP message IE, the UE handles the EAP message received in the EAP message IE and the ABBA of the AUTHENTICATION REQUEST message.

5.4.1.2.4.3 EAP message reliable transport procedure accepted by the UE

The UE shall create an AUTHENTICATION RESPONSE message.

If the received EAP message is an EAP-request message, the UE shall set the EAP message IE of the AUTHENTICATION RESPONSE message to the EAP-response message responding to the received EAP-request message.

The UE shall send the AUTHENTICATION RESPONSE message to the AMF.

Upon receipt of an AUTHENTICATION RESPONSE message, the AMF shall stop timer T3560. If the EAP message IE is included in the AUTHENTICATION RESPONSE message, the AMF handles the EAP message received in the EAP message IE of the AUTHENTICATION RESPONSE message.

5.4.1.2.4.4 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3560.

The AMF shall, on the first expiry of the timer T3560, retransmit the AUTHENTICATION REQUEST message and shall reset and start timer T3560. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3560, the AMF shall abort the EAP based primary authentication and key agreement procedure and any ongoing 5GMM specific procedure, and release the N1 NAS signalling connection.

b) Lower layers indication of non-delivered NAS PDU due to handover.

If the AUTHENTICATION REQUEST message could not be delivered due to an intra AMF handover and the target TA is included in the TAI list, then upon successful completion of the intra AMF handover the AMF shall retransmit the AUTHENTICATION REQUEST message. If a failure of handover procedure is reported by the lower layer and the N1 NAS signalling connection exists, the AMF shall retransmit the AUTHENTICATION REQUEST message.

AMF

5.4.1.2.4.5 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Authentication failure (5GMM cause #71 "ngKSI already in use").

The UE shall send an AUTHENTICATION FAILURE message, with 5GMM cause #71 "ngKSI already in use", to the network and start the timer T3520 (see example in figure 5.4.1.3.7.1). Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon the first receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #71 "ngKSI already in use", the network performs necessary actions to select a new ngKSI and send the same EAP-request message to the UE.

NOTE 1: Upon receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #71 "ngKSI already in use", the network can also re-initiate the EAP based primary authentication and key agreement procedure (see subclause 5.4.1.2.2.2).

Upon receiving a new AUTHENTICATION REQUEST message with the EAP message IE containing an EAP-request message from the network, the UE shall stop timer T3520, if running, process the EAP-request message as normal.

If the network is validated successfully (an AUTHENTICATION REQUEST message that contains a valid ngKSI and EAP-request message is received), the UE shall send the AUTHENTICATION RESPONSE message to the network and shall start any retransmission timers (e.g. T3510, T3517 or T3521) if they were running and stopped when the UE received the first failed AUTHENTICATION REQUEST message.

b) Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication from lower layers (if the EAP based primary authentication and key agreement procedure is triggered by a registration procedure).

The UE shall stop the timer T3520, if running, and re-initiate the registration procedure.

c) Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication with change in the current TAI (if the EAP based primary authentication and key agreement procedure is triggered by a service request procedure).

The UE shall stop the timer T3520, if running.

If the current TAI is not in the TAI list, the EAP based primary authentication and key agreement procedure shall be aborted and a registration procedure for mobility and periodic registration update shall be initiated.

If the current TAI is still part of the TAI list, it is up to the UE implementation how to re-run the ongoing procedure that triggered the EAP based primary authentication and key agreement procedure.

d) Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication without change in the current TAI (if the authentication procedure is triggered by a service request procedure).

The UE shall stop the timer T3520, if running. It is up to the UE implementation how to re-run the ongoing procedure that triggered the EAP based primary authentication and key agreement procedure.

e) Network failing the authentication check.

If the UE deems that the network has failed the authentication check, then it shall request RRC to locally release the RRC connection and treat the active cell as barred (see 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]). The UE shall start any retransmission timers (e.g. T3510, T3517 or T3521), if they were running and stopped when the UE received the first AUTHENTICATION REQUEST message containing an ngKSI that was already in use.

f) Change in the current TAI.

If that the current TAI is not in the TAI list before the AUTHENTICATION RESPONSE message is sent, the UE may discard sending the AUTHENTICATION RESPONSE message to the network and continue with the initiation of the registration procedure for mobility and periodic registration as described in subclause 5.5.1.3.2.

For item e, if no emergency service is started or is ongoing:

The UE shall stop timer T3520, if the timer is running and the UE enters 5GMM-IDLE mode, e.g. upon detection of a lower layer failure, release of the N1 NAS signalling connection, or as the result of an inter-system change in 5GMM-CONNECTED mode from N1 mode to S1 mode.

The UE shall deem that the network has failed the authentication check or assume that the authentication is not genuine and proceed as described in item e above if any of the following occurs:

- the timer T3520 expires;
- the UE detects any combination of the EAP-based authentication failures: transmission of AUTHENTICATION FAILURE message with 5GMM cause #71 "ngKSI already in use", transmission of AUTHENTICATION RESPONSE message with an EAP-response message after detecting an error as described in subclause 5.4.1.2.2.4, with an EAP-response message after not accepting of the server certificate as described in subclause 5.4.1.2.3.1 or with an EAP-response message after failing to authenticate the network as described in subclause 5.4.1.2.3A.1, during three consecutive authentication challenges. The EAP-request/AKA'-challenge challenges shall be considered as consecutive only, if the EAP-request/AKA'challenge challenges causing the second and third EAP-based authentication failure are received by the UE, while the timer T3520 started after the previous EAP-based authentication failure is running. Not accepting of the server certificate shall be considered as consecutive only, if the EAP-request detecting and third not accepting of the server certificate are received by the UE, while the timer T3520 started after the previous EAP-based authentication failure is running. Not accepting of the server certificate shall be considered as consecutive only, if the EAP-request messages causing the second and third not accepting of the server certificate are received by the UE, while the timer T3520 started after the previous EAP request message causing the previous not accepting of the server certificate is running.
- NOTE 2: Reception of an EAP-failure message is not considered when determining the three consecutive authentication challenges or three consecutive not accepting of the server certificate.

For item e if there is an emergency service started or is ongoin:

The UE shall stop timer T3520, if the timer is running and the UE enters 5GMM-IDLE mode, e.g. upon detection of a lower layer failure, release of the N1 NAS signalling connection, or as the result of an inter-system change in 5GMM-CONNECTED mode from N1 mode to S1 mode.

If a UE has an emergency PDU session established or is establishing an emergency PDU session, and sends an AUTHENTICATION FAILURE message to the AMF with the 5GMM cause appropriate for this cases (i.e. #71) or an AUTHENTICATION RESPONSE message containing an EAP-response message as described in subclause 5.4.1.2.2.4, containing an EAP-response message after not accepting of the server certificate as described in subclause 5.4.1.2.3.1 or containing an EAP-response message after failing to authenticate the network as described in subclause 5.4.1.2.3A.1, and receives the SECURITY MODE COMMAND message before the timeout of timer T3520, the UE shall deem that the network has passed the authentication check successfully, stop timer T3520, respectively, and execute the security mode control procedure.

If a UE has an emergency PDU session established or is establishing an emergency PDU session when timer T3520 expires, the UE shall not deem that the network has failed the authentication check and not behave as described in item e. Instead the UE shall continue using the current security context, if any, release all non-emergency PDU sessions, if any, by initiating UE-requested PDU session release procedure. If there is an ongoing PDU session establishment procedure, the UE shall release all non-emergency PDU sessions upon completion of the PDU session establishment procedure.

The UE shall start any retransmission timers (e.g. T3510, T3517 or T3521) if:

- they were running and stopped when the UE received the AUTHENTICATION REQUEST message and detected an authentication failure; and
- the procedures associated with these timers have not yet been completed.

The UE shall consider itself to be registered for emergency services.

5.4.1.2.5 EAP result message transport procedure

5.4.1.2.5.1 General

The purpose of the EAP result message transport procedure is to provide an EAP-success message or an EAP-failure message, and ngKSI from the network to the UE, when the EAP message cannot be piggybacked by another NAS message.

The EAP result message transport procedure is initiated:

- by an AUTHENTICATION RESULT message with the EAP message IE carrying the EAP-success message or the EAP-failure message; or
- by an AUTHENTICATION REJECT message with the EAP message IE carrying the EAP-failure message.

5.4.1.2.5.2 EAP result message transport procedure initiation by the network

In order to initiate the EAP result message transport procedure, the AMF shall create an AUTHENTICATION RESULT message or an AUTHENTICATION REJECT message.

The AMF shall set the EAP message IE of the AUTHENTICATION RESULT message to an EAP-success message or an EAP-failure message to be sent to the UE. The AMF shall set the EAP message IE of the AUTHENTICATION REJECT message to an EAP-failure message to be sent to the UE. The AMF shall set the ngKSI IE of the AUTHENTICATION RESULT message or the AUTHENTICATION REJECT message to the ngKSI value selected in subclause 5.4.1.2.2.2, subclause 5.4.1.2.3.1 or subclause 5.4.1.2.3A.1.

The AMF shall send the AUTHENTICATION RESULT message or the AUTHENTICATION REJECT message to the UE (see example in figure 5.4.1.2.5.2.1).

UE AMF
AUTHENTICATION RESULT
OR
AUTHENTICATION REJECT

Figure 5.4.1.2.5.2.1: EAP result message transport procedure

Upon receipt of an AUTHENTICATION RESULT message or an AUTHENTICATION REJECT message with the EAP message IE, the UE handles the EAP message received in the EAP message IE and the ABBA if received of the AUTHENTICATION RESULT message or in the AUTHENTICATION REJECT message.

5.4.1.3 5G AKA based primary authentication and key agreement procedure

5.4.1.3.1 General

The purpose of the 5G AKA based primary authentication and key agreement procedure is to provide mutual authentication between the UE and the network and to agree on the keys K_{AUSF} , K_{SEAF} and K_{AMF} (see 3GPP TS 33.501 [24]). The cases when the 5G AKA based primary authentication and key agreement procedure is used are defined in 3GPP TS 33.501 [24].

The network initiates the 5G AKA based primary authentication and key agreement procedure by sending an AUTHENTICATION REQUEST message to the UE without the EAP message IE. The network shall include the ngKSI and the ABBA in AUTHENTICATION REQUEST message.

The 5G AKA based primary authentication and key agreement procedure is always initiated and controlled by the network. However, the UE can reject the 5G authentication challenge sent by the network.

The UE shall proceed with a 5G authentication challenge only if a USIM is present.

A partial native 5G NAS security context is established in the UE and the network when a 5G authentication is successfully performed. During a successful 5G AKA based primary authentication and key agreement procedure, the CK and IK are computed by the USIM. CK and IK are then used by the ME as key material to compute new keys K_{AUSF} , K_{SEAF} and K_{AMF} . K_{AMF} is stored in the 5G NAS security contexts (see 3GPP TS 33.501 [24]) of both the network and in the volatile memory of the ME while registered to the network, and is the root for the 5GS integrity protection and ciphering key hierarchy.

NOTE 1: Generation of the new K_{AUSF} and the new K_{SEAF} does not result into deletion of the valid K_{AUSF} and the valid K_{SEAF} , if any.

The 5G AKA based primary authentication and key agreement procedure is initiated by an AUTHENTICATION REQUEST message without the EAP message IE.

Upon successful completion of the 5G AKA based primary authentication, the AMF shall initiate a security mode control procedure (see subclause 5.4.2) to take the new partial native 5G NAS security context into use.

NOTE 2: The AMF shall immediately initiate a security mode control procedure (see subclause 5.4.2) after 5G AKA primary authentication is successful to avoid K_{AUSF} key mismatch between the UE and the network.

5.4.1.3.2 Authentication initiation by the network

The network may initiate a 5G AKA based primary authentication and key agreement procedure for a UE in 5GMM-CONNECTED mode at any time. For restrictions applicable after handover or inter-system change to N1 mode in 5GMM-CONNECTED mode, see subclause 5.5.1.3.3.

The network initiates the 5G AKA based primary authentication and key agreement procedure by sending an AUTHENTICATION REQUEST message to the UE and starting the timer T3560 (see example in figure 5.4.1.3.2.1). The AUTHENTICATION REQUEST message shall contain the parameters necessary to calculate the authentication response (see 3GPP TS 33.501 [24]). This message shall include the ngKSI that will be used by the UE and AMF to identify the K_{AMF} and the partial native security context that is created if the authentication is successful. This message shall also include the ABBA parameter. In this release of specification, the network shall set the length of ABBA IE to 2 and the ABBA contents to be 2 octets in length with value 0000H as described in subclause 9.11.3.10.

If an ngKSI is contained in an initial NAS message during a 5GMM procedure, the network shall include a different ngKSI value in the AUTHENTICATION REQUEST message when it initiates a 5G AKA based primary authentication and key agreement procedure.

UE		AMF
•	AUTHENTICATION REQUEST	Start T3560
	AUTHENTICATION RESPONSE	Stop T3560
4	AUTHENTICATION REJECT	if authentication failed

Figure 5.4.1.3.2.1: 5G AKA based primary authentication and key agreement procedure

5.4.1.3.3 Authentication response by the UE

The UE shall respond to an AUTHENTICATION REQUEST message. With the exception of the cases described in subclause 5.4.1.3.6 and 5.4.1.3.7 case l, the UE shall process the 5G authentication challenge data and respond with an AUTHENTICATION RESPONSE message to the network.

Upon a successful 5G authentication challenge, the UE shall determine the PLMN identity to be used for the calculation of the new K_{AMF} from the 5G authentication challenge data according to the following rules:

- a) When the UE moves from 5GMM-IDLE mode to 5GMM-CONNECTED mode, until the first handover, the UE shall use the PLMN identity of the selected PLMN; and
- b) After handover or inter-system change to N1 mode in 5GMM-CONNECTED mode,
 - 1) if the target cell is not a shared network cell, the UE shall use the PLMN identity received as part of the broadcast system information;
 - 2) if the target cell is a shared network cell and the UE has a valid 5G-GUTI, the UE shall use the PLMN identity that is part of the 5G-GUTI; and
 - 3) if the target cell is a shared network cell and the UE has a valid 4G-GUTI, but not a valid 5G-GUTI, the UE shall use the PLMN identity that is part of the 4G-GUTI.

Upon a successful 5G authentication challenge, the new K_{AMF} calculated from the 5G authentication challenge data shall be stored in a new 5G NAS security context in the volatile memory of the ME.

The USIM will compute the authentication response (RES) using the 5G authentication challenge data received from the ME, and pass RES to the ME. From the RES, RES* is then generated according to Annex A of 3GPP TS 33.501 [24].

In order to avoid a synchronisation failure, when the UE receives an AUTHENTICATION REQUEST message, the UE shall store the received RAND together with the RES*, in the volatile memory of the ME. When the UE receives a subsequent AUTHENTICATION REQUEST message, if the stored RAND value is equal to the new received value in the AUTHENTICATION REQUEST message, then the ME shall not pass the RAND to the USIM, but shall send the AUTHENTICATION RESPONSE message with the stored RES*. If there is no valid stored RAND in the ME or the stored RAND is different from the new received value in the AUTHENTICATION REQUEST message, the ME shall pass the RAND to the USIM, shall override any previously stored RAND and RES* with the new ones and start, or reset and restart timer T3516.

The RAND and RES* values stored in the ME shall be deleted and timer T3516, if running, shall be stopped:

- a) upon receipt of a
 - 1) SECURITY MODE COMMAND message,
 - 2) SERVICE REJECT message,
 - 3) REGISTRATION REJECT message,
 - 4) **REGISTRATION ACCEPT** message,
 - 5) AUTHENTICATION REJECT message, or
 - 6) SERVICE ACCEPT message;
- b) upon expiry of timer T3516;
- c) if the UE enters the 5GMM state 5GMM-DEREGISTERED or 5GMM-NULL; or
- d) if the UE enters 5GMM-IDLE mode.

5.4.1.3.4 Authentication completion by the network

Upon receipt of an AUTHENTICATION RESPONSE message, the network stops the timer T3560 and checks the correctness of RES* (see 3GPP TS 33.501 [24]).

If the 5G AKA based primary authentication and key agreement procedure has been completed successfully and the related ngKSI is stored in the 5G NAS security context of the network, the network shall include a different ngKSI value in the AUTHENTICATION REQUEST message when it initiates a new 5G AKA based primary authentication and key agreement procedure.

Upon receipt of an AUTHENTICATION FAILURE message, the network stops the timer T3560. In the case where the 5GMM cause #21 "synch failure" is received, the core network may renegotiate with the UDM/AUSF and provide the UE with new authentication parameters.

5.4.1.3.5 Authentication not accepted by the network

If the authentication response (RES) returned by the UE is not valid, the network response depends upon the type of identity used by the UE in the initial NAS message, that is:

- if the 5G-GUTI was used; or
- if the SUCI was used.

If the 5G-GUTI was used, the network should initiate an identification procedure to retrieve SUCI from the UE and restart the 5G AKA based primary authentication and key agreement procedure with the received SUCI.

If the SUCI was used for identification in the initial NAS message or in a restarted 5G AKA based primary authentication and key agreement procedure, or the network decides not to initiate the identification procedure to retrieve SUCI from the UE after an unsuccessful 5G AKA based primary authentication and key agreement procedure, the network should send an AUTHENTICATION REJECT message to the UE.

Upon receipt of an AUTHENTICATION REJECT message,

1) if the AUTHENTICATION REJECT message has been successfully integrity checked by the NAS:

the UE shall set the update status to 5U3 ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.

In case of PLMN, the USIM shall be considered invalid until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN until switching off or the UICC containing the USIM is removed.

In case of SNPN, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the entry until switching off or the UICC containing the USIM is removed.

In case of SNPN, if the UE is registered for onboarding services in SNPN or is performing initial registration for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]; and

- if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall set:

- i) the counter for "SIM/USIM considered invalid for GPRS services" events, the counter for "USIM considered invalid for 5GS services over non-3GPP access" events, and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
- ii) the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN;

to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value; and

- if the UE is operating in single-registration mode, the UE shall handle EMM parameters, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the
authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and

2) if the AUTHENTICATION REJECT message is received without integrity protection and if timer T3516 or T3520 is running, the UE shall start timer T3247 with a random value uniformly drawn from the range between 30 minutes and 60 minutes, if the timer is not running (see subclause 5.3.20).

Additionally, if the UE is neither registered for onboarding services in SNPN nor performing initial registration for onboarding services in SNPN, the UE shall:

- a) if the AUTHENTICATION REJECT message is received over 3GPP access, and the counter for "SIM/USIM considered invalid for GPRS services" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-a) of subclause 5.3.20.2 (if the UE is not operating in SNPN access operation mode) or list item a)-1) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3;
- b) if the AUTHENTICATION REJECT message is received over non-3GPP access, and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN or the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN has a value less than a UE implementation-specific maximum value, proceed as specified in subclause 5.3.20, list item 1)-b) of subclause 5.3.20.2 (if the UE is not operating in SNPN access operation mode) or list item a)-2) of subclause 5.3.20.3 (if the UE is operating in SNPN access operation mode) for the case that the 5GMM cause value received is #3.
- c) otherwise:
 - i) if the AUTHENTICATION REJECT message is received over 3GPP access:
 - The UE shall set the update status for 3GPP access to 5U3 ROAMING NOT ALLOWED, delete for 3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI.
 - In case of PLMN, the UE shall consider the USIM as invalid for 5GS services via 3GPP access and non-EPS service until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN shall be considered invalid for 3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN via 3GPP access until switching off or the UICC containing the USIM is removed.

- The UE shall set:
 - the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "SIM/USIM considered invalid for non-GPRS services" events if maintained by the UE, in case of PLMN; or
 - the counter for "the entry for the current SNPN considered invalid for 3GPP access" events in case of SNPN;

to UE implementation-specific maximum value.

- If the UE is operating in single-registration mode, the UE shall handle 4G-GUTI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the authentication procedure is not accepted by the network. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed; and
- ii) if the AUTHENTICATION REJECT message is received over non-3GPP access:
 - the UE shall set the update status for non-3GPP access to 5U3 ROAMING NOT ALLOWED, delete for non-3GPP access only the stored 5G-GUTI, TAI list, last visited registered TAI and ngKSI;
 - in case of PLMN, the UE shall consider the USIM as invalid for 5GS services via non-3GPP access until switching off the UE or the UICC containing the USIM is removed.

In case of SNPN, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for non-3GPP access until the UE is switched off or the entry is updated. Additionally, the UE shall consider the USIM as invalid for the current SNPN and for non-3GPP access until switching off or the UICC containing the USIM is removed; and

- the UE shall set:
 - the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN; or
 - the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN.

If the UE is registered for onboarding services in SNPN or performing initial registration for onboarding services in SNPN, the UE shall:

- 1) if the SNPN-specific attempt counter for the SNPN sending the AUTHENTICATION REJECT message has a value less than a UE implementation-specific maximum value, increment the SNPN-specific attempt counter for the SNPN; or
- 2) otherwise, the UE shall set the update status to 5U3.ROAMING NOT ALLOWED, delete the stored 5G-GUTI, TAI list, last visited registered TAI, and ngKSI, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the AUTHENTICATION REJECT message is received by the UE, the UE shall abort any 5GMM signalling procedure, stop any of the timers T3510, T3516, T3517, T3519, T3520 or T3521 (if they were running), enter state 5GMM-DEREGISTERED and delete any stored SUCI.

Depending on local requirements or operator preference for emergency services, if the UE initiates a registration procedure with 5GS registration type IE set to "emergency registration" and the AMF is configured to allow emergency registration without user identity, the AMF needs not follow the procedures specified for the authentication failure in the present subclause. The AMF may continue a current 5GMM specific procedure.

5.4.1.3.6 Authentication not accepted by the UE

In the 5G authentication challenge, the UE shall check the 5G authentication challenge data (RAND, AUTN and ngKSI) received in the AUTHENTICATION REQUEST message to verify authenticity of the 5G core network.

The ME shall check that ngKSI received in the AUTHENTICATION REQUEST message is not already in use. The ME shall forward the RAND and AUTN to the USIM to check.

The UE may reject the core network due to an incorrect AUTN or ngKSI parameter. If the UE has to reject the 5G authentication challenge, the UE shall return AUTHENTICATION FAILURE message to the network with a cause value indicating the reason for the failure (see 3GPP TS 33.501 [24]).

Incorrect 5G authentication challenge data contains four possible causes for authentication failure:

a) MAC code failure:

If the UE finds the MAC code (supplied by the core network in the AUTN parameter) to be invalid, the UE shall send an AUTHENTICATION FAILURE message to the network, with the 5GMM cause #20 "MAC failure". The UE shall then follow the procedure described in subclause 5.4.1.3.7, item c.

b) Non-5G authentication unacceptable:

If the UE finds that the "separation bit" in the AMF field of AUTN supplied by the core network is set to 0, the UE shall send an AUTHENTICATION FAILURE message to the network, with the 5GMM cause #26 "non-5G authentication unacceptable" (see subclause 6.1.3 in 3GPP TS 33.501 [24]). The UE shall then follow the procedure described in subclause 5.4.1.3.7, item d.

c) ngKSI already in use:

If the UE detects that ngKSI received in the AUTHENTICATION REQUEST message is already in use in the UE shall send an AUTHENTICATION FAILURE message to the network, with the 5GMM cause #71 "ngKSI already in use". The UE shall then follow the procedure described in subclause 5.4.1.3.7, item e.

d) SQN failure:

If the UE finds the sequence number SQN (supplied by the core network in the AUTN parameter) to be out of range, the UE shall send an AUTHENTICATION FAILURE message to the network, with the 5GMM cause #21 "synch failure" and a re-synchronization token AUTS provided by the USIM (see 3GPP TS 33.102 [23]). The UE shall then follow the procedure described in subclause 5.4.1.3.7, item f.

If the UE returns an AUTHENTICATION FAILURE message to the network, the UE shall delete any previously stored RAND and RES* and shall stop timer T3516, if running.

If the UE has an emergency PDU session established or is establishing such a PDU session, additional UE requirements are specified in subclause 5.4.1.3.7, under "for items c, d, e and f".

5.4.1.3.7 Abnormal cases

a) Lower layer failure.

Upon detection of lower layer failure before the AUTHENTICATION RESPONSE message is received, the network shall abort the procedure.

b) Expiry of timer T3560.

The network shall, on the first expiry of the timer T3560, retransmit the AUTHENTICATION REQUEST message and shall reset and start timer T3560. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3560, the network shall abort the 5G AKA based primary authentication and key agreement procedure and any ongoing 5GMM specific procedure and release the N1 NAS signalling connection.

c) Authentication failure (5GMM cause #20 "MAC failure").

The UE shall send an AUTHENTICATION FAILURE message, with 5GMM cause #20 "MAC failure" according to subclause 5.4.1.3.6, to the network and start timer T3520 (see example in figure 5.4.1.3.7.1). Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon the first receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #20 "MAC failure", the network may initiate the identification procedure described in subclause 5.4.3. This is to allow the network to obtain the SUCI from the UE. The network may then check that the 5G-GUTI originally used in the 5G authentication challenge corresponded to the correct SUPI. Upon receipt of the IDENTITY REQUEST message from the network, the UE shall proceed as specified in subclause 5.4.3.3.

NOTE 1: Upon receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #20 "MAC failure", the network may also terminate the 5G AKA based primary authentication and key agreement procedure (see subclause 5.4.1.3.5).

If the mapping of 5G-GUTI to SUPI in the network was incorrect, the network should respond by sending a new AUTHENTICATION REQUEST message to the UE. Upon receiving the new AUTHENTICATION REQUEST message from the network, the UE shall stop the timer T3520, if running, and then process the 5G challenge information as normal. If the mapping of 5G-GUTI to SUPI in the network was correct, the network should terminate the 5G AKA based primary authentication and key agreement procedure by sending an AUTHENTICATION REJECT message (see subclause 5.4.1.3.5).

If the network is validated successfully (an AUTHENTICATION REQUEST message that contains a valid SQN and MAC is received), the UE shall send the AUTHENTICATION RESPONSE message to the network and shall start any retransmission timers (e.g. T3510, T3517 or T3521) if they were running and stopped when the UE received the first failed AUTHENTICATION REQUEST message.

If the UE receives the second AUTHENTICATION REQUEST message, and the MAC value cannot be resolved, the UE shall follow the procedure specified in this subclause, item c, starting again from the beginning, or if the message contains a UMTS authentication challenge, the UE shall follow the procedure specified in item d. If the SQN is invalid, the UE shall proceed as specified in item f.

UE		AMF
Start T3520	AUTHENTICATION REQUEST	Start T3560 Stop T3560
Stop T3520	AUTHENTICATION REQUEST	Start T3560 Stop T3560

Figure 5.4.1.3.7.1: Authentication failure during 5G AKA based primary authentication and key agreement procedure

d) Authentication failure (5GMM cause #26 "non-5G authentication unacceptable").

The UE shall send an AUTHENTICATION FAILURE message, with 5GMM cause #26 "non-5G authentication unacceptable", to the network and start the timer T3520 (see example in figure 5.4.1.3.7.1). Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon the first receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #26 "non-5G authentication unacceptable", the network may initiate the identification procedure described in subclause 5.4.3. This is to allow the network to obtain the SUCI from the UE. The network may then check that the 5G-GUTI originally used in the 5G authentication challenge corresponded to the correct SUPI. Upon receipt of the IDENTITY REQUEST message from the network, the UE shall proceed as specified in subclause 5.4.3.3.

NOTE 2: Upon receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #26 "non-5G authentication unacceptable", the network may also terminate the 5G AKA based primary authentication and key agreement procedure (see subclause 5.4.1.3.5).

If the mapping of 5G-GUTI to SUPI in the network was incorrect, the network should respond by sending a new AUTHENTICATION REQUEST message to the UE. Upon receiving the new AUTHENTICATION REQUEST message from the network, the UE shall stop the timer T3520, if running, and then process the 5G challenge information as normal. If the mapping of 5G-GUTI to SUPI in the network was correct, the network should terminate the 5G AKA based primary authentication and key agreement authentication procedure by sending an AUTHENTICATION REJECT message (see subclause 5.4.1.3.5).

If the network is validated successfully (an AUTHENTICATION REQUEST message that contains a valid 5G authentication challenge is received), the UE shall send the AUTHENTICATION RESPONSE message to the network and shall start any retransmission timers (e.g. T3510, T3517 or T3521) if they were running and stopped when the UE received the first failed AUTHENTICATION REQUEST message.

e) Authentication failure (5GMM cause #71 "ngKSI already in use").

The UE shall send an AUTHENTICATION FAILURE message, with 5GMM cause #71 "ngKSI already in use", to the network and start the timer T3520 (see example in figure 5.4.1.3.7.1). Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon the first receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #71 "ngKSI already in use", the network performs necessary actions to select a new ngKSI and send the same 5G authentication challenge to the UE.

NOTE 3: Upon receipt of an AUTHENTICATION FAILURE message from the UE with 5GMM cause #71 "ngKSI already in use", the network may also re-initiate the 5G AKA based primary authentication and key agreement procedure (see subclause 5.4.1.3.2).

Upon receiving the new AUTHENTICATION REQUEST message from the network, the UE shall stop the timer T3520, if running, and then process the 5G challenge information as normal.

If the network is validated successfully (an AUTHENTICATION REQUEST message that contains a valid ngKSI, SQN and MAC is received), the UE shall send the AUTHENTICATION RESPONSE message to the network and shall start any retransmission timers (e.g. T3510, T3517 or T3521) if they were running and stopped when the UE received the first failed AUTHENTICATION REQUEST message.

f) Authentication failure (5GMM cause #21 "synch failure").

The UE shall send an AUTHENTICATION FAILURE message, with 5GMM cause #21 "synch failure", to the network and start the timer T3520 (see example in figure 5.4.1.3.7.1). Furthermore, the UE shall stop any of the retransmission timers that are running (e.g. T3510, T3517 or T3521). Upon the first receipt of an AUTHENTICATION FAILURE message from the UE with the 5GMM cause #21 "synch failure", the network shall use the returned AUTS parameter from the authentication failure parameter IE in the AUTHENTICATION FAILURE message, to re-synchronise. The re-synchronisation procedure requires the AMF to delete all unused authentication vectors for that SUPI and obtain new vectors from the UDM/AUSF. When re-synchronisation is complete, the network shall initiate the 5G AKA based primary authentication and key agreement procedure. Upon receipt of the AUTHENTICATION REQUEST message, the UE shall stop the timer T3520, if running.

NOTE 4: Upon receipt of two consecutive AUTHENTICATION FAILURE messages from the UE with 5GMM cause #21 "synch failure", the network may terminate the 5G AKA based primary authentication and key agreement procedure by sending an AUTHENTICATION REJECT message.

If the network is validated successfully (a new AUTHENTICATION REQUEST message is received which contains a valid SQN and MAC) while T3520 is running, the UE shall send the AUTHENTICATION RESPONSE message to the network and shall start any retransmission timers (e.g. T3510, T3517 or T3521), if they were running and stopped when the UE received the first failed AUTHENTICATION REQUEST message.

Upon receipt of an AUTHENTICATION REJECT message, the UE shall perform the actions as specified in subclause 5.4.1.3.5.

g) Network failing the authentication check.

If the UE deems that the network has failed the authentication check, then it shall request RRC to locally release the RRC connection and treat the active cell as barred (see 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]). The UE shall start any retransmission timers (e.g. T3510, T3517 or T3521), if they were running and stopped when the UE received the first AUTHENTICATION REQUEST message containing an incorrect authentication challenge data causing authentication failure.

 h) Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication from lower layers (if the 5G AKA based primary authentication and key agreement procedure is triggered by a registration procedure).

The UE shall stop the timer T3520, if running, and re-initiate the registration procedure.

 Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication with change in the current TAI (if the 5G AKA based primary authentication and key agreement procedure is triggered by a service request procedure).

The UE shall stop the timer T3520, if running.

If the current TAI is not in the TAI list, the 5G AKA based primary authentication and key agreement procedure shall be aborted and a registration procedure for mobility and periodic registration update shall be initiated.

If the current TAI is still part of the TAI list, it is up to the UE implementation how to re-run the ongoing procedure that triggered the 5G AKA based primary authentication and key agreement procedure.

 j) Transmission failure of AUTHENTICATION RESPONSE message or AUTHENTICATION FAILURE message indication without change in the current TAI (if the authentication procedure is triggered by a service request procedure).

The UE shall stop the timer T3520, if running. It is up to the UE implementation how to re-run the ongoing procedure that triggered the 5G AKA based primary authentication and key agreement procedure.

k) Lower layers indication of non-delivered NAS PDU due to handover.

If the AUTHENTICATION REQUEST message could not be delivered due to an intra AMF handover and the target TA is included in the TAI list, then upon successful completion of the intra AMF handover the AMF shall retransmit the AUTHENTICATION REQUEST message. If a failure of handover procedure is reported by the lower layer and the N1 NAS signalling connection exists, the AMF shall retransmit the AUTHENTICATION REQUEST message.

l) Change in the current TAI.

If the current TAI is not in the TAI list before the AUTHENTICATION RESPONSE message is sent, the UE may discard sending the AUTHENTICATION RESPONSE message to the network and continue with the initiation of the registration procedure for mobility and periodic registration as described in subclause 5.5.1.3.2.

m) AUTHENTICATION REJECT message is received without integrity protection and neither timer T3516 nor T3520 is running.

If an AUTHENTICATION REJECT message is received without integrity protection and if neither timer T3516 nor T3520 is running, then the UE shall discard the AUTHENTICATION REJECT message. Additionally, the UE may request RRC to locally release the RRC connection and treat the active cell as barred (see 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C]).

For items c, d, e, and f if no emergency service is started or is ongoing:

The UE shall stop timer T3520, if the timer is running and the UE enters 5GMM-IDLE mode, e.g. upon detection of a lower layer failure, release of the N1 NAS signalling connection, or as the result of an inter-system change in 5GMM-CONNECTED mode from N1 mode to S1 mode.

The UE shall deem that the network has failed the authentication check or assume that the authentication is not genuine and proceed as described in item g above if any of the following occurs:

- the timer T3520 expires;
- the UE detects any combination of the 5G authentication failures: 5GMM causes #20 "MAC failure", #21 "synch failure", #26 "non-5G authentication unacceptable" or #71 "ngKSI already in use", during three consecutive authentication challenges. The 5G authentication challenges shall be considered as consecutive only, if the 5G authentication challenges causing the second and third 5G authentication failure are received by the UE, while the timer T3520 started after the previous 5G authentication failure is running.

For items c, d, e, and f if there is an emergency service started or is ongoing:

The UE shall stop timer T3520, if the timer is running and the UE enters 5GMM-IDLE mode, e.g. upon detection of a lower layer failure, release of the N1 NAS signalling connection, or as the result of an inter-system change in 5GMM-CONNECTED mode from N1 mode to S1 mode.

If there is an ongoing service request procedure for emergency services fallback the UE shall abort the service request procedure, stop timer T3517 and locally release any resources allocated for the service request procedure and enters state 5GMM-REGISTERED. The UE shall attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

Depending on local requirements or operator preference for emergency services, if the UE has an emergency PDU session established or is establishing an emergency PDU session, the AMF need not follow the procedures specified for the authentication failure specified in the present subclause. The AMF may respond to the AUTHENTICATION FAILURE message by initiating the security mode control procedure selecting the "null integrity protection algorithm" 5G-IA0, "null ciphering algorithm" 5G-EA0 or may abort the 5G AKA based primary authentication and key agreement procedure and continue using the current security context, if any. The AMF shall indicate to the SMF to perform the release of all non-emergency PDU sessions, if any. If there is an ongoing PDU session establishment procedure, the AMF shall indicate to the SMF to perform the release of all non-emergency PDU sessions. The network shall behave as if the UE is registered for emergency services.

If a UE has an emergency PDU session established or is establishing an emergency PDU session and sends an AUTHENTICATION FAILURE message to the AMF with the 5GMM cause appropriate for these cases (#20,

#21, #26, or #71 respectively) and receives the SECURITY MODE COMMAND message before the timeout of timer T3520, the UE shall deem that the network has passed the authentication check successfully, stop timer T3520, respectively, and execute the security mode control procedure.

If a UE has an emergency PDU session established or is establishing an emergency PDU session when timer T3520 expires, the UE shall not deem that the network has failed the authentication check and not behave as described in item g. Instead the UE shall continue using the current security context, if any, release all non-emergency PDU sessions, if any, by initiating UE-requested PDU session release procedure. If there is an ongoing PDU session establishment procedure, the UE shall release all non-emergency PDU session supon completion of the PDU session establishment procedure.

The UE shall start any retransmission timers (e.g. T3510, T3517 or T3521) if:

- they were running and stopped when the UE received the AUTHENTICATION REQUEST message and detected an authentication failure; and
- the procedures associated with these timers have not yet been completed.

The UE shall behave as if the UE is registered for emergency services.

5.4.2 Security mode control procedure

5.4.2.1 General

The purpose of the NAS security mode control procedure is to take a 5G NAS security context into use, and initialise and start NAS signalling security between the UE and the AMF with the corresponding 5G NAS keys and 5G NAS security algorithms.

Furthermore, the network may also initiate the security mode control procedure in the following cases:

- a)- in order to change the 5G NAS security algorithms for a current 5G NAS security context already in use;
- b) in order to change the value of uplink NAS COUNT used in the latest SECURITY MODE COMPLETE message as described in 3GPP TS 33.501 [24], subclause 6.9.4.4; and
- c) in order to provide the Selected EPS NAS security algorithms to the UE.

For restrictions concerning the concurrent running of a security mode control procedure with other security related procedures in the AS or inside the core network see 3GPP TS 33.501 [24], subclause 6.9.5.

If the security mode control procedure is initiated after successful 5G AKA based primary authentication and key agreement procedure and the security mode control procedure intends to bring into use the partial native 5G NAS security context created by the 5G AKA based primary authentication and key agreement procedure and the UE accept received security mode command (see subclause 5.4.2.3), the ME shall:

- a) delete the valid K_{AUSF} and the valid K_{SEAF} , if any; and
- b) consider the new K_{AUSF} to be the valid K_{AUSF}, and the new K_{SEAF} to be the valid K_{SEAF}, reset the SOR counter and the UE parameter update counter to zero, and store the valid K_{AUSF}, the valid K_{SEAF}, the SOR counter and the UE parameter update counter as specified in annex C and use the valid K_{AUSF} in the verification of SOR transparent container and UE parameters update transparent container, if any are received.
- NOTE: The AMF does not perform a security mode control procedure when the 5G AKA based primary authentication procedure successfully authenticates a 5G ProSe layer-3 remote UE accessing the network via a 5G ProSe layer-3 UE-to-network relay UE served by the AMF.

5.4.2.2 NAS security mode control initiation by the network

The AMF initiates the NAS security mode control procedure by sending a SECURITY MODE COMMAND message to the UE and starting timer T3560 (see example in figure 5.4.2.2).

The AMF shall reset the downlink NAS COUNT counter and use it to integrity protect the initial SECURITY MODE COMMAND message if the security mode control procedure is initiated:

- a) to take into use the security context created after a successful execution of the 5G AKA based primary authentication and key agreement procedure or the EAP based primary authentication and key agreement procedure; or
- b) upon receipt of REGISTRATION REQUEST message, if the AMF needs to create a mapped 5G NAS security context (i.e. the type of security context flag is set to "mapped security context" in the ngKSI IE included in the SECURITY MODE COMMAND message).

The AMF shall send the SECURITY MODE COMMAND message unciphered, but shall integrity protect the message with the 5G NAS integrity key based on K_{AMF} or mapped K'_{AMF} indicated by the ngKSI included in the message. The AMF shall set the security header type of the message to "integrity protected with new 5G NAS security context".

The AMF shall create a locally generated K_{AMF} and send the SECURITY MODE COMMAND message including an ngKSI value in the ngKSI IE set to "000" and 5G-IA0 and 5G-EA0 as the selected NAS security algorithms only when the security mode control procedure is initiated:

- a) during an initial registration procedure for emergency services if no valid 5G NAS security context is available;
- b) during a registration procedure for mobility and periodic registration update for a UE that has an emergency PDU session if no valid 5G NAS security context is available;
- c) during a service request procedure for a UE that has an emergency PDU session if no valid 5G NAS security context is available; or
- d) after a failed primary authentication and key agreement procedure for a UE that has an emergency PDU session or is establishing an emergency PDU session, if continued usage of a valid 5G NAS security context is not possible.

When the AMF sends the SECURITY MODE COMMAND message including an ngKSI value in the ngKSI IE set to "000" and 5G-IA0 and 5G-EA0 as the selected NAS security algorithms, if:

- a) the AMF supports N26 interface;
- b) the UE set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- c) the security mode control procedure is initiated during an initial registration procedure for emergency services, during a registration procedure for mobility and periodic registration update for a UE that has an emergency PDU session, or during a service request procedure for a UE that has an emergency PDU session,

the SECURITY MODE COMMAND message shall also include the Selected EPS NAS security algorithms IE. The selected EPS NAS security algorithms shall be set to EIA0 and EEA0.

The UE shall process a SECURITY MODE COMMAND message including an ngKSI value in the ngKSI IE set to "000" and 5G-IA0 and 5G-EA0 as the selected NAS security algorithms and, if accepted, create a locally generated K_{AMF} when the security mode control procedure is initiated:

- a) during an initial registration procedure for emergency services;
- b) during a registration procedure for mobility and periodic registration update for a UE that has an emergency PDU session;
- c) during a service request procedure for a UE that has an emergency PDU session; or
- d) after a primary authentication and key agreement procedure for a UE that has an emergency PDU session or is establishing an emergency PDU session.
- NOTE 1: The process for creation of the locally generated K_{AMF} by the AMF and the UE is implementation dependent. The K_{AMF} is specified in 3GPP TS 33.501 [24].

Upon receipt of a REGISTRATION REQUEST message, if the AMF does not have the valid current 5G NAS security context indicated by the UE, the AMF shall either:

a) indicate the use of the new mapped 5G NAS security context to the UE by setting the type of security context flag in the ngKSI IE to "mapped security context" and the KSI value related to the security context of the source system; or

3GPP TS 24.501 version 17.16.0 Release 17

b) set the ngKSI value to "000" in the ngKSI IE if the AMF sets 5G-IA0 and 5G-EA0 as the selected NAS security algorithms for a UE that has an emergency PDU session.

Upon receipt of a REGISTRATION REQUEST message, if the AMF has the valid current 5G NAS security context indicated by the UE, the AMF supports N26 interface and the UE set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the UE is not registered for disaster roaming services, the AMF shall check whether the selected EPS NAS security algorithms was successfully provided to the UE. If not, the AMF shall initiate the NAS security mode control procedure by sending a SECURITY MODE COMMAND message with the Selected EPS NAS security algorithms IE to the UE.

While having a current mapped 5G NAS security context with the UE, if the AMF needs to take the native 5G NAS security context into use, the AMF shall include the ngKSI that indicates the native 5G NAS security context in the SECURITY MODE COMMAND message.

The AMF shall include the replayed security capabilities of the UE (including the security capabilities with regard to NAS, RRC and UP (user plane) ciphering as well as NAS and RRC integrity, and other possible target network security capabilities, i.e. E-UTRAN if the UE included them in the message to network), the selected 5GS ciphering and integrity algorithms and the ngKSI.

If a UE is already registered over one access to a PLMN and the AMF decides to skip primary authentication and key agreement procedure when the UE attempts to register over the other access to the same PLMN, the AMF shall take into use the UE's current 5G NAS security context over the other access that the UE is registering. In this case, SECURITY MODE COMMAND message is not sent to the UE.

If the UE is registered to the same AMF and the same PLMN over both 3GPP access and non-3GPP access, and the UE is in 5GMM-CONNECTED mode over both the 3GPP and non-3GPP accesses, then at any time the primary authentication and key agreement procedure has successfully completed over:

- a) the 3GPP access, the AMF includes the ngKSI in the SECURITY MODE COMMAND message over the 3GPP access. When the AMF sends the SECURITY MODE COMMAND message to UE over the non-3GPP access to take into use the new 5G NAS security context, the AMF shall include the same ngKSI in the SECURITY MODE COMMAND message to identify the new 5G NAS security context; or
- b) the non-3GPP access, the AMF includes the ngKSI in the SECURITY MODE COMMAND message over the non-3GPP access. When the AMF sends the SECURITY MODE COMMAND message to UE over the 3GPP access to take into use the new 5G NAS security context, the AMF shall include the same ngKSI in the SECURITY MODE COMMAND message to identify the new 5G NAS security context.

The AMF may initiate a SECURITY MODE COMMAND in order to change the 5G security algorithms for a current 5G NAS security context already in use. The AMF re-derives the 5G NAS keys from K_{AMF} with the new 5G algorithm identities as input and provides the new 5GS algorithm identities within the SECURITY MODE COMMAND message. The AMF shall set the security header type of the message to "integrity protected with new 5G NAS security context".

If, during an ongoing registration procedure, the AMF is initiating a SECURITY MODE COMMAND (i.e. after receiving the REGISTRATION REQUEST message, but before sending a response to that message) and:

- a) the REGISTRATION REQUEST message does not successfully pass the integrity check at the AMF; or
- b) the AMF can not decipher the value part of the NAS message container IE in the REGISTRATION REQUEST message;

the AMF shall include the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message requesting the UE to send the entire REGISTRATION REQUEST message in the SECURITY MODE COMPLETE message as described in 3GPP TS 33.501 [24].

NOTE 2: The AMF uses the UE security capability which was provided by the UE.

If, during an ongoing service request procedure for a UE with an emergency PDU session, the AMF is initiating a SECURITY MODE COMMAND (i.e. after receiving the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message, but before sending a response to that message) and the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message does not successfully pass the integrity check at the AMF, the AMF shall include the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message requesting the UE to send the entire:

- a) SERVICE REQUEST message; or
- b) CONTROL PLANE SERVICE REQUEST message excluding non-cleartext IEs, except the Uplink data status IE if needed (see subclause 5.4.2.3);

in the SECURITY MODE COMPLETE message as described in 3GPP TS 33.501 [24].

Additionally, the AMF may request the UE to include its IMEISV in the SECURITY MODE COMPLETE message.

If the AMF supports N26 interface and the UE set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the AMF needs to provide the selected EPS NAS security algorithms to the UE, the AMF shall select ciphering and integrity algorithms to be used in the EPS and indicate them to the UE via the Selected EPS NAS security algorithms IE in the SECURITY MODE COMMAND message.

NOTE 3: The AS and NAS security capabilities are the same, i.e. if the UE supports one algorithm for NAS, the same algorithm is also supported for AS.

If the AMF performs horizontal key derivation e.g. during the mobility and periodic registration update or when the UE is already registered in the PLMN with another access type as described in 3GPP TS 33.501 [24], the AMF shall include horizontal derivation parameter in the SECURITY MODE COMMAND message.

If the security mode control procedure is initiated after successful EAP based primary authentication and key agreement procedure and the security mode control procedure intends to bring into use the partial native 5G NAS security context created by the EAP based primary authentication and key agreement procedure, the AMF shall set the EAP message IE of the SECURITY MODE COMMAND message to an EAP-success message to be sent to the UE.



Figure 5.4.2.2: Security mode control procedure

5.4.2.3 NAS security mode command accepted by the UE

Upon receipt of the SECURITY MODE COMMAND message, the UE shall check whether the security mode command can be accepted or not. This is done by performing the integrity check of the message, and by checking that the received Replayed UE security capabilities IE has not been altered compared to the latest values that the UE sent to the network.

When the SECURITY MODE COMMAND message includes an EAP-success message the UE handles the EAP-success message and the ABBA as described in subclause 5.4.1.2.2.8, 5.4.1.2.3.1, 5.4.1.2.3A.1 and 5.4.1.2.3B.1.

If:

- a) the UE is registered for emergency services, performing initial registration for emergency services, establishing an emergency PDU session or has an emergency PDU session established;
- b) the W-AGF acts on behalf of the FN-RG; or
- c) the W-AGF acts on behalf of the N5GC device,

and the SECURITY MODE COMMAND message is received with ngKSI value "000" and 5G-IA0 and 5G-EA0 as selected 5G NAS security algorithms, the UE shall locally derive and take in use 5G NAS security context. The UE shall delete existing current 5G NAS security context.

The UE shall accept a SECURITY MODE COMMAND message indicating the "null integrity protection algorithm" 5G-IA0 as the selected 5G NAS integrity algorithm only if the message is received when

- a) the UE is registered for emergency services, performing initial registration for emergency services, establishing an emergency PDU session or has an emergency PDU session established; or
- b) the W-AGF acts on behalf of the FN-RG; or
- c) the W-AGF acts on behalf of the N5GC device.

If the type of security context flag included in the SECURITY MODE COMMAND message is set to "native security context" and if the ngKSI matches a valid non-current native 5G NAS security context held in the UE while the UE has a mapped 5G NAS security context as the current 5G NAS security context, the UE shall take the non-current native 5G NAS security context into use which then becomes the current native 5G NAS security context and delete the mapped 5G NAS security context.

The UE shall ignore the Replayed S1 UE security capabilities IE if this IE is included in the SECURITY MODE COMMAND message.

If the SECURITY MODE COMMAND message can be accepted, the UE shall take the 5G NAS security context indicated in the message into use. The UE shall in addition reset the uplink NAS COUNT counter if:

- a) the SECURITY MODE COMMAND message is received in order to take a 5G NAS security context into use created after a successful execution of the 5G AKA based primary authentication and key agreement procedure or the EAP based primary authentication and key agreement procedure; or
- b) the SECURITY MODE COMMAND message received includes the type of security context flag set to "mapped security context" in the NAS key set identifier IE the ngKSI does not match the current 5G NAS security context, if it is a mapped 5G NAS security context.

If the SECURITY MODE COMMAND message can be accepted and a new 5G NAS security context is taken into use and SECURITY MODE COMMAND message does not indicate the "null integrity protection algorithm" 5G-IA0 as the selected NAS integrity algorithm, the UE shall:

- if the SECURITY MODE COMMAND message has been successfully integrity checked using an estimated downlink NAS COUNT equal to 0, then the UE shall set the downlink NAS COUNT of this new 5G NAS security context to 0;
- otherwise the UE shall set the downlink NAS COUNT of this new 5G NAS security context to the downlink NAS COUNT that has been used for the successful integrity checking of the SECURITY MODE COMMAND message.

If the SECURITY MODE COMMAND message includes the horizontal derivation parameter indicating " K_{AMF} derivation is required", the UE shall derive a new K'_{AMF}, as specified in 3GPP TS 33.501 [24] for K_{AMF} to K'_{AMF} derivation in mobility, and set both uplink and downlink NAS COUNTs to zero. When the new 5G NAS security context is taken into use for current access and the UE is registered with the same PLMN over the 3GPP access and the non-3GPP access:

- a) the UE is in 5GMM-IDLE mode over the non-current access, the AMF and the UE shall activate the new 5G NAS security context over the non-current access as described in 3GPP TS 33.501 [24]. The AMF and the UE shall set the downlink NAS COUNT and uplink NAS COUNT to zero for the non-current access; or
- b) the UE is in 5GMM-CONNECTED mode over the non-current access, the AMF shall send the SECURITY MODE COMMAND message over the non-current access to activate the new 5G NAS security context that was activated over the current access as described in 3GPP TS 33.501 [24]. The AMF shall include the same ngKSI in the SECURITY MODE COMMAND message to identify the new 5G NAS security context.

If the SECURITY MODE COMMAND message includes the horizontal derivation parameter indicating " K_{AMF} derivation is not required" or the Additional 5G security information IE is not included in the message, the UE is registered with the same PLMN over the 3GPP access and non-3GPP access, then after the completion of a security mode control procedure over the current access:

- a) the UE is in 5GMM-IDLE mode over the non-current access, the AMF and the UE shall activate the new 5G NAS security context for the non-current access. If a primary authentication and key agreement procedure was completed before the security mode control procedure, the AMF and the UE shall set the downlink NAS COUNT and uplink NAS COUNT to zero for the non-current access, otherwise the downlink NAS COUNT and uplink NAS COUNT for the non-3GPP access are not changed; or
- b) the UE is in 5GMM-CONNECTED mode over the non-current access, the AMF shall send the SECURITY MODE COMMAND message over the non-current access to activate the new 5G NAS security context that was activated over the current access as described in 3GPP TS 33.501 [24]. The AMF shall include the same ngKSI in the SECURITY MODE COMMAND message to identify the new 5G NAS security context.

If the SECURITY MODE COMMAND message can be accepted, the UE shall send a SECURITY MODE COMPLETE message integrity protected with the selected 5GS integrity algorithm and the 5G NAS integrity key based on the K_{AMF} or mapped K'_{AMF} if the type of security context flag is set to "mapped security context" indicated by the ngKSI. When the SECURITY MODE COMMAND message includes the type of security context flag set to "mapped security context" in the NAS key set identifier IE, then the UE shall check whether the SECURITY MODE COMMAND message indicates the ngKSI of the current 5GS security context, if it is a mapped 5G NAS security context, in order not to re-generate the K'_{AMF}.

Furthermore, if the SECURITY MODE COMMAND message can be accepted, the UE shall cipher the SECURITY MODE COMPLETE message with the selected 5GS ciphering algorithm and the 5GS NAS ciphering key based on the K_{AMF} or mapped K'_{AMF} indicated by the ngKSI. The UE shall set the security header type of the message to "integrity protected and ciphered with new 5G NAS security context".

From this time onward the UE shall cipher and integrity protect all NAS signalling messages with the selected 5GS integrity and ciphering algorithms.

If the AMF indicated in the SECURITY MODE COMMAND message that the IMEISV is requested and:

- a) if the UE:
 - 1) supports at least one 3GPP access technology, the UE shall include its IMEISV in the IMEISV IE of the SECURITY MODE COMPLETE message; or
 - 2) does not support any 3GPP access technology (i.e. satellite NG-RAN, NG-RAN, satellite E-UTRAN, E-UTRAN, UTRAN or GERAN) and supports NAS over untrusted or trusted non-3GPP access, the UE shall include its EUI-64 in the non-IMEISV PEI IE of the SECURITY MODE COMPLETE message; or
- b) if the 5G-RG contains neither an IMEISV nor an IMEI or when the W-AGF acts on behalf of the FN-RG (or on behalf of the N5GC device), the 5G-RG or the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) shall include the MAC address and the MAC address usage restriction indication determined as specified in subclause 5.3.2 in the non-IMEISV PEI IE in the SECURITY MODE COMPLETE message.

If during an ongoing registration procedure or service request procedure, the UE receives a SECURITY MODE COMMAND message which includes the Additional 5G security information IE with the RINMR bit set to "Retransmission of the initial NAS message requested", the UE shall include the entire unciphered REGISTRATION REQUEST message or SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message, which the UE had previously included in the NAS message container IE of the initial NAS message (i.e. REGISTRATION REQUEST message or SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message, respectively), in the NAS message container IE of the SECURITY MODE COMPLETE message. The retransmitted CONTROL PLANE SERVICE REQUEST message:

- a) shall not include any non-cleartext IE, except the Uplink data status IE; and
- b) may include the Uplink data status IE.

If, prior to receiving the SECURITY MODE COMMAND message, the UE without a valid 5G NAS security context had sent a REGISTRATION REQUEST message the UE shall include the entire REGISTRATION REQUEST message in the NAS message container IE of the SECURITY MODE COMPLETE message as described in subclause 4.4.6.

If the UE operating in the single-registration mode receives the Selected EPS NAS security algorithms IE, the UE shall use the IE according to 3GPP TS 33.501 [24].

For a UE operating in single-registration mode in a network supporting N26 interface after an inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, the UE shall set the value of the Selected EPS NAS security algorithms IE in the 5G NAS security context to the NAS security algorithms that were received from the source MME when the UE was in S1 mode.

5.4.2.4 NAS security mode control completion by the network

The AMF shall, upon receipt of the SECURITY MODE COMPLETE message, stop timer T3560. From this time onward the AMF shall integrity protect and encipher all signalling messages with the selected 5GS integrity and ciphering algorithms.

If the SECURITY MODE COMPLETE message contains a NAS message container IE with a REGISTRATION REQUEST message, the AMF shall complete the ongoing registration procedure by considering the REGISTRATION REQUEST message contained in the NAS message container IE as the message that triggered the procedure.

If the SECURITY MODE COMPLETE message contains a NAS message container IE with a REGISTRATION REQUEST message, the 5GMM capability IE included in the REGISTRATION REQUEST message indicates "S1 mode supported" and the AMF supports N26 interface, the AMF shall initiate another NAS security mode control procedure in order to provide the selected EPS NAS security algorithms to the UE as described in subclause 5.4.2.2. This second NAS security mode control procedure should be initiated as part of 5GMM common procedures of the ongoing registration procedure.

If the SECURITY MODE COMPLETE message contains a NAS message container IE with a SERVICE REQUEST message, the AMF shall complete the ongoing service request procedure by considering the SERVICE REQUEST message contained in the NAS message container IE as the message that triggered the procedure.

If the SECURITY MODE COMPLETE message contains a NAS message container IE with a CONTROL PLANE SERVICE REQUEST message, the AMF shall complete the ongoing service request procedure by considering the CONTROL PLANE SERVICE REQUEST message contained in the NAS message container IE as the message that triggered the procedure.

5.4.2.5 NAS security mode command not accepted by the UE

If the security mode command cannot be accepted, the UE shall send a SECURITY MODE REJECT message. The SECURITY MODE REJECT message contains a 5GMM cause that typically indicates one of the following cause values:

- #23 UE security capabilities mismatch.
- #24 security mode rejected, unspecified.

If the UE detects that the received Replayed UE security capabilities IE has been altered compared to the latest values that the UE sent to the network, the UE shall set the cause value to #23 "UE security capabilities mismatch".

Upon receipt of the SECURITY MODE REJECT message, the AMF shall stop timer T3560. The AMF shall also abort the ongoing procedure that triggered the initiation of the NAS security mode control procedure.

Both the UE and the AMF shall apply the 5G NAS security context in use before the initiation of the security mode control procedure, if any, to protect the SECURITY MODE REJECT message and any other subsequent messages according to the rules in subclause 4.4.4 and 4.4.5.

5.4.2.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Transmission failure of SECURITY MODE COMPLETE message or SECURITY MODE REJECT message indication from lower layers (if the security mode control procedure is triggered by a registration procedure).

The UE shall abort the security mode control procedure and re-initiate the registration procedure.

b) Transmission failure of SECURITY MODE COMPLETE message or SECURITY MODE REJECT message indication with change in the current TAI (if the security mode control procedure is triggered by a service request procedure). If the current TAI is not in the TAI list, the security mode control procedure shall be aborted and a registration procedure shall be initiated.

If the current TAI is still part of the TAI list, the security mode control procedure shall be aborted and it is up to the UE implementation how to re-run the ongoing procedure that triggered the security mode control procedure.

c) Transmission failure of SECURITY MODE COMPLETE message or SECURITY MODE REJECT message indication without change in the current TAI (if the security mode control procedure is triggered by a service request procedure).

The security mode control procedure shall be aborted and it is up to the UE implementation how to re-run the ongoing procedure that triggered the security mode control procedure.

5.4.2.7 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Lower layer failure before the SECURITY MODE COMPLETE or SECURITY MODE REJECT message is received.

The network shall abort the security mode control procedure.

b) Expiry of timer T3560.

The network shall, on the first expiry of the timer T3560, retransmit the SECURITY MODE COMMAND message and shall reset and start timer T3560. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3560, the procedure shall be aborted.

c) Collision between security mode control procedure and registration, service request or de-registration procedure not indicating switch off.

The network shall abort the security mode control procedure and proceed with the UE initiated procedure.

d) Collision between security mode control procedure and other 5GMM procedures than in item c.

The network shall progress both procedures.

e) Lower layers indication of non-delivered NAS PDU due to handover:

If the SECURITY MODE COMMAND message could not be delivered due to an intra AMF handover and the target TA is included in the TAI list, then upon successful completion of the intra AMF handover the AMF shall retransmit the SECURITY MODE COMMAND message. If a failure of the handover procedure is reported by the lower layer and the N1 signalling connection exists, the AMF shall retransmit the SECURITY MODE COMMAND message.

5.4.3 Identification procedure

5.4.3.1 General

The purpose of this procedure is to request a particular UE to provide specific identification parameters, e.g. the SUCI, the IMEI, the IMEISV, the EUI-64 or the MAC address. The SUCI is a privacy preserving identifier containing the concealed SUPI and the IMEI, the IMEISV, the EUI-64 and the MAC address are formats of PEI.

5.4.3.2 Identification initiation by the network

The AMF initiates the identification procedure by sending an IDENTITY REQUEST message to the UE and starting timer T3570 (see example in figure 5.4.3.2.1). The IDENTITY REQUEST message specifies the requested identification parameters in the Identity type information element.

UE

AMF

•	IDENTITY REQUEST	Start T3570
	IDENTITY RESPONSE	Stop T3570

Figure 5.4.3.2.1: Identification procedure

5.4.3.3 Identification response by the UE

A UE shall be ready to respond to an IDENTITY REQUEST message at any time whilst in 5GMM-CONNECTED mode.

Upon receipt of the IDENTITY REQUEST message:

- a) if the Identity type IE in the IDENTITY REQUEST message is not set to "SUCI", the UE shall send an IDENTITY RESPONSE message to the network. The IDENTITY RESPONSE message shall contain the identification parameters as requested by the network; and
- b) if the Identity type IE in the IDENTITY REQUEST message is set to "SUCI", the UE shall:
 - if timer T3519 is not running, generate a fresh SUCI as specified in 3GPP TS 33.501 [24], send an IDENTITY RESPONSE message with the SUCI, start timer T3519 and store the value of the SUCI sent in the IDENTITY RESPONSE message; and
 - 2) if timer T3519 is running, send an IDENTITY RESPONSE message with the stored SUCI.

5.4.3.4 Identification completion by the network

Upon receipt of the IDENTITY RESPONSE the network shall stop the timer T3570.

5.4.3.5 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Transmission failure of the IDENTITY RESPONSE message (if the identification procedure is triggered by a registration procedure).

The UE shall re-initiate the registration procedure.

b) Requested identity is not available

If the UE cannot encode the requested identity in the IDENTITY RESPONSE message, e.g. because no valid USIM is available, then it shall encode the identity type as "No identity".

5.4.3.6 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Lower layer failure.

Upon detection of a lower layer failure before the IDENTITY RESPONSE is received, the network shall abort any ongoing 5GMM procedure.

b) Expiry of timer T3570.

The network shall, on the first expiry of the timer T3570, retransmit the IDENTITY REQUEST message and reset and restart the timer T3570. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3570, the network shall abort the identification procedure and any ongoing 5GMM procedure.

c) Collision of an identification procedure with a registration procedure for initial registration.

If the network receives a REGISTRATION REQUEST message indicating either "initial registration" or "emergency registration" in the 5GS registration type IE before the ongoing identification procedure has been completed and no registration procedure is pending on the network (i.e. no REGISTRATION ACCEPT/REJECT message has still to be sent as an answer to a REGISTRATION REQUEST message), the network shall proceed with the registration procedure for initial registration.

d) Collision of an identification procedure with a registration procedure for initial registration when the identification procedure has been caused by a registration procedure for initial registration.

If the network receives a REGISTRATION REQUEST message indicating either "initial registration" or "emergency registration" in the 5GS registration type IE before the ongoing identification procedure has been completed and a registration procedure for initial registration is pending (i.e. a REGISTRATION ACCEPT/REJECT message has to be sent as an answer to an earlier REGISTRATION REQUEST message), then:

- If one or more of the information elements in the REGISTRATION REQUEST message differ from the ones received within the previous REGISTRATION REQUEST message, the network shall proceed with the new registration procedure for initial registration; or
- If the information elements do not differ, then the network shall not treat any further this new REGISTRATION REQUEST message.
- e) Collision of an identification procedure with a registration procedure for mobility and periodic registration update.

If the network receives a REGISTRATION REQUEST message indicating either "mobility registration updating" or "periodic registration updating" in the 5GS registration type IE before the ongoing identification procedure has been completed, the network shall progress both procedures.

f) Collision of an identification procedure with a UE initiated de-registration procedure.

If the network receives a DEREGISTRATION REQUEST message with "switch off" indication in the Deregistration type IE before the ongoing identification procedure has been completed, the network shall abort the identification procedure and shall progress the UE-initiated de-registration procedure;

Else the network shall complete the identification procedure and shall respond to the UE-initiated de-registration procedure as described in subclause 5.5.2.2.

5.4.4 Generic UE configuration update procedure

5.4.4.1 General

The purpose of this procedure is to:

- a) allow the AMF to update the UE configuration for access and mobility management-related parameters decided and provided by the AMF by providing new parameter information within the command;
- b) request the UE to perform a registration procedure for mobility and periodic registration update towards the network to update access and mobility management-related parameters decided and provided by the AMF (see subclause 5.5.1.3);
- c) deliver the UAV authorization information to the UE, as described in 3GPP TS 23.256 [6AB]; or
- d) update the PEIPS assistance information in the UE (see subclause 5.3.25).

This procedure is initiated by the network and can only be used when the UE has an established 5GMM context, and the UE is in 5GMM-CONNECTED mode. When the UE is in 5GMM-IDLE mode, the AMF may use the paging or

notification procedure to initiate the generic UE configuration update procedure. The AMF can request a confirmation response in order to ensure that the parameter has been updated by the UE.

This procedure shall be initiated by the network to assign a new 5G-GUTI to the UE after:

- a) a successful service request procedure invoked as a response to a paging request from the network and before the:
 - 1) release of the N1 NAS signalling connection; or
 - 2) suspension of the N1 NAS signalling connection due to user plane CIoT 5GS optimization i.e. before the UE and the AMF enter 5GMM-IDLE mode with suspend indication; or
- b) the AMF receives an indication from the lower layers that it has received the NGAP UE context resume request message as specified in 3GPP TS 38.413 [31] for a UE in 5GMM-IDLE mode with suspend indication and this resumption is a response to a paging request from the network, and before the:
 - 1) release of the N1 NAS signalling connection; or
 - 2) suspension of the N1 NAS signalling connection due to user plane CIoT 5GS optimization i.e. before the UE and the AMF enter 5GMM-IDLE mode with suspend indication.

If the service request procedure was triggered due to 5GSM downlink signalling pending, the procedure for assigning a new 5G-GUTI can be initiated by the network after the transport of the 5GSM downlink signalling.

The following parameters are supported by the generic UE configuration update procedure without the need to request the UE to perform the registration procedure for mobility and periodic registration update:

- a) 5G-GUTI;
- b) TAI list;
- c) Service area list;
- d) Network identity and time zone information (Full name for network, short name for network, local time zone, universal time and local time zone, network daylight saving time);
- e) LADN information;
- f) Rejected NSSAI;
- NOTE: A cause value associated with a rejected S-NSSAI can be included in the Rejected NSSAI IE or in the Extended rejected NSSAI IE and a back-off timer value associated with rejected S-NSSAI(s) can be included in the Extended rejected NSSAI IE.
- g) void;
- h) Operator-defined access category definitions;
- i) SMS indication;
- j) "CAG information list";
- k) UE radio capability ID;
- 1) 5GS registration result;
- m) Truncated 5G-S-TMSI configuration;
- n) T3447 value;
- o) "list of PLMN(s) to be used in disaster condition";
- p) disaster roaming wait range;
- q) disaster return wait range; and
- r) PEIPS assistance information;

- s) Priority indicator; and
- t) NSAG information.

The following parameters can be sent to the UE with or without a request to perform the registration procedure for mobility and periodic registration update:

- a) Allowed NSSAI;
- b) Configured NSSAI;
- c) Network slicing subscription change indication; or
- d) NSSRG information.

The following parameters are sent to the UE with a request to perform the registration procedure for mobility and periodic registration update:

- a) MICO indication;
- b) UE radio capability ID deletion indication; and
- c) Additional configuration indication.

The following parameters can be included in the Service-level-AA container IE to be sent to the UE without a request to perform the registration procedure for mobility and periodic registration update:

- a) Service-level device ID;
- b) Service-level-AA payload type;
- c) Service-level-AA payload;
- d) Service-level-AA response; or
- e) Service-level-AA service status indication.

The following parameters are sent over 3GPP access only:

- a) LADN information;
- b) MICO indication;
- c) TAI list;
- d) Service area list;
- e) "CAG information list";
- f) UE radio capability ID;
- g) UE radio capability ID deletion indication;
- h) Truncated 5G-S-TMSI configuration;
- i) Additional configuration indication;
- j) T3447 value;
- k) Service-level-AA container; and
- 1) NSAG information.

The following parameters are managed and sent per access type i.e., independently over 3GPP access or non-3GPP access:

a) Allowed NSSAI;

- b) Rejected NSSAI (when the NSSAI is rejected for the current registration area or is rejected for the maximum number of UEs reached); and
- c) If the UE is not registered to the same PLMN or SNPN over 3GPP and non-3GPP access:
 - 5G-GUTI;
 - Network identity and time zone information;
 - Rejected NSSAI (when the NSSAI is rejected for the current PLMN or SNPN or rejected for the failed or revoked NSSAA);
 - Configured NSSAI;
 - NSSRG information;- SMS indication;
 - 5GS registration result; and
 - PEIPS assistance information.

If the UE is registered to the same PLMN or SNPN over 3GPP and non-3GPP access, the following parameters are managed commonly and sent over 3GPP access or non-3GPP access:

- a) 5G-GUTI;
- b) Network identity and time zone information;
- c) Rejected NSSAI (when the NSSAI is rejected for the current PLMN or SNPN or rejected for the failed or revoked NSSAA);
- d) Configured NSSAI;
- e) SMS indication; and
- f) 5GS registration result;
- g) "list of PLMN(s) to be used in disaster condition";
- h) disaster roaming wait range;
- i) disaster return wait range;
- j) PEIPS assistance information; and
- k) NSSRG information;



5.4.4.2 Generic UE configuration update procedure initiated by the network

The AMF shall initiate the generic UE configuration update procedure by sending the CONFIGURATION UPDATE COMMAND message to the UE.

The AMF shall in the CONFIGURATION UPDATE COMMAND message either:

- a) include one or more of the following parameters: 5G-GUTI, TAI list, allowed NSSAI that may include the mapped S-NSSAI(s), LADN information, service area list, MICO indication, NITZ information, configured NSSAI that may include the mapped S-NSSAI(s), NSSRG information, rejected S-NSSAI(s) in the Rejected NSSAI IE or in the Extended rejected NSSAI IE, network slicing subscription change indication, operator-defined access category definitions, SMS indication, service gap time value, "CAG information list", UE radio capability ID, 5GS registration result, UE radio capability ID deletion indication, truncated 5G-S-TMSI configuration, T3447 value, "list of PLMN(s) to be used in disaster condition", disaster roaming wait range, disaster return wait range, PEIPS assistance information, the priority indicator or the NSAG information;
- b) include the Configuration update indication IE with the Registration requested bit set to "registration requested"; or
- c) include a combination of both a) and b).

If the UE is registering or registered for onboarding services in SNPN, the serving SNPN shall not provide the configured NSSAI, the allowed NSSAI or the rejected NSSAI to the UE.

If the UE supports extended rejected NSSAI, the rejected S-NSSAI(s) shall be included in the Extended rejected NSSAI IE. Otherwise the rejected S-NSSAI(s) shall be included in the Rejected NSSAI IE.

If an acknowledgement from the UE is requested, the AMF shall indicate "acknowledgement requested" in the Acknowledgement bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message and shall start timer T3555. Acknowledgement shall be requested for all parameters except when only NITZ is included.

To initiate parameter re-negotiation between the UE and network, the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message.

NOTE 1: Generic UE configuration update procedure can be initiated by the AMF for updating the emergency number list, the extended emergency number list or both by indicating "registration requested" in the Registration requested bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message to the UE.

If a new allowed NSSAI information or AMF re-configuration of supported S-NSSAIs requires an AMF relocation, the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE and include the Allowed NSSAI IE in the CONFIGURATION UPDATE COMMAND message.

If the AMF includes a new allowed NSSAI in the CONFIGURATION UPDATE COMMAND message and the subscription information includes the NSSRG information, then the S-NSSAIs of the allowed NSSAI shall be associated with at least one common NSSRG value. If the network has pending NSSAI, the S-NSSAIs in the pending NSSAI and allowed NSSAI shall be associated with at least one common NSSRG value.

If the AMF includes a new configured NSSAI in the CONFIGURATION UPDATE COMMAND message and the new configured NSSAI requires an AMF relocation as specified in 3GPP TS 23.501 [8], the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE in the message.

If the AMF includes a new configured NSSAI in the CONFIGURATION UPDATE COMMAND message, the subscription information includes the NSSRG information, and the UE has set the NSSRG bit in the 5GMM capability IE of the REGISTRATION REQUEST message to:

- a) "NSSRG supported", then the AMF shall include the NSSRG information in the CONFIGURATION UPDATE COMMAND message; or
- b) "NSSRG not supported", then the configured NSSAI shall include one or more S-NSSAIs each of which is associated with all the NSSRG value(s) of the default S-NSSAI(s), or the configured NSSAI shall include, based on the indication received from the UDM as specified in 3GPP TS 23.501 [8], all subscribed S-NSSAIs even if these S-NSSAIs do not share any common NSSRG value.

If the AMF needs to update the NSSRG information and the UE has set the NSSRG bit to "NSSRG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, then the AMF shall include the new NSSRG information in the CONFIGURATION UPDATE COMMAND message.

If the AMF includes a new NSSRG information in the CONFIGURATION UPDATE COMMAND message and the AMF determines that the UE needs to provide a new requested NSSAI due to no NSSRG value common to all the S-NSSAI(s) of the allowed NSSAI based on the new NSSRG information, then the CONFIGURATION UPDATE COMMAND message shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE.

If the CONFIGURATION UPDATE COMMAND message is initiated only due to changes to the allowed NSSAI and these changes require the UE to initiate a registration procedure, but the AMF is unable to determine an allowed NSSAI for the UE as specified in 3GPP TS 23.501 [8], then the CONFIGURATION UPDATE COMMAND message shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE, and shall not contain any other parameters.

If:

- the AMF needs to enforce a change in the restriction on the use of enhanced coverage or use of CE mode B as described in subclause 5.3.18; or
- the AMF decides to inform a UE in 5GMM-CONNECTED mode and registered for disaster roaming services, that a disaster condition is no longer applicable;
- NOTE 1A: The case of the AMF triggering a generic UE configuration update procedure to inform a UE registered for disaster roaming services that a disaster condition is no longer applicable, is only applicable for a UE already in 5GMM-CONNECTED mode.

the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE and "release of N1 NAS signalling connection not required" in the Signalling connection maintain request bit of the Additional configuration indication IE in the CONFIGURATION UPDATE COMMAND message.

If a network slice-specific authentication and authorization procedure for an S-NSSAI is completed as a:

- a) success, the AMF shall include this S-NSSAI in the allowed NSSAI over the same access of the requested S-NSSAI; or
- b) failure, the AMF shall include this S-NSSAI in the rejected NSSAI for the failed or revoked NSSAA with the rejection cause "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" over either 3GPP access or non-3GPP access.

If authorization is revoked for an S-NSSAI that is in the current allowed NSSAI for an access type, the AMF shall:

- a) provide a new allowed NSSAI to the UE, excluding the S-NSSAI for which authorization is revoked; and
- b) provide a new rejected NSSAI for the failed or revoked NSSAA, including the S-NSSAI in the rejected NSSAI for which the authorization is revoked, with the rejection cause "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization".

The allowed NSSAI and the rejected NSSAI shall be included in the CONFIGURATION UPDATE COMMAND message to reflect the result of the procedures subject to network slice-specific authentication and authorization.

NOTE 2: If there are multiple S-NSSAIs subject to network slice-specific authentication and authorization, it is implementation specific if the AMF informs the UE about the outcome of the procedures in one or more CONFIGURATION UPDATE COMMAND messages.

If the AMF includes the Network slicing indication IE in the CONFIGURATION UPDATE COMMAND message with the Network slicing subscription change indication set to "Network slicing subscription changed", and changes to the allowed NSSAI require the UE to initiate a registration procedure, but the AMF is unable to determine an allowed NSSAI for the UE as specified in 3GPP TS 23.501 [8], then the CONFIGURATION UPDATE COMMAND message shall additionally indicate "registration requested" in the Registration requested bit of the Configuration update indication IE and shall not include an allowed NSSAI.

If EAC mode is activated for an S-NSSAI, the AMF shall perform NSAC for the S-NSSAI subject to NSAC before such S-NSSAI is included in the allowed NSSAI in the CONFIGURATION UPDATE COMMAND message. If EAC

mode is deactivated for an S-NSSAI, the AMF shall perform NSAC for the S-NSSAI subject to NSAC after such S-NSSAI is included in the allowed NSSAI in the CONFIGURATION UPDATE COMMAND message.

If the UE supports extended rejected NSSAI and the AMF determines that maximum number of UEs reached for one or more S-NSSAI(s) in the allowed NSSAI as specified in subclause 4.6.2.5, the AMF shall include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE in the CONFIGURATION UPDATE COMMAND message. In addition, the AMF may include a back-off timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" included in the Extended rejected NSSAI IE of the CONFIGURATION UPDATE COMMAND message. To available due to maximum number of UEs reached" included in the Extended rejected NSSAI IE of the CONFIGURATION UPDATE COMMAND message. To avoid that large numbers of UEs simultaneously initiate deferred requests, the network should select the value for the backoff timer for each S-NSSAI for the informed UEs so that timeouts are not synchronised.

If the UE does not indicate support for extended rejected NSSAI and the maximum number of UEs has been reached, the AMF should include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available in the current registration area" in the Rejected NSSAI IE and should not include these S-NSSAIs in the allowed NSSAI in the CONFIGURATION UPDATE COMMAND message. In addition, the AMF may based on the network policies start a local implementation specific timer for the UE per rejected S-NSSAI and upon expiration of the local implementation specific timer, the AMF may remove the rejected S-NSSAI from the rejected NSSAI and update to the UE by initiating the generic UE configuration update procedure.

NOTE 3: Based on network policies, the AMF can include the S-NSSAI(s) for which the maximum number of UEs has been reached in the rejected NSSAI with rejection causes other than "S-NSSAI not available in the current registration area".

If the UE has set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the AMF may include the NSAG information IE in the CONFIGURATION UPDATE COMMAND message. Up to 4 NSAG entries are allowed to be associated with a TAI list in the NSAG information IE.

NOTE 3a: How the AMF selects NSAG entries to be included in the NSAG information IE is implementation specific, e.g. take the NSAG priority and the current registration area into accountNOTE 3b: If the NSAG for the PLMN and its equivalent PLMN(s) have different associations with S-NSSAIs, then the AMF includes a TAI list for the NSAG entry in the NSAG information IE.

If the AMF needs to update the LADN information, the AMF shall include the LADN information in the LADN information IE of the CONFIGURATION UPDATE COMMAND message.

If the AMF needs to update the "CAG information list", the AMF shall include the CAG information list IE or the Extended CAG information list IE in the CONFIGURATION UPDATE COMMAND message.

NOTE 4: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If the AMF needs to update the "CAG information list", the UE has an emergency PDU session, and the AMF can determine that the UE is in

- a) a CAG cell and none of the CAG-ID(s) supported by the CAG cell is included in the "allowed CAG list" for the current PLMN in the updated "CAG information list"; or
- b) a non-CAG cell and the entry for the current PLMN in the updated "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells";

the AMF may indicate to the SMF to perform a local release of:

- a) all non-emergency single access PDU sessions associated with 3GPP access;
- b) all MA PDU sessions without a PDN connection established as a user-plane resource and without user plane resources established on non-3GPP access; and
- c) the 3GPP access user plane resources of all those MA PDU sessions with user plane resources established on both accesses.

The AMF shall not indicate to the SMF to release the emergency PDU session. If the AMF indicated to the SMF to perform a local release of:

- a) all single access non-emergency PDU sessions associated with 3GPP access;
- b) all MA PDU sessions without a PDN connection established as a user-plane resource and without user plane resources established on non-3GPP access; and
- c) the 3GPP access user plane resources of all those MA PDU sessions with user plane resources established on both accesses;

the network shall behave as if the UE is registered for emergency services over 3GPP access and shall set the 5GS registration result IE value to "Registered for emergency services" in the CONFIGURATION UPDATE COMMAND message.

If the AMF is initiating the generic UE configuration update procedure to indicate to a UE which is registered for disaster roaming services, and which has an ongoing emergency PDU session, that the UE is registered for emergency services as described in subclause 4.24, the AMF shall set the 5GS registration result IE value to "Registered for emergency services" in the CONFIGURATION UPDATE COMMAND message.

If the AMF:

- updated the "CAG information list" to remove one or more CAG-ID(s) in the Allowed CAG list for the serving PLMN or an equivalent PLMN; or
- updated the "CAG information list" to set the "indication that the UE is only allowed to access 5GS via CAG cells" for the serving PLMN or an equivalent PLMN which was not set before,

then upon completion of the configuration update procedure and if the UE does not have an emergency PDU session, the AMF shall initiate the release of the N1 NAS signalling connection according to subclause 5.3.1.3.

If the AMF needs to update the truncated 5G-S-TMSI configuration for a UE in NB-N1 mode using control plane CIoT 5GS optimization, the AMF shall include the Truncated 5G-S-TMSI configuration IE in the CONFIGURATION UPDATE COMMAND message.

If the AMF includes a UE radio capability ID deletion indication IE in the CONFIGURATION UPDATE COMMAND message, the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE.

If the AMF needs to redirect the UE to EPC as described in subclause 4.8.4A.2, the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE and "release of N1 NAS signalling connection not required" in the Signalling connection maintain request bit of the Additional configuration indication IE in the CONFIGURATION UPDATE COMMAND message.

If the UE is not in NB-N1 mode and the UE supports RACS, the AMF may include either a UE radio capability ID IE or a UE radio capability ID deletion indication IE in the CONFIGURATION UPDATE COMMAND message.

During an established 5GMM context, the network may send none, one, or more CONFIGURATION UPDATE COMMAND messages to the UE. If more than one CONFIGURATION UPDATE COMMAND message is sent, the messages need not have the same content.

Upon receipt of the result of the UUAA-MM procedure from the UAS-NF, the AMF shall include:

- a) the service-level-AA response with the SLAR field set to:
 - 1) "Service level authentication and authorization was successful" if the AMF detects the UUAA-MM procedure has succeeded; or
 - 2) "Service level authentication and authorization was not successful or service level authorization is revoked" if the AMF detects the UUAA-MM procedure has failed;
- b) if the CAA-Level UAV ID is provided by the UAS-NF, the service-level device ID with the value set to the CAA-Level UAV ID; and;
- c) if ta payload is received from the UAS-NF:

- 1) the service-level-AA payload with the value set to the payload; and
- 2) if a payload type associated with the payload is received, the service-level-AA payload type with the values set to the payload type

in the Service-level-AA container IE of the CONFIGURATION UPDATE COMMAND message.

- NOTE 5: UAS security information can be included in the UUAA payload by the USS as specified in 3GPP TS 33.256 [24B].
- NOTE 6: If the AMF receives the HTTP code set to "4xx" or "5xx" as specified in 3GPP TS 29.500 [20AA] or the AMF detects that the UUAA-MM failure as specified in 3GPP TS 29.256 [21B], then the AMF considers the UUAA-MM procedure has failed.

If the AMF needs to deliver to the UE the UUAA revocation notification received from the UAS-NF, the AMF shall include the service-level-AA response with SLAR set to "Service level authentication and authorization was not successful or service level authorization is revoked" in the Service-level-AA container IE of the CONFIGURATION UPDATE COMMAND message.

If the UE supports UAS services and UAS services become enabled for the UE (e.g. because of the aerial subscription becomes a part of the UE subscription data retrieved from the UDM), the AMF may include the service-level-AA service status indication with UAS field set to "UAS services enabled" in the Service-level-AA container IE of the CONFIGURATION UPDATE COMMAND message.

If the UE supports MINT, the AMF may include the List of PLMNs to be used in disaster condition IE in the CONFIGURATION UPDATE COMMAND message.

If the UE supports MINT, the AMF may include the Disaster roaming wait range IE in the CONFIGURATION UPDATE COMMAND message.

If the UE supports MINT, the AMF may include the Disaster return wait range IE in the CONFIGURATION UPDATE COMMAND message.

NOTE 7: The AMF can determine the content of the "list of PLMN(s) to be used in disaster condition", the value of the disaster roaming wait range and the value of the disaster return wait range based on the network local configuration.

If the UE supports and the network supports and accepts the use of the PEIPS assistance information, and the AMF needs to update the PEIPS assistance information, the AMF may include the PEIPS assistance information in the Updated PEIPS assistance information IE of the CONFIGURATION UPDATE COMMAND message.

If the AMF needs to inform the UE that the use of access identity 1 is valid or is no longer valid then:

1) if the UE supports MPS indicator update via the UE configuration update procedure, the AMF:

a) informs the UE by setting the MPS indicator bit of the Priority indicator IE to "Access identity 1 valid" or "Access identity 1 not valid" respectively, in the CONFIGURATION UPDATE COMMAND message. Based on operator policy, the AMF sets the MPS indicator bit in the CONFIGURATION UPDATE COMMAND message based on the MPS priority information in the user's subscription context obtained from the UDM; or

- b) indicates "registration requested" in the Registration requested bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message;
- 2) otherwise, the AMF shall indicate "registration requested" in the Registration requested bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message.

5.4.4.3 Generic UE configuration update accepted by the UE

Upon receiving the CONFIGURATION UPDATE COMMAND message, the UE shall stop timer T3346 if running and use the contents to update appropriate information stored within the UE.

If "acknowledgement requested" is indicated in the Acknowledgement bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message, the UE shall send a CONFIGURATION UPDATE COMPLETE message.

If the UE receives a new 5G-GUTI in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new 5G-GUTI as valid, the old 5G-GUTI as invalid, stop timer T3519 if running, and delete any stored SUCI; otherwise, the UE shall consider the old 5G-GUTI as valid. The UE shall provide the 5G-GUTI to the lower layer of 3GPP access if the CONFIGURATION UPDATE COMMAND message is sent over the non-3GPP access, and the UE is in 5GMM-REGISTERED in both 3GPP access and non-3GPP access in the same PLMN.

If the UE receives a new TAI list in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new TAI list as valid and the old TAI list as invalid; otherwise, the UE shall consider the old TAI list as valid. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, and

- a) the UE already has stored allowed NSSAI for the current registration area, the UE shall store the allowed NSSAI for the current registration area in each of the allowed NSSAIs which are associated with each of the PLMNs in the registration area; and
- b) the UE already has stored rejected NSSAI for the current registration area, the UE shall store the rejected NSSAI for the current registration area in each of the rejected NSSAIs which are associated with each of the PLMNs in the registration area.

If the UE receives a new truncated 5G-S-TMSI configuration in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new truncated 5G-S-TMSI configuration as valid and the old truncated 5G-S-TMSI configuration as invalid; otherwise, the UE shall consider the old truncated 5G-S-TMSI configuration as valid.

If the UE receives a new service area list in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new service area list as valid and the old service area list as invalid; otherwise, the UE shall consider the old service area list, if any, as valid.

If the UE receives new NITZ information in the CONFIGURATION UPDATE COMMAND message, the UE considers the new NITZ information as valid and the old NITZ information as invalid; otherwise, the UE shall consider the old NITZ information as valid.

If the UE receives a LADN information IE in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the old LADN information as invalid and the new LADN information as valid, if any; otherwise, the UE shall consider the old LADN information as valid.

If the UE receives a new allowed NSSAI for the associated access type in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new allowed NSSAI as valid for the associated access type, store the allowed NSSAI for the associated access type as specified in subclause 4.6.2.2 and consider the old allowed NSSAI for the associated access type as invalid; otherwise, the UE shall consider the old allowed NSSAI as valid for the associated access type.

If the UE receives a new configured NSSAI in the CONFIGURATION UPDATE COMMAND message, the UE shall consider the new configured NSSAI for the registered PLMN or SNPN as valid and the old configured NSSAI for the registered PLMN or SNPN as invalid; otherwise, the UE shall consider the old configured NSSAI for the registered PLMN or SNPN as valid The UE shall store the new configured NSSAI as specified in subclause 4.6.2.2. In addition, if the CONFIGURATION UPDATE COMMAND message contains an NSSRG information IE, the UE shall store the contents of the NSSRG information IE as specified in subclause 4.6.2.2.

If the UE receives the Network slicing indication IE in the CONFIGURATION UPDATE COMMAND message with the Network slicing subscription change indication set to "Network slicing subscription changed", the UE shall delete the network slicing information for each and every PLMN except for the current PLMN as specified in subclause 4.6.2.2.

If the UE receives Operator-defined access category definitions IE in the CONFIGURATION UPDATE COMMAND message and the Operator-defined access category definitions IE contains one or more operator-defined access category definitions, the UE shall delete any operator-defined access category definitions stored for the RPLMN and shall store the received operator-defined access category definitions for the RPLMN. If the UE receives the Operator-defined access category definitions IE in the CONFIGURATION UPDATE COMMAND message and the Operator-defined access category definitions IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions, the UE shall delete any operator-defined access category definitions. IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions. IE contains no operator-defined access category definitions to the RPLMN. If the CONFIGURATION UPDATE COMMAND message and the Operator-defined access category definitions. IE contains no operator-defined access categor

message does not contain the Operator-defined access category definitions IE, the UE shall not delete the operatordefined access category definitions stored for the RPLMN.

If the UE receives the SMS indication IE in the CONFIGURATION UPDATE COMMAND message with the SMS availability indication set to:

- a) "SMS over NAS not available", the UE shall consider that SMS over NAS transport is not allowed by the network; and
- b) "SMS over NAS available", the UE may request the use of SMS over NAS transport by performing a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3, after the completion of the generic UE configuration update procedure.

If the UE receives the CAG information list IE in the CONFIGURATION UPDATE COMMAND message, the UE shall:

- a) replace the "CAG information list" stored in the UE with the received CAG information list IE when received in the HPLMN or EHPLMN;
- NOTE 1: When the UE receives the CAG information list IE in the HPLMN derived from the IMSI, the EHPLMN list is present and is not empty and the HPLMN is not present in the EHPLMN list, the UE behaves as if it receives the CAG information list IE in a VPLMN.
- b) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE when the UE receives the CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 2: When the UE receives the CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE are ignored.
- c) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE does not contain the serving VPLMN's entry.

The UE shall store the "CAG information list" received in the CAG information list IE as specified in annex C.

If the received "CAG information list" includes an entry containing the identity of the current PLMN and the UE had set the CAG bit to "CAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the UE shall operate as follows.

- a) If the UE receives the CONFIGURATION UPDATE COMMAND message via a CAG cell, the entry for the current PLMN in the received "CAG information list" does not include any of the CAG-ID(s) supported by the current CAG cell, and:
 - the entry for the current PLMN in the received "CAG information list" does not include an "indication that the UE is only allowed to access 5GS via CAG cells", then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list"; or
 - 2) the entry for the current PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:
 - i) if the entry for the current PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
 - ii) if the entry for the current PLMN in the received "CAG information list" does not include any CAG-ID and:
 - A) the UE does not have an emergency PDU session, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or

- B) the UE has an emergency PDU session, then the UE shall perform a local release of all PDU sessions associated with 3GPP access except for the emergency PDU session and enter the state 5GMM-REGISTERED.LIMITED-SERVICE; or
- b) If the UE receives the CONFIGURATION UPDATE COMMAND message via a non-CAG cell and the entry for the current PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:
 - if the "allowed CAG list" for the current PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
 - 2) if the entry for the current PLMN in the received "CAG information list" does not include any CAG-ID and:
 - i) the UE does not have an emergency PDU session, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
 - ii) the UE has an emergency PDU session, then the UE shall perform a local release of all PDU sessions associated with 3GPP access except for the emergency PDU session and enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

If the received "CAG information list" does not include an entry containing the identity of the current PLMN and the UE receives the CONFIGURATION UPDATE COMMAND message via a CAG cell, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".

If the CONFIGURATION UPDATE COMMAND message indicates "registration requested" in the Registration requested bit of the Configuration update indication IE and:

- a) contains no other parameters or contains at least one of the following parameters: a new allowed NSSAI, a new configured NSSAI, a new NSSRG information or the Network slicing subscription change indication, and:
 - an emergency PDU session exists, the UE shall, after the completion of the generic UE configuration update procedure and the release of the emergency PDU session, release the existing N1 NAS signalling connection. If any Tsor-cm timer(s) were running and have stopped, the UE shall attempt to obtain service on a higher priority PLMN (see 3GPP TS 23.122 [5]). Otherwise the UE start a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3; or
 - 2) no emergency PDU Session exists, the UE shall, after the completion of the generic UE configuration update procedure and the release of the existing N1 NAS signalling connection. If any Tsor-cm timer(s) were running and have stopped, the UE shall attempt to obtain service on a higher priority PLMN (see 3GPP TS 23.122 [5]). Otherwise the UE start a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3;
- b) a MICO indication is included without a new allowed NSSAI, a new configured NSSAI, a new NSSRG information or the Network slicing subscription change indication, the UE shall, after the completion of the generic UE configuration update procedure, start a registration procedure for mobility and registration update as specified in subclause 5.5.1.3 to re-negotiate MICO mode with the network;
- c) an Additional configuration indication IE is included, and:
 - 1) "release of N1 NAS signalling connection not required" is indicated in the Signalling connection maintain request bit of the Additional configuration indication IE; and
 - 2) a new allowed NSSAI, a new configured NSSAI, a new NSSRG information or the Network slicing subscription change indication is not included in the CONFIGURATION UPDATE COMMAND message,

the UE shall, after the completion of the generic UE configuration update procedure, start a registration procedure for mobility and registration update as specified in subclause 5.5.1.3; or

- d) a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested" is included, and:
 - 1) the UE is not in NB-N1 mode;

- 2) a new allowed NSSAI, a new configured NSSAI, a new NSSRG information or a Network slicing subscription change indication is not included; and
- the UE has set the RACS bit to "RACS supported" in the 5GMM capability IE of the REGISTRATION REQUEST message,

the UE shall, after the completion of the generic UE configuration update procedure, start a registration procedure for mobility and registration update as specified in subclause 5.5.1.3.

The UE receiving the rejected NSSAI in the CONFIGURATION UPDATE COMMAND message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

Unless the back-off timer value received along with the S-NSSAI is zero, the UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

NOTE 3: If the back-off timer value received along with the S-NSSAI in the rejected NSSAI for the maximum number of UEs reached is zero as specified in subclause 10.5.7.4a of TS 24.008, the UE does not consider the S-NSSAI as the rejected S-NSSAI.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or
 - 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

If the UE receives the NSAG information IE in the CONFIGURATION UPDATE COMMAND message, the UE shall store the NSAG information as specified in subclause 4.6.2.2.

If the UE receives a T3447 value IE in the CONFIGURATION UPDATE COMMAND message and has indicated "service gap control supported" in the REGISTRATION REQUEST, then the UE shall replace the stored T3447 value with the received value in the T3447 value IE, and if neither zero nor deactivated use the received T3447 value with the timer T3447 next time it is started. If the received T3447 value is zero or deactivated, then the UE shall stop the timer T3447 if running.

If the UE is not in NB-N1 mode, the UE has set the RACS bit to "RACS supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the CONFIGURATION UPDATE COMMAND message includes:

- a) a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested", the UE shall delete any network-assigned UE radio capability IDs associated with the RPLMN or RSNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription stored at the UE; or
- b) a UE radio capability ID IE, the UE shall store the UE radio capability ID as specified in annex C.

If the UE is not currently registered for emergency services and the 5GS registration result IE value in the CONFIGURATION UPDATE COMMAND message is set to "Registered for emergency services", the UE shall consider itself registered for emergency services and shall locally release all non-emergency PDU sessions, if any.

If the UE receives the service-level-AA container IE of the CONFIGURATION UPDATE COMMAND message, the UE passes it to the upper layer.

If the CONFIGURATION UPDATE COMMAND message includes the service-level-AA response in the Service-level-AA container IE with the SLAR field set to "Service level authentication and authorization was not successful or service level authorization is revoked", the UE shall forward the service-level-AA response to the upper layers, so the UUAA authorization data is deleted as specified in 3GPP TS 33.256 [24B].

If the UE receives the List of PLMNs to be used in disaster condition IE in the CONFIGURATION UPDATE COMMAND message and the UE supports MINT, the UE shall delete the "list of PLMN(s) to be used in disaster condition" stored in the ME together with the PLMN ID of the RPLMN, if any, and may store the "list of PLMN(s) to be used in disaster condition" included in the List of PLMNs to be used in disaster condition IE in the ME together with the PLMN ID of the RPLMN ID of the RPLMN.

If the UE receives the Disaster roaming wait range IE in the CONFIGURATION UPDATE COMMAND message and the UE supports MINT, the UE shall delete the disaster roaming wait range stored in the ME, if any, and store the disaster roaming wait range included in the Disaster roaming wait range IE in the ME.

If the UE receives the Disaster return wait range IE in the CONFIGURATION UPDATE COMMAND message and the UE supports MINT, the UE shall delete the disaster roaming wait range stored in the ME, if any, and store the disaster roaming wait range included in the Disaster roaming wait range IE in the ME.

If the UE receives the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message and the UE supports NR paging subgrouping, the UE shall use the PEIPS assistance information included in the Updated PEIPS assistance information IE.

If the UE receives a CONFIGURATION UPDATE COMMAND message with the MPS indicator bit set to "Access identity 1 valid", the UE shall act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2, in all NG-RAN of the registered PLMN and its equivalent PLMNs or in the case of SNPN, as described in subclause 4.5.2A, in all NG-RAN of the registered SNPN. The MPS indicator bit in the Priority indicator IE provided in the CONFIGURATION UPDATE COMMAND message is valid until the UE receives a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 not valid" or the UE receives a CONFIGURATION UPDATE COMMAND message with the Priority indicator IE set to "Access identity 1 not valid" or until the UE selects a non-equivalent PLMN (or in the case of SNPN, selects another SNPN). Access identity 1 is only applicable while the UE is in N1 mode.

If the UE supporting UAS services is not currently registered for UAS services and the CONFIGURATION UPDATE COMMAND message includes the service-level-AA service status indication in the Service-level-AA container IE with the UAS field set to "UAS services enabled", then the UE passes the service-level-AA service status indication to the upper layers.

5.4.4.4 Generic UE configuration update completion by the network

Upon receipt of the CONFIGURATION UPDATE COMPLETE message, the AMF shall stop the timer T3555.

If a new 5G-GUTI was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new 5G-GUTI as valid and the old 5G-GUTI as invalid.

If a new TAI list was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new TAI list as valid and the old TAI list as invalid.

If a new truncated 5G-S-TMSI configuration was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new truncated 5G-S-TMSI configuration as valid and the old truncated 5G-S-TMSI configuration as invalid.

If a new service area list was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new service area list as valid and the old service area list as invalid.

If new allowed NSSAI information was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new allowed NSSAI information as valid and the old allowed NSSAI information as invalid. If new configured NSSAI information was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new configured NSSAI information as valid and the old configured information as invalid. If there are active PDU sessions associated with S-NSSAI(s) not included in the new allowed NSSAI, the AMF shall notify the SMF(s) associated with these PDU sessions to initiate the network-requested PDU session release procedure according to subclause 6.3.3 in the present specification and subclause 5.15.5.2.2 in 3GPP TS 23.501 [8].

If "registration requested" was indicated in the Registration requested bit of the Configuration update indication IE in the CONFIGURATION UPDATE COMMAND message and:

- a) the CONFIGURATION UPDATE COMMAND message contained:
 - 1) an allowed NSSAI, a configured NSSAI or both;
 - 2) the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed"; or
 - 3) no other parameters; and
- b) no emergency PDU session has been established for the UE;

then the AMF shall initiate the release of the N1 NAS signalling connection.

If a LADN information IE was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the old LADN information as invalid and the new LADN information as valid, if any. In this case, if the tracking area identity list in the new LADN information does not include the current TA, the AMF shall indicate UE presence in LADN service area to the SMF (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]).

If a T3447 value was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the T3447 value as valid and if neither zero nor deactivated use the T3447 value with the timer T3447 next time it is started. If the T3447 value included in the CONFIGURATION UPDATE COMMAND message contained an indication that the timer is deactivated or timer value zero, then the AMF shall stop the timer T3447 if running.

If a CAG information IE was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new "CAG information list" as valid and the old "CAG information list" as invalid.

If a UE radio capability ID IE was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new UE radio capability ID as valid and the old UE radio capability ID as invalid.

If an Updated PEIPS assistance information IE was included in the CONFIGURATION UPDATE COMMAND message, the AMF shall consider the new PEIPS assistance information as valid and the old PEIPS assistance information, if any, as invalid.

5.4.4.5 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Transmission failure of the CONFIGURATION UPDATE COMPLETE message with TAI change from lower layers

If the current TAI is not in the TAI list, the generic UE configuration update procedure shall be aborted and a registration procedure for mobility and periodic registration update shall be initiated.

If the current TAI is still part of the TAI list, it is up to the UE implementation how to re-run the ongoing procedure that triggered the generic UE configuration update procedure.

b) Transmission failure of CONFIGURATION UPDATE COMPLETE message indication without TAI change from lower layers

It is up to the UE implementation how to re-run the ongoing procedure that triggered the generic UE configuration update procedure.

c) Generic UE configuration update and de-registration procedure collision

If the UE receives CONFIGURATION UPDATE COMMAND message after sending a DEREGISTRATION REQUEST message and the access type included in the DEREGISTRATION REQEUST message is same as the access in which the CONFIGURATION UPDATE COMMAND message is received, then the UE shall ignore the CONFIGURATION UPDATE COMMAND message and proceed with the de-registration procedure. Otherwise, the UE shall proceed with both the procedures.

- d) Void
- e) Generic UE configuration update and service request procedure collision

If the SERVICE REQUEST message does not include UE request type IE with Request type value set to "NAS signalling connection release" and the UE receives a CONFIGURATION UPDATE COMMAND message before the ongoing service request procedure has been completed, the UE shall proceed with both the procedures.

If the SERVICE REQUEST message includes UE request type IE with Request type value set to "NAS signalling connection release" and the UE receives a CONFIGURATION UPDATE COMMAND message before the ongoing service request procedure has been completed, the UE shall ignore the CONFIGURATION UPDATE COMMAND message and proceed with the service request procedure.

f) "CAG information list" is received and the UE is operating in SNPN access operation mode

If the UE receives the CAG information list IE in the CONFIGURATION UPDATE COMMAND message and the UE is operating in SNPN access operation mode, the UE shall ignore the content of CAG information list IE.

5.4.4.6 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3555.

The network shall, on the first expiry of the timer T3555, retransmit the CONFIGURATION UPDATE COMMAND message and shall reset and start timer T3555. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3555, the procedure shall be aborted. In addition, if the CONFIGURATION UPDATE COMMAND message includes the 5G-GUTI IE, the network shall behave as described in case b)-1) below.

b) Lower layer failure.

If a lower layer failure is detected before the CONFIGURATION UPDATE COMPLETE message is received and:

 if the CONFIGURATION UPDATE COMMAND message includes the 5G-GUTI IE, the old and the new 5G-GUTI shall be considered as valid until the old 5G-GUTI can be considered as invalid by the AMF. If a new TAI list was provided in the CONFIGURATION UPDATE COMMAND message, the old and new TAI list shall also be considered as valid until the old TAI list can be considered as invalid by the AMF.

Additionally, if the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message includes a new Paging subgroup ID and the UE is previously assigned a different

Paging subgroup ID then, the AMF shall consider both, the old and new Paging subgroup IDs as valid until the old Paging subgroup ID can be considered as invalid by the AMF.

During this period the AMF:

- i) may first use the old 5G-S-TMSI from the old 5G-GUTI and the old Paging subgroup ID, if any, for paging within the area defined by the old TAI list for an implementation dependent number of paging attempts for network originated transactions. If a new TAI list was provided in the CONFIGURATION UPDATE COMMAND message, the new TAI list should also be used for paging. Upon response from the UE, the AMF may re-initiate the CONFIGURATION UPDATE COMMAND. If the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message includes a new Paging subgroup ID, then the AMF may re-initiate the CONFIGURATION UPDATE COMMAND message includes a new Paging subgroup ID, then the AMF may re-initiate the CONFIGURATION UPDATE COMMAND. If the response is received from a tracking area within the old and new TAI list, the network shall re-initiate the CONFIGURATION UPDATE COMMAND. If the response is received from a tracking area within the old and new TAI list, the network shall re-initiate the CONFIGURATION UPDATE COMMAND message. If no response is received to the paging attempts, the network may use the new 5G-S-TMSI from the new 5G-GUTI and the new Paging subgroup ID, if any, for paging for an implementation dependent number of paging attempts. In this case, if a new TAI list was provided with new 5G-GUTI in the CONFIGURATION UPDATE COMMAND message, the new TAI list shall be used instead of the old TAI list. Upon response from the UE the AMF shall consider the new 5G-GUTI and the new PEIPS assistance information as valid and the old 5G-GUTI and the old PEIPS assistance information as invalid.
- ii) shall consider the new 5G-GUTI as valid if it is used by the UE and, additionally, the new TAI list as valid if it was provided with this 5G-GUTI in the CONFIGURATION UPDATE COMMAND message; and
- iii) may use the identification procedure followed by a new generic UE configuration update procedure if the UE uses the old 5G-GUTI; or
- 2) if the CONFIGURATION UPDATE COMMAND message does not include the 5G-GUTI IE and:
 - the CONFIGURATION UPDATE COMMAND message does not contain the Allowed NSSAI IE, the Rejected NSSAI IE, the Extended rejected NSSAI IE, or the Updated PEIPS assistance information IE, the network shall abort the procedure; or
 - ii) the CONFIGURATION UPDATE COMMAND message contains the Allowed NSSAI IE, the Rejected NSSAI IE, the Extended rejected NSSAI IE, or the Updated PEIPS assistance information IE, the network shall either abort the procedure or retransmit the CONFIGURATION UPDATE COMMAND message on expiry of the timer T3555. The retransmission shall not be repeated more than four times. If the retransmission is repeated for four times, the network shall abort the procedure.
- c) Generic UE configuration update and UE initiated de-registration procedure collision.

If the network receives a DEREGISTRATION REQUEST message before the ongoing generic UE configuration update procedure has been completed, the network shall abort the generic UE configuration update procedure and shall progress the de-registration procedure.

d) Generic UE configuration update and registration procedure for mobility and periodic registration update collision

If the network receives a REGISTRATION REQUEST message before the ongoing generic UE configuration update procedure has been completed, the network shall abort the generic UE configuration update procedure and shall progress the registration procedure for mobility and periodic registration update procedure.

e) Generic UE configuration update and service request procedure collision

If the network receives a SERVICE REQUEST message before the ongoing generic UE configuration update procedure has been completed and the SERVICE REQUEST message does not include UE request type IE with Request type value set to "NAS signalling connection release", both the procedures shall be progressed.

If the network receives a SERVICE REQUEST message before the ongoing generic UE configuration update procedure has been completed and the SERVICE REQUEST message includes UE request type IE with Request type value set to "NAS signalling connection release", the network shall abort the generic UE configuration update procedure and shall progress the service request procedure.

5.4.5 NAS transport procedure(s)

5.4.5.1 General

The purpose of the NAS transport procedures is to provide a transport of payload between the UE and the AMF. The type of the payload is identified by the Payload container type IE and includes one of the following:

- a) a single 5GSM message;
- b) SMS;
- c) an LPP message (see 3GPP TS 37.355 [26]);
- d) an SOR transparent container;
- e) a UE policy container;
- f) a UE parameters update transparent container;
- g) a location services message (see 3GPP TS 24.080 [13A]);
- h) a CIoT user data container;
- i) a Service-level-AA container; or
- j) Multiple payloads.

For payload type a) to e), g) and h), along with the payload, the NAS transport procedure may transport the associated information (e.g. PDU session information for 5GSM message payload).

For payload type j), the Payload container IE consists a list of payload container entries, where each of payload container entry contains the payload and optional associated information (e.g. PDU session information for 5GSM message payload).

NOTE: Payload type can be set to "Multiple payloads" if there are more than one payloads to be transported using the NAS transport procedures.

5.4.5.2 UE-initiated NAS transport procedure

5.4.5.2.1 General

The purpose of the UE-initiated NAS transport procedure is to provide a transport of:

- a) a single 5GSM message as defined in subclause 8.3;
- b) SMS (see 3GPP TS 24.011 [13]);
- c) an LPP message;
- d) an SOR transparent container;
- e) a UE policy container; or
- f) a UE parameters update transparent container;
- g) a location services message;
- h) a CIoT user data container;
- i) a Service-level-AA container; or
- j) multiple of the above types.

and:

- for a) to e), g) and h), optional associated payload routing information from the UE to the AMF in a 5GMM message; and
- for j), the Payload container IE consists a list of payload container entries, where each of the payload container entry contains the payload and optional associated payload routing information (e.g. PDU session information for 5GSM message payload).

5.4.5.2.2 UE-initiated NAS transport procedure initiation

In the connected mode, the UE initiates the NAS transport procedure by sending the UL NAS TRANSPORT message to the AMF, as shown in figure 5.4.5.2.2.1.

In case a) in subclause 5.4.5.2.1, the UE shall:

- include the PDU session information (PDU session ID, old PDU session ID, S-NSSAI, mapped S-NSSAI (if available in roaming scenarios), DNN, request type), if available;
- set the Payload container type IE to "N1 SM information"; and
- set the Payload container IE to the 5GSM message.

The UE shall set the PDU session ID IE to the PDU session ID. If an old PDU session ID is to be included, the UE shall set the Old PDU session ID IE to the old PDU session ID.

If an S-NSSAI is to be included, the UE shall set the S-NSSAI IE to the S-NSSAI selected for the PDU session from the allowed NSSAI for the current PLMN or SNPN, associated with the mapped S-NSSAI (if available in roaming scenarios).

If a DNN is to be included, the UE shall set the DNN IE to the DNN. 5GSM procedures specified in clause 6 describe conditions for inclusion of the S-NSSAI, mapped S-NSSAI (if available in roaming scenarios), and the DNN.

If a request type is to be included, the UE shall set the Request type IE to the request type. The request type is not provided along 5GSM messages other than the PDU SESSION ESTABLISHMENT REQUEST message and the PDU SESSION MODIFICATION REQUEST message.

If an MA PDU session information is to be included, the UE shall set the MA PDU session information IE to the MA PDU session information. The MA PDU session information is not provided along 5GSM messages other than the PDU SESSION ESTABLISHMENT REQUEST message and the PDU SESSION MODIFICATION REQUEST message as specified in 3GPP TS 24.193 [13B].

In case b) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "SMS"; and
- set the Payload container IE to the SMS payload.

Based on the UE preferences regarding access selection for mobile originated (MO) transmission of SMS over NAS as described in 3GPP TS 23.501 [8]:

- when SMS over NAS is preferred to be sent over 3GPP access: the UE attempts to deliver MO SMS over NAS via the 3GPP access if the UE is registered over both 3GPP access and non-3GPP access. If the delivery of SMS over NAS via the 3GPP access is not available, the UE attempts to deliver MO SMS over NAS via the non-3GPP access; and
- when SMS over NAS is preferred to be sent over non-3GPP access: the UE attempts to deliver MO SMS over NAS via the non-3GPP access if the UE is registered over both 3GPP access and non-3GPP access. If the delivery of SMS over NAS via the non-3GPP access is not available, the UE attempts to deliver MO SMS over NAS via the 3GPP access.

In case c) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "LTE Positioning Protocol (LPP) message container";
- set the Payload container IE to the LPP message payload; and

- set the Additional information IE to the routing information provided by the upper layer location services application.

In case d) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "SOR transparent container"; and
- set the Payload container IE to the UE acknowledgement due to successful reception of steering of roaming information, and set the ME support of SOR-CMCI indicator to "SOR-CMCI supported by the ME" in the Payload container IE carrying the acknowledgement (see 3GPP TS 23.122 [5]).

In case e) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "UE policy container"; and
- set the contents of the Payload container IE as specified in Annex D.

In case f) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "UE parameters update transparent container"; and
- set the contents of the Payload container IE to the UE acknowledgement due to successful reception of UE parameters update data (see 3GPP TS 23.502 [9]).

In case g) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "Location services message container";
- set the Payload container IE to the Location services message payload; and
- set the Additional information IE to the routing information, if provided by the upper layer location services application.

In case h) in subclause 5.4.5.2.1, the UE shall:

- include the PDU session ID, and Release assistance indication (if available);
- set the Payload container type IE to "CIoT user data container"; and
- set the Payload container IE to the user data container.

In case i) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "Service-level-AA container"; and
- set the Payload container IE to the Service-level-AA container.

In case j) in subclause 5.4.5.2.1, the UE shall:

- set the Payload container type IE to "Multiple payloads"; and
- set each payload container entry of the Payload container IE (see subclause 9.11.3.39), as follows:
 - i) set the payload container type field of the payload container entry to a payload container type value set in the Payload container type IE as specified in cases a) to i) above;
 - ii) set the payload container entry contents field of the payload container entry to the payload container contents set in the Payload container IE as specified in cases a) to i) above, and
 - iii) set the optional IE fields, if any, to the optional associated payload routing information as specified in casesa) to i) above.

UE AMF

Figure 5.4.5.2.2.1: UE-initiated NAS transport procedure

5.4.5.2.3 UE-initiated NAS transport of messages accepted by the network

Upon reception of a UL NAS TRANSPORT message, if the Payload container type IE is set to:

- a) "N1 SM information", the AMF looks up a PDU session routing context for:
 - 1) the UE and the PDU session ID IE in case the Old PDU session ID IE is not included, and:
- NOTE 1: If the Old PDU session ID IE is not included in the UL NAS TRANSPORT message and the AMF has received a reallocation requested indication from the SMF, the AMF needs to ignore the reallocation requested indication.
 - i) if the AMF has a PDU session routing context for the PDU session ID and the UE, and the Request type IE is either not included or is included but set to other value than "initial request", "existing PDU session", "initial emergency request", "existing emergency PDU session" or "MA PDU request", the AMF shall send the 5GSM message, and the PDU session ID IE towards the SMF identified by the SMF ID of the PDU session routing context;
 - ii) if the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is not an emergency PDU session, the Request type IE is included and is set to "existing PDU session" or "MA PDU request", and the S-NSSAI associated with the PDU session identified by the PDU session ID is allowed for the target access type, the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN (if received) and the request type towards the SMF identified by the SMF ID of the PDU session routing context;
 - iii) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, and the Request type IE is included and is set to "initial request" or "MA PDU request":
 - A) the AMF shall select an SMF with following handlings in case the UE is not registered for onboarding services in SNPN:

If the S-NSSAI IE is not included and the allowed NSSAI contains:

- one S-NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI as the S-NSSAI;
- two or more S-NSSAIs and the user's subscription context obtained from UDM contains only one default S-NSSAI that is included in the allowed NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI that matches the default S-NSSAI as the S-NSSAI; or
- two or more S-NSSAIs and the user's subscription context obtained from UDM contains zero, two or more default S-NSSAI(s) included in the allowed NSSAI, the AMF shall use an S-NSSAI in the allowed NSSAI selected based on operator policy as the S-NSSAI.

If the DNN IE is included, the AMF shall use the UE requested DNN as the DNN determined by the AMF; and

If the DNN IE is not included, and the user's subscription context obtained from UDM:

- contains the default DNN for the S-NSSAI, the AMF shall use the default DNN as the DNN determined by the AMF; and
- does not contain the default DNN for the S-NSSAI, the AMF shall use a locally configured DNN as the DNN determined by the AMF;
- A1) the AMF shall select an SMF with following handlings in case the UE is registered for onboarding services in SNPN:
 - if the AMF onboarding configuration data does not contain a configured SMF used for onboarding services in SNPN and contains the S-NSSAI used for onboarding services in SNPN, the AMF shall use the S-NSSAI used for onboarding services in SNPN as the S-NSSAI;
 - if the AMF onboarding configuration data does not contain a configured SMF used for onboarding services in SNPN and contains the DNN used for onboarding services in SNPN, the AMF shall use the DNN used for onboarding services in SNPN as the DNN;
 - if the AMF onboarding configuration data does not contain the S-NSSAI used for onboarding services in SNPN, does not contain the DNN used for onboarding services in SNPN, and contains a configured SMF used for onboarding services in SNPN, the AMF shall select the configured SMF used for onboarding services in SNPN;
 - if the AMF onboarding configuration data contains the S-NSSAI used for onboarding services in SNPN, the DNN used for onboarding services in SNPN, or both, and contains a configured SMF used for onboarding services in SNPN, the AMF shall use the S-NSSAI used for onboarding services in SNPN, if any, as the S-NSSAI, and use the DNN used for onboarding services in SNPN, if any, as the DNN or shall select the configured SMF used for onboarding services in SNPN, according to local policy; and
 - if the AMF onboarding configuration data contains none of the S-NSSAI used for onboarding services in SNPN, the DNN used for onboarding services in SNPN and a configured SMF used for onboarding services in SNPN, the AMF handling is implementation specific; and
- NOTE 2: The AMF can e.g. use a locally configured DNN used for onboarding services in SNPN as the DNN determined by the AMF.
- NOTE 3: SMF selection is outside the scope of the present document.
- NOTE 4: As part of SMF selection, the PCF can provide the AMF with a DNN selected by the network different from the DNN determined by the AMF.
 - B) if the SMF selection is successful:
 - if the DNN selected by the network is a LADN DNN, the AMF shall determine the UE presence in LADN service area;
 - the AMF shall store a PDU session routing context for the PDU session ID and the UE, shall set the SMF ID in the stored PDU session routing context to the SMF ID corresponding to the DNN in the user's subscription context obtained from the UDM; and
 - the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN determined by the AMF, DNN selected by the network (if different from DNN determined by the AMF), the request type, the MA PDU session information, UE presence in LADN service area (if DNN received corresponds to an LADN DNN, and the onboarding indication (if the UE is registered for onboarding services in SNPN) towards the SMF identified by the SMF ID of the PDU session routing context;
- NOTE 5: The MA PDU session information is not sent towards the SMF if the DNN received corresponds to an LADN DNN.
 - iv) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "existing PDU session" or "MA PDU request", and the AMF retrieves an SMF ID associated with:
 - A) the PDU session ID matching the PDU session ID received from the UE, if any; or
 - B) the DNN matching the DNN received from the UE, otherwise;

such that the SMF ID includes a PLMN identity corresponding to the UE's HPLMN or the current PLMN, then:

- A) the AMF shall store a PDU session routing context for the PDU session ID and the UE, shall set the SMF ID in the stored PDU session routing context to the retrieved SMF ID; and
- B) the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN (if received) and the request type towards the SMF identified by the SMF ID of the PDU session routing context;
- v) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "initial emergency request", and the AMF does not have a PDU session routing context for another PDU session ID of the UE indicating that the PDU session is an emergency PDU session:
 - A) the AMF shall select an SMF. The AMF shall use the emergency DNN from the AMF emergency configuration data as the DNN, if configured. The AMF shall derive the SMF from the emergency DNN or use the statically configured SMF from the AMF emergency configuration data, if configured; and
 - B) if the SMF selection is successful:
 - the AMF shall store a PDU session routing context for the PDU session ID and the UE, shall set the SMF ID in the stored PDU session routing context to the SMF ID of the selected SMF, and shall store an indication that the PDU session is an emergency PDU session in the stored PDU session routing context; and
 - the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data), and the request type towards the SMF identified by the SMF ID of the PDU session routing context; and
- vi) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "initial emergency request", and the AMF has a PDU session routing context indicating that the PDU session is an emergency PDU session for another PDU session ID of the UE:
 - A) the AMF shall store a PDU session routing context for the PDU session ID and the UE and shall set the SMF ID in the stored PDU session routing context to the SMF ID of the PDU session routing context for the other PDU session ID of the UE; and
 - B) the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data) and the request type towards the SMF identified by the SMF ID of the PDU session routing context; or
- vii)if the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is an emergency PDU session, and the Request type IE is included and is set to "existing emergency PDU session", the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data), and the request type towards the SMF identified by the SMF ID of the PDU session routing context; and
- viii) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "existing emergency PDU session", and the AMF retrieves an SMF ID associated with emergency services such that the SMF ID includes a PLMN identity corresponding to the current PLMN, then:
 - A) the AMF shall store a PDU session routing context for the PDU session ID and the UE, shall set the SMF ID in the stored PDU session routing context to the retrieved SMF ID; and
 - B) the AMF shall send the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data), and the request type towards the SMF identified by the SMF ID of the PDU session routing context; or

- 2) the UE and the Old PDU session ID IE in case the Old PDU session ID IE is included, and:
 - i) the AMF has a PDU session routing context for the old PDU session ID and the UE and does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "initial request", and the AMF received a reallocation requested indication from the SMF indicating that the SMF is to be reused, the AMF shall store a PDU session routing context for the PDU session ID and the UE, set the SMF ID in the stored PDU session routing context to the SMF ID of the PDU session routing context for the old PDU session ID and the UE. If the DNN is a LADN DNN, the AMF shall determine the UE presence in LADN service area. The AMF shall send the 5GSM message, the PDU session ID, the old PDU session ID, the S-NSSAI (if received), the mapped S-NSSAI (in roaming scenarios), the DNN, the request type and UE presence in LADN service area (if DNN received corresponds to an LADN DNN) towards the SMF identified by the SMF ID of the PDU session routing context;
 - ii) the AMF has a PDU session routing context for the old PDU session ID and the UE and does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is included and is set to "initial request", and the AMF received a reallocation requested indication from the SMF indicating that the SMF is to be reallocated:
 - A) the AMF shall select an SMF with the following handling;

If the S-NSSAI IE is not included and the allowed NSSAI contains:

- one S-NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI as the S-NSSAI;
- two or more S-NSSAIs and the user's subscription context obtained from UDM contains only one default S-NSSAI that is included in the allowed NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI that matches the default S-NSSAI; or
- two or more S-NSSAIs and the user's subscription context obtained from UDM contains zero, two
 or more default S-NSSAI(s) included in the allowed NSSAI, the AMF shall use an S-NSSAI in the
 allowed NSSAI selected based on operator policy as the S-NSSAI.

If the DNN is a LADN DNN, the AMF shall determine the UE presence in LADN service area.

- B) if the SMF selection is successful:
 - the AMF shall store a PDU session routing context for the PDU session ID and the UE and set the SMF ID of the PDU session routing context to the SMF ID of the selected SMF; and
 - the AMF shall send the 5GSM message, the PDU session ID, the old PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN, the request type, the MA PDU session information and UE presence in LADN service area (if DNN received corresponds to an LADN DNN) towards the SMF identified by the SMF ID of the PDU session routing context for the PDU session ID and the UE;
- NOTE 6: The MA PDU session information is not sent towards the SMF if the DNN received corresponds to an LADN DNN.
- b) "SMS", the AMF shall send the content of the Payload container IE to the SMSF associated with the UE;
- c) "LTE Positioning Protocol (LPP) message container", the AMF shall send the Payload container type and the content of the Payload container IE to the LMF associated with the routing information included in the Additional information IE of the UL NAS TRANSPORT message;
- d) "SOR transparent container", the AMF shall send the content of the Payload container IE to the UDM (see 3GPP TS 29.503 [20AB]);
- e) "UE policy container", the AMF shall send the content of the Payload container IE to the PCF.
- f) "UE parameters update transparent container", the AMF shall send the content of the Payload container IE to the UDM.
- g) "Location services message container":

- if the Additional information IE is not included in the UL NAS TRANSPORT message, the AMF shall provide the Payload container type and the content of the Payload container IE to the location services application; and
- 2) if the Additional information IE is included in the UL NAS TRANSPORT message, the AMF shall send the Payload container type and the content of the Payload container IE to an LMF associated with routing information included in the Additional information IE of the UL NAS TRANSPORT message.
- h) "CIoT user data container", the AMF shall look up a PDU session routing context for the UE and the PDU session ID, and
 - 1) send the content of the Payload container IE towards the SMF identified by the SMF ID of the PDU session routing context; and
 - 2) initiate the release of the N1 NAS signalling connection:
 - i) if the Release assistance indication IE is included in the UL NAS TRANSPORT message and the DDX field of the Release assistance indication IE indicates "No further uplink and no further downlink data transmission subsequent to the uplink data transmission is expected" and if there is no downlink signalling or downlink data for the UE; or
 - ii) upon subsequent delivery of the next received downlink data transmission to the UE if the Release assistance indication IE is included in the UL NAS TRANSPORT message and the DDX field of the Release assistance indication IE indicates "Only a single downlink data transmission and no further uplink data transmission subsequent to the uplink data transmission is expected" and if there is no additional downlink signalling or downlink data for the UE.
- i) "Service-level-AA container" and the Service-level-AA container is included in the Payload container IE of the UL NAS TRANSPORT message, and the Service-level device ID included in the Service-level-AA container is set to a CAA-level UAV ID, the AMF shall send the content of the Payload container IE to the UAS-NF corresponding to the CAA-level UAV ID. If the Service-level device ID is not included in the Service-level-AA container and a CAA-level UAV ID is included in the 5GMM context of the UE, then the AMF shall send the content of the Payload container IE to the UAS-NF corresponding to the CAA-level UAV ID is included in the 5GMM context of the UE.
- j) "Multiple payloads", the AMF shall first decode the content of the Payload container IE (see subclause 9.11.3.39) to obtain the number of payload container entries and for each payload container entry, the AMF shall:
 - i) decode the payload container type field;
 - ii) decode the optional IE fields and the payload container contents field in the payload container entry; and
 - iii) handle the content of each payload container entry the same as the content of the Payload container IE and the associated optional IEs as specified in bullets a) to i) above according to the payload container type field.

5.4.5.2.4 UE-initiated NAS transport of messages not accepted by the network

Upon reception of an UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information" and the UE is not configured for high priority access in selected PLMN, and:

- a) if the Request type IE is set to "initial request", "existing PDU session" or "MA PDU request";
 - DNN based congestion control is activated for the DNN included in the UL NAS TRANSPORT message, or DNN based congestion control is activated for the selected DNN in case of no DNN included in the UL NAS TRANSPORT message, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #22 "congestion" as specified in subclause 5.4.5.3.1 case f);
 - 2) S-NSSAI and DNN based congestion control is activated for the S-NSSAI and DNN included in the UL NAS TRANSPORT message, or S-NSSAI and DNN based congestion control is activated for the S-NSSAI included in the UL NAS TRANSPORT message and the selected DNN in case of no DNN included in the UL NAS TRANSPORT message, or S-NSSAI and DNN based congestion control is activated for the selected S-NSSAI in case of no S-NSSAI included in the UL NAS TRANSPORT message and the DNN

included in the UL NAS TRANSPORT message, or S-NSSAI and DNN based congestion control is activated for the selected S-NSSAI and the selected DNN in case of no S-NSSAI and no DNN included in the UL NAS TRANSPORT message, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #67 "insufficient resources for specific slice and DNN" as specified in subclause 5.4.5.3.1 case f);

3) S-NSSAI only based congestion control is activated for the S-NSSAI included in the UL NAS TRANSPORT message, or S-NSSAI based congestion control is activated for the selected S-NSSAI in case of no S-NSSAI included in the UL NAS TRANSPORT message, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #69 "insufficient resources for specific slice" as specified in subclause 5.4.5.3.1 case f);

b) void;

- c) if the Request type IE is set to "modification request" and the PDU session is not an emergency PDU session;
 - DNN based congestion control is activated for the stored DNN, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #22 "congestion" as specified in subclause 5.4.5.3.1 case f);
 - S-NSSAI and DNN based congestion control is activated for the stored S-NSSAI and DNN, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #67 "insufficient resources for specific slice and DNN" as specified in subclause 5.4.5.3.1 case f);
 - 3) S-NSSAI only based congestion control is activated for the stored S-NSSAI, e.g. configured by operation and maintenance, the AMF shall send back to the UE the 5GSM message which was not forwarded, a back-off timer value and 5GMM cause #69 "insufficient resources for specific slice" as specified in subclause 5.4.5.3.1 case f); or
- d) the timer T3447 is running and the UE does not support service gap control:
 - 1) the Request type IE:
 - i) is set to "initial request";
 - ii) is set to "existing PDU session"; or
 - iii) is set to "modification request" and the PDU session being modified is a non-emergency PDU session;
 - 2) the current NAS signalling connection was not triggered by paging; and
 - 3) mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection,

the AMF shall send back to the UE the message which was not forwarded, send the 5GMM cause #22 "Congestion", and may include a back-off timer set to the remaining time of the timer T3447 as specified in subclause 5.4.5.3.1 case f).

Upon reception of a UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information", the Request type IE is set to "initial request", "existing PDU session" or "MA PDU request", and the AMF determines that the PLMN's maximum number of PDU sessions has already been reached for the UE, the AMF shall send back to the UE the 5GSM message which was not forwarded and 5GMM cause #65 "maximum number of PDU sessions reached" as specified in subclause 5.4.5.3.1 case h).

Upon reception of a UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information", the Request type IE is set to "initial request", and

- a) the UE is in NB-N1 mode;
- b) the UE has indicated preference for user plane CIoT 5GS optimization;
- c) the network accepted the use of user plane CIoT 5GS optimization; and
- d) the AMF determines that there are user-plane resources established for a number of PDU sessions that is equal to the UE's maximum number of supported user-plane resources (see 3GPP TS 23.501 [8]),

the AMF shall either:

- a) send back to the UE the message which was not forwarded as specified in in subclause 5.4.5.3.1 case h1); or
- b) proceed with the PDU session establishment and include the Control Plane CIoT 5GS Optimisation indication or Control Plane Only indicator to the SMF.

Upon reception of an UL NAS TRANSPORT message, if the Payload container type IE is set to "CIoT user data container", the UE is not configured for high priority access in selected PLMN, and:

- a) the timer T3447 is running and the UE does not support service gap control;
- b) the current NAS signalling connection was not triggered by paging; and
- c) mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection;

the AMF shall send back to the UE the CIoT user data which was not forwarded, send the 5GMM cause #22 "Congestion", and include a back-off timer set to the remaining time of the timer T3447 as specified in subclause 5.4.5.3.1 case 12).

Upon reception of a UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information", the Request type IE is set to "existing PDU session", and

- a) the UE is in NB-N1 mode;
- b) the UE has indicated preference for user plane CIoT 5GS optimization;
- c) the network accepted the use of user plane CIoT 5GS optimization; and
- d) the AMF determines that there are user-plane resources established for a number of PDU sessions that equals to the UE's maximum number of supported user-plane resources (see 3GPP TS 23.501 [8]),

the AMF shall send back to the UE the message which was not forwarded as specified in in subclause 5.4.5.3.1 case h1).

Upon reception of an UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information", the Request type IE is set to "initial request" or "modification request", the associated S-NSSAI that the AMF determined through the S-NSSAI IE or the PDU session ID IE is an S-NSSAI for which the AMF is performing NSSAA, and the AMF determines to not forward the 5GSM message to the SMF based on local policy, the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case h2).

Upon reception of an UL NAS TRANSPORT message, if the Payload container type IE is set to "SMS" or "LTE Positioning Protocol (LPP) message container", the UE is not configured for high priority access in selected PLMN, and:

- a) the timer T3447 is running and the UE does not support service gap control;
- b) the current NAS signalling connection was not triggered by paging; and
- c) mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection;

the AMF shall abort the procedure.

NOTE 1: In this state the NAS signalling connection can be released by the network.

Upon reception of an UL NAS TRANSPORT message, if the Payload container type IE is set to "N1 SM information", the Request type IE is set to "initial request", and:

- a) the determined DNN, S-NSSAI or both DNN and S-NSSAI are identified for UAS services; and
- b) the UE is marked in the UE's 5GMM context that it is not allowed to request UAS services;

the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case h4).

NOTE 2: The UE marked in the UE's 5GMM context as not allowed to request UAS services happens in the case that the UUAA-MM procedure needs to be performed during the registration procedure according to operator policy.

5.4.5.2.5 Abnormal cases on the network side

The following abnormal cases in AMF are identified:

- a) If the Payload container type IE is set to "N1 SM information" and:
 - if the Old PDU session ID IE is not included in the UL NAS TRANSPORT message, the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "initial request" or "MA PDU request", and the SMF selection fails, then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
 - 2) if the Old PDU session ID IE is included in the UL NAS TRANSPORT message, the AMF has a PDU session routing context for the old PDU session ID and the UE and does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "initial request", the AMF received a reallocation requested indication from the SMF indicating that the SMF is to be reallocated, and the SMF selection fails, then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
 - 3) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "existing PDU session" or "MA PDU request", and the user's subscription context obtained from the UDM does not contain an SMF ID for the PDU session ID matching the PDU session ID received from the UE or for the DNN matching the DNN received from the UE such that the SMF ID includes a PLMN identity corresponding to the UE's HPLMN or the current PLMN or the PLMN ID part of the current SNPN, then the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f).
 - 4) if the Old PDU session ID IE is included in the UL NAS TRANSPORT message, and the AMF has a PDU session routing context for the old PDU session ID and the UE and does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "initial request" and the AMF has not received a reallocation requested indication, the AMF should select an SMF with following handlings:
 - i) if the S-NSSAI IE is not included and the allowed NSSAI contains:
 - A) one S-NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI as the S-NSSAI;
 - B) two or more S-NSSAIs and the user's subscription context obtained from UDM contains only one default S-NSSAI that is included in the allowed NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI as the S-NSSAI; or
 - C) two or more S-NSSAIs and the user's subscription context obtained from UDM contains zero, two or more default S-NSSAI(s) included in the allowed NSSAI, the AMF shall use an S-NSSAI in the allowed NSSAI selected based on operator policy as the S-NSSAI;
 - ii) if the DNN IE is not included, and the user's subscription context obtained from UDM:
 - A) contains the default DNN for the S-NSSAI, the AMF shall use the default DNN as the DNN; and
 - B) does not contain the default DNN for the S-NSSAI, the AMF shall use a locally configured DNN as the DNN;
 - iii) if the DNN selected by the network is a LADN DNN, the AMF shall determine the UE presence in LADN service area;
 - iv) if the SMF selection is successful, the AMF should store a PDU session routing context for the PDU session ID and the UE, set the SMF ID in the stored PDU session routing context to the selected SMF ID, and forward the 5GSM message, the PDU session ID, the old PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN determined by the AMF, DNN selected by the network (if different from DNN determined by the AMF), the request type and UE presence in LADN service area (if DNN selected by the network corresponds to an LADN DNN) towards the SMF ID of the PDU session routing context; and

- v) if the SMF selection fails, then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
- 5) if the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is an emergency PDU session, the Request type IE is set to "initial emergency request", the AMF should forward the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data) and the request type towards the SMF ID of the PDU session routing context;
- 6) if the Request type IE is set to "initial emergency request" and the S-NSSAI or the DNN is received, the AMF ignores the received S-NSSAI or the DNN and uses the emergency DNN from the AMF emergency configuration data, if any;
- 7) if the AMF does not have a PDU session routing context for the PDU session ID and the UE, and the Request type IE of the UL NAS TRANSPORT message is either not provided or is provided but set to other value than "initial request", "existing PDU session", "initial emergency request", "existing emergency PDU session" and "MA PDU request", then the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
- 8) if the AMF unsuccessfully attempted to forward the 5GSM message, the PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN and the request type (if received) towards a SMF ID, then the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f).
- 9) if the Old PDU session ID IE is included in the UL NAS TRANSPORT message, the AMF does not have a PDU session routing context for the old PDU session ID and the UE, the AMF does not have a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "initial request", the AMF should select an SMF with following handlings:
 - i) if the S-NSSAI IE is not included and the allowed NSSAI contains:
 - A) one S-NSSAI, the AMF shall use the S-NSSAI in the allowed NSSAI as the S-NSSAI;
 - B) two or more S-NSSAIs and the user's subscription context obtained from UDM contains only one default S-NSSAI that is included in the allowed NSSAI, the AMF shall use the default S-NSSAI in the allowed NSSAI as the S-NSSAI; or
 - C) two or more S-NSSAIs and the user's subscription context obtained from UDM contains zero, two or more default S-NSSAI(s) included in the allowed NSSAI, the AMF shall use an S-NSSAI in the allowed NSSAI selected based on operator policy as the S-NSSAI.
 - ii) if the DNN IE is not included, and the user's subscription context obtained from UDM:
 - A) contains the default DNN for the S-NSSAI, the AMF shall use the default DNN as the DNN; and
 - B) does not contain the default DNN for the S-NSSAI, the AMF shall use a locally configured DNN as the DNN;
 - iii) if the DNN selected by the network is a LADN DNN, the AMF shall determine the UE presence in LADN service area;
 - iv) if the SMF selection is successful, the AMF should store a PDU session routing context for the PDU session ID and the UE, set the SMF ID in the stored PDU session routing context to the selected SMF ID, and forward the 5GSM message, the PDU session ID, the old PDU session ID, the S-NSSAI, the mapped S-NSSAI (in roaming scenarios), the DNN determined by the AMF, DNN selected by the network (if different from DNN determined by the AMF), the request type and UE presence in LADN service area (if DNN selected by the network corresponds to an LADN DNN) towards the SMF ID of the PDU session routing context; and
 - v) if the SMF selection fails, then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
- 10) if the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is not an emergency PDU session, and the Request type IE is included

and is set to "existing emergency PDU session", the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);

- 11)if the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is an emergency PDU session, and the Request type IE is included and is set to "existing PDU session", the AMF may forward the 5GSM message, the PDU session ID, the S-NSSAI (if configured in the AMF emergency configuration data), the DNN (if configured in the AMF emergency configuration data), the SMF identified by the SMF ID of the PDU session routing context;
- 12) if the AMF has a PDU session routing context for the PDU session ID and the UE, the Request type IE is set to "initial request", then the AMF shall perform a local release of the PDU session identified by the PDU session ID and shall request the SMF to perform a local release of the PDU session, and proceed as specified in subclause 5.4.5.2.3;
- 13) if the Request type IE is set to "initial request" or "modification request", and the S-NSSAI IE contains an S-NSSAI that is not allowed by the network, then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e), case f) or h4);
- 14) if the Request type IE is set to "existing PDU session", the AMF has a PDU session routing context for the PDU session ID and the UE, the PDU session routing context indicates that the PDU session is not an emergency PDU session, and the S-NSSAI associated with the PDU session identified by the PDU session ID is not allowed for the target access type, the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e), case f) or h4);
- 15) if the Request type IE is set to "initial request", "existing PDU session", "modification request" or "MA PDU request", the UE is not configured for high priority access in selected PLMN, and the UE is in non-allowed area or is not in allowed area, the AMF shall send back to the UE the 5GSM message which was not forwarded, and 5GMM cause #28 "Restricted service area" as specified in subclause 5.4.5.3.1 case i);
- 15a) if the Request type IE is set to "initial request" or "initial emergency request" and the AMF determines that the UE has registered to a PLMN via a satellite NG-RAN cell that is not allowed to operate at the present UE location, then the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case i1); and
- 16) if the Request type IE is set to "initial request" or "MA PDU request", the AMF is pending the receipt of a REGISTRATION REQUEST message indicating "mobility registration updating" in the 5GS registration type IE, and an emergency PDU session exists for the UE (see subclause 5.4.4.3), the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
- 17) if the timer T3447 is running and the UE supports service gap control and:
 - i) the Request type IE:
 - A) is set to "initial request";
 - B) is set to "existing PDU session"; or
 - C is set to "modification request" and the PDU session being modified is a non-emergency PDU session;
 - ii) the UE is not configured for high priority access in selected PLMN;
 - iii) the current NAS signalling connection was not triggered by paging; and
 - iv) mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection,

then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);

18) if the AMF has a PDU session routing context for the PDU session ID and the UE, the Request type IE is not included, the UE is not configured for high priority access in selected PLMN, and the PDU session is not an emergency PDU session, then the AMF shall forward the 5GSM message, and the PDU session ID IE towards the SMF identified by the SMF ID of the PDU session routing context with:

- i) an exemptionInd attribute indicating "message was exempted from the DNN based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], if DNN based congestion control is activated for the selected DNN;
- ii) an exemptionInd attribute indicating "message was exempted from the S-NSSAI and DNN based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], if S-NSSAI and DNN based congestion control is activated for the selected S-NSSAI and the selected DNN; or
- iii) an exemptionInd attribute indicating "message was exempted from the S-NSSAI only based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], if S-NSSAI only based congestion control is activated for the selected S-NSSAI;
- 19) if the Request type IE is set to "MA PDU request" and the S-NSSAI IE contains an S-NSSAI that is not allowed by the network on neither access, then the AMF shall send to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f);
- 20) if the Request type IE is set to "initial request" and the UE is registered for emergency services over the current access, then the AMF may send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e) or case f); and
- 21) if the Request type IE is set to "existing PDU session", the UE is attempting to transfer a PDU session from 3GPP access to non-3GPP access, and the PDU session is associated with control plane only indication then the AMF shall send back to the UE the 5GSM message which was not forwarded as specified in subclause 5.4.5.3.1 case e).
- 22) if the Request type IE is set to "MA PDU request" and the UE requested DNN corresponds to an LADN DNN, the AMF shall send back to the UE the 5GSM message which was not forwarded and 5GMM cause #90 "payload was not forwarded" as specified in subclause 5.4.5.3.1 case hx).
- 23) if the Request type IE is set to "initial request", the UE requested DNN corresponds to an LADN DNN, and the MA PDU session information IE is included, the AMF shall not forward the MA PDU session information towards the SMF.
- 24) if the Request type IE is set to "modification request", the DNN associated with the PDU session corresponds to an LADN DNN, and MA PDU session information IE is included, the AMF shall not forward the MA PDU session information towards the SMF.
- b) If the Payload container type IE is set to "SMS" and the AMF does not have an SMSF address associated with the UE or the AMF cannot forward the content of the Payload container IE to the SMSF associated with the SMSF address available in the AMF, the AMF shall abort the procedure.
- c) If the Payload container type IE is set to "LTE Positioning Protocol (LPP) message container" and if the Additional information IE is not included in the UL NAS TRANSPORT message or the AMF cannot forward the content of the Payload container IE to the LMF associated with the routing information included in the Additional information IE, the AMF shall abort the procedure.
- d) If the Payload container type IE is set to "UE policy container" and the AMF does not have a PCF address associated with the UE or the AMF cannot forward the content of the Payload container IE to the PCF associated with the PCF address available in the AMF, the AMF shall abort the procedure.
- e) If the Payload container type IE is set to "Location services message container" and if the Additional information IE is included in the UL NAS TRANSPORT message and the AMF cannot forward the content of the Payload container IE to an LMF associated with the routing information included in the Additional information IE, the AMF shall abort the procedure.
- f) If the Payload container type IE is set to "SMS" or "LTE Positioning Protocol (LPP) message container":
 - 1) the timer T3447 is running and the UE supports service gap control;
 - 2) the UE is not configured for high priority access in selected PLMN;
 - 3) the current NAS signalling connection was not triggered by paging; and
 - mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection,

the AMF shall abort the procedure.

- NOTE: In this state the N1 NAS signalling connection can be released by the network.
- g) If the Payload container type IE is set to "CIoT user data container" and:
 - 1) if the AMF does not have a PDU session routing context for the PDU session ID and the UE; or
 - 2) if the AMF unsuccessfully attempted to forward the user data container and the PDU session ID,

then the AMF may send back to the UE the CIoT user data container which was not forwarded as specified in subclause 5.4.5.3.1 case 11).

- h) If the Payload container type IE is set to "CIoT user data container":
 - 1) if the timer T3447 is running and the UE supports service gap control;
 - 2) the UE is not configured for high priority access in selected PLMN;
 - 3) the current N1 NAS signalling connection was not triggered by paging; and
 - 4) mobile terminated signalling has not been sent or no user-plane resources have been established for any PDU session after the establishment of the current NAS signalling connection,

then the AMF shall send back to the UE the CIoT user data container which was not forwarded as specified in subclause 5.4.5.3.1 case 11).

5.4.5.2.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) The lower layers indicate that the access attempt is barred.

The UE shall not start the UE-initiated NAS transport procedure. The UE stays in the current serving cell and applies the normal cell reselection process.

If the access category for the access attempt is 6 due to a request from upper layers to send a mobile originated SMS over NAS and the UE is registered to the network via both 3GPP access and non-3GPP access, the UE may transmit the UL NAS TRANSPORT message via non-3GPP access, if available.

Otherwise, the UE-initiated NAS transport procedure is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

aa) The lower layers indicate that:

- 1) access barring is applicable for all access categories except categories 0 and 2 and the access category with which the access attempt was associated is other than 0 and 2; or
- 2) access barring is applicable for all access categories except category 0 and the access category with which the access attempt was associated is other than 0.

The UE shall proceed as specified for case a. For additional UE requirements see subclause 4.5.5.

- b) If the Payload container type IE is set to "N1 SM information", the Request type IE is set to "initial request" or "MA PDU request" and registration procedure for mobility and periodic registration update is pending due to receipt by the UE of new network slicing information via the generic UE configuration update procedure with reregistration request; and an emergency PDU session exists then:
 - 1) The UE shall not send the UL NAS TRANSPORT message; and
 - The UL NAS TRANSPORT message can be sent, if still necessary, after a successful procedure for mobility and periodic registration update.
- c) Transmission failure of the UL NAS TRANSPORT message with change in the current TAI.

If the current TAI is not in the TAI list, the UE-initiated NAS transport procedure shall be aborted and a registration procedure for mobility and periodic registration update shall be initiated. The UL NAS

TRANSPORT message can be sent, if still necessary, after a successful procedure for mobility and periodic registration update.

If the current TAI is still part of the TAI list, it is up to the UE implementation how to re-run the ongoing procedure that triggered the UE-initiated NAS transport procedure.

d) Transmission failure of the UL NAS TRANSPORT message indication without change in the current TAI.

It is up to the UE implementation how to re-run the ongoing procedure that triggered the UE-initiated NAS transport procedure.

e) Void.

f) Timer T3447 is running.

The UE shall not send the UL NAS TRANSPORT message unless:

- 1) the Payload container type IE is set to "N1 SM information" and:
 - i) the Request type IE is set to:
 - A) "initial emergency request";
 - B) "existing emergency PDU session"; or
 - C) "modification request" and the PDU session being modified is an emergency PDU session (see error cases described in subclause 6.4.1.3 and subclause 6.3.2.3); or
 - ii) the Request type IE is not included and the PDU session modification procedure is used to indicate a change of 3GPP PS data off UE status for a PDU session;
- 2) the UE is a UE configured for high priority access in selected PLMN;
- 3) a paging request triggered the establishment of the current NAS signalling connection; or
- 4) the UE in 5GMM-CONNECTED mode receives mobile terminated signalling or downlink data over the user-plane.

The UL NAS TRANSPORT message can be sent, if still necessary, when timer T3447 expires or timer T3447 is stopped.

g) The lower layers indicate that the RRC connection has been suspended.

The UE shall abort the UE-initiated NAS transport procedure.

h) Timer T3346 is running.

The UE shall not send the UL NAS TRANSPORT message unless:

- 1) the Payload container type IE is set to "N1 SM information" and:
 - i) the Request type IE is set to:
 - A) "initial emergency request";
 - B) "existing emergency PDU session"; or
 - C) "modification request" and the PDU session being modified is an emergency PDU session; or
 - ii) the Request type IE is not included and the PDU session modification procedure is used to indicate a change of 3GPP PS data off UE status for a PDU session; or
- 2) the UE is a UE configured for high priority access in selected PLMN.

The UL NAS TRANSPORT message can be sent, if still necessary, when timer T3346 expires.

i) NAS MAC calculation indication from lower layers.

If lower layers indicate to calculate an NAS MAC, the UE shall calculate an NAS MAC as specified in 3GPP TS 33.501 [24] and then provide the calculated NAS MAC and 5 least significant bits of the uplink NAS COUNT used to calculate the NAS MAC to lower layers (see 3GPP TS 36.331 [25A]). The UE shall increase the uplink NAS COUNT by one after the calculation of the NAS MAC.

5.4.5.3 Network-initiated NAS transport procedure

5.4.5.3.1 General

The purpose of the network-initiated NAS transport procedure is to provide a transport of:

- a) a single 5GSM message;
- b) SMS;
- c) an LPP message;
- d) an SOR transparent container;
- e) a single uplink 5GSM message which was not forwarded due to routing failure;
- f) a single uplink 5GSM message which was not forwarded due to congestion control;
- g) a UE policy container;
- h) a single uplink 5GSM message which was not forwarded, because the PLMN's maximum number of PDU sessions has been reached;
- h1) a single uplink 5GSM message which was not forwarded, because the maximum number of PDU sessions with active user-plane resources has been reached;
- h2) a single uplink 5GSM message which was not forwarded, because of ongoing network slice-specific authentication and authorization procedure for the S-NSSAI that is requested;
- h3) a single uplink 5GSM message which was not forwarded, because the UE requested to establish an MA PDU session for LADN DNN;
- h4) a single uplink 5GSM message which was not forwarded, because the maximum number of UEs for a network slice has been reached;
- h5) a single uplink 5GSM message which was not forwarded because the UE is marked in the UE's 5GMM context that it is not allowed to request UAS services;
- i) a single uplink 5GSM message which was not forwarded due to service area restrictions;
- i1) a single uplink 5GSM message which was not forwarded because the UE is registered to a PLMN via a satellite NG-RAN cell that is not allowed to operate at the present UE location;
- j) a UE parameters update transparent container;
- k) a location services message;
- l) a CIoT user data container;
- 11) a single uplink CIoT user data container or control plane user data which was not forwarded due to routing failure;
- 12) a single uplink CIoT user data container which was not forwarded due to congestion control;
- m) a service-level-AA container;
- m1) an event notification for upper layers; or
- n) multiple of the above types.

from the AMF to the UE in a 5GMM message.

5.4.5.3.2 Network-initiated NAS transport procedure initiation

In 5GMM-CONNECTED mode, the AMF initiates the NAS transport procedure by sending the DL NAS TRANSPORT message, as shown in figure 5.4.5.3.2.1.

In case a) in subclause 5.4.5.3.1, i.e. upon reception from an SMF of a 5GSM message without an N1 SM delivery skip allowed indication for a UE or a 5GSM message with an N1 SM delivery skip allowed indication for a UE in the 5GMM-CONNECTED mode, the AMF shall:

- a) include the PDU session information (PDU session ID) in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information"; and
- c) set the Payload container IE to the 5GSM message.

In case b) in subclause 5.4.5.3.1, i.e. upon reception from an SMSF of an SMS payload, the AMF shall:

- a) set the Payload container type IE to "SMS";
- b) set the Payload container IE to the SMS payload; and
- c) select the access type to deliver the DL NAS TRANSPORT message as follows in case the access type selection is required:
 - if the UE to receive the DL NAS TRANSPORT message is registered to the network via both 3GPP access and non-3GPP access, the 5GMM context of the UE indicates that SMS over NAS is allowed, the UE is in MICO mode, and the UE is in 5GMM-IDLE mode for 3GPP access and in 5GMM-CONNECTED mode for non-3GPP access, then the AMF selects non-3GPP access. Otherwise, the AMF selects either 3GPP access or non-3GPP access.

If the delivery of the DL NAS TRANSPORT message over 3GPP access has failed, the AMF may re-send the DL NAS TRANSPORT message over the non-3GPP access.

If the delivery of the DL NAS TRANSPORT message over non-3GPP access has failed, the AMF may resend the DL NAS TRANSPORT message over the 3GPP access; and

2) otherwise, the AMF selects 3GPP access.

NOTE 1: The AMF selects an access type between 3GPP access and non-3GPP access based on operator policy.

In case c) in subclause 5.4.5.3.1 i.e. upon reception from an LMF of an LPP message payload, the AMF shall:

- a) set the Payload container type IE to "LTE Positioning Protocol (LPP) message container";
- b) set the Payload container IE to the LPP message payload received from the LMF;
- c) set the Additional information IE to an LCS correlation identifier received from the LMF from which the LPP message was received.
- NOTE 2: The LCS Correlation Identifier is assigned originally by the AMF except for LPP message transfer associated with event reporting for periodic or triggered location as described in subclause 6.3.1 of 3GPP TS 23.273 [6B], where the LMF assigns the correlation identifier. AMF and LMF assigned correlation identifiers can be distinguished by an implementation specific convention (e.g. use of a different number of octets) to enable an AMF to distinguish one from the other when received in the Additional Information IE in an UL NAS Transport message.

In case d) in subclause 5.4.5.3.1 i.e. upon reception of a steering of roaming information (see 3GPP TS 23.122 [5]) from the UDM to be forwarded to the UE, the AMF shall:

- a) set the Payload container type IE to "SOR transparent container"; and
- b) set the Payload container IE to the steering of roaming information received from the UDM (see 3GPP TS 29.503 [20AB]).

In case e) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded due to routing failure, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded;
- d) set the 5GMM cause IE to the 5GMM cause #90 "payload was not forwarded" or 5GMM cause #91 "DNN not supported or not subscribed in the slice".

The AMF sets the 5GMM cause IE to the 5GMM cause #91 "DNN not supported or not subscribed in the slice", if the 5GSM message could not be forwarded since SMF selection fails because:

- 1) the DNN is not supported in the slice identified by the S-NSSAI used by the AMF; or
- 2) neither the DNN provided by the UE nor the wildcard DNN are in the subscribed DNN list of the UE for the S-NSSAI used by the AMF.

Otherwise, the AMF sets the 5GMM cause IE to the 5GMM cause #90 "payload was not forwarded"; and

e) optionally include the Back-off timer value IE if the 5GMM cause IE is set to 5GMM cause #91 "DNN not supported or not subscribed in the slice" due to the DNN is not supported in the slice.

In case f) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded due to congestion control, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded;
- d) set the 5GMM cause IE to the 5GMM cause #22 "Congestion", the 5GMM cause #67 "insufficient resources for specific slice and DNN" or the 5GMM cause #69 "insufficient resources for specific slice"; and
- e) include the Back-off timer value IE.

In case g) in subclause 5.4.5.3.1, i.e. upon reception of a UE policy container from the PCF to be forwarded to the UE, the AMF shall:

- a) set the Payload container type IE to "UE policy container"; and
- b) set the Payload container IE to the UE policy container received from the PCF.

In case h) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded, because the PLMN's maximum number of PDU sessions has been reached, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #65 "maximum number of PDU sessions reached".

In case h1) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded, because the maximum number of PDU sessions with active user-plane resources has been reached, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #92 "insufficient user-plane resources for the PDU session".

In case h2) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded because the UE requested to establish a PDU session associated with an S-NSSAI or to modify a PDU session associated with an S-NSSAI for which:

- a) the AMF is performing network slice-specific authentication and authorization and determined to reject the request based on local policy; or
- b) the network slice-specific authentication and authorization has failed or the authorization has been revoked;

the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #90 "payload was not forwarded".

In case h3) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded because the UE requested to establish an MA PDU session for LADN DNN, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #90 "payload was not forwarded".

In case h4) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded, because the maximum number of UEs for a network slice has been reached, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded;
- d) set the 5GMM cause IE to the 5GMM cause #69 "insufficient resources for specific slice"; and
- e) include the Back-off timer value IE.

For case h5) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded because the UE is marked in the UE's 5GMM context that it is not allowed to request UAS services, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #79 "UAS services not allowed".

In case i) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded due to service area restrictions, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";
- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #28 "Restricted service area".

In case i1) in subclause 5.4.5.3.1, i.e. upon sending a single uplink 5GSM message which was not forwarded because the UE is registered to a PLMN via a satellite NG-RAN cell that is not allowed to operate at the present UE location, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "N1 SM information";

- c) set the Payload container IE to the 5GSM message which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #78 "PLMN not allowed to operate at the present UE location".

In case j) in subclause 5.4.5.3.1 i.e. upon reception of UE parameters update data (see 3GPP TS 23.502 [9]) from the UDM to be forwarded to the UE, the AMF shall:

- a) set the Payload container type IE to "UE parameters update transparent container"; and
- b) set the contents of the Payload container IE to the UE parameters update data (see 3GPP TS 23.502 [9]) received from the UDM.

For case k) in subclause 5.4.5.3.1 upon reception from a location services application of a Location services message payload, the AMF shall:

- a) set the Payload container type IE to "Location services message container"; and
- b) set the Payload container IE to the Location services message payload.

For case k) in subclause 5.4.5.3.1 upon reception from an LMF of a Location services message payload, the AMF shall:

- a) set the Payload container type IE to "Location services message container";
- b) set the Payload container IE to the Location services message payload; and
- c) set the Additional information IE to routing information associated with the LMF from which the Location services message payload was received.
- NOTE 3: Case k) in subclause 5.4.5.3.1 supports transport of a Location services message container between a UE and an AMF and between a UE and an LMF. For transport between a UE and an LMF, the Additional information IE is included and provides routing information for the LMF. For transport between a UE and an AMF, the Additional information IE is not included.

In case l) in subclause 5.4.5.3.1, i.e. upon reception from an SMF of a user data container payload, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "CIoT user data container"; and
- c) set the Payload container IE to the user data container.

For case 11) in subclause 5.4.5.3.1, i.e. upon sending a single uplink CIoT user data container or control plane user data which was not forwarded due to routing failure, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to "CIoT user data container";
- c) set the Payload container IE to the CIoT user data container or control plane user data which was not forwarded; and
- d) set the 5GMM cause IE to the 5GMM cause #90 "payload was not forwarded".
- NOTE 4: For case 11) in subclause 5.4.5.3.1, this is also applied for a single uplink CIoT user data container or control plane user data in the CONTROL PLANE SERVICE REQUEST message which was not forwarded due to routing failure.

For case 12) in subclause 5.4.5.3.1, i.e. upon sending a single uplink CIoT user data container which was not forwarded due to congestion control, the AMF shall:

- a) include the PDU session ID in the PDU session ID IE;
- b) set the Payload container type IE to " CIoT user data container";
- c) set the Payload container IE to the CIoT user data container which was not forwarded;

 d) set the 5GMM cause IE to the 5GMM cause #22 "Congestion", the 5GMM cause #67 "insufficient resources for specific slice and DNN" or the 5GMM cause #69 "insufficient resources for specific slice", and include the Back-off timer value IE.

In case m) in subclause 5.4.5.3.1, during UUAA-MM procedure, if the AMF receives a payload from the UAS-NF, the AMF shall:

- a) include the service-level-AA payload with the value set to the payload; and
- b) if a payload type associated with the payload is received, include the service-level-AA payload type with the value set to the payload type

in the Service-level-AA container IE.

In case m1) in subclause 5.4.5.3.1, i.e. if the AMF needs to send an event notification indicator for upper layers to the UE which set the "EventNotification" bit of the 5GMM capability IE in the last REGISTRATION REQUEST message to "Event notification supported", the AMF shall:

- a) set the Payload container type IE to "Event notification"; and
- b) set the Payload container IE to the event notification indicator.

In case n) in subclause 5.4.5.3.1, the AMF shall:

- a) set the Payload container type IE to "Multiple payloads";
- b) set each payload container entry of the Payload container IE (see subclause 9.11.3.39) as follows:
 - i) set the payload container type field of the payload container entry to a payload container type value set in the Payload container type IE as specified for cases a) to m) above;
 - ii) set the payload container entry contents field of the payload container entry to the payload container contents set in the Payload container IE as specified for cases a) to m) above;
 - iii) set the optional IE fields, if any, to the optional associated information as specified for cases a) to m) above.



Figure 5.4.5.3.2.1: Network-initiated NAS transport procedure

5.4.5.3.3 Network-initiated NAS transport of messages

Upon reception of a DL NAS TRANSPORT message, the UE shall stop the timer T3346 if running.

Upon reception of a DL NAS TRANSPORT message, if the Payload container type IE is set to:

- a) "N1 SM information" and the 5GMM cause IE is not included in the DL NAS TRANSPORT message, the 5GSM message in the Payload container IE and the PDU session ID are handled in the 5GSM procedures specified in clause 6;
- b) "SMS", the UE shall forward the content of the Payload container IE to the SMS stack entity;
- c) "LTE Positioning Protocol (LPP) message container", the UE shall forward the payload container type, the content of the Payload container IE and the routing information included in the Additional information IE to the upper layer location services application;
- d) "SOR transparent container" and if the Payload container IE:

- 1) successfully passes the integrity check (see 3GPP TS 33.501 [24]), the ME shall store the received SOR counter as specified in annex C and proceed as follows:
 - i) If the Payload container IE indicates a list of preferred PLMN/access technology combinations is provided and the list type indicates "PLMN ID and access technology list", then the ME shall replace the highest priority entries in the "Operator Controlled PLMN Selector with Access Technology" list stored in the ME;
 - ii) If the list type indicates "secured packet", then the ME shall behave as if a SMS is received with protocol identifier set to SIM data download, data coding scheme set to class 2 message and SMS payload as secured packet contents of SOR transparent container IE. The SMS payload is forwarded to UICC as specified in 3GPP TS 23.040 [4A];
 - iii) If the Payload container IE includes SOR-SNPN-SI, the ME shall replace SOR-SNPN-SI of the selected entry of the "list of subscriber data" or associated with the selected PLMN subscription, as specified in 3GPP TS 23.122 [5] with the received SOR-SNPN-SI; and
 - iv) If the SOR-CMCI is present, in plain text, and the Store SOR-CMCI in ME indicator is set to "Store SOR-CMCI in ME" then the UE shall store or delete the SOR-CMCI in the non-volatile memory of the ME as described in annex C.1;

If the ACK bit of the SOR header for SOR data type in the SOR transparent container is set to "acknowledgement requested" and the list type indicates:

- A) "PLMN ID and access technology list"; or
- B) "secured packet" and the ME receives status bytes from the UICC indicating that the UICC has received the secured packet successfully;

then the ME shall send an acknowledgement in the Payload container IE of an UL NAS TRANSPORT message with Payload type IE set to "SOR transparent container" as specified in subclause 5.4.5.2.2. In the Payload container IE carrying the acknowledgement, the UE shall set the ME support of SOR-CMCI indicator to "SOR-CMCI supported by the ME". Additionally, if the UE supports access to an SNPN using credentials from a credentials holder and the UE is not operating in SNPN access operation mode, the UE may set the ME support of SOR-SNPN-SI indicator to "SOR-SNPN-SI supported by the ME".

The UE shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C; or

- does not successfully pass the integrity check (see 3GPP TS 33.501 [24]) then the UE shall discard the content of the payload container IE and proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C.
- e) Void;
- f) Void;
- g) "N1 SM information" and:
 - the 5GMM cause IE is set to the 5GMM cause #22 "Congestion", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to DNN based congestion control along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
 - 2) the 5GMM cause IE is set to the 5GMM cause #28 "Restricted service area", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to service area restrictions along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message, enters the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE and, if the DL NAS TRANSPORT message is received over 3GPP access, performs the registration procedure for mobility and periodic registration update without waiting for the release of the N1 NAS signalling connection (see subclauses 5.3.5 and 5.5.1.3);
 - 3) the 5GMM cause IE is set to the 5GMM cause #65 "maximum number of PDU sessions reached", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded because the PLMN's maximum number of PDU sessions has been reached, along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message;

- 4) the 5GMM cause IE is set to the 5GMM cause #67 "insufficient resources for specific slice and DNN", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to S-NSSAI and DNN based congestion control along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
- 5) the 5GMM cause IE is set to the 5GMM cause #69 "insufficient resources for specific slice", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to S-NSSAI only based congestion control along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
- 5a) the 5GMM cause IE is set to the 5GMM cause #78 "PLMN not allowed to operate at the present UE location", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded because the UE is registered to a PLMN via a satellite NG-RAN cell that is not allowed to operate at the present UE location along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message;
- 6) the 5GMM cause IE is set to the 5GMM cause #90 "payload was not forwarded", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to routing failure along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message;
- 7) the 5GMM cause IE is set to the 5GMM cause #91 "DNN not supported or not subscribed in the slice", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded because the DNN is not supported or not subscribed in a slice along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE, if any;
- 8) the 5GMM cause IE is set to the 5GMM cause #92 "insufficient user-plane resources for the PDU session", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded due to insufficient user-plane resources along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message.
- 9) the 5GMM cause IE is set to the 5GMM cause #79 "UAS services not allowed", the UE passes to the 5GSM sublayer an indication that the 5GSM message was not forwarded because the UE is marked in the UE's 5GMM context that it is not allowed to request UAS services along with the 5GSM message from the Payload container IE of the DL NAS TRANSPORT message.
- h) "UE policy container", the UE policy container in the Payload container IE is handled in the UE policy delivery procedures specified in Annex D;
- i) "UE parameters update transparent container" and if the Payload container IE
 - 1) successfully passes the integrity check (see 3GPP TS 33.501 [24]), the ME shall store the received UE parameter update counter as specified in annex C and proceed as follows:
 - i) if the UE parameters update list includes a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data",
 - A) the ME shall behave as if an SMS is received with protocol identifier set to SIM data download, data coding scheme set to class 2 message and SMS payload as secured packet contents of UE parameters update transparent container IE. The SMS payload is forwarded to UICC as specified in 3GPP TS 23.040 [4A]; and
 - B) if the ACK bit of the UE parameters update header in the UE parameters update transparent container is set to "acknowledgment requested" and if the ME receives status bytes from the UICC indicating that the UICC has received the secured packet successfully, the ME shall send an acknowledgement in the Payload container IE of an UL NAS TRANSPORT message with Payload type IE set to "UE parameters update transparent container" as specified in subclause 5.4.5.2.2; and
 - C) if the ME receives a REFRESH command from the UICC as specified in 3GPP TS 31.111 [22A] and if the REG bit of the UE parameters update header in the UE parameters update transparent container IE is set to "re-registration requested", and:
 - C1) the UE is registered over 3GPP access, then the UE shall wait until the emergency services over 3GPP access, if any, are completed, enter 5GMM-IDLE mode over 3GPP access or 5GMM-CONNECTED mode with RRC inactive indication, perform a de-registration procedure, and then delete its 5G-GUTI if the UE is registered to different PLMN or SNPN on non-3GPP access or the

UE is not registered over non-3GPP access, or wait until the de-registration procedure over non-3GPP access specified in case C2) or C3) is completed before deleting its 5G-GUTI if the UE is registered to same PLMN or SNPN on non-3GPP access, and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2;

- C2) the UE is registered over non-3GPP access and does not have emergency services ongoing over non-3GPP access, then the UE shall locally release the N1 NAS signalling connection and enter 5GMM-IDLE mode over non-3GPP access, perform a de-registration procedure, and then delete its 5G-GUTI if the UE is registered to different PLMN or SNPN on 3GPP access or the UE is not registered over 3GPP access, or wait until the de-registration procedure over 3GPP access specified in case C1) is completed before deleting its 5G-GUTI if the UE is registered to same PLMN or SNPN on 3GPP access, and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2; and
- C3) the UE is registered over non-3GPP access and has an emergency services ongoing over non-3GPP access, then the UE shall wait until the emergency services are completed before locally releasing the N1 NAS signalling connection and enter 5GMM-IDLE mode over non-3GPP access, perform a de-registration procedure, and then delete its 5G-GUTI if the UE is registered to different PLMN or SNPN on 3GPP access or if the UE is not registered over 3GPP access, or wait until the de-registration procedure over 3GPP access specified in case C1) is completed before deleting its 5G-GUTI if the UE is registered to same PLMN or SNPN on 3GPP access, and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2.
- ii) if the UE parameters update list includes a UE parameters update data set with UE parameters update data set type indicating "Default configured NSSAI update data",
 - A) if the ACK bit of the UE parameters update header in the UE parameters update transparent container is set to "acknowledgment requested" and if the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data", the ME shall send an acknowledgement in the Payload container IE of an UL NAS TRANSPORT message with Payload type IE set to "UE parameters update transparent container" as specified in subclause 5.4.5.2.2
 - B) the ME shall replace the stored default configured NSSAI with the default configured NSSAI included in the default configured NSSAI update data. In case of SNPN, the ME shall replace the stored default configured NSSAI associated with the selected entry of the "list of subscriber data" or the PLMN subscription with the default configured NSSAI included in the default configured NSSAI update data; and
 - C) if the REG bit of the UE parameters update header in the UE parameters update transparent container is set to "re-registration requested" and the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data", the UE shall wait until it enters 5GMM-IDLE mode and then the UE shall initiate a registration procedure for mobility registration update as specified in subclause 5.5.1.3.

if the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data", the UE used the old default configured NSSAI to create the requested NSSAI in a REGISTRATION REQUEST message, the UE does not have a configured NSSAI for the current PLMN or SNPN and the UE has an allowed NSSAI for the current PLMN or SNPN which contains one or more S-NSSAIs that are not included in the new default configured NSSAI, the UE shall wait until it enters 5GMM-IDLE mode and then the UE shall initiate a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3; and

- iii) if the UE parameters update list includes a UE parameters update data set with UE parameters update data set type indicating "Disaster roaming information update data",
 - A) if the ACK bit of the UE parameters update header in the UE parameters update transparent container is set to "acknowledgment requested" and if the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data" or a UE parameters update data set with UE parameters update data set type indicating "Default configured NSSAI update data", the ME shall send an acknowledgement in the Payload container IE of an UL NAS TRANSPORT message with Payload type IE set to "UE parameters update transparent container" as specified in subclause 5.4.5.2.2;

- B) the UE shall delete the indication of whether disaster roaming is enabled in the UE stored in the ME, if any, and store the indication of whether disaster roaming is enabled in the UE included in the disaster roaming information update data in the ME;
- C) the UE shall delete the indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN' stored in the ME, if any, and store the indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN' included in the disaster roaming information update data in the ME; and
- D) if the REG bit of the UE parameters update header in the UE parameters update transparent container is set to "re-registration requested" and the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Routing indicator update data", the UE shall wait until it enters 5GMM-IDLE mode and then the UE shall initiate a registration procedure for mobility registration update as specified in subclause 5.5.1.3.
- iv) if the UE parameters update list includes a UE parameters update data set with UE parameters update data set type indicating "ME routing indicator update data":
 - A) if the ACK bit of the UE parameters update header in the UE parameters update transparent container is set to "acknowledgment requested" and the UE parameters update list does not include a UE parameters update data set with UE parameters update data set type indicating "Default configured NSSAI update data", the ME shall send an acknowledgement in the Payload container IE of an UL NAS TRANSPORT message with Payload type IE set to "UE parameters update transparent container" as specified in subclause 5.4.5.2.2;
 - B) the UE shall set or replace the routing indicator of the selected entry of the "list of subscriber data" with the routing indicator included in the ME routing indicator update data; and
 - C) if the REG bit of the UE parameters update header in the UE parameters update transparent container IE is set to "re-registration requested", and:
 - C1) the UE is registered over 3GPP access and is not registered over non-3GPP access, then the UE shall wait until the emergency services over 3GPP access, if any, are completed, enter 5GMM-IDLE mode over 3GPP access or 5GMM-CONNECTED mode with RRC inactive indication, perform a de-registration procedure, delete its 5G-GUTI, and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2;
 - C2) the UE is registered over non-3GPP access and is not registered over 3GPP access, then the UE shall locally release the N1 NAS signalling connection and enter 5GMM-IDLE mode over non-3GPP access, perform a de-registration procedure, delete its 5G-GUTI, and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2; or
 - C3) the UE is registered over 3GPP access and non-3GPP access to same SNPN, then the UE shall wait until the emergency services over 3GPP access, if any, are completed, enter 5GMM-IDLE mode over 3GPP access or 5GMM-CONNECTED mode with RRC inactive indication over 3GPP access, perform a de-registration procedure over 3GPP access, locally release the N1 NAS signalling connection and enter 5GMM-IDLE mode over non-3GPP access, perform a de-registration procedure over non-3GPP access, delete its 5G-GUTI and then initiate a registration procedure for initial registration as specified in subclause 5.5.1.2.
- NOTE: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
 - 2) does not successfully pass the integrity check (see 3GPP TS 33.501 [24]) then the UE shall discard the content of the payload container IE;
- j) "Location services message container" and the 5GMM cause IE is not included in the DL NAS TRANSPORT message, the UE shall forward the payload container type, the content of the Payload container IE and the routing information in the Additional information IE if included to the upper layer location services application;
- k) "CIoT user data container", the UE shall forward the content of the Payload container IE and the PDU session ID to the 5GSM sublayer;
- 1) "CIoT user data container" and:

- the 5GMM cause IE is set to the 5GMM cause #22 "Congestion", the UE passes to the 5GSM sublayer an indication that the CIoT user data was not forwarded due to DNN based congestion control along with the CIoT user data from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
- 2) the 5GMM cause IE is set to the 5GMM cause #67 "insufficient resources for specific slice and DNN", the UE passes to the 5GSM sublayer an indication that the CIoT user data was not forwarded due to S-NSSAI and DNN based congestion control along with the CIoT user data from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
- 3) the 5GMM cause IE is set to the 5GMM cause #69 "insufficient resources for specific slice", the UE passes to the 5GSM sublayer an indication that the CIoT user data was not forwarded due to S-NSSAI only based congestion control along with the CIoT user data from the Payload container IE of the DL NAS TRANSPORT message, and the time value from the Back-off timer value IE;
- 4) the 5GMM cause IE is set to the 5GMM cause #90 "payload was not forwarded", the UE passes to the 5GSM sublayer an indication that the user data container was not forwarded due to routing failure along with the user data container from the Payload container IE and the PDU session ID from the PDU session ID IE of the DL NAS TRANSPORT message.
- m) "service-level-AA container", the UE shall forward the content of the Payload container IE to the upper layers;
- m1) "Event notification", the UE shall forward the received event notification indicator(s) to the upper layers (see 3GPP TS 23.216 [6A] and 3GPP TS 24.237 [14AA] for the "SRVCC handover cancelled, IMS session re-establishment required" indicator); or
- n) "Multiple payloads", the UE shall first decode the content of the Payload container IE (see subclause 9.11.3.39) to obtain the number of payload container entries and for each payload container entry, the UE shall:
 - 1) decode the payload container type field;
 - 2) decode the optional IE fields and the payload container contents field in the payload container entry; and
 - 3) handle the content of each payload container entry the same as the content of the Payload container IE and the associated optional IEs as specified in bullets a) to m) above according to the payload container type field.

5.4.6 5GMM status procedure

5.4.6.1 General

The purpose of the 5GMM status procedure is to report at any time in the 5GMM STATUS message certain error conditions detected upon receipt of 5GMM protocol data in the AMF or in the UE. The 5GMM STATUS message can be sent by both the AMF and the UE (see example in figure 5.4.6.1).

UE		AMF
	5GMM STATUS	
•		_
	OR	
-	5GMM STATUS	•
		-



5.4.6.2 5GMM status received in the UE

On receipt of a 5GMM STATUS message, no state transition and no specific action shall be taken as seen from the radio interface, i.e. local actions are possible. The local actions to be taken by UE on receipt of a 5GMM STATUS message are implementation dependent.

5.4.6.3 5GMM status received in the network

On receipt of a 5GMM STATUS message in the AMF, no state transition and no specific action shall be taken as seen from the radio interface, i.e. local actions are possible. The local actions to be taken by the AMF on receipt of a 5GMM STATUS message are implementation dependent.

5.4.7 Network slice-specific authentication and authorization procedure

5.4.7.1 General

The purpose of the network slice-specific authentication and authorization procedure is to enable the authentication, authorization and accounting server (AAA-S) via the Network Slice Specific and SNPN Authentication and Authorization Function (NSSAAF) to (re-)authenticate or (re-)authorize the upper layers of the UE.

The network slice-specific authentication and authorization procedure can be invoked for a UE supporting network slice-specific authentication and authorization procedure and for a HPLMN S-NSSAI or an SNPN S-NSSAI (see subclauses 5.15.10 and 5.30.2.9 in 3GPP TS 23.501 [8] and subclause 4.2.9.2 of 3GPP TS 23.502 [9]).

The network (re-)authenticates the UE using the EAP as specified in IETF RFC 3748 [34].

EAP has defined four types of EAP messages:

- a) an EAP-request message;
- b) an EAP-response message;
- c) an EAP-success message; and
- d) an EAP-failure message.

The EAP-request message is transported from the network to the UE using the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message of the network slice-specific EAP message reliable transport procedure.

The EAP-response message to the EAP-request message is transported from the UE to the network using the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message of the network slice-specific EAP message reliable transport procedure.

If the (re-)authentication of the UE completes successfully or unsuccessfully, the EAP-success message or the EAP-failure message, respectively, is transported from the network to the UE using the NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message of the network slice-specific result message transport procedure.

There can be several rounds of exchange of an EAP-request message and a related EAP-response message for the AAA-S via the NSSAAF to complete the (re-)authentication and (re-)authorization of the request for an S-NSSAI (see example in figure 5.4.7.1.1).

The AMF shall set the authenticator retransmission timer specified in subclause 4.3 of IETF RFC 3748 [34] to infinite value.

NOTE: The network slice-specific authentication and authorization procedure provides a reliable transport of EAP messages and therefore retransmissions at the EAP layer of the AMF do not occur.

UE AMF Start NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND T3575 <EAP-request message A> NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE Stop <EAP-response message to EAP-request message A> T3575 Start NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND T3575 <EAP-request message B> NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE Stop <EAP-response message to EAP-request message B> T3575 NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT <EAP-success message> OR NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT

<EAP-failure message>

Figure 5.4.7.1.1: Network slice-specific authentication and authorization procedure

5.4.7.2 Network slice-specific EAP message reliable transport procedure

5.4.7.2.1 Network slice-specific EAP message reliable transport procedure initiation

In order to initiate the network slice-specific EAP message reliable transport procedure, the AMF shall create a NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message.

The AMF shall set the EAP message IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message to the EAP-request message which is generated by the AMF or provided by the AAA-S via the NSSAAF.

The AMF shall set the S-NSSAI IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message to the HPLMN S-NSSAI or the SNPN S-NSSAI to which the EAP-request message is related.

The AMF shall send the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message and start timer T3575 per S-NSSAI (see example in figure 5.4.7.1.1).

Upon receipt of a NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message, the UE shall stop timer T3346 if running. The UE shall pass:

- a) the EAP-request message received in the EAP message IE; and
- b) the HPLMN S-NSSAI or the SNPN S-NSSAI in the S-NSSAI IE;

to the upper layers. Apart from this action, the network slice-specific authentication and authorization procedure is transparent to the 5GMM layer of the UE.

5.4.7.2.2 Network slice-specific EAP message reliable transport procedure accepted by the UE

When the upper layers provide an EAP-response message associated with the HPLMN S-NSSAI or the SNPN S-NSSAI, the UE shall create a NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message.

The UE shall set the EAP message IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message to the EAP-response message.

The UE shall set the S-NSSAI IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message to the HPLMN S-NSSAI or the SNPN S-NSSAI associated with the EAP-response message.

The UE shall send the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message. Apart from this action, the network slice-specific authentication and authorization procedure is transparent to the 5GMM layer of the UE.

Upon receipt of a NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message, the AMF shall stop timer T3575 and:

- a) pass the EAP-response message received in the EAP message IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message associated with the HPLMN S-NSSAI or the SNPN S-NSSAI in the S-NSSAI IE to the upper layers; or
- b) provide the EAP-response message received in the EAP message IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message associated with the HPLMN S-NSSAI or the SNPN S-NSSAI in the S-NSSAI IE to the AAA-S via the NSSAAF.

5.4.7.2.3 Abnormal cases on the network side

The following abnormal cases can be identified:

a) T3575 expiry

The AMF shall, on the first expiry of the timer T3575, retransmit the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message and shall reset and start timer T3575. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3575, the AMF shall abort the network slice-specific authentication and authorization procedure for the S-NSSAI. The AMF shall consider that the network slice-specific authentication and authorization procedure for the S-NSSAI is completed as a failure.

b) Lower layers indication of non-delivered NAS PDU due to handover

If the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message could not be delivered due to an intra AMF handover and the target TAI is included in the TAI list, then upon successful completion of the intra AMF handover the AMF shall retransmit the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message. If a failure of handover procedure is reported by the lower layer and the N1 NAS signalling connection exists, the AMF shall retransmit the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message.

c) Network slice-specific authentication and authorization procedure and de-registration procedure collision

If the network receives a DEREGISTRATION REQUEST message before the ongoing network slice-specific authentication and authorization procedure has been completed and the access type included in the DEREGISTRATION REQUEST message is the same as the one for which the network slice-specific authentication and authorization procedure is ongoing, the network shall abort the network slice-specific authentication and authorization procedure and shall progress the UE-initiated de-registration procedure. The AMF may initiate the network slice-specific authentication and authorization procedure and shall progress the UE-initiated de-registration procedure. The scale as a failure, if available. If the access type included in the DEREGISTRATION REQUEST message is different from the one for which the network slice-specific authentication and authorization procedure is ongoing, the network slice-specific authentication and authorization procedure is ongoing, the network slice-specific authentication and authorization procedure for the s-NSSAI which is completed as a failure, if available. If the access type included in the DEREGISTRATION REQUEST message is different from the one for which the network slice-specific authentication and authorization procedure is ongoing, the network shall proceed with both procedures.

5.4.7.2.4 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Transmission failure of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message with change in the current TAI

If the current TAI is not in the TAI list, the network slice-specific authentication and authorization procedure shall be aborted and:

- if the UE is in 5GMM-REGISTERED state, a registration procedure for mobility and periodic registration update indicating "mobility registration updating" in the 5GS registration type IE of the REGISTRATION REQUEST message shall be initiated; and
- otherwise a registration procedure for initial registration shall be initiated.
- b) Transmission failure of NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message indication without change in the current TAI

It is up to the UE implementation how to re-run the ongoing procedure that triggered the network slice-specific authentication and authorization procedure.

c) Network slice-specific authentication and authorization procedure and de-registration procedure collision

If the UE receives NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message after sending a DEREGISTRATION REQUEST message and the access type included in the DEREGISTRATION REQUEST message is the same as the access in which the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message is received, then the UE shall ignore the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message and proceed with the de-registration procedure. Otherwise, the UE shall proceed with both procedures.

5.4.7.3 Network slice-specific EAP result message transport procedure

5.4.7.3.1 Network slice-specific EAP result message transport procedure initiation

In order to initiate the network slice-specific EAP result message transport procedure, the AMF shall create a NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message.

The AMF shall set the EAP message IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message to the EAP-success or EAP-failure message provided by the AAA-S via the NSSAAF.

The AMF shall set the S-NSSAI IE of the NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message to the HPLMN S-NSSAI or the SNPN S-NSSAI to which the EAP-success or EAP-failure message is related.

The AMF shall send the NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message. The AMF shall retain the authentication result for the UE and the HPLMN S-NSSAI or the SNPN S-NSSAI while the UE is registered to the PLMN (see subclause 5.15.10 in 3GPP TS 23.501 [8]).

Upon receipt of a NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message, the UE shall pass:

- a) the EAP-success or EAP-failure message received in the EAP message IE; and
- b) the HPLMN S-NSSAI or the SNPN S-NSSAI in the S-NSSAI IE;

to the upper layers. Apart from this action, the network slice-specific authentication and authorization procedure is transparent to the 5GMM layer of the UE.

5.5 5GMM specific procedures

5.5.1 Registration procedure

5.5.1.1 General

The registration procedure is always initiated by the UE and used for initial registration as specified in subclause 5.5.1.2.2 or mobility and periodic registration update as specified in subclause 5.5.1.3.2.

When the UE needs to initiate registration over both 3GPP access and non-3GPP access in the same PLMN (e.g. the 3GPP access and the selected N3IWF are located in the same PLMN), the UE:

- a) in 5GMM-REGISTERED-INITIATED over 3GPP access shall not initiate registration over non-3GPP access; or
- b) in 5GMM-REGISTERED-INITIATED over non-3GPP access shall not initiate registration over 3GPP access.
- NOTE 1: To which access (i.e. 3GPP access or non-3GPP access) the UE initiates registration first is up to UE implementation.

When the UE is registered with a PLMN over a non-3GPP access, the AMF and the UE maintain:

- a) registration state and state machine over non-3GPP access;
- b) 5G NAS security context;
- c) 5G-GUTI;
- d) registration area for non-3GPP access, which is associated with a single TAI; and
- e) non-3GPP de-registration timer in the UE and non-3GPP implicit de-registration timer in the AMF.

A registration attempt counter is used to limit the number of subsequently rejected registration attempts. The registration attempt counter shall be incremented as specified in subclause 5.5.1.2.7 or subclause 5.5.1.3.7. Depending on the value of the registration attempt counter, specific actions shall be performed. The registration attempt counter shall be reset when:

- the UE is powered on;
- a USIM is inserted;
- a registration procedure is successfully completed;
- an EPS attach or combined EPS attach procedure is successfully completed in S1 mode and the UE is operating in single-registration mode. In this case, the UE shall reset the registration attempt counter for 3GPP access;

NOTE 2: The registration attempt counter for non-3GPP access is not impacted by the EPS attach and the combined EPS attach procedure.

- a registration procedure is rejected with cause #11, #12, #13, #15, #27, #31, #62, #72, #73, #74, #75, #76, #77 or #78;
- a registration procedure is rejected with cause #3, #6 or #7, the REGISTRATION REJECT message is received without integrity protection and the counter for "SIM/USIM considered invalid for GPRS services" events has a value less than a UE implementation-specific maximum value.
- a network initiated deregistration procedure is completed with cause #11, #12, #13, #15, #27; #62, #72, #74, #75, #76, #77 or #78; or
- a new PLMN or SNPN is selected.

Additionally, the registration attempt counter shall be reset when the UE is in substate 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION or 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE, and:

- the current TAI is changed;
- timer T3502 expires; or
- timer T3346 is started.

When the registration attempt counter is reset, the UE shall stop timer T3519 if running, and delete any stored SUCI.

The lower layers indicate to NAS whether the network supports emergency services for the UE in limited service state (see 3GPP TS 38.331 [30]). This information is taken into account when deciding whether to initiate an initial registration for emergency services.

5.5.1.2 Registration procedure for initial registration

5.5.1.2.1 General

This procedure can be used by a UE for initial registration for 5GS services.

When the UE initiates the registration procedure for initial registration, the UE shall indicate "initial registration" in the 5GS registration type IE. When the UE initiates the registration procedure for emergency services, the UE shall indicate "emergency registration" in the 5GS registration type IE. When the UE initiates the initial registration for onboarding services in SNPN, the UE shall indicate "SNPN onboarding registration" in the 5GS registration type IE. When the UE initiates the initial registration type IE. When the UE initiates the initial registration type IE. When the UE initiates the initial registration type IE. When the UE initiates the initial registration type IE. When the UE initiates the initial registration type IE. When the UE initiates the initial registration procedure for disaster roaming services, the UE shall indicate "disaster roaming initial registration" in the 5GS registration type IE.

If the MUSIM UE initiates the registration procedure for initial registration and indicates "emergency registration" in the 5GS registration type IE in the REGISTRATION REQUEST message, the network shall not indicate the support of:

- the N1 NAS signalling connection release;
- the paging indication for voice services;
- the reject paging request; or
- the paging restriction;

in the REGISTRATION ACCEPT message.

5.5.1.2.2 Initial registration initiation

The UE in state 5GMM-DEREGISTERED shall initiate the registration procedure for initial registration by sending a REGISTRATION REQUEST message to the AMF,

- a) when the UE performs initial registration for 5GS services;
- b) when the UE performs initial registration for emergency services;
- c) when the UE performs initial registration for SMS over NAS;
- d) when the UE moves from GERAN to NG-RAN coverage or the UE moves from a UTRAN to NG-RAN coverage and the following applies:
 - 1) the UE initiated a GPRS attach or routing area updating procedure while in A/Gb mode or Iu mode; or
 - 2) the UE has performed 5G-SRVCC from NG-RAN to UTRAN as specified in 3GPP TS 23.216 [6A],

and since then the UE did not perform a successful EPS attach or tracking area updating procedure in S1 mode or registration procedure in N1 mode;

- e) when the UE performs initial registration for onboarding services in SNPN; and
- f) when the UE performs initial registration for disaster roaming services;

with the following clarifications to initial registration for emergency services:

- a) the UE shall not initiate an initial registration for emergency services over the current access, if the UE is already registered for emergency services over the non-current access, unless the initial registration has to be initiated to perform handover of an existing emergency PDU session from the non-current access to the current access; and
- NOTE 1: Transfer of an existing emergency PDU session between 3GPP access and non-3GPP access is needed e.g. if the UE determines that the current access is no longer available.
- b) the UE can only initiate an initial registration for emergency services over non-3GPP access if it cannot register for emergency services over 3GPP access.

The UE initiates the registration procedure for initial registration by sending a REGISTRATION REQUEST message to the AMF, starting timer T3510. If timer T3502 is currently running, the UE shall stop timer T3502. If timer T3511 is currently running, the UE shall stop timer T3511.

During initial registration the UE handles the 5GS mobile identity IE in the following order:

a) if:

- 1) the UE:
 - i) was previously registered in S1 mode before entering state EMM-DEREGISTERED; and
 - ii) has received an "interworking without N26 interface not supported" indication from the network; and
- 2) EPS security context and a valid native 4G-GUTI are available;

then the UE shall create a 5G-GUTI mapped from the valid native 4G-GUTI as specified in 3GPP TS 23.003 [4] and indicate the mapped 5G-GUTI in the 5GS mobile identity IE. The UE shall include the UE status IE with the EMM registration status set to "UE is not in EMM-REGISTERED state" and shall include an ATTACH REQUEST message as specified in 3GPP TS 24.301 [15] in the EPS NAS message container IE.

Additionally, if the UE holds a valid 5G-GUTI, the UE shall include the 5G-GUTI in the Additional GUTI IE in the REGISTRATION REQUEST message in the following order:

- 1) a valid 5G-GUTI that was previously assigned by the same PLMN with which the UE is performing the registration, if available;
- 2) a valid 5G-GUTI that was previously assigned by an equivalent PLMN, if available; and
- 3) a valid 5G-GUTI that was previously assigned by any other PLMN, if available;

b) if:

- the UE is registering with a PLMN and the UE holds a valid 5G-GUTI that was previously assigned, over 3GPP access or non-3GPP access, by the same PLMN with which the UE is performing the registration, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE; or
- 2) the UE is registering with a SNPN, the UE holds a valid 5G-GUTI that was previously assigned, over 3GPP access or non-3GPP access, by the same SNPN with which the UE is performing the registration, and the UE is not initiating the initial registration for onboarding services in SNPN, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE;
- c) if the UE holds a valid 5G-GUTI that was previously assigned, over 3GPP access or non-3GPP access, by an equivalent PLMN, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE;

d) if:

- the UE is registering with a PLMN and the UE holds a valid 5G-GUTI that was previously assigned, over 3GPP access or non-3GPP access, by any other PLMN, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE; or
- 2) the UE is registering with an SNPN, the UE holds a valid 5G-GUTI that was previously assigned, over 3GPP access or non-3GPP access, by any other SNPN, and the UE is not initiating the initial registration for onboarding services in SNPN, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE and shall additionally include the NID of the other SNPN in the NID IE;
- e) if a SUCI other than an onboarding SUCI is available, and the UE is not initiating the initial registration for onboarding services in SNPN, the UE shall include the SUCI other than an onboarding SUCI in the 5GS mobile identity IE;
- f) if the UE does not hold a valid 5G-GUTI or SUCI other than an onboarding SUCI, and is initiating the initial registration for emergency services, the PEI shall be included in the 5GS mobile identity IE; and
- g) if the UE is initiating the initial registration for onboarding services in SNPN, an onboarding SUCI shall be included in the 5GS mobile identity IE.

NOTE 2: The AMF in ON-SNPN uses the onboarding SUCI as specified in 3GPP TS 23.501 [8].

If the SUCI is included in the 5GS mobile identity IE and the timer T3519 is not running, the UE shall start timer T3519 and store the value of the SUCI sent in the REGISTRATION REQUEST message. The UE shall include the stored SUCI in the REGISTRATION REQUEST message while timer T3519 is running.

If the UE is operating in the dual-registration mode and it is in EMM state EMM-REGISTERED, the UE shall include the UE status IE with the EMM registration status set to "UE is in EMM-REGISTERED state".

- NOTE 3: Inclusion of the UE status IE with this setting corresponds to the indication that the UE is "moving from EPC" as specified in 3GPP TS 23.502 [9].
- NOTE 4: The value of the 5GMM registration status included by the UE in the UE status IE is not used by the AMF.

If the last visited registered TAI is available, the UE shall include the last visited registered TAI in the REGISTRATION REQUEST message.

If the UE requests the use of SMS over NAS, the UE shall include the 5GS update type IE in the REGISTRATION REQUEST message with the SMS requested bit set to "SMS over NAS supported". When the 5GS update type IE is included in the REGISTRATION REQUEST for reasons other than requesting the use of SMS over NAS, and the UE does not need to register for SMS over NAS, the UE shall set the SMS requested bit of the 5GS update type IE to "SMS over NAS not supported" in the REGISTRATION REQUEST message.

If the UE supports MICO mode and requests the use of MICO mode, then the UE shall include the MICO indication IE in the REGISTRATION REQUEST message. If the UE requests to use an active time value, it shall include the active time value in the T3324 IE in the REGISTRATION REQUEST message. If the UE includes the T3324 IE, it may also request a particular T3512 value by including the Requested T3512 IE in the REGISTRATION REQUEST message. Additionally, if the UE supports strictly periodic registration timer, the UE shall set the Strictly Periodic Registration Timer Indication bit of the MICO indication IE in the REGISTRATION REQUEST message to "strictly periodic registration timer supported".

If the UE needs to use the UE specific DRX parameters, the UE shall include the Requested DRX parameters IE in the REGISTRATION REQUEST message.

If the UE is in NB-N1 mode and if the UE needs to use the UE specific DRX parameters for NB-N1 mode, the UE shall include the Requested NB-N1 mode DRX parameters IE in the REGISTRATION REQUEST message.

If the UE supports eDRX and requests the use of eDRX, the UE shall include the Requested extended DRX parameters IE in the REGISTRATION REQUEST message.

If the UE needs to request LADN information for specific LADN DNN(s) or indicates a request for LADN information as specified in 3GPP TS 23.501 [8], the UE shall include the LADN indication IE in the REGISTRATION REQUEST message and:

- request specific LADN DNNs by including a LADN DNN value in the LADN indication IE for each LADN DNN for which the UE requests LADN information; or
- to indicate a request for LADN information by not including any LADN DNN value in the LADN indication IE.

The UE shall include the requested NSSAI containing the S-NSSAI(s) corresponding to the slice(s) to which the UE intends to register with and shall include the mapped S-NSSAI(s) for the requested NSSAI, if available, in the REGISTRATION REQUEST message. If the UE has allowed NSSAI or configured NSSAI or both for the current PLMN or SNPN, the requested NSSAI shall be either:

- a) the configured NSSAI for the current PLMN or SNPN, or a subset thereof as described below;
- b) the allowed NSSAI for the current PLMN or SNPN, or a subset thereof as described below; or
- c) the allowed NSSAI for the current PLMN or SNPN, or a subset thereof as described below, plus one or more S-NSSAIs from the configured NSSAI for which no corresponding S-NSSAI is present in the allowed NSSAI and those are neither in the rejected NSSAI nor in the pending NSSAI.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and has a default configured NSSAI, the UE shall:

a) include the S-NSSAI(s) in the Requested NSSAI IE of the REGISTRATION REQUEST message using the default configured NSSAI; and

3GPP TS 24.501 version 17.16.0 Release 17

b) include the Network slicing indication IE with the Default configured NSSAI indication bit set to "Requested NSSAI created from default configured NSSAI" in the REGISTRATION REQUEST message.

If the UE has no allowed NSSAI for the current PLMN or SNPN, no configured NSSAI for the current PLMN or SNPN, and no default configured NSSAI, the UE shall not include a requested NSSAI in the REGISTRATION REQUEST message.

If all the S-NSSAI(s) corresponding to the slice(s) to which the UE intends to register are included in the pending NSSAI, the UE shall not include a requested NSSAI in the REGISTRATION REQUEST message.

The subset of configured NSSAI provided in the requested NSSAI consists of one or more S-NSSAIs in the configured NSSAI applicable to the current PLMN or SNPN, if the S-NSSAI is neither in the rejected NSSAI nor associated to the S-NSSAI(s) in the rejected NSSAI. In addition, if the NSSRG information is available, the subset of configured NSSAI provided in the requested NSSAI shall be associated with at least one common NSSRG value. If the UE is in 5GMM-REGISTERED state over the other access and has already an allowed NSSAI for the other access in the same PLMN or in different PLMNs, all the S-NSSAI(s) in the requested NSSAI for the current access shall share at least an NSSRG value common to all the S-NSSAI(s) of the allowed NSSAI for the other access. If the UE is simultaneously performing the registration procedure on the other access in different PLMNs, the UE shall include S-NSSAIs that share at least a common NSSRG value across all access types. The S-NSSAIs in the pending NSSAI and requested NSSAI shall be associated with at least one common NSSRG value.

- NOTE 5: If the UE has stored mapped S-NSSAI(s) for the rejected NSSAI, and one or more S-NSSAIs in the stored mapped S-NSSAI(s) for the configured NSSAI are not included in the stored mapped S-NSSAI(s) for the rejected NSSAI, then a S-NSSAI in the configured NSSAI associated to one or more of these mapped S-NSSAI(s) for the configured NSSAI are available to be included in the requested NSSAI together with their mapped S-NSSAI.
- NOTE 6: If one or more mapped S-NSSAIs in the stored mapped S-NSSAI(s) for the configured NSSAI are not included in the stored rejected NSSAI for the failed or revoked NSSAA, a S-NSSAI in the configured NSSAI associated to one or more of these mapped S-NSSAI(s) for the configured NSSAI are available to be included in the registration request together with their mapped S-NSSAI.
- NOTE 7: There is no need to consider the case that the UE is simultaneously performing the registration procedure on the other access in the same PLMN, due to that the UE is not allowed to initiate the registration procedure over one access when the registration over the other access to the same PLMN is going on.

The subset of allowed NSSAI provided in the requested NSSAI consists of one or more S-NSSAIs in the allowed NSSAI for the current PLMN.

- NOTE 8: How the UE selects the subset of configured NSSAI or allowed NSSAI to be provided in the requested NSSAI is implementation specific. The UE can take preferences indicated by the upper layers (e.g. policies like URSP, applications) and UE local configuration into account.
- NOTE 9: The number of S-NSSAI(s) included in the requested NSSAI cannot exceed eight.

If the UE initiates an initial registration for onboarding services in SNPN, the UE shall not include the Requested NSSAI IE in the REGISTRATION REQUEST message.

If the UE supports NSAG, the UE shall set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE initiates an initial registration for emergency services or needs to prolong the established NAS signalling connection after the completion of the initial registration procedure (e.g. due to uplink signalling pending), the UE shall set the Follow-on request indicator to "Follow-on request pending".

NOTE 10: The UE does not have to set the Follow-on request indicator to 1, even if the UE has to request resources for V2X communication over PC5 reference point, 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5.

If the UE supports S1 mode and the 5GS registration type IE in the REGISTRATION REQUEST message is not set to "disaster roaming initial registration", the UE shall:

 set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message;

- include the S1 UE network capability IE in the REGISTRATION REQUEST message; additionally, if the UE supports EPS-UPIP, the UE shall set the EPS-UPIP bit to "EPS-UPIP supported" in the S1 UE network capability IE in the REGISTRATION REQUEST message; and
- if the UE supports sending an ATTACH REQUEST message containing a PDN CONNECTIVITY REQUEST message with request type set to "handover" to transfer a PDU session from N1 mode to S1 mode, set the HO attach bit to "attach request message containing PDN connectivity request with request type set to handover to transfer PDU session from N1 mode to S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the LTE positioning protocol (LPP) in N1 mode as specified in 3GPP TS 37.355 [26], the UE shall set the LPP bit to "LPP in N1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the Location Services (LCS) notification mechanisms in N1 mode as specified in 3GPP TS 23.273 [6B], the UE shall set the 5G-LCS bit to "LCS notification mechanisms supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE is in NB-N1 mode, then the UE shall set the Control plane CIoT 5GS optimization bit to "Control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE is capable of NB-S1 mode, then the UE shall set the Control plane CIoT EPS optimization bit to "Control plane CIoT EPS optimization supported" in the S1 UE network capability IE of the REGISTRATION REQUEST message.

If the UE supports N3 data transfer and multiple user-plane resources in NB-N1 mode (see 3GPP TS 36.306 [25D], 3GPP TS 36.331 [25A]), then the UE shall set the Multiple user-plane resources support bit to "Multiple user-plane resources supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports 5G-SRVCC from NG-RAN to UTRAN as specified in 3GPP TS 23.216 [6A], the UE shall:

- set the 5G-SRVCC from NG-RAN to UTRAN capability bit to "5G-SRVCC from NG-RAN to UTRAN supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- include the Mobile station classmark 2 IE and the Supported codecs IE in the REGISTRATION REQUEST message.

If the UE supports service gap control, then the UE shall set the SGC bit to "service gap control supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the restriction on use of enhanced coverage, the UE shall set the RestrictEC bit to "Restriction on use of enhanced coverage supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports network slice-specific authentication and authorization, the UE shall set the NSSAA bit to "network slice-specific authentication and authorization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports CAG feature, the UE shall set the CAG bit to "CAG Supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports extended CAG information list, the UE shall set the Ex-CAG bit to "Extended CAG information list supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports sending of REGISTRATION COMPLETE message for acknowledging the reception of Negotiated PEIPS assistance information IE in the REGISTRATION ACCEPT message, the UE shall set the RCMP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS assistance information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

When the UE is not in NB-N1 mode, if the UE supports RACS, the UE shall:

- a) set the RACS bit to "RACS supported" in the 5GMM capability IE of the REGISTRATION REQUEST message;
- b) if the UE has an applicable network-assigned UE radio capability ID for the current UE radio configuration in the selected PLMN or SNPN, include the applicable network-assigned UE radio capability ID in the UE radio capability ID IE of the REGISTRATION REQUEST message; and
- c) if the UE:

- 1) does not have an applicable network-assigned UE radio capability ID for the current UE radio configuration in the selected PLMN or SNPN; and
- 2) has an applicable manufacturer-assigned UE radio capability ID for the current UE radio configuration,

include the applicable manufacturer-assigned UE radio capability ID in the UE radio capability ID IE of the REGISTRATION REQUEST message.

If the UE has one or more stored UE policy sections:

- identified by a UPSI with the PLMN ID part indicating the HPLMN or the selected PLMN; or
- identified by a UPSI with the PLMN ID part indicating the PLMN ID part of the SNPN identity of the selected SNPN and associated with the NID of the selected SNPN;

then the UE shall set the Payload container type IE to "UE policy container" and include the UE STATE INDICATION message (see annex D) in the Payload container IE of the REGISTRATION REQUEST message.

NOTE 11:In this version of the protocol, the UE can only include the Payload container IE in the REGISTRATION REQUEST message to carry a payload of type "UE policy container".

If the UE does not have a valid 5G NAS security context, the UE shall send the REGISTRATION REQUEST message without including the NAS message container IE. The UE shall include the entire REGISTRATION REQUEST message (i.e. containing cleartext IEs and non-cleartext IEs, if any) in the NAS message container IE that is sent as part of the SECURITY MODE COMPLETE message as described in subclauses 4.4.6 and 5.4.2.3.

If the UE has a valid 5G NAS security context and the UE needs to send non-cleartext IEs, the UE shall send a REGISTRATION REQUEST message including the NAS message container IE as described in subclause 4.4.6. If the UE does not need to send non-cleartext IEs, the UE shall send a REGISTRATION REQUEST message without including the NAS message container IE.

If the UE supports ciphered broadcast assistance data and needs to obtain new ciphering keys, the UE shall include the Additional information requested IE with the CipherKey bit set to "ciphering keys for ciphered broadcast assistance data requested" in the REGISTRATION REQUEST message.

The UE shall set the WUSA bit to "WUS assistance information reception supported" in the 5GMM capability IE if the UE supports WUS assistance information. The UE may include its UE paging probability information in the Requested WUS assistance information IE if the UE has set the WUSA bit to "WUS assistance information reception supported" in the 5GMM capability IE and the UE is not performing the initial registration for emergency services.

The UE shall set the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE if the UE supports PEIPS assistance information and the 5GS registration type IE in the REGISTRATION REQUEST message is not set to "emergency registration". The UE may include its UE paging probability information in the Requested PEIPS assistance information IE if the UE has set the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE.

If the REGISTRATION REQUEST message includes a NAS message container IE, the AMF shall process the REGISTRATION REQUEST message that is obtained from the NAS message container IE as described in subclause 4.4.6.

If the UE supports V2X as specified in 3GPP TS 24.587 [19B], the UE shall set the V2X bit to "V2X supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports V2X communication over E-UTRA-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCEPC5 bit to "V2X communication over E-UTRA-PC5 supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports V2X communication over NR-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCEPC5 bit to "V2X communication over NR-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCNPC5 bit to "V2X communication over NR-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCNPC5 bit to "V2X communication over NR-PC5 supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

The UE shall set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the NSSRG, then the UE shall set the NSSRG bit to "NSSRG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the W-AGF acting on behalf of an N5GC device initiates an initial registration as specified in 3GPP TS 23.316 [6D], the W-AGF acting on behalf of the N5GC device shall include the N5GC indication IE with the N5GC device indication bit set to "N5GC device registration is requested" in the REGISTRATION REQUEST message.

If the UE supports UAS services, the UE shall set the UAS bit to "UAS services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

When the UE supporting UAS services initiates an initial registration for UAS services, the UE shall include the service-level device ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and set the value to the CAA-level UAV ID. The UE shall include the service-level-AA server address in the Service-level-AA container IE of the REGISTRATION REQUEST message and set the value to the USS address, if it is provided by the upper layers.

If the UE supports 5G ProSe direct discovery as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-dd bit to "5G ProSe direct discovery supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports 5G ProSe direct communication as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-dc bit to "5G ProSe direct communication supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network relay UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2relay bit to "Acting as a 5G ProSe layer-2 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE shall set the 5G ProSe-l2relay bit to "Acting as a 5G ProSe layer-3 UE-to-network relay UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l3relay bit to "Acting as a 5G ProSe layer-3 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-3 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l3relay bit to "Acting as a 5G ProSe layer-3 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2rmt bit to "Acting as a 5G ProSe layer-2 UE-to-network remote UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-3 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l3rmt bit to "Acting as a 5G ProSe layer-3 UE-to-network remote UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports

If the MUSIM UE supports the N1 NAS signalling connection release, then the UE shall set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the MUSIM UE supports the paging indication for voice services, then the UE shall set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the MUSIM UE supports the reject paging request, then the UE shall set the reject paging request bit to "reject paging request supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the reject paging request bit to "reject paging request supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the MUSIM UE sets:

- the reject paging request bit to "reject paging request supported";
- the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported"; or
- both of them;

and supports the paging restriction, then the UE shall set the paging restriction bit to "paging restriction supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the paging restriction bit to "paging restriction supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports MINT, the UE shall set the MINT bit to "MINT supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE initiates the registration procedure for disaster roaming services, the UE has determined the MS determined PLMN with disaster condition as specified in 3GPP TS 23.122 [5] and:

- a) the MS determined PLMN with disaster condition is the HPLMN and:
 - 1) the Additional GUTI IE is included in the REGISTRATION REQUEST message and does not contain a valid 5G-GUTI that was previously assigned by the HPLMN; or
 - 2) the Additional GUTI IE is not included in the REGISTRATION REQUEST message and the 5GS mobile identity IE contains neither the SUCI nor a valid 5G-GUTI that was previously assigned by the HPLMN; or
- b) the MS determined PLMN with disaster condition is not the HPLMN and:
 - 1) the Additional GUTI IE is included in the REGISTRATION REQUEST message and does not contain a valid 5G-GUTI that was previously assigned by the MS determined PLMN with disaster condition; or
 - the Additional GUTI IE is not included in the REGISTRATION REQUEST message and the 5GS mobile identity IE does not contain a valid 5G-GUTI that was previously assigned by the MS determined PLMN with disaster condition;

the UE shall include in the REGISTRATION REQUEST message the MS determined PLMN with disaster condition IE indicating the MS determined PLMN with disaster condition.

NOTE 12: If the UE initiates the registration procedure for disaster roaming services, and the MS determined PLMN with disaster condition cannot be determined when an NG-RAN cell of the PLMN broadcasts the disaster related indication as specified in 3GPP TS 23.122 [5], the UE does not include in the REGISTRATION REQUEST message the MS determined PLMN with disaster condition IE but includes the Additional GUTI IE or the 5GS mobile identity IE or both as specified in subclauses 5.5.1.2.2.

If the UE supports event notification, the UE shall set the EventNotification bit to "Event notification supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports access to an SNPN using credentials from a credentials holder and the UE is in its HPLMN or EHPLMN or a subscribed SNPN, the UE shall set the SSNPNSI bit to "SOR-SNPN-SI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports MPS indicator update via the UE configuration update procedure, the UE shall set the MPSIU bit to "MPS indicator update supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.


Figure 5.5.1.2.2.1: Registration procedure for initial registration

5.5.1.2.3 5GMM common procedure initiation

The network may initiate 5GMM common procedures, e.g. the identification, authentication and security procedures during the registration procedure, depending on the information received in the REGISTRATION REQUEST message.

If two NAS security mode control procedures are needed to signal an entire unciphered REGISTRATION REQUEST message followed by signalling of selected EPS NAS security algorithms, both NAS security mode control procedures should be initiated as part of 5GMM common procedures of the ongoing registration procedure (see clause 5.4.2.4).

During a registration procedure with 5GS registration type IE set to "emergency registration", if the AMF is configured to support emergency registration for unauthenticated SUCIs, the AMF may choose to skip the authentication procedure even if no 5G NAS security context is available and proceed directly to the execution of the security mode control procedure.

5.5.1.2.4 Initial registration accepted by the network

During a registration procedure with 5GS registration type IE set to "emergency registration", the AMF shall not check for mobility and access restrictions, regional restrictions or subscription restrictions, or CAG restrictions when processing the REGISTRATION REQUEST message.

If the initial registration request is accepted by the network, the AMF shall send a REGISTRATION ACCEPT message to the UE.

For each of the information elements: 5GMM capability, S1 UE network capability, and UE security capability, the AMF shall store all octets received from the UE in the REGISTRATION REQUEST message, up to the maximum length defined for the respective information element.

NOTE 1: This information is forwarded to the new AMF during inter-AMF handover or to the new MME during inter-system handover to S1 mode.

The AMF shall assign and include a TAI list as a registration area the UE is registered to in the REGISTRATION ACCEPT message. The AMF shall not assign a TAI list containing both tracking areas in NB-N1 mode and tracking areas not in NB-N1 mode. The UE, upon receiving a REGISTRATION ACCEPT message, shall delete its old TAI list and store the received TAI list. If the REGISTRATION REQUEST message was received over non-3GPP access, the AMF shall include a single TAI in the TAI list.

- NOTE 2: For non-3GPP access, the operator can allocate a TAI per N3IWF, TNGF, TWIF or W-AGF. Each N3IWF, TNGF, TWIF or W-AGF is locally configured with a TAI. Each N3IWF, TNGF, TWIF or W-AGF can be configured with its own TAI value, or with the same TAI value as other N3IWFs, TNGFs, TWIFs or W-AGFs.
- NOTE 3: When assigning the TAI list, the AMF can take into account the eNodeB's capability of support of CIoT 5GS optimization.

The AMF may include service area restrictions in the Service area list IE in the REGISTRATION ACCEPT message. The UE, upon receiving a REGISTRATION ACCEPT message with the service area restrictions shall act as described in subclause 5.3.5.

The AMF may also include a list of equivalent PLMNs in the REGISTRATION ACCEPT message. Each entry in the list contains a PLMN code (MCC+MNC). The UE shall store the list as provided by the network, and if the initial registration procedure is not for emergency services, the UE shall remove from the list any PLMN code that is already in the forbidden PLMN list as specified in subclause 5.3.13A. In addition, the UE shall add to the stored list the PLMN code of the registered PLMN that sent the list. The UE shall replace the stored list on each receipt of the REGISTRATION ACCEPT message. If the REGISTRATION ACCEPT message does not contain a list, then the UE shall delete the stored list.

If the initial registration procedure is not for emergency services, the UE is not registered for disaster roaming services, and if the PLMN identity of the registered PLMN is a member of the forbidden PLMN list as specified in subclause 5.3.13A, any such PLMN identity shall be deleted from the corresponding list(s).

If the Service area list IE is not included in the REGISTRATION ACCEPT message, any tracking area in the registered PLMN and its equivalent PLMN(s) in the registration area is considered as an allowed tracking area as described in subclause 5.3.5.

If the REGISTRATION REQUEST message contains the LADN indication IE, based on the LADN indication IE, UE subscription information, UE location and local configuration about LADN and:

- if the LADN indication IE includes requested LADN DNNs, the UE subscribed DNN list includes the requested LADN DNNs or the wildcard DNN, and the LADN service area of the requested LADN DNN has an intersection with the current registration area, the AMF shall determine the requested LADN DNNs included in the LADN indication IE as LADN DNNs for the UE;
- if no requested LADN DNNs included in the LADN indication IE and the wildcard DNN is included in the UE subscribed DNN list, the AMF shall determine the LADN DNN(s) configured in the AMF whose LADN service area has an intersection with the current registration area as LADN DNNs for the UE; or
- if no requested LADN DNNs included in the LADN indication IE and the wildcard DNN is not included in the UE subscribed DNN list, or if the UE subscribed DNN list does not include any of the DNNs in the LADN indication IE, the AMF shall determine the LADN DNN(s) included in the UE subscribed DNN list whose LADN service area has an intersection with the current registration area as LADN DNNs for the UE.

If the LADN indication IE is not included in the REGISTRATION REQUEST message, the AMF shall determine the LADN DNN(s) included in the UE subscribed DNN list whose service area has an intersection with the current registration area as LADN DNNs for the UE, except for the wildcard DNN included in the UE subscribed DNN list.

If the UE supports WUS assistance information and the AMF supports and accepts the use of WUS assistance information for the UE, then the AMF shall determine the negotiated UE paging probability information for the UE, store it in the 5GMM context of the UE, and if the UE is not performing the initial registration for emergency services, the AMF shall include it in the Negotiated WUS assistance information IE in the REGISTRATION ACCEPT message. The AMF may consider the UE paging probability information received in the Requested WUS assistance information IE when determining the negotiated UE paging probability information for the UE.

NOTE 4: Besides the UE paging probability information requested by the UE, the AMF can take local configuration or previous statistical information for the UE into account when determining the negotiated UE paging probability information for the UE.

If the UE sets the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE in the REGISTRATION REQUEST message and the AMF supports and accepts the use of PEIPS assistance information for the UE, then the AMF shall determine the Paging subgroup ID for the UE, store it in the 5GMM context of the UE, and shall include it in the Negotiated PEIPS assistance information IE in the REGISTRATION ACCEPT message or in the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message as part of the registration procedure. The AMF may consider the UE paging probability information received in the Requested PEIPS assistance information IE when determining the Paging subgroup ID for the UE.

NOTE 5: Besides the UE paging probability information when provided by the UE, the AMF can also take local configuration, whether the UE is likely to receive IMS voice over PS session calls, UE mobility pattern or previous statistical information for the UE or information provided by the NG-RAN into account when determining the Paging subgroup ID for the UE.

The AMF shall include the LADN information which consists of the determined LADN DNNs for the UE and LADN service area(s) available in the current registration area in the LADN information IE of the REGISTRATION ACCEPT message.

The UE, upon receiving the REGISTRATION ACCEPT message with the LADN information, shall store the received LADN information. If there exists one or more LADN DNNs which are included in the LADN indication IE of the REGISTRATION REQUEST message and are not included in the LADN information IE of the REGISTRATION ACCEPT message, the UE considers such LADN DNNs as not available in the current registration area.

The 5G-GUTI reallocation shall be part of the initial registration procedure. During the initial registration procedure, if the AMF has not allocated a new 5G-GUTI by the generic UE configuration update procedure, the AMF shall include in the REGISTRATION ACCEPT message the new assigned 5G-GUTI together with the assigned TAI list.

If the UE has set the CAG bit to "CAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the AMF needs to update the "CAG information list" stored in the UE, the AMF shall include the CAG information list IE or the Extended CAG information list IE in the REGISTRATION ACCEPT message.

- NOTE 6: The "CAG information list" can be provided by the AMF and include no entry if no "CAG information list" exists in the subscription.
- NOTE 7: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If a 5G-GUTI or the SOR transparent container IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the Operator-defined access category definitions IE, the Extended emergency number list IE, the CAG information list IE or the Extended CAG information list IE are included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE has set the RCMAP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS assistance information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if the Negotiated PEIPS assistance information IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE is not in NB-N1 mode and the UE has set the RACS bit to "RACS supported" in the 5GMM Capability IE of the REGISTRATION REQUEST message, the AMF may include either a UE radio capability ID IE or a UE radio capability ID deletion indication IE in the REGISTRATION ACCEPT message. If the UE radio capability ID IE or the UE radio capability ID deletion indication IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

The AMF shall include the MICO indication IE in the REGISTRATION ACCEPT message only if the MICO indication IE was included in the REGISTRATION REQUEST message, the AMF supports and accepts the use of MICO mode. If the AMF supports and accepts the use of MICO mode, the AMF may indicate "all PLMN registration"

area allocated" in the MICO indication IE in the REGISTRATION ACCEPT message. If "all PLMN registration area allocated" is indicated in the MICO indication IE, the AMF shall not assign and include the TAI list in the REGISTRATION ACCEPT message. If the REGISTRATION ACCEPT message included an MICO indication IE indicating "all PLMN registration area allocated", the UE shall treat all TAIs in the current PLMN as a registration area and delete its old TAI list. If "strictly periodic registration timer supported" is indicated in the MICO indication IE in the REGISTRATION REQUEST message, the AMF may indicate "strictly periodic registration timer supported" in the MICO indication IE in the REGISTRATION ACCEPT message.

The AMF shall include an active time value in the T3324 IE in the REGISTRATION ACCEPT message if the UE requested an active time value in the REGISTRATION REQUEST message and the AMF accepts the use of MICO mode and the use of active time.

If the AMF supports and accepts the use of MICO, and the UE included the Requested T3512 value IE in the REGISTRATION REQUEST message, then the AMF shall take into account the T3512 value requested when providing the T3512 value IE in the REGISTRATION ACCEPT message.

NOTE 8: The T3512 value assigned to the UE by AMF can be different from the T3512 value requested by the UE. AMF can take several factors into account when assigning the T3512 value, e.g. local configuration, expected UE behaviour, UE requested T3512 value, UE subscription data, network policies.

The AMF shall include the T3512 value IE in the REGISTRATION ACCEPT message only if the REGISTRATION REQUEST message was sent over the 3GPP access.

The AMF shall include the non-3GPP de-registration timer value IE in the REGISTRATION ACCEPT message only if the REGISTRATION REQUEST message was sent over the non-3GPP access.

If the UE requests "control plane CIoT 5GS optimization" in the 5GS update type IE, indicates support of control plane CIoT 5GS optimization in the 5GMM capability IE and the AMF decides to accept the requested CIoT 5GS optimization and the registration request, the AMF shall indicate "control plane CIoT 5GS optimization supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

The AMF may include the T3447 value IE set to the service gap time value in the REGISTRATION ACCEPT message if:

- the UE has indicated support for service gap control in the REGISTRATION REQUEST message; and
- a service gap time value is available in the 5GMM context.

If there is a running T3447 timer in the AMF and the Follow-on request indicator is set to "Follow-on request pending" in the REGISTRATION REQUEST message, the AMF shall ignore the flag and proceed as if the flag was not received except for the following cases:

- a) the UE is configured for high priority access in the selected PLMN; or
- b) the 5GS registration type IE in the REGISTRATION REQUEST message is set to "emergency registration".

If the UE has indicated support for the control plane CIoT 5GS optimizations, and the AMF decides to activate the congestion control for transport of user data via the control plane, then the AMF shall include the T3448 value IE in the REGISTRATION ACCEPT message.

If:

- the UE in NB-N1 mode is using control plane CIoT 5GS optimization; and
- the network is configured to provide the truncated 5G-S-TMSI configuration for control plane CIoT 5GS optimizations;

the AMF shall include the Truncated 5G-S-TMSI configuration IE in the REGISTRATION ACCEPT message and set the "Truncated AMF Set ID value" and the "Truncated AMF Pointer value" in the Truncated 5G-S-TMSI configuration IE based on network policies. The AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message, and if:

- the UE has a valid aerial UE subscription information;

- the UUAA procedure is to be performed during the registration procedure according to operator policy;
- there is no valid successful UUAA result for the UE in the UE 5GMM context; and
- the REGISTRATION REQUEST message was not received over non-3GPP access,

then the AMF shall initiate the UUAA-MM procedure with the UAS-NF as specified in TS 23.256 [6AB] and shall include a service-level-AA pending indication in the Service-level-AA container IE of the REGISTRATION ACCEPT message. The AMF shall store in the UE 5GMM context that a UUAA procedure is pending. The AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3. If the REGISTRATION REQUEST message was received over non-3GPP access, the AMF shall not initiate UUAA-MM procedure.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message, and if:

- the UE has a valid aerial UE subscription information;
- the UUAA procedure is to be performed during the registration procedure according to operator policy; and
- there is a valid successful UUAA result for the UE in the UE 5GMM context,

then the AMF shall include a service-level-AA response in the Service-level-AA container IE of the REGISTRATION ACCEPT message and set the SLAR field in the service-level-AA response to "Service level authentication and authorization was successful".

If the AMF determines that the UUAA-MM procedure needs to be performed for a UE, the AMF has not received the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message from the UE and the AMF decides to accept the UE to be registered for other services than UAS services based on the user's subscription data and the operator policy, the AMF shall accept the initial registration request and shall mark in the UE's 5GMM context that the UE is not allowed to request UAS services.

If the UE supports MINT, the AMF may include the List of PLMNs to be used in disaster condition IE in the REGISTRATION ACCEPT message.

If the UE supports MINT, the AMF may include the Disaster roaming wait range IE in the REGISTRATION ACCEPT message.

If the UE supports MINT, the AMF may include the Disaster return wait range IE in the REGISTRATION ACCEPT message.

NOTE 9: The AMF can determine the contents of the "list of PLMN(s) to be used in disaster condition", the value of the disaster roaming wait range and the value of the disaster return wait range based on the network local configuration.

If the AMF received the list of TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, any but not all TAIs in the received list of TAIs is forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

- a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or
- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or
- c) both;

in the REGISTRATION ACCEPT message.

NOTE 10: Void.

Upon receipt of the REGISTRATION ACCEPT message, the UE shall reset the registration attempt counter, enter state 5GMM-REGISTERED and set the 5GS update status to 5U1 UPDATED.

If the UE receives the REGISTRATION ACCEPT message from a PLMN, then the UE shall reset the PLMN-specific attempt counter for that PLMN for the specific access type for which the message was received. The UE shall also reset the PLMN-specific N1 mode attempt counter for that PLMN for the specific access type for which the message was received. If the message was received via 3GPP access, the UE shall reset the counter for "SIM/USIM considered

invalid for GPRS services" events and the counter for "SIM/USIM considered invalid for non-GPRS services" events, if any. If the message was received via non-3GPP access, the UE shall reset the counter for "USIM considered invalid for 5GS services over non-3GPP" events.

If the UE receives the REGISTRATION ACCEPT message from an SNPN, then the UE shall reset the SNPN-specific attempt counter for the current SNPN for the specific access type for which the message was received. If the message was received via 3GPP access, the UE shall reset the counter for "the entry for the current SNPN considered invalid for 3GPP access" events. If the message was received via non-3GPP access, the UE shall reset the counter for "the entry for the current SNPN considered invalid for the current SNPN considered invalid for non-3GPP access" events.

If the REGISTRATION ACCEPT message included a T3512 value IE, the UE shall use the value in the T3512 value IE as periodic registration update timer (T3512).

If the REGISTRATION ACCEPT message include a T3324 value IE, the UE shall use the value in the T3324 value IE as active timer (T3324).

If the REGISTRATION ACCEPT message included a non-3GPP de-registration timer value IE, the UE shall use the value in non-3GPP de-registration timer value IE as non-3GPP de-registration timer.

If the REGISTRATION ACCEPT message contains the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed", or contains a Configured NSSAI IE with a new configured NSSAI for the current PLMN or SNPN and optionally the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN, or contains an NSSRG information IE with a new NSSRG information, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge the successful update of the network slicing information. If the UE has set the RCMAN bit to "Sending of REGISTRATION COMPLETE message for NSAG information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if REGISTRATION ACCEPT message contains the NSAG information IE, the UE shall return REGISTRATION COMPLETE message to the AMF to acknowledge the successful update of the REGISTRATION ACCEPT message contains the NSAG information IE, the UE shall return REGISTRATION COMPLETE message to the AMF to acknowledge the reception of the NSAG information IE.

If the REGISTRATION ACCEPT message contains the CAG information list IE or the Extended CAG information list IE and the UE had set the CAG bit to "CAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the UE shall:

- a) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
- NOTE 11: When the UE receives the CAG information list IE or the Extended CAG information list IE in the HPLMN derived from the IMSI, the EHPLMN list is present and is not empty and the HPLMN is not present in the EHPLMN list, the UE behaves as if it receives the CAG information list IE or the Extended CAG information list IE in a VPLMN.
- b) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 12: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
- c) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

The UE shall store the "CAG information list" received in the CAG information list IE or the Extended CAG information list IE as specified in annex C.

If the received "CAG information list" includes an entry containing the identity of the registered PLMN, the UE shall operate as follows:

a) if the UE receives the REGISTRATION ACCEPT message via a CAG cell, the entry for the registered PLMN in the received "CAG information list" does not include any of the CAG-ID(s) supported by the current CAG cell, and:

- the entry for the registered PLMN in the received "CAG information list" does not include an "indication that the UE is only allowed to access 5GS via CAG cells", then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list"; or
- 2) the entry for the registered PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:
 - i) if the entry for the registered PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
 - ii) if the entry for the registered PLMN in the received "CAG information list" does not include any CAG-ID, the UE has not set the 5GS registration type IE in the REGISTRATION REQUEST message to "emergency registration", and the initial registration was not initiated to perform handover of an existing emergency PDU session from the non-current access to the current access, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
- b) if the UE receives the REGISTRATION ACCEPT message via a non-CAG cell and the entry for the registered PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:
 - if the "allowed CAG list" for the registered PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
 - 2) if the entry for the registered PLMN in the received "CAG information list" does not include any CAG-ID, the UE has not set the 5GS registration type IE in the REGISTRATION REQUEST message to "emergency registration", and the initial registration was not initiated to perform handover of an existing emergency PDU session from the non-current access to the current access, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".

If the received "CAG information list" does not include an entry containing the identity of the registered PLMN and the UE receives the REGISTRATION ACCEPT message via a CAG cell, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".

If the REGISTRATION ACCEPT message contains the Operator-defined access category definitions IE, the Extended emergency number list IE the CAG information list IE or the Extended CAG information list IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the operator-defined access category definitions, the extended local emergency numbers list or the "CAG information list".

If the UE has set the RCMAP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS parameters supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if REGISTRATION ACCEPT message contains the Negotiated PEIPS assistance information IE, the UE shall return REGISTRATION COMPLETE message to the AMF to acknowledge the reception of the Negotiated PEIPS assistance information IE.

If the REGISTRATION ACCEPT message contains the UE radio capability ID IE or the UE radio capability ID deletion indication IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the UE radio capability ID IE or the UE radio capability ID deletion indication IE.

Upon receiving a REGISTRATION COMPLETE message, the AMF shall stop timer T3550 and change to state 5GMM-REGISTERED. The 5G-GUTI, if sent in the REGISTRATION ACCEPT message, shall be considered as valid, the PEIPS assistance information, if sent in the REGISTRATION ACCEPT message, shall be considered as valid, and the UE radio capability ID, if sent in the REGISTRATION ACCEPT, shall be considered as valid.

If the 5GS update type IE was included in the REGISTRATION REQUEST message with the SMS requested bit set to "SMS over NAS supported", and SMSF selection is successful, then the AMF shall send the REGISTRATION ACCEPT message after the SMSF has confirmed that the activation of the SMS service was successful. When sending the REGISTRATION ACCEPT message, the AMF shall:

- a) set the SMS allowed bit of the 5GS registration result IE to "SMS over NAS allowed" in the REGISTRATION ACCEPT message, if the UE has set the SMS requested bit of the 5GS update type IE to "SMS over NAS supported" in the REGISTRATION REQUEST message and the network allows the use of SMS over NAS for the UE; and
- b) store the SMSF address and the value of the SMS allowed bit of the 5GS registration result IE in the UE 5GMM context and consider the UE available for SMS over NAS.

If:

- a) the SMSF selection in the AMF is not successful;
- b) the SMS activation via the SMSF is not successful;
- c) the AMF does not allow the use of SMS over NAS;
- d) the SMS requested bit of the 5GS update type IE was set to "SMS over NAS not supported" in the REGISTRATION REQUEST message; or
- e) the 5GS update type IE was not included in the REGISTRATION REQUEST message;

then the AMF shall set the SMS allowed bit of the 5GS registration result IE to "SMS over NAS not allowed" in the REGISTRATION ACCEPT message.

When the UE receives the REGISTRATION ACCEPT message, if the UE is also registered over another access to the same PLMN, the UE considers the value indicated by the SMS allowed bit of the 5GS registration result IE as applicable for both accesses over which the UE is registered.

The AMF shall include the 5GS registration result IE in the REGISTRATION ACCEPT message. If the 5GS registration result IE value indicates:

- a) "3GPP access", the UE:
 - shall consider itself as being registered to 3GPP access only; and
 - if in 5GMM-REGISTERED state over non-3GPP access and on the same PLMN as 3GPP access, shall enter state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION over non-3GPP access and set the 5GS update status to 5U2 NOT UPDATED over non-3GPP access;
- b) "Non-3GPP access", the UE:
 - shall consider itself as being registered to non-3GPP access only; and
 - if in the 5GMM-REGISTERED state over 3GPP access and is on the same PLMN as non-3GPP access, shall enter the state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION over 3GPP access and set the 5GS update status to 5U2 NOT UPDATED over 3GPP access; or
- c) "3GPP access and Non-3GPP access", the UE shall consider itself as being registered to both 3GPP access and non-3GPP access.

In roaming scenarios, the AMF shall provide mapped S-NSSAI(s) for the configured NSSAI, the allowed NSSAI, the rejected NSSAI (if Extended rejected NSSAI IE is used), the pending NSSAI or NSSRG information when included in the REGISTRATION ACCEPT message.

The AMF shall include the allowed NSSAI for the current PLMN or SNPN and, in roaming scenarios, shall include the mapped S-NSSAI(s) for the allowed NSSAI contained in the requested NSSAI from the UE, in the REGISTRATION ACCEPT message if the UE included the requested NSSAI in the REGISTRATION REQUEST message and the AMF allows one or more S-NSSAIs in the requested NSSAI. Additionally, if the AMF allows one or more subscribed S-NSSAIs for the UE, the AMF may include the allowed subscribed S-NSSAI(s) in the allowed NSSAI in the REGISTRATION ACCEPT message.

The AMF may also include rejected NSSAI in the REGISTRATION ACCEPT message if the initial registration request is not for onboarding services in SNPN. If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected NSSAI shall be included in the Extended rejected NSSAI IE in the REGISTRATION ACCEPT message; otherwise the rejected NSSAI shall be

included in the Rejected NSSAI IE in the REGISTRATION ACCEPT message. If the initial registration request is for onboarding services in SNPN, the AMF shall not include rejected NSSAI in the REGISTRATION ACCEPT message.

If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected NSSAI contains S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with rejection cause(s); otherwise the rejected NSSAI contains S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with rejection cause(s); which was included in the requested NSSAI but rejected by the network associated with rejection cause(s) with the following restrictions:

- a) rejected NSSAI for the current PLMN or SNPN shall not include an S-NSSAI for the current PLMN or SNPN which is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are not allowed; and
- b) rejected NSSAI for the current registration area shall not include an S-NSSAI for the current PLMN or SNPN which is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are not allowed.
- NOTE 13: The UE that does not support extended rejected NSSAI can avoid requesting an S-NSSAI associated with a mapped S-NSSAI, which was included in the previous requested NSSAI but neither in the allowed NSSAI nor in the rejected NSSAI in the consequent registration procedures.

If the UE indicated the support for network slice-specific authentication and authorization, and if the Requested NSSAI IE includes one or more S-NSSAIs subject to network slice-specific authentication and authorization, the AMF shall in the REGISTRATION ACCEPT message include:

- a) the allowed NSSAI containing the S-NSSAI(s) or the mapped S-NSSAI(s), if any:
 - which are not subject to network slice-specific authentication and authorization and are allowed by the AMF; or
 - 2) for which the network slice-specific authentication and authorization has been successfully performed;
- b) optionally, the rejected NSSAI;
- c) pending NSSAI containing one or more S-NSSAIs for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing, and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing, if any; and
- d) the "NSSAA to be performed" indicator in the 5GS registration result IE set to indicate that the network slicespecific authentication and authorization procedure will be performed by the network, if the allowed NSSAI is not included in the REGISTRATION ACCEPT message.

If the initial registration request is not for onboarding services in SNPN, the UE indicated the support for network slice-specific authentication and authorization, and:

- a) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed;
- b) all default S-NSSAIs are subject to network slice-specific authentication and authorization; and
- c) the network slice-specific authentication and authorization procedure has not been successfully performed for any of the default S-NSSAIs,

the AMF shall in the REGISTRATION ACCEPT message include:

- a) the "NSSAA to be performed" indicator in the 5GS registration result IE to indicate that the network slicespecific authentication and authorization procedure will be performed by the network;
- b) pending NSSAI containing one or more default S-NSSAIs for which network slice-specific authentication and authorization will be performed or is ongoing and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing (if any); and
- c) optionally, the rejected NSSAI.

If the initial registration request is not for onboarding services in SNPN, the UE indicated the support for network slice-specific authentication and authorization, and:

- a) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed; and
- b) one or more default S-NSSAIs are not subject to network slice-specific authentication and authorization or the network slice-specific authentication and authorization procedure has been successfully performed for one or more default S-NSSAIs;

the AMF shall in the REGISTRATION ACCEPT message include:

- a) pending NSSAI containing one or more default S-NSSAIs for which network slice-specific authentication and authorization will be performed or is ongoing (if any) and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing (if any);
- b) allowed NSSAI containing S-NSSAI(s) for the current PLMN each of which corresponds to a default S-NSSAI which are not subject to network slice-specific authentication and authorization or for which the network slicespecific authentication and authorization has been successfully performed;
- c) allowed NSSAI containing one or more default S-NSSAIs, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, which are not subject to network slice-specific authentication and authorization or for which the network slice-specific authentication and authorization has been successfully performed; and
- d) optionally, the rejected NSSAI.

If the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed, the allowed NSSAI shall not contain default S-NSSAI(s) that are subject to NSAC. If the subscription information includes the NSSRG information, the S-NSSAIs of the allowed NSSAI shall be associated with at least one common NSSRG value.

When the REGISTRATION ACCEPT message includes a pending NSSAI, the pending NSSAI shall contain all S-NSSAIs for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing from the requested NSSAI of the REGISTRATION REQUEST message that was received over the 3GPP access, non-3GPP access, or both the 3GPP access and non-3GPP access.

If the UE supports extended rejected NSSAI and the AMF determines that maximum number of UEs reached for one or more S-NSSAI(s) in the requested NSSAI as specified in subclause 4.6.2.5, the AMF shall include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE in the REGISTRATION ACCEPT message. In addition, the AMF may include a back-off timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" included in the Extended rejected NSSAI IE of the REGISTRATION ACCEPT message. To avoid that large numbers of UEs simultaneously initiate deferred requests, the network should select the value for the backoff timer for each S-NSSAI for the informed UEs so that timeouts are not synchronised.

If the UE does not indicate support for extended rejected NSSAI and the maximum number of UEs has been reached, the AMF should include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available in the current registration area" in the Rejected NSSAI IE and should not include these S-NSSAIs in the allowed NSSAI in the REGISTRATION ACCEPT message.

NOTE 14:Based on network policies, the AMF can include the S-NSSAI(s) for which the maximum number of UEs has been reached in the rejected NSSAI with rejection causes other than "S-NSSAI not available in the current registration area".

If the AMF has a new configured NSSAI for the current PLMN or SNPN, the AMF shall include the configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message.

NOTE 14A: A new configured NSSAI can be available at the AMF following an indication that the subscription data for network slicing has changed.

The AMF may include a new configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message if:

- a) the REGISTRATION REQUEST message did not include the requested NSSAI and the initial registration request is not for onboarding services in SNPN;
- b) the REGISTRATION REQUEST message included the requested NSSAI containing an S-NSSAI that is not valid in the serving PLMN or SNPN;
- c) the REGISTRATION REQUEST message included the requested NSSAI containing S-NSSAI(s) with incorrect mapped S-NSSAI(s);
- d) the REGISTRATION REQUEST message included the Network slicing indication IE with the Default configured NSSAI indication bit set to "Requested NSSAI created from default configured NSSAI";
- e) the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message are not associated with any common NSSRG value, except for the case that the AMF, based on the indication received from the UDM as specified in 3GPP TS 23.501 [8], has provided all subscribed S-NSSAIs in the configured NSSAI to a UE who does not support NSSRG; or
- NOTE 15: If the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message are not associated with any common NSSRG value, it is possible that at least one of the S-NSSAIs is not included in any of new allowed NSSAI, new (extended) rejected NSSAI (if applicable), and new pending NSSAI (if applicable).
- f) the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message over the current access and the allowed NSSAI over the other access are not associated with any common NSSRG value.

If a new configured NSSAI for the current PLMN or SNPN is included in the REGISTRATION ACCEPT message and the UE is roaming, the AMF shall also include the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message. In this case the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If a new configured NSSAI for the current PLMN is included in the REGISTRATION ACCEPT message, the subscription information includes the NSSRG information, and the NSSRG bit in the 5GMM capability IE of the REGISTRATION REQUEST message is set to:

- a) "NSSRG supported", then the AMF shall include the NSSRG information in the REGISTRATION ACCEPT message; or
- b) "NSSRG not supported", then the configured NSSAI shall include one or more S-NSSAIs each of which is associated with all the NSSRG value(s) of the default S-NSSAI(s), or the configured NSSAI shall include, based on the indication received from the UDM as specified in 3GPP TS 23.501 [8], all subscribed S-NSSAIs even if these S-NSSAIs do not share any common NSSRG value.

If the AMF needs to update the NSSRG information and the UE has set the NSSRG bit to "NSSRG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, then the AMF shall include the new NSSRG information in the REGISTRATION ACCEPT message. In addition, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE has set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message over 3GPP access, the AMF may include the NSAG information IE in the REGISTRATION ACCEPT message. Up to 4 NSAG entries are allowed to be associated with a TAI list in the NSAG information IE.

- NOTE 15a: How the AMF selects NSAG entries to be included in the NSAG information IE is implementation specific, e.g. take the NSAG priority and the current registration area into account.
- NOTE 15b: If the NSAG for the PLMN and its equivalent PLMN(s) have different associations with S-NSSAIs, then the AMF includes a TAI list for the NSAG entry in the NSAG information IE.

If the UE receives the NSAG information IE in the REGISTRATION ACCEPT message, the UE shall store the NSAG information as specified in subclause 4.6.2.2.

If the UE requests ciphering keys for ciphered broadcast assistance data in the REGISTRATION REQUEST message and the AMF has valid ciphering key data applicable to the UE's subscription and current tracking area, then the AMF shall include the ciphering key data in the Ciphering key data IE of the REGISTRATION ACCEPT message.

The AMF shall include the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed" in the REGISTRATION ACCEPT message if the UDM has indicated that the subscription data for network slicing has changed. In this case the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

The UE that has indicated the support for network slice-specific authentication and authorization receiving the pending NSSAI in the REGISTRATION ACCEPT message shall store the S-NSSAI(s) in the pending NSSAI as specified in subclause 4.6.2.2. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the UE shall store the received pending NSSAI for each of the equivalent PLMNs as specified in subclause 4.6.2.2. If the pending NSSAI is not included in the REGISTRATION ACCEPT message and the "NSSAA to be performed" indicator is not set to "Network slice-specific authentication and authorization is to be performed" in the 5GS registration result IE of the REGISTRATION ACCEPT message, then the UE shall delete the pending NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), if existing, as specified in subclause 4.6.2.2.

The UE receiving the rejected NSSAI in the REGISTRATION ACCEPT message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

Unless the back-off timer value received along with the S-NSSAI is zero, the UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

NOTE 16:If the back-off timer value received along with the S-NSSAI in the rejected NSSAI for the maximum number of UEs reached is zero as specified in subclause 10.5.7.4a of TS 24.008, the UE does not consider the S-NSSAI as the rejected S-NSSAI.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if a back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or

- 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

If the UE sets the NSSAA bit in the 5GMM capability IE to "Network slice-specific authentication and authorization not supported", and:

- a) if the Requested NSSAI IE only includes the S-NSSAI(s) subject to network slice-specific authentication and authorization and one or more default S-NSSAIs (containing one or more S-NSSAIs each of which may be associated with a new S-NSSAI) which are not subject to network slice-specific authentication and authorization are available, the AMF shall in the REGISTRATION ACCEPT message include:
 - 1) the allowed NSSAI containing S-NSSAI(s) for the current PLMN or SNPN each of which corresponds to a default S-NSSAI which are not subject to network slice-specific authentication and authorization;
 - 2) the allowed NSSAI containing the default S-NSSAIs, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, which are not subject to network slice-specific authentication and authorization; and
 - 3) the rejected NSSAI containing the S-NSSAI(s) subject to network slice specific authentication and authorization with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN", except if the UE has not set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the S-NSSAI(s) is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are subject to NSSAA; or
- b) if the Requested NSSAI IE includes one or more S-NSSAIs subject to network slice-specific authentication and authorization, the AMF shall in the REGISTRATION ACCEPT message include:
 - 1) the allowed NSSAI containing the S-NSSAI(s) or the mapped S-NSSAI(s) which are not subject to network slice-specific authentication and authorization; and
 - 2) the rejected NSSAI containing:
 - the S-NSSAI(s) subject to network slice specific authentication and authorization with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN", except if the UE has not set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the S-NSSAI is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are subject to NSSAA; and
 - ii) the S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN" or the rejection cause indicating "S-NSSAI not available in the current registration area", if any.

If the UE does not indicate support for network slice-specific authentication and authorization, the initial registration request is not for onboarding services in SNPN, and if:

- a) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message; or
- b) none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed;

and one or more default S-NSSAIs (containing one or more S-NSSAIs each of which may be associated with a new S-NSSAI) which are not subject to network slice-specific authentication and authorization are available, the AMF shall:

- a) put the allowed S-NSSAI(s) for the current PLMN or SNPN each of which corresponds to a default S-NSSAI and not subject to network slice-specific authentication and authorization in the allowed NSSAI of the REGISTRATION ACCEPT message;
- b) put the default S-NSSAIs and not subject to network slice-specific authentication and authorization, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, in the allowed NSSAI of the REGISTRATION ACCEPT message; and
- c) determine a registration area such that all S-NSSAIs of the allowed NSSAI are available in the registration area.

If the REGISTRATION ACCEPT message contains the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed", the UE shall delete the network slicing information for each and every PLMN except for the current PLMN as specified in subclause 4.6.2.2.

If the REGISTRATION ACCEPT message contains the allowed NSSAI, then the UE shall store the included allowed NSSAI together with the PLMN identity of the registered PLMN or the SNPN identity of the registered SNPN and the registration area as specified in subclause 4.6.2.2. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the UE shall store the received allowed NSSAI in each of allowed NSSAIs which are associated with each of the PLMNs.

If the REGISTRATION ACCEPT message contains a configured NSSAI IE with a new configured NSSAI for the current PLMN or SNPN and optionally the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN, the UE shall store the contents of the configured NSSAI IE as specified in subclause 4.6.2.2. In addition, if the REGISTRATION ACCEPT message contains an NSSRG information IE, the UE shall store the contents of the NSSRG information IE as specified in subclause 4.6.2.2.

If the REGISTRATION ACCEPT message:

- a) includes the 5GS registration result IE with the "NSSAA to be performed" indicator set to "Network slice-specific authentication and authorization is to be performed";
- b) includes a pending NSSAI; and
- c) does not include an allowed NSSAI,

the UE shall delete the stored allowed NSSAI, if any, as specified in subclause 4.6.2.2, and the UE:

- a) shall not initiate a 5GSM procedure except for emergency services ; and
- b) shall not initiate a service request procedure except for cases f), i), m) and o) in subclause 5.6.1.1;
- c) shall not initiate an NAS transport procedure except for sending SMS, an LPP message, a location service message, an SOR transparent container, a UE policy container, a UE parameters update transparent container or a CIoT user data container;

until the UE receives an allowed NSSAI.

If the UE included S1 mode supported indication in the REGISTRATION REQUEST message, the AMF supporting interworking with EPS shall set the IWK N26 bit to either:

- a) "interworking without N26 interface not supported" if the AMF supports N26 interface; or
- b) "interworking without N26 interface supported" if the AMF does not support N26 interface

in the 5GS network feature support IE in the REGISTRATION ACCEPT message.

The UE supporting S1 mode shall operate in the mode for interworking with EPS as follows:

- a) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface not supported", the UE shall operate in single-registration mode;
- b) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface supported" and the UE supports dual-registration mode, the UE may operate in dual-registration mode; or

NOTE 17: The registration mode used by the UE is implementation dependent.

c) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface supported" and the UE only supports single-registration mode, the UE shall operate in single-registration mode.

The UE shall treat the received interworking without N26 interface indicator for interworking with EPS as valid in the entire PLMN and its equivalent PLMN(s).

The network informs the UE about the support of specific features, such as IMS voice over PS session, location services (5G-LCS), emergency services, emergency services fallback and ATSSS, in the 5GS network feature support information element. In a UE with IMS voice over PS session capability, the IMS voice over PS session indicator, the Emergency services support indicator, and the Emergency services fallback indicator shall be provided to the upper

layers. The upper layers take the IMS voice over PS session indicator into account when selecting the access domain for voice sessions or calls. In a UE with LCS capability, location services indicator (5G-LCS) shall be provided to the upper layers. When initiating an emergency call, the upper layers also take the IMS voice over PS session indicator, the Emergency services support indicator, and the Emergency services fallback indicator into account for the access domain selection. In a UE with the capability for ATSSS, the network support for ATSSS shall be provided to the upper layers. If the UE receives the 5GS network feature support IE with the ATSSS support indicator set to "ATSSS not supported", the UE shall perform a local release of the MA PDU session, if any.

The AMF shall set the EMF bit in the 5GS network feature support IE to:

- a) "Emergency services fallback supported in NR connected to 5GCN and E-UTRA connected to 5GCN" if the network supports the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN or an E-UTRA cell connected to 5GCN;
- b) "Emergency services fallback supported in NR connected to 5GCN only" if the network supports the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN and does not support the emergency services fallback procedure when the UE is in an E-UTRA cell connected to 5GCN;
- c) "Emergency services fallback supported in E-UTRA connected to 5GCN only" if the network supports the emergency services fallback procedure when the UE is in an E-UTRA cell connected to 5GCN and does not support the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN; or
- d) "Emergency services fallback not supported" if network does not support the emergency services fallback procedure when the UE is in any cell connected to 5GCN.
- NOTE 18: If the emergency services are supported in neither the EPS nor the 5GS homogeneously, based on operator policy, the AMF will set the EMF bit in the 5GS network feature support IE to "Emergency services fallback not supported".
- NOTE 19: Even though the AMF's support of emergency services fallback is indicated per RAT, the UE's support of emergency services fallback is not per RAT, i.e. the UE's support of emergency services fallback is the same for both NR connected to 5GCN and E-UTRA connected to 5GCN.

If the UE is not operating in SNPN access operation mode:

- a) the network informs the UE that the use of access identity 1 is valid in the RPLMN or equivalent PLMN by setting the MPS indicator bit of the 5GS network feature support IE to "Access identity 1 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MPS indicator bit in the REGISTRATION ACCEPT message based on the MPS priority information in the user's subscription context obtained from the UDM;
- b) upon receiving a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 valid", the UE shall act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2, in all NG-RAN of the registered PLMN and its equivalent PLMNs. The MPS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 not valid" or until the UE selects a non-equivalent PLMN. Access identity 1 is only applicable while the UE is in N1 mode.
- c) the network informs the UE that the use of access identity 2 is valid in the RPLMN or equivalent PLMN by setting the MCS indicator bit of the 5GS network feature support IE to "Access identity 2 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MCS indicator bit in the REGISTRATION ACCEPT message based on the MCS priority information in the user's subscription context obtained from the UDM; and
- d) upon receiving a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 valid", the UE shall act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2, in all NG-RAN of the registered PLMN and its equivalent PLMNs. The MCS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 not valid" or until the UE selects a non-equivalent PLMN. Access identity 2 is only applicable while the UE is in N1 mode.
- If the UE is operating in SNPN access operation mode:
 - a) the network informs the UE that the use of access identity 1 is valid in the RSNPN by setting the MPS indicator bit of the 5GS network feature support IE to "Access identity 1 valid", in the REGISTRATION ACCEPT

message. Based on operator policy, the AMF sets the MPS indicator bit in the REGISTRATION ACCEPT message based on the MPS priority information in the user's subscription context obtained from the UDM;

- b) upon receiving a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 valid", the UE shall act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2A, in all NG-RAN of the registered SNPN. The MPS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 not valid" or until the UE selects another SNPN. Access identity 1 is only applicable while the UE is in N1 mode.
- c) the network informs the UE that the use of access identity 2 is valid in the RSNPN by setting the MCS indicator bit of the 5GS network feature support IE to "Access identity 2 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MCS indicator bit in the REGISTRATION ACCEPT message based on the MCS priority information in the user's subscription context obtained from the UDM; and
- d) upon receiving a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 valid", the UE shall act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2A, in all NG-RAN of the registered SNPN. The MCS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 not valid" or until the UE selects another SNPN. Access identity 2 is only applicable while the UE is in N1 mode.

If the UE indicates support for restriction on use of enhanced coverage in the REGISTRATION REQUEST message and:

- a) in WB-N1 mode, the AMF decides to restrict the use of CE mode B for the UE, then the AMF shall set the RestrictEC bit to "CE mode B is restricted";
- b) in WB-N1 mode, the AMF decides to restrict the use of both CE mode A and CE mode B for the UE, then the AMF shall set the RestrictEC bit to " Both CE mode A and CE mode B are restricted"; or
- c) in NB-N1 mode, the AMF decides to restrict the use of enhanced coverage for the UE, then the AMF shall set the RestrictEC bit to "Use of enhanced coverage is restricted",

in the 5GS network feature support IE in the REGISTRATION ACCEPT message.

If the UE indicates support of the N1 NAS signalling connection release in the REGISTRATION REQUEST message and the network decides to accept the N1 NAS signalling connection release, then the AMF shall set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE indicates support of the paging indication for voice services in the REGISTRATION REQUEST message and the network decides to accept the paging indication for voice services, then the AMF shall set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message. Upon receipt of REGISTRATION ACCEPT message with the paging indication for voice services supported", the UE NAS layer informs the lower layers that paging indication for voice services is supported.

If the UE indicates support of the reject paging request in the REGISTRATION REQUEST message and the network decides to accept the reject paging request, then the AMF shall set the reject paging request bit to "reject paging request supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE indicates support of the paging restriction in the REGISTRATION REQUEST message, and the AMF sets:

- the reject paging request bit to "reject paging request supported";
- the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported"; or
- both of them;

in the 5GS network feature support IE of the REGISTRATION ACCEPT message, and the network decides to accept the paging restriction, then the AMF shall set the paging restriction bit to "paging restriction supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE has set the Follow-on request indicator to "Follow-on request pending" in the REGISTRATION REQUEST message, or the network has downlink signalling pending, the AMF shall not immediately release the NAS signalling connection after the completion of the registration procedure.

If the UE is authorized to use V2X communication over PC5 reference point based on:

- a) at least one of the following bits in the 5GMM capability IE of the REGISTRATION REQUEST message set by the UE, or already stored in the 5GMM context in the AMF during the previous registration procedure as follows:
 - 1) the V2XCEPC5 bit to "V2X communication over E-UTRA-PC5 supported"; or
 - 2) the V2XCNPC5 bit to "V2X communication over NR-PC5 supported"; and
- b) the user's subscription context obtained from the UDM as defined in 3GPP TS 23.287 [6C];

the AMF should not immediately release the NAS signalling connection after the completion of the registration procedure.

If the UE is authorized to use 5G ProSe services based on:

- a) at least one of the following bits in the 5GMM capability IE of the REGISTRATION REQUEST message set by the UE, or already stored in the 5GMM context in the AMF during the previous registration procedure as follows:
 - 1) the 5G ProSe direct discovery bit to "5G ProSe direct discovery supported"; or
 - 2) the 5G ProSe direct communication bit to "5G ProSe direct communication supported"; and
- b) the user's subscription context obtained from the UDM as defined in 3GPP TS 23.304 [6E];

the AMF should not immediately release the NAS signalling connection after the completion of the registration procedure.

If the Requested DRX parameters IE was included in the REGISTRATION REQUEST message, the AMF shall include the Negotiated DRX parameters IE in the REGISTRATION ACCEPT message. The AMF may set the Negotiated DRX parameters IE based on the received Requested DRX parameters IE and operator policy if available.

If the Requested NB-N1 mode DRX parameters IE was included in the REGISTRATION REQUEST message, the AMF shall include the Negotiated NB-N1 mode DRX parameters IE in the REGISTRATION ACCEPT message. The AMF may set the Negotiated NB-N1 mode DRX parameters IE based on the received Requested NB-N1 mode DRX parameters IE and operator policy if available.

The AMF shall include the Negotiated extended DRX parameters IE in the REGISTRATION ACCEPT message only if the Requested extended DRX parameters IE was included in the REGISTRATION REQUEST message, and the AMF supports and accepts the use of eDRX. The AMF may set the Negotiated extended DRX parameters IE based on the received Requested extended DRX parameters IE, operator policy, information from NG-RAN and the user's subscription context obtained from the UDM if available.

If:

- a) the UE's USIM is configured with indication that the UE is to receive the SOR transparent container IE, the SOR transparent container IE included in the REGISTRATION ACCEPT message does not successfully pass the integrity check (see 3GPP TS 33.501 [24]); and
- b) if the UE attempts obtaining service on another PLMNs as specified in 3GPP TS 23.122 [5] annex C;

then the UE shall locally release the established N1 NAS signalling connection after sending a REGISTRATION COMPLETE message.

If:

- a) the UE's USIM is configured with indication that the UE is to receive the SOR transparent container IE, the SOR transparent container IE is not included in the REGISTRATION ACCEPT message; and
- b) the UE attempts obtaining service on another PLMNs as specified in 3GPP TS 23.122 [5] annex C;

then the UE shall locally release the established N1 NAS signalling connection.

If:

- a) the UE operates in SNPN access operation mode;
- b) the ME is configured to indicate that the UE shall expect to receive the steering of roaming information during initial registration procedure for the selected entry of the "list of subscriber data" or the selected PLMN subscription;
- c) the SOR transparent container IE included in the REGISTRATION ACCEPT message does not successfully pass the integrity check (see 3GPP TS 33.501 [24]); and
- d) the UE attempts obtaining service on another SNPN as specified in 3GPP TS 23.122 [5] annex C;

then the UE shall locally release the established N1 NAS signalling connection after sending a REGISTRATION COMPLETE message.

If:

- a) the UE operates in SNPN access operation mode;
- b) the ME is configured to indicate that the UE shall expect to receive the steering of roaming information during initial registration procedure for the selected entry of the "list of subscriber data" or the selected PLMN subscription;
- c) the SOR transparent container IE is not included in the REGISTRATION ACCEPT message; and
- d) the UE attempts obtaining service on another SNPN as specified in 3GPP TS 23.122 [5] annex C;

then the UE shall locally release the established N1 NAS signalling connection.

If the REGISTRATION ACCEPT message includes the SOR transparent container IE and the SOR transparent container IE successfully passes the integrity check (see 3GPP TS 33.501 [24]), the ME shall store the received SOR counter as specified in annex C and proceed as follows:

- a) the UE shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C; and
- b) if the registration procedure is performed over 3GPP access and the UE attempts obtaining service on another PLMNs or SNPNs as specified in 3GPP TS 23.122 [5] annex C, then the UE may locally release the established N1 NAS signalling connection after sending a REGISTRATION COMPLETE message. Otherwise the UE shall send a REGISTRATION COMPLETE message and not release the current N1 NAS signalling connection locally. If an acknowledgement is requested in the SOR transparent container IE of the REGISTRATION ACCEPT message, the UE acknowledgement is included in the SOR transparent container IE of the REGISTRATION COMPLETE message. In the SOR transparent container IE carrying the acknowledgement, the UE shall set the ME support of SOR-CMCI indicator to "SOR-CMCI supported by the ME". Additionally, if the UE supports access to an SNPN using credentials from a credentials holder and the UE is not operating in SNPN access operation mode, the UE may set the ME support of SOR-SNPN-SI indicator to "SOR-SNPN-SI supported by the ME".

If the SOR transparent container IE successfully passes the integrity check (see 3GPP TS 33.501 [24]) and:

- a) the list type indicates:
 - "PLMN ID and access technology list", and the SOR transparent container IE indicates a list of preferred PLMN/access technology combinations is provided, then the ME shall replace the highest priority entries in the "Operator Controlled PLMN Selector with Access Technology" list stored in the ME and shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C; or
 - 2) "secured packet", then the ME shall behave as if a SMS is received with protocol identifier set to SIM data download, data coding scheme set to class 2 message and SMS payload as secured packet contents of SOR transparent container IE. The SMS payload is forwarded to UICC as specified in 3GPP TS 23.040 [4A] and the ME shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C; or
- b) the list type indicates "PLMN ID and access technology list" and the SOR transparent container IE indicates "HPLMN indication that 'no change of the "Operator Controlled PLMN Selector with Access Technology" list

stored in the UE is needed and thus no list of preferred PLMN/access technology combinations is provided", the UE operates in SNPN access operation mode and the SOR transparent container IE includes SOR-SNPN-SI, the ME shall replace SOR-SNPN-SI of the selected entry of the "list of subscriber data" or associated with the selected PLMN subscription, as specified in 3GPP TS 23.122 [5] with the received SOR-SNPN-SI.

If the SOR-CMCI is present and the Store SOR-CMCI in ME indicator is set to "Store SOR-CMCI in ME" then the UE shall store or delete the SOR-CMCI in the non-volatile memory of the ME as described in annex C.1.

The UE shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C.

If the SOR transparent container IE does not pass the integrity check successfully, then the UE shall discard the content of the SOR transparent container IE.

If required by operator policy, the AMF shall include the NSSAI inclusion mode IE in the REGISTRATION ACCEPT message (see table 4.6.2.3.1 of subclause 4.6.2.3). Upon receipt of the REGISTRATION ACCEPT message:

- a) if the message includes the NSSAI inclusion mode IE, the UE shall operate in the NSSAI inclusion mode indicated in the NSSAI inclusion mode IE over the current access within the current PLMN or SNPN and its equivalent PLMN(s), if any, in the current registration area; or
- b) otherwise:
 - 1) if the UE has NSSAI inclusion mode for the current PLMN or SNPN and access type stored in the UE, the UE shall operate in the stored NSSAI inclusion mode;
 - 2) if the UE does not have NSSAI inclusion mode for the current PLMN or SNPN and the access type stored in the UE and if the UE is performing the registration procedure over:
 - i) 3GPP access, the UE shall operate in NSSAI inclusion mode D in the current PLMN or SNPN and the current access type;
 - ii) untrusted non-3GPP access, the UE shall operate in NSSAI inclusion mode B in the current PLMN and the current access type; or
 - iii) trusted non-3GPP access, the UE shall operate in NSSAI inclusion mode D in the current PLMN and the current access type; or
 - 3) if the 5G-RG does not have NSSAI inclusion mode for the current PLMN and wireline access stored in the 5G-RG, and the 5G-RG is performing the registration procedure over wireline access, the 5G-RG shall operate in NSSAI inclusion mode B in the current PLMN and the current access type.

The AMF may include operator-defined access category definitions in the REGISTRATION ACCEPT message.

If the UE receives Operator-defined access category definitions IE in the REGISTRATION ACCEPT message and the Operator-defined access category definitions IE contains one or more operator-defined access category definitions, the UE shall delete any operator-defined access category definitions stored for the RPLMN and shall store the received operator-defined access category definitions for the RPLMN. If the UE receives the Operator-defined access category definitions IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions, the UE shall delete any operator-defined access category definitions IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions stored for the RPLMN. If the REGISTRATION ACCEPT message does not contain the Operator-defined access category definitions access category definitions access category definitions access category definitions access category definitions. The UE shall delete any operator-defined access category definitions access category

If the UE has indicated support for service gap control in the REGISTRATION REQUEST message and:

- the REGISTRATION ACCEPT message contains the T3447 value IE, then the UE shall store the new T3447 value, erase any previous stored T3447 value if exists and use the new T3447 value with the timer T3447 next time it is started; or
- the REGISTRATION ACCEPT message does not contain the T3447 value IE, then the UE shall erase any previous stored T3447 value if exists and stop the timer T3447 if running.

If the T3448 value IE is present in the received REGISTRATION ACCEPT message and the value indicates that this timer is neither zero nor deactivated, the UE shall:

a) stop timer T3448 if it is running; and

b) start timer T3448 with the value provided in the T3448 value IE.

If the UE is using 5GS services with control plane CIoT 5GS optimization, the T3448 value IE is present in the REGISTRATION ACCEPT message and the value indicates that this timer is either zero or deactivated, the UE shall ignore the T3448 value IE and proceed as if the T3448 value IE was not present.

If the REGISTRATION ACCEPT message contains the Truncated 5G-S-TMSI configuration IE, then the UE shall store the included truncated 5G-S-TMSI configuration and return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the truncated 5G-S-TMSI configuration.

NOTE 20: The UE provides the truncated 5G-S-TMSI configuration to the lower layers.

If the UE is not in NB-N1 mode, the UE has set the RACS bit to "RACS supported" in the 5GMM Capability IE of the REGISTRATION REQUEST message and the REGISTRATION ACCEPT message includes:

- a) a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested", the UE shall delete any network-assigned UE radio capability IDs associated with the RPLMN or RSNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription stored at the UE, then the UE shall, after the completion of the ongoing registration procedure, initiate a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3.2 over the existing N1 NAS signalling connection; or
- b) a UE radio capability ID IE, the UE shall store the UE radio capability ID as specified in annex C.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and the REGISTRATION ACCEPT message contains the service-level-AA pending indication in the Service-level-AA container IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the service-level-AA pending indication, and the UE shall not attempt to perform another registration procedure for UAS services until the UUAA-MM procedure is completed, or to establish a PDU session for USS communication or a PDU session for C2 communication until the UUAA-MM procedure is completed successfully.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and the REGISTRATION ACCEPT message does not contain the service-level-AA pending indication in the Service-level-AA container IE, the UE shall consider the UUAA-MM procedure is not triggered.

If the REGISTRATION REQUEST message includes the 5GS registration type IE set to "SNPN onboarding registration" or the network determines that the UE's subscription only allows for configuration of SNPN subscription parameters in PLMN via the user plane, the AMF may start an implementation specific timer for onboarding services when the network considers that the UE is in 5GMM-REGISTERED (i.e. the network receives the REGISTRATION COMPLETE message from UE).

- NOTE 21:If the AMF considers that the UE is in 5GMM-IDLE, when the implementation specific timer for onboarding services expires and the network considers that the UE is still in state 5GMM-REGISTERED, the AMF can locally de-register the UE; or if the UE is in 5GMM-CONNECTED, the AMF can initiate the network-initiated de-registration procedure (see subclause 5.5.2.3).
- NOTE 22: The value of the implementation specific timer for onboarding services needs to be large enough to allow a UE to complete the configuration of one or more entries of the "list of subscriber data" taking into consideration that configuration of SNPN subscription parameters in PLMN via the user plane or onboarding services in SNPN involves third party entities outside of the operator's network.

If the UE receives the List of PLMNs to be used in disaster condition IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the "list of PLMN(s) to be used in disaster condition" stored in the ME together with the PLMN ID of the RPLMN, if any, and may store the "list of PLMN(s) to be used in disaster condition" included in the List of PLMNs to be used in disaster condition IE in the ME together with the PLMN ID of the RPLMN.

If the UE receives the Disaster roaming wait range IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the disaster roaming wait range stored in the ME, if any, and store the disaster roaming wait range included in the Disaster roaming wait range IE in the ME.

If the UE receives the Disaster return wait range IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range included in the Disaster return wait range IE in the ME.

If the 5GS registration type IE in the REGISTRATION REQUEST message is set to "disaster roaming initial registration" and:

- a) the MS determined PLMN with disaster condition IE is included in the REGISTRATION REQUEST message, the AMF shall determine the PLMN with disaster condition in the MS determined PLMN with disaster condition IE;
- b) the MS determined PLMN with disaster condition IE is not included in the REGISTRATION REQUEST message and the Additional GUTI IE is included in the REGISTRATION REQUEST message and contains 5G-GUTI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the 5G-GUTI;
- c) the MS determined PLMN with disaster condition IE and the Additional GUTI IE are not included in the REGISTRATION REQUEST message and:
 - the 5GS mobile identity IE contains 5G-GUTI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the 5G-GUTI; or
 - the 5GS mobile identity IE contains SUCI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the SUCI; or
- d) the MS determined PLMN with disaster condition IE is not included in the REGISTRATION REQUEST message, NG-RAN of the PLMN providing disaster roaming broadcasts disaster roaming indication and:
 - the Additional GUTI IE is included in the REGISTRATION REQUEST message and contains 5G-GUTI of a PLMN of a country other than the country of the PLMN providing disaster roaming services; or
 - the Additional GUTI IE is not included and the 5GS mobile identity IE contains 5G-GUTI or SUCI of a PLMN of a country other than the country of the PLMN providing disaster roaming services;

the AMF shall determine the PLMN with disaster condition based on the disaster roaming agreement arrangement between mobile network operators.

NOTE 23: The disaster roaming agreement arrangement between mobile network operators is out scope of 3GPP.

If the AMF determines that a disaster condition applies to the PLMN with disaster condition, and the UE is allowed to be registered for disaster roaming services, the AMF shall set the Disaster roaming registration result value bit in the 5GS registration result IE to "no additional information" in the REGISTRATION ACCEPT message. If the AMF determines that the UE can be registered to the PLMN for normal service, the AMF shall set the Disaster roaming registration result value bit in the 5GS registration result IE to "request for registration for disaster roaming services accepted as registration not for disaster roaming services " in the REGISTRATION ACCEPT message.

If the UE indicates "disaster roaming initial registration" in the 5GS registration type IE in the REGISTRATION REQUEST message and the 5GS registration result IE value in the REGISTRATION ACCEPT message is set to:

- "request for registration for disaster roaming service accepted as registration not for disaster roaming services", the UE shall consider itself registered for normal service. If the PLMN identity of the registered PLMN is a member of the forbidden PLMN list as specified in subclause 5.3.13A, any such PLMN identity shall be deleted from the corresponding list(s); or
- "no additional information", the UE shall consider itself registered for disaster roaming services.

If the UE receives the forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the REGISTRATION ACCEPT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for roaming".

If the UE receives the forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the REGISTRATION ACCEPT message and if the TAI(s) included in the IE is not part of the list of "5GS

forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for regional provision of service".

5.5.1.2.5 Initial registration not accepted by the network

If the initial registration request cannot be accepted by the network, the AMF shall send a REGISTRATION REJECT message to the UE including an appropriate 5GMM cause value.

If the initial registration request is rejected due to general NAS level mobility management congestion control, the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

In NB-N1 mode, if the registration request is rejected due to operator determined barring (see 3GPP TS 29.503 [20AB]), the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

If the REGISTRATION REJECT message with 5GMM cause #76 or #78 was received without integrity protection, then the UE shall discard the message. If the REGISTRATION REJECT message with 5GMM cause #62 was received without integrity protected, the behaviour of the UE is specified in subclause 5.3.20.2.

Based on operator policy, if the initial registration request is rejected due to core network redirection for CIoT optimizations, the network shall set the 5GMM cause value to #31 "Redirection to EPC required".

NOTE 1: The network can take into account the UE's S1 mode capability, the EPS CIoT network behaviour supported by the UE or the EPS CIoT network behaviour supported by the EPC to determine the rejection with the 5GMM cause value #31 "Redirection to EPC required".

If the initial registration request is rejected because:

- a) all the S-NSSAI(s) included in the requested NSSAI are rejected; and
- b) the UE set the NSSAA bit in the 5GMM capability IE to:
 - 1) "Network slice-specific authentication and authorization supported" and:
 - i) there are no default S-NSSAIs;
 - ii) all default S-NSSAIs are not allowed; or
 - iii) network slice-specific authentication and authorization has failed or been revoked for all default S-NSSAIs and based on network local policy, the network decides not to initiate the network slice-specific re-authentication and re-authorization procedures for any subscribed S-NSSAI marked as default requested by the UE; or
 - 2) "Network slice-specific authentication and authorization not supported"; and
 - i) there are no default S-NSSAIs; or
 - ii) all default S-NSSAIs are either not allowed or are subject to network slice-specific authentication and authorization;

the network shall set the 5GMM cause value to #62 "No network slices available" and shall include the rejected S-NSSAI(s) in the rejected NSSAI of the REGISTRATION REJECT message. Otherwise, the network may include the rejected S-NSSAI(s) in the rejected NSSAI of the REGISTRATION REJECT message.

If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected S-NSSAI(s) shall be included in the Extended rejected NSSAI IE of the REGISTRATION REJECT message. Otherwise the rejected S-NSSAI(s) shall be included in the Rejected NSSAI IE of the REGISTRATION REJECT message.

If the UE supports extended rejected NSSAI and the AMF determines that maximum number of UEs reached for one or more S-NSSAIs in the requested NSSAI as specified in subclause 4.6.2.5, the AMF shall include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE in the REGISTRATION REJECT message. In addition, the AMF may include a back-off timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI iE of the REGISTRATION REJECT message.

If the AMF receives the initial registration request along with the authenticated indication over N2 reference point on non-3GPP access and does not receive the indication that authentication by the home network is not required over N12 reference point, the network shall set the 5GMM cause value to #72 "Non-3GPP access to 5GCN not allowed".

If the initial registration request from a UE supporting CAG is rejected due to CAG restrictions, the network shall set the 5GMM cause value to #76 "Not authorized for this CAG or authorized for CAG cells only" and should include the "CAG information list" in the CAG information list IE or the Extended CAG information list IE in the REGISTRATION REJECT message.

- NOTE 2: The network cannot be certain that "CAG information list" stored in the UE is updated as result of sending of the REGISTRATION REJECT message with the CAG information list IE or the Extended CAG information list IE, as the REGISTRATION REJECT message is not necessarily delivered to the UE (e.g. due to abnormal radio conditions).
- NOTE 3: The "CAG information list" can be provided by the AMF and include no entry if no "CAG information list" exists in the subscription.
- NOTE 4: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If the initial registration request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall operate as described in bullet j) of subclause 5.5.1.2.8.

If the UE's initial registration request is via a satellite NG-RAN cell and the network using the User Location Information provided by the NG-RAN, see 3GPP TS 38.413 [31], is able to determine that the UE is in a location where the network is not allowed to operate, the network shall set the 5GMM cause value in the REGISTRATION REJECT message to #78 "PLMN not allowed to operate at the present UE location".

NOTE 5: When the UE is accessing network for emergency services, it is up to operator and regulatory policies whether the network needs to determine if the UE is in a location where network is not allowed to operate.

If the AMF receives the initial registration request including the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE and the AMF determines that the UE is not allowed to use UAS services via 5GS based on the user's subscription data and the operator policy, the AMF shall return a REGISTRATION REJECT message with 5GMM cause #79 (UAS services not allowed).

If the UE initiates the registration procedure for disaster roaming services and the AMF determines that it does not support providing disaster roaming services for the determined PLMN with disaster condition to the UE, then the AMF shall send a REGISTRATION REJECT message with 5GMM cause #80 (Disaster roaming for the determined PLMN with disaster condition not allowed).

If the AMF received multiple TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, all of the received TAIs are forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

- a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or
- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or
- c) both;

in the REGISTRATION REJECT message.

Regardless of the 5GMM cause value received in the REGISTRATION REJECT message via satellite NG-RAN,

- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the REGISTRATION REJECT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for roaming"; and
- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the REGISTRATION REJECT message and if the TAI(s) included in the IE is not part of the list

of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for regional provision of service".

Furthermore, the UE shall take the following actions depending on the 5GMM cause value received in the REGISTRATION REJECT message.

#3 (Illegal UE); or

#6 (Illegal ME).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure was performed in the current SNPN, the UE shall consider the USIM as invalid for the current SNPN until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not performing initial registration for onboarding services in SNPN and the UE shall delete the list of equivalent PLMNs (if any) and enter the state 5GMM-DEREGISTERED.NO-SUPI. If the message has been successfully integrity checked by the NAS, then the UE shall:

- set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN if the UE maintains these counters; or
- set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN if the UE maintains these counters;

to a UE implementation-specific maximum value.

3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value. The USIM shall be considered as invalid also for non-EPS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.7a in 3GPP TS 24.301 [15]. If the message has been successfully integrity checked by the NAS and the UE maintains a counter for "SIM/USIM considered invalid for non-GPRS services", then the UE shall set this counter to a UE implementation-specific maximum value.

If the UE is performing initial registration for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#7 (5GS services not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not performing initial registration for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 5GS services until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not performing initial registration for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure was performed in the current SNPN, the UE shall consider the USIM as invalid for the current SNPN until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not performing initial registration for onboarding services in SNPN, the UE shall enter the state 5GMM-DEREGISTERED.NO-SUPI. If the message has been successfully integrity checked by the NAS, then the UE shall:

- set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events in case of PLMN if the UE maintains these counters; or
- set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events in case of SNPN if the UE maintains these counters;

to a UE implementation-specific maximum value.

3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

If the UE is performing initial registration for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#11 (PLMN not allowed).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs and reset the registration attempt counter and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. For 3GPP access the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE and perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been

successfully integrity checked by the NAS and the UE maintains the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same PLMN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#12 (Tracking area not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

#13 (Roaming not allowed in this tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall delete the list of equivalent PLMNs (if available) and reset the registration attempt counter.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE or optionally 5GMM-DEREGISTERED.PLMN-SEARCH. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE or optionally 5GMM-DEREGISTERED.PLMN-SEARCH. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN using credentials from a credentials holder, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder,

the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

For 3GPP access, if the UE is registered in S1 mode and operating in dual-registration mode, the PLMN that the UE chooses to register in is specified in subclause 4.8.3. Otherwise the UE shall perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5].

For non-3GPP access, the UE shall perform network selection as defined in 3GPP TS 24.502 [18].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

#15 (No suitable cells in tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

The UE shall search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

If received over non-3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.2.7.

#22 (Congestion).

If the T3346 value IE is present in the REGISTRATION REJECT message and the value indicates that this timer is neither zero nor deactivated, the UE shall proceed as described below; otherwise it shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.2.7.

The UE shall abort the initial registration procedure, set the 5GS update status to 5U2 NOT UPDATED, reset the registration attempt counter and enter state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION.

The UE shall stop timer T3346 if it is running.

If the REGISTRATION REJECT message is integrity protected, the UE shall start timer T3346 with the value provided in the T3346 value IE.

If the REGISTRATION REJECT message is not integrity protected, the UE shall start timer T3346 with a random value from the default range specified in 3GPP TS 24.008 [12].

The UE stays in the current serving cell and applies the normal cell reselection process. The initial registration procedure is started if still needed when timer T3346 expires or is stopped.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value.

If the UE is registering for onboarding services in SNPN, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

#27 (N1 mode not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter and shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the message has been successfully integrity checked by the NAS, the UE shall set:

- 1) the PLMN-specific N1 mode attempt counter for 3GPP access and the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN; or
- 2) the SNPN-specific attempt counter for 3GPP access for the current SNPN in case of SNPN and the SNPN-specific attempt counter for non-3GPP access for the current SNPN;

to the UE implementation-specific maximum value.

The UE shall disable the N1 mode capability for the specific access type for which the message was received (see subclause 4.9).

If the message has been successfully integrity checked by the NAS, the UE shall disable the N1 mode capability also for the other access type (see subclause 4.9).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI. Additionally, the UE shall reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#31 (Redirection to EPC required).

5GMM cause #31 received by a UE that has not indicated support for CIoT optimizations or not indicated support for S1 mode or received by a UE over non-3GPP access is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter.

The UE shall enable the E-UTRA capability if it was disabled, disable the N1 mode capability for 3GPP access (see subclause 4.9.2) and enter the 5GMM-DEREGISTERED.NO-CELL-AVAILABLE.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach procedure is rejected with the EMM cause with the same value.

#36 (IAB-node operation not authorized).

This cause value is only applicable when received over 3GPP access by a UE operating as an IAB-node. This cause value received from a 5G access network other than 3GPP access or received by a UE not operating as an IAB-node is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

If:

- 1) the UE is not operating in SNPN access operation mode,
 - i) the UE shall delete the list of equivalent PLMNs and reset the registration attempt counter and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS and the UE maintains the PLMN-specific attempt counter for 3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter for 3GPP access for that PLMN to the UE implementation-specific maximum value; and
 - ii) If the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value; or
- 2) the UE is operating in SNPN access operation mode,
 - i) the UE shall reset the registration attempt counter and store the SNPN identity in the "temporarily forbidden SNPNs" list for 3GPP access and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN attempt counter for 3GPP access for the current SNPN to the UE implementation-specific maximum value.
- #62 (No network slices available).

The UE shall abort the initial registration procedure, set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-DEREGISTERED. ATTEMPTING-REGISTRATION or 5GMM-DEREGISTERED.PLMN-SEARCH. Additionally, the UE shall reset the registration attempt counter.

The UE receiving the rejected NSSAI in the REGISTRATION REJECT message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as described in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

Unless the back-off timer value received along with the S-NSSAI is zero, the UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

NOTE 6: If the back-off timer value received along with the S-NSSAI in the rejected NSSAI for the maximum number of UEs reached is zero as specified in subclause 10.5.7.4a of TS 24.008, the UE does not consider the S-NSSAI as the rejected S-NSSAI.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if a back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or
 - 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

If the UE has an allowed NSSAI or configured NSSAI that contains S-NSSAI(s) which are not included in the rejected NSSAI the UE may stay in the current serving cell, apply the normal cell reselection process and start an initial registration with a requested NSSAI that includes any S-NSSAI from the allowed NSSAI or the configured NSSAI that is not in the rejected NSSAI. Otherwise the UE may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if the UE does not have an allowed NSSAI and each S-NSSAI in configured NSSAI, if available, was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and,

- 1) if at least one S-NSSAI in the default configured NSSAI is not rejected, the UE may stay in the current serving cell, apply the normal cell reselection process, and start an initial registration with a requested NSSAI with that default configured NSSAI; or
- 2) if all the S-NSSAI(s) in the default configured NSSAI are rejected and at least one S-NSSAI is rejected due to "S-NSSAI not available in the current registration area",
 - i) if the REGISTRATION REJECT message is integrity protected and the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; or
 - ii) if the REGISTRATION REJECT message is integrity protected and the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE.

Otherwise, the UE may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if each S-NSSAI in the default configured NSSAI was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and has rejected NSSAI for the reached maximum number of UEs, and the UE wants to obtain services in the current serving cell without performing a PLMN selection or SNPN selection, the UE may stay in

the current serving cell and attempt to use the rejected S-NSSAI(s) for the maximum number of UEs reached in the current serving cell after the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#72 (Non-3GPP access to 5GCN not allowed).

When received over non-3GPP access the UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter and enter the state 5GMM-DEREGISTERED. If the message has been successfully integrity checked by the NAS, the UE shall set:

1) the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN: or

2) the SNPN-specific attempt counter for non-3GPP access for that SNPN in case of SNPN;

to the UE implementation-specific maximum value.

NOTE 7: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).

The UE shall disable the N1 mode capability for non-3GPP access (see subclause 4.9.3).

As an implementation option, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5].

If received over 3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.2.7.

#73 (Serving network not authorized).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs, reset the registration attempt counter, store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A. For 3GPP access the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE and perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been successfully integrity checked by the NAS, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI. Additionally, the UE shall reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#74 (Temporarily not authorized for this SNPN).

5GMM cause #74 is only applicable when received from a cell belonging to an SNPN. 5GMM cause #74 received from a cell not belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and store the SNPN identity in the "temporarily forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the registration request is not for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the registration request is for onboarding services in SNPN, the UE shall enter state 5GMM-

DEREGISTERED.PLMN-SEARCH and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access and the SNPN-specific attempt counter for non-3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 8: When 5GMM cause #74 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 9: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #75 (Permanently not authorized for this SNPN).

5GMM cause #75 is only applicable when received from a cell belonging to an SNPN with a globally-unique SNPN identity. 5GMM cause #75 received from a cell not belonging to an SNPN or a cell belonging to an SNPN with a non-globally-unique SNPN identity is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and store the SNPN identity in the "permanently forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the registration request is not for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the registration request is for onboarding services in SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access and the SNPN-specific attempt counter for non-3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 10: When 5GMM cause #75 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 11: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #76 (Not authorized for this CAG or authorized for CAG cells only).

This cause value received via non-3GPP access or from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED, store the 5GS update status according to clause 5.1.3.2.2, and reset the registration attempt counter.

If 5GMM cause #76 is received from:

- 1) a CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the REGISTRATION REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when

the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or

- NOTE 12: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, then the UE shall delete the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN. In the case the "allowed CAG list" for the current PLMN only contains a range of CAG-IDs, how the UE deletes the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN is up to UE implementation. In addition:

- i) if the entry in the "CAG information list" for the current PLMN does not include an "indication that the UE is only allowed to access 5GS via CAG cells" or if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list";
- ii) if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
- iii) if the "CAG information list" does not include an entry for the current PLMN, then the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".
- 2) a non-CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the REGISTRATION REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 13: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN, if any. If the "CAG information list" stored in the UE does not include the current PLMN's entry, the UE shall add an entry for the current PLMN to the "CAG information list" and store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN. If the UE does not have a stored "CAG information list", the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells".

In addition:

- i) if the "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated CAG information; or
- ii) if the "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#77 (Wireline access area not allowed).

5GMM cause #77 is only applicable when received from a wireline access network by the 5G-RG or the W-AGF acting on behalf of the FN-CRG. 5GMM cause #77 received from a 5G access network other than a wireline access network and 5GMM cause #77 received by the W-AGF acting on behalf of the FN-BRG are considered as abnormal cases and the behaviour of the UE is specified in subclause 5.5.1.2.7.

When received over wireline access network, the 5G-RG and the W-AGF acting on behalf of the FN-CRG shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2), shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI, shall reset the registration attempt counter, shall enter the state 5GMM-DEREGISTERED and shall act as specified in subclause 5.3.23.

- NOTE 14: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).
- #78 (PLMN not allowed to operate at the present UE location).

This cause value received from a non-satellite NG-RAN cell is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.2.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter. The UE shall store the PLMN identity and, if it is known, the current geographical location in the list of "PLMNs not allowed to operate at the present UE location" and shall start a corresponding timer instance (see subclause 4.23.2). The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach procedure is rejected with the EMM cause with the same value.

#79 (UAS services not allowed).

The UE shall abort the initial registration procedure, set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-DEREGISTERED. ATTEMPTING-REGISTRATION or 5GMM-DEREGISTERED.PLMN-SEARCH. Additionally, the UE shall reset the registration attempt counter. If the UE re-attempt the registration procedure to the current PLMN, the UE shall not include the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of REGISTRATION REQUEST message unless the UE receives a CONFIGURATION UPDATE COMMAND message including the service-level-AA service status indication in the Service-level-AA container IE with the UAS field set to "UAS services enabled".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#80 (Disaster roaming for the determined PLMN with disaster condition not allowed).

The UE shall abort the initial registration procedure, set the 5GS update status to 5U2 NOT UPDATED, enter state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter. The UE shall not attempt to register for disaster roaming services on this PLMN for the determined PLMN with disaster

condition for a period in the range of 12 to 24 hours. The UE shall not attempt to register for disaster roaming services on this PLMN for a period in the range of 3 to 10 minutes. The UE shall perform PLMN selection as described in 3GPP TS 23.122 [6].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

Other values are considered as abnormal cases. The behaviour of the UE in those cases is specified in subclause 5.5.1.2.7.

5.5.1.2.6 Initial registration for emergency services not accepted by the network

Upon receiving the REGISTRATION REJECT message including 5GMM cause #5 "PEI not accepted", the UE shall enter the state 5GMM-DEREGISTERED.NO-SUPI. If the REGISTRATION REJECT message is received,

- over 3GPP access; or
- over non-3GPP access and is integrity protected;

and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

Upon receiving the REGISTRATION REJECT message including 5GMM cause value which is not #5 "PEI not accepted", the UE shall perform the actions as described in subclause 5.5.1.2.5 with the following addition: the UE shall inform the upper layers of the failure of the procedure.

NOTE 1: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

If the initial registration request for emergency services fails due to abnormal cases, the UE shall perform the actions as described in subclause 5.5.1.2.7 and inform the upper layers of the failure to access the network or the failure of the procedure.

NOTE 2: This can result in the upper layers requesting other implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

In a shared network, upon receiving the REGISTRATION REJECT message, the UE shall perform the actions as described in subclause 5.5.1.2.5, and shall:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 3: The upper layers can request implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] that can result in the emergency call being attempted to another IP-CAN.
- b) attempt to perform a PLMN selection in the shared network and, if an initial registration for emergency services was not already attempted with the selected PLMN and the REGISTRATION REQUEST message, for which the REGISTRATION REJECT message was received, is:
 - not for sending a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing emergency PDU session", initiate an initial registration for emergency services with the selected PLMN; or
 - for sending a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing emergency PDU session", and:
 - i) the selected PLMN is an equivalent PLMN, initiate an initial registration for emergency services with the selected PLMN; and
 - ii) the selected PLMN is not an equivalent PLMN, perform a PLMN selection and initiate an initial registration for emergency services with the selected PLMN if an initial registration for emergency services was not already attempted with the selected PLMN.

In a shared network, if the initial registration request for emergency services fails due to abnormal cases, the UE shall perform the actions as described in subclause 5.5.1.2.7 and shall:

a) inform the upper layers of the failure of the procedure; or

- NOTE 4: The upper layers can request implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] that can result in the emergency call being attempted to another IP-CAN.
- b) attempt to perform a PLMN selection in the shared network and, if an initial registration for emergency services was not already attempted with the selected PLMN and the REGISTRATION REQUEST message is:
 - not for sending a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing emergency PDU session", initiate an initial registration for emergency services with the selected PLMN; or
 - for sending a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing emergency PDU session", and:
 - i) the selected PLMN is an equivalent PLMN, initiate an initial registration for emergency services with the selected PLMN; and
 - ii) the selected PLMN is not an equivalent PLMN, perform a PLMN selection and initiate an initial registration for emergency services with the selected PLMN if an initial registration for emergency services was not already attempted with the selected PLMN.

5.5.1.2.6A Initial registration for initiating an emergency PDU session not accepted by the network

If the network cannot accept an initial registration request with 5GS registration type IE set to "initial registration" and for sending a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "initial emergency request", the UE shall perform the procedures as described in subclause 5.5.1.2.5. Then if the UE is in the same selected PLMN where the last initial registration request was attempted, the UE shall:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 1: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.
- b) attempt initial registration for emergency services.

If the network cannot accept an initial registration request with 5GS registration type IE set to "initial registration", for establishing an emergency PDU session and the PDU session needs to be established due to handover of an existing PDN connection for emergency bearer services, the UE shall perform the procedures as described in subclause 5.5.1.2.5. Then if the UE is in the same selected PLMN or equivalent PLMN where the last initial registration request was attempted, the UE shall attempt initial registration for emergency services.

If the initial registration request, with 5GS registration type IE set to "initial registration" and for initiating an emergency PDU session, fails due to abnormal case b) in subclause 5.5.1.2.7, the UE shall perform the actions as described in subclause 5.5.1.2.7 and inform the upper layers of the failure to access the network.

NOTE 2: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

If the initial registration request, with 5GS registration type IE set to "initial registration" and for initiating an emergency PDU session, fails due to abnormal cases c), d) or e) in subclause 5.5.1.2.7, the UE shall perform the actions as described in subclause 5.5.1.2.7. Then if the UE is in:

- a) the same selected PLMN where the last initial registration request was attempted and the PDU session does not need to be established due to handover of an existing PDN connection for emergency bearer services, the UE shall:
 - inform the upper layers of the failure of the procedure; or
- NOTE 3: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.
 - attempt initial registration for emergency services; or
- b) the same selected PLMN or equivalent PLMN where the last initial registration request was attempted and the PDU session needs to be established due to handover of an existing PDN connection for emergency bearer services, attempt initial registration for emergency services.
5.5.1.2.7 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Timer T3346 is running.

The UE shall not start the registration procedure for initial registration unless:

- 1) the UE is a UE configured for high priority access in selected PLMN;
- 2) the UE needs to perform the registration procedure for initial registration for emergency services;
- 3) the UE receives a DEREGISTRATION REQUEST message with the "re-registration required" indication;
- 4) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and:
 - the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]); and
 - timer T3346 was not started when N1 NAS signalling connection was established with RRC establishment cause set to "mo-ExceptionData"; or
- 5) the UE needs to perform the registration procedure with 5GS registration type IE set to "initial registration" for initiating of an emergency PDU session, upon request of the upper layers to establish the emergency PDU session.

The UE stays in the current serving cell and applies the normal cell reselection process.

- NOTE 1: It is considered an abnormal case if the UE needs to initiate a registration procedure for initial registration while timer T3346 is running independent on whether timer T3346 was started due to an abnormal case or a non-successful case.
- b) The lower layers indicate that the access attempt is barred.

The UE shall not start the initial registration procedure. The UE stays in the current serving cell and applies the normal cell reselection process. Receipt of the access barred indication shall not trigger the selection of a different core network type (EPC or 5GCN).

The initial registration procedure is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

ba) The lower layers indicate that:

- 1) access barring is applicable for all access categories except categories 0 and 2 and the access category with which the access attempt was associated is other than 0 and 2; or
- 2) access barring is applicable for all access categories except category 0 and the access category with which the access attempt was associated is other than 0.

If the REGISTRATION REQUEST message has not been sent, the UE shall proceed as specified for case b. If the REGISTRATION REQUEST message has been sent, the UE shall proceed as specified for case e and, additionally, the registration procedure for initial registration is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

c) T3510 timeout.

The UE shall abort the registration procedure for initial registration and the NAS signalling connection, if any, shall be released locally if the initial registration request is neither for emergency services nor for initiating a PDU session for emergency services with request type set to "existing emergency PDU session". The UE shall proceed as described below.

d) REGISTRATION REJECT message, other 5GMM cause values than those treated in subclause 5.5.1.2.5, and cases of 5GMM cause values #11, #15, #22, #31, #72, #73, #74, #75, #76, #77 and #78, if considered as abnormal cases according to subclause 5.5.1.2.5.

If the registration request is neither an initial registration request for emergency services nor an initial registration request for initiating a PDU session for emergency services with request type set to "existing emergency PDU session", upon reception of the 5GMM causes #95, #96, #97, #99 and #111 the UE should set the registration attempt counter to 5.

The UE shall proceed as described below.

e) Lower layer failure or release of the NAS signalling connection received from lower layers before the REGISTRATION ACCEPT or REGISTRATION REJECT message is received.

The UE shall abort the registration procedure for initial registration and proceed as described below.

f) UE initiated de-registration required.

The registration procedure for initial registration shall be aborted, and the UE initiated de-registration procedure shall be performed.

g) De-registration procedure collision.

If the UE receives a DEREGISTRATION REQUEST message from the network in state 5GMM-REGISTERED-INITIATED the de-registration procedure shall be aborted and the initial registration procedure shall be progressed.

- NOTE 2: The above collision case is valid if the DEREGISTRATION REQUEST message indicates the access type over which the initial registration procedure is attempted otherwise both the procedures are progressed.
- h) Change in the current TAI.

If the current TAI is changed before the registration procedure for initial registration is completed, the registration procedure for initial registration shall be aborted and re-initiated immediately.

If the REGISTRATION COMPLETE message needs to be sent and a tracking area border is crossed when the REGISTRATION ACCEPT message has been received but before a REGISTRATION COMPLETE message is sent and:

- 1) if the current TAI is in the TAI list, the UE sends the REGISTRATION COMPLETE message to the network; and
- 2) otherwise, the registration procedure for initial registration shall be aborted and the registration procedure for mobility registration update shall be initiated.

If a 5G-GUTI was allocated during the registration procedure, this 5G-GUTI shall be used in the registration procedure.

- i) Transmission failure of REGISTRATION COMPLETE message indication with change in the current TAI.
 - 1) If the current TAI is still part of the TAI list, the UE resends the REGISTRATION COMPLETE message to the network; and
 - 2) otherwise, the registration procedure for initial registration shall be aborted and the registration procedure for mobility registration update shall be initiated.
- j) Transmission failure of REGISTRATION COMPLETE message indication without change in the current TAI from lower layers.

It is up to the UE implementation how to re-run the ongoing procedure.

k) Transmission failure of REGISTRATION REQUEST message indication from the lower layers.

The registration procedure for initial registration shall be aborted and re-initiated immediately.

1) Timer T3447 is running.

The UE shall not start the registration procedure for initial registration with Follow-on request indicator set to "Follow-on request pending" unless:

- 1) the UE is a UE configured for high priority access in selected PLMN; or
- 2) the UE needs to perform the registration procedure for initial registration for emergency services.

The UE stays in the current serving cell and applies the normal cell reselection process. The registration procedure for initial registration is started, if still necessary, when timer T3447 expires or timer T3447 is stopped.

For the cases c, d and e, the UE shall proceed as follows:

Timer T3510 shall be stopped if still running.

If the registration procedure is neither an initial registration for emergency services nor for establishing an emergency PDU session with registration type not set to "emergency registration", the registration attempt counter shall be incremented, unless it was already set to 5.

If the registration attempt counter is less than 5:

 if the initial registration request is not for emergency services, timer T3511 is started and the state is changed to 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION. When timer T3511 expires the registration procedure for initial registration shall be restarted, if still required.

If the registration attempt counter is equal to 5

- the UE shall delete 5G-GUTI, TAI list, last visited registered TAI, list of equivalent PLMNs (if any), and ngKSI, start timer T3502 and shall set the 5GS update status to 5U2 NOT UPDATED. The state is changed to 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION or optionally to 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection, SNPN selection according to 3GPP TS 23.122 [5].
- if the procedure is performed via 3GPP access and the UE is operating in single-registration mode:
 - the UE shall in addition handle the EMM parameters EPS update status, EMM state, 4G-GUTI, TAI list, last visited registered TAI, list of equivalent PLMNs and eKSI as specified in 3GPP TS 24.301 [15] for the abnormal cases when an EPS attach procedure fails and the attach attempt counter is equal to 5; and
 - the UE shall attempt to select E-UTRAN radio access technology and proceed with appropriate EMM specific procedures. Additionally, The UE may disable the N1 mode capability as specified in subclause 4.9.

5.5.1.2.8 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Lower layer failure

If a lower layer failure occurs before the REGISTRATION COMPLETE message has been received from the UE and timer T3550 is running, the AMF shall locally abort the registration procedure for initial registration, enter state 5GMM-REGISTERED and shall not resend the REGISTRATION ACCEPT message. If a new 5G-GUTI was assigned to the UE in the registration procedure for initial registration, the AMF shall consider both the old and the new 5G-GUTI as valid until the old 5G-GUTI can be considered as invalid by the AMF or the 5GMM context which has been marked as deregistered in the AMF is released. If the old 5G-GUTI was allocated by an AMF other than the current AMF, the current AMF does not need to retain the old 5G-GUTI. During this period the network may use the identification procedure followed by a generic UE configuration update procedure if the old 5G-GUTI is used by the UE in a subsequent message.

b) Protocol error

If the REGISTRATION REQUEST message is received with a protocol error, the AMF shall return a REGISTRATION REJECT message with one of the following 5GMM cause values:

- #96 invalid mandatory information;
- #99 information element non-existent or not implemented;
- #100 conditional IE error; or

- #111 protocol error, unspecified.
- c) T3550 time-out

On the first expiry of the timer, the AMF shall retransmit the REGISTRATION ACCEPT message and shall reset and restart timer T3550.

This retransmission is repeated four times, i.e. on the fifth expiry of timer T3550, the registration procedure for initial registration shall be aborted and the AMF enters state 5GMM-REGISTERED. If a new 5G-GUTI was allocated in the REGISTRATION ACCEPT message, the AMF shall consider both the old and the new 5G-GUTIs as valid until the old 5G-GUTI can be considered as invalid by the AMF or the 5GMM context which has been marked as de-registered in the AMF is released. If the old 5G-GUTI was allocated by an AMF other than the current AMF, the current AMF does not need to retain the old 5G-GUTI. During this period, if the old 5G-GUTI is used by the UE in a subsequent message, the AMF acts as specified for case a) above.

- REGISTRATION REQUEST message received after the REGISTRATION ACCEPT message has been sent and before the REGISTRATION COMPLETE message is received, if the REGISTRATION COMPLETE message is expected.
 - If one or more of the information elements in the REGISTRATION REQUEST message differ from the ones received within the previous REGISTRATION REQUEST message, the previously initiated the registration procedure for initial registration shall be aborted if the REGISTRATION COMPLETE message has not been received and the new registration procedure for initial registration shall be progressed; or
 - if the information elements do not differ, then the REGISTRATION ACCEPT message shall be resent and the timer T3550 shall be restarted. In that case, the retransmission counter related to T3550 is not incremented.
- e) More than one REGISTRATION REQUEST message with 5GS registration type IE set to "initial registration" received and no REGISTRATION ACCEPT message or REGISTRATION REJECT message has been sent.
 - If one or more of the information elements in the REGISTRATION REQUEST message with 5GS registration type IE set to "initial registration" differs from the ones received within the previous REGISTRATION REQUEST message with 5GS registration type IE set to "initial registration", the previously initiated the registration procedure for initial registration shall be aborted and the new the registration procedure for initial registration shall be executed;
 - 2) if the information elements do not differ, then the network shall continue with the previous the registration procedure for initial registration and shall ignore the second REGISTRATION REQUEST message.
- f) REGISTRATION REQUEST message with 5GS registration type IE set to "initial registration" received in state 5GMM-REGISTERED.

If a REGISTRATION REQUEST message with 5GS registration type IE set to "initial registration" is received in state 5GMM-REGISTERED the network may initiate the 5GMM common procedures; if it turned out that the REGISTRATION REQUEST message was sent by a genuine UE that has already been registered, the 5GMM context, if any, are deleted and the new REGISTRATION REQUEST is progressed.

- NOTE 1: The network can determine that the UE is genuine by executing the authentication procedure as described in subclause 5.4.1.
- g) REGISTRATION REQUEST message with 5GS registration type IE set to "mobility registration updating" or "periodic registration updating" received before REGISTRATION COMPLETE message, if the REGISTRATION COMPLETE message is expected.

Timer T3550 shall be stopped. The allocated 5G-GUTI in the registration procedure for initial registration shall be considered as valid and the registration procedure for mobility and periodic update shall be progressed as described in subclause 5.5.1.3.

h) DEREGISTRATION REQUEST message received before REGISTRATION COMPLETE message, if the REGISTRATION COMPLETE message is expected.

The AMF shall abort the registration procedure for initial registration and shall progress the de-registration procedure as described in subclause 5.5.2.2.

i) UE security capabilities invalid or unacceptable

If the REGISTRATION REQUEST message is received with invalid or unacceptable UE security capabilities (e.g. no 5GS encryption algorithms (all bits zero), no 5GS integrity algorithms (all bits zero), mandatory 5GS encryption algorithms not supported or mandatory 5GS integrity algorithms not supported, etc.), the AMF shall return a REGISTRATION REJECT message.

- NOTE 2: 5GMM cause value to be used in REGISTRATION REJECT message is up to the network implementation.
- j) Based on operator policy, if the initial registration request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall reject the initial registration with a 5GMM cause value other than the 5GMM cause #76 (Not authorized for this CAG or authorized for CAG cells only).
- NOTE 3: 5GMM cause #7 (5GS services not allowed), 5GMM cause #11 (PLMN not allowed), 5GMM cause #27 (N1 mode not allowed), 5GMM cause #73 (Serving network not authorized) can be used depending on the subscription of the UE and whether the UE roams or not.

5.5.1.3 Registration procedure for mobility and periodic registration update

5.5.1.3.1 General

This procedure is used by a UE for both mobility and periodic registration update of 5GS services. This procedure, when used for periodic registration update of 5GS services, is performed only in 3GPP access.

This procedure used for periodic registration update of 5GS services is controlled in the UE by timer T3512. When timer T3512 expires, the registration procedure for mobility and periodic registration area updating is started. Start and reset of timer T3512 is described in subclause 10.2.

If the MUSIM UE is registered for emergency services and initiates a registration procedure for mobility and periodic registration update, the network shall not indicate the support of:

- the NAS signalling connection release;
- the paging indication for voice services;
- the reject paging request; or
- the paging restriction;

in the REGISTRATION ACCEPT message.

5.5.1.3.2 Mobility and periodic registration update initiation

The UE in state 5GMM-REGISTERED shall initiate the registration procedure for mobility and periodic registration update by sending a REGISTRATION REQUEST message to the AMF,

- a) when the UE detects that the current TAI is not in the list of tracking areas that the UE previously registered in the AMF;
- b) when the periodic registration updating timer T3512 expires in 5GMM-IDLE mode;
- c) when the UE receives a CONFIGURATION UPDATE COMMAND message indicating "registration requested" in the Registration requested bit of the Configuration update indication IE as specified in subclauses 5.4.4.3;
- d) when the UE in state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE either receives a paging or the UE receives a NOTIFICATION message with access type indicating 3GPP access over the non-3GPP access for PDU sessions associated with 3GPP access;
- NOTE 1: As an implementation option, MUSIM UE is allowed to not respond to paging based on the information available in the paging message, e.g. voice service indication.
- e) upon inter-system change from S1 mode to N1 mode and if the UE previously had initiated an attach procedure or a tracking area updating procedure when in S1 mode;

- f) when the UE receives an indication of "RRC Connection failure" from the lower layers and does not have signalling pending (i.e. when the lower layer requests NAS signalling connection recovery) except for the case specified in subclause 5.3.1.4;
- g) when the UE changes the 5GMM capability or the S1 UE network capability or both;
- h) when the UE's usage setting changes;
- i) when the UE needs to change the slice(s) it is currently registered to;
- j) when the UE changes the UE specific DRX parameters;
- k) when the UE in state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE receives a request from the upper layers to establish an emergency PDU session or perform emergency services fallback;
- 1) when the UE needs to register for SMS over NAS, indicate a change in the requirements to use SMS over NAS, or de-register from SMS over NAS;
- m) when the UE needs to indicate PDU session status to the network after performing a local release of PDU session(s) as specified in subclauses 6.4.1.5 and 6.4.3.5;
- n) when the UE in 5GMM-IDLE mode changes the radio capability for NG-RAN or E-UTRAN;
- o) when the UE receives a fallback indication from the lower layers and does not have signalling pending (i.e. when the lower layer requests NAS signalling connection recovery, see subclauses 5.3.1.4 and 5.3.1.2);
- p) void;
- q) when the UE needs to request new LADN information;
- r) when the UE needs to request the use of MICO mode or needs to stop the use of MICO mode or to request the use of new T3324 value or new T3512 value;
- s) when the UE in 5GMM-CONNECTED mode with RRC inactive indication enters a cell in the current registration area belonging to an equivalent PLMN of the registered PLMN and not belonging to the registered PLMN;
- t) when the UE receives over 3GPP access a SERVICE REJECT message or a DL NAS TRANSPORT message, with the 5GMM cause value set to #28 "Restricted service area";
- u) when the UE needs to request the use of eDRX, when a change in the eDRX usage conditions at the UE requires different extended DRX parameters, or needs to stop the use of eDRX;
- NOTE 2: A change in the eDRX usage conditions at the UE can include e.g. a change in the UE configuration, a change in requirements from upper layers or the battery running low at the UE.
- v) when the UE supporting 5G-SRVCC from NG-RAN to UTRAN changes the mobile station classmark 2 or the supported codecs;
- w) when the UE in state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE decides to request new network slices after being rejected due to no allowed network slices requested, or request S-NSSAI(s) which have been removed from the rejected NSSAI for the maximum number of UEs reached;
- x) when the UE is not in NB-N1 mode and the applicable UE radio capability ID for the current UE radio configuration changes due to a revocation of the network-assigned UE radio capability IDs by the serving PLMN or SNPN;
- y) when the UE receives a REGISTRATION REJECT message with 5GMM cause values #3, #6 or #7 without integrity protection over another access;
- z) when the UE needs to request new ciphering keys for ciphered broadcast assistance data;
- za) when due to manual CAG selection the UE has selected a CAG-ID which is not included in the "allowed CAG list" for the selected PLMN or a CAG-ID in a PLMN for which the entry in the "CAG information list" does not exist or when the UE has selected, without selecting a CAG-ID, a PLMN for which the entry in the "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells";

- zb) when the UE needs to start, stop or change the conditions for using the WUS assistance information or PEIPS assistance information;
- zc) when the UE changes the UE specific DRX parameters in NB-N1 mode;
- zd) when the UE in 5GMM-CONNECTED mode with RRC inactive indication enters a new cell with different RAT in current TAI list or not in current TAI list;
- ze) when the UE enters state 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) over 3GPP access after the UE has sent a NOTIFICATION RESPONSE message over non-3GPP access in response to reception of a NOTIFICATION message over non-3GPP access as specified in subclause 5.6.3.1;
- zf) when the UE supporting UAS services is not registered for UAS services and needs to register to the 5GS for UAS services;
- zg) when the UE supporting MINT needs to perform the registration procedure for mobility and periodic registration update to register to the PLMN offering disaster roaming services;
- zh) when the MUSIM UE supporting the paging timing collision control needs to request a new 5G-GUTI assignment and the UE is not registered for emergency services;
- NOTE 3: Based on implementation, the MUSIM UE can request a new 5G-GUTI assignment (e.g. when the lower layers request to modify the timing of the paging occasions).
- zi) when the network supports the paging restriction and the MUSIM UE in state 5GMM-REGISTERED.NON-ALLOWED-SERVICE needs to requests the network to remove the paging restriction;
- zj) when the UE changes the 5GS Preferred CIoT network behaviour or the EPS Preferred CIoT network behaviour;
- zk) when the UE that has entered 5GMM-REGISTERED.NO-CELL-AVAILABLE and it has one or more S-NSSAI(s) in pending NSSAI, finds a suitable cell according to 3GPP TS 38.304 [28]; or
- zl) when the UE is registered for disaster roaming services and receives a request from the upper layers to establish an emergency PDU session or perform emergency services fallback.

If case b) is the only reason for initiating the registration procedure for mobility and periodic registration update, the UE shall indicate "periodic registration updating" in the 5GS registration type IE; otherwise, if the UE initiates the registration procedure for mobility and periodic registration update due to case Zg), the UE shall indicate "disaster roaming mobility registration updating" in the 5GS registration type IE; otherwise the UE shall indicate "mobility registration updating".

If case zl) is the reason for initiating the registration procedure for mobility and periodic registration update and if the UE supports S1 mode, the UE shall:

- set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- include the S1 UE network capability IE in the REGISTRATION REQUEST message;

If the UE which is not registered for disaster roaming services indicates "mobility registration updating" in the 5GS registration type IE and the UE supports S1 mode, the UE shall:

- set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message;
- include the S1 UE network capability IE in the REGISTRATION REQUEST message; and
- if the UE supports sending an ATTACH REQUEST message containing a PDN CONNECTIVITY REQUEST message with request type set to "handover" to transfer a PDU session from N1 mode to S1 mode, set the HO attach bit to "attach request message containing PDN connectivity request with request type set to handover to transfer PDU session from N1 mode to S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the LTE positioning protocol (LPP) in N1 mode as specified in 3GPP TS 37.355 [26], the UE shall set the LPP bit to "LPP in N1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the Location Services (LCS) notification mechanisms in N1 mode as specified in 3GPP TS 23.273 [6B], the UE shall set the 5G-LCS bit to "LCS notification mechanisms supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For all cases except case b), when the UE is not in NB-N1 mode and the UE supports RACS, the UE shall set the RACS bit to "RACS supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports 5G-SRVCC from NG-RAN to UTRAN as specified in 3GPP TS 23.216 [6A], the UE shall set:

- the 5G-SRVCC from NG-RAN to UTRAN capability bit to "5G-SRVCC from NG-RAN to UTRAN supported" in the 5GMM capability IE of the REGISTRATION REQUEST message for all cases except case b; and
- include the Mobile station classmark 2 IE and the Supported codecs IE in the REGISTRATION REQUEST message for all cases except case b.

If the UE supports the restriction on use of enhanced coverage, the UE shall set the RestrictEC bit to "Restriction on use of enhanced coverage supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports network slice-specific authentication and authorization, the UE shall set the NSSAA bit to "network slice-specific authentication and authorization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message for all cases except case b.

If the UE supports CAG feature, the UE shall set the CAG bit to "CAG Supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports extended CAG information list, the UE shall set the Ex-CAG bit to "Extended CAG information list supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports sending of REGISTRATION COMPLETE message for acknowledging the reception of Negotiated PEIPS assistance inforation IE, the UE shall set the RCMAP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS assistance information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE operating in the single-registration mode performs inter-system change from S1 mode to N1 mode and has one or more stored UE policy sections identified by a UPSI with the PLMN ID part indicating the HPLMN or the selected PLMN, the UE shall set the Payload container type IE to "UE policy container" and include the UE STATE INDICATION message (see annex D) in the Payload container IE of the REGISTRATION REQUEST message.

NOTE 4: In this version of the protocol, the UE can only include the Payload container IE in the REGISTRATION REQUEST message to carry a payload of type "UE policy container".

The UE in state 5GMM-REGISTERED shall initiate the registration procedure for mobility and periodic update by sending a REGISTRATION REQUEST message to the AMF when the UE needs to request the use of SMS over NAS transport or the current requirements to use SMS over NAS transport change in the UE. The UE shall set the SMS requested bit of the 5GS update type IE in the REGISTRATION REQUEST message as specified in subclause 5.5.1.2.2.

When initiating a registration procedure for mobility and periodic registration update and the UE needs to send the 5GS update type IE for a reason different than indicating a change in requirement to use SMS over NAS, the UE shall set the SMS requested bit of the 5GS update type IE in the REGISTRATION REQUEST message to the same value as indicated by the UE in the last REGISTRATION REQUEST message.

If the UE no longer requires the use of SMS over NAS, then the UE shall include the 5GS update type IE in the REGISTRATION REQUEST message with the SMS requested bit set to "SMS over NAS not supported".

After sending the REGISTRATION REQUEST message to the AMF the UE shall start timer T3510. If timer T3502 is currently running, the UE shall stop timer T3502. If timer T3511 is currently running, the UE shall stop timer T3511.

If the last visited registered TAI is available, the UE shall include the last visited registered TAI in the REGISTRATION REQUEST message.

The UE shall handle the 5GS mobile identity IE in the REGISTRATION REQUEST message as follows:

- a) if the UE is operating in the single-registration mode, performs inter-system change from S1 mode to N1 mode, and the UE holds a valid native 4G-GUTI, the UE shall create a 5G-GUTI mapped from the valid native 4G-GUTI as specified in 3GPP TS 23.003 [4] and indicate the mapped 5G-GUTI in the 5GS mobile identity IE. Additionally, if the UE holds a valid 5G-GUTI, the UE shall include the 5G-GUTI in the Additional GUTI IE in the REGISTRATION REQUEST message in the following order:
 - 1) a valid 5G-GUTI that was previously assigned by the same PLMN with which the UE is performing the registration, if available;
 - 2) a valid 5G-GUTI that was previously assigned by an equivalent PLMN, if available; and
 - 3) a valid 5G-GUTI that was previously assigned by any other PLMN, if available; and

NOTE 5: The 5G-GUTI included in the Additional GUTI IE is a native 5G-GUTI.

b) for all other cases, if the UE holds a valid 5G-GUTI, the UE shall indicate the 5G-GUTI in the 5GS mobile identity IE. If the UE is registering with an SNPN and the valid 5G-GUTI was previously assigned by another SNPN, the UE shall additionally include the NID of the other SNPN in the NID IE.

If the UE does not operate in SNPN access operation mode, holds two valid native 5G-GUTIs assigned by PLMNs and:

- one of the valid native 5G-GUTI was assigned by the PLMN with which the UE is performing the registration, then the UE shall indicate the valid native 5G-GUTI assigned by the PLMN with which the UE is performing the registration. In addition, the UE shall include the other valid native 5G-GUTI in the Additional GUTI IE; or
- 2) none of the valid native 5G-GUTI was assigned by the PLMN with which the UE is performing the registration, then the UE shall indicate the valid native 5G-GUTI assigned over the same access via which the UE is performing the registration.

If the UE supports MICO mode and requests the use of MICO mode, then the UE shall include the MICO indication IE in the REGISTRATION REQUEST message. If the UE requests to use an active time value, it shall include the active time value in the T3324 IE in the REGISTRATION REQUEST message. If the UE includes the T3324 IE, it may also request a particular T3512 value by including the Requested T3512 IE in the REGISTRATION REQUEST message. Additionally, if the UE supports strictly periodic registration timer, the UE shall set the Strictly Periodic Registration Timer Indication bit of the MICO indication IE in the REGISTRATION REQUEST message to "strictly periodic registration timer supported". If the UE needs to stop the use of MICO mode, then the UE shall not include the MICO indication IE in the REGISTRATION REQUEST message.

If the UE needs to use or change the UE specific DRX parameters, the UE shall include the Requested DRX parameters IE in the REGISTRATION REQUEST message.

If the UE is in NB-N1 mode and if the UE needs to use or change the UE specific DRX parameters for NB-N1 mode, the UE shall include the Requested NB-N1 mode DRX parameters IE in the REGISTRATION REQUEST message.

If the UE supports eDRX and requests the use of eDRX, the UE shall include the Requested extended DRX parameters IE in the REGISTRATION REQUEST message.

If the UE needs to request LADN information for specific LADN DNN(s) or indicates a request for LADN information as specified in 3GPP TS 23.501 [8], the UE shall include the LADN indication IE in the REGISTRATION REQUEST message and:

- request specific LADN DNNs by including a LADN DNN value in the LADN indication IE for each LADN DNN for which the UE requests LADN information; or
- to indicate a request for LADN information by not including any LADN DNN value in the LADN indication IE.

If the UE is initiating the registration procedure for mobility and periodic registration update, the UE may include the Uplink data status IE to indicate which PDU session(s) that is:

- not associated with control plane only indication;
- associated with the access type the REGISTRATION REQUEST message is sent over; and
- have pending user data to be sent over user plane.

If the UE has one or more active always-on PDU sessions associated with the access type over which the REGISTRATION REQUEST message is sent and the user-plane resources for these PDU sessions are not established, and for cases triggering the REGISTRATION REQUEST message except b), the UE shall include the Uplink data status IE and indicate that the UE has pending user data to be sent for those PDU sessions. If the UE is located outside the LADN service area, the UE shall not include the PDU session for LADN in the Uplink data status IE. If the UE is in a non-allowed area or is not in an allowed area as specified in subclause 5.3.5, the UE shall not include the Uplink data status IE except for emergency services or for high priority access. If the MUSIM UE requests the network to release the NAS signalling connection, the UE shall not include the Uplink data status IE in the REGISTRATION REQUEST message.

If the UE has one or more active PDU sessions which are not accepted by the network as always-on PDU sessions and no uplink user data pending to be sent for those PDU sessions, the UE shall not include those PDU sessions in the Uplink data status IE in the REGISTRATION REQUEST message.

When the registration procedure for mobility and periodic registration update is initiated in 5GMM-IDLE mode, the UE may include a PDU session status IE in the REGISTRATION REQUEST message, indicating:

- which single access PDU sessions associated with the access type the REGISTRATION REQUEST message is sent over are active in the UE; and
- which MA PDU sessions are active and having user plane resources established in the UE on the access the REGISTRATION REQUEST message is sent over.

If the UE received a paging message with the access type indicating non-3GPP access, the UE shall include the Allowed PDU session status IE in the REGISTRATION REQUEST message. If the UE has established the PDU session(s) over the non-3GPP access for which the associated S-NSSAI(s) are included in the allowed NSSAI for 3GPP access, the UE shall indicate the PDU session(s) for which the UE allows to re-establish the user-plane resources over 3GPP access in the Allowed PDU session status IE. Otherwise, the UE shall not indicate any PDU session(s) in the Allowed PDU session status IE.

When the Allowed PDU session status IE is included in the REGISTRATION REQUEST message, the UE shall indicate that a PDU session is not allowed to be transferred to the 3GPP access if the 3GPP PS data off UE status is "activated" for the corresponding PDU session and the UE is not using the PDU session to send uplink IP packets for any of the 3GPP PS data off exempt services (see subclause 6.2.10).

If the UE operating in the single-registration mode performs inter-system change from S1 mode to N1 mode, the UE:

- a) shall include the UE status IE with the EMM registration status set to "UE is in EMM-REGISTERED state" in the REGISTRATION REQUEST message;
- NOTE 6: Inclusion of the UE status IE with this setting corresponds to the indication that the UE is "moving from EPC" as specified in 3GPP TS 23.502 [9], subclause 4.11.1.3.3 and 4.11.2.3.
- NOTE 7: The value of the 5GMM registration status included by the UE in the UE status IE is not used by the AMF.
- b) may include the PDU session status IE in the REGISTRATION REQUEST message indicating the status of the PDU session(s) mapped during the inter-system change from S1 mode to N1 mode from the PDN connection(s) for which the EPS indicated that interworking to 5GS is supported, if any (see subclause 6.1.4.1);
- c) shall include a TRACKING AREA UPDATE REQUEST message as specified in 3GPP TS 24.301 [15] in the EPS NAS message container IE in the REGISTRATION REQUEST message if the registration procedure is initiated in 5GMM-IDLE mode and the UE has received an "interworking without N26 interface not supported" indication from the network;
- c1) may include a TRACKING AREA UPDATE REQUEST message as specified in 3GPP TS 24.301 [15] in the EPS NAS message container IE in the REGISTRATION REQUEST message if the registration procedure is initiated in 5GMM-IDLE mode and the UE has received an "interworking without N26 interface supported" indication from the network; and
- d) shall include an EPS bearer context status IE in the REGISTRATION REQUEST message indicating which EPS bearer contexts are active in the UE, if the UE has locally deactivated EPS bearer context(s) for which interworking to 5GS is supported while the UE was in S1 mode without notifying the network.

For a REGISTRATION REQUEST message with a 5GS registration type IE indicating "mobility registration updating", if the UE:

- a) is in NB-N1 mode and:
 - 1) the UE needs to change the slice(s) it is currently registered to within the same registration area; or
 - 2) the UE has entered a new registration area; or
- b) the UE is not in NB-N1 mode and is not registered for onboarding services in SNPN;

the UE shall include the Requested NSSAI IE containing the S-NSSAI(s) corresponding to the network slices to which the UE intends to register and associated mapped S-NSSAI(s), if available, in the REGISTRATION REQUEST message as described in this subclause. When the UE is entering a visited PLMN and intends to register to the slices for which the UE has only HPLMN S-NSSAI(s) available, the UE shall include these HPLMN S-NSSAI(s) in the Requested mapped NSSAI IE. This is also applicable when the UE is entering an EHPLMN whose PLMN code is not derived from the IMSI.

NOTE 8: The REGISTRATION REQUEST message can include both the Requested NSSAI IE and the Requested mapped NSSAI IE as described below.

If the UE is registered for onboarding services in SNPN, the UE shall not include the Requested NSSAI IE in the REGISTRATION REQUEST message.

If the UE has allowed NSSAI or configured NSSAI or both for the current PLMN, the Requested NSSAI IE shall include either:

- a) the configured NSSAI for the current PLMN or SNPN, or a subset thereof as described below;
- b) the allowed NSSAI for the current PLMN or SNPN, or a subset thereof as described below; or
- c) the allowed NSSAI for the current PLMN or SNPN, or a subset thereof as described below, plus one or more S-NSSAIs from the configured NSSAI for which no corresponding S-NSSAI is present in the allowed NSSAI and those are neither in the rejected NSSAI nor in the pending NSSAI.

and in addition the Requested NSSAI IE shall include S-NSSAI(s) applicable in the current PLMN or SNPN, and if available the associated mapped S-NSSAI(s) for:

- a) each PDN connection that is established in S1 mode when the UE is operating in the single-registration mode and the UE is performing an inter-system change from S1 mode to N1 mode; or
- b) each active PDU session.

If the UE does not have S-NSSAI(s) applicable in the current PLMN or SNPN, then the Requested mapped NSSAI IE shall include HPLMN S-NSSAI(s) (e.g. mapped S-NSSAI(s), if available) for:

- a) each PDN connection established in S1 mode when the UE is operating in the single-registration mode and the UE is performing an inter-system change from S1 mode to N1 mode to a visited PLMN; or
- b) each active PDU session when the UE is performing mobility from N1 mode to N1 mode to a visited PLMN.
- NOTE 9: The Requested NSSAI IE is used instead of Requested mapped NSSAI IE in REGISTRATION REQUEST message when the UE enters HPLMN and the EHPLMN list is not present or is empty; or when the UE enters a PLMN whose PLMN code is derived from the IMSI and the EHPLMN list is not empty. The Requested mapped NSSAI IE is used when the UE enters an EHPLMN whose PLMN code is not derived from the IMSI.

For a REGISTRATION REQUEST message with a 5GS registration type IE indicating "mobility registration updating", if the UE is in NB-N1 mode and the procedure is initiated for all cases except case a), c), e), i), s), t), w), and x), the REGISTRATION REQUEST message shall not include the Requested NSSAI IE.

If the UE has:

- no allowed NSSAI for the current PLMN or SNPN;
- no configured NSSAI for the current PLMN or SNPN;

- neither active PDU session(s) nor PDN connection(s) to transfer associated with an S-NSSAI applicable in the current PLMN or SNPN; and
- neither active PDU session(s) nor PDN connection(s) to transfer associated with mapped S-NSSAI(s);

and has a default configured NSSAI, then the UE shall:

- a) include the S-NSSAI(s) in the Requested NSSAI IE of the REGISTRATION REQUEST message using the default configured NSSAI; and
- b) include the Network slicing indication IE with the Default configured NSSAI indication bit set to "Requested NSSAI created from default configured NSSAI" in the REGISTRATION REQUEST message.

If the UE has:

- no allowed NSSAI for the current PLMN or SNPN;
- no configured NSSAI for the current PLMN or SNPN;
- neither active PDU session(s) nor PDN connection(s) to transfer associated with an S-NSSAI applicable in the current PLMN or SNPN
- neither active PDU session(s) nor PDN connection(s) to transfer associated with mapped S-NSSAI(s); and
- no default configured NSSAI,

the UE shall include neither Requested NSSAI IE nor Requested mapped NSSAI IE in the REGISTRATION REQUEST message.

If all the S-NSSAI(s) corresponding to the slice(s) to which the UE intends to register are included in the pending NSSAI, the UE shall not include a requested NSSAI in the REGISTRATION REQUEST message.

When the UE storing a pending NSSAI intends to register to additional S-NSSAI(s) over the same access type, the UE shall send the requested NSSAI containing the additional S-NSSAI(s) that the UE intends to register to in the REGISTRATION REQUEST message. The requested NSSAI shall not include any S-NSSAI from the pending NSSAI.

The subset of configured NSSAI provided in the requested NSSAI consists of one or more S-NSSAIs in the configured NSSAI applicable to this PLMN or SNPN, if the S-NSSAI is neither in the rejected NSSAI nor associated to the S-NSSAI(s) in the rejected NSSAI. In addition, if the NSSRG information is available, the subset of configured NSSAI provided in the requested NSSAI shall be associated with at least one common NSSRG value. If the UE is in 5GMM-REGISTERED state over the other access and has already an allowed NSSAI for the other access in the same PLMN or in different PLMNs, all the S-NSSAI(s) in the requested NSSAI for the current access shall share at least an NSSRG value common to all the S-NSSAI(s) of the allowed NSSAI for the other access. If the UE is simultaneously performing the registration procedure on the other access in different PLMNs, the UE shall include S-NSSAIs that share at least a common NSSRG value across all access types. The S-NSSAIs in the pending NSSAI and requested NSSAI shall be associated with at least one common NSSRG value.

- NOTE 10:If the UE has stored mapped S-NSSAI(s) for the rejected NSSAI, and one or more S-NSSAIs in the stored mapped S-NSSAI(s) for the configured NSSAI are not included in the stored mapped S-NSSAI(s) for the rejected NSSAI, then a S-NSSAI in the configured NSSAI associated to one or more of these mapped S-NSSAI(s) for the configured NSSAI are available to be included in the requested NSSAI together with their mapped S-NSSAI.
- NOTE 11:If one or more mapped S-NSSAIs in the stored mapped S-NSSAI(s) for the configured NSSAI are not included in the stored rejected NSSAI for the failed or revoked NSSAA, a S-NSSAI in the configured NSSAI associated to one or more of these mapped S-NSSAI(s) for the configured NSSAI are available to be included in the registration request together with their mapped S-NSSAI.
- NOTE 12: There is no need to consider the case that the UE is simultaneously performing the registration procedure on the other access in the same PLMN, due to that the UE is not allowed to initiate the registration procedure over one access when the registration over the other access to the same PLMN is going on.

The subset of allowed NSSAI provided in the requested NSSAI consists of one or more S-NSSAIs in the allowed NSSAI for this PLMN.

- NOTE 13:How the UE selects the subset of configured NSSAI or allowed NSSAI to be provided in the requested NSSAI is implementation specific. The UE can take preferences indicated by the upper layers (e.g. policies like URSP, applications) and UE local configuration into account.
- NOTE 14: The number of S-NSSAI(s) included in the requested NSSAI cannot exceed eight.

If the UE supports NSAG, the UE shall set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports sending of REGISTRATION COMPLETE message for acknowledging the reception of NSAG information IE in the REGISTRATION ACCEPT message, the UE shall set the RCMAN bit to "Sending of REGISTRATION COMPLETE message for NSAG information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

The UE shall set the Follow-on request indicator to "Follow-on request pending", if the UE:

- a) initiates the mobility and periodic registration updating procedure upon request of the upper layers to establish an emergency PDU session;
- b) initiates the mobility and periodic registration updating procedure upon receiving a request from the upper layers to perform emergency services fallback; or
- c) needs to prolong the established NAS signalling connection after the completion of the registration procedure for mobility and periodic registration update (e.g. due to uplink signalling pending but no user data pending).
- NOTE 15: The UE does not have to set the Follow-on request indicator to 1 even if the UE has to request resources for V2X communication over PC5 reference point, 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5.

For case n), the UE shall include the 5GS update type IE in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to " UE radio capability update needed". Additionally, if the UE is not in NB-N1 mode, the UE supports RACS and the UE has an applicable UE radio capability ID for the new UE radio configuration in the serving PLMN or SNPN, the UE shall include the applicable UE radio capability ID in the UE radio capability ID of the REGISTRATION REQUEST message.

If the UE is in the 5GMM-CONNECTED mode and the UE changes the radio capability for NG-RAN or E-UTRAN, the UE may locally release the established N1 NAS signalling connection and enter the 5GMM-IDLE mode. Then, the UE shall initiate the registration procedure for mobility and periodic updating including the 5GS update type IE in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to "UE radio capability update needed".

For case o), the UE shall include the Uplink data status IE in the REGISTRATION REQUEST message indicating the PDU session(s) without active user-plane resources for which the UE has pending user data to be sent, if any, and the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any. If the UE is in a non-allowed area or if the UE is not in allowed area, the UE shall not include the Uplink data status IE in REGISTRATION REQUEST message, except if the PDU session for which user-plane resources were active prior to receiving the fallback indication is an emergency PDU session, or if the UE is configured for high priority access in the selected PLMN as specified in subclause 5.3.5.

For case f), the UE shall include the Uplink data status IE in the REGISTRATION REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving "RRC Connection failure" indication from the lower layers, if any. If the UE is in non-allowed area or not in allowed area, the UE shall not include the Uplink data status IE in REGISTRATION REQUEST message, except that the PDU session(s) for which user-plane resources were active prior to receiving the "RRC Connection failure" indication is emergency PDU session(s), or that the UE is configured for high priority access in selected PLMN, as specified in subclause 5.3.5.

If the UE supports service gap control, then the UE shall set the SGC bit to "service gap control supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For case a), x) or if the UE operating in the single-registration mode performs inter-system change from S1 mode to N1 mode, the UE shall:

- a) if the UE has an applicable network-assigned UE radio capability ID for the current UE radio configuration in the selected PLMN or SNPN, include the applicable network-assigned UE radio capability ID in the UE radio capability ID IE of the REGISTRATION REQUEST message; and
- b) if the UE:

1) does not have an applicable network-assigned UE radio capability ID for the current UE radio configuration in the selected PLMN or SNPN; and

2) has an applicable manufacturer-assigned UE radio capability ID for the current UE radio configuration,

include the applicable manufacturer-assigned UE radio capability ID in the UE radio capability ID IE of the REGISTRATION REQUEST message.

For all cases except cases b and z, if the UE supports ciphered broadcast assistance data and the UE needs to obtain new ciphering keys, the UE shall include the Additional information requested IE with the CipherKey bit set to "ciphering keys for ciphered broadcast assistance data requested" in the REGISTRATION REQUEST message.

For case z, the UE shall include the Additional information requested IE with the CipherKey bit set to "ciphering keys for ciphered broadcast assistance data requested" in the REGISTRATION REQUEST message.

For case a, if the UE supports ciphered broadcast assistance data and the UE detects that one or more ciphering keys stored at the UE is not applicable in the current TAI, the UE should include the Additional information requested IE with the CipherKey bit set to "ciphering keys for ciphered broadcast assistance data requested" in the REGISTRATION REQUEST message.

For case b, if the UE supports ciphered broadcast assistance data and the remaining validity time for one or more ciphering keys stored at the UE is less than timer T3512, the UE should include the Additional information requested IE with the CipherKey bit set to "ciphering keys for ciphered broadcast assistance data requested" in the REGISTRATION REQUEST message.

The UE shall set the WUSA bit to "WUS assistance information reception supported" in the 5GMM capability IE if the UE supports WUS assistance information. The UE may include its UE paging probability information in the Requested WUS assistance information IE if the UE has set the WUSA bit to "WUS assistance information reception supported" in the 5GMM capability IE and does not have an active emergency PDU session.

The UE shall set the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE if the UE supports PEIPS assistance information, is not registered for emergency services and does not have an active emergency PDU session. The UE may include its UE paging probability information in the Requested PEIPS assistance information IE if the UE has set the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE.

If the network supports the N1 NAS signalling connection release, and the MUSIM UE requests the network to release the NAS signalling connection, the UE shall set Request type to "NAS signalling connection release" in the UE request type IE, set the Follow-on request indicator to "No follow-on request pending" and, if the network supports the paging restriction, may set the paging restriction preference in the Paging restriction IE in the REGISTRATION REQUEST message. In addition, the UE shall not include the Uplink data status IE or the Allowed PDU session status IE in the REGISTRATION REQUEST message even if the UE has one or more active always-on PDU sessions associated with the 3GPP access.

- NOTE 16:If the network has already indicated support for N1 NAS signalling connection release in the current stored registration area and the UE doesn't have an emergency PDU session established, the MUSIM UE is allowed to request the network to release the NAS signalling connection during mobility registration update procedure that is due to mobility outside the registration area even before detecting whether the network supports the N1 NAS signalling connection release in the current TAI.
- NOTE 17: If the network has already indicated support for paging restriction in the current stored registration area and the UE doesn't have an emergency PDU session established, the MUSIM UE is allowed to include paging restriction together with the request to the network to release the NAS signalling connection during mobility registration update procedure that is due to mobility outside the registration area even before detecting whether the network supports the paging restriction in the current TAI.

For case zi), the UE shall not include the Paging restriction IE in the REGISTRATION REQUEST message. If the UE is in 5GMM-IDLE mode and the network supports the N1 NAS signalling connection release, the UE may include the UE request type IE and set Request type to "NAS signalling connection release" to remove the paging restriction and request the release of the NAS signalling connection at the same time. In addition, the UE shall not include the Uplink data status IE in the REGISTRATION REQUEST message.

If the UE does not have a valid 5G NAS security context and the UE is sending the REGISTRATION REQUEST message after an inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, the UE shall send the REGISTRATION REQUEST message without including the NAS message container IE. The UE shall include the

entire REGISTRATION REQUEST message (i.e. containing cleartext IEs and non-cleartext IEs, if any) in the NAS message container IE that is sent as part of the SECURITY MODE COMPLETE message as described in subclauses 4.4.6 and 5.4.2.3.

If the UE indicates "mobility registration updating" in the 5GS registration type IE and supports V2X as specified in 3GPP TS 24.587 [19B], the UE shall set the V2X bit to "V2X supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE indicates "mobility registration updating" in the 5GS registration type IE and supports V2X communication over E-UTRA-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCEPC5 bit to "V2X communication over E-UTRA-PC5 supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE indicates "mobility registration updating" in the 5GS registration type IE and supports V2X communication over E-UTRA-PC5 supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE indicates "mobility registration updating" in the 5GS registration type IE and supports V2X communication over NR-PC5 as specified in 3GPP TS 24.587 [19B], the UE shall set the V2XCNPC5 bit to "V2X communication over NR-PC5 supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

The UE shall send the REGISTRATION REQUEST message including the NAS message container IE as described in subclause 4.4.6:

- a) when the UE is sending the message from 5GMM-IDLE mode, the UE has a valid 5G NAS security context, and needs to send non-cleartext IEs; or
- b) when the UE is sending the message after an inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode and the UE has a valid 5G NAS security context and needs to send non-cleartext IEs.

The UE with a valid 5G NAS security context shall send the REGISTRATION REQUEST message without including the NAS message container IE when the UE does not need to send non-cleartext IEs and the UE is sending the message:

- a) from 5GMM-IDLE mode; or
- b) after an inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode.

If the UE is sending the REGISTRATION REQUEST message after an inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode and the UE needs to send non-cleartext IEs, the UE shall cipher the NAS message container IE using the mapped 5G NAS security context and send the REGISTRATION REQUEST message including the NAS message container IE as described in subclause 4.4.6. If the UE does not need to send non-cleartext IEs, the UE shall send the REGISTRATION REQUEST message without including the NAS message container IE.

If the REGISTRATION REQUEST message includes a NAS message container IE, the AMF shall process the REGISTRATION REQUEST message that is obtained from the NAS message container IE as described in subclause 4.4.6.

If the UE is in NB-N1 mode, then the UE shall set the Control plane CIoT 5GS optimization bit to "Control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE is capable of NB-S1 mode, then the UE shall set the Control plane CIoT EPS optimization bit to "Control plane CIoT EPS optimization supported" in the S1 UE network capability IE of the REGISTRATION REQUEST message.

If the registration procedure for mobility and periodic registration update is initiated and there is request from the upper layers to perform "emergency services fallback" pending, the UE shall send a REGISTRATION REQUEST message without an Uplink data status IE.

If the UE supports N3 data transfer and multiple user-plane resources in NB-N1 mode (see 3GPP TS 36.306 [25D], 3GPP TS 36.331 [25A]), then the UE shall set the Multiple user-plane resources support bit to "Multiple user-plane resources supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

The UE shall set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports the NSSRG, then the UE shall set the NSSRG bit to "NSSRG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For case zf), the UE shall include the service-level device ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and set the value to the CAA-level UAV ID. The UE shall include the service-level-AA server address in the Service-level-AA container IE of the REGISTRATION REQUEST message and set the value to the USS address, if it is provided by the upper layers.

3GPP TS 24.501 version 17.16.0 Release 17

339

If the UE supports 5G ProSe direct discovery as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-dd bit to "5G ProSe direct discovery supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports 5G ProSe discovery communication supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network relay UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-dc bit to "5G ProSe discovery communication supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network relay UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2relay bit to "Acting as a 5G ProSe layer-2 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-3 UE-to-network relay UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l3relay bit to "Acting as a 5G ProSe layer-3 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as a 5G ProSe layer-3 UE-to-network relay UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-2 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2rmt bit to "Acting as a 5G ProSe layer-2 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2rmt bit to "Acting as a 5G ProSe layer-2 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l2rmt bit to "Acting as a 5G ProSe layer-2 UE-to-network remote UE supported" in the 5GMM capability IE of the REGISTRATION REQUEST message. If the UE supports acting as 5G ProSe layer-3 UE-to-network remote UE as specified in 3GPP TS 24.554 [19E], the UE shall set the 5G ProSe-l3rmt bit to "Acting as a 5G ProSe layer-3 UE-to-ne

For all cases except case b, if the MUSIM UE supports the N1 NAS signalling connection release, then the UE shall set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For all cases except case b, if the MUSIM UE supports the paging indication for voice services, then the UE shall set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For all cases except case b, if the MUSIM UE supports the reject paging request, then the UE shall set the reject paging request bit to "reject paging request supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the reject paging request bit to "reject paging request supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For all cases except case b, if the MUSIM UE sets:

- the reject paging request bit to "reject paging request supported";
- the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported"; or
- both of them;

and supports the paging restriction, then the UE shall set the paging restriction bit to "paging restriction supported" in the 5GMM capability IE of the REGISTRATION REQUEST message otherwise the UE shall not set the paging restriction bit to "paging restriction supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports MINT, the UE shall set the MINT bit to "MINT supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports UAS services, the UE shall set the UAS bit to "UAS services supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

For case zg), if the UE has determined the MS determined PLMN with disaster condition as specified in 3GPP TS 23.122 [5], and:

- a) the MS determined PLMN with disaster condition is the HPLMN and:
 - 1) the Additional GUTI IE is included in the REGISTRATION REQUEST message and does not contain a valid 5G-GUTI that was previously assigned by the HPLMN; or
 - 2) the Additional GUTI IE is not included in the REGISTRATION REQUEST message and the 5GS mobile identity IE contains neither the SUCI nor a valid 5G-GUTI that was previously assigned by the HPLMN; or
- b) the MS determined PLMN with disaster condition is not the HPLMN and:

- 1) the Additional GUTI IE is included in the REGISTRATION REQUEST message and does not contain a valid 5G-GUTI that was previously assigned by the MS determsined PLMN with disaster condition; or
- the Additional GUTI IE is not included in the REGISTRATION REQUEST message and the 5GS mobile identity IE does not contain a valid 5G-GUTI that was previously assigned by the MS determined PLMN with disaster condition;

the UE shall include in the REGISTRATION REQUEST message the MS determined PLMN with disaster condition IE indicating the MS determined PLMN with disaster condition.

NOTE 18:If the UE initiates the registration procedure for disaster roaming services, and the MS determined PLMN with disaster condition cannot be determined when an NG-RAN cell of the PLMN broadcasts the disaster related indication as specified in 3GPP TS 23.122 [5], the UE does not include in the REGISTRATION REQUEST message the MS determined PLMN with disaster condition IE but includes the Additional GUTI IE or the 5GS mobile identity IE or both as specified in subclauses 5.5.1.2.2.

For case zh) the UE shall indicate "mobility registration updating" in the 5GS registration type IE of the REGISTRATION REQUEST message.

If the UE supports event notification, the UE shall set the EventNotification bit to "Event notification supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports access to an SNPN using credentials from a credentials holder and the UE is in its HPLMN or EHPLMN or a subscribed SNPN, the UE shall set the SSNPNSI bit to "SOR-SNPN-SI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.

If the UE supports MPS indicator update via the UE configuration update procedure, the UE shall set the MPSIU bit to "MPS indicator update supported" in the 5GMM capability IE of the REGISTRATION REQUEST message.



Figure 5.5.1.3.2.1: Registration procedure for mobility and periodic registration update

5.5.1.3.3 5GMM common procedure initiation

The AMF may initiate 5GMM common procedures, e.g. the identification, authentication and security procedures during the registration procedure, depending on the information received in the REGISTRATION REQUEST message.

The AMF may be configured to skip the authentication procedure even if no 5GS security context is available and proceed directly to the execution of the security mode control procedure as specified in subclause 5.4.2, during the registration procedure for mobility and periodic registration update for a UE that has only an emergency PDU session.

The AMF shall not initiate a 5GMM authentication procedure before completion of the registration procedure for mobility and periodic registration update, if the following conditions apply:

- a) the UE initiated the registration procedure for mobility and periodic registration update after handover or intersystem change to N1 mode in 5GMM-CONNECTED mode;
- b) the target cell is a shared network cell; and
- c.1) the UE has provided its 5G-GUTI in the 5GS mobile identity IE or the Additional GUTI IE in the REGISTRATION REQUEST message, and the PLMN identity included in the 5G-GUTI is different from the selected PLMN identity of the target cell; or
- c.2) the UE has included the 5G-GUTI mapped from the 4G-GUTI in the 5GS mobile identity IE and not included an Additional GUTI IE in the REGISTRATION REQUEST message, and the PLMN identity included in the 5G-GUTI is different from the selected PLMN identity of the target cell.

5.5.1.3.4 Mobility and periodic registration update accepted by the network

If the registration update request has been accepted by the network, the AMF shall send a REGISTRATION ACCEPT message to the UE.

If timer T3513 is running in the AMF, the AMF shall stop timer T3513 if a paging request was sent with the access type indicating non-3GPP and the REGISTRATION REQUEST message includes the Allowed PDU session status IE.

If timer T3565 is running in the AMF, the AMF shall stop timer T3565 when a REGISTRATION REQUEST message is received.

For each of the information elements: 5GMM capability, S1 UE network capability, and UE security capability, the AMF shall store all octets received from the UE in the REGISTRATION REQUEST message, up to the maximum length defined for the respective information element.

NOTE 1: This information is forwarded to the new AMF during inter-AMF handover or to the new MME during inter-system handover to S1 mode.

The 5G-GUTI reallocation shall be part of the registration procedure for mobility registration update. The 5G-GUTI reallocation should be part of the registration procedure for periodic registration update. During the registration procedure for mobility registration update, if the AMF has not allocated a new 5G-GUTI by the generic UE configuration update procedure, the AMF shall include in the REGISTRATION ACCEPT message the new assigned 5G-GUTI.

If the UE has set the CAG bit to "CAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the AMF needs to update the "CAG information list" stored in the UE, the AMF shall include the CAG information list IE or the Extended CAG information list IE in the REGISTRATION ACCEPT message.

- NOTE 2: The "CAG information list" can be provided by the AMF and include no entry if no "CAG information list" exists in the subscription.
- NOTE 2A: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If a 5G-GUTI or the SOR transparent container IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the Operator-defined access category definitions IE or the Extended emergency number list IE, the CAG information list IE or the Extended CAG information list IE are included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE has set the RCMP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS parameters supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if the PEIPS assistance information IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE is not in NB-N1 mode and the UE has set the RACS bit to "RACS supported" in the 5GMM Capability IE of the REGISTRATION REQUEST message, the AMF may include either a UE radio capability ID IE or a UE radio capability ID deletion indication IE in the REGISTRATION ACCEPT message. If the UE radio capability ID IE or the UE radio capability ID deletion indication IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

The AMF may include a new TAI list for the UE in the REGISTRATION ACCEPT message. The new TAI list shall not contain both tracking areas in NB-N1 mode and tracking areas not in NB-N1 mode. The UE, upon receiving a REGISTRATION ACCEPT message, shall delete its old TAI list and store the received TAI list. If there is no TAI list received, the UE shall consider the old TAI list as valid. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, and

a) the UE already has stored allowed NSSAI for the current registration area, the UE shall store the allowed NSSAI for the current registration area in each of the allowed NSSAIs which are associated with each of the PLMNs in the registration area; and

- b) the UE already has stored rejected NSSAI for the current registration area, the UE shall store the rejected NSSAI for the current registration area in each of the rejected NSSAIs which are associated with each of the PLMNs in the registration area.
- NOTE 3: When assigning the TAI list, the AMF can take into account the eNodeB's capability of support of CIoT 5GS optimization.

The AMF may also include a list of equivalent PLMNs in the REGISTRATION ACCEPT message. Each entry in the list contains a PLMN code (MCC+MNC). The UE shall store the list as provided by the network, and if there is no emergency PDU session established, the UE shall remove from the list any PLMN code that is already in the forbidden PLMN list as specified in subclause 5.3.13A. If the UE is not registered for emergency services and there is an emergency PDU session established, the UE shall remove from the list of equivalent PLMNs any PLMN code present in the forbidden PLMN list as specified in subclause 5.3.13A, when the emergency PDU session is released. In addition, the UE shall add to the stored list the PLMN code of the registered PLMN that sent the list. The UE shall replace the stored list on each receipt of the REGISTRATION ACCEPT message. If the REGISTRATION ACCEPT message does not contain a list, then the UE shall delete the stored list.

If the UE is not registered for emergency services, and if the PLMN identity of the registered PLMN is a member of the forbidden PLMN list as specified in subclause 5.3.13A, any such PLMN identity shall be deleted from the corresponding list(s).

The AMF may include new service area restrictions in the Service area list IE in the REGISTRATION ACCEPT message. The UE, upon receiving a REGISTRATION ACCEPT message with new service area restrictions shall act as described in subclause 5.3.5.

If the Service area list IE is not included in the REGISTRATION ACCEPT message, any tracking area in the registered PLMN and its equivalent PLMN(s) in the registration area is considered as an allowed tracking area as described in subclause 5.3.5.

The AMF shall include the MICO indication IE in the REGISTRATION ACCEPT message only if the MICO indication IE was included in the REGISTRATION REQUEST message, the AMF supports and accepts the use of MICO mode. If the AMF supports and accepts the use of MICO mode, the AMF may indicate "all PLMN registration area allocated" in the MICO indication IE in the REGISTRATION ACCEPT message. If "all PLMN registration area allocated" is indicated in the MICO indication IE, the AMF shall not assign and include the TAI list in the REGISTRATION ACCEPT message. If the REGISTRATION ACCEPT message includes an MICO indication IE indicating "all PLMN registration area allocated", the UE shall treat all TAIs in the current PLMN as a registration area and delete its old TAI list. If "strictly periodic registration timer supported" is indicated in the MICO indication IE in the REGISTRATION ACCEPT message. If the timer supported is indicated in the MICO indication is the AMF may indicate "strictly periodic registration area and delete its old TAI list. If "strictly periodic registration timer supported" is indicated in the MICO indication IE in the REGISTRATION REQUEST message, the AMF may indicate "strictly periodic registration timer supported" is indicated in the MICO indication IE in the REGISTRATION REQUEST message, the AMF may indicate "strictly periodic registration timer supported" in the MICO indication IE in the REGISTRATION ACCEPT message. If the timer value received in T3512 IE is different from the already stored value of the timer T3512 and the timer T3512 is running, the UE shall restart T3512 with the new value received in the T3512 value IE.

The AMF shall include an active time value in the T3324 IE in the REGISTRATION ACCEPT message if the UE requested an active time value in the REGISTRATION REQUEST message and the AMF accepts the use of MICO mode and the use of active time.

If the UE does not include MICO indication IE in the REGISTRATION REQUEST message, then the AMF shall disable MICO mode if it was already enabled.

If the AMF supports and accepts the use of MICO, and the UE included the Requested T3512 value IE in the REGISTRATION REQUEST message, then the AMF shall take into account the T3512 value requested when providing the T3512 value IE in the REGISTRATION ACCEPT message.

NOTE 4: The T3512 value assigned to the UE by AMF can be different from the T3512 value requested by the UE. AMF can take several factors into account when assigning the T3512 value, e.g. local configuration, expected UE behaviour, UE requested T3512 value, UE subscription data, network policies.

The AMF may include the T3512 value IE in the REGISTRATION ACCEPT message only if the REGISTRATION REQUEST message was sent over the 3GPP access.

The AMF may include the non-3GPP de-registration timer value IE in the REGISTRATION ACCEPT message only if the REGISTRATION REQUEST message was sent for the non-3GPP access.

If the UE indicates support of the N1 NAS signalling connection release in the REGISTRATION REQUEST message and the network decides to accept the N1 NAS signalling connection release, then the AMF shall set the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE indicates support of the paging indication for voice services in the REGISTRATION REQUEST message and the network decides to accept the paging indication for voice services, then the AMF shall set the paging indication for voice services bit to "paging indication for voice services supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message. If the UE receives the REGISTRATION ACCEPT message with the paging indication for voice services supported", the UE NAS layer informs the lower layers that paging indication for voice services is supported. Otherwise, the UE NAS layer informs the lower layers that paging indication for voice services is not supported.

If the UE indicates support of the reject paging request in the REGISTRATION REQUEST message and the network decides to accept the reject paging request, then the AMF shall set the reject paging request bit to "reject paging request supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE indicates support of the paging restriction in the REGISTRATION REQUEST message, and the AMF sets:

- the reject paging request bit to "reject paging request supported";
- the N1 NAS signalling connection release bit to "N1 NAS signalling connection release supported"; or
- both of them;

in the 5GS network feature support IE of the REGISTRATION ACCEPT message, and the network decides to accept the paging restriction, then the AMF shall set the paging restriction bit to "paging restriction supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the MUSIM UE does not include the Paging restriction IE in the REGISTRATION REQUEST message, the AMF shall delete any stored paging restriction for the UE and stop restricting paging.

If the MUSIM UE requests the release of the NAS signalling connection, by setting Request type to "NAS signalling connection release" in the UE request type IE included in the REGISTRATION REQUEST message, and the AMF supports the N1 NAS signalling connection release, the AMF shall initiate the release of the NAS signalling connection after the completion of the registration procedure for mobility and periodic registration update. If the UE requests restriction of paging by including the Paging restriction IE and the AMF supports the paging restriction, the AMF:

- if accepts the paging restriction, shall include the 5GS additional request result IE in the REGISTRATION ACCEPT message and set the Paging restriction decision to "paging restriction is accepted". The AMF shall store the paging restriction of the UE and enforce these restrictions in the paging procedure as described in clause 5.6.2; or
- if rejects the paging restriction, shall include the 5GS additional request result IE in the REGISTRATION ACCEPT message and set the Paging restriction decision to "paging restriction is rejected", and shall discard the received paging restriction. The AMF shall delete any stored paging restriction for the UE and stop restricting paging.

If the UE requests "control plane CIoT 5GS optimization" in the 5GS update type IE, indicates support of control plane CIoT 5GS optimization in the 5GMM capability IE and the AMF decides to accept the requested CIoT 5GS optimization and the registration request, the AMF shall indicate "control plane CIoT 5GS optimization supported" in the 5GS network feature support IE of the REGISTRATION ACCEPT message.

If the UE has indicated support for the control plane CIoT 5GS optimizations, and the AMF decides to activate the congestion control for transport of user data via the control plane, then the AMF shall include the T3448 value IE in the REGISTRATION ACCEPT message.

If the AMF decides to deactivate the congestion control for transport of user data via the control plane, then the AMF shall delete the stored control plane data back-off time for the UE and the AMF shall not include timer T3448 value IE in the REGISTRATION ACCEPT message.

If:

- the UE in NB-N1 mode is using control plane CIoT 5GS optimization; and

- the network is configured to provide the truncated 5G-S-TMSI configuration for control plane CIoT 5GS optimizations;

the AMF shall include the Truncated 5G-S-TMSI configuration IE in the REGISTRATION ACCEPT message and set the "Truncated AMF Set ID value" and the "Truncated AMF Pointer value" in the Truncated 5G-S-TMSI configuration IE based on network policies. The AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

For inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, if the UE has included a ngKSI indicating a current 5G NAS security context in the REGISTRATION REQUEST message by which the REGISTRATION REQUEST message is integrity protected, the AMF shall take one of the following actions:

- a) if the AMF retrieves the current 5G NAS security context as indicated by the ngKSI and 5G-GUTI sent by the UE, the AMF shall integrity check the REGISTRATION REQUEST message using the current 5G NAS security context and integrity protect the REGISTRATION ACCEPT message using the current 5G NAS security context;
- b) if the AMF cannot retrieve the current 5G NAS security context as indicated by the ngKSI and 5G-GUTI sent by the UE, the AMF shall treat the REGISTRATION REQUEST message fails the integrity check and take actions as specified in subclause 4.4.4.3; or
- c) if the UE has not included an Additional GUTI IE, the AMF may treat the REGISTRATION REQUEST message as in the previous item, i.e. as if it cannot retrieve the current 5G NAS security context.
- NOTE 5: The handling described above at failure to retrieve the current 5G NAS security context or if no Additional GUTI IE was provided does not preclude the option for the AMF to perform a primary authentication and key agreement procedure and create a new native 5G NAS security context.

For inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode, the AMF shall integrity check REGISTRATION REQUEST message using the current K'_{AMF} as derived when triggering the handover to N1 mode (see subclause 4.4.2.2). The AMF shall verify the received UE security capabilities in the REGISTRATION REQUEST message. The AMF shall then take one of the following actions:

- a) if the REGISTRATION REQUEST does not contain a valid KSI_{AMF} in the Non-current native NAS key set identifier IE, the AMF shall remove the non-current native 5G NAS security context, if any, for any 5G-GUTI for this UE. The AMF shall then integrity protect and cipher the REGISTRATION ACCEPT message using the security context based on K'_{AMF} and take the mapped 5G NAS security context into use; or
- b) if the REGISTRATION REQUEST contains a valid KSI_{AMF} in the Non-current native NAS key set identifier IE and:
 - the AMF decides to take the native 5G NAS security context into use, the AMF shall initiate a security mode control procedure to take the corresponding native 5G NAS security context into use and then integrity protect and cipher the REGISTRATION ACCEPT message using the corresponding native 5G NAS security context; and
 - 2) otherwise, the AMF shall then integrity protect and cipher the REGISTRATION ACCEPT message using the security context based on K'_{AMF} and take the mapped 5G NAS security context into use.
- NOTE 6: In above bullet b), it is recommended for the AMF to initiate a security mode control procedure to take the corresponding native 5G NAS security context into use.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message, and if:

- the UE has a valid aerial UE subscription information; and
- the UUAA procedure is to be performed during the registration procedure according to operator policy; and
- there is no valid successful UUAA result for the UE in the UE 5GMM context,

then the AMF shall initiate the UUAA-MM procedure with the UAS-NF as specified in TS 23.256 [6AB] and shall include a service-level-AA pending indication in the Service-level-AA container IE of the REGISTRATION ACCEPT message. The AMF shall store in the UE 5GMM context that a UUAA procedure is pending. The AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message, and if:

- the UE has a valid aerial UE subscription information;
- the UUAA procedure is to be performed during the registration procedure according to operator policy; and
- there is a valid successful UUAA result for the UE in the UE 5GMM context,

then the AMF shall include a service-level-AA response in the Service-level-AA container IE of the REGISTRATION ACCEPT message and set the SLAR field in the service-level-AA response to "Service level authentication and authorization was successful".

If the AMF determines that the UUAA-MM procedure needs to be performed for a UE, the AMF has not received the service -level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message from the UE and the AMF decides to accept the UE to be registered for other services than UAS services based on the user's subscription data and the operator policy, the AMF shall accept the registration update request and shall mark in the UE's 5GMM context that the UE is not allowed to request UAS services.

If the UE supports MINT, the AMF may include the List of PLMNs to be used in disaster condition IE in the REGISTRATION ACCEPT message.

If the UE supports MINT, the AMF may include the Disaster roaming wait range IE in the REGISTRATION ACCEPT message.

If the UE supports MINT, the AMF may include the Disaster return wait range IE in the REGISTRATION ACCEPT message.

NOTE 7: The AMF can determine the content of the "list of PLMN(s) to be used in disaster condition", the value of the disaster roaming wait range and the value of the disaster return wait range based on the network local configuration.

If the AMF received the list of TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, any but not all TAIs in the received list of TAIs is forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

- a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or
- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or
- c) both;

in the REGISTRATION ACCEPT message.

NOTE 8: Void.

Upon receipt of the REGISTRATION ACCEPT message, the UE shall reset the registration attempt counter and service request attempt counter, enter state 5GMM-REGISTERED and set the 5GS update status to 5U1 UPDATED.

If the UE receives the REGISTRATION ACCEPT message from a PLMN, then the UE shall reset the PLMN-specific attempt counter for that PLMN for the specific access type for which the message was received. The UE shall also reset the PLMN-specific N1 mode attempt counter for that PLMN for the specific access type for which the message was received. If the message was received via 3GPP access, the UE shall reset the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "SIM/USIM considered invalid for non-GPRS services", if any. If the message was received via non-3GPP access, the UE shall reset the counter for "USIM considered invalid for 5GS services over non-3GPP" events.

If the UE receives the REGISTRATION ACCEPT message from an SNPN, then the UE shall reset the SNPN-specific attempt counter for the current SNPN for the specific access type for which the message was received. If the message was received via 3GPP access, the UE shall reset the counter for "the entry for the current SNPN considered invalid for 3GPP access" events. If the message was received via non-3GPP access, the UE shall reset the counter for "the entry for the current SNPN considered invalid for the current SNPN considered invalid for non-3GPP access" events.

If the REGISTRATION ACCEPT message included a T3512 value IE, the UE shall use the value in T3512 value IE as periodic registration update timer (T3512). If the T3512 value IE is not included, the UE shall use the value currently stored, e.g. from a prior REGISTRATION ACCEPT message.

If the REGISTRATION ACCEPT message include a T3324 value IE, the UE shall use the value in the T3324 value IE as active time timer (T3324). If the REGISTRATION ACCEPT message does not include a T3324 value IE, UE shall not start the timer T3324 until a new value is received from the network.

If the REGISTRATION ACCEPT message included a non-3GPP de-registration timer value IE, the UE shall use the value in non-3GPP de-registration timer value IE as non-3GPP de-registration timer. If non-3GPP de-registration timer value IE is not included, the UE shall use the value currently stored, e.g. from a prior REGISTRATION ACCEPT message. If non-3GPP de-registration timer value IE is not included and there is no stored non-3GPP de-registration timer value in the UE, the UE shall use the default value of the non-3GPP de-registration timer.

If the REGISTRATION ACCEPT message contains a 5G-GUTI, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge the received 5G-GUTI, stop timer T3519 if running, and delete any stored SUCI. The UE shall provide the 5G-GUTI to the lower layer of 3GPP access if the REGISTRATION ACCEPT message is sent over the non-3GPP access, and the UE is in 5GMM-REGISTERED in both 3GPP access and non-3GPP access in the same PLMN.

If the REGISTRATION ACCEPT message contains the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed", or contains a Configured NSSAI IE with a new configured NSSAI for the current PLMN or SNPN and optionally the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN, or contains an NSSRG information IE with a new NSSRG information, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge the successful update of the network slicing information. If the UE has set the RCMAN bit to "Sending of REGISTRATION COMPLETE message for NSAG information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if REGISTRATION ACCEPT message contains the NSAG information IE, the UE shall return REGISTRATION COMPLETE message to the AMF to acknowledge the successful update of the REGISTRATION ACCEPT message contains the NSAG information IE, the UE shall return REGISTRATION COMPLETE message to the AMF to acknowledge the reception of the NSAG information IE.

If the REGISTRATION ACCEPT message contains the CAG information list IE or the Extended CAG information list IE and the UE had set the CAG bit to "CAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the UE shall:

- a) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
- b) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 9: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
- c) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

The UE shall store the "CAG information list" received in the CAG information list IE or the Extended CAG information list IE as specified in annex C.

If the received "CAG information list" includes an entry containing the identity of the registered PLMN, the UE shall operate as follows.

- a) if the UE receives the REGISTRATION ACCEPT message via a CAG cell, the entry for the registered PLMN in the received "CAG information list" does not include any of the CAG-ID(s) supported by the current CAG cell, and:
 - the entry for the registered PLMN in the received "CAG information list" does not include an "indication that the UE is only allowed to access 5GS via CAG cells", then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list"; or
 - 2) the entry for the registered PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:

- i) if the entry for the registered PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
- ii) if the entry for the registered PLMN in the received "CAG information list" does not include any CAG-ID and:
 - A) the UE does not have an emergency PDU session, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
 - B) the UE has an emergency PDU session, then the UE shall perform a local release of all PDU sessions associated with 3GPP access except for the emergency PDU session and enter the state 5GMM-REGISTERED.LIMITED-SERVICE; or
- b) if the UE receives the REGISTRATION ACCEPT message via a non-CAG cell and the entry for the registered PLMN in the received "CAG information list" includes an "indication that the UE is only allowed to access 5GS via CAG cells" and:
 - if the "allowed CAG list" for the registered PLMN in the received "CAG information list" includes one or more CAG-IDs, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated "CAG information list"; or
 - 2) if the entry for the registered PLMN in the received "CAG information list" does not include any CAG-ID and:
 - the UE does not have an emergency PDU session, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
 - ii) the UE has an emergency PDU session, then the UE shall perform a local release of all PDU sessions associated with 3GPP access except for the emergency PDU session and enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

If the received "CAG information list" does not include an entry containing the identity of the registered PLMN and the UE receives the REGISTRATION ACCEPT message via a CAG cell, the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".

If the REGISTRATION ACCEPT message contains the Operator-defined access category definitions IE, the Extended emergency number list IE, the CAG information list IE or the Extended CAG information list IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the operator-defined access category definitions or the extended local emergency numbers list or the CAG information list.

If the UE has set the RCMAP bit to "Sending of REGISTRATION COMPLETE message for negotiated PEIPS assistance information supported " in the 5GMM capability IE of the REGISTRATION REQUEST message and if REGISTRATION ACCEPT message contains the Negotiated PEIPS assistance parameters IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the Negotiated PEIPS assistance parameters IE.

If the REGISTRATION ACCEPT message contains the UE radio capability ID IE or the UE radio capability ID deletion indication IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the UE radio capability ID IE or the UE radio capability ID deletion indication IE.

If the T3448 value IE is present in the received REGISTRATION ACCEPT message and the value indicates that this timer is neither zero nor deactivated, the UE shall:

- a) stop timer T3448 if it is running; and
- b) start timer T3448 with the value provided in the T3448 value IE.

If the UE is using 5GS services with control plane CIoT 5GS optimization, the T3448 value IE is present in the REGISTRATION ACCEPT message and the value indicates that this timer is either zero or deactivated, the UE shall ignore the T3448 value IE and proceed as if the T3448 value IE was not present.

If the UE in 5GMM-IDLE mode initiated the registration procedure for mobility and periodic registration update and the REGISTRATION ACCEPT message does not include the T3448 value IE and if timer T3448 is running, then the UE shall stop timer T3448.

Upon receiving a REGISTRATION COMPLETE message, the AMF shall stop timer T3550 and change to state 5GMM-REGISTERED. The 5G-GUTI, if sent in the REGISTRATION ACCEPT message, shall be considered as valid, the PEIPS assistance information, if sent in the REGISTRATION ACCEPT message, shall be considered as valid, and the UE radio capability ID, if sent in the REGISTRATION ACCEPT message, shall be considered as valid.

If the 5GS update type IE was included in the REGISTRATION REQUEST message with the SMS requested bit set to "SMS over NAS supported" and:

- a) the SMSF address is stored in the UE 5GMM context and:
 - 1) the UE is considered available for SMS over NAS; or
 - 2) the UE is considered not available for SMS over NAS and the SMSF has confirmed that the activation of the SMS service is successful; or
- b) the SMSF address is not stored in the UE 5GMM context, the SMSF selection is successful and the SMSF has confirmed that the activation of the SMS service is successful;

then the AMF shall set the SMS allowed bit of the 5GS registration result IE in the REGISTRATION ACCEPT message as specified in subclause 5.5.1.2.4. If the UE 5GMM context does not contain an SMSF address or the UE is not considered available for SMS over NAS, then the AMF shall:

- a) store the SMSF address in the UE 5GMM context if not stored already; and
- b) store the value of the SMS allowed bit of the 5GS registration result IE in the UE 5GMM context and consider the UE available for SMS over NAS.

If SMSF selection in the AMF or SMS activation via the SMSF is not successful, or the AMF does not allow the use of SMS over NAS, then the AMF shall set the SMS allowed bit of the 5GS registration result IE to "SMS over NAS not allowed" in the REGISTRATION ACCEPT message.

If the 5GS update type IE was included in the REGISTRATION REQUEST message with the SMS requested bit set to "SMS over NAS not supported" or the 5GS update type IE was not included in the REGISTRATION REQUEST message, then the AMF shall:

- a) mark the 5GMM context to indicate that the UE is not available for SMS over NAS; and
- NOTE 10: The AMF can notify the SMSF that the UE is deregistered from SMS over NAS based on local configuration.
- b) set the SMS allowed bit of the 5GS registration result IE to "SMS over NAS not allowed" in the REGISTRATION ACCEPT message.

When the UE receives the REGISTRATION ACCEPT message, if the UE is also registered over another access to the same PLMN, the UE considers the value indicated by the SMS allowed bit of the 5GS registration result IE as applicable for both accesses over which the UE is registered.

If the 5GS update type IE was included in the REGISTRATION REQUEST message with the NG-RAN-RCU bit set to "UE radio capability update needed", the AMF shall delete the stored UE radio capability information or the UE radio capability ID, if any.

The AMF shall include the 5GS registration result IE in the REGISTRATION ACCEPT message. If the 5GS registration result IE value indicates:

- a) "3GPP access", the UE:
 - shall consider itself as being registered to 3GPP access only; and
 - if in 5GMM-REGISTERED state over non-3GPP access and on the same PLMN as 3GPP access, shall enter state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION over non-3GPP access and set the 5GS update status to 5U2 NOT UPDATED over non-3GPP access;

- b) "Non-3GPP access", the UE:
 - shall consider itself as being registered to non-3GPP access only; and
 - if in the 5GMM-REGISTERED state over 3GPP access and is on the same PLMN as non-3GPP access, shall enter the state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION over 3GPP access and set the 5GS update status to 5U2 NOT UPDATED over 3GPP access; or
- c) "3GPP access and Non-3GPP access", the UE shall consider itself as being registered to both 3GPP access and non-3GPP access.

If the UE is not currently registered for emergency services and the 5GS registration result IE value in the REGISTRATION ACCEPT message is set to "Registered for emergency services", the UE shall consider itself registered for emergency services and shall locally release all non-emergency PDU sessions, if any.

In roaming scenarios, the AMF shall provide mapped S-NSSAI(s) for the configured NSSAI, the allowed NSSAI, the rejected NSSAI (if Extended rejected NSSAI IE is used), the pending NSSAI or NSSRG information when included in the REGISTRATION ACCEPT message.

The AMF shall include the allowed NSSAI for the current PLMN or SNPN and, in roaming scenarios, shall include the mapped S-NSSAI(s) for the allowed NSSAI contained in the requested NSSAI (i.e. Requested NSSAI IE or Requested mapped NSSAI IE) from the UE, in the REGISTRATION ACCEPT message if the UE included the requested NSSAI in the REGISTRATION REQUEST message and the AMF allows one or more S-NSSAIs for the current PLMN or SNPN in the Requested NSSAI IE or one or more mapped S-NSSAIs in the Requested NSSAI IE or Requested mapped NSSAI IE. Additionally, if the AMF allows one or more subscribed S-NSSAIs for the UE, the AMF may include the allowed subscribed S-NSSAI(s) in the allowed NSSAI in the REGISTRATION ACCEPT message. The S-NSSAI associated with each of the active PDN connections for which interworking to 5GS is supported, shall be included in the allowed NSSAI if the UE included the UE status IE with the EMM registration status set to "UE is in EMM-REGISTERED state" in the REGISTRATION REQUEST message and the AMF supports N26 interface.

The AMF may also include rejected NSSAI in the REGISTRATION ACCEPT message if the UE is not registered for onboarding services in SNPN. If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected NSSAI shall be included in the Extended rejected NSSAI IE in the REGISTRATION ACCEPT message; otherwise the rejected NSSAI shall be included in the Registered for onboarding services in SNPN, the AMF shall not include rejected NSSAI in the REGISTRATION ACCEPT message.

If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected NSSAI contains S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with rejection cause(s); otherwise the rejected NSSAI contains S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with rejection cause(s) with the following restrictions:

- a) rejected NSSAI for the current PLMN or SNPN shall not include an S-NSSAI for the current PLMN or SNPN which is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are not allowed; and
- b) rejected NSSAI for the current registration area shall not include an S-NSSAI for the current PLMN or SNPN which is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are not allowed.
- NOTE 11: The UE that does not support extended rejected NSSAI can avoid requesting an S-NSSAI associated with a mapped S-NSSAI, which was included in the previous requested NSSAI but neither in the allowed NSSAI nor in the rejected NSSAI in the consequent registration procedures.

If the UE indicated the support for network slice-specific authentication and authorization, and if the requested NSSAI (i.e. the Requested NSSAI IE) includes one or more S-NSSAIs subject to network slice-specific authentication and authorization, the AMF shall in the REGISTRATION ACCEPT message include:

- a) the allowed NSSAI containing the S-NSSAI(s) or the mapped S-NSSAI(s), if any:
 - i) which are not subject to network slice-specific authentication and authorization and are allowed by the AMF; or
 - ii) for which the network slice-specific authentication and authorization has been successfully performed;

- b) optionally, the rejected NSSAI;
- c) pending NSSAI containing one or more S-NSSAIs for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing, and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing, if any; and
- d) the "NSSAA to be performed" indicator in the 5GS registration result IE set to indicate that the network slicespecific authentication and authorization procedure will be performed by the network, if the allowed NSSAI is not included in the REGISTRATION ACCEPT message.

If the UE is not registered for onboarding services in SNPN, the UE indicated the support for network slice-specific authentication and authorization, and:

- a) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed;
- b) all default S-NSSAIs are subject to network slice-specific authentication and authorization; and
- c) the network slice-specific authentication and authorization procedure has not been successfully performed for any of the default S-NSSAIs,

the AMF shall in the REGISTRATION ACCEPT message include:

- a) the "NSSAA to be performed" indicator in the 5GS registration result IE to indicate that the network slicespecific authentication and authorization procedure will be performed by the network; and
- b) pending NSSAI containing one or more default S-NSSAIs for which network slice-specific authentication and authorization will be performed or is ongoing and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing (if any); and
- c) optionally, the rejected NSSAI.

If the UE is not registered for onboarding services in SNPN, the UE indicated the support for network slice-specific authentication and authorization, and:

- a) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed; and
- b) one or more default S-NSSAIs are not subject to network slice-specific authentication and authorization or the network slice-specific authentication and authorization procedure has been successfully performed for one or more default S-NSSAIs;

the AMF shall in the REGISTRATION ACCEPT message include:

- a) pending NSSAI containing one or more default S-NSSAIs for which network slice-specific authentication and authorization will be performed or is ongoing (if any) and one or more S-NSSAIs from the pending NSSAI which the AMF provided to the UE during the previous registration procedure for which network slice-specific authentication and authorization will be performed or is ongoing (if any);
- b) allowed NSSAI containing S-NSSAI(s) for the current PLMN or SNPN each of which corresponds to a default S-NSSAI which are not subject to network slice-specific authentication and authorization or for which the network slice-specific authentication has been successfully performed;
- c) allowed NSSAI containing one or more default S-NSSAIs, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, which are not subject to network slice-specific authentication and authorization or for which the network slice-specific authentication and authorization has been successfully performed; and
- d) optionally, the rejected NSSAI.

If the UE did not include the requested NSSAI in the REGISTRATION REQUEST message or none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed, the allowed NSSAI shall not contain default S-NSSAI(s) that are subject to NSAC. If the subscription information includes the NSSRG information, the S-NSSAIs of the allowed NSSAI shall be associated with at least one common NSSRG value.

When the REGISTRATION ACCEPT includes a pending NSSAI, the pending NSSAI shall contain all S-NSSAIs for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing from the requested NSSAI of the REGISTRATION REQUEST message that was received over the 3GPP access, non-3GPP access, or both the 3GPP access and non-3GPP access.

If the UE supports extended rejected NSSAI and the AMF determines that maximum number of UEs reached for all S-NSSAIs in the requested NSSAI as specified in subclause 4.6.2.5, the AMF shall include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE in the REGISTRATION ACCEPT message. In addition, the AMF may include a backoff timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" included in the Extended rejected NSSAI IE of the REGISTRATION ACCEPT message. To avoid that large numbers of UEs simultaneously initiate deferred requests, the network should select the value for the backoff timer for each S-NSSAI for the informed UEs so that timeouts are not synchronised.

If the UE does not indicate support for extended rejected NSSAI and the maximum number of UEs has been reached, the AMF should include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available in the current registration area" in the Rejected NSSAI IE and should not include these S-NSSAIs in the allowed NSSAI in the REGISTRATION ACCEPT message.

NOTE 12: Based on network policies, the AMF can include the S-NSSAI(s) for which the maximum number of UEs has been reached in the rejected NSSAI with rejection causes other than "S-NSSAI not available in the current registration area".

If the AMF has a new configured NSSAI for the current PLMN or SNPN, the AMF shall include the configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message.

NOTE 14A: A new configured NSSAI can be available at the AMF following an indication that the subscription data for network slicing has changed.

The AMF may include a new configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message if:

- a) the REGISTRATION REQUEST message did not include a requested NSSAI and the UE is not registered for onboarding services in SNPN;
- b) the REGISTRATION REQUEST message included a requested NSSAI containing an S-NSSAI that is not valid in the serving PLMN or SNPN;
- c) the REGISTRATION REQUEST message included a requested NSSAI containing an S-NSSAI with incorrect mapped S-NSSAI(s);
- d) the REGISTRATION REQUEST message included the Network slicing indication IE with the Default configured NSSAI indication bit set to "Requested NSSAI created from default configured NSSAI";
- e) the REGISTRATION REQUEST message included the requested mapped NSSAI;
- f) the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message are not associated with any common NSSRG value, except for the case that the AMF, based on the indication received from the UDM as specified in 3GPP TS 23.501 [8], has provided all subscribed S-NSSAIs in the configured NSSAI to a UE who does not support NSSRG; or
- NOTE 13:If the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message are not associated with any common NSSRG value, it is possible that at least one of the S-NSSAIs is not included in any of new allowed NSSAI, new (extended) rejected NSSAI (if applicable), and new pending NSSAI (if applicable).
- g) the S-NSSAIs of the requested NSSAI in the REGISTRATION REQUEST message over the current access and the allowed NSSAI over the other access are not associated with any common NSSRG value.

If a new configured NSSAI for the current PLMN or SNPN is included and the UE is roaming, the AMF shall also include the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN in the REGISTRATION ACCEPT message. In this case the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If a new configured NSSAI for the current PLMN or SNPN is included, the subscription information includes the NSSRG information, and the NSSRG bit in the 5GMM capability IE of the REGISTRATION REQUEST message is set to:

- a) "NSSRG supported", then the AMF shall include the NSSRG information in the REGISTRATION ACCEPT message; or
- b) "NSSRG not supported", then the configured NSSAI shall include S-NSSAIs each of which is associated with all the NSSRG value(s) of the default S-NSSAI(s), or the configured NSSAI shall include, based on the indication received from the UDM as specified in 3GPP TS 23.501 [8], all subscribed S-NSSAIs even if these S-NSSAIs do not share any common NSSRG value.

If the AMF needs to update the NSSRG information and the UE has set the NSSRG bit to "NSSRG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, then the AMF shall include the new NSSRG information in the REGISTRATION ACCEPT message. In addition, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

The AMF shall include the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed" in the REGISTRATION ACCEPT message if the UDM has indicated that the subscription data for network slicing has changed. In this case the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.3.

If the S-NSSAI(s) associated with the existing PDU session(s) of the UE is not included in the requested NSSAI (i.e. Requested NSSAI IE or Requested mapped NSSAI IE) of the REGISTRATION REQUEST message, the AMF shall perform a local release of the PDU session(s) associated with the S-NSSAI(s) except for a PDU session associated with DNN and S-NSSAI in the AMF onboarding configuration data and shall request the SMF to perform a local release of those PDU session(s).

The UE that has indicated the support for network slice-specific authentication and authorization receiving the pending NSSAI in the REGISTRATION ACCEPT message shall store the S-NSSAI(s) in the pending NSSAI as specified in subclause 4.6.2.2. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the UE shall store the received pending NSSAI for each of the equivalent PLMNs as specified in subclause 4.6.2.2. If the pending NSSAI is not included in the REGISTRATION ACCEPT message and the "NSSAA to be performed" indicator is not set to "Network slice-specific authentication and authorization is to be performed" in the 5GS registration result IE of the REGISTRATION ACCEPT message, then the UE shall delete the pending NSSAI for the current PLMN or SNPN and its equivalent PLMN(s), if existing, as specified in subclause 4.6.2.2.

The UE receiving the rejected NSSAI in the REGISTRATION ACCEPT message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

Unless the back-off timer value received along with the S-NSSAI is zero, the UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

NOTE 14:If the back-off timer value received along with the S-NSSAI in the rejected NSSAI for the maximum number of UEs reached is zero as specified in subclause 10.5.7.4a of TS 24.008, the UE does not consider the S-NSSAI as the rejected S-NSSAI.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if a back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or
 - 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

If the UE sets the NSSAA bit in the 5GMM capability IE to "Network slice-specific authentication and authorization not supported", and:

- a) if the Requested NSSAI IE only includes the S-NSSAI(s) subject to network slice-specific authentication and authorization and one or more default S-NSSAIs (containing one or more S-NSSAIs each of which may be associated with a new S-NSSAI) which are not subject to network slice-specific authentication and authorization are available, the AMF shall in the REGISTRATION ACCEPT message include:
 - 1) the allowed NSSAI containing S-NSSAI(s) for the current PLMN or SNPN each of which corresponds to a default S-NSSAI which are not subject to network slice-specific authentication and authorization;
 - 2) the allowed NSSAI containing the default S-NSSAIs, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, which are not subject to network slice-specific authentication and authorization; and
 - 3) the rejected NSSAI containing the S-NSSAI(s) subject to network slice specific authentication and authorization with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN", except if the UE has not set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the S-NSSAI(s) is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are subject to NSSAA; or
- b) if the Requested NSSAI IE includes one or more S-NSSAIs subject to network slice-specific authentication and authorization, the AMF shall in the REGISTRATION ACCEPT message include:
 - 1) the allowed NSSAI containing the S-NSSAI(s) or the mapped S-NSSAI(s) which are not subject to network slice-specific authentication and authorization; and
 - 2) the rejected NSSAI containing:
 - i) the S-NSSAI(s) subject to network slice specific authentication and authorization with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN", except if the UE has not set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the S-NSSAI(s) is associated to multiple mapped S-NSSAIs and some of these but not all mapped S-NSSAIs are subject to NSSAA; and
 - ii) the S-NSSAI(s) which was included in the requested NSSAI but rejected by the network associated with the rejection cause indicating "S-NSSAI not available in the current PLMN or SNPN" or the rejection cause indicating "S-NSSAI not available in the current registration area", if any.

For a REGISTRATION REQUEST message with a 5GS registration type IE indicating "mobility registration updating", if the UE does not indicate support for network slice-specific authentication and authorization, the UE is not registered for onboarding services in SNPN, and:

a) the UE is not in NB-N1 mode; and

b) if:

- 1) the UE did not include the requested NSSAI in the REGISTRATION REQUEST message; or
- 2) none of the S-NSSAIs in the requested NSSAI in the REGISTRATION REQUEST message are allowed;

and one or more default S-NSSAIs which are not subject to network slice-specific authentication and authorization are available, the AMF shall:

- a) put the allowed S-NSSAI(s) for the current PLMN or SNPN each of which corresponds to a default S-NSSAI and not subject to network slice-specific authentication and authorization in the allowed NSSAI of the REGISTRATION ACCEPT message;
- b) put the default S-NSSAIs and not subject to network slice-specific authentication and authorization, as the mapped S-NSSAI(s) for the allowed NSSAI in roaming scenarios, in the allowed NSSAI of the REGISTRATION ACCEPT message; and
- c) determine a registration area such that all S-NSSAIs of the allowed NSSAI are available in the registration area.

During a registration procedure for mobility and periodic registration update for which the 5GS registration type IE indicates:

- a) "periodic registration updating"; or
- b) "mobility registration updating" and the UE is in NB-N1 mode;

and the UE is not registered for onboarding services in SNPN, the AMF:

- a) may provide a new allowed NSSAI to the UE;
- b) shall provide a pending NSSAI to the UE if the UE has indicated the support for network slice-specific authentication and authorization and there are S-NSSAIs for which network slice-specific authentication and authorization (except for re-NSSAA) will be performed or is ongoing for the current PLMN or SNPN; or
- c) may provide both a new allowed NSSAI and a pending NSSAI to the UE;

in the REGISTRATION ACCEPT message. Additionally, if a pending NSSAI is provided without an allowed NSSAI and no S-NSSAI is currently allowed for the UE, the REGISTRATION ACCEPT message shall include the 5GS registration result IE with the "NSSAA to be performed" indicator set to "Network slice-specific authentication and authorization is to be performed".

If the REGISTRATION ACCEPT message contains the Network slicing indication IE with the Network slicing subscription change indication set to "Network slicing subscription changed", the UE shall delete the network slicing information for each and every PLMN except for the current PLMN as specified in subclause 4.6.2.2.

If the REGISTRATION ACCEPT message contains the allowed NSSAI, then the UE shall store the included allowed NSSAI together with the PLMN identity of the registered PLMN or the SNPN identity of the registered SNPN and the registration area as specified in subclause 4.6.2.2. If the registration area contains TAIs belonging to different PLMNs, which are equivalent PLMNs, the UE shall store the received allowed NSSAI in each of allowed NSSAIs which are associated with each of the PLMNs.

For each of the PDU session(s) active in the UE:

- if the allowed NSSAI contains an HPLMN S-NSSAI (e.g. mapped S-NSSAI, in roaming scenarios) matching to the HPLMN S-NSSAI of the PDU session, the UE shall locally update the S-NSSAI associated with the PDU session to the corresponding S-NSSAI received in the allowed NSSAI; and
- if the allowed NSSAI does not contain an HPLMN S-NSSAI (e.g. mapped S-NSSAI, in roaming scenarios) matching to the HPLMN S-NSSAI of the PDU session, the UE may perform a local release of the PDU session

except for an emergency PDU session, if any, and except for a PDU session established when the UE is registered for onboarding services in SNPN, if any.

NOTE 15: According to 3GPP TS 23.501 [8], also the AMF will determine which PDU sessions can no longer be supported based on the new allowed NSSAI, and it will cause a release on the UE side either by indicating in the PDU session status IE which PDU sessions are inactive on the network side or by triggering the SMF to initiate a release via 5GSM signalling.

If the REGISTRATION ACCEPT message contains a configured NSSAI IE with a new configured NSSAI for the current PLMN or SNPN and optionally the mapped S-NSSAI(s) for the configured NSSAI for the current PLMN or SNPN, the UE shall store the contents of the configured NSSAI IE as specified in subclause 4.6.2.2. In addition, if the REGISTRATION ACCEPT message contains an NSSRG information IE, the UE shall store the contents of the NSSRG information IE as specified in subclause 4.6.2.2.

If the UE has set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message over 3GPP access, the AMF may include the NSAG information IE in the REGISTRATION ACCEPT message. Up to 4 NSAG entries are allowed to be associated with a TAI list in the NSAG information IE. If the UE has set the RCMAN bit to "Sending of REGISTRATION COMPLETE message for NSAG information supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and if the NSAG information IE is included in the REGISTRATION ACCEPT message, the AMF shall start timer T3550 and enter state 5GMM-COMMON-PROCEDURE-INITIATED as described in subclause 5.1.3.2.3.

- NOTE 15a: How the AMF selects NSAG entries to be included in the NSAG information IE is implementation specific, e.g. take the NSAG priority and the current registration area into account.
- NOTE 15b: If the NSAG for the PLMN and its equivalent PLMN(s) have different associations with S-NSSAIs, then the AMF includes a TAI list for the NSAG entry in the NSAG information IE.

If the UE receives the NSAG information IE in the REGISTRATION ACCEPT message, the UE shall store the NSAG information as specified in subclause 4.6.2.2.

If the REGISTRATION ACCEPT message:

- a) includes the 5GS registration result IE with the "NSSAA to be performed" indicator set to "Network slice-specific authentication and authorization is to be performed";
- b) includes a pending NSSAI; and
- c) does not include an allowed NSSAI;

the UE:

- a) shall not perform the registration procedure for mobility and registration update with the Uplink data status IE except for emergency services;
- b) shall not initiate a service request procedure except for emergency services, for responding to paging or notification over non-3GPP access, for cases f), i), m) and o) in subclause 5.6.1.1;
- c) shall not initiate a 5GSM procedure except for emergency services, indicating a change of 3GPP PS data off UE status, or to request the release of a PDU session; and
- d) shall not initiate the NAS transport procedure except for sending a CIoT user data container, SMS, an LPP message, a location services message, an SOR transparent container, a UE policy container or a UE parameters update transparent container;

until the UE receives an allowed NSSAI.

During a registration procedure for mobility and periodic registration update for which the 5GS registration type IE indicates:

- a) "mobility registration updating" and the UE is in NB-N1 mode; or
- b) "periodic registration updating";

if the REGISTRATION ACCEPT message includes the 5GS registration result IE with the "NSSAA to be performed" indicator not set to "Network slice-specific authentication and authorization is to be performed" and the message does not contain an allowed NSSAI and no new allowed NSSAI, the UE shall consider the previously received allowed NSSAI as valid.

During a registration procedure for mobility and periodic registration update for which the 5GS registration type IE indicates:

- a) "mobility registration updating"; or
- b) "periodic registration updating";

if the REGISTRATION ACCEPT message includes the 5GS registration result IE with the "NSSAA to be performed" indicator set to "Network slice-specific authentication and authorization is to be performed" and the message contains a pending NSSAI, the UE shall delete any stored allowed NSSAI as specified in subclause 4.6.2.2.

If the Uplink data status IE is included in the REGISTRATION REQUEST message:

- a) if the AMF determines that the UE is in non-allowed area or is not in allowed area, and the PDU session(s) indicated by the Uplink data status IE is non-emergency PDU session(s) or the UE is not configured for high priority access in selected PLMN, the AMF shall include the PDU session reactivation result IE in the REGISTRATION ACCEPT message indicating that user-plane resources for the corresponding PDU session(s) cannot be re-established, and shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #28 "Restricted service area";
- b) otherwise, the AMF shall:
 - 1) indicate the SMF to re-establish the user-plane resources for the corresponding PDU session;
 - 2) include PDU session reactivation result IE in the REGISTRATION ACCEPT message to indicate the userplane resources re-establishment result of the PDU sessions for which the UE requested to re-establish the user-plane resources; and
 - 3) determine the UE presence in LADN service area and forward the UE presence in LADN service area towards the SMF, if the corresponding PDU session is a PDU session for LADN.

If the Uplink data status IE is not included in the REGISTRATION REQUEST message and the REGISTRATION REQUEST message is sent for the trigger d) in subclause 5.5.1.3.2, the AMF may indicate the SMF to re-establish the user-plane resources for the PDU sessions.

If a PDU session status IE is included in the REGISTRATION REQUEST message:

- a) for single access PDU sessions, the AMF shall:
 - perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE on the AMF side associated with the access type the REGISTRATION REQUEST message is sent over, but are indicated by the UE as being in 5GSM state PDU SESSION INACTIVE. If any of those PDU sessions is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and
 - include a PDU session status IE in the REGISTRATION ACCEPT message to indicate which PDU sessions associated with the access type the REGISTRATION ACCEPT message is sent over are not in 5GSM state PDU SESSION INACTIVE in the AMF; and
- b) for MA PDU sessions:
 - for all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE and have user plane resources established on the access the REGISTRATION REQUEST message is sent over on the AMF side, but are indicated by the UE as no user plane resources established:
 - i) for PDU sessions having user plane resources established only on the access the REGISTRATION REQUEST message is sent over, the AMF shall perform a local release of all those PDU sessions. If the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and

- ii) for PDU sessions having user plane resources established on both accesses, the AMF shall perform a local release on the user plane resources associated with the access type the REGISTRATION REQUEST message is sent over. If the REGISTRATION REQUEST message is sent over 3GPP access and the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and
- the AMF shall include a PDU session status IE in the REGISTRATION ACCEPT message to indicate which MA PDU sessions having user plane resources established on the AMF side on the access the REGISTRATION ACCEPT message is sent over.

If the Allowed PDU session status IE is included in the REGISTRATION REQUEST message, the AMF shall:

- a) for a 5GSM message from each SMF that has indicated pending downlink signalling only, forward the received 5GSM message via 3GPP access to the UE after the REGISTRATION ACCEPT message is sent;
- b) for each SMF that has indicated pending downlink data only:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE; and
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE.
- c) for each SMF that have indicated pending downlink signalling and data:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE;
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE; and
 - 3) discard the received 5GSM message for PDU session(s) associated with non-3GPP access; and
- d) include the PDU session reactivation result IE in the REGISTRATION ACCEPT message to indicate the successfully re-established user-plane resources for the corresponding PDU sessions, if any.

If the PDU session reactivation result IE is included in the REGISTRATION ACCEPT message indicating that the user-plane resources have been successfully reactivated for a PDU session that was requested by the UE in the Allowed PDU session status IE, the UE considers the corresponding PDU session to be associated with the 3GPP access. If the user-plane resources of a PDU session have been successfully reactivated over the 3GPP access, the AMF and SMF update the associated access type of the corresponding PDU session.

If the PDU session reactivation result IE is included in the REGISTRATION ACCEPT message indicating that the user-plane resources cannot be established for a PDU session that was requested by the UE in the Allowed PDU session status IE, the UE considers the corresponding PDU session to be associated with the non-3GPP access.

If an EPS bearer context status IE is included in the REGISTRATION REQUEST message, the AMF handles the received EPS bearer context status IE as specified in 3GPP TS 23.502 [9].

If the EPS bearer context status information is generated for the UE during the inter-system change from S1 mode to N1 mode as specified in 3GPP TS 23.502 [9] and the AMF supports N26 interface, the AMF shall include an EPS bearer context status IE in the REGISTRATION ACCEPT message to indicate the UE which mapped EPS bearer contexts are active in the network.

If the user-plane resources cannot be established for a PDU session, the AMF shall include the PDU session reactivation result IE in the REGISTRATION ACCEPT message indicating that user-plane resources for the corresponding PDU session cannot be re-established, and:

a) if the user-plane resources cannot be established because the SMF indicated to the AMF that the UE is located out of the LADN service area (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #43 "LADN not available";

- b) if the user-plane resources cannot be established because the SMF indicated to the AMF that only prioritized services are allowed (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #28 "restricted service area";
- c) if the user-plane resources cannot be established because the SMF indicated to the AMF that the resource is not available in the UPF (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #92 "insufficient user-plane resources for the PDU session";
- d) if the user-plane resources cannot be established because the SMF indicated to the AMF that the S-NSSAI associated with the PDU session is unavailable due to NSAC (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #69 "insufficient resources for specific slice"; or
- e) otherwise, the AMF may include the PDU session reactivation result error cause IE to indicate the cause of failure to re-establish the user-plane resources.
- NOTE 16:It is up to UE implementation when to re-send a request for user-plane re-establishment for the associated PDU session after receiving a PDU session reactivation result error cause IE with a 5GMM cause set to #92 "insufficient user-plane resources for the PDU session".
- NOTE 17: The UE can locally start a back-off timer after receiving a PDU session reactivation result error cause IE with a 5GMM cause set to #69 "insufficient resources for specific slice". The value of the back-off timer is up to UE implementation. Upon expiry of the back-off timer, the UE can re-send a request for user-plane re-establishment for the associated PDU session.

If the AMF needs to initiate PDU session status synchronization the AMF shall include a PDU session status IE in the REGISTRATION ACCEPT message to indicate the UE:

- which single access PDU sessions associated with the access the REGISTRATION ACCEPT message is sent over are not in 5GSM state PDU SESSION INACTIVE in the AMF; and
- which MA PDU sessions are not in 5GSM state PDU SESSION INACTIVE and having user plane resources established in the AMF on the access the REGISTRATION ACCEPT message is sent over.

The AMF may include the LADN information IE in the REGISTRATION ACCEPT message as described in subclause 5.5.1.2.4. The UE, upon receiving the REGISTRATION ACCEPT message with the LADN information IE, shall delete its old LADN information (if any) and store the received new LADN information.

If the AMF does not include the LADN information IE in the REGISTRATION ACCEPT message during registration procedure for mobility and registration update, the UE shall delete its old LADN information.

If the PDU session status IE is included in the REGISTRATION ACCEPT message:

- a) for single access PDU sessions, the UE shall perform a local release of all those PDU sessions associated with the access type the REGISTRATION ACCEPT message is sent over which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING on the UE side, but are indicated by the AMF as being in 5GSM state PDU SESSION INACTIVE. If a locally released PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions; and
- b) for MA PDU sessions, for all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING and have user plane resources established in the UE on the access the REGISTRATION ACCEPT message is sent over, but are indicated by the AMF as no user plane resources established:
 - for MA PDU sessions having user plane resources established only on the access the REGISTRATION ACCEPT message is sent over, the UE shall perform a local release of those MA PDU sessions. If a locally released MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions; and
 - 2) for MA PDU sessions having user plane resources established on both accesses, the UE shall perform a local release on the user plane resources on the access the REGISTRATION ACCEPT message is sent over. If the user plane resources over 3GPP access are released and the MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions.
- a) the UE included a PDU session status IE in the REGISTRATION REQUEST message;
- b) the UE is operating in the single-registration mode;
- c) the UE is performing inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode; and
- d) the UE has received the IWK N26 bit set to "interworking without N26 interface supported";

the UE shall ignore the PDU session status IE if received in the REGISTRATION ACCEPT message.

If the EPS bearer context status IE is included in the REGISTRATION ACCEPT message, the UE shall locally delete all those QoS flow descriptions and all associated QoS rules, if any, which are associated with inactive EPS bearer contexts as indicated by the AMF in the EPS bearer context status IE.

If the UE included S1 mode supported indication in the REGISTRATION REQUEST message, the AMF supporting inter-system change with EPS shall set the IWK N26 bit to either:

- a) "interworking without N26 interface not supported" if the AMF supports N26 interface; or
- b) "interworking without N26 interface supported" if the AMF does not support N26 interface

in the 5GS network feature support IE in the REGISTRATION ACCEPT message.

The UE supporting S1 mode shall operate in the mode for inter-system interworking with EPS as follows:

- a) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface not supported", the UE shall operate in single-registration mode;
- b) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface supported" and the UE supports dual-registration mode, the UE may operate in dual-registration mode; or

NOTE 18: The registration mode used by the UE is implementation dependent.

c) if the IWK N26 bit in the 5GS network feature support IE is set to "interworking without N26 interface supported" and the UE only supports single-registration mode, the UE shall operate in single-registration mode.

The UE shall treat the received interworking without N26 interface indicator for inter-system change with EPS as valid in the entire PLMN and its equivalent PLMN(s).

The network informs the UE about the support of specific features, such as IMS voice over PS session, location services (5G-LCS), emergency services, emergency services fallback and ATSSS, in the 5GS network feature support information element. In a UE with IMS voice over PS session capability, the IMS voice over PS session indicator, Emergency services support indicator and Emergency services fallback indicator shall be provided to the upper layers. The upper layers take the IMS voice over PS session indicator into account when selecting the access domain for voice sessions or calls. When initiating an emergency call, the upper layers take the IMS voice over PS session indicator, Emergency services support indicator and Emergency services fallback indicator into account for the access domain selection. When the UE determines via the IMS voice over PS session indicator that the network does not support IMS voice over PS sessions in N1 mode, then the UE shall not perform a local release of any persistent PDU session if the AMF does not indicate that the PDU session is in 5GSM state PDU SESSION INACTIVE via the PDU session status IE. When the UE determines via the Emergency services support indicator that the network does not support emergency services in N1 mode, then the UE shall not perform a local release of any emergency PDU session if user-plane resources associated with that emergency PDU session are established if the AMF does not indicate that the PDU session is in 5GSM state PDU SESSION INACTIVE via the PDU session status IE. In a UE with LCS capability, location services indicators (5G-LCS) shall be provided to the upper layers. In a UE with the capability for ATSSS, the network support for ATSSS shall be provided to the upper layers. If the UE receives the 5GS network feature support IE with the ATSSS support indicator set to "ATSSS not supported", the UE shall perform a local release of the MA PDU session, if any. If a locally released MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions.

The AMF shall set the EMF bit in the 5GS network feature support IE to:

 a) "Emergency services fallback supported in NR connected to 5GCN and E-UTRA connected to 5GCN" if the network supports the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN or an E-UTRA cell connected to 5GCN;

- b) "Emergency services fallback supported in NR connected to 5GCN only" if the network supports the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN and does not support the emergency services fallback procedure when the UE is in an E-UTRA cell connected to 5GCN;
- c) "Emergency services fallback supported in E-UTRA connected to 5GCN only" if the network supports the emergency services fallback procedure when the UE is in an E-UTRA cell connected to 5GCN and does not support the emergency services fallback procedure when the UE is in an NR cell connected to 5GCN; or
- d) "Emergency services fallback not supported" if network does not support the emergency services fallback procedure when the UE is in any cell connected to 5GCN.
- NOTE 19:If the emergency services are supported in neither the EPS nor the 5GS homogeneously, based on operator policy, the AMF will set the EMF bit in the 5GS network feature support IE to "Emergency services fallback not supported".
- NOTE 20: Even though the AMF's support of emergency services fallback is indicated per RAT, the UE's support of emergency services fallback is not per RAT, i.e. the UE's support of emergency services fallback is the same for both NR connected to 5GCN and E-UTRA connected to 5GCN.

If the UE is not operating in SNPN access operation mode:

- a) the network informs the UE that the use of access identity 1 is valid in the RPLMN or equivalent PLMN by setting the MPS indicator bit of the 5GS network feature support IE to "Access identity 1 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MPS indicator bit in the REGISTRATION ACCEPT message based on the MPS priority information in the user's subscription context obtained from the UDM;
- b) upon receiving a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 valid", the UE shall act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2, in all NG-RAN of the registered PLMN and its equivalent PLMNs. The MPS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 not valid" or until the UE selects a non-equivalent PLMN. Access identity 1 is only applicable while the UE is in N1 mode;
- c) during ongoing active PDU sessions that were set up relying on the MPS indicator bit being set to "Access identity 1 valid", if the network indicates in a registration update that the MPS indicator bit is reset to "Access identity 1 not valid", then the UE shall no longer act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2 unless the USIM contains a valid configuration for access identity 1 in RPLMN or equivalent PLMN. In the UE, the ongoing active PDU sessions are not affected by the change of the MPS indicator bit;
- d) the network informs the UE that the use of access identity 2 is valid in the RPLMN or equivalent PLMN by setting the MCS indicator bit of the 5GS network feature support IE to "Access identity 2 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MCS indicator bit in the REGISTRATION ACCEPT message based on the MCS priority information in the user's subscription context obtained from the UDM;
- e) upon receiving a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 valid", the UE shall act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2, in all NG-RAN of the registered PLMN and its equivalent PLMNs. The MCS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 not valid" or until the UE selects a non-equivalent PLMN. Access identity 2 is only applicable while the UE is in N1 mode; and
- f) during ongoing active PDU sessions that were set up relying on the MCS indicator bit being set to "Access identity 2 valid", if the network indicates in a registration update that the MCS indicator bit is reset to "Access identity 2 not valid", then the UE shall no longer act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2 unless the USIM contains a valid configuration for access identity 2 in RPLMN or equivalent PLMN. In the UE, the ongoing active PDU sessions are not affected by the change of the MCS indicator bit.

If the UE indicates support for restriction on use of enhanced coverage in the REGISTRATION REQUEST message and:

- a) in WB-N1 mode, the AMF decides to restrict the use of CE mode B for the UE, then the AMF shall set the RestrictEC bit to "CE mode B is restricted";
- b) in WB-N1 mode, the AMF decides to restrict the use of both CE mode A and CE mode B for the UE, then the AMF shall set the RestrictEC bit to " Both CE mode A and CE mode B are restricted"; or
- c) in NB-N1 mode, the AMF decides to restrict the use of enhanced coverage for the UE, then the AMF shall set the RestrictEC bit to "Use of enhanced coverage is restricted",

in the 5GS network feature support IE in the REGISTRATION ACCEPT message.

If the UE is operating in SNPN access operation mode:

- a) the network informs the UE that the use of access identity 1 is valid in the RSNPN by setting the MPS indicator bit of the 5GS network feature support IE to "Access identity 1 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MPS indicator bit in the REGISTRATION ACCEPT message based on the MPS priority information in the user's subscription context obtained from the UDM;
- b) upon receiving a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 valid", the UE shall act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2A, in all NG-RAN of the registered SNPN. The MPS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MPS indicator bit set to "Access identity 1 not valid" or until the UE selects another SNPN. Access identity 1 is only applicable while the UE is in N1 mode;
- c) during ongoing active PDU sessions that were set up relying on the MPS indicator bit being set to "Access identity 1 valid", if the network indicates in a registration update that the MPS indicator bit is reset to "Access identity 1 not valid", then the UE shall no longer act as a UE with access identity 1 configured for MPS as described in subclause 4.5.2A unless the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]) indicates the UE is configured for access identity 1 in the RSNPN. In the UE, the ongoing active PDU sessions are not affected by the change of the MPS indicator bit;
- d) the network informs the UE that the use of access identity 2 is valid in the RSNPN by setting the MCS indicator bit of the 5GS network feature support IE to "Access identity 2 valid", in the REGISTRATION ACCEPT message. Based on operator policy, the AMF sets the MCS indicator bit in the REGISTRATION ACCEPT message based on the MCS priority information in the user's subscription context obtained from the UDM;
- e) upon receiving a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 valid", the UE shall act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2A, in all NG-RAN of the registered SNPN. The MCS indicator bit in the 5GS network feature support IE provided in the REGISTRATION ACCEPT message is valid until the UE receives a REGISTRATION ACCEPT message with the MCS indicator bit set to "Access identity 2 not valid" or until the UE selects another SNPN. Access identity 2 is only applicable while the UE is in N1 mode; and
- f) during ongoing active PDU sessions that were set up relying on the MCS indicator bit being set to "Access identity 2 valid", if the network indicates in a registration update that the MCS indicator bit is reset to "Access identity 2 not valid", then the UE shall no longer act as a UE with access identity 2 configured for MCS as described in subclause 4.5.2A unless the unified access control configuration in the "list of subscriber data" stored in the ME (see 3GPP TS 23.122 [5]) indicates the UE is configured for access identity 2 in the RSNPN. In the UE, the ongoing active PDU sessions are not affected by the change of the MCS indicator bit.

If the UE has set the Follow-on request indicator to "Follow-on request pending" in the REGISTRATION REQUEST message, or the network has downlink signalling pending, the AMF shall not immediately release the NAS signalling connection after the completion of the registration procedure.

If the UE is authorized to use V2X communication over PC5 reference point based on:

- a) at least one of the following bits in the 5GMM capability IE of the REGISTRATION REQUEST message set by the UE, or already stored in the 5GMM context in the AMF during the previous registration procedure as follows:
 - 1) the V2XCEPC5 bit to "V2X communication over E-UTRA-PC5 supported"; or
 - 2) the V2XCNPC5 bit to "V2X communication over NR-PC5 supported"; and

b) the user's subscription context obtained from the UDM as defined in 3GPP TS 23.287 [6C];

the AMF should not immediately release the NAS signalling connection after the completion of the registration procedure.

If the UE is authorized to use 5G ProSe services based on:

- a) at least one of the following bits in the 5GMM capability IE of the REGISTRATION REQUEST message set by the UE, or already stored in the 5GMM context in the AMF during the previous registration procedure as follows:
 - 1) the 5G ProSe direct discovery bit to "5G ProSe direct discovery supported"; or
 - 2) the 5G ProSe direct communication bit to "5G ProSe direct communication supported"; and
- b) the user's subscription context obtained from the UDM as defined in 3GPP TS 23.304 [6E];

the AMF should not immediately release the NAS signalling connection after the completion of the registration procedure.

If the Requested DRX parameters IE was included in the REGISTRATION REQUEST message, the AMF shall include the Negotiated DRX parameters IE in the REGISTRATION ACCEPT message. The AMF may set the Negotiated DRX parameters IE based on the received Requested DRX parameters IE and operator policy if available.

If the Requested NB-N1 mode DRX parameters IE was included in the REGISTRATION REQUEST message, the AMF shall include the Negotiated NB-N1 mode DRX parameters IE in the REGISTRATION ACCEPT message. The AMF may set the Negotiated NB-N1 mode DRX parameters IE based on the received Requested NB-N1 mode DRX parameters IE and operator policy if available.

The AMF shall include the Negotiated extended DRX parameters IE in the REGISTRATION ACCEPT message only if the Requested extended DRX parameters IE was included in the REGISTRATION REQUEST message, and the AMF supports and accepts the use of eDRX. The AMF may set the Negotiated extended DRX parameters IE based on the received Requested extended DRX parameters IE, operator policy, information from NG-RAN and the user's subscription context obtained from the UDM if available.

If the UE included in the REGISTRATION REQUEST message the UE status IE with the EMM registration status set to "UE is in EMM-REGISTERED state" and the AMF does not support N26 interface, the AMF shall operate as described in subclause 5.5.1.2.4.

If the UE has indicated support for service gap control in the REGISTRATION REQUEST message, a service gap time value is available in the 5GMM context, the AMF may include the T3447 value IE set to the service gap time value in the REGISTRATION ACCEPT message.

If the UE requests ciphering keys for ciphered broadcast assistance data in the REGISTRATION REQUEST message and the AMF has valid ciphering key data applicable to the UE's subscription and current tracking area, then the AMF shall include the ciphering key data in the Ciphering key data IE of the REGISTRATION ACCEPT message.

If the UE supports WUS assistance information and the AMF supports and accepts the use of WUS assistance information for the UE, then the AMF shall determine the negotiated UE paging probability information for the UE, store it in the 5GMM context of the UE, and if the UE does not have an active emergency PDU session, the AMF shall include it in the Negotiated WUS assistance information IE in the REGISTRATION ACCEPT message. The AMF may consider the UE paging probability information received in the Requested WUS assistance information IE when determining the negotiated UE paging probability information for the UE.

NOTE 21:Besides the UE paging probability information requested by the UE, the AMF can take local configuration or previous statistical information for the UE into account when determining the negotiated UE paging probability information for the UE.

If the UE sets the NR-PSSI bit to "NR paging subgrouping supported" in the 5GMM capability IE in the REGISTRATION REQUEST message and the AMF supports and accepts the use of PEIPS assistance information for the UE, then the AMF shall determine the Paging subgroup ID for the UE, store it in the 5GMM context of the UE, and include it in the Negotiated PEIPS assistance information IE in the REGISTRATION ACCEPT message or in the Updated PEIPS assistance information IE in the CONFIGURATION UPDATE COMMAND message as part of the registration procedure. The AMF may consider the UE paging probability information received in the Requested PEIPS assistance information IE when determining the Paging subgroup ID for the UE.

NOTE 22: Besides the UE paging probability information when provided by the UE, the AMF can also take local configuration, whether the UE is likely to receive IMS voice over PS session calls, UE mobility pattern or previous statistical information for the UE or information provided by the NG-RAN into account when determining the Paging subgroup ID for the UE.

If due to regional subscription restrictions or access restrictions the UE is not allowed to access the TA or due to CAG restrictions the UE is not allowed to access the cell, but the UE has an emergency PDU session established, the AMF may accept the REGISTRATION REQUEST message and indicate to the SMF to perform a local release of all non-emergency PDU sessions (associated with 3GPP access if it is due to CAG restrictions) and informs the UE via the PDU session status IE in the REGISTRATION ACCEPT message. The AMF shall not indicate to the SMF to release the emergency PDU session. If the AMF indicated to the SMF to perform a local release of all non-emergency PDU sessions (associated with 3GPP access if it is due to CAG restrictions), the network shall behave as if the UE is registered for emergency services and shall set the 5GS registration result IE value to "Registered for emergency services" in the REGISTRATION ACCEPT message.

If the REGISTRATION ACCEPT message includes the PDU session reactivation result error cause IE with the 5GMM cause set to #28 "Restricted service area", the UE shall enter the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE and behave as specified in subclause 5.3.5.

If the REGISTRATION ACCEPT message includes the SOR transparent container IE and:

- a) the SOR transparent container IE does not successfully pass the integrity check (see 3GPP TS 33.501 [24]); and
- b) if the UE attempts obtaining service on another PLMNs or SNPNs as specified in 3GPP TS 23.122 [5] annex C;

then the UE shall release locally the established NAS signalling connection after sending a REGISTRATION COMPLETE message.

If the REGISTRATION ACCEPT message includes the SOR transparent container IE and the SOR transparent container IE successfully passes the integrity check (see 3GPP TS 33.501 [24]), the ME shall store the received SOR counter as specified in annex C and proceed as follows:

- a) the UE shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C; and
- b) if the registration procedure is performed over 3GPP access and the UE attempts obtaining service on another PLMNs or SNPNs as specified in 3GPP TS 23.122 [5] annex C then the UE may release locally the established NAS signalling connection after sending a REGISTRATION COMPLETE message. Otherwise the UE shall send a REGISTRATION COMPLETE message and not release the current N1 NAS signalling connection locally. If an acknowledgement is requested in the SOR transparent container IE of the REGISTRATION ACCEPT message, the UE acknowledgement is included in the SOR transparent container IE of the REGISTRATION COMPLETE message. In the SOR transparent container IE carrying the acknowledgement, the UE shall set the ME support of SOR-CMCI indicator to "SOR-CMCI supported by the ME". Additionally, if the UE supports access to an SNPN using credentials from a credentials holder and the UE is not operating in SNPN access operation mode, the UE may set the ME support of SOR-SNPN-SI indicator to "SOR-SNPN-SI supported by the ME".

If the SOR transparent container IE successfully passes the integrity check (see 3GPP TS 33.501 [24]), and:

 a) the SOR transparent container IE indicates a list of preferred PLMN/access technology combinations is provided and the list type indicates "PLMN ID and access technology list", then the ME shall replace the highest priority entries in the "Operator Controlled PLMN Selector with Access Technology" list stored in the ME and shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C.

If the SOR-CMCI is present and the Store SOR-CMCI in ME indicator is set to "Store SOR-CMCI in ME" then the UE shall store or delete the SOR-CMCI in the non-volatile memory of the ME as described in annex C.1;

- b) the list type indicates "secured packet", then the ME shall behave as if a SMS is received with protocol identifier set to SIM data download, data coding scheme set to class 2 message and SMS payload as secured packet contents of SOR transparent container IE. The SMS payload is forwarded to UICC as specified in 3GPP TS 23.040 [4A]; or
- c) the SOR transparent container IE indicates "HPLMN indication that 'no change of the "Operator Controlled PLMN Selector with Access Technology" list stored in the UE is needed and thus no list of preferred PLMN/access technology combinations is provided", the UE operates in SNPN access operation mode and the SOR transparent container IE includes SOR-SNPN-SI, the ME shall replace SOR-SNPN-SI of the selected entry

of the "list of subscriber data" or associated with the selected PLMN subscription, as specified in 3GPP TS 23.122 [5] with the received SOR-SNPN-SI.

If the SOR-CMCI is present and the Store SOR-CMCI in ME indicator is set to "Store SOR-CMCI in ME" then the UE shall store or delete the SOR-CMCI in the non-volatile memory of the ME as described in annex C.1;

and the UE shall proceed with the behaviour as specified in 3GPP TS 23.122 [5] annex C.

If the SOR transparent container IE does not pass the integrity check successfully, then the UE shall discard the content of the SOR transparent container IE.

If required by operator policy, the AMF shall include the NSSAI inclusion mode IE in the REGISTRATION ACCEPT message (see table 4.6.2.3.1 of subclause 4.6.2.3). Upon receipt of the REGISTRATION ACCEPT message:

- a) if the message includes the NSSAI inclusion mode IE, the UE shall operate in the NSSAI inclusion mode indicated in the NSSAI inclusion mode IE over the current access within the current PLMN or SNPN and its equivalent PLMN(s), if any, in the current registration area; or
- b) otherwise:
 - 1) if the UE has NSSAI inclusion mode for the current PLMN or SNPN and access type stored in the UE, the UE shall operate in the stored NSSAI inclusion mode;
 - 2) if the UE does not have NSSAI inclusion mode for the current PLMN or SNPN and the access type stored in the UE and if the UE is performing the registration procedure over:
 - i) 3GPP access, the UE shall operate in NSSAI inclusion mode D in the current PLMN or SNPN and the current access type;
 - ii) untrusted non-3GPP access, the UE shall operate in NSSAI inclusion mode C in the current PLMN and the current access type; or
 - iii) trusted non-3GPP access, the UE shall operate in NSSAI inclusion mode D in the current PLMN and the current access type; or
 - 3) if the 5G-RG does not have NSSAI inclusion mode for the current PLMN and wireline access stored in the 5G-RG, and the 5G-RG is performing the registration procedure over wireline access, the 5G-RG shall operate in NSSAI inclusion mode B in the current PLMN and the current access type.

The AMF may include operator-defined access category definitions in the REGISTRATION ACCEPT message.

If there is a running T3447 timer in the AMF and the Uplink data status IE is included or the Follow-on request indicator is set to "Follow-on request pending" in the REGISTRATION REQUEST message, the AMF shall ignore the Uplink data status IE or that the Follow-on request indicator is set to "Follow-on request pending" and proceed as if the Uplink data status IE was not received or the Follow-on request indicator was not set to "Follow-on request pending" except for the following case:

- the PDU session(s) indicated by the Uplink data status IE is emergency PDU session(s);
- the UE is configured for high priority access in selected PLMN;
- the REGISTRATION REQUEST message is as a paging response; or
- the UE is establishing an emergency PDU session or performing emergency services fallback.

If the UE receives Operator-defined access category definitions IE in the REGISTRATION ACCEPT message and the Operator-defined access category definitions IE contains one or more operator-defined access category definitions, the UE shall delete any operator-defined access category definitions stored for the RPLMN and shall store the received operator-defined access category definitions for the RPLMN. If the UE receives the Operator-defined access category definitions IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions, the UE shall delete any operator-defined access category definitions IE contains no operator-defined access category definitions, the UE shall delete any operator-defined access category definitions stored for the RPLMN. If the REGISTRATION ACCEPT message does not contain the Operator-defined access category definitions access category definitions access category definitions access category definitions access category definitions. The UE shall delete any operator-defined access category definitions access category

If the UE has indicated support for service gap control in the REGISTRATION REQUEST message and:

- the REGISTRATION ACCEPT message contains the T3447 value IE, then the UE shall store the new T3447 value, erase any previous stored T3447 value if exists and use the new T3447 value with the timer T3447 next time it is started; or
- the REGISTRATION ACCEPT message does not contain the T3447 value IE, then the UE shall erase any previous stored T3447 value if exists and stop the timer T3447 if running.

If the REGISTRATION ACCEPT message contains the Truncated 5G-S-TMSI configuration IE, then the UE shall store the included truncated 5G-S-TMSI configuration and return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the truncated 5G-S-TMSI configuration.

NOTE 23: The UE provides the truncated 5G-S-TMSI configuration to the lower layers.

If the UE is not in NB-N1 mode, the UE has set the RACS bit to "RACS supported" in the 5GMM Capability IE of the REGISTRATION REQUEST message, and the REGISTRATION ACCEPT message includes:

- a) a UE radio capability ID deletion indication IE set to "Network-assigned UE radio capability IDs deletion requested", the UE shall delete any network-assigned UE radio capability IDs associated with the RPLMN or RSNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription stored at the UE, then the UE shall initiate a registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3.2 over the existing N1 NAS signalling connection; or
- b) a UE radio capability ID IE, the UE shall store the UE radio capability ID as specified in annex C.

If the registration procedure for mobility and periodic registration update was initiated and there is a request from the upper layers to perform "emergency services fallback" pending, the UE shall restart the service request procedure after the successful completion of the mobility and periodic registration update.

When AMF re-allocation occurs in the registration procedure for mobility and periodic registration update, if the new AMF receives in the 5GMM context of the UE the indication that the UE is registered for onboarding services in SNPN, the new AMF may start an implementation specific timer for onboarding services when the registration procedure for mobility and periodic registration update is successfully completed.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and the REGISTRATION ACCEPT message contains the service-level-AA pending indication in the Service-level-AA container IE, the UE shall return a REGISTRATION COMPLETE message to the AMF to acknowledge reception of the service-level-AA pending indication, and the UE shall not attempt to perform another registration procedure for UAS services until the UUAA-MM procedure is completed, or to establish a PDU session for USS communication or a PDU session for C2 communication until the UUAA-MM procedure is completed successfully.

If the UE has included the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the REGISTRATION REQUEST message and the REGISTRATION ACCEPT message does not contain the service-level-AA pending indication in the Service-level-AA container IE, the UE shall consider the UUAA-MM procedure is not triggered.

If the UE is registered for onboarding services in SNPN or the network determines that the UE's subscription only allows for configuration of SNPN subscription parameters in PLMN via the user plane, the AMF may start an implementation specific timer for onboarding services, if not running already, when the network considers that the UE is in 5GMM-REGISTERED (i.e. the network receives the REGISTRATION COMPLETE message from UE).

- NOTE 24: If the AMF considers that the UE is in 5GMM-IDLE, when the implementation specific timer for onboarding services expires and the network considers that the UE is still in state 5GMM-REGISTERED, the AMF can locally de-register the UE; or if the UE is in 5GMM-CONNECTED, the AMF can initiate the network-initiated de-registration procedure (see subclause 5.5.2.3).
- NOTE 25: The value of the implementation specific timer for onboarding services needs to be large enough to allow a UE to complete the configuration of one or more entries of the "list of subscriber data" taking into consideration that configuration of SNPN subscription parameters in PLMN via the user plane or onboarding services in SNPN involves third party entities outside of the operator's network.

If the UE receives the List of PLMNs to be used in disaster condition IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the "list of PLMN(s) to be used in disaster condition" stored in the ME together with the PLMN ID of the RPLMN, if any, and may store the "list of PLMN(s) to be used in disaster condition"

included in the List of PLMNs to be used in disaster condition IE in the ME together with the PLMN ID of the RPLMN.

If the UE receives the Disaster roaming wait range IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the disaster roaming wait range stored in the ME, if any, and store the disaster roaming wait range included in the Disaster roaming wait range IE in the ME.

If the UE receives the Disaster return wait range IE in the REGISTRATION ACCEPT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range stored included in the Disaster return wait range IE in the ME.

If the 5GS registration type IE is set to "disaster roaming mobility registration updating" and:

- a) the MS determined PLMN with disaster condition IE is included in the REGISTRATION REQUEST message, the AMF shall determine the PLMN with disaster condition in the MS determined PLMN with disaster condition IE;
- b) the MS determined PLMN with disaster condition IE is not included in the REGISTRATION REQUEST message and the Additional GUTI IE is included in the REGISTRATION REQUEST message and contains 5G-GUTI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the 5G-GUTI;
- c) the MS determined PLMN with disaster condition IE and the Additional GUTI IE are not included in the REGISTRATION REQUEST message and:
 - the 5GS mobile identity IE contains 5G-GUTI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the 5G-GUTI; or
 - the 5GS mobile identity IE contains SUCI of a PLMN of the country of the PLMN providing disaster roaming services, the AMF shall determine the PLMN with disaster condition in the PLMN identity of the SUCI; or
- d) the MS determined PLMN with disaster condition IE is not included in the REGISTRATION REQUEST message, NG-RAN of the PLMN providing disaster roaming services broadcasts disaster roaming indication and:
 - the Additional GUTI IE is included in the REGISTRATION REQUEST message and contains 5G-GUTI of a PLMN of a country other than the country of the PLMN providing disaster roaming services; or
 - the Additional GUTI IE is not included and the 5GS mobile identity IE contains 5G-GUTI or SUCI of a PLMN of a country other than the country of the PLMN providing disaster roaming services;

the AMF shall determine the PLMN with disaster condition based on the disaster roaming agreement arrangement between mobile network operators.

NOTE 26: The disaster roaming agreement arrangement between mobile network operators is out scope of 3GPP.

If the AMF determines that a disaster condition applies to the PLMN with disaster condition, and the UE is allowed to be registered for disaster roaming services, the AMF shall set the Disaster roaming registration result value bit in the 5GS registration result IE to "no additional information" in the REGISTRATION ACCEPT message. If the AMF determines that the UE can be registered to the PLMN for normal service, the AMF shall set the Disaster roaming registration result value bit in the 5GS registration result IE to "request for registration for disaster roaming services accepted as registration not for disaster roaming service" in the REGISTRATION ACCEPT message.

If the UE indicates "disaster roaming mobility registration updating" in the 5GS registration type IE in the REGISTRATION REQUEST message and the 5GS registration result IE value in the REGISTRATION ACCEPT message is set to:

- "request for registration for disaster roaming services accepted as registration not for disaster roaming service", the UE shall consider itself registered for normal service. If the PLMN identity of the registered PLMN is a member of the forbidden PLMN list as specified in subclause 5.3.13A, any such PLMN identity shall be deleted from the corresponding list(s); or
- "no additional information", the UE shall consider itself registered for disaster roaming services.

If the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the REGISTRATION ACCEPT message and the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE into the list of "5GS forbidden tracking areas for roaming" and remove the TAI(s) from the stored TAI list if present.

If the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the REGISTRATION ACCEPT message and the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE into the list of "5GS forbidden tracking areas for regional provision of service" and remove the TAI(s) from the stored TAI list if present.

5.5.1.3.5 Mobility and periodic registration update not accepted by the network

If the mobility and periodic registration update request cannot be accepted by the network, the AMF shall send a REGISTRATION REJECT message to the UE including an appropriate 5GMM cause value.

If the mobility and periodic registration update request is rejected due to general NAS level mobility management congestion control, the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

In NB-N1 mode, if the mobility and periodic registration update request is rejected due to operator determined barring (see 3GPP TS 29.503 [20AB]), the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

When the UE performs inter-system change from S1 mode to N1 mode, if the AMF is informed that verification of the integrity protection of the TRACKING AREA UPDATE REQUEST message included by the UE in the EPS NAS message container IE of the REGISTRATION REQUEST message has failed in the MME, then:

- a) If the AMF can retrieve the current 5G NAS security context as indicated by the ngKSI and 5G-GUTI sent by the UE, the AMF shall proceed as specified in subclause 5.5.1.3.4;
- b) if the AMF cannot retrieve the current 5G NAS security context as indicated by the ngKSI and 5G-GUTI sent by the UE, or the ngKSI or 5G-GUTI was not sent by the UE, the AMF may initiate the identification procedure by sending the IDENTITY REQUEST message with the "Type of identity" of the 5GS identity type IE set to "SUCI" before taking actions as specified in subclause 4.4.4.3; or
- c) If the AMF needs to reject the mobility and periodic registration update procedure, the AMF shall send REGISTRATION REJECT message including 5GMM cause #9 "UE identity cannot be derived by the network".

If the REGISTRATION REJECT message with 5GMM cause #76 or #78 was received without integrity protection, then the UE shall discard the message. If the REGISTRATION REJECT message with 5GMM cause #62 was received without integrity protected, the behaviour of the UE is specified in subclause 5.3.20.2.

Based on operator policy, if the mobility and periodic registration update request is rejected due to core network redirection for CIoT optimizations, the network shall set the 5GMM cause value to #31 "Redirection to EPC required".

NOTE 1: The network can take into account the UE's S1 mode capability, the EPS CIoT network behaviour supported by the UE or the EPS CIoT network behaviour supported by the EPC to determine the rejection with the 5GMM cause value #31 "Redirection to EPC required".

If the mobility and periodic registration update request is rejected because:

- a) all the S-NSSAI(s) included in the requested NSSAI (i.e. Requested NSSAI IE or Requested mapped NSSAI IE) are rejected;
- b) the UE set the NSSAA bit in the 5GMM capability IE to:
 - 1) "Network slice-specific authentication and authorization supported" and;
 - i) there are no default S-NSSAIs;
 - ii) all default S-NSSAIs are not allowed; or
 - iii) network slice-specific authentication and authorization has failed or been revoked for all subscribed S-NSSAIs marked as default and based on network local policy, the network decides not to initiate the

network slice-specific re-authentication and re-authorization procedures for any subscribed S-NSSAI marked as default requested by the UE; or

- 2) "Network slice-specific authentication and authorization not supported" and;
 - i) there are no subscribed S-NSSAIs which are marked as default; or
 - ii) all subscribed S-NSSAIs marked as default are either not allowed or are subject to network slice-specific authentication and authorization; and
- c) no emergency PDU session has been established for the UE;

the network shall set the 5GMM cause value to #62 "No network slices available". If the UE had included requested NSSAI in the REGISTRATION REQUEST message, then the network shall include the rejected S-NSSAI(s) in the rejected NSSAI of the REGISTRATION REJECT message. Otherwise, the network may include the rejected S-NSSAI(s) in the rejected NSSAI of the REGISTRATION REJECT message.

If the UE has set the ER-NSSAI bit to "Extended rejected NSSAI supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, the rejected S-NSSAI(s) shall be included in the Extended rejected NSSAI IE of the REGISTRATION REJECT message. Otherwise the rejected S-NSSAI(s) shall be included in the Rejected NSSAI IE of the REGISTRATION REJECT message.

If the UE supports extended rejected NSSAI and the AMF determines that maximum number of UEs reached for one or more S-NSSAI(s) in the requested NSSAI as specified in subclause 4.6.2.5, the AMF shall include the rejected NSSAI containing one or more S-NSSAIs with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE in the REGISTRATION REJECT message. In addition, the AMF may include a back-off timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI iE of the REGISTRATION REJECT message.

If the mobility and periodic registration update request from a UE supporting CAG is rejected due to CAG restrictions, the network shall set the 5GMM cause value to #76 "Not authorized for this CAG or authorized for CAG cells only" and should include the "CAG information list" in the CAG information list IE or the Extended CAG information list IE in the REGISTRATION REJECT message.

- NOTE 2: The network cannot be certain that "CAG information list" stored in the UE is updated as result of sending of the REGISTRATION REJECT message with the CAG information list IE or the Extended CAG information list IE, as the REGISTRATION REJECT message is not necessarily delivered to the UE (e.g due to abnormal radio conditions).
- NOTE 3: The "CAG information list" can be provided by the AMF and include no entry if no "CAG information list" exists in the subscription.
- NOTE 3A: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If the mobility and periodic registration update request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall operate as described in bullet i) of subclause 5.5.1.3.8.

If the UE's mobility and periodic registration update request is via a satellite NG-RAN cell and the network determines that the UE is in a location where the network is not allowed to operate, see 3GPP TS 23.502 [9], the network shall set the 5GMM cause value in the REGISTRATION REJECT message to #78 "PLMN not allowed at the present UE location".

NOTE 4: When the UE accessing network for emergency services, it is up to operator and regulatory policies whether the network needs to determine if the UE is in a location where network is not allowed to operate.

If the AMF receives the mobility and periodic registration update request including the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE and the AMF determines that the UE is not allowed to use UAS services via 5GS based on the user's subscription data and the operator policy, the AMF shall return a REGISTRATION REJECT message with 5GMM cause #79 (UAS services not allowed).

If the mobility and periodic registration update request from a UE supporting MINT is rejected due to a disaster condition no longer being applicable in the current location of the UE, the network shall set the 5GMM cause value to #11 "PLMN not allowed" or #13 "Roaming not allowed in this tracking area" and may include a disaster return wait range in the Disaster return wait range IE in the REGISTRATION REJECT message.

If the UE initiates the registration procedure for disaster roaming services and the AMF determines that it does not support providing disaster roaming services for the determined PLMN with disaster condition to the UE, then the AMF shall send a REGISTRATION REJECT message with 5GMM cause #80 (Disaster roaming for the determined PLMN with disaster condition not allowed).

If the AMF received multiple TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, all of the received TAIs are forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

- a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or
- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE;
- c) both;

in the REGISTRATION REJECT message.

Regardless of the 5GMM cause value received in the REGISTRATION REJECT message via satellite access,

- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the REGISTRATION REJECT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for roaming"; and
- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the REGISTRATION REJECT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for regional provision of service".

Furthermore, the UE shall take the following actions depending on the 5GMM cause value received in the REGISTRATION REJECT message.

- #3 (Illegal UE); or
- #6 (Illegal ME).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1.

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure using f, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall delete the list of equivalent PLMNs (if any) and shall move to 5GMM-DEREGISTERED.NO-SUPI state. If the message has been successfully integrity checked by the NAS, then the UE shall:

- 1) set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN if the UE maintains these counters;
- set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN if the UE maintains these counters; and
- 3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.
- 3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.
- to UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value. The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.7a in 3GPP TS 24.301 [15]. If the UE is in EMM-REGISTERED state, the UE shall move to EMM-DEREGISTERED state. If the message has been successfully integrity checked by the NAS and the UE maintains a counter for "SIM/USIM considered invalid for non-GPRS services", then the UE shall set this counter to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#7 (5GS services not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 5GS services until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE supports access to an SNPN using credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure using for the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall move to 5GMM-DEREGISTERED.NO-SUPI state. If the message has been successfully integrity checked by the NAS, then the UE shall:

 set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN if the UE maintains these counters;

- 2) set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN if the UE maintains these counters; and
- 3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

If the UE is registered for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#9 (UE identity cannot be derived by the network).

The UE shall set the 5GS update status to 5U2 NOT UPDATED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall enter the state 5GMM-DEREGISTERED.

If the UE has initiated the registration procedure in order to enable performing the service request procedure for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

If the rejected request was neither for initiating an emergency PDU session nor for emergency services fallback, the UE shall subsequently, automatically initiate the initial registration procedure.

NOTE 5: User interaction is necessary in some cases when the UE cannot re-establish the PDU session(s) automatically.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

#10 (implicitly de-registered).

The UE shall enter the state 5GMM-DEREGISTERED.NORMAL-SERVICE. The UE shall delete any mapped 5G NAS security context or partial native 5G NAS security context.

If the UE has initiated the registration procedure in order to enable performing the service request procedure for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

If the rejected request was neither for initiating an emergency PDU session nor for emergency services fallback, the UE shall perform a new registration procedure for initial registration.

NOTE 6: User interaction is necessary in some cases when the UE cannot re-establish the PDU session(s) automatically.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM state as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

#11 (PLMN not allowed).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1, delete the list of equivalent PLMNs, reset the registration attempt counter. For 3GPP access, the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5]. For non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE and perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been successfully integrity checked by the NAS and the UE maintains the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN, the UE shall set the PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same PLMN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

If the UE receives the Disaster return wait range IE in the REGISTRATION REJECT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range IE in the ME.

#12 (Tracking area not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

#13 (Roaming not allowed in this tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete the list of equivalent PLMNs (if available). The UE shall reset the

registration attempt counter. For 3GPP acess the UE shall change to state 5GMM-REGISTERED.PLMN-SEARCH, and for non-3GPP access the UE shall change to state 5GMM-REGISTERED.LIMITED-SERVICE.

If the UE is registered in S1 mode and operating in dual-registration mode, the PLMN that the UE chooses to register in is specified in subclause 4.8.3. Otherwise if:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and shall remove the current TAI from the stored TAI list if present. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials holder, the selected PLMN subscription areas for roaming of the "list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

For 3GPP access the UE shall perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall perform network selection as defined in 3GPP TS 24.502 [18].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

If the UE receives the Disaster return wait range IE in the REGISTRATION REJECT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range included in the Disaster return wait range IE in the ME.

#15 (No suitable cells in tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2). The UE shall reset the registration attempt counter and shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

If the UE has initiated the registration procedure in order to enable performing the service request procedure for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GC according to the emergency services support indicator (see 3GPP TS 36.331 [25A]). If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access. Otherwise, the UE shall search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and shall remove the current TAI from the stored TAI list, if present. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and shall remove the current TAI from the stored TAI list, if present. If the REGISTRATION REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status and tracking area updating attempt counter as

specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

If received over non-3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.3.7.

#22 (Congestion).

If the T3346 value IE is present in the REGISTRATION REJECT message and the value indicates that this timer is neither zero nor deactivated, the UE shall proceed as described below, otherwise it shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.3.7.

The UE shall abort the registration procedure for mobility and periodic registration update. If the rejected request was not for initiating an emergency PDU session, the UE shall set the 5GS update status to 5U2 NOT UPDATED, reset the registration attempt counter and change to state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE.

The UE shall stop timer T3346 if it is running.

If the REGISTRATION REJECT message is integrity protected, the UE shall start timer T3346 with the value provided in the T3346 value IE.

If the REGISTRATION REJECT message is not integrity protected, the UE shall start timer T3346 with a random value from the default range specified in 3GPP TS 24.008 [12].

The UE stays in the current serving cell and applies the normal cell reselection process. The registration procedure for mobility and periodic registration update is started, if still necessary, when timer T3346 expires or is stopped.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

If the registration procedure for mobility and periodic registration update was initiated for an MO MMTEL voice call (i.e. access category 4), or an MO MMTEL video call (i.e. access category 5), or an MO IMS registration related signalling (i.e. access category 9) or for NAS signalling connection recovery during an ongoing MO MMTEL voice call (i.e. access category 4), or during an ongoing MO MMTEL video call (i.e. access category 5) or during an ongoing MO IMS registration related signalling (i.e. access category 4), or during an ongoing MO MMTEL video call (i.e. access category 5) or during an ongoing MO IMS registration related signalling (i.e. access category 9), then a notification that the request was not accepted due to network congestion shall be provided to upper layers.

NOTE 7: Upper layers specified in 3GPP TS 24.173 [13C] and 3GPP TS 24.229 [14] handle the notification that the request was not accepted due to network congestion.

If the UE is registered for onboarding services in SNPN, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

#27 (N1 mode not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2). Additionally, the UE shall reset the registration attempt counter and shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE. If the message has been successfully integrity checked by the NAS, the UE shall set:

- 1) the PLMN-specific N1 mode attempt counter for 3GPP access and the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN; or
- 2) the SNPN-specific attempt counter for 3GPP access for the current SNPN and the SNPN-specific attempt counter for non-3GPP access for the current SNPN in case of SNPN;

to the UE implementation-specific maximum value.

The UE shall disable the N1 mode capability for the specific access type for which the message was received (see subclause 4.9).

If the message has been successfully integrity checked by the NAS, the UE shall disable the N1 mode capability also for the other access type (see subclause 4.9).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED. Additionally, the UE shall reset the tracking area updating attempt counter and enter the state EMM-REGISTERED.

#31 (Redirection to EPC required).

5GMM cause #31 received by a UE that has not indicated support for CIoT optimizations or not indicated support for S1 mode or received by a UE over non-3GPP access is considered an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2). The UE shall reset the registration attempt counter and enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

The UE shall enable the E-UTRA capability if it was disabled and disable the N1 mode capability for 3GPP access (see subclause 4.9.2).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

#36 (IAB-node operation not authorized).

This cause value is only applicable when received over 3GPP access by a UE operating as an IAB-node. This cause value received from a 5G access network other than 3GPP access or received by a UE not operating as an IAB-node is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

If:

- 1) the UE is not operating in SNPN access operation mode,
 - i) the UE shall delete the list of equivalent PLMNs and reset the registration attempt counter and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS and the UE maintains the PLMN-specific attempt counter for 3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter for 3GPP access for that PLMN to the UE implementation-specific maximum value; and
 - ii) If the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when the EPS attach request procedure is rejected with the EMM cause with the same value; or
- 2) the UE is operating in SNPN access operation mode,
 - i) the UE shall reset the registration attempt counter and store the SNPN identity in the "temporarily forbidden SNPNs" list for 3GPP access and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN attempt counter for 3GPP access for the current SNPN to the UE implementation-specific maximum value.

#62 (No network slices available).

The UE shall abort the registration procedure for mobility and periodic registration update procedure, set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE. Additionally, the UE shall reset the registration attempt counter.

The UE receiving the rejected NSSAI in the REGISTRATION REJECT message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

Unless the back-off timer value received along with the S-NSSAI is zero, the UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

NOTE 8: If the back-off timer value received along with the S-NSSAI in the rejected NSSAI for the maximum number of UEs reached is zero as specified in subclause 10.5.7.4a of TS 24.008, the UE does not consider the S-NSSAI as the rejected S-NSSAI.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if a back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or
 - 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

If the UE has an allowed NSSAI or configured NSSAI that contains S-NSSAIs which are not included in the rejected NSSAI, the UE may stay in the current serving cell, apply the normal cell reselection process and start a registration procedure for mobility and periodic registration update with a requested NSSAI that includes any S-NSSAI from the allowed S-NSSAI or the configured NSSAI that is not in the rejected NSSAI. Otherwise the UE

may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if the UE does not have an allowed NSSAI and each S-NSSAI in the configured NSSAI, if available, was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and,

- 1) if at least one S-NSSAI in the default configured NSSAI is not rejected, the UE may stay in the current serving cell, apply the normal cell reselection process, and start a registration procedure for mobility and periodic registration update with a requested NSSAI with that default configured NSSAI; or
- 2) if all the S-NSSAI(s) in the default configured NSSAI are rejected and at least one S-NSSAI is rejected due to "S-NSSAI not available in the current registration area",
 - i) if the REGISTRATION REJECT message is integrity protected and the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and enter the state 5GMM-REGISTERED.LIMITED-SERVICE; or
 - ii) If the REGISTRATION REJECT message is integrity protected and the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

Otherwise, the UE may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if each S-NSSAI in the default configured NSSAI was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and has rejected NSSAI for the reached maximum number of UEs, and the UE wants to obtain services in the current serving cell without performing a PLMN selection or SNPN selection, the UE may stay in the current serving cell and attempt to use the rejected S-NSSAI(s) for the maximum number of UEs reached in the current serving cell after rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the tracking area updating attempt counter and enter the state EMM-REGISTERED.

#72 (Non-3GPP access to 5GCN not allowed).

When received over non-3GPP access the UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter and enter the state 5GMM-DEREGISTERED. If the message has been successfully integrity checked by the NAS, the UE shall set:

- 1) the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN; or
- 2) the SNPN-specific attempt counter for non-3GPP access for that SNPN in case of SNPN;

to the UE implementation-specific maximum value.

NOTE 9: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).

The UE shall disable the N1 mode capability for non-3GPP access (see subclause 4.9.3).

As an implementation option, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5].

If received over 3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.1.3.7.

#73 (Serving network not authorized).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs, reset the registration attempt counter, store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A. For 3GPP access the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE and perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been successfully integrity checked by the NAS, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI. Additionally, the UE shall reset the tracking area updating attempt counter and enter the state EMM-DEREGISTERED.

#74 (Temporarily not authorized for this SNPN).

5GMM cause #74 is only applicable when received from a cell belonging to an SNPN. 5GMM cause #74 received from a cell not belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and store the SNPN identity in the "temporarily forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 10: When 5GMM cause #74 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 11: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #75 (Permanently not authorized for this SNPN).

5GMM cause #75 is only applicable when received from a cell belonging to an SNPN with a globally-unique SNPN identity. 5GMM cause #75 received from a cell not belonging to an SNPN or a cell belonging to an SNPN with a non-globally-unique SNPN identity is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and store the SNPN identity in the "permanently forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection for onboarding services according to

3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access and the SNPN-specific attempt counter for non-3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 12: When 5GMM cause #75 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 13: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #76 (Not authorized for this CAG or authorized for CAG cells only).

This cause value received via non-3GPP access or from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3.ROAMING NOT ALLOWED, store the 5GS update status according to clause 5.1.3.2.2, and reset the registration attempt counter.

If 5GMM cause #76 is received from:

- 1) a CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the REGISTRATION REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 14: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall delete the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN. In the case the "allowed CAG list" for the current PLMN only contains a range of CAG-IDs, how the UE deletes the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN is up to UE implementation. In addition:

- i) if the entry in the "CAG information list" for the current PLMN does not include an "indication that the UE is only allowed to access 5GS via CAG cells" or if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list";
- ii) if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or

- iii) if the "CAG information list" does not include an entry for the current PLMN, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".
- 2) a non-CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the REGISTRATION REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 15: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN, if any. If the "CAG information list" stored in the UE does not include the current PLMN's entry, the UE shall add an entry for the current PLMN to the "CAG information list" and store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN. If the UE does not have a stored "CAG information list", the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN.

In addition:

- i) if the "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated CAG information; or
- ii) if the "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED, reset the tracking area updating attempt counter and enter the state EMM-REGISTERED.

#77 (Wireline access area not allowed).

5GMM cause #77 is only applicable when received from a wireline access network by the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device). 5GMM cause #77 received from a 5G access network other than a wireline access network and 5GMM cause #77 received by the W-AGF acting on behalf of the FN-BRG are considered as abnormal cases and the behaviour of the UE is specified in subclause 5.5.1.3.7.

When received over wireline access network, the 5G-RG and the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2), shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI, shall reset the registration attempt counter, shall enter the state 5GMM-DEREGISTERED and shall act as specified in subclause 5.3.23.

- NOTE 16: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).
- #78 (PLMN not allowed to operate at the present UE location).

This cause value received from a non-satellite NG-RAN cell is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.1.3.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter. The UE shall store the PLMN identity and, if it is known, the current geographical location in the list of "PLMNs not allowed to operate at the present UE location" and shall start a corresponding timer instance (see subclause 4.23.2). The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, and tracking area updating attempt counter as specified in 3GPP TS 24.301 [15] for the case when the normal tracking area updating procedure is rejected with the EMM cause with the same value.

#79 (UAS services not allowed).

The UE shall abort the registration procedure for mobility and periodic registration update procedure, set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE. Additionally, the UE shall reset the registration attempt counter. The UE may reattempt the registration procedure to the current PLMN for services other than UAS services and shall not include the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of REGISTRATION REQUEST message unless the UE receives a CONFIGURATION UPDATE COMMAND message including the service-level-AA service status indication in the Service-level-AA container IE with the UAS field set to "UAS services enabled".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the tracking area updating attempt counter and enter the state EMM-REGISTERED.

#80 (Disaster roaming for the determined PLMN with disaster condition not allowed).

The UE shall abort the registration procedure for mobility and periodic registration update procedure, set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE. Additionally, the UE shall reset the registration attempt counter. The UE shall not attempt to register for disaster roaming services on this PLMN for the determined PLMN with disaster condition for a period in the range of 12 to 24 hours. The UE shall not attempt to register for disaster roaming services on this PLMN for a period in the range of 3 to 10 minutes. The UE shall perform PLMN selection as described in 3GPP TS 23.122 [6].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the tracking area updating attempt counter and enter the state EMM-REGISTERED.

Other values are considered as abnormal cases. The behaviour of the UE in those cases is specified in subclause 5.5.1.3.7.

5.5.1.3.6 Mobility and periodic registration update for initiating an emergency PDU session not accepted by the network

If the mobility and periodic registration update request for initiating an emergency PDU session cannot be accepted by the network, the UE shall perform the procedures as described in subclause 5.5.1.3.5. Then if the UE is in the same selected PLMN where the last mobility and periodic registration update request was attempted, the UE shall:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 1: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.
- b) perform de-registration locally, if not de-registered already, and attempt initial registration for emergency services.

If the mobility and periodic registration update request for initiating an emergency PDU session fails due to abnormal case b) in subclause 5.5.1.3.7, the UE shall perform the actions as described in subclause 5.5.1.3.7 and inform the upper layers of the failure to access the network.

NOTE 2: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

If the mobility and periodic registration update request for initiating an emergency PDU session fails due to abnormal cases c) or d) in subclause 5.5.1.3.7, the UE shall perform the procedures as described in subclause 5.5.1.3.7. Then if the UE is in the same selected PLMN where the last mobility and periodic registration update request was attempted, the UE shall:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 3: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.
- b) perform de-registration locally, if not de-registered already, and attempt initial registration for emergency services.

5.5.1.3.6A Mobility and periodic registration update for an emergency services fallback not accepted by the network

If the mobility and periodic registration update request for initiating an emergency services fallback:

- fails due to an abnormal case described in subclause 5.5.1.3.7, the UE shall perform the procedures as described in subclause 5.5.1.3.7; or
- cannot be accepted by the network as described in subclause 5.5.1.3.5, the UE shall perform the procedures as described in subclause 5.5.1.3.5.

If the mobility and periodic registration update request for initiating an emergency services fallback fails due to abnormal case b) in subclause 5.5.1.3.7, the UE shall inform the upper layers of the failure to access the network.

NOTE 1: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

If the mobility and periodic registration update request for initiating an emergency services fallback fails due to abnormal cases c) or d) or cannot be accepted and the UE does not attempt to select an E-UTRA cell connected to EPC or 5GCN as described in subclause 5.5.1.3.5 and the UE is camped on NR or E-UTRA cell connected to 5GCN in the same PLMN where the last mobility and periodic registration update request was attempted, the UE shall inform the upper layers of the failure of the procedure.

NOTE 2: This can result in the upper layers requesting implementation specific mechanisms, e.g. procedures specified in 3GPP TS 24.229 [14] can result in the emergency call being attempted to another IP-CAN.

5.5.1.3.7 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Timer T3346 is running.

The UE shall not start the registration procedure for mobility and periodic registration update unless:

- 1) the UE is in 5GMM-CONNECTED mode;
- 2) the UE received a paging;
- 3) the UE receives a NOTIFICATION message over non-3GPP access when the UE is in 5GMM-CONNECTED mode over non-3GPP access and in 5GMM-IDLE mode over 3GPP access;
- 4) the UE is a UE configured for high priority access in selected PLMN;
- 5) the UE has an emergency PDU session established or is establishing an emergency PDU session;

- 6) the UE receives a request from the upper layers to perform emergency services fallback;
- 7) the UE receives the CONFIGURATION UPDATE COMMAND message as specified in subclause 5.4.4.3;
- the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and:
 - the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]); and
 - timer T3346 was not started when N1 NAS signalling connection was established with RRC establishment cause set to "mo-ExceptionData"; or
- 9) the MUSIM UE needs to request a new 5G-GUTI assignment as specified in subclause 5.5.1.3.2.

The UE stays in the current serving cell and applies the normal cell reselection process.

NOTE 1: It is considered an abnormal case if the UE needs to initiate a registration procedure for mobility and periodic registration update while timer T3346 is running independent on whether timer T3346 was started due to an abnormal case or a non-successful case.

If the registration procedure for mobility and periodic registration update was initiated for an MO MMTEL voice call (i.e. access category 4), for an MO MMTEL video call (i.e. access category 5), for an MO IMS registration related signalling (i.e. access category 9) or for NAS signalling connection recovery during an ongoing MO MMTEL voice call (i.e. access category 4), or during an MO MMTEL video call (i.e. access category 5) or during an ongoing MO IMS registration related signalling (i.e. access category 4), or during an MO MMTEL video call (i.e. access category 5) or during an ongoing MO IMS registration related signalling (i.e. access category 9), then a notification that the procedure was not initiated due to network congestion shall be provided to upper layers.

b) The lower layers indicate that the access attempt is barred.

The UE shall not start the registration procedure for mobility and periodic registration update. The UE stays in the current serving cell and applies the normal cell reselection process. Receipt of the access barred indication shall not trigger the selection of a different core network type (EPC or 5GCN).

The registration procedure for mobility and periodic registration update is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

ba) The lower layers indicate that:

- 1) access barring is applicable for all access categories except categories 0 and 2 and the access category with which the access attempt was associated is other than 0 and 2; or
- 2) access barring is applicable for all access categories except category 0 and the access category with which the access attempt was associated is other than 0.

If the REGISTRATION REQUEST message has not been sent, the UE shall proceed as specified for case b. If the REGISTRATION REQUEST message has been sent, the UE shall proceed as specified for case e and, additionally, the registration procedure for mobility and periodic registration update is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated. For additional UE requirements for both cases see subclause 4.5.5.

c) T3510 timeout.

The UE shall abort the registration update procedure and the N1 NAS signalling connection, if any, shall be released locally.

If the UE has initiated the registration procedure in order to enable performing the service request procedure for emergency services fallback, the UE shall inform the upper layers of the failure of the emergency services fallback (see 3GP P TS 24.229 [14]). Otherwise, the UE shall proceed as described below.

d) REGISTRATION REJECT message, other 5GMM cause values than those treated in subclause 5.5.1.3.5, and cases of 5GMM cause values #11, #15, #22, #31, #72, #73, #74, #75, #76, #77 and #78, if considered as abnormal cases according to subclause 5.5.1.3.5.

Upon reception of the 5GMM causes #95, #96, #97, #99 and #111 the UE should set the registration attempt counter to 5.

The UE shall proceed as described below.

 e) Lower layer failure, release of the NAS signalling connection received from lower layers or the lower layers indicate that the RRC connection has been suspended without a cell change before the REGISTRATION ACCEPT or REGISTRATION REJECT message is received.

The UE shall abort the registration procedure and proceed as described below.

f) Change in the current TAI.

If the current TAI is changed before the registration procedure for mobility and periodic registration update is completed, the registration procedure for mobility and periodic registration update shall be aborted and reinitiated immediately. The UE shall set the 5GS update status to 5U2 NOT UPDATED.

g) Registration procedure for mobility and periodic registration update and de-registration procedure collision.

If the UE receives a DEREGISTRATION REQUEST message without 5GMM cause value #11, #12, #13 or #15 before the registration procedure for mobility and periodic registration update has been completed, the registration procedure for mobility and periodic registration update shall be aborted and the de-registration procedure shall be progressed.

If the UE receives a DEREGISTRATION REQUEST message with 5GMM cause value #11, #12, #13 or #15 before the registration procedure for mobility and periodic registration update has been completed, the registration procedure for mobility and periodic registration update shall be progressed and the de-registration procedure shall be aborted.

NOTE 2: The registration procedure for mobility and periodic registration update shall be aborted only if the DEREGISTRATION REQUEST message indicates in the access type that the access in which the registration procedure for mobility and periodic registration update was attempted shall be de-registered. Otherwise both the procedures shall be progressed.

h) Void

i) Transmission failure of REGISTRATION REQUEST message indication from the lower layers or the lower layers indicate that the RRC connection has been suspended with a cell change.

The registration procedure for mobility and periodic registration update shall be aborted and re-initiated immediately. The UE shall set the 5GS update status to 5U2 NOT UPDATED.

j) Transmission failure of REGISTRATION COMPLETE message indication with change in the current TAI.

If the current TAI is not in the TAI list, the registration procedure for mobility and periodic registration update shall be aborted and re-initiated immediately. The UE shall set the 5GS update status to 5U2 NOT UPDATED.

If the current TAI is still part of the TAI list, it is up to the UE implementation how to re-run the ongoing procedure.

k) Transmission failure of REGISTRATION COMPLETE message indication without change in the current TAI.

It is up to the UE implementation how to re-run the ongoing procedure.

1) UE-initiated de-registration required.

De-registration due to removal of USIM or entry update in the "list of subscriber data" or due to switch off:

The registration procedure for mobility and periodic registration update shall be aborted, and the UE initiated de-registration procedure shall be performed.

De-registration not due to removal of USIM or entry update in the "list of subscriber data" and not due to switch off:

the UE initiated de-registration procedure shall be initiated after successful completion of the registration procedure for mobility and periodic registration update.

m) Timer T3447 is running

The UE shall not start any mobility and periodic registration update procedure with Uplink data status IE or Follow-on request indicator set to "Follow-on request pending" unless:

- the UE received a paging;
- the UE is a UE configured for high priority access in selected PLMN;
- the UE has an emergency PDU session established or is establishing an emergency PDU session;
- the UE receives a request from the upper layers to perform emergency services fallback; or
- the MUSIM UE needs to request a new 5G-GUTI assignment as specified in subclause 5.5.1.3.2.

The UE stays in the current serving cell and applies the normal cell reselection process. The mobility and periodic registration update procedure is started, if still necessary, when timer T3447 expires or timer T3447 is stopped.

n) Timer T3448 is running

The UE in 5GMM-IDLE mode shall not start any mobility and periodic registration update procedure with Follow-on request indicator set to "Follow-on request pending" unless:

- 1) the UE is a UE configured for high priority access in selected PLMN;
- 2) the UE which is only using 5GS services with control plane CIoT 5GS optimization received a paging request; or
- 3) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]).

The UE stays in the current serving cell and applies the normal cell reselection process. The mobility and periodic registration update procedure is started, if still necessary, when timer T3448 expires.

For the cases c, d and e the UE shall proceed as follows:

Timer T3510 shall be stopped if still running.

If the registration procedure is not for initiating an emergency PDU session, the registration attempt counter shall be incremented, unless it was already set to 5.

If the registration attempt counter is less than 5:

- if the TAI of the current serving cell is not included in the TAI list or the 5GS update status is different to 5U1 UPDATED or if the registration procedure was triggered due to cases c, g, n, v in subclause 5.5.1.3.2, the UE shall start timer T3511, shall set the 5GS update status to 5U2 NOT UPDATED and change to state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE. When timer T3511 expires, the registration update procedure is triggered again.
- if the TAI of the current serving cell is included in the TAI list, the 5GS update status is equal to 5U1
 UPDATED, and the UE is not performing the registration procedure after an inter-system change from S1
 mode to N1 mode, the UE shall keep the 5GS update status to 5U1 UPDATED and enter state 5GMM REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2). The UE shall start timer T3511. If in addition the REGISTRATION REQUEST
 message did not include the MICO indication IE or the Extended DRX IE, and:
 - the REGISTRATION REQUEST message indicated "periodic registration updating";
 - the registration procedure was initiated to recover the NAS signalling connection due to "RRC Connection failure" from the lower layers; or
 - the registration procedure was initiated by the UE in 5GMM-CONNECTED mode with RRC inactive indication entering a cell in the current registration area belonging to an equivalent PLMN of the registered PLMN and not belonging to the registered PLMN,

and none of the other reasons for initiating the registration updating procedure listed in subclause 5.5.1.3.2 was applicable, the timer T3511 may be stopped when the UE enters 5GMM-CONNECTED mode.

- if the TAI of the current serving cell is included in the TAI list, the 5GS update status is equal to 5U1 UPDATED and the UE is performing the registration procedure after an inter-system change from S1 mode to N1 mode, the UE shall change the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE. The UE shall start timer T3511.
- If the procedure is performed via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition handle the EPS update status as specified in 3GPP TS 24.301 [15] for the abnormal cases when a normal or periodic tracking area updating procedure fails and the tracking area attempt counter is less than 5 and the EPS update status is different from EU1 UPDATED.

If the registration attempt counter is equal to 5

- the UE shall start timer T3502, shall set the 5GS update status to 5U2 NOT UPDATED.
- the UE shall delete the list of equivalent PLMNs (if any) and shall change to state 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE or optionally to 5GMM-REGISTERED.PLMN-SEARCH in order to perform a PLMN selection, SNPN selection or SNPN selection for onboarding services according to 3GPP TS 23.122 [5].
- if the procedure is performed via 3GPP access and the UE is operating in single-registration mode:
 - the UE shall in addition handle the EPS update status as specified in 3GPP TS 24.301 [15] for the abnormal cases when a normal or periodic tracking area updating procedure fails and the tracking area attempt counter is equal to 5; and
 - if the UE does not change to state 5GMM-REGISTERED.PLMN-SEARCH, the UE shall attempt to select E-UTRAN radio access technology. The UE may disable the N1 mode capability as specified in subclause 4.9.

5.5.1.3.8 Abnormal cases on the network side

The following abnormal cases can be identified:

a) If a lower layer failure occurs before the message REGISTRATION COMPLETE has been received from the UE and timer T3550 is running, the AMF shall abort the procedure, enter 5GMM-IDLE mode.

If a new 5G-GUTI was assigned to the UE in the REGISTRATION ACCEPT message, the AMF shall consider both, the old and new 5G-GUTIs as valid until the old 5G-GUTI can be considered as invalid by the AMF. If a new TAI list was provided in the REGISTRATION ACCEPT message, both the old and new TAI lists shall also be considered valid until the old TAI list can be considered invalid by the AMF. If the old 5G-GUTI was allocated by an AMF other than the current AMF, the current AMF does not need to retain the old 5G-GUTI.

Additionally, if the REGISTRATION ACCEPT message includes:

- 1) Negotiated PEIPS assistance information IE containing a new Paging subgroup ID and the UE is previously assigned a different Paging subgroup ID then, the AMF shall consider both, the old and new Paging subgroup IDs as valid until the old Paging subgroup ID can be considered as invalid by the AMF; or
- Negotiated PEIPS assistance information IE containing no Paging subgroup ID or no Negotiated PEIPS assistance information IE, then the AMF shall delete any old Paging subgroup ID stored in the 5GMM context of the UE.

During this period:

- 1) if the new 5G-GUTI is used by the UE in a subsequent message, then:
 - i) the AMF shall consider the old 5G-GUTI as invalid and, additionally, the old TAI list as invalid if a new TAI list was provided with the new 5G-GUTI in the REGISTRATION ACCEPT message; and
 - ii) if the AMF assigns a new Paging subgroup ID to the UE in the REGISTRATION ACCEPT message, then, the AMF shall consider the new Paging subgroup ID as valid and the old Paging subgroup ID, if any, as invalid.

- if the old 5G-GUTI is used by the UE in a subsequent message, the AMF may use the identification procedure followed by a generic UE configuration update procedure. If the AMF in the REGISTRATION ACCEPT message:
 - i) assigns a new Paging subgroup ID to the UE, then, the AMF shall include the PEIPS assistance information; or
 - ii) does not assign a Paging subgroup ID to the UE then, the AMF shall not include the PEIPS assistance information;

and initiate the generic UE configuration update procedure; and

- 3) if the UE needs to be paged:
 - i) if in the REGISTRATION ACCEPT message:
 - a new Paging subgroup ID is assigned to the UE that previously has no Paging subgroup ID assigned then, the AMF shall use the new Paging subgroup ID for paging the UE;
 - a new Paging subgroup ID is assigned to the UE that is same as the old Paging subgroup ID then, the AMF shall use the same Paging subgroup ID for paging the UE; and
 - a new Paging subgroup ID is assigned to the UE that is different than the old Paging subgroup ID then, the AMF may first use the old Paging subgroup ID followed by the new Paging subgroup ID for paging the UE.
 - ii) the AMF selects the 5G-GUTI and TAI list as follows:
 - the AMF may first use the old 5G-S-TMSI from the old 5G-GUTI for paging within the area defined by the old TAI list for an implementation dependent number of paging attempts using the selected Paging subgroup ID. If a new TAI list was provided in the REGISTRATION ACCEPT message, the new TAI list should also be used for paging. Upon response from the UE, the AMF may initiate the generic UE configuration update procedure. If the response is received from a tracking area within the old and new TAI list, the network shall initiate the generic UE configuration update procedure. If in the REGISTRATION ACCEPT message a new Paging subgroup ID was assigned to the UE that is different than the old Paging subgroup ID then the network shall initiate the generic UE configuration update procedure; and
 - if no response is received to the paging attempts using the old 5G-S-TMSI from the old 5G-GUTI and the old Paging subgroup ID, the AMF may use the new 5G-S-TMSI from the new 5G-GUTI and the new Paging subgroup ID, if any, for paging, for an implementation dependent number of paging attempts. In this case, if a new TAI list was provided with the new 5G-GUTI in the REGISTRATION ACCEPT message, the new TAI list shall be used instead of the old TAI list.
- b) Protocol error.

If the REGISTRATION REQUEST message has been received with a protocol error, the AMF shall return a REGISTRATION REJECT message with one of the following 5GMM cause values:

- #96 invalid mandatory information;
- #99 information element non-existent or not implemented;
- #100 conditional IE error; or
- #111 protocol error, unspecified.
- c) T3550 time out.

On the first expiry of the timer, the AMF shall retransmit the REGISTRATION ACCEPT message and shall reset and restart timer T3550. The retransmission is performed four times, i.e. on the fifth expiry of timer T3550, the registration procedure for mobility and periodic update procedure is aborted.

If a new 5G-GUTI was assigned to the UE in the REGISTRATION ACCEPT message, both, the old and new 5G-GUTI shall be considered as valid until the old 5G-GUTI can be considered as invalid by the AMF. If a new TAI list was provided in the REGISTRATION ACCEPT message, both the old and new TAI lists shall also be

considered valid until the old TAI list can be considered invalid by the AMF. If the old 5G-GUTI was allocated by an AMF other than the current AMF, the current AMF does not need to retain the old 5G-GUTI. During this period the AMF acts as described for case a) above.

- d) REGISTRATION REQUEST with 5GS registration type IE set to "mobility registration updating" or "periodic registration updating" received after the REGISTRATION ACCEPT message has been sent and before the REGISTRATION COMPLETE message is received, if the REGISTRATION COMPLETE message is expected.
 - If one or more of the information elements in the REGISTRATION REQUEST message differ from the ones received within the previous REGISTRATION REQUEST message, the previously initiated registration procedure for mobility and periodic registration update shall be aborted if the REGISTRATION COMPLETE message has not been received and the new registration procedure for mobility and periodic registration update shall be progressed; or
 - if the information elements do not differ, then the REGISTRATION ACCEPT message shall be resent and timer T3550 shall be restarted. In that case, the retransmission counter related to timer T3550 is not incremented.
- e) More than one REGISTRATION REQUEST message with 5GS registration type IE set to "mobility registration updating" or "periodic registration updating" received and neither REGISTRATION ACCEPT message nor REGISTRATION REJECT message has been sent.
 - If one or more of the information elements in the REGISTRATION REQUEST message differs from the ones received within the previous REGISTRATION REQUEST message, the previously initiated registration procedure for mobility and periodic registration update shall be aborted and the new registration procedure for mobility and periodic registration update shall be progressed; or
 - if the information elements do not differ, then the network shall continue with the previous registration procedure for mobility and periodic registration update and shall not treat any further this REGISTRATION REQUEST message.
- f) Lower layers indication of non-delivered NAS PDU due to handover.

If the REGISTRATION ACCEPT message or REGISTRATION REJECT message could not be delivered due to an intra AMF handover and the target TA is included in the TAI list, then upon successful completion of the intra AMF handover the AMF shall retransmit the REGISTRATION ACCEPT message or REGISTRATION REJECT message. If a failure of the handover procedure is reported by the lower layer and the N1 NAS signalling connection exists, the AMF shall retransmit the REGISTRATION ACCEPT message or REGISTRATION REJECT message.

g) DEREGISTRATION REQUEST message received before REGISTRATION COMPLETE message is received, if the REGISTRATION COMPLETE message is expected.

If the De-registration type IE is set to "switch off":

The AMF shall abort the signalling for the registration procedure for mobility and periodic update towards the UE and shall progress the de-registration procedure as described in subclause 5.5.2.2.

NOTE 1: Internally in the AMF, before processing the de-registration request, the AMF can perform the necessary signalling procedures for the registration procedure for mobility and periodic update before progressing the de-registration procedure.

If the De-registration type IE is set to other type than "switch off":

The AMF shall proceed with registration procedure for mobility and periodic update and shall progress the de-registration procedure after successful completion of the registration procedure for mobility and periodic update.

- h) If the REGISTRATION REQUEST message with 5GS registration type IE indicating "periodic registration updating" is received by the new AMF which does not have the 5GMM context data related to the subscription, the new AMF may send the REGISTRATION REJECT message with 5GMM cause #10 "implicitly deregistered".
- i) Based on operator policy, if the mobility and periodic registration update request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall reject the mobility and periodic registration update request

with a 5GMM cause value other than the 5GMM cause #76 (Not authorized for this CAG or authorized for CAG cells only).

NOTE 2: 5GMM cause #7 (5GS services not allowed), 5GMM cause #11 (PLMN not allowed), 5GMM cause #27 (N1 mode not allowed), 5GMM cause #73 (Serving network not authorized) can be used depending on the subscription of the UE and whether the UE roams or not.

5.5.2 De-registration procedure

5.5.2.1 General

The de-registration procedure is used:

- a) by the UE to de-register for 5GS services over 3GPP access when the UE is registered over 3GPP access;
- b) by the UE to de-register for 5GS services over non-3GPP access when the UE is registered over non-3GPP access;
- c) by the UE to de-register for 5GS services over 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN over both accesses;
- d) by the network to inform the UE that it is deregistered for 5GS services over 3GPP access when the UE is registered over 3GPP access;
- e) by the network to inform the UE that it is deregistered for 5GS services over non-3GPP access when the UE is registered over non-3GPP access;
- f) by the network to inform the UE that it is deregistered for 5GS services over 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN over both accesses;
- g) by the network to inform the UE to re-register to the network; and
- h) by the network to inform the UE supporting UAS service that it is deregistered for UAS services in 5GS.

The de-registration procedure with appropriate de-registration type shall be invoked by the UE:

- a) if the UE is switched off;
- b) as part of the eCall inactivity procedure defined in subclause 5.5.3; and
- c) as part of USIM removal.

The de-registration procedure with appropriate de-registration type shall be invoked by the network:

a) if the network informs whether the UE should re-register to the network.

The de-registration procedure with appropriate access type shall be invoked by the UE:

- a) if the UE needs to de-register for 5GS services over 3GPP access when the UE is registered over 3GPP access;
- b) if the UE needs to de-register for 5GS services over non-3GPP access when the UE is registered over non-3GPP access;
- c) the UE needs to de-register for 5GS services over 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN over both accesses; or
- d) to de-register for 5GS services over 3GPP access, if the UE is registered for disaster roaming services over 3GPP access and has successfully registered over non-3GPP access on another PLMN.

The de-registration procedure with appropriate access type shall be invoked by the network:

- a) if the network needs to inform the UE that it is deregistered over 3GPP access when the UE is registered over 3GPP access;
- b) if the network needs to inform the UE that it is deregistered over non-3GPP access when the UE is registered over non-3GPP access; or

3GPP TS 24.501 version 17.16.0 Release 17

c) if the network needs to inform the UE that it is deregistered over 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN over both accesses.

If the de-registration procedure is triggered due to USIM removal, the UE shall indicate "switch off" in the deregistration type IE.

If the de-registration procedure is requested by the network for a UE that has an emergency PDU session, the AMF shall not send a DEREGISTRATION REQUEST message to the UE and indicate to the SMF to release all non-emergency PDU sessions as specified in 3GPP TS 23.502 [9].

If the de-registration procedure for 5GS services is performed, a local release of the PDU sessions over the indicated access(es), if any, for this particular UE is performed. If a PDU session that will be released is associated with one or more multicast MBS sessions, the UE is considered as removed from the associated multicast MBS sessions.

The UE is allowed to initiate the de-registration procedure even if the timer T3346 is running.

NOTE 1: When the UE has no PDU sessions over non-3GPP access, or the UE moves all the PDU sessions over a non-3GPP access to a 3GPP access, the UE and the AMF need not initiate de-registration over the non-3GPP access.

The AMF shall provide the UE with a non-3GPP de-registration timer.

When the AMF enters the state 5GMM-DEREGISTERED for 3GPP access, the AMF shall delete the stored UE radio capability information or the UE radio capability ID, if any.

When upper layers indicate that emergency services are no longer required, the UE if still registered for emergency services, may perform UE-initiated de-registration procedure followed by a re-registration to regain normal services, if the UE is in or moves to a suitable cell.

If the UE is registered for onboarding services in SNPN, after completing the configuration of one or more entries of the "list of subscriber data", the UE should perform UE-initiated de-registration procedure. To prevent the UE registered for onboarding services in SNPN from staying on the ON-SNPN or to prevent a UE whose subscription only allows for configuration of SNPN subscription parameters in PLMN via the user plane, from staying registered indefinitely, when an implementation specific timer expires:

- if the AMF considers that the UE is in 5GMM-IDLE, the AMF shall locally de-registers the UE; or
- if the UE is in 5GMM-CONNECTED, the AMF shall initiate the network-initiated de-registration procedure (see subclause 5.5.2.3).
- NOTE 2: The value of the implementation specific timer for onboarding services needs to be large enough to allow a UE to complete the configuration of one or more entries of the "list of subscriber data" taking into consideration that configuration of SNPN subscription parameters in PLMN via the user plane or onboarding services in SNPN involves third party entities outside of the operator's network.
- NOTE 3: How to determine the completion of the configuration of one or more entries of the "list of subscriber data" is UE implementation specific.

If the de-registration procedure is not due to switch off and upper layers request establishing an emergency PDU session before the de-registration procedure has been completed, the UE shall abort the deregistration procedure, perform local de-registration and initiate a registration procedure.

If the UE in 5GMM-CONNECTED mode determines to obtain service on a higher priority PLMN due to SOR, then based on the conditions as specified in 3GPP TS 23.122 [5] annex C, where applicable, the UE shall perform UE-initiated de-registration procedure.

If the UE is registered for disaster roaming services, and the UE receives an indication of whether disaster roaming is enabled in the UE set to "Disaster roaming is disabled in the UE" in a UE parameters update transparent container, the UE shall perform UE-initiated de-registration procedure.

5.5.2.2 UE-initiated de-registration procedure

5.5.2.2.1 UE-initiated de-registration procedure initiation

The de-registration procedure is initiated by the UE by sending a DEREGISTRATION REQUEST message (see example in figure 5.5.2.2.1). The De-registration type IE included in the message indicates whether the de-registration procedure is due to a "switch off" or not. The access type included in the message indicates whether the de-registration procedure is:

- a) for 5GS services over 3GPP access when the UE is registered over 3GPP access;
- b) for 5GS services over non-3GPP access when the UE is registered over non-3GPP access; or
- c) for 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN over both accesses.

If the UE has a valid 5G-GUTI, the UE shall populate the 5GS mobile identity IE with the valid 5G-GUTI. If the UE does not have a valid 5G-GUTI, the UE shall populate the 5GS mobile identity IE with its SUCI as follows:

- a) if timer T3519 is not running, generate a fresh SUCI as specified in 3GPP TS 33.501 [24], send a DEREGISTRATION REQUEST message with the SUCI, start timer T3519 and store the value of the SUCI sent in the DEREGISTRATION REQUEST message; and
- b) if timer T3519 is running, send a DEREGISTRATION REQUEST message with the stored SUCI.

If the UE does not have a valid 5G-GUTI and it does not have a valid SUCI, then the UE shall populate the5GS mobile identity IE with its PEI.

If the de-registration request is not due to switch off and the UE is in the state 5GMM-REGISTERED or 5GMM-REGISTERED-INITIATED, timer T3521 shall be started in the UE after the DEREGISTRATION REQUEST message has been sent. The UE shall enter the state 5GMM-DEREGISTERED-INITIATED.

If the UE is to be switched off, the UE shall try for a period of 5 seconds to send the DEREGISTRATION REQUEST message. During this period, the UE may be switched off as soon as the DEREGISTRATION REQUEST message has been sent.

UE

AMF

Start T3521 DEREGISTRATION REQUEST Stop T3521

DEREGISTRATION ACCEPT

or UE at switch off:

DEREGISTRATION REQUEST

Figure 5.5.2.2.1.1: UE-initiated de-registration procedure

5.5.2.2.2 UE-initiated de-registration procedure completion

When the DEREGISTRATION REQUEST message is received by the AMF, the AMF shall send a DEREGISTRATION ACCEPT message to the UE, if the De-registration type IE does not indicate "switch off". Otherwise, the procedure is completed when the AMF receives the DEREGISTRATION REQUEST message.

The UE, when receiving the DEREGISTRATION ACCEPT message, shall stop timer T3521, stop timer T3519 if running, and delete any stored SUCI.

5.5.2.2.3 UE-initiated de-registration procedure completion for 5GS services over 3GPP access

If the access type in the DEREGISTRATION REQUEST message indicates that the de-registration procedure is for 3GPP access, the AMF shall trigger the SMF to perform a local release of the PDU session(s) established over 3GPP access, if any, for this UE. The UE shall perform a local release of the PDU session(s) established over 3GPP access, if any. If a PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions and the SMF shall consider the UE as removed from the associated multicast MBS session. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access in the same PLMN or in different PLMNs, the AMF shall trigger SMF to perform release of user plane resources on 3GPP access, and the UE shall consider the user plane resources on 3GPP access as released. If there is an MA PDU session with user plane resources only, the AMF shall trigger the SMF to perform a local release of the MA PDU session, and the UE shall perform a local release of the MA PDU session. If the MA PDU session is associated with one or more MBS multicast sessions, the UE shall locally leave the associated MBS multicast sessions. The UE is marked as inactive in the AMF for 5GS services for 3GPP access. The AMF shall enter the state 5GMM-DEREGISTERED for 3GPP access.

If the UE supports N1 mode only and the de-registration request is not due to switch off, then:

- a) if the de-registration procedure was performed due to disabling of 5GS services, then the UE shall enter the 5GMM-NULL state for 3GPP access;
- b) otherwise, the UE shall enter the 5GMM-DEREGISTERED state for 3GPP access.
- NOTE: Case b) is applicable when the UE is also registered over non-3GPP access.

If the access type indicates that the de-registration procedure is for 3GPP access or for 3GPP access and non-3GPP access, and the UE is operating in single-registration mode, it shall additionally proceed as specified in 3GPP TS 24.301 [15], subclause 5.5.2.2.2, for the case when the UE receives an EMM message DETACH ACCEPT. Furthermore, if the UE supports A/Gb or Iu mode, it shall disable the N1 mode capability for 3GPP access.

5.5.2.2.4 UE-initiated de-registration procedure completion for 5GS services over non-3GPP access

If the access type in the DEREGISTRATION REQUEST indicates that the de-registration procedure is for non-3GPP access, the AMF shall trigger SMF to perform a local release of the PDU session(s) established over non-3GPP access, if any, for this UE. The UE shall perform a local release of the PDU session(s) established over non-3GPP access, if any. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access in the same PLMN or in different PLMNs, the AMF shall trigger SMF to perform release of user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access as released. If there is an MA PDU session with a PDN connection as a user-plane resource and user plane resources established on non-3GPP access, the AMF shall trigger SMF to perform release of user plane resources established on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access, and the UE shall consider the user plane resources on non-3GPP access as released. If the UE has an MA PDU session with user plane resources established on non-3GPP access only, the AMF shall trigger the SMF to perform a local release of the MA PDU session, and the UE shall perform a local release of the MA PDU session. The UE is marked as inactive in the AMF for 5GS services for non-3GPP access. The AMF shall enter the state 5GMM-DEREGISTERED over non-3GPP access.

If the de-registration request is not due to switch off, the UE shall:

- a) if the de-registration procedure was performed due to disabling of 5GS services, enter the 5GMM-NULL state for non-3GPP access;
- b) otherwise, enter the 5GMM-DEREGISTERED state for non-3GPP access.
- NOTE: Case b) is applicable when the UE is also registered over 3GPP access.

5.5.2.2.5 UE-initiated de-registration procedure completion for 5GS services over both 3GPP access and non-3GPP access

If the access type in the DEREGISTRATION REQUEST indicates that the de-registration procedure is for both 3GPP access and non-3GPP access when the UE is registered in the same PLMN over both accesses, the descriptions for UE-initiated de-registration procedure completion for 5GS services over 3GPP access and over non-3GPP access, as specified in subclauses 5.5.2.2.3 and 5.5.2.2.4, shall be followed.

5.5.2.2.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Lower layer failure or release of the N1 NAS signalling connection before reception of DEREGISTRATION ACCEPT message.

The de-registration procedure shall be aborted and the UE proceeds as follows:

- 1) if the de-registration procedure was performed due to disabling of 5GS services, the UE shall enter the 5GMM-NULL state; or
- 2) if the de-registration type "normal de-registration" was requested for reasons other than disabling of 5GS services, the UE shall enter the 5GMM-DEREGISTERED state.
- b) The lower layers indicate that the access attempt is barred.

The UE shall not start the de-registration signalling procedure. The UE stays in the current serving cell and applies the normal cell reselection process. Receipt of the access barred indication shall not trigger the selection of a different core network type (EPC or 5GCN).

The UE may perform a local de-registration either immediately or after an implementation-dependent time.

The de-registration signalling procedure is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

ba) The lower layers indicate that:

- 1) access barring is applicable for all access categories except categories 0 and 2 and the access category with which the access attempt was associated is other than 0 and 2; or
- 2) access barring is applicable for all access categories except category 0 and the access category with which the access attempt was associated is other than 0.

If the DEREGISTRATION REQUEST message has not been sent, the UE shall proceed as specified for case b. If the DEREGISTRATION REQUEST message has been sent, the UE shall proceed as specified for case a.

c) T3521 timeout.

If the de-registration procedure was performed based on conditions specified in 3GPP TS 23.122 [5] annex C, on the expiry of timer T3521 the de-registration procedure shall be aborted and the UE shall locally release the established N1 NAS signalling connection and enter the 5GMM-DEREGISTERED state.

Otherwise, on the first four expiries of the timer, the UE shall retransmit the DEREGISTRATION REQUEST message and shall reset and restart timer T3521. On the fifth expiry of timer T3521, the de-registration procedure shall be aborted and the UE proceeds as follows:

- 1) if the de-registration procedure was performed due to disabling of 5GS services, the UE shall enter the 5GMM-NULL state; or
- 2) if the de-registration type "normal de-registration" was requested for reasons other than disabling of 5GS services, the UE shall enter the 5GMM-DEREGISTERED state.
- d) De-registration procedure collision.

De-registration containing de-registration type "switch off":

- If the UE receives a DEREGISTRATION REQUEST message before the UE-initiated de-registration procedure has been completed, this message shall be ignored and the UE-initiated de-registration procedure shall continue.

Otherwise:

- If the UE receives a DEREGISTRATION REQUEST message before the UE-initiated de-registration procedure has been completed, it shall treat the message as specified in subclause 5.5.2.3.2 with the following modification:

- If the DEREGISTRATION REQUEST message received by the UE contains de-registration type "reregistration required", and the UE-initiated de-registration procedure is with de-registration type "normal de-registration", the UE need not initiate the registration procedure for initial registration.
- e) De-registration and 5GMM common procedure collision.

De-registration containing de-registration type "switch off":

- If the UE receives a message used in a 5GMM common procedure before the de-registration procedure has been completed, this message shall be ignored and the de-registration procedure shall continue.

Otherwise:

- If the UE receives a message used in a 5GMM common procedure before the de-registration procedure has been completed, both the 5GMM common procedure and the de-registration procedure shall continue; or
- If the UE receives a DL NAS TRANSPORT message containing payload container type "Service-level-AA container" before the de-registration procedure has been completed, this message shall be ignored and the de-registration procedure shall continue.
- f) Change in the current TAI.

If that the current TAI is not in the stored TAI list before the UE-initiated de-registration procedure is completed, the UE proceeds as follows:

- if the de-registration procedure was initiated for reasons other than removal of the USIM, the UE is to be switched off or due to the last Tsor-cm timer expiry or stopped (see 3GPP TS 23.122 [5]), the de-registration procedure shall be aborted and re-initiated after successfully performing a registration procedure for mobility or periodic update used for mobility (i.e. the 5GS registration type IE set to "mobility registration updating" in the REGISTRATION REQUEST message); or
- 2) if the de-registration procedure was initiated due to removal of the USIM or the UE is to be switched off or due to the last Tsor-cm timer expiry or stopped (see 3GPP TS 23.122 [5]), the UE shall abort the de-registration procedure, perform a local de-registration and enter the state 5GMM-DEREGISTERED.
- g) Transmission failure of DEREGISTRATION REQUEST message indication with change in the current TAI.

If the current TAI is not in the TAI list, the UE proceeds as follows:

- if the de-registration procedure was initiated for reasons other than removal of the USIM, the UE is to be switched off or due to the last Tsor-cm timer expiry or stopped (see 3GPP TS 23.122 [5]), the de-registration procedure shall be aborted and re-initiated after successfully performing a registration procedure for mobility or periodic update; or
- 2) if the de-registration procedure was initiated due to removal of the USIM or the UE is to be switched off or due to the last Tsor-cm timer expiry or stopped (see 3GPP TS 23.122 [5]), the UE shall abort the deregistration procedure, perform a local de-registration and enter the state 5GMM-DEREGISTERED.

If the current TAI is still part of the TAI list, the UE shall restart the de-registration procedure.

h) Transmission failure of DEREGISTRATION REQUEST message indication without change in the current TAI.

The UE shall restart the de-registration procedure.

i) The lower layers indicate that the RRC connection has been suspended.

De-registration containing de-registration type "switch off":

- The UE may perform a local de-registration either immediately or after an implementation-dependent time.

Otherwise:

- The UE shall wait for an implementation-dependent time and shall restart the de-registration procedure, if still needed, upon expiry of the implementation-dependent time.

For the cases a, f, g and i:
- Timer T3521 shall be stopped if still running.

5.5.2.2.7 Abnormal cases in the network side

The following abnormal cases can be identified:

a) De-registration request received in a cell belonging to an SNPN with a non-globally-unique SNPN identity for which the UE has no valid subscription

If the UE initiates a de-registration procedure in a cell belonging to an SNPN with a non-globally-unique SNPN identity for which the UE has no valid subscription, and the de-registration procedure is not due to "switch off", the network shall initiate the de-registration procedure. The AMF shall send a DEREGISTRATION REQUEST message including the 5GMM cause #74 "Temporarily not authorized for this SNPN".

b) De-registration request received in a cell belonging to an SNPN with a globally-unique SNPN identity for which the UE has no valid subscription

If the UE initiates a de-registration procedure in a cell belonging to an SNPN with a globally-unique SNPN identity for which the UE has no valid subscription, and the de-registration procedure is not due to "switch off", the network shall initiate the de-registration procedure. The AMF shall send a DEREGISTRATION REQUEST message including the 5GMM cause #75 "Permanently not authorized for this SNPN".

c) De-registration request received in a CAG cell and none of the CAG ID broadcasted by the CAG cell is included in the UE's "allowed CAG list" for the current PLMN

If the UE initiates a de-registration procedure in a CAG cell and none of the CAG ID broadcasted by the CAG cell is included in the UE's "allowed CAG list" for the current PLMN and the de-registration procedure is not due to "switch off", the AMF shall initiate the de-registration procedure. The AMF shall send a DEREGISTRATION REQUEST message including the 5GMM cause #76 "Not authorized for this CAG or authorized for CAG cells only" if the UE supports CAG. Otherwise, the network shall operate as described in bullet g) of subclause 5.5.2.3.5.

d) De-registration request received in a non-CAG cell from a UE whose "CAG information list" includes an entry associated with the current PLMN, where the entry contains an "indication that the UE is only allowed to access 5GS via CAG cells"

If the UE initiates a de-registration procedure in a non-CAG cell from a UE whose "CAG information list" includes an entry associated with the current PLMN, where the entry contains an "indication that the UE is only allowed to access 5GS via CAG cells" and the de-registration procedure is not due to "switch off", the AMF shall initiate the de-registration procedure. The AMF shall send a DEREGISTRATION REQUEST message including the 5GMM cause #76 "Not authorized for this CAG or authorized for CAG cells only" if the UE supports CAG. Otherwise, the network shall operate as described in bullet g) of subclause 5.5.2.3.5.

e) De-registration and registration procedure for initial registration collision

If the network receives a REGISTRATION REQUEST message indicating either "initial registration" or "emergency registration" in the 5GS registration type IE from the UE before the UE-initiated de-registration procedure, which is not due to switch off, has been completed, the network shall abort the de-registration procedure and the registration procedure shall be progressed.

5.5.2.3 Network-initiated de-registration procedure

5.5.2.3.1 Network-initiated de-registration procedure initiation

The network initiates the de-registration procedure by sending a DEREGISTRATION REQUEST message to the UE (see example in figure 5.5.2.3.1.1).

NOTE 1: If the AMF performs a local de-registration, it will inform the UE with a 5GMM messages (e.g. SERVICE REJECT message or REGISTRATION REJECT message) with 5GMM cause #10 "implicitly de-registered" only when the UE initiates a 5GMM procedure.

The network may include a 5GMM cause IE to specify the reason for the DEREGISTRATION REQUEST message. The network shall start timer T3522. The network shall indicate whether re-registration is needed or not in the Deregistration type IE. The network shall also indicate via the access type whether the de-registration procedure is:

- a) for 3GPP access only;
- b) for non-3GPP access only; or
- c) for 3GPP access, non-3GPP access or both when the UE is registered in the same PLMN for both accesses.

If the network de-registration is triggered due to network slice-specific authentication and authorization failure or revocation as specified in subclause 4.6.2.4, then the network shall set the 5GMM cause value to #62 "No network slices available" in the DEREGISTRATION REQUEST message. In addition, if the UE supports extended rejected NSSAI, the AMF shall include the Extended rejected NSSAI IE in the DEREGISTRATION REQUEST message; otherwise the AMF shall include the Rejected NSSAI IE in the DEREGISTRATION REQUEST message.

If the UE supports extended rejected NSSAI and the network de-registration is triggered due to mobility management based network slice admission control as specified in subclause 4.6.2.5, then the network shall set the 5GMM cause value to #62 "No network slices available" in the DEREGISTRATION REQUEST message. In addition, the network may include a back-off timer value for each S-NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached" in the Extended rejected NSSAI IE of the DEREGISTRATION REQUEST message.

If the network de-registration is triggered for a UE supporting CAG due to CAG restrictions, the network shall set the 5GMM cause value to #76 "Not authorized for this CAG or authorized for CAG cells only" and should include the "CAG information list" in the CAG information list IE or the Extended CAG information list IE in the DEREGISTRATION REQUEST message.

If the network de-registration is triggered for a UE not supporting CAG due to CAG restrictions, the network shall operate as described in bullet g) of subclause 5.5.2.3.5.

If the network de-registration is triggered because the network determines that the UE is in a location where the network is not allowed to operate, see 3GPP TS 23.502 [9], the network shall set the 5GMM cause value in the DEREGISTRATION REQUEST message to #78 "PLMN not allowed to operate at the present UE location".

If the network de-registration is triggered due to:

- a) an unsuccessful outcome of an ongoing UUAA-MM procedure;
- b) an UUAA revocation notification received from the UAS-NF for a UE supporting UAS service requesting UAS services; or
- c) the UE not allowed to use UAS services via 5GS due to a change of the aerial UE subscription information,

then the network shall set the 5GMM cause value in the DEREGISTRATION REQUEST message to #79 "UAS services not allowed".

NOTE 3: If the UE supporting UAS service has requested other services than UAS services, or if there are other ongoing network slice-specific authentication and authorization on pending NSSAIs, it is then an operator policy or configuration decision whether to keep the UE supporting UAS service registered to the network, but that UE supporting UAS services is not allowed to access UAS services via 5GS as specified in 3GPP TS 23.256 [6AB].

If the network de-registration is triggered for a UE supporting MINT due to a disaster condition no longer being applicable in the current location of the UE, the network shall set the 5GMM cause value to #11 "PLMN not allowed" and may include a disaster return wait range in the Disaster return wait range IE in the DEREGISTRATION REQUEST message.

If the network de-registration is triggered because the AMF determines that, by UE subscription and operator's preferences, all of the TAIs received from the satellite NG-RAN are forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or

NOTE 2: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or
- c) both;

in the DEREGISTRATION REQUEST message.

If the network de-registration is triggered because the AMF determines that the UE operating as an IAB-node by subscription is not authorized for IAB-node operation, the AMF shall set the 5GMM cause value to #36 "IAB-node operation not authorized" in the DEREGISTRATION REQUEST message.

The AMF shall trigger the SMF to release locally the PDU session(s) over the indicated access(es), if any, for the UE and enter state 5GMM-DEREGISTERED-INITIATED. If a PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions.



Figure 5.5.2.3.1.1: Network-initiated de-registration procedure

5.5.2.3.2 Network-initiated de-registration procedure completion by the UE

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration required" and the de-registration request is for 3GPP access, the UE shall perform a local release of the PDU sessions over 3GPP access, if any. If a PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access in the same PLMN or in different PLMNs, the UE shall perform a local release of the user plane resources on 3GPP access. If there is an MA PDU session with user plane resources established on 3GPP access only, the UE shall perform a local release of the MA PDU session. If the MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. The UE shall stop the timer(s) T3346, T3396, T3584, T3585 and 5GSM back-off timer(s) not related to congestion control (see subclause 6.2.12), if running. If the UE is operating in single-registration mode, the UE shall also stop the ESM back-off timer(s) not related to congestion control (see subclause 6.3.6 in 3GPP TS 24.301 [15]), if running. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for 3GPP access. Furthermore, the UE shall, after the completion of the deregistration procedure, and the release of the existing NAS signalling connection, if any Tsor-cm timer(s) were running and have stopped, the UE shall attempt to obtain service on a higher priority PLMN (see 3GPP TS 23.122 [5]) on 3GPP access, otherwise initiate an initial registration. The UE should also re-establish any previously established PDU sessions over 3GPP access. For any previously established MA PDU sessions with user plane resources established on both accesses the UE should also re-establish the user plane resources over 3GPP access, and for any previously established MA PDU sessions with user plane resources established only on the 3GPP access the UE should re-establish the MA PDU session over 3GPP access.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration required" and the de-registration request is for non-3GPP access, the UE shall perform a local release of the PDU sessions over non-3GPP access, if any. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access in the same PLMN or in different PLMNs, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with a PDN connection as a user-plane resource and user plane resources established on non-3GPP access, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with user plane resources established on non-3GPP access, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session. The UE shall stop the timer(s) T3346, T3396, T3584 and T3585, if it is running. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for non-3GPP access. Furthermore, the UE shall, after the completion of the de-registration procedure, and the release of the existing NAS signalling connection, initiate an initial registration over non-3GPP. The UE should also re-establish any previously established PDU sessions over non-3GPP access the UE

should also re-establish the user plane resources over non-3GPP access, and for any previously established MA PDU sessions with user plane resources established only on the non-3GPP access the UE should re-establish the MA PDU session over non-3GPP access, and for any previously established MA PDU sessions with a PDN connection as a userplane resource and user plane resources established on non-3GPP access the UE should re-establish the user plane resources over non-3GPP access.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration required" and the de-registration request is for both 3GPP access and non-3GPP access when the UE is registered in the same PLMN for both accesses, the UE shall perform a local release of the MA PDU sessions and PDU sessions over both 3GPP access and non-3GPP access, if any. If an MA PDU session or a PDU sessions is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. The UE shall stop the timer(s) T3346, T3396, T3584 and T3585, if it is running. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for both 3GPP access and non-3GPP access. Furthermore, the UE shall, after the completion of the de-registration procedure, and the release of the existing NAS signalling connection, if any Tsor-cm timer(s) were running and have stopped, the UE shall attempt to obtain service on a higher priority PLMN (see 3GPP TS 23.122 [5]) on 3GPP access, otherwise initiate an initial registration over both 3GPP access and non-3GPP access. The UE should also re-establish any previously established MA PDU sessions the UE should also re-establish the MA PDU session and the user plane resources which were established previously.

NOTE 1: When the de-registration type indicates "re-registration required", user interaction is necessary in some cases when the UE cannot re-establish the PDU session (s), if any, automatically.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration not required" and the de-registration request is for 3GPP access, the UE shall perform a local release of the PDU sessions over 3GPP access, if any. If a PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access. If there is an MA PDU session with user plane resources established on 3GPP access on 3GPP access. If there is an MA PDU session with user plane resources established on 3GPP access only, the UE shall perform a local release of the MA PDU session. If the MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for 3GPP access.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration not required" and the de-registration request is for non-3GPP access, the UE shall perform a local release of the PDU sessions over non-3GPP access, if any. If there is an MA PDU session with user plane resources established on both 3GPP access and non-3GPP access in the same PLMN or in different PLMNs, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with a PDN connection as a user-plane resource and user plane resources established on non-3GPP access, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with user plane resources established on non-3GPP access, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with user plane resources established on non-3GPP access, the UE shall perform a local release of the user plane resources on non-3GPP access. If there is an MA PDU session with user plane resources established on non-3GPP access on the user plane resources on non-3GPP access. If there is an MA PDU session with user plane resources established on non-3GPP access only, the UE shall perform a local release of the MA PDU session. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for non-3GPP access.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message indicates "re-registration not required" and the de-registration request is for both 3GPP access and non-3GPP access when the UE is registered in the same PLMN for both accesses, the UE shall perform a local release of the MA PDU sessions and PDU sessions over both 3GPP access and non-3GPP access, if any. If an MA PDU session or a PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions. The UE shall send a DEREGISTRATION ACCEPT message to the network and enter the state 5GMM-DEREGISTERED for both 3GPP access and non-3GPP access.

Upon receiving the DEREGISTRATION REQUEST message, if the DEREGISTRATION REQUEST message includes the rejected NSSAI, the UE takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN until switching off

the UE, the UICC containing the USIM is removed, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as described in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

If there is one or more S-NSSAIs in the rejected NSSAI with the rejection cause "S-NSSAI not available due to maximum number of UEs reached", then for each S-NSSAI, the UE shall behave as follows:

- a) stop the timer T3526 associated with the S-NSSAI, if running;
- b) start the timer T3526 with:
 - 1) the back-off timer value received along with the S-NSSAI, if a back-off timer value is received along with the S-NSSAI that is neither zero nor deactivated; or
 - 2) an implementation specific back-off timer value, if no back-off timer value is received along with the S-NSSAI; and
- c) remove the S-NSSAI from the rejected NSSAI for the maximum number of UEs reached when the timer T3526 associated with the S-NSSAI expires.

Upon sending a DEREGISTRATION ACCEPT message, the UE shall delete the rejected NSSAI as specified in subclause 4.6.2.2.

Regardless of the 5GMM cause value received in the DEREGISTRATION REQUEST message via satellite NG-RAN,

- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the DEREGISTRATION REQUEST message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for roaming"; and
- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the DEREGISTRATION REQUEST message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for regional provision of service".

If the de-registration type indicates "re-registration required", then the UE shall ignore the 5GMM cause IE if received.

If the de-registration type indicates "re-registration not required", the UE shall take the actions depending on the received 5GMM cause value:

#3 (Illegal UE);

#6 (Illegal ME)

The message was received via 3GPP access and the UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

- In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure was performed in the current SNPN, the UE shall consider the USIM as invalid for the current SNPN until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall delete the list of equivalent PLMNs (if any) and shall enter the state 5GMM-DEREGISTERED.NO-SUPI.

If the UE is not registered for onboarding services in SNPN, the UE shall delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required". The USIM shall be considered as invalid also for non-EPS services until switching off or the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.7a in 3GPP TS 24.301 [15].

If the UE is registered for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#7 (5GS services not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 5GS services until the UE is switched off, or the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE supports access to an SNPN using credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure was performed in the current SNPN, the UE shall consider the USIM as invalid for the current SNPN until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall enter the state 5GMM-DEREGISTERED.NO-SUPI.

If the UE is not registered for onboarding services in SNPN, the UE shall delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

If the UE is registered for onboarding services in SNPN, the UE shall reset the registration attempt counter, store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

If the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#11 (PLMN not allowed).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs, shall reset the registration attempt counter. For 3GPP access the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH, and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE.

The UE shall store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1.

For 3GPP access the UE shall perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall perform network selection as defined in 3GPP TS 24.502 [18].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

If the UE also supports the registration procedure over the other access to the same PLMN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

If the UE receives the Disaster return wait range IE in the DEREGISTRATION REQUEST message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range IE in the ME.

#12 (Tracking area not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE.

If the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service". Otherwise, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters, EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

#13 (Roaming not allowed in this tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs (if available), reset the registration attempt counter. For 3GPP access the UE shall change to state 5GMM-DEREGISTERED.PLMN-SEARCH, and for non-3GPP access the UE shall change to state 5GMM-DEREGISTERED.LIMITED-SERVICE.

If the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming". Otherwise, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription.

For 3GPP access the UE shall perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall perform network selection as defined in 3GPP TS 24.502 [18].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

#15 (No suitable cells in tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE.

If the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming". Otherwise the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription.

The UE shall search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

If received over non-3GPP access and de-registration request is for non-3GPP access only, the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.2.3.4.

#22 (Congestion).

If the T3346 value IE is present in the DEREGISTRATION REQUEST message and the value indicates that this timer is neither zero nor deactivated, the UE shall proceed as described below, otherwise it shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.2.3.4.

The UE shall stop timer T3346 if it is running, set the 5GS update status to 5U2 NOT UPDATED, reset the registration attempt counter and enter the state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION.

The UE shall start timer T3346 with the value provided in the T3346 value IE.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and shall enter the state EMM-DEREGISTERED.

#27 (N1 mode not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter and shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE.

The UE shall disable the N1 mode capability for both 3GPP access and non-3GPP access (see subclause 4.9).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI. Additionally, the UE shall reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#36 (IAB-node operation not authorized).

This cause value is only applicable when received over 3GPP access by a UE operating as an IAB-node. This cause value received from a 5G access network other than 3GPP access or received by a UE not operating as an IAB-node is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

If:

1) the UE is not operating in SNPN access operation mode,

- the UE shall delete the list of equivalent PLMNs and reset the registration attempt counter and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5]; and
- ii) If the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required"; or
- 2) the UE is operating in SNPN access operation mode,
 - the UE shall reset the registration attempt counter and store the SNPN identity in the "temporarily forbidden SNPNs" list for 3GPP access. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5].

#62 (No network slices available).

The UE shall set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-DEREGISTERED.NORMAL-SERVICE or 5GMM-DEREGISTERED.PLMN-SEARCH. Additionally, the UE shall reset the registration attempt counter.

The UE receiving the rejected NSSAI in the DEREGISTRATION REQUEST message takes the following actions based on the rejection cause in the rejected S-NSSAI(s):

"S-NSSAI not available in the current PLMN or SNPN"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current PLMN or SNPN as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current PLMN or SNPN until switching off the UE, the UICC containing the USIM is removed, an entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available in the current registration area"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the current registration area as described in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI(s) in the current registration area until switching off the UE, the UE moving out of the current registration area, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.2.2.

"S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization"

The UE shall store the rejected S-NSSAI(s) in the rejected NSSAI for the failed or revoked NSSAA as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over any access until switching off the UE, the UICC containing the USIM is removed, the entry of the

"list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed or deleted as described in subclause 4.6.1 and 4.6.2.2.

"S-NSSAI not available due to maximum number of UEs reached"

The UE shall add the rejected S-NSSAI(s) in the rejected NSSAI for the maximum number of UEs reached as specified in subclause 4.6.2.2 and shall not attempt to use this S-NSSAI in the current PLMN or SNPN over the current access until switching off the UE, the UICC containing the USIM is removed, the entry of the "list of subscriber data" with the SNPN identity of the current SNPN is updated, or the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

If the UE has an allowed NSSAI or configured NSSAI that contains S-NSSAI(s) which are not included in the rejected NSSAI, the UE may stay in the current serving cell, apply the normal cell reselection process and start an initial registration with a requested NSSAI that includes any S-NSSAI from the allowed NSSAI or the configured NSSAI that is not in the rejected NSSAI. Otherwise the UE may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if the UE does not have an allowed NSSAI and each S-NSSAI in configured NSSAI, if available, was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" or "S-NSSAI not available due to maximum number of UEs reached" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and,

- if at least one S-NSSAI in the default configured NSSAI is not rejected, the UE may stay in the current serving cell, apply the normal cell reselection process, and start an initial registration with a requested NSSAI with that default configured NSSAI; or
- 2) if all the S-NSSAI(s) in the default configured NSSAI are rejected and at least one S-NSSAI is rejected due to "S-NSSAI not available in the current registration area",
 - i) if the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE; or
 - ii) if the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE.

Otherwise, the UE may perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5] and additionally, the UE may disable the N1 mode capability for the current PLMN or SNPN if each S-NSSAI in the default configured NSSAI was rejected with cause "S-NSSAI not available in the current PLMN or SNPN" or "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization" as described in subclause 4.9.

If the UE has neither allowed NSSAI for the current PLMN or SNPN nor configured NSSAI for the current PLMN or SNPN and has rejected NSSAI for the reached maximum number of UEs, and the UE wants to obtain services in the current serving cell without performing a PLMN selection or SNPN selection, the UE may stay in the current serving cell and attempt to use the rejected S-NSSAI(s) for the maximum number of UEs reached in the current serving cell after the rejected S-NSSAI(s) are removed as described in subclause 4.6.2.2.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#72 (Non-3GPP access to 5GCN not allowed).

If received over non-3GPP access when the UE is registered over non-3GPP access, or received over 3GPP access and de-registration request is for non-3GPP access when the UE is registered in the same PLMN for both accesses, the UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI for non-3GPP access. Additionally, the UE shall reset the registration attempt counter and enter the state 5GMM-DEREGISTERED for non-3GPP access.

NOTE 2: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).

The UE shall disable the N1 mode capability for non-3GPP access (see subclause 4.9.3).

As an implementation option, if the UE is not currently registered over 3GPP access, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5].

If received over 3GPP access and de-registration request is for 3GPP access only, the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.5.2.3.4.

#74 (Temporarily not authorized for this SNPN).

5GMM cause #74 is only applicable when received from a cell belonging to an SNPN. 5GMM cause #74 received from a cell not belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and shall store the SNPN identity in the "temporarily forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection for onboarding services according to 3GPP TS 23.122 [5].

#75 (Permanently not authorized for this SNPN).

5GMM cause #75 is only applicable when received from a cell belonging to an SNPN with a globally-unique SNPN identity. 5GMM cause #75 received from a cell not belonging to an SNPN or a cell belonging to an SNPN with a non-globally-unique SNPN identity is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall reset the registration attempt counter and store the SNPN identity in the "permanently forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN, selection for onboarding services according to 3GPP TS 23.122 [5].

#76 (Not authorized for this CAG or authorized for CAG cells only).

This cause value received via non-3GPP access or from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3.ROAMING NOT ALLOWED, store the 5GS update status according to clause 5.1.3.2.2, and reset the registration attempt counter.

If 5GMM cause #76 is received from:

- 1) a CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the DEREGISTRATION REQUEST message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when

the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or

- NOTE 3: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall delete the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN. In the case the "allowed CAG list" for the current PLMN only contains a range of CAG-IDs, how the UE deletes the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN is up to UE implementation. In addition:

- i) if the entry in the "CAG information list" for the current PLMN does not include an "indication that the UE is only allowed to access 5GS via CAG cells" or if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list"; or
- ii) if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".
- 2) a non-CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the DEREGISTRATION REQUEST message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received CAG information list IE or the Extended CAG information list IE when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 4: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN, if any. If the "CAG information list" stored in the UE does not include the current PLMN's entry, the UE shall add an entry for the current PLMN to the "CAG information list" and store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN. If the UE does not have a stored "CAG information list", the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN.

In addition:

 i) if the "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated CAG information; or ii) if the "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#77 (Wireline access area not allowed).

5GMM cause #77 is only applicable when received from a wireline access network by the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device). 5GMM cause #77 received from a 5G access network other than a wireline access network and 5GMM cause #77 received by the W-AGF acting on behalf of the FN-BRG are considered as abnormal cases and the behaviour of the UE is specified in subclause 5.5.2.3.4.

When received over wireline access network, the 5G-RG and the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2), shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI, shall reset the registration attempt counter, shall enter the state 5GMM-DEREGISTERED and shall act as specified in subclause 5.3.23.

- NOTE 5: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).
- #78 (PLMN not allowed to operate at the present UE location).

This cause value received from a non-satellite NG-RAN cell is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter. The UE shall store the PLMN identity and, if it is known, the current geographical location in the list of "PLMNs not allowed to operate at the present UE location" and shall start a corresponding timer instance (see subclause 4.23.2). The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list, eKSI and attach attempt counter as specified in 3GPP TS 24.301 [15] for the case when a DETACH REQUEST is received with the EMM cause with the same value and with detach type set to "re-attach not required".

#79 (UAS services not allowed).

- A UE which is not a UE supporting UAS services receiving this cause value shall considered it as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

A UE supporting UAS service shall set the 5GS update status to 5U2 NOT UPDATED and enter state 5GMM-DEREGISTERED.NORMAL-SERVICE or 5GMM-DEREGISTERED.PLMN-SEARCH. Additionally, the UE shall reset the registration attempt counter. If the UE re-attempt the registration procedure to the current PLMN, the UE shall not include the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of REGISTRATION REQUEST message.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU2 NOT UPDATED, reset the attach attempt counter and enter the state EMM-DEREGISTERED.

#93 (Onboarding services terminated).

If the UE is not registered for onboarding services in SNPN, this cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.5.2.3.4.

If the UE is not operating in SNPN access operation mode, the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5].

If the UE is operating in SNPN access operation mode, the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5].

NOTE 6: In case the configuration of one or more entries of the "list of subscriber data" was not completed at the time of network-initiated de-registration procedure, the UE can retry registration after the de-registration procedure is completed.

5.5.2.3.3 Network-initiated de-registration procedure completion by the network

The network shall stop timer T3522 upon receipt of the DEREGISTRATION ACCEPT message. The network shall enter state 5GMM-DEREGISTERED for 3GPP access if the de-registration request is for 3GPP access. The network shall enter state 5GMM-DEREGISTERED for non-3GPP access if the de-registration request is for non-3GPP access. The network shall enter state 5GMM-DEREGISTERED for both 3GPP access and non-3GPP access if the de-registration request is for both 3GPP access and non-3GPP access.

5.5.2.3.4 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Transmission failure of DEREGISTRATION ACCEPT message indication from lower layers.

The de-registration procedure shall be progressed and the UE shall send the DEREGISTRATION ACCEPT message.

b) DEREGISTRATION REQUEST, other 5GMM cause values than those treated in subclause 5.5.2.3.2, cases of 5GMM cause value#11, #15, #22, #72, #74, #75, #76, #77, #78, #79 and #93 that are considered as abnormal cases according to subclause 5.5.2.3.2 or no 5GMM cause IE is included, and the De-registration type IE indicates "re-registration not required".

The UE shall delete 5G-GUTI, TAI list, last visited registered TAI, list of equivalent PLMNs (if any), ngKSI, shall set the 5GS update status to 5U2 NOT UPDATED and shall start timer T3502.

A UE not supporting S1 mode may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5]; otherwise the UE shall enter the state 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall:

- enter the state 5GMM-DEREGISTERED and attempt to select E-UTRAN radio access technology and proceed with the appropriate EMM specific procedures. In this case, the UE may disable the N1 mode capability (see subclause 4.9); or
- enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5].

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall set the EPS update status to EU2 NOT UPDATED, enter the state EMM-DEREGISTERED and shall delete the EMM parameters 4G-GUTI, last visited registered TAI, TAI list and eKSI.

5.5.2.3.5 Abnormal cases in the network side

The following abnormal cases can be identified:

a) T3522 time-out

On the first expiry of the timer, the network shall retransmit the DEREGISTRATION REQUEST message and shall start timer T3522. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3522, the deregistration procedure shall be aborted. The network shall change to the state 5GMM-DEREGISTERED for the access type which the de-registration procedure is intended for.

b) Lower layer failure

The de-registration procedure is aborted. The network shall change to the state 5GMM-DEREGISTERED for the access type which the de-registration procedure is intended for.

c) De-registration procedure collision

If the network receives a DEREGISTRATION REQUEST message with "switch off" indication, before the network-initiated de-registration procedure has been completed, both procedures shall be considered completed.

If the network receives a DEREGISTRATION REQUEST message without "switch off" indication, before the network-initiated de-registration procedure has been completed, the network shall send a DEREGISTRATION ACCEPT message to the UE.

d) De-registration and registration procedure for initial registration collision

If the network receives a REGISTRATION REQUEST message indicating either "initial registration" or "emergency registration" in the 5GS registration type IE before the network-initiated de-registration procedure has been completed, the network shall abort the de-registration procedure and the registration procedure shall be progressed after the PDU sessions associated with the access type the REGISTRATION REQUEST message is sent over have been deleted.

- NOTE 1: The above collision case is valid if the DEREGISTRATION REQUEST message indicates the access type over which the initial registration procedure is attempted otherwise both the procedures are progressed.
- e) De-registration and registration procedure for mobility and periodic registration update collision

If the network sent a DEREGISTRATION REQUEST message without 5GMM cause value #11, #12, #13 or #15 and the network receives a REGISTRATION REQUEST message indicating either "mobility registration updating" or "periodic registration updating" in the 5GS registration type IE before the network-initiated deregistration procedure has been completed, the de-registration procedure shall be progressed, i.e. the REGISTRATION REQUEST message shall be ignored.

If the network sent a DEREGISTRATION REQUEST message with 5GMM cause value #11, #12, #13 or #15 and the network receives a REGISTRATION REQUEST message indicating either "mobility registration updating" or "periodic registration updating" in the 5GS registration type IE before the network-initiated deregistration procedure has been completed, the de-registration procedure shall be aborted and the registration procedure shall be progressed.

- NOTE 2: The above collision case is valid if the DEREGISTRATION REQUEST message indicates the access type over which the mobility and periodic registration procedure is attempted otherwise both the procedures are progressed.
- f) De-registration and service request procedure collision

If the network receives a SERVICE REQUEST message or a CONTROL PLANE SERVICE REQUEST message before the network-initiated de-registration procedure has been completed (e.g. the DEREGISTRATION REQUEST message is pending to be sent to the UE), the network shall progress the de-registration procedure.

- NOTE 3: The above collision case is valid if the DEREGISTRATION REQUEST message indicates the access type over which the service request procedure is attempted otherwise both the procedures are progressed.
- g) De-registration requested for a UE not supporting CAG due to CAG restrictions

Based on operator policy, if the network-initiated de-registration procedure is triggered for a UE not supporting CAG due to CAG restrictions, the network shall send the DEREGISTRATION REQUEST message including a 5GMM cause value other than the 5GMM cause #76 (Not authorized for this CAG or authorized for CAG cells only).

NOTE 4: 5GMM cause #7 (5GS services not allowed), 5GMM cause #11 (PLMN not allowed), 5GMM cause #27 (N1 mode not allowed), 5GMM cause #73 (Serving network not authorized) can be used depending on the subscription of the UE and whether the UE roams or not.

5.5.3 eCall inactivity procedure

The eCall inactivity procedure is performed only in 3GPP access and applicable only to a UE configured for eCall only mode as specified in 3GPP TS 31.102 [22]. The procedure shall be started when:

- a) the UE is in any 5GMM-REGISTERED substate except substates 5GMM-REGISTERED.PLMN-SEARCH or 5GMM-REGISTERED.NO-CELL-AVAILABLE;
- b) the UE is in 5GMM-IDLE mode; and
- c) one of the following conditions applies:
 - 1) timer T3444 expires or is found to have already expired and timer T3445 is not running;
 - 2) timer T3445 expires or is found to have already expired and timer T3444 is not running; or
 - 3) timers T3444 and T3445 expire or are found to have already expired.

The UE shall then perform the following actions:

- a) stop other running timers (e.g. T3511, T3512);
- b) if the UE is currently registered to the network for 5GS services, perform a de-registration procedure;
- c) delete any 5G-GUTI, TAI list, last visited registered TAI, list of equivalent PLMNs, and ngKSI; and
- d) enter 5GMM-DEREGISTERED.eCALL-INACTIVE state.

5.5.4 Authentication and key agreement procedure for 5G ProSe UE-tonetwork relay

5.5.4.1 General

The purpose of the authentication and key agreement procedure for 5G ProSe UE-to-network relay is to perform the authentication for 5G ProSe remote UE initiated by the 5G ProSe UE-to-network relay and to agree on the K_{AUSF_P} and K_{NR_ProSe} when the security for 5G ProSe communication via 5G ProSe UE-to-network relay is performed over control plane as specified in 3GPP TS 33.503 [56].

The procedure as shown in figure 5.5.4.1.1 is initiated by the UE when the UE receives the ProSe direct link establishment request including the SUCI or the CP-PRUK ID of the 5G ProSe remote UE from the 5G ProSe remote UE, for establishing secure PC5 unicast link as specified in 3GPP TS 24.554 [19E].

If the network decides to process the relay key request message, the EAP based authentication and key agreement procedure is initiated and controlled by the network. The exchanges of EAP messages between the 5G ProSe remote UE and the network are relayed by the UE.

UE		AMF
Start T3527 —	RELAY KEY REQUEST	
2001010021		,
Stop T3527 -	RELAY KEY REJECT	
1	<key is="" rejected="" request=""></key>	
	OR	
Stop T3527	RELAY AUTHENTICATION REQUEST	
Stop 15527	<eap-request a="" massage=""></eap-request>	
Start T3527	RELAY AUTHENTICATION RESPONSE	
	<eap-response a="" eap-request="" message="" to=""></eap-response>	
	i	
	I DEL AV AUTHENTICATION DEQUEST	
Stop T3527 ←	<pre></pre>	
	1	
Start T3527	RELAY AUTHENTICATION RESPONSE	>
Start 15527	<eap-response b="" eap-request="" message="" to=""></eap-response>	
	i	
G4 T2527	I RELAY KEY ACCEPT	
Stop 1352/	<eap-success message=""></eap-success>	
	OR	
Stop T2527	RELAY KEY REJECT	
Sup 15527 -	<eap-failure message=""></eap-failure>	

Figure 5.5.4.1.1: Authentication and key agreement procedure for 5G ProSe UE-to-network relay

5.5.4.2 ProSe relay transaction identity (PRTI)

Upon receiving a ProSe direct link establishment request from a 5G ProSe remote UE for establishing a secure PC5 unicast link as specified in 3GPP TS 24.554 [19E], the UE shall allocate an available PRTI value for the authentication and key agreement procedure for 5G ProSe UE-to-network relay and associate this PRTI value with the 5G ProSe remote UE.

The UE shall release the PRTI value allocated to the authentication and key agreement procedure for 5G ProSe UE-tonetwork relay when the authentication and key agreement procedure for 5G ProSe UE-to-network relay completes or is aborted.

5.5.4.3 UE-initiated authentication and key agreement procedure initiation

Upon receiving a ProSe direct link establishment request from the 5G ProSe remote UE including the SUCI or the CP-PRUK ID of the 5G ProSe remote UE, for establishing a secure PC5 unicast link as specified in 3GPP TS 24.554 [19E] when the security for 5G ProSe communication via 5G ProSe UE-to-network relay is performed over control plane as specified in 3GPP TS 33.503 [56], the UE shall:

- a) allocate a PRTI value as specified in clause 5.5.4.2;
- b) create a RELAY KEY REQUEST message;
- c) set the PRTI IE of the RELAY KEY REQUEST message to the allocated PRTI value;
- d) set the relay key request parameters IE of the RELAY KEY REQUEST message with SUCI or the CP-PRUK ID, relay service code, and nonce_1 received from the of the 5G ProSe remote UE;
- e) send the RELAY KEY REQUEST message; and
- f) start the timer T3527 upon sending the RELAY KEY REQUEST message.

5.5.4.4 UE-initiated authentication and key agreement procedure accepted by the network

Upon receiving the RELAY KEY REQUEST message, the AMF processes the message and interacts with the AUSF of the 5G ProSe remote UE as specified in 3GPP TS 33.503 [56]. If EAP-AKA' authentication for the 5G ProSe remote UE is initiated by the network, the AMF shall:

- a) create a RELAY AUTHENTICATION REQUEST message;
- b) set the PRTI IE of the RELAY AUTHENTICATION REQUEST message to the PRTI value of the received RELAY KEY REQUEST message;
- c) set the EAP message IE of the RELAY AUTHENTICATION REQUEST message to EAP request message received from the AUSF; and
- d) send the RELAY AUTHENTICATION REQUEST message to the UE.

Upon receiving the RELAY AUTHENTICATION REQUEST message, the UE stops the timer T3527 and forwards the EAP message to the 5G ProSe remote UE as specified in 3GPP TS 24.554 [19E].

Upon receiving the EAP response message from the 5G ProSe remote UE as specified in 3GPP TS 24.554 [19E], the UE shall:

- a) create a RELAY AUTHENTICATION RESPONSE message;
- b) set the PRTI IE of the RELAY AUTHENTICATION RESPONSE message to the PRTI value of the received RELAY AUTHENTICATION REQUEST message;
- c) set the EAP message IE of the RELAY AUTHENTICATION RESPONSE message to EAP response message received from the 5G ProSe remote UE; and
- d) start a timer T3527 upon sending the RELAY AUTHENTICATION RESPONSE message to the AMF.

After receiving the RELAY AUTHENTICATION RESPONSE message, the AMF may send a new RELAY AUTHENTICATION REQUEST message carrying EAP request message according to further handling of EAP-AKA' authentication from the AUSF as specified in 3GPP TS 33.503 [56]. The UE repeats the handling of the RELAY AUTHENTICATION REQUEST message as described above.

Upon receiving the message from the AUSF that the authentication is successful, the AMF shall:

- a) create a RELAY KEY ACCEPT message;
- b) set the PRTI IE of the RELAY KEY ACCEPT message to the PRTI value of the RELAY KEY REQUEST message;
- c) include the EAP message IE of the RELAY KEY ACCEPT message set to EAP-success message received from the AUSF, if any;
- d) include the relay key response parameters IE of the RELAY KEY ACCEPT message set to K_{NR_ProSe} and nonce_2 received from AUSF; and
- e) include the CP-PRUK ID in the relay key response parameters IE of the RELAY KEY ACCEPT message.

Upon receiving the RELAY KEY ACCEPT message, the UE shall forward the EAP-success message, if any, and nonce_2 to the 5G ProSe remote UE as specified in 3GPP TS 24.554 [19E], and consider the authentication is completed successfully. The UE shall store the CP-PRUK ID to be used in the remote UE report procedure as specified in clause 6.6.2.2.

5.5.4.5 UE-initiated authentication and key agreement procedure not accepted by the network

If the UE-initiated authentication and key agreement procedure is not accepted by the network, the AMF shall:

- a) create a RELAY KEY REJECT message;
- b) set the PRTI IE of the RELAY KEY REJECT message to the PRTI value of the received RELAY KEY REQUEST message if the network decides to reject the RELAY KEY REQUEST message; or
- NOTE: The network decides to reject the RELAY KEY REQUEST message when e.g. the CP-PRUK is not found in the network.

set the PRTI IE of the RELAY KEY REJECT message to the PRTI value of the received RELAY AUTHENTICATION RESPONSE message and include the EAP message IE set with EAP-failure message if the AMF receives an EAP-failure message from the AUSF; and

c) send the RELAY KEY REJECT message to the UE.

Upon receiving the RELAY KEY REJECT message, the UE shall consider the authentication has failed and perform the PC5 signalling protocol procedure as specified in subclause 7.2.2.5 of 3GPP 24.554 [19E].

5.5.4.6 Abnormal cases in the UE

The following abnormal cases in the UE can be identified:

a) Transmission failure of RELAY KEY REQUEST message or RELAY KEY AUTHENTICATION RESPONSE message indication from lower layers.

The UE shall abort the authentication and key agreement procedure for 5G ProSe UE-to-network relay and perform the PC5 signalling protocol procedure as specified in subclause 7.2.2.5 of 3GPP 24.554 [19E].

b) Expiry of timer T3527.

The UE shall, on the first expiry of the timer T3527, retransmit the RELAY KEY REQUEST message or the RELAY KEY AUTHENTICATION RESPONSE message and shall reset and start timer T3527. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3527, the procedure shall be aborted.

c) Collision between the authentication and key agreement procedure for 5G ProSe UE-to-network relay and deregistration procedure.

The UE shall abort the authentication and key agreement procedure for 5G ProSe UE-to-network relay, proceed with the network initiated de-registration procedure, and perform the PC5 signalling protocol procedure as specified in subclause 7.2.2.5 of 3GPP 24.554 [19E].

5.5.4.7 Abnormal cases on the network side

The following abnormal cases on the network side can be identified:

a) Lower layer failure before the RELAY KEY AUTHENTICATION RESPONSE message is received.

The network shall abort the authentication and key agreement procedure for 5G ProSe UE-to-network relay.

b) Collision between the authentication and key agreement procedure for 5G ProSe UE-to-network relay and deregistration procedure.

The network shall abort the authentication and key agreement procedure for 5G ProSe UE-to-network relay and proceed with the UE-initiated de-registration procedure.

c) Collision between the authentication and key agreement procedure for 5G ProSe UE-to-network relay and other 5GMM procedures other than in item b.

The network shall progress both procedures.

5.6 5GMM connection management procedures

5.6.1 Service request procedure

5.6.1.1 General

The purpose of the service request procedure is to change the 5GMM mode from 5GMM-IDLE to 5GMM-CONNECTED mode. If the UE is not using 5GS services with control plane CIoT 5GS optimization, this procedure is used to request the establishment of user-plane resources for PDU sessions which are established without user-plane resources. In latter case, the 5GMM mode can be the 5GMM-IDLE mode or the 5GMM-CONNECTED mode if the UE requires to establish user-plane resources for PDU sessions. If the UE is using 5GS services with control plane CIoT 5GS optimization, this procedure can be used for UE initiated transfer of user data via the control plane from 5GMM-IDLE mode.

NOTE 1: The lower layer indicates when the user-plane resources for PDU sessions are successfully established or released.

This procedure is used when:

- the network has downlink signalling pending over 3GPP access and the UE is in 5GMM-IDLE mode over 3GPP access;
- the network has downlink signalling pending over non-3GPP access, the UE is in 5GMM-IDLE mode over non-3GPP access and in 5GMM-IDLE or 5GMM-CONNECTED mode over 3GPP access;
- the UE has uplink signalling pending over 3GPP access and the UE is in 5GMM-IDLE mode over 3GPP access;
- the network has downlink user data pending over 3GPP access and the UE is in 5GMM-IDLE mode over 3GPP access;
- the network has downlink user data pending over non-3GPP access, the UE is in 5GMM-IDLE mode over non-3GPP access and in 5GMM-IDLE or 5GMM-CONNECTED mode over 3GPP access;
- the UE has user data pending over 3GPP access and the UE is in 5GMM-IDLE or 5GMM-CONNECTED mode over 3GPP access;
- the UE has user data pending over non-3GPP access and the UE is in 5GMM-CONNECTED mode over non-3GPP access;
- the UE in 5GMM-IDLE mode over non-3GPP access, receives an indication from the lower layers of non-3GPP access, that the access stratum connection is established between the UE and the network, if T3346 is not running;
- the UE in 5GMM-IDLE or 5GMM-CONNECTED mode over 3GPP access receives a request from the upper layers to perform emergency services fallback and performs emergency services fallback as specified in subclause 4.13.4.2 of 3GPP TS 23.502 [9];
- the UE has to request resources for V2X communication over PC5;
- the UE has to request resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5;
- the MUSIM UE in 5GMM-IDLE mode requests the network to remove the paging restriction; or
- the MUSIM UE requests the release of the NAS signalling connection or rejects the paging request from the network.

This procedure shall not be used for:

- a) initiating user data transfer; or
- b) PDU session management related signalling other than for performing UE-requested PDU session release procedure related to a PDU session for LADN or for performing the UE-requested PDU session modification procedure to indicate a change of 3GPP PS data off UE status;

when the UE is located outside the LADN service area.

- In NB-N1 mode, this procedure shall not be used to request the establishment of user-plane resources:
 - a) for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources if there is currently:
 - 1) no user-plane resources established for the UE;
 - 2) user-plane resources established for:
 - i) one PDU session and the Multiple user-plane resources support bit was set to "Multiple user-plane resources not supported" in the 5GMM capability IE; or
 - ii) two PDU sessions and the Multiple user-plane resources support bit was set to "Multiple user-plane resources supported" in the 5GMM capability IE; or
 - b) for additional PDU sessions, if the number of PDU sessions for which user-plane resources are currently established is equal to the UE's maximum number of supported user-plane resources.

The service request procedure is initiated by the UE, however, it can be triggered by the network by means of:

- the paging procedure (see subclause 5.6.2) for the transfer of downlink signalling or user data pending over 3GPP access to a UE in 5GMM-IDLE mode over 3GPP access;
- the paging procedure (see subclause 5.6.2) for the transfer of downlink signalling or user data pending over non-3GPP access to a UE in 5GMM-IDLE mode over 3GPP access and in 5GMM-IDLE mode over non-3GPP access;
- the notification procedure (see subclause 5.6.3) for the transfer of downlink signalling or user data pending over non-3GPP access to a UE in 5GMM-CONNECTED mode over 3GPP access and in 5GMM-IDLE mode over non-3GPP access; or
- the notification procedure (see subclause 5.6.3) for the transfer of downlink signalling or user data pending over 3GPP access to a UE in 5GMM-IDLE mode over 3GPP access and in 5GMM-CONNECTED mode over non-3GPP access.
- NOTE 2: In case the UE is in 5GMM-IDLE mode over 3GPP access and in 5GMM-CONNECTED mode over non-3GPP access and downlink signalling or user data pending over 3GPP access needs to be transferred, the AMF can trigger either the notification procedure or the paging procedure based on implementation.

The UE shall invoke the service request procedure when:

- a) the UE, in 5GMM-IDLE mode over 3GPP access, receives a paging request from the network;
- NOTE 3: As an implementation option, the MUSIM UE is allowed to not invoke service request to respond to paging based on the information available in the paging message, e.g. voice service indication.
- b) the UE, in 5GMM-CONNECTED mode over 3GPP access, receives a notification from the network with access type indicating non-3GPP access;
- c) the UE, in 5GMM-IDLE mode over 3GPP access, has uplink signalling pending (except in case i);
- d) the UE, in 5GMM-IDLE mode over 3GPP access, has uplink user data pending (except in case j);
- e) the UE, in 5GMM-CONNECTED mode or in 5GMM-CONNECTED mode with RRC inactive indication, has user data pending due to no user-plane resources established for PDU session(s) used for user data transport;
- f) the UE in 5GMM-IDLE mode over non-3GPP access, with T3346 not active or upon expiry of T3346, receives or has already received an indication from the lower layers of non-3GPP access, that the access stratum connection is established between the UE and the network;

- g) the UE, in 5GMM-IDLE mode over 3GPP access, receives a notification from the network with access type indicating 3GPP access when the UE is in 5GMM-CONNECTED mode over non-3GPP access;
- h) the UE, in 5GMM-IDLE, 5GMM-CONNECTED mode over 3GPP access, or 5GMM-CONNECTED mode with RRC inactive indication, receives a request from the upper layers to perform emergency services fallback and performs emergency services fallback as specified in subclause 4.13.4.2 of 3GPP TS 23.502 [9];
- i) the UE, in 5GMM-CONNECTED mode over 3GPP access or in 5GMM-CONNECTED mode with RRC inactive indication, receives a fallback indication from the lower layers (see subclauses 5.3.1.2 and 5.3.1.4) and the UE has a pending NAS procedure other than a registration, service request, or de-registration procedure;
- j) the UE, in 5GMM-CONNECTED mode over 3GPP access or in 5GMM-CONNECTED mode with RRC inactive indication, receives a fallback indication from the lower layers (see subclauses 5.3.1.2 and 5.3.1.4) and the UE has pending uplink user data for PDU session(s) with user-plane resources already established but no pending NAS procedure;
- k) the UE, in 5GMM-CONNECTED mode and has a NAS signalling connection only, is using 5GS services with control plane CIoT 5GS optimization and has pending user data to be sent via user-plane resources;
- 1) the UE in 5GMM-IDLE mode over 3GPP access has to request resources for V2X communication over PC5 (see 3GPP TS 23.287 [6C]);
- m) the network supports the paging restriction, and the MUSIM UE in 5GMM-IDLE mode is requesting the network to remove the paging restriction;
- n) the UE in 5GMM-IDLE mode over 3GPP access
 - has to request resources for 5G ProSe direct discovery over PC5 or 5G ProSe direct communication over PC5 (see 3GPP TS 23.304 [6E]); or
 - acts as a 5G ProSe layer-2 UE-to-network relay UE and receives a trigger from lower layers to establish the NAS signalling connection (see 3GPP TS 23.304 [6E]);
- o) the network supports the N1 NAS signalling connection release, the MUSIM UE,
 - is in 5GMM-CONNECTED mode, requests the network to release the NAS signalling connection and, if the network supports the paging restriction, optionally includes paging restriction;
 - is in 5GMM-CONNECTED mode with RRC inactive indication, requests the network to release the NAS signalling connection and, if the network supports the paging restriction, optionally includes paging restriction; or
 - is in 5GMM-CONNECTED mode with RRC inactive indication, rejects the RAN paging, requests the network to release the NAS signalling connection and, if the network supports the paging restriction, optionally includes paging restriction; or
- p) the network supports the reject paging request, the MUSIM UE in 5GMM-IDLE mode when responding to paging rejects the paging request from the network, requests the network to release the NAS signalling connection and, if the network supports the paging restriction, optionally includes paging restriction.

If one of the above criteria to invoke the service request procedure is fulfilled, then the service request procedure shall only be initiated by the UE when the following conditions are fulfilled:

- its 5GS update status is 5U1 UPDATED, and the TAI of the current serving cell is included in the TAI list; and
- no 5GMM specific procedure is ongoing.

The UE shall not invoke the service request procedure when the UE is in the state 5GMM-SERVICE-REQUEST-INITIATED.

The MUSIM UE shall not initiate service request procedure for requesting the network to release the N1 NAS signalling connection if the UE is registered for emergency services or if the UE has an emergency PDU session established. To enable the emergency call back, the UE shall not initiate service request procedure for requesting the network to release the NAS signalling connection for a UE implementation-specific duration of time after the completion of the emergency services.

UE	AS	AMF
Start T3517	SERVICE REQUEST	
Stop T3517	SERVICE ACCEPT	
	OR	
Start T3517	SERVICE REQUEST (for emergency services fallback)	
Stop T3517	AS indication	
	OR	
Start T3517	SERVICE REQUEST	
Stop T3517	SERVICE REJECT	
	Figure 5.6.1.1.1: Service Request procedure (Part 1)	
UE	AS	AMF
Start T3517	CONTROL PLANE SERVICE REQUEST	
Stop T3517	SERVICE ACCEPT	_
	OR	
Start T3517	CONTROL PLANE SERVICE REQUEST	
Stop T3517	AS indication about release of RRC connection	
Start T3517	CONTROL PLANE SERVICE REQUEST	
G. 50517	SERVICE REJECT	



A service request attempt counter is used to limit the number of service request attempts and no response from the network. The service request attempt counter shall be incremented as specified in subclause 5.6.1.7.

The service request attempt counter shall be reset when:

- a registration procedure for mobility and periodic registration update is successfully completed;
- a service request procedure is successfully completed;
- a service request procedure is rejected as specified in subclause 5.6.1.5 or subclause 5.3.20; or
- the UE moves to 5GMM-DEREGISTERED state.

5.6.1.2 Service request procedure initiation

5.6.1.2.1 UE is not using 5GS services with control plane CIoT 5GS optimization

The UE initiates the service request procedure by sending a SERVICE REQUEST message to the AMF. The UE shall start timer T3517 and enter the state 5GMM-SERVICE-REQUEST-INITIATED.

If the UE is sending the SERVICE REQUEST message from 5GMM-IDLE mode and the UE needs to send noncleartext IEs, the UE shall send the SERVICE REQUEST message including the NAS message container IE as described in subclause 4.4.6.

For cases a), b), and g) in subclause 5.6.1.1, the service type IE in the SERVICE REQUEST message shall be set to "mobile terminated services".

For cases c), d), e), f), i), j), l) m) and n) in subclause 5.6.1.1, if the UE is a UE configured for high priority access in selected PLMN, the service type IE in the SERVICE REQUEST message shall be set to "high priority access".

For case a) in subclause 5.6.1.1:

- a) if the paging request includes an indication for non-3GPP access type, the Allowed PDU session status IE shall be included in the SERVICE REQUEST message. If the UE has established the PDU session(s) over the non-3GPP access for which the associated S-NSSAI(s) are included in the allowed NSSAI for 3GPP access, the UE shall indicate the PDU session(s) for which the UE allows the user-plane resources to be re-established over 3GPP access in the Allowed PDU session status IE. Otherwise, the UE shall not indicate any PDU session(s) in the Allowed PDU session status IE; and
- b) if the UE has uplink user data pending to be sent over 3GPP access, the Uplink data status IE shall be included in the SERVICE REQUEST message to indicate the PDU session(s) for which the UE has pending user data to be sent. Otherwise, the Uplink data status IE shall not be included in the SERVICE REQUEST message.

For case b) in subclause 5.6.1.1:

- a) the Allowed PDU session status IE shall be included in the SERVICE REQUEST message. If the UE has the PDU session(s) over the non-3GPP access for which the associated S-NSSAI(s) are included in the allowed NSSAI for 3GPP access, the UE shall indicate the PDU session(s) for which the UE allows the user-plane resources to be re-established over 3GPP access in the Allowed PDU session status IE. Otherwise, the UE shall not indicate any PDU session(s) in the Allowed PDU session status IE; and
- b) if the UE has uplink user data pending to be sent over 3GPP access, the Uplink data status IE shall be included in the SERVICE REQUEST message to indicate the PDU session(s) for which the UE has pending user data to be sent. Otherwise, the Uplink data status IE shall not be included in the SERVICE REQUEST message.

When the Allowed PDU session status IE is included in the SERVICE REQUEST message, the UE shall indicate that a PDU session is not allowed to be transferred to the 3GPP access if the 3GPP PS data off UE status is "activated" for the corresponding PDU session and the UE is not using the PDU session to send uplink IP packets for any of the 3GPP PS data off exempt services (see subclause 6.2.10).

For case c) in subclause 5.6.1.1, the Uplink data status IE shall not be included in the SERVICE REQUEST message except if the UE has one or more active always-on PDU sessions associated with the access type over which the SERVICE REQUEST message is sent. If the UE is not a UE configured for high priority access in selected PLMN and:

- a) if the SERVICE REQUEST message is triggered by a request for emergency services from the upper layer, the UE shall set the service type IE in the SERVICE REQUEST message to "emergency services"; or
- b) otherwise, the UE shall set the service type IE to "signalling".

When the UE is in a non-allowed area or is not in an allowed area as specified in subclause 5.3.5 and:

- a) if the uplink signalling pending is to indicate a change of 3GPP PS data off UE status for a PDU session, the UE shall set the service type IE in the SERVICE REQUEST message to "elevated signalling", and shall not include the Uplink data status IE in the SERVICE REQUEST message even if the UE has one or more active always-on PDU sessions associated with the access type over which the SERVICE REQUEST message is sent; or
- b) otherwise, the UE shall not initiate service request procedure except for emergency services, high priority access or responding to paging or notification.

For cases d) and e) in subclause 5.6.1.1, the Uplink data status IE shall be included in the SERVICE REQUEST message to indicate the PDU session(s) the UE has pending user data to be sent. If the UE is not a UE configured for high priority access in selected PLMN:

- a) if there exists an emergency PDU session which is indicated in the Uplink data status IE the service type IE in the SERVICE REQUEST message shall be set to "emergency services"; or
- b) otherwise, the service type IE in the SERVICE REQUEST message shall be set to "data".
- NOTE 1: For a UE in NB-N1 mode, the Uplink data status IE cannot be used to request the establishment of userplane resources such that there will be user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources.

For case f) in subclause 5.6.1.1:

- a) if the UE has uplink user data pending to be sent, the Uplink data status IE shall be included in the SERVICE REQUEST message to indicate the PDU session(s) the UE has pending user data to be sent. If the UE is not a UE configured for high priority access in selected PLMN, the service type IE in the SERVICE REQUEST message shall be set to "data";
- b) otherwise, if the UE is not a UE configured for high priority access in selected PLMN, the service type IE in the SERVICE REQUEST message shall be set to "signalling".

For case g) in subclause 5.6.1.1, if the UE has uplink user data pending to be sent, the Uplink data status IE shall be included in the SERVICE REQUEST message to indicate the PDU session(s) the UE has pending user data to be sent.

For case h) in subclause 5.6.1.1, the UE shall send a SERVICE REQUEST message with service type set to "emergency services fallback" and without an Uplink data status IE.

For case i) in subclause 5.6.1.1, if the UE is not configured for high priority access in selected PLMN, the UE shall set the Service type IE in the SERVICE REQUEST message as follows:

- a) if the pending message is an UL NAS TRANSPORT message with the Request type IE set to "initial emergency request" or "existing emergency PDU session", the UE shall set the Service type IE in the SERVICE REQUEST message to "emergency services"; or
- b) otherwise, the UE shall set the Service type IE in the SERVICE REQUEST message to "signalling".

For case j) in subclause 5.6.1.1:

- a) the UE shall include the Uplink data status IE in the SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any; and
- b) if the UE is not a UE configured for high priority access in selected PLMN, the UE shall set the Service type IE in the SERVICE REQUEST message as follows:
 - 1) if there is an emergency PDU session which is indicated in the Uplink data status IE, the UE shall set the Service type IE in the SERVICE REQUEST message to "emergency services"; or
 - 2) if there is no emergency PDU session which is indicated in the Uplink data status IE, the UE shall set the Service type IE in the SERVICE REQUEST message to "data".

For case l) and n) in subclause 5.6.1.1, if the UE is not a UE configured for high priority access in selected PLMN:

- a) if there exists an emergency PDU session which is indicated in the Uplink data status IE the service type IE in the SERVICE REQUEST message shall be set to "emergency services"; or
- b) otherwise, the service type IE in the SERVICE REQUEST message shall be set to "signalling".

For case m) in subclause 5.6.1.1, the UE shall not include the Paging restriction IE in the SERVICE REQUEST message and set Service type to "signalling". The UE may include the UE request type IE and set Request type to "NAS signalling connection release" to remove the paging restriction and request the release of the NAS signalling connection at the same time. If the UE requests the release of the NAS signalling connection, the UE shall not include the Uplink data status IE in the SERVICE REQUEST message.

For cases o and p in subclause 5.6.1.1, the UE shall not include the Uplink data status IE and the Allowed PDU session status IE in the SERVICE REQUEST message. Further,

- for case o in subclause 5.6.1.1, the UE shall set Request type to "NAS signalling connection release" in the UE request type IE and Service type to "signalling";
- for case p in subclause 5.6.1.1, the UE shall set Request type to "Rejection of paging" in the UE request type IE and Service type to "mobile terminated services"; and

may include its paging restriction preference in the Paging restriction IE in the SERVICE REQUEST message.

The UE shall include a valid 5G-S-TMSI in the 5G-S-TMSI IE of the SERVICE REQUEST message.

For all cases except cases o) and p) in subclause 5.6.1.1, if the UE has one or more active always-on PDU sessions associated with the access type over which the SERVICE REQUEST message is sent and the user-plane resources for these PDU sessions are not established, the UE shall include the Uplink data status IE in the SERVICE REQUEST message and indicate that the UE has pending user data to be sent for those PDU sessions.

If the UE has one or more active PDU sessions which are not accepted by the network as always-on PDU sessions and no uplink user data pending to be sent for those PDU sessions, the UE shall not include those PDU sessions in the Uplink data status IE in the SERVICE REQUEST message.

The Uplink data status IE may be included in the SERVICE REQUEST message to indicate which PDU session(s) associated with the access type the SERVICE REQUEST message is sent over have pending user data to be sent.

The PDU session status information element may be included in the SERVICE REQUEST message to indicate:

- the single access PDU session(s) not in 5GSM state PDU SESSION INACTIVE in the UE associated with the access type the SERVICE REQUEST message is sent over; and
- the MA PDU session(s) not in 5GSM state PDU SESSION INACTIVE and having user plane resources established in the UE on the access the SERVICE REQUEST message is sent over.

If the SERVICE REQUEST message includes a NAS message container IE, the AMF shall process the SERVICE REQUEST message that is obtained from the NAS message container IE as described in subclause 4.4.6.

If the UE has an emergency PDU session over the non-current access, it shall not initiate the SERVICE REQUEST message with the service type IE set to "emergency services" over the current access, unless the SERVICE REQUEST message has to be initiated to perform handover of an existing emergency PDU session from the non-current access to the current access.

NOTE 2: Transfer of an existing emergency PDU session between 3GPP access and non-3GPP access is needed e.g. if the UE determines that the current access is no longer available.

5.6.1.2.2 UE is using 5GS services with control plane CIoT 5GS optimization

The UE shall send a CONTROL PLANE SERVICE REQUEST message, start T3517 and enter the state 5GMM-SERVICE-REQUEST-INITIATED.

For case a), and case b) in subclause 5.6.1.1, the Control plane service type of the CONTROL PLANE SERVICE REQUEST message shall indicate "mobile terminating request". If:

a) the UE only has uplink CIoT user data or SMS to be sent, the UE shall:

- 1) if the data size is not more than 254 octets and there is no other optional IE to be included in the message:
 - i) for sending CIoT user data, set the Data type field to "control plane user data", include the PDU session ID, data, and Downlink data expected (DDX) (if available), in the CIoT small data container IE; and
 - ii) for sending SMS, set the Data type field to "SMS", include SMS in the CIoT small data container IE; and
- 2) otherwise if the data size is more than 254 octets or there are other optional IEs to be included in the message:
 - i) for sending CIoT user data, set the Payload container type IE to "CIoT user data container", include the PDU session ID in the PDU session ID IE and include data in the Payload container IE as described in subclause 5.4.5.2.2; and
 - ii) for sending SMS, set the Payload container type IE to "SMS" and include data in the Payload container IE as described in subclause 5.4.5.2.2; and
- b) the paging request or the notification includes an indication for non-3GPP access type, the UE has at least one PDU session that is not associated with control plane only indication, the Allowed PDU session status IE shall be included in the CONTROL PLANE SERVICE REQUEST message.
- NOTE 1: The term DDX used in the present document corresponds to the term NAS RAI used in 3GPP TS 23.502 [9].

For case c), and case d) if the UE has pending CIoT user data that is to be sent via the control plane in subclause 5.6.1.1, the UE shall set the Control plane service type of the CONTROL PLANE SERVICE REQUEST message to "mobile originating request". If the UE has only uplink CIoT user data, SMS or location services message to be sent, the UE shall:

- a) if the data size is not more than 254 octets, there is no other optional IE to be included in the CONTROL PLANE SERVICE REQUEST message, and the data being sent is:
 - 1) CIoT user data, set the Data type field to "control plane user data", include the PDU session ID, data, and Downlink data expected (DDX) (if available), in the CIoT small data container IE;
 - 2) location services message, set the Data type field to "Location services message container" and Downlink data expected (DDX), if available, in the CIoT small data container IE, and:
 - i) if routing information is provided by upper layers:
 - A) set the length of additional information field in the CIoT small data container IE to the length of routing information provided by upper layer location services application (see subclause 9.11.3.67), and set the additional information field in the CIoT small data container IE to the routing information provided by upper layer location services application (see subclause 9.11.3.67); or
 - B) otherwise set the length of additional information field in the CIoT small data container IE to zero. In this case the Additional information field of the CIoT small data container IE shall not be included; and
 - set the Data contents field of the CIoT small data container IE to the location services message payload; or
 - 3) SMS, set the Data type field to "SMS", include SMS in the CIoT small data container IE; or
- b) otherwise if the data size is more than 254 octets or there are other optional IEs to be included in the CONTROL PLANE SERVICE REQUEST message, and the data being sent is:
 - 1) CIoT user data, set the Payload container type IE to "CIoT user data container", include the PDU session ID in the PDU session ID IE and include data in the Payload container IE as described in subclause 5.4.5.2.2;
 - 2) location services message, set the Payload container type IE to "Location services message container", include data in the Payload container IE as described in subclause 5.4.5.2.2. If the upper layer location services application provides the routing information set the Additional information IE to the routing information as described in subclause 5.4.5.2.2; or

3) SMS, set the Payload container type IE to "SMS" and include data in the Payload container IE as described in subclause 5.4.5.2.2.

For case a), and case b) in subclause 5.6.1.1, if the UE has pending user data that is to be sent via the user plane, the UE shall set the Control plane service type of the CONTROL PLANE SERVICE REQUEST message to "mobile terminating request". The UE shall include the Uplink data status IE in the CONTROL PLANE SERVICE REQUEST message to indicate which PDU session(s) have pending user data to be sent via user-plane resources.

For case c) in subclause 5.6.1.1, if the UE is in WB-N1 mode and the CONTROL PLANE SERVICE REQUEST message is triggered by a request for emergency services from the upper layer, the UE shall set the Control plane service type of the CONTROL PLANE SERVICE REQUEST message to "emergency services".

For cases d) and k), if the UE has pending user data that is to be sent via the user plane in subclause 5.6.1.1:

- a) and if there exists an emergency PDU session which is indicated in the Uplink data status IE, the UE shall set the Control plane service type of the CONTROL PLANE SERVICE REQUEST message to "emergency services"; or
- b) otherwise, the UE shall set the Control plane service type to "mobile originating request".

The UE shall include the Uplink data status IE in the CONTROL PLANE SERVICE REQUEST message to indicate which PDU session(s) have pending user data to be sent via user-plane resources.

NOTE 2: For a UE in NB-N1 mode, the Uplink data status IE cannot be used to request the establishment of userplane resources such that there will be user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources.

For case h) in subclause 5.6.1.1, if the UE is in WB-N1 mode and the UE does not have any PDU session that is associated with control plane only indication, the UE shall send a CONTROL PLANE SERVICE REQUEST message with the Control plane service type set to "emergency services fallback" and without an Uplink data status IE.

For case i) in subclause 5.6.1.1, the Control plane service type of the CONTROL PLANE SERVICE REQUEST message shall indicate "mobile originating request". If the pending message is an UL NAS TRANSPORT message with the Payload container type IE set to:

- a) "SMS", "Location services message container", or "CIoT user data container", the UE shall send the CONTROL PLANE SERVICE REQUEST and include the SMS, location services message, or CIoT user data as described in this subclause; or
- b) otherwise, the UE shall send the CONTROL PLANE SERVICE REQUEST:
 - 1) without including the CIoT small data container IE and without including the NAS message container IE if the UE has no other optional IE to be sent; or
 - 2) with the NAS message container IE if the UE has an optional IE to be sent as described in this subclause.

For case j) in subclause 5.6.1.1, the Control plane service type of the CONTROL PLANE SERVICE REQUEST message shall indicate "mobile originating request". The UE shall include the Uplink data status IE in the CONTROL PLANE SERVICE REQUEST message indicating the PDU session(s) for which user-plane resources were active prior to receiving the fallback indication, if any.

For cases o) and p) in subclause 5.6.1.1, the UE shall not include the Uplink data status IE and the Allowed PDU session status IE in the CONTROL PLANE SERVICE REQUEST message. Further,

- for case o) in subclause 5.6.1.1, the UE shall set Request type to "NAS signalling connection release" in the UE request type IE and Control plane service type to "mobile originating request";
- for case p) in subclause 5.6.1.1, the UE shall set Request type to "Rejection of paging" in the UE request type IE and Control plane service type to "mobile terminating request"; and

may include its paging restriction preferences in the Paging restriction IE in the CONTROL PLANE SERVICE REQUEST message.

For case m) in clause 5.6.1.1, the Control plane service type of the CONTROL PLANE SERVICE REQUEST message shall indicate "mobile originating request". The UE shall not include the Paging restriction IE in the CONTROL PLANE SERVICE REQUEST message. The UE may include the UE request type IE and set Request type to "NAS"

signalling connection release" to remove the paging restriction and request the release of the NAS signalling connection at the same time. If the UE requests the release of the NAS signalling connection, the UE shall not include the Uplink data status IE in the SERVICE REQUEST message.

The UE may include the PDU session status IE in the CONTROL PLANE SERVICE REQUEST message to indicate which PDU session(s) associated with the access type the CONTROL PLANE SERVICE REQUEST message is sent over are active in the UE.

5.6.1.3 Common procedure initiation

Upon receipt of the SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message, the AMF may initiate the common procedures e.g. the 5G AKA based primary authentication and key agreement procedure or the EAP based primary authentication and key agreement procedure.

5.6.1.4 Service request procedure accepted by the network

5.6.1.4.1 UE is not using 5GS services with control plane CIoT 5GS optimization

For cases other than h) in subclause 5.6.1.1, the UE shall treat the reception of the SERVICE ACCEPT message as successful completion of the procedure. The UE shall reset the service request attempt counter, stop timer T3517 and enter the state 5GMM-REGISTERED.

For case h) in subclause 5.6.1.1,

- a) the UE shall treat the indication from the lower layers when the UE has changed to S1 mode or E-UTRA connected to 5GCN (see 3GPP TS 23.502 [9]) as successful completion of the procedure and stop timer T3517;
- b) if a UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access (see subclause 4.9.2); and
- c) the AMF shall not check for CAG restrictions.

If the PDU session status information element is included in the SERVICE REQUEST message, then:

- a) for single access PDU sessions, the AMF shall:
 - perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE on the AMF side associated with the access type the SERVICE REQUEST message is sent over, but are indicated by the UE as being in 5GSM state PDU SESSION INACTIVE; and
 - 2) request the SMF to perform a local release of all those PDU sessions. If any of those PDU sessions is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and
- b) for MA PDU sessions, the AMF shall:
 - 1) for MA PDU sessions having user plane resources established in the AMF only on the access the SERVICE REQUEST message is sent over, but are indicated by the UE as no user plane resources established:
 - i) for all those MA PDU sessions without a PDN connection established as a user-plane resource, perform a local release of all those MA PDU sessions and request the SMF to perform a local release of all those MA PDU sessions. If the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and
 - ii) for all those MA PDU sessions with a PDN connection established as a user-plane resource, perform a local release of user plane resources of all those PDU sessions on the access the SERVICE REQUEST message is sent over and request the SMF to perform a local release of user plane resources of all those PDU sessions on the access type the SERVICE REQUEST message is sent over; and
 - 2) for MA PDU sessions having user plane resources established on both accesses in the AMF, but are indicated by the UE as no user plane resources established:
 - i) perform a local release of user plane resources of all those PDU sessions on the access the SERVICE REQUEST message is sent over; and

ii) request the SMF to perform a local release of user plane resources of all those PDU sessions on the access type the SERVICE REQUEST message is sent over. If the SERVICE REQUEST message is sent over 3GPP access and the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions.

If the AMF needs to initiate PDU session status synchronization or a PDU session status IE was included in the SERVICE REQUEST message, the AMF shall include a PDU session status IE in the SERVICE ACCEPT message to indicate:

- which single access PDU sessions associated with the access type the SERVICE ACCEPT message is sent over are not in 5GSM state PDU SESSION INACTIVE in the AMF; and
- which MA PDU sessions are not in 5GSM state PDU SESSION INACTIVE and having user plane resources established in the AMF on the access the SERVICE ACCEPT message is sent over.

If the PDU session status information element is included in the SERVICE ACCEPT message, then:

- a) for single access PDU sessions, the UE shall perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING on the UE side associated with the access type the SERVICE ACCEPT message is sent over, but are indicated by the AMF as in 5GSM state PDU SESSION INACTIVE. If a locally released PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions; and
- b) for MA PDU sessions, for all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING and have user plane resources established on the UE side associated with the access the SERVICE ACCEPT message is sent over, but are indicated by the AMF as no user plane resources established:
 - for MA PDU sessions having user plane resources established only on the access type the SERVICE ACCEPT message is sent over, the UE shall perform a local release of those MA PDU sessions. If a locally released MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions; and
 - 2) for MA PDU sessions having user plane resources established on both accesses, the UE shall perform a local release on the user plane resources on the access type the SERVICE ACCEPT message is sent over. If the user plane resources over 3GPP access are released and the MA PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions.

If the Uplink data status IE is included in the SERVICE REQUEST message and the UE is:

- a) not in NB-N1 mode; or
- b) in NB-N1 mode and the UE does not indicate a request to have user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources;

the AMF shall:

- a) indicate the SMF to re-establish the user-plane resources for the corresponding PDU sessions;
- b) include the PDU session reactivation result IE in the SERVICE ACCEPT message to indicate the user-plane resources re-establishment result of the PDU sessions for which the UE requested to re-establish the user-plane resources; and
- c) determine the UE presence in LADN service area and forward the UE presence in LADN service area towards the SMF, if the corresponding PDU session is a PDU session for LADN.

If the Allowed PDU session status IE is included in the SERVICE REQUEST message, the AMF shall:

- a) for a 5GSM message from each SMF that has indicated pending downlink signalling only, forward the received 5GSM message via 3GPP access to the UE after the SERVICE ACCEPT message is sent;
- b) for each SMF that has indicated pending downlink data only:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE; and

- 2) notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if:
 - i) for a UE not in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE; or
 - ii) for a UE in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE, and the resulting number of PDU sessions with established user-plane resources does not exceed the UE's maximum number of supported user-plane resources;
- c) for each SMF that have indicated pending downlink signalling and data:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE;
 - 2) notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if:
 - i) for a UE not in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE; or
 - ii) for a UE in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE, and the resulting number of PDU sessions with established user-plane resources does not exceed the UE's maximum number of supported user-plane resources; and
 - 3) discard the received 5GSM message for PDU session(s) associated with non-3GPP access; and
- d) include the PDU session reactivation result IE in the SERVICE ACCEPT message to indicate the successfully re-established user-plane resources for the corresponding PDU sessions, if any.

If due to regional subscription restrictions or access restrictions the UE is not allowed to access the TA or due to CAG restrictions the UE is not allowed to access the cell, but the UE has an emergency PDU session established, the AMF may accept the SERVICE REQUEST message and indicate to the SMF to perform a local release of all non-emergency PDU sessions (associated with 3GPP access if it is due to CAG restrictions) and informs the UE via the PDU session status IE in the SERVICE ACCEPT message. The AMF shall not indicate to the SMF to release the emergency PDU session. If the AMF indicated to the SMF to perform a local release of all non-emergency PDU sessions (associated with 3GPP access if it is due to CAG restrictions), the network shall behave as if the UE is registered for emergency services.

If the PDU session reactivation result IE is included in the SERVICE ACCEPT message indicating that the user-plane resources have been successfully reactivated for a PDU session that was requested by the UE in the Allowed PDU session status IE, the UE considers the corresponding PDU session to be associated with the 3GPP access. If the user-plane resources of a PDU session have been successfully reactivated over the 3GPP access, the AMF and SMF update the associated access type of the corresponding PDU session.

If the user-plane resources cannot be established for a PDU session, the AMF shall include the PDU session reactivation result IE in the SERVICE ACCEPT message indicating that user-plane resources for the corresponding PDU session cannot be re-established, and:

- a) if the user-plane resources cannot be established because the SMF indicated to the AMF that the UE is located out of the LADN service area (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #43 "LADN not available";
- b) if the user-plane resources cannot be established because the SMF indicated to the AMF that only prioritized services are allowed (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #28 "restricted service area";
- c) if the user-plane resources cannot be established because the SMF indicated to the AMF that the resource is not available in the UPF (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #92 "insufficient user-plane resources for the PDU session";
- e) if the user-plane resources cannot be established because the SMF indicated to the AMF that the S-NSSAI associated with the PDU session is unavailable due to NSAC (see 3GPP TS 29.502 [20A]), the AMF shall

include the PDU session reactivation result error cause IE with the 5GMM cause set to #69 "insufficient resources for specific slice"; or

- d) otherwise, the AMF may include the PDU session reactivation result error cause IE to indicate the cause of failure to re-establish the user-plane resources.
- NOTE 1: It is up to UE implementation when to re-send a request for user-plane re-establishment for the associated PDU session after receiving a PDU session reactivation result error cause IE with a 5GMM cause set to #92 "insufficient user-plane resources for the PDU session".
- NOTE 2: The UE can locally start a back-off timer after receiving a PDU session reactivation result error cause IE with a 5GMM cause set to #69 "insufficient resources for specific slice". The value of the back-off timer is up to UE implementation. Upon expiry of the back-off timer, the UE can re-send a request for user-plane re-establishment for the associated PDU session.

If the PDU session reactivation result IE is included in the SERVICE ACCEPT message indicating that the user-plane resources cannot be established for a PDU session that was requested by the UE in the Allowed PDU session status IE, the UE considers the corresponding PDU session to be associated with the non-3GPP access.

If the MUSIM UE does not include the Paging restriction IE in the SERVICE REQUEST message, the AMF shall delete any stored paging restriction for the UE and stop restricting paging.

For case m in subclause 5.6.1.1 when the MUSIM UE sets the Request type to "NAS signalling connection release" in the SERVICE REQUEST message, the AMF shall initiate the release of the N1 NAS signalling connection after the completion of the service request procedure.

For cases o and p in subclause 5.6.1.1 when the MUSIM UE sets the Request type to "NAS signalling connection release" or to "Rejection of paging" in the UE request type IE in the SERVICE REQUEST message and if the UE requests restriction of paging by including the Paging restriction IE, the AMF:

- if accepts the paging restriction, shall include the 5GS additional request result IE in the SERVICE ACCEPT message and set the Paging restriction decision to "paging restriction is accepted". The AMF shall store the paging restriction of the UE and enforce these restrictions in the paging procedure as described in clause 5.6.2; or
- if rejects the paging restriction, shall include the 5GS additional request result IE in the SERVICE ACCEPT message and set the Paging restriction decision to "paging restriction is rejected", and shall discard the received paging restriction. The AMF shall delete any stored paging restriction for the UE and stop restricting paging; and

the AMF shall initiate the release of the N1 NAS signalling connection as follows:

- for case o in subclause 5.6.1.1, after the completion of the service request procedure;
- for case p in subclause 5.6.1.1, after the completion of the generic UE configuration update procedure that is triggered after the completion of the service request procedure.

If the SERVICE REQUEST message is for emergency services fallback, the AMF triggers the emergency services fallback procedure as specified in subclause 4.13.4.2 of 3GPP TS 23.502 [9].

If the UE having an emergency PDU session sent the SERVICE REQUEST message via:

- a) a CAG cell and none of the CAG-IDs of the CAG cell are included in the "Allowed CAG list" for the current PLMN in the UE's subscription; or
- b) a non-CAG cell in a PLMN for which the UE's subscription contains an "indication that the UE is only allowed to access 5GS via CAG cells";

the network shall accept the SERVICE REQUEST message and release all non-emergency PDU sessions locally. The emergency PDU session shall not be released.

If the AMF received the list of TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, any but not all TAIs in the received list of TAIs is forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or

b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or

c) both;

in the SERVICE ACCEPT message.

NOTE 9: Void.

If the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the SERVICE ACCEPT message and the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE into the list of "5GS forbidden tracking areas for roaming" and remove the TAI(s) from the stored TAI list if present.

If the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the SERVICE ACCEPT message and the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE into the list of "5GS forbidden tracking areas for regional provision of service" and remove the TAI(s) from the stored TAI list if present.

5.6.1.4.2 UE is using 5GS services with control plane CIoT 5GS optimization

For case a in subclause 5.6.1.1, upon receipt of the CONTROL PLANE SERVICE REQUEST message with Control plane service type indicating "mobile terminating request", after completion of the 5GMM common procedures (if initiated) according to subclause 5.6.1.3, the AMF shall send a SERVICE ACCEPT message.

For cases c, d and m in subclause 5.6.1.1, upon receipt of the CONTROL PLANE SERVICE REQUEST message with Control plane service type indicating "mobile originating request", after completion of the 5GMM common procedures (if initiated) according to subclause 5.6.1.3, the AMF shall send a SERVICE ACCEPT message, except for case d when the DDX field of the Release assistance indication IE or the DDX field of the CIoT small data container IE indicates "No further uplink and no further downlink data transmission subsequent to the uplink data transmission is expected".

For case a, c and d:

- a) if the CIoT small data container IE is included in the message, the AMF shall decipher the value part of the CIoT small data container IE and:
 - if the Data type field indicates "control plane user data", extract the PDU session ID and data content from the CIoT small data container IE, look up a PDU session routing context for the UE and the PDU session ID, and forward the content of the CIoT small data container IE to the SMF associated with the UE;
 - 2) if the Data type field indicates "SMS", forward the content of the CIoT small data container IE to the SMSF associated with the UE; or
 - 3) if the Data type field indicates "Location services message container", and if
 - i) length of additional information field in the CIoT small data container IE is zero, forward the value of Data type field and the content of the CIoT small data container IE to the to the location services application; or
 - ii) otherwise forward the value of Data type field and the content of the CIoT small data container IE to the LMF associated with the routing information that is included in the additional information field of the CIoT small data container IE; or
- NOTE 1: If the AMF determines there is no pending data or signalling for the UE, the AMF provides an indication of control plane CIoT 5GS Optimisation to the LMF as specified in 3GPP TS 29.518 [20B].
- b) otherwise, the AMF shall decipher the value part of NAS message container IE and:
 - if the Payload container IE is included in the CONTROL PLANE SERVICE REQUEST message and if the Payload container type IE is set to "CIoT user data container", the AMF shall look up a PDU session routing context for the UE and the PDU session ID, and forward the content of the Payload container IE to the SMF associated with the UE;
 - if the Payload container IE is included in the CONTROL PLANE SERVICE REQUEST message and if the Payload container type IE is set to "SMS", the AMF shall forward the content of the Payload container IE to the SMSF associated with the UE;

- 3) if the PDU session status IE is included in the CONTROL PLANE SERVICE REQUEST message or the AMF needs to perform a PDU session status synchronization, the AMF shall include a PDU session status IE in the SERVICE ACCEPT message to indicate which PDU sessions associated with the access type the SERVICE ACCEPT message is sent over are active in the AMF;
- 4) if the Uplink data status IE is included in the CONTROL PLANE SERVICE REQUEST message and the UE is:
 - i) not in NB-N1 mode; or
 - ii) in NB-N1 mode and the UE does not indicate a request to have user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources;

the AMF shall:

- i) indicate the SMF to re-establish the user-plane resources for the corresponding PDU sessions; and
- ii) include the PDU session reactivation result IE in the SERVICE ACCEPT message to indicate the userplane resources re-establishment result of the PDU sessions for which the UE requested to re-establish the user-plane resources;
- 5) if the Uplink data status IE is included in the CONTROL PLANE SERVICE REQUEST, the UE is in NB-N1 mode, and the UE indicates a request to have user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources, the AMF shall not indicate to the SMF to re-establish the user-plane resources for the corresponding PDU sessions; or
- 6) otherwise, if the Payload container IE is included in the message and if the Payload container type IE is set to "Location services message container", the AMF shall forward the Payload container type and the content of the Payload container IE to the LMF associated with the routing information included in the Additional information IE of the CONTROL PLANE SERVICE REQUEST message.
- NOTE 2: If the AMF determines there is no pending data or signalling for the UE, the AMF provides an indication of control plane CIoT 5GS Optimisation to the LMF as specified in 3GPP TS 29.518 [20B].

For case k) in subclause 5.6.1.1, if the Uplink data status IE is included in the CONTROL PLANE SERVICE REQUEST message and the UE is:

- a) not in NB-N1 mode; or
- b) in NB-N1 mode and the UE does not indicate a request to have user-plane resources established for a number of PDU sessions that exceeds the UE's maximum number of supported user-plane resources,

the AMF shall:

- a) indicate the SMF to re-establish the user-plane resources for the corresponding PDU sessions; and
- b) include the PDU session reactivation result IE in the SERVICE ACCEPT message to indicate the user-plane resources re-establishment result of the PDU sessions for which the UE requested to re-establish the user-plane resources.

If the Allowed PDU session status IE is included in the CONTROL PLANE SERVICE REQUEST message, the AMF shall:

- a) for a 5GSM message from each SMF that has indicated pending downlink signalling only, forward the received 5GSM message via 3GPP access to the UE after the SERVICE ACCEPT message is sent;
- b) for each SMF that has indicated pending downlink data only:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE; and
 - 2) notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if:

- i) for a UE not in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE; or
- ii) for a UE in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE and the resulting number of PDU sessions with established user-plane resources does not exceed the UE's maximum number of supported user-plane resources;
- c) for each SMF that have indicated pending downlink signalling and data:
 - notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access cannot be performed if the corresponding PDU session ID(s) are not indicated in the Allowed PDU session status IE;
 - 2) notify the SMF that reactivation of the user-plane resources for the corresponding PDU session(s) associated with non-3GPP access can be performed if:
 - i) for a UE not in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE; or
 - ii) for a UE in NB-N1 mode, the corresponding PDU session ID(s) are indicated in the Allowed PDU session status IE and the resulting number of PDU sessions with established user-plane resources does not exceed the UE's maximum number of supported user-plane resources; and
 - 3) discard the received 5GSM message for PDU session(s) associated with non-3GPP access; and
- d) include the PDU session reactivation result IE in the SERVICE ACCEPT message to indicate the successfully re-established user-plane resources for the corresponding PDU sessions, if any.

If the DDX field in the CIoT small data container IE or the DDX field of the Release assistance indication IE indicates:

- 1) "No further uplink and no further downlink data transmission subsequent to the uplink data transmission is expected" and if there is no downlink signalling or downlink data for the UE; or
- 2) "Only a single downlink data transmission and no further uplink data transmission subsequent to the uplink data transmission is expected" and upon subsequent delivery of the next received downlink data transmission to the UE and if there is no additional downlink signalling or downlink data for the UE,

the AMF initiates the release of the N1 NAS signalling connection (see 3GPP TS 23.502 [9]).

If the MUSIM UE does not include the Paging restriction IE in the CONTROL PLANE SERVICE REQUEST message, the AMF shall delete any stored paging restriction for the UE and stop restricting paging.

For case m in subclause 5.6.1.1 when the MUSIM UE sets the Request type to "NAS signalling connection release" in the CONTROL PLANE SERVICE REQUEST message, the AMF shall initiate the release of the N1 NAS signalling connection after the completion of the service request procedure.

For cases o and p in subclause 5.6.1.1 when the MUSIM UE sets the Request type to "NAS signalling connection release" or to "Rejection of paging" in the UE request type IE in the CONTROL PLANE SERVICE REQUEST message and if the UE requests restriction of paging by including the Paging restriction IE, the AMF:

- if accepts the paging restriction, shall include the 5GS additional request result IE in the SERVICE ACCEPT message and set the Paging restriction decision to "paging restriction is accepted". The AMF shall store the paging restriction of the UE and enforce these restrictions in the paging procedure as described in clause 5.6.2; or
- if rejects the paging restriction, shall include the 5GS additional request result IE in the SERVICE ACCEPT message and set the Paging restriction decision to "paging restriction is rejected", and shall discard the received paging restriction. The AMF shall delete any stored paging restriction for the UE and stop restricting paging; and

the AMF shall send the SERVICE ACCEPT message and initiate the release of the N1 NAS signalling connection as follows:

- for case o in subclause 5.6.1.1, after the completion of the service request procedure;
- for case p in subclause 5.6.1.1, after the completion of the generic UE configuration update procedure that is triggered after the completion of the service request procedure.

Upon successful completion of the procedure, the UE shall reset the service request attempt counter, stop the timer T3517 and enter the state 5GMM-REGISTERED.

If the PDU session status information element is included in the CONTROL PLANE SERVICE REQUEST message, then the AMF:

- a) shall perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE on the AMF side associated with the access type the CONTROL PLANE SERVICE REQUEST message is sent over, but are indicated by the UE as being inactive, and
- b) request the SMF to perform a local release of all those PDU sessions. If any of those PDU sessions is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions.

If the PDU session status information element is included in the SERVICE ACCEPT message, then the UE shall perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING on the UE side associated with the 3GPP access but are indicated by the AMF as being inactive. If a locally released PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions.

If the user-plane resources cannot be established for a PDU session, the AMF shall include the PDU session reactivation result IE in the SERVICE ACCEPT message indicating that user-plane resources for the corresponding PDU session cannot be re-established, and:

- a) if the user-plane resources cannot be established because the SMF indicated to the AMF that the UE is located out of the LADN service area (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #43 "LADN not available";
- b) if the user-plane resources cannot be established because the SMF indicated to the AMF that only prioritized services are allowed (see 3GPP TS 29.502 [20A]), the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #28 "restricted service area"; or
- c) if the user-plane resources cannot be established because:
 - 1) the SMF indicated to the AMF that the resource is not available in the UPF (see 3GPP TS 29.502 [20A]); or
 - 2) the UE is in NB-N1 mode and the result will lead to user-plane resources established for more than two PDU sessions (see 3GPP TS 23.502 [9])

the AMF shall include the PDU session reactivation result error cause IE with the 5GMM cause set to #92"insufficient user-plane resources for the PDU session":

NOTE 3: For a UE that is not in NB-N1 mode, it is up to UE implementation when to re-send a request for userplane re-establishment for the associated PDU session after receiving a PDU session reactivation result error cause IE with a 5GMM cause set to #92 "insufficient user-plane resources for the PDU session".

If the PDU session reactivation result IE is included in the SERVICE ACCEPT message indicating that the user-plane resources cannot be established for a PDU session that was requested by the UE in the Allowed PDU session status IE, the UE considers the corresponding PDU session to be associated with the non-3GPP access.

For case d) in subclause 5.6.1.1, the UE shall also treat the indication from the lower layers that the RRC connection has been released as successful completion of the procedure. The UE shall reset the service request attempt counter, stop the timer T3517 and enter the state 5GMM-REGISTERED.

Upon receipt of the CONTROL PLANE SERVICE REQUEST message with uplink data:

- if the DDX field of the Release assistance indication IE or the DDX field of the CIoT small data container IE is set to "No further uplink and no further downlink data transmission subsequent to the uplink data transmission is expected" in the message;
- if the AMF decides to forward the uplink data piggybacked in the CONTROL PLANE SERVICE REQUEST message; and
- if the AMF decides to activate the congestion control for transport of user data via the control plane,

then the AMF shall send SERVICE ACCEPT message with the T3448 value IE included.
If the AMF decides to deactivate the congestion control for transport of user data via the control plane, then the AMF shall delete the stored control plane data back-off time for the UE and the AMF shall not include timer T3448 value IE in the SERVICE ACCEPT message.

If the T3448 value IE is present in the received SERVICE ACCEPT message and the value indicates that this timer is neither zero nor deactivated, the UE shall:

- a) stop timer T3448 if it is running;
- b) consider the transport of user data via the control plane as successful; and
- c) start timer T3448 with the value provided in the T3448 value IE.

If the UE is using 5GS services with control plane CIoT 5GS optimization, the T3448 value IE is present in the SERVICE ACCEPT message and the value indicates that this timer is either zero or deactivated, the UE shall ignore the T3448 value IE and proceed as if the T3448 value IE was not present.

If the UE in 5GMM-IDLE mode initiated the service request procedure by sending a CONTROL PLANE SERVICE REQUEST message and the SERVICE ACCEPT message does not include the T3448 value IE and if timer T3448 is running, then the UE shall stop timer T3448.

For case h) in subclause 5.6.1.1,

- a) the UE shall treat the indication from the lower layers when the UE has changed to S1 mode as successful completion of the procedure and stop timer T3517;
- b) if a UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access (see subclause 4.9.2); and
- c) the AMF shall not check for CAG restrictions.

If the CONTROL PLANE SERVICE REQUEST message is for emergency services fallback, the AMF triggers the emergency services fallback procedure as specified in subclause 4.13.4.2 of 3GPP TS 23.502 [9].

5.6.1.5 Service request procedure not accepted by the network

If the service request cannot be accepted, the network shall return a SERVICE REJECT message to the UE including an appropriate 5GMM cause value.

If the SERVICE REJECT message with 5GMM cause #76 or #78 was received without integrity protection, then the UE shall discard the message.

If the AMF needs to initiate PDU session status synchronisation or a PDU session status IE was included in the SERVICE REQUEST message, the AMF shall include a PDU session status IE in the SERVICE REJECT message to indicate which PDU sessions associated with the access type the SERVICE REJECT message is sent over are active in the AMF. If the PDU session status IE is included in the SERVICE REJECT message and if the message is integrity protected, then:

- a) for single access PDU sessions, the UE shall perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING on the UE side associated with the access type the SERVICE REJECT message is sent over, but are indicated by the AMF as being in 5GSM state PDU SESSION INACTIVE. If a locally released PDU session is associated with one or more MBS sessions, the UE shall locally leave the associated MBS multicast sessions; and
- b) for MA PDU sessions, for all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE or PDU SESSION ACTIVE PENDING and have user plane resources established on the UE side associated with the access the SERVICE REJECT message is sent over, but are indicated by the AMF as no user plane resources established:
 - for MA PDU sessions having user plane resources established only on the access type the SERVICE REJECT message is sent over, the UE shall perform a local release of those MA PDU sessions. If a locally released MA PDU session is associated with one or more MBS multicast sessions, the UE shall locally leave the associated MBS multicast sessions; and

2) for MA PDU sessions having user plane resources established on both accesses, the UE shall perform a local release on the user plane resources on the access type the SERVICE REJECT message is sent over. If the user plane resources over 3GPP access are released and the MA PDU session is associated with one or more MBS multicast sessions, the UE shall locally leave the associated MBS multicast sessions.

If the service request for mobile originated services is rejected due to general NAS level mobility management congestion control, the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

In NB-N1 mode, if the service request for mobile originated services is rejected due to operator determined barring (see 3GPP TS 29.503 [20AB]), the network shall set the 5GMM cause value to #22 "congestion" and assign a value for back-off timer T3346.

If the service request from a UE supporting CAG is rejected due to CAG restrictions, the network shall set the 5GMM cause value to #76 "Not authorized for this CAG or authorized for CAG cells only" and should include the "CAG information list" in the CAG information list IE or the Extended CAG information list IE in the SERVICE REJECT message.

- NOTE 1: The network cannot be certain that "CAG information list" stored in the UE is updated as result of sending of the SERVICE REJECT message with the CAG information list IE or the Extended CAG information list IE, as the SERVICE REJECT message is not necessarily delivered to the UE (e.g., due to abnormal radio conditions).
- NOTE 2: The "CAG information list" can be provided by the AMF and include no entry if no "CAG information list" exists in the subscription.
- NOTE 2A: If the UE supports extended CAG information list, the CAG information list can be included either in the CAG information list IE or Extended CAG information list IE.

If the UE does not support extended CAG information list, the CAG information list shall not be included in the Extended CAG information list IE.

If the service request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall operate as described in bullet h) of subclause 5.6.1.8.

Upon receipt of the CONTROL PLANE SERVICE REQUEST message with uplink data:

- if the AMF decides to not forward the uplink data piggybacked in the CONTROL PLANE SERVICE REQUEST message; and
- if the AMF decides to activate the congestion control for transport of user data via the control plane,

then the AMF shall send a SERVICE REJECT message and set the 5GMM cause value to #22 "congestion" and assign a value for control plane data back-off timer T3448.

If the AMF determines that the UE is in a non-allowed area or is not in an allowed area as specified in subclause 5.3.5, then:

- a) if the service type IE in the SERVICE REQUEST message is set to "signalling" or "data", the AMF shall send a SERVICE REJECT message with the 5GMM cause value set to #28 "Restricted service area";
- b) otherwise, if the service type IE in the SERVICE REQUEST message is set to "mobile terminated services", "emergency services", "emergency services fallback", "high priority access" or "elevated signalling", the AMF shall continue the process as specified in subclause 5.6.1.4 unless for other reasons the service request cannot be accepted.

If the service request for mobile originated services is rejected due to service gap control as specified in subclause 5.3.17, i.e. the T3447 timer is running in AMF, the network shall set the 5GMM cause value to #22 "Congestion" and may include T3346 value IE in the SERVICE REJECT message set to the remaining time of the running T3447 timer.

Based on operator policy, if the service request procedure is rejected due to core network redirection for CIoT optimizations, the network shall set the 5GMM cause value to #31 "Redirection to EPC required".

NOTE 3: The network can take into account the UE's S1 mode capability, the EPS CIoT network behaviour supported by the UE or the EPS CIoT network behaviour supported by the EPC to determine the rejection with the 5GMM cause value #31 "Redirection to EPC required".

If the service request is via a satellite NG-RAN cell, and the network determines that the UE is in a location where the network is not allowed to operate, see 3GPP TS 23.502 [9], the network shall set the 5GMM cause value in the SERVICE REJECT message to #78 "PLMN not allowed to operate at the present UE location".

If the service request from a UE supporting MINT is rejected due to a disaster condition no longer being applicable in the current location of the UE, the network shall set the 5GMM cause value to #11 "PLMN not allowed" or #13 "Roaming not allowed in this tracking area" and may include a disaster return wait range in the Disaster return wait range IE in the SERVICE REJECT message.

On receipt of the SERVICE REJECT message, if the UE is in state 5GMM-SERVICE-REQUEST-INITIATED, the UE shall reset the service request attempt counter and stop timer T3517 if running.

If the AMF received multiple TAIs from the satellite NG-RAN as described in 3GPP TS 23.501 [8], and determines that, by UE subscription and operator's preferences, all of the received TAIs are forbidden for roaming or for regional provision of service, the AMF shall include the TAI(s) in:

- a) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE; or
- b) the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE; or
- c) both;

in the SERVICE REJECT message.

Regardless of the 5GMM cause value received in the SERVICE REJECT message via satellite NG-RAN,

- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE in the SERVICE REJECT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for roaming", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for roaming"; and
- if the UE receives the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE in the SERVICE REJECT message and if the TAI(s) included in the IE is not part of the list of "5GS forbidden tracking areas for regional provision of service", the UE shall store the TAI(s) included in the IE, if not already stored, into the list of "5GS forbidden tracking areas for regional provision of service".

Furthermore, the UE shall take the following actions depending on the 5GMM cause value received in the SERVICE REJECT message.

- #3 (Illegal UE);
- #6 (Illegal ME);

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE supports access to an SNPN using credentials from a credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure using the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall delete the list of equivalent PLMNs (if any) and shall enter the state 5GMM-DEREGISTERED.NO-SUPI. If the message has been successfully integrity checked by the NAS, then the UE shall:

- set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN if the UE maintains these counters;
- 2) set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN if the UE maintains these counters; and
- 3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value. The USIM shall be considered as invalid also for non-EPS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.7a in 3GPP TS 24.301 [15]. If the message has been successfully integrity checked by the NAS and the UE maintains a counter for "SIM/USIM considered invalid for non-GPRS services", then the UE shall set this counter to UE implementation-specific maximum value.

If the UE is registered for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

#7 (5GS services not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

In case of PLMN, the UE shall consider the USIM as invalid for 5GS services until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.1;

In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall consider the entry of the "list of subscriber data" with the SNPN identity of the current SNPN as invalid for 5GS services until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. In case of SNPN, if the UE is not registered for onboarding services in SNPN and the UE supports access to an SNPN using credentials holder, the UE shall consider the selected entry of the "list of subscriber data" as invalid for 3GPP access until the UE is switched off, the entry is updated or the timer T3245 expires as described in clause 5.3.19a.2. Additionally, if EAP based primary authentication and key agreement procedure using EAP-AKA' or 5G AKA based primary authentication and key agreement procedure using the USIM as invalid for the current SNPN until switching off, the UICC containing the USIM is removed or the timer T3245 expires as described in clause 5.3.19a.2.

If the UE is not registered for onboarding services in SNPN, the UE shall enter the state 5GMM-DEREGISTERED.NO-SUPI. If the message has been successfully integrity checked by the NAS, then the UE shall:

- set the counter for "SIM/USIM considered invalid for GPRS services" events and the counter for "USIM considered invalid for 5GS services over non-3GPP access" events to UE implementation-specific maximum value in case of PLMN if the UE maintains these counters; or
- set the counter for "the entry for the current SNPN considered invalid for 3GPP access" events and the counter for "the entry for the current SNPN considered invalid for non-3GPP access" events to UE implementation-specific maximum value in case of SNPN if the UE maintains these counters; and
- 3) delete the 5GMM parameters stored in non-volatile memory of the ME as specified in annex C.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and

eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

If the UE is registered for onboarding services in SNPN, the UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for onboarding services, enter state 5GMM-DEREGISTERED.PLMN-SEARCH, and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 4: The possibility to configure a UE so that the radio transceiver for a specific radio access technology is not active, although it is implemented in the UE, is outside the scope of the present document.
- #9 (UE identity cannot be derived by the network).

The UE shall set the 5GS update status to 5U2 NOT UPDATED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall enter the state 5GMM-DEREGISTERED.

If the service request was initiated for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

If the service request was initiated for any reason other than emergency services fallback or initiating an emergency PDU session, the UE shall perform a new initial registration procedure.

NOTE 5: User interaction is necessary in some cases when the UE cannot re-establish the PDU session(s) automatically.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

#10 (Implicitly de-registered).

The UE shall enter the state 5GMM-DEREGISTERED.NORMAL-SERVICE. The UE shall delete any mapped 5G NAS security context or partial native 5G NAS security context.

If the service request was initiated for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

If the rejected request was neither for initiating an emergency PDU session nor for emergency services fallback, the UE shall perform a new initial registration procedure.

NOTE 6: User interaction is necessary in some cases when the UE cannot re-establish the PDU session(s) automatically.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall handle the EMM state as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

#11 (PLMN not allowed).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. For 3GPP access, the UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE and perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been successfully integrity checked by the NAS and the UE mantains the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same PLMN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

If the UE receives the Disaster return wait range IE in the SERVICE REJECT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range IE in the ME.

#12 (Tracking area not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and enter the state 5GMM-DEREGISTERED.LIMITED-SERVICE. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for regional provision of service" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

#13 (Roaming not allowed in this tracking area).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2). For 3GPP access the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH, and for non-3GPP access the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

If:

 the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and remove the current TAI from the stored TAI list if present. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or

2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and remove the current TAI from the stored TAI list if present. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

For 3GPP access the UE shall perform a PLMN selection or SNPN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall perform network selection as defined in 3GPP TS 24.502 [18].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state and EPS update status as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

If the UE receives the Disaster return wait range IE in the SERVICE REJECT message and the UE supports MINT, the UE shall delete the disaster return wait range stored in the ME, if any, and store the disaster return wait range IE in the ME.

#15 (No suitable cells in tracking area).

The UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

If:

- the UE is not operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" and remove the current TAI from the stored TAI list if present. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for non-integrity protected NAS reject message; or
- 2) the UE is operating in SNPN access operation mode, the UE shall store the current TAI in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, and remove the current TAI from the stored TAI list if present. If the SERVICE REJECT message is not integrity protected, the UE shall memorize the current TAI was stored in the list of "5GS forbidden tracking areas for roaming" for the current SNPN and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription, for non-integrity protected NAS reject message.

If the UE initiated service request for emergency services fallback, the UE shall attempt to select an E-UTRA cell connected to EPC or 5GC according to the emergency services support indicator (see 3GPP TS 36.331 [25A]). If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

If the service request was not initiated for emergency services fallback, the UE shall search for a suitable cell in another tracking area according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall handle the EMM parameters EMM state and EPS update status as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

If received over non-3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.6.1.7.

#22 (Congestion).

If the T3346 value IE is present in the SERVICE REJECT message and the value indicates that this timer is neither zero nor deactivated, the UE shall proceed as described below, otherwise it shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.6.1.7.

If the rejected request was not for initiating an emergency PDU session, the UE shall abort the service request procedure and enter state 5GMM-REGISTERED and stop timer T3517 if still running.

The UE shall stop timer T3346 if it is running.

If the SERVICE REJECT message is integrity protected, the UE shall start timer T3346 with the value provided in the T3346 value IE.

If the SERVICE REJECT message is not integrity protected, the UE shall start timer T3346 with a random value from the default range specified in 3GPP TS 24.008 [12].

For all other cases the UE stays in the current serving cell and applies normal cell reselection process. The service request procedure is started, if still necessary, when timer T3346 expires or is stopped.

If the message was received via 3GPP access and the UE is operating in the single-registration mode, the UE shall handle the EMM parameters EMM state and EPS update status as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

If the service request procedure was initiated for an MO MMTEL voice call (i.e. access category 4), or for an MO MMTEL video call (i.e. access category 5) or for an MO IMS registration related signalling (i.e. access category 9), a notification that the service request was not accepted due to congestion shall be provided to the upper layers.

If the UE is using 5GS services with control plane CIoT 5GS optimization and if the T3448 value IE is present in the SERVICE REJECT message and the value indicates that this timer is neither zero nor deactivated, the UE shall:

- a) stop timer T3448 if it is running;
- b) consider the transport of user data via the control plane as unsuccessful; and
- c) start timer T3448:
 - 1) with the value provided in the T3448 value IE if the SERVICE REJECT message is integrity protected; or
 - 2) with a random value from the default range specified in 3GPP TS 24.301 [15] table 10.2.1 if the SERVICE REJECT message is not integrity protected.

If the UE is using 5GS services with control plane CIoT 5GS optimization, the T3448 value IE is present in the SERVICE REJECT message and the value indicates that this timer is either zero or deactivated, the UE shall ignore the T3448 value IE and:

- a) stop timer T3448 if it is running; and
- b) consider the transport of user data via the control plane as unsuccessful.

If the UE is using 5GS services with control plane CIoT 5GS optimization and if the T3448 value IE is not present in the SERVICE REJECT message, it shall be considered as an abnormal case and the behaviour of UE for this case is specified in subclause 5.6.1.7.

#27 (N1 mode not allowed).

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE. If the message has been successfully integrity checked by the NAS, the UE shall set:

- 1) the PLMN-specific N1 mode attempt counter for 3GPP access and the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN; or
- 2) the SNPN-specific attempt counter for 3GPP access for the current SNPN and the SNPN-specific attempt counter for non-3GPP access for the current SNPN in case of SNPN

to the UE implementation-specific maximum value.

The UE shall disable the N1 mode capability for the specific access type for which the message was received (see subclause 4.9).

If the message has been successfully integrity checked by the NAS, the UE shall disable the N1 mode capability also for the other access type (see subclause 4.9).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED and enter the state EMM-REGISTERED.

#28 (Restricted service area).

The UE shall enter the state 5GMM-REGISTERED.NON-ALLOWED-SERVICE, wait for the release of the N1 NAS signalling connection and perform the registration procedure for mobility and periodic registration update if the service type IE in the SERVICE REQUEST message was not set to "elevated signalling" and the SERVICE REJECT message is received over 3GPP access (see subclause 5.3.5 and 5.5.1.3).

If the service type IE in the SERVICE REQUEST message was set to "elevated signalling", the UE shall not reinitiate service request procedure until the UE enters an allowed area or leaves a non-allowed area, except for emergency services, high priority access or responding to paging or notification.

#31 (Redirection to EPC required).

5GMM cause #31 received by a UE that has not indicated support for CIoT optimizations or received by a UE over non-3GPP access is considered an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2). The UE shall reset the service request attempt counter and enter the state 5GMM-REGISTERED.LIMITED-SERVICE.

The UE shall enable the E-UTRA capability if it was disabled and disable the N1 mode capability for 3GPP access (see subclause 4.9.2).

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters, EMM state, and EPS update status as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

#36 (IAB-node operation not authorized).

This cause value is only applicable when received over 3GPP access by a UE operating as an IAB-node. This cause value received from a 5G access network other than 3GPP access or received by a UE not operating as an IAB-node is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI.

If:

- 1) the UE is not operating in SNPN access operation mode,
 - i) the UE shall delete the list of equivalent PLMNs and store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A and if the UE is configured to use timer T3245 then the UE shall start timer T3245 and proceed as described in clause 5.3.19a.1. The UE shall enter the state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS and the UE maintains the PLMNspecific attempt counter for 3GPP access for that PLMN, the UE shall set the PLMN-specific attempt counter for 3GPP access for that PLMN to the UE implementation-specific maximum value; and
 - ii) If the UE is operating in single-registration mode, the UE shall in addition handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value; or
- 2) the UE is operating in SNPN access operation mode,

- i) the UE shall store the SNPN identity in the "temporarily forbidden SNPNs" list for 3GPP access and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access for the current SNPN to the UE implementation-specific maximum value.
- #72 (Non-3GPP access to 5GCN not allowed).

If the UE initiated the service request procedure over non-3GPP access, the UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI for non-3GPP access. Additionally, the UE shall enter the state 5GMM-DEREGISTERED for non-3GPP access. If the message has been successfully integrity checked by the NAS, the UE shall set:

- 1) the PLMN-specific N1 mode attempt counter for non-3GPP access for that PLMN in case of PLMN; or
- 2) the SNPN-specific attempt counter for non-3GPP access for that SNPN in case of SNPN;

to the UE implementation-specific maximum value.

NOTE 7: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).

The UE shall disable the N1 mode capability for non-3GPP access (see subclause 4.9.3).

As an implementation option, if the UE is not currently registered over 3GPP access, the UE may enter the state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5].

If received over 3GPP access the cause shall be considered as an abnormal case and the behaviour of the UE for this case is specified in subclause 5.6.1.7.

#73 (Serving network not authorized).

This cause value received from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall delete the list of equivalent PLMNs, store the PLMN identity in the forbidden PLMN list as specified in subclause 5.3.13A. For 3GPP access the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH in order to perform a PLMN selection according to 3GPP TS 23.122 [5], and for non-3GPP access the UE shall enter state 5GMM-DEREGISTERED.LIMITED-SERVICE in order to perform network selection as defined in 3GPP TS 24.502 [18]. If the message has been successfully integrity checked by the NAS, the UE shall set the PLMN-specific attempt counter and the PLMN-specific attempt counter for non-3GPP access for that PLMN to the UE implementation-specific maximum value.

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED, enter the state EMM-DEREGISTERED and shall delete any 4G-GUTI, last visited registered TAI, TAI list and eKSI.

#74 (Temporarily not authorized for this SNPN).

5GMM cause #74 is only applicable when received from a cell belonging to an SNPN. 5GMM cause #74 received from a cell not belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall store the SNPN identity in the "temporarily forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform

an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection or an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access and the SNPN-specific attempt counter for non-3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 8: When 5GMM cause #74 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 9: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #75 (Permanently not authorized for this SNPN).

5GMM cause #75 is only applicable when received from a cell belonging to an SNPN with a globally-unique SNPN identity. 5GMM cause #75 received from a cell not belonging to an SNPN or a cell belonging to an SNPN with a non-globally-unique SNPN identity is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete any 5G-GUTI, last visited registered TAI, TAI list and ngKSI. The UE shall store the SNPN identity in the "permanently forbidden SNPNs" list for the specific access type for which the message was received and, if the UE supports access to an SNPN using credentials from a credentials holder, the selected entry of the "list of subscriber data" or the selected PLMN subscription. If the UE is not registered for onboarding services in SNPN, the UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform an SNPN selection according to 3GPP TS 23.122 [5]. If the UE is registered for onboarding services in SNPN, the UE shall enter state 5GMM-SEARCH and perform an SNPN selection for onboarding services according to 3GPP TS 23.122 [5]. If the message has been successfully integrity checked by the NAS, the UE shall set the SNPN-specific attempt counter for 3GPP access and the SNPN-specific attempt counter for non-3GPP access for the current SNPN to the UE implementation-specific maximum value.

If the message has been successfully integrity checked by the NAS and the UE also supports the registration procedure over the other access to the same SNPN, the UE shall in addition handle 5GMM parameters and 5GMM state for this access, as described for this 5GMM cause value.

- NOTE 10: When 5GMM cause #75 is received over 3GPP access, the term "other access" in "the UE also supports the registration procedure over the other access to the same SNPN" is used to express access to SNPN services via a PLMN.
- NOTE 11: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- #76 (Not authorized for this CAG or authorized for CAG cells only).

This cause value received via non-3GPP access or from a cell belonging to an SNPN is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3.ROAMING NOT ALLOWED, store the 5GS update status according to clause 5.1.3.2.2.

If 5GMM cause #76 is received from:

- 1) a CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the SERVICE REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received "CAG information list" when received in the HPLMN or EHPLMN;

- ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 12: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall delete the CAG-ID from the "allowed CAG list" for the current PLMN. In the case the "allowed CAG list" for the current PLMN only contains a range of CAG-IDs, how the UE deletes the CAG-ID(s) of the cell from the "allowed CAG list" for the current PLMN is up to UE implementation. In addition:

- i) if the entry in the "CAG information list" for the current PLMN does not include an "indication that the UE is only allowed to access 5GS via CAG cells" or if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list";
- ii) if the entry in the "CAG information list" for the current PLMN includes an "indication that the UE is only allowed to access 5GS via CAG cells" and the updated "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list"; or
- iii) if the "CAG information list" does not include an entry for the current PLMN, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C] with the updated "CAG information list".
- 2) a non-CAG cell, and if the UE receives a "CAG information list" in the CAG information list IE or the Extended CAG information list IE included in the SERVICE REJECT message, the UE shall:
 - i) replace the "CAG information list" stored in the UE with the received "CAG information list" when received in the HPLMN or EHPLMN;
 - ii) replace the serving VPLMN's entry of the "CAG information list" stored in the UE with the serving VPLMN's entry of the received CAG information list IE or the Extended CAG information list IE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN; or
- NOTE 13: When the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN, entries of a PLMN other than the serving VPLMN, if any, in the received CAG information list IE or the Extended CAG information list IE are ignored.
 - iii) remove the serving VPLMN's entry of the "CAG information list" stored in the UE when the UE receives the CAG information list IE or the Extended CAG information list IE in a serving PLMN other than the HPLMN or EHPLMN and the CAG information list IE or the Extended CAG information list IE does not contain the serving VPLMN's entry.

Otherwise, the UE shall store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN, if any. If the "CAG information list" stored in the UE does not include the current PLMN's entry, the UE shall add an entry for the current PLMN to the "CAG information list" and store an "indication that the UE is only allowed to access 5GS via CAG cells" in the entry of the "CAG information list" for the current PLMN. If the UE does not have a stored "CAG information list", the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" in the UE is only allowed to access 5GS via CAG cells" information list", the UE shall create a new "CAG information list" and add an entry with an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN.

In addition:

- i) if the "allowed CAG list" for the current PLMN includes one or more CAG-IDs, then the UE shall enter the state 5GMM-REGISTERED.LIMITED-SERVICE and shall search for a suitable cell according to 3GPP TS 38.304 [28] with the updated CAG information; or
- ii) if the "allowed CAG list" for the current PLMN does not include any CAG-ID, then the UE shall enter the state 5GMM-REGISTERED.PLMN-SEARCH and shall apply the PLMN selection process defined in 3GPP TS 23.122 [5] with the updated "CAG information list".

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall in addition set the EPS update status to EU3 ROAMING NOT ALLOWED, reset the service request attempt counter and enter the state EMM-REGISTERED.

#77 (Wireline access area not allowed).

5GMM cause #77 is only applicable when received from a wireline access network by the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device). 5GMM cause #77 received from a 5G access network other than a wireline access network and 5GMM cause #77 received by the W-AGF acting on behalf of the FN-BRG are considered as abnormal cases and the behaviour of the UE is specified in subclause 5.6.1.7.

When received over wireline access network, the 5G-RG and the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2), shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI, shall enter the state 5GMM-DEREGISTERED and shall act as specified in subclause 5.3.23.

- NOTE 14: The 5GMM sublayer states, the 5GMM parameters and the registration status are managed per access type independently, i.e. 3GPP access or non-3GPP access (see subclauses 4.7.2 and 5.1.3).
- #78 (PLMN not allowed to operate at the present UE location).

This cause value received from a non-satellite NG-RAN cell is considered as an abnormal case and the behaviour of the UE is specified in subclause 5.6.1.7.

The UE shall set the 5GS update status to 5U3 ROAMING NOT ALLOWED (and shall store it according to subclause 5.1.3.2.2) and shall delete 5G-GUTI, last visited registered TAI, TAI list and ngKSI. Additionally, the UE shall reset the registration attempt counter. The UE shall store the PLMN identity and, if it is known, the current geographical location in the list of "PLMNs not allowed to operate at the present UE location" and shall start a corresponding timer instance (see subclause 4.23.2). The UE shall enter state 5GMM-DEREGISTERED.PLMN-SEARCH and perform a PLMN selection according to 3GPP TS 23.122 [5].

If the message was received via 3GPP access and the UE is operating in single-registration mode, the UE shall handle the EMM parameters EMM state, EPS update status, 4G-GUTI, last visited registered TAI, TAI list and eKSI as specified in 3GPP TS 24.301 [15] for the case when the service request procedure is rejected with the EMM cause with the same value.

5.6.1.6 Service request procedure for initiating an emergency PDU session not accepted by the network

If the service request for initiating an emergency PDU session cannot be accepted by the network, the UE shall perform the procedures as described in subclause 5.6.1.5. Then if the UE is in the same selected PLMN where the last service request was attempted, the UE shall:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 1: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6].
- b) de-register locally, if not de-registered already, attempt initial registration for emergency services.

If the service request for initiating an emergency PDU session fails due to abnormal cases a) in subclause 5.6.1.7, the UE shall perform the procedures as described in subclause 5.6.1.7. Then if the UE is in the same selected PLMN where the last SERVICE REQUEST message was attempted, the UE shall:

a) inform the upper layers of the failure of the procedure; or

- NOTE 2: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6] and 3GPP TS 24.229 [14].
- b) de-register locally, if not de-registered already, attempt initial registration for emergency services.

5.6.1.6A Service request procedure for an emergency services fallback not accepted by the network

If the service request for initiating an emergency services fallback cannot be accepted by the network, the UE shall perform the procedures as described in subclause 5.6.1.5 and if the UE does not attempt to select an E-UTRA cell connected to EPC or 5GCN as described in subclause 5.6.1.5 and is camped on NR or E-UTRA cell connected to 5GCN in the same PLMN where the last service request was attempted, the UE shall inform the upper layers of the failure of the procedure.

NOTE 1: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6].

If the service request for initiating an emergency services fallback fails due to abnormal cases a) in subclause 5.6.1.7, the UE shall inform the upper layers of the failure of the emergency services fallback.

NOTE 2: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6].

If the service request procedure for initiating an emergency services fallback fails due to abnormal cases other than a) in subclause 5.6.1.7, the UE may abort the service request procedure, if not aborted already and attempt to select an E-UTRA cell connected to EPC or 5GCN according to the domain priority and selection rules specified in 3GPP TS 23.167 [6]. If the UE finds a suitable E-UTRA cell, it then proceeds with the appropriate EMM or 5GMM procedures. If the UE operating in single-registration mode has changed to S1 mode, it shall disable the N1 mode capability for 3GPP access.

5.6.1.7 Abnormal cases in the UE

The following abnormal cases can be identified:

a) T3517 expired.

The UE shall enter the state 5GMM-REGISTERED.

If the UE triggered the service request procedure in 5GMM-IDLE mode sending a:

- 1) SERVICE REQUEST message and the service type of the SERVICE REQUEST message was not set to "emergency services fallback"; or
- 2) CONTROL PLANE SERVICE REQUEST message and the control plane service type of the CONTROL PLANE SERVICE REQUEST message was not set to "emergency services fallback";

then the 5GMM sublayer shall increment the service request attempt counter, abort the procedure and release locally any resources allocated for the service request procedure. The service request attempt counter shall not be incremented, if:

- 1) the service request procedure is initiated to establish an emergency PDU session;
- 2) the UE has an emergency PDU session established;
- 3) the UE is a UE configured for high priority access in selected PLMN;
- 4) the service request procedure is initiated in response to paging or notification from the network; or
- 5) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]).

If the service request attempt counter is greater than or equal to 5, the UE shall start timer T3525. Additionally, if the service request procedure was initiated for an MO MMTEL voice call or for an MO MMTEL video call or

for an MO IMS registration related signalling, a notification that the service request was not initiated due to the UE having started timer T3525 shall be provided to the upper layers.

NOTE 1: This can result in the upper layers requesting implementation specific mechanisms, e.g. the MMTEL voice call being attempted to another IP-CAN, or establishment of a CS voice call (if supported and not already attempted in the CS domain).

The UE shall not attempt service request until expiry of timer T3525 unless:

- 1) the service request procedure is initiated in response to paging or notification from the network;
- 2) the UE is a UE configured for high priority access in selected PLMN;
- 3) the service request procedure is initiated to establish an emergency PDU session;
- 4) the UE has an emergency PDU session established;
- 5) the service request procedure is initiated for emergency services fallback;
- 6) the UE is registered in a new PLMN; or
- NOTE 2: According to Table 10.2.1, when "UE camped on a new PLMN other than the PLMN on which timer started", timer T3525 is stopped, hence this check may be skipped.
 - 7) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]).
- NOTE 3: The NAS signalling connection can also be released if the UE deems that the network has failed the authentication check as specified in subclause 5.4.1.3.7.

If the UE triggered the service request procedure in 5GMM-CONNECTED mode sending a:

- SERVICE REQUEST message and the service type of the SERVICE REQUEST message was not set to "emergency services fallback"; or
- 2) CONTROL PLANE SERVICE REQUEST message and the control plane service type of the CONTROL PLANE SERVICE REQUEST message was not set to "emergency services fallback",

the 5GMM sublayer shall abort the procedure, and stay in 5GMM-CONNECTED mode.

If the service type of the SERVICE REQUEST message was set to "emergency services fallback" or the control plane service type of the CONTROL PLANE SERVICE REQUEST message was set to "emergency services fallback" and:

- 1) the service request procedure was triggered in 5GMM-IDLE mode, the 5GMM sublayer shall abort the procedure, release locally any resources allocated for the service request procedure; or
- 2) the service request procedure was triggered in 5GMM-CONNECTED mode, the 5GMM sublayer shall abort the procedure, stay in 5GMM-CONNECTED mode.
- b) The lower layers indicate that the access attempt is barred.

The UE shall not start the service request procedure. The UE stays in the current serving cell and applies the normal cell reselection process. Receipt of the access barred indication shall not trigger the selection of a different core network type (EPC or 5GCN).

The service request procedure is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

ba) The lower layers indicate that:

- 1) access barring is applicable for all access categories except categories 0 and 2 and the access category with which the access attempt was associated is other than 0 and 2; or
- 2) access barring is applicable for all access categories except category 0 and the access category with which the access attempt was associated is other than 0.

If the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST has not been sent, the UE shall proceed as specified for case b.

If the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST has been sent:

- 1) the UE shall abort the service request procedure and stop timer T3517. The UE stays in the current serving cell and applies the normal cell reselection process; and
- 2) the service request procedure is started, if still needed, when the lower layers indicate that the barring is alleviated for the access category with which the access attempt was associated.

For additional UE requirements for both cases see subclause 4.5.5.

c) Timer T3346 is running.

The UE shall not start the service request procedure unless:

- 1) the UE receives a paging;
- 2) the UE receives a NOTIFICATION message over non-3GPP access when the UE is in 5GMM-CONNECTED mode over non-3GPP access and in 5GMM-IDLE mode over 3GPP access;
- 3) the UE receives a NOTIFICATION message over 3GPP access when the UE is in 5GMM-CONNECTED mode over 3GPP access and in 5GMM-IDLE mode over non-3GPP access;
- 4) the UE is a UE configured for high priority access in selected PLMN;
- 5) the UE has an emergency PDU session established or is establishing an emergency PDU session;
- 6) the service request procedure is initiated for emergency services fallback;
- 7) the service request procedure is initiated for elevated signalling;
- 8) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and:
 - the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]); and
 - timer T3346 was not started when N1 NAS signalling connection was established with RRC establishment cause set to "mo-ExceptionData"; or
- 9) the MUSIM UE is in 5GMM-CONNECTED mode and requests the network to release the NAS signalling connection (see case 0 in subclause 5.6.1.1).

If the UE is in 5GMM-IDLE mode, the UE stays in the current serving cell and applies normal cell reselection process. The service request procedure is started, if still necessary, when timer T3346 expires or is stopped.

If the service request procedure was triggered for an MO MMTEL voice call (i.e. access category 4), or for an MO MMTEL video call (i.e. access category 5) or for an MO IMS registration related signalling (i.e. access category 9), a notification that the service request procedure was not initiated due to congestion shall be provided to the upper layers.

If the UE receives a paging with access type set to "Non-3GPP access" and the non-3GPP access is available and UE is in 5GMM-REGISTERED.NORMAL SERVICE over non-3GPP access, the UE shall stop timer T3346 and send the SERVICE REQUEST message over non-3GPP access.

d) Registration procedure for mobility and periodic registration update is triggered.

The UE shall abort the service request procedure, stop timer T3517, if running and perform the registration procedure for mobility and periodic registration update. The Follow-on request indicator in the REGISTRATION REQUEST message shall be handled as specified in clause 5.5.1.3.2.

e) Switch off.

If the UE is in state 5GMM-SERVICE-REQUEST-INITIATED at switch off, the de-registration procedure shall be performed.

f) De-registration procedure collision.

If the UE receives a DEREGISTRATION REQUEST message from the network in state 5GMM-SERVICE-REQUEST-INITIATED, the UE shall progress the DEREGISTRATION REQUEST message and the service request procedure shall be aborted.

- NOTE 4: The above collision case is valid if the DEREGISTRATION REQUEST message indicates the access type over which the service request procedure is attempted otherwise both the procedures are progressed.
- g) Transmission failure of SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message indication with change in the current TAI.

If the current TAI is not in the TAI list, UE shall abort the service request procedure to perform the registration procedure for mobility and periodic registration update as specified in subclause 5.5.1.3.2.

If the current TAI is part of the TAI list, the UE shall restart the service request procedure unless the service request procedure is initiated for case o) or p) in subclause 5.6.1.1. For case o) and p) in subclause 5.6.1.1 the UE shall abort the service request procedure, enters state 5GMM-REGISTERED, locally release the N1 NAS signalling connection, stop timer T3517 and locally release any resources allocated for the service request procedure.

h) Transmission failure of SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST message indication without change in the current TAI.

The UE shall restart the service request procedure unless the service request procedure is initiated for case o) or p) in subclause 5.6.1.1. For case o) and p) in subclause 5.6.1.1 the UE shall abort the service request procedure, enters state 5GMM-REGISTERED, locally release the N1 NAS signalling connection, stop timer T3517 and locally release any resources allocated for the service request procedure.

i) SERVICE REJECT message received with other 5GMM cause values than those treated in subclause 5.6.1.5, and cases of 5GMM cause values #11, #15, #22, #31, #72, #73, #74, #75, #76, #77 and #78 that are considered as abnormal cases according to subclause 5.6.1.5.

The UE shall enter state 5GMM-REGISTERED.

The UE shall abort the service request procedure, stop timer T3517 and locally release any resources allocated for the service request procedure.

j) The UE in 5GMM-CONNECTED mode with RRC inactive indication over the 3GPP access, and in 5GMM-CONNECTED mode over non-3GPP access, receives a NOTIFICATION message over the non-3GPP access with access type indicating 3GPP access.

The UE shall transition from 5GMM-CONNECTED mode with RRC inactive indication to 5GMM-IDLE mode over 3GPP access and initiate the service request procedure over the 3GPP access.

k) Timer T3447 is running

The UE shall not start any service request procedure unless:

- 1) the UE in 5GMM-IDLE receives a paging request;
- 2) the UE is a UE configured for high priority access;
- 3) the UE has a PDU session for emergency services established or is establishing a PDU session for emergency services;
- 4) the service request procedure is initiated for emergency services fallback;
- 5) the UE in 5GMM-CONNECTED mode receives mobile terminated signalling or downlink data over the user-plane;
- 6) the service request procedure is initiated for elevated signalling; or

7) the MUSIM UE requests the network to release the NAS signalling connection (see case 0 in subclause 5.6.1.1).

The UE stays in the current serving cell and applies the normal cell reselection process. The service request procedure is started, if still necessary, when timer T3447 expires or timer T3447 is stopped.

 Lower layer failure, release of the N1 signalling connection received from lower layers or the lower layers indicate that the RRC connection has been suspended before the service request procedure is completed or SERVICE REJECT message is received.

The UE shall abort the service request procedure, stop timer T3517, locally release any resources allocated for the service request procedure and enters state 5GMM-REGISTERED. For case m) in subclause 5.6.1.1 the UE may retry the service request procedure a certain number of times (maximum re-attempts 5).

m) Timer T3448 is running

The UE in 5GMM-IDLE mode shall not initiate the service request procedure for transport of user data via the control plane unless:

- 1) the UE is a UE configured for high priority access in selected PLMN;
- 2) the UE which is only using 5GS services with control plane CIoT 5GS optimization received a paging request;
- 3) the UE in NB-N1 mode is requested by the upper layer to transmit user data related to an exceptional event and the UE is allowed to use exception data reporting (see the ExceptionDataReportingAllowed leaf of the NAS configuration MO in 3GPP TS 24.368 [17] or the USIM file EF_{NASCONFIG} in 3GPP TS 31.102 [22]); or
- 4) the UE is initiating the service request procedure to request emergency services or emergency services fallback.

The UE stays in the current serving cell and applies the normal cell reselection process. The service request procedure is started, if still necessary, when timer T3448 expires.

5.6.1.8 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Lower layer failure.

If a lower layer failure occurs before a SERVICE REJECT message has been sent to the UE or the service request procedure has been completed by the AMF, the AMF enters/stays in 5GMM-IDLE.

b) Protocol error.

If the SERVICE REQUEST message or the CONTROL PLANE SERVICE REQUEST message is received with a protocol error, the AMF shall return a SERVICE REJECT message with one of the following 5GMM cause values:

- #96 invalid mandatory information;
- #99 information element non-existent or not implemented;
- #100 conditional IE error; or
- #111 protocol error, unspecified.

The AMF stays in the current 5GMM mode.

- c) More than one SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message received before the procedure has been completed (i.e., before SERVICE REJECT message has been sent or service request procedure has been completed).
 - If one or more of the information elements in the SERVICE REQUEST message or CONTROL PLANE SERVICE REQUEST message differs from the ones received within the previous SERVICE REQUEST

message or CONTROL PLANE SERVICE REQUEST message, the previously initiated service request procedure shall be aborted, and the new service request procedure shall be progressed;

- If the information elements do not differ, then the AMF shall continue with the previous service request procedure and shall not treat any further this SERVICE REQUEST message or this CONTROL PLANE SERVICE REQUEST message.
- d) REGISTRATION REQUEST message received with "initial registration" or "emergency registration" in the 5GS registration type IE before a SERVICE REJECT message has been sent or the service request procedure has been completed.

If a REGISTRATION REQUEST message with "initial registration" or "emergency registration" in the 5GS registration type IE is received and the service request procedure has not been completed or a SERVICE REJECT message has not been sent, the AMF may initiate the 5GMM common procedures, e.g. the primary authentication and key agreement procedure. The AMF may e.g. after a successful primary authentication and key agreement procedure execution, abort the service request procedure, delete the 5GMM context, indicate towards the SMF that the 5GMM context has been deleted and progress the new REGISTRATION REQUEST message.

e) REGISTRATION REQUEST message received with "mobility registration updating" or "periodic registration updating" in the 5GS registration type IE received before the service request procedure has been completed or a SERVICE REJECT message has been sent.

If a REGISTRATION REQUEST message with "mobility registration updating" or "periodic registration updating" in the 5GS registration type IE is received and the service request procedure has not been completed or a SERVICE REJECT message has not been sent, the AMF may initiate the 5GMM common procedures, e.g. the primary authentication and key agreement procedure. The AMF may e.g. after a successful primary authentication and key agreement procedure execution, abort the service request procedure and progress the new REGISTRATION REQUEST message.

- f) If a CONTROL PLANE SERVICE REQUEST message with Control plane service type indicating "mobile originating request" is received after the AMF initiated a paging procedure, the AMF shall treat this CONTROL PLANE SERVICE REQUEST as a paging response and handle the message according to subclauses 5.6.1.4 and 5.6.1.5.
- g) CONTROL PLANE SERVICE REQUEST message received with the Data type field indicates "control plane user data" in the CIoT small data container IE or received with Payload container type IE set to "CIoT user data container" and:
 - 1) the AMF does not have a PDU session routing context for the PDU session ID and the UE; or
 - 2) the AMF unsuccessfully attempted to forward the user data container and the PDU session ID,

then the AMF may send back to the UE the CIoT user data container or control plane user data which was not forwarded as specified in subclause 5.4.5.3.1 case 11) or case 12).

- h) Based on operator policy, if the service request from a UE not supporting CAG is rejected due to CAG restrictions, the network shall reject the service request with a 5GMM cause value other than the 5GMM cause #76 (Not authorized for this CAG or authorized for CAG cells only).
- NOTE: 5GMM cause #7 (5GS services not allowed), 5GMM cause #11 (PLMN not allowed), 5GMM cause #27 (N1 mode not allowed), 5GMM cause #73 (Serving network not authorized) can be used depending on the subscription of the UE and whether the UE roams or not.

5.6.2 Paging procedure

5.6.2.1 General

The paging procedure is performed only in 3GPP access and used by the network to request the establishment of a NAS signalling connection to the UE. The paging procedure is also used by the network to request the UE to re-establish the user-plane resources of PDU sessions for downlink user data transport. Another purpose of the paging procedure is to request the UE to re-establish the user-plane resources of PDU session(s) associated with non-3GPP access over 3GPP access.

Additionally, the network can use the paging procedure to initiate the mobile terminating SMS.

For the UE using eDRX, the network initiates the paging procedure when NAS signalling messages or user data is pending to be sent to the UE within the paging time window and requests the lower layers to include the eDRX cycle length and paging time window length in the paging message. If NAS signalling messages or user data is pending to be sent to the UE outside the paging time window and the eDRX value that the network provides to the UE in the Negotiated extended DRX parameters IE during the last registration procedure indicates:

- a) the eDRX cycle length duration of the E-UTRA cell connected to 5GCN, is higher than 5.12 seconds; or
- b) the eDRX cycle length duration of the NR cell connected to 5GCN, is higher than 10.24 seconds,

the network initiates the paging procedure at T time ahead of the beginning of the next paging time window.

NOTE: T time is a short time period based on implementation. The operator can take possible imperfections in the synchronization between the 5GCN and the UE into account when choosing T time.

If the network detects that the pending user data to be sent to the UE is related to the voice service as specified in 3GPP TS 23.502 [9] and the network decides to initiate the paging procedure based on the stored paging restriction information, if any, the AMF should request the lower layer to include the Voice Service Indication in the paging message when the UE and the network support the paging indication for voice services.

5.6.2.2 Paging for 5GS services

5.6.2.2.1 General

The network shall initiate the paging procedure for 5GS services when NAS signalling messages or user data is pending to be sent to the UE in 5GMM-IDLE mode over 3GPP access (see example in figure 5.6.2.2.1.1) and there is no paging restriction applied in the network for that paging.



Figure 5.6.2.2.1.1: Paging procedure

To initiate the procedure the 5GMM entity in the AMF requests the lower layer to start paging and shall start timer T3513.

If downlink signalling or user data is pending to be sent over non-3GPP access, the 5GMM entity in the AMF shall indicate to the lower layer that the paging is associated to non-3GPP access.

The network shall not page the UE to re-establish user-plane resources of PDU session(s) associated with non-3GPP access over 3GPP access if all the PDU sessions of the UE that are established over the 3GPP access are associated with control plane only indication.

If the network has downlink user data pending for a UE, the AMF has stored paging restriction of the UE and the Paging restriction type in the stored paging restriction is set to:

- a) "All paging is restricted", the network should not page the UE;
- b) "All paging is restricted except for voice service", the network should page the UE only when:
 - 1) the pending downlink user data for the UE is considered as voice service related by the network;

- c) "All paging is restricted except for specified PDU session(s)", the network should page the UE only when:
 - 1) for PDU session(s) that paging is not restricted based on the stored paging restriction, the network has downlink user data pending; or
- d) "All paging is restricted except for voice service and specified PDU session(s)", the network should page the UE only when:
 - 1) the pending downlink user data for the UE is considered as voice service related by the network; or
 - 2) for PDU session(s) that paging is not restricted based on the stored paging restriction, the network has downlink user data pending.

If the network has downlink signalling pending for a UE and the AMF has stored paging restriction of the UE and the Paging restriction type in the stored paging restriction is set to:

- a) "All paging is restricted", the network should not page the UE;
- b) "All paging is restricted except for voice service", the network should page the UE only when:
 - 1) the pending downlink signalling for the UE is 5GMM signalling or 5GSM signalling of the PDU session of voice service;
- c) "All paging is restricted except for specified PDU session(s)", the network should page the UE only when:
 - 1) the pending downlink signalling for the UE is 5GMM signalling; or
 - 2) for PDU session(s) that paging is not restricted based on the stored paging restriction, the network has downlink 5GSM signalling pending; or
- d) "All paging is restricted except for voice service and specified PDU session(s)", the network should page the UE only when:
 - 1) the pending downlink signalling for the UE is 5GMM signalling or 5GSM signalling of the PDU session of voice service; or
 - 2) for PDU session(s) that paging is not restricted based on the stored paging restriction, the network has downlink 5GSM signalling pending.
- NOTE 1: If the network pages the UE due to downlink signalling pending, the network initiates the release of the N1 NAS signalling connection after network-requested procedure is completed.

The 5GMM entity in the AMF may provide the lower layer with the "allowed CAG list" and an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN, if available, and with the "allowed CAG list" and an "indication that the UE is only allowed to access 5GS via CAG cells" per equivalent PLMN, if available. If there is an active emergency PDU session, the 5GMM entity in the AMF shall not provide the lower layer with the "allowed CAG list" and an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN, even if available, or with the "allowed CAG list" and an "indication that the UE is only allowed to access 5GS via CAG cells" for the current PLMN, even if available, or with the "allowed CAG list" and an "indication that the UE is only allowed to access 5GS via CAG cells" per equivalent PLMN, even if available.

Upon reception of a paging indication, the UE shall stop the timer T3346, if running, and:

- a) if control plane CIoT 5GS optimization is not used by the UE, the UE shall:
 - initiate a service request procedure over 3GPP access to respond to the paging as specified in subclauses 5.6.1.2.1 if the UE is in 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) state and the UE is in the 5GMM-IDLE mode without suspend indication;
 - 2) initiate a service request procedure over non-3GPP access to respond to the paging as specified in subclauses 5.6.1;
 - 3) initiate a registration procedure for mobility and periodic registration update over 3GPP access to respond to the paging as specified in subclauses 5.5.1.3.2; or
 - 4) proceed as specified in subclause 5.3.1.5 if the UE is in the 5GMM-IDLE mode with suspend indication; or

- b) if control plane CIoT 5GS optimization is used by the UE, the UE shall:
 - 1) initiate a service request procedure as specified in subclause 5.6.1.2.2 if the UE is in the 5GMM-IDLE mode without suspend indication;
 - 2) initiate a registration procedure for mobility and periodic registration update over 3GPP access as specified in subclauses 5.5.1.3.2; or
 - 3) proceed as specified in subclause 5.3.1.5 if the UE is in the 5GMM-IDLE mode with suspend indication.
- NOTE 2: If the UE is in the 5GMM-IDLE mode without suspend indication and has an uplink user data to be sent to the network using control plane CIoT 5GS optimization when receiving the paging indication, the UE can piggyback the uplink user data during the service request procedure initiated to respond to the paging, as specified in subclause 5.6.1.2.2.

The MUSIM UE based on implementation may use the paging cause indicated by lower layers (see 3GPP TS 38.331 [30]), if any, to accept the paging, reject the paging or ignore the paging indication.

Upon reception of a paging indication, if the network supports the rejection of paging request and if a MUSIM UE decides not to accept the paging, the UE may initiate a service request procedure to reject the paging as specified in clause 5.6.1.1.

NOTE 3: As an implementation option, MUSIM UE is allowed to not respond to paging based on the information available in the paging message, e.g. voice service indication.

If TMGI is used as paging identity and the TMGI matches with multicast MBS session which the has UE joined, the UE shall respond to the paging. Otherwise, the UE shall not respond to the paging.

The network shall stop timer T3513 for the paging procedure when an integrity-protected response is received from the UE and successfully integrity checked by the network or when the 5GMM entity in the AMF receives an indication from the lower layer that it has received the NGAP UE context resume request message as specified in 3GPP TS 38.413 [31]. If the response received is not integrity protected, or the integrity check is unsuccessful, timer T3513 for the paging procedure shall be kept running unless:

- a) the UE is registered for emergency services;
- b) the UE has an emergency PDU session; or
- c) the response received is a REGISTRATION REQUEST message for mobility and periodic registration update and the security mode control procedure or authentication procedure performed during mobility and periodic registration update has completed successfully.

Upon expiry of timer T3513, the network may reinitiate paging.

If the network, while waiting for a response to the paging sent without paging priority, receives downlink signalling or downlink data associated with priority user-plane resources for PDU sessions, the network shall stop timer T3513, and then initiate the paging procedure with paging priority.

5.6.2.2.2 Abnormal cases on the network side

The following abnormal case can be identified:

a) Void.

5.6.2.2.3 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Paging message received with access type set to non-3GPP access while the UE is in 5GMM-CONNECTED mode over non-3GPP access.

The UE shall not respond to paging message.

b) Paging message received with access type set to 3GPP access when UE-initiated 5GMM specific procedure or service request procedure is ongoing.

The UE shall proceed with 5GMM specific procedure or service request procedure. If for these procedures lower layers indicate that the access attempt is barred, then the UE shall handle the pending paging message as specified in subclause 5.6.2.2.1. Otherwise, the UE shall ignore the paging once lower layers confirm the establishment of the signalling connection.

5.6.3 Notification procedure

5.6.3.1 General

The notification procedure is used by the network:

- a) to request the UE, by sending the NOTIFICATION message over 3GPP access, to re-establish the user-plane resources of PDU session(s) associated with non-3GPP access over 3GPP access or to deliver 5GSM downlink signalling messages associated with non-3GPP access over 3GPP access when the UE is in 5GMM-IDLE mode over non-3GPP access and in 5GMM-CONNECTED mode over 3GPP access; or
- b) to request the UE, by sending the NOTIFICATION message over non-3GPP access, to re-establish user-plane resources of the PDU session(s) or to deliver downlink signalling associated with 3GPP access over 3GPP access when the UE is in 5GMM-CONNECTED mode over non-3GPP access and:
 - 1) in 5GMM-IDLE mode over 3GPP access when the UE is not in MICO mode; or
 - 2) in 5GMM-IDLE mode with suspend indication over 3GPP access when the UE is not in MICO mode.

The network shall not use the NOTIFICATION message to re-establish user-plane resources of PDU session(s) associated with non-3GPP access over 3GPP access if all the PDU sessions of the UE that are established over the 3GPP access are associated with control plane only indication.

The network shall not use the NOTIFICATION message over non-3GPP access, if:

- a) the MUSIM UE supports the paging indication for voice services;
- b) the network has indicated "paging indication for voice services supported" to the UE; and
- c) the AMF detects that the downlink data is related to voice service (see 3GPP TS 23.501 [8]).

5.6.3.2 Notification procedure initiation

The network shall initiate the notification procedure by sending the NOTIFICATION message to the UE and start timer T3565 (see example in figure 5.6.3.2.1).

For case a) in subclause 5.6.3.1, the NOTIFICATION message is sent from the network to the UE via 3GPP access with access type indicating non-3GPP access.

For case b) in subclause 5.6.3.1, the NOTIFICATION message is sent from the network to the UE via non-3GPP access with access type indicating 3GPP access when the UE is not in MICO mode.

UE			AMF
-	NOTIFICATION		Start T3565
	SERVICE REQUEST or CONTROL PLANE SERVICE REQUEST or REGISTRATION REQUEST	►	Stop T3565
	OR		
◄-	NOTIFICATION		Start T3565
	NOTIFICATION RESPONSE	►	Stop T3565

Figure 5.6.3.2.1: Notification procedure

Upon reception of a NOTIFICATION message, the UE shall stop the timer T3346, if running.

For case a) in subclause 5.6.3.1, upon reception of NOTIFICATION message, the UE shall initiate a service request procedure over 3GPP access as specified in subclauses 5.6.1.

NOTE: For a UE in NB-NI mode, if there is DL user data pending for a PDU session associated with non-3GPP access then the AMF notifies the SMF that reactivation of user plane resources cannot be performed if the number of PDU sessions that currently has user-plane resources established equals to the UE's maximum number of supported user-plane resources.

For case b) in subclause 5.6.3.1, upon reception of NOTIFICATION message:

- a) if control plane CIoT 5GS optimization is not used by the UE, the UE shall:
 - initiate a service request procedure over 3GPP access as specified in subclause 5.6.1.2.1, if the UE is in 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) state over 3GPP access or 5GMM-REGISTERED.NON-ALLOWED-SERVICE state (see subclause 5.3.5.2), and the UE is in the 5GMM-IDLE mode without suspend indication;
 - initiate a registration procedure for mobility and periodic registration update over 3GPP access as specified in subclause 5.5.1.3.2, if the UE is in 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE state over 3GPP access; or
 - 3) proceed as specified in subclause 5.3.1.5 if the UE is in the 5GMM-IDLE mode with suspend indication;
- b) if control plane CIoT 5GS optimization is used by the UE, the UE shall:
 - initiate a service request procedure over 3GPP access as specified in subclause 5.6.1.2.2, if the UE is in 5GMM-REGISTERED.NORMAL-SERVICE or 5GMM-REGISTERED.NON-ALLOWED-SERVICE (as described in subclause 5.3.5.2) state and the UE is in the 5GMM-IDLE mode without suspend indication;
 - initiate a registration procedure for mobility and periodic registration update over 3GPP access as specified in subclause 5.5.1.3.2, if the UE is in 5GMM-REGISTERED.ATTEMPTING-REGISTRATION-UPDATE state; or
 - 3) proceed as specified in subclause 5.3.1.5 if the UE is in the 5GMM-IDLE mode with suspend indication; or

c) if:

 the UE is in 5GMM-REGISTERED.NO-CELL-AVAILABLE state, 5GMM-REGISTERED.PLMN-SEARCH state, 5GMM-REGISTERED.LIMITED-SERVICE state or 5GMM-REGISTERED.UPDATE-NEEDED state over 3GPP access; or 2) the MUSIM UE is not able to respond the NOTIFICATION message as specified in case a) and b) above, e.g., due to UE implementation constraints;

the UE shall respond with NOTIFICATION RESPONSE message over non-3GPP access indicating inability of the UE to initiate a service request procedure or a registration procedure over 3GPP access and may include the PDU session status information element to indicate:

- 1) the single access PDU session(s) not in 5GSM state PDU SESSION INACTIVE in the UE associated with the 3GPP access type; and
- 2) the MA PDU session(s) not in 5GSM state PDU SESSION INACTIVE in the UE and having user plane resources established associated with the 3GPP access type.

Upon reception of NOTIFICATION message:

For case b) in subclause 5.6.3.1, if the UE is in 5GMM-REGISTERED.NO-CELL-AVAILABLE state or 5GMM-REGISTERED.PLMN-SEARCH state and a local release was performed in the UE for the single access PDU sessions associated with the 3GPP access or for user plane resources on the 3GPP access of MA PDU sessions;

then the UE shall respond with NOTIFICATION RESPONSE message over non-3GPP access indicating with the PDU session status information element that:

- the local release of its single access PDU sessions associated with the 3GPP access was performed; and
- the local release of its 3GPP access user plane resources of MA PDU sessions was performed.

5.6.3.3 Notification procedure completion

Upon reception of SERVICE REQUEST message, CONTROL PLANE SERVICE REQUEST message or REGISTRATION REQUEST message, the AMF shall stop timer T3565 and proceed service request procedure as specified in subclauses 5.6.3.1 or registration procedure for mobility and periodic registration update as specified in subclauses 5.5.1.3. If no user-plane resources of PDU session(s) need to be re-established, the AMF should notify the SMF that the UE was reachable but did not accept to re-establish the user-plane resources of PDU session(s).

When the 5GMM entity in the AMF receives an indication from the lower layer that it has received the NGAP UE context resume request message as specified in 3GPP TS 38.413 [31], the AMF shall stop timer T3565.

Upon reception of NOTIFICATION RESPONSE message over non-3GPP access, the AMF shall stop timer T3565 and should notify the SMF that the UE is unreachable.

If the NOTIFICATION RESPONSE message includes the PDU session status information element, then:

- a) for single access PDU sessions, the AMF shall:
 - perform a local release of all those PDU sessions which are not in 5GSM state PDU SESSION INACTIVE on the AMF side associated with 3GPP access, but are indicated by the UE in the PDU session status information element in the NOTIFICATION RESPONSE message as being in 5GSM state PDU SESSION INACTIVE; and
 - 2) shall request the SMF to perform a local release of all those PDU sessions associated with 3GPP access. If any of those PDU sessions is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and
- b) For MA PDU sessions, the AMF shall:
 - for MA PDU sessions having user plane resources established only on the 3GPP access in the AMF side, but are indicated by the UE in the PDU session status information element in the NOTIFICATION RESPONSE message as no user plane resources established on the 3GPP access:
 - i) perform a local release of all those MA PDU sessions; and
 - ii) request the SMF to perform a local release of all those MA PDU sessions. If the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions; and

- 2) for MA PDU sessions having user plane resources established on both accesses in the AMF side, but are indicated by the UE in the PDU session status information element in the NOTIFICATION RESPONSE message as no user plane resources established on the 3GPP access:
 - i) perform a local release of 3GPP access user plane resources of all those MA PDU sessions; and
 - ii) request the SMF to perform a local release of 3GPP access user plane resources of all those MA PDU sessions. If the MA PDU session is associated with one or more multicast MBS sessions, the SMF shall consider the UE as removed from the associated multicast MBS sessions.

5.6.3.4 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3565.

The network shall, on the first expiry of the timer T3565, retransmit the NOTIFICATION message and shall reset and start timer T3565. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3565, the procedure shall be aborted. In addition, upon the fifth expiry of timer T3565:

For case a) in subclause 5.6.3.1, the AMF should notify the SMF that the UE is unreachable. The AMF may enter 5GMM-IDLE mode over 3GPP access.

For case b) in subclause 5.6.3.1, the AMF may either:

- 1) perform the paging procedure over the 3GPP access; or
- 2) notify the SMF that the UE is unreachable.

NOTE: Whether the AMF performs the paging procedure or notifies the SMF is up to operator's policies.

b) De-registration procedure collision

If the network receives a DEREGISTRATION REQUEST message before it receives a SERVICE REQUEST message or REGISTRATION REQUEST message, the AMF shall stop timer T3565 and proceed de-registration procedure as specified in subclause 5.5.2.

5.6.3.5 Abnormal cases on the UE side

The following abnormal cases can be identified:

a) NOTIFICATION message received via non-3GPP access with access type indicating 3GPP access when UEinitiated 5GMM specific procedure or service request procedure over 3GPP access is ongoing.

The UE shall proceed with 5GMM specific procedure or service request procedure. If for these procedures lower layers indicate that the access attempt is barred, then the UE shall handle the pending NOTIFICATION message as specified in subclause 5.6.3.2. Otherwise, the UE shall ignore the NOTIFICATION message once lower layers confirms the establishment of the signalling connection.

6 Elementary procedures for 5GS session management

6.1 Overview

6.1.1 General

This clause describes the procedures used for 5GS session management (5GSM) performed over an N1 NAS signalling connection.

The main function of the 5GSM sublayer is to support the PDU session handling in the UE and in the SMF (transferred via the AMF).

The 5GSM comprises procedures for:

- the authentication and authorization, establishment, modification and release of PDU sessions; and
- request for performing handover of an existing PDU session between 3GPP access and non-3GPP access, or to transfer an existing PDN connection in the EPS to the 5GS.

Each PDU session represents a PDU session established between the UE and an SMF. PDU sessions can remain established even if the radio and network resources constituting the corresponding PDU session between the UE and the SMF are temporarily released.

5GSM procedures can be performed only if a 5GMM context has been established between the UE and the AMF, and the secure exchange of NAS messages has been initiated by the AMF by use of the 5GMM procedures described in clause 5. Once the UE is successfully registered to a PLMN, a PDU session can be established. If no 5GMM context has been established, the 5GMM sublayer has to initiate the establishment of a 5GMM context by use of the 5GMM procedures as described in clause 5.

The UE can request the network to modify or release PDU sessions. The network can fulfil such a request from the UE by modifying a PDU session or releasing a PDU session using network-requested procedures (see subclause 6.3).

6.1.2 Types of 5GSM procedures

Three types of 5GSM procedures can be distinguished:

a) Procedures related to PDU sessions:

These procedures are initiated by the network and are used for authentication and authorization or manipulation of PDU sessions:

- 1) PDU authentication and authorization;
- 2) network-requested PDU session modification;
- 3) network-requested PDU session release; and
- 4) service-level authentication and authorization.

This procedure is initiated by the UE and to request for establishment of PDU sessions or to perform handover of an existing PDU session between 3GPP access and non-3GPP access, or to transfer an existing PDN connection in the EPS to the 5GS:

UE-requested PDU session establishment.

b) Transaction related procedures:

These procedures are initiated by the UE to request for handling of PDU sessions, i.e. to modify a PDU session, or to release a PDU session:

- 1) UE-requested PDU session modification; and
- 2) UE-requested PDU session release.

This procedure is initiated by the 5G ProSe UE-to-network relay UE and is used for the manipulation of PDU sessions:

remote UE report.

A successful transaction related procedure initiated by the UE triggers the network to execute one of the following procedures related to PDU session; network-requested PDU session modification procedure or network-requested PDU session release procedure. The UE treats the start of the procedure related to the PDU session as completion of the transaction related procedure.

c) Common procedure:

The following 5GSM procedure can be related to a PDU session or to a procedure transaction:

5GSM status procedure.

6.1.3 5GSM sublayer states

6.1.3.1 General

In the following subclauses, the possible states of a PDU session in the UE and the network side are described.

6.1.3.2 5GSM sublayer states in the UE

6.1.3.2.1 Overview

In the following subclauses, the possible 5GSM sublayer states of the UE are described and shown in figure 6.1.3.2.1.1.



NOTE 1: Not all possible transitions are shown in this figure.
NOTE 2: Some transitions shown in this figure are not applicable to the MA PDU session.
Figure 6.1.3.2.1.1: The 5GSM sublayer states for PDU session handling in the UE (overview)

6.1.3.2.2 PDU SESSION INACTIVE

No PDU session exists.

6.1.3.2.3 PDU SESSION ACTIVE PENDING

The UE has initiated a PDU session establishment procedure towards the network and is waiting for a response from the network.

6.1.3.2.4 PDU SESSION ACTIVE

The PDU session is active in the UE.

6.1.3.2.5 PDU SESSION INACTIVE PENDING

The UE has initiated a PDU session release procedure towards the network and is waiting for a response from the network.

6.1.3.2.6 PDU SESSION MODIFICATION PENDING

The UE has initiated a PDU session modification procedure towards the network and is waiting for a response from the network.

6.1.3.2.7 PROCEDURE TRANSACTION INACTIVE

No procedure transaction exists.

6.1.3.2.8 PROCEDURE TRANSACTION PENDING

The UE has initiated a procedure transaction towards the network.



Figure 6.1.3.2.8.1: The procedure transaction states in the UE (overview)

6.1.3.3 5GSM sublayer states in the network side

6.1.3.3.1 Overview

In the following subclauses, the possible 5GSM sublayer states of the network are described and shown in Figure 6.1.3.3.1.1.



NOTE 1: Not all possible transitions are shown in this figure. NOTE 2: Some transitions shown in this figure are not applicable to the MA PDU session. Figure 6.1.3.3.1.1: The 5GSM sublayer states for PDU session handling in the network (overview)

6.1.3.3.2 PDU SESSION INACTIVE

No PDU session exists.

6.1.3.3.3 PDU SESSION ACTIVE

The PDU session is active in the network.

6.1.3.3.4 PDU SESSION INACTIVE PENDING

The network has initiated a PDU session release procedure towards the UE and is waiting for a response from the UE.

6.1.3.3.5 PDU SESSION MODIFICATION PENDING

The network has initiated a PDU session modification procedure towards the UE and is waiting for a response from the UE.

6.1.3.3.6 PROCEDURE TRANSACTION INACTIVE

No procedure transaction exists.

6.1.3.3.7 PROCEDURE TRANSACTION PENDING

The network has initiated a procedure transaction towards the UE.



Figure 6.1.3.3.7.1: The procedure transaction states in the network (overview)

6.1.4 Coordination between 5GSM and ESM

6.1.4.1 Coordination between 5GSM and ESM with N26 interface

Interworking with EPS is supported for a PDU session, if the PDU session includes the mapped EPS bearer context(s) or has association(s) between QoS flow and mapped EPS bearer after inter-system change from S1 mode to N1 mode. The SMF shall not include any mapped EPS bearer contexts associated with a PDU session for LADN and with a PDU session which is a multi-homed IPv6 PDU session. See coding of the Mapped EPS bearer contexts IE in subclause 9.11.4.8. In an MA PDU session, the UE shall have one set of the mapped EPS bearer contexts. The network can provide the set of the mapped EPS bearer contexts of the MA PDU session via either access of the MA PDU session. In an MA PDU session, the UE shall support modification or deletion via an access of a mapped EPS bearer context of the MA PDU session created via the same or the other access.

Upon inter-system change from N1 mode to S1 mode, the UE shall create the default EPS bearer context and the dedicated EPS bearer context(s) based on the parameters of the mapped EPS bearer contexts or the associations between QoS flow and mapped EPS bearer in the PDU session, if available. The EPS bearer identity assigned for the QoS flow of the default QoS rule becomes the EPS bearer identity of the default bearer in the corresponding PDN connection. If there is no EPS bearer identity assigned to the QoS flow of the default QoS rule of a PDU session associated with 3GPP access, or if there is no corresponding mapped EPS bearer contexts associated with the EPS bearer identity assigned to the QoS flow of the default QoS rule of a PDU session associated with 3GPP access:

- a) the PDU session is not an MA PDU session established over both 3GPP access and non-3GPP access, the UE shall perform a local release of the PDU session; or
- b) the PDU session is an MA PDU session established over both 3GPP access and non-3GPP access, the UE shall perform a local release of the PDU session over 3GPP access and consider that the MA PDU session is established over non-3GPP access only.

If there is no EPS bearer identity assigned to the QoS flow(s) of a PDU session associated with 3GPP access which is not associated with the default QoS rule, or if there is no corresponding mapped EPS bearer contexts associated with the EPS bearer identity assigned to the QoS flow of the non-default QoS rule of a PDU session associated with 3GPP access, unless the PDU session is an MA PDU session established over 3GPP access and over non-3GPP access, the UE shall locally delete the QoS rules and the QoS flow description(s). The UE uses the parameters from each PDU session for which interworking with EPS is supported to create corresponding default EPS bearer context and optionally dedicated EPS bearer context(s) as follows:

- a) the PDU session type of the PDU session shall be mapped to the PDN type of the default EPS bearer context as follows:
 - 1) the PDN type shall be set to "non-IP" if the PDU session type is "Unstructured";
 - 2) the PDN type shall be set to "IPv4" if the PDU session type is "IPv4";
 - 3) the PDN type shall be set to "IPv6" if the PDU session type is "IPv6";
 - 4) the PDN type shall be set to "IPv4v6" if the PDU session type is "IPv4v6";
 - 5) the PDN type shall be set to "non-IP" if the PDU session type is "Ethernet", and the UE, the network or both of them do not support Ethernet PDN type in S1 mode; and

- 6) the PDN type shall be set to "Ethernet" if the PDU session type is "Ethernet" and the UE and the network support Ethernet PDN type in S1 mode;
- b) the PDU address of the PDU session shall be mapped to the PDN address of the default EPS bearer context as follows:
 - 1) the PDN address of the default EPS bearer context is set to the PDU address of the PDU session, if the PDU session type is "IPv4", "IPv6" or "IPv4v6"; and
 - the PDN address of the default EPS bearer context is set to zero, if the PDU session type is "Ethernet" or "Unstructured";
- c) the DNN of the PDU session shall be mapped to the APN of the default EPS bearer context, unless the PDU session is an emergency PDU session;
- d) the APN-AMBR and extended APN-AMBR received in the parameters of the default EPS bearer context of the mapped EPS bearer contexts shall be mapped to the APN-AMBR and extended APN-AMBR of the default EPS bearer context;
- e) for each PDU session in state PDU SESSION ACTIVE, PDU SESSION MODIFICATION PENDING or PDU SESSION INACTIVE PENDING:
 - 1) if the UE is performing an inter-system change from N1 mode to WB-S1 mode, the UE shall set the state of the mapped EPS bearer context(s) to BEARER CONTEXT ACTIVE; or
 - 2) if the UE is performing an inter-system change from N1 mode to NB-S1 mode, for the mapped EPS bearer context corresponding to the default EPS bearer, the UE shall set the state of the mapped EPS bearer context to BEARER CONTEXT ACTIVE. Additionally, if the UE is performing an inter-system change from WB-N1 mode to NB-S1 mode, for the mapped EPS bearer context corresponding to a dedicated EPS bearer, if any, the UE shall set the state of the mapped EPS bearer context to BEARER CONTEXT INACTIVE; and
- f) for any other PDU session the UE shall set the state of the mapped EPS bearer context(s) to BEARER CONTEXT INACTIVE.

Additionally, for each mapped EPS bearer context or the association between QoS flow and mapped EPS bearer in the PDU session:

- a) the EPS bearer identity shall be set to the EPS bearer identity received in the mapped EPS bearer context, or the EPS bearer identity associated with the QoS flow;
- b) the EPS QoS parameters shall be set to the mapped EPS QoS parameters of the EPS bearer received in the mapped EPS bearer context, or the EPS QoS parameters associated with the QoS flow;
- c) the extended EPS QoS parameters shall be set to the mapped extended EPS QoS parameters of the EPS bearer received in the mapped EPS bearer context, or the extended EPS QoS parameters associated with the QoS flow; and
- d) the traffic flow template shall be set to the mapped traffic flow template of the EPS bearer received in the mapped EPS bearer context, or the stored traffic flow template associated with the QoS flow, if available.

After inter-system change from N1 mode to S1 mode, the UE shall associate the PDU session identity, the S-NSSAI, and the session-AMBR with the default EPS bearer context, and for each EPS bearer context mapped from one or more QoS flows, associate the QoS rule(s) for the QoS flow(s) and the QoS flow description(s) for the QoS flow(s) with the EPS bearer context.

If the PDU session is associated with the control plane only indication and supports interworking with EPS, after intersystem change from N1 mode to S1 mode, the UE shall associate the EPS bearer context(s) of the PDN connection corresponding to the PDU session with the control plane only indication.

If the PDU session is associated with a PDU session pair ID, after inter-system change from N1 mode to S1 mode, the UE shall associate the default EPS bearer context of the PDN connection corresponding to the PDU session with the PDU session pair ID. If the PDU session is associated with an RSN, after inter-system change from N1 mode to S1 mode, the UE shall associate the default EPS bearer context of the PDN connection corresponding to the PDU session with the RSN.

After inter-system change from N1 mode to S1 mode, the UE and the SMF shall maintain the PDU session type of the PDU session until the PDN connection corresponding to the PDU session is released if the UE supports non-IP PDN type and the PDU session type is "Ethernet" or "Unstructured".

After inter-system change from N1 mode to S1 mode, the UE and the SMF shall maintain the following 5GSM attributions and capabilities associated with the PDU session until the PDN connection corresponding to the PDU session is released:

- a) the always-on PDU session indication;
- b) the maximum number of supported packet filters;
- c) the support of reflective QoS;
- d) the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink and the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink;
- e) the support of multi-homed IPv6 PDU session; and
- f) if the PDU session is an MA PDU session established over 3GPP access, the PDN connection of the default EPS bearer corresponding to the MA PDU session shall be considered as a user-plane resource of the MA PDU session.

After inter-system change from N1 mode to S1 mode, the UE operating in single-registration mode in a network supporting N26 interface shall deem that the following features are supported by the network on the PDN connection corresponding to the PDU session:

- a) PS data off; and
- b) Local address in TFT.

If there is a QoS flow used for IMS signalling, after inter-system change from N1 mode to S1 mode, the EPS bearer associated with the QoS flow for IMS signalling becomes the EPS bearer for IMS signalling.

When the UE is provided with a new session-AMBR in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message, the UE shall discard the corresponding association(s) and associate the new value(s) with the EPS bearer context.

The network may provide the UE with one or more QoS rules by including either one QoS rules parameter, or one QoS rules with the length of two octets parameter, but not both, in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message. The network may provide the UE with one or more QoS flow descriptions corresponding to the EPS bearer context being modified, by including either one QoS flow descriptions parameter, or one QoS flow descriptions with the length of two octets parameter, but not both, in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message.

When the UE is provided with one or more QoS flow descriptions or the EPS bearer identity of an existing QoS flow description is modified in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message, the UE shall check the EPS bearer identity included in the QoS flow description; and:

- a) if the EPS bearer identity corresponds to the EPS bearer context being modified or the EPS bearer identity is not included, the UE shall store the QoS flow description and all the associated QoS rules, if any, for the EPS bearer context being modified for use during inter-system change from S1 mode to N1 mode; and
- b) otherwise the UE shall locally delete the QoS flow description and all the associated QoS rules, if any, and include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #84 "syntactical error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

When the UE is provided with one or more QoS rules, or one or more QoS flow descriptions in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message, the UE shall process the QoS rules sequentially starting with the first QoS rule and shall process the QoS flow descriptions sequentially starting with the first QoS flow description. The UE shall check the QoS rules and QoS flow descriptions for different types of errors as follows:

- NOTE 1: If an error is detected in a QoS rule or a QoS flow description which requires sending a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause value, then the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter and the QoS flow descriptions with the length of two octets parameter included in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message are discarded, if any.
- NOTE 2: If the EPS bearer context modification procedure is rejected, then the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter and the QoS flow descriptions with the length of two octets parameter included in the Protocol configuration options IE or Extended protocol configuration options IE in the MODIFY EPS BEARER CONTEXT REQUEST message are discarded, if any.
- a) Semantic errors in QoS operations:
 - When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters" on the default QoS rule and the DQR bit is set to "the QoS rule is not the default QoS rule".
 - 2) When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters" on a QoS rule which is not the default QoS rule and the DQR bit is set to "the QoS rule is the default QoS rule".
 - 3) When the rule operation is "Create new QoS rule" and the DQR bit is set to "the QoS rule is the default QoS rule" when there's already a default QoS rule with different QoS rule identifier.
 - 4) When the rule operation is "Delete existing QoS rule" on the default QoS rule.
 - 5) When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters", or "Modify existing QoS rule without modifying packet filters" and two or more QoS rules associated with this PDU session would have identical precedence values.
 - 6) When the rule operation is "Modify existing QoS rule and delete packet filters", the QoS rule is a QoS rule of a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the packet filter list in the resultant QoS rule is empty.
 - 7) When the rule operation is "Create new QoS rule", and there is already an existing QoS rule with the same QoS rule identifier and the existing QoS rule is associated with a QoS flow description stored for the EPS bearer context being modified or the existing QoS rule is not associated with any QoS flow description.
 - 8) When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters", or "Modify existing QoS rule without modifying packet filters" and there is no existing QoS rule with the same QoS rule identifier associated with a QoS flow description stored for the EPS bearer context being modified.
 - 9) When the rule operation is "Delete existing QoS rule" and there is no existing QoS rule with the same QoS rule identifier associated with a QoS flow description stored for the EPS bearer context being modified.
 - 10) When the flow description operation is "Create new QoS flow description" and there is already an existing QoS flow description with the same QoS flow identifier stored for the EPS bearer context being modified.
 - 11) When the flow description operation is "Modify existing QoS flow description" and there is no existing QoS flow description with the same QoS flow identifier stored for the EPS bearer context being modified.
 - 12) When the flow description operation is "Delete existing QoS flow description" and there is no existing QoS flow description with the same QoS flow identifier stored for the EPS bearer context being modified.
 - 13) When the UE determines that:
 - i) the default EPS bearer context is associated with one or more QoS flows but the default EPS bearer context is not associated with the default QoS rule.

- ii) a dedicated EPS bearer context is associated with one or more QoS flows but the dedicated EPS bearer context is associated with the default QoS rule.
- 14) When the rule operation is "Create new QoS rule" and there is already an existing QoS rule with the same QoS rule identifier associated with a QoS flow description stored for an EPS bearer context different from the EPS bearer context being modified and belonging to the same PDN connection as the EPS bearer context being modified.
- 15) When the flow description operation is "Create new QoS flow description", and there is already an existing QoS flow description with the same QoS flow identifier stored for an EPS bearer context different from the EPS bearer context being modified and belonging to the same PDN connection as the EPS bearer context being modified.
- 16) When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters", or "Modify existing QoS rule without modifying packet filters" and the resultant QoS rule is associated with a QoS flow description stored for an EPS bearer context different from the EPS bearer context being modified.
- 17) When the rule operation is "Create new QoS rule", the DQR bit is set to "the QoS rule is not the default QoS rule", the QoS rule is provided for a PDN connection of PDN type "non-IP" and there is locally available information associated with the PDN connection that is set to "Unstructured".
- 18) When the flow description operation is "Create new QoS flow description" or "Modify existing QoS flow description", the QFI associated with the QoS flow description is not the same as the QFI of the default QoS rule, the QoS flow description is provided for a PDN connection of PDN type "non-IP" and there is locally available information associated with the PDN connection that is set to "Unstructured".
- In case 5, if the old QoS rule (i.e. the QoS rule that existed before the MODIFY EPS BEARER CONTEXT REQUEST message was received) is not the default QoS rule and the old QoS rule is associated with a QoS flow description stored for the EPS bearer context being modified, the UE shall not diagnose an error, shall further process the new request and, if it was processed successfully, shall delete the old QoS rule which has identical precedence value. Otherwise, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #83 "semantic error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

In case 6, if the QoS rule is not the default QoS rule, the UE shall delete the QoS rule. If the QoS rule is the default QoS rule, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #83 "semantic error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

In case 7, if the existing QoS rule is not the default QoS rule and the DQR bit of the new QoS rule is set to "the QoS rule is not the default QoS rule", the UE shall not diagnose an error, further process the create request and, if it was processed successfully, delete the old QoS rule (i.e. the QoS rule that existed before the MODIFY EPS BEARER CONTEXT REQUEST message was received). If the existing QoS rule is the default QoS rule or the DQR bit of the new QoS rule is set to "the QoS rule is the default QoS rule", the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause #83 "semantic error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

In case 9, the UE shall not diagnose an error, further process the delete request and, if it was processed successfully, consider the respective QoS rule as successfully deleted.

In case 10, the UE shall not diagnose an error, further process the create request and, if it was processed successfully, delete the old QoS flow description (i.e. the QoS flow description that existed before the MODIFY EPS BEARER CONTEXT REQUEST message was received).

In case 12, the UE shall not diagnose an error, further process the delete request and, if it was processed successfully, consider the respective QoS flow description as successfully deleted.

Otherwise, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #83 "semantic error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

- b) Syntactical errors in QoS operations:
 - When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters" or "Modify existing QoS rule and delete packet filters", the packet filter list in the QoS rule is empty, and the QoS rule is provided for a PDN connection of PDN type IPv4, IPv6, IPv4v6 or Ethernet, or for a PDN connection of PDN type "non-IP" and there is locally available information associated with the PDN connection that is set to "Ethernet".
 - 2) When the rule operation is "Delete existing QoS rule" or "Modify existing QoS rule without modifying packet filters" with a non-empty packet filter list in the QoS rule.
 - 3) When the rule operation is "Modify existing QoS rule and delete packet filters" and the packet filter to be deleted does not exist in the original QoS rule.
 - 4) Void.
 - 5) When there are other types of syntactical errors in the coding of the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter or the QoS flow descriptions with the length of two octets parameter, such as a mismatch between the number of packet filters subfield, and the number of packet filters in the packet filter list when the rule operation is "delete existing QoS rule" or "create new QoS rule", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list (i.e., there is no QoS rule precedence subfield included in the QoS rule IE), the QoS Rule Identifier is set to "no QoS rule identifier assigned", or the QoS flow identifier is set to "no QoS flow identifier assigned".
 - 6) When, the
 - A) rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters", the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1), and there is no QoS flow description with a QFI corresponding to the QFI of the resulting QoS rule.
 - B) flow description operation is "Delete existing QoS flow description", and the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) with a QFI corresponding to the QFI of the QoS flow description that is deleted (i.e. there is no associated QoS flow description with the same QFI).
 - 7) When the flow description operation is "Create new QoS flow description" or "Modify existing QoS flow description", and the UE determines that there is a QoS flow description of a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) which lacks at least one of the mandatory parameters (i.e., GFBR uplink, GFBR downlink, MFBR uplink and MFBR downlink). If the QoS flow description does not include a 5QI, the UE determines this by using the QFI as the 5QI.

In case 3 the UE shall not diagnose an error, further process the deletion request and, if no error according to items c and d was detected, consider the respective packet filter as successfully deleted.

Otherwise the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #84 "syntactical error in the QoS operation" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

- NOTE 3: It is not considered an error if the UE determines that after processing all QoS operations on QoS rules and QoS flow descriptions there is a QoS flow description that is not associated with any QoS rule and the UE is not in NB-N1 mode.
- c) Semantic errors in packet filters:
 - 1) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.

The UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #44 "semantic errors in packet filter(s)" in the MODIFY EPS BEARER CONTEXT ACCEPT message.
d) Syntactical errors in packet filters:

- 1) When the rule operation is "Modify existing QoS rule and add packet filters" or "Modify existing QoS rule and replace all packet filters", and two or more packet filters in the resultant QoS rule would have identical packet filter identifiers.
- 2) When the rule operation is "Create new QoS rule", and two or more packet filters in the resultant QoS rule would have identical packet filter identifiers.
- 3) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.
- 4) When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters" or "Modify existing QoS rule and replace all packet filters" with a non-empty packet filter list in the QoS rule, and the DQR bit is set to "the QoS rule is the default QoS rule", the QoS rule is provided for a PDN connection of PDN type "non-IP" and there is locally available information associated with the PDN connection that is set to "Unstructured".

In case 1, if two or more packet filters with identical packet filter identifiers are contained in the MODIFY EPS BEARER CONTEXT REQUEST message, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #45 "syntactical error in packet filter(s)" in the MODIFY EPS BEARER CONTEXT ACCEPT message. Otherwise, the UE shall not diagnose an error, further process the MODIFY EPS BEARER CONTEXT REQUEST message and, if it was processed successfully, delete the old packet filters which have the identical packet filter identifiers.

Otherwise the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #45 "syntactical error in packet filter(s)" in the MODIFY EPS BEARER CONTEXT ACCEPT message.

If the UE detects different errors in the QoS rules and QoS flow descriptions as described in this subclause which requires sending a 5GSM cause parameter in the MODIFY EPS BEARER CONTEXT ACCEPT message, the UE shall include a single 5GSM cause parameter in the MODIFY EPS BEARER CONTEXT ACCEPT message.

NOTE 4: The 5GSM cause to use cannot be different from #44 "semantic error in packet filter(s)", #45 "syntactical errors in packet filter(s)", #83 "semantic error in the QoS operation" or #84 "syntactical error in the QoS operation". The selection of a 5GSM cause is up to UE implementation.

Upon successful completion of an EPS attach procedure or tracking area updating procedure after inter-system change from N1 mode to S1 mode (see 3GPP TS 24.301 [15]), unless the PDU session is an MA PDU session established over 3GPP access and over non-3GPP access,

- a) the UE shall delete any UE derived QoS rules of each PDU session which has been transferred to EPS;
- b) the UE and the SMF shall perform a local release of the PDU session(s) associated with 3GPP access which have not been transferred to EPS; and
- c) the UE and the SMF shall perform a local release of QoS flow(s) which have not been transferred to EPS, of the PDU session(s) which have been transferred to EPS. The UE and the SMF shall also perform a local release of any QoS flow description not associated with any QoS rule and not associated with any mapped EPS bearer context. If there is a QoS flow description not associated with any QoS rule, but associated with a mapped EPS bearer context, and after the inter-system change from N1 mode to S1 mode the respective EPS bearer context is active, then the UE shall associate the QoS flow description with the EPS bearer context.

For PDU session(s) associated with non-3GPP access in 5GS, if present, the UE may:

- a) keep some or all of these PDU sessions still associated with non-3GPP access in 5GS, if supported;
- b) release some or all of these PDU sessions explicitly by initiating the UE requested PDU session release procedure(s); or
- c) attempt to transfer some or all of these PDU sessions from N1 mode to S1 mode by initiating the UE requested PDN connectivity procedure(s) with the PDN CONNECTIVITY REQUEST message created as follows:
 - 1) if the PDU session is an emergency PDU session, the request type shall be set to "handover of emergency bearer services". Otherwise the request type shall be set to "handover";

- 2) the PDU session type of the PDU session shall be mapped to the PDN type of the default EPS bearer context as follows:
 - i) the PDN type shall be set to "non-IP" if the PDU session type is "Unstructured";
 - ii) the PDN type shall be set to "IPv4" if the PDU session type is "IPv4";
 - iii) the PDN type shall be set to "IPv6" if the PDU session type is "IPv6";
 - iv) the PDN type shall be set to "IPv4v6" if the PDU session type is "IPv4v6";
 - v) the PDN type shall be set to "non-IP" if the PDU session type is "Ethernet" and the UE, the network or both of them do not support Ethernet PDN type in S1 mode; and
 - vi) the PDN type shall be set to "Ethernet" if the PDU session type is "Ethernet" and the UE and the network support Ethernet PDN type in S1 mode;
- 3) the DNN of the PDU session shall be mapped to the APN of the default EPS bearer context, unless the PDN connection is an emergency PDN connection; and
- 4) the PDU session ID parameter in the Protocol configuration options IE or the Extended protocol configuration options IE shall be set to the PDU session identity of the PDU session.

If a PDU session associated with non-3GPP access is transferred to EPS, the UE shall associate the PDU session identity with the default EPS bearer context and shall delete any UE derived QoS rules of such PDU session.

Interworking to 5GS is supported for a PDN connection, if the corresponding default EPS bearer context includes a PDU session identity, an S-NSSAI, if the PDN connection is a non-emergency PDN connection, session AMBR and one or more QoS flow descriptions received in the Protocol configuration options IE or Extended protocol configuration options IE (see 3GPP TS 24.301 [15]), or the default EPS bearer context has association with the PDU session identity, the S-NSSAI, if the PDU session is a non-emergency PDU session, the session-AMBR and one or more QoS flow descriptions after inter-system change from N1 mode to S1 mode.

For a PDN connection established in S1 mode, to enable the UE to attempt to transfer the PDN connection from S1 mode to N1 mode in case of inter-system change, the UE shall allocate a PDU session identity, indicate the allocated PDU session identity in the PDU session ID parameter in the Protocol configuration options IE of the PDN CONNECTIVITY REQUEST message and associate the allocated PDU session identity with the default EPS bearer context of the PDN connection. If an N5CW device supports 3GPP access and establishes a new PDN connection in S1 mode, the N5CW device shall refrain from allocating "PDU session identity value 15".

For a PDN connection established in S1 mode, the SMF assigning the QoS rules shall consider that the UE supports 16 packet filters for the corresponding PDU session until the UE indicates a higher number (as specified in subclause 6.4.2.2).

The network may provide the UE with one or more QoS rules by including either one QoS rules parameter, or one QoS rules with the length of two octets parameter, but not both, in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message. The network may provide the UE with one or more QoS flow descriptions corresponding to the EPS bearer context being activated, by including either one QoS flow descriptions parameter, or one QoS flow descriptions with the length of two octets parameter, but not both, in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEFINITION options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message.

When the UE is provided with one or more QoS flow descriptions in the Protocol configuration options IE or Extended protocol configuration options IE of the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message, the UE shall check the EPS bearer identity included in the QoS flow description; and:

- a) if the EPS bearer identity corresponds to the EPS bearer context being activated or the EPS bearer identity is not included, the UE shall store the QoS flow description and all the associated QoS rules, if any, for the EPS bearer context being activated for use during inter-system change from S1 mode to N1 mode; and
- b) otherwise the UE shall locally delete the QoS flow description and all the associated QoS rules, if any, and include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause

parameter set to 5GSM cause #84 "syntactical error in the QoS operation" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

When the UE is provided with one or more QoS rules, or one or more QoS flow descriptions in the Protocol configuration options IE or Extended protocol configuration options IE of the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message, the UE shall process the QoS rules sequentially starting with the first QoS rule and shall process the QoS flow descriptions sequentially starting with the first QoS flow description. The UE shall check QoS rules and QoS flow descriptions for different types of errors as follows:

- NOTE 5: If an error is detected in a QoS rule or a QoS flow description which requires sending a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause value, then the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter and the QoS flow descriptions with the length of two octets parameter included in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message are discarded, if any.
- NOTE 6: If the default EPS bearer context activation procedure or the dedicated EPS bearer context activation procedure is rejected, then the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter and the QoS flow descriptions with the length of two octets parameter included in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST or ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message are discarded, if any.
- a) Semantic errors in QoS operations:
 - 1) When the rule operation is "Create new QoS rule" and the DQR bit is set to "the QoS rule is the default QoS rule" when there's already a default QoS rule.
 - 2) When the rule operation is received in an ACTIVATE DEFAULT EPS BEARER CONTEXT REQUEST message, the rule operation is "Create new QoS rule", and there is no rule with the DQR bit set to "the QoS rule is the default QoS rule".
 - 3) When the rule operation is "Create new QoS rule" and two or more QoS rules associated with this PDU session would have identical precedence values.
 - 4) When the rule operation is an operation other than "Create new QoS rule".
 - 5) When the flow description operation is an operation other than "Create new QoS flow description".
 - 6) When the UE determines that:
 - i) the default EPS bearer context is associated with one or more QoS flows but the default EPS bearer context is not associated with the default QoS rules.
 - ii) a dedicated EPS bearer context is associated with one or more QoS flows but the dedicated EPS bearer context is associated with the default QoS rule.
 - 7) When the flow description operation is received in an ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message, the flow description operation is "Create new QoS flow description" and there is already an existing QoS flow description with the same QoS flow identifier stored for an EPS bearer context different from the EPS bearer context being activated and belonging to the same PDN connection as the EPS bearer context being activated.
 - 8) When the rule operation is received in an ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message, the rule operation is "Create new QoS rule" and there is already an existing QoS rule with the same QoS rule identifier stored for an EPS bearer context different from the EPS bearer context being activated and belonging to the same PDN connection as the EPS bearer context being activated.
 - 9) When the rule operation is received in an ACTIVATE DEDICATED EPS BEARER CONTEXT REQUEST message, the rule operation is "Create new QoS rule" and the resultant QoS rule is associated with a QoS flow description stored for an EPS bearer context different from the EPS bearer context being activated and belonging to the same PDN connection as the EPS bearer context being activated.

In case 4, if the rule operation is for a non-default QoS rule, the UE shall delete the QoS rule. If the QoS rule is the default QoS rule, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #83 "semantic error in the QoS operation" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

Otherwise for all the cases above, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #83 "semantic error in the QoS operation" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

- b) Syntactical errors in QoS operations:
 - 1) When the rule operation is "Create new QoS rule", the packet filter list in the QoS rule is empty, and the QoS rule is provided for a PDN connection of PDN type IPv4, IPv6, IPv4v6 or Ethernet, or for a PDN connection of PDN type "non-IP" and there is locally available information associated with the PDN connection that is set to "Ethernet".
 - 2) Void.
 - 3) When there are other types of syntactical errors in the coding of the QoS rules parameter, the QoS rules with the length of two octets parameter, the QoS flow descriptions parameter or the QoS flow descriptions with the length of two octets parameter, such as a mismatch between the number of packet filters subfield, and the number of packet filters in the packet filter list when the rule operation is "delete existing QoS rule" or "create new QoS rule", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list (i.e., there is no QoS rule precedence subfield included in the QoS rule IE), the QoS Rule Identifier is set to "no QoS rule identifier assigned", or the QoS flow identifier is set to "no QoS flow identifier assigned.
 - 4) When, the
 - A) rule operation is "Create new QoS rule", the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1), and there is no QoS flow description with a QFI corresponding to the QFI of the resulting QoS rule.
 - 5) When the flow description operation is "Create new QoS flow description", and the UE determines that there is a QoS flow description of a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) which lacks at least one of the mandatory parameters (i.e., GFBR uplink, GFBR downlink, MFBR uplink and MFBR downlink). If the QoS flow description does not include a 5QI, the UE determines this by using the QFI as the 5QI.

In case 1, case 3 or case 4, if the QoS rule is not the default QoS rule, the UE shall delete the QoS rule. If the QoS rule is the default QoS rule, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #84 "syntactical error in the QoS operation" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

In case 5, the UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #84 "syntactical error in the QoS operation" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

- NOTE 7: It is not considered an error if the UE determines that after processing all QoS operations on QoS rules and QoS flow descriptions there is a QoS flow description that is not associated with any QoS rule and the UE is not in NB-N1 mode.
- c) Semantic errors in packet filters:
 - 1) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.

The UE shall include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #44 "semantic errors in packet filter(s)" in the ACTIVATE

DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

- d) Syntactical errors in packet filters:
 - 1) When the rule operation is "Create new QoS rule" and two or more packet filters in the resultant QoS rule would have identical packet filter identifiers.
 - 2) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.

The UE shall delete the QoS rule and include a Protocol configuration options IE or Extended protocol configuration options IE with a 5GSM cause parameter set to 5GSM cause #45 "syntactical error in packet filter(s)" in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

If the UE detects different errors in the QoS rules and QoS flow descriptions as described in this subclause which requires sending a 5GSM cause parameter in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message, the UE shall include a single 5GSM cause parameter in the ACTIVATE DEFAULT EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT or ACTIVATE DEDICATED EPS BEARER CONTEXT ACCEPT message.

NOTE 8: The 5GSM cause to use cannot be different from #44 "semantic error in packet filter(s)", #45 "syntactical errors in packet filter(s)", #83 "semantic error in the QoS operation" or #84 "syntactical error in the QoS operation". The selection of a 5GSM cause is up to UE implementation.

Upon inter-system change from S1 mode to N1 mode, the UE uses the parameters from the default EPS bearer context of each PDN connection for which interworking to 5GS is supported to create a corresponding PDU session associated with 3GPP access as follows, unless the PDN connection is a user-plane resource of an MA PDU session:

- a) the PDN type of the default EPS bearer context shall be mapped to the PDU session type of the PDU session as follows:
 - 1) if the PDN type is "non-IP":
 - the PDU session type is set to the locally available information associated with the PDN connection (either "Ethernet" or "Unstructured"), if available; or
 - otherwise, the PDU session type is set to "Unstructured";
 - 2) if the PDN type is "IPv4" the PDU session type is set to "IPv4";
 - 3) if the PDN type is "IPv6", the PDU session type is set to "IPv6";
 - 4) if the PDN type is "IPv4v6", the PDU session type is set to "IPv4v6"; and
 - 5) if the PDN type is "Ethernet", the PDU session type is "Ethernet";
- b) the PDN address of the default EPS bearer context shall be mapped to PDU address of the PDU session, if the PDN type is "IPv4", "IPv6" or "IPv4v6";
- c) the APN of the default EPS bearer context shall be mapped to the DNN of the PDU session, unless the PDN connection is an emergency PDN connection;
- d) for each default EPS bearer context in state BEARER CONTEXT ACTIVE the UE shall set the state of the mapped PDU session to PDU SESSION ACTIVE; and
- e) for any other default EPS bearer context the UE shall set the state of the mapped PDU session to PDU SESSION INACTIVE.

Additionally, the UE shall set:

 a) the PDU session identity of the PDU session to the PDU session identity included by the UE in the Protocol configuration options IE or Extended protocol configuration options IE in the PDN CONNECTIVITY REQUEST message, or the PDU session identity associated with the default EPS bearer context;

- b) the S-NSSAI of the PDU session to the S-NSSAI included by the network in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message, or the S-NSSAI associated with the default EPS bearer context, if the PDN connection is a nonemergency PDN connection;
- c) the session-AMBR of the PDU session to the session-AMBR included by the network in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message or the MODIFY EPS BEARER CONTEXT REQUEST message, or the session-AMBR associated with the default EPS bearer context;
- d) the SSC mode of the PDU session to "SSC mode 1"; and
- e) the always-on PDU session indication to the always-on PDU session indication maintained in the UE, if any.

Upon inter-system change from S1 mode to N1 mode, for each PDN connection which is a user-plane resource of MA PDU session and for which interworking to 5GS is supported, the UE shall consider that the MA PDU session is established over 3GPP access and, unless the MA PDU session is established over non-3GPP access too, the UE shall set the session-AMBR of the PDU session to the session-AMBR included by the network in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message or the MODIFY EPS BEARER CONTEXT REQUEST message, or the session-AMBR associated with the default EPS bearer context of the PDN connection.

Additionally, for each EPS bearer context of the PDN connection, the UE shall create QoS flow(s) each of which is associated with the QoS flow description received in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message, ACTIVATE DEDICATED EPS BEARER REQUEST message, and/or MODIFY EPS BEARER REQUEST message (see 3GPP TS 24.301 [15]), or the QoS flow description associated with EPS bearer context, unless:

- a) the PDU session is an MA PDU session which:
 - 1) is established over non-3GPP access; and
 - 2) has a PDN connection as a user-plane resource; and
- b) the QoS flow already exists over the non-3GPP access.

Additionally, for each EPS bearer context of the PDN connection, the UE shall create QoS rules(s), if any, each of which is associated with the QoS rule received in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message, ACTIVATE DEDICATED EPS BEARER REQUEST message, or MODIFY EPS BEARER CONTEXT REQUEST message (see 3GPP TS 24.301 [15]), or the QoS rules associated with EPS bearer context, unless:

- a) the PDU session is an MA PDU session which:
 - 1) is established over non-3GPP access; and
 - 2) has a PDN connection as a user-plane resource; and
- b) the QoS rule already exists over the non-3GPP access.
- NOTE 9: For a QoS rule which does not exist over non-3GPP access, the UE does not create the QoS rule if the QoS rule is the default QoS rule, or the precedence value of the QoS rule equals to the precedence value of a QoS rule exists over the non-3GPP access.

Additionally, for each PDU session which was created at inter-system change from S1 mode to N1 mode from a corresponding PDN connection of the "Ethernet" PDN type, the UE shall consider that Ethernet PDN type in S1 mode is supported by the network and the SMF shall consider that Ethernet PDN type in S1 mode is supported by the UE.

The UE and the network shall locally release the PDN connection(s) and EPS bearer context(s) associated with the 3GPP access which have not been transferred to 5GS.

After inter-system change from S1 mode to N1 mode, for each QoS flow mapped from an EPS bearer context the UE shall associate the EPS bearer identity, the EPS QoS parameters, the extended EPS QoS parameters, and the traffic flow template, if available, of the EPS bearer context with the QoS flow.

After inter-system change from S1 mode to N1 mode, for each QoS flow of an MA PDU session which:

- a) is established over non-3GPP access; and
- b) has a PDN connection as a user-plane resource;

such that the QoS flow was received in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message, ACTIVATE DEDICATED EPS BEARER REQUEST message, MODIFY EPS BEARER CONTEXT REQUEST message, (see 3GPP TS 24.301 [15]), or associated with EPS bearer context, the UE shall associate the EPS bearer identity, the EPS QoS parameters, the extended EPS QoS parameters, and the traffic flow template, if available, of the EPS bearer context with the QoS flow.

If the EPS bearer context(s) of the PDN connection are associated with the control plane only indication, and the PDN connection supports interworking to 5GS, after inter-system change from S1 mode to N1 mode, the UE shall associate the PDU session corresponding to the PDN connection with the control plane only indication.

If the default EPS bearer context of the PDN connection is associated with the PDU session pair ID, and the PDN connection supports interworking to 5GS, after inter-system change from S1 mode to N1 mode, the UE shall associate the PDU session corresponding to the PDN connection with the PDU session pair ID. If the default EPS bearer context of the PDN connection is associated with the RSN, and the PDN connection supports interworking to 5GS, after inter-system change from S1 mode to N1 mode, the UE shall associate the PDU session corresponding to the PDN connection supports interworking to 5GS, after inter-system change from S1 mode to N1 mode, the UE shall associate the PDU session corresponding to the PDN connection with the RSN.

If there is an EPS bearer used for IMS signalling, after inter-system change from S1 mode to N1 mode, the QoS flow of the default QoS rule in the corresponding PDU session is used for IMS signalling.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, the SMF shall determine the PDU session indication as specified in subclause 6.3.2.2.

When the UE is provided with one or more mapped EPS bearer contexts in the Mapped EPS bearer contexts IE of the PDU SESSION MODIFICATION COMMAND message, the UE shall process the mapped EPS bearer contexts sequentially starting with the first mapped EPS bearer context.

When the UE is provided with a new EPS bearer identity, a new EPS QoS parameters, a new extended EPS QoS parameters, a new APN-AMBR or a new extended APN-AMBR in the Mapped EPS bearer context IE of the PDU SESSION MODIFICATION COMMAND message for a QoS flow, the UE shall discard the corresponding association(s) and associate the new value(s) with the QoS flow.

When the UE is provided with a new traffic flow template in the Mapped EPS bearer contexts IE of the PDU SESSION MODIFICATION COMMAND message for a QoS flow, the UE shall check the traffic flow template for different types of TFT IE errors as specified in subclause 6.3.2.3.

When a QoS flow is deleted, the associated EPS bearer context information that are mapped from the deleted QoS flow shall be deleted from the UE and the network if there is no other existing QoS flow associated with this EPS bearer context. When the EPS bearer identity of a QoS flow is deleted, the associated EPS bearer context information that are mapped from the deleted EPS bearer identity shall be deleted from the UE and the network if there is no other existing QoS flow associated with this EPS bearer context. When an EPS bearer is released, all the associated QoS flow descriptions and QoS rules that are mapped from the released EPS bearer shall be deleted from the UE and the network.

NOTE 10:If T3584 is running or deactivated for the S-NSSAI and optionally the DNN combination, the UE is allowed to initate ESM procedures in EPS with or without APN corresponding to that DNN, and if the APN is congested in EPS, the MME can send a back-off timer for the APN to the UE as specified in 3GPP TS 24.301 [15].

Upon inter-system change from N1 mode to S1 mode, if the UE has any PDU sessions associated with one or more multicast MBS multicast sessions, the UE shall locally leave the associated multicast MBS sessions and the network shall consider the UE as removed from the associated multicast MBS sessions.

For the case of handover of an existing PDU session from 3GPP access to non-3GPP access,

- upon receipt of the PDU SESSION ESTABLISHMENT ACCEPT message, the UE locally deletes the EPS bearer identities for the PDU session, if any (see subclause 6.4.1.3); and
- after successful handover, the network shall locally delete the EPS bearer identities for the PDU session, if any.

6.1.4.2 Coordination between 5GSM and ESM without N26 interface

When the network does not support N26 interface, the SMF does not provide the UE with the mapped EPS bearer context for a PDU session.

- NOTE 1: Since the SMF does not provide the UE with the mapped EPS bearer context for a PDU session, the UE does not know whether interworking with EPS is supported for a PDU session before attempting to transfer the PDU session from N1 mode to S1 mode.
- NOTE 2: It is up to UE implementation to decide which PDU session(s) to be attempted to transfer from N1 mode to S1 mode, e.g. based on UE policy or UE local configuration.

Upon inter-system change from N1 mode to S1 mode in EMM-IDLE mode, the UE shall not transfer a PDU session for LADN to EPS.

Upon inter-system change from N1 mode to S1 mode in EMM-IDLE mode, the UE shall not transfer a multi-homed IPv6 PDU session to EPS.

Upon inter-system change from N1 mode to S1 mode in EMM-IDLE mode, the UE shall use the parameters from each PDU session which the UE intends to transfer to EPS to create the contents of a PDN CONNECTIVITY REQUEST message as follows:

- a) if the PDU session is an emergency PDU session, the request type shall be set to "handover of emergency bearer services". Otherwise the request type shall be set to "handover";
- b) the PDU session type of the PDU session shall be mapped to the PDN type of the default EPS bearer context as follows:
 - 1) the PDN type shall be set to "non-IP" if the PDU session type is "Unstructured";
 - 2) the PDN type shall be set to "IPv4" if the PDU session type is "IPv4";
 - 3) the PDN type shall be set to "IPv6" if the PDU session type is "IPv6";
 - 4) the PDN type shall be set to "IPv4v6" if the PDU session type is "IPv4v6";
 - 5) the PDN type shall be set to "non-IP" if the PDU session type is "Ethernet" and the UE, the network or both of them do not support Ethernet PDN type in S1 mode; and
 - 6) the PDN type shall be set to "Ethernet" if the PDU session type is "Ethernet" and the UE and the network support Ethernet PDN type in S1 mode;
- c) the DNN of the PDU session shall be mapped to the APN of the default EPS bearer context, unless the PDU session is an emergency PDU session;
- d) the PDU session ID parameter in the Protocol configuration options IE or Extended protocol configuration options IE shall be set to the PDU session identity of the PDU session; and
- e) if the PDU session is an MA PDU session established over 3GPP access, the ATSSS request parameter shall be included in the Protocol configuration options IE or Extended protocol configuration options IE.

After inter-system change from N1 mode to S1 mode, the UE shall associate the PDU session identity with the default EPS bearer context. If the PDU session being transferred is a non-emergency PDU session, the UE shall in addition associate the S-NSSAI and the PLMN ID of the current PLMN with the default EPS bearer context.

Upon successful completion of an EPS attach procedure after inter-system change from N1 mode to S1 mode (see 3GPP TS 24.301 [15]), the UE shall delete any UE derived QoS rules except when the PDU session is an MA PDU session established over 3GPP access and non-3GPP access.

The UE shall perform a local release of the PDU session(s) and QoS flow(s) associated with the 3GPP access which have not been transferred to EPS. The UE shall also perform a local release of any QoS flow description not associated with any QoS rule.

For PDU session(s) associated with non-3GPP access in 5GS, if present, the UE may:

a) keep some or all of these PDU sessions still associated with non-3GPP access in 5GS, if supported;

3GPP TS 24.501 version 17.16.0 Release 17

476

- b) release some or all of these PDU sessions explicitly by initiating the UE requested PDU session release procedure(s); or
- c) attempt to transfer some or all of these PDU sessions from N1 mode to S1 mode by initiating the UE requested PDN connectivity procedure(s) with the PDN CONNECTIVITY REQUEST message created as above.

When the network does not support N26 interface, the MME does not provide the UE with the mapped PDU session for a PDN connection but provides the UE with an S-NSSAI if the PDN connection is not for emergency bearer services. When establishing a new PDN connection in S1 mode, to enable the UE to attempt to transfer the PDN connection from S1 mode to N1 mode in case of inter-system change, the UE shall allocate a PDU session identity, indicate the allocated PDU session identity in the PDU session ID parameter in the Protocol configuration options IE of the PDN CONNECTIVITY REQUEST message and associate the allocated PDU session identity with the default EPS bearer context of the PDN connection. If an N5CW device supports 3GPP access and establishes a new PDN connection in S1 mode, the N5CW device shall refrain from allocating "PDU session identity value 15". The network provides the UE with an S-NSSAI and the related PLMN ID in the Protocol configuration options IE or Extended protocol configuration options IE of the ACTIVATE DEFAULT EPS BEARER REQUEST message, the UE shall delete the stored S-NSSAI and the related PLMN ID, if any, and shall store the S-NSSAI and the related PLMN ID provided in the ACTIVATE DEFAULT EPS BEARER REQUEST message.

- NOTE 3: Since the MME does not provide the UE with the mapped PDU session for a PDN connection, the UE does not know whether interworking to 5GS is supported for a PDN connection for which the UE assigned a PDU Session identity before attempting to transfer the PDN connection from S1 mode to N1 mode.
- NOTE 4: It is up to UE implementation to decide which PDN connection(s) to be attempted to transfer from S1 mode to N1 mode, e.g. based on UE policy or UE local configuration.
- NOTE 5: If the PDN connection has been transferred from a PDN connection established via non-3GPP access to EPC, it is possible that the network provided the S-NSSAI already during the establishment via non-3GPP access (see 3GPP TS 24.302 [16]).

Upon inter-system change from S1 mode to N1 mode in 5GMM-IDLE mode, the UE uses the parameters from the default EPS bearer context of each PDN connection which the UE intends to transfer to 5GS and for which the UE has allocated a PDU session identity to create a PDU SESSION ESTABLISHMENT REQUEST message as follows:

- a) if the PDN connection is for emergency bearer services, the request type shall be set to "existing emergency PDU session". Otherwise the request type shall be set to:
 - 1) "MA PDU request", if the PDN connection to be transferred is a user-plane resource of an MA PDU session; or
 - 2) "existing PDU session";
- b) the PDN type of the default EPS bearer context shall be mapped to the PDU session type of the PDU session as follows:
 - 1) if the PDN type is "non-IP":
 - the PDU session type is set to the locally available information associated with the PDN connection (either "Ethernet" or "Unstructured"), if available; or
 - otherwise, the PDU session type is set to "Unstructured";
 - 2) if the PDN type is "IPv4" the PDU session type is set to "IPv4";
 - 3) if the PDN type is "IPv6", the PDU session type is set to "IPv6";
 - 4) if the PDN type is "IPv4v6", the PDU session type is set to "IPv4v6"; and
 - 5) if the PDN type is "Ethernet", the PDU session type is set to "Ethernet"; and
- c) the APN of the default EPS bearer context shall be mapped to the DNN of the PDU session, unless the PDN connection is an emergency PDN connection;

- d) the PDU session ID shall be set to the PDU session identity included by the UE in the Protocol configuration options IE or Extended protocol configuration options IE in the PDN CONNECTIVITY REQUEST message, or to the PDU session ID associated with the default EPS bearer context; and
- e) if the PDU session is not an emergency PDU session, the S-NSSAI of the PDU session shall be set to the S-NSSAI included by the network in the Protocol configuration options IE or Extended protocol configuration options IE in the ACTIVATE DEFAULT EPS BEARER REQUEST message, if provided by the network, or the S-NSSAI associated with the default EPS bearer context, if available.
- NOTE 6: If T3584 is running or deactivated for the S-NSSAI and optionally the DNN combination, the UE is allowed to initiate ESM procedures in EPS with or without APN corresponding to that DNN, and if the APN is congested in EPS, the MME can send a back-off timer for the APN to the UE as specified in 3GPP TS 24.301 [15].

The UE shall locally release the PDN connection(s) and EPS bearer(s) associated with the 3GPP access which have not been transferred to 5GS.

Upon inter-system change from N1 mode to S1 mode, if the UE has any PDU sessions associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions and the network shall consider the UE as removed from the associated multicast MBS sessions.

6.1.4a Coordination between 5GSM and SM

Coordination between 5GSM and SM states is not required.

After the 5G-SRVCC handover from NG-RAN to UTRAN (see 3GPP TS 23.216 [6A]), all the PDU sessions of the UE are locally released at the UE and the network. If the UE has any PDU sessions associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions and the network shall consider the UE as removed from the associated multicast MBS sessions.

6.1.5 Coordination for interworking with ePDG connected to EPC

When the UE establishes a new PDN connection via an ePDG connected to EPC, to enable the transfer of the PDN connection to N1 mode in case of inter-system change, the UE allocates a PDU session identity and indicates its value in the PDU session ID field in the N1_MODE_CAPABILITY Notify payload of the IKE_AUTH request message (see 3GPP TS 24.302 [16]). The network provides the UE with an S-NSSAI in the N1_MODE_INFORMATION Notify payload of the IKE_AUTH response message (see 3GPP TS 24.302 [16]).

Upon inter-system change to N1 mode, for PDN connection(s) established via an ePDG connected to EPC, if present, the UE may:

- a) keep some or all of these PDN connections still via ePDG connected to EPC, if supported;
- b) release some or all of these PDN connections explicitly by initiating the UE initiated tunnel disconnection procedure(s) as specified in 3GPP TS 24.302 [16]; or
- c) attempt to transfer some or all of these PDN connections to N1 mode using the parameters of the PDN connection for which the UE has allocated a PDU session identity by initiating the PDU session establishment procedure(s) with the PDU SESSION ESTABLISHMENT REQUEST message created. In that case, for each and every PDN connection to be transferred:
 - 1) if the PDN connection is for emergency bearer services, the request type shall be set to "existing emergency PDU session". Otherwise the request type shall be set to "existing PDU session";
 - if the previously allocated home address information for a PDN connection consists of an IPv4 address only for an ePDG connected to EPC according to 3GPP TS 24.302 [16], the PDU session type shall be set to "IPv4";
 - if the previously allocated home address information for a PDN connection consists of an IPv6 prefix only for an ePDG connected to EPC according to 3GPP TS 24.302 [16], the PDU session type shall be set to "IPv6";

- 4) if the previously allocated home address information for a PDN connection consists of both an IPv4 address and an IPv6 prefix for an ePDG connected to EPC according to 3GPP TS 24.302 [16], the PDU session type shall be set to "IPv4v6";
- 5) the APN of the PDN connection shall be mapped to the DNN of the PDU session;
- 6) the PDU session ID shall be set to the PDU session identity in the N1_MODE_CAPABILITY Notify payload of the IKE_AUTH request message establishing IPsec tunnel of the PDN connection; and
- 7) if the PDN connection is not for emergency bearer services, the S-NSSAI of the PDU session shall be set to the S-NSSAI associated with the PDN connection as specified in 3GPP TS 24.302 [16]. The UE shall not request to perform handover of an existing PDN connection to N1 mode if the associated S-NSSAI is not included in the allowed NSSAI for the target access.

If an existing PDU session is transferred from 3GPP access to an ePDG connected to EPC connection and that existing PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions and the SMF shall consider the UE as removed from the associated multicast MBS sessions.

6.2 General on elementary 5GSM procedures

6.2.1 Principles of PTI handling for 5GSM procedures

When the UE or the network initiates a transaction related procedure (i.e. a procedure consisting of more than one message and the messages are related), it shall include a valid PTI value in the message header of the request message or of the command message.

If a response message is sent as result of a received request message or a received command message, the sending entity shall include in the response message the PTI value received within the request message or within the command message (see examples in figure 6.2.1.1, figure 6.2.1.2, and figure 6.2.1.3).

If a command message is sent as result of a received request message, the sending entity shall include in the command message the PTI value received with the request message (see examples in figure 6.2.1.3).

If a command message is not sent as result of a received request message, the sending entity shall include in the command message the PTI value set to "no procedure transaction identity assigned" (see examples in figure 6.2.1.4).



Figure 6.2.1.1: UE-requested transaction related procedure accepted by the network





transaction related command (PTI = unassigned)

transaction related response (PTI = unassigned)

6.2.2 PDU session types

The following PDU Session types are supported:

- a) IPv4;
- b) IPv6;

- c) IPv4v6;
- d) Ethernet (EtherType as defined in IEEE Std 802.3 [31A]); and
- e) Unstructured.

IP address allocation for IPv4, IPv6 and IPv4v6 PDU session types is described in subclause 6.2.4.

Neither a MAC nor an IP address is allocated by the 5GCN to the UE for Ethernet PDU session type.

6.2.3 PDU session management

The SMF is responsible for the session management functions to provide the PDU connectivity service to the UE via the 5GSM signalling between UE and SMF. The session management procedures includes:

- a) the UE-requested PDU session establishment procedure;
- b) the PDU session authentication and authorization procedure;
- c) the UE-requested PDU session modification procedure;
- d) the network-requested PDU session modification procedure;
- e) the UE-requested PDU session release procedure; and
- f) the network-requested PDU session release procedure.

A UE may establish multiple PDU sessions, to the same data network or to different data networks, via 3GPP and via and Non-3GPP access networks at the same time.

The session management messages between UE and SMF are transferred via AMF as specified in subclause 8.3.

6.2.4 IP address allocation

6.2.4.1 General

This clause specifies IP address allocation for the PDU session.

In this release of specification, PDU session can be initiated with one IP version, i.e. IPv4 PDU session type or IPv6 PDU session type, or with both IP versions, i.e. IPv4v6 PDU session type.

IP address allocation to the UE shall be performed by SMF based on one or both the selected IP versions and operator policies. If IPv4 PDU session type is selected, an IPv4 address is allocated to the UE. If IPv6 PDU session type is selected, an IPv6 prefix except when the SMF acts according to subclause 6.2.4.3, and an interface identifier for the IPv6 link local address are allocated to the UE. If IPv4v6 PDU session type is selected, an IPv4 address, an IPv6 prefix except when the SMF acts according to subclause 6.2.4.4, and an interface identifier for the IPv6 link local address are allocated to the UE. If IPv4v6 PDU session type is selected in a PDv4 address, an IPv6 prefix except when the SMF acts according to rIPv4v6 PDU session type is selected in a PDU session established by the W-AGF acting on behalf of the FN-RG and the PDU SESSION ESTABLISHMENT REQUEST message contains the Suggested interface identifier IE, the SMF shall allocate to the UE the interface identifier for the IPv6 link local address indicated in the Suggested interface identifier IE.

For IPv4 PDU session type and for IPv4v6 PDU session type, the UE:

- a) shall obtain an IPv4 address via:
 - 1) NAS signalling as specified in subclause 6.2.4.2; or
 - 2) DHCPv4; and
- b) may obtain IPv4 configuration parameters (e.g. DNS server address) via DHCPv4 or may receive IPv4 configuration parameters (e.g. DNS server address) as specified in subclause 6.4.1 and subclause 6.3.2.

For IPv6 PDU session type and for IPv4v6 PDU session type, the UE:

a) shall build an IPv6 link local address based on the allocated interface identifier for the IPv6 link local address;

- b) shall obtain /64 IPv6 prefix via IPv6 stateless address autoconfiguration as specified in 3GPP TS 23.501 [8] and IETF RFC 4862 [39], except when the 5G-RG or the W-AGF act according to subclause 6.2.4.3; and
- c) may obtain IPv6 configuration parameters via stateless DHCPv6 as specified in IETF RFC 8415 [33D], except when the 5G-RG or the W-AGF act according to subclause 6.2.4.3, may receive IPv6 configuration parameters (e.g. DNS server address) as specified in subclause 6.4.1 and subclause 6.3.2, or may receive DNS server IPv6 addresses in a Router Advertisement Message as specified in IETF RFC 4861 [38B] with recursive DNS server option as specified in IETF RFC 8106 [52].

6.2.4.2 IP address allocation via NAS signalling

The UE shall set the PDU session type IE in the PDU SESSION ESTABLISHMENT REQUEST message, based on its IP stack capabilities if the UE requests IP connectivity as follows:

- a) A UE:
 - 1) which is IPv6 and IPv4 capable, shall set the PDU session type IE to IPv4, IPv6 or IPv4v6 according to UE configuration or received policy.
 - 2) which is only IPv6 capable, shall set the PDU session type IE to IPv6.
 - 3) which is only IPv4 capable, shall set the PDU session type IE to IPv4.
- b) When the IP version capability of the UE is unknown in the UE (as in the case when the MT and TE are separated and the capability of the TE is not known in the MT), the UE shall set the PDU session type IE to IPv4v6.

If the UE wants to use DHCPv4 for IPv4 address assignment, it shall indicate that to the network within the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST.

On receipt of the PDU SESSION ESTABLISHMENT REQUEST message sent by the UE, the network when allocating an IP address shall take into account the PDU session type IE, the operator's policies of the network, and the user's subscription data and:

- a) if the network sets the Selected PDU session type IE to IPv4, the network shall include an IPv4 address in the PDU address IE;
- b) if the network sets the Selected PDU session type IE to IPv6, the network shall include an interface identifier for the IPv6 link local address in the PDU address IE; and
- c) if the network sets the Selected PDU session type IE to IPv4v6, the network shall include an IPv4 address and an interface identifier for the IPv6 link local address in the PDU address IE.

6.2.4.3 Additional RG related requirements for IP address allocation

If IPv6 PDU session type or IPv4v6 PDU session type is selected, an IPv6 address, one or more IPv6 prefixes or both are allocated to the 5G-RG or the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device).

If the 5G-RG or the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) receives a Router Advertisement Message as specified in IETF RFC 4861 [38B] with the "Managed address configuration" flag set to zero, the 5G-RG and the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device):

- a) shall obtain /64 IPv6 prefix via IPv6 stateless address autoconfiguration as specified in 3GPP TS 23.501 [8] and IETF RFC 4862 [39];
- b) may obtain IPv6 configuration parameters via stateless DHCPv6 as specified in IETF RFC 8415 [33D]; and
- c) may request additional IPv6 prefixes using DHCPv6. If the 5G-RG and the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) request IPv6 prefixes using DHCPv6, the 5G-RG and the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) shall act as a "Requesting Router" as described in IETF RFC 3633 [33C], shall obtain IPv6 prefixes using the DHCPv6 Identity association for prefix delegation option as specified in IETF RFC 8415 [33D], may include DHCPv6 Rapid commit option as specified in IETF RFC 8415 [33D] in a DHCP message, and may include DHCPv6 OPTION_ORO option with the OPTION_PD_EXCLUDE option code as specified in IETF RFC 6603 [40A] in the DHCP message.

3GPP TS 24.501 version 17.16.0 Release 17

NOTE 1: The 5G-RG and the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) can include several DHCP options in a DHCP message.

If the 5G-RG or the W-AGF acting on behalf of the FN-RG receives a Router Advertisement Message as specified in IETF RFC 4861 [38B] with the "Managed address configuration" flag set to one, the 5G-RG and the W-AGF acting on behalf of the FN-RG:

- a) shall obtain an IPv6 address via DHCPv6 and the DHCPv6 Identity association for non-temporary addresses option as specified in IETF RFC 8415 [33D];
- b) may obtain IPv6 configuration parameters via DHCPv6 as specified in IETF RFC 8415 [33D]; and
- c) may request IPv6 prefixes using DHCPv6. If the 5G-RG and the W-AGF acting on behalf of the FN-RG requests IPv6 prefixes using DHCPv6, the 5G-RG and the W-AGF acting on behalf of the FN-RG shall act as a "Requesting Router" as described in IETF RFC 8415 [33D], shall obtain IPv6 prefixes using the DHCPv6 Identity association for prefix delegation option as specified in IETF RFC 8415 [33D], may include DHCPv6 Rapid commit option as specified in IETF RFC 8415 [33D] in a DHCP message, and may include DHCPv6 OPTION_ORO option with the OPTION_PD_EXCLUDE option code as specified in IETF RFC 6603 [40A] in the DHCP message.
- NOTE 2: The 5G-RG and the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) can include several DHCP options in a DHCP message.

The 5G-RG may obtain ACS information via DHCP as specified in clause 3.1 of BBF TR-069 [49] or in BBF TR-369 [50] R-DIS.1 and R-DIS.2.

6.2.4.4 Additional requirements of the UE acting as 5G ProSe layer-3 UE-to-network relay UE for IP address allocation

If a UE acting as 5G ProSe layer-3 UE-to-network relay needs to indicate "IPv6 Router" or "DHCPv4 Server & IPv6 Router" in the IP address configuration IE as specified in 3GPP 24.554 [19E], the UE shall support acting as a "Requesting Router" as described in IETF RFC 8415 [33D] to request additional IPv6 prefixes (i.e. prefixes in addition to the /64 default prefix which was allocated via stateless IPv6 address autoconfiguration as specified in subclause 6.2.4.1) from the SMF as specified in subclause 5.5.2 of 3GPP TS 23.304 [6E].

When the UE acting as 5G ProSe layer-3 UE-to-network relay UE requests additional prefixes using DHCPv6, the UE:

- a) shall include the OPTION_ORO option with the OPTION_PD_EXCLUDE option code as specified in IETF RFC 6603 [40A] in the DHCP message; and
- b) may include the Rapid Commit Option as specified in IETF RFC 8415 [33D] in the DHCP message.

6.2.5 Quality of service

- 6.2.5.1 General
- 6.2.5.1.1 QoS rules
 - 6.2.5.1.1.1 General

In a PDU session of IPv4, IPv6, IPv4v6 and Ethernet PDU session type, the NAS protocol enables different forwarding treatments of UL user data packets in one or more QoS flows based on signalled QoS rules, derived QoS rules or any combination of them.

In a PDU session of Unstructured PDU session type, all UL user data packets are associated with the same QoS flow.

6.2.5.1.1.2 Signalled QoS rules

The NAS protocol enables the network to provide the UE with signalled QoS rules associated with a PDU session.

The network can provide the UE with one or more signalled QoS rules associated with a PDU session at the PDU session establishment or at the PDU session modification.

Each signalled QoS rule contains:

- a) an indication of whether the QoS rule is the default QoS rule;
- b) a QoS rule identifier (QRI);
- c) a QoS flow identifier (QFI);
- d) optionally, a set of packet filters; and
- e) a precedence value.

NOTE 1: The default QoS rule indication (DQR) of a signalled QoS rule cannot be changed.

For case d) above:

- If the QoS rule is the default QoS rule of a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, the set of packet filters contains zero or more packet filters for DL direction, and may additionally contain one of the following:
 - A) a match-all packet filter for UL direction;
 - B) a match-all packet filter for UL and DL directions;
 - C) zero or more packet filters for UL direction (other than the match-all packet filter for UL direction);
 - D) zero or more packet filters for UL and DL directions (other than the match-all packet filter for UL and DL directions); or
 - E) one or more packet filters for UL direction (other than the match-all packet filter for UL direction) and one or more packet filters for UL and DL directions (other than the match-all packet filter for UL and DL directions).

The set of packet filters for the default QoS rule shall not be empty. If the default QoS rule contains a matchall packet filter, then the highest precedence value shall be used for the default QoS rule.

- 2) If the QoS rule is a QoS rule of a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type and is not the default QoS rule, the set of packet filters contains zero or more packet filters for the DL direction, and may additionally contain one of the following:
 - A) zero or more packet filters for UL direction (other than the match-all packet filter for UL direction); and
 - B) zero or more packet filters for both UL and DL directions (other than the match-all packet filter for UL and DL directions).

The set of packet filters for a QoS rule which is not the default QoS rule shall not be empty.

3) For PDU session of unstructured PDU session type, there is only one QoS rule associated with it and the set of packet filters of that QoS rule is empty.

If the UE requests a new QoS rule, it shall assign a precedence value for the signalled QoS rule which is not in the range from 70 to 99 (decimal).

- NOTE 2: In this release of the specification, there is no support for a match-all packet filter for DL direction.
- NOTE 3: In order to support QoS differentiation in case of access to PLMN services via an SNPN, the UE, within the SNPN, can construct packet filters based on the destination IP address to reach the N3IWF in the PLMN and the security parameters index (SPI) for the IPsec SA.
- NOTE 4: In order to support QoS differentiation in case of access to SNPN services via a PLMN, the UE, within the PLMN, can construct packet filters based on the destination IP address to reach the N3IWF in the SNPN and the security parameters index (SPI) for the IPsec SA.
- NOTE 5: The above described condition of assigning a precedence value for the signalled QoS rule is applied to the UE when the UE requests a QoS rule for network to bind service data flows described by the QoS rule to a dedicated QoS flow by setting the segregation bit to 1.

In NB-N1 mode, there is only one QoS rule associated with a PDU session and that is the default QoS rule. As described in 3GPP TS 23.501 [8], when the SMF determines that the UE has:

- a) moved from a tracking area in WB-N1 mode into a tracking area in NB-N1 mode;
- b) moved from a tracking area in WB-S1 mode into a tracking area in NB-N1 mode; or
- c) moved from a tracking area in NR connected to 5GCN into a tracking area in NB-N1 mode;

the SMF shall, for each PDU session that is kept active, initiate the PDU session modification procedure (see subclause 6.3.3.2) to delete every QoS rule that is not the default QoS rule, if any.

Within a PDU session:

- a) each signalled QoS rule has a unique QRI;
- b) there is at least one signalled QoS rule;
- c) one signalled QoS rule is the default QoS rule; and
- d) there can be zero, one or more signalled QoS rules associated with a given QFI.

6.2.5.1.1.3 Derived QoS rules

Derived QoS rules are applicable only for PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type.

The reflective QoS in the UE creates derived QoS rules associated with a PDU session based on DL user data packets received via the PDU session.

Each derived QoS rule contains:

- a) a QoS flow identifier (QFI);
- b) a packet filter for UL direction; and
- c) a precedence value of 80 (decimal).
- NOTE: On the network side, the corresponding QoS rule can be associated with a different precedence value in the range from 70 to 99 (decimal).

Within a PDU session:

- a) there can be zero, one or more derived QoS rules associated with a given QFI; and
- b) there can be up to one derived QoS rule associated with a given packet filter for UL direction.

In the UE, a timer T3583 runs for each derived QoS rule.

Reflective QoS is not supported in NB-N1 mode. Reflective QoS is not applicable for a PDU session with control plane only indication.

6.2.5.1.1.4 QoS flow descriptions

The network can also provide the UE with one or more QoS flow descriptions associated with a PDU session at the PDU session establishment or at the PDU session modification.

Each QoS flow description contains:

- a) a QoS flow identifier (QFI);
- b) if the flow is a GBR QoS flow:
 - 1) Guaranteed flow bit rate (GFBR) for UL;
 - 2) Guaranteed flow bit rate (GFBR) for DL;
 - 3) Maximum flow bit rate (MFBR) for UL;

- 4) Maximum flow bit rate (MFBR) for DL; and
- 5) optionally averaging window, applicable for both UL and DL;
- c) 5QI, if the QFI is not the same as the 5QI of the QoS flow identified by the QFI; and
- d) optionally, an EPS bearer identity (EBI) if the QoS flow can be mapped to an EPS bearer as specified in subclause 4.11.1 of 3GPP TS 23.502 [9].

If the averaging window is not included in a QoS flow description for a GBR QoS flow with a 5QI indicated in 3GPP TS 23.501 [8] table 5.7.4-1, the averaging window associated with the 5QI in 3GPP TS 23.501 [8] table 5.7.4-1 applies for the averaging window.

If the averaging window is not included in a QoS flow description for a GBR QoS flow with a 5QI not indicated in 3GPP TS 23.501 [8] table 5.7.4-1, the standardized value of two seconds is used as the averaging window.

6.2.5.1.2 Session-AMBR

The NAS protocol enables the network to provide the UE with the session-AMBR associated with a PDU session.

The standardized value of two seconds is used as the averaging window for the UE's enforcement of the UL rate limitation indicated by the session-AMBR.

6.2.5.1.2A Void

6.2.5.1.3 UL user data packet matching

For PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, upon receiving an UL user data packet from the upper layers for transmission via a PDU session, the UE shall attempt to associate the UL user data packet with:

- a) the QFI of a signalled QoS rule associated with the PDU session which has a set of packet filters containing a packet filter for UL direction matching the UL user data packet or containing a packet filter for both UL and DL directions matching the UL user data packet; or
- b) the QFI of a derived QoS rule associated with the PDU session which has the packet filter for UL direction matching the UL user data packet;

by evaluating the QoS rules in increasing order of their precedence values until the UL user data packet is associated with a QFI or all QoS rules are evaluated.

For PDU session of unstructured PDU session type, upon receiving an UL user data packet from the upper layers for transmission via a PDU session, the UE shall associate the UL user data packet with the QFI of the default QoS rule associated with the PDU session.

If the UL user data packet is associated with a QFI, the UE shall pass the QFI along the UL user data packet to the lower layers for transmission.

NOTE: Marking of the UL user data packet with the QFI is performed by the lower layers.

If all QoS rules are evaluated and the UL user data packet is not associated with a QFI, the UE shall discard the UL user data packet.

6.2.5.1.4 Reflective QoS

6.2.5.1.4.1 General

The UE may support reflective QoS.

If the UE supports the reflective QoS, the UE shall support the procedures in the following subclauses.

The reflective QoS is applicable in a PDU session of IPv4, IPv6, IPv4v6 and Ethernet PDU session type. The reflective QoS is not applicable in a PDU session of Unstructured PDU session type. Reflective QoS is not applicable for a PDU session with control plane only indication.

The UE may request to revoke the usage of reflective QoS for an existing PDU session for which the UE had previously indicated support for reflective QoS.

6.2.5.1.4.2 Derivation of packet filter for UL direction from DL user data packet

If the UE needs to derive a packet filter for UL direction from the DL user data packet (see subclause 6.2.5.1.4.3 and 6.2.5.1.4.4), the UE shall proceed as follows:

- a) if the received DL user data packet belongs to a PDU session of IPv4 or IPv4v6 PDU session type and is an IPv4 packet and:
 - 1) the protocol field of the received DL user data packet indicates TCP as specified in IETF RFC 793 [33];
 - 2) the protocol field of the received DL user data packet indicates UDP as specified in IETF RFC 768 [32]; or
 - the protocol field of the received DL user data packet indicates ESP as specified in IETF RFC 4303 [38] and an uplink IPSec SA corresponding to a downlink IPSec SA indicated in the security parameters index field of the received DL user data packet exists;

then the packet filter for UL direction contains the following packet filter components:

- an IPv4 remote address component set to the value of the source address field of the received DL user data packet;
- an IPv4 local address component set to the value of the destination address field of the received DL user data packet;
- a protocol identifier/next header type component set to the value of the protocol field of the received DL user data packet;
- 4) if the protocol field of the received DL user data packet indicates TCP as specified in IETF RFC 793 [33] or UDP as specified in IETF RFC 768 [32]:
 - i) a single local port type component set to the value of the destination port field of the received DL user data packet; and
 - ii) a single remote port type component set to the value of the source port field of the received DL user data packet;
- 5) if the protocol field of the received DL user data packet indicates ESP as specified in IETF RFC 4303 [38], an uplink IPSec SA corresponding to a downlink IPSec SA of the SPI in the DL user data packet exists and the SPI of the uplink IPSec SA is known to the NAS layer:
 - i) a security parameter index type component set to the security parameters index of the uplink IPSec SA corresponding to the downlink IPSec SA indicated in the security parameters index field of the received DL user data packet; and
- 6) if the protocol field of the received DL user data packet indicates UDP and the received DL user data packet contains a UDP-encapsulated ESP header as specified in IETF RFC 3948 [55], an uplink IPSec SA corresponding to a downlink IPSec SA of the SPI in the DL user data packet exists and the SPI of the uplink IPSec SA is known to the NAS layer:
 - i) a security parameter index type component set to the security parameters index of the uplink IPSec SA corresponding to the downlink IPSec SA indicated in the security parameters index field of the ESP header field of the UDP-encapsulated ESP header as specified in IETF RFC 3948 [55] of the received DL user data packet;

otherwise it is not possible to derive a packet filter for UL direction from the DL user data packet;

- b) if the received DL user data packet belongs to a PDU session of IPv6 or IPv4v6 PDU session type and is an IPv6 packet and:
 - the last next header field of the received DL user data packet indicates TCP as specified in IETF RFC 793 [33];

- 2) the last next header field of the received DL user data packet indicates UDP as specified in IETF RFC 768 [32]; or
- the last next header field of the received DL user data packet indicates ESP as specified in IETF RFC 4303 [38] and an uplink IPSec SA corresponding to a downlink IPSec SA indicated in the security parameters index field of the received DL user data packet exists;

then the packet filter for UL direction contains the following packet filter components:

- 1) an IPv6 remote address/prefix length component set to the value of the source address field of the received DL user data packet;
- 2) an IPv6 local address/prefix length component set to the value of the destination address field of the received DL user data packet;
- 3) a protocol identifier/next header type component set to the value of the last next header field of the received DL user data packet;
- 4) if the last next header field of the received DL user data packet indicates TCP as specified in IETF RFC 793 [33] or UDP as specified in IETF RFC 768 [32]:
 - i) a single local port type component set to the value of the destination port field of the received DL user data packet; and
 - ii) a single remote port type component set to the value of the source port field of the received DL user data packet;
- 5) if the last next header field of the received DL user data packet indicates ESP as specified in IETF RFC 4303 [38], an uplink IPSec SA corresponding to a downlink IPSec SA of the SPI in the DL user data packet exists and the SPI of the uplink IPSec SA is known to the NAS layer:
 - i) a security parameter index type component set to the security parameters index of the uplink IPSec SA corresponding to the downlink IPSec SA indicated in the security parameters index field of the received DL user data packet; and
- 6) if the protocol field of the received DL user data packet indicates UDP, and the received DL user data packet contains a UDP-encapsulated ESP header as specified in IETF RFC 3948 [55], an uplink IPSec SA corresponding to a downlink IPSec SA of the SPI in the DL user data packet exists and the SPI of the uplink IPSec SA is known to the NAS layer:
 - a security parameter index type component set to the security parameters index of the uplink IPSec SA corresponding to the downlink IPSec SA indicated in the security parameters index field of the ESP header field of the UDP-encapsulated ESP header as specified in IETF RFC 3948 [55] of the received DL user data packet;

otherwise it is not possible to derive a packet filter for UL direction from the DL user data packet;

- c) if the received DL user data packet belongs to a PDU session of Ethernet PDU session type, the packet filter for UL direction contains the following packet filter components:
 - 1) a destination MAC address component set to the source MAC address of the received DL user data packet;
 - 2) a source MAC address component set to the destination MAC address of the received DL user data packet;
 - if one or more 802.1Q C-TAG is included in the received DL user data packet, an 802.1Q C-TAG VID component set to the outermost 802.1Q C-TAG VID of the received DL user data packet and an 802.1Q C-TAG PCP/DEI component set to the outermost 802.1Q C-TAG PCP/DEI of the received DL user data packet;
 - 4) if one or more 802.1Q S-TAG is included in the received DL user data packet, an 802.1Q S-TAG VID component set to the outermost 802.1Q S-TAG VID of the received DL user data packet and an 802.1Q S-TAG PCP/DEI component set to the outermost 802.1Q S-TAG PCP/DEI of the received DL user data packet;
 - 5) If the Ethertype field of the received DL user data packet is set to a value of 1536 or above, an Ethertype component set to the Ethertype of the received DL user data packet;

- 6) if the Ethertype field of the Ethernet frame header indicates that the data carried in the Ethernet frame is IPv4 data, the UE shall also add to the packet filter for UL direction the IP-specific components based on the contents of the IP header of the received DL user data packet as described in bullet a) above; and
- 7) if the Ethertype field of the Ethernet frame header indicates that the data carried in the Ethernet frame is IPv6 data, the UE shall also add to the packet filter for UL direction the IP-specific components based on the contents of the IP header of the received DL user data packet as described in bullet b) above; and
- d) if the received DL user data packet belongs to a PDU session of PDU session type other than Ethernet, IPv4, IPv6 and IPv4v6, it is not possible to derive a packet filter for UL direction from the DL user data packet.

6.2.5.1.4.3 Creating a derived QoS rule by reflective QoS in the UE

If the UE receives a DL user data packet marked with a QFI and an RQI, the DL user data packet belongs to a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the UE does not have a derived QoS rule with the same packet filter for UL direction as the packet filter for UL direction derived from the DL user data packet as specified in subclause 6.2.5.1.4.2, then the UE shall create a new derived QoS rule as follows:

- a) the QFI of the derived QoS rule is set to the received QFI;
- b) the precedence value of the derived QoS rule is set to 80 (decimal); and
- c) the packet filter for UL direction of the derived QoS rule is set to the derived packet filter for UL direction;

and the UE shall start the timer T3583 associated with the derived QoS rule with the RQ timer value last received during the UE-requested PDU session establishment procedure of the PDU session (see subclause 6.4.1) or the network-requested PDU session modification procedure of the PDU session (see subclause 6.4.2). If the RQ timer value was received neither in the UE-requested PDU session establishment procedure of the PDU session nor in any network-requested PDU session modification procedure of the PDU session, the default standardized RQ timer value is used.

6.2.5.1.4.4 Updating a derived QoS rule by reflective QoS in the UE

If the UE receives a DL user data packet associated with a QFI and an RQI, the DL user data packet belongs to a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the UE has a derived QoS rule with the same packet filter for UL direction as the packet filter for UL direction derived from the DL user data packet as specified in subclause 6.2.5.1.4.2:

- a) the UE shall re-start the timer T3583 associated with the derived QoS rule with the RQ timer value last received during the UE-requested PDU session establishment procedure of the PDU session (see subclause 6.4.1) or the network-requested PDU session modification procedure of the PDU session (see subclause 6.4.2). If the RQ timer value was received neither in the UE-requested PDU session establishment procedure of the PDU session nor in any network-requested PDU session modification procedure of the PDU session, the default standardized RQ timer value is used; and
- b) if the QFI value associated with the DL user data packet is different from the QFI value stored for the derived QoS rule, the UE shall replace the QFI value stored for the derived QoS rule with the new QFI value for the derived QoS rule.

6.2.5.1.4.5 Deleting a derived QoS rule in the UE

Upon expiry of timer T3583 associated with a derived QoS rule, the UE shall remove the derived QoS rule.

Upon release of the PDU session, the UE shall remove the derived QoS rule(s) associated with the PDU session.

If the network accepts the request from the UE to revoke the usage of reflective QoS and sets the value of the RQ timer to "deactivated" or zero, the UE shall remove the derived QoS rule(s) associated with the PDU session.

Upon inter-system mobility from WB-N1 mode to NB-N1 mode or from NR connected to 5GCN to NB-N1 mode, the UE shall remove the derived QoS rule(s) associated with the PDU session that is kept active.

When a derived QoS rule is deleted, the timer T3583 associated with the derived QoS rule shall be stopped.

6.2.5.1.4.6 Ignoring RQI in the UE

If the UE receives a DL user data packet marked with a QFI and an RQI and it is not possible to derive a packet filter for UL direction from the DL user data packet as specified in subclause 6.2.5.1.4.2, the UE shall ignore the RQI and shall handle the received DL user data packet.

6.2.5.2 QoS in MA PDU session

In an MA PDU session, unless it :

- a) is established over non-3GPP access; and
- b) has a PDN connection as a user-plane resource;

the UE shall have one set of QoS rules, one set of QoS flow descriptions and one session-AMBR. The network can provide the set of QoS rules, the set of QoS flow descriptions and the session-AMBR of the MA PDU session via either access of the MA PDU session. In an MA PDU session, the UE shall support:-

- modification or deletion via an access of a QoS rule or a QoS flow description; and
- modification via an access of the session-AMBR;

of the MA PDU session created via the same or the other access.

In an MA PDU session:

- a) established over non-3GPP access; and
- b) with a PDN connection as a user-plane resource;

the UE shall have two sets of QoS rules, two sets of QoS flow descriptions and two session-AMBR values - one is maintained via non-3GPP access and the other is associated with EPS bearer contexts of the PDN connection and maintained via extended protocol configuration options IE parameters received via the PDN connection.

6.2.6 Local area data network (LADN)

The UE can receive the local area data network (LADN) information consisting of LADN DNNs and LADN service area information (a set of tracking areas that belong to the current registration area) during the registration procedure or the generic UE configuration update procedure (see subclause 5.5.1 and subclause 5.4.4).

If the UE is not operating in SNPN access operation mode, the UE considers the received LADN information to be valid only in the TAIs of the registered PLMN that are in the LADN service area information, and in the TAIs of the equivalent PLMNs if the LADN service area information includes TAIs for the equivalent PLMNs. When the AMF provides the UE with LADN service area information containing TAIs for the equivalent PLMNs, the AMF shall include these TAIs of the equivalent PLMNs in the UE's registration area.

If the UE is operating in SNPN access operation mode, the UE considers the received LADN information to be valid only in the TAIs of the registered SNPN that are in the LADN service area information.

The LADN DNN(s) received by the UE is also considered as LADN DNN(s) in the equivalent PLMNs.

The UE shall consider itself to be located inside the LADN service area based on the LADN service area information. If the UE does not have a LADN service area information for the LADN DNN, the UE shall consider itself to be located outside the LADN service area.

When the UE is located in the LADN service area and the UE is in substate 5GMM-REGISTERED.NORMAL-SERVICE, the UE may initiate:

- the UE-requested PDU session establishment procedure with a LADN DNN to establish a PDU session for LADN;
- the UE-requested PDU session modification procedure to modify the PDU session for LADN; and
- the service request procedure to re-establish the user-plane resources for the PDU session for LADN.

NOTE 1: If both the Service area list IE and the LADN information IE were received by the UE, the Service area list IE is evaluated first.

When the UE is located outside the LADN service area, the UE is allowed:

- to initiate the UE-requested PDU session release procedure to release a PDU session for LADN; or
- to initiate the UE-requested PDU session modification procedure to indicate a change of 3GPP PS data off UE status.

If the UE has moved out of the LADN service area, the SMF shall:

- a) release the PDU session for LADN; or
- b) release the user-plane resources for the PDU session for LADN and maintain the PDU session for LADN;

according to operator's policy.

In case b):

- if the UE has returned to the LADN service area, and the network has downlink user data pending, the network re-establishes the user-plane resources for the PDU session for LADN; and.
- if the UE has not returned to the LADN service area after a period of time according to operator's policy, the SMF may release the PDU session for LADN.

When the UE moves to 5GMM-DEREGISTERED state, the UE shall delete the stored LADN information, if any.

NOTE 2: In this release, LADNs apply only to 3GPP access.

Upon inter-system change from N1 mode to S1 mode in EMM-IDLE mode, the UE shall not transfer a PDU session for LADN to EPS.

6.2.7 Handling of DNN based congestion control

The network may detect and start performing DNN based congestion control when one or more DNN congestion criteria as specified in 3GPP TS 23.501 [8] are met. If the UE does not provide a DNN for a non-emergency PDU session, then the network uses the selected DNN.

In the UE, 5GS session management timers T3396 for DNN based congestion control are started and stopped on a per DNN basis except for an LADN DNN in case of PLMN. For an LADN DNN, 5GS session management timers T3396 for DNN based congestion control is applied to the registered PLMN and its equivalent PLMNs. In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, in the UE, 5GS session management timers T3396 for DNN based congestion control are started and stopped on a per DNN and SNPN basis. If the UE supports access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3396 for DNN based congestion control are started and stopped on a per DNN and SNPN basis. If the UE supports access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3396 for DNN based congestion control are started and stopped on a per DNN, SNPN and selected entry of the "list of subscriber data" or selected PLMN subscription basis. Upon receipt of a 5GMM message or 5GSM message from the network for which the UE needs to stop the running timers T3396 associated with an LADN DNN as specified in subclause 6.3.2.3, 6.3.3.3, 6.4.1.4.2 and 6.4.2.4.2, only the running timer T3396 which is associated with the PLMN and equivalent PLMNs where the timer was started is stopped.

The DNN associated with T3396 is the DNN provided by the UE during the PDU session establishment. If no DNN is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST, then T3396 is associated with no DNN. For this purpose, the UE shall memorize the DNN provided to the network during the PDU session establishment. The timer T3396 associated with no DNN will never be started due to any 5GSM procedure related to an emergency PDU session. If the timer T3396 associated with no DNN is running, it does not affect the ability of the UE to request an emergency PDU session.

If T3396 is running or is deactivated, then the UE is not allowed to initiate the:

- a) PDU session establishment procedure;
- b) PDU session modification procedure; or
- c) NAS transport procedure for sending CIoT user data;

for the respective DNN or without a DNN unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the network does not include timer T3396 with 5GSM cause #26 "insufficient resources", the UE may use a local back-off timer that has exponential value or a default value, which is provisioned using implementation specific means, to:

- a) prevent sending any 5GSM procedure for the respective DNN till its expiry; and
- b) allow sending any 5GSM procedure for the respective DNN after its expiry.

6.2.8 Handling of S-NSSAI based congestion control

The network may detect and start performing S-NSSAI based congestion control when one or more S-NSSAI congestion criteria as specified in 3GPP TS 23.501 [8] are met. If the UE does not provide a DNN for a non-emergency PDU session, then the network uses the selected DNN. If the UE does not provide an S-NSSAI for a non-emergency PDU session, then the network uses the selected S-NSSAI.

In case of PLMN, in the UE, 5GS session management timers T3584 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI, DNN and PLMN basis. If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the 5GSM message with the 5GSM cause value #67 "insufficient resources for specific slice and DNN", then the UE applies the timer T3584 for all the PLMNs. Otherwise, the UE applies the timer T3584 for the registered PLMN. If the timer T3584 applies for all the PLMNs, the timer T3584 starts when the UE is registered in a VPLMN and the S-NSSAI is provided by the UE during the PDU session establishment, the timer T3584 is associated with the [mapped S-NSSAI, DNN] combination of the PDU session.

In case of PLMN, in the UE, 5GS session management timers T3585 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI and PLMN basis. If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the 5GSM message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE applies the timer T3585 for all the PLMNs. Otherwise, the UE applies the timer T3585 for the registered PLMN. If the timer T3585 applies for all the PLMNs, the timer T3585 starts when the UE is registered in a VPLMN and the S-NSSAI is provided by the UE during the PDU session establishment, the timer T3585 is associated with the mapped S-NSSAI of the PDU session. Additionally, if the 5GSM congestion re-attempt indicator IE with the CATBO bit set to "The back-off timer is applied in the current access type" is included in the 5GSM message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE applies the timer T3585 for the current access type. Otherwise, the UE applies the timer T3585 for both 3GPP access type and non-3GPP access type.

In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3584 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI, DNN and SNPN basis. If the UE supports access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3584 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI, DNN, SNPN and selected entry of the "list of subscriber data" or selected PLMN subscription basis.

In case of SNPN, if the UE does not support access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3585 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI and SNPN basis. If the UE supports access to an SNPN using credentials from a credentials holder, in the UE 5GS session management timers T3585 for the S-NSSAI based congestion control are started and stopped on a per S-NSSAI, SNPN and selected entry of the "list of subscriber data" or selected PLMN subscription basis.

The 5GSM congestion re-attempt indicator IE shall not be applicable in an SNPN.

If the timer T3584 or timer T3585 was provided during the PDU session establishment procedure, the S-NSSAI associated with T3584 or T3585, respectively is the S-NSSAI, including no S-NSSAI, provided by the UE during the PDU session establishment.

If the timer T3584 is provided during the PDU session modification or PDU session release procedure, the UE behaves as follows: The DNN associated with T3584 is the DNN provided by the UE during the PDU session establishment. If no S-NSSAI but DNN is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3584 is associated with no S-NSSAI and the DNN provided to the network during the PDU session establishment. If the PDN connection was established when in the S1 mode, then T3584 is associated with no S-NSSAI. If no DNN but S-NSSAI is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3584 is associated with no S-NSSAI is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3584 is associated with no S-NSSAI is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3584 is

associated with no DNN and the S-NSSAI of the PDU session. If no DNN and no S-NSSAI is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3584 is associated with no DNN and no S-NSSAI. For this purpose, the UE shall memorize the DNN and the S-NSSAI provided to the network during the PDU session establishment. The timer T3584 associated with no DNN and an S-NSSAI will never be started due to any 5GSM procedure related to an emergency PDU session. If the timer T3584 associated with no DNN and an S-NSSAI is running, it does not affect the ability of the UE to request an emergency PDU session.

If the timer T3585 was provided during the PDU session modification or PDU session release procedure, the UE behaves as follows: if an S-NSSAI was provided by the UE during the PDU session establishment, then T3585 is associated with the S-NSSAI of the PDU session. If no S-NSSAI is provided by the UE along the PDU SESSION ESTABLISHMENT REQUEST message, then T3585 is associated with no S-NSSAI. If the PDN connection was established when in the S1 mode, then T3585 is associated with no S-NSSAI.

If T3584 is running or is deactivated, then the UE is not allowed to initiate the:

- a) PDU session establishment procedure;
- b) PDU session modification procedure; or
- c) NAS transport procedure for sending CIoT user data;

for the respective [S-NSSAI, no DNN] or [S-NSSAI, DNN] combination unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the timer T3584 is running or is deactivated for all the PLMNs and is associated with an S-NSSAI other than no S-NSSAI, then

- a) the UE registered in the HPLMN is not allowed to initiate the:
 - 1) PDU session establishment procedure;
 - 2) PDU session modification procedure; or
 - 3) NAS transport procedure for sending CIoT user data;

when the [S-NSSAI, no DNN] or [S-NSSAI, DNN] combination provided by the UE during the PDU session establishment is the same as the [S-NSSAI, no DNN] or [S-NSSAI, DNN] combination associated with the timer T3584 unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status; and

- b) the UE registered in a VPLMN is not allowed to initiate the:
 - 1) PDU session establishment procedure;
 - 2) PDU session modification procedure; or
 - 3) NAS transport procedure for sending CIoT user data;

when the [mapped S-NSSAI, no DNN] or [mapped S-NSSAI, DNN] combination provided by the UE during the PDU session establishment is the same as the [S-NSSAI, no DNN] or [S-NSSAI, DNN] combination associated with the timer T3584 unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the timer T3584 is running or is deactivated for all the PLMNs and is associated with [no S-NSSAI, no DNN] or [no S-NSSAI, DNN] combination, then the UE is not allowed to initiate the:

- a) PDU session establishment procedure;
- b) PDU session modification procedure; or
- c) NAS transport procedure for sending CIoT user data;

for [no S-NSSAI, no DNN] or [no S-NSSAI, DNN] combination in any PLMN unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If T3585 is running or is deactivated, then the UE is neither allowed to initiate the PDU session establishment procedure nor the PDU session modification procedure for the respective S-NSSAI unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the timer T3585 is running or is deactivated for all the PLMNs and is associated with an S-NSSAI other than no S-NSSAI, then

- a) the UE registered in the HPLMN is not allowed to initiate the:
 - 1) PDU session establishment procedure;
 - 2) PDU session modification procedure; or
 - 3) NAS transport procedure for sending CIoT user data;

when the S-NSSAI provided by the UE during the PDU session establishment is the same as the S-NSSAI associated with timer T3585 unless the UE is a UE configured for high priority access in selected PLMNs or to report a change of 3GPP PS data off UE status; and

- b) the UE registered in a VPLMN is not allowed to initiate the:
 - 1) PDU session establishment procedure;
 - 2) PDU session modification procedure; or
 - 3) NAS transport procedure for sending CIoT user data;

when the mapped S-NSSAI provided by the UE during the PDU session establishment is the same as the S-NSSAI associated the timer T3585 unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the timer T3585 is running or is deactivated for all the PLMNs and is associated with no S-NSSAI, then the UE is not allowed to initiate the:

- a) PDU session establishment procedure;
- b) PDU session modification procedure;
- c) NAS transport procedure for sending CIoT user data;

for no S-NSSAI in any PLMN unless the UE is a UE configured for high priority access in selected PLMN or to report a change of 3GPP PS data off UE status.

If the network does not include timer T3584 with 5GSM cause #67 "insufficient resources for specific slice and DNN", the UE may use a local back-off timer that has exponential value or a default value, which is provisioned using implementation specific means, to:

- a) prevent sending any 5GSM procedure for the respective DNN and S-NSSAI till its expiry; and
- b) allow sending any 5GSM procedure for the respective DNN and S-NSSAI after its expiry.

If the network does not include timer T3585 with 5GSM cause #69 "insufficient resources for specific slice", the UE may use a local back-off timer that has exponential value or a default value, which is provisioned using implementation specific means, to:

- a) prevent sending any 5GSM procedure for the respective S-NSSAI till its expiry; and
- b) allow sending any 5GSM procedure for the respective S-NSSAI after its expiry.

6.2.9 Interaction with upper layers

6.2.9.1 General

A 5GSM entity may interact with upper layers. Subclause 6.2.9.2 describes how the 5GSM entity interacts with upper layers with respect to the URSP. Subclause 6.2.9.3 describes how the 5GSM entity interacts with upper layers with respect to the ProSeP.

6.2.9.2 URSP

The URSP requires interaction between upper layers and the 5GSM entities in the UE (see 3GPP TS 24.526 [19] for further details). Each of the 5GSM entities in the UE shall indicate attributes (e.g. PDU session identity, SSC mode, S-NSSAI, DNN, PDU session type, access type, PDU address) of a newly established PDU session to the upper layers. If a PDU session is released, the 5GSM entity handling the PDU session shall inform the PDU session identity of the released PDU session to the upper layers. The upper layers may request a 5GSM entity:

- a) to establish a PDU session indicating one or more PDU session attributes;
- b) to release an existing PDU session; or
- c) to establish a PDU session indicating one or more PDU session attributes, and to release an existing PDU session.

6.2.9.3 ProSeP

The ProSeP requires interaction between upper layers and the 5GSM entities in the UE acting as a 5G ProSe layer-3 UE-to-network relay UE (see 3GPP TS 24.554 [19E] for further details). The upper layers may request the 5GSM entity:

- a) to establish a PDU session indicating one or more PDU session attributes; or
- b) to release the existing PDU session; or
- c) to establish a PDU session indicating one or more PDU session attributes, and to release the existing PDU session.

Each of the 5GSM entities in the UE acting as a 5G ProSe layer-3 UE-to-network relay UE shall indicate attributes (e.g. PDU session identity, SSC mode, S-NSSAI, DNN, PDU session type, access type, PDU address) of the newly established PDU session to the upper layers. If the PDU session is released, the 5GSM entity handling the PDU session shall inform the PDU session identity of the released PDU session to the upper layers.

6.2.10 Handling of 3GPP PS data off

In case of PLMN, a UE, which supports 3GPP PS data off (see 3GPP TS 23.501 [8]), can be configured with up to two lists of 3GPP PS data off exempt services as specified in 3GPP TS 24.368 [17] or in the EF_{3GPPPSDATAOFF} USIM file as specified in 3GPP TS 31.102 [22]:

- a) a list of 3GPP PS data off exempt services to be used in the HPLMN or EHPLMN; and
- b) a list of 3GPP PS data off exempt services to be used in the VPLMN.

If only the list of 3GPP PS data off exempt services to be used in the HPLMN or EHPLMN is configured at the UE, this list shall be also used in the VPLMN.

In case of SNPN, a UE, which supports 3GPP PS data off (see 3GPP TS 23.501 [8]), can be configured with:

- a) up to two lists of 3GPP PS data off exempt services as specified in 3GPP TS 24.368 [17] for each subscribed SNPN whose entry exists in the "list of subscriber data":
 - 1) a list of 3GPP PS data off exempt services to be used in the subscribed SNPN; and
 - 2) a list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN; and

- b) one list of 3GPP PS data off exempt services as specified in 3GPP TS 24.368 [17] for PLMN subscription:
 - 1) a list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN.

If only the list of 3GPP PS data off exempt services to be used in the subscribed SNPN is configured for the selected entry of "list of subscriber data", this list shall be also used in the non-subscribed SNPN.

If the UE supports 3GPP PS data off, the UE shall provide the 3GPP PS data off UE status in the Extended protocol configuration options IE during UE-requested PDU session establishment procedure except for the transfer of a PDU session from non-3GPP access to 3GPP access and except for the establishment of user plane resources on the other access for the MA PDU session(see subclause 6.4.1), and during UE-requested PDU session modification procedure (see subclause 6.4.2), regardless of associated access type of the PDU session. If the UE requests a PDU session establishment procedure in order to transfer a PDU session from non-3GPP access to 3GPP access, or in order to establish user plane resources on the other access for the MA PDU session over 3GPP access or non-3GPP access, and:

- a) if the 3GPP PS data off UE status has changed since the last providing to the network, the UE shall provide the 3GPP PS data off UE status in the Extended protocol configuration options IE; or
- b) if the 3GPP PS data off UE status has not changed since the last providing to the network, the UE need not provide the 3GPP PS data off UE status.

The network shall support of 3GPP PS data off.

The UE shall indicate change of the 3GPP PS data off UE status for the PDU session by using the UE-requested PDU session modification procedure as specified in subclause 6.4.2.

When the 3GPP PS data off UE status is "activated":

- a) the UE does not send uplink IP packets via 3GPP access except:
 - 1) for those services indicated in the list of 3GPP PS data off exempt services to be used in the HPLMN or EHPLMN as specified in 3GPP TS 24.368 [17] when the UE is in its HPLMN or EHPLMN;
 - for those services indicated in the list of 3GPP PS data off exempt services to be used in the subscribed SNPN, configured for the selected entry of "list of subscriber data", as specified in 3GPP TS 24.368 [17] when the UE is in the subscribed SNPN;
 - 3) for those services indicated in the list of 3GPP PS data off exempt services to be used in the HPLMN or EHPLMN when the UE is in the VPLMN, if only the list of 3GPP PS data off exempt services to be used in the HPLMN or EHPLMN is configured to the UE as specified in 3GPP TS 24.368 [17];
 - 4) for those services indicated in the list of 3GPP PS data off exempt services to be used in the subscribed SNPN, configured for the selected entry of "list of subscriber data", when the UE is in a non-subscribed SNPN and only the list of 3GPP PS data off exempt services to be used in the subscribed SNPN is configured for the selected entry of "list of subscriber data" as specified in 3GPP TS 24.368 [17];
 - 5) for those services indicated in the list of 3GPP PS data off exempt services to be used in the VPLMN when the UE is in the VPLMN, if the list of 3GPP PS data off exempt services to be used in the VPLMN is configured to the UE as specified in 3GPP TS 24.368 [17];
 - 6) for those services indicated in the list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN, configured for the selected entry of "list of subscriber data", when the UE is in a non-subscribed SNPN and the list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN is configured for the selected entry of "list of subscriber data" as specified in 3GPP TS 24.368 [17];
 - 7) for those services indicated in the list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN, configured for the selected PLMN subscription, when the UE is in the non-subscribed SNPN and the list of 3GPP PS data off exempt services to be used in the non-subscribed SNPN is configured for the selected PLMN subscription as specified in 3GPP TS 24.368 [17];
 - 8) for those services indicated in the EF_{3GPPPSDATAOFF} USIM file as specified in 3GPP TS 31.102 [22];
 - 9) any uplink traffic due to procedures specified in 3GPP TS 24.229 [14]; and

10) any uplink traffic due to procedures specified in 3GPP TS 24.623 [20];

- b) the UE does not send uplink Ethernet user data packets via 3GPP access; and
- c) the UE does not send uplink Unstructured user data packets via 3GPP access.

Otherwise the UE sends uplink user data packets without restriction.

NOTE: If the UE supports 3GPP PS data off, uplink IP packets are filtered as specified in 3GPP TS 24.229 [14] in U.3.1.5.

3GPP PS data off does not restrict sending of uplink user data packets via non-3GPP access of a single access PDU session or an MA PDU session.

6.2.11 Multi-homed IPv6 PDU session

The UE supporting IPv6 may support multi-homed IPv6 PDU session.

If the UE supports the multi-homed IPv6 PDU session:

- a) the UE shall support acting as a type C host as specified in IETF RFC 4191 [36]; and
- b) the UE indicates support of the multi-homed IPv6 PDU session:
 - 1) during the UE-requested PDU session establishment of a PDU session of "IPv6" or "IPv4v6" PDU session type; and
 - 2) during the UE-requested PDU session modification performed after an inter-system change from S1 mode to N1 mode, for a PDU session associated with a PDN connection established when in S1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is of "IPv6" or "IPv4v6" PDU session type, and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication.

6.2.12 Handling of network rejection not due to congestion control

The network may include a back-off timer value in a 5GS session management reject message to regulate the time interval at which the UE may retry the same procedure for 5GSM cause values other than #26 "insufficient resources", #28 "unknown PDU session type", #39 "reactivation requested", #46 "out of LADN service area", #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #54 "PDU session does not exist", #57 "PDU session type IPv4v6 only allowed", #67 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode" and #69 "insufficient resources for specific slice". For 5GSM cause values other than #26 "insufficient resources", #28 "unknown PDU session type", #39 "reactivation requested", #46 "out of LADN service area", #54 "PDU session type Ethernet only allowed", #67 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode" and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice". and #86 "UAS services not allowed", the network may also include the re-attempt indicator to indicate whether the UE is allowed to re-attempt the corresponding session management procedure for the same DNN in S1 mode after inter-system change.

- NOTE 1: In a PLMN, if the network includes this back-off timer value for 5GSM cause values other than #27 "missing or unknown DNN", then the UE is blocked from sending another 5GSM request for the same procedure for the same [PLMN, DNN, S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, S-NSSAI], or [PLMN, no DNN, no S-NSSAI] combination for the specified duration. If the network includes this back-off timer value for 5GSM cause value #27 "missing or unknown DNN", then the UE is blocked from sending another 5GSM request for the same procedure for the same [PLMN, DNN], or [PLMN, no DNN] combination for the specified duration. In an SNPN, if the network includes this backoff timer value for 5GSM cause values other than #27 "missing or unknown DNN", then the UE is blocked from sending another 5GSM request for the same procedure for the same [SNPN, DNN, S-NSSAI], [SNPN, DNN, no S-NSSAI], [SNPN, no DNN, S-NSSAI], or [SNPN, no DNN, no S-NSSAI] combination for the specified duration if the UE does not support access to an SNPN using credentials from a credentials holder, and the UE is blocked from sending another 5GSM request for the same procedure for the same [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, S-NSSAI, [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, no S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, S-NSSAI, or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, no S-NSSAI] combination for the specified duration if the UE supports access to an SNPN using credentials from a credentials holder . If the network includes this back-off timer value for 5GSM cause value #27 "missing or unknown DNN", then the UE is blocked from sending another 5GSM request for the same procedure for the same [SNPN, DNN], or [SNPN, no DNN] combination for the specified duration if the UE does not support access to an SNPN using credentials from a credentials holder, and the UE is blocked from sending another 5GSM request for the same procedure for the same [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN] combination for the specified duration if the UE supports access to an SNPN using credentials from a credentials holder. Therefore, the operator needs to exercise caution in determining the use of this timer value.
- NOTE 2: If the re-attempt indicator is not provided by the network, a UE registered in its HPLMN or in an EHPLMN can use the configured SM_RetryAtRATChange value specified in the NAS configuration MO or in the USIM NAS_{CONFIG} file to derive the re-attempt indicator as specified in subclauses 6.4.1.4.3 and 6.4.2.4.3.

If re-attempt in S1 mode is allowed for 5GSM cause values other than #27 "missing or unknown DNN", the UE shall consider the back-off timer to be applicable only to the 5GS session management in N1 mode for the rejected 5GS session management procedure and the given [PLMN, DNN, S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, S-NSSAI], or [PLMN, no DNN, no S-NSSAI] combination. If re-attempt in S1 mode is allowed for 5GSM cause value #27 "missing or unknown DNN", the UE shall consider the back-off timer to be applicable only to the 5GS session management in N1 mode for the rejected 5GS session management procedure and the given [PLMN, DNN], or [PLMN, no DNN] combination. If re-attempt in S1 mode is not allowed, the UE shall consider the back-off timer to be applicable to both NAS protocols, i.e. applicable to the 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management in N1 mode for the rejected 5GS session management procedure and to the EPS session management in S1 mode for the corresponding session management procedure and the given [PLMN, DNN] or [PLMN, no DNN] combination.

NOTE 3: In the present subclause the terms DNN and APN are referring to the same parameter.

In a PLMN, if the back-off timer was provided during the PDU session establishment procedure, the UE behaves as follows: for 5GSM cause values other than #27 "missing or unknown DNN", when the UE is registered in a HPLMN, the DNN and the S-NSSAI of the [PLMN, DNN, S-NSSAI] combination associated with the back-off timer is the DNN and the S-NSSAI provided by the UE when the PDU session is established. When the UE is registered in a VPLMN, the DNN and the S-NSSAI of the [PLMN, DNN, S-NSSAI] combination associated with the back-off timer is the DNN and the S-NSSAI provided by the UE when the PDU session is established. For 5GSM cause value #27 "missing or unknown DNN", the DNN of the [PLMN, DNN] combination associated with the back-off timer is the DNN provided by the UE when the PDU session established. For 5GSM cause value #27 "missing or unknown DNN", the DNN of the [PLMN, DNN] combination associated with the back-off timer is the DNN provided by the UE when the PDU session established. If no DNN or no S-NSSAI was provided to the network during the PDU session establishment, then the back-off timer is associated with the [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination, dependent on which parameters were provided for 5GSM cause values other than #27 "missing or unknown DNN". If no DNN was provided to the network during the PDU session establishment, then the back-off timer is associated with the [PLMN, no DNN] combination for 5GSM cause value #27 "missing or unknown DNN". For this purpose, the UE shall memorize the DNN and the S-NSSAI provided to the network during the PDU session establishment.

In a PLMN, if the back-off timer was provided during the PDU session modification procedure, the UE behaves as follows: the DNN associated with the back-off timer is the DNN, including no DNN, provided by the UE when the

PDU session is established. If an S-NSSAI was provided by the UE during the PDU session establishment, when the UE is registered in a HPLMN, then the S-NSSAI associated with the back-off timer is the S-NSSAI of the PDU session. If an S-NSSAI was provided by the UE during the PDU session establishment, when the UE is registered in a VPLMN, then the S-NSSAI associated with the back-off timer is the mapped S-NSSAI of the PDU session. If no S-NSSAI was provided by the UE during the PDU session establishment, then the back-off timer is associated with no S-NSSAI. For this purpose, the UE shall memorize the DNN and the S-NSSAI provided to the network during the PDU session establishment.

In a PLMN, the back-off timer associated with the [PLMN, no DNN, no S-NSSAI], or [PLMN, no DNN] combination will never be started due to any 5GSM procedure related to an emergency PDU session. If the back-off timer associated with the [PLMN, no DNN, no S-NSSAI], or [PLMN, no DNN] combination is running, it does not affect the ability of the UE to request an emergency PDU session.

In an SNPN, the back-off timer associated with the [SNPN, no DNN, no S-NSSAI], or [SNPN, no DNN] combination if the UE does not support access to an SNPN using credentials from a credentials holder, or the back-off timer associated with the [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN, no S-NSSAI], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN] combination if the UE supports access to an SNPN using credentials from a credentials holder, will never be started due to any 5GSM procedure related to an emergency PDU session. If the back-off timer associated with the [SNPN, no DNN, no S-NSSAI], or [SNPN, no DNN] combination if the UE does not support access to an SNPN using credentials from a credentials holder, will never be started due to any 5GSM procedure related to an emergency PDU session. If the back-off timer associated with the [SNPN, no DNN, no S-NSSAI], or [SNPN, no DNN] combination if the UE does not support access to an SNPN using credentials from a credentials holder, or the back-off timer associated with the [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN, no S-NSSAI], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN, no S-NSSAI], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN, no S-NSSAI], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subscription, no DNN] combination if the UE supports access to an SNPN using credentials from a credentials holder is running, it does not affect the ability of the UE to request an emergency PDU session.

In a PLMN, the network may additionally indicate in the re-attempt indicator that a command to back-off is applicable not only for the PLMN in which the UE received the 5GS session management reject message, but for each PLMN included in the equivalent PLMN list at the time when the 5GS session management reject message was received.

In a PLMN, if the back-off timer is running or is deactivated for a given [PLMN, DNN, S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, no S-NSSAI], [PLMN, no DNN], or [PLMN, no DNN] combination, and the UE is a UE configured for high priority access in selected PLMN, then the UE is allowed to initiate 5GSM procedures for the [PLMN, DNN, S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, S-NSSAI], [PLMN, no DNN, no S-NSSAI], [PLMN, no DNN, s-NSSAI], [PLMN, no DNN, no S-NSSAI], [PLMN, no DNN, s-NSSAI], [PLMN, no DNN, no S-NSSAI], [PLMN, no DNN, s-NSSAI], [PLMN, no DNN, no S-NSSAI], [PLMN, no DNN], or [PLMN, no DNN] combination.

In an SNPN, if the back-off timer is running or is deactivated for a given [SNPN, DNN, S-NSSAI], [SNPN, DNN, no S-NSSAI], [SNPN, no DNN, S-NSSAI], [SNPN, no DNN, no S-NSSAI], [SNPN, DNN], or [SNPN, no DNN] combination if the UE does not support access to an SNPN using credentials from a credentials holder, or the back-off timer is running or is deactivated for a given [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, S-NSSAIJ, [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, no S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, no S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN] combination if the UE supports access to an SNPN using credentials from a credentials holder, and the UE is a UE configured for high priority access in selected SNPN, then the UE is allowed to initiate 5GSM procedures for the [SNPN, DNN, S-NSSAI], [SNPN, DNN, no S-NSSAI], [SNPN, no DNN, S-NSSAI], [SNPN, no DNN, no S-NSSAI], [SNPN, DNN], or [SNPN, no DNN] combination if the UE does not support access to an SNPN using credentials from a credentials holder, or the UE is allowed to initiate 5GSM procedures for the [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN, no S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN, no S-NSSAI], [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, DNN], or [SNPN, selected entry of the "list of subscriber data" or selected PLMN subcription, no DNN] combination if the UE supports access to an SNPN using credentials from a credentials holder.

Neither the re-attempt indicator IE nor re-attempt indicator derivation shall be applicable in an SNPN.

If the network does not include a back-off timer with 5GSM cause #29 "insufficient resources", #31 "request rejected, unspecified", and #38 "network failure", the UE may use a local back-off timer that has exponential value or a default value, which is provisioned using implementation specific means, to:

- prevent sending the rejected 5GSM procedure for the respective DNN till its expiry; and

- allow sending the rejected 5GSM procedure for the respective DNN after its expiry.

6.2.13 Handling of Small data rate control

Small data rate control is applicable only to NB-N1 mode and WB-N1 mode.

Small data rate control controls the maximum number of uplink user data messages including uplink exception data reporting sent by the UE in a time interval for the PDU session in accordance with 3GPP TS 23.501 [8]. The UE shall limit the rate at which it generates uplink user data messages to comply with the small data rate control policy. The NAS shall provide the indicated rates to upper layers for enforcement. The indicated rates in a NAS procedure applies to the PDU session the NAS procedure corresponds to, and the indicated rates are valid until a new value is indicated or the PDU session is released.

If the UE indicates support for CIoT 5GS optimizations, the network may provide the small data rate control parameters to the UE and may provide the small data rate control parameters for exception data to the UE if and only if the small data rate control parameters is provided to the UE. Small data rate control parameters and small data rate control parameters for exception data can also be provided to the UE in S1 mode as specified in 3GPP TS 24.301 [15].

If an allowed indication of additional exception reports is provided with the small data rate control parameters and:

- the additional small data rate control parameters for exception data is provided and the limit for additional rate for exception data reporting is not reached; or
- the additional small data rate control parameters for exception data is not provided,

the UE is allowed to send uplink exception reports even if the limit for the small data rate control has been reached.

During a PDU session release procedure, if the small data rate control was applied to the PDU session that is being released, the network may store the small data rate control status for the released PDU session as specified in 3GPP TS 23.501 [8].

If:

- a) the UE indicates support for CIoT 5GS optimizations; and
- b) the small data rate control status was stored for the PDU session and is still valid,

the network may provide the remaining small data rate control status as initial small data rate control parameters to the UE and initial small data rate control parameters for exception data to the UE during a subsequent PDU session establishment procedure.

If received during the establishment of a PDU session, the UE shall apply the initial small data rate control parameters and the initial small data rate control parameters for exception data for the duration of the validity period. When the validity period expires the small data rate control parameters and the small data rate control parameters for exception data shall be applied (see 3GPP TS 23.501 [8]).

NOTE 1: The HPLMN can discard or delay user data that exceeds the limit provided for small data rate control.

Upon inter-system change from N1 mode to S1 mode, the UE shall store the current small data rate control status for PDU sessions to be transferred from N1 mode to S1 mode as specified in 3GPP TS 23.501 [8].

NOTE 2: How long the UE stores the current small data rate control status is implementation specific.

Upon inter-system change from S1 mode to N1 mode, the UE shall use the stored small data rate control status, if any, to comply with the small data rate control policy for PDU sessions transferred from S1 mode to N1 mode as specified in 3GPP TS 23.501 [8], if the validity period of the stored small data rate control status has not expired.

6.2.14 Handling of Serving PLMN rate control

Serving PLMN rate control is applicable only for PDU sessions established for control plane CIoT 5GS optimization.

Serving PLMN rate control protect its AMF from the load generated by user data over control plane.

The SMF can inform the UE of any local serving PLMN rate control during the PDU session establishment procedure (see subclause 6.4.1) or the PDU session modification procedure (see subclause 6.4.2). If serving PLMN rate control is

enabled, the SMF shall start the serving PLMN rate control for the PDU session when the first control plane user data is received over the PDU session. The UE shall limit the rate at which it generates uplink control plane user data to comply with the serving PLMN policy provided by the network. The indicated rate in a NAS procedure applies to the PDU session the NAS procedure corresponds to, and the indicated rate is valid until the PDU session is released.

Any Serving PLMN rate control information provided by the network to the UE is only applicable for the PLMN which provided this information. This serving PLMN rate control information shall be discarded when the UE successfully registers to another PLMN.

NOTE: The serving PLMN can discard or delay control plane user data that exceed the limit provided for Serving PLMN rate control.

6.2.15 Handling of Reliable Data Service

If the UE supports Reliable Data Service (see 3GPP TS 23.501 [8] and 3GPP TS 24.250 [14A]), the UE may request data transfer using Reliable Data Service for a PDU session in the Extended protocol configuration options IE during UE-requested PDU session establishment procedure (see subclause 6.4.1).

The Reliable Data Service may only be used with PDU sessions for which the "Control Plane CIoT 5GS Optimisation" indicator is set or with PDU sessions using the control plane CIoT 5GS optimization when the AMF does not move such PDU sessions to the user plane.

The network shall inform the UE about the acceptance of UE's request for Reliable Data Service usage during the PDU session establishment procedure (see subclause 6.4.1) in the Extended protocol configuration options IE.

If the network accepts the use of Reliable Data Service to transfer data for the specified PDU session, the UE shall use this PDU session exclusively for data transfer using Reliable Data Service; otherwise the UE shall not use this PDU session for data transfer using Reliable Data Service.

6.2.16 Handling of header compression for control plane CIoT optimizations

The UE and the SMF may use:

- IP header compression for PDU sessions of "IPv4", "IPv6" or "IPv4v6" PDU session type; and
- Ethernet header compression for PDU sessions of "Ethernet" PDU session type.

Both the UE and the AMF indicate whether header compression for control plane CIoT 5GS optimization is supported during registration procedures (see subclause 5.5.1). If both the UE and the network support header compression, the header compression configuration for each PDU session is negotiated during the PDU session establishment procedure and PDU session modification procedure as specified in subclauses 6.3.2, 6.4.1 and 6.4.2.

For IP header compression, ROHC protocol specified in IETF RFC 5795 [39B] is used. The IP header compression configuration used for IP header compression is (re-)negotiated between the UE and the SMF using the IP header compression configuration IE as specified in subclauses 6.3.2.2, 6.4.1.2, 6.4.1.3 and 6.4.2.2, respectively.

For Ethernet header compression, Ethernet Header Compression (EHC) protocol specified in 3GPP TS 38.323 [29] is used. The Ethernet header compression configuration used for Ethernet header compression is (re-)negotiated between the UE and the SMF using the Ethernet header compression configuration IE as specified in subclauses 6.3.2.2, 6.4.1.2, 6.4.1.3 and 6.4.2.2, respectively.

6.2.17 Handling of edge computing enhancements

EAS discovery, EAS rediscovery and ECS address provisioning provide enhanced edge computing support in 5GS (see 3GPP TS 23.548 [10A]).

If the network supports the session breakout connectivity model or distributed anchor connectivity model to enable edge computing enhancements and the UE generated DNS message is to be handled by an edge application server discovery function (EASDF) for EAS discovery as specified in 3GPP TS 23.548 [10A], the SMF selects the EASDF and it provides its IP address to the UE as the DNS server to be used for the PDU session in the Extended protocol

configuration options IE during the UE-requested PDU session establishment procedure as described in subclause 6.4.1.3.

NOTE 1: EASDF selection is outside the scope of the present document.

If the network supports the session breakout connectivity model to enable edge computing enhancements and the UE generated DNS message is to be handled by a local DNS server for EAS discovery as specified in 3GPP TS 23.548 [10A], the SMF selects the local DNS server, obtains its IP address and can provide the IP address of the local DNS server to the UE as the DNS server to be used for the PDU session in the Extended protocol configuration options IE during the UE-requested PDU session establishment procedure or the network-requested PDU session modification procedure as described in subclauses 6.4.1.3 and 6.3.2.2, respectively.

NOTE 2: Local DNS server selection and the acquisition of its IP address is outside the scope of the present document.

If the UE supports EAS rediscovery and the SMF decides to trigger the EAS rediscovery as specified in 3GPP TS 23.548 [10A], the SMF initiates a network-requested PDU session modification procedure to provide the EAS rediscovery information to the UE as described in subclauses 6.3.2.2. Upon receipt of the EAS rediscovery information, the UE provides the received information to the upper layers.

NOTE 3: The upper layers of the UE uses the EAS rediscovery information to trigger the EAS discovery procedure to get the new EAS information as specified in 3GPP TS 23.548 [10A].

If the UE supports ECS address provisioning over NAS as specified in 3GPP TS 23.548 [10A], the UE indicates its support of ECS configuration information provisioning over NAS in the Extended protocol configuration options IE either during the UE-requested PDU session establishment procedure as described in subclause 6.4.1.2 or while in S1 mode as described in 3GPP TS 24.301 [15], respectively.

If the UE indicated support of ECS configuration information address provisioning over NAS, the SMF can provide the ECS configuration information in the Extended protocol configuration options IE during the network-requested PDU session modification procedure, UE-requested PDU session establishment procedure or the UE-requested PDU session modification procedure as described in subclauses 6.3.2.2, 6.4.1.3 and 6.4.2.3, respectively.

NOTE 4: The SMF can obtain the ECS configuration information based on the local configuration, the UE's location, and/or UE's subscription information.

If the UE supports the edge DNS client (EDC) as specified in 3GPP TS 23.548 [10A], the UE indicates its support of EDC in the Extended protocol configuration options IE during the UE-requested PDU session establishment procedure as described in subclause 6.4.1.2 or the UE-requested PDU session modification procedure as described in subclause 6.4.2.2.

If the UE indicates support of EDC, the SMF can indicate in the Extended protocol configuration options IE during the UE-requested PDU session establishment procedure as described in subclause 6.4.1.3 or the network-requested PDU session modification procedure as described in subclause 6.3.2.2, that the network allows the use of EDC for the applications which are mapped onto the PDU session and explicitly requested the use of EDC, or that the network requires the use of EDC for all applications mapped onto the PDU session.

6.2.18 Support of redundant PDU sessions

The 5GSM sublayer may support establishment of redundant PDU sessions (see clause 5.33.2 of 3GPP TS 23.501 [8]).

In order to establish a set of two redundant PDU sessions, a UE can include a PDU session pair ID, an RSN, or both in a PDU SESSION ESTABLISHMENT REQUEST message for each of the two redundant PDU sessions (see subclause 6.4.1.2). The UE can set the PDU session pair ID, the RSN, or both according to URSP or UE local configuration (see 3GPP TS 24.526 [19]).

An SMF receiving a PDU session pair ID, an RSN, or both via a PDU SESSION ESTABLISHMENT REQUEST message operates as specified in clause 5.33.2 of 3GPP TS 23.501 [8]. In addition, an SMF can handle two PDU sessions as redundant even if the UE provides neither a PDU session pair ID nor an RSN in a PDU SESSION ESTABLISHMENT REQUEST message for each of the PDU sessions (see clause 5.33.2 of 3GPP TS 23.501 [8]).

6.3 Network-requested 5GSM procedures

6.3.1 PDU session authentication and authorization procedure

6.3.1.1 General

The purpose of the PDU session authentication and authorization procedure is to enable the DN:

- a) to authenticate the upper layers of the UE, when establishing the PDU session;
- b) to authorize the upper layers of the UE, when establishing the PDU session;
- c) both of the above; or
- d) to re-authenticate the upper layers of the UE after establishment of the PDU session.

The PDU session authentication and authorization procedure can be performed only during or after the UE-requested PDU session procedure establishing a non-emergency PDU session. The PDU session authentication and authorization procedure shall not be performed during or after the UE-requested PDU session establishment procedure establishing an emergency PDU session.

The upper layers store the association between a DNN and corresponding credentials, if any, for the PDU session authentication and authorization.

If the UE is registered for onboarding services in SNPN the SMF may initiate the PDU session authentication and authorization procedure based on local policy with a DCS as specified in 3GPP TS 33.501 [24] clause I.9.2.4.1 or a DN-AAA server as specified in 3GPP TS 33.501 [24] clause I.9.2.4.2.

If the UE is registered for onboarding services in SNPN and the network initiates the PDU session authentication and authorization procedure, the UE shall use the default UE credentials for secondary authentication for the PDU session authentication and authorization procedure.

The network authenticates the UE using the Extensible Authentication Protocol (EAP) as specified in IETF RFC 3748 [34].

EAP has defined four types of EAP messages:

- a) an EAP-request message;
- b) an EAP-response message;
- c) an EAP-success message; and
- d) an EAP-failure message.

The EAP-request message is transported from the network to the UE using the PDU SESSION AUTHENTICATION COMMAND message of the PDU EAP message reliable transport procedure.

The EAP-response message to the EAP-request message is transported from the UE to the network using the PDU SESSION AUTHENTICATION COMPLETE message of the PDU EAP message reliable transport procedure.

If the PDU session authentication and authorization procedure is performed during the UE-requested PDU session establishment procedure:

- a) and the DN authentication of the UE completes successfully, the EAP-success message is transported from the network to the UE as part of the UE-requested PDU session establishment procedure in the PDU SESSION ESTABLISHMENT ACCEPT message.
- b) and the DN authentication of the UE completes unsuccessfully, the EAP-failure message is transported from the network to the UE as part of the UE-requested PDU session establishment procedure in the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU session authentication and authorization procedure is performed after the UE-requested PDU session establishment procedure:

- a) and the DN authentication of the UE completes successfully, the EAP-success message is transported from the network to the UE using the PDU SESSION AUTHENTICATION RESULT message of the PDU EAP result message transport procedure.
- b) and the DN authentication of the UE completes unsuccessfully, the EAP-failure message is transported from the network to the UE using the PDU SESSION RELEASE COMMAND message of the network-requested PDU session release procedure.

There can be several rounds of exchange of an EAP-request message and a related EAP-response message for the DN to complete the authentication and authorization of the request for a PDU session (see example in figure 6.3.1.1).

The SMF shall set the authenticator retransmission timer specified in IETF RFC 3748 [34] subclause 4.3 to infinite value.

NOTE: The PDU session authentication and authorization procedure provides a reliable transport of EAP messages and therefore retransmissions at the EAP layer of the SMF do not occur.
UE		SMF	
•	PDU SESSION AUTHENTICATION COMMAND <eap-request a="" message=""></eap-request>		Start T3590
	PDU SESSION AUTHENTICATION COMPLETE <eap-response a="" eap-request="" message="" to=""> </eap-response>		Stop T3590
•	 PDU SESSION AUTHENTICATION COMMAND <eap-request b="" message=""></eap-request>		Start T3590
	PDU SESSION AUTHENTICATION COMPLETE <eap-response b="" eap-request="" message="" to=""></eap-response>		Stop T3590
4	 PDU SESSION ESTABLISHMENT ACCEPT <eap-success message=""></eap-success>		
•	OR PDU SESSION ESTABLISHMENT REJECT <eap-failure message=""></eap-failure>		
◀	OR PDU SESSION AUTHENTICATION RESULT <eap-success message=""></eap-success>		
•	OR PDU SESSION RELEASE COMMAND <eap-failure message=""></eap-failure>		
•	OR REMOTE UE REPORT RESPONSE <eap-success eap-failure="" message="" or=""></eap-success>		

Figure 6.3.1.1: PDU session authentication and authorization procedure

6.3.1.2 PDU EAP message reliable transport procedure

6.3.1.2.1 PDU EAP message reliable transport procedure initiation

In order to initiate the PDU EAP message reliable transport procedure, the SMF shall create a PDU SESSION AUTHENTICATION COMMAND message.

The SMF shall set the PTI IE of the PDU SESSION AUTHENTICATION COMMAND message to "No procedure transaction identity assigned".

The SMF shall set the EAP message IE of the PDU SESSION AUTHENTICATION COMMAND message to the EAP-request message provided by the DN or generated locally.

The SMF shall send the PDU SESSION AUTHENTICATION COMMAND message, and the SMF shall start timer T3590 (see example in figure 6.3.1.1).

Upon receipt of the PDU SESSION AUTHENTICATION COMMAND message, if the UE provided a DNN during the PDU session establishment, the UE shall stop timer T3396, if it is running for the DNN provided by the UE. If the UE did not provide a DNN during the PDU session establishment, the UE shall stop the timer T3396 associated with no DNN if it is running.

Upon receipt of the PDU SESSION AUTHENTICATION COMMAND message, if the UE provided an S-NSSAI and a DNN during the PDU session establishment, the UE shall stop timer T3584, if it is running for the [S-NSSAI of the PDU session, DNN] combination. If the UE provided a DNN but did not provide an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, DNN] combination provided by the UE. If the UE provided an S-NSSAI but did not provide a DNN during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [S-NSSAI, no DNN] combination provided by the UE. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, no DNN] combination provided by the UE. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, no DNN] combination provided by the UE. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, no DNN] combination provided by the UE. The timer T3584 to be stopped includes the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running.

Upon receipt of the PDU SESSION AUTHENTICATION COMMAND message, if the UE provided an S-NSSAI during the PDU session establishment, the UE shall stop timer T3585, if it is running for the S-NSSAI of the PDU session. If the UE did not provide an S-NSSAI during the PDU session establishment, the UE shall stop the timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running.

- NOTE 1: Upon receipt of the PDU SESSION AUTHENTICATION COMMAND message for a PDU session, if the UE provided a DNN (or no DNN) and an S-NSSAI (or no S-NSSAI) when the PDU session is established, timer T3396 associated with the DNN (or no DNN, if no DNN was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3396 and the timer T3584.
- NOTE 2: Upon receipt of the PDU SESSION AUTHENTICATION COMMAND message for a PDU session, if the UE provided a DNN (or no DNN) and an S-NSSAI (or no S-NSSAI) when the PDU session is established, timer T3585 associated with the S-NSSAI (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI (or no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3585 and the timer T3584.

Upon receipt of a PDU SESSION AUTHENTICATION COMMAND message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE passes to the upper layers the EAP message received in the EAP message IE of the PDU SESSION AUTHENTICATION COMMAND message. Apart from this action and the stopping of timers T3396, T3584 and T3485 (if running), the authentication and authorization procedure initiated by the DN is transparent to the 5GSM layer of the UE.

6.3.1.2.2 PDU EAP message reliable transport procedure accepted by the UE

The UE shall create a PDU SESSION AUTHENTICATION COMPLETE message when the upper layers provide an EAP-response message responding to the received EAP-request message.

The UE shall set the EAP message IE of the PDU SESSION AUTHENTICATION COMPLETE message to the EAPresponse message.

The UE shall transport the PDU SESSION AUTHENTICATION COMPLETE message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5. Apart from this action, the authentication and authorization procedure initiated by the DN is transparent to the 5GSM layer of the UE.

Upon receipt of a PDU SESSION AUTHENTICATION COMPLETE message, the SMF shall stop timer T3590 and provides the EAP message received in the EAP message IE of the PDU SESSION AUTHENTICATION COMPLETE message to the DN or handles it locally.

6.3.1.2.3 Abnormal cases on the network side

The following abnormal cases can be identified:

a) T3590 expired.

The SMF shall, on the first expiry of the timer T3590, retransmit the PDU SESSION AUTHENTICATION COMMAND message and shall reset and start timer T3590. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3590, the SMF shall abort the procedure.

b) Collision of UE-requested PDU session release procedure and a PDU session authentication and authorization procedure.

When the SMF receives a PDU SESSION RELEASE REQUEST message during the PDU session authentication and authorization procedure, and the PDU session indicated in the PDU SESSION RELEASE REQUEST message is the PDU session that the SMF had requested to authenticate, the SMF shall abort the PDU session authentication and authorization procedure and proceed with the UE-requested PDU session release procedure.

6.3.1.2.4 Abnormal cases in the UE

The following abnormal cases can be identified:

a) PDU session inactive for the received PDU session ID.

If the PDU session ID in the PDU SESSION AUTHENTICATION COMMAND message belongs to any PDU session in state PDU SESSION INACTIVE in the UE, the UE shall send a 5GSM STATUS message with the 5GSM cause IE set to #43 "Invalid PDU session identity".

b) Collision of UE-requested PDU session release procedure and a PDU session authentication and authorization procedure.

When the UE receives a PDU SESSION AUTHENTICATION COMMAND message during the UE-requested PDU session release procedure, and the PDU session indicated in PDU SESSION AUTHENTICATION COMMAND message is the PDU session that the UE had requested to release, the UE shall ignore the PDU SESSION AUTHENTICATION COMMAND message and proceed with the UE-requested PDU session release procedure.

6.3.1.3 PDU EAP result message transport procedure

6.3.1.3.1 PDU EAP result message transport procedure initiation

PDU EAP result message transport procedure is initiated by the SMF if the PDU session authentication and authorization procedure is performed after the PDU session is established and the DN authentication of the UE completes successfully.

In order to initiate the PDU EAP result message transport procedure, the SMF shall create a PDU SESSION AUTHENTICATION RESULT message.

The SMF shall set the PTI IE of the PDU SESSION AUTHENTICATION RESULT message to "No procedure transaction identity assigned".

The SMF shall set the EAP message IE of the PDU SESSION AUTHENTICATION RESULT message to the EAPsuccess message provided by the DN.

The SMF shall send the PDU SESSION AUTHENTICATION RESULT message.

Upon receipt of a PDU SESSION AUTHENTICATION RESULT message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE passes to the upper layers the EAP message received in the EAP message IE of the PDU SESSION AUTHENTICATION RESULT message. Apart from this action, the authentication and authorization procedure initiated by the DN is transparent to the 5GSM layer of the UE.

6.3.1.3.2 Abnormal cases in the UE

The following abnormal cases can be identified:

a) PDU session inactive for the received PDU session ID.

If the PDU session ID in the PDU SESSION AUTHENTICATION RESULT message belongs to any PDU session in state PDU SESSION INACTIVE in the UE, the UE shall send a 5GSM STATUS message with the 5GSM cause IE set to #43 "Invalid PDU session identity".

b) Collision of UE-requested PDU session release procedure and a PDU EAP result message transport procedure.

When the UE receives a PDU SESSION AUTHENTICATION RESULT message during the UE-requested PDU session release procedure, and the PDU session indicated in PDU SESSION AUTHENTICATION RESULT message is the PDU session that the UE had requested to release, the UE shall ignore the PDU SESSION AUTHENTICATION RESULT message and proceed with the UE-requested PDU session release procedure.

6.3.1A Service-level authentication and authorization procedure

6.3.1A.1 General

The purpose of the service-level authentication and authorization (service-level-AA) procedure is to enable the DN using NEF services for authentication:

- a) to authenticate the upper layers of the UE, when establishing the PDU session;
- b) to authorize the upper layers of the UE, when establishing the PDU session;
- c) both of the above; or
- d) to re-authenticate the upper layers of the UE after establishment of the PDU session.

The service-level authentication and authorization procedure is used for UUAA as specified in TS 23.256 [6AB].

NOTE 1: The authentication protocol for UUAA is out of scope of 3GPP in this release of specification.

The service-level authentication and authorization procedure can be performed only during or after the UE-requested PDU session procedure establishing a non-emergency PDU session. The service-level authentication and authorization procedure shall not be performed during or after the UE-requested PDU session establishment procedure establishing an emergency PDU session.

If the service-level authentication and authorization procedure is performed during the UE-requested PDU session establishment procedure:

- a) and the service-level-AA procedure of the UE completes successfully, the service-level-AA response is transported from the network to the UE as a part of the UE-requested PDU session establishment procedure in the PDU SESSION ESTABLISHMENT ACCEPT message; or
- b) and the service-level-AA procedure of the UE completes unsuccessfully, the service-level-AA response is transported from the network to the UE as a part of the UE-requested PDU session establishment procedure in the PDU SESSION ESTABLISHMENT REJECT message.
- NOTE 2: If the SMF receives the HTTP code set to "4xx" or "5xx" as specified in 3GPP TS 29.500 [20AA] or the SMF detects a UUAA-SM failure as specified in 3GPP TS 29.256 [21B], then the SMF considers that the UUAA-SM procedure has completed unsuccessfully.

If the service-level authentication and authorization procedure is performed for the established PDU session with reauthentication purpose:

a) and the service-level-AA procedure of the UE completes successfully, the service-level-AA response is transported from the network to the UE as a part of the network-requested PDU session modification procedure in the PDU SESSION MODIFICATION COMMAND message; or

b) and the service-level-AA procedure of the UE completes unsuccessfully, the service-level-AA response is transported from the network to the UE as a part of the network-requested PDU session release procedure in the PDU SESSION RELEASE COMMAND message.

There can be several rounds of exchange of a service-level-AA payload for the service to complete the service-level authentication and authorization of the request for a PDU session (see example in figure 6.3.1A.1-1).

If the UE receives the service-level-AA response in the PDU SESSION ESTABLISHMENT ACCEPT message or the PDU SESSION ESTABLISHMENT REJECT message, the UE passes it to the upper layer.



Figure 6.3.1A.1-1: Service-level authentication and authorization procedure

6.3.1A.2 Service-level authentication and authorization procedure initiation

In order to initiate the service-level authentication and authorization procedure, the SMF shall create a SERVICE-LEVEL AUTHENTICATION COMMAND message.

The SMF shall set the PTI IE of the SERVICE-LEVEL AUTHENTICATION COMMAND message to "No procedure transaction identity assigned".

The SMF shall set the service-level-AA payload in the Service-level-AA container IE of the SERVICE-LEVEL AUTHENTICATION COMMAND message to the payload provided by the DN via the NEF. If a payload type associated with the payload is provided by the DN via the NEF, the SMF shall set the service-level-AA payload type with the value set to the payload type.

NOTE: In case of UUAA, the service-level-AA payload is provided by the DN via the UAS-NF.

The SMF shall send the SERVICE-LEVEL AUTHENTICATION COMMAND message, and the SMF shall start timer T3594 (see example in figure 6.3.1A.1-1).

Upon receipt of a SERVICE-LEVEL AUTHENTICATION COMMAND message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE passes to the upper layers the service-level-AA payload received in the Service-level-AA container IE of the SERVICE-LEVEL AUTHENTICATION COMMAND message. Apart from this action, the service-level authentication and authorization procedure initiated by the DN is transparent to the 5GSM layer of the UE.

6.3.1A.3 Service-level authentication and authorization procedure accepted by the UE

When the upper layers provide a service-level-AA payload, the UE shall create a SERVICE-LEVEL AUTHENTICATION COMPLETE message and set the service-level-AA payload of the Service-level-AA container IE to the service-level-AA payload received from the upper layers, and if the service-level-AA payload type is received in the SERVICE-LEVEL AUTHENTICATION COMMAND message from the SMF, set the service-level-AA payload type of the Service-level-AA container IE to the service-level-AA payload type received from the SMF.

The UE shall transport the SERVICE-LEVEL AUTHENTICATION COMPLETE message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5. Apart from this action, the service-level authentication and authorization procedure initiated by the DN is transparent to the 5GSM layer of the UE.

Upon receipt of a SERVICE-LEVEL AUTHENTICATION COMPLETE message, the SMF shall stop timer T3594 and provides the service-level-AA payload received in the Service-level-AA container IE of the SERVICE-LEVEL AUTHENTICATION COMPLETE message to the DN.

6.3.1A.4 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3594.

On the first expiry of the timer T3594, the SMF shall resend the SERVICE-LEVEL AUTHENTICATION COMMAND message and shall reset and restart timer T3594. This retransmission is repeated four times, i.e., on the fifth expiry of timer T3594, the SMF shall abort the procedure and send PDU SESSION ESTABLISHMENT REJECT message with the 5GSM cause #29 "user authentication or authorization failed" as specified in subclause 6.4.1.4.1.

6.3.1A.5 Abnormal cases in the UE

The following abnormal case can be identified:

a) Collision of UE-requested PDU session release procedure and a service-level authentication and authorization procedure.

When the UE receives a SERVICE-LEVEL AUTHENTICATION COMMAND message during the UErequested PDU session release procedure, and the PDU session indicated in SERVICE-LEVEL AUTHENTICATION COMMAND message is the PDU session that the UE has requested to release, the UE shall ignore the SERVICE-LEVEL AUTHENTICATION COMMAND message and proceed with the UErequested PDU session release procedure.

6.3.2 Network-requested PDU session modification procedure

6.3.2.1 General

The purpose of the network-requested PDU session modification procedure is to enable the network to modify a PDU session, re-negotiate header compression configuration associated to a PDU session, convey a port management information container, to trigger EAS rediscovery, provide updated DNS server address(es) due to the newly selected local DNS server or the newly selected EASDF, provide updated ECS configuration information, remove joined UE from one or more multicast MBS sessions associated with a PDU session, update ATSSS parameters (e.g. ATSSS rules), update the MBS service area or the security information of multicast MBS session that the UE has joined or to inform about the result of service-level AA procedure or C2 authorization for UAS services.

6.3.2.2 Network-requested PDU session modification procedure initiation

In order to initiate the network-requested PDU session modification procedure, the SMF shall create a PDU SESSION MODIFICATION COMMAND message.

If the authorized QoS rules of the PDU session is modified or is marked as to be synchronised with the UE, the SMF shall set the Authorized QoS rules IE of the PDU SESSION MODIFICATION COMMAND message to the authorized QoS rules of the PDU session. The SMF shall ensure that the number of the packet filters used in the authorized QoS rules of the PDU Session does not exceed the maximum number of packet filters supported by the UE for the PDU session. The SMF may bind service data flows for which the UE has requested traffic segregation to a dedicated QoS flow for the PDU session, if possible. Otherwise the SMF may bind the service data flows to an existing QoS flow. The SMF shall use only one dedicated QoS flow for traffic segregation. If the UE has requested traffic segregation for multiple service data flows with different QoS handling, the SMF shall bind all these service data flows to a single QoS flow. If the SMF allows traffic segregation for service data flows in a QoS rule, then the SMF shall create a new authorized QoS rule for these service data flows and shall delete packet filters corresponding to these service data flows from the other authorized QoS rules.

If the authorized QoS flow descriptions of the PDU session is modified or is marked as to be synchronised with the UE, the SMF shall set the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message to the authorized QoS flow descriptions of the PDU session.

If SMF creates a new authorized QoS rule for a new QoS flow, then SMF shall include the authorized QoS flow description for that QoS flow in the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message, if:

- a) the newly created authorized QoS rules is for a new GBR QoS flow;
- b) the QFI of the new QoS flow is not the same as the 5QI of the QoS flow identified by the QFI;
- c) the new QoS flow can be mapped to an EPS bearer as specified in subclause 4.11.1 of 3GPP TS 23.502 [9]; or
- d) the new QoS flow is established for the PDU session used for relaying, as specified in subclause 5.6.2.1 of 3GPP TS 23.304 [6E].
- NOTE 0: In cases other than above listed cases, it is up to the SMF implementation to include the authorized QoS flow description of the new QoS flow for the new authorized QoS rule in the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message.

If the session-AMBR of the PDU session is modified, the SMF shall set the selected Session-AMBR IE of the PDU SESSION MODIFICATION COMMAND message to the session-AMBR of the PDU session.

If interworking with EPS is supported for the PDU session and if the mapped EPS bearer contexts of the PDU session is modified, the SMF shall set the Mapped EPS bearer contexts IE of the PDU SESSION MODIFICATION COMMAND message to the mapped EPS bearer contexts of the PDU session. If the association between a QoS flow and the mapped EPS bearer context is changed, the SMF shall set the EPS bearer identity parameter in Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message to the new EPS bearer identity associated with the QoS flow.

If the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure and the PDU SESSION MODIFICATION REQUEST message includes a 5GSM capability IE, the SMF shall:

- a) if the RQoS bit is set to:
 - 1) "Reflective QoS supported", consider that the UE supports reflective QoS for this PDU session; or
 - 2) "Reflective QoS not supported", consider that the UE does not support reflective QoS for this PDU session; and;
- b) if the MH6-PDU bit is set to:
 - 1) "Multi-homed IPv6 PDU session supported", consider that this PDU session is supported to use multiple IPv6 prefixes; or

3GPP TS 24.501 version 17.16.0 Release 17

2) "Multi-homed IPv6 PDU session not supported", consider that this PDU session is not supported to use multiple IPv6 prefixes.

If the SMF considers that reflective QoS is supported for QoS flows belonging to this PDU session, the SMF may include the RQ timer IE set to an RQ timer value in the PDU SESSION MODIFICATION COMMAND message.

If a port management information container needs to be delivered (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]) and the UE has set the TPMIC bit to "Transport of port management information container supported" in the 5GSM capability IE, the SMF shall include a Port management information container IE in the PDU SESSION MODIFICATION COMMAND message.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the PDU session type is "IPv4", "IPv6", "IPv4v6" or "Ethernet" and the PDU SESSION MODIFICATION REQUEST message includes a Maximum number of supported packet filters IE, the SMF shall consider this number as the maximum number of packet filters that can be supported by the UE for this PDU session. Otherwise the SMF considers that the UE supports 16 packet filters for this PDU session.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the SMF shall consider that the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink and the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink are valid for the lifetime of the PDU session.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure and the SMF determines, based on local policies or configurations in the SMF and the Always-on PDU session requested IE in the PDU SESSION MODIFICATION REQUEST message (if available), that either:

- a) the requested PDU session needs to be an always-on PDU session, the SMF shall include the Always-on PDU session indication IE in the PDU SESSION MODIFICATION COMMAND message and shall set the value to "Always-on PDU session required"; or
- b) the requested PDU session shall not be an always-on PDU session and:
 - 1) if the UE included the Always-on PDU session requested IE, the SMF shall include the Always-on PDU session indication IE in the PDU SESSION MODIFICATION COMMAND message and shall set the value to "Always-on PDU session not allowed"; or
 - 2) if the UE did not include the Always-on PDU session requested IE, the SMF shall not include the Always-on PDU session indication IE in the PDU SESSION MODIFICATION COMMAND message.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the UE supports EDC and the network allows the use of EDC, then the SMF shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message with the EDC usage allowed indicator.

For a PDN connection established when in S1 mode, upon the first inter-system change from S1 mode to N1 mode, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the UE supports EDC and the network requires the use of EDC, then the SMF shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message with the EDC usage required indicator.

If a QoS flow for URLLC is created in a PDU session and the SMF has not provided the Always-on PDU session indication IE with the value set to "Always-on PDU session required" in the UE-requested PDU session establishment procedure or a network-requested PDU session modification procedure for the PDU session, the SMF shall include the Always-on PDU session indication IE in the PDU SESSION MODIFICATION COMMAND message and shall set the value to "Always-on PDU session required".

If the value of the RQ timer is set to "deactivated" or has a value of zero, the UE considers that RQoS is not applied for this PDU session and remove the derived QoS rule(s) associated with the PDU session, if any.

If the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the SMF shall set the PTI IE of the PDU SESSION MODIFICATION COMMAND message to the PTI of

the PDU SESSION MODIFICATION REQUEST message received as part of the UE-requested PDU session modification procedure.

If the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure and the UE has included the Requested MBS container IE in the PDU SESSION MODIFICATION REQUEST message with the MBS operation set to "Join multicast MBS session", the SMF:

- a) shall include the TMGI for the multicast MBS session IDs that the UE is allowed to join, if any, in the Received MBS container IE, shall set the MBS decision to "MBS join is accepted" for each of those Received MBS information, may include the MBS start time to indicate the time when the multicast MBS session starts, and shall include the MBS security container in each of those Received MBS information if security protection is applied for that multicast MBS session and the control plane security procedure is used as specified in annex W.4.1.2 in 3GPP TS 33.501 [24], and shall use separate QoS flows dedicated for multicast by including the Authorized QoS flow descriptions IE if no separate QoS flows dedicated for multicast exist or if the SMF wants to establish new QoS flows dedicated for multicast;
- NOTE 1: The network determines whether security protection applies or not for the multicast MBS session as specified in 3GPP TS 33.501 [24].
- b) shall include the TMGI for multicast MBS session IDs that the UE is rejected to join, if any, in the Received MBS container IE, shall set the MBS decision to "MBS join is rejected" for each of those Received MBS information, shall set the Rejection cause for each of those Received MBS information with the reason of rejection and, if the Rejection cause is set to "multicast MBS session has not started or will not start soon", may include an MBS back-off timer value; and
- c) may include in the Received MBS container IE the MBS service area for each multicast MBS session and include in it the MBS TAI list, the NR CGI list or both, that identify the service area(s) for the local MBS service;
- NOTE 2: For an multicast MBS session that has multiple MBS service areas, the MBS service areas are indicated to the UE using MBS service announcement as described in 3GPP TS 23.247 [53], which is out of scope of this specification.

in the PDU SESSION MODIFICATION COMMAND message. If the UE has set the Type of multicast MBS session ID to "Source specific IP multicast address" in the Requested MBS container IE for certain multicast MBS session(s) in the PDU SESSION MODIFICATION REQUEST message, the SMF shall include the Source IP address information and Destination IP address information in the Received MBS information together with the TMGI for each of those multicast MBS sessions.

- NOTE 3: Including the Source IP address information and Destination IP address information in the Received MBS information in that case is to allow the UE to perform the mapping between the requested multicast MBS session ID and the provided TMGI.
- NOTE 4: In SNPN, TMGI is used together with NID to identify an MBS Session.

If:

- a) the SMF wants to remove joined UE from one or more multicast MBS sessions; or
- b) the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure and the UE has included the Requested MBS container IE in the PDU SESSION MODIFICATION REQUEST message with the MBS operation set to "Leave multicast MBS session",

the SMF shall include the multicast MBS session IDs that the UE is removed from, if any, in the Received MBS container IE in the PDU SESSION MODIFICATION COMMAND message and shall set the MBS decision to "Remove UE from multicast MBS session" for each of those Received MBS information. The SMF may include the updated MBS service area in each of the Received MBS information, if any. The SMF may delete the QoS flows associated for the multicast by including the Authorized QoS flow descriptions IE in the PDU SESSION MODIFICATION COMMAND message. If the UE is removed from multicast MBS session due to the multicast MBS session release, the SMF shall set the Rejection cause to "MBS session is released". The SMF shall include the Rejection cause for each of the Received MBS information, if any, and set its value with the reason of removing the UE from the corresponding multicast MBS session.

NOTE 5: based on operator's policy, e.g. after a locally configured time period, the SMF is allowed to trigger the removal of joined UE from a multicast MBS session when the UE moves outside all the MBS service area(s) of that multicast MBS session.

If the SMF wants to update the MBS security information of an multicast MBS session that the UE has joined, the SMF shall include the corresponding multicast MBS session ID and the MBS security container in the Received MBS container IE in the PDU SESSION MODIFICATION COMMAND message, and shall set the MBS Decision to "MBS security information update" in the Received MBS information.

If the SMF wants to update the MBS service area of an multicast MBS session that the UE has joined, the SMF shall include the corresponding multicast MBS session ID and the updated MBS service area in the Received MBS container IE in the PDU SESSION MODIFICATION COMMAND message, and shall set the MBS decision to "MBS service area update" in the Received MBS information.

NOTE 6: The MBS service area of an multicast MBS session is also allowed to be updated to the UE using the MBS service announcement as described in 3GPP TS 23.247 [53], which is out of scope of this specification.

If the network needs to update ATSSS parameters (see subclause 5.2.4 of 3GPP TS 24.193 [13B]), the SMF shall include the ATSSS container IE with the updates of ATSSS parameters in the PDU SESSION MODIFICATION COMMAND message.

If the network-requested PDU session modification procedure is not triggered by a UE-requested PDU session modification procedure, the SMF shall set the PTI IE of the PDU SESSION MODIFICATION COMMAND message to "No procedure transaction identity assigned".

If the selected SSC mode of the PDU session is "SSC mode 3" and the SMF requests the relocation of SSC mode 3 PDU session anchor with multiple PDU sessions as specified in 3GPP TS 23.502 [9], the SMF shall include 5GSM cause #39 "reactivation requested", in the PDU SESSION MODIFICATION COMMAND message, and may include the PDU session address lifetime in a PDU session address lifetime parameter in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message.

The SMF shall send the PDU SESSION MODIFICATION COMMAND message, and the SMF shall start timer T3591 (see example in figure 6.3.2.2.1).

NOTE 7: If the SMF requests the relocation of SSC mode 3 PDU session anchor with multiple PDU sessions as specified in 3GPP TS 23.502 [9], the reallocation requested indication indicating whether the SMF is to be reallocated or the SMF is to be reused is provided to the AMF.

If the control plane CIoT 5GS optimization is enabled for a PDU session and the IP header compression configuration IE was included in the PDU SESSION ESTABLISHMENT REQUEST message or the PDU SESSION MODIFICATION REQUEST message, and the SMF supports control plane CIoT 5GS optimization and IP header compression for control plane CIoT 5GS optimization, the SMF may include the IP header compression configuration IE in the PDU SESSION MODIFICATION COMMAND message to re-negotiate IP header compression configuration associated to the PDU session.

If the control plane CIoT 5GS optimization is enabled for a PDU session and the Ethernet header compression configuration IE was included in the PDU SESSION ESTABLISHMENT REQUEST message or the PDU SESSION MODIFICATION REQUEST message, and the SMF supports control plane CIoT 5GS optimization and Ethernet header compression for control plane CIoT 5GS optimization, the SMF may include the Ethernet header compression configuration IE in the PDU SESSION MODIFICATION COMMAND message to re-configure Ethernet header compression configuration associated with the PDU session.

If the network-requested PDU session modification procedure is associated with C2 authorization procedure, the SMF shall send the PDU SESSION MODIFICATION COMMAND message by including the Service-level-AA container IE containing:

- a) the service-level-AA response with the value of C2AR field set to the "C2 authorization was successful";
- b) if apayload is provided from the UAS-NF, the service-level-AA payload with the value set to the payload; and
- c) if a payload type associated with the payload is provided from the UAS-NF, the service-level-AA payload type with the value set to the payload type; and

- d) if the CAA-level UAV ID is provided from the UAS-NF, the service-level device ID set to the CAA-level UAV ID.
- NOTE 8: The C2 authorization payload in the service-level-AA payload can include one or both of the C2 session security information and C2 pairing information.

If the service-level-AA procedure is triggered for the established PDU session for UAS services with re-authentication purpose, and the SMF is provided by the UAS-NF with the successful UUAA-SM result, the SMF shall transmit a PDU SESSION MODIFICATION COMMAND message to the UE, where the PDU SESSION MODIFICATION COMMAND message shall include the Service-level-AA container IE containing:

- a) the service-level-AA response with the value of SLAR field set to "Service level authentication and authorization was successful";
- b) if received the CAA-level UAV ID from the UAS-NF, the service-level device ID with the value set to the CAA-level UAV ID;
- c) if received a payload from the UAS-NF, the service-level-AA payload with the value set to the UUAA payload; and
- d) if received a payload type associated with the payload, the service-level-AA payload type with the value set to the payload type.

If the SMF needs to provide new ECS configuration information to the UE and the UE has indicated support for ECS configuration information provisioning in the PDU SESSION ESTABLISHMENT REQUEST message or while in S1 mode, then the SMF may include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message with:

- at least one of ECS IPv4 Address(es), ECS IPv6 Address(es), ECS FQDN(s);
- at least one associated ECSP identifier; and
- optionally, spatial validity conditions associated with the ECS address;
- NOTE 9: The IP address(es) and/or FQDN(s) are associated with the ECSP identifier and replace previously provided ECS configuration information associated with the same ECSP identifier, if any.

If the SMF needs to provide DNS server address(es) to the UE and the UE has provided the DNS server IPv4 address request, the DNS server IPv6 address request or both of them, in the PDU SESSION ESTABLISHMENT REQUEST message or a PDU SESSION MODIFICATION REQUEST message, then the SMF shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message with one or more DNS server IPv4 address(es), one or more DNS server IPv6 address(es) or both of them.

If the SMF needs to trigger EAS rediscovery and the UE has indicated support of the EAS rediscovery in the PDU SESSION ESTABLISHMENT REQUEST message or the PDU SESSION MODIFICATION REQUEST message, then the SMF shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message:

- a) with the EAS rediscovery indication without indicated impact; or
- b) with the following:
 - 1) one or more EAS rediscovery indication(s) with impacted EAS IPv4 address range, if the UE supports EAS rediscovery indication(s) with impacted EAS IPv4 address range;
 - 2) one or more EAS rediscovery indication(s) with impacted EAS IPv6 address range, if the UE supports EAS rediscovery indication(s) with impacted EAS IPv6 address range;
 - 3) one or more EAS rediscovery indication(s) with impacted EAS FQDN, if the UE supports EAS rediscovery indication(s) with impacted EAS FQDN; or
 - 4) any combination of the above.

When UE has requested P-CSCF IPv6 address or P-CSCF IPv4 address and the SMF has provided P-CSCF address(es) during the PDU session establishment procedure, if the network-requested PDU session modification procedure is triggered for P-CSCF restoration, the SMF shall include the P-CSCF IP address(es) in the Extended protocol

configuration options IE in the PDU SESSION MODIFICATION COMMAND message as specified in subclause 5.8.2.2 of 3GPP TS 23.380 [54].



Figure 6.3.2.2.1: Network-requested PDU session modification procedure

6.3.2.3 Network-requested PDU session modification procedure accepted by the UE

Upon receipt of the PDU SESSION MODIFICATION COMMAND message, if the UE provided a DNN during the PDU session establishment, the UE shall stop timer T3396, if it is running for the DNN provided by the UE. If the UE did not provide a DNN during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3396 associated with no DNN if it is running. If the PDU SESSION MODIFICATION COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3396 associated with no DNN if it is running.

Upon receipt of the PDU SESSION MODIFICATION COMMAND message, if the UE provided an S-NSSAI and a DNN during the PDU session establishment, the UE shall stop timer T3584, if it is running for the [S-NSSAI of the PDU session, DNN] combination provided by the UE. If the UE provided a DNN and did not provide an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, DNN] combination provided by the UE. If the UE provided an S-NSSAI and did not provide a DNN during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, DNN] combination provided by the UE. If the UE provided an S-NSSAI and did not provide a DNN during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [S-NSSAI, no DNN] combination provided by the UE. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment, the UE shall stop timer T3584, if it is running for the same [no S-NSSAI, no DNN] combination provided by the UE. The timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running.

Upon receipt of the PDU SESSION MODIFICATION COMMAND message, if the UE provided an S-NSSAI during the PDU session establishment, the UE shall stop timer T3585, if it is running for the S-NSSAI of the PDU session. If the UE did not provide an S-NSSAI during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. If the PDU SESSION MODIFICATION COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3585 associated with no S-NSSAI if it is running.

NOTE 1: Upon receipt of the PDU SESSION MODIFICATION COMMAND message for a PDU session, if the UE provided a DNN (or no DNN) and an S-NSSAI (or no S-NSSAI) when the PDU session is established, timer T3396 associated with the DNN (or no DNN, if no DNN was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3396 and the timer T3584.

NOTE 2: Upon receipt of the PDU SESSION MODIFICATION COMMAND message for a PDU session, if the UE provided a DNN (or no DNN) and an S-NSSAI (or no S-NSSAI) when the PDU session is established, timer T3585 associated with the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3585 and the timer T3584.

If the PDU SESSION MODIFICATION COMMAND message includes the Authorized QoS rules IE, the UE shall process the QoS rules sequentially starting with the first QoS rule.

If the PDU SESSION MODIFICATION COMMAND message includes the Mapped EPS bearer contexts IE, the UE shall process the mapped EPS bearer contexts sequentially starting with the first mapped EPS bearer context.

If the PDU SESSION MODIFICATION COMMAND message includes the Authorized QoS flow descriptions IE, the UE shall process the QoS flow descriptions sequentially starting with the first QoS flow description.

The UE shall replace the stored authorized QoS rules, authorized QoS flow descriptions and session-AMBR of the PDU session with the received value(s), if any, in the PDU SESSION MODIFICATION COMMAND message.

If the PDU SESSION MODIFICATION COMMAND message includes a Mapped EPS bearer contexts IE, the UE shall check each mapped EPS bearer context for different types of errors as follows:

- NOTE 3: An error detected in a mapped EPS bearer context does not cause the UE to discard the Authorized QoS rules IE and Authorized QoS flow descriptions IE included in the PDU SESSION MODICATION COMMAND message, if any.
- a) Semantic error in the mapped EPS bearer operation:
 - 1) operation code = "Create new EPS bearer" and there is already an existing mapped EPS bearer context with the same EPS bearer identity associated with any PDU session.
 - 2) operation code = "Delete existing EPS bearer" and there is no existing mapped EPS bearer context with the same EPS bearer identity associated with the PDU session that is being modified.
 - 3) operation code = "Modify existing EPS bearer" and there is no existing mapped EPS bearer context with the same EPS bearer identity associated with the PDU session that is being modified.
 - 4) operation code = "Create new EPS bearer" or "Modify existing EPS bearer" and the resulting mapped EPS bearer context has invalid mandatory parameters or missing mandatory parameters (e.g., mapped EPS QoS parameters or traffic flow template for a dedicated EPS bearer context).

In case 1, if the existing mapped EPS bearer context is associated with the PDU session that is being modified, the UE shall not diagnose an error, further process the create request and, if it was process successfully, delete the old EPS bearer context.

In case 2, the UE shall not diagnose an error, further process the delete request and, if it was processed successfully, consider the mapped EPS bearer context as successfully deleted.

Otherwise, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

- b) if the mapped EPS bearer context includes a traffic flow template, the UE shall check the traffic flow template for different types of TFT IE errors as follows:
 - 1) Semantic errors in TFT operations:
 - i) TFT operation = "Create new TFT" when there is already an existing TFT for the EPS bearer context.
 - ii) When the TFT operation is an operation other than "Create a new TFT" and there is no TFT for the EPS bearer context.
 - iii) TFT operation = "Delete packet filters from existing TFT" when it would render the TFT empty.
 - iv) TFT operation = "Delete existing TFT" for a dedicated EPS bearer context.

In case iv, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #41 "semantic error in the TFT operation".

In the other cases the UE shall not diagnose an error and perform the following actions to resolve the inconsistency:

In case i, the UE shall further process the new activation request to create a new TFT and, if it was processed successfully, delete the old TFT.

In case ii, the UE shall:

- process the new request and if the TFT operation is "Delete existing TFT" or "Delete packet filters from existing TFT", and if no error according to items 2, 3, and 4 was detected, consider the TFT as successfully deleted;
- process the new request as an activation request, if the TFT operation is "Add packet filters in existing TFT" or "Replace packet filters in existing TFT".

In case iii, if the packet filters belong to a dedicated EPS bearer context, the UE shall process the new deletion request and, if no error according to items 2, 3, and 4 was detected, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #41 "semantic error in the TFT operation".

In case iii, if the packet filters belong to the default EPS bearer context, the UE shall process the new deletion request and if no error according to items 2, 3, and 4 was detected then delete the existing TFT, this corresponds to using match-all packet filter for the default EPS bearer context.

- 2) Syntactical errors in TFT operations:
 - i) When the TFT operation = "Create new TFT", "Add packet filters in existing TFT", "Replace packet filters in existing TFT" or "Delete packet filters from existing TFT" and the packet filter list in the TFT IE is empty.
 - ii) TFT operation = "Delete existing TFT" or "No TFT operation" with a non-empty packet filter list in the TFT IE.
 - iii) TFT operation = "Replace packet filters in existing TFT" when the packet filter to be replaced does not exist in the original TFT.
 - iv) TFT operation = "Delete packet filters from existing TFT" when the packet filter to be deleted does not exist in the original TFT.
 - v) Void.
 - vi) When there are other types of syntactical errors in the coding of the TFT IE, such as a mismatch between the number of packet filters subfield, and the number of packet filters in the packet filter list when the TFT operation is "delete existing TFT" or "create new TFT", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list.

In case iii, the UE shall not diagnose an error, further process the replace request and, if no error according to items 3 and 4 was detected, include the packet filters received to the existing TFT.

In case iv, the UE shall not diagnose an error, further process the deletion request and, if no error according to items 3 and 4 was detected, consider the respective packet filter as successfully deleted.

Otherwise, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #42 "syntactical error in the TFT operation".

3) Semantic errors in packet filters:

- i) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.
- ii) When the resulting TFT, which is assigned to a dedicated EPS bearer context, does not contain any packet filter applicable for the uplink direction among the packet filters created on request from the network.

After sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #44 "semantic errors in packet filter(s)".

- 4) Syntactical errors in packet filters:
 - i) When the TFT operation = "Create new TFT", "Add packet filters to existing TFT", or "Replace packet filters in existing TFT" and two or more packet filters in the resultant TFT would have identical packet filter identifiers.
 - ii) When the TFT operation = "Create new TFT", "Add packet filters to existing TFT" or "Replace packet filters in existing TFT", and two or more packet filters among all TFTs associated with this PDN connection would have identical packet filter precedence values.
 - iii) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.

In case i, if two or more packet filters with identical packet filter identifiers are contained in the new request, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #45 "syntactical error in packet filter(s)". Otherwise, the UE shall not diagnose an error, further process the new request and, if it was processed successfully, delete the old packet filters which have the identical packet filter identifiers.

In case ii, if the old packet filters do not belong to the default EPS bearer context, the UE shall not diagnose an error, shall further process the new request and, if it was processed successfully, shall delete the old packet filters which have identical filter precedence values.

In case ii, if one or more old packet filters belong to the default EPS bearer context, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #45 "syntactical errors in packet filter(s)".

Otherwise, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #45 "syntactical error in packet filter(s)".

And if a new EPS bearer identity parameter in Authorized QoS flow descriptions IE is received for a QoS flow which can be transferred to EPS, the UE shall update the association between the QoS flow and the mapped EPS bearer context, based on the new EPS bearer identity and the mapped EPS bearer contexts. If the "Delete existing EPS bearer" operation code in the Mapped EPS bearer contexts IE was received, the UE shall discard the association between the QoS flow and the corresponding mapped EPS bearer context.

If:

- a) the UE detects different errors in the mapped EPS bearer contexts as described above which requires sending a PDU SESSION MODIFICATION REQUEST message to delete the erroneous mapped EPS bearer contexts; and
- b) optionally, if the UE detects errors in QoS rules that require to delete at least one QoS rule as described in subclause 6.3.2.4 which requires sending a PDU SESSION MODIFICATION REQUEST message to delete the erroneous QoS rules;

the UE, after sending the PDU SESSION MODIFICATION COMPLETE message for the ongoing PDU session modification procedure, may send a single PDU SESSION MODIFICATION REQUEST message to delete the

erroneous mapped EPS bearer contexts, and optionally to delete the erroneous QoS rules. The UE shall include a 5GSM cause IE in the PDU SESSION MODIFICATION REQUEST message.

NOTE 4: The 5GSM cause to use cannot be different from #41 "semantic error in the TFT operation", #42 "syntactical error in the TFT operation", #44 "semantic error in packet filter(s)", #45 "syntactical errors in packet filter(s)", #83 "semantic error in the QoS operation", #84 "syntactical error in the QoS operation", or #85 "Invalid mapped EPS bearer identity". The selection of a 5GSM cause is up to UE implementation.

Upon receipt of a PDU SESSION MODIFICATION COMMAND message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, if the UE accepts the PDU SESSION MODIFICATION COMMAND message, the UE considers the PDU session as modified and the UE shall create a PDU SESSION MODIFICATION COMPLETE message.

If the PDU SESSION MODIFICATION COMMAND message contains the PTI value allocated in the UE-requested PDU session modification procedure, the UE shall stop the timer T3581. The UE should ensure that the PTI value assigned to this procedure is not released immediately.

NOTE 5: The way to achieve this is implementation dependent. For example, the UE can ensure that the PTI value assigned to this procedure is not released during the time equal to or greater than the default value of timer T3591.

While the PTI value is not released, the UE regards any received PDU SESSION MODIFICATION COMMAND message with the same PTI value as a network retransmission (see subclause 7.3.1).

If the selected SSC mode of the PDU session is "SSC mode 3" and the PDU SESSION MODIFICATION COMMAND message includes 5GSM cause #39 "reactivation requested", the UE can provide to the upper layers the PDU session address lifetime if received in the PDU session address lifetime parameter of the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message. After the completion of the network-requested PDU session modification procedure:

- a) if the PDU session is an MA PDU session:
 - 1) established over both 3GPP access and non-3GPP access, and:
 - the UE is registered over both 3GPP access and non-3GPP access in the same PLMN:
 - the UE should re-initiate a UE-requested PDU session establishment procedure as specified in subclause 6.4.1 over the access the PDU SESSION MODIFICATION COMMAND message is received; or
 - the UE is registered over both 3GPP access and non-3GPP access in different PLMNs:
 - the UE should re-initiate UE-requested PDU session establishment procedures as specified in subclause 6.4.1 over both accesses. The UE should re-initiate the UE-requested PDU session establishment procedure over the access the PDU SESSION MODIFICATION COMMAND message is received first; or
 - 2) established over only single access:
 - the UE should re-initiate a UE-requested PDU session establishment procedure as specified in subclause 6.4.1 over the access the user plane resources were established; or
- b) if the PDU session is a single access PDU session:
 - the UE should re-initiate a UE-requested PDU session establishment procedure as specified in subclause 6.4.1 over the access the PDU session was associated with; and

for the re-initiated UE-requested PDU session establishment procedure(s) the UE should set a new PDU session ID different from the PDU session ID associated with the present PDU session and should set:

- a) the PDU session type to the PDU session type associated with the present PDU session;
- b) the SSC mode to the SSC mode associated with the present PDU session;
- c) the DNN to the DNN associated with the present PDU session; and

d) the S-NSSAI to the S-NSSAI associated with (if available in roaming scenarios) a mapped S-NSSAI if provided in the UE-requested PDU session establishment procedure of the present PDU session.

If the UE has indicated support for CIoT 5GS optimizations and receives a small data rate control parameters container in the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message, the UE shall store the small data rate control parameters value and use the stored small data rate control parameters value as the maximum allowed limit of uplink user data for the PDU session in accordance with 3GPP TS 23.501 [8]. If the UE has a previously stored small data rate control parameter value for the PDU session, the UE shall replace the stored small data rate control parameters value in the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message.

If the UE has indicated support for CIoT 5GS optimizations and receives an additional small data rate control parameters for exception data container in the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message, the UE shall store the additional small data rate control parameters for exception data value and use the stored additional small data rate control parameters for exception data value as the maximum allowed limit of uplink exception data for the PDU session in accordance with 3GPP TS 23.501 [8]. If the UE has a previously stored additional small data rate control parameters for exception data value for the PDU session, the UE shall replace the stored additional small data rate control parameters for exception data value for the PDU session with the received additional small data rate control parameters for exception data value in the Extended protocol configuration options IE in the PDU SESSION MODIFICATION COMMAND message.

The UE shall include the PDU session ID of the old PDU session which is about to get released in the old PDU session ID IE of the UL NAS TRANSPORT message that transports the PDU SESSION ESTABLISHMENT REQUEST message.

NOTE 6: The UE is expected to maintain the PDU session for which the PDU SESSION MODIFICATION COMMAND message including 5GSM cause #39 "reactivation requested" is received during the time indicated by the PDU session address lifetime value or until receiving an indication from upper layers (e.g. that the old PDU session is no more needed).

If the selected PDU session type of the PDU session is "Unstructured", the UE supports inter-system change from N1 mode to S1 mode, the UE does not support establishment of a PDN connection for the PDN type set to "non-IP" in S1 mode, and the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message contains an EPS bearer identity (EBI), then the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions. After sending the PDU SESSION MODIFICATION COMPLETE message for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

If the selected PDU session type of the PDU session is "Ethernet", the UE supports inter-system change from N1 mode to S1 mode, the UE does not support establishment of a PDN connection for the PDN type set to "non-IP" in S1 mode, the UE, the network or both of them do not support Ethernet PDN type in S1 mode, and the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message contains an EPS bearer identity (EBI), the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions. After sending the PDU SESSION MODIFICATION COMPLETE message for the ongoing PDU session modification procedure, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

For a UE which is registered for disaster roaming services and for a PDU session which is not a PDU session for emergency services:

- a) if the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION MODIFICATION COMMAND message contains an EPS bearer identity (EBI), then the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions; and
- b) the UE shall locally delete the contents of the Mapped EPS bearer contexts IE if it is received in the PDU SESSION MODIFICATION COMMAND message.

If the Always-on PDU session indication IE is included in the PDU SESSION MODIFICATION COMMAND message and:

- a) the value of the IE is set to "Always-on PDU session required", the UE shall consider the established PDU session as an always-on PDU session; or
- b) the value of the IE is set to "Always-on PDU session not allowed", the UE shall not consider the established PDU session as an always-on PDU session.

If the UE does not receive the Always-on PDU session indication IE in the PDU SESSION MODIFICATION COMMAND message:

- a) if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure upon an inter-system change from S1 mode to N1 mode for a PDN connection established when in S1 mode, the UE shall not consider the modified PDU session as an always-on PDU session; or
- b) otherwise:
 - if the UE has received the Always-on PDU session indication IE with the value set to "Always-on PDU session required" for this PDU session, the UE shall consider the PDU session as an always-on PDU session; or
 - 2) otherwise the UE shall not consider the PDU session as an always-on PDU session.

If the PDU SESSION MODIFICATION COMMAND message contains a Port management information container IE, the UE shall forward the contents of the Port management information container IE to the DS-TT (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]).

If the UE receives a Serving PLMN rate control IE in the PDU SESSION MODIFICATION COMMAND message, the UE shall store the Serving PLMN rate control IE value, replacing any existing value, and use the stored serving PLMN rate control value as the maximum allowed limit of uplink control plane user data for the corresponding PDU session in accordance with 3GPP TS 23.501 [8].

If the PDU SESSION MODIFICATION COMMAND message includes the Received MBS container IE, for each of the Received MBS informations:

- a) if MBS decision is set to "MBS join is accepted", the UE shall consider that it has successfully joined the multicast MBS session. The UE shall store the received TMGI and shall use it for any further operation on that multicast MBS session. The UE shall store the received MBS service area associated with the received TMGI, if any, and provide the received TMGI to lower layers. The UE may provide the MBS start time if it is included in the Received MBS information to upper layers;
- b) if MBS decision is set to "MBS join is rejected", the UE shall consider the requested join as rejected. The UE shall store the received MBS service area associated with the received TMGI, if any. If the received Rejection cause is set to "User is outside of local MBS service area", the UE shall not request to join the same multicast MBS session if neither current TAI nor CGI of the current cell is part of the received MBS service area. If the received Rejection cause is set to "multicast MBS session has not started or will not start soon" and an MBS back-off timer value is included with value that indicates neither zero nor deactivated, the UE shall start a back-off timer T3587 with the value provided in the MBS back-off timer value for the received TMGI, and shall not attempt to join the multicast MBS session with the same TMGI until the expiry of T3587. If the MBS back-off timer value indicates that this timer is deactivated, the UE shall not attempt to join the multicast MBS session with the same TMGI until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated. If the MBS back-off timer value indicates zero, the UE may attempt to join the multicast MBS session with the same TMGI;
- c) if the MBS decision is set to "Remove UE from multicast MBS session", the UE shall consider that it has successfully left the multicast MBS session, and if the received Rejection cause is set to "multicast MBS session is released", the UE shall consider the multicast MBS session as released. Then the UE shall indicate to lower layers to delete the stored TMGI;
- d) if the MBS decision is set to "MBS service area update", the UE shall store the received MBS service area associated with the received TMGI and replace the current MBS service area with the received one. or
- e) if the MBS decision is set to "MBS security information update", the UE shall replace the current MBS security information with the MBS security information received in the MBS security container associated with the received TMGI.

If the UE has indicated support for ECS configuration information provisioning in the SESSION ESTABLISHMENT REQUEST message or while in S1 mode, then upon receiving

- one or more ECS IPv4 address(es), ECS IPv6 address(es), ECS FQDN(s);
- one or more associated ECSP identifier(s);and
- optionally spatial validity conditions associated with the ECS address

in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message, then the UE shall pass them to the upper layers.

If the UE supports receiving DNS server addresses in protocol configuration options and receives one or more DNS server IPv4 address(es), one or more DNS server IPv6 address(es) or both of them, in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message, then the UE shall pass the received DNS server IPv4 address(es), if any, and the received DNS server IPv6 address(es), if any, to upper layers.

NOTE 7: The received DNS server address(es) replace previously provided DNS server address(es), if any.

If the UE supports the EAS rediscovery and receives:

- a) the EAS rediscovery indication without indicated impact; or
- b) the following:
 - 1) one or more EAS rediscovery indication(s) with impacted EAS IPv4 address range, if supported by the UE;
 - 2) one or more EAS rediscovery indication(s) with impacted EAS IPv6 address range, if supported by the UE;
 - 3) one or more EAS rediscovery indication(s) with impacted EAS FQDN, if supported by the UE; or
 - 4) any combination of the above;

in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message, then the UE shall pass the EAS rediscovery indication and the received impacted EAS IPv4 address range(s), if supported and included, the received EAS IPv6 address range(s), if supported and included, and the received EAS FQDN(s), if supported and included, to upper layers.

NOTE 8: The upper layers handle the EAS rediscovery indication and the impacted EAS IPv4 address range(s), if any, the impacted EAS IPv6 address range(s), if any, and the received EAS FQDN(s), if any, according to 3GPP TS 23.548 [10A].

Upon receipt of PDU SESSION MODIFICATION COMMAND message, if the network-requested PDU session modification procedure is triggered by a UE-requested PDU session modification procedure, the Service-level-AA container IE is included, then the UE shall forward the service-level-AA contents of the Service-level-AA container IE to the upper layers.

If the UE supports EDC and receives the EDC usage allowed indicator in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message, the UE shall indicate to upper layers that network allows the use of EDC.

If the UE supports EDC and receives the EDC usage required indicator in the Extended protocol configuration options IE of the PDU SESSION MODIFICATION COMMAND message, the UE shall indicate to upper layers that network requires the use of EDC.

NOTE 9: Handling of indication that network allows the use of EDC or that network requires the use of EDC is specified in 3GPP TS 23.548 [182].

The UE shall transport the PDU SESSION MODIFICATION COMPLETE message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5.

After sending the PDU SESSION MODIFICATION COMPLETE message, if the "Create new EPS bearer" operation code in the Mapped EPS bearer contexts IE was received in the PDU SESSION MODIFICATION COMMAND message and there is neither a corresponding Authorized QoS flow descriptions IE in the PDU SESSION MODIFICATION COMMAND message nor an existing QoS flow description corresponding to the EPS bearer identity

included in the mapped EPS bearer context, the UE shall send a PDU SESSION MODIFICATION REQUEST message including a Mapped EPS bearer contexts IE to delete the mapped EPS bearer context.

After sending the PDU SESSION MODIFICATION COMPLETE message, if for the PDU session being modified, there are mapped EPS bearer context(s) but none of them is associated with the default QoS rule, the UE shall locally delete the mapped EPS bearer context(s) and shall locally delete the stored EPS bearer identity (EBI) in all the QoS flow descriptions of the PDU session, if any.

If a port management information container needs to be delivered (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]), the UE shall include a Port management information container IE in the PDU SESSION MODIFICATION COMPLETE message.

Upon receipt of a PDU SESSION MODIFICATION COMPLETE message, the SMF shall stop timer T3591 and shall consider the PDU session as modified. If the selected SSC mode of the PDU session is "SSC mode 3" and the PDU SESSION MODIFICATION COMMAND message included 5GSM cause #39 "reactivation requested", the SMF shall start timer T3593. If the PDU Session Address Lifetime value is sent to the UE in the PDU SESSION MODIFICATION COMMAND message then timer T3593 shall be started with the same value, otherwise it shall use a default value. If the PDU SESSION MODIFICATION COMPLETE message contains a Port management information container IE, the SMF shall handle the contents of the Port management information container IE as specified in 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9].

6.3.2.4 Network-requested PDU session modification procedure not accepted by the UE

Upon receipt of a PDU SESSION MODIFICATION COMMAND message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, if the UE rejects the PDU SESSION MODIFICATION COMMAND message, the UE shall create a PDU SESSION MODIFICATION COMMAND REJECT message.

If the PDU SESSION MODIFICATION COMMAND message contains the PTI value allocated in the UE-requested PDU session modification procedure, the UE shall stop the timer T3581. The UE should ensure that the PTI value assigned to this procedure is not released immediately.

NOTE 1: The way to achieve this is implementation dependent. For example, the UE can ensure that the PTI value assigned to this procedure is not released during the time equal to or greater than the default value of timer T3591.

While the PTI value is not released, the UE regards any received PDU SESSION MODIFICATION COMMAND message with the same PTI value as a network retransmission (see subclause 7.3.1).

The UE shall set the 5GSM cause IE of the PDU SESSION MODIFICATION COMMAND REJECT message to indicate the reason for rejecting the PDU session modification.

The 5GSM cause IE typically indicates one of the following 5GSM cause values:

- #26 insufficient resources;
- #44 semantic error in packet filter(s);
- #45 syntactical error in packet filter(s);
- #83 semantic error in the QoS operation; or
- #84 syntactical error in the QoS operation.

If the selected SSC mode of the PDU session is "SSC mode 3" and the PDU SESSION MODIFICATION COMMAND messages includes 5GSM cause #39 "reactivation requested", while the UE does not have sufficient resources for initiating the PDU session establishment procedure as specified in subclause 6.4.1 then the UE shall set cause IE to #26 "insufficient resources".

If the PDU SESSION MODIFICATION COMMAND message includes a request to add a new authorized QoS rule, or a request to modify the authorized QoS rules, or both, and the UE decides to reject the request due to e.g. the supported number of authorized QoS rules or number of packet filters associated with a PDU session having reached the maximum number, then the UE shall set the 5GSM cause IE to #26 "insufficient resources".

NOTE 2: The maximum number of supported authorized QoS rules or packet filters associated with a PDU session is implementation specific.

If the PDU SESSION MODIFICATION COMMAND message includes a request to add a new authorized QoS flow description, or a request to modify the authorized QoS flow descriptions, or both and the UE decides to reject the request due to e.g. the supported number of authorized QoS flow descriptions, then the UE shall set the 5GSM cause IE to #26 "insufficient resources".

NOTE 3: The maximum number of supported authorized QoS flow descriptions associated with a PDU session is implementation specific.

If the PDU SESSION MODIFICATION COMMAND message includes the Authorized QoS rules IE, the UE shall process the QoS rules sequentially starting with the first QoS rule. The UE shall check the QoS rule and the QoS flow description provided in the PDU SESSION MODIFICATION COMMAND message for different types of errors as follows:

- NOTE 4: If an error is detected in a QoS rule or a QoS flow description which requires rejecting the PDU SESSION MODIFICATION COMMAND message, then the Authorized QoS rules IE, the Authorized QoS flow descriptions IE, the Mapped EPS bearer contexts IE and any other IE (RQ timer value IE, Always-on PDU session indication IE, etc) included in the PDU SESSION MODIFICATION COMMAND message are discarded, if any.
- a) Semantic errors in QoS operations:
 - 1) When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters" on the default QoS rule and the DQR bit is set to "the QoS rule is not the default QoS rule".
 - 2) When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters" on a QoS rule which is not the default QoS rule and the DQR bit is set to "the QoS rule is the default QoS rule".
 - 3) When the rule operation is "Create new QoS rule" and the DQR bit is set to "the QoS rule is the default QoS rule" when there's already a default QoS rule with different QoS rule identifier.
 - 4) When the rule operation is "Delete existing QoS rule" on the default QoS rule.
 - 5) When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters ", or "Modify existing QoS rule without modifying packet filters" and two or more QoS rules associated with this PDU session would have identical precedence values, and the UE is not in NB-N1 mode.
 - 6) When the rule operation is "Modify existing QoS rule and delete packet filters", the QoS rule is a QoS rule of a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the packet filter list in the resultant QoS rule is empty.
 - 7) When the rule operation is "Create new QoS rule", there is already an existing QoS rule with the same QoS rule identifier and the UE is not in NB-N1 mode.
 - 8) When the rule operation is "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters", the associated QoS rule does not exist and the UE is not in NB-N1 mode.
 - 9) When the rule operation is different than "Delete existing QoS rule", the DQR bit of the QoS rule is set to "the QoS rule is not the default QoS rule" and the UE is in NB-N1 mode.
 - 10) When the rule operation is "Create new QoS rule", the DQR bit is set to "the QoS rule is not the default QoS rule", and the PDU session type of the PDU session is "Unstructured".
 - 11) When the rule operation is "Delete existing QoS rule" and there is no existing QoS rule with the same QoS rule identifier.

- 12) When the flow description operation is "Create new QoS flow description", there is already an existing QoS flow description with the same QoS flow identifier and the UE is not in NB-N1 mode.
- 13) When the flow description operation is "Modify existing QoS flow description", the associated QoS flow description does not exist and the UE is not in NB-N1 mode.
- 14) When the flow description operation is "Delete existing QoS flow description" and there is no existing QoS flow description with the same QoS flow identifier.
- 15) When the flow description operation is different than "Delete existing QoS flow description", the QFI is not the same as the QFI of the default QoS rule and the UE is in NB-N1 mode.
- 16) When the flow description operation is "Create new QoS flow description" or "Modify existing QoS flow description", the QFI associated with the QoS flow description is not the same as the QFI of the default QoS rule, and the PDU session type of the PDU session is "Unstructured".
- 17) When the rule operation is "Modify existing QoS rule and add packet filters", the "packet filter list" field contains a match-all packet filter, the resultant QoS rule is the default QoS rule and there is already an existing match-all packet filter associated with the default QoS rule.
- 18) When the rule operation is "Create new QoS rule" and the DQR bit is set to "the QoS rule is not the default QoS rule", or the rule operation is "Modify existing QoS rule and add packet filters" on a QoS rule which is not the default QoS rule or "Modify existing QoS rule and replace all packet filters" on a QoS rule which is not the default QoS rule, and one match-all packet filter is to be associated with the resultant QoS rule.

In case 4, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #83 "semantic error in the QoS operation".

In case 5, if the old QoS rule (i.e. the QoS rule that existed before the PDU SESSION MODIFICATION COMMAND message was received) is not the default QoS rule, the UE shall not diagnose an error, shall further process the new request and, if it was processed successfully, shall delete the old QoS rule which has identical precedence value. Furthermore, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall send a PDU SESSION MODIFICATION REQUEST message with 5GSM cause #83 "semantic error in the QoS operation" to delete the QoS rule.

In case 5, if the old QoS rule (i.e. the QoS rule that existed before the PDU SESSION MODIFICATION COMMAND message was received) is the default QoS rule, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #83 "semantic error in the QoS operation".

In case 6, if the QoS rule is not the default QoS rule, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall send a PDU SESSION MODIFICATION REQUEST message with 5GSM cause #83 "semantic error in the QoS operation" to delete the QoS rule.

In case 6, if the QoS rule is the default QoS rule, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #83 "semantic error in the QoS operation".

In case 7, if the existing QoS rule is not the default QoS rule and the DQR bit of the new QoS rule is set to "the QoS rule is not the default QoS rule", the UE shall not diagnose an error, further process the create request and, if it was processed successfully, delete the old QoS rule. If the existing QoS rule is the default QoS rule or the DQR bit of the new QoS rule is set to "the QoS rule is the default QoS rule", the UE shall reject the PDU SESSION MODIFICATION COMMAND message with 5GSM cause #83 "semantic error in the QoS operation".

In case 9 or case 10, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall send a PDU SESSION MODIFICATION REQUEST message with 5GSM cause #83 "semantic error in the QoS operation" to delete the QoS rule.

In case 11, the UE shall not diagnose an error, further process the delete request and, if it was processed successfully, consider the respective QoS rule as successfully deleted.

In case 12, the UE shall not diagnose an error, further process the create request and, if it was processed successfully, delete the old QoS flow description.

In case 14, the UE shall not diagnose an error, further process the delete request and, if it was processed successfully, consider the respective QoS flow description as successfully deleted.

In case 15 or case 16, after sending the PDU SESSION MODIFICATION COMPLETE for the ongoing PDU session modification procedure, the UE shall send a PDU SESSION MODIFICATION REQUEST message with 5GSM cause #83 "semantic error in the QoS operation" to delete the QoS flow description.

Otherwise, the UE shall reject the PDU SESSION MODIFICATION COMMAND message with 5GSM cause #83 "semantic error in the QoS operation".

- b) Syntactical errors in QoS operations:
 - When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters" or "Modify existing QoS rule and delete packet filters", the PDU session type of the PDU session is IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the packet filter list in the QoS rule is empty.
 - 2) When the rule operation is "Delete existing QoS rule" or "Modify existing QoS rule without modifying packet filters" with a non-empty packet filter list in the QoS rule.
 - 3) When the rule operation is "Modify existing QoS rule and delete packet filters" and the packet filter to be deleted does not exist in the original QoS rule.
 - 4) Void.
 - 5) When there are other types of syntactical errors in the coding of the Authorized QoS rules IE or the Authorized QoS flow descriptions IE, such as: a mismatch between the number of packet filters subfield and the number of packet filters in the packet filter list when the rule operation is "delete existing QoS rule" or "create new QoS rule", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list (i.e., there is no QoS rule precedence subfield included in the QoS rule IE), the QoS Rule Identifier is set to "no QoS rule identifier assigned", or the QoS flow identifier is set to "no QoS flow identifier assigned".
 - 6) When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters" or "Modify existing QoS rule and replace all packet filters", the DQR bit is set to "the QoS rule is the default QoS rule", the PDU session type of the PDU session is "Unstructured", and the packet filter list in the QoS rule is not empty.
 - 7) When, the
 - A) rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters", "Modify existing QoS rule and replace all packet filters", "Modify existing QoS rule and delete packet filters" or "Modify existing QoS rule without modifying packet filters", there is no QoS flow description with a QFI corresponding to the QFI of the resulting QoS rule and the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1).
 - B) flow description operation is "Delete existing QoS flow description", and the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) with a QFI corresponding to the QFI of the QoS flow description that is deleted (i.e. there is no associated QoS flow description with the same QFI).
 - 8) When the flow description operation is "Create new QoS flow description" or "Modify existing QoS flow description", and the UE determines that there is a QoS flow description of a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) which lacks at least one of the mandatory parameters (i.e., GFBR uplink, GFBR downlink, MFBR uplink and MFBR downlink). If the QoS flow description does not include a 5QI, the UE determines this by using the QFI as the 5QI,

In case 3 the UE shall not diagnose an error, further process the deletion request and, if no error according to items c and d was detected, consider the respective packet filter as successfully deleted.

In case 6, after completion of the PDU session modification procedure, the UE shall send a PDU SESSION MODIFICATION REQUEST message with 5GSM cause #84 "syntactical error in the QoS operations" to delete all the packet filters for the default QoS rule.

In case 7, if the Authorized QoS rules IE contains at least one other valid QoS rule, the UE shall not diagnose an error and shall further process the request, if no error according to items c and d was detected. After completion of the PDU session modification procedure, the UE shall delete the QoS rule for which no corresponding QoS flow description is available and initiate UE-requested PDU session modification procedure with 5GSM cause #84 "syntactical error in the QoS operation" to delete the QoS rule for which it has deleted.

In case 8, if the default QoS rule is associated with the QoS flow description which lacks at least one of the mandatory parameters, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #84 "syntactical error in the QoS operation". Otherwise, if the Authorized QoS rules IE contains at least one other valid QoS rule or the QoS flow description IE contains at least one other valid QoS rule or the QoS flow description IE contains at least one other valid QoS flow description, the UE shall not diagnose an error and shall further process the request, if no error according to items c and d was detected. After completion of the PDU session modification procedure, the UE shall delete the QoS flow description which lacks at least one of the mandatory parameters and the associated QoS rule(s), if any, and initiate UE-requested PDU session modification procedure with 5GSM cause #84 "syntactical error in the QoS operation" to delete the QoS flow description and the associated QoS rule(s), if any, which it has deleted.

Otherwise the UE shall reject the PDU SESSION MODIFICATION COMMAND message with 5GSM cause #84 "syntactical error in the QoS operation".

- NOTE 5: It is not considered an error if the UE determines that after processing all QoS operations on QoS rules and QoS flow descriptions there is a QoS flow description that is not associated with any QoS rule and the UE is not in NB-N1 mode.
- c) Semantic errors in packet filters:
 - 1) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.

The UE shall reject the PDU SESSION MODIFICATION COMMAND message with 5GSM cause #44 "semantic error in packet filter(s)".

- d) Syntactical errors in packet filters:
 - When the rule operation is "Create new QoS rule", "Modify existing QoS rule and add packet filters" or "Modify existing QoS rule and replace all packet filters", and two or more packet filters in the resultant QoS rule would have identical packet filter identifiers.
 - 2) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.

In case 1, if two or more packet filters with identical packet filter identifiers are contained in the PDU SESSION MODIFICATION COMMAND message, the UE shall reject the PDU SESSION MODIFICATION COMMAND with 5GSM cause #45 "syntactical errors in packet filter(s)". Otherwise, the UE shall not diagnose an error, further process the PDU SESSION MODIFICATION COMMAND message and, if it was processed successfully, replace the old packet filter with the new packet filter which have the identical packet filter identifiers.

Otherwise the UE shall reject the PDU SESSION MODIFICATION COMMAND message with 5GSM cause #45 "syntactical errors in packet filter(s)".

If:

- a) the UE detects errors in QoS rules that require to delete at least one QoS rule as described above which requires sending a PDU SESSION MODIFICATION REQUEST message to delete the erroneous mapped EPS bearer contexts; and
- b) optionally, if the UE detects different errors in the mapped EPS bearer contexts as described in subclause 6.3.2.3 which requires sending a PDU SESSION MODIFICATION REQUEST message to delete the erroneous QoS rules;

the UE, after sending the PDU SESSION MODIFICATION COMPLETE message for the ongoing PDU session modification procedure, may send a single PDU SESSION MODIFICATION REQUEST message to delete the

erroneous QoS rules, and optionally to delete the erroneous mapped EPS bearer contexts. The UE shall include a 5GSM cause IE in the PDU SESSION MODIFICATION REQUEST message.

NOTE 6: The 5GSM cause to use cannot be different from #41 "semantic error in the TFT operation", #42 "syntactical error in the TFT operation", #44 "semantic error in packet filter(s)", #45 "syntactical errors in packet filter(s)", #83 "semantic error in the QoS operation", #84 "syntactical error in the QoS operation", or #85 "Invalid mapped EPS bearer identity". The selection of a 5GSM cause is up to UE implementation.

The UE shall transport the PDU SESSION MODIFICATION COMMAND REJECT message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5.

Upon receipt of a PDU SESSION MODIFICATION COMMAND REJECT message with 5GSM cause value in state PDU SESSION MODIFICATION PENDING, the SMF shall stop timer T3591, enter the state PDU SESSION ACTIVE and abort the PDU session modification procedure.

6.3.2.5 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3591.

On the first expiry of the timer T3591, the SMF shall resend the PDU SESSION MODIFICATION COMMAND message and shall reset and restart timer T3591. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3591, the SMF shall abort the procedure and enter the state PDU SESSION ACTIVE.

The SMF may continue to use the previous configuration of the PDU session or initiate the network-requested PDU session release procedure. If the SMF decides to continue to use the previous configuration of the PDU session and

- the Authorized QoS rules IE is included in the PDU SESSION MODIFICATION COMMAND message, the SMF may mark the corresponding authorized QoS rule(s) of the PDU session as to be synchronised with the UE; and
- ii) the Authorized QoS flow descriptions IE is included in the PDU SESSION MODIFICATION COMMAND message, the SMF may mark the corresponding authorized QoS flow description(s) of the PDU session as to be synchronised with the UE.
- b) Void.
- c) Collision of UE-requested PDU session release procedure and network-requested PDU session modification procedure.

If the SMF receives a PDU SESSION RELEASE REQUEST message during the network-requested PDU session modification procedure, and the PDU session indicated in the PDU SESSION RELEASE REQUEST message is the PDU session that the SMF had requested to modify, the SMF shall abort the PDU session modification procedure and proceed with the UE-requested PDU session release procedure.

d) Collision of UE-requested PDU session modification procedure and network-requested PDU session modification procedure.

If the network receives a PDU SESSION MODIFICATION REQUEST message during the network-requested PDU session modification procedure, and the PDU session indicated in the PDU SESSION MODIFICATION REQUEST message is the PDU session that the network had requested to modify, the network shall ignore the PDU SESSION MODIFICATION REQUEST message received in the state PDU SESSION MODIFICATION PENDING. The network shall proceed with the network-requested PDU session modification procedure as if no PDU SESSION MODIFICATION REQUEST message was received from the UE.

e) 5G access network cannot forward the message.

If the SMF determines based on content of the n2SmInfo attribute specified in 3GPP TS 29.502 [20A] that the DL NAS TRANSPORT message carrying the PDU SESSION MODIFICATION COMMAND message was not forwarded to the UE by the 5G access network, then the SMF shall abort the procedure and enter the state PDU SESSION ACTIVE.

f) 5G access network cannot forward the message due to handover.

If the SMF determines based on content of the n2SmInfo attribute specified in 3GPP TS 29.502 [20A] that the DL NAS TRANSPORT message carrying the PDU SESSION MODIFICATION COMMAND message was not forwarded to the UE by the 5G access network due to handover, then the SMF shall abort the procedure and enter the state PDU SESSION ACTIVE.

The SMF may re-initiate, up to a pre-configured number of times, the network-requested PDU session modification procedure when the SMF detects that the handover is completed successfully or has failed or at the expiry of the configured guard timer as specified in 3GPP TS 23.502 [9].

g) Collision of re-establishment of the user-plane resources and network-requested PDU session modification procedure for the same PDU session.

If the SMF receives an indication from the AMF to re-establish the user-plane resources during the network-requested PDU session modification procedure for the same PDU session, the network shall abort the network-requested PDU session modification procedure and proceed with re-establishment of the user-plane resources for the PDU session as specified in 3GPP TS 29.502 [20A] subclause 5.2.2.3.2.2.

- NOTE: After the completion of re-establishment of the user-plane resources for the PDU session, the SMF can reinitiate the network-requested PDU session modification procedure for the PDU session.
- h) Collision of UE-requested PDU session establishment procedure and network-requested PDU session modification procedure.

If the network receives a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing PDU session" or "existing emergency PDU session" during the network-requested PDU session modification procedure, and the PDU session ID indicated in the PDU SESSION ESTABLISHMENT REQUEST message is the PDU session that the network had requested to modify, the network shall abort the network-requested PDU session modification procedure and proceed with the UE-requested PDU session establishment procedure.

6.3.2.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) PDU session inactive for the received PDU session ID.

If the PDU session ID in the PDU SESSION MODIFICATION COMMAND message belongs to any PDU session in state PDU SESSION INACTIVE in the UE, the UE shall set the 5GSM cause IE to #43 "Invalid PDU session identity" in the 5GSM STATUS message, and set the PDU session ID to the received PDU session ID in the UL NAS TRANSPORT message as specified in subclause 5.4.5.

b) Collision of network-requested PDU session modification procedure and UE-requested PDU session modification procedure.

If the UE receives a PDU SESSION MODIFICATION COMMAND message during the UE-requested PDU session modification procedure, the PTI IE of the PDU SESSION MODIFICATION COMMAND message is set to "No procedure transaction identity assigned", and the PDU session indicated in the PDU SESSION MODIFICATION COMMAND message is the PDU session that the UE had requested to modify, the UE shall abort internally the UE-requested PDU session modification procedure, enter the state PDU SESSION ACTIVE and proceed with the network-requested PDU session modification procedure.

6.3.3 Network-requested PDU session release procedure

6.3.3.1 General

The purpose of the network-requested PDU session release procedure is to enable the network to release a PDU session or the user-plane resources on a single access of an MA PDU session.

6.3.3.2 Network-requested PDU session release procedure initiation

In order to initiate the network-requested PDU session release procedure, the SMF shall create a PDU SESSION RELEASE COMMAND message.

The SMF shall set the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message to indicate the reason for releasing the PDU session.

The 5GSM cause IE typically indicates one of the following 5GSM cause values:

#8 operator determined barring;

- #26 insufficient resources;
- #29 user authentication or authorization failed;
- #36 regular deactivation;
- #38 network failure;
- #39 reactivation requested;
- #46 out of LADN service area;
- #67 insufficient resources for specific slice and DNN;
- #69 insufficient resources for specific slice.

If the selected SSC mode of the PDU session is "SSC mode 2" and the SMF requests the relocation of SSC mode 2 PDU session anchor with different PDU sessions as specified in 3GPP TS 23.502 [9], the SMF shall include 5GSM cause #39 "reactivation requested".

If the network-requested PDU session release procedure is triggered by a UE-requested PDU session release procedure, the SMF shall set the PTI IE of the PDU SESSION RELEASE COMMAND message to the PTI of the PDU SESSION RELEASE REQUEST message received as part of the UE-requested PDU session release procedure and shall not include the Access type IE in the PDU SESSION RELEASE COMMAND.

If the network-requested PDU session release procedure is not triggered by a UE-requested PDU session release procedure, the SMF shall set the PTI IE of the PDU SESSION RELEASE COMMAND message to "No procedure transaction identity assigned".

If the PDU session ID included in PDU SESSION RELEASE COMMAND message is associated with one or more multicast MBS sessions and either the Access type IE is not included or the Access type IE indicates "3GPP access", the SMF shall consider the UE as removed from the associated multicast MBS sessions.

Based on the local policy and user's subscription data, if the SMF decides to release the PDU session after determining:

- a) the UE has moved between a tracking area in NB-N1 mode and a tracking area in WB-N1 mode;
- b) the UE has moved between a tracking area in NB-S1 mode and a tracking area in WB-N1 mode;
- c) the UE has moved between a tracking area in WB-S1 mode and a tracking area in NB-N1 mode; or
- d) a PDU session is not only for control plane CIoT 5GS optimization any more,

the SMF shall:

- a) include the 5GSM cause value #39 "reactivation requested" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message; or
- b) include a 5GSM cause value other than #39 "reactivation requested" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message.

NOTE: The included 5GSM cause value is up to the network implementation.

If the SMF receives UE presence in LADN service area from the AMF indicating that the UE is out of the LADN service area and the SMF decides to release the PDU session, the SMF shall include the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message. Upon receipt of the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause the PDU SESSION RELEASE COMMAND message. Upon receipt of the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message. Upon receipt of the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message.

The SMF may include a Back-off timer value IE in the PDU SESSION RELEASE COMMAND message when the 5GSM cause value #26 "insufficient resources" is included in the PDU SESSION RELEASE COMMAND message. If the 5GSM cause value is #26 "insufficient resources" and the PDU SESSION RELEASE COMMAND message is sent to a UE configured for high priority access in selected PLMN or the request type was set to "initial emergency request" or "existing emergency PDU session" for the establishment of the PDU session, the network shall not include a Back-off timer value IE.

The SMF may include a Back-off timer value IE in the PDU SESSION RELEASE COMMAND message when the 5GSM cause value #67 "insufficient resources for specific slice and DNN" is included in the PDU SESSION RELEASE COMMAND message. If the 5GSM cause value is #67 "insufficient resources for specific slice and DNN" and the PDU SESSION RELEASE COMMAND message is sent to a UE configured for high priority access in selected PLMN or the request type was set to "initial emergency request" or "existing emergency PDU session" for the establishment of the PDU session, the network shall not include a Back-off timer value IE.

The SMF may include a Back-off timer value IE in the PDU SESSION RELEASE COMMAND message when the 5GSM cause #69 "insufficient resources for specific slice" is included in the PDU SESSION RELEASE COMMAND message. If the 5GSM cause value is #69 "insufficient resources for specific slice" and the PDU SESSION RELEASE COMMAND message is sent to a UE configured for high priority access in selected PLMN or the request type was set to "initial emergency request" or "existing emergency PDU session" for the establishment of the PDU session, the network shall not include a Back-off timer value IE.

The SMF should include a Back-off timer value IE in the PDU SESSION RELEASE COMMAND message when the 5GSM cause value #29 "user authentication or authorization failed" is included in the PDU SESSION RELEASE COMMAND message.

If the service-level-AA procedure is triggered for the established PDU session for UAS services with re-authentication purpose, and the SMF is informed by the UAS-NF that UUAA-SM is unsuccessful or if the SMF receives UUAA revocation notification message from the UAS-NF as described in 3GPP TS 23.256 [6AB], the SMF shall transmit the PDU SESSION RELEASE COMMAND message to the UE, including:

- a) the service-level-AA response in the Service-level-AA container IE, with the SLAR field set to the value of "Service level authentication and authorization was not successful or service level authorization is revoked"; and
- b) the 5GSM cause value #29 "user authentication or authorization failed" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message.

If the PDU session was established for C2 communication and the SMF is informed by UAS-NF that C2 authorization is revoked, the SMF shall include:

- a) the service-level-AA response with the value of the C2AR field set to the "C2 authorization was not successful or C2 authorization is revoked" in the service-level-AA container IE of the PDU SESSION RELEASE COMMAND message, and
- b) the 5GSM cause value #29 "user authentication or authorization failed" in the 5GSM cause IE of the PDU SESSION RELEASE COMMAND message.

The SMF shall send:

- a) the PDU SESSION RELEASE COMMAND message; and
- b) the N1 SM delivery skip allowed indication:
 - if the SMF allows the AMF to skip sending the N1 SM container to the UE and the 5GSM cause IE is not set to #39 "reactivation requested"; or
 - 2) if the SMF allows the AMF to skip sending the N1 SM container to the UE and the Access type IE is not included

towards the AMF, and the SMF shall start timer T3592 (see example in figure 6.3.3.2.1).

UE		SMF	
◄	PDU SESSION RELEASE COMMAND		Start T3592
	PDU SESSION RELEASE COMPLETE		Stop T3592

Figure 6.3.3.2.1: Network-requested PDU session release procedure

6.3.3.3 Network-requested PDU session release procedure accepted by the UE

For a single access PDU session, upon receipt of a PDU SESSION RELEASE COMMAND message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE considers the PDU session as released and the UE shall create a PDU SESSION RELEASE COMPLETE message.

For an MA PDU session, upon receipt of the PDU SESSION RELEASE COMMAND, the UE shall behave as follows:

- a) if the PDU SESSION RELEASE COMMAND includes the Access type IE and the MA PDU session has userplane resources established on both 3GPP access and non-3GPP access, the UE shall consider the user-plane resources on the access indicated in the Access type IE as released and shall create a PDU SESSION RELEASE COMPLETE message. If the Access type IE indicates "3GPP access" and there is one or more multicast MBS sessions associated with the MA PDU session, the UE shall locally leave these associated multicast MBS sessions;
- b) if the PDU SESSION RELEASE COMMAND includes the Access type IE and the MA PDU session has userplane resources established on only the access indicated in the Access type IE, the UE shall consider the MA PDU session as released and shall create a PDU SESSION RELEASE COMPLETE message; and
- c) if the PDU SESSION RELEASE COMMAND does not include the Access type IE, the UE shall consider the MA PDU session as released and shall create a PDU SESSION RELEASE COMPLETE message.

If there is one or more multicast MBS sessions associated with the PDU session the UE considers as released, the UE shall locally leave these associated multicast MBS sessions.

If the PDU SESSION RELEASE COMMAND message contains the PTI value allocated in the UE-requested PDU session release procedure, the UE shall stop the timer T3582. The UE should ensure that the PTI value assigned to this procedure is not released immediately.

NOTE 1: The way to achieve this is implementation dependent. For example, the UE can ensure that the PTI value assigned to this procedure is not released during the time equal to or greater than the default value of timer T3592.

While the PTI value is not released, the UE regards any received PDU SESSION RELEASE COMMAND message with the same PTI value as a network retransmission (see subclause 7.3.1).

If the PDU SESSION RELEASE COMMAND message includes 5GSM cause #39 "reactivation requested", then after completion of the network-requested PDU session release procedure, the UE should re-initiate the UE-requested PDU session establishment procedure as specified in subclause 6.4.1 for:

- a) the PDU session type associated with the released PDU session;
- b) the SSC mode associated with the released PDU session;

- c) the DNN associated with the released PDU session; and
- d) the S-NSSAI associated with (if available in roaming scenarios) a mapped S-NSSAI if provided in the UErequested PDU session establishment procedure of the released PDU session.
- NOTE 2: User interaction is necessary in some cases when the UE cannot re-initiate the UE-requested PDU session establishment procedure automatically.

If the PDU SESSION RELEASE COMMAND message is received without the Back-off timer value IE or includes 5GSM cause #39 "reactivation requested", and the UE provided an S-NSSAI during the PDU session establishment, the UE shall stop timer T3585 if it is running for the S-NSSAI of the PDU session. If the UE did not provide an S-NSSAI during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3585 associated with no S-NSSAI if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running.

If the PDU SESSION RELEASE COMMAND message is received without the Back-off timer value IE or includes 5GSM cause #39 "reactivation requested", and the UE provided a DNN during the PDU session establishment, the UE shall stop timer T3396 if it is running for the DNN provided by the UE. If the UE did not provide a DNN during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3396 associated with no DNN if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3396 associated with no DNN if it is running.

If the PDU SESSION RELEASE COMMAND message is received without the Back-off timer value IE or includes 5GSM cause #39 "reactivation requested", and the UE provided an S-NSSAI and a DNN during the PDU session establishment, the UE shall stop timer T3584 if it is running for the [S-NSSAI of the PDU session, DNN] combination provided by the UE. If the UE did not provide an S-NSSAI during the PDU session establishment, the UE shall stop the timer T3584 associated with [no S-NSSAI, DNN] if it is running. If the UE did not provide a DNN during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3584 associated with [S-NSSAI of the PDU session, no DNN] combination, if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3584 associated with [S-NSSAI of the PDU session, no DNN] if it is running. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment and the request type was different from "existing emergency PDU session", the UE shall stop the timer T3584 associated with [S-NSSAI of the PDU session, no DNN] if it is running. If the UE provided neither a DNN nor an S-NSSAI during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop the timer T3584 associated with [No S-NSSAI, no DNN] if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3584 associated with [no S-NSSAI, no DNN] if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T3584 associated with [no S-NSSAI, no DNN] if it is running. If the PDU SESSION RELEASE COMMAND message was received for an emergency PDU session, the UE shall not stop the timer T35

- NOTE 3: If the PDU SESSION RELEASE COMMAND message is received without the Back-off timer value IE or includes 5GSM cause #39 "reactivation requested" for a PDU session, the UE provided a DNN (or no DNN) and an S-NSSAI (or no S-NSSAI) when the PDU session is established, timer T3396 associated with the DNN (or no DNN, if no DNN was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3396 and the timer T3584.
- NOTE 4: If the PDU SESSION RELEASE COMMAND message is received without the Back-off timer value IE or includes 5GSM cause #39 "reactivation requested" for a PDU session, the UE provided a DNN (or no DNN) and an S-NSSAI of the PDU session (or no S-NSSAI) when the PDU session is established, timer T3585 associated with the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, and timer T3584 associated with the DNN (or no DNN, if no DNN was provided by the UE) and the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) and the S-NSSAI of the PDU session (or no S-NSSAI, if no S-NSSAI was provided by the UE) is running, then the UE stops both the timer T3585 and the timer T3584.

If the PDU SESSION RELEASE COMMAND message includes 5GSM cause #26 "insufficient resources" and the Back-off timer value IE, the UE shall ignore the 5GSM congestion re-attempt indicator IE provided by the network, if any, and the UE shall take different actions depending on the timer value received for timer T3396 in the Back-off timer value:

a) If the timer value indicates neither zero nor deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the

timer value indicates neither zero nor deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE shall then start timer T3396 with the value provided in the Back-off timer value IE and:

- shall not send a PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN that was sent by the UE, until timer T3396 expires or timer T3396 is stopped; and
- 2) shall not send a PDU SESSION ESTABLISHMENT REQUEST message without an DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until timer T3396 expires or timer T3396 is stopped.

The UE shall not stop timer T3396 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the timer value indicates that this timer is deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE:
 - shall not send a PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the same DNN from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the same DNN, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for the same DNN from the network; and
 - 2) shall not send a PDU SESSION ESTABLISHMENT REQUEST message without a DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without a DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without a DNN provided by the UE, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established without a DNN provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for a non-emergency PDU session established without a DNN provided by the UE.

The timer T3396 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero, the UE:
 - shall stop timer T3396 associated with the corresponding DNN, if running, and may send a PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN; and
 - 2) if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN, if running, and may send a PDU SESSION ESTABLISHMENT REQUEST message without a DNN, or a PDU SESSION MODIFICATION REQUEST message without an DNN provided by the UE.

If the PDU SESSION RELEASE COMMAND message includes 5GSM cause #26 "insufficient resources" and the Back-off timer value IE is not included, then the UE may send a PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN or without a DNN.

When the timer T3396 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3396 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3396 is associated (if any) is not updated, then timer T3396 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3396 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3396 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3396 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the 5GSM cause value is #39 "reactivation requested", the UE shall ignore the Back-off timer value IE and 5GSM congestion re-attempt indicator IE provided by the network, if any.

If the 5GSM cause value is #67 "insufficient resources for specific slice and DNN" and the Back-off timer value IE is included, the UE shall take different actions depending on the timer value received for timer T3584 in the Back-off timer value:

- a) If the timer value indicates neither zero nor deactivated, and both an S-NSSAI and a DNN were provided by the UE during the PDU session establishment the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated, an S-NSSAI and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with [S-NSSAI of the PDU session, no DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated, no S-NSSAI and a DNN was provided during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated and neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency request" and different from "ue indicates neither zero nor deactivated and neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination, if it is running. The timer T3584 to be stopped includes the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running. The UE shall then start timer T3584 with the value provided in the Back-off timer value IE.
 - The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, DNN] combination, until timer T3584 expires or timer T3584 is stopped;
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, no DNN] combination, if no DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped;
 - 3) shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, DNN] combination, if no S-NSSAI was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped; and
 - 4) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, no DNN] combination, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped.

The UE shall not stop timer T3584 upon a PLMN change or inter-system change;

b) if the timer value indicates that this timer is deactivated:

- 1) if both S-NSSAI and DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, DNN] combination that was sent by the UE, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the [S-NSSAI of the PDU session, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the [S-NSSAI of the PDU session, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for the [S-NSSAI of the PDU session, DNN] combination from the network;
- 2) if an S-NSSAI was provided but a DNN was not provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, no DNN] combination, if no DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives an PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the [S-NSSAI of the PDU session, no DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established the [S-NSSAI of the PDU session, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for a non-emergency PDU session established for the [S-NSSAI of the PDU session, no DNN] combination from the network;
- 3) if an S-NSSAI was not provided but a DNN was provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message, or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, DNN], if no S-NSSAI was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives an PDU SESSION MODIFICATION COMMAND message for the [no S-NSSAI, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the [no S-NSSAI, DNN] combination from the network, or a PDU SESSION AUTHENTICATION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for the same [no S-NSSAI, DNN] combination from the network; and
- 4) if neither S-NSSAI nor DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, no DNN] combination, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives an PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network.

The timer T3584 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero:
 - if both S-NSSAI and DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the [S-NSSAI of the PDU session, DNN] combination;
 - 2) if an S-NSSAI was provided but a DNN was not provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the [S-NSSAI of the PDU session, no DNN] combination if the request type was different from "initial emergency request" and different from "existing emergency PDU session";
 - 3) if an S-NSSAI was not provided but a DNN was provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the [no S-NSSAI, DNN] combination; and
 - 4) if neither S-NSSAI nor DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the [no S-NSSAI, no DNN] combination if the request type was different from "initial emergency request" and different from "existing emergency PDU session".

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION RELEASE COMMAND message with the 5GSM cause value #67 "insufficient resources for specific slice and DNN", then the UE shall apply the timer T3584 for all the PLMNs. Otherwise, the UE shall apply the timer T3584 for the registered PLMN.

If the 5GSM cause value is #67 "insufficient resources for specific slice and DNN" and the Back-off timer value IE is not included, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same [S-NSSAI, DNN] combination.

When the timer T3584 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3584 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3584 is associated (if any) is not updated, then timer T3584 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3584 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3584 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3584 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the 5GSM cause value is #69 "insufficient resources for specific slice" and the Back-off timer value IE is included, the UE shall take different actions depending on the timer value received for timer T3585 in the Back-off timer value:

a) If the timer value indicates neither zero nor deactivated and an S-NSSAI was provided during the PDU session establishment, the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session, if it is running. If the timer value indicates neither zero nor deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585

applied for the registered PLMN, if running. The UE shall then start timer T3585 with the value provided in the Back-off timer value IE and:

- if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the S-NSSAI of the PDU session, until timer T3585 expires or timer T3585 is stopped; and
- 2) if the request type was different from "initial emergency request" and from "existing emergency PDU session", and an S-NSSAI was not provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, until timer T3585 expires or timer T3585 is stopped.

The UE shall not stop timer T3585 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated and an S-NSSAI was provided during the PDU session establishment, the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session, if it is running. If the timer value indicates that this timer is deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. In addition:
 - 1) if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST, or PDU SESSION MODIFICATION REQUEST with exception of those identified in subclause 6.4.2.1, for the S-NSSAI of the PDU session until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or a PDU SESSION MODIFICATION COMMAND message for the S-NSSAI of the PDU session from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the S-NSSAI of the PDU session from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for the S-NSSAI of the PDU session from the network; and
 - 2) if the request type was different from "initial emergency request" and from "existing emergency PDU session", and an S-NSSAI was not provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, , until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE or including 5GSM cause #39 "reactivation requested" for a non-emergency PDU session established without an S-NSSAI provided by the UE.

The timer T3585 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero:
 - if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), if running, and may send another PDU SESSION ESTABLISHMENT REQUEST, or PDU SESSION MODIFICATION REQUEST message for the S-NSSAI of the PDU session; and
 - 2) if no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), if running, and may send another PDU

SESSION ESTABLISHMENT REQUEST message without an S-NSSAI, or another PDU SESSION MODIFICATION REQUEST message without an S-NSSAI provided by the UE.

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION RELEASE COMMAND message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE shall apply the timer T3585 for all the PLMNs. Otherwise, the UE shall apply the timer T3585 for the registered PLMN.

If the 5GSM cause value is #69 "insufficient resources for specific slice" and the Back-off timer value IE is not included, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same S-NSSAI or without an S-NSSAI.

When the timer T3585 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3585 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3585 is associated (if any) is not updated, then timer T3585 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3585 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3585 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

- let t1 be the time remaining for T3585 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.
- NOTE 5: As described in this subclause, upon PLMN change or inter-system change, the UE does not stop the timer T3584 or T3585. This means the timer T3584 or T3585 can still be running or be deactivated for the given 5GSM procedure, the PLMN, the S-NSSAI and optionally the DNN combination when the UE returns to the PLMN or when it performs inter-system change back from S1 mode to N1 mode. Thus the UE can still be prevented from sending another PDU SESSION ESTABLISHMENT REQUEST or PDU SESSION MODIFICATION REQUEST message in the PLMN for the same S-NSSAI and optionally the same DNN.

Upon PLMN change, if T3584 is running or is deactivated for an S-NSSAI, a DNN, and old PLMN, but T3584 is not running and is not deactivated for the S-NSSAI, the DNN, and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI and the same DNN in the new PLMN.

Upon PLMN change, if T3585 is running or is deactivated for an S-NSSAI and old PLMN, but T3585 is not running and is not deactivated for the S-NSSAI and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI in the new PLMN.

If the PDU SESSION RELEASE COMMAND message includes 5GSM cause #29 "user authentication or authorization failed "and the Back-off timer value IE, the UE shall behave as follows:

- a) if the timer value indicates neither zero nor deactivated and:
 - if the UE provided a DNN and S-NSSAI to the network during the PDU session establishment, the UE shall start the back-off timer with the value provided in the Back-off timer value IE for the PDU session establishment procedure and [PLMN, DNN, (mapped) HPLMN S-NSSAI] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN and (mapped) HPLMN S-NSSAI in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
 - 2) if the UE provided a DNN to the network during the PDU session establishment, the UE shall start the back-off timer with the value provided in the Back-off timer value IE for the PDU session establishment procedure and [PLMN, DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the
selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder;

- 3) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment, it shall start the back-off timer accordingly for the PDU session establishment procedure and the [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination. Dependent on the combination, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder; or
- 4) if the UE did not provide a DNN to the network during the PDU session establishment, it shall start the back-off timer accordingly for the PDU session establishment procedure and the [PLMN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, no DNN] in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
- b) if the timer value indicates that this timer is deactivated and:
 - if the UE provided a DNN and S-NSSAI to the network during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN and (mapped) HPLMN S-NSSAI in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder;
 - 2) if the UE provided a DNN to the network during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder;
 - 3) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder; or
 - 4) if the UE did not provide a DNN to the network during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, no DNN] in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder; and
- c) if the timer value indicates zero, the UE may send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI], or [PLMN, no DNN, no S-NSSAI] in the current PLMN.

The UE shall not stop any back-off timer:

- a) upon a PLMN change;
- b) upon an inter-system change; or

c) upon registration over another access type.

If the PDU SESSION RELEASE COMMAND message includes:

- a) 5GSM cause #29 "user authentication or authorization failed"; and
- b) the service-level-AA response in the Service-level-AA container IE with the SLAR field set to the value of "Service level authentication and authorization was not successful or service level authorization is revoked",

the UE shall forward the service-level-AA response to the upper layers, so the UUAA authorization data is deleted as specified in 3GPP TS 33.256 [24B].

The UE shall transport the PDU SESSION RELEASE COMPLETE message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5.

Upon receipt of a PDU SESSION RELEASE COMPLETE message, the SMF shall stop timer T3592 and shall consider the PDU session as released.

6.3.3.4 N1 SM delivery skipped

If the PDU SESSION RELEASE COMMAND message was sent along the N1 SM delivery skip allowed indication towards AMF, then upon receipt of an indication that N1 SM delivery was skipped, the SMF shall stop timer T3592 and shall consider the PDU session as released.

6.3.3.5 Abnormal cases on the network side

The following abnormal cases can be identified:

a) Expiry of timer T3592.

The SMF shall, on the first expiry of the timer T3592, retransmit the PDU SESSION RELEASE COMMAND message and shall reset and start timer T3592. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3592, the SMF shall abort the procedure.

b) Collision of network-requested PDU session release procedure and UE-requested PDU session modification procedure.

When the SMF receives a PDU SESSION MODIFICATION REQUEST message during the network-requested PDU session release procedure, and the PDU session indicated in PDU SESSION MODIFICATION REQUEST message is the PDU session that the SMF had requested to release, the SMF shall ignore the PDU SESSION MODIFICATION REQUEST mossage and proceed with the PDU session release procedure.

c) Collision of network-requested PDU session release procedure and UE-requested PDU session release procedure.

If the SMF receives a PDU SESSION RELEASE REQUEST message after sending a PDU SESSION RELEASE COMMAND message to the UE, and the PDU session ID in the PDU SESSION RELEASE REQUEST message is the same as the PDU session ID in the PDU SESSION RELEASE COMMAND message, the SMF shall ignore the PDU SESSION RELEASE REQUEST message and proceed with the network-requested PDU session release procedure.

d) Collision of re-establishment of the user-plane resources and network-requested PDU session release procedure for the same PDU session.

If the SMF receives an indication from the AMF to re-establish the user-plane resources during the network-requested PDU session release procedure for the same PDU session, the SMF shall not re-establish the user-plane resources for the PDU session as specified in 3GPP TS 29.502 [20A] subclause 5.2.2.3.2.2 and proceed with the network-requested PDU session release procedure.

6.3.3.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) PDU session inactive for the received PDU session ID.

If the PDU session ID in the PDU SESSION RELEASE COMMAND message belongs to any PDU session in state PDU SESSION INACTIVE in the UE, the UE shall include the 5GSM cause #43 "Invalid PDU session identity" in the 5GSM STATUS message, and set the PDU session ID to the received PDU session ID in the UL NAS TRANSPORT message as specified in subclause 5.4.5.

b) User-plane resources of the MA PDU session on the access indicated in the Access type IE not established.

If the PDU session is an MA PDU session and has user-plane resources established on a single access different from the access indicated in the Access type IE, the UE shall not diagnose an error, further process the release command and consider the user-plane resources of the MA PDU session on the access indicated in the Access type IE as successfully released.

6.4 UE-requested 5GSM procedures

6.4.1 UE-requested PDU session establishment procedure

6.4.1.1 General

The purpose of the UE-requested PDU session establishment procedure is to establish a new PDU session with a DN, to perform handover of an existing PDU session between 3GPP access and non-3GPP access, to transfer an existing PDN connection in an untrusted non-3GPP access connected to the EPC to the 5GS, or to establish an MA PDU session to support ATSSS (see 3GPP TS 24.193 [13B]), or to relay the service associated with the RSC for 5G ProSe layer-3 UE-to-network relay (see 3GPP TS 24.554 [19E]). If accepted by the network, the PDU session enables exchange of PDUs between the UE and the DN.

The UE shall not request a PDU session establishment:

- a) for an LADN when the UE is located outside the LADN service area;
- b) to transfer a PDU session from non-3GPP access to 3GPP access when the 3GPP PS data off UE status is "activated" and the UE is not using the PDU session to send uplink IP packets for any of the 3GPP PS data off exempt services (see subclause 6.2.10);
- c) when the UE is in NB-N1 mode, the UE has indicated preference for user plane CIoT 5GS optimization, the network has accepted the use of user plane CIoT 5GS optimization for the UE, and the number of PDU sessions that currently has user-plane resources established equals to the UE's maximum number of supported user-plane resources;
- d) to transfer a PDU session from 3GPP access to non-3GPP access when the UE has indicated preference for control plane CIoT 5GS optimization, the network has accepted the use of control plane CIoT 5GS optimization for the UE, and the Control plane only indication IE was received in the PDU SESSION ESTABLISHMENT ACCEPT message; or
- e) to transfer a PDU session from the non-3GPP access to the 3GPP access when the UE is in NB-N1 mode, the UE has indicated preference for user plane CIoT 5GS optimization, the network has accepted the use of user plane CIoT 5GS optimization for the UE, and the number of PDU sessions that currently has user-plane resources established equals to the UE's maximum number of supported user-plane resources.

6.4.1.2 UE-requested PDU session establishment procedure initiation

In order to initiate the UE-requested PDU session establishment procedure, the UE shall create a PDU SESSION ESTABLISHMENT REQUEST message.

NOTE 0: When IMS voice is available over either 3GPP access or non-3GPP access, the "voice centric" UE in 5GMM-REGISTERED state will receive a request from upper layers to establish the PDU session for IMS signalling, if the conditions for performing an initial registration with IMS indicated in 3GPP TS 24.229 [14] subclause U.3.1.2 are satisfied.

If the UE requests to establish a new PDU session, the UE shall allocate a PDU session ID which is not currently being used by another PDU session over either 3GPP access or non-3GPP access. If the N5CW device supports 3GPP access and requests to establish a new PDU session via 3GPP access, the N5CW device shall refrain from allocating "PDU"

session identity value 15". If the TWIF acting on behalf of the N5CW device requests to establish a new PDU session, the TWIF acting on behalf of the N5CW device shall allocate the "PDU session identity value 15".

The UE shall allocate a PTI value currently not used and shall set the PTI IE of the PDU SESSION ESTABLISHMENT REQUEST message to the allocated PTI value.

If the UE is registered for emergency services over the current access, the UE shall not request establishing a nonemergency PDU session over the current access. If the UE is registered for emergency services over the current access it shall not request establishing an emergency PDU session over the non-current access except if the request is for transferring the emergency PDU session to the non-current access. Before transferring an emergency PDU session from non-3GPP access to 3GPP access, or before transferring a PDN connection for emergency bearer services from untrusted non-3GPP access connected to EPC to 3GPP access, the UE shall check whether emergency services are supported in the NG-RAN cell (either an NR cell or an E-UTRA cell) on which the UE is camping.

NOTE 1: Transfer of an existing emergency PDU session or PDN connection for emergency bearer services between 3GPP access and non-3GPP access is needed e.g. if the UE determines that the current access is no longer available.

If the UE requests to establish a new emergency PDU session, the UE shall include the PDU session type IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall set the IE to the IP version capability as specified in subclause 6.2.4.2.

If the UE requests to establish a new non-emergency PDU session with a DN, the UE shall include the PDU session type IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall set the IE to one of the following values: the IP version capability as specified in subclause 6.2.4.2, "Ethernet" or "Unstructured" based on the URSP rules or based on UE local configuration (see 3GPP TS 24.526 [19]).

NOTE 2: When the UE initiates the UE-requested PDU session establishment procedure to transfer an existing non-IP PDN connection in the EPS to the 5GS, the UE can use locally available information associated with the PDN connection to select the PDU session type between "Ethernet" and "Unstructured".

If the UE requests to establish a new non-emergency PDU session with a DN and the UE requests an SSC mode, the UE shall set the SSC mode IE of the PDU SESSION ESTABLISHMENT REQUEST message to the SSC mode. If the UE requests to establish a PDU session of "IPv4", "IPv6" or "IPv4v6" PDU session type, the UE shall either omit the SSC mode IE or set the SSC mode IE to "SSC mode 1", "SSC mode 2", or "SSC mode 3". If the UE requests to establish a PDU session of "IPv4", "IPv6" or "IPv4v6" PDU session de 3". If the UE requests to establish a PDU session of "Ethernet" or "Unstructured" PDU session type, the UE shall either omit the SSC mode IE or set the SSC mode 1" or "SSC mode 2". If the UE requests transfer of an existing PDN connection in the EPS to the 5GS or the UE requests transfer of an existing PDN connection in an untrusted non-3GPP access connected to the EPC to the 5GS, the UE shall set the SSC mode IE to "SSC mode 1".

If the UE requests to establish a new emergency PDU session, the UE shall set the SSC mode IE of the PDU SESSION ESTABLISHMENT REQUEST message to "SSC mode 1".

A UE supporting PDU connectivity service shall support SSC mode 1 and may support SSC mode 2 and SSC mode 3 as specified in 3GPP TS 23.501 [8].

If the UE requests to establish a new non-emergency PDU session with a DN, the UE may include the SM PDU DN request container IE with a DN-specific identity of the UE complying with network access identifier (NAI) format as specified in IETF RFC 7542 [37].

NOTE 3: The UE can avoid including both the SM PDU DN request container IE and the extended protocol configuration options IE with PAP/CHAP protocol identifiers in the PDU SESSION ESTABLISHMENT REQUEST message. The way to achieve this is implementation dependent.

If the UE requests to:

- a) establish a new PDU session;
- b) perform handover of an existing PDU session from non-3GPP access to 3GPP access;
- c) transfer an existing PDN connection in the EPS to the 5GS according to subclause 4.8.2.3.1;
- d) transfer an existing PDN connection in untrusted non-3GPP access connected to the EPC to the 5GS; or
- e) establish user plane resources over 3GPP access of an MA PDU session established over non-3GPP access only;

and the UE at the same time intends to join one or more multicast MBS sessions that is associated to the PDU session, the UE should include the Requested MBS container IE in the PDU SESSION ESTABLISHMENT REQUEST message. In that case, the UE shall set the MBS operation to "Join multicast MBS session" and include the multicast MBS session information(s) and shall set the Type of multicast MBS session ID for each of the multicast MBS session information to either "Temporary Mobile Group Identity (TMGI)" or "Source specific IP multicast address" depending on the type of the multicast MBS session ID available in the UE. Then the remaining values of each of the multicast MBS session information shall be set as following:

- a) if the Type of multicast MBS session ID is set to "Temporary Mobile Group Identity (TMGI)", the UE shall set the multicast MBS session ID to the TMGI; or
- b) if the Type of multicast MBS session ID is set to "Source specific IP multicast address for IPv4" or " Source specific IP multicast address for IPv6", the UE shall set the Source IP address information and the Destination IP address information to the corresponding values.

The UE should not request to join a multicast MBS session for local MBS service if neither current TAI nor CGI of the current cell is part of the MBS service area(s) of the multicast MBS session, if the UE has valid information of the MBS service area(s) of the multicast MBS session.

NOTE 4: The UE obtains the details of the multicast MBS session ID(s) i.e. TMGI, Source IP address information and Destination IP address information as a pre-configuration in the UE or during the MBS service announcement, which is out of scope of this specification.

The UE should set the RQoS bit to "Reflective QoS supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message if the UE supports reflective QoS and:

- a) the UE requests to establish a new PDU session of "IPv4", "IPv6", "IPv4v6" or "Ethernet" PDU session type;
- b) the UE requests to transfer an existing PDN connection in the EPS of "IPv4", "IPv6", "IPv4v6" or "Ethernet" PDN type or of "Non-IP" PDN type mapping to "Ethernet" PDU session type, to the 5GS; or
- c) the UE requests to transfer an existing PDN connection in an untrusted non-3GPP access connected to the EPC of "IPv4", "IPv6" or "IPv4v6" PDN type to the 5GS.
- NOTE 5: The determination to not request the usage of reflective QoS by the UE for a PDU session is implementation dependent.

The UE shall indicate the maximum number of packet filters that can be supported for the PDU session in the Maximum number of supported packet filters IE of the PDU SESSION ESTABLISHMENT REQUEST message if:

- a) the UE requests to establish a new PDU session of "IPv4", "IPv6", "IPv4v6", or "Ethernet" PDU session type, and the UE can support more than 16 packet filters for this PDU session;
- b) the UE requests to transfer an existing PDN connection in the EPS of "IPv4", "IPv6", "IPv4v6", or "Ethernet" PDN type or of "Non-IP" PDN type mapping to "Ethernet" PDU session type, to the 5GS and the UE can support more than 16 packet filters for this PDU session; or
- c) the UE requests to transfer an existing PDN connection in an untrusted non-3GPP access connected to the EPC of "IPv4", "IPv6" or "IPv4v6" PDN type to the 5GS and the UE can support more than 16 packet filters for this PDU session.

The UE shall include the Integrity protection maximum data rate IE in the PDU SESSION ESTABLISHMENT REQUEST message to indicate the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink and the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink.

The UE shall set the MH6-PDU bit to "Multi-homed IPv6 PDU session supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message if the UE supports multi-homed IPv6 PDU session and:

- a) the UE requests to establish a new PDU session of "IPv6" or "IPv4v6" PDU session type; or.
- b) the UE requests to transfer an existing PDN connection of "IPv6" or "IPv4v6" PDN type in the EPS or in an untrusted non-3GPP access connected to the EPC to the 5GS.

The UE shall set the EPT-S1 bit to "Ethernet PDN type in S1 mode supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message if the UE supports Ethernet PDN type in S1 mode and requests "Ethernet" PDU session type.

If the UE requests to establish a new PDU session as an always-on PDU session (e.g. because the PDU session is for time synchronization or TSC), the UE shall include the Always-on PDU session requested IE and set the value of the IE to "Always-on PDU session requested" in the PDU SESSION ESTABLISHMENT REQUEST message.

NOTE 6: Determining whether a PDU session is for time synchronization or TSC is UE implementation dependent.

If the UE has an emergency PDU session, the UE shall not perform the UE-requested PDU session establishment procedure to establish another emergency PDU session. The UE may perform the UE-requested PDU session establishment procedure to transfer an existing emergency PDU session or an existing PDN connection for emergency services.

If:

- a) the UE requests to perform handover of an existing PDU session between 3GPP access and non-3GPP access;
- b) the UE requests to perform transfer an existing PDN connection in the EPS to the 5GS; or
- c) the UE requests to perform transfer an existing PDN connection in an untrusted non-3GPP access connected to the EPC to the 5GS;

the UE shall:

- a) set the PDU session ID in the PDU SESSION ESTABLISHMENT REQUEST message and in the UL NAS TRANSPORT message to the stored PDU session ID corresponding to the PDN connection; and
- b) set the S-NSSAI in the UL NAS TRANSPORT message to the stored S-NSSAI associated with the PDU session ID of a non-emergency PDU session. The UE shall not request to perform handover of an existing nonemergency PDU session between 3GPP access and non-3GPP access if the S-NSSAI is not included in the allowed NSSAI for the target access.

If the N5CW device supports 3GPP access and requests to perform handover of an existing PDU session from non-3GPP access to 3GPP access, the N5CW device shall set the PDU session ID in the PDU SESSION ESTABLISHMENT REQUEST message and in the UL NAS TRANSPORT message to "PDU session identity value 15".

If the UE is registered to a network which supports ATSSS and the UE requests to establish a new PDU session the UE may allow the network to upgrade the requested PDU session to an MA PDU session. In order to allow the network to upgrade the requested PDU session information IE and shall set "MA PDU session network upgrade is allowed" in the MA PDU session information IE and shall set the request type to "initial request" in the UL NAS TRANSPORT message. If the UE is registered to a network which does not support ATSSS, the UE shall not perform the procedure to allow the network to upgrade the requested PDU session to an MA PDU session.

If the UE is registered to a network which supports ATSSS, the UE may request to establish an MA PDU session. If the UE requests to establish an MA PDU session, the UE shall set the request type to "MA PDU request" in the UL NAS TRANSPORT message. If the UE is registered to a network which does not support ATSSS, the UE shall not request to establish an MA PDU session.

When the UE is registered over both 3GPP access and non-3GPP access in the same PLMN and the UE requests to establish a new MA PDU session, the UE may provide an S-NSSAI in the UL NAS TRANSPORT message only if the S-NSSAI is included in the allowed NSSAIs of both accesses.

NOTE 7: If the UE requested DNN corresponds to an LADN DNN, the AMF does not forward the MA PDU session information IE to the SMF but sends the message back to the UE to inform of the unhandled request (see subclause 5.4.5.2.5).

If the UE is registered to a network which supports ATSSS and the UE has already an MA PDU session established over one access, the UE may perform the UE-requested PDU session establishment procedure to establish user-plane resources over the other access for the MA PDU session as specified in subclause 4.22 of 3GPP TS 23.502 [9] and the S-NSSAI associated with the MA PDU session is included in the allowed NSSAI of the other access. If the UE establishes user-plane resources over the other access for the other access for the MA PDU session, the UE shall:

- a) set the request type to "MA PDU request" in the UL NAS TRANSPORT message;
- b) set the PDU session ID to the stored PDU session ID corresponding to the established MA PDU session in the PDU SESSION ESTABLISHMENT REQUEST message and in the UL NAS TRANSPORT message; and

3GPP TS 24.501 version 17.16.0 Release 17

546

c) set the S-NSSAI in the UL NAS TRANSPORT message to the stored S-NSSAI associated with the PDU session ID.

If the UE requests to establish a new MA PDU session or if the UE requests to establish a new PDU session and the UE allows the network to upgrade the requested PDU session to an MA PDU session:

- a) if the UE supports ATSSS Low-Layer functionality with any steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "ATSSS Low-Layer functionality with any steering mode supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message;
- b) if the UE supports MPTCP functionality with any steering mode and ATSSS-LL functionality with only activestandby steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "MPTCP functionality with any steering mode and ATSSS-LL functionality with only active-standby steering mode supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message;
- c) if the UE supports MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message; and
- d) if a performance measurement function in the UE can perform access performance measurements using the QoS flow of the non-default QoS rule as specified in subclause 5.32.5 of 3GPP TS 23.501 [8], the UE shall set the APMQF bit to "Access performance measurements per QoS flow supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message.

Upon receipt of a PDU SESSION ESTABLISHMENT REQUEST message for MA PDU session establishment, the SMF shall check if the 5GSM capability IE in the PDU SESSION ESTABLISHMENT REQUEST message, includes:

- a) the ATSSS-ST bits set to "MPTCP functionality with any steering mode and ATSSS-LL functionality with only active-standby steering mode supported" and:
 - i) if the DNN configuration allows for the MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode, the SMF shall ensure that the established PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with any steering mode in the downlink and MPTCP with any steering mode and ATSSS-LL with only active-standby steering mode in the uplink; or
 - ii) if the DNN configuration allows for the MPTCP functionality with any steering mode and ATSSS-LL functionality with only active-standby steering mode, the SMF shall ensure that the established PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with only active-standby steering mode in the downlink and the uplink;
- b) the ATSSS-ST bits set to "ATSSS Low-Layer functionality with any steering mode allowed for ATSSS-LL supported" and if the DNN configuration allows for the ATSSS-LL functionality with any steering mode, the SMF shall ensure that the established PDU session has the capability of ATSSS-LL with any steering mode in the downlink and the uplink; or
- c) the ATSSS-ST bits set to "MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode allowed for ATSSS-LL supported" and if the DNN configuration allows for the MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode, the SMF shall ensure that the established PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with any steering mode in the downlink and the uplink.

If the UE requests to establish a new MA PDU session and the UE supports to establish a PDN connection as the user plane resource of an MA PDU session, the UE shall include the ATSSS request parameter in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message.

If the UE is registered to a network which does not support ATSSS and the UE has already an MA PDU session established over one access, the UE shall not attempt to establish user-plane resources for the MA PDU session over the network which does not support ATSSS as specified in subclause 4.22 of 3GPP TS 23.502 [9].

If the UE supports 3GPP PS data off, except for the transfer of a PDU session from non-3GPP access to 3GPP access and except for the establishment of user plane resources on the other access for the MA PDU session, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and include the 3GPP PS data off UE status. The UE behaves as described in subclause 6.2.10.

If the UE supports Reliable Data Service, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and include the Reliable Data Service request indicator. The UE behaves as described in subclause 6.2.15.

If the UE supports DNS over (D)TLS (see 3GPP TS 33.501 [24]), the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and include DNS server security information indicator and optionally, if the UE wishes to indicate which security protocol type(s) are supported by the UE, it may include the DNS server security protocol support.

NOTE 8: Support of DNS over (D)TLS is based on the informative requirements as specified in 3GPP TS 33.501 [24].

If:

- a) the PDU session type value of the PDU session type IE is set to "IPv4", "IPv6" or "IPv4v6";
- b) the UE indicates "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- c) the network indicates "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message;

the UE shall include the IP header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message.

If:

- a) the PDU session type value of the PDU session type IE is set to "Ethernet";
- b) the UE indicates "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- c) the network indicates "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message;

the UE shall include the Ethernet header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message.

If the UE supports transfer of port management information containers, the UE shall:

- a) set the TPMIC bit to "Transfer of port management information containers supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message;
- b) if the UE requests to establish a PDU session of "Ethernet" PDU session type, include the DS-TT Ethernet port MAC address IE in the PDU SESSION ESTABLISHMENT REQUEST message and set its contents to the MAC address of the DS-TT Ethernet port used for the PDU session;
- c) if the UE-DS-TT residence time is available at the UE, include the UE-DS-TT residence time IE and set its contents to the UE-DS-TT residence time; and
- d) if a Port management information container is provided by the DS-TT, include the Port management information container IE in the PDU SESSION ESTABLISHMENT REQUEST message.

NOTE 9: Only SSC mode 1 is supported for a PDU session which is for time synchronization or TSC.

If the UE supporting S1 mode supports receiving QoS rules with the length of two octets or QoS flow descriptions with the length of two octets via the Extended protocol configuration options IE, the UE shall include the QoS rules with the length of two octets support indicator or the QoS flow descriptions with the length of two octets support indicator, respectively, in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message.

If:

- the UE is operating in single-registration mode;
- the UE supports local IP address in traffic flow aggregate description and TFT filter in S1 mode; and
- the PDU session Type requested is different from "Unstructured".

the UE shall indicate the support of local address in TFT in S1 mode in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message.

If the W-AGF acting on behalf of the FN-RG requests to establish a PDU session of "IPv6" or "IPv4v6" PDU session type, the W-AGF acting on behalf of the FN-RG may include in the PDU SESSION ESTABLISHMENT REQUEST message the Suggested interface identifier IE with the PDU session type value field set to "IPv6" and containing the interface identifier for the IPv6 link local address associated with the PDU session suggested to be allocated to the FN-RG.

If the UE supports provisioning of ECS configuration information to the EEC in the UE, then the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall include the ECS configuration information provisioning support indicator.

If the UE supports receiving DNS server addresses in protocol configuration options, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and in the Extended protocol configuration options IE:

- a) if the UE requests to establish a PDU session of "IPv4" or "IPv4v6" PDU session type, the UE shall include the DNS server IPv4 address request; and
- b) if the UE requests to establish a PDU session of "IPv6" or "IPv4v6" PDU session type, the UE shall include the DNS server IPv6 address request.

If the UE supporting UAS services requests to establish a PDU session for C2 communication, the UE shall include the Service-level-AA container IE in the PDU SESSION ESTABLISHMENT REQUEST message. In the Service-level-AA container IE, the UE shall include:

- a) the service-level device ID with the value set to the CAA-level UAV ID of the UE; and
- b) if available, the service-level-AA payload with the value set to the C2 authorization payload and the service-level-AA payload type with the value set to "C2 authorization payload".
- NOTE 10: The C2 authorization payload in the service-level-AA payload can include the pairing information for C2 communication and the UAV flight authorization information.

If the UE supports the EAS rediscovery, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall include the EAS rediscovery support indication in the Extended protocol configuration options IE.

If the UE needs to include a PDU session pair ID based on the matching URSP rule or UE local configuration, the UE shall include the PDU session pair ID IE in the PDU SESSION ESTABLISHMENT REQUEST message. If the UE needs to include an RSN based on the matching URSP rule or UE local configuration, the UE shall include the RSN IE in the PDU SESSION ESTABLISHMENT REQUEST message.

If the UE is not registered for onboarding services in SNPN and needs PVS information, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and include the PVS information request in the Extended protocol configuration options IE.

If the UE supports the EDC, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall include the EDC support indicator in the Extended protocol configuration options IE.

If the UE supports a "destination MAC address range type" packet filter component and a "source MAC address range type" packet filter component, the UE shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT REQUEST message and shall include the MS support of MAC address range in 5GS indicator in the Extended protocol configuration options IE.

The UE shall transport:

a) the PDU SESSION ESTABLISHMENT REQUEST message;

- b) the PDU session ID of the PDU session being established, being handed over, being transferred, or been established as an MA PDU session;
- c) if the request type is set to:
 - 1) "initial request" or "MA PDU request" and the UE determined to establish a new PDU session or an MA PDU session based on either a URSP rule including one or more S-NSSAIs in the URSP (see subclause 6.2.9) or UE local configuration, according to subclause 4.2.2 of 3GPP TS 24.526 [19]:
 - i) if the UE is in the HPLMN or the subscribed SNPN, an S-NSSAI in the allowed NSSAI which corresponds to one of the S-NSSAI(s) in the matching URSP rule, if any, or else to the S-NSSAI(s) in the UE local configuration or in the default URSP rule, if any, according to the conditions given in subclause 4.2.2 of 3GPP TS 24.526 [19];
 - ii) if the UE is in a non-subscribed SNPN, the UE determined according to the conditions given in subclause 4.2.2 of 3GPP TS 24.526 [19] to establish a new PDU session or an MA PDU session based on a URSP rule including one or more S-NSSAIs, and the URSP rule is a part of a non-subscribed SNPN signalled URSP (see 3GPP TS 24.526 [19]):
 - A) an S-NSSAI in the allowed NSSAI, which is one of the S-NSSAI(s) in the URSP rule; and
 - B) a mapped S-NSSAI associated with the S-NSSAI in A); or
 - iii) otherwise:
 - A) one of the mapped S-NSSAI(s) which is equal to one of the S-NSSAI(s) in the matching URSP rule, if any, or else to the S-NSSAI(s) in the UE local configuration or in the default URSP rule, if any, according to the conditions given in subclause 4.2.2 of 3GPP TS 24.526 [19]; and
 - B) the S-NSSAI in the allowed NSSAI associated with the mapped S-NSSAI in A); or
- NOTE 11: When the UE is roaming, an AMF compliant with earlier versions of the specification can omit providing to the UE a mapped S-NSSAI for one or more S-NSSAIs in the allowed NSSAI and the UE then locally sets the mapped S-NSSAI as described in clause 4.6.2.1.
 - 1a) "initial request" and the UE determined to establish a new PDU session based on the PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE including an S-NSSAI in the UE policies for 5G ProSe UE-tonetwork relay UE as defined in 3GPP TS 24.555 [19F]:
 - i) in case of a non-roaming scenario, an S-NSSAI in the allowed NSSAI which corresponds to the S-NSSAI in the selected PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE, if any; or
 - ii) in case of a roaming scenario:
 - A) one of the mapped S-NSSAI(s) which corresponds to the S-NSSAI in the selected PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE, if any; and
 - B) the S-NSSAI in the allowed NSSAI associated with the S-NSSAI in A); or
 - "existing PDU session", an S-NSSAI, which is an S-NSSAI associated with the PDU session and (in roaming scenarios) a mapped S-NSSAI, with exception when S-NSSAI is not provided by the network in subclause 6.1.4.2;
- d) if the request type is set to:
 - "initial request" or "MA PDU request" and the UE determined to establish a new PDU session or an MA PDU session based on either a URSP rule including one or more DNNs in the URSP (see subclause 6.2.9) or UE local configuration, according to subclause 4.2.2 of 3GPP TS 24.526 [19], a DNN which corresponds to one of the DNN(s) in the matching URSP rule, if any, or else to the DNN(s) in the UE local configuration or in the default URSP rule, if any, according to the conditions given in subclause 4.2.2 of 3GPP TS 24.526 [19];
 - 1a) "initial request" and the UE determined to establish a new PDU session based on the PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE including a DNN in the UE policies for 5G ProSe UE-tonetwork relay UE as defined in 3GPP TS 24.555 [19F], a DNN which corresponds to the DNN in the selected PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE, if any; or

- 2) "existing PDU session", a DNN which is a DNN associated with the PDU session;
- e) the request type which is set to:
 - 1) "initial request", if the UE is not registered for emergency services and the UE requests to establish a new non-emergency PDU session;
 - 2) "existing PDU session", if the UE is not registered for emergency services and the UE requests:
 - i) handover of an existing non-emergency PDU session between 3GPP access and non-3GPP access;
 - ii) transfer of an existing PDN connection for non-emergency bearer services in the EPS to the 5GS; or
 - iii) transfer of an existing PDN connection for non-emergency bearer services in an untrusted non-3GPP access connected to the EPC to the 5GS;
 - 3) "initial emergency request", if the UE requests to establish a new emergency PDU session;
 - 4) "existing emergency PDU session", if the UE requests:
 - i) handover of an existing emergency PDU session between 3GPP access and non-3GPP access;
 - ii) transfer of an existing PDN connection for emergency bearer services in the EPS to the 5GS; or
 - iii) transfer of an existing PDN connection for emergency bearer services in an untrusted non-3GPP access connected to the EPC to the 5GS; or
 - 5) "MA PDU request", if:
 - i) the UE requests to establish an MA PDU session;
 - ii) the UE requests to establish user plane resources over other access of an MA PDU session established over one access only; or
 - iii) the UE performs inter-system change from S1 mode to N1 mode according to subclause 4.8.2.3.1 and requests transfer of a PDN connection which is a user plane resource of an MA PDU session; and
- f) the old PDU session ID which is the PDU session ID of the existing PDU session, if the UE initiates the UErequested PDU session establishment procedure upon receiving the PDU SESSION MODIFICATION COMMAND messages with the 5GSM cause IE set to #39 "reactivation requested";

using the NAS transport procedure as specified in subclause 5.4.5, and the UE shall start timer T3580 (see example in figure 6.4.1.2.1).

For bullet c) 1), if the matching URSP rule does not have an associated S-NSSAI, or if the UE does not have any matching URSP rule and there is no S-NSSAI in the UE local configuration or in the default URSP rule, the UE shall not provide any S-NSSAI in a PDU session establishment procedure.

For bullet c) 1a), if the selected PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE do not have an associated S-NSSAI, the UE shall not provide any S-NSSAI in a PDU session establishment procedure.

For bullet d) 1), if the matching URSP rule does not have an associated DNN, or if the UE does not have any matching URSP rule and there is no DNN in the UE local configuration or in the default URSP rule and:

- a) if the UE requests a connectivity to the default DNN for the S-NSSAI and the requested connectivity requires PAP/CHAP, the UE should provide a DNN in a PDU session establishment procedure; or
- b) otherwise, the UE shall not provide any DNN in a PDU session establishment procedure.

For bullet d) 1a), if the selected the PDU session parameters for 5G ProSe layer-3 UE-to-network relay UE do not have an associated DNN, the UE shall not provide any DNN in a PDU session establishment procedure.

If the request type is set to "initial emergency request" or "existing emergency PDU session" or the UE is registered for onboarding services in SNPN, neither DNN nor S-NSSAI is transported by the UE using the NAS transport procedure as specified in subclause 5.4.5.



Figure 6.4.1.2.1: UE-requested PDU session establishment procedure

Upon receipt of a PDU SESSION ESTABLISHMENT REQUEST message, a PDU session ID, optionally an S-NSSAI associated with (in roaming scenarios) a mapped S-NSSAI, optionally a DNN determined by the AMF, optionally a DNN selected by the network (if different from the DNN determined by the AMF), the request type, and optionally an old PDU session ID, the SMF checks whether connectivity with the requested DN can be established. If the requested DNN is not included, the SMF shall use the default DNN.

If the PDU session being established is a non-emergency PDU session, the request type is not set to "existing PDU session" and the PDU session authentication and authorization by the external DN is required due to local policy, the SMF shall check whether the PDU SESSION ESTABLISHMENT REQUEST message includes the SM PDU DN request container IE or the Service-level-AA container IE.

If the PDU session being established is a non-emergency PDU session, the request type is not set to "existing PDU session", the SM PDU DN request container IE is included in the PDU SESSION ESTABLISHMENT REQUEST message, the PDU session authentication and authorization by the external DN is required due to local policy and user's subscription data, and:

- a) the information for the PDU session authentication and authorization by the external DN in the SM PDU DN request container IE is compliant with the local policy and user's subscription data, the SMF shall proceed with the EAP Authentication procedure specified in 3GPP TS 33.501 [24] and refrain from accepting or rejecting the PDU SESSION ESTABLISHMENT REQUEST message until the EAP Authentication procedure finalizes; or
- b) the information for the PDU session authentication and authorization by the external DN in the SM PDU DN request container IE is not compliant with the local policy and user's subscription data, the SMF shall consider it as an abnormal case and proceed as specified in subclause 6.4.1.7.

If the PDU session being established is a non-emergency PDU session, the request type is not set to "existing PDU session", the SM PDU DN request container IE is not included in the PDU SESSION ESTABLISHMENT REQUEST message and the PDU session authentication and authorization by the external DN is required due to local policy and user's subscription data, the SMF shall proceed with the EAP Authentication procedure specified in 3GPP TS 33.501 [24] and refrain from accepting or rejecting the PDU SESSION ESTABLISHMENT REQUEST message until the EAP Authentication procedure finalizes.

If the SMF receives the old PDU session ID from the AMF and a PDU session exists for the old PDU session ID, the SMF shall consider that the request for the relocation of SSC mode 3 PDU session anchor with multiple PDU sessions as specified in 3GPP TS 23.502 [9] is accepted by the UE.

If the SMF receives the onboarding indication from the AMF, the SMF shall consider that the PDU session is established for onboarding services in SNPN.

If the UE has set the TPMIC bit to "Transfer of port management information containers supported" in the 5GSM capability IE of the PDU SESSION ESTABLISHMENT REQUEST message and has included a DS-TT Ethernet port MAC address IE (if the PDU session type is "Ethernet"), the Port management information container IE, and,

optionally, the UE-DS-TT residence time IE in the PDU SESSION ESTABLISHMENT REQUEST message, the SMF shall operate as specified in 3GPP TS 23.502 [9] subclause 4.3.2.2.1.

If requested by the upper layers, the UE supporting UAS services shall initiate a request to establish a PDU session for UAS services, where the UE:

- a) shall include the service-level device ID with the value set to the CAA-level UAV ID;
- b) if provided by the upper layers, shall include the service-level-AA server address, with the value set to the USS address; and
- c) if provided by the upper layers, shall include:
 - i) the service-level-AA payload type, with the value set to "UUAA payload"; and
 - ii) the service-level-AA payload, with the value set to UUAA payload,

in the Service-level-AA container IE of the PDU SESSION ESTABLISHMENT REQUEST message.

If the PDU session being established is a non-emergency PDU session, the request type is not set to "existing PDU session", the Service-level-AA container IE is included in the PDU SESSION ESTABLISHMENT REQUEST message, and

- a) the service-level authentication and authorization by the external DN is required due to local policy;
- b) there is a valid user's subscription information for the requested DNN or for the requested DNN and S-NSSAI; and
- c) the information for the service-level authentication and authorization by the external DN in the Service-level-AA container IE includes CAA-level UAV ID,

then the SMF shall proceed with the UUAA-SM procedure as specified in 3GPP TS 23.256 [6AB] and refrain from accepting or rejecting the PDU SESSION ESTABLISHMENT REQUEST message until the service-level authentication and authorization procedure is completed.

The UE supporting UAS services shall not request a PDU session establishment procedure to the same DNN (or no DNN, if no DNN was indicated by the UE) and the same S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) for which the UE has requested a service level authentication and authorization procedure which is ongoing.

If the PDU SESSION ESTABLISHMENT REQUEST message includes the PDU session pair ID IE, the RSN IE, or both, the SMF shall operate as specified in clause 5.33.2 of 3GPP TS 23.501 [8].

6.4.1.3 UE-requested PDU session establishment procedure accepted by the network

If the connectivity with the requested DN is accepted by the network, the SMF shall create a PDU SESSION ESTABLISHMENT ACCEPT message.

If the UE requests establishing an emergency PDU session, the network shall not check for service area restrictions or subscription restrictions when processing the PDU SESSION ESTABLISHMENT REQUEST message.

The SMF shall set the Authorized QoS rules IE of the PDU SESSION ESTABLISHMENT ACCEPT message to the authorized QoS rules of the PDU session and may include the authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message set to the authorized QoS flow descriptions of the PDU session.

NOTE 1: This is applicable also if the PDU session establishment procedure was initiated to perform handover of an existing PDU session between 3GPP access and non-3GPP access, and even if the authorized QoS rules and authorized QoS flow descriptions for source and target access of the handover are the same.

The SMF shall ensure that the number of the packet filters used in the authorized QoS rules of the PDU Session does not exceed the maximum number of packet filters supported by the UE for the PDU session. If the received request type is "initial emergency request", the SMF shall set the Authorized QoS flow descriptions IE according to the initial QoS parameters used for establishing emergency services configured in the SMF emergency configuration data.

SMF shall set the Authorized QoS flow descriptions IE to the authorized QoS flow descriptions of the PDU session, if:

- a) the Authorized QoS rules IE contains at least one GBR QoS flow;
- b) the QFI is not the same as the 5QI of the QoS flow identified by the QFI;
- c) the QoS flow can be mapped to an EPS bearer as specified in subclause 4.11.1 of 3GPP TS 23.502 [9]; or
- d) the QoS flow is established for the PDU session used for relaying, as specified in subclause 5.6.2.1 of 3GPP TS 23.304 [6E].
- NOTE 2: In cases other than above listed cases, it is up to the SMF implementation to include the authorized QoS flow description for the QoS flow in the Authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message.

If interworking with EPS is supported for the PDU session, the SMF shall set in the PDU SESSION ESTABLISHMENT ACCEPT message:

- a) the Mapped EPS bearer contexts IE to the EPS bearer contexts mapped from one or more QoS flows of the PDU session; and
- b) the EPS bearer identity parameter in the Authorized QoS flow descriptions IE to the EPS bearer identity corresponding to the QoS flow, for each QoS flow which can be transferred to EPS.

If the "Create new EPS bearer" operation code in the Mapped EPS bearer contexts IE was received, and there is no corresponding Authorized QoS flow descriptions IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall send a PDU SESSION MODIFICATION REQUEST message including a Mapped EPS bearer contexts IE to delete the mapped EPS bearer context. If the EPS bearer identity parameter in the Authorized QoS flow descriptions IE was received, the operation code is "Create new QoS flow description" and there is no corresponding Mapped EPS bearer contexts IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall not diagnose an error, and shall keep storing the association between the QoS flow and the corresponding EPS bearer identity.

Furthermore, the SMF shall store the association between the QoS flow and the mapped EPS bearer context, for each QoS flow which can be transferred to EPS.

The SMF shall set the selected SSC mode IE of the PDU SESSION ESTABLISHMENT ACCEPT message to:

- a) the received SSC mode in the SSC mode IE included in the PDU SESSION ESTABLISHMENT REQUEST message based on one or more of the PDU session type, the subscription and the SMF configuration;
- b) either the default SSC mode for the data network listed in the subscription or the SSC mode associated with the SMF configuration, if the SSC mode IE is not included in the PDU SESSION ESTABLISHMENT REQUEST message.
- NOTE 3: For bullet b), to avoid issues for UEs not supporting all SSC modes, the network operator can, in the subscription data and local configuration, include at least SSC mode 1 in the allowed SSC modes, and set the default SSC mode to "SSC mode 1" as per 3GPP TS 23.501 [8].

If the PDU session is an emergency PDU session, the SMF shall set the Selected SSC mode IE of the PDU SESSION ESTABLISHMENT ACCEPT message to "SSC mode 1". If the PDU session is a non-emergency PDU session of "Ethernet" or "Unstructured" PDU session type, the SMF shall set the Selected SSC mode IE to "SSC mode 1" or "SSC mode 2". If the PDU session is a non-emergency PDU session of "IPv4", "IPv6" or "IPv4v6" PDU session type, the SMF shall set the selected SSC mode 3".

If the PDU session is a non-emergency PDU session and the UE is not registered for onboarding services in SNPN, the SMF shall set the S-NSSAI IE of the PDU SESSION ESTABLISHMENT ACCEPT message to:

- a) the S-NSSAI of the PDU session; and
- b) the mapped S-NSSAI (in roaming scenarios).

The SMF shall set the Selected PDU session type IE of the PDU SESSION ESTABLISHMENT ACCEPT message to the selected PDU session type, i.e. the PDU session type of the PDU session.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv4v6", the SMF shall select "IPv4", "IPv6" or "IPv4v6" as the Selected PDU session type. If the subscription, the SMF configuration, or both, are limited to IPv4 only or IPv6 only for the DNN selected by the network, the SMF shall

include the 5GSM cause value #50 "PDU session type IPv4 only allowed", or #51 "PDU session type IPv6 only allowed", respectively, in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT ACCEPT message.

If the selected PDU session type is "IPv4", the SMF shall include the PDU address IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall set the PDU address IE to an IPv4 address is allocated to the UE in the PDU session.

If the selected PDU session type is "IPv6", the SMF shall include the PDU address IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall set the PDU address IE to an interface identifier for the IPv6 link local address allocated to the UE in the PDU session.

If the selected PDU session type is "IPv4v6", the SMF shall include the PDU address IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall set the PDU address IE to an IPv4 address and an interface identifier for the IPv6 link local address, allocated to the UE in the PDU session.

If the selected PDU session type of a PDU session established by the W-AGF acting on behalf of the FN-RG is "IPv4v6" or "IPv6", the SMF shall also indicate the SMF's IPv6 link local address in the PDU address IE of the PDU SESSION ESTABLISHMENT ACCEPT message.

If the PDU session is a non-emergency PDU session and the UE is not registered for onboarding services in SNPN, the SMF shall set the DNN IE of the PDU SESSION ESTABLISHMENT ACCEPT message to the DNN determined by the AMF of the PDU session.

The SMF shall set the Session-AMBR IE of the PDU SESSION ESTABLISHMENT ACCEPT message to the Session-AMBR of the PDU session.

If the selected PDU session type is "IPv4", "IPv40", "IPv4v6" or "Ethernet" and if the PDU SESSION ESTABLISHMENT REQUEST message includes a 5GSM capability IE with the RQoS bit set to "Reflective QoS supported", the SMF shall consider that reflective QoS is supported for QoS flows belonging to this PDU session and may include the RQ timer IE set to an RQ timer value in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the selected PDU session type is "IPv4", "IPv6", "IPv4v6" or "Ethernet" and if the PDU SESSION ESTABLISHMENT REQUEST message includes a Maximum number of supported packet filters IE, the SMF shall consider this number as the maximum number of packet filters that can be supported by the UE for this PDU session. Otherwise the SMF considers that the UE supports 16 packet filters for this PDU session.

The SMF shall consider that the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink and the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink are valid for the lifetime of the PDU session.

If the value of the RQ timer is set to "deactivated" or has a value of zero, the UE considers that RQoS is not applied for this PDU session.

NOTE 4: If the 5G core network determines that reflective QoS is to be used for a QoS flow, the SMF sends reflective QoS indication (RQI) to UPF to activate reflective QoS. If the QoS flow is established over 3GPP access, the SMF also includes reflective QoS Attribute (RQA) in QoS profile of the QoS flow during QoS flow establishment.

If the selected PDU session type is "IPv6" or "IPv4v6" and if the PDU SESSION ESTABLISHMENT REQUEST message includes a 5GSM capability IE with the MH6-PDU bit set to "Multi-homed IPv6 PDU session supported", the SMF shall consider that this PDU session is supported to use multiple IPv6 prefixes.

If the selected PDU session type is "Ethernet", the PDU SESSION ESTABLISHMENT REQUEST message includes a 5GSM capability IE with the EPT-S1 bit set to "Ethernet PDN type in S1 mode supported" and the network supports Ethernet PDN type in S1 mode, the SMF shall set the EPT-S1 bit of the 5GSM network feature support IE of the PDU SESSION ESTABLISHMENT ACCEPT message to "Ethernet PDN type in S1 mode supported".

If the DN authentication of the UE was performed and completed successfully, the SMF shall set the EAP message IE of the PDU SESSION ESTABLISHMENT ACCEPT message to an EAP-success message as specified in IETF RFC 3748 [34], provided by the DN.

Based on local policies or configurations in the SMF and the Always-on PDU session requested IE in the PDU SESSION ESTABLISHMENT REQUEST message (if available), if the SMF determines that either:

- a) the requested PDU session needs to be established as an always-on PDU session (e.g. because the PDU session is for time synchronization or TSC, for URLLC, or for both), the SMF shall include the Always-on PDU session indication IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall set the value to "Alwayson PDU session required"; or
- b) the requested PDU session shall not be established as an always-on PDU session and:
 - if the UE included the Always-on PDU session requested IE, the SMF shall include the Always-on PDU session indication IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall set the value to "Always-on PDU session not allowed"; or
 - ii) if the UE did not include the Always-on PDU session requested IE, the SMF shall not include the Always-on PDU session indication IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the PDU session is an MA PDU session, the SMF shall include the ATSSS container IE in the PDU SESSION ESTABLISHMENT ACCEPT message. The SMF shall set the content of the ATSSS container IE as specified in 3GPP TS 24.193 [13B]. If the UE requests to establish user plane resources over the second access of an MA PDU session which has already been established over the first access and the parameters associated with ATSSS previously provided to the UE are not to be updated, the "ATSSS container contents" shall not be included in the ATSSS container IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the PDU session is a single access PDU session containing the MA PDU session information IE with the value set to "MA PDU session network upgrade is allowed" and:

- a) if the SMF decides to establish a single access PDU session, the SMF shall not include the ATSSS container IE in the PDU SESSION ESTABLISHMENT ACCEPT message; or
- b) if the SMF decides to establish an MA PDU session, the SMF shall include the ATSSS container IE in the PDU SESSION ESTABLISHMENT ACCEPT message, which indicates to the UE that the requested single access PDU session was established as an MA PDU Session.

If the network decides that the PDU session is only for control plane CIoT 5GS optimization, the SMF shall include the control plane only indication in the PDU SESSION ESTABLISHMENT ACCEPT message.

If:

- a) the UE provided the IP header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message; and
- b) the SMF supports IP header compression for control plane CIoT 5GS optimization;

the SMF shall include the IP header compression configuration IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

If:

- a) the UE provided the Ethernet header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message; and
- b) the SMF supports Ethernet header compression for control plane CIoT 5GS optimization;

the SMF shall include the Ethernet header compression configuration IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the PDU SESSION ESTABLISHMENT REQUEST included the Requested MBS container IE with the MBS operation set to "Join multicast MBS session", the SMF:

a) shall include the TMGI for the multicast MBS session IDs that the UE is allowed to join, if any, in the Received MBS container IE, shall set the MBS decision to "MBS join is accepted" for each of those Received MBS information, may include the MBS start time to indicate the time when the multicast MBS session starts and shall include the MBS security container in each of those Received MBS information if security protection is applied for that multicast MBS session and the control plane security procedure is used as specified in annex W.4.1.2 in 3GPP TS 33.501 [24], and shall use separate QoS flows dedicated for multicast by including the Authorized QoS flow descriptions IE if no separate QoS flows dedicated for multicast exist or if the SMF wants to establish new QoS flows dedicated for multicast;

- NOTE 5: The network determines whether security protection applies or not for the multicast MBS session as specified in 3GPP TS 33.501 [24].
- b) shall include the TMGI for multicast MBS session IDs that the UE is not allowed to join, if any, in the Received MBS container IE, shall set the MBS decision to "MBS join is rejected" for each of those Received MBS information, shall set the Rejection cause for each of those Received MBS information with the reason of rejection, and if the Rejection cause is set to "multicast MBS session has not started or will not start soon", may include an MBS back-off timer value; and
- c) may include in the Received MBS container IE the MBS service area for each multicast MBS session and include in it the MBS TAI list, the NR CGI list or both, that identify the service area(s) for the local MBS service
- NOTE 6: For an multicast MBS session that has multiple MBS service areas, the MBS service areas are indicated to the UE using MBS service announcement as described in 3GPP TS 23.247 [53], which is out of scope of this specification.

in the PDU SESSION ESTABLISHMENT ACCEPT message. If the UE has set the Type of multicast MBS session ID to "Source specific IP multicast address" in the Requested MBS container IE for certain multicast MBS session(s) in the PDU SESSION MODIFICATION REQUEST message, the SMF shall include the Source IP address information and Destination IP address information in the Received MBS information together with the TMGI for each of those multicast MBS sessions.

- NOTE 7: Including the Source IP address information and Destination IP address information in the Received MBS information in that case is to allow the UE to perform the mapping between the requested multicast MBS session ID and the provided TMGI.
- NOTE 8: In SNPN, TMGI is used together with NID to identify an MBS Session.

If the request type is "existing PDU session", the SMF shall not perform network slice admission control for the PDU session, except for the following cases:

- a) when EPS counting is not required for the S-NSSAI of the PDU session for network slice admission control and the PDU session is established due to transfer the PDN connection from S1 mode to N1 mode in case of intersystem change; or
- b) handover of an existing PDU session between 3GPP access and non-3GPP access is performed.

The SMF shall send the PDU SESSION ESTABLISHMENT ACCEPT message.

Upon receipt of a PDU SESSION ESTABLISHMENT ACCEPT message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE shall stop timer T3580, shall release the allocated PTI value and shall consider that the PDU session was established.

If the PDU session establishment procedure was initiated to perform handover of an existing PDU session between 3GPP access and non-3GPP access, then upon receipt of the PDU SESSION ESTABLISHMENT ACCEPT message the UE shall locally delete any authorized QoS rules, authorized QoS flow descriptions, the session-AMBR and the parameters provided in the Protocol configuration options IE when in S1 mode or the Extended protocol configuration options IE stored for the PDU session before processing the new received authorized QoS rules, authorized QoS flow descriptions, the session-AMBR and the parameters provided in the Parameters provided in the parameters provided in the PDU session before processing the new received authorized QoS rules, authorized QoS flow descriptions, the session-AMBR and the parameters provided in the Extended protocol configuration options IE, if any.

NOTE 9: For the case of handover from 3GPP access to non-3GPP access, deletion of the QoS flow descriptions implies deletion of the associated EPS bearer identities, if any, and according to subclause 6.1.4.1 also deletion of the associated EPS bearer contexts. Regarding the reverse direction, for PDU sessions via non-3GPP access the network does not allocate associated EPS bearer identities (see 3GPP TS 23.502 [9], subclause 4.11.1.4.1).

If the PDU session establishment procedure was initiated to perform handover of an existing PDU session from 3GPP access to non-3GPP access and that existing PDU session is associated with one or more multicast MBS sessions, the UE shall locally leave the associated multicast MBS sessions and the SMF shall consider the UE as removed from the associated multicast MBS sessions.

For an MA PDU session already established on a single access, except for all those MA PDU sessions with a PDN connection established as a user-plane resource, upon receipt of PDU SESSION ESTABLISHMENT ACCEPT message over the other access:

- a) the UE shall delete the stored authorized QoS rules and the stored session-AMBR;
- b) if the authorized QoS flow descriptions IE is included in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall delete the stored authorized QoS flow descriptions; and
- c) if the mapped EPS bearer contexts IE is included in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall delete the stored mapped EPS bearer contexts.

The UE shall store the authorized QoS rules, and the session-AMBR received in the PDU SESSION ESTABLISHMENT ACCEPT message for the PDU session. The UE shall also store the authorized QoS flow descriptions if it is included in the Authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message for the PDU session.

If the number of the authorized QoS rules, the number of the packet filters, or the number of the authorized QoS flow descriptions associated with the PDU session have reached the maximum number supported by the UE upon receipt of a PDU SESSION ESTABLISHMENT ACCEPT message, then the UE may initiate the PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #26 "insufficient resources".

For a PDU session that is being established with the request type set to "initial request", "initial emergency request" or "MA PDU request", or a PDU session that is being transferred from EPS to 5GS and established with the request type set to "existing PDU session" or "existing emergency PDU session" or a PDU session that is being handed over between non-3GPP access and 3GPP access and established with the request type set to "existing PDU session" or "existing emergency PDU session" or "exi

- a) Semantic errors in QoS operations:
 - 1) When the rule operation is "Create new QoS rule", and the DQR bit is set to "the QoS rule is the default QoS rule" when there's already a default QoS rule.
 - 2) When the rule operation is "Create new QoS rule", and there is no rule with the DQR bit set to "the QoS rule is the default QoS rule".
 - 3) When the rule operation is "Create new QoS rule" and two or more QoS rules associated with this PDU session would have identical precedence values.
 - 4) When the rule operation is an operation other than "Create new QoS rule".
 - 5) When the rule operation is "Create new QoS rule", the DQR bit is set to "the QoS rule is not the default QoS rule", and the UE is in NB-N1 mode.
 - 6) When the rule operation is "Create new QoS rule" and two or more QoS rules associated with this PDU session would have identical QoS rule identifier values.
 - 7) When the rule operation is "Create new QoS rule", the DQR bit is set to "the QoS rule is not the default QoS rule", and the PDU session type of the PDU session is "Unstructured".
 - 8) When the flow description operation is an operation other than "Create new QoS flow description".
 - 9) When the flow description operation is "Create new QoS flow description", the QFI associated with the QoS flow description is not the same as the QFI of the default QoS rule and the UE is NB-N1 mode.
 - 10) When the flow description operation is "Create new QoS flow description", the QFI associated with the QoS flow description is not the same as the QFI of the default QoS rule, and the PDU session type of the PDU session is "Unstructured".

In case 4, case 5, or case 7 if the rule operation is for a non-default QoS rule, the UE shall send a PDU SESSION MODIFICATION REQUEST message to delete the QoS rule with 5GSM cause #83 "semantic error in the QoS operation".

In case 8, case 9, or case 10, the UE shall send a PDU SESSION MODIFICATION REQUEST message to delete the QoS flow description with 5GSM cause #83 "semantic error in the QoS operation".

Otherwise for all the cases above, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #83 "semantic error in the QoS operation".

- b) Syntactical errors in QoS operations:
 - 1) When the rule operation is "Create new QoS rule", the QoS rule is a QoS rule of a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type, and the packet filter list in the QoS rule is empty.
 - 2) When the rule operation is "Create new QoS rule", the DQR bit is set to "the QoS rule is the default QoS rule", the PDU session type of the PDU session is "Unstructured", and the packet filter list in the QoS rule is not empty.
 - 3) When there are other types of syntactical errors in the coding of the Authorized QoS rules IE or the Authorized QoS flow descriptions IE, such as: a mismatch between the number of packet filters subfield and the number of packet filters in the packet filter list when the rule operation is "delete existing QoS rule" or "create new QoS rule", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list (i.e., there is no QoS rule precedence subfield included in the QoS rule IE), the QoS Rule Identifier is set to "no QoS rule identifier assigned", or the QoS flow identifier is set to "no QoS flow identifier assigned".
 - 4) When, the rule operation is "Create new QoS rule", there is no QoS flow description with a QFI corresponding to the QFI of the resulting QoS rule and the UE determines, by using the QoS rule's QFI as the 5QI, that there is a resulting QoS rule for a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1).
 - 5) When the flow description operation is "Create new QoS flow description", and the UE determines that there is a QoS flow description of a GBR QoS flow (as described in 3GPP TS 23.501 [8] table 5.7.4-1) which lacks at least one of the mandatory parameters (i.e., GFBR uplink, GFBR downlink, MFBR uplink and MFBR downlink). If the QoS flow description does not include a 5QI, the UE determines this by using the QFI as the 5QI.

In case 1, case 3 or case 4, if the QoS rule is not the default QoS rule, the UE shall send a PDU SESSION MODIFICATION REQUEST message including a requested QoS rule IE to delete the QoS rule with 5GSM cause #84 "syntactical error in the QoS operation". Otherwise, if the QoS rule is the default QoS rule, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #84 "syntactical error in the QoS operation".

In case 2, if the QoS rule is the default QoS rule, the UE shall send a PDU SESSION MODIFICATION REQUEST message including a requested QoS rule IE to delete all the packet filters of the default QoS rule. The UE shall include the 5GSM cause #84 "syntactical error in the QoS operation".

In case 5, if the default QoS rule is associated with the QoS flow description which lacks at least one of the mandatory parameters, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #84 "syntactical error in the QoS operation". Otherwise, the UE shall send a PDU SESSION MODIFICATION REQUEST message to delete the QoS flow description which lacks at least one of the mandatory parameters and the associated QoS rule(s), if any, with 5GSM cause #84 "syntactical error in the QoS operation".

- NOTE 10:It is not considered an error if the UE determines that after processing all QoS operations on QoS rules and QoS flow descriptions there is a QoS flow description that is not associated with any QoS rule and the UE is not in NB-N1 mode.
- c) Semantic errors in packet filters:
 - 1) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.

If the QoS rule is the default QoS rule, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #44 "semantic error in packet filter(s)". Otherwise, the UE shall send a PDU SESSION MODIFICATION REQUEST message to delete the QoS rule with 5GSM cause #44 "semantic error in packet filter(s)".

- d) Syntactical errors in packet filters:
 - 1) When the rule operation is "Create new QoS rule" and two or more packet filters in the resultant QoS rule would have identical packet filter identifiers.

2) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.

If the QoS rule is the default QoS rule, the UE shall initiate a PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #45 "syntactical errors in packet filter(s)". Otherwise, the UE shall send a PDU SESSION MODIFICATION REQUEST message to delete the QoS rule with 5GSM cause #45 "syntactical errors in packet filter(s)".

If the Always-on PDU session indication IE is included in the PDU SESSION ESTABLISHMENT ACCEPT message and:

- a) the value of the IE is set to "Always-on PDU session required", the UE shall consider the established PDU session as an always-on PDU session; or
- b) the value of the IE is set to "Always-on PDU session not allowed", the UE shall not consider the established PDU session as an always-on PDU session.

The UE shall not consider the established PDU session as an always-on PDU session if the UE does not receive the Always-on PDU session indication IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

The UE shall store the mapped EPS bearer contexts, if received in the PDU SESSION ESTABLISHMENT ACCEPT message. Furthermore, the UE shall also store the association between the QoS flow and the mapped EPS bearer context, for each QoS flow which can be transferred to EPS, based on the received EPS bearer identity parameter in Authorized QoS flow descriptions IE and the mapped EPS bearer contexts. The UE shall check each mapped EPS bearer context for different types of errors as follows:

- NOTE 11: An error detected in a mapped EPS bearer context does not cause the UE to discard the Authorized QoS rules IE and Authorized QoS flow descriptions IE included in the PDU SESSION ESTABLISHMENT ACCEPT, if any.
- a) Semantic error in the mapped EPS bearer operation:
 - 1) When the operation code is an operation code other than "Create new EPS bearer".
 - 2) When the operation code is "Create new EPS bearer" and there is already an existing mapped EPS bearer context with the same EPS bearer identity associated with any PDU session.
 - 3) When the operation code is "Create new EPS bearer" and the resulting mapped EPS bearer context has invalid mandatory parameters or missing mandatory parameters (e.g., mapped EPS QoS parameters or traffic flow template for a dedicated EPS bearer context).

In case 2, if the existing mapped EPS bearer context is associated with the PDU session that is being established, the UE shall not diagnose an error, further process the create request and, if it was process successfully, delete the old EPS bearer context.

Otherwise, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

- b) if the mapped EPS bearer context includes a traffic flow template, the UE shall check the traffic flow template for different types of TFT IE errors as follows:
 - 1) Semantic errors in TFT operations:
 - i) When the TFT operation is an operation other than "Create new TFT"

The UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #41 "semantic error in the TFT operation".

- 2) Syntactical errors in TFT operations:
 - i) When the TFT operation = "Create new TFT" and the packet filter list in the TFT IE is empty.
 - ii) When there are other types of syntactical errors in the coding of the TFT IE, such as a mismatch between the number of packet filters subfield, and the number of packet filters in the packet filter list when the

TFT operation is "delete existing TFT" or "create new TFT", or the number of packet filters subfield is larger than the maximum possible number of packet filters in the packet filter list.

The UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message with to delete the mapped EPS bearer context 5GSM cause #42 "syntactical error in the TFT operation".

- 3) Semantic errors in packet filters:
 - i) When a packet filter consists of conflicting packet filter components which would render the packet filter ineffective, i.e. no IP packet will ever fit this packet filter. How the UE determines a semantic error in a packet filter is outside the scope of the present document.
 - ii) When the resulting TFT does not contain any packet filter which applicable for the uplink direction.

The UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #44 "semantic errors in packet filter(s)".

- 4) Syntactical errors in packet filters:
 - i) When the TFT operation = "Create new TFT" and two or more packet filters in the resultant TFT would have identical packet filter identifiers.
 - ii) When the TFT operation = "Create new TFT" and two or more packet filters in all TFTs associated with this PDN connection would have identical packet filter precedence values.
 - iii) When there are other types of syntactical errors in the coding of packet filters, such as the use of a reserved value for a packet filter component identifier.

In case ii, if the old packet filters do not belong to the default EPS bearer context, the UE shall not diagnose an error and shall delete the old packet filters which have identical filter precedence values.

In case ii, if one or more old packet filters belong to the default EPS bearer context, the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #45 "syntactical errors in packet filter(s)".

In cases i and iii the UE shall initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #45 "syntactical error in packet filter(s)".

If the UE detects different errors in the mapped EPS bearer contexts, QoS rules or QoS flow descriptions, the UE may send a single PDU SESSION MODIFICATION REQUEST message to delete the erroneous mapped EPS bearer contexts, QoS rules or QoS flow descriptions. In that case, the UE shall include a single 5GSM cause in the PDU SESSION MODIFICATION REQUEST message.

NOTE 12: The 5GSM cause to use cannot be different from: #41 "semantic error in the TFT operation", #42 "syntactical error in the TFT operation", #44 "semantic error in packet filter(s)", #45 "syntactical errors in packet filter(s)", #83 "semantic error in the QoS operation", #84 "syntactical error in the QoS operation", and #85 "Invalid mapped EPS bearer identity". The selection of a 5GSM cause is up to the UE implementation.

If there are mapped EPS bearer context(s) associated with a PDU session, but none of them is associated with the default QoS rule, the UE shall locally delete the mapped EPS bearer context(s) and shall locally delete the stored EPS bearer identity (EBI) in all the QoS flow descriptions of the PDU session, if any.

The UE shall only use the Control plane CIoT 5GS optimization for this PDU session if the Control plane only indication is included in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the UE requests the PDU session type "IPv4v6" and:

- a) the UE receives the selected PDU session type set to "IPv4" and does not receive the 5GSM cause value #50 "PDU session type IPv4 only allowed"; or
- b) the UE receives the selected PDU session type set to "IPv6" and does not receive the 5GSM cause value #51 "PDU session type IPv6 only allowed";

the UE may subsequently request another PDU session for the other IP version using the UE-requested PDU session establishment procedure to the same DNN (or no DNN, if no DNN was indicated by the UE) and the same S-NSSAI associated with (in roaming scenarios) a mapped S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) with a single address PDN type (IPv4 or IPv6) other than the one already activated.

If the UE requests the PDU session type "IPv4v6", receives the selected PDU session type set to "IPv4" and the 5GSM cause value #50 "PDU session type IPv4 only allowed", the UE shall not subsequently request another PDU session using the UE-requested PDU session establishment procedure to the same DNN (or no DNN, if no DNN was indicated by the UE) and the same S-NSSAI associated with (in roaming scenarios) a mapped S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) to obtain a PDU session type different from the one allowed by the network until any of the following conditions is fulfilled:

- a) the UE is registered to a new PLMN;
- b) the UE is switched off; or
- c) the USIM is removed or the entry in the "list of subscriber data" for the current SNPN is updated.

If the UE requests the PDU session type "IPv4v6", receives the selected PDU session type set to "IPv6" and the 5GSM cause value #51 "PDU session type IPv6 only allowed", the UE shall not subsequently request another PDU session using the UE-requested PDU session establishment procedure to the same DNN (or no DNN, if no DNN was indicated by the UE) and the same S-NSSAI associated with (in roaming scenarios) a mapped S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) to obtain a PDU session type different from the one allowed by the network until any of the following conditions is fulfilled:

- a) the UE is registered to a new PLMN;
- b) the UE is switched off; or
- c) the USIM is removed or the entry in the "list of subscriber data" for the current SNPN is updated.
- NOTE 13:For the 5GSM cause values #50 "PDU session type IPv4 only allowed", and #51 "PDU session type IPv6 only allowed", re-attempt in S1 mode for the same DNN (or no DNN, if no DNN was indicated by the UE) is only allowed using the PDU session type(s) indicated by the network.

If the selected PDU session type of the PDU session is "Unstructured" or "Ethernet", the UE supports inter-system change from N1 mode to S1 mode, the UE does not support establishment of a PDN connection for the PDN type set to "non-IP" in S1 mode, and the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message contains an EPS bearer identity (EBI), then the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions. Additionally the UE shall also initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

If the selected PDU session type of the PDU session is "Ethernet", the UE supports inter-system change from N1 mode to S1 mode, the UE does not support establishment of a PDN connection for the PDN type set to "non-IP" in S1 mode, the UE, the network or both of them do not support Ethernet PDN type in S1 mode, and the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message contains an EPS bearer identity (EBI), then the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions. Additionally, the UE shall also initiate a PDU session modification procedure by sending a PDU SESSION MODIFICATION REQUEST message to delete the mapped EPS bearer context with 5GSM cause #85 "Invalid mapped EPS bearer identity".

For a UE which is registered for disaster roaming services and for a PDU session which is not a PDU session for emergency services:

- a) if the parameters list field of one or more authorized QoS flow descriptions received in the Authorized QoS flow descriptions IE of the PDU SESSION ESTABLISHMENT ACCEPT message contains an EPS bearer identity (EBI), then the UE shall locally remove the EPS bearer identity (EBI) from the parameters list field of such one or more authorized QoS flow descriptions; and
- b) the UE shall locally delete the contents of the Mapped EPS bearer contexts IE if it is received in the PDU SESSION ESTABLISHMENT ACCEPT message.

If the UE receives an IPv4 Link MTU parameter, an Ethernet Frame Payload MTU parameter, an Unstructured Link MTU parameter, or a Non-IP Link MTU parameter in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall pass to the upper layer the received IPv4 link MTU size, the received Ethernet frame payload MTU size, the unstructured link MTU size, or the non-IP link MTU size.

- NOTE 14: The IPv4 link MTU size corresponds to the maximum length of user data packet that can be sent either via the control plane or via N3 interface for a PDU session of the "IPv4" PDU session type.
- NOTE 15: The Ethernet frame payload MTU size corresponds to the maximum length of a payload of an Ethernet frame that can be sent either via the control plane or via N3 interface for a PDU session of the "Ethernet" PDU session type.
- NOTE 16: The unstructured link MTU size correspond to the maximum length of user data packet that can be sent either via the control plane or via N3 interface for a PDU session of the "Unstructured" PDU session type.
- NOTE 17: A PDU session of "Ethernet" or "Unstructured" PDU session type can be transferred to a PDN connection of "non-IP" PDN type, thus the UE can request the non-IP link MTU parameter in the PDU session establishment procedure. The non-IP link MTU size corresponds to the maximum length of user data that can be sent either in the user data container in the ESM DATA TRANSPORT message or via S1-U interface as specified in 3GPP TS 24.301 [15].

If the 5G-RG receives an ACS information parameter in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message, the 5G-RG shall pass the ACS URL in the received ACS information parameter to the upper layer.

If the UE has indicated support for CIoT 5GS optimizations and receives a small data rate control parameters container in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall store the small data rate control parameters value and use the stored small data rate control parameters value as the maximum allowed limit of uplink user data for the PDU session in accordance with 3GPP TS 23.501 [8].

If the UE has indicated support for CIoT 5GS optimizations and receives an additional small data rate control parameters for exception data container in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall store the additional small data rate control parameters for exception data value and use the stored additional small data rate control parameters for exception data value as the maximum allowed limit of uplink exception data for the PDU session in accordance with 3GPP TS 23.501 [8].

If the UE has indicated support for CIoT 5GS optimizations and receives an initial small data rate control parameters container or an initial additional small data rate control parameters for exception data container in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall use these parameters for the newly established PDU Session. When the validity period of the initial parameters expire, the parameters received in a small data rate control parameters container or an additional small data rate control parameters for exception data container shall be used.

If the UE receives a Serving PLMN rate control IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall store the Serving PLMN rate control IE value and use the stored serving PLMN rate control value as the maximum allowed limit of uplink control plane user data for the corresponding PDU session in accordance with 3GPP TS 23.501 [8].

If the UE receives an APN rate control parameters container or an additional APN rate control for exception data parameters container in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall store these parameters and use them to limit the rate at which it generates uplink user data messages for the PDN connection corresponding to the PDU session if the PDU session is transferred to EPS upon inter-system change from N1 mode to S1 mode in accordance with 3GPP TS 24.301 [15]. The received APN rate control parameters and additional APN rate control for exception data parameters shall replace any previously stored APN rate control parameters and additional APN rate control for exception data parameters, respectively, for this PDN connection.

If the UE receives an initial APN rate control parameters container or an initial additional APN rate control for exception data parameters container in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall store these parameters in the APN rate control status and use them to limit the rate at which it generates exception data messages for the PDN connection corresponding to the PDU session if the PDU session is transferred to EPS upon inter-system change from N1 mode to S1 mode in accordance with 3GPP TS 24.301 [15]. The received APN rate control status shall replace any previously stored APN rate control status for this PDN connection.

- NOTE 18: In the PDU SESSION ESTABLISHMENT ACCEPT message, the SMF provides either APN rate control parameters container, or initial APN rate control parameters container, in the Extended protocol configuration options IE, but not both.
- NOTE 19:In the PDU SESSION ESTABLISHMENT ACCEPT message, the SMF provides either additional APN rate control for exception data parameters container, or initial additional APN rate control for exception data parameters container, in the Extended protocol configuration options IE, but not both.

If the network accepts the use of Reliable Data Service to transfer data for the PDU session, the network shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and include the Reliable Data Service accepted indicator. The UE behaves as described in subclause 6.2.15.

If the UE indicates support of DNS over (D)TLS by providing DNS server security information indicator to the network and optionally, if the UE wishes to indicate which security protocol type(s) are supported by the UE, providing the DNS server security protocol support and the network wants to enforce the use of DNS over (D)TLS, the network may include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and include the DNS server security information with length of two octets. Upon receiving the DNS server security information, the UE shall pass it to the upper layer. The UE shall use this information to send the DNS over (D)TLS (See 3GPP TS 33.501 [24]).

If the PDU SESSION ESTABLISHMENT REQUEST message includes the Service-level-AA container IE with the service-level device ID set to the CAA-level UAV ID, and the SMF is provided by the UAS-NF the successful UUAA-SM result and the CAA-level UAV ID, the SMF shall store the successful result together with the authorized CAA-level UAV ID and transmit the PDU SESSION ESTABLISHMENT ACCEPT message to the UE, where the PDU SESSION ESTABLISHMENT ACCEPT message shall include the Service-level-AA container IE containing:

- a) the service-level-AA response, with the SLAR field set to "Service level authentication and authorization was successful";
- b) the service-level device ID with the value set to the CAA-level UAV ID;
 - c) if a payload is received from the UAS-NF, the service-level-AA payload with the value set to the payload; and
- d) if a payload type associated with the payload is received from the UAS-NF, the service-level-AA payload type with the values set to the associated payload type.

NOTE 21:UAS security information can be included in the UUAA payload by the USS as specified in 3GPP TS 33.256 [24B].

If the network accepts the request of the PDU session establishment for C2 communication, the network shall send the PDU SESSION ESTABLISHMENT ACCEPT message including the Service-level-AA container IE containing:

- a) the service-level-AA response with the value of C2AR field set to the "C2 authorization was successful";
- b) if a payload is provided from the UAS-NF, the service-level-AA payload with the value set to the payload; and
- c) if a payload type associated with the payload is provided from the UAS-NF, the service-level-AA payload type with the value set to the payload type; and
- d) if the CAA-level UAV ID is provided from the UAS-NF, the service-level device ID with the value set to the CAA-level UAV ID.
- NOTE 22: The C2 authorization payload in the service-level-AA payload can include the C2 session security information.

Upon receipt of the PDU SESSION ESTABLISHMENT ACCEPT message of the PDU session for C2 communication, if the Service-level-AA container IE is included, the UE shall forward the service-level-AA contents of the Service-level-AA container IE to the upper layers.

The SMF may be configured with one or more PVS IP addresses or PVS names or both associated with the DNN and S-NSSAI used for onboarding services in SNPN, for configuration of SNPN subscription parameters in PLMN via the user plane, or for configuration of a UE via the user plane with credentials for NSSAA or PDU session authentication

NOTE 20:Support of DNS over (D)TLS is based on the informative requirements as specified in 3GPP TS 33.501 [24] and it is implemented based on the operator requirement.

and authorization procedure. If the PDU session was established for onboarding services in SNPN, or the PVS information request is included in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message establishing a PDU session providing connectivity for configuration of SNPN subscription parameters in PLMN via the user plane, the network may include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and include the PVS IP address(es) or the PVS name(s) or both associated with the DNN and S-NSSAI of the established PDU session, if available. If the PVS information request is included in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message establishing the PDU session providing connectivity for configuration of a UE via the user plane with credentials for PDU session authentication and authorization procedure, based on the subscribed DNN(s) and S-NSSAI(s) of the UE and the DNN and S-NSSAI of the established PDU session, the network should include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and include the PVS IP address(es) or the PVS name(s) or both, which are associated with the established PDU session and per subscribed DNN(s) and S-NSSAI(s) of the UE, if available. If the PVS information request is included in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message establishing the PDU session providing connectivity for configuration of a UE via the user plane with credentials for NSSAA, based on the subscribed S-NSSAI(s) of the UE and the S-NSSAI of the established PDU session, the network should include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and include the PVS IP address(es) or the PVS name(s) or both, which are associated with the established PDU session and per subscribed S-NSSAI(s) of the UE, if available.

NOTE 23: If the PVS information request is included in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message establishing a PDU session providing connectivity for configuration of SNPN subscription parameters in SNPN via the user plane by a UE which is not registered for onboarding services in SNPN, the SMF can include the PVS IP address(es) or the PVS name(s) or both, associated with the DNN and S-NSSAI of the established PDU session, if available, in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message.

The UE upon receiving one or more PVS IP address(es), if any, one or more the PVS name(s), if any, or both shall pass them to the upper layers.

NOTE 24:If several PVS IP addresses, several PVS name(s), or one or more PVS IP addresses and one or more PVS name(s) are received, how the UE uses this information is up to UE implementation.

If the UE indicates support for ECS configuration information provisioning by providing the ECS configuration information provisioning support indicator in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT REQUEST message, then the SMF may include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message with

- at least one of ECS IPv4 Address(es), ECS IPv6 Address(es), and ECS FQDN(s);
- at least one associated ECSP identifier; and
- optionally, spatial validity conditions associated with the ECS address.

The UE upon receiving one or more ECS IPv4 address(es), if any, ECS IPv6 address(es), if any, or ECS FQDN(s), if any, with the associated spatial validity condition, if any, and an ECSP identifier shall pass them to the upper layers.

NOTE 25: The IP address(es) and/or FQDN(s) are associated with the ECSP identifier and replace previously provided ECS configuration information associated with the same ECSP identifier, if any.

If the SMF needs to provide DNS server address(es) to the UE and the UE has provided the DNS server IPv4 address request, the DNS server IPv6 address request or both of them, in the PDU SESSION ESTABLISHMENT REQUEST message, then the SMF shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message with one or more DNS server IPv4 address(es), one or more DNS server IPv6 address(es) or both of them. If the UE supports receiving DNS server addresses in protocol configuration options and receives one or more DNS server IPv4 address(es), one or more DNS server IPv6 address(es) or both of them, in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message, then the UE shall pass the received DNS server IPv4 address(es), if any, and the received DNS server IPv6 address(es), if any, to upper layers.

NOTE 26: The received DNS server address(es) replace previously provided DNS server address(es), if any.

If the PDU SESSION ESTABLISHMENT ACCEPT message includes the Received MBS container IE, for each of the Received MBS information:

- a) if MBS decision is set to "MBS join is accepted", the UE shall consider that it has successfully joined the multicast MBS session. The UE shall store the received TMGI and shall use it for any further operation on that multicast MBS session. The UE shall store the received MBS service area associated with the received TMGI, if any, and provide the received TMGI to lower layers. The UE may provide the MBS start time if it is included in the Received MBS information to upper layers; or
- b) if MBS decision is set to "MBS join is rejected", the UE shall consider the requested join as rejected. The UE shall store the received MBS service area associated with the received TMGI, if any. If the received Rejection cause is set to "User is outside of local MBS service area", the UE shall not request to join the same multicast MBS session if neither current TAI nor CGI of the current cell is part of the received MBS service area. If the received Rejection cause is set to "multicast MBS session has not started or will not start soon" and an MBS back-off timer value is included with value that indicates neither zero nor deactivated, the UE shall start a back-off timer T3587 with the value provided in the MBS back-off timer value for the received TMGI, and shall not attempt to join the multicast MBS session with the same TMGI until the expiry of T3587. If the MBS back-off timer value indicates that this timer is deactivated, the UE shall not attempt to join the multicast MBS session with the same TMGI, or the Destination IP address information of the TMGI until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated. If the MBS back-off timer value indicates zero, the UE may attempt to join the multicast MBS session with the same TMGI.

If the PDU session is established for IMS signalling and the UE has requested P-CSCF IPv6 address or P-CSCF IPv4 address, the SMF shall include P-CSCF IP address(es) in the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message.

NOTE 27: The P-CSCF selection functionality is specified in subclause 5.16.3.11 of 3GPP TS 23.501 [8].

Upon receipt of the PDU SESSION ESTABLISHMENT ACCEPT message, if the UE included the PDU session pair ID in the PDU SESSION ESTABLISHMENT REQUEST message, the UE shall associate the PDU session with the PDU session pair ID. If the UE included the RSN in the PDU SESSION ESTABLISHMENT REQUEST message, the UE shall associate the PDU session with the RSN.

If the UE supports EDC and the network allows the use of EDC, the SMF shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message with the EDC usage allowed indicator. If the UE supports EDC and receives the EDC usage allowed indicator in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall indicate to upper layers that network allows the use of EDC.

If the UE supports EDC and the network requires the use of EDC, the SMF shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message with the EDC usage required indicator. If the UE supports EDC and receives the EDC usage required indicator in the Extended protocol configuration options IE of the PDU SESSION ESTABLISHMENT ACCEPT message, the UE shall indicate to upper layers that network requires the use of EDC.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a MS support of MAC address range in 5GS indicator in the Extended protocol configuration options IE, the SMF:

- a) shall consider that the UE supports a "destination MAC address range type" packet filter component and a "source MAC address range type" packet filter component; and
- b) if the SMF supports a "destination MAC address range type" packet filter component and a "source MAC address range type" packet filter component and enables the UE to request QoS rules with a "destination MAC address range type" packet filter component and a "source MAC address range type" packet filter component and a "source MAC address range type" packet filter component, shall include the Extended protocol configuration options IE in the PDU SESSION ESTABLISHMENT ACCEPT message and shall include the Network support of MAC address range in 5GS indicator in the Extended protocol configuration options IE.

If the PDU SESSION ESTABLISHMENT ACCEPT message includes a Network support of MAC address range in 5GS indicator in the Extended protocol configuration options IE, the UE shall consider that the network supports a "destination MAC address range type" packet filter component and a "source MAC address range type" packet filter component.

NOTE 28: Handling of indication that network allows the use of EDC or that network requires the use of EDC is specified in 3GPP TS 23.548 [182].

6.4.1.4 UE-requested PDU session establishment procedure not accepted by the network

6.4.1.4.1 General

If the connectivity with the requested DN is rejected by the network, the SMF shall create a PDU SESSION ESTABLISHMENT REJECT message.

The SMF shall set the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message to indicate the reason for rejecting the PDU session establishment.

The 5GSM cause IE typically indicates one of the following SM cause values:

#8 operator determined barring;

- #26 insufficient resources;
- #27 missing or unknown DNN;
- #28 unknown PDU session type;
- #29 user authentication or authorization failed;
- #31 request rejected, unspecified;
- #32 service option not supported;
- #33 requested service option not subscribed;
- #35 PTI already in use;
- #38 network failure;
- #39 reactivation requested;
- #46 out of LADN service area;
- #50 PDU session type IPv4 only allowed;
- #51 PDU session type IPv6 only allowed;
- #54 PDU session does not exist;
- #57: PDU session type IPv4v6 only allowed;
- #58: PDU session type Unstructured only allowed;
- #61: PDU session type Ethernet only allowed;
- #67 insufficient resources for specific slice and DNN;
- #68 not supported SSC mode;
- #69 insufficient resources for specific slice;
- #70 missing or unknown DNN in a slice;
- #82 maximum data rate per UE for user-plane integrity protection is too low;
- #86 UAS services not allowed; or
- #95 111 protocol errors.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv6", and the subscription, the SMF configuration, or both, are limited to IPv4 only for the requested DNN, the SMF shall include the 5GSM cause value #50 "PDU session type IPv4 only allowed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv6", and the subscription, the SMF configuration, or both, support none of "IPv4" and "IPv6" PDU session types for the requested DNN, the SMF shall include the 5GSM cause value #28 "unknown PDU session type" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv4", and the subscription, the SMF configuration, or both, are limited to IPv6 only for the requested DNN, the SMF shall include the 5GSM cause value #51 "PDU session type IPv6 only allowed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv4", and the subscription, the SMF configuration, or both, support none of "IPv4" and "IPv6" PDU session types for the requested DNN, the SMF shall include the 5GSM cause value #28 "unknown PDU session type" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "IPv4v6", and the subscription, the SMF configuration, or both, support none of "IPv4v6", "IPv4" and "IPv6" PDU session types for the requested DNN, the SMF shall include the 5GSM cause value #28 "unknown PDU session type" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message includes a PDU session type IE set to "Unstructured" or "Ethernet", and the subscription, the SMF configuration, or both, do not support the PDU session type for the requested DNN, the SMF shall include the 5GSM cause value #28 "unknown PDU session type" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message is to establish an MA PDU session and includes a PDU session type IE set to "Unstructured", and the SMF configuration does not support the PDU session type, the SMF shall include the 5GSM cause value #28 "unknown PDU session type" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message contains the SSC mode IE indicating an SSC mode not supported by the subscription, the SMF configuration, or both of them, and the SMF decides to rejects the PDU session establishment, the SMF shall include the 5GSM cause value #68 "not supported SSC mode" in the 5GSM cause IE and the SSC modes allowed by SMF in the Allowed SSC mode IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message is to establish an MA PDU session and MA PDU session is not allowed due to operator policy and subscription, and the SMF decides to reject the PDU session establishment, the SMF shall include the 5GSM cause value #33 "requested service option not subscribed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the PDU SESSION ESTABLISHMENT REQUEST message is identified to be for C2 communication and:

- a) does not include the Service-level-AA container IE with the service-level device ID set to the CAA-level UAV ID;
- b) does not include the Service-level-AA container IE with the service-level-AA payload and the Service-level-AA payload type; or-
- c) the SMF is informed by the UAS-NF that the UAS service is not allowed,

the SMF shall reject the PDU SESSION ESTABLISHMENT REQUEST message by transmitting a PDU SESSION ESTABLISHMENT REJECT message with 5GSM cause IE set to 5GSM cause value #86 "UAS services not allowed".

In 3GPP access, if the operator's configuration requires user-plane integrity protection for the PDU session and, the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink or the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink, or both, are lower than required by the operator's configuration, the SMF shall include the 5GSM cause value #82 "maximum data rate per UE for user-plane integrity protection is too low" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the UE requests a PDU session establishment for an LADN when the UE is located outside of the LADN service area, the SMF shall include the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

If the DN authentication of the UE was performed with the PDU session authentication and authorization procedure and completed unsuccessfully, the SMF shall include the 5GSM cause value #29 "user authentication or authorization failed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message and shall set the EAP message IE of the PDU SESSION ESTABLISHMENT REJECT message to an EAP-failure message as specified in IETF RFC 3748 [34], provided by the DN.

If the DN authentication of the UE was performed with the service-level authentication and authorization procedure and completed unsuccessfully, the SMF shall include the 5GSM cause value #29 "user authentication or authorization failed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message and shall include the service-level-AA response provided by DN in the Service-level-AA container IE of the PDU SESSION ESTABLISHMENT REJECT message.

Based on the local policy and user's subscription data, if a PDU session is being established with the request type set to "existing PDU session" and the SMF determines the UE has:

- a) moved between a tracking area in NB-N1 mode and a tracking area in WB-N1 mode;
- b) moved between a tracking area in NB-S1 mode and a tracking area in WB-N1 mode; or
- c) moved between a tracking area in WB-S1 mode and a tracking area in NB-N1 mode,

the SMF may reject the PDU SESSION ESTABLISHMENT REQUEST message and:

- a) include the 5GSM cause value #39 "reactivation requested" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message; or
- b) include a 5GSM cause value other than #39 "reactivation requested" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.
- NOTE 1: The included 5GSM cause value is up to the network implementation.

If the PDU session cannot be established due to resource unavailability in the UPF, the SMF shall include the 5GSM cause value #26 "insufficient resources" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

Based on the user's subscription data and the operator policy, if the SMF determines that the UUAA-SM procedure needs to be performed for a UE but the SMF does not receives the service-level device ID set to the CAA-level UAV ID in the Service-level-AA container IE of the PDU SESSION ESTABLISHMENT REQUEST message from the UE, the SMF shall include the 5GSM cause value #86 "UAS services not allowed" in the 5GSM cause IE of the PDU SESSION ESTABLISHMENT REJECT message.

The network may include a Back-off timer value IE in the PDU SESSION ESTABLISHMENT REJECT message.

If the 5GSM cause value is #26 "insufficient resources", #67 "insufficient resources for specific slice and DNN", or #69 "insufficient resources for specific slice" and the PDU SESSION ESTABLISHMENT REQUEST message was received from a UE configured for high priority access in selected PLMN or the request type provided during the PDU session establishment is set to "initial emergency request" or "existing emergency PDU session", the network shall not include a Back-off timer value IE.

If the 5GSM cause value is #29 "user authentication or authorization failed ", the network should include a Back-off timer value IE.

If the Back-off timer value IE is included and the 5GSM cause value is different from #26 "insufficient resources", #28 "unknown PDU session type", #46 "out of LADN service area", "#50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #54 "PDU session does not exist", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", #61 "PDU session type Ethernet only allowed", #67 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice", the network may include the Re-attempt indicator IE to indicate whether the UE is allowed to attempt a PDN connectivity procedure in the PLMN for the same DNN in S1 mode, and whether another attempt in S1 mode or in N1 mode is allowed in an equivalent PLMN.

If the 5GSM cause value is #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", or #61 "PDU session type Ethernet only allowed", the network may include the Re-attempt indicator IE without Back-off timer value IE to indicate whether the UE is allowed to attempt a PDU session establishment procedure in an equivalent PLMN in N1 mode using the same PDU session type for the same DNN (or no DNN, if no DNN was indicated by the UE) and the same S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE).

The SMF shall send the PDU SESSION ESTABLISHMENT REJECT message.

Upon receipt of a PDU SESSION ESTABLISHMENT REJECT message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE shall stop timer T3580 shall release the allocated PTI value and shall consider that the PDU session was not established.

If the PDU SESSION ESTABLISHMENT REQUEST message was sent with request type set to "initial emergency request" or "existing emergency PDU session" and the UE receives a PDU SESSION ESTABLISHMENT REJECT message, then the UE may:

- a) inform the upper layers of the failure of the procedure; or
- NOTE 2: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6].
- b) de-register locally, if not de-registered already, attempt initial registration for emergency services.

If the PDU SESSION ESTABLISHMENT REJECT message includes 5GSM cause #39 "reactivation requested" and the PDU session is being transferred from EPS to 5GS and established with the request type set to "existing PDU session", the UE should re-initiate the UE-requested PDU session establishment procedure as specified in subclause 6.4.1 for:

- a) the PDU session type associated with the transferred PDU session;
- b) the SSC mode associated with the transferred PDU session;
- c) the DNN associated with the transferred PDU session; and
- d) the S-NSSAI associated with (if available in roaming scenarios) a mapped S-NSSAI if provided in the UErequested PDU session establishment procedure of the transferred PDU session.

If the PDU SESSION ESTABLISHMENT REJECT message includes 5GSM cause #86 "UAS services not allowed" and the UE has not included the service-level device ID in the Service-level-AA container IE of the PDU SESSION ESTABLISHMENT REQUEST message and set the value to the CAA-level UAV ID:

- a) the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for UAS services without including the CAA-level UAV ID in the service-level device ID of the Service-level-AA container IE; and
- b) upon receipt of the request from the upper layers to establish a PDU session for UAS services, the UE shall initiate the UE-requested PDU session establishment procedure by including the service-level device ID in the Service-level-AA container IE of the PDU SESSION ESTABLISHMENT REQUEST message and set the value to the CAA-level UAV ID as specified in subclause 6.4.1.2.

6.4.1.4.2 Handling of network rejection due to congestion control

If:

- the 5GSM cause value #26 "insufficient resources" and the Back-off timer value IE are included in the PDU SESSION ESTABLISHMENT REJECT message; or
- an indication that the 5GSM message was not forwarded due to DNN based congestion control is received along a Back-off timer value and a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the 5GSM congestion re-attempt indicator or the Re-attempt indicator IE provided by the network, if any, and the UE shall take different actions depending on the timer value received for timer T3396 in the Back-off

timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.7):

- a) If the timer value indicates neither zero nor deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the timer value indicates neither zero nor deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE shall then start timer T3396 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN that was sent by the UE, until timer T3396 expires or timer T3396 is stopped; and
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without a DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until timer T3396 expires or timer T3396 is stopped.

The UE shall not stop timer T3396 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the timer value indicates that this timer is deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE:
 - 1) shall not send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the same DNN from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the same DNN from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the same DNN from the network; and
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without a DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without a DNN provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established without a DNN provided by the UE.

The timer T3396 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero, the UE:
 - shall stop timer T3396 associated with the corresponding DNN, if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN; and

2) if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN, if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN, or another PDU SESSION MODIFICATION REQUEST message without a DNN provided by the UE.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN or without a DNN as specified in clause 6.2.7.

When the timer T3396 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3396 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3396 is associated (if any) is not updated, then timer T3396 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3396 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3396 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3396 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If:

- the 5GSM cause value #67 "insufficient resources for specific slice and DNN" and the Back-off timer value IE are included in the PDU SESSION ESTABLISHMENT REJECT message; or
- an indication that the 5GSM message was not forwarded due to S-NSSAI and DNN based congestion control is received along a Back-off timer value and a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the Re-attempt indicator IE provided by the network, if any, and take different actions depending on the timer value received for timer T3584 in the Back-off timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.8):

- a) If the timer value indicates neither zero nor deactivated, the UE shall stop timer T3584 associated with the same [S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment, if it is running. If the timer value indicates neither zero nor deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with [S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment, if it is running. If the timer value indicates neither zero nor deactivated and no S-NSSAI was provided during the PDU session establishment, the UE shall stop timer T3584 associated with [no S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment, if it is running. If the timer value indicates neither zero nor deactivated and neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with [no S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment, if it is running. The timer T3584 to be stopped includes the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running. The UE shall then start timer T3584 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [S-NSSAI, DNN] combination that was sent by the UE, until timer T3584 expires or timer T3584 is stopped;
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST with exception of those identified in subclause 6.4.2.1, message for

the same [S-NSSAI, no DNN] combination that was sent by the UE, if no DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped;

- 3) shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, DNN] combination that was sent by the UE, if no S-NSSAI was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped; and
- 4) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, no DNN] combination that was sent by the UE, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped.

The UE shall not stop timer T3584 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated, the UE:
 - 1) shall stop timer T3584 associated with the same [S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [S-NSSAI, DNN] combination that was sent by the UE, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the same [S-NSSAI, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the same [S-NSSAI, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the same [S-NSSAI, DNN] combination from the network;
 - 2) shall stop timer T3584 associated with the same [S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [S-NSSAI, no DNN] combination that was sent by the UE, if no DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives an PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the same [S-NSSAI, no DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established for the same [S-NSSAI, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established for the same [S-NSSAI, no DNN] combination from the network;
 - 3) shall stop timer T3584 associated with the same [no S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message, or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, DNN] combination that was sent by the UE, if no S-NSSAI was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the same [no S-NSSAI, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established for the same [no S-NSSAI, no DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the same [no S-NSSAI, DNN] combination from the network; and

4) shall stop timer T3584 associated with the same [no S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [no S-NSSAI, no DNN] combination that was sent by the UE, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives an PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the same [no S-NSSAI, no DNN] combination from the network or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established for the same [no S-NSSAI, no DNN] combination from the network.

The timer T3584 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero, the UE:
 - shall stop timer T3584 associated with the same [S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same [S-NSSAI, DNN] combination;
 - 2) shall stop timer T3584 associated with the same [S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same [S-NSSAI, no DNN] combination if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session";
 - 3) shall stop timer T3584 associated with the same [no S-NSSAI, DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the same [no S-NSSAI, DNN] combination if no NSSAI was provided during the PDU session establishment; and
 - 4) shall stop timer T3584 associated with the same [no S-NSSAI, no DNN] combination as that the UE provided during the PDU session establishment (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the same [no S-NSSAI, no DNN] combination if neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session".

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION ESTABLISHMENT REJECT message with the 5GSM cause value #67 "insufficient resources for specific slice and DNN", then the UE shall apply the timer T3584 for all the PLMNs. Otherwise, the UE shall apply the timer T3584 for the registered PLMN.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same [S-NSSAI, DNN] combination, or for the same [S-NSSAI, no DNN] combination, or for the same [no S-NSSAI, no DNN] combination as specified in clause 6.2.8.

When the timer T3584 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3584 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3584 is associated (if any) is not updated, then timer T3584 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3584 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3584 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3584 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If:

- the 5GSM cause value #69 "insufficient resources for specific slice" and the Back-off timer value IE are included in the PDU SESSION ESTABLISHMENT REJECT message; or
- an indication that the 5GSM message was not forwarded due to S-NSSAI only based congestion control is received along a Back-off timer value and a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the Re-attempt indicator IE provided by the network, if any, and take different actions depending on the timer value received for timer T3585 in the Back-off timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.8):

- a) If the timer value indicates neither zero nor deactivated and an S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with the corresponding S-NSSAI, if it is running. If the timer value indicates neither zero nor deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. The UE shall then start timer T3585 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session for the same S-NSSAI that was sent by the UE, until timer T3585 expires or timer T3585 is stopped; and
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, if no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until timer T3585 expires or timer T3585 is stopped.

The UE shall not stop timer T3585 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated and an S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with the corresponding S-NSSAI, if it is running. If the timer value indicates that this timer is deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. The UE:
 - 1) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU

SESSION MODIFICATION REQUEST with exception of those identified in subclause 6.4.2.1, for a nonemergency PDU session for the same S-NSSAI until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session for the same S-NSSAI from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a nonemergency PDU session for the same S-NSSAI from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the same S-NSSAI from the network; and

2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, if no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established without an S-NSSAI provided by the UE.

The timer T3585 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero, the UE:
 - shall stop timer T3585 associated with the corresponding S-NSSAI (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same S-NSSAI; and
 - 2) if no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request " and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI, if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), or another PDU SESSION MODIFICATION REQUEST message without an S-NSSAI provided by the UE.

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION ESTABLISHMENT REJECT message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE shall apply the timer T3585 for all the PLMNs. Otherwise, the UE shall apply the timer T3585 for the registered PLMN. Additionally, if the 5GSM congestion re-attempt indicator IE with the CATBO bit set to "The back-off timer is applied in the current access type" is included in the PDU SESSION ESTABLISHMENT REJECT message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE shall apply the timer T3585 for the current access type. Otherwise, the UE shall apply the timer T3585 for both 3GPP access type.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same S-NSSAI or without an S-NSSAI as specified in clause 6.2.8.

When the timer T3585 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3585 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3585 is associated (if any) is not updated, then timer T3585 is kept running until it expires or it is stopped.
If the UE is switched off when the timer T3585 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" to which timer T3585 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3585 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 - t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

NOTE: As described in this subclause, upon PLMN change or inter-system change, the UE does not stop the timer T3584 or T3585. This means the timer T3584 or T3585 can still be running or be deactivated for the given 5GSM procedure, the PLMN, the S-NSSAI and optionally the DNN combination when the UE returns to the PLMN or when it performs inter-system change back from S1 mode to N1 mode. Thus the UE can still be prevented from sending another PDU SESSION ESTABLISHMENT REQUEST or PDU SESSION MODIFICATION REQUEST message in the PLMN for the same S-NSSAI and optionally the same DNN.

Upon PLMN change, if T3584 is running or is deactivated for an S-NSSAI, a DNN, and old PLMN, but T3584 is not running and is not deactivated for the S-NSSAI, the DNN, and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI and the same DNN in the new PLMN.

Upon PLMN change, if T3585 is running or is deactivated for an S-NSSAI and old PLMN, but T3585 is not running and is not deactivated for the S-NSSAI and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI in the new PLMN.

6.4.1.4.3 Handling of network rejection not due to congestion control

If the 5GSM cause value is different from #26 "insufficient resources", #28 "unknown PDU session type", #39 "reactivation requested", #46 "out of LADN service area", #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #54 "PDU session does not exist", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", #61 "PDU session type Ethernet only allowed", #67 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", and #69 "insufficient resources for specific slice and DNN", #68 "not supported service option not subscribed" upon sending PDU SESSION ESTABLISHMENT REQUEST to establish an MA PDU session, and the Back-off timer value IE is included, the UE shall behave as follows: (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.12):

- a) if the timer value indicates neither zero nor deactivated and:
 - 1) if the UE provided a DNN and S-NSSAI to the network during the PDU session establishment and the 5GSM cause value is different from #27 "missing or unknown DNN", the UE shall start the back-off timer with the value provided in the Back-off timer value IE for the PDU session establishment procedure and [PLMN, DNN, (mapped) HPLMN S-NSSAI] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN and (mapped) HPLMN S-NSSAI in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
 - 2) if the UE provided a DNN to the network during the PDU session establishment and the 5GSM cause value is #27 "missing or unknown DNN", the UE shall start the back-off timer with the value provided in the Back-off timer value IE for the PDU session establishment procedure and [PLMN, DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
 - 3) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment and the 5GSM cause value is different from #27 "missing or unknown DNN", it shall start the back-off timer accordingly for the PDU session establishment procedure and the [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination. Dependent on the combination, the UE shall not send another PDU SESSION ESTABLISHMENT

REQUEST message for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder; or

- 4) if the UE did not provide a DNN to the network during the PDU session establishment and the 5GSM cause value is #27 "missing or unknown DNN", it shall start the back-off timer accordingly for the PDU session establishment procedure and the [PLMN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, no DNN] in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
- b) if the timer value indicates that this timer is deactivated and:
 - if the UE provided a DNN and S-NSSAI to the network during the PDU session establishment and the 5GSM cause value is different from #27 "missing or unknown DNN", the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN and (mapped) HPLMN S-NSSAI in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder;
 - 2) if the UE provided a DNN to the network during the PDU session establishment and the 5GSM cause value is #27 "missing or unknown DNN", the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder;
 - 3) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment and the 5GSM cause value is different from #27 "missing or unknown DNN", the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder; or
 - 4) if the UE did not provide a DNN to the network during the PDU session establishment and the 5GSM cause value is #27 "missing or unknown DNN", the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same [PLMN, no DNN] in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials from a credentials holder; and
- c) if the timer value indicates zero and the 5GSM cause value is different from #27 "missing or unknown DNN", the UE may send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI], or [PLMN, no DNN, no S-NSSAI] in the current PLMN. If the timer value indicates zero and the 5GSM cause value is #27 "missing or unknown DNN", the UE may send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN], or [PLMN, no DNN] in the current PLMN.

If the Back-off timer value IE is not included, then the UE shall ignore the Re-attempt indicator IE provided by the network in the PDU SESSION ESTABLISHMENT REJECT message, if any.

 a) Additionally, if the 5GSM cause value is #8 "operator determined barring", #32 "service option not supported", #33 "requested service option not subscribed" upon sending PDU SESSION ESTABLISHMENT REQUEST not to establish an MA PDU session, or #70 "missing or unknown DNN in a slice", then:

- 1) the UE not operating in SNPN access operation mode shall proceed as follows:
 - i) if the UE is registered in the HPLMN or in a PLMN that is within the EHPLMN list, the UE shall behave as described above in the present subclause using the configured SM Retry Timer value as specified in 3GPP TS 24.368 [17] or in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22], if available, as back-off timer value; and
- NOTE 1: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if the UE is configured with:
 - an SM Retry Timer value in ME as specified in 3GPP TS 24.368 [17]; and
 - an SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22].
 - ii) otherwise, if the UE is not registered in its HPLMN or in a PLMN that is within the EHPLMN list, or if the SM Retry Timer value is not configured, the UE shall behave as described above in the present subclause, using the default value of 12 minutes for the back-off timer; or
 - 2) the UE operating in SNPN access operation mode shall proceed as follows:
 - i) if:
 - A) the SM Retry Timer value for the current SNPN as specified in 3GPP TS 24.368 [17] is available; or
 - B) the UE used the USIM for registration to the current SNPN and the SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22] is available;

then the UE shall behave as described above in the present subclause using the configured SM Retry Timer value as back-off timer value; or

- NOTE 2: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if both conditions in bullets A) and B) above are satisfied.
 - ii) otherwise, the UE shall behave as described above in the present subclause, using the default value of 12 minutes for the back-off timer.
- b) For 5GSM cause value #27 "missing or unknown DNN", then:
 - 1) the UE not operating in SNPN access operation mode shall proceed as follows:
 - i) if the UE is registered in the HPLMN or in a PLMN that is within the EHPLMN list, the UE shall start the back-off timer with the configured SM Retry Timer value as specified in 3GPP TS 24.368 [17] or in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22], if available, as back-off timer value for the PDU session establishment procedure and the [PLMN, DNN] or [PLMN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the back-off timer expires, the UE is switched off or the USIM is removed; and
- NOTE 3: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if the UE is configured with:
 - an SM Retry Timer value in ME as specified in 3GPP TS 24.368 [17]; and
 - an SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22].
 - ii) otherwise, if the UE is not registered in its HPLMN or in a PLMN that is within the EHPLMN list, or if the SM Retry Timer value is not configured, the UE shall start the back-off timer with the default value of 12 minutes as back-off timer value for the PDU session establishment procedure and the [PLMN, DNN] or [PLMN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current PLMN, until the back-off timer expires, the UE is switched off or the USIM is removed; or
 - 2) the UE operating in SNPN access operation mode shall proceed as follows:
 - i) if:
 - A) the SM Retry Timer value for the current SNPN as specified in 3GPP TS 24.368 [17] is available; or
 - B) the UE used the USIM for registration to the current SNPN and the SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22] is available;

then:

- if the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall start the back-off timer with the configured SM Retry Timer value as back-off timer value for the PDU session establishment procedure and the [SNPN, DNN] or [SNPN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current SNPN, until the back-off timer expires, the UE is switched off, or the entry in the "list of subscriber data" for the current SNPN is updated; and
- if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall start
 the back-off timer with the configured SM Retry Timer value as back-off timer value for the PDU
 session establishment procedure and the [SNPN, selected entry of the "list of subscriber data" or
 selected PLMN subscription, DNN] or [SNPN, selected entry of the "list of subscriber data" or
 selected PLMN subscription, no DNN] combination. The UE shall not send another PDU SESSION
 ESTABLISHMENT REQUEST message for the same DNN in the current SNPN using the selected
 entry in the "list of subscriber data" or selected PLMN subscription, until the back-off timer expires,
 the UE is switched off, the UICC containing the USIM is removed or the selected entry of the "list of
 subscriber data" is updated; or
- NOTE 4: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if both conditions in bullets A) and B) above are satisfied.
 - ii) otherwise:
 - if the UE does not support access to an SNPN using credentials from a credentials holder, the UE shall start the back-off timer with the default value of 12 minutes as back-off timer value for the PDU session establishment procedure and the [SNPN, DNN] or [SNPN, no DNN] combination. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN in the current SNPN, until the back-off timer expires, the UE is switched off, or the entry in the "list of subscriber data" for the current SNPN is updated; and
 - if the UE supports access to an SNPN using credentials from a credentials holder, the UE shall start
 the back-off timer with the default value of 12 min as back-off timer value for the PDU session
 establishment procedure and the [SNPN, selected entry of the "list of subscriber data" or selected
 PLMN subscription, DNN] or [SNPN, selected entry in the "list of subscriber data" or selected PLMN
 subscription, no DNN] combination. The UE shall not send another PDU SESSION
 ESTABLISHMENT REQUEST message for the same DNN in the current SNPN using the selected
 entry of the "list of subscriber data", until the back-off timer expires, the UE is switched off, the UICC
 containing the USIM is removed or the selected entry of the "list of subscriber data" is updated; and
- c) For 5GSM cause values different from #8 "operator determined barring", #27 "missing or unknown DNN", #32 "service option not supported", #33 "requested service option not subscribed" and #70 "missing or unknown DNN in a slice", the UE behaviour regarding the start of a back-off timer is specified in clause 6.2.12.

The UE shall not stop any back-off timer:

- a) upon a PLMN change;
- b) upon an inter-system change; or
- c) upon registration over another access type.

If the network indicates that a back-off timer for the PDU session establishment procedure is deactivated, then it remains deactivated;

- a) upon a PLMN change;
- b) upon an inter-system change; or
- c) upon registration over another access type.

NOTE 5: This means the back-off timer can still be running or be deactivated for the given 5GSM procedure when the UE returns to the PLMN or when it performs inter-system change back from S1 mode to N1 mode. Thus, the UE can still be prevented from sending another PDU SESSION ESTABLISHMENT REQUEST message for the combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI], [PLMN, no S-NSSAI], [PLMN, no DNN] in the PLMN.

If the back-off timer is started upon receipt of a PDU SESSION ESTABLISHMENT REJECT (i.e. the timer value was provided by the network, a configured value is available or the default value is used as explained above) or the back-off timer is deactivated, the UE behaves as follows:

- a) after a PLMN change:
 - the UE may send a PDU SESSION ESTABLISHMENT REQUEST message for the combination of [new PLMN, DNN, (mapped) HPLMN S-NSSAI], [new PLMN, DNN, no S-NSSAI], [new PLMN, no DNN, (mapped) HPLMN S-NSSAI], or [new PLMN, no DNN, no S-NSSAI] in the new PLMN, if the back-off timer is not running and is not deactivated for the PDU session establishment procedure and the combination of [new PLMN, DNN, (mapped) HPLMN S-NSSAI], [new PLMN, DNN, no S-NSSAI], [new PLMN, no DNN, (mapped) HPLMN S-NSSAI], or [new PLMN, no DNN, no S-NSSAI];
 - 2) as an implementation option, for the 5GSM cause value #8 "operator determined barring", #32 "service option not supported", #33 "requested service option not subscribed" and #70 "missing or unknown DNN in a slice", if the network does not include a Re-attempt indicator IE, the UE may decide not to automatically send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI], or [PLMN, no DNN, no S-NSSAI] using the same PDU session type if the UE is registered to a new PLMN which is in the list of equivalent PLMNs; and
 - 3) as an implementation option, for the 5GSM cause value #27 "missing or unknown DNN", if the network does not include a Re-attempt indicator IE, the UE may decide not to automatically send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN] or [PLMN, no DNN] using the same PDU session type if the UE is registered to a new PLMN which is in the list of equivalent PLMNs;
- b) if the network does not include the Re-attempt indicator IE to indicate whether re-attempt in S1 mode is allowed, or the UE ignores the Re-attempt indicator IE, e.g. because the Back-off timer value IE is not included, then:
 - if the UE is registered in its HPLMN or in a PLMN that is within the EHPLMN list and the back-off timer is running for the combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI] or [PLMN, DNN, no S-NSSAI], the UE shall apply the configured SM_RetryAtRATChange value as specified in 3GPP TS 24.368 [17] or in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22], if available, to determine whether the UE may attempt a PDN connectivity procedure for the same [PLMN, DNN] combination in S1 mode. If the back-off timer is running for the combination of [PLMN, no DNN, (mapped) HPLMN S-NSSAI] or [PLMN, no DNN, no S-NSSAI], the same applies for the PDN connectivity procedure for the [PLMN, no DNN] combination in S1 mode accordingly; and
- NOTE 6: The way to choose one of the configured SM_RetryAtRATChange values for back-off timer value is up to UE implementation if the UE is configured with:
 - an SM_RetryAtRATChange value in ME as specified in 3GPP TS 24.368 [17]; and
 - an SM_RetryAtRATChange value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22].
 - 2) if the UE is not registered in its HPLMN or in a PLMN that is within the EHPLMN list, or if the NAS configuration MO as specified in 3GPP TS 24.368 [17] is not available and the value for inter-system change is not configured in the USIM file NAS_{CONFIG}, then the UE behaviour regarding a PDN connectivity procedure for the same [PLMN, DNN] or [PLMN, no DNN] combination in S1 mode is unspecified; and
- c) if the network includes the Re-attempt indicator IE indicating that re-attempt in an equivalent PLMN is not allowed, then depending on the timer value received in the Back-off timer value IE, for each combination of a PLMN from the equivalent PLMN list and the respective [DNN, (mapped) HPLMN S-NSSAI], [DNN, no S-NSSAI], [no DNN, (mapped) HPLMN S-NSSAI], or [no DNN, no S-NSSAI] combination, the UE shall start a back-off timer for the PDU session establishment procedure with the value provided by the network, or deactivate the respective back-off timer as follows:

- 1) if the Re-attempt indicator IE additionally indicates that re-attempt in S1 mode is allowed, the UE shall start or deactivate the back-off timer for N1 mode only; and
- 2) otherwise, the UE shall start or deactivate the back-off timer for S1 and N1 mode.

If the back-off timer for a [PLMN, DNN] or [PLMN, no DNN] combination, was started or deactivated in S1 mode upon receipt of PDN CONNECTIVITY REJECT message (see 3GPP TS 24.301 [15]) and the network indicated that re-attempt in N1 mode is allowed, then this back-off timer does not prevent the UE from sending a PDU SESSION ESTABLISHMENT REQUEST message in this PLMN for the same DNN, or without DNN, after inter-system change to N1 mode. If the network indicated that re-attempt in N1 mode is not allowed, the UE shall not send any PDU SESSION ESTABLISHMENT REQUEST message in this PLMN for the same DNN in combination with any S-NSSAI or without S-NSSAI, or in this PLMN without DNN in combination with any S-NSSAI or without S-NSSAI, after inter-system change to N1 mode until the timer expires, the UE is switched off or the USIM is removed.

- NOTE 7: The back-off timer is used to describe a logical model of the required UE behaviour. This model does not imply any specific implementation, e.g. as a timer or timestamp.
- NOTE 8: Reference to back-off timer in this section can either refer to use of timer T3396 or to use of a different packet system specific timer within the UE. Whether the UE uses T3396 as a back-off timer or it uses different packet system specific timers as back-off timers is left up to UE implementation.

When the back-off timer is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure if the procedure is for emergency services.

If the 5GSM cause value is #28 "unknown PDU session type" and the PDU SESSION ESTABLISHMENT REQUEST message contained a PDU session type IE indicating a PDU session type, the UE shall ignore the Back-off timer value IE and Re-attempt indicator IE provided by the network, if any. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message to establish a new PDU session with the PDU session type IE indicating another PDU session type, e.g. using another value which can be used for the rejected component in the same route selection descriptor as specified in 3GPP TS 24.526 [19]. The behaviour of the UE for 5GSM cause value #28 also applies if the PDU session is a MA PDU Session.

If the 5GSM cause value is #39 "reactivation requested", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any.

NOTE 9: Further UE behaviour upon receipt of 5GSM cause value #39 is up to the UE implementation.

If the 5GSM cause value is #46 "out of LADN service area", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any. If the UE is not located inside the LADN service area, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message or another PDU SESSION MODIFICATION REQUEST message for the LADN DNN provided by the UE during the PDU session establishment procedure until the LADN information for the specific LADN DNN is updated as described in subclause 5.4.4 and subclause 5.5.1. If the UE is not located inside the LADN service area, the UE shall not indicate the PDU session(s) for the LADN DNN provided by the UE during the PDU session establishment procedure in the Uplink data status IE included in the SERVICE REQUEST message until the LADN information for the specific LADN DNN is updated as described in subclause 5.4.4 and subclause 5.5.1.

NOTE 10: Based on UE implementation, the UE locating inside the LADN service area can send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the LADN DNN which was rejected with the 5GSM cause value #46 "out of LADN service area".

If the 5GSM cause value is #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", or #61 "PDU session type Ethernet only allowed", the UE shall ignore the Back-off timer value IE provided by the network, if any. The UE shall evaluate the URSP rules if available as specified in 3GPP TS 24.526 [19]. The UE shall not subsequently send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN (or no DNN, if no DNN was indicated by the UE) and the same (mapped) HPLMN S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) to obtain a PDU session type different from the one allowed by the network until any of the following conditions is fulfilled:

a) the UE is registered to a new PLMN which was not in the list of equivalent PLMNs at the time when the PDU SESSION ESTABLISHMENT REJECT message was received;

- b) the UE is registered to a new PLMN which was in the list of equivalent PLMNs at the time when the PDU SESSION ESTABLISHMENT REJECT message was received, and either the network did not include a Reattempt indicator IE in the PDU SESSION ESTABLISHMENT REJECT message or the Re-attempt indicator IE included in the message indicated that re-attempt in an equivalent PLMN is allowed;
- c) void;
- d) the UE is switched off; or
- e) the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials holder.

For the 5GSM cause values #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", and #61 "PDU session type Ethernet only allowed", the UE shall ignore the value of the RATC bit in the Re-attempt indicator IE provided by the network, if any.

NOTE 11:For the 5GSM cause values #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", and #61 "PDU session type Ethernet only allowed", re-attempt in S1 mode for the same DNN (or no DNN, if no DNN was indicated by the UE) is only allowed using the PDU session type(s) indicated by the network.

If the 5GSM cause value is #54 "PDU session does not exist", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any. If the PDU session establishment procedure is to perform handover of an existing PDU session between 3GPP access and non-3GPP access, the UE shall release locally the existing PDU session with the PDU session ID included in the PDU SESSION ESTABLISHMENT REJECT message. The UE may initiate another UE-requested PDU session establishment procedure with the request type set to "initial request" in the subsequent PDU SESSION ESTABLISHMENT REQUEST message to establish a PDU session with the same DNN (or no DNN, if no DNN was indicated by the UE) and the same (mapped) HPLMN S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE).

NOTE 12:User interaction is necessary in some cases when the UE cannot re-establish the PDU session(s) automatically.

If the 5GSM cause value is #68 "not supported SSC mode", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any. The UE shall evaluate the URSP rules if available as specified in 3GPP TS 24.526 [19]. The UE shall not subsequently send another PDU SESSION ESTABLISHMENT REQUEST message for the same DNN (or no DNN, if no DNN was indicated by the UE) and the same (mapped) HPLMN S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) using the same SSC mode or an SSC mode which was not included in the Allowed SSC mode IE until any of the following conditions is fulfilled:

- a) the UE is registered to a new PLMN which was not in the list of equivalent PLMNs at the time when the PDU SESSION ESTABLISHMENT REJECT message was received;
- b) the SSC mode which is used to access to the DNN (or no DNN, if no DNN was indicated by the UE) and the (mapped) HPLMN S-NSSAI (or no S-NSSAI, if no S-NSSAI was indicated by the UE) is changed by the UE which subsequently requests a new SSC mode in the Allowed SSC mode IE or no SSC mode;
- c) the UE is switched off; or
- d) the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder.

If the UE receives the 5GSM cause value is #33 "requested service option not subscribed" upon sending PDU SESSION ESTABLISHMENT REQUEST to establish an MA PDU session, the UE shall ignore the Back-off timer value IE and Re-attempt indicator IE provided by the network, if any. The UE shall evaluate URSP rules, if available, as specified in 3GPP TS 24.526 [19] and the UE may send PDU SESSION ESTABLISHMENT REQUEST after evaluating those URSP rules.

If the 5GSM cause value is #86 "UAS services not allowed", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any, and shall behave as specified in subcluase 6.4.1.4.1.

Upon receipt of an indication from 5GMM sublayer that the 5GSM message was not forwarded because the DNN is not supported or not subscribed in a slice along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session, the UE shall stop timer T3580, shall abort the procedure and shall behave as follows:

- a) if the timer value indicates neither zero nor deactivated, the UE shall start the back-off timer with the value received from the 5GMM sublayer for the PDU session establishment procedure and the [PLMN, DNN, S-NSSAI] combination or the [PLMN, DNN, no S-NSSAI] combination, if no S-NSSAI was provided during the PDU session establishment. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message in the PLMN for the same DNN and the same S-NSSAI that were sent by the UE, or for the same DNN and no S-NSSAI if S-NSSAI that was not sent by the UE, until:
 - 1) the back-off timer expires;
 - 2) the UE is switched off;
 - 3) the USIM is removed the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder; or
 - 4) the DNN is included in the LADN information and the network provides the LADN information during the registration procedure or the generic UE configuration update procedure;
- b) if the timer value is not received from the 5GMM sublayer or the timer value indicates that this timer is deactivated, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message in the PLMN for the same DNN and the same S-NSSAI that were sent by the UE, or for the same DNN and no S-NSSAI if S-NSSAI that was not sent by the UE, until:
 - 1) the UE is switched off;
 - 2) the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated if the UE does not support access to an SNPN using credentials from a credentials holder, or the selected entry of the "list of subscriber data" is updated if the UE supports access to an SNPN using credentials from a credentials holder; or
 - 3) the DNN is included in the LADN information and the network provides the LADN information during the registration procedure or the generic UE configuration update procedure; and
- c) if the timer value indicates zero, the UE may send another PDU SESSION ESTABLISHMENT REQUEST message for the same combination of [PLMN, DNN, S-NSSAI], [PLMN, DNN, no S-NSSAI] in the current PLMN.

6.4.1.5 Handling the maximum number of established PDU sessions

The maximum number of PDU sessions which a UE can establish in a PLMN or SNPN is limited by whichever is the lowest of: the maximum number of PDU session IDs allowed by the protocol (as specified in 3GPP TS 24.007 [11] subclause 11.2.3.1b), the PLMN's or SNPN's maximum number of PDU sessions and the UE's implementation-specific maximum number of PDU sessions.

If during a UE-requested PDU session establishment procedure the 5GSM sublayer in the UE receives an indication that the 5GSM message was not forwarded because:

- a) the PLMN's maximum number of PDU sessions has been reached, then the UE determines the PLMN's maximum number of PDU sessions as the number of active PDU sessions it has; or
- b) the SNPN's maximum number of PDU sessions has been reached, then the UE determines the SNPN's maximum number of PDU sessions as the number of active PDU sessions it has and associates the determined maximum number of PDU sessions with:
 - 1) the entry in the "list of subscriber data" for the current SNPN if the UE does not support access to an SNPN using credentials from a credentials holder; or
 - 2) the selected entry of the "list of subscriber data" or the selected PLMN subscription if the UE supports access to an SNPN using credentials from a credentials holder.

- NOTE 1: In some situations, when attempting to establish multiple PDU sessions, the number of active PDU sessions that the UE has when 5GMM cause #65 is received is not equal to the maximum number of PDU sessions reached in the network.
- NOTE 2: When the network supports emergency services, it is not expected that 5GMM cause #65 is returned by the network when the UE requests an emergency PDU session.
- NOTE 3: There is only one maximum number of PDU sessions for a PLMN regardless of which access the PDU session exists in.
- NOTE 4: An MA PDU session which (only) has a PDN connection established as a user-plane resource is counted as an active PDU session when determining the PLMN's maximum number of PDU sessions.

The PLMN's maximum number of PDU sessions applies to the PLMN in which the 5GMM cause #65 "maximum number of PDU sessions reached" is received. When the UE is switched off or when the USIM is removed, the UE shall clear all previous determinations representing PLMN's maximum number of PDU sessions.

The SNPN's maximum number of PDU sessions applies to the SNPN in which the 5GMM cause #65 "maximum number of PDU sessions reached" is received. When the UE is switched off, the UE shall clear all previous determinations representing SNPN's maximum number of PDU sessions. In addition:

- a) if the UE does not support access to an SNPN using credentials from a credentials holder and the entry in the "list of subscriber data" for the current SNPN is updated, then the UE shall clear all previous determinations representing SNPN's maximum number of PDU sessions associated with the entry in the "list of subscriber data" for the current SNPN; and
- b) if the UE supports access to an SNPN using credentials from a credentials holder and:
 - the selected entry of the "list of subscriber data" is updated, then UE shall clear all previous determinations representing SNPN's maximum number of PDU sessions associated with the selected entry in the "list of subscriber data"; or
 - the USIM associated with the selected PLMN subscription is removed, then UE shall clear all previous determinations representing SNPN's maximum number of PDU sessions associated with the selected PLMN subscription.

Upon successful registration with a new PLMN or SNPN, the UE may clear previous determinations representing any PLMN's or SNPN's maximum number(s) of PDU sessions.

If the maximum number of established PDU sessions is reached at the UE and the upper layers of the UE request connectivity to a DNN the UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message unless an established PDU session is released. If the UE needs to release an established PDU session, choosing which PDU session to release is implementation specific, however the UE shall not release the emergency PDU session.

If the UE needs to release a PDU session in order to request an emergency PDU session, it shall either perform a local release of a PDU session or release a PDU session via explicit signalling. If the UE performs a local release, the UE shall:

- a) if the PDU session is an MA PDU session:
 - 1) perform a registration procedure for mobility and periodic registration update to indicate PDU session status to the network over each access that user plane resources have been established; and
 - perform a normal and periodic tracking area updating to indicate EPS bearer context status to the network as specified in clause 5.5.3.2.2 of 3GPP TS 24.301 [15] when a PDN connection has been established as a user plance resource; or
- b) if the PDU session is a single access PDU session, perform a registration procedure for mobility and periodic registration update to indicate PDU session status to the network over the access the PDU session is associated with.

6.4.1.5A Handling the maximum number of allowed active user-plane resources for PDU sessions of UEs in NB-N1 mode

For a UE in NB-N1 mode, the UE's maximum number of supported user-plane resources is two (as defined in 3GPP TS 36.300 [25B]) when the UE sets the Multiple user-plane resources support bit to "Multiple user-plane resources supported" during the registration procedure for initial registration or for mobility and periodic registration update, and one otherwise.

For a UE operating in NB-N1 mode, if:

- a) the UE's maximum number of supported user-plane resources is one, then only one PDU session can have active user-plane resources even though that UE might have established more than one PDU session; or
- b) the UE's maximum number of supported user-plane resources is two, then only two PDU sessions can have active user-plane resources even though that UE might have established more than two PDU sessions.

When the maximum number of active user-plane resources is reached and upper layers request for more user-plane resources for PDU sessions other than the PDU sessions with those active user-plane resources, the UE can choose to release one or more of the PDU sessions with active user-plane resources to cater for the upper layer request. The choice of which PDU sessions to be released is implementation specific. However if there is a PDU session with an active user-plane that is used for exception data reporting (see subclause 6.2.13), that PDU session shall not be released.

If the maximum number of active user-plane resources is reached and the upper layers of the UE request user-plane resources for exception data reporting (see subclause 6.2.13), the UE shall release a PDU session that has user-plane resources to cater for the request for exception data reporting. The choice of which PDU session to be released is implementation specific.

If the UE decides to release one or more active user-plane resources to cater for upper layer request, the UE shall release the PDU session via explicit 5GSM signalling.

6.4.1.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Expiry of timer T3580

The UE shall, on the first expiry of the timer T3580:

- if the PDU SESSION ESTABLISHMENT REQUEST message was sent with request type set to "initial emergency request" or "existing emergency PDU session", then the UE may:
 - a) inform the upper layers of the failure of the procedure; or
- NOTE 1: This can result in the upper layers requesting another emergency call attempt using domain selection as specified in 3GPP TS 23.167 [6].
 - b) de-register locally, if not de-registered already, attempt initial registration for emergency services.

If the UE sent the PDU SESSION ESTABLISHMENT REQUEST message in order to perform a handover of an existing emergency PDU session between 3GPP access and non-3GPP access, the UE shall consider that the emergency PDU session is associated with the source access type.

- otherwise, retransmit the PDU SESSION ESTABLISHMENT REQUEST message and the PDU session information which was transported together with the initial transmission of the PDU SESSION ESTABLISHMENT REQUEST message and shall reset and start timer T3580, if still needed. This retransmission can be repeated up to four times, i.e. on the fifth expiry of timer T3580, the UE shall abort the procedure, release the allocated PTI and enter the state PROCEDURE TRANSACTION INACTIVE. If the UE sent the PDU SESSION ESTABLISHMENT REQUEST message in order to perform a handover of an existing non-emergency PDU session between 3GPP access and non-3GPP access, the UE shall consider that the PDU session is associated with the source access type.
- b) Upon receiving an indication that the 5GSM message was not forwarded due to routing failure along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3580 and shall abort the procedure. If the UE

sent the PDU SESSION ESTABLISHMENT REQUEST message in order to perform a handover of an existing PDU session between 3GPP access and non-3GPP access, the UE shall consider that the PDU session is associated with the source access type.

- b1)Upon receiving an indication that the 5GSM message was not forwarded due to service area restrictions along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3580 and shall abort the procedure. If the UE sent the PDU SESSION ESTABLISHMENT REQUEST message in order to perform a handover of an existing PDU session between 3GPP access and non-3GPP access, the UE shall consider that the PDU session is associated with the source access type.
- b2)Upon receiving an indication that the 5GSM message was not forwarded because the UE is registered to a PLMN via a satellite NG-RAN cell that is not allowed to operate at the present UE location along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3580 and shall abort the procedure. The UE shall not trigger the PDU session establishment procedure until the UE is deregistered from the PLMN.
- b3)Upon receiving an indication that the 5GSM message was not forwarded because the UE is marked in the UE's 5GMM context that it is not allowed to request UAS services along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3580 and shall abort the procedure. The UE shall not trigger the PDU session establishment procedure until the UE is deregistered from the PLMN.
- b4) Upon receiving an indication that the 5GSM message was not forwarded because the PLMN's maximum number of PDU sessions has been reached along with a PDU SESSION ESTABLISHMENT REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3580 and shall abort the procedure.
- c) Collision of UE-requested PDU session establishment procedure and network-requested PDU session release procedure.

If the UE receives a PDU SESSION RELEASE COMMAND message after sending a PDU SESSION ESTABLISHMENT REQUEST message to the network, and the PDU session ID in the PDU SESSION RELEASE COMMAND message is the same as the PDU session ID in the PDU SESSION ESTABLISHMENT REQUEST message:

- i) if the UE-requested PDU session establishment procedure was to request the establishment of user plane resources on the second access for an MA PDU session established on a first access and the Access type IE is not included in PDU SESSION RELEASE COMMAND or the Access type IE included in PDU SESSION RELEASE COMMAND indicates the first access, the UE shall proceed with the network-requested PDU session release procedure, abort the UE-requested PDU session establishment procedure, stop timer T3580, release the allocated PTI and enter the state PROCEDURE TRANSACTION INACTIVE;
- ii) if the PDU SESSION ESTABLISHMENT REQUEST message was sent with request type set to "existing PDU session" or "existing emergency PDU session" in order to perform a handover of an existing PDU session between 3GPP access and non-3GPP access, the UE shall abort the PDU session establishment procedure and proceed with the network-requested PDU session release procedure; or
- iii) otherwise, the UE shall ignore the PDU SESSION RELEASE COMMAND message and proceed with the UE-requested PDU session establishment procedure.
- d) Inter-system change from N1 mode to S1 mode triggered during UE-requested PDU session establishment procedure.

If the UE-requested PDU session establishment procedure is triggered for handover of an existing PDU session from non-3GPP access to 3GPP access, and the inter-system change from N1 mode to S1 mode is triggered by the NG-RAN and the UE did not receive response to PDU session establishment request, then the UE shall abort the procedure, stop timer T3580, and notify the upper layer of the handover failure.

- NOTE 2: This can result in the upper layer requesting re-initiation of handover from non-3GPP access to 3GPP access after the inter-system change is completed, if still required.
- e) For an MA PDU session established on a single access, upon receipt of a PDU SESSION ESTABLISHMENT ACCEPT message over the other access, if any value of the selected PDU session type, selected SSC mode,

5GSM cause, PDU address, S-NSSAI, DNN IEs in the PDU SESSION ESTABLISHMENT ACCEPT message is different from the corresponding stored value, the UE shall perform a local release of the MA PDU session, and perform the registration procedure for mobility and periodic registration update with a REGISTRATION REQUEST message including the PDU session status IE over both accesses.

- f) For an MA PDU session has a PDN connection as a user-plane resource, upon receipt of a PDU SESSION ESTABLISHMENT ACCEPT message over non-3GPP access, if any value of the selected PDU session type, selected SSC mode, 5GSM cause, PDU address, S-NSSAI, DNN IEs in the PDU SESSION ESTABLISHMENT ACCEPT message is different from the corresponding stored mapped value, the UE shall perform a local release of the MA PDU session, perform the registration procedure for mobility and periodic registration update with a REGISTRATION REQUEST message including the PDU session status IE over non-3GPP access, and perform the tracking area updating procedure as specified in clause 5.5.3.2.2 of 3GPP TS 24.301 [15] with a TRACKING AREA UPDATE REQUEST message including EPS bearer context status IE.
- g) Collision of UE-requested PDU session establishment procedure initiated to perform handover of an existing PDU session from non-3GPP access to 3GPP access and a notification from the network with access type indicating non-3GPP access.

If the UE receives a notification from the network with access type indicating non-3GPP access after sending a PDU SESSION ESTABLISHMENT REQUEST message to perform handover of an existing PDU session from non-3GPP access to 3GPP access, the UE shall abort the PDU session establishment procedure, stop timer T3580, proceed with the service request procedure to perform handover of existing PDU session(s) from non-3GPP access to 3GPP access.

h) Collision of UE-requested PDU session establishment procedure and N1 NAS signalling connection release

The UE may immediately retransmit the PDU SESSION ESTABLISHMENT REQUEST message and stop, reset and restart timer T3580, if the following conditions apply:

- 1) The original UE-requested PDU session establishment procedure was initiated over an existing N1 NAS signalling connection;
- 2) the previous transmission of the PDU SESSION ESTABLISHMENT REQUEST message was not initiated due to timer T3580 expiry; and
- 3) no 5GSM message related to the PDU session (e.g. PDU SESSION AUTHENTICATION COMMAND message) was received after the PDU SESSION ESTABLISHMENT REQUEST message was transmitted.
- i) Collision of UE-requested PDU session establishment procedure and network-requested PDU session modification procedure

If the UE receives a PDU SESSION MODIFICATION COMMAND message after sending a PDU SESSION ESTABLISHMENT REQUEST message to the network, and the PDU session ID in the PDU SESSION MODIFICATION COMMAND message is the same as the PDU session ID in the PDU SESSION ESTABLISHMENT REQUEST message:

- i) if the UE-requested PDU session establishment procedure was to request the establishment of user plane resources on the second access for an MA PDU session established on a first access, the UE shall proceed with both the UE-requested PDU session establishment procedure and the network-requested PDU session modification procedure; or
- ii) if the PDU SESSION ESTABLISHMENT REQUEST message was sent with request type set to "existing PDU session" or "existing emergency PDU session" in order to perform a handover of an existing PDU session between 3GPP access and non-3GPP access, the UE shall proceed with the UE-requested PDU session establishment procedure and abort the network-requested PDU session modification procedure.

6.4.1.7 Abnormal cases on the network side

The following abnormal cases can be identified:

a) If the received request type is "initial emergency request" and there is an existing emergency PDU session for the UE, regardless whether the PDU session ID in the PDU SESSION ESTABLISHMENT REQUEST message is identical to the PDU session ID of the existing PDU session, the SMF shall locally release the existing emergency PDU session and proceed the new PDU SESSION ESTABLISHMENT REQUEST message b) The information for the PDU session authentication and authorization by the external DN in PDU DN request container is not compliant with local policy and user's subscription data

If the PDU session being established is a non-emergency PDU session, the request type is not set to "existing PDU session", the PDU session authentication and authorization by the external DN is required due to local policy and user's subscription data and the information for the PDU session authentication and authorization by the external DN in the SM PDU DN request container IE is not compliant with the local policy and user's subscription data, the SMF shall reject the PDU session establishment request including the 5GSM cause #29 "user authentication or authorization failed", in the PDU SESSION ESTABLISHMENT REJECT message.

c) UE-requested PDU session establishment with request type set to "initial request" for an existing PDU session:

If the SMF receives a PDU SESSION ESTABLISHMENT REQUEST message with a PDU session ID identical to the PDU session ID of an existing PDU session and with request type set to "initial request", the SMF shall locally release the existing PDU session and proceed with the PDU session establishment procedure.

d) UE-requested PDU session establishment with request type "existing PDU session" or "existing emergency PDU session" for a PDU session that does not exist:

If the SMF receives a PDU SESSION ESTABLISHMENT REQUEST message with request type set to "existing PDU session" or "existing emergency PDU session", and the SMF does not have any information about that PDU session, then the SMF shall reject the PDU session establishment procedure with the 5GSM cause set to #54 "PDU session does not exist" in the PDU SESSION ESTABLISHMENT REJECT message.

e) 5G access network cannot forward the message:

If the SMF determines based on content of the n2SmInfo attribute specified in 3GPP TS 29.502 [20A] that the DL NAS TRANSPORT message carrying the PDU SESSION ESTABLISHMENT ACCEPT was not forwarded to the UE by the 5G access network, then the SMF shall reject the PDU session establishment procedure with the 5GSM cause set to #26 "insufficient resources" in the PDU SESSION ESTABLISHMENT REJECT message.

6.4.2 UE-requested PDU session modification procedure

6.4.2.1 General

The purpose of the UE-requested PDU session modification procedure is:

- a) to enable the UE to request modification of a PDU session;
- b) to indicate a change of 3GPP PS data off UE status for a PDU session;
- c) to revoke the previously indicated support for reflective QoS;
- d) to request specific QoS handling and segregation of service data flows;
- e) to indicate to the network the relevant 5GSM parameters and capabilities (e.g. the UE's 5GSM capabilities, whether the UE supports more than 16 packet filters, the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink, the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink and whether the UE requests the PDU session to be an always-on PDU session in the 5GS) for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface and the UE has not previously successfully performed the UE-requested PDU session modification to indicate to the network the relevant 5GSM parameters and capabilities;
- f) to delete one or more mapped EPS bearer contexts;
- g) to convey a port management information container;
- h) to re-negotiate header compression configuration associated to a PDU session using control plane CIoT 5GS optimization; or
- i) to enable the UE to request to join or leave one or more multicast MBS sessions associated with a PDU session.

NOTE 1: The case c), d), e), f) and g) do not apply to PDU sessions associated with the control plane only indication.

NOTE 2: For case e), the procedure is attempted after the first inter-system change from S1 mode to N1 mode.

When the UE-requested PDU session modification procedure is used to indicate a change of 3GPP PS data off UE status for a PDU session (see subclause 6.2.10), the UE shall initiate the UE-requested PDU session modification procedure even if the UE is outside the LADN service area or the timer T3396, T3584, T3585 or the back-off timer is running or is deactivated.

If the UE needs to revoke the previously indicated support for reflective QoS for a PDU session and timer T3396, T3584, T3585 or the back-off timer is running or is deactivated, the UE shall not initiate the UE-requested PDU session modification procedure and shall instead initiate the UE-requested PDU session release procedure.

If the UE needs to initiate the UE-requested PDU session modification procedure to indicate to the network the relevant 5GSM parameters and capabilities (e.g. the UE's 5GSM capabilities, whether the UE supports more than 16 packet filters, the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink, the maximum data rate per UE for user-plane integrity protection supported by the UE for uplink, the maximum data rate per UE for user-plane integrity protection supported by the UE for downlink and whether the UE requests the PDU session to be an always-on PDU session in the 5GS) for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, the UE is a UE operating in single-registration mode in the network supporting N26 interface, the UE has not previously successfully performed the UE-requested PDU session modification to indicate to the network the relevant 5GSM parameters and capabilities, and:

- a) timer T3396, T3584, T3585 or the back-off timer is running, the UE shall initiate the UE-requested PDU session modification procedure after expiry of timer T3396, T3584 or T3585 or after expiry of the back-off timer; or
- b) the UE is in substate 5GMM-REGISTERED.NON-ALLOWED-SERVICE and has not performed the UE-requested PDU session modification procedure (see subclause 5.3.5), the UE shall initiate the UE-requested PDU session modification procedure after entering substate 5GMM-REGISTERED.NORMAL-SERVICE.

6.4.2.2 UE-requested PDU session modification procedure initiation

In order to initiate the UE-requested PDU session modification procedure, the UE shall create a PDU SESSION MODIFICATION REQUEST message.

The UE shall allocate a PTI value currently not used and shall set the PTI IE of the PDU SESSION MODIFICATION REQUEST message to the allocated PTI value.

The UE shall not perform the UE-requested PDU session modification procedure for an emergency PDU session, except for a procedure initiated according to subclause 6.4.2.1, item e) only, and for the error cases described in subclause 6.4.1.3 and subclause 6.3.2.3.

The UE shall not perform the UE-requested PDU session modification procedure for a PDU session for LADN when the UE is located outside the LADN service area except for indicating a change of 3GPP PS data off UE status.

If the UE requests a specific QoS handling and the PDU session is not associated with the control plane only indication, the UE shall include the Requested QoS rules IE indicating requested QoS rules or the Requested QoS flow descriptions IE indicating requested QoS flow descriptions or both for the specific QoS handling. The Requested QoS rules IE includes the packet filters which describe the service data flows requested by the UE. The specific QoS parameters requested by the UE are specified in the Requested QoS flow descriptions IE. If the UE requests the network to bind specific service data flows to a dedicated QoS flow, the UE shall create a new QoS rule by setting the rule operation code to "Create new QoS rule" and shall set the segregation bit to "Segregation requested" for the corresponding QoS rule in the Requested QoS rules IE. The UE shall set the QRI values to "no QoS rule identifier assigned" in the Requested QoS rules IE, if the QoS rules are newly created; otherwise, the UE shall set the QRI values to those of the existing QoS rules for which the specific QoS handling applies. The UE shall set the QFI values to "no QoS flow identifier assigned" in the Requested OoS flow descriptions IE, if the OoS flow descriptions are newly created; otherwise, the UE shall set the QFI values to the QFIs of the existing QoS flow descriptions for which the specific QoS handling applies. The UE shall not request to create more than one QoS flow in a UE-requested PDU session modification procedure. If the SMF receives a PDU SESSION MODIFICATION REQUEST message with a Requested QoS rules IE containing more than one QoS rule with the rule operation code set to "Create new QoS rule", the SMF shall assign the same QFI to all the QoS rules which are created.

If the UE requests to join or leave one or more multicast MBS sessions associated with a PDU session, the UE shall include the Requested MBS container IE in the PDU SESSION MODIFICATION REQUEST message and shall set the

MBS operation to "Join multicast MBS session" for the join case or to "Leave multicast MBS session" for the leave case. The UE shall include the multicast MBS session information(s) and shall set the Type of multicast MBS session ID for each of the multicast MBS session information to either "Temporary Mobile Group Identity (TMGI)" or "Source specific IP multicast address" depending on the type of the multicast MBS session ID available in the UE. Then the remaining values of each of the multicast MBS session informations shall be set as following:

- a) if the Type of multicast MBS session ID is set to "Temporary Mobile Group Identity (TMGI)", the UE shall set the multicast MBS session ID to the TMGI; or
- b) if the Type of multicast MBS session ID is set to "Source specific IP multicast address for IPv4" or " Source specific IP multicast address for IPv6", the UE shall set the Source IP address information and the Destination IP address information to the corresponding values.

The UE should not request to join a multicast MBS session for local MBS service if neither current TAI nor CGI of the current cell is part of the MBS service area(s) of the multicast MBS session, if the UE has valid information of the MBS service area(s) of the multicast MBS session.

NOTE 1: The UE obtains the details of the multicast MBS session ID(s) i.e. TMGI, Source IP address information and Destination IP address information as a pre-configuration in the UE or during the MBS service announcement which is out of scope of this specification.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is of "IPv4", "IPv6", "IPv4v6", or "Ethernet" PDU session type, the PDU session is not associated with the control plane only indication:

- a) the UE is performing the PDU session modification procedure to indicate the support of reflective QoS and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication, the UE shall set the RQoS bit to "Reflective QoS supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message; or
- b) the UE is performing the PDU session modification procedure to indicate that reflective QoS is not supported and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication, the UE shall set the RQoS bit to "Reflective QoS not supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message.

If the UE is performing the PDU session modification procedure to revoke the previously indicated support of reflective QoS and the PDU session is not associated with the control plane only indication, the UE shall set the RQoS bit to "Reflective QoS not supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message. The UE shall not indicate support for reflective QoS for this PDU Session for the remaining lifetime of the PDU Session.

NOTE 2: The determination to revoke the usage of reflective QoS by the UE for a PDU session is implementation dependent.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is of "IPv6" or "IPv4v6" PDU session type, the PDU session is not associated with the control plane only indication:

- a) the UE is performing the PDU session modification procedure to indicate the support of Multi-homed IPv6 PDU session and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication, the UE shall set the MH6-PDU bit to "Multi-homed IPv6 PDU session supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message; or
- b) the UE is performing the PDU session modification procedure to indicate that Multi-homed IPv6 PDU session is not supported and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication, the UE shall set the MH6-PDU bit to "Multi-homed IPv6 PDU session not supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is of "IPv4", "IPv6", "IPv4v6", or "Ethernet" PDU session type, the PDU session is not associated with the control plane only indication, the UE supports more than 16 packet filters for this PDU session, and the UE has not previously successfully performed the UE-requested PDU session modification to provide this indication, the UE shall indicate the maximum

number of packet filters supported for the PDU session in the Maximum number of supported packet filters IE of the PDU SESSION MODIFICATION REQUEST message.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is not associated with the control plane only indication, and the UE has not previously successfully performed the UE-requested PDU session modification to include the Integrity protection maximum data rate IE in the PDU SESSION MODIFICATION REQUEST message, the UE shall include the Integrity protection maximum data rate IE in the PDU SESSION MODIFICATION MODIFICATION REQUEST message.

If the UE is performing the PDU session modification procedure

- a) to request the deletion of a non-default QoS rule due to errors in QoS operations or packet filters;
- b) to request the deletion of a QoS flow description due to errors in QoS operations; or
- c) to request the deletion of a mapped EPS bearer context due to errors in mapped EPS bearer operation, TFT operation or packet filters,

the UE shall include the 5GSM cause IE in the PDU SESSION MODIFICATION REQUEST message as described in subclauses 6.3.2.3, 6.3.2.4 and 6.4.1.3.

When the UE-requested PDU session modification procedure is used to indicate a change of 3GPP PS data off UE status for a PDU session, the UE shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION REQUEST message and setting the 3GPP PS data off UE status.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the PDU session is not associated with the control plane only indication, the UE requests the PDU session to be an always-on PDU session in the 5GS and the UE has not previously successfully performed the UE-requested PDU session modification to request this, the UE shall include the Always-on PDU session requested IE and set the value of the IE to "Always-on PDU session requested" in the PDU SESSION MODIFICATION REQUEST message.

If a port management information container needs to be delivered (see 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9]), the UE shall include a Port management information container IE in the PDU SESSION MODIFICATION REQUEST message.

To request re-negotiation of IP header compression configuration, the UE shall include the IP header compression configuration IE in the PDU SESSION MODIFICATION REQUEST message if the network indicated "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature support IE.

To request re-negotiation of Ethernet header compression configuration, the UE shall include the Ethernet header compression configuration IE in the PDU SESSION MODIFICATION REQUEST message if the network indicated "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization support in the 5GS network support feature support IE.

After an inter-system change from S1 mode to N1 mode, if:

- a) the UE is operating in single-registration mode in the network supporting N26 interface;
- b) the PDU session type value of the PDU session type IE is set to "IPv4", "IPv6" or "IPv4v6";
- c) the UE indicates "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- d) the network indicates "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message;

the UE shall initiate the PDU session modification procedure to negotiate the IP header compression configuration and include the IP header compression configuration IE in the PDU SESSION MODIFICATION REQUEST message.

The UE shall include the Service-level-AA container IE in the PDU SESSION MODIFICATION REQUEST message, when requesting to modify an established PDU session for C2 communication. In the Service-level-AA container IE, the UE shall include:

- a) the service-level device ID with the value set to the CAA-level UAV ID of the UE; and
- b) if available, the service-level-AA payload with the value set to the C2 authorization payload and the service-level-AA payload type with the value set to "C2 authorization payload".
- NOTE 3: The C2 authorization payload in the service-level-AA payload can include the pairing information for C2 communication and the UAV flight authorization information.

After an inter-system change from S1 mode to N1 mode, if:

- a) the UE is operating in single-registration mode in a network that supports N26 interface;
- b) the PDU session type value of the PDU session type IE is set to "Ethernet";
- c) the UE indicates "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- d) the network indicates "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message;

the UE shall initiate the PDU session modification procedure to negotiate the Ethernet header compression configuration and include the Ethernet header compression configuration IE in the PDU SESSION MODIFICATION REQUEST message.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, and if the UE is a UE operating in single-registration mode in a network supporting N26 interface, and the UE supports receiving DNS server addresses in protocol configuration options and the UE has not previously successfully performed the UE-requested PDU session modification to indicate this support, the UE shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION REQUEST message and:

- a) if the PDU session is of "IPv4" or "IPv4v6" PDU session type, the UE shall include the DNS server IPv4 address request; and
- b) if the PDU session is of "IPv6" or "IPv4v6" PDU session type, the UE shall include the DNS server IPv6 address request.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, and if the UE is a UE operating in single-registration mode in a network supporting N26 interface, and the UE supports the EAS rediscovery and the UE has not previously successfully performed the UE-requested PDU session modification to indicate this support, the UE shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION REQUEST message and shall include the EAS rediscovery support indication in the Extended protocol configuration options IE.

For a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, and if the UE is a UE operating in single-registration mode in a network supporting N26 interface, and the UE supports the EDC and the UE has not previously successfully performed the UE-requested PDU session modification to indicate this support, then the UE shall include the Extended protocol configuration options IE in the PDU SESSION MODIFICATION REQUEST message and shall include the EDC support indicator in the Extended protocol configuration options IE.

The UE shall transport:

- a) the PDU SESSION MODIFICATION REQUEST message;
- b) the PDU session ID; and
- c) if the UE-requested PDU session modification:
 - 1) is not initiated to indicate a change of 3GPP PS data off UE status associated to a PDU session, then the request type set to "modification request"; and

3GPP TS 24.501 version 17.16.0 Release 17

593

2) is initiated to indicate a change of 3GPP PS data off UE status associated to a PDU session, then without transporting the request type;

using the NAS transport procedure as specified in subclause 5.4.5, and the UE shall start timer T3581 (see example in figure 6.4.2.2.1).

For a PDN connection established when in S1 mode and not associated with the control plane only indication, after inter-system change from S1 mode to N1 mode, if the UE is registered in a network supporting the ATSSS,

- a) the UE may request to modify a PDU session to an MA PDU session; or
- b) the UE may allow the network to upgrade the PDU session to an MA PDU session. In order for the UE to allow the network to upgrade the PDU session to an MA PDU session, the UE shall set "MA PDU session network upgrade is allowed" in the MA PDU session information IE and set the request type to "modification request" in the UL NAS TRANSPORT message.
- NOTE 4: If the DNN corresponds to an LADN DNN, the AMF does not forward the MA PDU session information IE to the SMF but sends the message back to the UE to inform of the unhandled request (see subclause 5.4.5.2.5).

In case the UE executes case a) or b):

- if the UE supports ATSSS Low-Layer functionality with any steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "ATSSS Low-Layer functionality with any steering mode supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message;
- 2) if the UE supports MPTCP functionality with any steering mode and ATSSS-LL functionality with only Active-Standby steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "MPTCP functionality with any steering mode and ATSSS-LL functionality with only Active-Standby steering mode supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message;
- 3) if the UE supports MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode as specified in subclause 5.32.6 of 3GPP TS 23.501 [8], the UE shall set the ATSSS-ST bits to "MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message; and
- 4) if a performance measurement function in the UE can perform access performance measurements using the QoS flow of the non-default QoS rule as specified in subclause 5.32.5 of 3GPP TS 23.501 [8], the UE shall set the APMQF bit to "Access performance measurements per QoS flow supported" in the 5GSM capability IE of the PDU SESSION MODIFICATION REQUEST message.

Upon receipt of a PDU SESSION MODIFICATION REQUEST message for MA PDU session modification, the SMF shall check if the 5GSM capability IE in the PDU SESSION MODIFICATION REQUEST message, includes:

- a) the ATSSS-ST bits set to "MPTCP functionality with any steering mode and ATSSS-LL functionality with only active-standby steering mode supported" and:
 - i) if the DNN configuration allows for the MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode, the SMF shall ensure that the modified PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with any steering mode in the downlink and MPTCP with any steering mode and ATSSS-LL with only active-standby steering mode in the uplink; or
 - ii) if the DNN configuration allows for the MPTCP functionality with any steering mode and ATSSS-LL functionality with only active-standby steering mode, the SMF shall ensure that the modified PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with only active-standby steering mode in the downlink and the uplink;
- b) the ATSSS-ST bits set to "ATSSS Low-Layer functionality with any steering mode allowed for ATSSS-LL supported" and if the DNN configuration allows for the ATSSS-LL functionality with any steering mode, the SMF shall ensure that the modified PDU session has the capability of ATSSS-LL with any steering mode in the downlink and the uplink; or
- c) the ATSSS-ST bits set to "MPTCP functionality with any steering mode and ATSSS-LL functionality with any steering mode allowed for ATSSS-LL supported" and if the DNN configuration allows for the MPTCP

functionality with any steering mode and ATSSS-LL functionality with any steering mode, the SMF shall ensure that the modified PDU session has the capability of MPTCP with any steering mode and ATSSS-LL with any steering mode in the downlink and the uplink.



Figure 6.4.2.2.1: UE-requested PDU session modification procedure

6.4.2.3 UE-requested PDU session modification procedure accepted by the network

Upon receipt of a PDU SESSION MODIFICATION REQUEST message, if the SMF accepts the request to modify the PDU session, the SMF shall perform the network-requested PDU session modification procedure as specified in subclause 6.3.2.

If the PDU SESSION MODIFICATION REQUEST message contains a Port management information container IE, the SMF shall handle the contents of the Port management information container IE as specified in 3GPP TS 23.501 [8] and 3GPP TS 23.502 [9].

6.4.2.4 UE-requested PDU session modification procedure not accepted by the network

6.4.2.4.1 General

Upon receipt of a PDU SESSION MODIFICATION REQUEST message, if the SMF does not accepts the request to modify the PDU session, the SMF shall create a PDU SESSION MODIFICATION REJECT message.

The SMF shall set the 5GSM cause IE of the PDU SESSION MODIFICATION REJECT message to indicate the reason for rejecting the PDU session modification.

The 5GSM cause IE typically indicates one of the following SM cause values:

- #26 insufficient resources;
- #29 user authentication or authorization failed;
- #31 request rejected, unspecified;
- #32 service option not supported;
- #33 requested service option not subscribed;
- #35 PTI already in use;
- #37 5GS QoS not accepted;

- #43 Invalid PDU session identity;
- #44 Semantic errors in packet filter(s);
- #45 Syntactical error in packet filter(s);
- #46 out of LADN service area;
- #59 unsupported 5QI value;
- #67 insufficient resources for specific slice and DNN;
- #69 insufficient resources for specific slice;
- #83 Semantic error in the QoS operation;
- #84 Syntactical error in the QoS operation; or
- #95 111 protocol errors.

If the UE requests a PDU session modification for an LADN when the UE is located outside of the LADN service area, the SMF shall include the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION MODIFICATION REJECT message.

If the Extended protocol configuration options IE of the PDU SESSION MODIFICATION REQUEST message indicates 3GPP PS data off UE status and the SMF detects the change of the 3GPP PS data off UE status, the SMF shall not include the 5GSM cause value #26 "insufficient resources", the 5GSM cause value #67 "insufficient resources for specific slice and DNN", the 5GSM cause value #69 "insufficient resources for specific slice" and the 5GSM cause value #46 "out of LADN service area" in the 5GSM cause IE of the PDU SESSION MODIFICATION REJECT message.

If the UE initiates UE-requested PDU session modification procedure to modify the PDU session transferred from EPS to an MA PDU session with the Request type IE set to "MA PDU request" in the UL NAS TRANSPORT message as specified in 3GPP TS 24.193 [13B] and the SMF determines, based on operator policy and subscription information, that the PDU SESSION MODIFICATION REQUEST message is to be rejected, the SMF shall include the 5GSM cause value #33 "requested service option not subscribed" in the 5GSM cause IE of the PDU SESSION MODIFICATION REJECT message.

NOTE: If the SMF determines, based on operator policy and subscription information, that the PDU SESSION MODIFICATION REQUEST message is to be accepted as single access PDU session, the ATSSS container IE cannot be included in the PDU SESSION MODIFICATION COMMAND message.

The network may include a Back-off timer value IE in the PDU SESSION MODIFICATION REJECT message.

If the 5GSM cause value is #26"insufficient resources", #67 "insufficient resources for specific slice and DNN", or #69 "insufficient resources for specific slice" and the PDU SESSION MODIFICATION REQUEST message was received from a UE configured for high priority access in selected PLMN or the request type provided during the PDU session establishment is set to "initial emergency request" or "existing emergency PDU session", the network shall not include a Back-off timer value IE.

The SMF shall send the PDU SESSION MODIFICATION REJECT message.

Upon receipt of a PDU SESSION MODIFICATION REJECT message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE shall stop timer T3581, release the allocated PTI value, and enter the state PROCEDURE TRANSACTION INACTIVE.

6.4.2.4.2 Handling of network rejection due to congestion control

If:

- the 5GSM cause value #26 "insufficient resources" and the Back-off timer value IE are included in the PDU SESSION MODIFICATION REJECT message; or
- an indication that the 5GSM message was not forwarded due to DNN based congestion control is received along a Back-off timer value and a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the Re-attempt indicator IE or the 5GSM congestion re-attempt indicator IE provided by the network, if any, and the UE shall take different actions depending on the timer value received for timer T3396 in the Back-off timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.7):

- a) If the timer value indicates neither zero nor deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the timer value indicates neither zero nor deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE shall then start timer T3396 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN that was sent by the UE, until timer T3396 expires or timer T3396 is stopped; and
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without a DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until timer T3396 expires or timer T3396 is stopped.

The UE shall not stop timer T3396 upon a PLMN change or inter-system change.

- b) if the timer value indicates that this timer is deactivated and a DNN was provided during the PDU session establishment, the UE shall stop timer T3396 associated with the corresponding DNN, if it is running. If the timer value indicates that this timer is deactivated and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3396 associated with no DNN if it is running. The UE:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST with exception of those identified in subclause 6.4.2.1, for the same DNN until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the same DNN from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the same DNN from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the same DNN from the network; and
 - 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without a DNN provided by the UE, if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without a DNN provided by the UE, a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established without a DNN provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established without a DNN provided by the UE.

The timer T3396 remains deactivated upon a PLMN change or inter-system change.

- c) if the timer value indicates zero, the UE:
 - shall stop timer T3396 associated with the corresponding DNN, if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN; and
 - 2) if no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer

T3396 associated with no DNN, if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message without a DNN, or another PDU SESSION MODIFICATION REQUEST message without a DNN provided by the UE.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same DNN or without a DNN as specified in clause 6.2.7.

If the timer T3396 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3396 is associated (if any) is not updated, then timer T3396 is kept running until it expires or it is stopped

When the timer T3396 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the UE is switched off when the timer T3396 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3396 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3396 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the UE is a UE operating in single-registration mode in a network supporting N26 interface and the PDU SESSION MODIFICATION REQUEST message was sent for a PDN connection established when in S1 mode after an intersystem change from S1 mode to N1 mode and timer T3396 associated with the corresponding DNN (or no DNN) is running, then the UE shall re-initiate the UE-requested PDU session modification procedure after expiry of timer T3396.

If:

- the 5GSM cause value #67 "insufficient resources for specific slice and DNN" and the Back-off timer value IE are included in the PDU SESSION MODIFICATION REJECT message; or
- an indication that the 5GSM message was not forwarded due to S-NSSAI and DNN based congestion control is received along a Back-off timer value and a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the Re-attempt indicator IE provided by the network, if any, and take different actions depending on the timer value received for timer T3584 in the Back-off timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.8):

- a) If the timer value indicates neither zero nor deactivated, and both an S-NSSAI and a DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated, an S-NSSAI and no DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with [S-NSSAI of the PDU session, no DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated, no S-NSSAI and a DNN was provided during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated and neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency request" and different from "session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination, if it is running. If the timer value indicates neither zero nor deactivated and neither S-NSSAI nor DNN was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination, if it is running. The timer T3584 to be stopped includes the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running. The UE shall then start timer T3584 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - shall not send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [S-NSSAI, DNN] combination, until timer T3584 expires or timer T3584 is stopped;

- 2) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, no DNN] combination, if no DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped;
- 3) shall not send another PDU SESSION ESTABLISHMENT REQUEST message, or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, DNN] combination, if no S-NSSAI was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped; and
- 4) shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with the exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, no DNN] combination, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until timer T3584 expires or timer T3584 is stopped.

The UE shall not stop timer T3584 upon a PLMN change or inter-system change;

- b) if the timer value indicates that this timer is deactivated:
 - 1) if both S-NSSAI and DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, DNN] combination that was sent by the UE, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the [S-NSSAI of the PDU session, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the [S-NSSAI of the PDU session, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the [S-NSSAI of the PDU session, DNN] combination from the network;
 - 2) if an S-NSSAI was provided but a DNN was not provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [S-NSSAI of the PDU session, no DNN] combination, if no DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the [S-NSSAI of the PDU session, no DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established for the [S-NSSAI of the PDU session, no DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established for the [S-NSSAI of the PDU session, no DNN] combination from the network;
 - 3) if an S-NSSAI was not provided but a DNN was provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message, or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, DNN] combination, if no S-NSSAI was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for the [no S-NSSAI, DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for the [no S-NSSAI, DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the [no S-NSSAI, DNN] combination from the network; and

4) if neither S-NSSAI nor DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE shall not send a PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or a PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the [no S-NSSAI, no DNN] combination, if neither S-NSSAI nor DNN was provided during the PDU session establishment, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established for the [no S-NSSAI, no DNN] combination from the network.

The timer T3584 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero:
 - if both S-NSSAI and DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the [S-NSSAI of the PDU session, DNN] combination;
 - 2) if an S-NSSAI was provided but a DNN was not provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [S-NSSAI of the PDU session, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the [S-NSSAI of the PDU session, no DNN] combination if the request type was different from "initial emergency request" and different from "existing emergency PDU session";
 - 3) if an S-NSSAI was not provided but a DNN was provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the [no S-NSSAI, DNN] combination; and
 - 4) if neither S-NSSAI nor DNN were provided by the UE during the PDU session establishment, the UE shall stop timer T3584 associated with the [no S-NSSAI, no DNN] combination (including the timer T3584 applied for all the PLMNs, if running, and the timer T3584 applied for the registered PLMN, if running), if it is running. The UE may send another PDU SESSION ESTABLISHMENT REQUEST message, or PDU SESSION MODIFICATION REQUEST message for the [no S-NSSAI, no DNN] combination and the request type was different from "initial emergency request" and different from "existing emergency PDU session".

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION MODIFICATION REJECT message with the 5GSM cause value #67 "insufficient resources for specific slice and DNN", then the UE shall apply the timer T3584 for all the PLMNs. Otherwise, the UE shall apply the timer T3584 for the registered PLMN.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same [S-NSSAI, DNN] combination, or for the same [S-NSSAI, no DNN] combination, or for the same [no S-NSSAI, no DNN] combination as specified in clause 6.2.8.

When the timer T3584 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3584 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3584 is associated (if any) is not updated, then timer T3584 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3584 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3584 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3584 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the UE is a UE operating in single-registration mode in a network supporting N26 interface and the PDU SESSION MODIFICATION REQUEST message was sent for a PDN connection established when in S1 mode after an intersystem change from S1 mode to N1 mode and timer T3584 associated with the corresponding [no S-NSSAI, DNN] combination or [no S-NSSAI, no DNN] combination is running, then the UE shall re-initiate the UE-requested PDU session modification procedure after expiry of timer T3584.

If:

- the 5GSM cause value #69 "insufficient resources for specific slice" and the Back-off timer value IE are included in the PDU SESSION MODIFICATION REJECT message; or
- an indication that the 5GSM message was not forwarded due to S-NSSAI only based congestion control is received along a Back-off timer value and a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the PDU session ID of the PDU session;

the UE shall ignore the bit "RATC" and the bit "EPLMNC" in the Re-attempt indicator IE provided by the network, if any, and take different actions depending on the timer value received for timer T3585 in the Back-off timer value IE or depending on the Back-off timer value received from the 5GMM sublayer (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.8):

- a) If the timer value indicates neither zero nor deactivated and an S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session, if it is running. If the timer value indicates neither zero nor deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. The UE shall then start timer T3585 with the value provided in the Back-off timer value IE or with the Back-off timer value received from the 5GMM sublayer and:
 - if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session for the S-NSSAI of the PDU session, until timer T3585 expires or timer T3585 is stopped; and
 - 2) if the request type was different from "initial emergency request" and from "existing emergency PDU session", and an S-NSSAI was not provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, , until timer T3585 expires or timer T3585 is stopped.

The UE shall not stop timer T3585 upon a PLMN change or inter-system change;

b) if the timer value indicates that this timer is deactivated and an S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session, if it is running. If the timer value indicates that this timer is deactivated and no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request" and different

from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI if it is running. The timer T3585 to be stopped includes the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running. In addition:

- 1) if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session for the S-NSSAI of the PDU session until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session for the S-NSSAI of the PDU session from the network, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session for the S-NSSAI of the PDU session from the network, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for the S-NSSAI of the PDU session from the network; and
- 2) if the request type was different from "initial emergency request" and from "existing emergency PDU session", and an S-NSSAI was not provided by the UE during the PDU session establishment, the UE shall not send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI and with request type different from "initial emergency request" and different from "existing emergency PDU session", or another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for a non-emergency PDU session established without an S-NSSAI provided by the UE, until the UE is switched off, the USIM is removed, the entry in the "list of subscriber data" for the current SNPN is updated, or the UE receives a PDU SESSION MODIFICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION AUTHENTICATION COMMAND message for a non-emergency PDU session established without an S-NSSAI provided by the UE, or a PDU SESSION RELEASE COMMAND message without the Back-off timer value IE for a non-emergency PDU session established without an S-NSSAI provided by the UE.

The timer T3585 remains deactivated upon a PLMN change or inter-system change; and

- c) if the timer value indicates zero:
 - if an S-NSSAI was provided by the UE during the PDU session establishment, the UE shall stop timer T3585 associated with the S-NSSAI of the PDU session (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the S-NSSAI of the PDU session; and
 - 2) if no S-NSSAI was provided during the PDU session establishment and the request type was different from "initial emergency request " and different from "existing emergency PDU session", the UE shall stop timer T3585 associated with no S-NSSAI (including the timer T3585 applied for all the PLMNs, if running, and the timer T3585 applied for the registered PLMN, if running), if running, and may send another PDU SESSION ESTABLISHMENT REQUEST message without an S-NSSAI, or another PDU SESSION MODIFICATION REQUEST message without an S-NSSAI provided by the UE.

If the 5GSM congestion re-attempt indicator IE with the ABO bit set to "The back-off timer is applied in all PLMNs" is included in the PDU SESSION MODIFICATION REJECT message with the 5GSM cause value #69 "insufficient resources for specific slice", then the UE shall apply the timer T3585 for all the PLMNs. Otherwise, the UE shall apply the timer T3585 for the registered PLMN.

If the Back-off timer value IE is not included or no Back-off timer value is received from the 5GMM sublayer, then the UE may send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the same S-NSSAI or without an S-NSSAI as specified in clause 6.2.8.

When the timer T3585 is running or the timer is deactivated, the UE is allowed to initiate a PDU session establishment procedure for emergency services.

If the timer T3585 is running when the UE enters state 5GMM-DEREGISTERED, the UE remains switched on, and the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3585 is associated (if any) is not updated, then timer T3585 is kept running until it expires or it is stopped.

If the UE is switched off when the timer T3585 is running, and if the USIM in the UE (if any) remains the same and the entry in the "list of subscriber data" for the SNPN to which timer T3585 is associated (if any) is not updated when the UE is switched on, the UE shall behave as follows:

let t1 be the time remaining for T3585 timeout at switch off and let t be the time elapsed between switch off and switch on. If t1 is greater than t, then the timer shall be restarted with the value t1 – t. If t1 is equal to or less than t, then the timer need not be restarted. If the UE is not capable of determining t, then the UE shall restart the timer with the value t1.

If the UE is a UE operating in single-registration mode in a network supporting N26 interface and the PDU SESSION MODIFICATION REQUEST message was sent for a PDN connection established when in S1 mode after an intersystem change from S1 mode to N1 mode and timer T3585 associated with no S-NSSAI is running, then the UE shall re-initiate the UE-requested PDU session modification procedure after expiry of timer T3585.

NOTE 3: As described in this subclause, upon PLMN change or inter-system change, the UE does not stop the timer T3584 or T3585. This means the timer T3584 or T3585 can still be running or be deactivated for the given 5GSM procedure, the PLMN, the S-NSSAI and optionally the DNN combination when the UE returns to the PLMN or when it performs inter-system change back from S1 mode to N1 mode. Thus the UE can still be prevented from sending another PDU SESSION ESTABLISHMENT REQUEST or PDU SESSION MODIFICATION REQUEST message in the PLMN for the same S-NSSAI and optionally the same DNN.

Upon PLMN change, if T3584 is running or is deactivated for an S-NSSAI, a DNN, and old PLMN, but T3584 is not running and is not deactivated for the S-NSSAI, the DNN, and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI and the same DNN in the new PLMN.

Upon PLMN change, if T3585 is running or is deactivated for an S-NSSAI and old PLMN, but T3585 is not running and is not deactivated for the S-NSSAI and new PLMN, then the UE is allowed to send a PDU SESSION ESTABLISHMENT REQUEST message for the same S-NSSAI in the new PLMN.

6.4.2.4.3 Handling of network rejection not due to congestion control

If the 5GSM cause value is different from #26 "insufficient resources", #37 "5GS QoS not accepted", #44 "Semantic errors in packet filter(s)", #45 "Syntactical error in packet filter(s)", #46 "out of LADN service area", #59 "unsupported 5QI value", #67 "insufficient resources for specific slice and DNN", #69 "insufficient resources for specific slice", #83 "Semantic error in the QoS operation", and #84 "Syntactical error in the QoS operation", and the Back-off timer value IE is included, the UE shall behave as follows: (if the UE is a UE configured for high priority access in selected PLMN, exceptions are specified in subclause 6.2.12):

- a) if the timer value indicates neither zero nor deactivated and:
 - if the UE provided DNN and S-NSSAI to the network during the PDU session establishment, the UE shall start the back-off timer with the value provided in the Back-off timer value IE for the PDU session modification procedure and [PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session] combination. The UE shall not send another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN and the (mapped) HPLMN S-NSSAI of the PDU session in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated; or
 - 2) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment, it shall start the back-off timer accordingly for the PDU session modification procedure and the [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session] or [PLMN, no DNN, no S-NSSAI] combination. Dependent on the combination, the UE shall not send another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the back-off timer expires, the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated;
- b) if the timer value indicates that this timer is deactivated and:
 - if the UE provided DNN and S-NSSAI to the network during the PDU session establishment, the UE shall not send another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same DNN and the (mapped) HPLMN S-NSSAI of the PDU session in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated; or

- 2) if the UE did not provide a DNN or S-NSSAI or any of the two parameters to the network during the PDU session establishment, the UE shall not send another PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, for the same [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session] or [PLMN, no DNN, no S-NSSAI] combination in the current PLMN, until the UE is switched off, the USIM is removed, or the entry in the "list of subscriber data" for the current SNPN is updated; and
- c) if the timer value indicates zero, the UE may send another PDU SESSION MODIFICATION REQUEST message for the same combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [PLMN, no DNN, no S-NSSAI] in the current PLMN.

If the Back-off timer value IE is not included, then the UE shall ignore the Re-attempt indicator IE provided by the network in the PDU SESSION MODIFICATION REJECT message, if any.

- a) Additionally, if the 5GSM cause value is #32 "service option not supported", or #33 "requested service option not subscribed", then:
 - 1) the UE not operating in SNPN access operation mode shall proceed as follows:
 - i) if the UE is registered in the HPLMN or in a PLMN that is within the EHPLMN list, the UE shall behave as described above in the present subclause using the configured SM Retry Timer value as specified in 3GPP TS 24.368 [17] or in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22], if available, as back-off timer value; and
- NOTE 1: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if the UE is configured with:
 - an SM Retry Timer value in the ME as specified in 3GPP TS 24.368 [17]; and
 - an SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22].
 - ii) otherwise, if the UE is not registered in its HPLMN or in a PLMN that is within the EHPLMN list, or if the SM Retry Timer value is not configured, the UE shall behave as described above in the present subclause, using the default value of 12 minutes for the back-off timer; or
 - 2) the UE operating in SNPN access operation mode shall proceed as follows:
 - i) if:
 - A) the SM Retry Timer value for the current SNPN as specified in 3GPP TS 24.368 [17] is available; or
 - B) the SM Retry Timer value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22] is available and the UE used the USIM for registration to the current SNPN;

then the UE shall behave as described above in the present subclause using the configured SM Retry Timer value as back-off timer value; or

- NOTE 2: The way to choose one of the configured SM Retry Timer values for back-off timer value is up to UE implementation if both conditions in bullets A) and B) above are satisfied.
 - ii) otherwise, the UE shall behave as described above in the present subclause, using the default value of 12 minutes for the back-off timer.
- b) For 5GSM cause values different from #32 "service option not supported", or #33 "requested service option not subscribed", the UE behaviour regarding the start of a back-off timer is specified in clause 6.2.12.

The UE shall not stop any back-off timer:

- a) upon a PLMN change;
- b) upon an inter-system change; or
- c) upon registration over another access type.

If the network indicates that a back-off timer for the PDU session modification procedure is deactivated, then it remains deactivated:

- a) upon a PLMN change;
- b) upon an inter-system change; or
- c) upon registration over another access type.
- NOTE 3: This means the back-off timer can still be running or be deactivated for the given 5GSM procedure when the UE returns to the PLMN or when it performs inter-system change back from S1 mode to N1 mode. Thus the UE can still be prevented from sending another PDU SESSION MODIFICATION REQUEST message for the combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [PLMN, no DNN, no S-NSSAI] in the PLMN.

If the back-off timer is started upon receipt of a PDU SESSION MODIFICATION REJECT (i.e. the timer value was provided by the network, a configured value is available or the default value is used as explained above) or the back-off timer is deactivated, the UE behaves as follows:

a) after a PLMN change the UE may send a PDU SESSION MODIFICATION REQUEST message for the combination of [new PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session], [new PLMN, DNN, no S-NSSAI], [new PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [new PLMN, no DNN, no S-NSSAI] in the new PLMN, if the back-off timer is not running and is not deactivated for the PDU session modification procedure and the combination of [new PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session], [new PLMN, DNN, no S-NSSAI], [new PLMN, DNN, no S-NSSAI], [new PLMN, no DNN, no S-NSSAI], [new PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [new PLMN, DNN, no S-NSSAI];

Furthermore, as an implementation option, for the 5GSM cause value #32 "service option not supported" or #33 "requested service option not subscribed", if the network does not include a Re-attempt indicator IE, the UE may decide not to automatically send another PDU SESSION MODIFICATION REQUEST message for the same combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session], [PLMN, DNN, no S-NSSAI], [PLMN, no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [PLMN, no DNN, no S-NSSAI], if the UE is registered to a new PLMN which is in the list of equivalent PLMNs.

- b) if the network does not include the Re-attempt indicator IE to indicate whether re-attempt in S1 mode is allowed, or the UE ignores the Re-attempt indicator IE, e.g. because the Back-off timer value IE is not included, then:
 - if the UE is registered in its HPLMN or in a PLMN that is within the EHPLMN list and the back-off timer is running for the combination of [PLMN, DNN, (mapped) HPLMN S-NSSAI of the PDU session] or [PLMN DNN, no S-NSSAI], the UE shall apply the configured SM_RetryAtRATChange value as specified in 3GPP TS 24.368 [17] or in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22], if available, to determine whether the UE may attempt an EPS bearer resource allocation procedure or an EPS bearer resource modification procedure for the same [PLMN, DNN] combination in S1 mode; and
- NOTE 4: The way to choose one of the configured SM_RetryAtRATChange values for back-off timer value is up to UE implementation if the UE is configured with:
 - an SM_RetryAtRATChange value in ME as specified in 3GPP TS 24.368 [17]; and
 - an SM_RetryAtRATChange value in USIM file NAS_{CONFIG} as specified in 3GPP TS 31.102 [22].
 - 2) if the UE is not registered in its HPLMN or in a PLMN that is within the EHPLMN list, or if the NAS configuration MO as specified in 3GPP TS 24.368 [17] is not available and the value for inter-system change is not configured in the USIM file NAS_{CONFIG}, then the UE behaviour regarding an EPS bearer resource allocation procedure or an EPS bearer resource modification procedure for the same [PLMN, DNN] combination in S1 mode is unspecified; and
- c) if the network includes the Re-attempt indicator IE indicating that re-attempt in an equivalent PLMN is not allowed, then depending on the timer value received in the Back-off timer value IE, for each combination of a PLMN from the equivalent PLMN list and the respective [DNN, (mapped) HPLMN S-NSSAI of the PDU session], [DNN, no S-NSSAI], [no DNN, (mapped) HPLMN S-NSSAI of the PDU session], or [no DNN, no S-NSSAI] combination, the UE shall start a back-off timer for the PDU session modification procedure with the value provided by the network, or deactivate the respective back-off timer as follows:
 - 1) if the Re-attempt indicator IE additionally indicates that re-attempt in S1 mode is allowed, the UE shall start or deactivate the back-off timer for N1 mode only; and
 - 2) otherwise, the UE shall start or deactivate the back-off timer for S1 and N1 mode.

If the back-off timer for a [PLMN, DNN] or [PLMN, no DNN] combination was started or deactivated in S1 mode upon receipt of BEARER RESOURCE ALLOCATION REJECT message or BEARER RESOURCE MODIFICATION REJECT message (see 3GPP TS 24.301 [15]) and the network indicated that re-attempt in N1 mode is allowed, then this back-off timer does not prevent the UE from sending a PDU SESSION MODIFICATION REQUEST message in this PLMN for the same DNN after inter-system change to N1 mode. If the network indicated that re-attempt in N1 mode is not allowed, the UE shall not send any PDU SESSION MODIFICATION REQUEST message with exception of those identified in subclause 6.4.2.1, in this PLMN for the same DNN in combination with any S-NSSAI or without S-NSSAI, after inter-system change to N1 mode until the timer expires, the UE is switched off or the USIM is removed.

- NOTE 5: The back-off timer is used to describe a logical model of the required UE behaviour. This model does not imply any specific implementation, e.g. as a timer or timestamp.
- NOTE 6: Reference to back-off timer in this section can either refer to use of timer T3396 or to use of a different packet system specific timer within the UE. Whether the UE uses T3396 as a back-off timer or it uses different packet system specific timers as back-off timers is left up to UE implementation.

If the 5GSM cause value is #46 "out of LADN service area", the UE shall ignore the Back-off timer value IE and Reattempt indicator IE provided by the network, if any. If the UE is not located inside the LADN service area, the UE shall not send another PDU SESSION MODIFICATION REQUEST message or another PDU SESSION ESTABLISHMENT REQUEST message for the LADN DNN provided by the UE during the PDU session establishment procedure until the LADN information for the specific LADN DNN is updated as described in subclause 5.4.4 and subclause 5.5.1. If the UE is not located inside the LADN service area, the UE shall not indicate the PDU session(s) for the LADN DNN provided by the UE during the PDU session establishment procedure in the Uplink data status IE included in the SERVICE REQUEST message until the LADN information for the specific LADN DNN is provided by network as described in subclause 5.4.4 and subclause 5.5.1.

NOTE 7: Based on UE implementation, the UE locating inside the LADN service area can send another PDU SESSION ESTABLISHMENT REQUEST message or PDU SESSION MODIFICATION REQUEST message for the LADN DNN which was rejected with the 5GSM cause value #46 "out of LADN service area".

If the 5GSM cause value is #37 "5GS QoS not accepted", #44 "Semantic errors in packet filter(s)", #45 "Syntactical error in packet filter(s)", #59 "unsupported 5QI value", #83 "Semantic error in the QoS operation" or #84 "Syntactical error in the QoS operation", the UE shall ignore the Back-off timer value IE and Re-attempt indicator IE provided by the network, if any. The UE should pass the corresponding error cause to the upper layers.

NOTE 8: How to solve the issues of not accepted 5GS QoS and unsupported 5QI value in the upper layers is UE implementation specific.

6.4.2.5 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Expiry of timer T3581.

The UE shall, on the first expiry of the timer T3581, retransmit the PDU SESSION MODIFICATION REQUEST message and the PDU session information which was transported together with the initial transmission of the PDU SESSION MODIFICATION REQUEST message and shall reset and start timer T3581. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3581, the UE shall abort the procedure and shall release the allocated PTI.

b) Invalid PDU session identity.

Upon receipt of the PDU SESSION MODIFICATION REJECT message including 5GSM cause #43 "invalid PDU session identity", the UE shall perform a local release of the existing PDU session and shall stop the timer T3581.

c) Collision of network-requested PDU session release procedure and UE-requested PDU session modification procedure.

If the UE receives a PDU SESSION RELEASE COMMAND message during the UE-requested PDU session modification procedure, and the PDU session indicated in the PDU SESSION RELEASE COMMAND message is the PDU session that the UE had requested to modify, the UE shall abort the PDU session modification procedure and proceed with the network-requested PDU session release procedure.

d) Handling DL user data packets marked with RQI when UE has already revoked the usage of reflective QoS

If the UE receives a DL user data packet marked with a RQI and the DL user data packet belongs to a PDU session of IPv4, IPv6, IPv4v6 or Ethernet PDU session type for which the UE has already revoked the usage of reflective QoS, then the UE shall ignore the RQI and shall handle the received DL user data packet.

e) Collision of network-requested PDU session modification procedure and UE-requested PDU session modification procedure.

The handling of the same abnormal case as described in subclause 6.3.2.6 applies.

- f) Upon receiving an indication that the 5GSM message was not forwarded due to service area restrictions along with a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall abort the procedure and shall stop the timer T3581.
- g) Upon receiving an indication that the 5GSM message was not forwarded due to routing failure along with a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3581 and shall abort the procedure.
- ga) Upon receiving an indication that the 5GSM message was not forwarded because the UE accessing via a satellite NG-RAN cell is informed that the PLMN is not allowed to operate at the present UE location along with a PDU SESSION MODIFICATION REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3581 and shall abort the procedure.
- h) Collision of UE-requested PDU session modification procedure and N1 NAS signalling connection release

The UE may immediately retransmit the PDU SESSION MODIFICATION REQUEST message and stop, reset and restart timer T3581, if the following conditions apply:

- 1) The original UE-requested PDU session modification procedure was initiated over an existing N1 NAS signalling connection; and
- 2) the previous transmission of the PDU SESSION MODIFICATION REQUEST message was not initiated due to timer T3581 expiry.

6.4.2.6 Abnormal cases on the network side

The following abnormal cases can be identified:

- a) If the PDU session is an emergency PDU session and the 5GSM cause IE is not included in the PDU SESSION MODIFICATION REQUEST message which is not triggered according to subclause 6.4.2.1, item e), or is set to a 5GSM cause other than the 5GSM causes #41, #42, #44, #45, #83, #84, and #85, the SMF shall reject the PDU SESSION MODIFICATION REQUEST message with 5GSM cause #31 "request rejected, unspecified".
- b) PDU session inactive for the received PDU session identity.

If the PDU session ID in the PDU SESSION MODIFICATION REQUEST message belongs to any PDU session in state PDU SESSION INACTIVE in the SMF, the SMF shall set the 5GSM cause IE to #43 "Invalid PDU session identity" in the PDU SESSION MODIFICATION REJECT message.

c) Collision of network-requested PDU session modification procedure and UE-requested PDU session modification procedure.

The handling of the same abnormal case as described in subclause 6.3.2.5 applies.

d) AMF provides a "message was exempted from the DNN based congestion activated in the AMF" but the UErequested PDU session modification procedure is not exempt from DNN based congestion control.

If the SMF receives an exemptionInd attribute indicating "message was exempted from the DNN based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], and the Extended protocol configuration options IE of the PDU SESSION MODIFICATION REQUEST message does not indicate 3GPP PS data off UE status, then the SMF shall set the 5GSM cause #26 "insufficient resources" in the PDU SESSION MODIFICATION REJECT message.

e) AMF provides a "message was exempted from the S-NSSAI and DNN based congestion activated in the AMF" but the UE-requested PDU session modification procedure is not exempt from S-NSSAI only based congestion control.

If the SMF receives an exemptionInd attribute indicating "message was exempted from the S-NSSAI and DNN based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], and the Extended protocol configuration options IE of the PDU SESSION MODIFICATION REQUEST message does not indicate 3GPP PS data off UE status, then the SMF shall set the 5GSM cause #67 "insufficient resources for specific slice and DNN" in the PDU SESSION MODIFICATION REJECT message.

f) AMF provides a "message was exempted from the S-NSSAI only based congestion activated in the AMF" but the UE-requested PDU session modification procedure is not exempt from S-NSSAI only based congestion control.

If the SMF receives an exemptionInd attribute indicating "message was exempted from the S-NSSAI only based congestion activated in the AMF" as specified in 3GPP TS 29.502 [20A], and the Extended protocol configuration options IE of the PDU SESSION MODIFICATION REQUEST message does not indicate 3GPP PS data off UE status, then the SMF shall set the 5GSM cause #69 "insufficient resources for specific slice" in the PDU SESSION MODIFICATION REJECT message.

g) 5G access network cannot forward the message.

If the SMF determines based on content of the n2SmInfo attribute specified in 3GPP TS 29.502 [20A] that the DL NAS TRANSPORT message carrying the PDU SESSION MODIFICATION COMMAND message was not forwarded to the UE by the 5G access network due to a cause other than handover procedure in progress, then the SMF shall reject the UE-requested PDU session modification procedure with an appropriate 5GSM cause value in the PDU SESSION MODIFICATION REJECT message.

NOTE: The use of an appropriate 5GSM cause value is implementation specific.

6.4.3 UE-requested PDU session release procedure

6.4.3.1 General

The purpose of the UE-requested PDU session release procedure is to enable by the UE to request a release of a PDU session.

The UE is allowed to initiate the PDU session release procedure even if the timer T3396 is running.

The UE is allowed to initiate the PDU session release procedure even if the timer T3584 is running.

The UE is allowed to initiate the PDU session release procedure even if the timer T3585 is running.

The UE is allowed to initiate the PDU session release procedure even if the UE is outside the LADN service area.

6.4.3.2 UE-requested PDU session release procedure initiation

In order to initiate the UE-requested PDU session release procedure, the UE shall create a PDU SESSION RELEASE REQUEST message.

The UE may set the 5GSM cause IE of the PDU SESSION RELEASE REQUEST message to indicate the reason for releasing the PDU session.

The 5GSM cause IE typically indicates one of the following 5GSM cause values:

- #26 insufficient resources;
- #36 regular deactivation;
- #44 Semantic errors in packet filter(s);
- #45 Syntactical error in packet filter(s);
- #83 Semantic error in the QoS operation;

- #84 Syntactical error in the QoS operation;
- #96 Invalid mandatory information.

The UE shall allocate a PTI value currently not used and shall set the PTI IE of the PDU SESSION RELEASE REQUEST message to the allocated PTI value.

The UE shall transport the PDU SESSION RELEASE REQUEST message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, and the UE shall start timer T3582 (see example in figure 6.4.3.2.1).

If the UE is releasing the PDU session due to:

- a) errors in QoS operations or packet filters; or
- b) the number of the authorized QoS rules, the number of the packet filters, or the number of the authorized QoS flow descriptions associated with the PDU session have reached the maximum number supported by the UE,

the UE shall include the 5GSM cause IE in the PDU SESSION RELEASE REQUEST message as described in subclauses 6.3.2.4 and 6.4.1.3.



Figure 6.4.3.2.1: UE-requested PDU session release procedure

6.4.3.3 UE-requested PDU session release procedure accepted by the network

Upon receipt of a PDU SESSION RELEASE REQUEST message and a PDU session ID, if the SMF accepts the request to release the PDU session, and shall perform the network-requested PDU session release procedure as specified in subclause 6.3.3.

6.4.3.4 UE-requested PDU session release procedure not accepted by the network

Upon receipt of a PDU SESSION RELEASE REQUEST message, if the SMF does not accept the request to release the PDU session, the SMF shall create a PDU SESSION RELEASE REJECT message.

The SMF shall set the 5GSM cause IE of the PDU SESSION RELEASE REJECT message to indicate the reason for rejecting the PDU session release.

The 5GSM cause IE typically indicates one of the following SM cause values:

- #35 PTI already in use; or
- #43 Invalid PDU session identity; or
- #95 111: protocol errors.

The SMF shall send the PDU SESSION RELEASE REJECT message.

Upon receipt of a PDU SESSION RELEASE REJECT message and a PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, the UE shall stop timer T3582, release the allocated PTI value and locally release the PDU session. If there is one or more multicast MBS sessions associated with the PDU session, the UE shall locally leave the associated multicast MBS sessions.

6.4.3.5 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Expiry of timer T3582.

The UE shall, on the first expiry of the timer T3582, retransmit the PDU SESSION RELEASE REQUEST message and the PDU session information which was transported together with the initial transmission of the PDU SESSION RELEASE REQUEST message and shall reset and start timer T3582. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3582, the UE shall abort the procedure, release the allocated PTI, perform a local release of the PDU session, and perform the registration procedure for mobility and periodic registration update by sending a REGISTRATION REQUEST message including the PDU session status IE over each access that user plane resources have been established if the PDU session is an MA PDU session, or over the access the PDU session is associated with if the PDU session is a single access PDU session. If there is one or more multicast MBS sessions associated with the PDU session, the UE shall locally leave the associated multicast MBS sessions.

b) Collision of UE-requested PDU session release procedure and network-requested PDU session modification procedure.

When the UE receives a PDU SESSION MODIFICATION COMMAND message during the UE-requested PDU session release procedure, and the PDU session indicated in PDU SESSION MODIFICATION COMMAND message is the PDU session that the UE had requested to release, the UE shall ignore the PDU SESSION MODIFICATION COMMAND modification models and proceed with the PDU session release procedure.

c) Collision of UE-requested PDU session release procedure and network-requested PDU session release procedure.

When the UE receives a PDU SESSION RELEASE COMMAND message with the PTI IE set to "No procedure transaction identity assigned" during the UE-requested PDU session release procedure, the PDU session indicated in the PDU SESSION RELEASE COMMAND message is the same as the PDU session that the UE requests to release:

- if the Access type IE is included in the PDU SESSION RELEASE COMMAND message and the PDU session is an MA PDU session and having user-plane resources established on the access different from the access indicated in the Access type IE in the PDU SESSION RELEASE COMMAND message, the UE shall proceed both the UE-requested PDU session release procedure and network-requested PDU session release procedure; or
- otherwise, the UE shall abort the UE-requested PDU session release procedure and shall stop the timer T3582 and proceed with the network-requested PDU session release procedure.
- d) Receipt of an indication that the 5GSM message was not forwarded due to routing failure

Upon receiving an indication that the 5GSM message was not forwarded due to routing failure along with a PDU SESSION RELEASE REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3582, abort the procedure, release the allocated PTI, perform a local release of the PDU session, and perform the registration procedure for mobility and periodic registration update by sending a REGISTRATION REQUEST message including the PDU session status IE over each access that user plane resources have been established if the PDU session is an MA PDU session, or over the access the PDU session is associated with if the PDU session is a single access PDU session. If there is one or more multicast MBS sessions associated with the released PDU session, the UE shall locally leave the associated multicast MBS sessions.

e) PDU session release signalling restricted due to service area restriction

The UE may delay the release of the PDU session until the UE is not restricted by service area restrictions, or it may release the allocated PTI, perform a local release of the PDU session, and include the PDU session status IE over each access that user plane resources have been established if the PDU session is an MA PDU session, or

over the access the PDU session is associated with if the PDU session is a single access PDU when performing the next registration procedure. If the UE performs the local release of the PDU session and there is one or more MBS multicast sessions associated with the released PDU session, the UE shall locally leave the associated MBS multicast sessions.

f) Collision of UE-requested PDU session release procedure and N1 NAS signalling connection release

The UE may immediately retransmit the PDU SESSION RELEASE REQUEST message and stop, reset and restart timer T3582, if the following conditions apply:

- 1) The original UE-requested PDU session release procedure was initiated over an existing N1 NAS signalling connection; and
- 2) the previous transmission of the PDU SESSION RELEASE REQUEST message was not initiated due to timer T3582 expiry.
- g) Receipt of an indication that the 5GSM message was not forwarded due to the PLMN is not allowed to operate at the present UE location

Upon receiving an indication that the 5GSM message was not forwarded because the UE accessing via a satellite NG-RAN cell is informed that the PLMN is not allowed to operate at the present UE location along with a PDU SESSION RELEASE REQUEST message with the PDU session ID IE set to the same value as the PDU session ID that was sent by the UE, the UE shall stop timer T3582, abort the procedure and locally release the PDU session.

6.4.3.6 Abnormal cases on the network side

The following abnormal cases can be identified:

a) PDU session inactive for the received PDU session identity.

If the PDU session ID in the PDU SESSION RELEASE REQUEST message belongs to any PDU session in state PDU SESSION INACTIVE in the SMF, the SMF shall send the PDU SESSION RELEASE REJECT message to the UE with the 5GSM cause #43 "Invalid PDU session identity".

6.5 5GSM status procedure

6.5.1 General

The purpose of the sending of the 5GSM STATUS message is to report at any time certain error conditions detected upon receipt of 5GSM protocol data. The 5GSM STATUS message can be sent by both the network and the UE (see example in figure 6.5.1.1).

UE		SMF
	5GSM STATUS	_
	OR	
	5GSM STATUS	•
		F

Figure 6.5.1.1: 5GSM status procedure

6.5.2 5GSM status received in the UE

If the 5GSM entity of the UE receives a 5GSM STATUS message the UE shall take different actions depending on the received 5GSM cause value:

#47 PTI mismatch.

The UE shall abort any ongoing 5GSM procedure related to the received PTI value and stop any related timer.

#81 invalid PTI value.

The UE shall abort any ongoing 5GSM procedure related to the received PTI value and stop any related timer.

#97 Message type non-existent or not implemented.

The UE shall abort any ongoing 5GSM procedure related to the PTI or PDU session ID and stop any related timer.

On receipt of a 5GSM STATUS message with any other 5GSM cause value no state transition and no specific action shall be taken as seen from the radio interface, i.e. local actions are possible.

6.5.3 5GSM status received in the SMF

If the SMF receives a 5GSM STATUS message the SMF shall take different actions depending on the received 5GSM cause value:

#43 invalid PDU session identity.

The SMF shall abort any ongoing 5GSM procedure related to the PTI or PDU session ID, stop any related timer and locally release the PDU session indicated in the 5GSM STATUS message.

#47 PTI mismatch.

The SMF shall abort any ongoing 5GSM procedure related to the received PTI value and stop any related timer.

If the PTI indicated in the 5GSM STATUS message is related to a PDU SESSION ESTABLISHMENT ACCEPT message, the SMF shall perform a local release of the PDU session indicated in the PDU SESSION ESTABLISHMENT ACCEPT message.

#81 invalid PTI value.

The SMF shall abort any ongoing 5GSM procedure related to the received PTI value and stop any related timer.

#96 invalid mandatory information.

The SMF shall abort any ongoing 5GSM procedure related to the PTI or PDU session ID and stop any related timer.

#97 message type non-existent or not implemented.

The SMF shall abort any ongoing 5GSM procedure related to the PTI or PDU session ID and stop any related timer.

The local actions to be taken by the SMF on receipt of a 5GSM STATUS message with any other 5GSM cause value are implementation dependent.

6.6 Miscellaneous procedures

6.6.1 Exchange of extended protocol configuration options

The UE and the SMF can exchange protocol configuration options via 5GSM procedures.

The protocol configuration options shall be exchanged via the Extended protocol configuration options IE.
6.6.2 Remote UE report procedure

6.6.2.1 General

The purpose of the 5G ProSe remote UE report procedure is for a UE acting as a 5G ProSe layer-3 UE-to-network UE relay to notify the network that a 5G ProSe remote UE is connected to the 5G ProSe layer-3 UE-to-network relay UE or disconnected from the 5G ProSe layer-3 UE-to-network relay UE as specified in 3GPP TS 23.304 [6E].

The UE does not initiate the remote UE report procedure if the timer T3396 is running.

The UE does not initiate the remote UE report procedure if the timer T3584 is running.

The UE does not initiate the remote UE report procedure if the timer T3585 is running.

6.6.2.2 Remote UE report procedure initiation

In order to initiate the 5G ProSe remote UE report procedure, the UE shall create a REMOTE UE REPORT message.

The UE shall include information of newly connected or disconnected 5G ProSe remote UEs to the network in the REMOTE UE REPORT message by setting the values of the Remote UE context connected IE or the Remote UE context disconnected IE to the 5G ProSe remote UE identities that are being connected or disconnected, respectively.

The UE shall set the Remote UE ID with:

- a) the UP-PRUK ID of the 5G ProSe remote UE, if the security for 5G ProSe communication via 5G ProSe UE-tonetwork relay is performed over user plane as specified in 3GPP TS 33.503 [56]; or
- b) the CP-PRUK ID of the 5G ProSe remote UE, if the security for 5G ProSe communication via 5G ProSe UE-tonetwork relay is performed over control plane as specified in 3GPP TS 33.503 [56].

If the UE sets the Remote UE ID with the PRUK ID of the 5G ProSe remote UE and the UP-PRUK ID is in 64-bit string format, the UE shall include the HPLMN ID of the 5G ProSe remote UE.

If the UE allocated an IPv4 address to a 5G ProSe remote UE and enabled UDP usage to the 5G ProSe remote UE, the UE shall include in the REMOTE UE REPORT message the UDP port range assigned to the 5G ProSe remote UE in the NAT function of 5G ProSe layer-3 UE-to-network relay. If the UE allocated an IPv4 address to a 5G ProSe remote UE and enabled TCP usage to the 5G ProSe remote UE, the UE shall include in the REMOTE UE REPORT message the TCP port range assigned to the 5G ProSe remote UE in the NAT function of 5G ProSe address to the 5G ProSe remote UE in the NAT function of 5G ProSe address to the 5G ProSe remote UE in the NAT function of 5G ProSe layer-3 UE-to-network relay.

The UE shall set the PDU session ID IE to the value of the PDU session associated with the 5G ProSe remote UE connected to the 5G ProSe layer-3 UE-to-network relay UE or disconnected from the 5G ProSe layer-3 UE-to-network relay UE.

The UE shall allocate a PTI value currently not used and shall set the PTI IE of the REMOTE UE REPORT message to the allocated PTI value.

The UE shall transport the REMOTE UE REPORT message and the PDU session ID, using the NAS transport procedure as specified in subclause 5.4.5, and the UE shall start timer T3586 (see example in figure 6.6.2.2.1).

UE Network
Start T3586 REMOTE UE REPORT
Stop T3586 REMOTE UE REPORT RESPONSE

Figure 6.6.2.2.1: Remote UE report procedure

6.6.2.3 Remote UE report procedure accepted by the network

Upon receipt of the REMOTE UE REPORT message, the SMF shall send a REMOTE UE REPORT RESPONSE message to the UE. The SMF shall include the PTI from the REMOTE UE REPORT message.

The SMF shall set the EAP message IE to an EAP-success message or an EAP-failure message to be sent to the 5G ProSe layer-3 remote UE if the EAP-success message or the EAP-failure message is received from the DN.

Upon receipt of the REMOTE UE REPORT RESPONSE message, the UE shall stop timer T3586 and enter the state PROCEDURE TRANSACTION INACTIVE.

6.6.2.4 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Expiry of timer T3586:

On the first expiry of the timer T3586, the UE shall resend the REMOTE UE REPORT message and shall reset and restart timer T3586. This retransmission is repeated two times, i.e. on the third expiry of timer T3586, the UE shall abort the procedure and release any resources for this procedure.

- NOTE: After the abortion of the remote UE report procedure, the remote UE report procedure for the remote UE(s) can be restarted and how to restart the procedure is left to UE implementation.
- b) Collision of network-requested PDU session release procedure and remote UE report procedure:

If the UE receives a PDU SESSION RELEASE COMMAND message during the remote UE report procedure, and the PDU session indicated in the PDU SESSION RELEASE COMMAND message is the PDU session ID that the UE had indicated in the REMOTE UE REPORT message, the UE shall abort the remote UE report procedure and proceed with the network-requested PDU session release procedure.

6.6.2.5 Abnormal cases on the network side

The following abnormal cases have been identified:

a) Collision of network-requested PDU session release procedure and remote UE report procedure:

If the SMF receives a REMOTE UE REPORT message during a network-requested PDU session release procedure, and the PDU session indicated in the PDU SESSION RELEASE COMMAND message is the PDU session ID that is in the REMOTE UE REPORT message, the SMF shall abort the remote UE report procedure and proceed with the network-requested PDU session release procedure.

b) PDU session inactive for the received PDU session identity:

If the PDU session ID in the REMOTE UE REPORT message belongs to any PDU session in state PDU SESSION INACTIVE in the SMF, the SMF shall send the 5GSM STATUS message to the UE with the 5GSM cause #43 "Invalid PDU session identity".

7 Handling of unknown, unforeseen, and erroneous protocol data

7.1 General

The procedures specified in the present document apply to those messages which pass the checks described in this subclause.

This subclause also specifies procedures for the handling of unknown, unforeseen, and erroneous protocol data by the receiving entity. These procedures are called "error handling procedures", but in addition to providing recovery mechanisms for error situations they define a compatibility mechanism for future extensions of the protocols.

Subclauses 7.1 to 7.8 shall be applied in order of precedence.

Detailed error handling procedures in the network are implementation dependent and may vary from PLMN to PLMN. However, when extensions of this protocol are developed, networks are assumed to have the error handling which is indicated in this subclause as mandatory ("shall") and that is indicated as strongly recommended ("should").

Also, the error handling of the network is only considered as mandatory or strongly recommended when certain thresholds for errors are not reached during a dedicated connection.

For definition of semantical and syntactical errors see 3GPP TS 24.007 [11], subclause 11.4.2.

7.2 Message too short or too long

7.2.1 Message too short

When a message is received that is too short to contain a complete message type information element, that message shall be ignored, cf. 3GPP TS 24.007 [11].

7.2.2 Message too long

The maximum size of a NAS message for NR connected to 5GCN is specified in 3GPP TS 38.323 [29].

The maximum size of a NAS message for E-UTRA connected to 5GCN is specified 3GPP TS 36.323 [25].

The maximum size of a NAS message for non-3GPP access connected to 5GCN is specified in 3GPP TS 24.502 [18]

7.3 Unknown or unforeseen procedure transaction identity or PDU Session identity

7.3.1 Procedure transaction identity

The following network procedures shall apply for handling an unknown, erroneous, or unforeseen PTI received in a 5GSM message:

a) In case the network receives a PDU SESSION MODIFICATION COMPLETE, a PDU SESSION RELEASE COMPLETE message or a PDU SESSION MODIFICATION COMMAND REJECT message in which the PTI value is an assigned or unassigned value that does not match any PTI in use, the network shall respond with a 5GSM STATUS message including 5GSM cause #47 "PTI mismatch".

- b) In case the network receives a PDU SESSION AUTHENTICATION COMPLETE message in which the PTI value is an assigned value, the network shall respond with a 5GSM STATUS message including 5GSM cause #81 "invalid PTI value".
- c) In case the network receives a PDU SESSION ESTABLISHMENT REQUEST message, a PDU SESSION MODIFICATION REQUEST message or a PDU SESSION RELEASE REQUEST message in which the PTI value is an unassigned value, the network shall respond with a 5GSM STATUS message including 5GSM cause #81 "invalid PTI value".
- d) In case the network receives a 5GSM message in which the PTI value is a reserved value, the network shall ignore the message.

The following UE procedures shall apply for handling an unknown, erroneous, or unforeseen PTI received in a 5GSM message:

- a) In case the UE receives a PDU SESSION MODIFICATION COMMAND message or a PDU SESSION MODIFICATION REJECT message in which the PTI value is an assigned value that does not match any PTI in use:
 - 1) if the UE detects that this PDU SESSION MODIFICATION COMMAND message is a network retransmission of an already accepted request (see subclause 6.3.2.3), the UE shall respond with a PDU SESSION MODIFICATION COMPLETE message;
 - 2) if the UE detects that this PDU SESSION MODIFICATION COMMAND message is a network retransmission of an already rejected request (see subclause 6.3.2.4), the UE shall respond with a PDU SESSION MODIFICATION COMAND REJECT message; or
 - 3) otherwise, the UE shall respond with a 5GSM STATUS message including 5GSM cause #47 "PTI mismatch".
- b) In case the UE receives a PDU SESSION RELEASE COMMAND message or a PDU SESSION RELEASE REJECT message in which the PTI value is an assigned value that does not match any PTI in use:
 - 1) if the UE detects that this PDU SESSION RELEASE COMMAND message is a network retransmission of an already accepted request (see subclause 6.3.3.3), the UE shall respond with a PDU SESSION RELEASE COMPLETE message; or
 - 2) otherwise, the UE shall respond with a 5GSM STATUS message including 5GSM cause #47 "PTI mismatch".
- c) In case the UE receives a PDU SESSION ESTABLISHMENT ACCEPT message or a PDU SESSION ESTABLISHMENT REJECT message in which the PTI value is an assigned value that does not match any PTI in use:
 - 1) the UE shall respond with a 5GSM STATUS message including 5GSM cause #47 "PTI mismatch".
- d) In case the UE receives a PDU SESSION AUTHENTICATION COMMAND message or a PDU SESSION AUTHENTICATION RESULT message in which the PTI value is an assigned value, the UE shall respond with a 5GSM STATUS message including 5GSM cause #81 "invalid PTI value".
- e) In case the UE receives a PDU SESSION ESTABLISHMENT ACCEPT message, a PDU SESSION ESTABLISHMENT REJECT message, a PDU SESSION MODIFICATION REJECT message or a PDU SESSION RELEASE REJECT message in which the PTI value is an unassigned value, the UE shall ignore the message.
- f) In case the UE receives a 5GSM message in which the PTI value is a reserved value, the UE shall ignore the message.

7.3.2 PDU Session identity

The following network procedures shall apply for handling an unknown, erroneous, or unforeseen PDU session identity received in the header of a 5GSM message (specified as the header of a standard L3 message, see 3GPP TS 24.007 [11]):

- a) If the network receives a PDU SESSION MODIFICATION REQUEST message which includes an unassigned or reserved PDU session identity value, the network shall respond with a PDU SESSION MODIFICATION REJECT message including 5GSM cause #43 "invalid PDU session identity".
- b) If the network receives PDU SESSION RELEASE REQUEST message which includes an unassigned or reserved PDU session identity value, the network shall respond with a PDU SESSION RELEASE REJECT message including 5GSM cause #43 "invalid PDU session identity".
- c) Upon receipt of an UL NAS TRANSPORT message, the network takes the following actions:
 - If the Request type IE is set to "initial request" or "initial emergency request" and the message includes a reserved PDU session identity value, the network shall respond with a DL NAS TRANSPORT message with 5GMM cause #90 "payload was not forwarded";
 - 2) otherwise, if the message includes an unassigned or reserved PDU session identity value, the network shall respond with a DL NAS TRANSPORT message with 5GMM cause #90 "payload was not forwarded".
- d) If the network receives a 5GSM message other than those listed in items a) through c) above in which the message includes a reserved PDU session identity value or an assigned value that does not match an existing PDU session, the network shall ignore the message.

The following UE procedures shall apply for handling an unknown, erroneous, or unforeseen PDU session identity received in the header of a 5GSM message:

- a) If the UE receives a 5GSM message which includes an unassigned or reserved PDU session identity value, the UE shall ignore the message.
- b) If the UE receives a 5GSM message which includes a PDU session identity belonging to any PDU session in state PDU SESSION INACTIVE in the UE, the UE shall respond with a 5GSM STATUS message including 5GSM cause #43 "invalid PDU session identity".

7.4 Unknown or unforeseen message type

If UE receives a 5GMM message or 5GSM message with message type not defined for the extended protocol discriminator (EPD) or not implemented by the receiver, it shall return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #97 "message type non-existent or not implemented".

If the network receives a 5GMM or 5GSM message with message type not defined for the EPD or not implemented by the receiver in a protocol state where reception of an unsolicited message with the given EPD from the UE is not foreseen in the protocol, the network actions are implementation dependent. Otherwise, if the network receives a message with message type not defined for the EPD or not implemented by the receiver, it shall ignore the message except that it should return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #97 "message type non-existent or not implemented".

NOTE: A message type not defined for the EPD in the given direction is regarded by the receiver as a message type not defined for the EPD, see 3GPP TS 24.007 [11].

If the UE receives a message not compatible with the protocol state, the UE shall return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #98 "message type not compatible with protocol state".

If the network receives a message not compatible with the protocol state, the network actions are implementation dependent.

7.5 Non-semantical mandatory information element errors

7.5.1 Common procedures

When on receipt of a message,

a) an "imperative message part" error; or

b) a "missing mandatory IE" error

is diagnosed or when a message containing:

- a) a syntactically incorrect mandatory IE;
- b) an IE unknown in the message, but encoded as "comprehension required" (see 3GPP TS 24.007 [11]); or
- c) an out of sequence IE encoded as "comprehension required" (see 3GPP TS 24.007 [11]) is received,

the UE shall proceed as follows:

If the message is not one of the messages listed in the UE procedures in subclause 7.5.3, item a), b) or c), the UE shall return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #96 "invalid mandatory information";

the network shall proceed as follows:

If the message is not one of the messages listed in the network procedures in subclause 7.5.3, item a), b) or c), the network shall either:

- 1) try to treat the message (the exact further actions are implementation dependent); or
- 2) ignore the message except that it should return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #96 "invalid mandatory information".

7.5.2 5GS mobility management

No exceptional cases are described for 5GS mobility management messages.

No semantical or syntactical diagnosis other than presence and length shall be performed on the EPS NAS message container information element in the REGISTRATION REQUEST message.

7.5.3 5GS session management

The following UE procedures shall apply for handling an error encountered with a mandatory information element in a 5GSM message:

- a) If the message is a PDU SESSION ESTABLISHMENT ACCEPT, the UE shall initiate PDU session release procedure by sending a PDU SESSION RELEASE REQUEST message with 5GSM cause #96 "invalid mandatory information".
- b) Void.
- c) If the message is a PDU SESSION RELEASE COMMAND, a PDU SESSION RELEASE COMPLETE message with 5GSM cause #96 "invalid mandatory information" shall be returned.

The following network procedures shall apply for handling an error encountered with a mandatory information element in a 5GSM message:

- a) If the message is a PDU SESSION ESTABLISHMENT REQUEST, a PDU SESSION ESTABLISHMENT REJECT message with 5GSM cause #96 "invalid mandatory information" shall be returned.
- b) If the message is a PDU SESSION MODIFICATION REQUEST, a PDU SESSION MODIFICATION REJECT message with 5GSM cause #96 "invalid mandatory information" shall be returned.
- c) If the message is a PDU SESSION RELEASE REQUEST, a PDU SESSION RELEASE REJECT message with 5GSM cause #96 "invalid mandatory information" shall be returned.

7.6 Unknown and unforeseen IEs in the non-imperative message part

7.6.1 IEIs unknown in the message

The UE shall ignore all IEs unknown in a message which are not encoded as "comprehension required" (see 3GPP TS 24.007 [11]).

The network shall take the same approach.

7.6.2 Out of sequence IEs

The UE shall ignore all out of sequence IEs in a message which are not encoded as "comprehension required" (see 3GPP TS 24.007 [11]).

The network should take the same approach.

7.6.3 Repeated IEs

If an information element with format T, TV, TLV, or TLV-E is repeated in a message in which repetition of the information element is not specified in clause 8 and clause 9 of the present document, the UE shall handle only the contents of the information element appearing first and shall ignore all subsequent repetitions of the information elements. When repetition of information elements is specified, the UE shall handle only the contents of specified repeated information elements. If the limit on repetition of information elements is exceeded, the UE shall handle the contents of information elements appearing first up to the limit of repetitions and shall ignore all subsequent repetitions of the information element.

The network should follow the same procedures.

7.7 Non-imperative message part errors

This category includes:

- a) syntactically incorrect optional IEs; and
- b) conditional IE errors.

7.7.1 Syntactically incorrect optional IEs

The UE shall treat all optional IEs that are syntactically incorrect in a message as not present in the message.

The network shall take the same approach.

7.7.2 Conditional IE errors

When upon receipt of a 5GMM or 5GSM message the UE diagnoses a "missing conditional IE" error or an "unexpected conditional IE" error, or when it receives a 5GMM or 5GSM message containing at least one syntactically incorrect conditional IE, the UE shall ignore the message and shall return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #100 "conditional IE error".

When the network receives a message and diagnoses a "missing conditional IE" error or an "unexpected conditional IE" error or when it receives a message containing at least one syntactically incorrect conditional IE, the network shall either:

- a) try to treat the message (the exact further actions are implementation dependent); or
- b) ignore the message except that it should return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #100 "conditional IE error".

7.8 Messages with semantically incorrect contents

When a message with semantically incorrect contents is received, the UE shall perform the foreseen reactions of the procedural part of the present document (i.e. of clauses 5, 6). If, however no such reactions are specified, the UE shall ignore the message except that it shall return a status message (5GMM STATUS or 5GSM STATUS depending on the EPD) with cause #95 "semantically incorrect message".

The network should follow the same procedure except that a status message is not normally transmitted.

8 Message functional definitions and contents

- 8.1 Overview
- 8.2 5GS mobility management messages
- 8.2.1 Authentication request

8.2.1.1 Message definition

The AUTHENTICATION REQUEST message is sent by the AMF to the UE to initiate authentication of the UE identity. See table 8.2.1.1.1.

Message type: AUTHENTICATION REQUEST

Significance: dual

Direction: network to UE

Table 8.2.1.1.1: AUTHENTICATION REQUEST message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Authentication request message identity	Message type 9.7	М	V	1
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	ABBA	ABBA 9.11.3.10	М	LV	3-n
21	Authentication parameter RAND (5G authentication challenge)	Authentication parameter RAND 9.11.3.16	0	TV	17
20	Authentication parameter AUTN (5G authentication challenge)	Authentication parameter AUTN 9.11.3.15	0	TLV	18
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503

8.2.1.2 Authentication parameter RAND

Authentication parameter RAND IE is included if the AUTHENTICATION REQUEST message is used in a 5G AKA authentication procedure.

8.2.1.3 Authentication parameter AUTN

Authentication parameter AUTN IE is included if the AUTHENTICATION REQUEST message is used in a 5G AKA authentication procedure.

8.2.1.4 Void

8.2.1.5 EAP message

EAP message IE is included if the AUTHENTICATION REQUEST message is used in an EAP based primary authentication and key agreement procedure.

8.2.2 Authentication response

8.2.2.1 Message definition

The AUTHENTICATION RESPONSE message is sent by the UE to the AMF to deliver a calculated authentication response to the network. See table 8.2.2.1.1.

Message type: AUTHENTICATION RESPONSE

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator	М	V	1
		9.2			
	Security header type	Security header type	М	V	1/2
		9.3			
	Spare half octet	Spare half octet	М	V	1/2
		9.5			
	Authentication response message	Message type	М	V	1
	identity	9.7			
2D	Authentication response	Authentication response parameter	0	TLV	18
	parameter	9.11.3.17			
78	EAP message	EAP message	0	TLV-E	7-1503
	_	9.11.2.2			

Table 8.2.2.1.1: AUTHENTICATION RESPONSE message content

8.2.2.2 Authentication response parameter

This IE is included if the message is sent in a 5G AKA based primary authentication and key agreement procedure.

8.2.2.3 EAP message

EAP message IE is included if the EAP message received in a related AUTHENTICATION REQUEST message was an EAP-request.

8.2.3 Authentication result

8.2.3.1 Message definition

The AUTHENTICATION RESULT message is sent by the AMF to the UE to provide result of EAP authentication of the UE identity. See table 8.2.3.1.1.

Message type: AUTHENTICATION RESULT

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator	М	V	1
		9.2			
	Security header type	Security header type	М	V	1/2
		9.3			
	Spare half octet	Spare half octet	М	V	1/2
		9.5			
	Authentication result message	Message type	М	V	1
	identity	9.7			
	ngKSI	NAS key set identifier	М	V	1/2
		9.11.3.32			
	Spare half octet	Spare half octet	М	V	1/2
		9.5			
	EAP message	EAP message	М	LV-E	6-1502
	_	9.11.2.2			
38	ABBA	ABBA	0	TLV	4-n
		9.11.3.10			

8.2.3.2 ABBA

This IE shall be included if the message contains an EAP message IE with an EAP-success message.

8.2.4 Authentication failure

8.2.4.1 Message definition

The AUTHENTICATION FAILURE message is sent by the UE to the AMF to indicate that authentication of the network has failed. See table 8.2.4.1.1.

Message type: AUTHENTICATION FAILURE

Significance: dual

Direction: UE to network

Table 8.2.4.1.1: AUTHENTICATION FAILURE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	Μ	V	1/2
	Authentication failure message identity	Message type 9.7	Μ	V	1
	5GMM cause	5GMM cause 9.11.3.2	Μ	V	1
30	Authentication failure parameter	Authentication failure parameter 9.11.3.14	0	TLV	16

8.2.4.2 Authentication failure parameter

This IE shall be included in a 5G AKA based primary authentication and key agreement procedure if and only if the 5GMM cause was #21 "synch failure". It shall include the response to the authentication challenge from the USIM, which is made up of the AUTS parameter (see 3GPP TS 33.501 [24]).

8.2.5 Authentication reject

8.2.5.1 Message definition

The AUTHENTICATION REJECT message is sent by the AMF to the UE to indicate that the authentication procedure has failed and that the UE shall abort all activities. See table 8.2.5.1.1.

Message type: AUTHENTICATION REJECT

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Authentication reject message identity	Message type 9.7	М	V	1
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503

Table 8.2.5.1.1: AUTHENTICATION REJECT message content

8.2.5.2 EAP message

EAP message IE is included if the AUTHENTICATION REJECT message is used to convey EAP-failure message.

8.2.6 Registration request

8.2.6.1 Message definition

The REGISTRATION REQUEST message is sent by the UE to the AMF. See table 8.2.6.1.1.

Message type: REGISTRATION REQUEST

Significance: dual

Direction: UE to network

Table 8.2.6.1.1: REGISTRATION REQUEST message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended Protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Registration request message	Message type 9.7	М	V	1
	5GS registration type	5GS registration type 9.11.3.7	М	V	1/2
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
	5GS mobile identity	5GS mobile identity 9.11.3.4	М	LV-E	6-n
C-	Non-current native NAS key set identifier	NAS key set identifier 9.11.3.32	0	TV	1
10	5GMM capability	5GMM capability 9.11.3.1	0	TLV	3-15
2E	UE security capability	UE security capability 9.11.3.54	0	TLV	4-10
2F	Requested NSSAI	NSSAI 9.11.3.37	0	TLV	4-74
52	Last visited registered TAI	5GS tracking area identity 9.11.3.8	0	TV	7
17	S1 UE network capability	S1 UE network capability 9.11.3.48	0	TLV	4-15
40	Uplink data status	Uplink data status 9.11.3.57	0	TLV	4-34
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
B-	MICO indication	MICO indication 9.11.3.31	0	TV	1
2B	UE status	UE status 9.11.3.56	0	TLV	3
77	Additional GUTI	5GS mobile identity 9.11.3.4	0	TLV-E	14
25	Allowed PDU session status	Allowed PDU session status 9.11.3.13	0	TLV	4-34
18	UE's usage setting	UE's usage setting 9.11.3.55	0	TLV	3
51	Requested DRX parameters	5GS DRX parameters 9.11.3.2A	0	TLV	3
70	EPS NAS message container	EPS NAS message container 9.11.3.24	0	TLV-E	4-n
74	LADN indication	LADN indication 9.11.3.29	0	TLV-E	3-811
8-	Payload container type	Payload container type 9.11.3.40	0	TV	1
7B	Payload container	Payload container 9.11.3.39	0	TLV-E	4-65538
9-	Network slicing indication	Network slicing indication 9.11.3.36	0	TV	1
53	5GS update type	5GS update type 9.11.3.9A	0	TLV	3
41	Mobile station classmark 2	Mobile station classmark 2 9.11.3.31C	0	TLV	5
42	Supported codecs	Supported codec list 9.11.3.51A	0	TLV	5-n
71	NAS message container	NAS message container 9.11.3.33	0	TLV-E	4-n
60	EPS bearer context status	EPS bearer context status 9.11.3.23A	0	TLV	4
6E	Requested extended DRX parameters	Extended DRX parameters 9.11.3.26A	0	TLV	3-4

6A	T3324 value	GPRS timer 3 9.11.2.5	0	TLV	3
67	UE radio capability ID	UE radio capability ID 9.11.3.68	0	TLV	3-n
35	Requested mapped NSSAI	Mapped NSSAI 9.11.3.31B	0	TLV	3-42
48	Additional information requested	Additional information requested 9.11.3.12A	0	TLV	3
1A	Requested WUS assistance information	WUS assistance information 9.11.3.71	0	TLV	3-n
A-	N5GC indication	N5GC indication 9.11.3.72	0	TV	1
30	Requested NB-N1 mode DRX parameters	NB-N1 mode DRX parameters 9.11.3.73	0	TLV	3
29	UE request type	UE request type 9.11.3.76	0	TLV	3
28	Paging restriction	Paging restriction 9.11.3.77	0	TLV	3-35
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538
32	NID	NID 9.11.3.79	0	TLV	8
16	MS determined PLMN with disaster condition	PLMN identity 9.11.3.85	0	TLV	5
2A	Requested PEIPS assistance information	PEIPS assistance information 9.11.3.80	0	TLV	3-n
3B	Requested T3512 value	GPRS timer 3 9.11.2.5	0	TLV	3

8.2.6.2 Non-current native NAS key set identifier

The UE shall include this IE if the UE has a valid non-current native 5G NAS security context when the UE performs an inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode and the UE uses a mapped 5G NAS security context to protect the REGISTRATION REQUEST message.

8.2.6.3 5GMM capability

The UE shall include this IE, unless the UE performs a periodic registration updating procedure.

8.2.6.4 UE security capability

The UE shall include this IE, unless the UE performs a periodic registration updating procedure.

8.2.6.5 Requested NSSAI

This IE shall be included by the UE when performing the registration procedure if the 5GS registration type IE indicates:

- a) "initial registration", according to the conditions specified in subclause 5.5.1.2.2; or
- b) "mobility registration updating", according to the conditions specified in subclause 5.5.1.3.2.

8.2.6.6 Last visited registered TAI

This IE shall be included if the UE holds a valid last visited registered TAI.

8.2.6.7 S1 UE network capability

A UE supporting S1 mode shall include this IE, unless the UE performs a periodic registration updating procedure.

8.2.6.8 Uplink data status

This IE shall be included if the UE has uplink user data pending to be sent, unless the UE performs a periodic registration updating procedure.

8.2.6.9 PDU session status

This IE shall be included when the UE needs to indicate the PDU sessions that are associated with the access type that the message is sent over, that are active within the UE.

8.2.6.10 MICO indication

The UE may include this IE to request the use of MICO mode.

8.2.6.11 UE status

This IE shall be included if the UE in single-registration mode performs the registration procedure due to inter-system change from S1 mode to N1 mode or if the UE in dual-registration mode and EMM state EMM-REGISTERED performs initial registration.

8.2.6.12 Additional GUTI

This IE shall be included:

- a) if the UE performs the registration procedure due to inter-system change from S1 mode to N1 mode, the UE operates in single-registration mode and the UE has a valid 5G-GUTI; or
- b) the UE holds two valid native 5G-GUTIs and one of the valid native 5G-GUTI was assigned by the PLMN with which the UE is performing the registration.

8.2.6.13 Allowed PDU session status

This IE shall be included if the REGISTRATION REQUEST message is sent as a response to paging with the access type indicating non-3GPP access and the UE wants to indicate the user-plane resources of PDU session(s) associated with non-3GPP access allowed to be re-established over 3GPP access.

8.2.6.14 UE's usage setting

This IE shall be included if the UE supports IMS voice.

8.2.6.15 Requested DRX parameters

If the UE wants to use or change the UE specific DRX parameters, the UE shall include the Requested DRX parameters IE in the REGISTRATION REQUEST message.

8.2.6.16 EPS NAS message container

The UE operating in the single-registration mode shall include this information element as specified in subclause 5.5.1.3.2 if the UE performs mobility from S1 mode to N1 mode in 5GMM-IDLE mode. The content of this message container is the complete integrity protected TRACKING AREA UPATE REQUEST message, using EPS security context.

The UE performing initial registration shall include this information element if

- a) the UE:
 - 1) was previously registered in S1 mode before entering state EMM-DEREGISTERED; and
 - 2) has received an "interworking without N26 interface not supported" indication from the network; and
- b) EPS security context and a valid 4G-GUTI are available.

The content of this message container is the complete integrity protected ATTACH REQUEST message, using EPS security context.

8.2.6.17 LADN indication

The UE shall include this information element when the UE needs to request LADN information for specific LADN DNN(s) or to indicate a request for LADN information.

8.2.6.17A Payload container type

This IE shall be included if the UE includes the Payload container IE.

NOTE: In this version of the protocol, the Payload container type IE in the REGISTRATION REQUEST message is set to "UE policy container" as described in subclauses 5.5.1.2.2 and 5.5.1.3.2.

8.2.6.18 Payload container

Within a PLMN, this IE shall be included if the UE has one or more stored UE policy sections identified by a UPSI with the PLMN ID part indicating the HPLMN or the selected PLMN for the registration procedure for mobility and periodic registration update due to inter-system change from S1 mode to N1 mode of a UE operating in the single-registration mode or for the registration procedure for initial registration.

Within an SNPN, this IE shall be included if the UE has one or more stored UE policy sections for the selected SNPN for the registration procedure for initial registration.

8.2.6.19 Network slicing indication

This IE shall be included when a requested NSSAI is included in the REGISTRATION REQUEST message and the requested NSSAI is created from the default configured NSSAI.

8.2.6.20 5GS update type

This IE shall be included when the UE is performing the registration procedure to indicate any of the following:

- a) the UE requests the use of SMS over NAS or there is a change in the UE's requirements to use SMS over NAS;
- b) a change in the UE's radio capability for NG-RAN; or
- c) the UE requests CIoT 5GS optimizations.

8.2.6.21 NAS message container

This IE shall be included if the UE is sending a REGISTRATION REQUEST message as an initial NAS message, the UE has a valid 5G NAS security context and the UE needs to send non-cleartext IEs.

8.2.6.22 Requested extended DRX parameters

The UE shall include this IE if the UE needs to use extended DRX or change the extended DRX parameters.

8.2.6.23 EPS bearer context status

The UE shall include this IE if the UE operating in the single-registration mode performs inter-system change from S1 mode to N1 mode and the UE has locally deactivated EPS bearer context(s) for which interworking to 5GS is supported while the UE was in S1 mode without notifying the network.

8.2.6.24 T3324 value

The UE may include this IE during the registration update procedure if it requests to use MICO mode and use the active time timer.

8.2.6.25 Mobile station classmark 2

This IE shall be included if the UE supports 5G-SRVCC from NG-RAN to UTRAN (see 3GPP TS 23.216 [6A]).

8.2.6.26 Supported codecs

This IE shall be included if the UE supports 5G-SRVCC from NG-RAN to UTRAN.

8.2.6.27 UE radio capability ID

This IE shall be included if the UE is not in NB-N1 mode, the UE supports RACS and the UE needs to signal a UE radio capability ID to the network.

8.2.6.28 Requested mapped NSSAI

This IE shall be included by the UE when the UE has a PDN connection or a PDU session to transfer to visited PLMN associated only with an S-NSSAI that is applicable in the HPLMN as specified in clause 5.5.1.3.2.

8.2.6.29 Additional information requested

The UE shall include this IE if the UE supports ciphered broadcast assistance data and the UE needs to obtain new ciphering keys for ciphered broadcast assistance data.

8.2.6.30 Requested WUS assistance information

The UE may include this IE if:

- the UE supports WUS assistance information;
- the UE is not performing initial registration for emergency services; and
- the UE does not have an active emergency PDU session.

8.2.6.31 Void

8.2.6.32 N5GC indication

This IE shall be included in the REGISTRATION REQUEST message when the W-AGF is acting on behalf of an N5GC device.

8.2.6.33 Requested NB-N1 mode DRX parameters

The UE shall include this IE if the UE wants to use or change the UE specific DRX parameters for NB-N1 mode.

8.2.6.34 UE request type

The UE shall include this IE if the MUSIM UE requests the release of the NAS signalling connection.

8.2.6.35 Paging restriction

The UE shall include this IE if the Request type is set to "NAS signalling connection release" in the UE request type IE and the UE requests the network to restrict paging.

8.2.6.35 Service-level-AA container

The UE shall include this IE if the UE supporting UAS services requests a registration for UAS services.

8.2.6.36 NID

The UE shall include this IE if the 5G-GUTI in the 5GS mobile identity IE was assigned by an SNPN other than the SNPN with which the UE is registering.

8.2.6.37 MS determined PLMN with disaster condition

The UE shall include this IE when the UE needs to indicate the MS determined PLMN with disaster condition determined as specified in 3GPP TS 23.122 [5].

8.2.6.38 Requested PEIPS assistance information

The UE may include this IE if the UE supports NR paging subgrouping, the UE is not performing initial registration for emergency services, is not registered for emergency services and does not have an active emergency PDU session.

8.2.6.39 Requested T3512 value

The UE may include this IE during the registration procedure if it requests to use MICO mode and T3324 IE is included, to request a particular T3512 timer value.

8.2.7 Registration accept

8.2.7.1 Message definition

The REGISTRATION ACCEPT message is sent by the AMF to the UE. See table 8.2.7.1.1.

Message type: REGISTRATION ACCEPT

Significance: dual

Direction: network to UE

Table 8.2.7.1.1: REGISTRATION ACCEPT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Registration accept message identity	Message type 9.7	М	V	1
	5GS registration result	5GS registration result 9.11.3.6	М	LV	2
77	5G-GUTI	5GS mobile identity 9.11.3.4	0	TLV-E	14
4A	Equivalent PLMNs	PLMN list 9.11.3.45	0	TLV	5-47
54	TAI list	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
15	Allowed NSSAI	NSSAI 9.11.3.37	0	TLV	4-74
11	Rejected NSSAI	Rejected NSSAI 9.11.3.46	0	TLV	4-42
31	Configured NSSAI	NSSAI 9.11.3.37	0	TLV	4-146
21	5GS network feature support	5GS network feature support 9.11.3.5	0	TLV	3-5
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
26	PDU session reactivation result	PDU session reactivation result 9.11.3.42	0	TLV	4-34
72	PDU session reactivation result error cause	PDU session reactivation result error cause	0	TLV-E	5-515
79	LADN information	LADN information	0	TLV-E	12-1715
B-	MICO indication	MICO indication	0	TV	1
9-	Network slicing indication	Network slicing indication	0	TV	1
27	Service area list	Service area list 9 11 3 49	0	TLV	6-114
5E	T3512 value	GPRS timer 3 9 11 2 5	0	TLV	3
5D	Non-3GPP de-registration timer	GPRS timer 2 9 11 2 4	0	TLV	3
16	T3502 value	GPRS timer 2 9 11 2 4	0	TLV	3
34	Emergency number list	Emergency number list 9.11.3.23	0	TLV	5-50
7A	Extended emergency number list	Extended emergency number list 9.11.3.26	0	TLV-E	7-65538
73	SOR transparent container	SOR transparent container 9.11.3.51	0	TLV-E	20-n
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
A-	NSSAI inclusion mode	NSSAI inclusion mode 9.11.3.37A	0	TV	1
76	Operator-defined access category definitions	Operator-defined access category definitions 9.11.3.38	0	TLV-E	3-8323
51	Negotiated DRX parameters	5GS DRX parameters 9.11.3.2A	0	TLV	3
D-	Non-3GPP NW policies	Non-3GPP NW provided policies 9.11.3.36A	0	TV	1
60	EPS bearer context status	EPS bearer context status 9.11.3.23A	0	TLV	4

6E	Negotiated extended DRX parameters	Extended DRX parameters 9.11.3.26A	0	TLV	3-4
6C	T3447 value	GPRS timer 3 9.11.2.5	0	TLV	3
6B	T3448 value	GPRS timer 2 9.11.2.4	0	TLV	3
6A	T3324 value	GPRS timer 3 9.11.2.5	0	TLV	3
67	UE radio capability ID	UE radio capability ID 9.11.3.68	0	TLV	3-n
E-	UE radio capability ID deletion indication	UE radio capability ID deletion indication 9.11.3.69	0	TV	1
39	Pending NSSAI	NSSAI 9.11.3.37	0	TLV	4-146
74	Ciphering key data	Ciphering key data 9.11.3.18C	0	TLV-E	34-n
75	CAG information list	CAG information list 9.11.3.18A	0	TLV-E	3-n
1B	Truncated 5G-S-TMSI configuration	Truncated 5G-S-TMSI configuration 9.11.3.70	0	TLV	3
1C	Negotiated WUS assistance information	WUS assistance information 9.11.3.71	0	TLV	3-n
29	Negotiated NB-N1 mode DRX parameters	NB-N1 mode DRX parameters 9.11.3.73	0	TLV	3
68	Extended rejected NSSAI	Extended rejected NSSAI 9.11.3.75	0	TLV	5-90
7B	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538
33	Negotiated PEIPS assistance information	PEIPS assistance information 9.11.3.80	0	TLV	3-n
35	5GS additional request result	5GS additional request result 9.11.3.81	0	TLV	3
70	NSSRG information	NSSRG information 9.11.3.82	0	TLV-E	7-4099
14	Disaster roaming wait range	Registration wait range 9.11.3.84	0	TLV	4
2C	Disaster return wait range	Registration wait range 9.11.3.84	0	TLV	4
13	List of PLMNs to be used in disaster condition	List of PLMNs to be used in disaster condition 9.11.3.83	0	TLV	2-n
1D	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
1E	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
71	Extended CAG information list	Extended CAG information list 9.11.3.86	0	TLV-E	3-n
7C	NSAG information	NSAG information 9.11.3.87	0	TLV-E	9-3143

8.2.7.2 5G-GUTI

This IE may be included to assign a 5G-GUTI to a UE.

8.2.7.3 Equivalent PLMNs

This IE may be included in order to assign a new equivalent PLMNs list to a UE.

8.2.7.4 TAI list

This IE may be included to assign a TAI list to a UE.

8.2.7.5 Allowed NSSAI

This IE shall be included:

a) if:

- 1) one or more S-NSSAIs in the requested NSSAI of the REGISTRATION REQUEST message are allowed by the AMF for a network not supporting NSSAA;
- 2) one or more S-NSSAIs in the requested NSSAI of the REGISTRATION REQUEST message are not subject to network slice-specific authentication and authorization and are allowed by the AMF; or
- 3) the network slice-specific authentication and authorization has been successfully performed for one or more S-NSSAIs in the requested NSSAI of the REGISTRATION REQUEST message; or
- b) if the initial registration request is not for onboarding services in SNPN or the UE is not registered for onboarding services in SNPN, the requested NSSAI was not included in the REGISTRATION REQUEST message or none of the requested NSSAI are allowed; and
 - 1) the network not supporting NSSAA has one or more default S-NSSAIs; or
 - 2) the network has one or more default S-NSSAIs which are not subject to network slice-specific authentication and authorization.

8.2.7.6 Rejected NSSAI

The network may include this IE to inform the UE of one or more S-NSSAIs that were included in the requested NSSAI in the REGISTRATION REQUEST message but were rejected by the network.

8.2.7.7 Configured NSSAI

The network may include this IE if the network needs to provide the UE with a new configured NSSAI for the current PLMN or SNPN and the UE is neither registering nor registered for onboarding services in SNPN.

8.2.7.8 5GS network feature support

The network may include this IE to inform the UE of the support of certain features. If this IE is not included then the UE shall interpret this as a receipt of an information element with all bits of the value part coded as zero.

8.2.7.9 PDU session status

This IE shall be included when the network needs to indicate the PDU sessions that are associated with the access type the message is sent over, that are active in the network.

8.2.7.10 PDU session reactivation result

This IE shall be included:

- if the Uplink data status IE is included in the REGISTRATION REQUEST message;
- if the Allowed PDU session status IE is included in the REGISTRATION REQUEST message and there is at least one PDU session indicated in the Allowed PDU session status IE for which the user-plane resources can be re-established over 3GPP access.

8.2.7.11 PDU session reactivation result error cause

This IE may be included, if the PDU session reactivation result IE is included and there exist one or more PDU sessions for which the user-plane resources cannot be re-established, to indicate the cause of failure to re-establish the user-plane resources.

8.2.7.12 LADN information

The network shall include this IE if there are valid LADN service area(s) for the subscribed DNN(s) of the UE in the current registration area.

8.2.7.13 MICO indication

The network shall include the MICO indication IE if:

- a)- the UE included the MICO indication IE in the REGISTRATION REQUEST message; and
- b) the network supports and accepts the use of MICO mode.

8.2.7.14 Network slicing indication

This IE shall be included if the user's network slicing subscription has changed in the UDM of a PLMN or an SNPN.

8.2.7.15 Service area list

This IE may be included to assign new service area restrictions to the UE.

8.2.7.16 T3512 value

The AMF shall include this IE during a registration procedure over 3GPP access when the 5GS registration type IE does not indicate "periodic registration updating". The AMF may include this IE during the mobility and periodic registration update procedure over 3GPP access when the 5GS registration type IE indicates "periodic registration updating".

8.2.7.17 Non-3GPP de-registration timer value

This IE may be included if the network needs to indicate to the UE registered over non-3GPP access the value of a non-3GPP de-registration timer value.

8.2.7.18 T3502 value

This IE may be included to indicate a value for timer T3502.

8.2.7.19 Emergency number list

This IE may be sent by the network. If this IE is sent, the contents of this IE indicates a list of emergency numbers valid within the same country as in the PLMN from which this IE is received.

8.2.7.20 Extended emergency number list

This IE may be sent by the network. If this IE is sent, the contents of this IE indicates a list of emergency numbers (with URN information) valid within the same country as in the PLMN from which this IE is received or valid only in the PLMN or SNPN from which this IE is received.

8.2.7.21 SOR transparent container

This IE may be sent by the network.

8.2.7.22 EAP message

EAP message IE is included if the REGISTRATION ACCEPT message is sent as part of registration for emergency services and is used to convey EAP-failure message.

8.2.7.23 NSSAI inclusion mode

This IE shall be included if required by operatory policy.

8.2.7.24 Operator-defined access category definitions

This IE may be included to assign new operator-defined access category definitions to the UE or delete the operatordefined access category definitions at the UE side.

8.2.7.25 Negotiated DRX parameters

The network shall include this IE if the Requested DRX parameters IE was included in the REGISTRATION REQUEST message.

8.2.7.26 Non-3GPP NW policies

The AMF shall not include this IE during a registration procedure over non-3GPP access.

This IE is included if the network needs to indicate whether emergency numbers provided via non-3GPP access can be used to initiate UE detected emergency calls (see 3GPP TS 24.302 [16]). If this IE is not included then the UE shall interpret this as a receipt of an information element with all bits of the value part coded as zero.

NOTE: In this version of the specification, this IE is applicable in case the UE is connected to a PLMN using an ePDG as specified in 3GPP TS 24.302 [16].

8.2.7.27 Negotiated extended DRX parameters

The network shall include the Negotiated extended DRX parameters IE if:

- the UE included the Requested extended DRX parameters IE in the REGISTRATION REQUEST message; and
- the network supports eDRX and accepts the use of eDRX.

8.2.7.28 T3447 value

The network may include T3447 value IE if:

- the UE has indicated support for service gap control in the REGISTRATION REQUEST message; and
- the 5GMM context contains a service gap time value.

8.2.7.29 T3448 value

The network may include this IE if the congestion control for transport of user data via the control plane is active and the UE supports the control plane CIoT 5GS optimizations.

8.2.7.30 T3324 value

The AMF shall include this IE if the UE has requested active time value in the REGISTRATION REQEUST message and the AMF decides to accept the use of MICO mode and the use of the active time.

8.2.7.31 EPS bearer context status

This IE shall be included when the network generated an EPS bearer context status information for the UE during the inter-system change from S1 mode to N1 mode and the network supports N26 interface.

8.2.7.32 UE radio capability ID

This IE may be included if the UE is not in NB-N1 mode, both the UE and the network support RACS and the network needs to assign a network-assigned UE radio capability ID to the UE.

8.2.7.33 UE radio capability ID deletion indication

This IE may be included if the UE is not in NB-N1 mode, both the UE and the network support RACS and the network needs to trigger the UE to delete all network-assigned UE radio capability IDs stored at the UE for the serving PLMN or SNPN.

8.2.7.34 Pending NSSAI

The network may include this IE to inform the UE of one or more S-NSSAIs that are pending as the network slice-specific authentication and authorization procedure is not completed.

8.2.7.35 Ciphering key data

This IE is included if the network needs to send ciphering key data to the UE for ciphered broadcast assistance data.

8.2.7.36 CAG information list

This IE may be included to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.7.37 Truncated 5G-S-TMSI configuration

The network shall include this IE if:

- the UE is in NB-N1 mode;
- the UE requests "control plane CIoT 5GS optimization" in the 5GS update type IE of REGISTRATION REQUEST message;
- the AMF decides to accept the requested CIoT 5GS optimization; and
- the network is configured to provide the truncated 5G-S-TMSI configuration for control plane CIoT 5GS optimizations.

8.2.7.38 Negotiated NB-N1 mode DRX parameters

The network shall include the Negotiated NB-N1 mode DRX parameters IE if the requested NB-N1 mode DRX parameters IE was included in the REGISTRATION REQUEST message.

8.2.7.39 Negotiated WUS assistance information

The network shall include the Negotiated WUS assistance information IE if:

- the UE supports WUS assistance information;
- the AMF supports and accepts the use of WUS assistance information;
- the UE is not performing the initial registration for emergency services; and
- the UE does not have an active emergency PDU session.

8.2.7.40 Extended rejected NSSAI

If the UE supports Extended rejected NSSAI, the network may include this IE to inform the UE of one or more S-NSSAIs that were included in the requested NSSAI in the REGISTRATION REQUEST message but were rejected by the network.

8.2.7.41 Service-level-AA container

The network shall include this IE if the UUAA procedure is triggered for the UE supporting UAS services. The network may include this IE if there is a valid successful UUAA result for the UE in the UE 5GMM context upon a registration for UAS services is requested from the UE.

8.2.7.42 Negotiated PEIPS assistance information

The network shall include the Negotiated PEIPS assistance information IE if:

- the UE supports NR paging subgrouping;
- the AMF supports and accepts the use of PEIPS assistance information for the UE; and
- the UE is not performing initial registration for emergency services and does not have an active emergency PDU session.

8.2.7.43 5GS additional request result

The network may include this IE to inform the UE about the result of additional request.

8.2.7.44 NSSRG information

This IE may be included to provide NSSRG information associated with the configured NSSAI.

8.2.7.45 Disaster roaming wait range

This IE may be included to assign a new disaster roaming wait range to the UE.

8.2.7.46 Disaster return wait range

This IE may be included to assign a new disaster return wait range to the UE.

8.2.7.47 List of PLMNs to be used in disaster condition

This IE may be included by an allowed PLMN to assign a new "list of PLMN(s) to be used in disaster condition" associated with the serving PLMN to the UE.

8.2.7.48 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for roaming". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.7.49 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for regional provision of service". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.7.50 Extended CAG information list

If the UE supports Extended CAG information list, the network may include this IE to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.7.51 NSAG information

If the UE has set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message over 3GPP access, the network may include this IE to provide NSAG information to the UE. Otherwise, the

network shall not include this IE (see the 'comprehension required' scheme in subclause 11.2.5 of 3GPP TS 24.007 [11]).

8.2.8 Registration complete

8.2.8.1 Message definition

The REGISTRATION COMPLETE message is sent by the UE to the AMF. See table 8.2.8.1.1.

Message type: REGISTRATION COMPLETE

Significance: dual

Direction: UE to network

Table 8.2.8.1.1: REGISTRATION COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Registration complete message identity	Message type 9.7	М	V	1
73	SOR transparent container	SOR transparent container 9.11.3.51	0	TLV-E	20

8.2.8.2 SOR transparent container

This IE may be sent by the UE. If this IE is sent, the contents of this IE indicates the UE acknowledgement of successful reception of the SOR transparent container IE in the REGISTRATION ACCEPT message. This IE shall indicate the ME support of SOR-CMCI.

8.2.9 Registration reject

8.2.9.1 Message definition

The REGISTRATION REJECT message is sent by the AMF to the UE. See table 8.2.9.1.1.

Message type: REGISTRATION REJECT

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Registration reject message identity	Message type 9.7	М	V	1
	5GMM cause	5GMM cause 9.11.3.2	М	V	1
5F	T3346 value	GPRS timer 2 9.11.2.4	0	TLV	3
16	T3502 value	GPRS timer 2 9.11.2.4	0	TLV	3
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
69	Rejected NSSAI	Rejected NSSAI 9.11.3.46	0	TLV	4-42
75	CAG information list	CAG information list 9.11.3.18A	0	TLV-E	3-n
68	Extended rejected NSSAI	Extended rejected NSSAI 9.11.3.75	0	TLV	5-90
2C	Disaster return wait range	Registration wait range 9.11.3.84	0	TLV	4
71	Extended CAG information list	Extended CAG information list 9.11.3.86	0	TLV-E	3-n
3A	Lower bound timer value	GPRS timer 3 9.11.2.5	0	TLV	3
1D	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
1E	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114

Table 8.2.9.1.1: REGISTRATION REJEC	CT message content
-------------------------------------	--------------------

NOTE: It is possible for AMFs compliant with version 17.7.0 or 17.8.0 of this specification to send the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE with IEI of value "3B" for this message or the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE with IEI of value "3C" for this message.

8.2.9.2 T3346 value

The AMF may include this IE when the general NAS level mobility management congestion control is active

8.2.9.3 T3502 value

This IE may be included to indicate a value for timer T3502 during the initial registration.

8.2.9.4 EAP message

EAP message IE is included if the REGISTRATION REJECT message is used to convey EAP-failure message.

8.2.9.5 Rejected NSSAI

The network may include this IE to inform the UE of one or more S-NSSAIs that were included in the requested NSSAI in the REGISTRATION REQUEST message but were rejected by the network.

8.2.9.6 CAG information list

This IE may be included to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.9.7 Extended rejected NSSAI

If the UE supports Extended rejected NSSAI, the network may include this IE to inform the UE of one or more S-NSSAIs that were included in the requested NSSAI in the REGISTRATION REQUEST message but were rejected by the network.

8.2.9.8 Disaster return wait range

This IE may be included to assign a new disaster return wait range to the UE.

8.2.9.9 Extended CAG information list

If the UE supports Extended CAG information list, the network may include this IE to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.9.10 Lower bound timer value

The AMF may include this IE when the 5GMM cause is set to #78 "PLMN not allowed to operate at the present UE location", to provide a minimum time value for an entry added to the list of "PLMNs not allowed to operate at the present UE location".

8.2.9.11 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for roaming". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.9.12 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for regional provision of service". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.10 UL NAS transport

8.2.10.1 Message definition

The UL NAS TRANSPORT message transports message payload and associated information to the AMF. See table 8.2.10.1.1.

Message type: UL NAS TRANSPORT

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	UL NAS TRANSPORT message identity	Message type 9.7	М	V	1
	Payload container type	Payload container type 9.11.3.40	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Payload container	Payload container 9.11.3.39	М	LV-E	3-65537
12	PDU session ID	PDU session identity 2 9.11.3.41	С	TV	2
59	Old PDU session ID	PDU session identity 2 9.11.3.41	0	TV	2
8-	Request type	Request type 9.11.3.47	0	TV	1
22	S-NSSAI	S-NSSAI 9.11.2.8	0	TLV	3-10
25	DNN	DNN 9.11.2.1B	0	TLV	3-102
24	Additional information	Additional information 9.11.2.1	0	TLV	3-n
A-	MA PDU session information	MA PDU session information 9.11.3.31A	0	TV	1
F-	Release assistance indication	Release assistance indication 9.11.3.46A	0	TV	1

Table 8.2.10.1.1: UL NAS TRANSPORT message content

8.2.10.2 PDU session ID

The UE shall include this IE when the Payload container type IE is set to "N1 SM information" or "CIoT user data.

8.2.10.3 Old PDU session ID

The UE shall include this IE if the UL NAS TRANSPORT message transports a PDU SESSION ESTABLISHMENT REQUEST message upon receiving the PDU SESSION MODIFICATION COMMAND message with the 5GSM cause IE set to #39 "reactivation requested" and the Payload container type IE is set to "N1 SM information".

8.2.10.4 Request type

The UE shall include this IE when the PDU session ID IE is included and the Payload container IE contains the PDU SESSION ESTABLISHMENT REQUEST message or the PDU SESSION MODIFICATION REQUEST which is not initiated to indicate a change of 3GPP PS data off UE status associated to a PDU session.

8.2.10.5 S-NSSAI

The UE may include this IE when the Request type IE is set to "initial request", "existing PDU session" or "MA PDU request", the Payload container type IE is set to "N1 SM information" and the UE is not registered for onboarding services in SNPN.

8.2.10.6 DNN

The UE may include this IE when the Request type IE is set to "initial request", "existing PDU session" or "MA PDU request", the Payload container type IE is set to "N1 SM information" and the UE is not registered for onboarding services in SNPN.

8.2.10.7 Additional information

The UE may include this IE when the Payload container type IE is set to "LTE Positioning Protocol (LPP) message container" or "Location services (LCS) message container".

8.2.10.8 MA PDU session information

The UE may include this IE if the Request type IE is included and is not set to "initial emergency request " or "existing emergency PDU session" in the UL NAS TRANSPORT message.

8.2.10.9 Release assistance indication

The UE may include this IE to inform the network whether:

- no further uplink and no further downlink data transmission is expected; or
- only a single downlink data transmission (e.g. acknowledgement or response to uplink data) and no further uplink data transmission subsequent to the uplink data transmission is expected.

8.2.11 DL NAS transport

8.2.11.1 Message definition

The DL NAS TRANSPORT message transports message payload and associated information to the UE. See table 8.2.11.1.1.

Message type: DL NAS TRANSPORT

Significance: dual

Direction: network to UE

Table 8.2.11.1.1: DL NAS TRANSPORT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	DL NAS TRANSPORT message identity	Message type 9.7	М	V	1
	Payload container type	Payload container type 9.11.3.40	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Payload container	Payload container 9.11.3.39	М	LV-E	3-65537
12	PDU session ID	PDU session identity 2 9.11.3.41	С	TV	2
24	Additional information	Additional information 9.11.2.1	0	TLV	3-n
58	5GMM cause	5GMM cause 9.11.3.2	0	TV	2
37	Back-off timer value	GPRS timer 3 9.11.2.5	0	TLV	3
ЗA	Lower bound timer value	GPRS timer 3 9.11.2.5	0	TLV	3

8.2.11.2 PDU session ID

The AMF shall include this IE when the Payload container type IE is set to "N1 SM information" or "CIoT user data container".

8.2.11.3 Additional information

The AMF may include this IE when the Payload container type IE is set to "LTE Positioning Protocol (LPP) message container" or "Location services (LCS) message container".

8.2.11.4 5GMM cause

The AMF shall include this IE when the Payload container IE contains an uplink payload which was not forwarded and the Payload container type IE is not set to "Multiple payloads".

8.2.11.5 Back-off timer value

The AMF shall include this IE when the Payload container IE contains an uplink 5GSM message which was not forwarded due to DNN based congestion control, S-NSSAI and DNN based congestion control, S-NSSAI only based congestion control or the DNN is not supported or not subscribed in a slice, and the Payload container type IE is not set to "Multiple payloads".

8.2.11.6 Lower bound timer value

The AMF may include this IE when the 5GMM cause is set to #78 "PLMN not allowed to operate at the present UE location", to provide a minimum time value for an entry added to the list of "PLMNs not allowed to operate at the present UE location".

8.2.12 De-registration request (UE originating de-registration)

8.2.12.1 Message definition

The DEREGISTRATION REQUEST message is sent by the UE to the AMF. See table 8.2.12.1.1.

Message type: DEREGISTRATION REQUEST

Significance: dual

Direction: UE to network

Table 8.2.12.1.1: DEREGISTRATION REQUEST	message content
--	-----------------

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	De-registration request message identity	Message type 9.7	М	V	1
	De-registration type	De-registration type 9.11.3.20	М	V	1/2
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
	5GS mobile identity	5GS mobile identity 9.11.3.4	М	LV-E	6-n

8.2.13 De-registration accept (UE originating de-registration)

8.2.13.1 Message definition

The DEREGISTRATION ACCEPT message is sent by the AMF to the UE. See table 8.2.13.1.1.

Message type: DEREGISTRATION ACCEPT

Significance: dual

Direction: network to UE

Table 8.2.13.1.1: DEREGISTRATION ACCEPT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	De-registration accept message identity	Message type 9.7	М	V	1

8.2.14 De-registration request (UE terminated de-registration)

8.2.14.1 Message definition

The DEREGISTRATION REQUEST message is sent by the AMF to the UE. See table 8.2.14.1.1.

Message type: DEREGISTRATION REQUEST

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	De-registration request message identity	Message type 9.7	М	V	1
	De-registration type	De-registration type 9.11.3.20	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
58	5GMM cause	5GMM cause 9.11.3.2	0	TV	2
5F	T3346 value	GPRS timer 2 9.11.2.4	0	TLV	3
6D	Rejected NSSAI	Rejected NSSAI 9.11.3.46	0	TLV	4-42
75	CAG information list	CAG information list 9.11.3.18A	0	TLV-E	3-n
68	Extended rejected NSSAI	Extended rejected NSSAI 9.11.3.75	0	TLV	5-90
2C	Disaster return wait range	Registration wait range 9.11.3.84	0	TLV	4
71	Extended CAG information list	Extended CAG information list 9.11.3.86	0	TLV-E	3-n
3A	Lower bound timer value	GPRS timer 3 9.11.2.5	0	TLV	3
1D	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
1E	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114

Table 8.2.14.1.1: DEREGISTRATION REQUEST message content

NOTE: It is possible for AMFs compliant with version 17.7.0 or 17.8.0 of this specification to send the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE with IEI of value "3B" for this message or the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE with IEI of value "3C" for this message.

8.2.14.2 5GMM cause

This information element is included if a 5GMM cause is provided.

8.2.14.3 T3346 value

The AMF may include this IE when the general NAS level mobility management congestion control is active.

8.2.14.4 Rejected NSSAI

The AMF may include this IE to inform the UE of one or more S-NSSAIs that were rejected by the network due to network slice-specific authentication and authorization failure or revocation as specified in subclause 4.6.2.4.

8.2.14.5 CAG information list

This IE may be included to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.14.6 Extended rejected NSSAI

If the UE supports Extended rejected NSSAI, the AMF may include this IE to inform the UE of one or more S-NSSAIs that were rejected by the network due to network slice-specific authentication and authorization failure or revocation as specified in subclause 4.6.2.4.

8.2.14.7 Disaster return wait range

This IE may be included to assign a new disaster return wait range to the UE.

8.2.14.7A Extended CAG information list

If the UE supports Extended CAG information list, the network may include this IE to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.14.8 Lower bound timer value

The AMF may include this IE when the 5GMM cause is set to #78 "PLMN not allowed to operate at the present UE location", to provide a minimum time value for an entry added to the list of "PLMNs not allowed to operate at the present UE location".

8.2.14.9 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for roaming". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.14.10 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for regional provision of service". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.15 De-registration accept (UE terminated de-registration)

8.2.15.1 Message definition

The DEREGISTRATION ACCEPT message is sent by the UE to the AMF. See table 8.2.15.1.1.

Message type: DEREGISTRATION ACCEPT

Significance: dual

Direction: UE to network

Table 8.2.15.1.1.1: DEREGISTRATION ACCEPT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	De-registration accept message identity	Message type 9.7	М	V	1

8.2.16 Service request

8.2.16.1 Message definition

The SERVICE REQUEST message is sent by the UE to the AMF in order to request the establishment of an N1 NAS signalling connection and/or to request the establishment of user-plane resources for PDU sessions which are established without user-plane resources. See table 8.2.16.1.1.

Message type: SERVICE REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Service request message identity	Message type 9.7	М	V	1
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
	Service type	Service type 9.11.3.50	М	V	1/2
	5G-S-TMSI	5GS mobile identity 9.11.3.4	М	LV-E	9
40	Uplink data status	Uplink data status 9.11.3.57	0	TLV	4-34
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
25	Allowed PDU session status	Allowed PDU session status 9.11.3.13	0	TLV	4-34
71	NAS message container	NAS message container 9.11.3.33	0	TLV-E	4-n
29	UE request type	UE request type 9.11.3.76	0	TLV	3
28	Paging restriction	Paging restriction 9.11.3.77	0	TLV	3-35

Table 8.2.16.1.1: SERVICE REQUEST message content

8.2.16.2 Uplink data status

This IE shall be included if the UE has uplink user data pending to be sent.

8.2.16.3 PDU session status

This IE shall be included when the UE needs to indicate the PDU sessions that are associated with the access type that the message is sent over, that are active within the UE.

8.2.16.4 Allowed PDU session status

This IE shall be included if the SERVICE REQUEST message is sent as a response to paging or notification via 3GPP access for PDU session(s) associated with non-3GPP access and the UE needs to indicate the user-plane resources of PDU session(s) associated with non-3GPP access allowed to be re-established over 3GPP access or if there is no PDU session(s) for which the UE allows the user-plane resources to be re-established over 3GPP access.
8.2.16.5 NAS message container

This IE shall be included if the UE is sending a SERVICE REQUEST message as an initial NAS message and the UE needs to send non-cleartext IEs.

8.2.16.6 UE request type

The UE shall include this IE if the MUSIM UE requests the release of the NAS signalling connection or rejects the paging request from the network.

8.2.16.7 Paging restriction

The UE shall include this IE if the Request type is set to "NAS signalling connection release" or to "Rejection of paging" in the UE request type IE and the UE requests the network to restrict paging.

8.2.17 Service accept

8.2.17.1 Message definition

The SERVICE ACCEPT message is sent by the AMF to the UE in order to accept the service request procedure. See table 8.2.17.1.1.

Message type: SERVICE ACCEPT

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator Extended protocol discriminat 9.2		М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Service accept message identity	Message type 9.7	М	V	1
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
26	PDU session reactivation result	PDU session reactivation result 9.11.3.42	0	TLV	4-34
72	PDU session reactivation result error cause	PDU session reactivation result error cause 9.11.3.43	0	TLV-E	5-515
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
6B	T3448 value	GPRS timer 2 9.11.2.4	0	TLV	3
34	5GS additional request result	5GS additional request result 9.11.3.81	0	TLV	3
1D	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
1E	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114

Table 8.2.17.1.1: SERVICE ACCEPT message content

8.2.17.2 PDU session status

This IE shall be included when the network needs to indicate the PDU sessions that are associated with the access type that the message is sent over that are active within the network.

8.2.17.3 PDU session reactivation result

This IE shall be included:

- if the Uplink data status IE is included in the SERVICE REQUEST message;
- if the Allowed PDU session status IE is included in the SERVICE REQUEST message and there is at least one PDU session indicated in the Allowed PDU session status IE for which user-plane resources can be re-established over 3GPP access.

8.2.17.4 PDU session reactivation result error cause

This IE may be included if the PDU session reactivation result IE is included and there exist one or more PDU sessions for which the user-plane resources cannot be re-established, to indicate the cause of failure to re-establish the user-plane resources.

8.2.17.5 EAP message

EAP message IE is included if the SERVICE ACCEPT message is sent to a UE registered for emergency services and is used to convey EAP-failure message.

8.2.17.6 T3448 value

The network may include this IE if the congestion control for transport of user data via the control plane is active and the UE supports the control plane CIoT 5GS optimizations.

8.2.17.7 5GS additional request result

The network may include this IE to inform the UE about the result of additional request.

8.2.17.8 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for roaming". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.17.9 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for regional provision of service". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.18 Service reject

8.2.18.1 Message definition

The SERVICE REJECT message is sent by the AMF to the UE in order to reject the service request procedure. See table 8.2.18.1.1.

Message type: SERVICE REJECT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Service reject message identity	Message type 9.7	М	V	1
	5GMM cause	5GMM cause 9.11.3.2	М	V	1
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
5F	T3346 value	GPRS timer 2 9.11.2.4	0	TLV	3
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
6B	T3448 value	GPRS timer 2 9.11.2.4	0	TLV	3
75	CAG information list	CAG information list 9.11.3.18A	0	TLV-E	3-n
2C	Disaster return wait range	Registration wait range 9.11.3.84	0	TLV	4
71	Extended CAG information list	Extended CAG information list 9.11.3.86	0	TLV-E	3-n
3A	Lower bound timer value	GPRS timer 3 9.11.2.5	0	TLV	3
1D	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
1E	Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"	5GS tracking area identity list 9.11.3.9	0	TLV	9-114

Table 8.2.18.1.1: SERVICE REJECT message content

NOTE: It is possible for AMFs compliant with version 17.7.0 or 17.8.0 of this specification to send the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming" IE with IEI of value "3B" for this message or the Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service" IE with IEI of value "3C" for this message.

8.2.18.2 PDU session status

This IE shall be included when the network needs to indicate the PDU sessions that are associated with the access type that the message is sent over, that are active within the network.

8.2.18.3 T3346 value

The AMF may include this IE when the general NAS level mobility management congestion control is active.

8.2.18.4 EAP message

EAP message IE is included if the SERVICE REJECT message is used to convey EAP-failure message.

8.2.18.5 T3448 value

The network may include this IE if the congestion control for transport of user data via the control plane is active and the UE supports the control plane CIoT 5GS optimizations.

8.2.18.6 CAG information list

This IE may be included to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.18.7 Disaster return wait range

This IE may be included to assign a new disaster return wait range to the UE.

8.2.18.8 Extended CAG information list

If the UE supports Extended CAG information list, the network may include this IE to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.18.9 Lower bound timer value

The AMF may include this IE when the 5GMM cause is set to #78 "PLMN not allowed to operate at the present UE location", to provide a minimum time value for an entry added to the list of "PLMNs not allowed to operate at the present UE location".

8.2.18.10 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for roaming"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for roaming". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.18.11 Forbidden TAI(s) for the list of "5GS forbidden tracking areas for regional provision of service"

This IE is included to indicate the forbidden TAI(s) to be stored in the list of "5GS forbidden tracking areas for regional provision of service". This IE is included only if the message is sent via satellite NG-RAN access.

8.2.19 Configuration update command

8.2.19.1 Message definition

The CONFIGURATION UPDATE COMMAND message is sent by the AMF to the UE. See table 8.2.19.1.1.

Message type: CONFIGURATION UPDATE COMMAND

Significance: dual

Table 8.2.19.1.1: CONFIGURATION UPDATE COMMAND message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator	М	V	1
	Security header type	Security header type	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Configuration update command message identity	Message type 9.7	М	V	1
D-	Configuration update indication	Configuration update indication 9.11.3.18	0	TV	1
77	5G-GUTI	5GS mobile identity 9.11.3.4	0	TLV-E	14
54	TAI list	5GS tracking area identity list 9.11.3.9	0	TLV	9-114
15	Allowed NSSAI	NSSAI 9.11.3.37	0	TLV	4-74
27	Service area list	Service area list 9.11.3.49	0	TLV	6-114
43	Full name for network	Network name 9.11.3.35	0	TLV	3-n
45	Short name for network	Network name 9.11.3.35	0	TLV	3-n
46	Local time zone	Time zone 9.11.3.52	0	TV	2
47	Universal time and local time zone	Time zone and time 9.11.3.53	0	TV	8
49	Network daylight saving time	Daylight saving time 9.11.3.19	0	TLV	3
79	LADN information	LADN information 9.11.3.30	0	TLV-E	3-1715
B-	MICO indication	MICO indication 9.11.3.31	0	TV	1
9-	Network slicing indication	Network slicing indication 9.11.3.36	0	TV	1
31	Configured NSSAI	NSSAI 9.11.3.37	0	TLV	4-146
11	Rejected NSSAI	Rejected NSSAI 9.11.3.46	0	TLV	4-42
76	Operator-defined access category definitions	Operator-defined access category definitions 9.11.3.38	0	TLV-E	3-8323
F-	SMS indication	SMS indication 9.11.3.50A	0	TV	1
6C	T3447 value	GPRS timer 3 9.11.2.5	0	TLV	3
75	CAG information list	CAG information list 9.11.3.18A	0	TLV-E	3-n
67	UE radio capability ID	UE radio capability ID 9.11.3.68	0	TLV	3-n
A-	UE radio capability ID deletion indication	UE radio capability ID deletion indication 9.11.3.69	0	TV	1
44	5GS registration result	5GS registration result 9.11.3.6	0	TLV	3
1B	Truncated 5G-S-TMSI configuration	Truncated 5G-S-TMSI configuration 9.11.3.70	0	TLV	3
C-	Additional configuration indication	Additional configuration indication 9.11.3.74	0	TV	1
68	Extended rejected NSSAI	Extended rejected NSSAI 9.11.3.75	0	TLV	5-90
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538
70	NSSRG information	NSSRG information 9.11.3.82	0	TLV-E	7-4099

14	Disaster roaming wait range	Registration wait range 9.11.3.84	0	TLV	4
2C	Disaster return wait range	Registration wait range 9.11.3.84	0	TLV	4
13	List of PLMNs to be used in disaster condition	List of PLMNs to be used in disaster condition 9.11.3.83	0	TLV	2-n
71	Extended CAG information list	Extended CAG information list 9.11.3.86	0	TLV-E	3-n
1F	Updated PEIPS assistance information	PEIPS assistance information 9.11.3.80	0	TLV	3-n
73	NSAG information	NSAG information 9.11.3.87	0	TLV-E	9-3143
E-	Priority indicator	Priority indicator 9.11.3.91	0	TV	1

8.2.19.2 Configuration update indication

The AMF shall include this IE if the AMF needs to request an acknowledgement or a registration procedure from the UE.

8.2.19.3 5G-GUTI

This IE may be included to assign a new 5G GUTI to the UE.

8.2.19.4 TAI list

This IE may be included to assign a new TAI list to the UE.

8.2.19.5 Allowed NSSAI

This IE may be included to assign a new allowed NSSAI to the UE not registered for onboarding services in SNPN.

8.2.19.6 Service area list

This IE may be included to assign a new service area list to the UE.

8.2.19.7 Full name for network

This IE may be included to assign a new full name for network to the UE.

8.2.19.8 Short name for network

This IE may be included to assign a new short name for network to the UE.

8.2.19.9 Local time zone

This IE may be included to assign a new local time zone to the UE.

8.2.19.10 Universal time and local time zone

This IE may be included to assign new universal time and local time zone to the UE.

8.2.19.11 Network daylight saving time

This IE may be included to assign new network daylight saving time to the UE.

8.2.19.12 LADN information

This IE may be included to assign new LADN information to the UE or delete the LADN information at the UE side.

8.2.19.13 MICO indication

This IE may be included to request the UE to re-negotiate MICO mode.

8.2.19.14 Network slicing indication

This IE shall be included if the user's network slicing subscription has changed in the UDM of a PLMN or an SNPN.

8.2.19.15 Configured NSSAI

The AMF shall include this IE when the AMF needs to provide the UE with a new configured NSSAI for the current PLMN or SNPN and the UE is neither registering nor registered for onboarding services in SNPN.

8.2.19.16 Rejected NSSAI

The network may include this IE to inform the UE of one or more S-NSSAIs that were previously sent to the UE in the allowed NSSAI or the pending NSSAI, but are now considered rejected by the network.

8.2.19.17 Operator-defined access category definitions

This IE may be included to assign new operator-defined access category definitions to the UE or delete the operatordefined access category definitions at the UE side.

8.2.19.18 SMS indication

This IE may be included to indicate that the ability for the UE to use SMS over NAS has changed.

8.2.19.19 T3447 value

This IE may be included to assign a new T3447 value to the UE.

8.2.19.20 CAG information list

This IE may be included to assign new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.19.21 UE radio capability ID

This IE may be included if the UE is not in NB-N1 mode, both the UE and the network support RACS and the network needs to assign a network-assigned UE radio capability ID to the UE.

8.2.19.22 UE radio capability ID deletion indication

This IE may be included if the UE is not in NB-N1 mode, both the UE and the network support RACS and the network needs to trigger the UE to delete all network-assigned UE radio capability IDs stored at the UE for the serving PLMN or serving SNPN.

8.2.19.23 5GS registration result

This IE shall be included if the network wants to indicate to the UE that the UE is registered for emergency services.

8.2.19.24 Truncated 5G-S-TMSI configuration

This IE may be included to provide a new truncated 5G-S-TMSI configuration to the UE in NB-N1 mode if the network is configured to provide the truncated 5G-S-TMSI configuration for control plane CIoT 5GS optimizations.

8.2.19.25 Additional configuration indication

The network may include this IE when requesting the UE to register without the release of the N1 NAS signalling connection.

8.2.19.26 Extended rejected NSSAI

If the UE supports Extended rejected NSSAI, the network may include this IE to inform the UE of one or more S-NSSAIs that were previously sent to the UE in the allowed NSSAI or the pending NSSAI, but are now considered rejected by the network.

8.2.19.27 Service-level-AA container

The network shall include this IE when the AMF receives the Service-level-AA payload or the UUAA-MM result from the UAS-NF during the UUAA-MM procedure or the UUAA revocation procedure. The network shall also include this IE if the AMF receives from the UAS-AF, the CAA-Level UAV ID as part of the UUAA-MM procedure.

8.2.19.28 NSSRG information

This IE may be included to provide NSSRG information associated with the configured NSSAI.

8.2.19.29 Disaster roaming wait range

This IE may be included to assign a new disaster roaming wait range to the UE.

8.2.19.30 Disaster return wait range

This IE may be included to assign a new disaster return wait range to the UE.

8.2.19.31 List of PLMNs to be used in disaster condition

This IE may be included by an allowed PLMN to assign a new "list of PLMN(s) to be used in disaster condition" associated with the serving PLMN to the UE.

8.2.19.32 Extended CAG information list

If the UE supports Extended CAG information list, the network may include this IE to assign a new "CAG information list" to the UE or delete the "CAG information list" at the UE side.

8.2.19.33 Updated PEIPS assistance information

The AMF may include this IE if the UE supports NR paging subgrouping, the AMF supports and accepts the use of PEIPIS assistance information for the UE, the UE is not registered for emergency services, the UE does not have an active emergency PDU session, and the network needs to update PEIPS assistance information for the UE.

8.2.19.34 NSAG information

If the UE has set the NSAG bit to "NSAG supported" in the 5GMM capability IE of the REGISTRATION REQUEST message and the CONFIGURATION UPDATE COMMAND message is sent over 3GPP access, the network may include this IE to provide NSAG information to the UE.

8.2.19.35 Priority indicator

The network shall include this IE when it needs to inform the UE that the use of access identity 1 is valid or is no longer valid.

8.2.20 Configuration update complete

8.2.20.1 Message definition

The CONFIGURATION UPDATE COMPLETE message is sent by the UE to the AMF. See table 8.2.20.1.1.

Message type: CONFIGURATION UPDATE COMPLETE

Significance: dual

Direction: UE to network

Table 8.2.20.1.1: CONFIGURATION UPDATE COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Configuration update complete message identity	Message type 9.7	М	V	1

8.2.20.2 Void

8.2.21 Identity request

8.2.21.1 Message definition

The IDENTITY REQUEST message is sent by the AMF to the UE to request the UE to provide specified identity. See table 8.2.21.1.1

Message type: IDENTITY REQUEST

Significance: dual

Direction: AMF to UE

Table 8.2.21.1.1: IDENTITY REQUEST message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Identity request message identity	Message type 9.7	М	V	1
	Identity type	5GS identity type 9.11.3.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2

8.2.22 Identity response

8.2.22.1 Message definition

The IDENTITY RESPONSE message is sent by the UE to the AMF to provide the requested identity. See table 8.2.22.1.

Message type: IDENTITY RESPONSE

Significance: dual

Direction: UE to AMF

Table 8.2.22.1.1: IDENTITY RESPONSE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Identity response message identity	Message type 9.7	М	V	1
	Mobile identity	5GS mobile identity 9.11.3.4	М	LV-E	3-n

8.2.23 Notification

8.2.23.1 Message definition

The NOTIFICATION message is sent by the AMF to the UE to notify the UE to initiate a service request procedure. See table 8.2.23.1.1.

Message type: NOTIFICATION

Significance: dual

Table 0.2.23.1.1. NOTIFICATION ITESSage content	Table	8.2.23.1.1	: NOTIFICAT	ION messag	ge content
---	-------	------------	-------------	------------	------------

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	Μ	V	1/2
	Notification message identity	Message type 9.7	Μ	V	1
	Access type	Access type 9.11.2.1A	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2

8.2.24 Notification response

8.2.24.1 Message definition

The NOTIFICATION RESPONSE message is sent by the UE to the AMF to notify the failure to initiate the service request procedure as a response of notification. See table 8.2.24.1.1.

Message type: NOTIFICATION RESPONSE

Significance: dual

Direction: UE to network

Table 8.2.2341.1: NOTIFICATION RESPONSE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Notification response message identity	Message type 9.7	М	V	1
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34

8.2.24.2 PDU session status

This information element shall be included when the UE needs to indicate over non-3GPP access the type of the PDU sessions that are associated with the 3GPP access type that are active within the UE.

8.2.25 Security mode command

8.2.25.1 Message definition

The SECURITY MODE COMMAND message is sent by the AMF to the UE to establish NAS signalling security. See table 8.2.25.1.1.

Message type: SECURITY MODE COMMAND

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Security mode command message identity	Message type 9.7	М	V	1
	Selected NAS security algorithms	NAS security algorithms 9.11.3.34	М	V	1
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Replayed UE security capabilities	UE security capability 9.11.3.54	М	LV	3-9
E-	IMEISV request	IMEISV request 9.11.3.28	0	TV	1
57	Selected EPS NAS security algorithms	EPS NAS security algorithms 9.11.3.25	0	TV	2
36	Additional 5G security information	Additional 5G security information 9.11.3.12	0	TLV	3
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
38	ABBA	ABBA 9.11.3.10	0	TLV	4-n
19	Replayed S1 UE security capabilities	S1 UE security capability 9.11.3.48A	0	TLV	4-7

Table 8.2.25.1.1: SECURITY MODE COMMAND message content

8.2.25.2 IMEISV request

The AMF may include this information element to request the UE to send its IMEISV with the corresponding SECURITY MODE COMPLETE message.

8.2.25.3 Void

8.2.25.4 Selected EPS NAS security algorithms

This IE shall be included if the AMF supports N26 interface, the UE set the S1 mode bit to "S1 mode supported" in the 5GMM capability IE of the REGISTRATION REQUEST message, and the AMF needs to provide the selected EPS NAS security algorithms to the UE.

8.2.25.5 Additional 5G security information

The network shall include this IE if:

- a) the network needs to provide the UE with horizontal derivation parameter; or
- b) the initial NAS message (i.e. REGISTRATION REQUEST or SERVICE REQUEST) does not successfully pass the integrity check at the AMF (see subclause 5.4.2.2).

8.2.25.6 EAP message

This IE is included when the EAP Success message is sent as part of the EAP based primary authentication and key agreement procedure, as specified in subclause 5.4.1.2.

8.2.25.7 ABBA

This IE shall be included if the message contains an EAP message IE with an EAP-success message.

8.2.25.8 Replayed S1 UE security capabilities

This IE shall be included if the Selected EPS NAS security algorithms information element is included.

8.2.26 Security mode complete

8.2.26.1 Message definition

The SECURITY MODE COMPLETE message is sent by the UE to the AMF in response to a SECURITY MODE COMMAND message. See table 8.2.26.1.1.

Message type: SECURITY MODE COMPLETE

Significance: dual

Direction: UE to network

Table 8.2.26.1.1: SECURITY MODE COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Security mode complete message identity	Message type 9.6	М	V	1
77	IMEISV	5GS mobile identity 9.11.3.4	0	TLV-E	12
71	NAS message container	NAS message container 9.11.3.33	0	TLV-E	4-n
78	non-IMEISV PEI	5GS mobile identity 9.11.3.4	0	TLV-E	7-n

8.2.26.2 IMEISV

The UE shall include this information element, if the IMEISV was requested within the corresponding SECURITY MODE COMMAND message.

8.2.26.3 NAS message container

The UE shall include this information element:

- a) if during an ongoing registration procedure or service request procedure, the AMF included the Additional 5G security information with the RINMR bit set to "Retransmission of the initial NAS message requested" in the SECURITY MODE COMMAND message as described in 3GPP TS 33.501 [24]; or
- b) if during an ongoing registration procedure, the UE does not have a valid 5G NAS security context.

8.2.26.4 non-IMEISV PEI

The 5G-RG or the W-AGF acting on behalf of the FN-RG (or on behalf of the N5GC device) shall include this information element, if the IMEISV was requested within the corresponding SECURITY MODE COMMAND message, the IMEISV is not available but MAC address is available.

The UE shall include this information element, if the IMEISV was requested within the corresponding SECURITY MODE COMMAND message, the IMEISV is not available but EUI-64 is available.

8.2.27 Security mode reject

8.6.27.1 Message definition

The SECURITY MODE REJECT message is sent by the UE to the AMF to indicate that the corresponding security mode command has been rejected. See table 8.2.27.1.1.

Message type: SECURITY MODE REJECT

Significance: dual

Direction: UE to network

Table 8.2.27.1.1: SECURITY MODE REJECT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Security mode reject message identity	Message type 9.6	М	V	1
	5GMM cause	5GMM cause 9.11.3.2	М	V	1

8.2.28 Security protected 5GS NAS message

8.2.28.1 Message definition

This message is sent by the UE or the network to transfer a plain 5GS NAS message as specified in subclause 8.2 together with the sequence number and the message authentication code protecting the message. See table 8.2.28.1.1.

Message type: SECURITY PROTECTED 5GS NAS MESSAGE

Significance: dual

Direction: both

Table 8.2.28.1.1: SECURITY PROTECTED 5GS NAS MESSAGE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Message authentication code	Message authentication code 9.8	М	V	4
	Sequence number	Sequence number 9.10	Μ	V	1
	Plain 5GS NAS message	Plain 5GS NAS message 9.9	М	V	3-n

NOTE: The minimum length of Plain 5GS NAS message IE can be 2 octets if it includes a Test Mode Control message specified in 3GPP TS 38.509 [31AA].

8.2.29 5GMM status

8.2.29.1 Message definition

The 5GMM STATUS message is sent by the UE or by the AMF at any time to report certain error conditions. See table 8.2.28.1.1.

Message type: 5GMM STATUS

Significance: local

Direction: both

Table 8.2.29.1.1: 5GMM STATUS message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	Μ	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	5GMM STATUS message identity	Message type 9.7	Μ	V	1
	5GMM cause	5GMM cause 9.11.3.2	М	V	1

8.2.30 Control Plane Service request

8.2.30.1 Message definition

The CONTROL PLANE SERVICE REQUEST message is sent by the UE to the AMF when the UE is using 5GS services with control plane CIoT 5GS optimization. See table 8.2.30.1.1.

Message type: CONTROL PLANE SERVICE REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Control plane service request message identity	Message type 9.7	М	V	1
	Control plane service type	Control plane service type 9.11.3.18D	М	V	1/2
	ngKSI	NAS key set identifier 9.11.3.32	М	V	1/2
6F	CloT small data container	CloT small data container 9.11.3.18B	0	TLV	4-257
8-	Payload container type	Payload container type 9.11.3.40	0	TV	1
7B	Payload container	Payload container 9.11.3.39	0	TLV-E	4-65538
12	PDU session ID	PDU session identity 2 9.11.3.41	С	TV	2
50	PDU session status	PDU session status 9.11.3.44	0	TLV	4-34
F-	Release assistance indication	Release assistance indication 9.11.3.46A	0	TV	1
40	Uplink data status	Uplink data status 9.11.3.57	0	TLV	4-34
71	NAS message container	NAS message container 9.11.3.33	0	TLV-E	4-n
24	Additional information	Additional information 9.11.2.1	0	TLV	3-n
25	Allowed PDU session status	Allowed PDU session status 9.11.3.13	0	TLV	4-34
29	UE request type	UE request type 9.11.3.76	0	TLV	3
28	Paging restriction	Paging restriction 9.11.3.77	0	TLV	3-35

Table 8.2.30.1.1: CONTROL PLANE SERVICE REQUEST message content

8.2.30.2 CloT small data container

This IE shall be included if the UE needs to send uplink small user data, SMS or location services message or uplink SMS that is not more than 254 bytes, and there is no other optional IE to be sent.

NOTE: When the UE determines to use the CIoT small data container IE to send uplink data in this message, there is no other optional IEs in this message.

8.2.30.3 Payload container type

This IE shall be included if the UE includes the Payload container IE.

8.2.30.4 Payload container

This IE shall be included if the UE needs to send uplink CIoT user data, SMS or location services message.

8.2.30.5 PDU session ID

The UE shall include this IE when the Payload container type IE is set to "CIoT user data container".

8.2.30.6 PDU session status

This IE shall be included when the UE needs to indicate the PDU sessions that are associated with the access type that the message is sent over, that are active within the UE.

8.2.30.7 Release assistance indication

The UE may include this IE to inform the network whether:

- no further uplink and no further downlink data transmission is expected; or
- only a single downlink data transmission (e.g. acknowledgement or response to uplink data) and no further uplink data transmission subsequent to the uplink data transmission is expected.

8.2.30.8 Uplink data status

This IE shall be included if the UE has uplink user data pending to be sent over the user plane.

8.2.30.9 NAS message container

This IE shall be included if the UE is sending a CONTROL PLANE SERVICE REQUEST message as an initial NAS message and the UE needs to send non-cleartext IEs.

8.2.30.10 Additional information

The UE may include this IE when the Payload container type IE is set to "Location services message container".

8.2.30.11 Allowed PDU session status

This IE shall be included if the CONTROL PLANE SERVICE REQUEST message is sent as a response to paging or notification via 3GPP access for PDU session(s) associated with non-3GPP access and the UE needs to indicate the user-plane resources of PDU session(s) associated with non-3GPP access allowed to be re-established over 3GPP access or if there is no PDU session(s) for which the UE allows the user-plane resources to be re-established over 3GPP access.

8.2.30.12 UE request type

The UE shall include this IE if the MUSIM UE requests the release of the NAS signalling connection or rejects the paging request from the network.

8.2.30.13 Paging restriction

The UE shall include this IE if the Request type is set to "NAS signalling connection release" or to "Rejection of paging" in the UE request type IE and the UE requests the network to restrict paging.

8.2.31 Network slice-specific authentication command

8.2.31.1 Message definition

The NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message is sent by the AMF to the UE for authentication of the upper layers of the UE. See table 8.2.31.1.1.

Message type: NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message identity	Message type 9.7	М	V	1
	S-NSSĂI	S-NSSAI 9.11.2.8	М	LV	2-5
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502

Table 8.2.31.1.1: NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message content

8.2.32 Network slice-specific authentication complete

8.2.32.1 Message definition

The NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message is sent by the UE to the AMF in response to the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message and indicates acceptance of the NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message. See table 8.2.32.1.1.

Message type: NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE

Significance: dual

Direction: UE to network

Table 8.2.32.1.1: NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message identity	Message type 9.7	М	V	1
	S-NSSAI	S-NSSAI 9.11.2.8	М	LV	2-5
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502

8.2.33 Network slice-specific authentication result

8.2.33.1 Message definition

The NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message is sent by the AMF to the UE for indicating the result of the network slice-specific authentication and authorization procedure. See table 8.2.33.1.1.

Message type: NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message identity	Message type 9.7	М	V	1
	S-NSSAI	S-NSSAI 9.11.2.8	М	LV	2-5
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502

Table 8.2.33.1.1: NETWORK SLICE-SPECIFIC AUTHENTICATION RESULT message content

8.2.34 Relay key request

8.2.34.1 Message definition

The RELAY KEY REQUEST message is sent by the UE to the AMF for initiation of PC5 keys establishment with the 5G ProSe remote UE as specified in 3GPP TS 33.503 [56]. See table 8.2.34.1.

Message type: RELAY KEY REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Relay key request message identity	Message type 9.7	М	V	1
	PRTI	ProSe relay transaction identity 9.11.3.88	М	V	1
	Relay key request parameters	Relay key request parameters 9.11.3.89	М	LV	22-65537

Table 8.2.34.1: RELAY KEY REQUEST message content

8.2.35 Relay key accept

8.2.35.1 Message definition

The RELAY KEY ACCEPT message is sent by the AMF to the UE as specified in 3GPP TS 33.503 [56]. See table 8.2.35.1.

Message type: RELAY KEY ACCEPT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Relay key accept message identity	Message type 9.7	М	V	1
	PRTI	ProSe relay transaction identity 9.11.3.88	М	V	1
	Relay key response parameters	Relay key response parameters 9.11.3.90	М	LV-E	51-65537
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503

Table 8.2.35.1: RELAY KEY ACCEPT message content

8.2.35.2 EAP message

The AMF shall include the EAP message IE if the AMF has received an EAP-success message from the AUSF.

8.2.36 Relay key reject

8.2.36.1 Message definition

The RELAY KEY REJECT message is sent by the AMF to the UE to indicate the rejection of the relay key request. See table 8.2.36.1.

Message type: RELAY KEY REJECT

Significance: dual

Direction: network to UE

Table 8.2.36.1: RELAY KEY REJECT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator	М	V	1
		9.2			
	Security header type	Security header type	Μ	V	1/2
		9.3			
	Spare half octet	Spare half octet	Μ	V	1/2
		9.5			
	Relay key reject message identity	Message type	М	V	1
		9.7			
	PRTI	ProSe relay transaction identity	Μ	V	1
		9.11.3.88			
78	EAP message	EAP message	0	TLV-E	7-1503
		9.11.2.2			

8.2.36.2 EAP message

EAP message IE is included if the RELAY KEY REJECT message is used to convey EAP-failure message.

8.2.37 Relay authentication request

8.2.37.1 Message definition

The RELAY AUTHENTICATION REQUEST message is sent by the network to the UE to initiate authentication of the 5G ProSe remote UE as specified in 3GPP TS 33.503 [56]. See table 8.2.37.1.

Message type: RELAY AUTHENTICATION REQUEST

Significance: dual

Direction: Network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Relay authentication request message identity	Message type 9.7	М	V	1
	PRTI	ProSe relay transaction identity 9.11.3.88	М	V	1
	EAP message	EAP message 9.11.2.2	М	LV-E	7-1503

Table 8.2.37.1: RELAY AUTHENTICATION REQUEST message content

8.2.38 Relay authentication response

8.2.38.1 Message definition

The RELAY AUTHENTICATION RESPONSE message is sent by the UE to the network to forward the authentication response from the 5G ProSe remote UE as specified in 3GPP TS 33.503 [56]. See table 8.2.38.1.

Message type: RELAY AUTHENTICATION RESPONSE

Significance: dual

Direction: UE to network

Table 8.2.38.1: RELAY AUTHENTICATION RESPONSE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	Security header type	Security header type 9.3	М	V	1/2
	Spare half octet	Spare half octet 9.5	М	V	1/2
	Relay authentication response message identity	Message type 9.7	М	V	1
	PRTI	ProSe relay transaction identity 9.11.3.88	М	V	1
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502

8.3 5GS session management messages

8.3.1 PDU session establishment request

8.3.1.1 Message definition

The PDU SESSION ESTABLISHMENT REQUEST message is sent by the UE to the SMF to initiate establishment of a PDU session. See table 8.3.1.1.1.

Message type: PDU SESSION ESTABLISHMENT REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Lenath
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION ESTABLISHMENT REQUEST message identity	Message type 9.7	М	V	1
	Integrity protection maximum data rate	Integrity protection maximum data rate 9.11.4.7	М	V	2
9-	PDU session type	PDU session type 9.11.4.11	0	TV	1
A-	SSC mode	SSC mode 9.11.4.16	0	TV	1
28	5GSM capability	5GSM capability 9.11.4.1	0	TLV	3-15
55	Maximum number of supported packet filters	Maximum number of supported packet filters 9.11.4.9	0	TV	3
B-	Always-on PDU session requested	Always-on PDU session requested 9.11.4.4	0	TV	1
39	SM PDU DN request container	SM PDU DN request container 9.11.4.15	0	TLV	3-255
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
66	IP header compression configuration	IP header compression configuration 9.11.4.24	0	TLV	5-257
6E	DS-TT Ethernet port MAC address	DS-TT Ethernet port MAC address 9.11.4.25	0	TLV	8
6F	UE-DS-TT residence time	UE-DS-TT residence time 9.11.4.26	0	TLV	10
74	Port management information container	Port management information container 9.11.4.27	0	TLV-E	8-65538
1F	Ethernet header compression configuration	Ethernet header compression configuration 9.11.4.28	0	TLV	3
29	Suggested interface identifier	PDU address 9.11.4.10	0	TLV	11
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538
70	Requested MBS container	Requested MBS container 9.11.4.30	0	TLV-E	8-65538
34	PDU session pair ID	PDU session pair ID 9.11.4.32	0	TLV	3
35	RSN	RSN 9.11.4.33	0	TLV	3

Table 8.3.1.1.1: PDU SESSION ESTABLISHMENT REQUEST message content

8.3.1.2 PDU session type

This IE shall be included in the message when the UE requests to establish a new PDU session.

8.3.1.3 SSC mode

This IE is included in the message when the UE requests to establish a new PDU session with a DN and requests an SSC mode.

8.3.1.4 Maximum number of supported packet filters

This IE shall be included in the message when the selected PDU session type is "IPv4", "IPv6", "IPv4v6" or "Ethernet" and the UE can support more than 16 packet filters for this PDU session.

8.3.1.5 5GSM capability

This IE is included in the message when the UE requests to establish a new PDU session or to transfer an existing PDN connection and any of the 5GSM capabilities supported by the UE is relevant for the PDU session.

8.3.1.6 Void

8.3.1.7 Always-on PDU session requested

The UE shall include this IE if the UE requests to establish a PDU session as an always-on PDU session.

8.3.1.8 SM PDU DN request container

This IE is included in the message when the UE requests to establish a new PDU session with a DN and needs to provide information for the PDU session authentication and authorization by the external DN.

8.3.1.9 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.1.10 IP header compression configuration

The UE shall include the IP header compression configuration IE if:

- the PDU session type value of the PDU session type IE is set to "IPv4", "IPv6" or "IPv4v6";
- the UE indicates "Control Plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- the network indicates "Control plane CIoT 5GS optimization supported" and "IP header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message.

8.3.1.11 DS-TT Ethernet port MAC address

This IE shall be included in the message if the UE supports transfer of port management information containers and the UE requests to establish a new PDU session of "Ethernet" PDU session type.

8.3.1.12 UE-DS-TT residence time

This IE shall be included in the message if:

- a) the UE supports transfer of port management information containers; and
- b) the UE-DS-TT residence time is available at the UE.

8.3.1.13 Port management information container

This IE shall be included in the message if the UE supports transfer of port management information containers.

8.3.1.14 Ethernet header compression configuration

The UE shall include the Ethernet header compression configuration IE if:

- the PDU session type value of the PDU session type IE is set to "Ethernet";
- the UE indicated "Control Plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GMM capability IE of the REGISTRATION REQUEST message; and
- the network indicated "Control plane CIoT 5GS optimization supported" and "Ethernet header compression for control plane CIoT 5GS optimization supported" in the 5GS network support feature IE of the REGISTRATION ACCEPT message.

8.3.1.15 Suggested interface identifier

This IE may be included by the W-AGF acting on behalf of the FN-RG.

8.3.1.16 Service-level-AA container

This IE shall be included in the message when the UE needs to establish a PDU session for the UAS services.

8.3.1.17 Requested MBS container

This IE is included in the message when the UE requests to join one or more multicast MBS sessions that are associated with the PDU session.

8.3.1.18 PDU session pair ID

This IE shall be included in the message when the UE needs to include a PDU session pair ID.

8.3.1.19 RSN

This IE shall be included in the message when the UE needs to include an RSN.

8.3.2 PDU session establishment accept

8.3.2.1 Message definition

The PDU SESSION ESTABLISHMENT ACCEPT message is sent by the SMF to the UE in response to PDU SESSION ESTABLISHMENT REQUEST message and indicates successful establishment of a PDU session. See table 8.3.2.1.1.

Message type: PDU SESSION ESTABLISHMENT ACCEPT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION ESTABLISHMENT ACCEPT message identity	Message type 9.7	М	V	1
	Selected PDU session type	PDU session type 9.11.4.11	М	V	1/2
	Selected SSC mode	SSC mode 9.11.4.16	М	V	1/2
	Authorized QoS rules	QoS rules 9.11.4.13	М	LV-E	6-65538
	Session AMBR	Session-AMBR 9 11 4 14	М	LV	7
59	5GSM cause	5GSM cause 9 11 4 2	0	TV	2
29	PDU address	PDU address 9 11 4 10	0	TLV	7-31
56	RQ timer value	GPRS timer 9 11 2 3	0	TV	2
22	S-NSSAI	S-NSSAI 9 11 2 8	0	TLV	3-10
8-	Always-on PDU session indication	Always-on PDU session indication	0	TV	1
75	Mapped EPS bearer contexts	Mapped EPS bearer contexts 9.11.4.8	0	TLV-E	7-65538
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
79	Authorized QoS flow descriptions	QoS flow descriptions 9 11 4 12	0	TLV-E	6-65538
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
25	DNN	DNN 9.11.2.1B	0	TLV	3-102
17	5GSM network feature support	5GSM network feature support 9.11.4.18	0	TLV	3-15
18	Serving PLMN rate control	Serving PLMN rate control 9.11.4.20	0	TLV	4
77	ATSSS container	ATSSS container 9.11.4.22	0	TLV-E	3-65538
C-	Control plane only indication	Control plane only indication 9.11.4.23	0	TV	1
66	IP header compression configuration	IP header compression configuration 9.11.4.24	0	TLV	5-257
1F	Ethernet header compression configuration	Ethernet header compression configuration 9.11.4.28	0	TLV	3
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538
71	Received MBS container	Received MBS container 9.11.4.31	0	TLV-E	9-65538

Table 8.3.2.1.1: PDU SESSION ESTABLISHMENT ACCEPT message content

8.3.2.2 5GSM cause

This IE is included when the selected PDU session type is different from the PDU session type requested by the UE.

8.3.2.3 PDU address

This IE is included when the selected PDU session type is "IPv4", "IPv6" or "IPv4v6".

8.3.2.4 RQ timer value

This IE is included when the network needs to provide the RQ timer value.

8.3.2.5 S-NSSAI

This IE shall be included in the message when the SMF received from the AMF an S-NSSAI together with the PDU SESSION ESTABLISHMENT REQUEST message, the PDU session is a non-emergency PDU session and the UE is not registered for onboarding services in SNPN.

8.3.2.6 Always-on PDU session indication

The network shall include this IE if the network decides to inform the UE whether the PDU session is established as an always-on PDU session.

8.3.2.7 Mapped EPS bearer contexts

This IE is included when interworking with EPS is supported for the PDU session.

8.3.2.8 EAP message

This IE is included when the external DN successfully performed authentication and authorization of the UE using EAP.

8.3.2.9 Authorized QoS flow descriptions

This IE is included when the network needs to provide authorized QoS flow descriptions.

8.3.2.10 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.2.11 DNN

The IE shall be included in the message when the PDU session is a non-emergency PDU session and the UE is not registered for onboarding services in SNPN.

8.3.2.12 5GSM network feature support

This IE is included when the network needs to indicate support of 5GSM network features.

8.3.2.13 Void

8.3.2.14 Serving PLMN rate control

This IE shall be included when the network needs to indicate the maximum uplink control plane user data the UE is allowed to send per 6 minute interval.

8.3.2.15 ATSSS container

The IE shall be included in the message when the PDU session is an MA PDU session.

8.3.2.16 Control plane only indication

The network shall include the control plane only indication IE if the network determines that the associated PDU session is only for control plane CIoT 5GS optimization.

8.3.2.17 IP header compression configuration

The SMF may include the IP header compression configuration IE if:

- the network accepts an IP PDU session type;
- control plane CIoT 5GS optimization is selected; and
- the UE provided the IP header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message.

8.3.2.18 Ethernet header compression configuration

The SMF may include the Ethernet header compression configuration IE if:

- the network accepts an Ethernet PDU session type;
- control plane CIoT 5GS optimization is selected; and
- the UE provided the Ethernet header compression configuration IE in the PDU SESSION ESTABLISHMENT REQUEST message.

8.3.2.19 Service-level-AA container

The SMF shall include the Service-level-AA container IE if the service-level authentication and authorization procedure is completed successfully by the external DN.

8.3.2.20 Received MBS container

The network shall include this IE if the UE has requested to join one or more multicast MBS sessions.

8.3.3 PDU session establishment reject

8.3.3.1 Message definition

The PDU SESSION ESTABLISHMENT REJECT message is sent by the SMF to the UE in response to PDU SESSION ESTABLISHMENT REQUEST message and indicates unsuccessful establishment of a PDU session. See table 8.3.3.1.1.

Message type: PDU SESSION ESTABLISHMENT REJECT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	Μ	V	1
	PDU session ID	PDU session identity 9.4	Μ	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION ESTABLISHMENT REJECT message identity	Message type 9.7	Μ	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1
37	Back-off timer value	GPRS timer 3 9.11.2.5	0	TLV	3
F-	Allowed SSC mode	Allowed SSC mode 9.11.4.5	0	TV	1
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
61	5GSM congestion re-attempt indicator	5GSM congestion re-attempt indicator 9.11.4.21	0	TLV	3
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
1D	Re-attempt indicator	Re-attempt indicator 9.11.4.17	0	TLV	3
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538

Table 8.3.3.1.1: PDU SESSION ESTABLISHMENT REJECT message content

8.3.3.2 Back-off timer value

The network may include this IE if the 5GSM cause is not #28 "unknown PDU session type", #39 "reactivation requested", #46 "out of LADN service area", #50 "PDU session type IPv4 only allowed", #51 "PDU session type IPv6 only allowed", #54 "PDU session does not exist", #57 "PDU session type IPv4v6 only allowed", #58 "PDU session type Unstructured only allowed", #61 "PDU session type Ethernet only allowed", #68 "not supported SSC mode", or #86 "UAS services not allowed" to request a minimum time interval before procedure retry is allowed.

8.3.3.3 Allowed SSC mode

This IE is included when the network rejects the PDU SESSION ESTABLISHMENT REQUEST with cause #68 "not supported SSC mode.

8.3.3.4 EAP message

This IE is included when the external DN unsuccessfully performed authentication and authorization of the UE using EAP.

8.3.3.4A 5GSM congestion re-attempt indicator

The network may include this IE only if it includes the Back-off timer value IE and the 5GSM cause value is either #67 "insufficient resources for specific slice and DNN" or #69 "insufficient resources for specific slice".

8.3.3.5 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.3.6 Re-attempt indicator

The network may include this IE if the network includes the Back-off timer value IE and the 5GSM cause value is not #26 "insufficient resources", #28 "unknown PDU session type", #39 "reactivation requested", #46 "out of LADN service area", #54 "PDU session does not exist", #67 "insufficient resources for specific slice and DNN", #68 "not supported SSC mode", or #69 "insufficient resources for specific slice".

8.3.3.7 Service-level-AA container

The network shall include this IE if the service-level authentication and authorization procedure has completed unsuccessfully and the 5GSM cause is #29 "user authentication or authorization failed". The network shall include the service-level-AA response if provided by the DN in the service-level-AA container.

8.3.4 PDU session authentication command

8.3.4.1 Message definition

The PDU SESSION AUTHENTICATION COMMAND message is sent by the SMF to the UE for authentication of the UE establishing the PDU session or of the UE participating in the PDU session. See table 8.3.4.1.1.

Message type: PDU SESSION AUTHENTICATION COMMAND

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION AUTHENTICATION COMMAND message identity	Message type 9.7	М	V	1
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

Table 8.3.4.1.1: PDU SESSION AUTHENTICATION COMMAND message content

8.3.4.2 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

NOTE: How the Extended protocol configuration options IE is used by the network and the UE during the PDU session authentication and authorization procedure is not specified in this release of the specification.

8.3.4.3 Void

8.3.5 PDU session authentication complete

8.3.5.1 Message definition

The PDU SESSION AUTHENTICATION COMPLETE message is sent by the UE to the SMF in response to the PDU SESSION AUTHENTICATION COMMAND message and indicates acceptance of the PDU SESSION AUTHENTICATION COMMAND message. See table 8.3.5.1.1.

Message type: PDU SESSION AUTHENTICATION COMPLETE

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION AUTHENTICATION COMPLETE message identity	Message type 9.7	Μ	V	1
	EAP message	EAP message 9.11.2.2	М	LV-E	6-1502
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

Table 8.3.5.1.1: PDU SESSION AUTHENTICATION COMPLETE message content

8.3.5.2 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

NOTE: How the Extended protocol configuration options IE is used by the network and the UE during the PDU session authentication and authorization procedure is not specified in this release of the specification.

8.3.5.3 Void

8.3.6 PDU session authentication result

8.3.6.1 Message definition

The PDU SESSION AUTHENTICATION RESULT message is sent by the SMF to the UE for indication of successful result of authentication of the UE participating in the PDU session. See table 8.3.6.1.1.

Message type: PDU SESSION AUTHENTICATION RESULT

Significance: dual

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION AUTHENTICATION RESULT message identity	Message type 9.7	М	V	1
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

Table 8.3.6.1.1: PDU SESSION AUTHENTICATION RESULT message content

8.3.6.2 EAP message

This IE shall be included when the external DN performs authentication and authorization of the UE using EAP and it completes successfully.

8.3.6.3 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

NOTE: How the Extended protocol configuration options IE is used by the network and the UE during the PDU session authentication and authorization procedure is not specified in this release of the specification.

8.3.7 PDU session modification request

8.3.7.1 Message definition

The PDU SESSION MODIFICATION REQUEST message is sent by the UE to the SMF to request a modification of a PDU session. See table 8.3.7.1.1.

Message type: PDU SESSION MODIFICATION REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION MODIFICATION REQUEST message identity	Message type 9.7	М	V	1
28	5GSM capability	5GSM capability 9.11.4.1	0	TLV	3-15
59	5GSM cause	5GSM cause 9.11.4.2	0	TV	2
55	Maximum number of supported packet filters	Maximum number of supported packet filters 9.11.4.9	0	TV	3
B-	Always-on PDU session requested	Always-on PDU session requested 9.11.4.4	0	TV	1
13	Integrity protection maximum data rate	Integrity protection maximum data rate 9.11.4.7	0	TV	3
7A	Requested QoS rules	QoS rules 9.11.4.13	0	TLV-E	7-65538
79	Requested QoS flow descriptions	QoS flow descriptions 9.11.4.12	0	TLV-E	6-65538
75	Mapped EPS bearer contexts	Mapped EPS bearer contexts 9.11.4.8	0	TLV-E	7-65538
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
74	Port management information container	Port management information container 9.11.4.27	0	TLV-E	4-65538
66	IP header compression configuration	Header compression configuration 9.11.4.24	0	TLV	5-257
1F	Ethernet header compression configuration	Ethernet header compression configuration 9.11.4.28	0	TLV	3
70	Requested MBS container	Requested MBS container 9.11.4.30	0	TLV-E	8-65538
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538

Table 8.3.7.1.1: PDU SESSION MODIFICATION REQUEST message content

NOTE: It is possible for UEs compliant with version 15.2.1 or earlier versions of this specification to send the Mapped EPS bearer contexts IE with IEI of value "7F" for this message.

8.3.7.2 5GSM capability

This IE is included in the message:

- for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the UE has not previously successfully performed the UE-requested PDU session modification to provide this capability, and:
 - a) if the PDU session is of "IPv4", "IPv6", "IPv4v6" or "Ethernet" PDU session type, and the UE supports reflective QoS; or
 - b) if the PDU session is of "IPv6" or "IPv4v6" PDU session type, and the UE supports multi-homed IPv6 PDU session; or
- 2) if the UE needs to revoke the previously indicated support of reflective QoS.

8.3.7.3 5GSM cause

This IE is included in the message to indicate the reason for the deletion of one or more non-default QoS rules, QoS flow descriptions or mapped EPS bearer contexts.

8.3.7.4 Maximum number of supported packet filters

This IE shall be included in the message for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the UE has not previously successfully performed the UE-requested PDU session modification to provide this capability, the PDU session type is "IPv4", "IPv6", "IPv4v6" or "Ethernet", and the UE can support more than 16 packet filters for this PDU session.

8.3.7.5 Always-on PDU session requested

This IE shall be included in the message for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, the UE has not previously successfully performed the UE-requested PDU session modification to provide this capability, and the UE requests the PDU session to be an always-on PDU session in the 5GS.

8.3.7.6 Integrity protection maximum data rate

This IE shall be included in the message for a PDN connection established when in S1 mode, after an inter-system change from S1 mode to N1 mode, if the UE is a UE operating in single-registration mode in a network supporting N26 interface, and the UE has not previously successfully performed the UE-requested PDU session modification to provide this capability.

8.3.7.7 Requested QoS rules

This IE is included in the message when the UE requests a specific QoS handling.

8.3.7.8 Requested QoS flow descriptions

This IE is included in the message when the UE requests a specific QoS flow descriptions.

8.3.7.9 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.7.10 Mapped EPS bearer contexts

This IE is included when the UE requests to delete one or more mapped EPS bearer contexts.

8.3.7.11 Port management information container

This IE shall be included when the UE needs to convey a port management information container.

8.3.7.12 IP header compression configuration

This IE is included in the message:

- a) if the UE wishes to re-negotiate IP header compression configuration associated to a PDU session and both the UE and the network supports Control plane CIoT 5GS optimization and IP header compression; or
- b) to negotiate IP header compression configuration associated to a PDU session after an inter-system change from S1 mode to N1 mode when both the UE and the network support control plane CIoT 5GS optimization and IP header compression, and the UE is operating in single-registration mode in the network supporting N26 interface.

8.3.7.13 Ethernet header compression configuration

This IE is included in the message:

- a) if the UE wishes to re-negotiate Ethernet header compression configuration associated to a PDU session and both the UE and the network support Control plane CIoT 5GS optimization and Ethernet header compression; or
- b) to negotiate Ethernet header compression configuration associated to a PDU session after an inter-system change from S1 mode to N1 mode.

8.3.7.14 Requested MBS container

This IE is included in the message when the UE requests to join or leave one or more multicast MBS sessions that are associated with the PDU session.

8.3.7.15 Service-level-AA container

This IE shall be included in the message when the UE needs to modify an established PDU session for C2 communication.

8.3.8 PDU session modification reject

8.3.8.1 Message definition

The PDU SESSION MODIFICATION REJECT message is sent by the SMF to the UE to indicate rejection of the PDU SESSION MODIFICATION REQUEST. See table 8.3.8.1.1.

Message type: PDU SESSION MODIFICATION REJECT

Significance: dual

Direction: network to UE

Table 8.3.8.1.1: PDU SESSION MODIFICATION REJECT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	Μ	V	1
	PDU session ID	PDU session identity 9.4	Μ	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION MODIFICATION REJECT message identity	Message type 9.7	Μ	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1
37	Back-off timer value	GPRS timer 3 9.11.2.5	0	TLV	3
61	5GSM congestion re-attempt indicator	5GSM congestion re-attempt indicator 9.11.4.21	0	TLV	3
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
1D	Re-attempt indicator	Re-attempt indicator 9.11.4.17	0	TLV	3

8.3.8.2 Back-off timer value

The network may include this IE if the 5GSM cause is not #26 "insufficient resources", #37 "5GS QoS not accepted", #44 "Semantic errors in packet filter(s)", #45 "Syntactical error in packet filter(s)", #46 "out of LADN service area", #59 "unsupported 5QI value", #67 "insufficient resources for specific slice and DNN", #69 "insufficient resources for

specific slice", #83 "Semantic error in the QoS operation", or #84 "Syntactical error in the QoS operation" to request a minimum time interval before procedure retry is allowed.

8.3.8.2A 5GSM congestion re-attempt indicator

The network may include this IE only if it includes the Back-off timer value IE and the 5GSM cause value is either #67 "insufficient resources for specific slice and DNN" or #69 "insufficient resources for specific slice".

8.3.8.3 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.8.4 Re-attempt indicator

The network may include this IE only if it includes the Back-off timer value IE and the 5GSM cause value is not #26 "insufficient resources", #37 "5GS QoS not accepted", #44 "Semantic errors in packet filter(s)", #45 "Syntactical error in packet filter(s)", #46 "out of LADN service area", #59 "unsupported 5QI value", #67 "insufficient resources for specific slice and DNN", #69 "insufficient resources for specific slice", #83 "Semantic error in the QoS operation", or #84 "Syntactical error in the QoS operation".

8.3.9 PDU session modification command

8.3.9.1 Message definition

The PDU SESSION MODIFICATION COMMAND message is sent by the SMF to the UE to indicate a modification of a PDU session. See table 8.3.9.1.1

Message type: PDU SESSION MODIFICATION COMMAND

Significance: dual
IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION MODIFICATION COMMAND message identity	Message type 9.7	М	V	1
59	5GSM cause	5GSM cause 9.11.4.2	0	TV	2
2A	Session AMBR	Session-AMBR 9.11.4.14	0	TLV	8
56	RQ timer value	GPRS timer 9.11.2.3	0	TV	2
8-	Always-on PDU session indication	Always-on PDU session indication 9.11.4.3	0	TV	1
7A	Authorized QoS rules	QoS rules 9.11.4.13	0	TLV-E	7-65538
75	Mapped EPS bearer contexts	Mapped EPS bearer contexts 9.11.4.8	0	TLV-E	7-65538
79	Authorized QoS flow descriptions	QoS flow descriptions 9.11.4.12	0	TLV-E	6-65538
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
77	ATSSS container	ATSSS container 9.11.4.22	0	TLV-E	3-65538
66	IP header compression configuration	IP header compression configuration 9.11.4.24	0	TLV	5-257
74	Port management information container	Port management information container 9.11.4.27	0	TLV-E	4-65538
1E	Serving PLMN rate control	Serving PLMN rate control 9.11.4.20	0	TLV	4
1F	Ethernet header compression configuration	Ethernet header compression configuration 9.11.4.28	0	TLV	3
71	Received MBS container	Received MBS container 9.11.4.31	0	TLV-E	9-65538
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538

Table 8.3.9.1.1: PDU SESSION MODIFICATION COMMAND message content

NOTE: It is possible for networks compliant with version 15.2.1 or earlier versions of this specification to send the Mapped EPS bearer contexts IE with IEI of value "7F" for this message.

8.3.9.2 5GSM cause

This IE is included when the network performs the PDU session anchor relocation for SSC mode 3.

8.3.9.3 Session-AMBR

This IE is included when the session-AMBR of the PDU session is modified.

8.3.9.4 RQ timer value

This IE is included when the network needs to provide the RQ timer value.

8.3.9.5 Always-on PDU session indication

The network shall include this IE if the network decides to inform the UE whether the PDU session is an always-on PDU session.

8.3.9.6 Authorized QoS rules

This IE is included when the authorized QoS rules of the PDU session are modified.

8.3.9.7 Mapped EPS bearer contexts

This IE is included when interworking with EPS is supported for the PDU session and the mapped EPS bearer contexts is modified.

8.3.9.8 Authorized QoS flow descriptions

This IE is included when the authorized QoS flow descriptions of the PDU session are modified.

8.3.9.9 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.9.10 Void

8.3.9.11 ATSSS container

The IE is included in the message when the network needs to indicate that the ATSSS parameters of the MA PDU session are modified.

8.3.9.12 IP header compression configuration

This IE is included in the message if the network wishes to re-negotiate IP header compression configuration associated to a PDU session and both the UE and the network support Control plane CIoT 5GS optimization and IP header compression.

8.3.9.13 Port management information container

This IE shall be included when the network needs to convey a port management information container.

8.3.9.14 Serving PLMN rate control

This IE shall be included when the network needs to indicate the maximum uplink control plane user data the UE is allowed to send per 6 minute interval.

8.3.9.15 Ethernet header compression configuration

This IE is included in the message if the network wishes to re-negotiate Ethernet header compression configuration associated to a PDU session and both the UE and the network support Control plane CIoT 5GS optimization and Ethernet header compression.

8.3.9.16 Received MBS container

The network shall include this IE if:

- the UE has requested to join or leave one or more multicast MBS sessions;
- the network wants to remove joined UE from one or more multicast MBS sessions; or

- the network wants to update the MBS service area of multicast MBS session that the UE has joined; or
- the network wants to update the MBS security information of multicast MBS session that the UE has joined.

8.3.9.17 Service-level-AA container

The SMF shall include the service-level-AA container IE if the service-level authentication and authorization procedure for re-authentication purpose is completed successfully by the external DN.

8.3.10 PDU session modification complete

8.3.10.1 Message definition

The PDU SESSION MODIFICATION COMPLETE message is sent by the UE to the SMF in response to the PDU SESSION MODIFICATION COMMAND message and indicates an acceptance of the PDU SESSION MODIFICATION COMMAND message. See table 8.3.10.1.1.

Message type: PDU SESSION MODIFICATION COMPLETE

Significance: dual

Direction: UE to network

Table 8.3.10.1.1: PDU SESSION MODIFICATION COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION MODIFICATION COMPLETE message identity	Message type 9.7	М	V	1
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
74	Port management information container	Port management information container 9.11.4.27	0	TLV-E	4-65538

NOTE: It is possible for UEs compliant with version 15.3.0 of this specification to include the 5GSM cause IE with IEI 59 in the PDU SESSION MODIFICATION COMPLETE message, and therefore the IEI 59 cannot be used for other optional IEs other than the 5GSM cause IE for future extensions of the PDU SESSION MODIFICATION COMPLETE message.

8.3.10.2 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.10.3 Port management information container

This IE shall be included when the UE needs to convey a port management information container.

8.3.11 PDU session modification command reject

8.3.11.1 Message definition

The PDU SESSION MODIFICATION COMMAND REJECT message is sent by the UE to the SMF to indicate rejection of the PDU SESSION MODIFICATION COMMAND message. See table 8.3.11.1.1

Message type: PDU SESSION MODIFICATION COMMAND REJECT

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION MODIFICATION COMMAND REJECT message identity	Message type 9.7	М	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

Table 8.3.11.1.1: PDU SESSION MODIFICATION COMMAND REJECT message content

8.3.11.2 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.12 PDU session release request

8.3.12.1 Message definition

The PDU SESSION RELEASE REQUEST message is sent by the UE to the SMF to request a release of a PDU session. See table 8.3.12.1.1.

Message type: PDU SESSION RELEASE REQUEST

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	Μ	V	1
	PDU session ID	PDU session identity 9.4	Μ	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION RELEASE REQUEST message identity	Message type 9.7	Μ	V	1
59	5GSM cause	5GSM cause 9.11.4.2	0	ΤV	2
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

Table 8.3.12.1.1: PDU SESSION RELEASE REQUEST message content

8.3.12.2 5GSM cause

This IE is included in the message to indicate the reason for releasing the PDU session.

8.3.12.3 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.13 PDU session release reject

8.3.13.1 Message definition

The PDU SESSION RELEASE REJECT message is sent by the SMF to the UE to indicate rejection of request a release of a PDU session. See table 8.3.13.1.1.

Message type: PDU SESSION RELEASE REJECT

Significance: dual

Direction: network to UE

Table 8.3.13.1.1: PDU SESSION RELEASE REJECT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION RELEASE REJECT message identity	Message type 9.7	М	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	Ō	TLV-E	4-65538

8.3.13.2 Extended protocol configuration options

This IE is included in the message when the network needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.14 PDU session release command

8.3.14.1 Message definition

The PDU SESSION RELEASE COMMAND message is sent by the SMF to the UE to indicate a release of a PDU session. See table 8.3.14.1.1.

Message type: PDU SESSION RELEASE COMMAND

Significance: dual

Direction: network to UE

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION RELEASE COMMAND message identity	Message type 9.7	М	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1
37	Back-off timer value	GPRS timer 3 9.11.2.5	0	TLV	3
78	EAP message	EAP message 9.11.2.2	0	TLV-E	7-1503
61	5GSM congestion re-attempt indicator	5GSM congestion re-attempt indicator 9.11.4.21	0	TLV	3
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538
D-	Access type	Access type 9.11.2.1A	0	TV	1
72	Service-level-AA container	Service-level-AA container 9.11.2.10	0	TLV-E	4-65538

Table 8.3.14.1.1: PDU SESSION RELEASE COMMAND message content

8.3.14.2 Back-off timer value

The network may include this IE to request a minimum time interval before procedure retry is allowed.

8.3.14.3 EAP message

This IE is included when the external DN performs re-authentication and re-authorization of the UE using EAP and it completes unsuccessfully.

8.3.14.4 Extended protocol configuration options

This IE is included in the message when the network wants to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the UE.

8.3.14.5 5GSM congestion re-attempt indicator

The network may include this IE only if it includes the Back-off timer value IE and the 5GSM cause value is either #67 "insufficient resources for specific slice and DNN" or #69 "insufficient resources for specific slice".

8.3.14.6 Access type

This IE is included in the message when the network releases user-plane reources of an MA PDU session specifically over either 3GPP access or non-3GPP access.

8.3.14.7 Service-level-AA container

The SMF shall include the service-level-AA container IE if the service-level authentication and authorization procedure for re-authentication purpose is completed unsuccessfully by the external DN.

8.3.15 PDU session release complete

8.3.15.1 Message definition

The PDU SESSION RELEASE COMPLETE message is sent by the UE to the SMF in response to the PDU SESSION RELEASE COMMAND message and indicates an acceptance of a release of the PDU session. See table 8.3.15.1.1.

Message type: PDU SESSION RELEASE COMPLETE

Significance: dual

Direction: UE to network

Table 8.3.15.1.1: PDU SESSION RELEASE COMPLETE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	PDU SESSION RELEASE COMPLETE message identity	Message type 9.7	Μ	V	1
59	5GSM cause	5GSM cause 9.11.4.2	0	TV	2
7B	Extended protocol configuration options	Extended protocol configuration options 9.11.4.6	0	TLV-E	4-65538

8.3.15.2 5GSM cause

This IE is included in the message when the UE needs to indicate to the network that an error encountered with a mandatory information element in the PDU SESSION RELEASE COMMAND message.

8.3.15.3 Extended protocol configuration options

This IE is included in the message when the UE needs to transmit (protocol) data (e.g. configuration parameters, error codes or messages/events) to the network.

8.3.16 5GSM status

8.3.16.1 Message definition

The 5GSM STATUS message is sent by the SMF or the UE to pass information on the status of the indicated PDU session and report certain error conditions. See table 8.3.16.1.1.

Message type: 5GSM STATUS

Significance: dual

Direction: both

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	5GSM STATUS message identity	Message type 9.7	М	V	1
	5GSM cause	5GSM cause 9.11.4.2	М	V	1

Table 8.3.16.1.1: 5GSM STATUS message content

8.3.17 Service-level authentication command

8.3.17.1 Message definition

The SERVICE-LEVEL AUTHENTICATION COMMAND message is sent by the SMF to the UE for service-level authentication and authorization procedure. See table 8.3.17.1.1.

Message type: SERVICE-LEVEL AUTHENTICATION COMMAND

Significance: dual

Direction: network to UE

Table 8.3.17.1.1: SERVICE-LEVEL AUTHENTICATION COMMAND message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	SERVICE-LEVEL AUTHENTICATION COMMAND message identity	Message type 9.7	М	V	1
	Service-level-AA container	Service-level-AA container 9.11.2.10	М	LV-E	5-n

8.3.18 Service-level authentication complete

8.3.18.1 Message definition

The SERVICE-LEVEL AUTHENTICATION COMPLETE message is sent by the UE to the SMF in response to the SERVICE-LEVEL AUTHENTICATION COMMAND message and indicates acceptance of the SERVICE-LEVEL AUTHENTICATION COMMAND message. See table 8.3.18.1.1.

Message type: SERVICE-LEVEL AUTHENTICATION COMPLETE

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	SERVICE-LEVEL AUTHENTICATION COMPLETE message identity	Message type 9.7	Μ	V	1
	Service-level-AA container	Service-level-AA container 9.11.2.10	М	LV-E	5-n

 Table 8.3.18.1.1: SERVICE-LEVEL AUTHENTICATION COMPLETE message content

8.3.19 Remote UE report

8.3.19.1 Message definition

The REMOTE UE REPORT message is sent by the UE to the network to report connection or disconnection of 5G ProSe remote UE(s). See table 8.3.19.1.

Message type: REMOTE UE REPORT

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator 9.2	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	Remote UE report message identity	Message type 9.7	М	V	1
76	Remote UE context connected	Remote UE context list 9.11.4.29	0	TLV-E	16-65538
70	Remote UE context disconnected	Remote UE context list 9.11.4.29	0	TLV-E	16-65538

Table 8.3.19.1: REMOTE UE REPORT message content

8.3.19.2 Remote UE context connected

This IE is included in the message by the UE acting as a 5G ProSe layer-3 UE-to-network UE relay to provide the network with newly connected 5G ProSe remote UE information as specified in 3GPP TS 23.304 [6E].

8.3.19.3 Remote UE context disconnected

This IE is included in the message by the UE acting as a 5G ProSe layer-3 UE-to-network UE relay to provide the network with disconnected 5G ProSe remote UE information as specified in 3GPP TS 23.304 [6E].

8.3.20 Remote UE report response

8.3.20.1 Message definition

The REMOTE UE REPORT RESPONSE message is sent by the network to the UE to acknowledge receipt of a remote UE report message. See table 8.3.20.1.

Message type: REMOTE UE REPORT RESPONSE

Significance: dual

Direction: network to UE

Table 8.3.20.1: REMOTE UE REPORT RESPONSE message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	Extended protocol discriminator	Extended protocol discriminator	М	V	1
	PDU session ID	PDU session identity 9.4	М	V	1
	PTI	Procedure transaction identity 9.6	М	V	1
	Remote UE report response message identity	Message type 9.7	М	V	1

- 8.3.20.2 Void
- 8.3.20.3 Void
- 8.3.20.4 Void

9 General message format and information elements coding

9.1 Overview

9.1.1 NAS message format

Within the protocols defined in the present document, every 5GS NAS message is a standard L3 message as defined in 3GPP TS 24.007 [11]. This means that the message consists of the following parts:

- 1) if the message is a plain 5GS NAS message:
 - a) extended protocol discriminator;
 - b) security header type associated with a half spare octet or PDU session identity;
 - c) procedure transaction identity;
 - d) message type;
 - e) other information elements, as required.
- 2) if the message is a security protected 5GS NAS message:
 - a) extended protocol discriminator;

- b) security header type associated with a half spare octet;
- c) message authentication code;
- d) sequence number;
- e) plain 5GS NAS message, as defined in item 1

The organization of a plain 5GS NAS message is illustrated in the example shown in figure 9.1.1.1.



Figure 9.1.1.1: General message organization example for a plain 5GS NAS message

The PDU session identity and the procedure transaction identity are only used in messages with extended protocol discriminator 5GS session management. Octet 2a with the procedure transaction identity shall only be included in these messages.

The organization of a security protected 5GS NAS message is illustrated in the example shown in figure 9.1.1.2.



Figure 9.1.1.2: General message organization example for a security protected 5GS NAS message

Unless specified otherwise in the message descriptions of clause 8 and annex D, a particular information element shall not be present more than once in a given message.

9.1.2 Field format and mapping

When a field is contained within a single octet, the lowest numbered bit of the field represents the least significant bit.

When a field extends over more than one octet, the order of bit values progressively decreases as the octet number increases. In that part of the field contained in a given octet, the lowest numbered bit represents the least significant bit. The most significant bit of the field is represented by the highest numbered bit of the lowest numbered octet of the field. The least significant bit of the field is represented by the lowest numbered bit of the highest numbered octet of the field.

For example, a bit number can be identified as a couple (o, b) where o is the octet number and b is the relative bit number within the octet. Figure 9.1.2.1 illustrates a field that spans from bit (1, 3) to bit (2, 7). The most significant bit of the field is mapped on bit (1, 3) and the least significant bit is mapped on bit (2, 7).





9.2 Extended protocol discriminator

Bits 1 to 8 of the first octet of every 5GS NAS message contain the Extended protocol discriminator (EPD) IE. The EPD and its use are defined in 3GPP TS 24.007 [11]. The extended protocol discriminator in the header (see 3GPP TS 24.007 [11]) of a security protected 5GS NAS message is encoded as "5GS mobility management messages".

9.3 Security header type

Bits 1 to 4 of the second octet of every 5GMM message contain the Security header type IE. This IE includes control information related to the security protection of a 5GMM message. The total size of the Security header type IE is 4 bits.

The Security header type IE can take the values shown in table 9.3.1.

Table 9.3.1: Security header type

Se	Security header type (octet 1)											
Bit	Bits											
4	4 3 2 1											
0	0	0	0	Plain 5GS NAS message, not security protected								
	Security protected 5GS NAS message:											
0	0 0 0 1 Integrity protected											
0	0 0 1 0 Integrity protected and ciphered											
0	0 0 1 1 Integrity protected with new 5G NAS security context (NOTE 1)											
0	0 1 0 0 Integrity protected and ciphered with new 5G NAS security context											
				(NOTE 2)								
All	All other values are reserved.											
NC	NOTE 1: This codepoint may be used only for a SECURITY MODE COMMAND											
			me	essage.								
NC	DTE	2:	Th	is codepoint may be used only for a SECURITY MODE COMPLETE								
			me	essage.								

A 5GMM message received with the security header type encoded as 0000 shall be treated as not security protected, plain 5GS NAS message. A protocol entity sending a not security protected 5GMM message shall send the message as plain 5GS NAS message and encode the security header type as 0000.

9.4 PDU session identity

Bits 1 to 8 of the second octet of every 5GSM message contain the PDU session identity IE. The PDU session identity and its use to identify a message flow are defined in 3GPP TS 24.007 [11].

9.5 Spare half octet

This element is used in the description of 5GMM and 5GSM messages when an odd number of half octet type 1 information elements are used. This element is filled with spare bits set to zero and is placed in bits 5 to 8 of the octet unless otherwise specified.

9.6 Procedure transaction identity

Bits 1 to 8 of the third octet of every 5GSM message contain the procedure transaction identity. Bits 1 to 8 of the first octet of every UE policy delivery message contain the procedure transaction identity. The procedure transaction identity and its use are defined in 3GPP TS 24.007 [11].

9.7 Message type

The Message type IE and its use are defined in 3GPP TS 24.007 [11]. Tables 9.7.1 and 9.7.2 define the value part of the message type IE used in the 5GS mobility management protocol and 5GS session management protocol.

Table 9.7.1: Message typ	es for 5GS mobilit	y management
--------------------------	--------------------	--------------

Rit	c							
8 8	5 7	6	5	4	3	2	1	
0	1	-	-	-	-	-	-	5GS mobility management messages
0 0 0 0 0 0 0	1 1 1 1 1 1	0 0 0 0 0 0	0 0 0 0 0 0 0	0 0 0 0 0 0	0 0 1 1 1 0	0 1 0 1 1 0	1 0 1 0 1 0 1	Registration request Registration accept Registration complete Registration reject Deregistration request (UE originating) Deregistration accept (UE originating) Deregistration request (UE terminated) Deregistration accept (UE terminated)
0 0 0 0	1 1 1 1	0 0 0 0	0 0 0 0	1 1 1 1	1 1 1 1	0 0 1 1	0 1 0 1	Service request Service reject Service accept Control plane service request
0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0 0 0 0 0 1 1 1 1 1 1 1	$\begin{array}{c} 0 \\ 0 \\ 1 \\ 1 \\ 1 \\ 0 \\ 0 \\ 0 \\ 1 \\ 1 \\$	$\begin{array}{c} 0 \\ 0 \\ 1 \\ 0 \\ 0 \\ 1 \\ 1 \\ 0 \\ 0 \\ 1 \\ 1$	0 1 0 1 0 1 0 1 0 1 0 1 0 1 0	Network slice-specific authentication command Network slice-specific authentication complete Network slice-specific authentication result Configuration update command Configuration update complete Authentication request Authentication response Authentication reject Authentication failure Authentication failure Authentication result Identity request Identity response Security mode command Security mode reject
0 0 0 0	1 1 1 1	1 1 1 1	0 0 0 0	0 0 0 1	1 1 1 1 0	0 0 1 1 0	0 1 0 1 0	5GMM status Notification Notification response UL NAS transport DL NAS transport
0 0 0 0	1 1 1 1	1 1 1 1	0 0 0 0	1 1 1 1	0 0 1 1	0 1 1 0 0	1 0 1 0 1	Relay key request Relay key accept Relay key reject Relay authentication request Relay authentication response

-							
.s	~	-		•	~		
1	6	5	4	3	2	1	
1							ECS cossion management messages
I	-	-	-	-	-	-	565 session management messages
1	0	0	0	0	0	1	PDU session establishment request
1	Ō	Ō	Ō	Ō	1	0	PDU session establishment accept
1	0	0	0	0	1	1	PDU session establishment reject
-	-	-	-	-	-	-	
1	0	0	0	1	0	1	PDU session authentication command
1	0	0	0	1	1	0	PDU session authentication complete
1	0	0	0	1	1	1	PDU session authentication result
1	0	0	1	0	0	1	PDU session modification request
1	0	0	1	0	1	0	PDU session modification reject
1	0	0	1	0	1	1	PDU session modification command
1	0	0	1	1	0	0	PDU session modification complete
1	0	0	1	1	0	1	PDU session modification command reject
1	0	1	0	0	0	1	PDU session release request
1	0	1	0	0	1	0	PDU session release reject
1	0	1	0	0	1	1	PDU session release command
1	0	1	0	1	0	0	PDU session release complete
4	^	4	0	4	4	0	ECSM status
I	0	1	0	1	1	0	565M status
1	0	1	1	0	0	0	Service-level authentication command
1	õ	1	1	ñ	ñ	1	Service-level authentication complete
	U	•	•	Ŭ	U		
1	0	1	1	0	1	0	Remote UE report
1	0	1	1	0	1	1	Remote UE report response
	-			-			
	ts 7 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	is 7 6 1 - 1 0	ts 7 6 5 1 - - 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 0 1	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	is 7 6 5 4 3 2 1 - - - - - - 1 0 0 0 0 0 1 1 0 0 0 0 1 1 1 0 0 0 1 1 1 1 1 0 0 0 1 <	Image: s 7 6 5 4 3 2 1 1 - - - - - - - 1 0 0 0 0 1 0 1 1 0 0 0 0 1 1 0 1 0 0 0 1 1 1 1 1 0 0 0 1 0 1 1 1 0 0 1 1 1 1 1 1 0 0 1 1 1 1 1 1 0 0 1 0 1 1 1 1 0 1 0 1 1 1 1 1 0 1 0 1 1 1 1 1 1 0 1 0 1 1 1 1 1 1 0 1 0 1 1 0

Table 9.7.2: Message types for 5GS session management

9.8 Message authentication code

The message authentication code (MAC) information element contains the integrity protection information for the message. The MAC IE shall be included in the SECURITY PROTECTED 5GS NAS MESSAGE message if a valid 5G NAS security context exists and security functions are started.

The message authentication code (MAC) is also included in the Intra N1 mode NAS transparent container IE and in the S1 mode to N1 mode NAS transparent container IE.

The usage of MAC is specified in subclause 4.4.3.3.

9.9 Plain 5GS NAS message

This IE includes a complete plain 5GS NAS message as specified in subclauses 8.2 and 8.3. The SECURITY PROTECTED 5GS NAS MESSAGE (see subclause 8.2.28) includes a complete plain 5GS NAS message as specified in subclauses 8.2. The SECURITY PROTECTED 5GS NAS MESSAGE message (see subclause 8.2.28) is not plain 5GS NAS messages and shall not be included in this IE.

9.10 Sequence number

This IE includes the NAS message sequence number (SN) which consists of the eight least significant bits of the NAS COUNT for a SECURITY PROTECTED 5GS NAS MESSAGE message.

The NAS message sequence number (SN) with the eight least significant bits of the NAS COUNT is also included in the Intra N1 mode NAS transparent container IE and in the N1 mode to S1 mode NAS transparent container IE.

The usage of SN is specified in subclause 4.4.3.

9.11 Other information elements

9.11.1 General

The different formats (V, LV, T, TV, TLV, LV-E, TLV-E) and the five categories of information elements (type 1, 2, 3, 4 and 6) are defined in 3GPP TS 24.007 [11].

The first octet of an information element in the non-imperative part contains the IEI of the information element. If this octet does not correspond to an IEI known in the message, the receiver shall determine whether this IE is of type 1 or 2 (i.e. it is an information element of one octet length) or an IE of type 4 (i.e. that the next octet is the length indicator indicating the length of the remaining of the information element) (see 3GPP TS 24.007 [11]).

This allows the receiver to jump over unknown information elements and to analyse any following information elements of a particular message.

The definitions of information elements which are:

- a) common for the 5GMM and 5GSM protocols;
- b) used by access stratum protocols; or
- c) sent to upper layers

are described in subclause 9.11.2.

The information elements of the 5GMM or 5GSM protocols can be defined by reference to an appropriate specification which provides the definition of the information element, e.g., "see subclause 10.5.6.3A in 3GPP TS 24.008 [12]".

9.11.2 Common information elements

9.11.2.1 Additional information

The purpose of the Additional information information element is to provide additional information to upper layers in relation to the NAS transport mechanism.

The Additional information information element is coded as shown in figure 9.11.2.1.1 and table 9.11.2.1.1.

The Additional information is a type 4 information element with a minimum length of 3 octets.

8	7	6	5	4	3	2	1				
Additional information IEI											
Additional information length											
Additional information value											

Figure 9.11.2.1.1: Additional information information element

Table 9.11.2.1.1 :	Additional	information	information	element
--------------------	------------	-------------	-------------	---------

Additional information value (octet 3 to octet n) The coding of the additional information value is dependent on the LCS application.

9.11.2.1A Access type

The purpose of the access type information element is to indicate the access type over which the signalling or user data is pending to be sent to the UE.

The access type is a type 1 information element.

The access type information element is coded as shown in figure 9.11.2.1A.1 and table 9.11.2.1A.1.

Access type 0 Access type octet	8	7	6	5	4	3	2	1	
IEI spare		Acces	s type		()	Acces	s type	octet 1
			El		spa	are			

Figure 9.11.2.1A.1: Access type information element

Ac	Access type value (octet 1, bit 1 to bit 2)										
Bit	s										
2	1										
0	1	3GPP access									
1	0	Non-3GPP access									
All	All other values are reserved.										

9.11.2.1B DNN

The purpose of the DNN information element is to identify the data network.

The DNN information element is coded as shown in figure 9.11.2.1B.1.

The DNN is a type 4 information element with a minimum length of 3 octets and a maximum length of 102 octets.



A DNN value field contains an APN as defined in 3GPP TS 23.003 [4].

9.11.2.2 EAP message

The purpose of the EAP message information element is to transport an EAP message as specified in IETF RFC 3748 [34].

The EAP message information element is coded as shown in figure 9.11.2.2.1 and table 9.11.2.2.1.

The EAP message is a type 6 information element with minimum length of 7 octets and maximum length of 1503 octets.



Figure 9.11.2.2.1: EAP message information element

Table 9.11.2.2.1: EAP message information element

EAP message (octet 4 to n) An EAP message as specified in IETF RFC 3748 [34].

9.11.2.3 GPRS timer

See subclause 10.5.7.3 in 3GPP TS 24.008 [12].

9.11.2.4 GPRS timer 2

See subclause 10.5.7.4 in 3GPP TS 24.008 [12].

9.11.2.5 GPRS timer 3

See subclause 10.5.7.4a in 3GPP TS 24.008 [12].

9.11.2.6 Intra N1 mode NAS transparent container

The purpose of the Intra N1 mode NAS transparent container information element is to provide the UE with parameters that enable the UE to handle the 5G NAS security context after N1 mode to N1 mode handover.

The Intra N1 mode NAS transparent container information element is coded as shown in figure 9.11.2.6.1 and table 9.11.2.6.1.

The Intra N1 mode NAS transparent container is a type 4 information element with a length of 9 octets.

The value part of the Intra N1 mode NAS transparent container information element is included in specific information elements within some RRC messages sent to the UE.

NOTE: For these cases the coding of the information element identifier and length information of RRC is defined in 3GPP TS 38.331 [30].

	1	3 2	3	4	5	6	7	8					
octet 1	Intra N1 mode NAS transparent container IEI												
octet 2	Length of Intra N1 mode NAS transparent container contents												
octet 3													
	Message authentication code												
octet 6													
octet 7	Type of ciphering algorithm Type of integrity protection algorithm												
	in 5G	Cey set identifier	Ke	TSC	KACF	0	0	0					
octet 8							Spare						
octet 9	Sequence number												

Figure 9.11.2.6.1: Intra N1 mode NAS transparent container information element

Table 9.11.2.6.1: Intra N1 mode NAS transparent container information element

Message authentication code (octet 3 to 6)
This field is coded as the Message authentication code information element (see subclause 9.8).
Type of integrity protection algorithm (octet 7, bit 1 to 4) and type of ciphering algorithm (octet 7, bit 5 to 8)
These fields are coded as the type of integrity protection algorithm and type of ciphering algorithm in the NAS security algorithms information element (see subclause 9.11.3.34).
K_AMF_change_flag (KACF) (octet 8, bit 5) Bit 5
 a new K_{AMF} has not been calculated by the network a new K_{AMF} has been calculated by the network
Key set identifier in 5G (octet 8, bit 1 to 3) and Type of security context flag (TSC) (octet 8, bit 4)
These fields are coded as the NAS key set identifier and type of security context flag in the NAS key set identifier information element (see subclause 9.11.3.32).
Sequence number (octet 9)
This field is coded as the Sequence number information element (see subclause 9.10)

9.11.2.7 N1 mode to S1 mode NAS transparent container

The purpose of the N1 mode to S1 mode NAS transparent container information element is to provide the UE with information that enables the UE to create a mapped EPS security context.

The N1 mode to S1 mode NAS transparent container information element is coded as shown in figure 9.11.2.7.1 and table 9.11.2.7.1.

The N1 mode to S1 mode NAS transparent container is a type 3 information element with a length of 2 octets.

The value part of the N1 mode to S1 mode NAS transparent container information element is included in specific information elements within some RRC messages sent to the UE; see 3GPP TS 38.331 [30]. For these cases the coding of the information element identifier and length information is defined in 3GPP TS 38.331 [30].

8	7	6	5	4	3	2	1				
N1 mode to S1 mode NAS transparent container IEI											
			Sequenc	e number				octet 2			

Figure 9.11.2.7.1: N1 mode to S1 mode NAS transparent container information element

Table 9.11.2.7.1: N1 mode to S1 mode NAS transparent container information element

Sequence number (octet 2)

This field is coded as the Sequence number information element (see subclause 9.10).

9.11.2.8 S-NSSAI

The purpose of the S-NSSAI information element is to identify a network slice.

The S-NSSAI information element is coded as shown in figure 9.11.2.8.1 and table 9.11.2.8.1.

The S-NSSAI is a type 4 information element with a minimum length of 3 octets and a maximum length of 10 octets.

8	7	6	5	4	3	2	1					
	S-NSSAI IEI											
		Leng	th of S-N	ISSAI cor	ntents			octet 2				
			S	ST				octet 3				
								octet 4*				
			S	SD								
								octet 6*				
	Mapped HPLMN SST											
		Ν	Napped H	HPLMN S	D							
								octet 10*				



Fable 9.11.2.8.1	S-NSSA	information	element
-------------------------	--------	-------------	---------

Length of S-NSSAI contents (octet 2) This field indicates the length of the included S-NSSAI contents, and it can have the following values. Depending on the value of the length field the following S-NSSAI contents are included: Bits 8 7 6 5 4 3 2 1 0 0 0 0 0 0 0 1 SST 0 0 0 0 0 0 1 0 SST and mapped HPLMN SST 0 0 0 0 0 1 0 0 SST and SD SST, SD and mapped HPLMN SST 0 0 0 0 0 1 0 1 0 0 0 0 1 0 0 0 SST, SD, mapped HPLMN SST and mapped HPLMN SD All other values are reserved. Slice/service type (SST) (octet 3) This field contains the 8 bit SST value. The coding of the SST value part is defined in 3GPP TS 23.003 [4]. If this IE is included during the network slice-specific authentication and authorization procedure, this field contains the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPLMN or the RSNPN. Slice differentiator (SD) (octet 4 to octet 6) This field contains the 24 bit SD value. The coding of the SD value part is defined in 3GPP TS 23.003 [4]. If this IE is included during the network slice-specific authentication and authorization procedure, this field contains the 24 bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN or the RSNPN. If the SST encoded in octet 3 is not associated with a valid SD value, and the sender needs to include a mapped HPLMN SST (octet 7) and a mapped HPLMN SD (octets 8 to 10), then the sender shall set the SD value (octets 4 to 6) to "no SD value associated with the SST". mapped HPLMN Slice/service type (SST) (octet 7) This field contains the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SST value is mapped. The coding of the SST value part is defined in 3GPP TS 23.003 [4]. mapped HPLMN Slice differentiator (SD) (octet 8 to octet 10) This field contains the 24 bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SD value is mapped. The coding of the SD value part is defined in 3GPP TS 23.003 [4]. NOTE 1: Octet 3 shall always be included. NOTE 2: If the octet 4 is included, then octet 5 and octet 6 shall be included. NOTE 3: If the octet 7 is included, then octets 8, 9, and 10 may be included. NOTE 4: If the octet 8 is included, then octet 9 and octet 10 shall be included. NOTE 5: If only HPLMN S-NSSAI or RSNPN S-NSSAI is included, then octets 7 to 10 shall not be included.

9.11.2.9 S1 mode to N1 mode NAS transparent container

The purpose of the S1 mode to N1 mode NAS transparent container information element is to provide the UE with parameters that enable the UE to create a mapped 5G NAS security context and take this context into use after intersystem change to N1 mode in 5GMM-CONNECTED mode.

The S1 mode to N1 mode NAS transparent container information element is coded as shown in figure 9.11.2.9.1 and table 9.11.2.9.1.

The S1 mode to N1 mode NAS transparent container is a type 4 information element with a length of 10 octets.

The value part of the S1 mode to N1 mode NAS transparent container information element is included in specific information elements within some RRC messages sent to the UE.

NOTE: For these cases the coding of the information element identifier and length information of RRC is defined in 3GPP TS 38.331 [30].

8	7	6	5	4	3	2	1	_				
	S1 mode to N1 mode NAS transparent container IEI											
Lengt	Length of S1 mode to N1 mode NAS transparent container contents											
		Messa	age authe	entication	code							
								octet 6				
Type	Type of ciphering algorithm Type of integrity protection											
					algor	ithm		octet 7				
0		NCC		TSC	Kev set	in 5G						
Spare					- ,			octet 8				
0	0	0	0	0	0	0	0					
Spare	Spare	Spare	Spare	Spare	Spare	Spare	Spare	octet 9				
0	0	0	0	0	0	0	0					
Spare	Spare	Spare	Spare	Spare	Spare	Spare	Spare	octet 10				

Figure 9.11.2.9.1: S1 mode to N1 mode NAS transparent container information element

Table 9.11.2.9.1: S1 mode to N1 mode NAS transparent container information element

NOTE: In earlier versions of this protocol, octets 9 and 10 can have any value. In this version of the protocol, octets 9 and 10 can always be ignored by the UE.								
Octets 9 and 10 are spare and shall be coded as zero.								
These fields are coded as the NAS key set identifier and type of security context flag in the NAS key set identifier information element (see subclause 9.11.3.32).								
Key set identifier in 5G (octet 8, bit 1 to 3) and type of security context flag (TSC) (octet 8, bit 4)								
This field contains the 3 bit Next hop chaining counter (see 3GPP TS 33.501 [24])								
NCC (octet 8, bits 5 to 7)								
These fields are coded as the type of integrity protection algorithm and type of ciphering algorithm in the NAS security algorithms information element (see subclause 9.11.3.34).								
Type of integrity protection algorithm (octet 7, bit 1 to 4) and type of ciphering algorithm (octet 7, bit 5 to 8)								
This field is coded as the Message authentication code information element (see subclause 9.8).								
Message authentication code (octet 3 to 6)								

9.11.2.10 Service-level-AA container

The purpose of the Service-level-AA container information element is to transfer upper layer information for authentication and authorization between the UE and the network.

The Service-level-AA container information element is coded as shown in figure 9.11.2.10.1, figure 9.11.2.10.2, figure 9.11.2.10.3, figure 9.11.2.10.4 and table 9.11.2.10.1.

The Service-level-AA container information element is a type 6 information element with a minimum length of 4 octets and a maximum length of 65538 octets.







Figure 9.11.2.10.2: Service-level-AA container contents















Figure 9.11.2.10.6: Service-level-AA parameter (when the type of service-level-AA parameter field contains an IEI of a type 1 information element as specified in 3GPP TS 24.007 [11])

Table 9.11.2.10.1: Service-level-AA container information element

Service-lev	vel-AA container conte	nts (octet 4 to octet n); max value of 65535 octets								
The error I and 7.7.1 s container o	The error handlings for service-level-AA parameters specified in subclauses 7.6.1, 7.6.3 and 7.7.1 shall apply to the service-level-AA parameters included in the service-level-AA container contents.									
Service-lev Type of se This field o	vel-AA parameters rvice-level-AA parame contains the IEI of the s	ter (octet xi +1) service-level-AA parameter.								
Length of s This field in	service-level-AA paran ndicates binary coded	neter length of the value of the service-level-AA parameter.								
Value of se This field o referred in	ervice-level-AA parame contains the value of th formation element bas	eter le service-level-AA parameter with the value part of the ed on following service-level-AA parameter reference.								
The receiv parameter	ring entity shall ignore s field containing an unl	service-level-AA parameter with type of service-level-AA known IEI.								
IEI (hexadec imal)	Service-level-AA parameter name	Service-level-AA parameter reference								
10	Service-level device	Service-level device ID (see subclause 9.11.2.11)								
20	Service-level-AA	Service-level-AA server address (see subclause 9.11.2.12)								
30	Service-level-AA	Service-level-AA response (see subclause 9.11.2.14)								
40	Service-level-AA	Service-level-AA payload type (see subclause 9.11.2.15)								
70	Service-level-AA	Service-level-AA payload (see subclause 9.11.2.13)								
A-	Service-level-AA	Service-level-AA pending indication (see subclause								
50	Service-level-AA service status indication	Service-level-AA service status indication (see subclause 9.11.2.18)								
NOTE:	A service-level-AA pay	yload type is always followed by the associated service-								
	level-AA payload as sl	hown in figure 9.11.2.10.5.								

9.11.2.11 Service-level device ID

The purpose of the Service-level device ID information element is to carry the necessary identity for authentication and authorization by the external DN.

The Service-level device ID information element is coded as shown in figure 9.11.2.11.1 and table 9.11.2.11.1.

The Service-level device ID information element is a type 4 information element with minimum length of 3 octets and maximum length of 257 octets.

8	7	6	5	4	3	2	1				
Service-level device ID IEI											
	Service-level device ID length										
	Service-level device ID										

Figure 9.11.2.11.1: Service-level device ID information element

Table 9.11.2.11.1: Service-level device ID information element

Service-level device ID (octet 3 to octet y)	
A service-level device ID encoded as UTF-8 string.	

9.11.2.12 Service-level-AA server address

The purpose of the Service-level-AA server address information element is to carry the address of the service level authentication and authorization server.

The Service-level-AA server address information element is coded as shown in figure 9.11.2.12.1 and table 9.11.2.12.1.

The Service-level-AA server address information element is a type 4 information element with minimum length of 4 octets and maximum length of 258 octets.

8	7	6	5	4	3	2	1					
	Service-level-AA server address IEI o											
Service-level-AA server address length												
Service-level-AA server address type												
		Service	-level-AA	server	address			octets 4-z				

Figure 9.11.2.12.1: Service-level-AA server address information element

Table 9.11.2.12.1: Service-level-AA	server address	information	element
-------------------------------------	----------------	-------------	---------

Ser	vice-	leve	el-AA	ser	ver a	ddre	ess ty	pe (octet 3):	
	' -	~	-		•	~			
ð	1	0	Э	4	3	2	1		
0	0	0	0	0	0	0	1	IPv4	
0	0	0	0	0	0	1	0	IPv6	
0	0	0	0	0	0	1	1	IPv4v6	
0	0	0	0	0	1	0	0	FQDN	
All (other	val	ues	are s	spare	e. If r	eceiv	ed they shall be ignored.	
lf th serv	ie se ver a	rvice ddre	e-lev ess f	el-A	A sei conta	rver ains	addre an IP [,]	ess type indicates IPv4, then the service-level-AA v4 address in octet 4 to octet 7.	
lf th serv	ie se ver a	rvice ddre	e-lev ess f	el-A	A sei conta	rver ains	addre an IP [,]	ess type indicates IPv6, then the service-level-AA v6 address in octet 4 to octet 19.	
lf th serv in o	If the service-level-AA server address type indicates IPv4v6, then the service-level-AA server address field contains two IP addresses. The first IP address is an IPv4 address in octet 4 to octet 7. The second IP address is an IPv6 address in octet 8 to octet 23.								
lf th enc	ie se odec	rvice d as	e-lev defii	vel-A ned i	A sei in su	rver bcla	addre use 1	ess type indicates FQDN, octet 4 to octet z is 9.4.2.1in 3GPP TS 23.003 [4].	

9.11.2.13 Service-level-AA payload

The purpose of the Service-level-AA payload information element is to carry the upper layer payload for authentication and authorization between the UE and the service-level-AA server.

The Service-level-AA payload information element is coded as shown in figure 9.11.2.13.1 and table 9.11.2.13.1.

The Service-level-AA payload information element is a type 6 information element with minimum length of 4 octets and maximum length of 65538 octets.

8	7	6	5	4	3	2	1	_			
		Servio	ce-level-A	AA payle	oad IEI			octet 1			
	Service-level-AA payload length										
		Serv	vice-level	-AA pa	yload			octets 4-s			
	Figure 9.	11.2.13.1	: Servic	ce-leve	el-AA pa	yload infe	ormatio	n element			

Table 9.11.2.13.1: Service-level-AA payload information element

Service-level-AA payload (octet 4 to octet s) A payload for authentication and authorization transparently transported and which is provided from/to the upper layers.

9.11.2.14 Service-level-AA response

The purpose of the Service-level-AA response information element is to provide information regarding the service level authentication and authorization request, e.g. to indicate that the authentication and authorization request to the service level authentication server was successful, or to notify that service level authorization is revoked.

The Service-level-AA response information element is coded as shown in figure 9.11.2.14.1 and table 9.11.2.14.1.

The Service-level-AA response information element is a type 4 information element with length of 3 octets.

8	7	6	5	4	3	2	1			
Service-level-AA response IEI										
Service-level-AA response length										
0 0 C2AR SLAR										
Spar	е	Spare	•							

Figure 9.11.2.14.1: Service-level-AA response information element

Table 9.11.2.14.1: Service-level-AA response information element

S	ervice-le	evel-AA result field (SLAR) (octet 3, bits 1 and 2)
Bi	ts	
1	2	
0	0	No information
0	1	Service level authentication and authorization was successful.
1	0	Service level authentication and authorization was not successful or service
		level authorization is revoked.
1	1	Reserved
C	2 autho	rization result field (C2AR) (octet 3, bits 3 and 4)
Bi	ts	
3	4	
0	0	No information
0	1	C2 authorization was successful.
1	0	C2 authorization was not successful or C2 authorization is revoked.
1	1	Reserved
Bi	ts 5 to 8	3 of octet 3 are spare and shall be coded as zero.

9.11.2.15 Service-level-AA payload type

The purpose of the Service-level-AA payload type information element is to indicates type of payload included in the Service-level-AA payload information element.

The Service-level-AA payload type information element is coded as shown in figure 9.11.2.15.1 and table 9.11.2.15.1.

The Service-level-AA payload type information element is a type 4 information element with length of 3 octets.

8	7	6	5	4	3	2	1	
		Service	-level-AA	payloa	d type IEI			octet 1
		Service-le	evel-AA p	ayload	type leng	th		octet 2
		Servic	e-level-A	A paylo	ad type			octet 3
	Figure 9.1	1.2.15.1:	Service-	level-/	AA payle	oad type	informati	ion element

Table 9.11.2.15.1: Service-level-AA payload type information element

-								
Se	rvice	lev	el-AA	A pay	yload	type	e (oct	tet 3):
Bit	s							
8	7	6	5	4	3	2	1	
0	0	0	0	0	0	0	1	UUAA payload (see NOTE 1)
0	0	0	0	0	0	1	0	C2 authorization payload (see NOTE 2)
All	othe	r va	lues	are	spare	, an	d the	receiving entity shall ignore the service-level-AA
pa	yload	typ	e va	lue s	set to	a sp	bare \	/alue.
	-					-		
NC	DTE 1	: t e t t t t f	f the he se leme JE su f the field f nform proce	servic ent is uppo servic servic for the natio	vice-lev s an a prting vice-lev ne sen on elev e betv	evel- el-A appli UAS evel- rvice emer weer	AA p A pay icatio S ser AA p e-leve nt is a n the	ayload type indicates UUAA payload, the field for yload of the Service-level AA payload information n layer payload for UUAA procedure between the vices and the USS. ayload type indicates C2 authorization payload, the I-AA payload of the Service-level-AA payload in application layer payload for C2 authorization UE supporting UAS services and the USS.

9.11.2.16 Void

_

9.11.2.17 Service-level-AA pending indication

The purpose of the Service-level-AA pending indication information element is to provide an indication that the service level authentication and authorization procedure is to be performed.

The Service-level-AA pending indication information element is coded as shown in figure 9.11.2.17.1 and table 9.11.2.17.1.

The Service-level-AA pending indication information element is a type 1 information element.

8	7	6	5	4	3	2	1	
Ser	vice-leve	I-AA pend	ling	0	0	0	SLAPI	octet 1
	indicat	tion IEI		Spare	Spare	Spare		
	Eigu	ro 0 11 1	2 4 7 4 . (Sanviaa		A nondi	na india	otion

Figure 9.11.2.17.1: Service-level-AA	pending	Indication
--------------------------------------	---------	------------

	Table 9.11.2.17.1:	Service-level-AA	pending indication
--	--------------------	------------------	--------------------

Se	rvice-level-AA pending indication (SLAPI) (octet 1, bit 1)
Bit	
1	
0	reserved
1	Service-level-AA procedure is to be performed

9.11.2.18 Service-level-AA service status indication

The purpose of the Service-level-AA service status indication information element is to provide an indication of the service availability to the UE.

The Service-level-AA service status indication information element is coded as shown in figure 9.11.2.18.1 and table 9.11.2.18.1.

The Service-level-AA service status indication information element is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3		2	1	
		Service-le	evel-AA s	service st	atus indic	ation IEI			octet 1
	Ler	ngth of Se	ervice-lev	el-AA ser	vice statu	us indicat	ion		octet 2
0	0	0	0	0	0	0	0	UAS	7
Spare	Spare	Spare	Spare	Spare	Spare	Spare	Spare		octet 3
Figure 9	0.11.2.18	3.1: Serv	vice-leve	el-AA-se	ervice-st	tatus in	dication	inform	ation element

Table 9.11.2.18.1: Service-level-AA-service-status indication information element

UAS (octet 3, bit 1)	
Bit	
1	
0 UAS services not enabled	
1 UAS services enabled	
Bits 2 to 8 of octet 3 are spare and shall be encoded as zero.	

9.11.3 5GS mobility management (5GMM) information elements

9.11.3.1 5GMM capability

The purpose of the 5GMM capability information element is to provide the network with information concerning aspects of the UE related to the 5GCN or interworking with the EPS. The contents might affect the manner in which the network handles the operation of the UE.

The 5GMM capability information element is coded as shown in figure 9.11.3.1.1 and table 9.11.3.1.1.

The 5GMM capability is a type 4 information element with a minimum length of 3 octets and a maximum length of 15 octets.

8	7	6	5	4	3	2	1	
			5GN	MM capabi	lity IEI			octet 1
			Length of 5	GMM capa	ability contents			octet 2
SGC	5G-	N3	5G-CP	Restrict	LPP	HO	S1	
	IPHC-	data	CloT	EC		attach	mode	octet 3
	CP							
	CloT							
RACS	NSSA	5G-	V2XCNPC	V2XCE	V2X	5G-UP	5GSRV	
	Α	LCS	5	PC5		CloT	CC	octet 4*
5G	5G	5G	ER-NSSAI	5G-	multipleUP	WUSA	CAG	
ProSe-	ProSe-	ProSe-		EHC-				octet 5*
l2relay	dc	dd		CP				
				CloT				
PR	RPR	PIV	NCR	NR-	5G ProSe-I3rmt	5G	5G	octet 6*
				PSSI		ProSe-	ProSe-	
						l2rmt	13relay	
MPSIU	UAS	NSAG	Ex-CAG	SSNPN	EventNotification	MINT	NSSRG	octet 7*
				SI				
0	0	0	0	0	0	RCMAN	RCMAP	octet 8*
spare	spare	spare	spare	spare	spare			
0	0	0	0	0	0	0	0	
				Spare				octet 9*-15*



Table 9.11.3.1.1: 5GMM capability information element

EPC NAS support	ted (S1 mode) (octet 3, bit 1)
0	S1 mode supported S1 mode supported
ATTACH REQUE	ST message containing PDN CONNECTIVITY REQUEST message for
0	(HO attach) (octet 3, bit 2) ATTACH REQUEST message containing PDN CONNECTIVITY REQUEST message with request type set to "handover" or "handover of emergency bearer services" to transfer PDU session from N1 mode to S1 mode not supported
	ATTACH REQUEST message containing PDN CONNECTIVITY REQUEST message with request type set to "handover" or "handover of emergency bearer services" to transfer PDU session from N1 mode to S1 mode supported
LTE Positioning P 0 1	rotocol (LPP) capability (octet 3, bit 3) LPP in N1 mode not supported LPP in N1 mode supported (see 3GPP TS 37.355 [26])
Restriction on use This bit indicates t 0 1	e of enhanced coverage support (RestrictEC) (octet 3, bit 4) the capability to support restriction on use of enhanced coverage. Restriction on use of enhanced coverage not supported Restriction on use of enhanced coverage supported
Control plane Clo This bit indicates t 0 1	T 5GS optimization (5G-CP CloT) (octet 3, bit 5) the capability for control plane CloT 5GS optimization. Control plane CloT 5GS optimization not supported Control plane CloT 5GS optimization supported
N3 data transfer (This bit indicates t 0 1	N3 data) (octet 3, bit 6) the capability for N3 data transfer. N3 data transfer supported N3 data transfer not supported
IP header compre	ession for control plane CIoT 5GS optimization (5G-IPHC-CP CIoT) (octet
This bit indicates to potimization.	the capability for IP header compression for control plane CIoT 5GS
0	IP header compression for control plane CIoT 5GS optimization not supported IP header compression for control plane CIoT 5GS optimization supported
Service gap contro 0 1	ol (SGC) (octet 3, bit 8) service gap control not supported service gap control supported
5G-SRVCC from I 0 1	NG-RAN to UTRAN (5GSRVCC) capability (octet 4, bit 1) 5G-SRVCC from NG-RAN to UTRAN not supported 5G-SRVCC from NG-RAN to UTRAN supported (see 3GPP TS 23.216 [6A])
User plane CloT 5 This bit indicates 1 0 1	5GS optimization (5G-UP CloT) (octet 4, bit 2) the capability for user plane CloT 5GS optimization. User plane CloT 5GS optimization not supported User plane CloT 5GS optimization supported
V2X capability (V2 This bit indicates t Bit 3	2X) (octet 4, bit 3) the capability for V2X, as specified in 3GPP TS 24.587 [19B].
0	V2X not supported V2X supported
V2X communicati This bit indicates t 3GPP TS 24.587 Bit 4	on over E-UTRA-PC5 capability (V2XCEPC5) (octet 4, bit 4) the capability for V2X communication over E-UTRA-PC5, as specified in [19B].

V2X communication over E-UTRA-PC5 not supported 0 V2X communication over E-UTRA-PC5 supported 1 V2X communication over NR-PC5 capability (V2XCNPC5) (octet 4, bit 5) This bit indicates the capability for V2X communication over NR-PC5, as specified in 3GPP TS 24.587 [19B]. Bit 5 0 V2X communication over NR-PC5 not supported 1 V2X communication over NR-PC5 supported Location Services (5G-LCS) notification mechanisms capability (octet 4, bit 6) LCS notification mechanisms not supported 0 LCS notification mechanisms supported (see 3GPP TS 23.273 [6B]) 1 Network slice-specific authentication and authorization (NSSAA) (octet 4, bit 7) This bit indicates the capability to support network slice-specific authentication and authorization. 0 Network slice-specific authentication and authorization not supported 1 Network slice-specific authentication and authorization supported Radio capability signalling optimisation (RACS) capability (octet 4, bit 8) 0 RACS not supported 1 **RACS** supported Closed Access Group (CAG) capability (octet 5, bit 1) CAG not supported CAG supported WUS assistance (WUSA) information reception capability (octet 5, bit 2) WUS assistance information reception not supported 0 WUS assistance information reception supported 1 Multiple user-plane resources support (multipleUP) (octet 5, bit 3) This bit indicates the capability to support multiple user-plane resources in NB-N1 mode. Multiple user-plane resources not supported 0 1 Multiple user-plane resources supported Ethernet header compression for control plane CIoT 5GS optimization (5G-EHC-CP CIoT) (octet 5, bit 4) Ethernet header compression for control plane CIoT 5GS optimization not supported Ethernet header compression for control plane CIoT 5GS optimization supported 1 Extended rejected NSSAI support (ER-NSSAI) (octet 5, bit 5) This bit indicates the capability to support extended rejected NSSAI. 0 Extended rejected NSSAI not supported Extended rejected NSSAI supported 1 5G ProSe direct discovery (5G ProSe-dd) (octet 5, bit 6) This bit indicates the capability for 5G ProSe direct discovery. 5G ProSe direct discovery not supported 0 5G ProSe direct discovery supported 1 5G ProSe direct communication (5G ProSe-dc) (octet 5, bit 7) This bit indicates the capability for 5G ProSe direct communication. 5G ProSe direct communication not supported 0 1 5G ProSe direct communication supported 5G ProSe layer-2 UE-to-network-relay (5G ProSe-I2relay) (octet 5, bit 8) This bit indicates the capability to act as a 5G ProSe layer-2 UE-to-network relay UE 0 Acting as a 5G ProSe layer-2 UE-to-network relay UE not supported 1 Acting as a 5G ProSe layer-2 UE-to-network relay UE supported 5G ProSe layer-3 UE-to-network-relay (5G ProSe-I3relay) (octet 6, bit 1) This bit indicates the capability to act as a 5G ProSe layer-3 UE-to-network relay UE

1	Acting as a 5G ProSe layer-3 UE-to-network relay UE not supported Acting as a 5G ProSe layer-3 UE-to-network relay UE supported
5G ProSe layer-2 This bit indicates t 0 1	UE-to-network-remote (5G ProSe-l2rmt) (octet 6, bit 2) he capability to act as a 5G ProSe layer-2 UE-to-network remote UE Acting as a 5G ProSe layer-2 UE-to-network remote UE not supported Acting as a 5G ProSe layer-2 UE-to-network remote UE supported
5G ProSe layer-3 This bit indicates t 0 1	UE-to-network-remote (5G ProSe-l3rmt) (octet 6, bit 3) he capability to act as a 5G ProSe layer-3 UE-to-network remote UE Acting as a 5G ProSe layer-3 UE-to-network remote UE not supported Acting as a 5G ProSe layer-3 UE-to-network remote UE supported
NR paging subgro This bit indicates t Bit	up support indication (NR-PSSI) (octet 6, bit 4) he capability to support NR paging subgrouping
0	NR paging subgrouping not supported NR paging subgrouping supported
N1 NAS signalling This bit indicates v Bit	connection release (NCR) (octet 6, bit 5) whether N1 NAS signalling connection release is supported.
0	N1 NAS signalling connection release not supported N1 NAS signalling connection release supported
Paging indication f This bit indicates v Bit	for voice services (PIV) (octet 6, bit 6) whether paging indication for voice services is supported.
6 0 1	paging indication for voice services not supported paging indication for voice services supported
Reject paging requ This bit indicates v Bit 7	uest (RPR) (octet 6, bit 7) whether reject paging request is supported.
0	reject paging request not supported
	reject paging request supported
Paging restriction This bit indicates v Bit	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported.
Paging restriction This bit indicates v Bit 8 0 1	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported
Paging restriction This bit indicates v Bit 8 0 1 NSSRG (octet 7, b This bit indicates t Bit	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported bit 1) he capability to support the NSSRG.
Paging restriction This bit indicates v Bit 8 0 1 NSSRG (octet 7, b This bit indicates t Bit 0 1	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported bit 1) he capability to support the NSSRG. NSSRG not supported NSSRG supported
Paging restriction This bit indicates v Bit 8 0 1 NSSRG (octet 7, k This bit indicates t Bit 0 1 Minimization of se This bit indicates t Bit 2	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported bit 1) he capability to support the NSSRG. NSSRG not supported NSSRG supported rvice interruption (MINT) (octet 7, bit 2) he capability to support Minimization of service interruption (MINT)
Paging restriction This bit indicates v Bit 8 0 1 NSSRG (octet 7, b This bit indicates t Bit 0 1 Minimization of se This bit indicates t Bit 2 0 1	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported bit 1) the capability to support the NSSRG. NSSRG not supported NSSRG supported rvice interruption (MINT) (octet 7, bit 2) he capability to support Minimization of service interruption (MINT) MINT not supported MINT supported
Paging restriction This bit indicates v Bit 8 0 1 NSSRG (octet 7, b This bit indicates t Bit 0 1 Minimization of se This bit indicates t Bit 2 0 1 Event notification (This bit indicates t Bit 2	reject paging request supported (PR) (octet 6, bit 8) whether paging restriction is supported. paging restriction not supported paging restriction supported bit 1) he capability to support the NSSRG. NSSRG not supported NSSRG supported rvice interruption (MINT) (octet 7, bit 2) he capability to support Minimization of service interruption (MINT) MINT not supported MINT supported (EventNotification) (octet 7, bit 3) he capability to support event notification for upper layers

SOR-SNPN-SI (SSNPNSI) (octet 7, bit 4) This bit indicates the capability to support SOR-SNPN-SI Bit 2 0 SOR-SNPN-SI not supported SOR-SNPN-SI supported 1 Extended CAG information list support (Ex-CAG) (octet 7, bit 5) This bit indicates the capability to support extended CAG information list. Bit 3 Extended CAG information list not supported 0 1 Extended CAG information list supported NSAG (octet 7, bit 6) This bit indicates the capability to support NSAG. Bit 0 NSAG not supported NSAG supported 1 UAS (octet 7, bit 7) This bit indicates the capability to support UAS services. Bit 0 UAS services not supported UAS services supported 1 MPS indicator update (MPSIU) (octet 7, bit 8) This bit indicates the capability to support MPS indicator update via the UE configuration update procedure. Bit 8 0 MPS indicator update not supported MPS indicator update supported 1 Registration complete message for acknowledging negotiated PEIPS assistance information (RCMAP) (octet 8, bit 1) This bit indicates the capability for sending REGISTRATION COMPLETE message when Negotiated PEIPS assistance information IE is included in the REGISTRATION ACCEPT message. Bit 1 0 Sending of REGISTRATION COMPLETE message for negotiated PEIPS assistance information not supported Sending of REGISTRATION COMPLETE message for negotiated 1 PEIPS assistance information supported Registration complete message for acknowledging NSAG information (RCMAN) (octet 8, bit This bit indicates the capability for sending REGISTRATION COMPLETE message when NSAG information IE is included in the REGISTRATION ACCEPT message. Bit 2 0 Sending of REGISTRATION COMPLETE message for NSAG information not supported 1 Sending of REGISTRATION COMPLETE message for NSAG information supported Bits in 3 to 8 in octet 8 and bits in octets 9 to 15 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.3.2 5GMM cause

The purpose of the 5GMM cause information element is to indicate the reason why a 5GMM request from the UE is rejected by the network.

The 5GMM cause information element is coded as shown in figure 9.11.3.2.1 and table 9.11.3.2.1.

The 5GMM cause is a type 3 information element with 2 octets length.



Table 9.11.3.2.1:	5GMM caus	se information	n element

Bits 8 7 6 5 4 3 2 1 0 0 0 0 0 1 0 1 Illegal UE 0 0 0 0 1 1 0 Illegal ME 0 0 0 0 1 1 1 SGS services not allowed 0 0 0 1 0 1 1 SGS services not allowed 0 0 0 1 1 0 1 PLMN not allowed 0 0 0 1 1 0 Tracking area not allowed 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 <td< th=""><th>Ca</th><th colspan="10">Cause value (octet 2)</th></td<>	Ca	Cause value (octet 2)									
8 7 6 5 4 3 2 1 0 0 0 0 0 0 0 1 1 PEI not accepted 0 0 0 0 1 1 1 SGS services not allowed 0 0 0 1 1 1 SGS services not allowed 0 0 0 1 1 1 DEI dentity cannot be derived by the network 0 0 0 1 1 1 Nacciaity cande and allowed 0 0 0 1 1 1 Nacciaity cande rejected, unspecified 0 0 1 0 1 1 Nacciaity cande rejected, unspecified 0 0 1 1 0 Nacciaitable Nacciaitable 0 0 1 1 1 Nacciaitable Nacciaitable 0 0 1 1 1 Nacciaitable Naciaitable	Bit	Bits									
0 0 0 0 1 1 Illegal ME 0 0 0 0 1 1 SGS services not allowed 0 0 0 0 1 0 1 1 0 0 0 1 0 1 0 1 0 0 0 1 0 1 1 1 0 0 0 1 1 1 0 0 0 0 0 0 1 1 1 0 0 0 0 0 0 0 1 0 1 0 0 0 0 0 0 0 0 1 0 1 0	8	7	6	5	4	3	2	1			
0 0 0 0 1 1 1 PEL not accepted 0 0 0 0 1 1 1 5GS services not allowed 0 0 0 1 0 1 0 1 1 1 0 0 0 1 0 1	0	0	0	0	0	0	1	1			
0 0 0 0 1 1 OSGS services not allowed 0 0 0 1 0 1 1 SGS services not allowed 0 0 0 1 0 1 1 PLMN not allowed 0 0 0 1 1 1 PLMN not allowed 0 0 0 1 1 1 Reaming not allowed 0 0 0 1 1 1 Reaming not allowed 0 0 1 0 1 1 Reaming not allowed 0 0 1 0 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 No suitable cells in tracking a	0	0	0	0	0	1	0	1	PEI not accepted		
0 0 0 0 1 1 Description 0 0 0 1 0 1 0 1 1 Description 1 Description 1 Description 1 Description 1 Description 1 Description 1 1 Description Description Description <tdd< td=""><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>1</td><td>1</td><td>1</td><td></td></tdd<>	0	0	0	0	0	1	1	1			
0 0 0 0 1 0 Implicitly de-registered 0 0 0 1 1 0 Tracking area not allowed 0 0 0 1 1 0 Tracking area not allowed 0 0 0 1 1 0 Tracking area not allowed 0 0 0 1 1 0 Tracking area not allowed 0 0 0 1 1 1 No suitable cells in tracking area 0 0 0 1 1 1 No area not allowed 0 0 1 1 1 No congestion No not congestion 0 0 1 1 1 No Scauthy machemic canuacceptable 0 0 1 1 1 No resource for specific slice and DNN 0 1 1 1 1 No network slices available 0 1 0 1 1 No network slices a	0	0	0	0	1	1	1	1	5GS services not allowed		
0 0 0 1 PLNN not allowed 0 0 0 1 0 1 PLNN not allowed 0 0 0 1 1 0 Roaming not allowed 0 0 0 1 1 0 No suitable cells in tracking area 0 0 0 1 1 0 MAC failure 0 0 0 1 1 0 Congestion 0 0 1 1 0 Congestion Non-5G authentication unacceptable 0 0 1 1 0 Non-5G authentication unacceptable 0 0 1 1 1 Non-5G authentication unacceptable 0 0 1 1 1 Non-5G authentication unacceptable 0 0 1 1 1 Non-5G authentication unacceptable 0 1 0 1 No statbole No authorized 0 1 <t< td=""><td>0</td><td>0</td><td>0</td><td>0</td><td>1</td><td>0</td><td>1</td><td>0</td><td>DE laenility cannot be derived by the network</td></t<>	0	0	0	0	1	0	1	0	DE laenility cannot be derived by the network		
0 0 0 1 1 0 Tracking area not allowed 0 0 0 1 1 0 1 Tracking area not allowed 0 0 0 1 1 0 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 No suitable cells in tracking area 0 0 1 1 1 Non-SG authentication unacceptable 0 0 1 1 1 It Redirection to EPC required 0 0 1 1 1 Redirection to EPC required 0 1 0 1 1 1 Redirection to EPC required 0 1 1 1 1 Redirection to EPC required 0 1 0 1 1 1 1 1 0 1 0 1 1 1	0	0	0	0	1	0	1	1	PI MN not allowed		
0 0 0 1 1 0 1 Reaming not allowed in this tracking area 0 0 0 1 1 1 No suitable cells in tracking area 0 0 0 1 0 1 1 1 No suitable cells in tracking area 0 0 1 0 1 0 0 MAC failure 0 0 1 0 1 0 0 Congestion 0 0 1 1 0 0 Security capabilities mismatch 0 0 1 1 0 1 Non-5G authentication unacceptable 0 0 1 1 0 Non-5G authentication unacceptable 0 1 1 1 0 Non-5G authentication unacceptable 0 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 </td <td>0</td> <td>0</td> <td>0</td> <td>0</td> <td>1</td> <td>1</td> <td>0</td> <td>0</td> <td>Tracking area not allowed</td>	0	0	0	0	1	1	0	0	Tracking area not allowed		
0 0 0 1 1 1 No suitable cells in tracking area 0 0 0 1 0 0 MAC failure 0 0 0 1 0 1 0 0 0 0 1 0 1 1 0 Congestion 0 0 1 0 1 1 0 Congestion 0 0 1 1 1 Usecurity mode rejected, unspecified 0 0 1 1 1 Non-5G authentication unacceptable 0 0 1 1 1 Restricted service area 0 0 1 1 1 Redirection to EPC required 0 1 0 1 No network slices available 0 1 1 1 ADN not available 0 1 0 1 Insufficient resources for specific slice and DNN 0 1 0 1	0	0	0	0	1	1	0	1	Roaming not allowed in this tracking area		
0 0 1 0 MAC failure 0 0 1 0 1 0 1 0 0 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 0 0 1 1 0 0 0 0 1 1 0 0 0 0 1 1 1 1 0 0 0 0 1 1 1 1 0 0 0 0 1 1 1 0 0 0 0 1 1 1 0 0 0 1 1 1 0 0 1 1 1 1 0 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 <td< td=""><td>0</td><td>0</td><td>0</td><td>0</td><td>1</td><td>1</td><td>1</td><td>1</td><td>No suitable cells in tracking area</td></td<>	0	0	0	0	1	1	1	1	No suitable cells in tracking area		
0 0 1 0 1 Synch failure 0 0 1 0 1 1 Congestion 0 0 1 1 0 Congestion Use security capabilities mismatch 0 0 1 1 0 0 Security mode rejected, unspecified 0 0 1 1 0 Non-5G authentication unacceptable 0 0 1 1 1 Restricted service area 0 0 1 1 1 Redirection to EPC required 0 1 0 1 AB-node operation not authorized 0 0 1 0 1 AB-node operation not authorized 0 1 0 0 1 Maximum number of PDU sessions reached 1 0 0 1 1 Insufficient resources for specific slice and DNN 0 1 0 0 1 1 Insufficient resources for specific slice 1 <td>õ</td> <td>Õ</td> <td>õ</td> <td>1</td> <td>0</td> <td>1</td> <td>0</td> <td>0</td> <td>MAC failure</td>	õ	Õ	õ	1	0	1	0	0	MAC failure		
0 0 1 0 1 1 UE Security capabilities mismatch 0 0 1 1 0 0 1 1 0 0 0 1 1 0 0 0 1 1 0 0 0 1 1 0 1 0 0 0 1 1 0 0 0 0 1 1 0 0 0 0 1 1 0 0 0 1 1 0 0 0 1 1 1 1 0 0 0 1 0 0 1 1 1 1 0 0 1 0 0 1 1 1 1 0 0 1 <t< td=""><td>0</td><td>0</td><td>0</td><td>1</td><td>0</td><td>1</td><td>0</td><td>1</td><td>Synch failure</td></t<>	0	0	0	1	0	1	0	1	Synch failure		
0 0 1 1 1 UE security capabilities mismatch 0 0 1 1 0 0 0 1 1 0 0 0 0 0 1 1 0 0 0 0 1 1 0 1 0 0 0 1 1 1 Non-56 authentication unacceptable 0 0 1 1 1 0 Restricted service area 0 0 1 1 1 1 Redirection to EPC required 0 0 1 0 1 0 1 0 1 0 1 0 0 1 1 1 0 0 1 0 1 0 1 0 1 1 1 1 1 0 1 0 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	0	0	0	1	0	1	1	0	Congestion		
0 0 1 1 0 0 Security mode rejected, unspecified 0 0 1 1 0 1 0 Non-5G authentication unacceptable 0 0 0 1 1 0 1 N1 mode not allowed 0 0 1 1 1 0 Restricted service area 0 0 1 0 1 1 1 Redirection to EPC required 0 0 1 0 1 1 1 Redirection to EPC required 0 0 1 0 1 1 No network slices available 0 1 0 0 0 1 Insufficient resources for specific slice 1 0 0 1 1 Insufficient resources for specific slice 1 0 1 0 1 1 Insufficient resources for specific slice 1 0 1 0 1 1 1 Insuffic	0	0	0	1	0	1	1	1	UE security capabilities mismatch		
0 0 1 1 0 Non-5G authentication unacceptable 0 0 0 1 1 0 Non-5G authentication unacceptable 0 0 0 1 1 0 Restricted service area 0 0 1 1 1 1 Restricted service area 0 1 0 1 0 1 0 Non-5G authentication unacceptable 0 0 1 1 1 1 1 Nonetwork slices area 0 0 1 0 1 0 Nonetwork slices available 0 1 0 0 0 1 Insufficient resources for specific slice and DNN 1 0 0 1 0 1 not anthorized 1 0 1 0 1 0 1 not anthorized 1 0 0 1 0 1 not anthorized notisisister 1	0	0	0	1	1	0	0	0	Security mode rejected, unspecified		
000111N1 mode not allowed0001110Restricted service area0010101110010111LAB-node operation not authorized001111LADN not available001111LADN not available001111No network slices available0100011Insufficient resources for specific slice and DNN010011Insufficient resources for specific slice010011Insufficient resources to specific slice010010Non-3GPP access to 5GCN not allowed010101010011Permanently not authorized for this SNPN01010101011Permanently not authorized for this SNPN0101110111Not authorized for this SAPN0101110111Not authorized for this SAPN0101110111Not authorized for this SAPN<	0	0	0	1	1	0	1	0	Non-5G authentication unacceptable		
0 0 1 1 1 1 1 Restricted service area 0 0 1 1 1 1 Redirection to EPC required 0 0 1 0 1 1 1 LAB. Node operation not authorized 0 0 1 0 1 1 1 LADN not available 0 0 1 1 1 1 0 No network slices available 0 1 0 0 0 1 Insufficient resources for specific slice 0 1 0 0 1 1 ngKSI already in use 0 1 0 1 0 1 Non-3GPP access to 5GCN not allowed 0 1 0 1 1 1 ngKSI already in use 0 1 0 1 1 1 Permanently not authorized for this SNPN 0 1 0 1 1 0 Not authorized for this CAG or authori	0	0	0	1	1	0	1	1	N1 mode not allowed		
0 0 1 1 1 1 Redirection to EPC required 0 0 1 0 1 0 1 1 LAB-node operation not authorized 0 0 1 0 1 1 1 1 1 LAB.not available 0 0 1 1 1 1 0 No network slices available 0 1 0 0 0 1 Insufficient resources for specific slice and DNN 0 1 0 0 1 1 ngKSI already in use 0 1 0 0 1 1 ngKSI already in use 0 1 0 1 0 1 ngKSI already in use 0 1 0 1 0 1 ngKSI already in use 0 1 0 1 0 1 ngKSI already in use 0 1 0 1 0 1 ngKSI already in use 0 1 0 1 0 1 ngKSI already in use <	0	0	0	1	1	1	0	0	Restricted service area		
010101IAB-node operation not authorized00101111LADN not available0111110No network slices available0100011Insufficient resources for specific slice and DNN010011Insufficient resources for specific slice010011Insufficient resources for specific slice010011Insufficient resources for specific slice010010Non-3GPP access to 5GCN not allowed010101010011Permanently not authorized for this SNPN01011010011Permanently not authorized for this SNPN01011010011Not authorized for this CAG or authorized for CAG cells only01011010011UAS services not allowed01011010110Disaster condition not allowed01011010111DNN not supported or not subscribed in the slice0<	0	0	0	1	1	1	1	1	Redirection to EPC required		
010111LADN not available0011110No network slices available0100011Maximum number of PDU sessions reached0100011Insufficient resources for specific slice0100111ngKSI already in use0100101ngKSI already in use0100101Serving network not authorized0101010Temporarily not authorized for this SNPN010110Non-3GPP access to 5GCN not allowed010110Non-3GPP access to 5GCN not authorized for this SNPN010110Not authorized for this CAG or authorized for CAG cells only010110Not authorized for this CAG or authorized for CAG cells only010110PluMn not allowed to operate at the present UE location010110PluMn not allowed to operate at the present UE location010110PluMn not allowed010110DNN not supported or not subscribed in the slice location010110Nohoarr	0	0	1	0	0	1	0	0	IAB-node operation not authorized		
0 1 1 1 1 1 0 No network slices available 0 1 0 0 0 1 Maximum number of PDU sessions reached 0 1 0 0 1 1 Insufficient resources for specific slice 0 1 0 0 1 1 Insufficient resources for specific slice 0 1 0 0 1 1 Insufficient resources for specific slice 0 1 0 0 1 1 Insufficient resources for specific slice 0 1 0 0 1 0 Non-3GPP access to 5GCN not allowed 0 1 0 1 0 1 0 No tauthorized for this SNPN 0 1 0 1 1 0 Not authorized for this SNPN 0 1 0 1 1 0 Not authorized for this SNPN 0 1 0 1 1 0 Not authorized for this SNPN 0 1 0 1 1 0	0	0	1	0	1	0	1	1	LADN not available		
0 1 0 0 0 1 Maximum number of PDU sessions reached 0 1 0 0 1 1 Insufficient resources for specific slice and DNN 0 1 0 0 1 1 1 number of PDU sessions reached 0 1 0 0 1 1 1 Insufficient resources for specific slice 0 1 0 0 1 1 1 number of PDU sessions reached 0 1 0 0 1 1 1 number of PDU sessions reached 0 1 0 0 1 1 0 number of PDU sessions reached 0 1 0 0 1 1 0 Non-3GPP access to 5GCN not allowed 0 1 0 1 1 0 Temporarily not authorized for this SNPN 0 1 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 1 1 0 Not authorized for this CAG or authorized f	0	0	1	1	1	1	1	0	No network slices available		
0 1 0 0 0 1 Insufficient resources for specific slice and DNN 0 1 0 0 1 1 1 Insufficient resources for specific slice 0 1 0 0 1 1 1 ngKSI already in use 0 1 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 1 0 1 0 1 0 1 0 0 1 1 0 Nor 3GPP access to 5GCN not allowed 0 1 0 1 1 0 Nor 3uthorized for this SNPN 0 1 0 1 1 0 Not authorized for this CAG or authorized for 0 1 0 1 1 1 UAS services not allowed Deltation 0 1 <td< td=""><td>0</td><td>1</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>1</td><td>Maximum number of PDU sessions reached</td></td<>	0	1	0	0	0	0	0	1	Maximum number of PDU sessions reached		
0 1 0 0 1 0 1 Insufficient resources for specific slice 0 1 0 0 1 1 ngKSI already in use 0 1 0 0 1 0 0 1 0 1 0 0 1 0 0 1 0 1 0 0 1 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 1 0 1 1 0 1 0 1 1 0 1 1 0 1 0 1 1 0 Nort authorized for this SNPN 0 1 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 1 1 1 UAS services not allowed 0 1 0 1 1 1 UAS services not allowed 0 1 0 <	0	1	0	0	0	0	1	1	Insufficient resources for specific slice and DNN		
0100111ngKSI already in use01001000Non-3GPP access to 5GCN not allowed0100101100101111Permanently not authorized for this SNPN010011011010110110101101101001111010011110100111101001111010111110110111101111111011111110111111101111111011111110111111101111111011111110111 <td< td=""><td>0</td><td>1</td><td>0</td><td>0</td><td>0</td><td>1</td><td>0</td><td>1</td><td>Insufficient resources for specific slice</td></td<>	0	1	0	0	0	1	0	1	Insufficient resources for specific slice		
0 1 0 0 0 1 Non-3GPP access to SeCrN not allowed 0 1 0 0 1 0 0 1 Serving network not authorized 0 1 0 0 1 0 1 0 1 0 1 0 1 0 0 1 1 0 Temporarily not authorized for this SNPN 0 1 0 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 1 1 1 UAS services not allowed 0 1 0 1 1 1 UAS services not allowed 0 1 0 1 <	0	1	0	0	0	1	1	1	ngKSI already in use		
010010010100101010100111Permanently not authorized for this SNPN010011001001101Permanently not authorized for this SNPN010011001001101Wireline access area not allowed01001110011110PLMN not allowed to operate at the present UE location010011100110110Payload was not forwarded0101100Disaster roaming for the determined PLMN with disaster condition not allowed010111DNN not supported or not subscribed in the slice010111DNN not supported or not subscribed in the slice010111DNN on supported or not subscribed in the slice010111Disaster roaming for the determined PLMN with disaster condition not allowed010111Onboarding services terminated010001Me	0	1	0	0	1	0	0	1	Non-3GPP access to 5GCN not allowed		
010010110100111Permanently not authorized for this SNPN01001100Not authorized for this CAG or authorized for CAG cells only010011011Wireline access area not allowed01001110PLMN not allowed to operate at the present UE location0101111UAS services not allowed010110Payload was not forwarded010110Disaster roaming for the determined PLMN with disaster condition not allowed010111DNN not supported or not subscribed in the slice010111DNN not supported or not subscribed in the slice010111Semantically incorrect message01111Semantically incorrect message011001Message type not compatible with the protocol state011001110011Information element non-existent or not implemented01101111001Message not compatible with the protocol state0	0	1	0	0	1	0	1	1	Serving network not authorized		
0 1 0 0 1 1 0 Not authorized for this CAG or authorized for CAG cells only 0 1 0 0 1 1 0 1 0 1 0 1 0 0 1 1 0 1 0 1 0 1 0 1 0 0 1 1 0 1 0 1 0 1 0 1 1 0 0 1 0 1 0 1 0 1 1 1 0	0	1	0	0	1	0	1	1	Permanently not authorized for this SNPN		
0 1 0 0 1 1 0 0 1 1 0 1 1 0 1 1 0 1 1 0 1 1 1 0 0 1 1 1 0 0 1 1 1 0 1 1 1 0 0 1 1 1 0 0 1 1 1 1 0 0 0 1 1 1 1 0	0	1	0	0	1	1	0	0	Not authorized for this CAG or authorized for		
0100110001001110001001111001011110001011110001011110001011100001011000010111000111000010111000111110101111010111101011110101111010001011001011001111001011001011001011001011001011001 <tr< td=""><td>Ŭ</td><td></td><td>U</td><td>U</td><td>•</td><td>'</td><td>U</td><td>0</td><td>CAG cells only</td></tr<>	Ŭ		U	U	•	'	U	0	CAG cells only		
0 1 0 0 1 1 1 0 0 1 0 0 1 1 1 0 0 0 0 0 0 0 0 0 1 1 1 0 1 0 1 1 1 1 0 1 0 0 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0 1 1 1 0 0 1 0 1 1 1 0 1 0 1 1 1 0 1 0 1 1 1 0 1 1 1 1 0 1	0	1	0	0	1	1	0	1	Wireline access area not allowed		
0 1 0 1 1 1 1 Ucation 0 1 0 1 1 1 1 UAS services not allowed 0 1 0 1 0 1 0 1 0 0 1 0 1 0 1 0 1 0 0 1 0 1 1 1 0 Payload was not forwarded 0 1 0 1 1 1 0 Disaster roaming for the determined PLMN with disaster condition not allowed 0 1 0 1 1 1 DNN not supported or not subscribed in the slice 0 1 0 1 1 1 DNN not supported or not subscribed in the slice 0 1 0 1 1 0 D Insufficient user-plane resources for the PDU session 0 1 1 1 1 1 D Onboarding services terminated 0 1 1 0 0 0 Invalid mandatory information	Ő	1	õ	õ	1	1	1	0	PLMN not allowed to operate at the present UF		
0101111UAS services not allowed01011010Payload was not forwarded0101000Disaster roaming for the determined PLMN with disaster condition not allowed010110110111011DNN not supported or not subscribed in the slice01011100Insufficient user-plane resources for the PDU session01011111Semantically incorrect message011111Semantically incorrect message0110001Message type non-existent or not implemented0110010Message type not compatible with the protocol state0110011Information element non-existent or not implemented0110101Nessage not compatible with the protocol state0110111Protocol error, unspecifiedAny other value received by the mobile station shall be treated as 01101111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110		-	-	•	-	-	-	-	location		
010110Payload was not forwarded0101000Disaster roaming for the determined PLMN with disaster condition not allowed0101111DNN not supported or not subscribed in the slice01011100Insufficient user-plane resources for the PDU session01011111011111Semantically incorrect message011111Semantically incorrect message011000Invalid mandatory information011001Message type non-existent or not implemented0110011Information element non-existent or not implemented0110011Message not compatible with the protocol state0110111Protocol error, unspecifiedAny other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 01101111, "protocol error, unspecified".	0	1	0	0	1	1	1	1	UAS services not allowed		
01010000Disaster roaming for the determined PLMN with disaster condition not allowed0101111DNN not supported or not subscribed in the slice01011100Insufficient user-plane resources for the PDU session01011111Semantically incorrect message010000Invalid mandatory information0110001Message type non-existent or not implemented0110011Information element non-existent or not implemented01101011Protocol error, unspecified01101111Protocol error, unspecifiedAny other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110	0	1	0	1	1	0	1	0	Payload was not forwarded		
disaster condition not allowed 0 1 0 1 1 1 0 1 1 0 1 0 1 1 1 0 0 0 1 0 1 1 1 1 0 0 0 1 0 1 1 1 1 0 1 0 1 0 1 1 1 1 0 1 0 1 0 1 1 1 1 1 0 0 1 0 0 0 0 0 0 1 1 0 0 0 0 0 0 0 1 1 0 0 0 0 0 0 1 1 0 0 0 0 0 1 0 1 1 0 0 0 0 1 0 0 1 1 0 0 0 0 1 0 0 1 1 0 0 0 0 1 1 0 1 1 0 0 0 1 0 0 0 1 1 0 0 1 0 1 0 1 1 0 1 1 1 1 0 0 0 1 0 1 0 0 0 0 0	0	1	0	1	0	0	0	0	Disaster roaming for the determined PLMN with		
0 1 0 1 1 0 1 1 0 1 1 1 0 0 Insufficient user-plane resources for the PDU session 0 1 0 1 1 1 0 1 0 1 1 1 0 1 1 1 0 1 0 1 1 1 1 0 0 0 1									disaster condition not allowed		
0 1 0 1 1 1 0 0 Insufficient user-plane resources for the PDU session 0 1 0 1 1 1 0 1 0 1 0 1 0 1 1 1 1 1 0 0 0 0 1 0 1 1 1 1 1 Semantically incorrect message 0 1 1 0 0 0 0 Invalid mandatory information 0 1 1 0 0 0 1 Message type non-existent or not implemented 0 1 1 0 0 1 Information element non-existent or not implemented 0 1 1 0 0 1 Information element non-existent or not implemented 0 1 1 0 0 1 Inscriptional IE error 0 1 1 0 1 1 Protocol error, unspecified	0	1	0	1	1	0	1	1	DNN not supported or not subscribed in the slice		
session010111000000000000000000000111 </td <td>0</td> <td>1</td> <td>0</td> <td>1</td> <td>1</td> <td>1</td> <td>0</td> <td>0</td> <td>Insufficient user-plane resources for the PDU</td>	0	1	0	1	1	1	0	0	Insufficient user-plane resources for the PDU		
0 1 0 1 1 0 1 0 1									session		
0 1 0 1 1 1 1 1 Semantically incorrect message 0 1 1 0 0 0 0 Invalid mandatory information 0 1 1 0 0 0 1 Message type non-existent or not implemented 0 1 1 0 0 0 1 Message type not compatible with the protocol state 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 0 Conditional IE error 0 1 1 0 1 1 Message not compatible with the protocol state 0 1 1 0 1 1 1 Protocol error, unspecified	0	1	0	1	1	1	0	1	Onboarding services terminated		
0 1 1 0 0 0 Invalid mandatory information 0 1 1 0 0 0 1 Message type non-existent or not implemented 0 1 1 0 0 1 0 Message type not compatible with the protocol state 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 0 Conditional IE error 0 1 1 0 1 1 Message not compatible with the protocol state 0 1 1 0 1 1 Protocol error, unspecified	0	1	0	1	1	1	1	1	Semantically incorrect message		
0 1 1 0 0 0 1 Message type non-existent or not implemented 0 1 1 0 0 1 0 Message type not compatible with the protocol state 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 0 Conditional IE error 0 1 1 0 1 0 1 Nessage not compatible with the protocol state 0 1 1 0 1 1 1 Protocol error, unspecified	0	1	1	0	0	0	0	0	Invalid mandatory information		
0 1 1 0 0 1 0 Message type not compatible with the protocol state 0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 0 1 0 Conditional IE error 0 1 1 0 1 0 1 Message not compatible with the protocol state 0 1 1 0 1 1 1 Protocol error, unspecified	0	1	1	0	0	0	0	1	Message type non-existent or not implemented		
State0110011Information element non-existent or not implemented0110010Conditional IE error0110101Message not compatible with the protocol state0110111Protocol error, unspecifiedAny other value received by the mobile station shall be treated as 0110 1111, "protocolerror, unspecified".01101111, "protocol error, unspecified".0000	0	1	1	0	0	0	1	0	Message type not compatible with the protocol		
0 1 1 0 0 1 1 Information element non-existent or not implemented 0 1 1 0 1 0 0 Conditional IE error 0 1 1 0 1 0 1 Message not compatible with the protocol state 0 1 1 0 1 1 1 Protocol error, unspecified	_			•	~	~			state		
Implemented 0 1 1 0 0 1 0 0 0 1 1 0 0 1 0 1 0 1 1 0 0 1 0 1 Any other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110 1111, "protocol error, unspecified".	0	1	1	0	U	U	1	1	information element non-existent or not		
0 1 0 0 1 0 0 1 0 0 1 0 0 1 0 1 0 1		4	4	0	0		0	0	Implemented		
Any other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified".	0	1 4	1 4	0	0	1 4	0	U 4	Conditional IE error		
Any other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110 1111, "protocol error, unspecified".	0	1	1	0	1	1	U 1	1	Reproduced a state with the protocol state		
Any other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110 1111, "protocol error, unspecified".	ľ	I	I	U	I	I	I	I	Frotocol error, unspecified		
Any other value received by the mobile station shall be treated as 0110 1111, "protocol error, unspecified". Any other value received by the network shall be treated as 0110 1111, "protocol error, unspecified".											
error, unspecified". Any other value received by the network shall be treated as 0110 1111, "protocol error, unspecified".	Δn	v ot	her	val	י פוו	ere	ive	d hv	the mobile station shall be treated as 0110 1111 "protocol		
1111, "protocol error, unspecified".	err	or	une	nec	ifier	1" L	Anv	a by othe	are meetice station shall be treated as 0110 1111, protocol		
	11	11.	"pro	otoc	ol e	rror	, . un	spe	cified".		

9.11.3.2A 5GS DRX parameters

The purpose of the 5GS DRX parameters information element is to indicate that the UE wants to use DRX and for the network to indicate the DRX cycle value to be used at paging.

The 5GS DRX parameters is a type 4 information element with a length of 3 octets.

The 5GS DRX parameters information element is coded as shown in figure 9.11.3.2A.1 and table 9.11.3.2A.1.

The value part of a DRX parameter information element is coded as shown in table 9.11.3.2A.1.

	1	2	3	4	5	6	7	8			
octet 1	5GS DRX parameters IEI										
octet 2	Length of 5GS DRX parameters contents										
	0 0 0 0 DRX value										
octet 3	spare										
	Figure 0.44.2.2.4.4. ECC DBV never store information clamer										

Figure 9.11.3.2A.1: 5GS DRX parameters information element

Table 9.11.3.2A.1: 5GS DRX parameters information element

D	DRX value (bits 4 to 1 of octet 3)									
Т З	This field represents the DRX cycle parameter 'T' as defined in 3GPP TS 38.304 [28] or 3GPP TS 36.304 [25C].									
в	its									
4	3	2	1							
0	0	0	0	DRX value not specified						
0	0	0	1	DRX cycle parameter $T = 32$						
0	0	1	0	DRX cycle parameter $T = 64$						
0	0	1	1	DRX cycle parameter T = 128						
0	1	0 0 DRX cycle parameter T = 256								
All other values shall be interpreted as "DRX value not specified" by this version of the protocol.										
в	Bits 5 to 8 of octet 3 are spare and shall be coded as zero.									

9.11.3.3 5GS identity type

The purpose of the 5GS identity type information element is to specify which identity is requested.

The 5GS identity type is a type 1 information element.

The 5GS identity type information element is coded as shown in figure 9.11.3.3.1 and table 9.11.3.3.1.



Ту	Type of identity (octet 1)									
Bit	Bits									
3	2	1								
0	0	1	SUCI							
0	1	0	5G-GUTI							
0	1	1	IMEI							
1	0	0	5G-S-TMSI							
1	0	1	IMEISV							
1	1 1 0 MAC address									
1	1	1	EUI-64							
All	All other values are unused and shall be interpreted as "SUCI", if received by the UE.									

Table 9.11.3.3.1: 5GS identity type information element

9.11.3.4 5GS mobile identity

The purpose of the 5GS mobile identity information element is to provide either the SUCI, the 5G-GUTI, the IMEI, the IMEISV, the 5G-S-TMSI, the MAC address or the EUI-64.

The 5GS mobile identity information element is coded as shown in figures 9.11.3.4.1, 9.11.3.4.2, 9.11.3.4.3, 9.11.3.4.4, 9.11.3.4.5, 9.11.3.4.6, 9.11.3.4.8 and 9.11.3.4.7, and table 9.11.3.4.1.

The 5GS mobile identity is a type 6 information element with a minimum length of 4.

8	7	6	5	4	3	2	1	_			
	5GS mobile identity IEI										
	Length of 5GS mobile identity contents										
1											
	I	I	I	spare	i ył		iity	octet 4			
	MCC	digit 2			MCC	digit 1		octet 5			
		al: a: 4 O			MOO	di a: t- 0		a stat C			
	MINC	aigit 3			MCC	aigit 3					
	MNC digit 2 MNC digit 1										
	inite digit 1										
	AMF Region ID										
	AME Set ID										
AMF	Set ID							-			
(conti	(continued) AMF Pointer										
			5G-	TMSI				octet 11			
	5G-TMSI (continued)										
	5G-TMSI (continued)										
	5G-TMSI (continued)										

Figure 9.11.3.4.1: 5GS mobile identity information element for type of identity "5G-GUTI"


Figure 9.11.3.4.2: 5GS mobile identity information element for type of identity "IMEI" or "IMEISV"

8 7 6 5 4 3 2 1											
		5G	S mobile	identity l	EI			octet 1			
								octet 2			
	Length of 5GS mobile identity contents										
					-			octet 3			
0	0 Type of identity										
spare	spare SUPI format spare										
	MCC digit 2 MCC digit 1										
MNC digit 3 MCC digit 3								octet 6			
								e et et 7			
MINU digit 2 MINU digit 1											
Routing indicator digit 2 Routing indicator digit 1							octet 8				
Routing indicator digit 4 Routing indicator digit 3								octet 9			
0	0 0 0 0										
Spare Spare Spare Spare Protection scheme Id								octet 10			
llama naturali muhlia kay idantifian											
	Home network public key identifier										
Scheme output											













Figure 9.11.3.4.5: 5GS mobile identity information element for type of identity "5G-S-TMSI"



Figure 9.11.3.4.6: 5GS mobile identity information element for type of identity "No identity"



Figure 9.11.3.4.7: 5GS mobile identity information element for type of identity "MAC address"





Table 9.11.3.4.1: 5GS mobile identity information element

Type of identity (octet 4) Bits 3 2 1 0 0 0 No identity (see NOTE 1) 0 SUCI 0 1 0 5G-GUTI 0 1 0 1 IMEI 1 1 0 0 5G-S-TMSI 0 1 IMEISV 1 1 0 MAC address 1 1 1 1 EUI-64 All other values are reserved. Odd/even indication (octet 4) Bit 4 0 even number of identity digits odd number of identity digits 1 For the 5G-GUTI, then bits 5 to 8 of octet 4 are coded as "1111", octet 5 through 7 contain the MCC and MNC values as specified below, octet 8 through 10 contain the AMF Region ID, the AMF Set ID and the AMF Pointer values and octet 11 through 14 contain the 5G-TMSI as defined in 3GPP TS 23.003 [4]. MCC, Mobile country code (octet 5, octet 6 bits 1 to 4) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code (octet 6 bits 5 to 8, octet 7) The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet 6 shall be coded as "1111". The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12]. AMF Region ID (octet 8) This field contains the binary encoding of the AMF Region ID. Bit 8 of octet 7 is the most significant bit and bit 1 of octet 7 is the least significant bit. AMF Set ID (octet 9, octet 10 bits 7 to 8) This field contains the binary encoding of the AMF Set ID. Bit 8 of octet 9 is the most significant bit and bit 7 of octet 10 is the least significant bit. AMF Pointer (octet 10 bits 1 to 6) This field contains the binary encoding of the AMF Pointer. Bit 6 of octet 9 is the most significant bit and bit 1 of octet 9 is the least significant bit. 5G-TMSI (octet 11 to 14) Bit 8 of octet 11 is the most significant bit and bit 1 of octet 14 is the least significant bit. Identity digit (octet 4 bits 5 to 8, octet 5 etc.) For the IMEI, Identity digit field is coded using BCD coding. If the number of identity digits is even then bits 5 to 8 of the last octet shall be filled with an end mark coded as "1111". The format of the IMEI is described in 3GPP TS 23.003 [4]. For the IMEISV, Identity digit field is coded using BCD coding. Bits 5 to 8 of the last octet shall be filled with an end mark coded as "1111". The format of the IMEISV is described in 3GPP TS 23.003 [4]. For the SUCI, bit 8 of octet 4 is spare and shall be coded as zero. Bits 5-7 of octet 4 contain the SUPI format and are coded as shown below.

SUPI format (octet 4, bits 5-7) Bits 7 6 5 0 0 0 IMSI 0 Network specific identifier 0 1 0 1 0 GCI 1 GLI 0 1 All other values are interpreted as IMSI by this version of the protocol. For the SUCI with SUPI format "IMSI", octets 5 through 7 contain the MCC and MNC values as specified below. For subsequent fields, bit 8 of octet 8 is the most significant bit and bit 1 of the last octet the least significant bit. The required fields for the SUCI are as defined in 3GPP TS 23.003 [4]. MCC, Mobile country code (octet 5, octet 6 bits 1 to 4) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code (octet 6 bits 5 to 8, octet 7) The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet 6 shall be coded as "1111". The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12]. Routing indicator (octets 8-9) Routing Indicator shall consist of 1 to 4 digits. The coding of this field is the responsibility of home network operator but BCD coding shall be used. If a network operator decides to assign less than 4 digits to Routing Indicator, the remaining digits shall be coded as "1111" to fill the 4 digits coding of Routing Indicator (see NOTE 2). If no Routing Indicator is configured in the USIM or the ME, the UE shall code bits 1 to 4 of octet 8 of the Routing Indicator as "0000" and the remaining digits as "1111". Protection scheme identifier (octet 10 bits 1 to 4) Rits 4 3 2 1 Λ 0 0 0 Null scheme 0 0 1 ECIES scheme profile A 0 0 0 1 0 ECIES scheme profile B 0 1 1 0 Reserved to 0 1 1 1 1 0 0 1 Operator-specific protection scheme to 1 1 1 1 Bits 5-8 of octet 10 are spare and shall be coded as zero. Home network public key identifier (octet 11) The Home network public key identifier (PKI) field is coded as defined in 3GPP TS 23.003 [4]. Home network public key identifier shall be coded as "00000000" when Protection scheme identifier is set to "0000" (i.e. Null scheme). Bits 8 7 6 5 4 3 2 1 0 0 0 0 0 0 0 0 Home network PKI value 0 0 0 0 0 0 0 0 1 Home network PKI value (1-254) to 1 1 1 0 1 1 1 1 1 1 1 1 1 1 1 Reserved 1

Scheme output (octets 12 to x)

For the SUCI with SUPI format set to "Network specific identifier", the SUCI NAI field contains an NAI constructed as specified in subclause 28.7.3 of 3GPP TS 23.003 [4] and encoded as UTF-8 string.

For the SUCI with SUPI format set to "GCI", the SUCI NAI field contains an NAI constructed as specified in subclause 28.15.5 of 3GPP TS 23.003 [4] and encoded as UTF-8 string.

For the SUCI with SUPI format set to "GLI", the SUCI NAI field contains an NAI constructed as specified in subclause 28.16.5 of 3GPP TS 23.003 [4] and encoded as UTF-8 string.

For the 5G-S-TMSI, bits 5 to 8 of octet 4 are coded as "1111". The coding of the 5G-S-TMSI is left open for each administration.

AMF Set ID (octet 5, octet 6 bits 7 to 8) This field contains the binary encoding of the AMF Set ID. Bit 8 of octet 5 is the most significant bit and bit 7 of octet 6 is the least significant bit.

AMF Pointer (octet 6 bits 1 to 6) This field contains the binary encoding of the AMF Pointer. Bit 6 of octet 6 is the most significant bit and bit 1 of octet 6 is the least significant bit.

5G-TMSI (octet 7 to 10) Bit 8 of octet 7 is the most significant bit and bit 1 of octet 10 is the least significant bit.

For Type of identity "No identity", the length of mobile identity contents parameter shall be set to 1 and the bits 4-8 of octet 4 are spare and shall be coded as zero.

MAC address usage restriction indication (MAURI) (octet 4 bit 4) Bit

4	
0	No restrictions
1	MAC address is not usable as an equipment identifier
MAC add This field Bit 8 of oc	ress (octets 5 to 10) contains the MAC address as defined in subclause 8 of IEEE Std 802 [43]. :tet 5 is the most significant bit and bit 1 of octet 10 is the least significant bit.
EUI-64 (o This field Bit 8 of oc	ctets 5 to 12) contains an EUI-64 as defined in [48]. tet 5 is the most significant bit and bit 1 of octet 12 is the least significant bit.
NOTE 1:	This can be used when the requested identity is not available at the UE during the identification procedure.
NOTE 2:	For a 3-digit Routing Indicator, e.g "567", bits 1 to 4 of octet 8 are coded as "0101", bits 5 to 8 of octet 8 are coded as "0110", bits 1 to 4 of octet 9 are coded as "0111", bits 5 to 8 of octet 9 are coded as "1111".

9.11.3.5 5GS network feature support

The purpose of the 5GS network feature support information element is to indicate whether certain features are supported by the network.

The 5GS network feature support information element is coded as shown in figure 9.11.3.5.1 and table 9.11.3.5.1.

The 5GS network feature support is a type 4 information element with a minimum length of 3 octets and a maximum length of 5 octets.

If the network does not include octet 4 as defined in figure 9.11.3.5.1 in the present version of the protocol, then the UE shall interpret this as a receipt of an information element with all bits of octet 4 coded as zero.

8	7	6	5	4	3	2	1		
		5GS ne	etwork fea	ature sup	port IEI			octet 1	
Length of 5GS network feature support contents									
MPSI	IWK	EN	ΛF	EN	/IC	IMS-	IMS-	octet 3	
	N26					VoPS-	VoPS-		
						N3GP	3GPP		
P									
5G-UP	5G-	N3	5G-CP	Restr	ictEC	MCSI	EMCN	octet 4*	
CloT	IPHC-	data	CloT				3		
	CP								
CloT									
0	PR	RPR	PIV	NCR	5G-	ATS-	5G-	octet 5*	
Spare					EHC-	IND	LCS		
					CP				
					CIoT				

Figure 9.11.3.5.1: 5GS network feature support information element

Table 9.11.3.5.1: 5GS network feature support information element

IMS voice over PS session over 3GPP access indicator (IMS-VoPS-3GPP) (octet 3, bit 1) This bit indicates the support of IMS voice over PS session over 3GPP access (see NOTE 1). Bit 1 0 IMS voice over PS session not supported over 3GPP access IMS voice over PS session supported over 3GPP access 1 IMS voice over PS session over non-3GPP access indicator (IMS-VoPS-N3GPP) (octet 3, bit 2) This bit indicates the support of IMS voice over PS session over non-3GPP access. Bit 2 0 IMS voice over PS session not supported over non-3GPP access IMS voice over PS session supported over non-3GPP access 1 Emergency service support indicator for 3GPP access (EMC) (octet 3, bit 3 and bit 4) These bits indicate the support of emergency services in 5GS for 3GPP access (see NOTE 1). Bits 4 3 0 0 Emergency services not supported 0 1 Emergency services supported in NR connected to 5GCN only 1 0 Emergency services supported in E-UTRA connected to 5GCN only Emergency services supported in NR connected to 5GCN and E-UTRA 1 1 connected to 5GCN Emergency services fallback indicator for 3GPP access (EMF) (octet 3, bit 5 and bit 6) These bits indicate the support of emergency services fallback for 3GPP access (see NOTE 1). Bits 65 0 0 Emergency services fallback not supported 0 Emergency services fallback supported in NR connected to 5GCN only 1 Emergency services fallback supported in E-UTRA connected to 5GCN 1 0 only Emergency services fallback supported in NR connected to 5GCN and 1 1 E-UTRA connected to 5GCN Interworking without N26 interface indicator (IWK N26) (octet 3, bit 7) This bit indicates whether interworking without N26 interface is supported. Bit 7 0 Interworking without N26 interface not supported Interworking without N26 interface supported 1 MPS indicator (MPSI) (octet 3, bit 8) This bit indicates the validity of MPS. Bit 8 0 Access identity 1 not valid Access identity 1 valid 1 Emergency service support for non-3GPP access indicator (EMCN3) (octet 4, bit 1) This bit indicates the support of emergency services in 5GS for non-3GPP access. Bit (see NOTE 2) 1 0 Emergency services not supported over non-3GPP access Emergency services supported over non-3GPP access 1 MCS indicator (MCSI) (octet 4, bit 2) This bit indicates the validity of MCS. Bit 2 0 Access identity 2 not valid 1 Access identity 2 valid

Restriction on enhanced coverage (RestrictEC) (octet 4, bit 3 and bit 4) These bits indicate enhanced coverage restricted information. In WB-N1 mode these bits are set as follows: Bits 4 3 0 0 Both CE mode A and CE mode B are not restricted 0 1 Both CE mode A and CE mode B are restricted CE mode B is restricted 1 0 Reserved 1 1 In NB-N1 mode these bits are set as follows Bits 4 3 0 0 Use of enhanced coverage is not restricted 0 1 Use of enhanced coverage is restricted Reserved 1 0 Reserved 1 1 Control plane CIoT 5GS optimization (5G-CP CIoT) (octet 4, bit 5) This bit indicates the capability for control plane CIoT 5GS optimization. Bit 5 0 Control plane CIoT 5GS optimization not supported Control plane CIoT 5GS optimization supported 1 N3 data transfer (N3 data) (octet 4, bit 6) This bit indicates the capability for N3 data transfer. Bit 6 0 N3 data transfer supported 1 N3 data transfer not supported IP header compression for control plane CIoT 5GS optimization (5G-IPHC-CP CIoT) (octet 4, bit 7) This bit indicates the capability for IP header compression for control plane CIoT 5GS optimization. Bit 7 0 IP header compression for control plane CIoT 5GS optimization not supported 1 IP header compression for control plane CIoT 5GS optimization supported User plane CIoT 5GS optimization (5G-UP CIoT) (octet 4, bit 8) This bit indicates the capability for user plane CIoT 5GS optimization. Bit 8 0 User plane CIoT 5GS optimization not supported 1 User plane CIoT 5GS optimization supported Location Services indicator in 5GC (5G-LCS) (octet 5, bit 1) Bit 1 0 Location services via 5GC not supported Location services via 5GC supported 1 ATSSS support indicator (ATS-IND) (octet 5, bit 2) This bit indicates the network support for ATSSS. Bit 2 0 ATSSS not supported 1 ATSSS supported Ethernet header compression for control plane CIoT 5GS optimization (5G-EHC-CP CIoT) (octet 5, bit 3) This bit indicates the capability for Ethernet header compression for control plane CIoT 5GS optimization

N1 NAS signalling connection release (NCR) (octet 5, bit 4) This bit indicates whether N1 NAS signalling connection release is supported. Bit 4 0 N1-NAS signalling connection release not supported 1 N1-NAS signalling connection release supported Paging indication for voice services (PIV) (octet 5, bit 5) This bit indicates whether paging indication for voice services is supported. Bit 5 0 paging indication for voice services not supported 1 paging indication for voice services supported Reject paging request (RPR) (octet 5, bit 6) This bit indicates whether reject paging request is supported. Bit 6 0 reject paging request not supported 1 reject paging request supported Paging restriction (PR) (octet 5, bit 7) This bit indicates whether paging restriction is supported. Bit 7 0 paging restriction not supported 1 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	Bit 3 0 1	Ethernet header compression for control plane CIoT 5GS optimization not supported Ethernet header compression for control plane CIoT 5GS optimization supported
0 N1-NAS signalling connection release not supported 1 N1-NAS signalling connection release supported Paging indication for voice services (PIV) (octet 5, bit 5) This bit indicates whether paging indication for voice services is supported. Bit 5 0 paging indication for voice services not supported 1 paging indication for voice services not supported 1 paging indication for voice services supported 1 paging request (RPR) (octet 5, bit 6) This bit indicates whether reject paging request is supported. Bit 6 0 reject paging request supported 1 reject paging request supported 1 reject paging restriction is supported. Bit 7 0 paging restriction not supported 1 paging restriction supported 1 paging restriction supported 1 paging restriction supported 1 paging restriction supported	N1 NAS signa This bit indica Bit 4	alling connection release (NCR) (octet 5, bit 4) ates whether N1 NAS signalling connection release is supported.
Paging indication for voice services (PIV) (octet 5, bit 5) This bit indicates whether paging indication for voice services is supported. Bit 5 0 paging indication for voice services not supported 1 paging indication for voice services supported Reject paging request (RPR) (octet 5, bit 6) This bit indicates whether reject paging request is supported. Bit 6 0 reject paging request not supported 1 reject paging request supported 1 reject paging request supported 1 reject paging request supported 1 reject paging restriction is supported 1 reject paging restriction is supported. Bit Bit indicates whether paging restriction is supported. Bit paging restriction not supported 1 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	0	N1-NAS signalling connection release not supported N1-NAS signalling connection release supported
 paging indication for voice services not supported paging indication for voice services supported Reject paging request (RPR) (octet 5, bit 6) This bit indicates whether reject paging request is supported. Bit reject paging request not supported reject paging request supported reject paging request supported reject paging request supported reject paging request supported paging restriction (PR) (octet 5, bit 7) This bit indicates whether paging restriction is supported. Bit paging restriction not supported paging restriction supported paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero. 	Paging indica This bit indica Bit 5	tion for voice services (PIV) (octet 5, bit 5) ates whether paging indication for voice services is supported.
Reject paging request (RPR) (octet 5, bit 6) This bit indicates whether reject paging request is supported. Bit 6 0 reject paging request not supported 1 reject paging request supported Paging restriction (PR) (octet 5, bit 7) This bit indicates whether paging restriction is supported. Bit 7 0 paging restriction not supported 1 paging restriction supported Bit 7 0 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	0 1	paging indication for voice services not supported paging indication for voice services supported
0 reject paging request not supported 1 reject paging request supported Paging restriction (PR) (octet 5, bit 7) This bit indicates whether paging restriction is supported. Bit 7 0 paging restriction not supported 1 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	Reject paging This bit indica Bit 6	g request (RPR) (octet 5, bit 6) ates whether reject paging request is supported.
Paging restriction (PR) (octet 5, bit 7) This bit indicates whether paging restriction is supported. Bit 7 0 paging restriction not supported 1 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	0	reject paging request not supported reject paging request supported
0 paging restriction not supported 1 paging restriction supported Bit 8 in octet 5 is spare and shall be coded as zero.	Paging restric This bit indica Bit 7	ction (PR) (octet 5, bit 7) ates whether paging restriction is supported.
Bit 8 in octet 5 is spare and shall be coded as zero.	0 1	paging restriction not supported paging restriction supported
	Bit 8 in octet	5 is spare and shall be coded as zero.
 NOTE 1: For a registration procedure over non-3GPP access, bit 1 of octet 3 and bits 3 to 6 of octet 3 are ignored. NOTE 2: For a registration procedure over 3GPP access, bit 2 of octet 3 and bit 1 of octet 4 are ignored. 	NOTE 1: Fo 3 tr NOTE 2: Fo oct	

9.11.3.6 5GS registration result

The purpose of the 5GS registration result information element is to specify the result of a registration procedure. The 5GS registration result information element is coded as shown in figure 9.11.3.6.1 and table 9.11.3.6.1. The 5GS registration result is a type 4 information element with a length of 3 octets.

	8	7	6	5	4	3	2	1		
			5GS	registrat	ion result	: IEI			octet 1	
Length of 5GS registration result contents										
	0	Disast	Emerg	NSSA	SMS	5GS reg	gistration	result		
	Spare	er	ency	Α	allowe		value		octet 3	
		roamin	registe	Perfor	d					
		g	red	med						
		registr								
		ation								
		result								
		value								

Figure 9.11.3.6.1: 5GS registration result information element

Table 9.11.3.6.1: 5GS registration result information element

5GS registration result value (octet 3, bits 1 to 3) (NOTE)								
Bits								
$3 \ 2 \ 1$								
0 1 0 Non-3GPP access								
0 1 1 3GPP access and non-3GPP access								
1 1 1 reserved								
All other values are unused and shall be treated as "3GPP access", if received by the UE.								
SMS over NAS transport allowed (SMS allowed) (octet 3, bit 4) (NOTI Bit	Ξ)							
SMS over NAS not allowed								
1 SMS over NAS allowed								
Network slice-specific authentication and authorization is to be perfor be performed) (octet 3, bit 5) (NOTE) Bit	med (NSSAA to							
0 Network slice-specific authentication and authorization	is not to be							
performed 1 Network slice-specific authentication and authorization performed Performed	is to be							
Emergency registered (octet 3, bit 6)								
Bit								
0 Not registered for emergency services								
1 Registered for emergency services								
Disaster roaming registration result value (octet 3, bit 7) Bit 7								
0 No additional information								
1 Request for registration for disaster roaming services a registration not for disaster roaming services	ccepted as							
Bit 8 of octet 3 is spare and shall be coded as zero.								
NOTE: All bits other than bit 6 in octet 3 shall be ignored by the U registration result IE is received in the CONFIGURATION COMMAND message	E when the 5GS UPDATE							

9.11.3.7 5GS registration type

The purpose of the 5GS registration type information element is to indicate the type of the requested registration. The 5GS registration type information element is coded as shown in figure 9.11.3.7.1 and table 9.11.3.7.1. The 5GS registration type is a type 1 information element with a length of 1 octet.



5G	S re	egistrat	ion type value (octet 1, bits 1 to 3)
Bit	s		
3	2	1	
0	0	1	initial registration
0	1	0	mobility registration updating
0	1	1	periodic registration updating
1	0	0	emergency registration
1	0	1	SNPN onboarding registration
1	1	0	disaster roaming mobility registration updating
1	1	1	disaster roaming initial registration
All by	oth the	er valu networ	es are unused and shall be interpreted as "initial registration", if received k.
Fo	llow	-on rec	uest bit (FOR) (octet 1, bit 4)
Bit			
4			
0			No follow-on request pending
1			Follow-on request pending

9.11.3.8 5GS tracking area identity

The purpose of the 5GS tracking area identity information element is to provide an unambiguous identification of tracking areas within the area covered by the 5GS.

The 5GS tracking area identity information element is coded as shown in figure 9.11.3.8.1 and table 9.11.3.8.1.

The 5GS tracking area identity is a type 3 information element with a length of 7 octets.

8	7	6	5	4	3	2	1		
5GS tracking area identity IEI									
MCC digit 2 MCC digit 1									
	MNC digit 3 MCC digit 3								
	MNC digit 2 MNC digit 1								
ТАС									
TAC (continued)									
TAC (continued)									

Figure 9.11.3.8.1: 5GS tracking area identity information element

MCC, Mobile country code (octets 2 and 3) The MCC field is coded as in ITU-T Rec. E212 [39], annex A.
If the TAI is deleted the MCC and MNC shall take the value from the deleted TAI.
In abnormal cases, the MCC stored in the UE can contain elements not in the set {0, 1 9}. In such cases the UE should transmit the stored values using full hexadecimal encoding. When receiving such an MCC, the network shall treat the TAI as deleted.
MNC, Mobile network code (octet 3 bits 5 to 8, octet 4) The coding of this field is the responsibility of each administration, but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. For PCS 1900 for NA, Federal regulation mandates that a 3-digit MNC shall be used. However, a network operator may decide to use only two digits in the MNC in the TAI over the radio interface. In this case, bits 5 to 8 of octet 3 shall be coded as "1111". Mobile equipment shall accept a TAI coded in such a way.
In abnormal cases, the MNC stored in the UE can have: - digit 1 or 2 not in the set {0, 1 9}, or - digit 3 not in the set {0, 1 9, F} hex. In such cases the UE shall transmit the stored values using full hexadecimal encoding. When receiving such an MNC, the network shall treat the TAI as deleted.
The same handling shall apply for the network, if a 3-digit MNC is sent by the UE to a network using only a 2-digit MNC.
TAC, Tracking area code (octets 5 to 7) In the TAC field bit 8 of octet 5 is the most significant bit and bit 1 of octet 7 the least significant bit. The coding of the tracking area code is the responsibility of each administration except that two values are used to mark the TAC, and hence the TAI, as deleted. Coding using full hexadecimal representation may be used. The tracking area code consists of 3 octets. If a TAI has to be deleted, then all bits of the tracking area code shall be set to one with the exception of the least significant bit which shall be set to zero. If a USIM is
inserted in a mobile equipment with the tracking area code containing all zeros, then the mobile equipment shall recognise this TAC as part of a deleted TAI.

Table 9.11.3.8.1: 5GS tracking area identity information element

9.11.3.9 5GS tracking area identity list

The purpose of the 5GS tracking area identity list information element is to transfer a list of tracking areas from the network to the UE.

The coding of the information element allows combining different types of lists. The lists of type "00" and "01" allow a more compact encoding, when the different TAIs are sharing the PLMN identity.

The 5GS tracking area identity list information element is coded as shown in figure 9.11.3.8.1, figure 9.11.3.8.2, figure 9.11.3.9.3, figure 9.11.3.9.4 and table 9.11.3.9.1.

The 5GS tracking area identity list is a type 4 information element, with a minimum length of 9 octets and a maximum length of 114 octets. The list can contain a maximum of 16 different tracking area identities.



Figure 9.11.3.9.1: 5GS tracking area identity list information element

8	7	6	5	4	3	2	1		
0 Spare	0 Type of list Number of elements Spare								
	MCC	digit 2			МСС	digit 1		octet 2	
	MNC	digit 3			MCC	digit 3		octet 3	
	MNC	digit 2			MNC	digit 1		octet 4	
	TAC 1								
		octet 6							
		octet 7							
	octet 3k+2*								
	octet 3k+3*								
	octet 3k+4*								

Figure 9.11.3.9.2: Partial tracking area identity list - type of list = "00"

8	7	6	5	4	3	2	1	
0	Туре	of list		Numb	per of eler	ments		octet 1
Spare								_
	MCC	digit 2			MCC	digit 1		octet 2
MNC digit 3					MCC	digit 3		octet 3
	MNC	digit 2			MNC	digit 1		octet 4
TAC 1								octet 5
TAC 1 (continued)							octet 6	
			TAC 1 (c	ontinued)			octet 7

Figure 9.11.3.9.3: Partial tracking area identity list – type of list = "01"

8	7 6	5	4	3	2	1	_
0	Type of list		Num	ber of elen	nents		octet 1
Spare			1				_
	MCC digit 2			MCC	ligit 1		octet 2
					ligit i		
	MNC digit 3			MCC o	ligit 3		octet 3
	MNC digit 2			MNC o	digit 1		octet 4
		TA	C 1				octet 5
		TAC 1 (co	ontinued	l)			octet 6
		TAC 1 (c	ontinuec	Ð			octet 7
				·)			
	MCC digit 2			MCC o	digit 1		octet 8*
	MNC digit 2			MCC	ligit 2		ootot 0*
	wine aigit 3				ligit 3		ociel 9
	MNC digit 2			MNC o	ligit 1		octet 10*
		τ.,	<u></u>				o ot ot 11*
		IA	62				
		TAC 2 (co	ontinued	I)			octet 12*
			ontinuer	n.			octot 13*
		17.0 2 (0		•)			
	MCC digit 2			MCC o	ligit 1		octet 6k-4*
	MNC digit 3			MCC o	ligit 3		octet 6k-3*
	MNC digit 2			MNC c	ligit 1		octet 6k-2*
		TA	Ck				octet 6k-1*
		octet 6k*					
		TAC k (co	ontinuec	I)			octet 6k+1*

Figure 9.11.3.9.4: Partial tracking area identity list – type of list = "10"

Table 9.11.3.9.1: Tracking area identity list information element

Value part of the Tracking area identity list information element (octets 3 to n) The value part of the Tracking area identity list information element consists of one or several partial tracking area identity lists. The length of each partial tracking area identity list can be determined from the 'type of list' field and the 'number of elements' field in the first octet of the partial tracking area identity list. The UE shall store the complete list received. If more than 16 TAIs are included in this information element, the UE shall store the first 16 TAIs and ignore the remaining octets of the information element. Partial tracking area identity list: Type of list (octet 1) Bits 76 0 0 list of TACs belonging to one PLMN or SNPN, with non-consecutive TAC values 1 list of TACs belonging to one PLMN or SNPN, with consecutive TAC values 0 1 0 list of TAIs belonging to different PLMNs (see NOTE) All other values are reserved. Number of elements (octet 1) Bits 54 3 2 1 0 0 0 0 0 1 element 0 0 0 0 1 2 elements 0 0 0 1 0 3 elements 1 1 0 1 14 elements 0 0 15 elements 0 1 1 1 0 1 1 1 1 16 elements All other values are unused and shall be interpreted as 16, if received by the UE. Bit 8 of octet 1 is spare and shall be coded as zero. For type of list = "00" and number of elements = k: octet 2 to 4 contain the MCC+MNC, and for j = 1, ..., k: octets 3j+2 to 3j+4 contain the TAC of the j-th TAI belonging to the partial list, For type of list = "01" and number of elements = k: octet 2 to 4 contain the MCC+MNC, and octets 5 to 7 contain the TAC of the first TAI belonging to the partial list. The TAC values of the other k-1 TAIs are TAC+1, TAC+2, ..., TAC+k-1. For type of list = "10" and number of elements = k: for j = 1, ..., k. octets 6j-4 to 6j-2 contain the MCC+MNC, and octets 6j-1 to 6j+1 contain the TAC of the j-th TAI belonging to the partial list. MCC, Mobile country code The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111". TAC, Tracking area code

In the TAC field bit 8 of the first octet is the most significant bit and bit 1 of third octet the least significant bit. The coding of the tracking area code is the responsibility of each administration. Coding using full hexadecimal representation may be used. The tracking area code consists of 3 octets.

NOTE: If the "list of TAIs belonging to different PLMNs" is used, the PLMNs included in the list need to be present in the list of "equivalent PLMNs". This type of list is not applicable in an SNPN.

9.11.3.9A 5GS update type

The purpose of the 5GS update type IE is to allow the UE to provide additional information to the network when performing a registration procedure.

The 5GS update type information element is coded as shown in figure 9.11.3.9A.1 and table 9.11.3.9A.1.

The 5GS update type is a type 4 information element with a length of 3 octects.

_	1	2	3	4	5	6	7	8				
octet 1	5GS update type IEI											
octet 2	Length of 5GS update type octe											
	SMS	NG-	CloT	5GS-PNB-	IB-CIoT	EPS- PN	0	0				
octet 3	reques	RAN-					Spare	Spare				
	ted	RCU										

Figure 9.11.3.9A.1: 5GS update type information element



SMS over NAS transport requested (SMS requested) (octet 3, bit 1) Bit 1 0 SMS over NAS not supported 1 SMS over NAS supported NG-RAN Radio Capability Update (NG-RAN-RCU) (octet 3, bit 2) Bit 2 0 UE radio capability update not needed 1 UE radio capability update needed For a list of RATs for which a radio capability update can be triggered by means of this indication see subclause 5.5.1.3.2, case n). 5GS Preferred CloT network behaviour (5GS PNB-CloT) (octet 3, bits 3 and 4) Bits 4 3 0 0 no additional information control plane CIoT 5GS optimization 0 1 0 user plane CIoT 5GS optimization 1 1 1 reserved EPS Preferred CloT network behaviour (EPS-PNB-CloT) (octet 3, bits 5 and 6) Bits 65 0 0 no additional information 0 1 control plane CIoT EPS optimization 0 user plane CIoT EPS optimization 1 1 reserved 1 Bits 7 to 8 of octet 3 are spare and shall be coded as zero.

9.11.3.10 ABBA

The purpose of the ABBA information element is to enable the bidding down protection of security features.

The ABBA information element is coded as shown in figure 9.11.3.10.1 and table 9.11.3.10.1.

The ABBA is a type 4 information element with a minimum length of 4 octets.



Figure 9.11.3.10.1: ABBA information element

Table 9.11.3.10.1: ABBA information element

ABBA contents (octet 3-n): indicate set of security features defined for 5GS as described in 3GPP TS 33.501 [24].								
NOTE 1:	If the UE receives the ABBA IE with a length that is set to a value of 2 and with a value of 0000H, the UE shall use the length and the contents of the ABBA IE as received from the network.							
NOTE 2: 2 or with a contents of	If the UE receives the ABBA IE with a length that is set to a value larger than a value that is different from 0000H, the UE shall use the length and the of the ABBA IE as received from the network.							

9.11.3.11 Void

9.11.3.12 Additional 5G security information

The purpose of the Additional 5G security information information element is to provide the UE with additional security parameters (e.g. horizontal derivation parameter) or to request the UE to retransmit an initial NAS message during a security mode control procedure as defined in 3GPP TS 33.501 [24]. The UE uses these parameters for completion of security mode control procedure.

The Additional 5G security information information element is coded as shown in figure 9.11.3.12.1 and table 9.11.3.12.1.

The Additional 5G security information is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1			
Additional 5G security information IEI										
	Length of Additional 5G security information contents									
0	0	0	0	0	0	RINM	HDP			
Spare	Spare	Spare	Spare	Spare	Spare	R		octet 3		
	44 0 40	A. A.J.	iti amal K		alder limfo		:			

Figure 9.11.3.12.1: Additional 5G security information information element

Table 9.11.3.12.1: Additional 5G security information information eleme	ent
---	-----

Horizo	ontal derivation parameter (HDP) (octet 3, bit 1)
0	K _{AMF} derivation is not required
1	K _{AMF} derivation is required
Retrar	nsmission of initial NAS message request (octet 3, bit 2)
0	Retransmission of the initial NAS message not requested
1	Retransmission of the initial NAS message requested
Bits 3	to 8 of octet 3 are spare and shall be coded as zero.

9.11.3.12A Additional information requested

The purpose of the Additional information requested information element is to enable the UE to request ciphering keys for deciphering of ciphered broadcast assistance data.

The Additional information requested information element is coded as shown in figure 9.11.3.12A.1 and table 9.11.3.12A.1.

The Additional information requested is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	_	
Additional information requested IEI									
Length of additional information requested contents									
0	0 0 0 0 0 0 0 Cipher								
			Spare				Key	octet 3	



Table 9.11.3.12A.1: Additional information requested information element

Ciphering key	/s for ciphered broadcast assistance data (CipherKey) (octet 3, bit 1)
Bit	
1	
0	ciphering keys for ciphered broadcast assistance data not requested
1	ciphering keys for ciphered broadcast assistance data requested
Bits 8 to 2 of	octet 3 are spare and shall be coded as zero.

9.11.3.13 Allowed PDU session status

The purpose of the Allowed PDU session status information element is to indicate to the network user-plane resources of PDU sessions associated with non-3GPP access that are allowed to be re-established over 3GPP access or if there is no PDU session(s) for which the UE allows the user-plane resources to be re-established over 3GPP access.

NOTE: Allowed PDU session status IE is not applicable for MA PDU session(s) in this release of specification.

The Allowed PDU session status information element is coded as shown in figure 9.11.3.13.1 and table 9.11.3.13.1.

The Allowed PDU session status is a type 4 information element with minimum length of 4 octets and maximum length of 34 octets.

	8	7	6	5	4	3	2	1		
Allowed PDU session status IEI										
	octet 2									
	PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 3	
	(7)	(6)	(5)	(4)	(3)	(2)	(1)	(0)		
	PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 4	
	(15)	(14)	(13)	(12)	(11)	(10)	(9)	(8)		
	0	0	0	0	0	0	0	0		
				Spa	are				octet 5* -34	

Figure 9.11.3.13.1: Allowed PDU session status information element

Table 9.11.3.13.1: Allowed PDU session status information element

PSI(x) shall be coded as follows:
PSI(0): Bit 1octet 3 is spare and shall be coded as zero.
PSI(1) – PSI(15):
0 indicates that the user-plane resources of corresponding PDU session is not allowed to be re-established over 3GPP access.
1 indicates that the user-plane resources of corresponding PDU session can be re-established over 3GPP access.
1 indicates that the user-plane resources of corresponding PDU session can be re-established over 3GPP access.
If there is no PDU session for which the user-plane resources can be re-established over 3GPP access, all bits in PSI(1) – PSI(15) shall be coded as zero.
All bits in octet 5 to 34 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.3.14 Authentication failure parameter

See subclause 10.5.3.2.2 in 3GPP TS 24.008 [12].

9.11.3.15 Authentication parameter AUTN

See subclause 10.5.3.1.1 in 3GPP TS 24.008 [12].

9.11.3.16 Authentication parameter RAND

See subclause 10.5.3.1 in 3GPP TS 24.008 [12].

9.11.3.17 Authentication response parameter

See subclause 9.9.3.4 in 3GPP TS 24.301 [15].

9.11.3.18 Configuration update indication

The purpose of the Configuration update indication information element is to indicate the additional information associated with the generic UE configuration update procedure.

The Configuration update indication information element is coded as shown in figure 9.11.3.18.1 and table 9.11.3.18.1.

The Configuration update indication is a type 1 information element.



Table 9.11.3.18.1: Configuration update indication

Acknowledgement (ACK) (octet 1, bit 1) Bit 1 0 acknowledgement not requested 1 acknowledgement requested

Registration requested (RED) (octet 1, bit 2) Bit 2	
0 registration not requested1 registration requested	
Bits 3 and 4 are spare and shall be coded as zero,	

9.11.3.18A CAG information list

The purpose of the CAG information list information element is to provide "CAG information list" or to delete the "CAG information list" at the UE.

The CAG information list information element is coded as shown in figures 9.11.3.18A.1 and 9.11.3.18A.2 and table 9.11.3.18A.1.

The CAG information list is a type 6 information element, with a minimum length of 3 octets.





8	7	6	5	4	3	2	1				
	octet q										
	MCC	digit 2			MCC	digit 1		octet q+1			
	MNC	digit 3			MCC	digit 3		octet q+2			
	MNC	digit 2			MNC	digit 1		octet q+3			
0	0	0	0	0	0	0	CAG	octet q+4			
Spare	Spare	Spare	Spare	Spare	Spare	Spare	only				
								octet q+5*			
			CAG	-ID 1							
								octet q+8*			
								octet q+9*			
			CAG	-ID 2							
								octet q+12*			
								octet q+13*			
								octet q+4m*			
			CAG	-ID m							
								octet q+4m+4			

Figure 9.11.3.18A.2: Entry n

MCC, Mobile country code (octet q+1 and bits 1 to 4 octet q+2) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.
MNC, Mobile network code (bits 5 to 8 of octet q+2 and octet q+3) The coding of this field is the responsibility of each administration, but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet q+2 shall be coded as "1111".
The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12].
Indication that the UE is only allowed to access 5GS via CAG cells (CAGonly) (bit 1 of octet q+4) Bit
 Indication that the UE is only allowed to access 5GS via CAG cells" is not set (i.e., the UE is allowed to access 5GS via non-CAG cells) Indication that the UE is only allowed to access 5GS via CAG cells" is set (i.e., the UE is not allowed to access 5GS via non-CAG cells)
CAG-ID m (octet q+4m+1 to octet q+4m+4) This field contains the 32 bit CAG-ID. The coding of the CAG-ID is defined as the CAG- Identifier in 3GPP TS 23.003 [4].
NOTE 1: The Length of CAG information list contents shall be 0 if no subscription data for CAG information list exists
NOTE 2: The Length of entry contents shall be 4 if there is no allowed CAG-ID for the PLMN.
NOTE 3: Bit 2 in octet q+4 may be set to 1 in the USIM (see 3GPP TS 31.102 [22]). NOTE 4: For a given PLMN ID, there shall be up to one Entry containing the MCC value and the MNC value of the PLMN ID.

Table 9.11.3.18A.1: CAG information list information element

9.11.3.18B CIoT small data container

This information element is used to encapsulate the CIoT user data, SMS, or location services message with a size that is not more than 254 octets between the UE and the AMF when the UE is using control plane CIoT 5GS optimization. The CIoT small data container information element is coded as shown in figure 9.11.3.18B.1, figure 9.11.3.18B.2, figure 9.11.3.18B.3, figure 9.11.3.18B.4 and table 9.11.3.18B.1.

The CIoT small data container is a type 4 information element with a minimum length of 4 octets and a maximum length of 257 octets.











Figure 9.11.3.18B.3: CloT small data container contents for Data type "Location services message container"



Figure 9.11.3.18B.4: CloT small data container contents for Data type "SMS"

Table 9.11.3.18B.1: CloT small data container information element

CIoT small data container contents (octet 3 to octet 257) These octets include user data to be delivered between UE and AMF. Data type (octet 3, bits 6 to 8) Bits 8 7 6 0 0 0 Control plane user data 0 0 1 SMS Location services message container 0 1 0 All other values are spare. If received they shall be ignored. When the Data type is "Control plane user data ", the PDU session identity and Downlink data expected (DDX) fields are encoded as follows: PDU session identity (octet 3, bits 1 to 4) Bit 3 2 1 0 0 0 No PDU session identity assigned PDU session identity value 1 0 0 1 PDU session identity value 2 0 1 0 PDU session identity value 3 0 1 1 1 0 0 PDU session identity value 4 1 0 1 PDU session identity value 5 0 PDU session identity value 6 1 1 1 1 PDU session identity value 7 1 Downlink data expected (DDX) (octet 3, bits 5 to 6) Bits 5 4 0 0 No information available No further uplink and no further downlink data transmission subsequent to the 0 1 uplink data transmission is expected 0 Only a single downlink data transmission and no further uplink data 1 transmission subsequent to the uplink data transmission is expected Reserved 1 1 NOTE: The DDX field is only used in the UE to network direction. Data contents (octet 4 to octet 257) This field contains the control plane user data. When the Data type is "SMS", Bits 1 to 5 of octet 3 are spare and shall be coded as zero. Data contents (octet 4 to octet 257) This field contains an SMS message. When the Data type is "Location services message container": Downlink data expected (DDX) (octet 3, bits 5 to 4) This field is encoded as described above for the case when the Data type is "Control plane user data". Bits 3 to 1 of octet 3 are spare and shall be encoded as zero. Length of Additional information (octet 4) (see NOTE) Indicates the length, in octets, of the Additional information field. Additional information (octets 5 to m) Contains additional information if provided by the upper layer location services application. Data contents (octets m+1 to n) Contains the location services message payload.

NOTE: The Length of Additional information shall be set to zero if the upper layer location service application does not provide routing information.

9.11.3.18C Ciphering key data

The purpose of the Ciphering key data information element is to transfer a list of ciphering data sets from the network to the UE for deciphering of ciphered assistance data.

The Ciphering key data information element is coded as shown in figure 9.11.3.18C.1, figure 9.11.3.18C.2 and table 9.11.3.18C.1.

The Ciphering key data is a type 6 information element, with a minimum length of 34 octets and a maximum length of 2675 octets. The list can contain a maximum of 16 ciphering data sets.



Figure 9.11.3.18C.1: Ciphering key data information element

8	7	6	5	4	3	2	1	_
	Ciphering set ID o						octet 1	
	Ciphering key							octet 2
			Ciprion	ing ito y				001010
								a at at 10
0	0	0			c0 lenath	1		octet 19
Ĵ	Spare	Ū			ee lengu			
			_	•				octet 20
			С	0				
				-				octet k
0	0	0	0	E-	UTRA po	sSIB len	gth	octet k+1
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	octet k+2
Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	00101 1112
1	2	3	4	5	6	7	8	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	octet k+3
1	2	3	4	5	6	7	8	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	
Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	
Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	
17 DooSIR	18 DooSIR	19 DooSIR	20 DooSIR	21 DooSIR	22	23	24	ootot n
Type2-	Tvpe3-	Tvpe4-	Type5-	Type1-	Tvpe1-	Spare	Spare	ociel p
25	1	1	1	9	10	-1	-1	
0	0	0	0		NR posS	IB length		octet p+1
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	octet p+2
Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	Type1-	
	2	3		5	6		8 DeeCID	octot p 13
Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	ociei p+3
1	2	3	4	5	6	7	8	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	
Typez- 9	1ypez-	1ypez-	Type∠- 12	13	1ypez-	1ypez-	1ypez-	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	
Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type2-	Type3-	
PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	PosSIB	
Type4-	Type5-	Type6-	Type6-	Type6-	Type6-	Type6-	Type6-	
			2	3	4	5	6	
Type1-	Type1-	Type2-	Type2-	Spare	Spare	Spare	Spare	octet q
9	10	24	25	opaio	oparo	oparo	oparo	
Volidity start time							octet q+1	
Validity start time						octe a+5		
Validity duration							octet q+6	
							octet q+7	
			TAIG	s list				octet q+8
1			1740					

Figure 9.11.3.18C.2: Ciphering data set

Table 9.11.3.18C.1: Ciphering key data information element

Value part of the Ciphering key data information element (octets 4 to n) The value part of the Ciphering key data information element consists of one or several ciphering data sets. The UE shall store the complete list received. If more than 16 ciphering data sets are included in this information element, the UE shall store the first 16 ciphering data sets and ignore the remaining octets of the information element. Ciphering data set: Ciphering set ID (octets 1 to 2) This field contains the binary encoding of the ID identifying the ciphering set. Ciphering key (octets 3 to octet 18) This field contains the 128 bit ciphering key. c0 length (octet 19, bits 5 to 1) This field contains the binary encoding of the length, in octets, of the c0 counter. The maximum value for the length of the c0 counter is 16 octets. Bits 8 to 6 of octect 19 are spare and shall be coded as zero. c0 (octets 20 to k) This field contains the binary encoding of the c0 counter. E-UTRA posSIB length (octet k+1, bits 4 to 1) This field contains the length in octets of the E -UTRA Positioning SIB types. A length of zero means E -UTRA Positioning SIB types are not included (see NOTE). E-UTRA Positioning SIB types for which the ciphering data set is applicable (octets k+2 to p). Unassigned bits shall be ignored by a UE. Non-included bits shall be assumed to be zero by a UE. Ciphering data set applicable for positioning SIB type 1-1 (octet k+2, bit 8) Ciphering data set not applicable to positioning SIB type 1-1 0 1 Ciphering data set applicable to positioning SIB type 1-1 Ciphering data set applicable for positioning SIB type 1-2 (octet k+2, bit 7) Ciphering data set not applicable to positioning SIB type 1-2 0 Ciphering data set applicable to positioning SIB type 1-2 1 Ciphering data set applicable for positioning SIB type 1-3 (octet k+2, bit 6) Ciphering data set not applicable to positioning SIB type 1-3 Ciphering data set applicable to positioning SIB type 1-3 1 Ciphering data set applicable for positioning SIB type 1-4 (octet k+2, bit 5) Ciphering data set not applicable to positioning SIB type 1-4 0 Ciphering data set applicable to positioning SIB type 1-4 1 Ciphering data set applicable for positioning SIB type 1-5 (octet k+2, bit 4) 0 Ciphering data set not applicable to positioning SIB type 1-5 Ciphering data set applicable to positioning SIB type 1-5 1 Ciphering data set applicable for positioning SIB type 1-6 (octet k+2, bit 3) Ciphering data set not applicable to positioning SIB type 1-6 0 Ciphering data set applicable to positioning SIB type 1-6 1 Ciphering data set applicable for positioning SIB type 1-7 (octet k+2, bit 2) 0 Ciphering data set not applicable to positioning SIB type 1-7 Ciphering data set applicable to positioning SIB type 1-7 1

Ciphering data set applicable for positioning SIB type 1-8 (octet k+2, bit 1)0Ciphering data set not applicable to positioning SIB type 1-81Ciphering data set applicable to positioning SIB type 1-8
Ciphering data set applicable for positioning SIB type 2-1 (octet k+3, bit 8) Ciphering data set not applicable to positioning SIB type 2-1 Ciphering data set applicable to positioning SIB type 2-1
Ciphering data set applicable for positioning SIB type 2-2 (octet k+3, bit 7) Ciphering data set not applicable to positioning SIB type 2-2 Ciphering data set applicable to positioning SIB type 2-2
Ciphering data set applicable for positioning SIB type 2-3 (octet k+3, bit 6) Ciphering data set not applicable to positioning SIB type 2-3 Ciphering data set applicable to positioning SIB type 2-3
Ciphering data set applicable for positioning SIB type 2-4 (octet k+3, bit 5) Ciphering data set not applicable to positioning SIB type 2-4 Ciphering data set applicable to positioning SIB type 2-4
Ciphering data set applicable for positioning SIB type 2-5 (octet k+3, bit 4)0Ciphering data set not applicable to positioning SIB type 2-51Ciphering data set applicable to positioning SIB type 2-5
Ciphering data set applicable for positioning SIB type 2-6 (octet k+3, bit 3)0Ciphering data set not applicable to positioning SIB type 2-61Ciphering data set applicable to positioning SIB type 2-6
Ciphering data set applicable for positioning SIB type 2-7 (octet k+3, bit 2)0Ciphering data set not applicable to positioning SIB type 2-71Ciphering data set applicable to positioning SIB type 2-7
Ciphering data set applicable for positioning SIB type 2-8 (octet k+3, bit 1)0Ciphering data set not applicable to positioning SIB type 2-81Ciphering data set applicable to positioning SIB type 2-8
Ciphering data set applicable for positioning SIB type 2-9 (octet k+4, bit 8)0Ciphering data set not applicable to positioning SIB type 2-91Ciphering data set applicable to positioning SIB type 2-9
Ciphering data set applicable for positioning SIB type 2-10 (octet k+4, bit 7)0Ciphering data set not applicable to positioning SIB type 2-101Ciphering data set applicable to positioning SIB type 2-10
Ciphering data set applicable for positioning SIB type 2-11 (octet k+4, bit 6)0Ciphering data set not applicable to positioning SIB type 2-111Ciphering data set applicable to positioning SIB type 2-11
Ciphering data set applicable for positioning SIB type 2-12 (octet k+4, bit 5)0Ciphering data set not applicable to positioning SIB type 2-121Ciphering data set applicable to positioning SIB type 2-12
Ciphering data set applicable for positioning SIB type 2-13 (octet k+4, bit 4)0Ciphering data set not applicable to positioning SIB type 2-131Ciphering data set applicable to positioning SIB type 2-13
Ciphering data set applicable for positioning SIB type 2-14 (octet k+4, bit 3)0Ciphering data set not applicable to positioning SIB type 2-141Ciphering data set applicable to positioning SIB type 2-14
Ciphering data set applicable for positioning SIB type 2-15 (octet k+4, bit 2)0Ciphering data set not applicable to positioning SIB type 2-151Ciphering data set applicable to positioning SIB type 2-15
Ciphering data set applicable for positioning SIB type 2-16 (octet k+4, bit 1) 0 Ciphering data set not applicable to positioning SIB type 2-16

1 Ciphering data set applicable to positioning SIB type 2-16
Ciphering data set applicable for positioning SIB type 2-17 (octet k+5, bit 8) 0 Ciphering data set not applicable to positioning SIB type 2-17
1 Ciphering data set applicable to positioning SIB type 2-17
Ciphering data set applicable for positioning SIB type 2-18 (octet k+5, bit 7) Ciphering data set not applicable to positioning SIB type 2-18 Ciphering data set applicable to positioning SIB type 2-18
Ciphering data set applicable for positioning SIB type 2-19 (octet k+5, bit 6) Ciphering data set not applicable to positioning SIB type 2-19 Ciphering data set applicable to positioning SIB type 2-19
Ciphering data set applicable for positioning SIB type 2-20 (octet k+5, bit 5)0Ciphering data set not applicable to positioning SIB type 2-201Ciphering data set applicable to positioning SIB type 2-20
Ciphering data set applicable for positioning SIB type 2-21 (octet k+5, bit 4)0Ciphering data set not applicable to positioning SIB type 2-211Ciphering data set applicable to positioning SIB type 2-21
Ciphering data set applicable for positioning SIB type 2-22 (octet k+5, bit 3) Ciphering data set not applicable to positioning SIB type 2-22 Ciphering data set applicable to positioning SIB type 2-22
Ciphering data set applicable for positioning SIB type 2-23 (octet k+5, bit 2) Ciphering data set not applicable to positioning SIB type 2-23 Ciphering data set applicable to positioning SIB type 2-23
Ciphering data set applicable for positioning SIB type 2-24 (octet k+5, bit 1)0Ciphering data set not applicable to positioning SIB type 2-241Ciphering data set applicable to positioning SIB type 2-24
Ciphering data set applicable for positioning SIB type 2-25 (octet k+6, bit 8)0Ciphering data set not applicable to positioning SIB type 2-251Ciphering data set applicable to positioning SIB type 2-25
Ciphering data set applicable for positioning SIB type 3-1 (octet k+6, bit 7)0Ciphering data set not applicable to positioning SIB type 3-11Ciphering data set applicable to positioning SIB type 3-1
Ciphering data set applicable for positioning SIB type 4-1 (octet k+6, bit 6)0Ciphering data set not applicable to positioning SIB type 4-11Ciphering data set applicable to positioning SIB type 4-1
Ciphering data set applicable for positioning SIB type 5-1 (octet k+6, bit 5)0Ciphering data set not applicable to positioning SIB type 5-11Ciphering data set applicable to positioning SIB type 5-1
Ciphering data set applicable for positioning SIB type 1-9 (octet k+6, bit 4) Ciphering data set not applicable to positioning SIB type 1-9 Ciphering data set applicable to positioning SIB type 1-9
Ciphering data set applicable for positioning SIB type 1-10 (octet k+6, bit 3) Ciphering data set not applicable to positioning SIB type 1-10 Ciphering data set applicable to positioning SIB type 1-10
Any unassigned bits shall be coded as zero.
NR posSIB length (octet p+1, bits 4 to 1)
This field contains the length in octets of the NR Positioning SIB types. A length of zero means NR Positioning SIB types are not included (see NOTE).
NR Positioning SIB types for which the ciphering data set is applicable (octets p+2 to q). Unassigned bits shall be ignored. Non-included bits shall be assumed to be zero.

	_
Ciphering data set applicable for positioning SIB type 1-1 (octet p+2, bit 8) Ciphering data set not applicable to positioning SIB type 1-1 Ciphering data set applicable to positioning SIB type 1-1	
Ciphering data set applicable for positioning SIB type 1-2 (octet p+2, bit 7)Ciphering data set not applicable to positioning SIB type 1-2Ciphering data set applicable to positioning SIB type 1-2	
Ciphering data set applicable for positioning SIB type 1-3 (octet p+2, bit 6)Ciphering data set not applicable to positioning SIB type 1-3Ciphering data set applicable to positioning SIB type 1-3	
Ciphering data set applicable for positioning SIB type 1-4 (octet p+2, bit 5)Ciphering data set not applicable to positioning SIB type 1-4Ciphering data set applicable to positioning SIB type 1-4	
Ciphering data set applicable for positioning SIB type 1-5 (octet p+2, bit 4)Ciphering data set not applicable to positioning SIB type 1-5Ciphering data set applicable to positioning SIB type 1-5	
Ciphering data set applicable for positioning SIB type 1-6 (octet p+2, bit 3)Ciphering data set not applicable to positioning SIB type 1-6Ciphering data set applicable to positioning SIB type 1-6	
Ciphering data set applicable for positioning SIB type 1-7 (octet p+2, bit 2)Ciphering data set not applicable to positioning SIB type 1-7Ciphering data set applicable to positioning SIB type 1-7	
Ciphering data set applicable for positioning SIB type 1-8 (octet p+2, bit 1)Ciphering data set not applicable to positioning SIB type 1-8Ciphering data set applicable to positioning SIB type 1-8	
Ciphering data set applicable for positioning SIB type 2-1 (octet p+3, bit 8)Ciphering data set not applicable to positioning SIB type 2-1Ciphering data set applicable to positioning SIB type 2-1	
Ciphering data set applicable for positioning SIB type 2-2 (octet p+3, bit 7)Ciphering data set not applicable to positioning SIB type 2-2Ciphering data set applicable to positioning SIB type 2-2	
Ciphering data set applicable for positioning SIB type 2-3 (octet p+3, bit 6)Ciphering data set not applicable to positioning SIB type 2-3Ciphering data set applicable to positioning SIB type 2-3	
Ciphering data set applicable for positioning SIB type 2-4 (octet p+3, bit 5)Ciphering data set not applicable to positioning SIB type 2-4Ciphering data set applicable to positioning SIB type 2-4	
Ciphering data set applicable for positioning SIB type 2-5 (octet p+3, bit 4)Ciphering data set not applicable to positioning SIB type 2-5Ciphering data set applicable to positioning SIB type 2-5	
Ciphering data set applicable for positioning SIB type 2-6 (octet p+3, bit 3)Ciphering data set not applicable to positioning SIB type 2-6Ciphering data set applicable to positioning SIB type 2-6	
Ciphering data set applicable for positioning SIB type 2-7 (octet p+3, bit 2)Ciphering data set not applicable to positioning SIB type 2-7Ciphering data set applicable to positioning SIB type 2-7	
Ciphering data set applicable for positioning SIB type 2-8 (octet p+3, bit 1)Ciphering data set not applicable to positioning SIB type 2-8Ciphering data set applicable to positioning SIB type 2-8	
Ciphering data set applicable for positioning SIB type 2-9 (octet p+4, bit 8) Ciphering data set not applicable to positioning SIB type 2-9	

l	1		Ciphering	data set	applic	able to	positio	ning SII	B type 2	<u>2</u> -9	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	BIB type e to pos position	2-10 (d sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 7) pe 2-10 2-10	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos positio	2-11 (d sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 6) pe 2-11 2-11	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos positio	2-12 (o sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 5) pe 2-12 2-12	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos position	2-13 (d sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 4) pe 2-13 2-13	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos positio	2-14 (o sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 3) pe 2-14 2-14	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	BIB type e to pos position	2-15 (d sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 2) pe 2-15 2-15	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos positio	2-16 (o sitioning ning SII	octet p+ g SIB ty B type 2	-4, bit 1) pe 2-16 2-16	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos positio	2-17 (d sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 8) pe 2-17 2-17	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos positio	2-18 (o sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 7) pe 2-18 2-18	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos positio	2-19 (d sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 6) pe 2-19 2-19	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos positio	2-20 (o sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 5) pe 2-20 2-20	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos position	2-21 (o sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 4) pe 2-21 2-21	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos position	2-22 (o sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 3) pe 2-22 2-22	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	IB type e to pos position	2-23 (o sitioning ning SII	octet p+ g SIB ty B type 2	-5, bit 2) pe 2-23 2-23	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	positio not ap applic	oning S oplicable able to	BIB type e to pos position	3-1 (oo sitioning ning SII	ctet p+5 g SIB ty B type 3	5, bit 1) pe 3-1 3-1	
	Ciphering 0 1	data	a set appli Ciphering Ciphering	cable for data set data set	position not ap applic	oning S oplicable able to	IB type e to pos position	e 4-1 (oo sitioning ning SII	ctet p+6 g SIB ty B type 4	5, bit 8) pe 4-1 I-1	
ľ	Ciphering	data	a set appli Ciphering	cable for data set	position	oning S	BIB type	5-1 (od	ctet p+6 3 SIB tv	6, bit 7) pe 5-1	

1 Ciphering data set applicable to positioning SIB type 5-1
Ciphering data set applicable for positioning SIB type 6-1 (octet p+6, bit 6)0Ciphering data set not applicable to positioning SIB type 6-11Ciphering data set applicable to positioning SIB type 6-1
Ciphering data set applicable for positioning SIB type 6-2 (octet p+6, bit 5) Ciphering data set not applicable to positioning SIB type 6-2 Ciphering data set applicable to positioning SIB type 6-3
Ciphering data set applicable for positioning SIB type 6-3 (octet p+6, bit 4) 0 Ciphering data set not applicable to positioning SIB type 6-3 1 Ciphering data set applicable to positioning SIB type 6-3
Ciphering data set applicable for positioning SIB type 6-4 (octet p+6, bit 3) 0 Ciphering data set not applicable to positioning SIB type 6-4 1 Ciphering data set applicable to positioning SIB type 6-4
Ciphering data set applicable for positioning SIB type 6-5 (octet p+6, bit 2)0Ciphering data set not applicable to positioning SIB type 6-51Ciphering data set applicable to positioning SIB type 6-5
Ciphering data set applicable for positioning SIB type 6-6 (octet p+6, bit 1)0Ciphering data set not applicable to positioning SIB type 6-61Ciphering data set applicable to positioning SIB type 6-6
Ciphering data set applicable for positioning SIB type 1-9 (octet p+7, bit 8)0Ciphering data set not applicable to positioning SIB type 1-91Ciphering data set applicable to positioning SIB type 1-9
Ciphering data set applicable for positioning SIB type 1-10 (octet p+7, bit 7)0Ciphering data set not applicable to positioning SIB type 1-101Ciphering data set applicable to positioning SIB type 1-10
Ciphering data set applicable for positioning SIB type 2-24 (octet p+7, bit 6)0Ciphering data set not applicable to positioning SIB type 2-241Ciphering data set applicable to positioning SIB type 2-24
Ciphering data set applicable for positioning SIB type 2-25 (octet p+7, bit 5) 0 Ciphering data set not applicable to positioning SIB type 2-25 1 Ciphering data set applicable to positioning SIB type 2-25
Any unassigned bits shall be coded as zero.
Validity start time (octets q+1 to q+5)
This field contains the UTC time when the ciphering data set becomes valid, encoded as octets 2 to 6 of the Time zone and time IE specified in 3GPP TS 24.008 [12].
Validity duration (octets q+6 to q+7)
This field contains the duration for which the ciphering data set is valid after the validity start time, in units of minutes.
TAIs list (octets q+8 to r)
This field contains the list of tracking areas for which the ciphering data set is applicable, encoded as octets 2 to n of the Tracking area identity list IE as specified in subclause 9.11.3.9. If the TAIs list is empty (as indicated by a zero length), the ciphering data set is applicable to the entire serving PLMN.
NOTE: The ciphering data set is always applicable to at least one of the E -UTRA Positioning SIB types or the NR Positioning SIB types.
9.11.3.18D Control plane service type

The purpose of the Control plane service type information element is to specify the purpose of the CONTROL PLANE SERVICE REQUEST message.

The Control plane service type information element is coded as shown in figure 9.11.3.18D.1 and table 9.11.3.18D.1.

The Control plane service type is a type 1 information element.





Table 9.9.3.18D.1: Control plane service type information element

Сс	ontro	ol plar	ne service type value (octet 1, bit 1 to 3)
Bi	ts		
3	2	1	
0	0	0	mobile originating request
0	0	1	mobile terminating request
0	1	0	emergency services
0	1	1	emergency services fallback
1	0	0	
	to		unused; shall be interpreted as " mobile originating request", if received by the network.
1	1	1	

9.11.3.19 Daylight saving time

See subclause 10.5.3.12 in 3GPP TS 24.008 [12].

9.11.3.20 De-registration type

The purpose of the De-registration type information element is to indicate the type of de-registration.

The De-registration type information element is coded as shown in figure 9.11.3.20.1 and table 9.11.3.20.1.

The De-registration type is a type 1 information element.



Figure 9.11.3.20.1: Deregistration type information element

Table 9.11.3.20.1: Deregistration type information element

S٧	vitch	off (octet 1, bit 4)
In Bit 4	the	UE to network direction:
0 1		Normal de-registration Switch off
In	the	network to UE direction bit 4 is spare. The network shall set this bit to zero.
Re	e-reg	jistration required (octet 1, bit 3)
In Bit 3	the	network to UE direction:
0 1		re-registration not required re-registration required
In	the	UE to network direction bit 3 is spare. The UE shall set this bit to zero.
Ac Bit	ces	s type (octet 1,bit 2, bit 1)
2	1	
0	1	3GPP access
1	0	Non-3GPP access
1	1	3GPP access and non-3GPP access
All	oth	er values are reserved.

- 9.11.3.21 Void
- 9.11.3.22 Void

9.11.3.23 Emergency number list

See subclause 10.5.3.13 in 3GPP TS 24.008 [12].

9.11.3.23A EPS bearer context status

See subclause 9.9.2.1 in 3GPP TS 24.301 [15].

9.11.3.24 EPS NAS message container

The purpose of the EPS NAS message container information element is to transport an EPS NAS message as specified in 3GPP TS 24.301 [15].

The EPS NAS message container information element is coded as shown in figure 9.11.3.24.1 and table 9.11.3.24.1.

The EPS NAS message container is a type 6 information element.



Figure 9.11.3.24.1: EPS NAS message container information element

Table 9.11.3.24.1: EPS NAS message container information element

EPS NAS message container (octet 4 to n) An EPS NAS message as specified in 3GPP TS 24.301 [15].

9.11.3.25 EPS NAS security algorithms

See subclause 9.9.3.23 in 3GPP TS 24.301 [15].

Extended emergency number list 9.11.3.26

See subclause 9.9.3.37A in 3GPP TS 24.301 [15].

9.11.3.26A Extended DRX parameters

See subclause 10.5.5.32 in 3GPP TS 24.008 [12].

9.11.3.27 Void

9.11.3.28 **IMEISV** request

See subclause 10.5.5.10 in 3GPP TS 24.008 [12].

9.11.3.29 LADN indication

The purpose of the LADN indication information element is to request the network for LADN information for specific LADN DNN(s) or to indicate a request for LADN information.

The LADN indication information element is coded as shown in figure 9.11.3.29.1 and table 9.11.3.29.1.

The LADN indication is a type 6 information element with a minimum length of 3 octets and a maximum length of 811 octets.

The LADN indication information element can contain a minimum of 0 and a maximum of 8 different LADN DNN values.

8	1	6	5	4	3	2	1	
		L	ADN inc	dication IE	1			octet 1
								octet 2
		Length of	LADN i	indication	contents	6		octet 3
								octet 4*
		L	ADN DN	N value	1			
								octet a*
								octet a+1*
		L	ADN DN	NN value	2			
								octet b*
								octet b+1*
								octet g*
								octet g+1*
		L	ADN DN	N value	n			
								octet h*
	Figure	9 11 3 20	λ 1 · Ι Δ	DN india	cation i	nformat	ion elei	ment

Figure 9.11.3.29.1: LADN indication information element

Value part of the LADN indication information element (octet 4 to h): The value part of the LADN indication information element consists of zero or more LADN DNN values. If the LADN indication information element conveys more than 8 LADN DNN values in this information element, the network shall consider the first 8 LADN DNN values and ignore the remaining octets of the information element. LADN DNN value: LADN DNN value: LADN DNN value is coded as the length and value part of DNN information element as specified in subclause 9.11.2.1B starting with the second octet.

Table 9.11.3.29.1: LADN indication information element

9.11.3.30 LADN information

~

The purpose of the LADN information information element is to provide the UE with the LADN service area for each available LADN in the current registration area or to delete the LADN information at the UE.

The LADN information information element is coded as shown in figure 9.11.3.30.1, figure 9.11.3.30.2 and table 9.11.3.30.1.

The LADN information is a type 6 information element with a minimum length of 3 octets and a maximum length of 1715 octets.

The LADN information information element can contain a minimum of 0 and a maximum of 8 different LADNs each including a DNN and a 5GS tracking area identity list.

~

~

8	1	0	Э	4	3	2	1	
		L	ADN info	rmation I	EI			octet 1
		Length of	LADN in	formatior	n conten	ts		octet 2 octet 3
			LAD	DN 1				octet 4
								octet a
			LAD	DN 2				octet a+1"
								_octet b*
								octet g*
			LAD	DN n				octet g+1 [*]
								octet h*
	Figure 9	9.11.3.30).1: LAC	ON infor	mation	informa	tion ele	ent



8 7 6 5 4 3 2 1

Length of DNN value	octet 4
	octet 5
DNN value	
	octet m
	octet m+1
5GS tracking area identity list	
	octet a

Figure 9.11.3.30.2: LADN

Table 9.11.3.30.1: LADN information information element

Value part of the LADN information information element (octet 4 to octet h) The value part of the LADN information information element consists of one or several LADNs. Each LADN (e.g. octet 4 to octet a) consists one DNN value and one 5GS tracking area identity list. The length of each LADN is determined by the length of DNN value field and the length of 5GS tracking area identity list field. The UE shall store the complete list as received. If more than 8 LADNs are included in this information element, the UE shall store the first 8 LADNs and ignore the remaining octets of the information element. DNN value (octet 5 to octet m): DNN value (octet 5 to octet m): SGS tracking area identity list (octet m+1 to octet a): 5GS tracking area identity list field is coded as the length and the value part of the 5GS Tracking area identity list field is coded as the length and the value part of the 5GS Tracking area identity list information element as specified in subclause 9.11.3.9 starting with the second octet.

9.11.3.31 MICO indication

The purpose of the MICO indication information element is to indicate the use of MICO mode or the re-negotiation of MICO mode.

The MICO indication information element is coded as shown in figure 9.11.3.31.1 and table 9.11.3.31.1.

The MICO indication is a type 1 information element.

	8	7	6	5	4	3	2	1	
Γ		MICO ind	ication IEI		0	0	SPRTI	RAAI	octet 1
					Spare	Spare			

|--|

Table 9.11.3.31.1: MICO indication

Re	Registration Area Allocation Indication (RAAI) (octet 1, bit 1)						
In Bit 1	he network to UE direction:						
0 1	all PLMN registration area not allocated all PLMN registration area allocated						
In	he UE to network direction bit 1 is spare. The UE shall set this bit to zero.						
Str	ctly Periodic Registration Timer Indication (SPRTI) (octet 1, bit 2)						
In Bit 2	he network to UE and the UE to network direction:						
0 1	strictly periodic registration timer not supported strictly periodic registration timer supported						
Bit	3 and 4 are spare and shall be coded as zero.						
NC	TE: In the network to UE direction in the CONFIGURATION UPDATE COMMAND message, bits 1 and 2 shall be coded as zero.						

9.11.3.31A MA PDU session information

The purpose of the MA PDU session information information element is to convey the MA-related information for the PDU session.

The MA PDU session information information element is coded as shown in figure 9.11.3.31A.1 and table 9.11.3.31A.1.

The MA PDU session information is a type 1 information element.

8	7	6	5	4	3	2	1	
MA PD	U sessior	n informa	ation IEI	MA P	DU sessi	ion inforn	nation	octet 1
					va	lue		

Table 9.11.3.31A.1: MA PDU session information information element

MA PDU session information value (octet 1, bit 1 to bit 4)Bits43000000001MA PDU session network upgrade is allowed

All other values are spare. If received they shall be ignored.

9.11.3.31B Mapped NSSAI

The purpose of the Mapped NSSAI information element is to transfer S-NSSAI(s) applicable in the HPLMN to the visited PLMN.

The Mapped NSSAI information element is coded as shown in figure 9.11.3.31B.1, figure 9.11.3.31B.2 and table 9.11.3.31B.1.

The Mapped NSSAI is a type 4 information element with a minimum length of 4 octets and a maximum length of 42 octets.

NOTE 1: The total number of S-NSSAI values in a requested mapped NSSAI cannot exceed eight.







Table 9.11.3.31B.1: Mapped NSSAI information element

Value part of the Mapped NSSAI information element (octet 3 to v) The value part of the Mapped NSSAI information element consists of one or more mapped S-NSSAI contents. Mapped S-NSSAI content: Length of S-NSSAI contents (octet 3) Mapped HPLMN Slice/service type (SST) (octet 4) This field contains the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SST value is mapped. The coding of the SST value part is defined in 3GPP TS 23.003 [4]. NOTE 1: Octet 4 (i.e. mapped HPLMN SST) shall always be included. Mapped HPLMN Slice differentiator (SD) (octet 5 to octet 7) This field contains a 24-bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SD value is mapped. The coding of the SD value part is defined in 3GPP TS 23.003 [4].

9.11.3.31C Mobile station classmark 2

See subclause 10.5.1.6 in 3GPP TS 24.008 [12].

9.11.3.32 NAS key set identifier

The NAS key set identifier is allocated by the network.

The NAS key set identifier information element is coded as shown in figure 9.11.3.32.1 and table 9.11.3.32.1.

The NAS key set identifier is a type 1 information element.

8	7	6	5	4	3	2	1	
NAS	S key set	identifier	· IEI	TSC	NAS I	key set id	entifier	octet 1

Figure 9.11.3.32.1: NAS key set identifier information element

Table 9.11.3.32.1: NAS key set identifier information element

Type of security context flag (TSC) (octet 1)
Bit
4 0 native security context (for KSI _{AMF}) 1 mapped security context (for KSI _{ASME})
TSC does not apply for NAS key set identifier value "111".
NAS key set identifier (octet 1)
Bits
3 2 1 0 0 0
hrough possible values for the NAS key set identifier
1 1 1 no key is available (UE to network); reserved (network to UE)

9.11.3.33 NAS message container

The purpose of the NAS message container IE is to encapsulate a plain 5GS NAS REGISTRATION REQUEST or SERVICE REQUEST message, or to encapsulate non-cleartext IEs of a CONTROL PLANE SERVICE REQUEST message.

The NAS message container information element is coded as shown in figure 9.11.3.33.1 and table 9.11.3.33.1.

The NAS message container is a type 6 information element.

8	7	6	5	4	3	2	1	
		NAS	message	e contai	ner IEI			octet 1
	Ler	ngth of NA	S messa	age cont	ainer cor	ntents		octet 2
								octet 3
								octet 4
		NAS me	essage co	ontainer	contents	6		
								octet n
	Figure 9	.11.3.33.	1: NAS	messa	ige cont	tainer info	ormation	ו element

Table 9.11.3.33.1: NAS message container information element

NAS message container contents (octet 4 to octet n); Max value of 65535 octets This IE can contain a REGISTRATION REQUEST message as defined in

subclause 5.5.1, or a SERVICE REQUEST message as defined in subclause 5.6.1, or non-cleartext IEs of a CONTROL PLANE SERVICE REQUEST message as defined in subclause 5.6.1.

9.11.3.34 NAS security algorithms

The purpose of the NAS security algorithms information element is to indicate the 5G algorithms to be used for ciphering and integrity protection.

The NAS security algorithms information element is coded as shown in figure 9.11.3.34.1 and table 9.11.3.34.1.

The NAS security algorithms is a type 3 information element with a length of 2 octets.



Figure 9.11.3.34.1: NAS security algorithms information element

Table 9.11.3.34.1: NAS security algorithms information element

Ту	pe c	of in	teg	rity protection algorithm (octet 2, bit 1 to 3)
Bit	s			
4	3	2	1	
0	0	0	0	5G integrity algorithm 5G-IA0 (null integrity protection algorithm)
0	0	0	1	5G integrity algorithm 128-5G-IA1
0	0	1	0	5G integrity algorithm 128-5G-IA2
0	0	1	1	5G integrity algorithm 128-5G-IA3
0	1	0	0	5G integrity algorithm 5G-IA4
0	1	0	1	5G integrity algorithm 5G-IA5
0	1	1	0	5G integrity algorithm 5G-IA6
0	1	1	1	5G integrity algorithm 5G-IA7
Typ Bit	pe c	of ci	phe	ring algorithm (octet 2, bit 5 to 7)
8	7	6	5	
0	0	0	0	5G encryption algorithm 5G-EA0 (null ciphering algorithm)
0	0	0	1	5G encryption algorithm 128-5G-EA1
0	0	1	0	5G encryption algorithm 128-5G-EA2
0	0	1	1	5G encryption algorithm 128-5G-EA3
0	1	0	0	5G encryption algorithm 5G-EA4
0	1	0	1	5G encryption algorithm 5G-EA5
0	1	1	0	5G encryption algorithm 5G-EA6
0	1	1	1	5G encryption algorithm 5G-EA7
All	oth	er v	alu	es are reserved.

9.11.3.35 Network name

See subclause 10.5.3.5a in 3GPP TS 24.008 [12].

9.11.3.36 Network slicing indication

The purpose of the Network slicing indication information element is to indicate additional information associated with network slicing in the generic UE configuration update procedure and the registration procedure, other than the user's configured NSSAI, allowed NSSAI, pending NSSAI and rejected NSSAI information.

The Network slicing indication information element is coded as shown in figure 9.11.3.36.1 and table 9.11.3.36.1.

The Network slicing indication is a type 1 information element.

8	7	6	5	4	3	2	1	
Netw	ork slicing	g indicatio	on IEI	0	0	DCNI	NSSCI	octet 1
		-		Spare	Spare			
		Figure	9.11.3.3	6.1: Net	work sl	icing in	dicatior	ì

Table 9.11.3.36.1: Network slicing indication

Netv	work slicing subscription change indication (NSSCI) (octet 1, bit 1)
Bit	
1	
0 1	Network slicing subscription not changed
1 I	Network slicing subscription changed
Def	ault configured NSSAI indication (DCNI) (octet 1, bit 2)
Bit	
2	
0	Requested NSSAI not created from default configured NSSAI
1	Requested NSSAI created from default configured NSSAI
]	
In th	ne UE to network direction bit 1 is spare. The UE shall set this bit to zero.
In th	ne network to UE direction bit 2 is spare. The network shall set this bit to zero.
Bits	3 and 4 are spare and shall be coded as zero.

9.11.3.36A Non-3GPP NW provided policies

See subclause 10.5.5.37 in 3GPP TS 24.008 [12].

9.11.3.37 NSSAI

The purpose of the NSSAI information element is to identify a collection of S-NSSAIs

The NSSAI information element is coded as shown in figure 9.11.3.37.1 and table 9.11.3.37.1.

The NSSAI is a type 4 information element with a minimum length of 4 octets and a maximum length of 146 octets.

NOTE: More than one S-NSSAIs in an NSSAI can have the same SST values, and optionally same SD values, which are associated with different mapped HPLMN SST values and optionally mapped HPLMN SD values.



Figure 9.11.3.37.1: NSSAI information element

Value part of the NSSAI information element (octet 3 to v)
The value part of the NSSAI information element consists of one or more S-NSSAI
values. Each S-NSSAI value consists of one S-NSSAI and optionally one mapped S-
NSSAI.
The recipient of this information element shall store the complete list received
(NOTE 1, NOTE 2, NOTE 3). If the NSSAI information element conveys an allowed
NSSAI and more than 8 S-NSSAI values are included in this information element, the
UE shall store the first 8 S-NSSAI values and ignore the remaining octets of the
information element.
If the NSSAI information element conveys a configured NSSAI or pending NSSAI and
more than 16 S-NSSAI values are included in this information element, the UE shall
store the first 16 S-NSSAI values and ignore the remaining octets of the information
element.
S-NSSAI value:
S-NSSAI value is coded as the length and value part of S-NSSAI information element
as specified in subclause 9.11.2.8 starting with the second octet.
NOTE 1: The total number of S-NSSAI values in a requested NSSAI shall not exceed
eight.
NOTE 2: The number of S-NSSAI values in an allowed NSSAI shall not exceed eight.
NOTE 3: The number of S-NSSAI values in a configured NSSAI or pending NSSAI
shall not exceed sixteen.

9.11.3.37A NSSAI inclusion mode

The purpose of the NSSAI inclusion mode information element is to indicate the NSSAI inclusion mode in which the UE shall operate.

The NSSAI inclusion mode is a type 1 information element.

The NSSAI inclusion mode information element is coded as shown in figure 9.11.3.37A.1 and table 9.11.3.37A.1.



	Table 9.11.3.37A	1: NSSAI	inclusion	mode	information	element
--	------------------	----------	-----------	------	-------------	---------

NS	SAI	I inclusion mode (octet 1, bit 1 to bit 2)	
Bit	s		
2	1		
0	0	NSSAI inclusion mode A	
0	1	NSSAI inclusion mode B	
1	0	NSSAI inclusion mode C	
1	1	NSSAI inclusion mode D	

9.11.3.38 Operator-defined access category definitions

The purpose of the Operator-defined access category definitions information element is to provide the UE with the operator-defined access category definitions or to delete the operator-defined access category definitions at the UE.

The Operator-defined access category definitions information element is coded as shown in figure 9.11.3.38.1, figure 9.11.3.38.2 and table 9.11.3.38.1.

The Operator-defined access category definitions is a type 6 information element with a minimum length of 3 octets, and maximum length of 8323 octets.



Figure 9.11.3.38.1: Operator-defined access category definitions information element

Leng	th of ope	rator-def	ined access category definition contents	octet 4
			Precedence value	octet 5
PSAC	0	0	Operator-defined access category number	octet 6
	Spare	Spare		
			Length of criteria	octet 7
				octet 8
			Criteria	
				octet a-1
0	0	0	Standardized access category	octet a*
Spare	Spare	Spare		

8 7 6 5 4 3 2 1

Figure 9.11.3.38.2: Operator-defined access category definition

Table 9.11.3.38.1: Operator-defined access category definitions information element

Value part of the Operator-defined access category definitions information element (octet 4 to h) The value part of the Operator-defined access category definitions information element consists of zero or no more than 32 operator-defined access category definition fields. Each operator-defined access category definition field is coded as described in figure 9.11.3.38.2. The length of each operator-defined access category definition field is determined by the length of operator-defined access category definition contents field. Operator-defined access category definition (octet 4 to octet a): Length of operator-defined access category definition contents (octet 4) Length of operator-defined access category definition contents indicates binary coded length of the operator-defined access category definition value field (octet 5 to octet a). Precedence value (octet 5) Bits 8 7 6 5 4 3 2 1 Precedence value 0 0 0 0 0 0 0 0 0 to 1 1 1 1 1 1 1 1 Precedence value 255 Operator-defined access category number (bits 5 to 1 of octet 6) Bits 5 4 3 2 1 0 0 0 0 0 Access category number 32 to 1 1 1 1 1 Access category number 63 Presence of standardized access category (PSAC) (bit 8 of octet 6) PSAC field indicates whether the standardized access category field is present or absent. Bit 8 Standardized access category field is not included 0 Standardized access category field is included Length of criteria (octet 7) Length of criteria field indicates binary coded length of the criteria field. Criteria (octets 8 to octet a-1) The criteria field contains one or more criteria components fields. Each criteria component field shall be encoded as a sequence of a one octet criteria type field and zero or more octets criteria value field. The criteria type field shall be transmitted first. Criteria type Bits 8 7 3 6 5 4 2 1 0 0 0 0 0 0 0 0 DNN type OS id + OS App Id type 0 0 0 0 0 0 0 1 0 0 0 0 0 1 0 S-NSSAI type 0 All other values are reserved. For "DNN type", the criteria value field shall be encoded as a sequence of one octet DNN length-value pair count field and one or more DNN length-value pair fields. The DNN length-value pair count field indicates the number of included DNN length-value pair fields. Each DNN length-value pair field is coded as a sequence of one octet DNN value length field and a DNN value field. The DNN value length field indicates the length in octets of the DNN value field. The DNN value field contains an APN as specified in 3GPP TS 23.003 [4].

Fo see Th Ea on val id IE ⁻ ap	r "O quei e ap ch a e oc ue l valu IF F plica	IS lo nce op io app ctet leng le fio RFC ation	d + of c d va id v OS gth f eld c 41 h id bcui	OS one llue app ield cont 22 [entil men	App Id type", the criteria value field shall be encoded as a octet app id value count field and one or more app id value fields. count field indicates the number of included app id value fields. a field is coded as a sequence of a sixteen octet OS id value field, id value length field and an OS app id value field. The OS app id indicates the length in octets of the OS app id value field. The OS app id value field in 35A]. The OS app id value field contains an OS specific fier. Coding of the OS app id value field is outside the scope of the t.
Fo oci val inc co val NS foll coi	r "S ue f lude ded ue f SAI owe ntair	-NS S-NS field ed S as field I va ed b ns a	SA SSA s. T S-N a se I. Th lue by th an S	I typ The SSA eque field field SD. S	e", the criteria value field shall be encoded as a sequence of one ngth-value pair count field and one or more S-NSSAI length-value S-NSSAI length-value pair count field indicates the number of I length-value pair fields. Each S-NSSAI length-value pair field is ence of one octet S-NSSAI value length field and an S-NSSAI -NSSAI value length field indicates the length in octets of the S- I. The S-NSSAI value field contains one octet SST field optionally octets SD field. The SST field contains a SST. The SD field SST and SD are specified in 3GPP TS 23.003 [4].
Sta Sta sta	anda anda inda dete	ardi ardi ardiz ardiz	zed zed zed ne	acc acc acc the	tess category (bits 5 to 1 of octet a) tess category field indicates the access category number of the ess category that is used in combination with the access identities establishment cause.
Bit	s	-	_		
5	4	3	2	1	
0	0	0	0	0	Access category number 0
_	_	to			
0	0	1	1	1	Access category number 7
0	1	0	0	1	Access category number 9
0	1	0	1	0	Access category number 10
All	oth	er v	alu	es a	re reserved.

9.11.3.39 Payload container

The purpose of the Payload container information element is to transport one or multiple payloads. If multiple payloads are transported, the associated information of each payload are also transported together with the payload.

The Payload container information element is coded as shown in figure 9.11.3.39.1, figure 9.11.3.39.1A, figure 9.11.3.39.1B, figure 9.11.3.39.2, figure 9.11.3.39.3, figure 9.11.3.39.4 and table 9.11.3.39.1.

The Payload container information element is a type 6 information element with a minimum length of 4 octets and a maximum length of 65538 octets.

8	7	6	5	4	3	2	1	
		Pa	ayload co	ontainer	IEI			octet 1
		Longth of	navload	contain	or conto	ote		octet 2
		Lengin of	payloau	contain		115		octet 3
		Povl	and contr	ainar ca	ntonto			octet 4
		Гаую			Internis			octet n

Figure 9.11.3.39.1: Payload container information element





Figure 9.11.3.39.2: Payload container contents with Payload container type "Multiple payloads"

octet n

772

8 7 6 5 4 3 2 1 Length of Payload container entry octet xi +1 octet xi +2 Number of optional IEs Payload container type octet xi +3 octet xi +4 Optional IE 1 octet y2 octet y2+1 **Optional IE 2** octet y3 ... octet yj+1 Optional IE j octet z octet z+1

Figure 9.11.3.39.3: Payload container entry

0 1 0 5 4 5 2	8	7	6	5	4	3	2	1
---------------	---	---	---	---	---	---	---	---

Payload container entry contents

	1
	octet y2
Value of optional IE	octet xi +6
	octet XI +5
Type of optional IE	octet xi +4
	-

Figure 9.11.3.39.4: Optional IE

Table 9.11.3.39.1: Payload container information element

Payload container contents (octet 4 to octet n); max value of 65535 octets

If the payload container type is set to "N1 SM information" and is included in the UL NAS TRANSPORT or DL NAS TRANSPORT message, the payload container contents contain a 5GSM message as defined in subclause 8.3.

If the payload container type is set to "SOR transparent container" and is included in the DL NAS TRANSPORT message, the payload container contents are coded the same way as the contents of the SOR transparent container IE (see subclause 9.11.3.51) for SOR data type is set to value "0" except that the first three octets are not included.

If the payload container type is set to "SOR transparent container" and is included in the UL NAS TRANSPORT message, the payload container contents are coded the same way as the contents of the SOR transparent container IE (see subclause 9.11.3.51) for SOR data type is set to value "1" except that the first three octets are not included.

If the payload container type is set to "UE policy container" and is included in the DL NAS TRANSPORT, UL NAS TRANSPORT or REGISTRATION REQUEST message, the payload container contents are coded as defined in subclause Annex D.

If the payload container type is set to "UE parameters update transparent container" and is included in the DL NAS TRANSPORT message, the payload container contents are coded the same way as the contents of the UE parameters update transparent container IE (see subclause 9.11.3.53A) for UE parameters update data type is set to value "0" except that the first three octets are not included.

If the payload container type is set to "UE parameters update transparent container" and is included in the UL NAS TRANSPORT message, the payload container contents are coded the same way as the contents of the UE parameters update transparent container IE (see subclause 9.11.3.53A) for UE parameters update data type is set to value "1" except that the first three octets are not included.

If the payload container type is set to "SMS" and is included in the UL NAS TRANSPORT or DL NAS TRANSPORT message, the payload container contents contain an SMS message (i.e. CP-DATA, CP-ACK or CP-ERROR) as defined in subclause 7.2 in 3GPP TS 24.011 [13].

If the payload container type is set to "CIoT user data container" and is included in the UL NAS TRANSPORT, DL NAS TRANSPORT or CONTROL PLANE SERVICE REQUEST message, the payload container contents are coded the same way as the contents of the user data container IE (see subclause 9.9.4.24 in 3GPP TS 24.301 [15]) except that the first three octets are not included.

If the payload container type is set to "SMS" and is included in the CONTROL PLANE SERVICE REQUEST message, the payload container contents are coded the same way as the contents of the NAS message container IE (see subclause 9.9.3.22 in 3GPP TS 24.301 [15]) except that the first two octets are not included.

If the payload container type is set to "Location services message container" and is included in the UL NAS TRANSPORT, DL NAS TRANSPORT or CONTROL PLANE SERVICE REQUEST message, the payload container contents include location services message payload.

If the payload container type is set to "LTE Positioning Protocol (LPP) message container" and is included in the UL NAS TRANSPORT or DL NAS TRANSPORT message, the payload container contents include LPP message payload.

If the payload container type is set to "Service-level-AA container" and is included in the UL NAS TRANSPORT or DL NAS TRANSPORT message, the payload container contents are coded the same way as the contents of service-level-AA container (see subclause 9.11.2.10).

If the payload container type is set to "Event notification", the payload container contents include one or more event notification indicators.

Туре о	f even	t noti	ncal		ator n (octet I+1)
Bits	6 5	4	,	0 1	
0 0	0 0	4 0	0 (2 1	"SRVCC handover cancelled, IMS session re- establishment required" indicator
0 0	0 0	0	0 (01	Unused shall be ignored if received by the UE
1 1	1 1	1	1	1 1	
If the ty session indicat	ype of n re-es or sha	an ev stabli: Il not	vent shm be i	notificat ent requ ncluded.	ion indicator is set to "SRVCC handover cancelled, IMS ired" indicator, the value of the event notification
If the p represe entry c figure s figure s	ents th ontent 9.11.3 9.11.3	d con le tota s fiel .39.2 .39.3	taine al nu d is d , with and	er type is imber of coded as n each p figure 9	s set to "Multiple payloads", the number of entries field payload container entries, and the payload container s a list of payload container entry according to ayload container entry is coded according to .11.3.39.4.
The co	ding o	f Pay	loac	l contain	er contents is dependent on the particular application.
Payloa	d cont	ainer	ent	ry	
For ea payloa payloa numbe the pay handlir the opt	ch pay d cont d cont er of op yload o ngs for tional l	load ainer ainer otiona conta optic Es in	conf type con al IEs iner onal clud	tainer en e value a tents fiel s field re entry co IEs spec ed in the	try, the payload container type field represents the as described in subclause 9.11.3.40, the coding of ld is dependent on the particular application, and the presents the total number of optional IEs associated with ntents field in the payload container entry. The error cified in subclauses 7.6.1, 7.6.3 and 7.7.1 shall apply to a payload container entry.
Option	al IEs				
Type o This fie	f optio	nal II Itains	∃ (oo s the	ctet xi +4 IEI of th	e optional IE.
Length This fie	of op eld ind	tional icates	IE (s bin	octet xi+ ary code	5) ed length of the value of the optional IE entry.
Value o This fie informa include 5 to 8 s value p bits 5 t value p bits 5 t	of option ation e add, the shall b part of o 8 sh part of o 8 sh	onal I Itains Ieme value e coc the F all be the N all be	IE (o the e pa ded a Relea coc //A P e coc	ctet xi+6 value of ased on rt of the as zero. ase assis led as ze DU sess led as ze	to octet y2) the optional IE entry with the value part of the referred following optional IE reference. If the Request type is Request type shall be encoded in the bits 1 to 4 and bits If the Release assistance indication is included, the stance indication shall be encoded in the bits 1 to 4 and ero. If the MA PDU session information is included, the sion information shall be encoded in the bits 1 to 4 and ero.
IEI	Optic	nal II	Ena	me	Optional IE reference PDU session identity 2 (see subclause 9.11.3.41)

9.11.3.40 Payload container type

The purpose of the Payload container type information element indicates type of payload included in the payload container information element.

The Payload container type information element is coded as shown in figure 9.11.3.40.1 and table 9.11.3.40.1.

The Payload container type information element is a type 1 information element.

8	7	6	5	4	3	2	1	
	Payload co	ontainer t	ype	Pa	ayload co	ntainer ty	pe value	octet 1
		EI						
	E'	44 0 40				· · · · · · · · · · · · · · · · · · ·		- I

Figure 9.11.3.40.1: Payload container type information element

Table 9.11.3.40.1: Payload	container type	information	element
----------------------------	----------------	-------------	---------

Pa	yloa	ad c	onta	ainer type value (octet 1)
Bit	S			
4	3	2	1	
0	0	0	1	N1 SM information
0	0	1	0	SMS
0	0	1	1	LTE Positioning Protocol (LPP) message container
0	1	0	0	SOR transparent container
0	1	0	1	UE policy container
0	1	1	0	UE parameters update transparent container
0	1	1	1	Location services message container (see 3GPP TS 23.273 [6B])
1	0	0	0	CloT user data container
1	0	0	1	Service-level-AA container
1	0	1	0	Event notification
1	1	1	1	Multiple payloads
All	oth	er v	alu	es are reserved.
NC)TE	:	Th	e value "Multiple payloads" is only used when the Payload container
			COI	ntents in figure 9.11.3.39.1 contains multiple payloads as shown in
			fig	ure 9.11.3.39.2.

9.11.3.41 PDU session identity 2

The purpose of the PDU session identity 2 information element is to indicate the identity of a PDU session in a 5GMM message.

The PDU session identity 2 information element is coded as shown in figure 9.11.3.41.1 and table 9.11.3.41.1.

The PDU session identity 2 is a type 3 information element with a length of 2 octets .

8	1	6	5	4	3	2	1	
PDU session identity 2 IEI oct							octet 1	
PDU session identity 2 value octe							octet 2	

Figure 9.11.3.41.1: PDU session identity 2 information element

Table 9.11.3.41.1: PDU sessio	n identity 2	2 information	element
-------------------------------	--------------	---------------	---------

PDU session identity 2 value (octet 2) The coding of the DU session identity 2 value is identical to the coding of the PDU session identity value as defined in 3GPP TS 24.007 [11].

9.11.3.42 PDU session reactivation result

The purpose of the PDU session reactivation result information element is to indicate the result of establishments of user-plane resources of PDU sessions.

The PDU session reactivation result information element is coded as shown in figure 9.11.3.42.1 and table 9.11.3.42.1.

The PDU session reactivation result is a type 4 information element with minimum length of 4 octets and maximum length of 34 octets.

	1	2	3	4	5	6	7	8
octet 1			result IEI	ctivation i	sion read	PDU ses		
octet 2		sult	vation res	on reactiv	DU sessi	ngth of Pl	Ler	
octet 3	PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI
	(0)	(1)	(2)	(3)	(4)	(5)	(6)	(7)
octet 4	PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI
	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
octet 5* -34'	0	0	0	0	0	0	0	0
				are	Sp			

Figure 9.11.3.42.1: PDU session reactivation result information element

Table 9.11.3.42.1: PDL	J session reactivation	n result information	element
------------------------	------------------------	----------------------	---------

PSI(x) shall be coded as follows:

PSI(0):

Bit 0 of octet 3 is spare and shall be coded as zero.

PSI(1) - PSI(15):

0 indicates establishment of user-plane resources of the PDU session was not requested in the Uplink data status IE or establishment of user-plane resources of the PDU session was not allowed in the Allowed PDU session status IE or establishment of user-plane resource of the PDU session is successful.

1 indicates either establishment of user-plane resources of the PDU session was requested in the Uplink data status IE but establishment of user-plane resource of the PDU session is not successful or indicates establishment of user-plane resources of the PDU session was allowed in the Allowed PDU session status IE but establishment of user-plane resource of the PDU session is either not performed or not successful.

All bits in octet 5 to 34 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.3.43 PDU session reactivation result error cause

The purpose of the PDU session reactivation result error cause information element is to indicate error causes for PDU session ID(s) where there was a failure to establish the user-plane resources.

The PDU session reactivation result error cause information element is coded as shown in figure 9.11.3.43.1 and table 9.11.3.43.1.

The PDU session reactivation result error cause is a type 6 information element with a minimum length of 5 octets and a maximum length of 515 octets.





Table 9.11.3.43.1: PDU session reactivation result error cause information element

PDU session ID is coded same as PDU session ID IE (see subclause 9.4).

The cause value is coded same as second octet of 5GMM cause information element (see subclause 9.11.3.2).

9.11.3.44 PDU session status

The purpose of the PDU session status information element is to indicate the state of each PDU session that can be identified by a PDU session identity.

The PDU session status information element is coded as shown in figure 9.11.3.44.1 and table 9.11.3.44.1.

The PDU session status information element is a type 4 information element with minimum length of 4 octets and a maximum length of 34 octets.

8	7	6	5	4	3	2	1	
		PD	U sessic	n status	IEI			octet 1
	Le	ength of I	DU ses	sion statu	is conten	ts		octet 2
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 3
(7)	(6)	(5)	(4)	(3)	(2)	(1)	(0)	
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 4
(15)	(14)	(13)	(12)	(11)	(10)	(9)	(8)	
0	0	0	0	0	0	0	0	octet 5*-
			spa	are				34*

Figure 9.11.3.44.1: PDU session status information element

Fable 9.11.3.44.1: PDL	session status	information	element
------------------------	----------------	-------------	---------

PSI(x) shall be coded as follows:

PSI(0):

Bit 1 of octet 3 is spare and shall be coded as zero.

PSI(1) – PSI(15):

0 indicates that the 5GSM state of the corresponding PDU session is PDU SESSION INACTIVE.

1 indicates that the 5GSM state of the corresponding PDU session is not PDU SESSION INACTIVE

All bits in octet 5 to 34 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.3.45 PLMN list

See subclause 10.5.1.13 in 3GPP TS 24.008 [12].

9.11.3.46 Rejected NSSAI

The purpose of the Rejected NSSAI information element is to identify a collection of rejected S-NSSAIs.

The Rejected NSSAI information element is coded as shown in figure 9.11.3.46.1, figure 9.11.3.46.2 and table 9.11.3.46.1.

The Rejected NSSAI is a type 4 information element with a minimum length of 4 octets and a maximum length of 42 octets.





Figure 9.11.3.46.2: Rejected S-NSSAI

Table 9.11.3.46.1: Rejected NSSAI information element

Value part of the Rejected NSSAI information element (octet 3 to v)
The value part of the Rejected NSSAI information element consists of one or more rejected S-NSSAIs. Each rejected S-NSSAI consists of one S-NSSAI and an associated cause value. The length of each rejected S-NSSAI can be determined by the 'length of rejected S-NSSAI' field in the first octet of the rejected S-NSSAI. The UE shall store the complete list received (NOTE 0). If more than 8 rejected S-NSSAIs are included in this information element, the UE shall store the first 8 rejected S-NSSAIs and ignore the remaining octets of the information element.
Rejected S-NSSAI:
Cause value (octet 3)
4 3 2 1 0 0 0 S-NSSAI not available in the current PLMN or SNPN 0 0 1 S-NSSAI not available in the current registration area 0 0 1 0 0 1 0 S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization All other values are reserved. S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization
Slice/service type (SST) (octet 4) This field contains the 8 bit SST value. The coding of the SST value part is defined in 3GPP TS 23.003 [4]. (NOTE 2)
Slice differentiator (SD) (octet 5 to octet 7) This field contains the 24 bit SD value. The coding of the SD value part is defined in 3GPP TS 23.003 [4]. (NOTE 3)
 NOTE 0: The number of rejected S-NSSAI(s) shall not exceed eight. NOTE 1: If octet 5 is included, then octet 6 and octet 7 shall be included. NOTE 2: If the Cause value is "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization", this field shall contain the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPI MN.
NOTE 3: If the Cause value is "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization", this field shall contain the 24 bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN.

9.11.3.46A Release assistance indication

See subclause 9.9.4.25 in 3GPP TS 24.301 [15].

9.11.3.47 Request type

The purpose of the Request type information element is to indicate the type of the 5GSM message.

The Request type information element is coded as shown in figure 9.11.3.47.1 and table 9.11.3.47.1.

The Request type is a type 1 information element.







Re	ques	st ty	vpe value (octet 1, bit 1 to bit 4)							
Bits	S									
3	2	1								
0	0	1	initial request							
0	1	0	existing PDU session							
0	1	1	initial emergency request							
1	0	0	existing emergency PDU session							
1	0	1	modification request							
1	1	0	MA PDU request (NOTE)							
1	1	1	reserved							
All the	All other values are unused and shall be interpreted as "initial request", if received by the network.									
NC	NOTE: This value shall be interpreted as "initial request", if received by a network not supporting MA PDU sessions.									

9.11.3.48 S1 UE network capability

See subclause 9.9.3.34 in 3GPP TS 24.301 [15].

9.11.3.48A S1 UE security capability

See subclause 9.9.3.36 in 3GPP TS 24.301 [15].

9.11.3.49 Service area list

The purpose of the Service area list information element is to transfer a list of allowed tracking areas for an allowed area or a list of non-allowed tracking areas for a non-allowed area from the network to the UE.

The coding of the information element allows combining different types of lists. The lists of type "00" and "01" allow a more compact encoding, when the different TAIs are sharing the PLMN identity. The lists of type "11" indicate all TAIs of the PLMNs in the registration area are allowed area.

The Service area list information element is coded as shown in figure 9.11.3.49.1, figure 9.11.3.49.2, figure 9.11.3.49.3, figure 9.11.3.49.4, figure 9.11.3.49.5 and table 9.11.3.49.1.

The Service area list is a type 4 information element with a minimum length of 6 octets and a maximum length of 114 octets. The list can contain a maximum of 16 different tracking area identities.

8	7	6	5	4	3	2	1		
	Service area list IEI								
		Length c	of service	area list	contents			octet 2	
								octet 3	
		Par	tial servi	ce area lis	st 1				
								octet i	
								octet i+1*	
		Par	tial servi	ce area li	st 2				
								octet I*	
								octet I+1*	
								octet m*	
		-						octet m+1	
		Par	tial servi	ce area li	st p			+	
								octet n°	



8	7	6	5	4	3	2	1	_
Allowed type	Туре	e of list		Numt	er of ele	ements		octet 1
	MCC	digit 2			МСС	digit 1		octet 2
	MNC	digit 3			MCC	digit 3		octet 3
	MNC	digit 2			MNC	digit 1		octet 4
			TA	C 1				octet 5
			TAC 1 (c	ontinued)			octet 6
		-	TAC 1 (c	ontinued				octet 7
			•					
	octet 3k+2							
TAC k (continued)								octet 3k+3
TAC k (continued)								octet 3k+4

Figure 9.11.3.49.2: Partial service area list – type of list = "00"



8	7 6	5	4	3	2	1	_
Allowed	Type of list		Numbe	r of elem	nents		octet 1
type							-
	MCC digit 2			MCC d	igit 1		octet 2
	MNC digit 3			MCC d	igit 3		octet 3
	MNC digit 2			MNC d	igit 1		octet 4
		TA	C 1				octet 5
		TAC 1 (co	ontinued)				octet 6
		TAC 1 (co	ontinued)				octet 7
	MCC digit 2			MCC d	igit 1		octet 8*
	MNC digit 3			MCC d	igit 3		octet 9*
	MNC digit 2			MNC d	igit 1		octet 10*
		TA	C 2				octet 11*
		TAC 2 (co	ontinued)				octet 12*
		TAC 2 (co	ontinued)				octet 13*
							-
	MCC digit 2			MCC d	igit 1		octet 6k-4*
	MNC digit 3			MCC d	igit 3		octet 6k-3*
	MNC digit 2			MNC d	igit 1		octet 6k-2*
		TA	Ck				octet 6k*-1
		TAC k (co	ontinued)				octet 6k*
		TAC k (co	ontinued)				octet 6k+1*

Figure 9.11.3.49.4: Partial service area list – type of list = "10"

8	7	6	5	4	3	2	1	
Allowed	Туре	of list		Numb	er of ele	ments		octet 1
type				1				_
	MCC o	digit 2			мсс	digit 1		octet 2
	MNC	diait 3			MCC	diait 3		octet 3
		uigit 3			NICC	uigit 3		OCIEL 3
	MNC (digit 2			MNC	digit 1		octet 4
Ei/		11 2 /0	5. Darti	al convid	o araa	lict two	na af li	ct _ "11

Figure 9.11.3.49.5: Partial service area list – type of list = "11"

Table 9.11.3.49.1: Service area list information element

Value part of the Service area list information element (octets 3 to n) The value part of the Service area list information element consists of one or several partial service area lists. The length of each partial service area list can be determined from the 'type of list' field and the 'number of elements' field in the first octet of the partial service area list. The "Allowed type" fields in all the partial service area lists shall have the same value. For allowed type "0", TAIs contained in all partial service area lists are in the allowed area. For allowed type "1", TAIs contained in all partial service area lists are in the nonallowed area. The UE shall store the complete list received. If more than 16 TAIs are included in this information element, the UE shall store the first 16 TAIs and ignore the remaining octets of the information element. Partial service area list: Allowed type (octet 1) Bit 8 0 TAIs in the list are in the allowed area 1 TAIs in the list are in the non-allowed area Type of list (octet 1) Bits 76 0 0 list of TACs belonging to one PLMN, with non-consecutive TAC values 0 1 list of TACs belonging to one PLMN, with consecutive TAC values 1 0 list of TAIs belonging to different PLMNs (see NOTE) 1 All TAIs belonging to the PLMNs in the registration area are in the allowed area 1 Number of elements (octet 1) Bits 5 4 3 2 1 0 0 0 0 0 1 element 0 0 0 0 1 2 elements 0 0 0 1 0 3 elements to 0 1 0 1 14 elements 1 15 elements 0 1 1 1 0 0 1 1 1 1 16 elements All other values are unused and shall be interpreted as 16, if received by the UE. For type of list = "00" and number of elements = k: octets 2 to 4 contain the MCC+MNC, and for j = 1, ..., k: octets 3j+2 to 3j+4 contain the TAC of the j-th TAI belonging to the partial list, For type of list = "01" and number of elements = k: octets 2 to 4 contain the MCC+MNC, and octets 5 to 7 contain the TAC of the first TAI belonging to the partial list. The TAC values of the other k-1 TAIs are TAC+1, TAC+2, ..., TAC+k-1. For type of list = "10" and number of elements = k: for j = 1, ..., k. octets 6j-4 to 6j-2 contain the MCC+MNC, and octets 6j-1 to 6j+1 contain the TAC of the j-th TAI belonging to the partial list. For type of list = "11": Allowed type shall be coded as "0" and number of elements shall be ignored, and octets 2 to 4 containing the MCC+MNC can be ignored. If allowed type is coded as "1", it shall be interpreted as "0".

Coding us	ing full hexadecimal representation may be used. The tracking area code if 3 octets.								
In the TA octet the I	C field bit 8 of the first octet is the most significant bit and bit 1 of the third east significant bit. g of the tracking area code is the responsibility of each administration								
TAC, Trac	cking area code								
The codin be used. only two c	g of this field is the responsibility of each administration but BCD coding shall The MNC shall consist of 2 or 3 digits. If a network operator decides to use ligits in the MNC, MNC digit 3 shall be coded as "1111".								
MNC, Mol	bile network code								
The MCC	The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.								
MCC, Mol	bile country code								

9.11.3.50 Service type

The purpose of the service type information element is to specify the purpose of the service request procedure.

The service type is a type 1 information element.

The service type information element is coded as shown in figure 9.11.3.50.1 and table 9.11.3.50.1.



Table 9.11.3.50.1: Service type information element

S	ervic	e ty	/pe	value (octet 1)						
s	Service type value									
В	ts		•							
4	3	2	1							
0	0	0	0	signalling						
0	0	0	1	data						
0	0	1	0	mobile terminated services						
0	0	1	1	emergency services						
0	1	0	0	emergency services fallback						
0	1	0	1	high priority access						
0	1	1	0	elevated signalling						
0	1	1	1	unused; shall be interpreted as "signalling", if received by the network						
1	0	0	0	unused; shall be interpreted as "signalling", if received by the network						
1	0	0	1	unused; shall be interpreted as "data", if received by the network						
1	0	1	0	unused; shall be interpreted as "data", if received by the network						
1	0	1	1	unused; shall be interpreted as "data", if received by the network						
А	All other values are reserved.									

9.11.3.50A SMS indication

The purpose of the SMS indication information element is to indicate that the ability for the UE to use SMS over NAS has changed.

The SMS indication information element is coded as shown in figure 9.11.3.50A.1 and table 9.11.3.50A.1.

The SMS indication is a type 1 information element.



Table	9.11.	3.50A.	1: \$	SMS	indication
-------	-------	--------	-------	-----	------------

SMS availability indication (SAI) (octet 1)
Bit
1 0 SMS over NAS not available
1 SMS over NAS available
Bits 2, 3 and 4 are spare and shall be coded as zero,

9.11.3.51 SOR transparent container

The purpose of the SOR transparent container information element in the REGISTRATION ACCEPT message is to provide the list of preferred PLMN/access technology combinations (or HPLMN indication that 'no change of the "Operator Controlled PLMN Selector with Access Technology" list stored in the UE is needed and thus no list of preferred PLMN/access technology combinations is provided'), or a secured packet (see 3GPP TS 23.122 [5] annex C) and optional indication of an acknowledgement request, SOR-CMCI, request the storage of the received SOR-CMCI in the ME, and SOR-SNPN-SI (or subscribed SNPN or HPLMN indication that 'no change of the SOR-SNPN-SI stored in the UE is needed and thus no SOR-SNPN-SI is provided'). The purpose of the SOR transparent container information element in the REGISTRATION COMPLETE message is to indicate the UE acknowledgement of successful reception of the SOR transparent container IE in the REGISTRATION ACCEPT message as well as to indicate the ME support of SOR-CMCI and the ME support of SOR-SNPN-SI.

NOTE: When used in NAS transport procedure, the contents of the SOR transparent container information element in the Payload container IE of the DL NAS TRANSPORT message are used to provide the list of preferred PLMN/access technology combinations and optional indication of an acknowledgement request, SOR-CMCI, request the storage of the received SOR-CMCI in the ME, and SOR-SNPN-SI. The contents of the SOR transparent container information element in the Payload container IE of the UL NAS TRANSPORT message are used to indicate the UE acknowledgement of successful reception of the SOR transparent container IE in the DL NAS TRANSPORT message as well as to indicate the ME support of SOR-CMCI and the ME support of SOR-SNPN-SI.

The SOR transparent container information element is coded as shown in figure 9.11.3.51.1, figure 9.11.3.51.2, figure 9.11.3.51.3, figure 9.11.3.51.4, figure 9.11.3.51.5, figure 9.11.3.51.6, figure 9.11.3.51.7, figure 9.11.3.51.8, figure 9.11.3.51.9, figure 9.11.3.51.10, figure 9.11.3.51.11, figure 9.11.3.51.12, figure 9.11.3.51.13, table 9.11.3.51.1, table 9.11.3.51.2, table 9.11.3.51.3, table 9.11.3.51.4, table 9.11.3.51.5 and table 9.11.3.51.6.

The SOR transparent container is a type 6 information element with a minimum length of 20 octets.

8	7	6	5	4	3	2	1		
		SOR t	ranspare	ent contai	ner IEI			octet 1	
	Length of SOR transparent container contents								
		-						octet 3	
			SOR I	header				octet 4	
			SOR-M	AC-IAUSF				octet 5-20	
			Coun	tersor				octet 21-22	
			Secure	d packet				octet 23* -	
				-				n*	



8	7	6	5	4	3	2	1	
		SOR t	ranspare	nt conta	iner IEI			octet 1
	Leng	th of SOF	R transpa	rent con	tainer con	tents		octet 2
	-							octet 3
			SOR h	neader				octet 4
			SOR-M	AC-IAUSF				octet 5-20
			Coun	tersor				octet 21-22
		PLMN ID	and acc	ess tech	nology list			octet 23*-m*

Figure 9.11.3.51.2: SOR transparent container information element for list type with value "1", SOR data type with value "0", and additional parameters with value "0"





PLMN ID 1	octet 23*-
	25*
access technology identifier 1	
access technology identifier i	
	27*
PLMN ID n	octet
	(18+5*n)*-
	(20,5*n)*
	(20+5 11)
access technology identifier n	octet
	(21+5*n)*-
	(22+5*n)*
Figure 9 11 3 51 3. PI MN ID and access technology	list (m-22 \pm 5*n)
i igure 3. i 1.3.3 i.3. F Livin ID and access technology	nsi (m=22+5 m)

8	7	6	5	4	3	2	1		
		SOR t	ranspare	ent contai	ner IEI			octet 1	
Length of SOR transparent container contents									
SOR header									

SOR-MAC-I _{UE}	octet 5 - 20
 51 4: COB transportant container information alament for COE	

Figure 9.11.3.51.4: SOR transparent container information element for SOR data type with value "1"

8	7	6	5	4	3	2	1	
0	0	0	AP	ACK	List	List	SOR	octet 4
Spare	Spare	Spare			type	indication	data	
							type	
						• • •		

Figure 9.11.3.51.5: SOR header for SOR data type with value "0"

8	7	6	5	4	3	2	1	
0	0	0	0	0	MSS	MSSI	SOR	octet 4
Spare	Spare	Spare	Spare	Spare	NPN		data	
					SI		type	

Figure 9.11.3.51.6: SOR header for SOR data type with value "1"

Table 9.11.3.51.1: SOR transparent container information element

SOR-MAC-IAUSF (see NOTE 1), SOR-MAC-IUE (see NOTE 2) and Countersor (see NOTE 1) are coded as specified in 3GPP TS 33.501 [24]. SOR data type (octet 4, bit 1) 0 The SOR transparent container carries steering of roaming information. The SOR transparent container carries acknowledgement of successful reception of 1 the steering of roaming information. List indication (octet 4, bit 2) (see NOTE 1 and NOTE 5) 0 HPLMN indication that 'no change of the "Operator Controlled PLMN Selector with Access Technology" list stored in the UE is needed and thus no list of preferred PLMN/access technology combinations is provided' 1 list of preferred PLMN/access technology combinations is provided List type (octet 4, bit 3) (see NOTE 1) 0 The list type is a secured packet. 1 The list type is a "PLMN ID and access technology list". Acknowledgement (ACK) value (octet 4, bit 4) (see NOTE 1) 0 acknowledgement not requested 1 acknowledgement requested Additional parameters (AP) value (octet 4, bit 5) Bit 5 0 Additional parameters not included 1 Additional parameters included (see NOTE 3) If the SOR data type is set to value "0", the list type bit is set to value "1", and the additional parameters bit is set to value "1" then: - the octet o is present. - if the list indication bit is set to "0" then the PLMN ID and access technology list field and the length of PLMN ID and access technology list field are absent. - if the list indication bit is set to "1" then the PLMN ID and access technology list field and the length of PLMN ID and access technology list field are present. The secure packet is coded as specified in 3GPP TS 31.115 [22B]. (see NOTE 1) The PLMN ID and access technology list consists of PLMN ID and access technology identifier and are coded as specified in 3GPP TS 31.102 [22] subclause 4.2.5. The PLMN ID and access technology identifier are provided in decreasing order of priority, i.e. PLMN ID 1 indicates highest priority and PLMN ID n indicates lowest priority. The PLMN ID and access technology list contains at minimum zero and at maximum 16 (decimal) PLMN IDs and access technology identifiers. (see NOTE 1) ME support of SOR-CMCI indicator (MSSI) value (octet 4, bit 2) (see NOTE 2, NOTE 4) 0 SOR-CMCI not supported by the ME 1 SOR-CMCI supported by the ME ME support of SOR-SNPN-SI indicator (MSSNPNSI) value (octet 4, bit 3) (see NOTE 2, NOTE 6) 0 SOR-SNPN-SI not supported by the ME 1 SOR-SNPN-SI supported by the ME SOR-CMCI indicator (SI) value (octet o, bit 1) Bit 0 SOR-CMCI absent 1 SOR-CMCI present If the SOR-CMCI indicator bit is set to "SOR-CMCI present", the SOR-CMCI field is present. If the SI bit is set to "SOR-CMCI absent", the SOR-CMCI field is absent. Store SOR-CMCI in ME indicator (SSCMI) value (octet o, bit 2) Bit 2 0 Do not store SOR-CMCI in ME 1 Store SOR-CMCI in ME

SOR-CMCI (octet o+1 to octet p) The SOR-CMCI field is coded according to figure 9.11.3.51.7 and table 9.11.3.51.2. SOR-SNPN-SI indicator (SSSI) value (octet o, bit 3) Bit 3 0 subscribed SNPN or HPLMN indication that 'no change of the SOR-SNPN-SI stored in the UE is needed and thus no SOR-SNPN-SI is provided' 1 SOR-SNPN-SI present If the SSSI bit is set to "SOR-SNPN-SI present", the SOR-SNPN-SI field is present. If the SSSI bit is set to "subscribed SNPN or HPLMN indication that 'no change of the SOR-SNPN-SI stored in the UE is needed and thus no SOR-SNPN-SI is provided", the SOR-SNPN-SI is absent. NOTE 1: This bit or field applies for SOR header with SOR data type with value "0". NOTE 2: This bit or field applies for SOR header with SOR data type with value "1". NOTE 3: Additional parameters can be set to value "1" only when the ME supports SOR-CMCI or SOR-SNPN-SI, and the list type bit is set to value "1". NOTE 4: The "SOR-CMCI supported by the ME" is not set by a UE compliant to an earlier release of the specification. NOTE 5: This bit or field applies for SOR header with list type with value "1". The "SOR-SNPN-SI supported by the ME" may only be set by a UE which NOTE 6: supports access to an SNPN using credentials from a credentials holder and which is not operating in SNPN access operation mode.

8	7	6	5	4	3	2	1		
			(octet (o+1)	
	octot (o 12)								
								0 ctet (0+2)	
			SOR-CM	ICI rule 1	1			00101 (010)	
								octet q*	
			SOR-CN	ACI rule 2	2			ootot r*	
								octet $(r+1)^*$	
								00101 (1+1)	
								octet s*	
								octet (s+1)*	
			SOR-CN	ICI rule r	ו			1 - 1 - +	
<u> </u>								octet p^	

Figure 9.11.3.51.7: SOR-CMCI

Table 9.11.3.51.2: SOR-CMCI

SOR-CMCI rule:

The SOR-CMCI rule is coded according to figure 9.11.3.51.8 and table 9.11.3.51.3.

If the length of SOR-CMCI contents field indicates a length bigger than indicated in figure 9.11.3.51.7, receiving entity shall ignore any superfluous octets located at the end of the SOR-CMCI.






Tsor-cm timer value The Tsor-cm timer value field is coded according to octet 2 of the GPRS timer information element as specified in 3GPP TS 24.008 [12] subclause 10.5.7.3 and indicates the Tsor-cm timer value. When the unit field of the Tsor-cm timer value field indicates that the timer is deactivated, the receiving entity shall consider that Tsor-cm timer value is set to the infinity value. Criterion type Bits 87654321 00000001 DNN 00000010 S-NSSAI SST S-NSSAI SST and SD 00000011 00000100 IMS registration related signalling MMTEL voice call 00000101 00000110 MMTEL video call 00000111 SMS over NAS or SMSoIP SOR security check not successful 00001000 11111111 match all All other values are spare. The receiving entity shall ignore SOR-CMCI rule with criterion of criterion type set to a spare value. For "DNN", the criterion value field shall be encoded as a DNN length-value pair field. For "S-NSSAI SST", the criterion value field shall be encoded as one octet SST field. For "S-NSSAI SST and SD", the criterion value field shall be encoded as a sequence of one octet SST field and three octets SD field. The SST field shall be transmitted first. The DNN length-value pair field shall be encoded as a sequence of one octet DNN value length field and a DNN value field. The DNN value length field shall be transmitted first. The DNN value length field indicates the length in octets of the DNN value field. The DNN value field contains an APN as specified in 3GPP TS 23.003 [4]. The SST field contains SST of HPLMN's S-NSSAI. The SD field contains SD of HPLMN's S-NSSAI. For "match all", "SOR security check not successful", "IMS registration related signalling", "MMTEL voice call", "MMTEL video call", and "SMS over NAS or SMSoIP", the criterion value field is zero octets long. If the length of SOR-CMCI rule contents field indicates a length bigger than indicated in figure 9.11.3.51.8, receiving entity shll ignore any superfluous octets located at the end of the SOR-CMCI rule. The UE applies SOR-CMCI rules as described in 3GPP TS 23.122 [5] annex C.



Figure 9.11.3.51.9: SOR-SNPN-SI

Table 9.11.3.51.4: SOR-SNPN-SI

CH controlled prioritized list of preferred SNPNs indicator (CLSI) value (octet p+3, bit 1) Bit 1 0 CH controlled prioritized list of preferred SNPNs absent 1 CH controlled prioritized list of preferred SNPNs present If the CLSI bit is set to "CH controlled prioritized list of preferred SNPNs present", the CH controlled prioritized list of preferred SNPNs field is present. If the CLSI bit is set to "CH controlled prioritized list of preferred SNPNs absent", the CH controlled prioritized list of preferred SNPNs field is absent. CH controlled prioritized list of GINs indicator (CLGI) value (octet p+3, bit 2) Bit 2 0 CH controlled prioritized list of GINs absent 1 CH controlled prioritized list of GINs present If the CLGI bit is set to "CH controlled prioritized list of GINs present", the CH controlled prioritized list of GINs field is present. If the CLGI bit is set to "CH controlled prioritized list of GINs absent", the CH controlled prioritized list of GINs field is absent. If the length of SOR-SNPN-SI contents field indicates a length bigger than indicated in figure 9.11.3.51.9, receiving entity shall ignore any superfluous octets located at the end of the SOR-SNPN-SI. 7 5 8 6 4 3 2 octet p+4 Length of CH controlled prioritized list of preferred SNPNs contents octet p+5 octet (p+6)* **SNPN** identity 1



Figure 9.11.3.51.10: CH controlled prioritized list of preferred SNPNs

8	7	6	5	4	3	2	1		
	MCC	digit 2			MCC	digit 1		octet p+15	
	MNC	digit 3			MCC digit 3				
	MNC	digit 2			MNC	digit 1		octet p+17	
0 0 0 0 Spare Spare Spare Spare				NI	D assigr	nment mo	ode	octet p+18	
	NID valu	ie digit 2			NID val	ue digit 1		octet p+19	
	NID valu	ie digit 4			NID val	ue digit 3		octet p+20	
	NID valu	ie digit 6			NID val	ue digit 5		octet p+21	
	NID valu	ie digit 8			NID val	ue digit 7		octet p+22	
	NID valu	e digit 10		NID val	ue digit 9		octet p+23		

Figure 9.11.3.51.11: SNPN identity

Table 9.11.3.51.5: CH controlled prioritized list of preferred SNPNs

Mobile country code (MCC): The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. Mobile network code (MNC): The coding of MNC field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111". NID assignment mode (see NOTE) NID assignment mode is coded as specified in 3GPP TS 23.003 [4]. NID value (see NOTE) NID value is coded as specified in 3GPP TS 23.003 [4]. NOTE: NID coding deviates from coding of value part of NID IE as specified in subclause 9.2.7 of 3GPP TS 24.502 [18].





8	7	6	5	4	3	2	1	
	MCC	digit 2			MCC	digit 1		octet t+12
	MNC	digit 3			MCC digit 3			
	MNC	digit 2			MNC	digit 1		octet t+14
0 Spare	0 Spare	0 Spare	0 Spare	NI	D assigr	nment mo	de	octet t+15
	NID valu	ie digit 2			NID val	ue digit 1		octet t+16
	NID valu	ie digit 4			NID val	ue digit 3		octet t+17
	NID valu	ie digit 6			NID val	ue digit 5		octet t+18
	NID valu	ie digit 8		NID value digit 7				octet t+19
	NID valu	e digit 10			NID val	ue digit 9		octet t+20

Figure 9.11.3.51.13: GIN

Table 9.11.3.51.6: CH controlled prioritized list of GINs

Mobile country code (MCC): The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.
Mobile network code (MNC): The coding of MNC field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111".
NID assignment mode (see NOTE) NID assignment mode is coded as specified in 3GPP TS 23.003 [4].
NID value (see NOTE) NID value is coded as specified in 3GPP TS 23.003 [4].
NOTE: NID coding deviates from coding of value part of NID IE as specified in subclause 9.2.7 of 3GPP TS 24.502 [18].

9.11.3.51A Supported codec list

See subclause 10.5.4.32 in 3GPP TS 24.008 [12].

9.11.3.52 Time zone

See subclause 10.5.3.8 in 3GPP TS 24.008 [12].

9.11.3.53 Time zone and time

See subclause 10.5.3.9 in 3GPP TS 24.008 [12].

9.11.3.53A UE parameters update transparent container

The purpose of the UE parameters update transparent container when sent from the network to the UE is to provide UE parameters update data, optional acknowledgement request and optional re-registration request. The purpose of the UE parameters update transparent container when sent from the UE to the network is to indicate the UE acknowledgement of successful reception of the UE parameters update transparent containers.

The UE parameters update transparent container information element is coded as shown in figure 9.11.3.53A.1, figure 9.11.3.53A.2, figure 9.11.3.53A.3, figure 9.11.3.53A.4, figure 9.11.3.53A.4B, figure 9.11.3.53A.5, figure 9.11.3.53A.6, figure 9.11.3.53A.7 and table 9.11.3.53A.1.

The UE parameters update transparent container is a type 6 information element with a minimum length of 20 octets.

8	7	6	5	4	3	2	1				
	UE parameters update transparent container IEI										
Leng	Length of UE parameters update transparent container contents										
								octet 3			
	octet 4										
	UPU-MAC-I _{AUSF}										
			Coun	terupu				octet 21-22			
		UE	paramete	ers update	e list			octet 23* -			
								n*			

Figure 9.11.3.53A.1: UE parameters update transparent container information element for UE parameters update data type with value "0"



Figure 9.11.3.53A.2: UE parameters update list



Figure 9.11.3.53A.3: UE parameters update data set for UE parameters update data set type with value "0001"



Figure 9.11.3.53A.4: UE parameters update data set for UE parameters update data set type with value "0010"

8	7	6	5	4	3	2	1	
0	0	0	0	0	0	AOL	DREI	octet d*
Spare	Spare	Spare	Spare	Spare	Spare			

Figure 9.11.3.53A.4A: UE parameters update data set for UE parameters update data set type with value "0011"





UE parameters update transparent container IEI	octet 1
Length of UE parameters update transparent container contents	octet 2
	octet 3
UE parameters update header	octet 4
	octet 5 -

 UPU-MAC-IUE
 octet 5 - 20

 Figure 9.11.3.53A.5: UE parameters update transparent container information element for UE parameters update data type with value "1"

8	7	6	5	4	3	2	1	
0	0	0	0	0	REG	ACK	UPU	octet 4
Spare	Spare	Spare	Spare	Spare			data	
							type	

Figure 9.11.3.53A.6: UE parameters update header for UE parameters update data type with value "0"

8	7	6	5	4	3	2	1	
0	0	0	0	0	0	0	UPU	octet 4
Spare	data							
							type	

Figure 9.11.3.53A.7: UE parameters update header for UE parameters update data type with value "1"

Table 9.11.3.53A.1: UE parameters update transparent container information element

UPU-MAC-I _{AUSF} , UPU-MAC-I _{UE} and Counter _{UPU} are coded as specified in 3GPP TS 33.501 [24]							
UPU data type (octet 4, bit 1) 0 The UE parameters update transparent container carries a UE parameters update							
 The UE parameters update transparent container carries an acknowledgement of successful reception of a UE parameters update list 							
Acknowledgement (ACK) value (octet 4, bit 2) 0 acknowledgement not requested 1 acknowledgement requested							
Re-registration (REG) value (octet 4, bit 3) 0 re-registration not requested 1 re-registration requested							
UE parameters update data set type Bits							
4 3 2 1 0 0 0 1 Routing indicator update data 0 0 1 0 Default configured NSSAI update data 0 0 1 1 Disaster roaming information update data 0 1 0 0 ME routing indicator update data							
All other values are reserved							
Disaster Roaming Enabled Indication (DREI) value (octet d*, bit 1) 0 Disaster roaming is disabled in the UE 1 Disaster roaming is enabled in the UE							
Indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN' (AOL) value (octet d*, bit 2) 0 false 1 true							
The secured packet is coded as specified in 3GPP TS 31.115 [22B].							
The default configured NSSAI is encoded as the value part of the NSSAI IE (see subclause 9.11.3.37).							
Routing indicator Routing indicator is encoded as the routing indicator field of the 5GS mobile identity IE (see subclause 9.11.3.4).							
NOTE: When the routing indicator is updated, if the SNPN uses the EAP based primary authentication and key agreement procedure using the EAP-AKA' or the 5G AKA based primary authentication and key agreement procedure, then UE parameter update data set type is set to "routing indicator update data", otherwise, UE parameter update data set type is set to "ME routing indicator update data".							

9.11.3.54 UE security capability

The UE security capability information element is used by the UE and by the network to indicate which security algorithms are supported by the UE in N1 mode for NAS security as well as which security algorithms are supported over NR and E-UTRA connected to 5GCN for AS security.

The UE security capability information element is coded as shown in figure 9.11.3.54.1 and table 9.11.3.54.1.

The UE security capability is a type 4 information element with a minimum length of 4 octets and a maximum length of 10 octets.

Octets 5 to 10 are optional. If octet 5 is included, then also octet 6 shall be included.

If the UE does not support any security algorithm for AS security over E-UTRA connected to 5GCN, it shall not include octets 5 and 6. The UE shall not include octets 7 to 10.

If the UE does not support any security algorithm for AS security over E-UTRA connected to 5GCN, and if the network includes octets 7 to 10, then the network shall also include octets 5 to 6.

If the network includes octet 7, then it shall include also octet 8. If the network includes octet 9, then it shall include also octet 10.

8	7	6	5	4	3	2	1	_				
	UE security capability IEI											
	octet 2											
	128-	128-	128-									
5G-	5G-	5G-	5G-	5G-	5G-	5G-	5G-	octet 3				
EA0	EA1	EA2	EA3	EA4	EA5	EA6	EA7					
	128-	128-	128-									
5G-IA0	5G-IA1	5G-IA2	5G-IA3	5G-IA4	5G-IA5	5G-IA6	5G-IA7	octet 4				
	128-	128-	128-									
EEA0	EEA1	EEA2	EEA3	EEA4	EEA5	EEA6	EEA7	octet 5*				
	128-	128-	128-									
EIA0	EIA1	EIA2	EIA3	EIA4	EIA5	EIA6	EIA7	octet 6*				
0	0	0	0	0	0	0	0					
			Spa	are				octet 7* -10*				

Figure 9.11.3.54.1: UE security capability information element

Table 9.11.3.54.1: UE security capability information element

5GS encryption algorithms supported (see NOTE 1) (octet 3)
5GS encryption algorithm 5G-EA0 supported (octet 3, bit 8)05GS encryption algorithm 5G-EA0 not supported15GS encryption algorithm 5G-EA0 supported
5GS encryption algorithm 128-5G-EA1 supported (octet 3, bit 7)05GS encryption algorithm 128-5G-EA1 not supported15GS encryption algorithm 128-5G-EA1 supported
5GS encryption algorithm 128-5G-EA2 supported (octet 3, bit 6)05GS encryption algorithm 128-5G-EA2 not supported15GS encryption algorithm 128-5G-EA2 supported
5GS encryption algorithm 128-5G-EA3 supported (octet 3, bit 5)05GS encryption algorithm 128-5G-EA3 not supported15GS encryption algorithm 128-5G-EA3 supported
5GS encryption algorithm 5G-EA4 supported (octet 3, bit 4)05GS encryption algorithm 5G-EA4 not supported15GS encryption algorithm 5G-EA4 supported
5GS encryption algorithm 5G-EA5 supported (octet 3, bit 3)05GS encryption algorithm 5G-EA5 not supported15GS encryption algorithm 5G-EA5 supported
5GS encryption algorithm 5G-EA6 supported (octet 3, bit 2)05GS encryption algorithm 5G-EA6 not supported15GS encryption algorithm 5G-EA6 supported
5GS encryption algorithm 5G-EA7 supported (octet 3, bit 1)05GS encryption algorithm 5G-EA7 not supported15GS encryption algorithm 5G-EA7 supported
5GS integrity algorithms supported (see NOTE 2) (octet 4)
5GS integrity algorithm 5G-IA0 supported (octet 4, bit 8)05GS integrity algorithm 5G-IA0 not supported15GS integrity algorithm 5G-IA0 supported
5GS integrity algorithm 128-5G-IA1 supported (octet 4, bit 7)05GS integrity algorithm 128-5G-IA1 not supported15GS integrity algorithm 128-5G-IA1 supported
5GS integrity algorithm 128-5G-IA2 supported (octet 4, bit 6)05GS integrity algorithm 128-5G-IA2 not supported15GS integrity algorithm 128-5G-IA2 supported
5GS integrity algorithm 128-5G-IA3 supported (octet 4, bit 5)05GS integrity algorithm 128-5G-IA3 not supported15GS integrity algorithm 128-5G-IA3 supported
5GS integrity algorithm 5G-IA4 supported (octet 4, bit 4)05GS integrity algorithm 5G-IA4 not supported15GS integrity algorithm 5G-IA4 supported
5GS integrity algorithm 5G-IA5 supported (octet 4, bit 3)05GS integrity algorithm 5G-IA5 not supported15GS integrity algorithm 5G-IA5 supported
5GS integrity algorithm 5G-IA6supported (octet 4, bit 2)05GS integrity algorithm 5G-IA6 not supported15GS integrity algorithm 5G-IA6 supported
5GS integrity algorithm 5G-IA7 supported (octet 4, bit 1)05GS integrity algorithm 5G-IA7 not supported15GS integrity algorithm 5G-IA7 supported

EPS encryption algorithms supported (see NOTE 3) (octet 5)
EPS encryption algorithm EEA0 supported (octet 5, bit 8)0EPS encryption algorithm EEA0 not supported1EPS encryption algorithm EEA0 supported
EPS encryption algorithm 128-EEA1 supported (octet 5, bit 7)0EPS encryption algorithm 128-EEA1 not supported1EPS encryption algorithm 128-EEA1 supported
EPS encryption algorithm 128-EEA2 supported (octet 5, bit 6)0EPS encryption algorithm 128-EEA2 not supported1EPS encryption algorithm 128-EEA2 supported
EPS encryption algorithm 128-EEA3 supported (octet 5, bit 5)0EPS encryption algorithm 128-EEA3 not supported1EPS encryption algorithm 128-EEA3 supported
EPS encryption algorithm EEA4 supported (octet 5, bit 4)0EPS encryption algorithm EEA4 not supported1EPS encryption algorithm EEA4 supported
EPS encryption algorithm EEA5 supported (octet 5, bit 3)0EPS encryption algorithm EEA5 not supported1EPS encryption algorithm EEA5 supported
EPS encryption algorithm EEA6 supported (octet 5, bit 2)0EPS encryption algorithm EEA6 not supported1EPS encryption algorithm EEA6 supported
EPS encryption algorithm EEA7 supported (octet 5, bit 1)0EPS encryption algorithm EEA7 not supported1EPS encryption algorithm EEA7 supported
EPS integrity algorithms supported (see NOTE 4) (octet 6)
EPS integrity algorithm EIA0 supported (octet 6, bit 8)0EPS integrity algorithm EIA0 not supported1EPS integrity algorithm EIA0 supported
EPS integrity algorithm 128-EIA1 supported (octet 6, bit 7)0EPS integrity algorithm 128-EIA1 not supported1EPS integrity algorithm 128-EIA1 supported
EPS integrity algorithm 128-EIA2 supported (octet 6, bit 6)0EPS integrity algorithm 128-EIA2 not supported1EPS integrity algorithm 128-EIA2 supported
EPS integrity algorithm 128-EIA3 supported (octet 6, bit 5)0EPS integrity algorithm 128-EIA3 not supported1EPS integrity algorithm 128-EIA3 supported
EPS integrity algorithm EIA4 supported (octet 6, bit 4)0EPS integrity algorithm EIA4 not supported1EPS integrity algorithm EIA4 supported
EPS integrity algorithm EIA5 supported (octet 6, bit 3)0EPS integrity algorithm EIA5 not supported1EPS integrity algorithm EIA5 supported
EPS integrity algorithm EIA6 supported (octet 6, bit 2)0EPS integrity algorithm EIA6 not supported1EPS integrity algorithm EIA6 supported
EPS integrity algorithm EIA7 supported (octet 6, bit 1)0EPS integrity algorithm EIA7 not supported1EPS integrity algorithm EIA7 supported

For the UE not supporting any security algorithm for AS security over E-UTRA connected to 5GCN, all bits in octets 5 to 10 are spare and shall be ignored, if the respective octet is received with the information element. For the UE supporting at least one security algorithm for AS security over E-UTRA connected to 5GCN all bits in octets 7 to 10 are spare and shall be ignored, if the respective octet is received with the information element. If the AMF receives any of the octets 7 to 10 (NOTE 5), it shall store the octets as received and include them when sending the UE security capability information element to the UE. NOTE 1: The code points in octet 3 are used to indicate support for 5GS encryption algorithms for NAS security in N1 mode and support for 5GS encryption algorithms for AS security over NR. NOTE 2: The code points in octet 4 are used to indicate support for 5GS integrity algorithms for NAS security in N1 mode and support for 5GS integrity algorithms for AS security over NR. NOTE 3: The code points in octet 5 are used to indicate support for EPS encryption algorithms for AS security over E-UTRA connected to 5GCN. NOTE 4: The code points in octet 6 are used to indicate support for EPS integrity algorithms for AS security over E-UTRA connected to 5GCN. NOTE 5: The AMF can receive this information element also from another AMF or MME during N1 mode to N1 mode or S1 mode to N1 mode handover preparation.

9.11.3.55 UE's usage setting

The purpose of the UE's usage setting information element is to provide the network with the UE's usage setting as defined in 3GPP TS 24.301 [15]. The network uses the UE's usage setting to select the RFSP index.

The UE's usage setting information element is coded as shown in figure 9.11.3.55.1 and table 9.11.3.55.1.

The UE's usage setting is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	_	
		UE	E's usage	setting I	EI			octet 1	
Length of UE's usage setting contents									
0	0	0	0	0	0	0	UE's		
Spare	Spare	Spare	Spare	Spare	Spare	Spare	usage	octet 3	
					_		setting		

Figure 9.11.3.55.1: UE's usage setting information element

Table 9.11.3.55.1: UE's	usage setting	information	element
-------------------------	---------------	-------------	---------

I	UE's usage setting (octet 3, bit 1)								
	0 voice centric								
	1 data centric								
	All other bits in the octet 3 are spare and shall be coded as zero,								

9.11.3.56 UE status

The purpose of the UE status information element is to provide the network with information concerning aspects of the current UE registration status which is used for interworking with EPS.

The UE status information element is coded as shown in figure 9.11.3.56.1 and table 9.11.3.56.1.

The UE status is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	_	
UE status IEI									
	Length of UE status contents								
0	0	0	0	0	0	N1	S1		
Spare	Spare	Spare	Spare	Spare	Spare	mode	mode	octet 3	
						reg	reg		

Figure 9.11.3.56.1: UE status information element

EMM registration status (S1 mode reg) (octet 3, bit 1)
0 UE is not in EMM-REGISTERED state
1 UE is in EMM-REGISTERED state
5GMM registration status (N1 mode reg) (octet 3, bit 2) 0 UE is not in 5GMM-REGISTERED state
1 UE is in 5GMM-REGISTERED state
All other bits in the octet 3 are spare and shall be coded as zero.

Table 9.11.3.56.1: UE status information element

9.11.3.57 Uplink data status

The purpose of the Uplink data status information element is to indicate to the network which preserved PDU sessions have uplink data pending.

The Uplink data status information element is coded as shown in figure 9.11.3.57.1 and table 9.11.3.57.1.

The Uplink data status information element is a type 4 information element with minimum length of 4 octets a maximum length of 34 octets.

8	7	6	5	4	3	2	1				
	octet 1										
	Length of uplink data status contents										
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 3			
(7)	(6)	(5)	(4)	(3)	(2)	(1)	(0)				
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 4			
(15)	(14)	(13)	(12)	(11)	(10)	(9)	(8)				
0	0	0	0	0	0	0	0				
	octet 5* -34*										
	Figure 9.11.3.57.1: Uplink data status information element										

Figure 5.11.5.57.1. Opinik data status information element



PSI(x) shall be coded as follows:

PSI(0):

Bit 1 of octet 3 is spare and shall be coded as zero.

PSI(1) – PSI(15):

0 indicates that no uplink data are pending for the corresponding PDU session identity or the PDU session is in PDU SESSION INACTIVE state or is in PDU SESSION ACTIVE state with user-plane resources already established.

1 indicates that uplink data are pending for the corresponding PDU session identity and the user-plane resources for the corresponding PDU session are not established.

All bits in octet 5 to 34 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.3.58	Void
9.11.3.59	Void
9.11.3.60	Void
9.11.3.61	Void
9.11.3.62	Void
9.11.3.63	Void

- 9.11.3.64 Void
- 9.11.3.65 Void
- 9.11.3.66 Void
- 9.11.3.67 Void

9.11.3.68 UE radio capability ID

The purpose of the UE radio capability ID information element is to carry a UE radio capability ID.

The UE radio capability ID information element is coded as shown in figure 9.11.3.68.1 and table 9.11.3.68.1.

The UE radio capability ID is a type 4 information element with a length of n octets.



Figure 9.11.3.68.1: UE radio capability ID information element



UE radio capability ID (octets 3 to n) The UE radio capability ID contents contain the UE radio capability ID as specified in 3GPP TS 23.003 [4], with each hexadecimal digit coded over 4 bits, starting with the first hexadecimal digit coded in bits 4 to 1 of octet 3, the second hexadecimal digit coded in bits 8 to 5 of octet 3, and so on. If the UE radio capability ID contains an odd number of hexadecimal digits, bits 8 to 5 of the last octet (octet n) shall be coded as "1111".

9.11.3.69 UE radio capability ID deletion indication

The purpose of the UE radio capability ID deletion indication information element is to indicate to the UE that deletion of UE radio capability IDs is requested.

The UE radio capability ID deletion indication is a type 1 information element.

The UE radio capability ID deletion indication information element is coded as shown in figure 9.11.3.69.1 and table 9.11.3.69.1.



Figure 9.11.3.69.1: UE radio capability ID deletion indication information element

Table 9.11.3.69.1: UE radio capability ID deletion indication information element

Deletion requested (octet 1) Bits **3 2 1** 0 0 0 UE radio capability ID deletion not requested 0 0 1 Network-assigned UE radio capability IDs deletion requested All other values are unused and shall be interpreted as "UE radio capability ID deletion not requested", if received by the UE.

9.11.3.70 Truncated 5G-S-TMSI configuration

The purpose of the Truncated 5G-S-TMSI configuration information element is to provide the size of the components of the truncated 5G-S-TMSI to the UE in NB-N1 mode to create the truncated 5G-S-TMSI.

The Truncated 5G-S-TMSI configuration information element is coded as shown in figure 9.11.3.70.1 and table 9.11.3.70.1.

The Truncated 5G-S-TMSI configuration is a type 4 information element with 3 octets length.



Figure 9.11.3.70.1: Truncated 5G-S-TMSI configuration information element

Table 9.11.3.70.1: Truncated 5G-S-TMSI configuration information element

Truncated AMF Pointer value (bits 4 to 1 of octet 3)										
Th	This field represents the size of the least significant bits of the AMF Pointer.									
Bit 4 0 0 0 0 0 0 0	s 0 0 0 1 1	2 0 1 1 0 1	 reserved 1 least significant bit of the AMF Pointer 2 least significant bits of the AMF Pointer 3 least significant bits of the AMF Pointer 4 least significant bits of the AMF Pointer 5 least significant bits of the AMF Pointer 6 least significant bits of the AMF Pointer 							
All this	oth s ve	ner v ersio	valu on c	es shall be interpreted as "6 least significant bits of the AMF Pointer" by of the protocol.						
Τrι	Inc	ateo	lA b	MF Set ID value (bits 8 to 5 of octet 3)						
Th	is fi	ield	rep	resents the size of the least significant bits of the AMF Set ID.						
Bit 4 0 0 0 0 0 0 0 1 1 1	Bits 4 3 2 1 0 0 0 reserved 0 1 1 least significant bit of the AMF Set ID 0 1 0 2 least significant bits of the AMF Set ID 0 1 0 2 least significant bits of the AMF Set ID 0 1 1 3 least significant bits of the AMF Set ID 0 1 0 4 least significant bits of the AMF Set ID 0 1 0 4 least significant bits of the AMF Set ID 0 1 0 1 5 least significant bits of the AMF Set ID 0 1 0 6 least significant bits of the AMF Set ID 0 1 1 7 least significant bits of the AMF Set ID 1 0 0 8 least significant bits of the AMF Set ID 1 0 0 8 least significant bits of the AMF Set ID 1 0 1 9 least significant bits of the AMF Set ID 1 0 1 9 least significant bits of the AMF Set ID 1 0 1 1 1 0 1									
All this	oth s ve	ner v ersio	valu on c	tes shall be interpreted as "10 least significant bits of the AMF Set ID" by of the protocol.						
NC	DTE	:	To Po 30	otal sum of the "Truncated AMF Set ID value" and the "Truncated AMF pinter value" in the Truncated 5G-S-TMSI configuration IE is specified in GPP TS 23.003 [4] and 3GPP TS 36.300 [25B].						

9.11.3.71 WUS assistance information

See subclause 9.9.3.62 in 3GPP TS 24.301 [15].

9.11.3.72 N5GC indication

The purpose of the N5GC indication information element is to indicate to the network that the registration request by the W-AGF is on behalf of an N5GC device.

The N5GC indication information element is coded as shown in figure 9.11.3.72.1.

The N5GC indication is a type 1 information element.



Table 9.11.3.72.1: N5GC indication

N5GC device indication bit (N5GCREG) (octet 1, bit 1) Bit 1 0 N5GC device registration is not requested

- 1 N5GC device registration is requested
- 1 NSGC device registration is requested

Bits 2 to 4 are spare and shall be coded as zero.

9.11.3.73 NB-N1 mode DRX parameters

The purpose of the NB-N1 mode DRX parameters information element is to indicate that the UE wants to use DRX in NB-N1 mode and for the network to indicate the DRX cycle value to be used at paging in NB-N1 mode.

The NB-N1 mode DRX parameters is a type 4 information element with a length of 3 octets.

The NB-N1 mode DRX parameters information element is coded as shown in figure 9.11.3.73.1 and table 9.11.3.73.1.

The value part of a DRX parameter information element is coded as shown in table 9.11.3.73.1.

8	7	6	5	4	3	2	1	
NB-N1 mode DRX parameters IEI								
Length of NB-N1 mode DRX parameters contents								
0	0	0	0	NB	-N1 mod	e DRX va	lue	
	Spa	are						octet 3
Figure 9 11 3 73 1: NB-N1 mode DBX parameters information element								

Figure 9.11.3.73.1: NB-N1 mode DRX parameters information element

Table 9.11.3.73.1: NB-N1 mode DRX parameters information element

NB-N1 mode DRX value (octet 3, bits 1 to 4) This field represents the DRX cycle parameter 'T', for NB-N1 mode, as defined in 3GPP TS 36.304 [25C]. Bits 2 4 3 1 0 DRX value not specified 0 0 0 DRX cycle parameter T = 320 0 0 1 0 DRX cycle parameter T = 64 0 0 1 0 0 1 1 DRX cycle parameter T = 128 0 1 0 0 DRX cycle parameter T = 256 0 1 0 1 DRX cycle parameter T = 512 0 1 1 1 DRX cycle parameter T = 1024 All other values shall be interpreted as "DRX value not specified" by this version of the protocol. Bits 5 to 8 of octet 3 are spare and shall be coded as zero.

9.11.3.74 Additional configuration indication

The purpose of the Additional configuration indication information element is to indicate additional information associated with the generic UE configuration update procedure.

The Additional configuration indication information element is coded as shown in figure 9.11.3.74.1 and table 9.11.3.74.1.

The Additional configuration indication is a type 1 information element.

8	7	6	5	4	3	2	1	_
Additiona	al configu	uration in	dication	0	0	0	SCMR	octet 1
	IE	1		Spare	Spare	Spare		

Figure 9.11.3.74.1: Additional configuration indication

Table 9.11.3.74.1	Additional	configuration	indication
-------------------	------------	---------------	------------

Signalling connection maintain request (SCMR) (octet 1, bit 1)							
Bit							
1							
0 no additional information							
1 release of N1 NAS signalling connection not required							
Bits 2 to 4 are spare and shall be coded as zero,							

9.11.3.75 Extended rejected NSSAI

The purpose of the Extended rejected NSSAI information element is to identify a collection of rejected S-NSSAIs if UE supports extended rejected NSSAI.

The Extended rejected NSSAI information element is coded as shown in figure 9.11.3.75.1, figure 9.11.3.75.2 and table 9.11.3.75.1.

The Extended rejected NSSAI is a type 4 information element with a minimum length of 5 octets and a maximum length of 90 octets.







Figure 9.11.3.75.2: Partial extended rejected NSSAI list – type of list = 000





Figure 9.11.3.75.4: Rejected S-NSSAI

Table 9.11.3.75.1: Extended rejected NSSAI information element

Value part of the Extended rejected NSSAI information element (octet 3 to v) The value part of the Extended rejected NSSAI information element consists of one or more partial extended rejected NSSAI lists. The length of each partial extended rejected NSSAI list can be determined from the 'type of list' field and the 'number of elements' field in the first octet of the partial extended rejected NSSAI list. Each rejected S-NSSAI consists of one S-NSSAI and an associated cause value. Each rejected S-NSSAI also includes the mapped HPLMN S-NSSAI if available. The length of each rejected S-NSSAI can be determined by the 'length of rejected S-NSSAI' field in the first octet of the rejected S-NSSAI. The UE shall store the complete list received (NOTE 0). If more than 8 rejected S-NSSAIs are included in this information element, the UE shall store the first 8 rejected S-NSSAIs and ignore the remaining octets of the information element. Partial extended rejected NSSAI list: Number of elements (octet 3, bits 1 to 4) Bits 4 3 2 1 0 0 0 0 1 element 0 0 0 1 2 element 0 1 1 0 7 element 0 1 1 1 8 element All other values are unused and shall be interpreted as 8, if received by the UE. Type of list (octet 3, bits 5 to 7) (NOTE 7) Bits 7 6 5 0 0 0 list of S-NSSAIs without any associated back-off timer value 0 0 1 list of S-NSSAIs with one associated back-off timer value that applies to all S-NSSAIs in the list All other values are reserved. Bit 8 of octet 3 is spare and shall be coded as zero. Back-off timer value (octet 4): Back-off timer value is coded as the value part of GPRS timer 3 in subclause 10.5.7.4a in 3GPP TS 24.008 [12]. Rejected S-NSSAI: Cause value (octet 4) Bits 4 3 2 1 0 0 0 0 S-NSSAI not available in the current PLMN or SNPN 0 0 0 1 S-NSSAI not available in the current registration area S-NSSAI not available due to the failed or revoked network slice-0 0 1 0 specific authentication and authorization 0 0 1 1 S-NSSAI not available due to maximum number of UEs reached All other values are reserved. Slice/service type (SST) (octet 5) This field contains the 8 bit SST value. The coding of the SST value part is defined in 3GPP TS 23.003 [4]. (NOTE 5) Slice differentiator (SD) (octet 6 to octet 8)

This field contains the 24 bit SD value. The coding of the SD value part is defined in 3GPP TS 23.003 [4]. (NOTE 6)

If the SST encoded in octet 4 is not associated with a valid SD value, and the sender needs to include a mapped HPLMN SST (octet 8) and a mapped HPLMN SD (octets 9 to 11), then the sender shall set the SD value (octets 5 to 7) to "no SD value associated with the SST".

mapped HPLMN Slice/service type (SST) (octet 9)

This field contains the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SST value is mapped. The coding of the SST value part is defined in 3GPP TS 23.003 [4].

mapped HPLMN Slice differentiator (SD) (octet 10 to octet 12)

This field contains the 24 bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN to which the SD value is mapped. The coding of the SD value part is defined in 3GPP TS 23.003 [4].

NOTE 0: The number of rejected S-NSSAI(s) shall not exceed eight.

- NOTE 1: Octet 4 and octet 5 shall always be included.
- NOTE 2: If the octet 6 is included, then octet 7 and octet 8 shall be included.
- NOTE 3: If the octet 9 is included, then octets 10, 11, and 12 may be included.
- NOTE 4: If the octet 10 is included, then octet 11 and octet 12 shall be included.
- NOTE 5: If the Cause value is "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization", this field shall contain the 8 bit SST value of an S-NSSAI in the S-NSSAI(s) of the HPLMN and octets 9, 10, 11, and 12 shall not be included.
- NOTE 6: If the Cause value is "S-NSSAI not available due to the failed or revoked network slice-specific authentication and authorization", this field shall contain the 24 bit SD value of an S-NSSAI in the S-NSSAI(s) of the HPLMN and octets 9, 10, 11, and 12 shall not be included.
- NOTE 7: The partial extended rejected NSSAI with type of list = 001 shall only be used for rejected S-NSSAI(s) with the rejection cause "S-NSSAI not available due to maximum number of UEs reached".

9.11.3.76 UE request type

See subclause 9.9.3.65 in 3GPP TS 24.301 [15].

9.11.3.77 Paging restriction

The purpose of the Paging restriction information element is to request the network to restrict paging.

The Paging restriction information element is coded as shown in figure 9.11.3.77.1, figure 9.11.3.77.2 and table 9.11.3.77.1.

The Paging restriction is a type 4 information element with a minimum length of 3 octets and a maximum length of 35 octets.



Figure 9.11.3.77.1: Paging restriction information element for Paging restriction type = "All paging is restricted" and for Paging restriction type = "All paging is restricted except voice"

8	7	6	5	4	3	2	1			
Paging restriction IEI										
Length of Paging restriction contents										
0 0 0 0 Paging restriction type										
Spare	Spare	Spare	Spare					octet 3		
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 4		
(7)	(6)	(5)	(4)	(3)	(2)	(1)	(0)			
PSI	PSI	PSI	PSI	PSI	PSI	PSI	PSI	octet 5		
(15)	(14)	(13)	(12)	(11)	(10)	(9)	(8)			
0	0	0	0	0	0	0	0	octet 6*-35*		
spare										

Figure 9.11.3.77.2: Paging restriction information element for Paging restriction type = "All paging is restricted except for specified PDU session(s)" and for Paging restriction type = "All paging is restricted except for voice service and specified PDU session(s)"



Pa	Paging restriction type (bits 4 to 1 of octet 3)														
Rit	Bits														
4	4 3 2 1														
0	0 0 0 reserved														
0	0	0 1 All paging is restricted													
0	0	1	0) All paging is restricted except for voice service											
0	0	1	1	All paging is restricted except for specified PDU session(s)											
0	1	 0 0 All paging is restricted except for voice service and specified PDU session(s) 													
All	oth	۱er	valu	es are reserved.											
Bit	s 5	to 8	3 of	octet 3 are spare and shall be coded as zero.											
PS Th res	SI(x) is fi stric) (bi eld ted	ts 8 ind	to 1 of octet 4 and octet 5): icates the PDU session identity of the PDU session for which paging is											
PS Sp	l(0) are): (b an	oit 1 d sł	of octet 4) nall be coded as zero.											
 PSI(1) – PSI(15): 0 indicates that paging is restricted for the PDU session associated with the PDU session identity. 1 indicates that paging is not restricted for the PDU session associated with the PDU session identity. All bits in octet 6 to 35 are spare and shall be coded as zero, if the respective octet is 															
inc	iua	ed I	n tr	e information element.											

9.11.3.78 Void

9.11.3.79 NID

See subclause 9.2.7 in 3GPP TS 24.502 [18].

9.11.3.80 PEIPS assistance information

The purpose of the PEIPS assistance information, information element is to transfer the required assistance information to indicate the paging subgroup used when paging the UE.

The coding of the information element allows combining different types of PEIPS assistance information.

The PEIPS assistance information, information element is coded as shown in figure 9.11.3.80.1, figure 9.11.3.80.2, figure 9.11.3.80.3 and table 9.11.3.80.1.

The PEIPS assistance information is a type 4 information element, with a minimum length of 3 octets.





8	7	6	5	4	3	2	1	_
Туре	of inform	ation		Paging s	ubgroup	ID value		octet 1

Figure 9.11.3.80.2: PEIPS assistance information type -type of information= "000"

8	7	6	5	4	3	2	1	_
Туре	of inform	ation	UE pa	ging prob	ability in	formatior	n value	octet 1

Figure 9.11.3.80.3: PEIPS assistance information type -type of information= "001"

Value part of the PEIPS assistance information information element (octets 3 to n)												
The value part of the PEIPS assistance information information element consists of one or several types of PEIPS assistance information.												
PEIPS assistance information type:												
Type of information (octet 1) Bits												
7 6												
0 0 Paging subgroup ID 0 0 1 UE paging probability information												
All other values are reserved.												
Paging subgroup ID value: (actet 1, bits 1-5)												
This field contains the value (in desimal) of paging subgroup ID that is assigned by the												
This need contains the value (in decimal) of paging subgroup to that is assigned by the												
AME for paging the UE. This field has a valid range of values from (0-7). All other values												
are reserved and shall be interpreted as 0 by this version of the protocol.												
UE paging probability information value: (octet 1, bits 1-5)												
This field contains the value of UE paging probability information provided by the UE to												
the AMF. It represents the probability of the UE receiving the paging.												
Bit												
5 4 3 2 1 UE paging probability information value												
0 0 0 0 0 p00												
0 0 0 0 1 p05												
0 0 0 1 0 p10												
0 0 0 1 1 p15												
0 0 1 0 0 p20												
0 0 1 0 1 p25												
0 0 1 1 0 p30												
0 0 1 1 1 p35												
0 1 0 0 0 p40												
0 1 0 0 1 p45												
0 1 0 1 0 p50												
0 1 0 1 1 p55												
0 1 1 0 0 p60												
0 1 1 0 1 p65												
0 1 1 1 0 p20												
0 1 1 1 1 p75												
All other values shall be interpreted as 10100 by this version of the protocol.												

9.11.3.81 5GS additional request result

The purpose of the 5GS additional request result information element is to inform the UE about the result of additional request.

The 5GS additional request result information element is coded as shown in figure 9.11.3.81.1 and table 9.11.3.81.1.

The 5GS additional request result is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1					
5GS additional request result IEI												
	Length of 5GS additional request result contents											
0	0	0	0	0	0	PR	D					
Spare	Spare	Spare	Spare	Spare	Spare			octet 3				

Figure 9.11.3.81.1: 5GS additional request result information element

Table 9.11.3.81.1: 5GS additional request result information element

Paging restriction decision (PRD) (bits 2 to 1 of octet 3)
Bits
2 1
0 0 no additional information
0 1 paging restriction is accepted
1 0 paging restriction is rejected
All other values are reserved.
Bits 3 to 8 of octet 3 are spare and shall be coded as zero.

9.11.3.82 NSSRG information

The purpose of the NSSRG information information element is to identify one or more NSSRG values associated with each of the HPLMN S-NSSAIs in a configured NSSAI.

The NSSRG information information element is coded as shown in figure 9.11.3.82.1, figure 9.11.3.82.2 and table 9.11.3.82.1.

The NSSRG information is a type 6 information element with minimum length of 7 octets and maximum length of 4099 octets.



Figure 9.11.3.82.1: NSSRG information information element

8	7	6	5	4	3	2	1						
	Length of NSSRG values for S-NSSAI												
								octet 5					
			S-NSSA	Al value									
								octet w					
		NSSRG	G value 1	for the S	S-NSSAI			octet w+1					
		NSSRG	3 value 2	for the S	S-NSSAI			octet w+2*					
								octet w+3*					
								octet m-1*					
		NSSRC	S value y	for the S	S-NSSAI			octet m*					
		Figure 9.11.3.82.2: NSSRG values for S-NSSAI											



/alue part of the NSSRG information information element (octet 4 to v)										
The value part of the NSSRG information information element consists of one or more NSSRG values for each S-NSSAI in the Configured NSSAI IE.										
S-NSSAI value (octet 5 to w) (see NOTE 2)										
S-NSSAI va as specified	S-NSSAI value is coded as the length and value part of S-NSSAI information element as specified in subclause 9.11.2.8 starting with the second octet. See NOTE 1.									
NSSRG va	lue for the S-NSSAI (octet w+1)									
This field co	ontains the 8 bit NSSRG value.									
NOTE 1: I	If a mapped HPLMN SST is included in a S-NSSAI value, then the NSSRG value(s) are associated with the Mapped HPLMN SST, and the Mapped HPLMN SD, if included.									
NOTE 2: i	The NSSRG information IE shall contain the complete set of S-NSSAI(s) included in the configured NSSAI.									

9.11.3.83 List of PLMNs to be used in disaster condition

The purpose of the list of PLMNs to be used in disaster condition information element is to provide the "list of PLMN(s) to be used in disaster condition" associated with the serving PLMN to the UE.

The list of PLMNs to be used in disaster condition information element is coded as shown in figures 9.11.3.83.1 and 9.11.3.83.2 and table 9.11.3.83.1.

The list of PLMNs to be used in disaster condition is a type 4 information element, with a minimum length of 2 octets.

8	7	6	5	4	3	2	1				
	List of PLMNs to be used in disaster condition list IEI										
Leng	Length of list of PLMNs to be used in disaster condition contents										
			PLM	N ID 1				octet 3*			
								octet 5*			
								octet 6*			
			PLM	N ID 2							
								octet 8*			
								octet 9*			
				••							
								octet q			
								octet q+1'			
			PLM	חטוא				octet q+3*			



8	7	6	5	4	3	2	1					
	MCC	digit 2			MCC digit 1							
	MNC	diait 3			octet a+2							
	WINC	aigit o			MOO	aigit o		00101 912				
	MNC	digit 2			octet q+3							
Figure 9.11.3.83.2: PLMN ID n												

Table 9.11.3.83.1: List of PLMNs to be used in disaster condition information element

MCC, Mobile country code (octet q+1 and bits 1 to 4 octet q+2)
The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.
MNC. Mobile network code (bits 5 to 8 of octet g+2 and octet g+3)
The coding of this field is the responsibility of each administration but BCD coding shall
be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use
only two digits in the MNC, bits 5 to 8 of octet q+2 shall be coded as "1111".
The MCC and MNC digite are coded as estate 6 to 8 of the Temperary mobile group
The MCC and MNC digits are coded as octers o to o of the Temporary mobile group
identity IE in figure 10.5.154 of 3GPP TS 24.008 [12].
NOTE: The PLMN IDs are provided in decreasing order of priority, i.e. PLMN ID 1
indicates highest priority and PLMN ID n indicates lowest priority.

9.11.3.84 Registration wait range

The purpose of the registration wait range information element is to provide the disaster roaming wait range or the disaster return wait range to the UE.

The registration wait range information element is coded as shown in figure 9.11.3.84.1 and table 9.11.3.84.1.

The registration wait range is a type 4 information element, with a length of 4 octets.

8	7	6	5	4	3	2	1						
Registration wait range IEI													
Length of registration wait range													
	Minimum registration wait time												
		Maxim	um regist	tration wa	ait time			octet 4					

Figure 9.11.3.84.1: Registration wait range information element

Table 9.11.3.84.1: Registration wait range information element

Minimum registration wait time (octet 3) The minimum registration wait time contains the minimum duration of the registration wait time, encoded as octet 2 of the GPRS timer information element (see 3GPP TS 24.008 [12] subclause 10.5.7.3).

Maximum registration wait time (octet 4) The maximum registration wait time contains the maximum duration of the registration wait time, encoded as octet 2 of the GPRS timer information element (see 3GPP TS 24.008 [12] subclause 10.5.7.3).

9.11.3.85 PLMN identity

The purpose of the PLMN identity information element is to provide a PLMN identity.

The PLMN identity information element is coded as shown in figure 9.11.3.85.1, and table 9.11.3.85.1.

The PLMN identity is a type 4 information element.

8	7	6	5	4	3	2	1	
			PLMN id	entity IEI				octet 1
		Length	of PLMN	identity of	contents			octet 2
	MCC	digit 2			octet 3			
	MNC	digit 3			MCC	digit 3		octet 4
								_
	MNC	digit 2			MNC	digit 1		octet 5
	Elaur	- 0 4 4 2	0E 4. DI		maine ind	armatia	-	0.001

Figure 9.11.3.85.1: PLMN identity information element



MCC, Mobile country code (octet 3, octet 4 bits 1 to 4) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code (octet 4 bits 5 to 8, octet 5) The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet 4 shall be coded as "1111". The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12].

9.11.3.86 Extended CAG information list

The purpose of the Extended CAG information list information element is to provide "CAG information list" or to delete the "CAG information list" at the UE.

The Extended CAG information list information element is coded as shown in figures 9.11.3.86.1 and 9.11.3.86.2 and table 9.11.3.86.1.

The Extended CAG information list is a type 6 information element, with a minimum length of 3 octets.



Figure 9.11.3.86.1: Extended CAG information list information element

8	7	6	5	4	3	2	1	_		
	octet q									
								octet q+1		
	MCC	digit 2			MCC	digit 1		octet q+2		
	MNC	digit 3			MCC	digit 3		octet q+3		
	MNC	digit 2			MNC	digit 1		octet q+4		
0	0	0	0	0	0	0	CAG	octet q+5		
Spare	Spare	Spare	Spare	Spare	Spare	Spare	only			
			CAG	-ID 1				octet q+6*		
								octet q+9*		
								octet q+10*		
			CAG	-ID 2						
								octet q+13*		
								octet q+14*		
	octet q+4m+1*									
	octet q+4m+2*									
	CAG-ID m									
								_octet q+4m+5*		

Figure 9.11.3.86.2: Entry n

Table 9.11.3.86.1: Extended CAG information list information element

Value part of the Extended CAG information list information element (octet 4 to h) The value part of the Extended CAG information list information element consists of one or more entries.

Entry n:

Length of entry contents (octet q and q+1)

MCC, Mobile country code (octet q+2 and bits 1 to 4 octet q+3) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.

MNC, Mobile network code (bits 5 to 8 of octet q+3 and octet q+4) The coding of this field is the responsibility of each administration, but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet q+2 shall be coded as "1111".

The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12].

Indication that the UE is only allowed to access 5GS via CAG cells (CAGonly) (bit 1 of octet q+5)

Bit 1

- 0 "Indication that the UE is only allowed to access 5GS via CAG cells" is not set (i.e., the UE is allowed to access 5GS via non-CAG cells)
- 1 "Indication that the UE is only allowed to access 5GS via CAG cells" is set (i.e., the UE is not allowed to access 5GS via non-CAG cells)

CAG-ID m (octet q+4m+2 to octet q+4m+5)

This field contains the 32 bit CAG-ID. The coding of the CAG-ID is defined as the CAG-Identifier in 3GPP TS 23.003 [4].

NOTE 1: The Length of Extended CAG information list contents shall be 0 if no subscription data for CAG information list exists.

NOTE 2: The Length of entry contents shall be 4 if there is no allowed CAG-ID for the PLMN.

NOTE 3: For a given PLMN ID, there shall be up to one Entry containing the MCC value and the MNC value of the PLMN ID.

NSAG information 9.11.3.87

The purpose of the NSAG information information element is to provide NSAG information to the UE.

The NSAG information information element is coded as shown in figures 9.11.3.87.1, 9.11.3.87.2 and 9.11.3.87.3, and table 9.11.3.87.1.

The NSAG information information element can contain a maximum of 32 NSAG entries.

In the NSAG information information element, at most 4 NSAG entries can contain a TAI list.

The NSAG information is a type 6 information element, with a minimum length of 9 octets and a maximum length of 3143 octets.

8	7	6	5	4	3	2	1	
		N	SAG info	rmation	IEI			octet 1
	I	_ength of	NSAG ir	nformatio	n content	S		octet 2 octet 3
			NSA	AG 1				octet 4
								octet m
			NSA	AG 2				octet m+1*
								octet n*
								octet n+1*
								octet u*
			NSA	AC v				octet u+1*
			1107					octet v*

Figure 9.11.3.87.1: NSAG information information element



Figure 9.11.3.87.2: NSAG





Table 9.11.3.87.1: NSAG information information element

9.11.3.88 ProSe relay transaction identity

The purpose of the ProSe relay transaction identity (PRTI) information element is to uniquely identify an authentication and key agreement procedure for 5G ProSe UE-to-network relay. The PRTI allows distinguishing up to 254 different bi-directional messages.

The ProSe relay transaction identity information element is coded as shown in figure 9.11.3.88.1 and table 9.11.3.88.1.

The ProSe relay transaction identity is a type 3 information element with a length of 2 octets.



Table 9.11.3.88.1: ProSe relay transaction identity information elem	ent
--	-----

Pr Bit	ProSe relay transaction identity value (octet 2)									
0		6	5	4	2	2	4			
0	0	0	0	4 0	0	0	0	No ProSe relay transaction identity assigned		
0	0	0	0	0	0	0	1	\		
to								} ProSe relay transaction identity value		
1	1	1	1	1	1	1	0	/		
1	1	1	1	1	1	1	1	Reserved		

9.11.3.89 Relay key request parameters

The purpose of the relay key request parameters information element is to transport the parameters of the key request for 5G ProSe UE-to-network relay as specified in 3GPP TS 33.503 [56].

The relay key request parameters information element is coded as shown in figure 9.11.3.89.1, figure 9.11.3.89.2 and table 9.11.3.89.1.

The relay key request parameters is a type 6 information element.







Figure 9.11.3.89.2: Remote UE identity



Relay service code (octet 4 to 6) The relay service code contains 24-bit relay service code as defined in 3GPP TS 24.554 [19E].
Nonce_1 (octet 7 to 22) Nonce_1 is the 128-bit nonce value as defined in 3GPP TS 24.554 [19E].
Remote UE ID type (RUIT) (octet 23, bit 1) Bit 1
0 SUCI 1 CP-PRUK ID
Remote UE ID (octet 23+1 to n)
Remote UE ID indicates the value of the 5G ProSe remote UE identity.
If the Remote UE ID type is set to SUCI, the Remote UE ID is coded as 5GS mobile
identity IE starting from octet 2 with the Type of identity set to "SUCI" (see
subclause 9.11.3.4).
If the Remote UE ID type is set to 5GPRUK ID, the Remote UE ID is coded as the
364KUK ID as delined in 3644 13 33.303 [30].

9.11.3.90 Relay key response parameters

The purpose of the relay key response parameters information element is to transport the parameters of the key response for 5G ProSe UE-to-network relay as specified in 3GPP TS 33.503 [56].

The relay key response parameters information element is coded as shown in figure 9.11.3.90.1 and table 9.11.3.90.1.

The relay key response parameters is a type 6 information element.

8	7	6	5	4	3	2	1	
		Relay ke	y respon	ise paran	neters IEI			octet 1
	Le	ngth of R	elay key	response	paramet	ers		octet 2
		-						octet 3
								octet 4
			Key K	NR_ProSe				
								octet 35
								octet 36
			Non	ce_2				
								octet 51
								octet 52
			CP-PF	ruk id				
								octet m

Figure 9.11.3.90.1: Relay key response parameters information element

Table 9.11.3.90.1: Rela	y ke	y response	parameters	information	element
-------------------------	------	------------	------------	-------------	---------

Key K_{NR_ProSe} (octet 5 to 35) Key K_{NR_ProSe} contains a 256-bit root key that is established between the two entities that communicating using NR PC5 unicast link as defined in 3GPP TS 33.503 [56]. Nonce_2 (octet 36 to 51) Nonce_2 is the 128-bit nonce value as defined in 3GPP TS 24.554 [19E]. CP-PRUK ID (octet 52 to m) The CP-PRUK ID is defined in 3GPP TS 33.503 [56].

9.11.3.91 Priority indicator

The purpose of the Priority indicator information element is to convey a priority indication to the UE.

The Priority indicator information element is coded as shown in figure 9.11.3.91.1 and table 9.11.3.91.1.

The Priority indicator is a type 1 information element.



Table 9.11.3.91.1: Priority indicator

MPS indicator (MPSI) (octet 1, bit 1)
Bit
1
0 Access identity 1 not valid
1 Access identity 1 valid
Bits 2, 3, 4 are spare and shall be coded as zero.

9.11.4 5GS session management (5GSM) information elements

9.11.4.1 5GSM capability

The purpose of the 5GSM capability information element is to indicate UE capability related to the PDU session management.

The 5GSM capability information element is coded as shown in figure 9.11.4.1.1 and table 9.11.4.1.1.

The 5GSM capability is a type 4 information element with a minimum length of 3 octets and a maximum length of 15 octets.

8	7	6	5	4	3	2	1				
		5	GSM cap	bability IE				octet 1			
		octet 2									
TPMIC		ATSS	S-ST		EPT-	MH6-	RqoS				
					S1	PDU		octet 3			
						_	1 51/0				
0	0	0	0	0	0	0	APIMQ	octet 4 [°]			
Spare	Spare	Spare	Spare	Spare	Spare	Spare	F				
0	0	0	0	0	0	0	0				
	Spare										





50 R	SSN	1 ca	pak	ility value								
Tł	This bit indicates the 5GSM capability to support reflective QoS.											
0) Reflective QoS not supported											
1				Reflective QoS supported								
М	ulti-l	hom	ned	IPv6 PDU session (MH6-PDU) (octet 3, bit 2)								
Th	nis b	oit ir	ndic	ates the 5GSM capability for Multi-homed IPv6 PDU session.								
1		Multi-homed IPv6 PDU session not supported										
Et	heri	net	PDI	N type in S1 mode (EPT-S1) (octet 3, bit 3)								
0	115 0	11 11	luic	Ethernet PDN type in S1 mode not supported								
1				Ethernet PDN type in S1 mode supported								
Sı	ippo	orte	d A	TSSS steering functionalities and steering modes (ATSSS-ST) (octet 3,								
bit	s 4	to 7	7)									
Th	nese	e bit	s in	dicate the 5GSM capability of ATSSS steering functionalities and steering								
0	0	0	0	ATSSS not supported								
0	0	0	1	ATSSS Low-Layer functionality with any steering mode supported								
0	0	1	0	MPTCP functionality with any steering mode and ATSSS-LL functionality								
	•			with only active-standby steering mode supported								
0	0	1	1	with any steering mode supported								
AI	l oth	ner v	valu	les are reserved.								
тг	200	for	of n	ort management information containers (TDMIC) (actor 3, bit 9)								
٦ŀ	nis b	oit ir	ndic	ates the 5GSM capability to support transfer of port management								
int	orm	natio	on c	containers								
0				Transfer of port management information containers not supported								
1				mansier of port management mormation containers supported								
Ac	ces	s p	erfo	ormance measurements per QoS flow rule (APMQF) (octet 4, bit1)								
	iis b	nt ir the	ndic Oo	ates the 5GSM capability to support access performance measurements								
(S	DF)	tra	ffic.									
Ò				Access performance measurements per QoS flow not supported.								
1				Access performance measurements per QoS flow supported.								
AI	l oth	ner l	oits	in octet 4 to 15 are spare and shall be coded as zero, if the respective								
oc	tet i	is in	clu	ded in the information element.								
1												

9.11.4.2 5GSM cause

The purpose of the 5GSM cause information element is to indicate the reason why a 5GSM request is rejected.

827

The 5GSM cause information element is coded as shown in figure 9.11.4.2.1 and table 9.11.4.2.1.

The 5GSM cause is a type 3 information element with 2 octets length.





Ca	use	valı	ne (d	octe	t 2)				
Bits	s _	~	-		~	~			
8	1	6	5	4	3	2	1		
0	0	0	0	1	0	0	0	Operator determined barring	
0	0	0	1	1	0	1	0	Insufficient resources	
0	0	0	1	1	0	1	1	Missing or unknown DNN	
0	0	0	1	1	1	0	0	Unknown PDU session type	
0	0	0	1	1	1	0	1	User authentication or authorization failed	
0	0	0	1	1	1	1	1	Request rejected, unspecified	
0	0	1	0	0	0	0	0	Service option not supported	
0	0	1	0	0	0	0	1	Requested service option not subscribed	
0	0	1	0	0	0	1	1	PTI already in use	
0	0	1	0	0	1	0	0	Regular deactivation	
0	0	1	0	0	1	0	1	5GS QoS not accepted	
0	0	1	0	0	1	1	0	Network failure	
0	0	1	0	0	1	1	1	Reactivation requested	
0	0	1	0	1	0	0	1	Semantic error in the TFT operation	
Ō	0	1	0	1	Õ	1	0	Syntactical error in the TFT operation	
õ	Õ	1	Õ	1	õ	1	1	Invalid PDU session identity	
õ	õ	1	Õ	1	1	0	0	Semantic errors in packet filter(s)	
õ	õ	1	Õ	1	1	Õ	1	Syntactical error in packet filter(s)	
ő	õ	1	õ	1	1	1	ò	Out of LADN service area	
õ	õ	1	ñ	1	1	1	1	PTI mismatch	
ő	õ	1	1	ò	ò	1	ò	PDLI session type IPv4 only allowed	
0	ň	1	1	ň	ñ	1	1	PDLL session type IPv6 only allowed	
0	ň	1	1	0	1	1	0	PDU session does not exist	
0	0	1	1	1	0	0	1	PDU session type IPv4v6 only allowed	
0	0	1	1	1	0	1	0	PDU session type II v4v0 only allowed	
0	0	' 1	1	1	0	1	1	Linsupported 50L value	
0	۰ ١	1	1	1	1	^	1	PDLL session type Ethernet only allowed	
0	1	0	0	0	0	1	1	I DO Session type Ethemet only allowed	
0	1	0	0	0	1	0	0	Not supported SSC mode	
0	1	0	0	0	1	0	1	Insufficient resources for epocific clice	
0	1	0	0	0	1	1	0	Missing or upknown DNN in a clica	
0	1	0	1	0	0	0	1	INISSING OF UNKNOWN DINN IN a Since	
0	1	0	1	0	0	0		Invalio PTI value	
0	1	0	1	0	0	1	0	maximum data rate per UE for user-plane	
~		~		0	~			Integrity protection is too low	
0	1	0	1	0	0	1	1	Semantic error in the QoS operation	
0	1	0	1	0	1	0	0	Syntactical error in the QoS operation	
0	1	0	1	0	1	0	1	Invalid mapped EPS bearer identity	
0	1	0	1	0	1	1	0	UAS services not allowed	
0	1	0	1	1	1	1	1	Semantically incorrect message	
0	1	1	0	0	0	0	0	Invalid mandatory information	
0	1	1	0	0	0	0	1	Message type non-existent or not implemented	
0	1	1	0	0	0	1	0	Message type not compatible with the protocol	
			_	_				state	
0	1	1	0	0	0	1	1	Information element non-existent or not	
								implemented	
0	1	1	0	0	1	0	0	Conditional IE error	
0	1	1	0	0	1	0	1	Message not compatible with the protocol state	
0	1	1	0	1	1	1	1	Protocol error, unspecified	
Any	y otł	ner	valu	e re	ceiv	ed	by t	he UE shall be treated as 0001 1111, " Request rejected,	
uns	spec	ifie	d ". /	Any	othe	er v	alue	e received by the network shall be treated as 0110 1111,	
"pro	"protocol error, unspecified".								
9.11.4.3 Always-on PDU session indication

The purpose of the Always-on PDU session indication information element is to indicate whether a PDU session is established as an always-on PDU session.

The Always-on PDU session indication information element is coded as shown in figure 9.11.4.3.1 and table 9.11.4.3.1.

The Always-on PDU session indication is a type 1 information element.

8	7	6	5	4	3	2	1	
Alw	/ays-on F	DU sess	ion	0	0	0	APSI	octet 1
	indication IEI				Spare	Spare		

Figure 9.11.4.3.1: Always-on PDU session indication

Always-on PDU session indication (APSI) (octet 1) Bit 1 0 Always-on PDU session not allowed 1 Always-on PDU session required Bits 2, 3 and 4 are spare and shall be coded as zero,

9.11.4.4 Always-on PDU session requested

The purpose of the Always-on PDU session requested information element is to indicate whether a PDU session is requested to be established as an always-on PDU session.

The Always-on PDU session requested information element is coded as shown in figure 9.11.4.4.1 and table 9.11.4.4.1.

The Always-on PDU session requested is a type 1 information element.



Table 9.11.4.4.1: Always-on PDU session requested

Always-on PDU session requested (APSR) (octet 1) Bit 1 0 Always-on PDU session not requested 1 Always-on PDU session requested Bits 2, 3 and 4 are spare and shall be coded as zero,

9.11.4.5 Allowed SSC mode

The purpose of the Allowed SSC mode information element is to indicate the SSC modes allowed to be used by the UE for the PDU session.

The Allowed SSC mode information element is coded as shown in figure 9.11.4.5.1 and table 9.11.4.5.1.

The Allowed SSC mode is a type 1 information element.



SSC1 (octet 1, bit 1	
Bit	
1	
0	SSC mode 1 not allowed
1	SSC mode 1 allowed
SSC2 (octet 1, bit 2) Bit)
2	
0	SSC mode 2 not allowed
1	SSC mode 2 allowed
SSC3 (octet 1, bit 3) Bit)
3	
0	SSC mode 3 not allowed
1	SSC mode 3 allowed
Bit 4 is spare and sh	nall be encoded as zero.

Table 9.11.4.5.1: Allowed SSC mode information element

9.11.4.6 Extended protocol configuration options

See subclause 10.5.6.3A in 3GPP TS 24.008 [12].

9.11.4.7 Integrity protection maximum data rate

The purpose of the integrity protection maximum data rate information element is for the UE to indicate to the network the maximum data rate per UE for user-plane integrity protection for uplink and the maximum data rate per UE for user-plane integrity protection for downlink that are supported by the UE.

The integrity protection maximum data rate is coded as shown in figure 9.11.4.7.1 and table 9.11.4.7.2.

The integrity protection maximum data rate is a type 3 information element with a length of 3 octets.



Figure 9.11.4.7.1: Integrity protection maximum data rate information element

Table 9.11.4.7.2: Integrity protection maximum data rate information element

Ma	axim	านm	da	ta ra	ate	per	UE	for user-plane integrity protection for uplink (octet 2)
Bi	ts							
8	7	6	5	4	3	2	1	
0	0	0	0	0	0	0	0	64 kbps (NOTE 3)
0	0	0	0	0	0	0	1	NULL (NOTE 1)
1	1	1	1	1	1	1	1	Full data rate (NOTE 2)
Al ve	l oth rsio	n of	/alu f thi	es a s sp	are Deci	spa ficat	re a tion.	nd shall not be used by a UE compliant to the present If received they shall be interpreted as "64 kbps".
Ma	axim	num	da	ta ra	ate	per	UE	for user-plane integrity protection for downlink (octet 3)
0	15 7	6	5	4	2	2	1	
0	6	0	0	4	 ∩	2	0	64 kbps (NOTE 2)
0	0	0	0	0	0	0	1	
1	1	1	1	1	1	1	1	Full data rate (NOTE 2)
				1	'	'		
AI	l oth	ner v	/alu	es a	are	spa	re a	nd shall not be used by a UE compliant to the present
ve	rsio	n of	fthi	s sp	eci	ficat	tion.	If received they shall be interpreted as "64 kbps".
N	DTE	1:	Th	is v	alu	e sh	all k	be used when N3 data transfer is not supported by the UE
			or	whe	en t	he I	JE d	loes not support standalone NR connected to 5GCN.
N	DTE	2:	lf t	he	UE	sup	port	s N3 data transfer and supports standalone NR connected
			to	5G	ĊN	(this	s inc	ludes UEs supporting NR-NR dual connectivity, NR-E-
			U	FRA	du	àl c	onn	ectivity with MN terminated bearers or both of them as
			de	scri	bec	l in	3GF	P TS 37.340 [51]), then the UE shall use this value.
N	OTE	3:	Th	ie n	etw	ork	can	receive this value from a UE compliant to an earlier version
			of	this	sp	ecifi	cati	on.

9.11.4.8 Mapped EPS bearer contexts

The purpose of the mapped EPS bearer contexts information element is to indicate a set of EPS bearer contexts for a PDU session, as described in subclause 6.1.4.1.

The mapped EPS bearer contexts information element is a type 6 information element with a minimum length of 7 octet and a maximum length of 65538 octets.

The mapped EPS bearer contexts information element is coded as shown in figure 9.11.4.8.1, figure 9.11.4.8.2, figure 9.11.4.8.3 and table 9.11.4.8.1.



Figure 9.11.4.8.1: Mapped EPS bearer contexts

8	7	6	5	4	3	2	1		
	EPS bearer identity								
	Len	gth of Ma	pped E	PS be	arer cor	itext		octet 5	
0									
Opera	ation	0	E bit		Number	of EPS	5	octet 7	
COC	de	Spare			paran	neters			
		EPS	S param	neters	list			octet 8*	
								octet u*	



8	7	6	5	4	3	2	1			
	EPS parameter identifier 1									
	Le	ngth of I	EPS pai	rameter	conten	ts 1		octet 9		
		EPS	barame	ter conte	ents 1			octet 10		
								octet h		
		EPS	paramet	ter ident	tifier 2			octet h+1		
	Length of EPS parameter contents 2									
	EPS parameter contents 2									
	·									
								octet i+1		
								octet j		
		EPSp	paramet	ter ident	ifier N			octet j+1		
Length of EPS parameter contents N							octet j+2			
		EPSp	paramet	ter conte	ents N			octet j+3		
								octet u		
C :		4 4 0 0.			4 a n a 1 ! .	- 4				

Figure 9.11.4.8.3: EPS parameters list

Table 9.11.4.8.1: Mapped EPS bearer contexts information element

EPS bearer identity (octet 4) Bits 5 to 8 contain the EPS bearer identity, and are coded as specified in subclause 9.3.2 of 3GPP TS 24.301 [15]. Bits 1 to 4 are spare and shall be coded as zero. Operation code (bits 8 to 7 of octet 7) Bits 87 0 0 Reserved 0 1 Create new EPS bearer 1 0 Delete existing EPS bearer 1 1 Modify existing EPS bearer Bit 6 of octet 7 is spare and shall be coded as zero. E bit (bit 5 of octet 7) For the "create new EPS bearer" operation, the E bit is encoded as follows: Bit 5 0 parameters list is not included (NOTE) parameters list is included 1 For the "modify existing EPS bearer" operation, the E bit is encoded as follows: Bit 5 extension of previously provided parameters list 0 replacement of all previously provided parameters list 1 If the E bit is set to "parameters list is included", the number of EPS parameters field has non-zero value. If the E bit is set to "extension of previously provided parameters list" or "replacement of previously provided parameters list", the number of parameters field has non-zero value. For the "create new EPS bearer" operation and "delete existing EPS bearer" operation, bit 5 of octet 7 is ignored. Number of EPS parameters (bits 4 to 1 of octet 7) The number of EPS parameters contains the binary coding for the number of EPS parameters in the EPS parameters list field. The number of EPS parameters field is encoded in bits 4 through 1 of octet 7 where bit 4 is the most significant and bit 1 is the least significant bit. EPS parameters list (octets 8 to u) The EPS parameters list contains a variable number of EPS parameters. Each EPS parameter included in the EPS parameters list is of variable length and consists of: an EPS parameter identifier (1 octet); the length of the EPS parameter contents (1 octet); and the EPS parameter contents itself (variable amount of octets). The EPS parameter identifier field is used to identify each EPS parameter included in the EPS parameters list and it contains the hexadecimal coding of the EPS parameter identifier. Bit 8 of the EPS parameter identifier field contains the most significant bit and bit 1 contains the least significant bit. In this version of the protocol, the following EPS parameter identifiers are specified: 01H (Mapped EPS QoS parameters); 02H (Mapped extended EPS QoS parameters); and 03H (Traffic flow template). 04H (APN-AMBR). 05H (extended APN-AMBR). If the EPS parameters list contains an EPS parameter identifier that is not supported by the receiving entity the corresponding EPS parameter shall be discarded.

The length of EPS parameter contents field contains the binary coded representation of the length of the EPS parameter contents field. The first bit in transmission order is the most significant bit.

When the parameter identifier indicates mapped EPS QoS parameters, the length and parameter contents field are coded as specified in subclause 9.9.4.3 of 3GPP TS 24.301 [15].

When the parameter identifier indicates mapped extended EPS QoS parameters, the length and parameter contents field are coded as specified in subclause 9.9.4.30 of 3GPP TS 24.301 [15].

When the parameter identifier indicates traffic flow template, the length and parameter contents field are coded from octet 2 as shown figure 10.5.144 and table 10.5.16.2 of 3GPP TS 24.008 [12].

When the parameter identifier indicates APN-AMBR, the length and parameter contents field are coded as specified in subclause 9.9.4.2 of 3GPP TS 24.301 [15].

When the parameter identifier indicates Extended APN-AMBR, the length and parameter contents field are coded as specified in subclause 9.9.4.29 of 3GPP TS 24.301 [15]. NOTE: This value shall not be used In this version of the specification.

9.11.4.9 Maximum number of supported packet filters

The purpose of the Maximum number of supported packet filters information element is for the UE to indicate to the network the maximum number of packet filters, associated with signaled QoS rules, that can be supported by the UE for the PDU session that is being established, when the PDU session type "IPv4", "IPv6", "IPv4v6" or "Ethernet".

The Maximum number of supported packet filters is coded as shown in figure 9.11.4.9.1 and table 9.11.4.9.1.

The Maximum number of supported packet filters is a type 3 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	
	Maxim	num num	ber of sup	ported p	acket filte	ers IEI		octet 1
	Max	imum nui	mber of s	upported	packet fi	lters		octet 2
Maxin support ((hum num ted packe continuec	ber of et filters I)	0 Spare	0 Spare	0 Spare	0 Spare	0 Spare	octet 3

Figure 9.11.4.9.1: Maximum number of supported packet filters information element

Table 9.11.4.9.1: Maximum number of supported packet filters information element

Maximum number of supported packet filters (octet 2 to 3)

In the Maximum number of supported packet filters field bit 8 of the first octet is the most significant bit and bit 6 of second octet is the least significant bit. Bit 5 to bit 1 of the second octet are spare bits and shall be coded as zero. The number of supported packet filters shall be in the range of 17 to 1024.

9.11.4.10 PDU address

The purpose of the PDU address information element is to assign to the UE:

- an IPv4 address associated with a PDU session;
- an interface identifier for the IPv6 link local address associated with the PDU session; or
- an IPv4 address and an interface identifier for the IPv6 link local address, associated with the PDU session.

This purpose of the PDU address information element is also to enable the W-AGF acting on behalf of the FN-RG to provide an interface identifier for the IPv6 link local address associated with the PDU session suggested to be allocated

to the FN-RG, and to enable the SMF to provide SMF's IPv6 link local address to the W-AGF acting on behalf of the FN-RG.

The PDU address information element is coded as shown in figure 9.11.4.10.1 and table 9.11.4.10.1.

The PDU address is a type 4 information element with minimum length of 7 octets and a maximum length of 31 octets.

8	7	6	5	4	3	2	1	
			PDU ad	dress IEI				octet 1
		Length	of PDU a	address c	ontents			octet 2
0	0	0	0	SI6LLA	PDU se	ession type	value	octet 3
Spare	Spare	Spare	Spare					
	octet 4							
	octet n							
		octet (n+1)*						
SMF's IPv6 link local address								
								octet (n+16

Figure 9.11.4.10.1: PDU address information element



PD	U s	ession	type value (octet 3, bits 1 to 3)
Bits	5		
3	2	1	
0	0	1	IPv4
0	1	0	IPv6
0	1	1	IPv4v6
All	oth	er valu	es are reserved.
SI6 Bit 4	LL/	A (SMF	's IPv6 link local address) bit (octet 3, bit 4) (see NOTE)
0			SMF's IPv6 link local address field is absent
1			SMF's IPv6 link local address field is present
Bits	s 5 '	to 8 of	octet 3 are spare and shall be coded as zero.
PD	U a	ddress	information (octet 4 to n)
lf th octo	ne F et 7	PDU se ' contai	ssion type value indicates IPv4, the PDU address information in octet 4 to ins an IPv4 address.
lf th octo	ne F et 1	PDU se 1 conta	ssion type value indicates IPv6, the PDU address information in octet 4 to ains an interface identifier for the IPv6 link local address.
lf th to c to c	ne F octe octe	PDU se et 11 cc et 15 cc	ession type value indicates IPv4v6, the PDU address information in octet 4 ontains an interface identifier for the IPv6 link local address and in octet 12 ontains an IPv4 address.
SM	F's	IPv6 li	nk local address (octet n+1 to n+16)
SM	F's	IPv6 li	nk local address field contains SMF's IPv6 link local address.
NO	TE	: In loc	the UE to network direction, the SI6LLA bit shall be set to "SMF's IPv6 link cal address field is absent".

9.11.4.11 PDU session type

The purpose of the PDU session type information element is to indicate type of the PDU session.

The PDU session type information element is coded as shown in figure 9.11.4.11.1 and table 9.11.4.11.1.

The PDU session type is a type 1 information element.

8	7	6	5	4	3	2	1	
	PDU sessi	on type II	ΞΙ	0	PDU se	ession typ	be value	octet 1
				Spare				

Figure 9.11.4.11.1: PDU session type information element

PDU session type value (octet 1, bit 1 to bit 3) Bits 32 1 0 0 IPv4 1 IPv6 0 1 0 0 1 IPv4v6 1 Unstructured 0 1 0 1 0 1 Ethernet 1 1 reserved 1 All other values are unused and shall be interpreted as "IPv4v6", if received by the UE or the network.

9.11.4.12 QoS flow descriptions

The purpose of the QoS flow descriptions information element is to indicate a set of QoS flow descriptions to be used by the UE, where each QoS flow description is a set of parameters as described in subclause 6.2.5.1.1.4.

The QoS flow descriptions information element is a type 6 information element with a minimum length of 6 octets. The maximum length for the information element is 65538 octets.

The QoS flow descriptions information element is coded as shown in figure 9.11.4.12.1, figure 9.11.4.12.2, figure 9.11.4.12.3, figure 9.11.4.12.4, and table 9.11.4.12.1.







Figure 9.11.4.12.2: QoS flow description



1	2	3	4	5	6	7	8
octet 7			[·] identifier	Parameter	F		
octet 8		tents	neter con	h of parar	Lengt		
octet 9			contents	Parameter	F		
octet m							

Figure 9.11.4.12.4: Parameter

Table 9.11.4.12.1: QoS flow descriptions information element

QoS flow identifier (QFI) (bits 6 to 1 of octet 4) QFI field contains the QoS flow identifier. Bits 6 5 4 3 2 1 0 0 0 0 0 0 0 0 QoS flow identifier assigned 0 0 0 0 0 1 QFI 1 to 1 1 1 1 1 1 QFI 63 The network shall not set the QFI value to 0. Operation code (bits 8 to 6 of octet 5) Bits 8 7 6 0 0 1 Create new QoS flow description 0 1 0 Delete existing QoS flow description 0 1 1 Modify existing QoS flow description All other values are reserved.

E bit (bit 7 of octet 6) For the "create new QoS flow description" operation, the E bit is encoded as follows: Bit 0 reserved 1 parameters list is included For the "Delete existing QoS flow description" operation, the E bit is encoded as follows: Bit 7 0 parameters list is not included 1 reserved For the "modify existing QoS flow description" operation, the E bit is encoded as follows: Bit 7 0 extension of previously provided parameters replacement of all previously provided parameters 1 If the E bit is set to "parameters list is not included", the number of parameters field has zero value. If the E bit is set to "parameters list is included", the number of parameters field has non-zero value. If the E bit is set to "extension of previously provided parameters" or "replacement of all previously provided parameters", the number of parameters field has non-zero value. If the E bit is set to "extension of previously provided parameters" and one of the parameters in the new parameters list already exists in the previously provided parameters, the parameter shall be set to the new value. Number of parameters (bits 6 to 1 of octet 6) The number of parameters field contains the binary coding for the number of parameters in the parameters list field. The number of parameters field is encoded in bits 6 through 1 of octet 6 where bit 6 is the most significant and bit 1 is the least significant bit. Parameters list (octets 7 to u) The parameters list contains a variable number of parameters. Each parameter included in the parameters list is of variable length and consists of: a parameter identifier (1 octet); the length of the parameter contents (1 octet); and the parameter contents itself (variable amount of octets). The parameter identifier field is used to identify each parameter included in the parameters list and it contains the hexadecimal coding of the parameter identifier. Bit 8 of the parameter identifier field contains the most significant bit and bit 1 contains the least significant bit. In this version of the protocol, the following parameter identifiers are specified: 01H (5QI); 02H (GFBR uplink); 03H (GFBR downlink); 04H (MFBR uplink); 05H (MFBR downlink); 06H (Averaging window); and 07H (EPS bearer identity). If the parameters list contains a parameter identifier that is not supported by the receiving entity the corresponding parameter shall be discarded. The length of parameter contents field contains the binary coded representation of the length of the parameter contents field. The first bit in transmission order is the most significant bit. When the parameter identifier indicates 5QI, the parameter contents field contains the binary representation of 5G QoS identifier (5QI) that is one octet in length. 5QI: Bits

87654321 00000000 Reserved

00000001	5QI 1
00000010	5QI 2
00000011	5QI 3
00000100	5QI 4
00000101	5QI 5
00000110	5QI 6
00000111	5QI 7
00001000	5QI 8
00001001	5QI 9
00001010	5QI 10
00001011	
to Spare	
01000000	
01000001	5QI 65
01000010	5QI 66
01000011	50167
01000100	Spare
01000101	50169
01000110	50170
01000110	50171
01000111	50172
01001000	50172
01001001	50173
01001010	5Q174
01001011	50176
01001100	50176
	FOI 70
	50179
01010000	5Q1 80
01010001	Spare
01010010	5Q1 82
01010011	5QI 83
01010100	5QI 84
01010101	5QI 85
01010110	5QI 86
01010111	5QI 87
01011000	5QI 88
01011001	5QI 89
01011010	5QI 90
01011011	
to Spare	
01111111	
10000000	
to Operator-	specific 5QIs
1111110	
11111111	Reserved
The network shal	I consider all other values not explicitly defined in this version of the
protocol as unsup	oported.
If the UE receives	s a 5QI value (excluding the reserved 5QI values) that it does not
understand, the l	JE shall choose a 5QI value from the set of 5QI values defined in this
version of the pro	otocol (see 3GPP TS 23.501 [8]) and associated with:
 GBR QoS 	flows, if the QoS flow includes a GFBR uplink parameter, a GFBR
downlink parame	ter, a MFBR uplink parameter and a MFBR downlink parameter; and
 non-GBR 	QoS flows, if the QoS flow does not include any one of a GFBR uplink
parameter, a GFI	3R downlink parameter, a MFBR uplink parameter or a MFBR downlink
parameter.	
The UE shall use	this chosen 5QI value for internal operations only. The UE shall use
the received 5QI	value in subsequent NAS signalling procedures.

When the parameter identifier indicates "GFBR uplink", the parameter contents field contains one octet indicating the unit of the guaranteed flow bit rate for uplink followed by two octets containing the value of the guaranteed flow bit rate for uplink. Unit of the guaranteed flow bit rate for uplink (octet 1) Bits

87654321	
00000000	value is not used (see NOTE 2)
00000001	value is incremented in multiples of 1 Kbps
00000010	value is incremented in multiples of 4 Kbps
	value is incremented in multiples of 16 Kbps
00000100	value is incremented in multiples of 64 Kbps
	value is incremented in multiples of 250 Kbps
	value is incremented in multiples of 1 Mbps
	value is incremented in multiples of 4 Mbps
00001000	value is incremented in multiples of 64 Mbps
00001001	value is incremented in multiples of 256 Mbps
00001010	value is incremented in multiples of 2.50 Mbps
00001011	value is incremented in multiples of 4 Gbps
00001101	value is incremented in multiples of 16 Gbps
00001110	value is incremented in multiples of 64 Gbps
00001111	value is incremented in multiples of 256 Gbps
00010000	value is incremented in multiples of 1 Tbps
00010001	value is incremented in multiples of 4 Tbps
00010010	value is incremented in multiples of 16 Tbps
00010011	value is incremented in multiples of 64 Tbps
00010100	value is incremented in multiples of 256 Tbps
00010101	value is incremented in multiples of 1 Pbps
00010110	value is incremented in multiples of 4 Pbps
00010111	value is incremented in multiples of 16 Pbps
00011000	value is incremented in multiples of 64 Pbps
00011001	value is incremented in multiples of 256 Pbps
Other values sha	Il be interpreted as multiples of 256 Pbps in this version of the protocol.
	ante ad flavo bit acta fan omlink (actata Oland Ol
Value of the guar	anteed flow bit rate for uplink (octets 2 and 3)
Octets 2 and 3 re	present the binary coded value of the guaranteed flow bit rate for
uplink in units dei	ined by the unit of the guaranteed now bit rate for uplink.
When the LIE ind	icates subscribed GEBR for unlink, the "GEBR unlink" parameter is not
included in the "P	Parametere liet"
When the parame	eter identifier indicates "GFBR downlink", the parameter contents field
contains one octe	t indicating the unit of the guaranteed flow bit rate for downlink
followed by two o	ctets containing the value of the guaranteed flow bit rate for downlink.
······	
Unit of the guarar	nteed flow bit rate for downlink (octet 1)
The coding is ide	ntical to that of the unit of the guaranteed flow bit rate for uplink.
J	3
Value of the guar	anteed flow bit rate for downlink (octets 2 and 3)
Octets 2 and 3 re	present the binary coded value of the guaranteed flow bit rate for
downlink in units	defined by the unit of the guaranteed flow bit rate for downlink.
When the UE ind	icates subscribed GFBR for downlink, the "GFBR downlink" parameter
is not included in	the "Parameters list".
When the parame	eter identifier indicates "MFBR uplink", the parameter contents field
contains the one	octet indicating the unit of the maximum flow bit rate for uplink followed
by two octets con	taining the value of maximum flow bit rate for uplink.
Unit of the maxim	um flow bit rate for uplink (octet 1)
The coding is ide	ntical to that of the unit of the guaranteed flow bit rate for uplink.
	man flam hit anta fam miliale (antata O an d O)
Value of the maxi	mum flow bit rate for uplink (octets 2 and 3)
Octets 2 and 3 re	present the binary coded value of the maximum flow bit rate for uplink
in units defined b	y the unit of the maximum now bit rate for uplink.
When the LIF ind	icates subscribed MEBR for unlink the "MERR unlink" parameter is not
included in the "P	arameters list"
When the parame	eter identifier indicates "MFBR downlink". the parameter contents field
contains one octe	et indicating the unit of the maximum flow bit rate for downlink followed
by two octets con	taining the value of the maximum flow bit rate for downlink.

Unit of the maximum flow bit rate for downlink (octet 1) The coding is identical to that of the unit of the guaranteed flow bit rate for uplink. Value of the maximum flow bit rate for downlink (octets 2 and 3) Octets 2 and 3 represent the binary coded value of the maximum flow bit rate for downlink in units defined by the unit of the maximum flow bit rate for downlink. When the UE indicates subscribed MFBR for downlink, the "MFBR downlink" parameter is not included in the "Parameters list". In this version of the protocol, for messages specified in the present document, the sending entity shall not request 0 kbps for both the maximum flow bit rate for downlink and the maximum flow bit rate for uplink at the same time. Any entity receiving a request for 0 kbps in both the maximum flow bit rate for downlink and the maximum flow bit rate for uplink shall consider that as a syntactical error (see clause 7). When the parameter identifier indicates "averaging window", the parameter contents field contains the binary representation of the averaging window for both uplink and downlink in milliseconds and the parameter contents field is two octets in length. When the parameter identifier indicates EPS bearer identity, the length of EPS bearer identity is one octet, bits 5 to 8 of the parameter contents contain the EPS bearer identity as specified in subclause 9.3.2 of 3GPP TS 24.301 [15] (see NOTE) and bits 1 to 4 of the parameter contents are spare and shall be coded as zero. The UE shall not include the EPS bearer identity parameter in any mobile originated 5GSM messages. NOTE 1: The total number of EPS bearer identities included in all QoS flow descriptions of a UE cannot exceed fifteen. In this release of the specifications if received it shall be interpreted as value NOTE 2: is incremented in multiples of 1 Kbps. In earlier releases of specifications, the interpretation of this value is up to implementation.

9.11.4.13 QoS rules

The purpose of the QoS rules information element is to indicate a set of QoS rules to be used by the UE, where each QoS rule is a set of parameters as described in subclause 6.2.5.1.1.2:

- a) for classification and marking of uplink user traffic; and
- b) for identification of a QoS flow which the network is to use for a particular downlink user traffic.
- NOTE: The UE needs to be aware of a QoS flow which the network is to use for a particular downlink user traffic e.g. to determine whether a resource is available for downlink media of a media stream of an SDP media description provided by the UE in an IMS session.

The QoS rules may contain a set of packet filters consisting of zero or more packet filters for UL direction, zero or more packet filters for DL direction, zero or more packet filters for both UL and DL directions or any combinations of these. The set of packet filters determine the traffic mapping to QoS flows.

The QoS rules information element is a type 6 information element with a minimum length of 7 octets. The maximum length for the information element is 65538 octets.

The QoS rules information element is coded as shown in figure 9.11.4.13.1, figure 9.11.4.13.2, figure 9.11.4.13.3, figure 9.11.4.13.4 and table 9.11.4.13.1.







Figure 9.11.4.13.2: QoS rule (u=m+2)





8	7	6	5	4	3	2	1		
0	0	Packe	et filter	Pa	cket filte	r identif	ier 1	octet 8	
S	pare	direc	tion 1						
		Length	of pack	et filter	content	ts 1		octet 9	
		Pa	acket filt	ter cont	ents 1			octet 10	
								octet m	
0	0	Packe	et filter	Pa	cket filte	r identif	ier 2	octet m+1	
S	oare	direc	tion 2						
	octet m+2								
		P	acket filf	ter cont	ents 2			octet m+3	
								octet n	
	octet n+1								
								octet y	
0	0	Packe	et filter	Pad	cket filte	r identif	ier N	octet y+1	
S	oare	direct	tion N						
	Length of packet filter contents N								
		Pa	acket filt	er cont	ents N			octet y+3	
								octet z	

Figure 9.11.4.13.4: Packet filter list when the rule operation is "create new QoS rule", or "modify existing QoS rule and add packet filters" or "modify existing QoS rule and replace all packet filters"

Table 9.11.4.13.1: QoS rules information element

QoS rule identifier (octet 4) The QoS rule identifier field is used to identify the QoS rule. Bits 87654321 00000000 no QoS rule identifier assigned 00000001 QRI 1 to 11111111 **QRI 255** The network shall not set the QRI value to 0. QoS rule precedence (octet m+1) The QoS rule precedence field is used to specify the precedence of the QoS rule among all QoS rules (both the signalled QoS rules as described in subclause 6.2.5.1.1.2 and the derived QoS rules as described in subclause 6.2.5.1.1.3) associated with the PDU session of the QoS flow. This field includes the binary coded value of the QoS rule precedence in the range from 0 to 255 (decimal). The higher the value of the QoS rule precedence field, the lower the precedence of that QoS rule is. For the "delete existing QoS rule" operation, the QoS rule precedence value field shall not be included. For the "create new QoS rule" operation, the QoS rule precedence value field shall be included. The value 80 (decimal) is reserved. Segregation bit (bit 7 of octet m+2) (see NOTE 1) In the UE to network direction the segregation bit indicates whether the UE is requesting the network to bind service data flows described by the QoS rule to a dedicated QoS Flow and it is encoded as follows. In the network to UE direction this bit is spare. Bit 7 0 Segregation not requested Segregation requested 1 QoS flow identifier (QFI) (bits 6 to 1 of octet m+2) (see NOTE 1) The QoS flow identifier (QFI) field contains the QoS flow identifier. Bits 654321 000001QFI1 to 1 1 1 1 1 1 1 QFI 63 The network shall not set the QFI value to 0. For the "delete existing QoS rule" operation, the QoS flow identifier value field shall not be included. For the "create new QoS rule" operation, the QoS flow identifier value field shall be included. DQR bit (bit 5 of octet 7) The DQR bit indicates whether the QoS rule is the default QoS rule and it is encoded as follows: Bit 5 0 the QoS rule is not the default QoS rule. the QoS rule is the default QoS rule. 1 Rule operation code (bits 8 to 6 of octet 7) Bits 876 000 Reserved 0 0 1 Create new QoS rule 0 1 0 Delete existing QoS rule 0 1 1 Modify existing QoS rule and add packet filters 1 0 0 Modify existing QoS rule and replace all packet filters Modify existing QoS rule and delete packet filters 101 1 1 0 Modify existing QoS rule without modifying packet filters 111 Reserved Number of packet filters (bits 4 to 1 of octet 7)

The number of packet filters contains the binary coding for the number of packet filters in the packet filter list. The number of packet filters field is encoded in bits 4 through 1 of octet 7 where bit 4 is the most significant and bit 1 is the least significant bit. For the "delete existing QoS rule" operation and for the "modify existing QoS rule without modifying packet filters" operation, the number of packet filters shall be coded as 0. For the "create new QoS rule" operation and the "modify existing QoS rule and replace all packet filters" operation, the number of packet filters shall be greater than or equal to 0 and less than or equal to 15. For all other operations, the number of packet filters shall be greater than or equal to 15.

Packet filter list (octets 8 to m)

The packet filter list contains a variable number of packet filters.

For the "delete existing QoS rule" operation, the length of QoS rule field is set to one.

For the "delete existing QoS rule" operation and the "modify existing QoS rule without modifying packet filters" operation, the packet filter list shall be empty.

For the "modify existing QoS rule and delete packet filters" operation, the packet filter list shall contain a variable number of packet filter identifiers. This number shall be derived from the coding of the number of packet filters field in octet 7.

For the "create new QoS rule" operation and for the "modify existing QoS rule and replace all packet filters" operation, the packet filter list shall contain 0 or a variable number of packet filters. This number shall be derived from the coding of the number of packet filters field in octet 7.

For the "modify existing QoS rule and add packet filters" operation, the packet filter list shall contain a variable number of packet filters. This number shall be derived from the coding of the number of packet filters field in octet 7.

Each packet filter is of variable length and consists of

- a packet filter direction (2 bits);
- a packet filter identifier (4 bits);
- the length of the packet filter contents (1 octet); and
- the packet filter contents itself (variable amount of octets).

The packet filter direction field is used to indicate for what traffic direction the filter applies.

Bits 6 5

0 0 reserved

0 1 downlink only (see NOTE 2)

- 1 0 uplink only
- 1 1 bidirectional

The packet filter identifier field is used to identify each packet filter in a QoS rule. The least significant 4 bits are used. When the UE requests to "create new QoS rule", "modify existing QoS rule and replace all packet filters" or "modify existing QoS rule and add packet filters", the packet filter identifier values shall be set to 0.

The length of the packet filter contents field contains the binary coded representation of the length of the packet filter contents field of a packet filter. The first bit in transmission order is the most significant bit.

The packet filter contents field is of variable size and contains a variable number (at least one) of packet filter components. Each packet filter component shall be encoded as a sequence of a one octet packet filter component type identifier and a fixed length packet filter component value field. The packet filter component type identifier shall be transmitted first.

In each packet filter, there shall not be more than one occurrence of each packet filter component type. Among the "IPv4 remote address type" and "IPv6 remote address/prefix length type" packet filter components, only one shall be present in one packet filter. Among the "IPv4 local address type" and "IPv6 local address/prefix length type" packet filter components, only one shall be present in one packet filter.

Among the "single local port type" and "local port range type" packet filter components, only one shall be present in one packet filter. Among the "single remote port type" and "remote port range type" packet filter components, only one shall be present in one packet filter. If the "match-all type" packet filter component is present in the packet filter, no other packet filter component shall be present in the packet filter and the length of the packet filter contents field shall be set to one. If the "Ethertype type" packet filter component is present in the packet filter and the "Ethertype type" packet filter component value is neither "0x0800" (for IPv4) nor "0x86DD" (for IPv6), no IP packet filter component shall be present in the packet filter. The term "IP packet filter component" refers to "IPv4 remote address type", "IPv4 local address type", "IPv6 remote address/prefix length type", "IPv6 local address/prefix length type", "Protocol identifier/Next header type", "Single local port type", "Local port range type", "Single remote port type", "Remote port range type", "Security parameter index type", "Type of service/Traffic class type" and "Flow label type". The term local refers to the UE and the term remote refers to an external network entity. Packet filter component type identifier Bits 87654321 00000001 Match-all type (see NOTE 2) 00010000 IPv4 remote address type 00010001 IPv4 local address type 00100001 IPv6 remote address/prefix length type 00100011 IPv6 local address/prefix length type 00110000 Protocol identifier/Next header type 01000000 Single local port type 0100001 Local port range type 01010000 Single remote port type 01010001 Remote port range type 01100000 Security parameter index type 01110000 Type of service/Traffic class type 10000000 Flow label type 1000001 Destination MAC address type 10000010 Source MAC address type 10000011 802.1Q C-TAG VID type 10000100 802.1Q S-TAG VID type 10000101 802.1Q C-TAG PCP/DEI type 10000110 802.1Q S-TAG PCP/DEI type 10000111 Ethertype type 10001000 Destination MAC address range type 10001001 Source MAC address range type All other values are reserved. The description and valid combinations of packet filter component type identifiers in a packet filter are defined in 3GPP TS 23.501 [8]. For "match-all type", the packet filter component shall not include the packet filter component value field. For "IPv4 remote address type", the packet filter component value field shall be encoded as a sequence of a four octet IPv4 address field and a four octet IPv4 address mask field. The IPv4 address field shall be transmitted first. For "IPv4 local address type", the packet filter component value field shall be encoded as defined for "IPv4 remote address type". For "IPv6 remote address/prefix length type", the packet filter component value field shall be encoded as a sequence of a sixteen octet IPv6 address field and one octet prefix length field. The IPv6 address field shall be transmitted first.

For "IPv6 local address/prefix length type", the packet filter component value field shall be encoded as defined for "IPv6 remote address /prefix length".

For "protocol identifier/Next header type", the packet filter component value field shall be encoded as one octet which specifies the IPv4 protocol identifier or Ipv6 next header.

For "single local port type" and "single remote port type", the packet filter component value field shall be encoded as two octets which specify a port number.

For "local port range type" and "remote port range type", the packet filter component value field shall be encoded as a sequence of a two octet port range low limit field and a two octet port range high limit field. The port range low limit field shall be transmitted first.

For "security parameter index", the packet filter component value field shall be encoded as four octets which specify the IPSec security parameter index.

For "type of service/traffic class type", the packet filter component value field shall be encoded as a sequence of a one octet type-of-service/traffic class field and a one octet type-of-service/traffic class field. The type-of-service/traffic class field shall be transmitted first.

For "flow label type", the packet filter component value field shall be encoded as three octets which specify the IPv6 flow label. The bits 8 through 5 of the first octet shall be spare whereas the remaining 20 bits shall contain the IPv6 flow label.

For "destination MAC address type" and "source MAC address type", the packet filter component value field shall be encoded as 6 octets which specify a MAC address. When the packet filter direction field indicates "bidirectional", the destination MAC address is the remote MAC address and the source MAC address is the local MAC address.

For "802.1Q C-TAG VID type", the packet filter component value field shall be encoded as two octets which specify the VID of the customer-VLAN tag (C-TAG). The bits 8 through 5 of the first octet shall be spare whereas the remaining 12 bits shall contain the VID. If there are more than one C-TAG in the Ethernet frame header, the outermost C-TAG is evaluated.

For "802.1Q S-TAG VID type", the packet filter component value field shall be encoded as two octets which specify the VID of the service-VLAN tag (S-TAG). The bits 8 through 5 of the first octet shall be spare whereas the remaining 12 bits shall contain the VID. If there are more than one S-TAG in the Ethernet frame header, the outermost S-TAG is evaluated.

For "802.1Q C-TAG PCP/DEI type", the packet filter component value field shall be encoded as one octet which specifies the 802.1Q C-TAG PCP and DEI. The bits 8 through 5 of the octet shall be spare, the bits 4 through 2 contain the PCP and bit 1 contains the DEI. If there are more than one C-TAG in the Ethernet frame header, the outermost C-TAG is evaluated.

For "802.1Q S-TAG PCP/DEI type", the packet filter component value field shall be encoded as one octet which specifies the 802.1Q S-TAG PCP. The bits 8 through 5 of the octet shall be spare, the bits 4 through 2 contain the PCP and bit 1 contains the DEI. If there are more than one S-TAG in the Ethernet frame header, the outermost S-TAG is evaluated.

For "ethertype type", the packet filter component value field shall be encoded as two octets which specify an ethertype.

For "destination MAC address range type", the packet filter component value field shall be encoded as a sequence of a 6 octet destination MAC address range low limit field and a 6 octet destination MAC address range high limit field. The destination MAC address range low limit field shall be transmitted first. When the packet filter direction field indicates "bidirectional", the destination MAC address range is the remote MAC address range.

For "source MAC address range type", the packet filter component value field shall be encoded as a sequence of a 6 octet source MAC address range low limit field and a 6 octet source MAC address range high limit field. The source MAC address

range low	limit field shall be transmitted first. When the packet filter direction field
indicates	"bidirectional", the source MAC address is the local MAC address range.
NOTE 1:	Octet m+2 shall not be included without octet m+1.
NOTE 2:	The "Match-all type" packet filter component type identifier shall not be
	used with packet filter direction "downlink only".

9.11.4.14 Session-AMBR

The purpose of the Session-AMBR information element is to indicate the initial subscribed PDU session aggregate maximum bit rate when the UE establishes a PDU session or to indicate the new subscribed PDU session aggregate maximum bit rate if it is changed by the network.

The Session-AMBR information element is coded as shown in figure 9.11.4.14.1 and table 9.11.4.14.1.

The Session-AMBR is a type 4 information element with a length of 8 octets.

8	7	6	5	4	3	2	1					
	Session-AMBR IEI											
	Length of Session-AMBR contents											
Unit for Session-AMBR for downlink												
		Sess	ion-AMB	R for dov	vnlink			octet 4-5				
	octet 6											
Session-AMBR for uplink												

Figure 9.11.4.14.1: Session-AMBR information element

Table 9.11.4.14.1:	Session-AMBR	information	element
--------------------	--------------	-------------	---------

Unit for Session-	AMBR for downlink (octet 3)					
00000000	value is not used (see NOTE)					
00000001	value is incremented in multiples of 1 Kbps					
00000010	value is incremented in multiples of 4 Kbps					
00000011	value is incremented in multiples of 16 Kbps					
00000100	value is incremented in multiples of 64 Kbps					
00000101	value is incremented in multiples of 256 kbps					
00000110	value is incremented in multiples of 1 Mbps					
00000111	value is incremented in multiples of 4 Mbps					
00001000	value is incremented in multiples of 16 Mbps					
00001001	value is incremented in multiples of 64 Mbps					
00001010	value is incremented in multiples of 256 Mbps					
00001011	value is incremented in multiples of 1 Gbps					
00001100	value is incremented in multiples of 4 Gbps					
00001101	value is incremented in multiples of 16 Gbps					
00001110	value is incremented in multiples of 64 Gbps					
00001111	value is incremented in multiples of 256 Gbps					
00010000	value is incremented in multiples of 1 Tbps					
00010001	value is incremented in multiples of 4 Tbps					
00010010	value is incremented in multiples of 16 Tbps					
00010011	value is incremented in multiples of 64 Tbps					
00010100	value is incremented in multiples of 256 Tbps					
00010101	value is incremented in multiples of 1 Pbps					
00010110	value is incremented in multiples of 4 Pbps					
00010111	value is incremented in multiples of 16 Pbps					
00011000	value is incremented in multiples of 64 Pbps					
00011001	value is incremented in multiples of 256 Pbps					
Other values sha protocol.	Il be interpreted as multiples of 256 Pbps in this version of the					
Session-AMBR f	or downlink (octets 4 and 5)					
Octets 4 and 5 re maximum bit rate	epresent the binary coded value of PDU session aggregated of r downlink in units defined by octet 3.					
Unit for Session-	AMBR for uplink (octet 6)					
The coding is ide (octet 3)	ntical to the unit coding defined for Session-AMBR for downlink					
Session-AMBR for uplink (octets 7 and 8)						
Octets 7 and 8 represent the binary coded value of PDU session aggregated						
maximum bit rate	e for uplink in units defined by octet 6.					
NOTE: In this	release of the specifications if received it shall be interpreted as					
value specif	is incremented in multiples of 1 Kbps. In earlier releases of ications, the interpretation of this value is up to implementation.					

9.11.4.15 SM PDU DN request container

The purpose of the SM PDU DN request container information element is to carry a DN-specific identity of the UE in the network access identifier (NAI) format.

The SM PDU DN request container information element is coded as shown in figure 9.11.4.15.1 and table 9.11.4.15.1.

The SM PDU DN request container is a type 4 information element with minimal length of 3 octets and maximum length of 255 octets.

8	7	6	5	4	3	2	1			
	octet 1									
	octet 2									
	octets 3*-n*									
Figure 9.11.4.15.1: SM PDU DN request container information ele										

Table 9.11.4.15.1: SM PDU DN request container information element

DN-specific identity (octet 3 to octet n) A DN-specific identity of the UE in the network access identifier (NAI) format according to IETF RFC 7542 [37], encoded as UTF-8 string.

9.11.4.16 SSC mode

The purpose of the SSC mode information element is to indicate SSC mode.

The SSC mode information element is coded as shown in figure 9.11.4.16.1 and table 9.11.4.16.1.

The SSC mode is a type 1 information element.

8	7	6	5	4	3	2	1			
SSC mode IEI				0	SSC	C mode v	alue	octet 1		
				Spare						
Figure 0.44.4.46.4. SSC mode information element										

Figure 9.11.4.16.1: SSC mode information element

SS	SSC mode value (octet 1, bit 1 to bit 4)								
Bits									
3	2	1							
0	0	1	SSC mode 1						
0	1	0	SSC mode 2						
0	1	1	SSC mode 3						
1	0	0	unused; shall be interpreted as "SSC mode 1", if received by the network						
1	0	1	unused; shall be interpreted as "SSC mode 2", if received by the network						
1	1	0	unused; shall be interpreted as "SSC mode 3", if received by the network						
All	Il other values are reserved.								

9.11.4.17 Re-attempt indicator

The purpose of the Re-attempt indicator information element is to indicate a condition under which the UE is allowed in the current PLMN or its equivalent PLMN(s) for the same DNN, to re-attempt a session management procedure (see 3GPP TS 24.301 [15]) corresponding to the 5GS session management procedure which was rejected by the network.

The Re-attempt indicator information element is coded as shown in figure 9.11.4.17.1 and table 9.11.4.17.1.

The Re-attempt indicator is a type 4 information element with a length of 3 octets.

	8	7	6	5	4	3	2	1		
	Re-attempt indicator IEI									
	Length of Re-attempt indicator contents									
	0	0	0	0	0	0	EPLM	RATC	octet 3	
	Spare	Spare	Spare	Spare	Spare	Spare	NC			
Figure 9 11 4 17 1: Re-attempt indicator										

Figure 9.11.4.17.1: Re-attempt indicator

	Table 9.11.4.17.1: Re-attempt indicator
RA	VTC (octet 3, bit 1)
Bit	
1	
0	UE is allowed to re-attempt the procedure in S1 mode
1	UE is not allowed to re-attempt the procedure in S1 mode
ΕP	PLMNC (octet 3, bit 2)
Bit	
2	
0	UE is allowed to re-attempt the procedure in an equivalent PLMN
1	UE is not allowed to re-attempt the procedure in an equivalent PLMN
.	
Bit	s 3 to 8 of octet 3 are spare and shall be encoded as zero.

Table 0.11 / 17 1. Do attempt indicator

9.11.4.18 5GSM network feature support

The purpose of the 5GSM network feature support information element is to indicate whether certain session management related features are supported by the network.

The 5GSM network feature support information element is coded as shown in figure 9.11.4.18.1 and table 9.11.4.18.1.

The 5GSM network feature support is a type 4 information element with a minimum length of 3 octets and a maximum length of 15 octets.

8	7	6	5	4	3	2	1	
	octet 1							
	Length	of 5GSM	l network	feature s	upport co	ontents		octet 2
0	0	0	0	0	0	0	EPT-	
Spare	Spare	Spare	Spare	Spare	Spare	Spare	S1	octet 3
0	0	0	0	0	0	0	0	
0	octet 4* -15							

Figure 9.11.4.18.1: 5GSM network feature support information element

Table 9.11.4.18.1: 5GSM network feature sup	pport information element
---	---------------------------

5GSM network feature support contents
Ethernet PDN type in S1 mode (IEPT-S1) (octet 3, bit 1)This bit indicates network's capability for Ethernet PDN type in S1 mode.0Ethernet PDN type in S1 mode not supported1Ethernet PDN type in S1 mode supported
All other bits in octet 3 to 15 are spare and shall be coded as zero, if the respective octet is included in the information element.

9.11.4.19 Void

9.11.4.20 Serving PLMN rate control

See subclause 9.9.4.28 in 3GPP TS 24.301 [15].

9.11.4.21 5GSM congestion re-attempt indicator

The purpose of the 5GSM congestion re-attempt indicator information element is to indicate whether the back-off timer is applied in the registered PLMN or all PLMNs, and additionally to indicate whether the back-off timer is applied in the current access type or both 3GPP access type and non-3GPP access type.

The 5GSM congestion re-attempt indicator information element is coded as shown in figure 9.11.4.21.1 and table 9.11.4.21.1.

The 5GSM congestion re-attempt indicator is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	_	
5GSM congestion re-attempt indicator IEI									
	Length of 5GSM congestion re-attempt indicator contents								
0	0	0	0	0	0	CATB	ABO	octet 3	
Spare	Spare	Spare	Spare	Spare	Spare	0			

Figure 9.11.4.21.1: 5GSM congestion re-attempt indicator

AB	O (All PLMNs Back-off timer) (octet 3, bit 1)
Bit	
1	
0	The back-off timer is applied in the registered PLMN.
1	The back-off timer is applied in all PLMNs.
CA Bit 2	TBO (Current Access Type Back-off Timer) (octet 3, bit 2)
0	The back-off timer is applied in both 3GPP access type and non-3GPP access type
1	The back-off timer is applied in the current access type
Bit	s 3 to 8 of octet 3 are spare and shall be encoded as zero.

Table 9.11.4.21.1: 5GSM congestion re-attempt indicator

9.11.4.22 ATSSS container

The purpose of the ATSSS container information element is to transfer parameters associated with ATSSS.

The ATSSS container information element is coded as shown in figure 9.11.4.22.1 and table 9.11.4.22.1.

The ATSSS container is a type 6 information element with a minimum length of 3 octets and a maximum length of 65538 octets.





ATSSS container contents are defined in 3GPP TS 24.193 [13B].

9.11.4.23 Control plane only indication

The purpose of the control plane only indication information element is to indicate that a PDU session is only for control plane CIoT 5GS optimization.

The control plane only indication information element is coded as shown in figure 9.11.4.23.1.

The control plane only indication is a type 1 information element.

8	7	6	5	4	3	2	1	
Contro	l plane or	nly indica	tion IEI	0	0	0	CPOI	octet 1
				Spare	Spare	Spare	value	

Figure 9.11.4.23.1: Control plane only indication information element

Table 9.11.4.23.1: Control plane only indication information element

Control plane only indication value (CPOI) (octet 1) Bit **1** 0 reserved 1 PDU session can be used for control plane CIoT 5GS optimization only The value 0 is reserved. If received, it shall be interpreted as if the control plane only indication IE was not included in the message. Bits 4 to 2 of octet 1 are spare and shall be all encoded as zero.

9.11.4.24 IP header compression configuration

The purpose of the IP header compression configuration information element is to negotiate ROHC channel setup parameters specified in IETF RFC 5795 [39B] and, optionally, provide additional header compression context setup parameters.

The IP header compression configuration information element is coded as shown in figure 9.11.4.24.1 and table 9.11.4.24.1.

The IP header compression configuration is a type 4 information element with a minimum length of 5 octets and a maximum length of 257 octets.

The optional Additional IP header compression parameters container field conveys the additional header compression context setup parameters as specified in 3GPP TS 23.501 [8] in a generic container. This field corresponds to the profile-specific information in the header of the ROHC IR packet type in IETF RFC 5795 [39B].

_	1	2	3	4	5	6	7	8				
octet 1	IP header compression configuration IEI											
octet 2		ontents	guration co	ssion confi	ler compre	of IP head	Length					
octet 3	P0x0002	P0x0003	P0x0004	P0x0006	P0x0102	P0x0103	P0x0104	Spare				
octet 4 octet 5	MAX_CID											
octet 6'	Additional IP header compression context setup parameters type											
octet 7'	Additional IP header compression context setup parameters container											
octet n'												

Figure 9.11.4.24.1: IP header compression configuration information element

Table 9.11.4.24.1: IP header compression configuration information element

ROHC Profiles (octet 3)

The ROHC Profiles shall indicate which of the ROHC profiles is supported. When a particular bit is set to 1, this indicates that the corresponding profile is supported. The No Compression profile 0x0000 (see IETF RFC 5795 [39B]) shall always be supported. When all the bits are set to 0, this indicates that only the No Compression profile 0x0000 is supported.

Profile 0x0002 support indicator (see IETF RFC 3095 [33A] and IETF RFC 4815 [38A]) (octet 3 bit 1)

- 0 RoHC profile 0x0002 (UDP/IP) is not supported
- 1 RoHC profile 0x0002 (UDP/IP) is supported

Profile 0x0003 support indicator (see IETF RFC 3095 [33A] and IETF RFC 4815 [38A]) (octet 3 bit 2)

- 0 RoHC profile 0x0003 (ESP/IP) is not supported
- 1 RoHC profile 0x0003 (ESP/IP) is supported

Profile 0x0004 support indicator (see IETF RFC 3843 [34A] and IETF RFC 4815 [38A]) (octet 3 bit 3)

- 0 RoHC profile 0x0004 (IP) is not supported
- 1 RoHC profile 0x0004 (IP) is supported

Profile 0x0006 support indicator (see IETF RFC 6846 [40B]) (octet 3 bit 4)

- 0 RoHC profile 0x0006 (TCP/IP) is not supported
- 1 RoHC profile 0x0006 (TCP/IP) is supported

Profile 0x0102 support indicator (see IETF RFC 5225 [39A]) (octet 3 bit 5)

- 0 RoHC profile 0x0102 (UDP/IP) is not supported
- 1 RoHC profile 0x0102 (UDP/IP) is supported

Profile 0x0103 support indicator (see IETF RFC 5225 [39A]) (octet 3 bit 6)

- 0 RoHC profile 0x0103 (ESP/IP) is not supported
- 1 RoHC profile 0x0103 (ESP/IP) is supported

Profile 0x0104 support indicator (see IETF RFC 5225 [39A]) (octet 3 bit 7)

- 0 RoHC profile 0x0104 (IP) is not supported
- 1 RoHC profile 0x0104 (IP) is supported

Bits 8 is spare and shall be set to 0.

MAX_CID (octet 4 and octet 5)

This is the MAX_CID value as specified in 3GPP TS 36.323 [25]. It is encoded in binary coding with a value in the range from 1 to 16383.

Additional IP header compression context parameters type (octet 6).

The Additional IP header compression context parameters type octet indicates the profile associated with the profilespecific information in the Additional IP header compression context parameters container.

Bits 8 7 6 54 3 2 1 Type

00000000 0x0000 (No Compression) 0x0002 (UDP/IP) 0000001 0000010 0x0003 (ESP/IP) 0000011 0x0004 (IP) 0x0006 (TCP/IP) 0000100 0x0102 (UDP/IP) 00000101 0000110 0x0103 (ESP/IP) 0000111 0x0104 (IP) 00001000 Other



9.11.4.25 DS-TT Ethernet port MAC address

The purpose of the DS-TT Ethernet port MAC address information element is to signal the MAC address of the DS-TT Ethernet port used for a PDU session of "Ethernet" PDU session type.

The DS-TT Ethernet port MAC address information element is coded as shown in figure 9.11.4.25.1 and table 9.11.4.25.1.

The DS-TT Ethernet port MAC address is a type 4 information element with a length of 8 octets.



Table 9.11.4.25.1: DS-TT Ethernet port MAC address information element

DS-TT Ethernet port MAC address contents (octets 3 to 8)

The DS-TT Ethernet port MAC address contents consist of the binary representation of the MAC address of the DS-TT Ethernet port used for the PDU session, starting with the LSB bit of the first octet of the MAC address included in bit 1 of octet 3.

9.11.4.26 UE-DS-TT residence time

The purpose of the UE-DS-TT residence time information element is to signal the time taken within the UE and DS-TT to forward a packet between the UE and the DS-TT port.

The UE- DS-TT residence time information element is coded as shown in figure 9.11.4.26.1 and table 9.11.4.26.1.

The UE-DS-TT residence time is a type 4 information element with a length of 10 octets.





Table 9.11.4.26.1: UE-DS-TT residence time information element

UE-DS-TT residence time contents (octets 3 to 10)

The UE-DS-TT residence time contents contain the UE-DS-TT residence time encoded as specified for the correctionField in IEEE Std 1588-2019 [43B], with the LSB bit of the first octet of the UE-DS-TT residence time included in bit 1 of octet 3. If the UE-DS-TT residence time is too big to be represented, all bits of octets 3 to 10 shall be coded as "1" except the MSB bit of octet 10.

9.11.4.27 Port management information container

The purpose of the Port management information container information element is to transport a port management service message as specified in clause 8 of 3GPP TS 24.539 [19BA].

The Port management information container information element is coded as shown in figure 9.11.4.27.1 and table 9.11.4.27.1.

The Port management information container is a type 6 information element with a minimum length of 4 octets and a maximum length of 65538 octets.



Figure 9.11.4.27.1: Port management information container information element

Table 9.11.4.27.1: Port management information container information element

Port management information container (octet 4 to n) A port management service message as specified in clause 8 of 3GPP TS 24.539 [19BA].

9.11.4.28 Ethernet header compression configuration

The purpose of the Ethernet header compression configuration information element is to negotiate the use of EHC and the length of the CID field in the EHC packet (see 3GPP TS 38.323 [29]).

The Ethernet header compression configuration information element is coded as shown in figure 9.11.4.28.1 and table 9.11.4.28.1.

The Ethernet header compression configuration is a type 4 information element with the length of 3 octets.

8	7	6	5	4	3	2	1			
Ethernet header compression configuration IEI										
	Length of Ethernet header compression configuration contents									
0	0	0	0	0	0		onath	ootot 2		
Spare	Spare	Spare	Spare	Spare	Spare		ociel 3			

Figure 9.11.4.28.1: Ethernet header compression configuration information element

Table 9.11.4.28.1: Ethernet header compression configuration information element

Length of CID field value (CID Length) (octet 3 bits 1 and 2)									
Bit									
2 1									
0 0 Ethernet header compression not used									
0 1 7 bits									
1 0 15 bits									
All other values shall be interpreted as "7 bits".									
Bits 3 to 8 of octet 3 are spare and shall be coded as zero.	Bits 3 to 8 of octet 3 are spare and shall be coded as zero.								

9.11.4.29 Remote UE context list

The purpose of the Remote UE context list information element is to provide identity and optionally IP address of a 5G ProSe remote UE connected to, or disconnected from, a UE acting as a 5G ProSe layer-3 UE-to-network relay.

The Remote UE context list information element is coded as shown in figure 9.11.4.29.1, figure 9.11.4.29.2, table 9.11.4.29.1 and table 9.11.4.29.2.

The Remote UE context list is a type 6 information element with a minimum length of 16 octets and a maximum length of 65538 octets.



Figure 9.11.4.29.1: Remote UE context list

Table 9.11.4.29.1: Remote UE context list

Remote UE context (octet 5 etc)

The contents of remote UE context are applicable for one individual UE and are coded as shown in figure 9.11.4.29.2 and table 9.11.4.29.2.

8	7	6	5	4	3	2	1	_	
		Lengt	h of remo	ote UE co	ontext			octet 5	
0	0	0	0	Remot				octet 6	
Spare	Spare	Spare	Spare	e UE	Remo	ote UE ID	type		
				ID					
				format					
		Ler	igth of re	mote UE	ID			octet 7	
								octet 8	
			Remote	UE ID					
								octet q	
				-	-			Octet j*	
	Spare		UPRI4	TPRI4I	Protoco	l used by	remote	octet j+1*	
						UE			
Address information									
								octet j+k*	
HPLMN ID									
								octet	
								(j+k+3)*	
]	

Figure 9.11.4.29.2: Remote UE context

Table 9.11.4.29.2: Remote UE context list information element
Remote UE ID type (bits 1 to 3 of octet 6) Bits 32 1 0 0 1 **UP-PRUK ID** 0 1 0 **CP-PRUK ID** All other values are reserved. Remote UE ID format (bit 4 of octet 6) (NOTE) Bit 4 0 Network access identifier (NAI) 64-bit string 1 Bits 5 to 8 of octet 6 are spare and shall be coded as zero. Remote UE ID (octet 8 to octet j) The UP-PRUK ID or the CP-PRUK ID of the 5G ProSe Remote UE. Protocol used by remote UE (octet j+1, bits 1 to 3) Bits 32 1 No IP info 0 0 0 0 1 IPv4 0 0 1 0 IPv6 Unstructured 1 0 0 Ethernet 0 1 1 All other values are reserved. TCP port range for IPv4 indicator (TPRI4I) (octet j+1, bits 4) Bit 4 TCP port range for IPv4 absent 0 TCP port range for IPv4 present 1 UDP port range for IPv4 indicator (UPRI4I) (octet j+1, bits 5) Bit 5 UDP port range for IPv4 absent 0 UDP port range for IPv4 present 1 Bits 4 to 8 of octet j+1 are spare and shall be coded as zero.

If the Protocol used by remote UE indicates IPv4 and: - TPRI4I bit indicates "TCP port range for IPv4 absent" and UPRI4I bit indicates "UDP port range for IPv4 absent", the Address information in octet j+2 to octet j+5
 contains the IPv4 address. TPRI4I bit indicates "TCP port range for IPv4 present" and UPRI4I bit indicates "UDP port range for IPv4 absent", the Address information in octet j+2 to octet j+9 contains the IPv4 address followed by the TCP port range field. TPRI4I bit indicates "TCP port range for IPv4 absent" and UPRI4I bit indicates "UDP port range for IPv4 present", the Address information in octet j+2 to octet j+9 contains the IPv4 address followed by the UDP port range field. TPRI4I bit indicates "TCP port range for IPv4 present" and UPRI4I bit indicates "UDP port range for IPv4 present", the Address information in octet j+2 to octet j+9 contains the IPv4 address followed by the UDP port range field. TPRI4I bit indicates "TCP port range for IPv4 present" and UPRI4I bit indicates "UDP port range for IPv4 present", the Address information in octet j+2 to octet j+13 contains the IPv4 address followed by the UDP port range field followed by the TCP port range field. See NOTE.
The UDP port range field consists of the lowest UDP port number field followed by the highest UDP port number field, of the UDP port range assigned to the remote UE in the NAT function of 5G ProSe layer-3 UE-to-network relay. The TCP port range field consists of the lowest TCP port number field followed by highest TCP port number field, of the TCP port range assigned to the remote UE in the NAT function of 5G ProSe layer-3 UE-to-network relay. Each port number field is two octets long and bit 8 of first octet of the port number field represents the most significant bit of the port number and bit 1 of second octet of the port number field the least significant bit.
If the Protocol used by remote UE indicates IPv6, the Address information in octet j+2 to octet j+9 contains the /64 IPv6 prefix of a remote UE. Bit 8 of octet j+2 represents the most significant bit of the /64 IPv6 prefix and bit 1 of octet j+9 the least significant bit.
If the Protocol used by remote UE indicates Ethernet, the Address information in octet j+2 to octet j+7 contains the remote UE MAC address. Bit 8 of octet j+2 represents the most significant bit of the MAC address and bit 1 of octet j+7 the least significant bit.
If the Protocol used by remote UE indicates Unstructured, the Address information octets are not included.
If the Protocol used by remote UE indicates No IP info, the Address information octets are not included
If the Remote UE ID type field indicates "UP-PRUK ID" and the Remote UE ID format field indicates "64-bit string", then the HPLMN ID field is present otherwise the HPLMN ID field is absent. The HPLMN ID field indicates HPLMN ID of the 5G ProSe remote UE and is coded as value part of the PLMN ID information element as specified in 3GPP TS 24.554 [19E] subclause 11.3.33 starting with the second octet.
NOTE: In the present release of the specification, providing information for IP

9.11.4.30 Requested MBS container

The purpose of the Requested MBS container information element is for UE to request to join or leave one or more multicast MBS sessions.

The Requested MBS container information element is coded as shown in figure 9.11.4.30.1, figure 9.11.4.30.2, figure 9.11.4.30.3, figure 9.11.4.30.4 and table 9.11.4.30.1.

protocols other than UDP or TCP is not specified

The Requested MBS container is a type 6 information element with a minimum length of 8 octets and a maximum length of 65538 octets.







Figure 9.11.4.30.2: multicast MBS session information



Figure 9.11.4.30.3: multicast MBS session ID for Type of multicast MBS session ID = "Temporary Mobile Group Identity (TMGI)"



Figure 9.11.4.30.4: multicast MBS session ID for Type of multicast MBS session ID = "Source specific IP multicast address for IPv4" or "Source specific IP multicast address for IPv6"

Table 9.11.4.30.1: Requested MBS container information element

9.11.4.31 Received MBS container

The purpose of the Received MBS container information element is to indicate to the UE the information of the multicast MBS sessions that the network accepts or rejects the UE to join, the information of the multicast MBS sessions that the UE is removed from, or the information of the updated MBS service area.

The Received MBS container information element is coded as shown in figure 9.11.4.31.1, figure 9.11.4.31.2, figure 9.11.4.31.3, figure 9.11.4.31.4, figure 9.11.4.31.5, figure 9.11.4.31.6, figure 9.11.4.31.7, figure 9.11.4.31.8, figure 9.11.4.31.9, figure 9.11.4.31.10, figure 9.11.4.31.11 and table 9.11.4.31.1.

The Received MBS container is a type 6 information element with a minimum length of 9 octets and a maximum length of 65538 octets.







Figure 9.11.4.31.2: Received MBS information



Figure 9.11.4.31.3: MBS service area for MBS service area indication = "MBS service area included as MBS TAI list"











Figure 9.11.4.31.6: NR CGI list

	1	2	3	4	5	6	7	8
octet k+1*								
				ell ID	NR C			
octet k+5*								
octet k+6*		digit 1	MCC			digit 2	MCC	
octet k+7*		digit 3	MCC			digit 3	MNC	
octet k+8*		digit 1	MNC			digit 2	MNC	

Figure 9.11.4.31.7: NR CGI



Figure 9.11.4.31.8: MBS timers for MBS timer indication = "MBS start time"



Figure 9.11.4.31.9: MBS timers for MBS timer indication = "MBS back-off timer"



Figure 9.11.4.31.10: MBS security container

8	7	6	5	4	3	2	1	
			0				MTKI	octet i+2*
			spare					
								octet i+3*
			Key do	main ID				
								octet i+5*
								octet i+6*
			MSł	< ID				
								octet i+9*
								octet i+10'
			MS	SK				
								octet i+25'
								octet i+26'
			MTł	< ID				
								octet i+27'
								octet i+28'
			Encrypt	ed MTK				
			-					octet i+43'

Figure 9.11.4.31.11: MBS security keys set

Table 9.11.4.31.1: Received MBS container information element

MBS decision (MD) (bits 1 to 3 of octet 4) The MD indicates the network decision of the join requested by the UE, the network requests to remove the UE from the multicast MBS session or the network request to update the MBS service area or the security information of multicast MBS multicast session. Bits 3 2 1 0 0 1 MBS service area update 0 MBS join is accepted 0 1 0 MBS join is rejected 1 1 1 0 0 Remove UE from multicast MBS session MBS security information update 0 1 All other values are unused in this version of the specification and interpreted as 000 if received. If MD is set to "MBS join is rejected" or "Remove UE from multicast MBS session", bits 6 to 8 of octet 4 shall contain the Rejection cause which indicates the reason of rejecting the MBS join request or the reason of removing the UE from multicast MBS session, respectively, otherwise bits 6 to 8 of octet 4 are spare and shall be coded as zero. MBS service area indication (MSAI) (bits 4 and 5 of octet 4) The MSAI indicates whether and how the MBS service area is included in the IE. Bits 54 0 MBS service area not included \cap 0 1 MBS service area included as MBS TAI list MBS service area included as NR CGI list 1 0 MBS service area included as MBS TAI list and NR CGI list 1 1 Rejection cause (bits 6 to 8 of octet 4) The Rejection cause indicates the reason of rejecting the join request or the reason of removing the UE from the multicast MBS session. Bits 8 7 6 0 0 0 No additional information provided Insufficient resources 0 0 1 0 1 0 User is not authorized to use MBS service 0 1 multicast MBS session has not started or will not start soon 1 0 0 User is outside of local MBS service area 1 Session context not found 0 1 1 multicast MBS session is released 1 0 1 All other values are unused in this version of the specification and interpreted as 000 if received. IP address existence (IPAE) (bit1 of octet 5) The IPAE indicates whether the Source IP address information and Destination IP address information are included in the IE or not. Bit 1 0 Source and destination IP address information not included Source and destination IP address information included 1 If IPAE is set to "Source and destination IP address information included", Source IP address information and Destination IP address information shall be included in the IE, otherwise Source IP address information and Destination IP address information shall not be included in the IE. MBS timer indication (MTI) (bits 2 and 3 of octet 5) The MTI indicates whether there is MBS timer included in the IE or not. Bit 32 0 0 No MBS timers included 0 1 MBS start time included MBS back-off timer included 1 0 All other values are unused in this version of the specification and interpreted as 00 if received

MBS security container indication (MSCI) (bit 4 of octet 5) The MSCI indicates whether the MBS security container is included in the IE or not Bit 4 0 MBS security container not included 1 MBS security container included IP address type (IPAT) (bit 5 of octet 5) The IPAT indicates the type of the source IP address information and destination IP address information. This field is ignored when IPAE is set to "Source and destination IP address information not included". Bit 5 0 Source IP address information and destination IP address information are IPv4 Source IP address information and destination IP address information are IPv6 1 Bits 6 to 8 of octet 5 are spare and shall be coded as zero. TMGI (octets 6 to i) The TMGI is coded as described in subclause 10.5.6.13 in 3GPP TS 24.008 [12] starting from octet 2. The structure of the TMGI is defined in 3GPP TS 23.003 [4]. Source IP address information (octet j+1 to v) This field contains the IP unicast address used as source address in IP packets for identifying the source of the multicast service. The value of this field is copied from the corresponding source IP address information in the requested MBS container. If the IPAT indicates "Source and destination IP address information are IPv4", the Source IP address information in octet j+1 to octet j+4 contains an IPv4 address. If the IPAT indicates "Source and destination IP address information are IPv6", the Source IP address information in octet j+1 to octet j+16 contains an IPv6 address. Destination IP address information (octet v+1 to k) This field contains the IP multicast address used as destination address in related IP packets for identifying a multicast service associated with the source. The value of this field is copied from the corresponding destination IP address information in the requested MBS container. If the IPAT indicates "Source and destination IP address information are IPv4", the Destination IP address information in octet v+1 to octet v+4 contains an IPv4 address. If the IPAT indicates "Source and destination IP address information are IPv6", the Destination IP address information in octet v+1 to octet v+16 contains an IPv6 address. MBS service area (octet k+1 to s) The MBS service area contains the MBS TAI list, the NR CGI list or both, that identifies the service area(s) for a local MBS service. MBS TAI list (octet k+1 to s) The MBS TAI list is coded as octet 2 and above of the 5GS tracking area identity list IE defined in subclause 9.11.3.9. NR CGI (octet k+2 to k+9) The NR CGI globally identifies an NR cell. It contains the NR Cell ID and the PLMN ID of that cell. NR Cell ID (octet k+2 to k+6) The NR Cell ID consists of 36 bits identifying an NR Cell ID as specified in subclause 9.3.1.7 of 3GPP TS 38.413 [31], in hexadecimal representation. Bit 8 of octet y+1 is the most significant bit and bit 5 of octet y+5 is the least significant bit. Bits 1 to 4 of octet y+5 are spare and shall be coded as zero. MCC, Mobile country code (octet k+6 and bits 1 to 4 octet k+7) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code (bits 5 to 8 of octet k+7 and octet k+8) The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, bits 5 to 8 of octet k+7 shall be coded as "1111".

The MCC and MNC digits are coded as octets 6 to 8 of the Temporary mobile group identity IE in figure 10.5.154 of 3GPP TS 24.008 [12]. MBS start time (octets s+1 to s+6) The MBS start time is coded as described in subclause 10.5.3.9 in 3GPP TS 24.008 [12] starting from octet 2 till octet 7. MBS back-off timer (octet s+1) The MBS back-off timer is coded as octet 3 described in subclause 10.5.7.4a in 3GPP TS 24.008 [12]. MTK indication (MTKI) (bit1 of octet i+2) The MTKI indicates whether the MTK ID and Encrypted MTK are included in the MBS security keys set or not. Bit 1 0 MTK ID and Encrypted MTK not included MTK ID and Encrypted MTK included 1 Bits 2 to 8 of octet i+2 are spare and shall be coded as zero

Key domain ID (octet i+3 to i+5) The key domain ID is 3 bytes long and is defined in 3GPP TS 33.246 [57].

MBS Service Key Identifier (MSK ID) (octets i+6 to i+9) The MSK ID is 4 bytes long and is defined in 3GPP TS 33.246 [57].

MBS Service Key (MSK) (octets i+10 to i+25) The MSK is 16 bytes long and is defined in 3GPP TS 33.246 [57].

MBS Traffic Key Identifier (MTK ID) (octets i+26 to i+27) The MTK ID is 2 bytes long and is defined in 3GPP TS 33.246 [57].

Encrypted MBS Traffic Key (Encrypted MTK) (octets i+28 to i+43) The Encrypted MTK is 16 bytes long and contains the encrypted version of MTK using MSK as defined in 3GPP TS 33.246 [57].

NOTE: The IPAE bit is not expected to be set to "Source and destination IP address information included" when the MBS decision (MD) indicates "Remove UE from multicast MBS session".

9.11.4.32 PDU session pair ID

The purpose of the PDU session pair ID information element is to indicate a PDU session pair ID.

The PDU session pair ID information element is coded as shown in figure 9.11.4.32.1 and table 9.11.4.32.1.

The PDU session pair ID is a type 4 information element with a length of 3 octets.

8	7	6	5	4	3	2	1	_	
PDU session pair ID IEI o									
Length of PDU session pair ID IE									
PDU session pair ID									
Figure 0.44.4.22.4. DDU accessor pair ID information along									

Figure 9.11.4.32.1: PDU session pair ID information element

Table 9.11.4.32.1: PDU session	pair ID	information	element
--------------------------------	---------	-------------	---------

2DU session pair ID (octet 3)										
All other values are reserved.										

9.11.4.33 RSN

The purpose of the RSN information element is to indicate an RSN.

The RSN information element is coded as shown in figure 9.11.4.33.1 and table 9.11.4.33.1.

The RSN is a type 4 information element with a length of 3 octets.



Figure 9.11.4.33.1: RSN information element

RSN (octet 3) Bits 8 7 6 5 4 3 2 1 0 0 0 0 0 0 0 0 v1 0 0 0 0 0 0 0 1 v2 All other values are spare and shall not be used by a UE compliant to the present version of this specification.

9.11.4.34 ECS address

The purpose of the ECS address information element is to indicate the ECS address (either IPv4 address, IPv6 address, or FQDN) and the associated spatial validity condition.

The ECS address information element is coded as shown in figure 9.11.4.34-1, figure 9.11.4.34-2, table 9.11.4.34-1, and table 9.11.4.34-2.

The ECS address information element is a type 6 information element with minimum length of 8 octets and a maximum length of 65538 octets.

8 7 6 5 4 3 2 1 ECS address IEI octet 1 octet 2 Length of ECS address contents octet 3 Type of ECS address Type of spatial validity condition octet 4 octet 5 ECS address octet a octet (a+1)* Spatial validity condition contents

octet n* Figure 9.11.4.34-1: ECS address information element



Figure 9.11.4.34-2: Spatial validity condition contents

Table 9.11.4.34-1: ECS address information element

Type of ECS address (octet 4, bit 1 to 4) Bits 4 3 1 2 0 0 0 0 IPv4 0 0 0 1 IPv6 0 1 0 FQDN 0 1 1 1 1 Unspecified All other values are spare. The receiving entity shall ignore an ECS address IE with type of ECS address containing a spare value. Type of spatial validity condition (octet 4, bit 5 to 8) Bits 8 7 6 5 0 0 0 0 No spatial validity condition 0 Geographical service area 0 0 1 Tracking area 0 0 1 0 0 0 1 1 Country-wide All other values are spare. The receiving entity shall ignore a spatial validity condition with type of spatial validity condition containing an unknown value. If the type of ECS address indicates IPv4, then the ECS address field contains an IPv4 address in octet 5 to octet 8. If the type of ECS address indicates IPv6, then the ECS address field contains an IPv6 address in octet 5 to octet 20 and is encoded according to IETF RFC 4291 [RFC4291]. If the type of ECS address indicates FQDN, then the ECS address field contains in octet 5 the length of FQDN value and in octet 6 to octet a an FQDN value encoded as defined in subclause 19.4.2 in 3GPP TS 23.003 [4]. If the type of ECS address indicates unspecified, then the remaining fields of ECS address information element shall be passed to the upper layers. Spatial validity condition contents (octet (a+1)* to n*) The spatial validity condition contents contain a variable number of spatial validity condition information.

Table 9.11.4.34-2: Spatial validity condition contents

If the type of spatial validity condition of the ECS address indicates No spatial validity condition, then the spatial validity condition information field is empty.

If the type of spatial validity condition of the ECS address indicates geographical service area, then the spatial validity condition information field contains a geographical service area which is specified by geographical descriptions as defined in 3GPP TS 23.032 [4B].

If the type of spatial validity condition of the ECS address indicates tracking area, then the spatial validity condition information field contains a TAI as defined in subclause 9.11.3.8 starting from octet 2.

If the type of spatial validity condition of the ECS address indicates country-wide, then the spatial validity condition information field contains an MCC as defined in in ITU-T Recommendation E.212 [42], annex A. The first MCC digit is coded in bit 1 to 4 of the octet b, the second MCC digit is coded in bit 5 to 8 of the octet b, and the third MCC digit is coded in bit 1 to 4 of the octet b+1. Bit 5 to bit 8 of the octet b+1 shall be padded with 1. If only two digits are used for for MCC, octet b+1 shall be padded with 1.

9.11.4.35 Void

9.12 3GPP specific coding information defined within present document

9.12.1 Serving network name (SNN)

The serving network name (SNN) is used:

- in the Network name field of the AT_KDF_INPUT attribute defined in IETF RFC 5448 [40];
- in K_{AUSF} derivation function as specified in 3GPP TS 33.501 [24] annex A; and
- in RES* and XRES* derivation function as specified in 3GPP TS 33.501 [24] annex A.

SNN shall contain a UTF-8 string without terminating null characters.

SNN is of maximum length of 1020 octets.

SNN consists of SNN-service-code and SNN-network-identifier, delimited by a colon.

SNN-network-identifier identifies the serving PLMN or the serving SNPN.

MCC and MNC in the SNN-PLMN-ID are MCC and MNC of the serving PLMN. If the MNC of the serving PLMN has two digits, then a zero is added at the beginning.

MCC and MNC in the SNN-SNPN-ID are MCC and MNC of the serving SNPN. If the MNC of the serving SNPN has two digits, then a zero is added at the beginning.

SNN-NID contains an NID in hexadecimal digits.

ABNF syntax of SNN is specified in table 9.12.1.1

Table 9.12.1.1: ABNF syntax of SNN

```
SNN = SNN-service-code ":" SNN-network-identifier
SNN-service-code = %x35.47 ; "5G"
SNN-network-identifier = SNN-PLMN-ID / SNN-SNPN-ID
```

SNN-PLMN-ID = SNN-mnc-string SNN-mnc-digits "." SNN-mcc-string SNN-mcc-digits "." SNN-3gppnetworkstring "." SNN-org-string ; applicable when not operating in SNPN access operation mode. SNN-SNPN-ID = SNN-mnc-string SNN-mnc-digits "." SNN-mcc-string SNN-mcc-digits "." SNN-3gppnetworkstring "." SNN-org-string ":" SNN-NID ; applicable when operating in SNPN access operation mode. SNN-mnc-digits = DIGIT DIGIT DIGIT ; MNC of the PLMN ID SNN-mcc-digits = DIGIT DIGIT DIGIT ; MCC of the PLMN ID SNN-mcc-string = %x6d.6e.63 ; "mnc" in lower case SNN-mcc-string = %x6d.63.63 ; "mcc" in lower case SNN-3gppnetwork-string = %x33.67.70.70.6e.65.74.77.6f.72.6b ; "3gppnetwork" in lower case SNN-org-string = %x6f.72.67 ; "org" in lower case SNN-NID = llSNN-hexadecimal-digit ; NID in hexadecimal digits SNN-hexadecimal-digit = DIGIT / %x41 / %x42 / %x43 / %x44 / %x45 / %x46

- NOTE: SNN-service-code allows for distinguishing of ANID specified in 3GPP TS 24.302 [16] and SNN as either of SNN or ANID can be carried in the AT_KDF_INPUT attribute.
- EXAMPLE 1: In case of a PLMN, if PLMN ID contains MCC = 234 and MNC = 15, SNN is 5G:mnc015.mcc234.3gppnetwork.org.
- EXAMPLE 2: In case of an SNPN, if SNPN ID contains a PLMN ID of MCC = 234 and MNC = 15 and an NID of 123456ABCDEH, SNN is 5G:mnc015.mcc234.3gppnetwork.org:123456ABCDE.

10 List of system parameters

10.1 General

The description of timers in the following tables should be considered a brief summary. The precise details are found in clauses 4 to 6, which should be considered the definitive descriptions.

10.2 Timers of 5GS mobility management

Timers of 5GS mobility management are shown in table 10.2.1 and table 10.2.2.

NOTE: Timers T3324, T3346, T3245 and T3247 are defined in 3GPP TS 24.008 [12]. Timers T3444, T3445, T3447 and T3448 are defined in 3GPP TS 24.301 [15].

Table 10.2.1: Timers of 5GS mobility management – UE side

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
T3502	Default 12 min. NOTE 1	5GMM- DEREGISTERED 5GMM- REGISTERED	At registration failure and the attempt counter is equal to 5	Transmission of REGISTRATION REQUEST message	Initiation of the registration procedure, if still required
T3510	15s NOTE 7 NOTE 8 In WB- N1/CE mode, 85s For access via a satellite NG-RAN cell, 27s NOTE 12	5GMM- REGISTERED- INITIATED	Transmission of REGISTRATION REQUEST message	REGISTRATION ACCEPT message received or REGISTRATION REJECT message received	Start T3511 or T3502 as specified in subclause 5.5.1.2.7 if T3510 expired during registration procedure for initial registration. Start T3511 or T3502 as specified in subclause 5.5.1.3.7 if T3510 expired during the registration procedure for mobility and periodic registration update
T3511	10s	5GMM- DEREGISTERED. ATTEMPTING- REGISTRATION 5GMM- REGISTRATION- UPDATE 5GMM- REGISTERED.NO RMAL-SERVICE or 5GMM- REGISTERED.NO N-ALLOWED- SERVICE	At registration failure due to lower layer failure, T3510 timeout or registration rejected with other 5GMM cause values than those treated in subclause 5.5.1.2.5 for initial registration or subclause 5.5.1.3.5 for mobility and periodic registration	Transmission of REGISTRATION REQUEST message 5GMM- CONNECTED mode entered (NOTE 5)	Retransmission of the REGISTRATION REQUEST message, if still required
T3512	Default 54 min NOTE 1 NOTE 2	5GMM- REGISTERED	In 5GMM-REGISTERED, when 5GMM-CONNECTED mode is left and if the NW does not indicate support for strictly periodic registration timer as specified in subclause 5.3.7. If the network indicates support for strictly periodic registration timer, T3512 is started after the successful completion of registration update procedure. T3512 is restarted if it expires in 5GMM-CONNECTED mode as specified in subclause 5.3.7.	When entering state 5GMM- DEREGISTERED When entering 5GMM- CONNECTED mode if the NW does not indicate support for strictly periodic registration timer as specified in subclause 5.3.7.	In 5GMM-IDLE mode, Initiation of the periodic registration procedure if the UE is not registered for emergency services. In 5GMM- CONNECTED mode, restart the timer T3512. Locally deregister if the UE is registered for emergency services

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
T3516	30s NOTE 7 NOTE 8 In WB- N1/CE mode, 48s For access via a satellite NG-RAN cell, 35s NOTE 12	5GMM- REGISTERED- INITIATED 5GMM- REGISTERED 5GMM- DEREGISTERED- INITIATED 5GMM-SERVICE- REQUEST- INITIATED	RAND and RES* stored as a result of an 5G authentication challenge	SECURITY MODE COMMAND message received SERVICE REJECT message received REGISTRATION ACCEPT message received AUTHENTICATION REJECT message received AUTHENTICATION FAILURE message sent 5GMM- DEREGISTERED, 5GMM-NULL or 5GMM-IDLE mode entered	Delete the stored RAND and RES*
T3517	(a) 5s for case h) in subclause 5.6.1.1; or (b) 15s for cases other than h) in subclause 5.6.1.1 NOTE 7 NOTE 7 NOTE 8 NOTE 10 In WB- N1/CE mode, 61s For access via a satellite NG-RAN cell, 27s NOTE 12	5GMM-SERVICE- REQUEST- INITIATED	Transmission of SERVICE REQUEST message, or CONTROL PLANE SERVICE REQUEST message	 (a) Indication from the lower layers that the UE has changed to S1 mode or E- UTRA connected to 5GCN for case h) in subclause 5.6.1.1; or (b) SERVICE ACCEPT message received, or SERVICE REJECT message received for cases other than h) in subclause 5.6.1.1 see subclause 5.6.1.4.2 	Abort the procedure
T3519	60s NOTE 7 NOTE 8 In WB- N1/CE mode, 90s For access via a satellite NG-RAN cell, 65s	5GMM- REGISTERED- INITIATED 5GMM- REGISTERED 5GMM- DEREGISTERED- INITIATED 5GMM-SERVICE- REQUEST- INITIATED (NOTE 6)	Transmission of IDENTITY RESPONSE message, REGISTRATION REQUEST message, or DEREGISTRATION REQUEST message with freshly generated SUCI	REGISTRATION ACCEPT message with new 5G-GUTI received CONFIGURATION UPDATE COMMAND message with new 5G-GUTI received DEREGISTRATION ACCEPT message	Delete stored SUCI

TIMER	TIMER	STATE	CAUSE OF START	NORMAL STOP	ON
NUM.	VALUE				EXPIRY
TIMER NUM. T3520	TIMER VALUE 15s NOTE 7 NOTE 8 In WB- N1/CE mode, 33s For access via a satellite NG-RAN cell, 20s NOTE 12	STATE 5GMM- REGISTERED- INITIATED 5GMM- DEREGISTERED- INITIATED 5GMM-SERVICE- REQUEST- INITIATED	CAUSE OF START Transmission of AUTHENTICATION FAILURE message with any of the 5GMM cause #20, #21, #26 or #71 Transmission of AUTHENTICATION RESPONSE message with an EAP-response message after detection of an error as described in subclause 5.4.1.2.2.4	NORMAL STOP AUTHENTICATION REQUEST message received or AUTHENTICATION REJECT message received or SECURITY MODE COMMAND message received when entering 5GMM-IDLE mode indication of transmission failure of AUTHENTICATION FAILURE message from lower layers	ON EXPIRY On first expiry during a 5G AKA based primary authentication and key agreement procedure, the UE should consider the network as false and follow item g of subclause 5.4.1.3.7, if the UE is not registered for emergency services. On first expiry during a 5G AKA based primary authentication and key agreement procedure, the UE will follow subclause 5.4.1.3.7 under "For items c, d, e and f:", if the UE is registered for emergency services. On first expiry during an EAP based primary authentication and key agreement procedure, the UE will follow services. On first expiry during an EAP based primary authentication and key agreement procedure, the UE should consider the network as false and follow item e of subclause 5.4.1.2.4. 5, if the UE is not registered for
					and follow item e of subclause 5.4.1.2.4. 5, if the UE is not registered for emergency services.
					On first expiry during an EAP based primary authentication and key agreement procedure, the UE will follow subclause 5.4.1.2.4. 5 under "For item e:", if the UE is registered for emergency services

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
T3521	15s NOTE 7 NOTE 8 In WB- N1/CE mode, 45s For access via a satellite NG-RAN cell, 27s NOTE 12	5GMM- DEREGISTERED- INITIATED	Transmission of DEREGISTRATION REQUEST message when de-registration procedure is not due to a "switch off"	DEREGISTRATION ACCEPT message received	Retransmission of DEREGISTRATION REQUEST message
T3525	Default 60s NOTE 3 NOTE 7 NOTE 8 In WB- N1/CE mode, default 120s For access via a satellite NG-RAN cell, default 72s NOTE 12	5GMM- REGISTERED.NO RMAL-SERVICE or 5GMM- REGISTERED.NO N-ALLOWED- SERVICE	T3517 expires and service request attempt counter is greater than or equal to 5	When entering state other than 5GMM- REGISTERED.NOR MAL-SERVICE state or 5GMM- REGISTERED.NON -ALLOWED- SERVICE, or UE camped on a new PLMN other than the PLMN on which timer started, or User-plane resources established with the network	The UE may initiate service request procedure
T3540		5GMM- DEREGISTERED 5GMM- REGISTERED	REGISTRATION REJECT message or DEREGISTRATION REQUEST message received with any of the 5GMM cause #3, #6, #7, #11, #12, #13, #15, #27, #31, #36, #62, #72, #73, #74, #75 or #76 SERVICE REJECT message received with any of the 5GMM cause #3, #6, #7, #11, #12, #13, #15, #27, #36, #72, #73, #74, #75 or #76. REGISTRATION ACCEPT message received as described in subclause 5.3.1.3 case b) and case h) SERVICE ACCEPT message received as described in subclause 5.3.1.3 case f) AUTHENTICATION REJECT message received DEREGISTRATION ACCEPT message received d DEREGISTRATION ACCEPT message received as described in subclause 5.3.1.3 case k)	N1 NAS signalling connection released PDU sessions have been set up except for the case the UE has set Request type to "NAS signalling connection release" in the UE request type IE in the REGISTRATION REQUEST message as described in subclause 5.3.1.3 case b) Other use cases see subclause 5.3.1.3	Release the NAS signalling connection for the cases a), b), f) and g) as described in subclause 5.3.1.3

TIMER	TIMER	STATE	CAUSE OF START	NORMAL STOP	ON
NUM.	VALUE				EXPIRY
	10s	5GMM-		N1 NAS signalling	Release the NAS
	NOTE /	REGISTERED	COMMAND message received	connection released	signalling
	(applicable		as described in	Other use cases	connection for the
	to case t) in		subclause 5.3.1.3 case e) and h)	See	case e) and perform
	subclause		SERVICE ACCEPT message	subclause 5.3.1.3	a new registration
	5.3.1.3)		received as described in		procedure as
			Subclause 5.5.1.5 case I)		
					Subciause 5.5.1.5.2
	mode 3/s				Release the NAS
	(applicable				signalling
	to case f) in				connection for the
	subclause				case h) and i) as
	5.3.1.3)				described in
	NOTE 11				subclause 5.3.1.3
	For access	5GMM-	REGISTRATION REJECT		Release the NAS
	via a	DEREGISTERED	message received with the		signalling
	satellite		5GMM cause #9 or #10		connection for the
	NG-RAN	5GMM-	SERVICE REJECT message		cases c) and d) as
	cell, default	DEREGISTERED.	received with the 5GMM cause		described in
	22s	NORMAL-	#9, #10 or #28		subclause 5.3.1.3
	(applicable	SERVICE			and initiation of the
	to case f) in				registration
	subclause	5GMM-			procedure as
	5.3.1.3)	REGISTERED.NO			specified in
	NOTE 12	N-ALLOWED-			subclause 5.5.1.2.2
	Defeult 54	SERVICE	Entering COMM IDLE mode even		OF 5.5.1.3.2
Non-SGPP	Delault 54	All SGIVIIVI State	Entering SGMM-IDLE mode over	NT NAS signalling	
			HOH-SOFF ACCESS	non-3GPP access	3GPP access on 1st
timer	NOTE 2	5GMM-		established or when	expiry
unior	NOTE 4	DEREGISTERED		entering state	схриу
		over non-3GPP		5GMM-	
		access		DEREGISTERED	
				over non-3GPP	
				access	
T3526	NOTE 9	5GMM-	Rejected S-NSSAI with rejection	The associated	Remove the S-
		DEREGISTERED	cause "maximum number of UEs	rejected S-NSSAI	NSSAI in the
		5GMM-	per network slice reached"	for the maximum	rejected NSSAI for
		REGISTERED	received.	number of UEs	the maximum
				reached as	number of UEs
				specified in	reached associated
				subclause 4.6.2.2	with the 13526
				deleted.	timer.
13527	155	5GMM-			Retransmission of
		REGISTERED.NO	REQUEST message	REJECT message	
		RIVIAL-SERVICE	Transmission of PELAV		
					messaye
			message	REQUEST	
				message received	
				or	
				RELAY KEY	
				ACCEPT message	
				received	

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
NOTE 1:	The value of t	his timer is provided	by the network operator during th	e registration proced	ure.
NOTE 2:	The default va	alue of this timer is u	used if the network does not indica	te a value in the REG	SISTRATION
	ACCEPT mes	sage and the UE do	pes not have a stored value for this	s timer.	
NOTE 3:	The value of t	his timer is UE impl	ementation specific, with a minimu	m value of 60 second	ls if not in NB-N1
	mode and if n	ot in WB-N1/CE mo	de.		
NOTE 4:	If the T3346 v	alue received in the	mobility management messages	is greater than the va	lue of the non-3GPP
	value of timer	T3346.	the non-3GPP de-registration time	er value to be 4 minut	es greater than the
NOTE 5:	The condition	s for which this appl	ies are described in subclause 5.5	5.1.3.7.	
NOTE 6:	The condition	s for which this appl	ies to the 5GMM-SERVICE-REQU	JEST-INITIATED stat	e are described in
	subclause 5.4	.1.3.7 case c) and c	ase d).		
NOTE 7:	In NB-N1 mod	le, the timer value s	hall be calculated as described in	subclause 4.17.	
NOTE 8:	In WB-N1 mo	de, if the UE suppor	ts CE mode B and operates in eith	her CE mode A or CE	mode B, then the
	timer value is	as described in this	table for the case of WB-N1/CE m	node (see subclause	4.19).
NOTE 9:	The value of t	his timer is provided	by the network operator during the	e registration proced	ure or the generic
	OE configurat	ion update procedu	re along with the rejected S-NSSA	i with rejection cause	ific with a minimum
	value of 12 m	inutes and used if the	. The default value of this time is		
	message the	REGISTRATION R	E IECT message or the CONFIGI		
	along with the	rejected S-NSSAL	with rejection cause "maximum nu	mber of UEs per netv	vork slice reached".
NOTE 10:	Based on imp	lementation, the tim	er may be set to a value between	250ms and 15s wher	the MUSIM UE
	indicates "NA	S signalling connect	tion release" in the UE request typ	e IE of the SERVICE	REQUEST message
	or CONTROL	PLĂNE SERVICE	REQUEST message.		0
NOTE 11:	Based on imp	lementation, the tim	er may be set to a value between	250ms and 10s wher	the MUSIM UE not
	in NB-N1 mod	le or WB-N1 mode i	ndicated "NAS signalling connecti-	on release" or "Rejec	tion of paging" in the
	UE request ty	pe IE of the SERVI	CE REQUEST message or CONT	ROL PLANE SERVIC	E REQUEST
	message; or i	ndicated "NAS signation	alling connection release" in the U	E request type IE of t	he REGISTRATION
	REQUEST m	essage.			
NOTE 12:	In satellite NG	-RAN access, this v	alue shall be selected when satel	lite NG-RAN RAT typ	e is NR(MEO) or
	NK(GEU).				

Table 10.2.2: Timers of 5GS mobility management – AMF side

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
T3513 NOTE 7 NOTE 9	NOTE 4	5GMM- REGISTERED	Paging procedure initiated	Paging procedure completed as specified in subclause 5.6.2.2.1	Network dependent
T3522 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 11s NOTE 12	5GMM- DEREGISTERED- INITIATED	Transmission of DEREGISTRATION REQUEST message	DEREGISTRATION ACCEPT message received	Retransmission of DEREGISTRATION REQUEST message
T3550 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 18s For access via a satellite NG-RAN cell, 11s NOTE 12	5GMM-COMMON- PROCEDURE- INITIATED	Transmission of REGISTRATION ACCEPT message as specified in subclause 5.5.1.2.4 and 5.5.1.3.4	REGISTRATION COMPLETE message received	Retransmission of REGISTRATION ACCEPT message
T3555 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 11s NOTE 12	5GMM- REGISTERED	Transmission of CONFIGURATION UPDATE COMMAND message with "acknowledgement requested" set in the Acknowledgement bit of the Configuration update indication IE	CONFIGURATION UPDATE COMPLETE message received	Retransmission of CONFIGURATION UPDATE COMMAND message
T3560 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 11s	5GMM-COMMON- PROCEDURE- INITIATED	Transmission of AUTHENTICATION REQUEST message Transmission of SECURITY MODE COMMAND message	AUTHENTICATION RESPONSE message received AUTHENTICATION FAILURE message received SECURITY MODE COMPLETE message received SECURITY MODE REJECT message received	Retransmission of AUTHENTICATION REQUEST message or SECURITY MODE COMMAND message

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
T3565 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 11s NOTE 12	5GMM- REGISTERED	Transmission of NOTIFICATION message	SERVICE REQUEST message received CONTROL PLANE SERVICE REQUEST message received NOTIFICATION RESPONSE message received REGISTRATION REQUEST Message received DEREGISTRATION REQUEST message received NGAP UE context resume request message as specified in 3GPP TS 38.413 [3 1] received	Retransmission of NOTIFICATION message
T3570 NOTE 6 NOTE 8	6s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 11s NOTE 12	5GMM-COMMON- PROCEDURE- INITIATED	Transmission of IDENTITY REQUEST message	IDENTITY RESPONSE message received	Retransmission of IDENTITY REQUEST message
T3575 NOTE 6 NOTE 8	15s In WB- N1/CE mode, 60s For access via a satellite NG-RAN cell, 27s NOTE 12	5GMM- REGISTERED	Transmission of NETWORK SLICE-SPECIFIC AUTHENTICATION COMMAND message	NETWORK SLICE- SPECIFIC AUTHENTICATION COMPLETE message received	Retransmission of NETWORK SLICE- SPECIFIC AUTHENTICATION COMMAND message
Active timer	NOTE 10	All except 5GMM- DEREGISTERED	Entering 5GMM-IDLE mode after indicating MICO mode activation to the UE with an active timer value.	N1 NAS signalling connection established	Activate MICO mode for the UE.
Implicit de- registration timer	NOTE 2	All except 5GMM- DEREGISTERED	The mobile reachable timer expires while the network is in 5GMM-IDLE mode Entering 5GMM-IDLE mode over 3GPP access if the MICO mode is activated and strictly periodic monitoring timer is not running The strictly periodic monitoring timer expires while the network is in 5GMM-IDLE mode	N1 NAS signalling connection established	Implicitly de-register the UE on 1 st expiry

TIMER NUM.	TIMER VALUE	STATE	CAUSE OF START	NORMAL STOP	ON EXPIRY
Mobile reachable timer	NOTE 1	All except 5GMM- DEREGISTERED	Entering 5GMM-IDLE mode	N1 NAS signalling connection established	Network dependent, but typically paging is halted on 1 st expiry, and start implicit de- registration timer, if the UE is not registered for emergency services. Implicitly de-register the UE which is registered for emergency services
Non-3GPP implicit de- registration timer	NOTE 3	All except 5GMM- DEREGISTERED	Entering 5GMM-IDLE mode over non-3GPP access	N1 NAS signalling connection over non-3GPP access established	Implicitly de-register the UE for non- 3GPP access on 1 ^s expiry
Strictly periodic monitoring timer	NOTE 5	All except 5GMM- DEREGISTERED	At the successful completion of registration update procedure if strictly periodic registration timer indication is supported as specified in subclause 5.3.7.	Entering 5GMM- DEREGISTERED.	In 5GMM-IDLE mode, start implicit de-registration timer as specified in subclause 5.3.7. In 5GMM- CONNECTED mode, Strictly periodic monitoring timer is started again as specified in subclause 5.3.7.
Implement ation specific timer for onboarding services	NOTE 11	5GMM- REGISTERED	At the successful completion of initial registration for onboarding services in SNPN or initial registration for the UE which the subscription is only for configuration of SNPN subscription parameters in PLMN via the user plane or successful completion of registration procedure for mobility and periodic registration update if the implementation specific timer for onboarding services is not running and: - the UE is registered for onboarding services in SNPN; or - the UE's subscription only allows for configuration of SNPN subscription parameters in PLMN via the user plane.	DEREGISTRATION REQUEST message received.	Network-initiated de-registration procedure performed

		STATE	CAUSE OF START	NORMAL STOP	
NOTE 1:	The default va	alue of this timer is 4	f minutes greater than the value of	timer T3512. If the U	E is registered for
-	emergency se	rvices, the value of	this timer is set equal to the value	of timer T3512. If the	T3346 value
	provided in the	e mobility managen	nent messages is greater than the	value of the timer T35	512, the AMF sets
	the mobile rea	achable timer and th	e implicit de-registration timer suc	h that the sum of the	timer values is
	greater than th	he value of timer T3	346.		
NOTE 2:	The value of t	his timer is network	dependent. If MICO is activated, t	he default value of thi	s timer is 4 minutes
	greater than th	he value of timer T3	512.		
NOTE 3:	The value of t	his timer is network	dependent. The default value of th	his timer is 4 minutes	greater than the
	non-3GPP de	-registration timer. I	t the 13346 value provided in the r	nobility management	messages is greater
	than the value	of the non-3GPP of	le-registration timer, the AMF sets	the non-3GPP implic	it de-registration
NOTE 4	timer value to	be 8 minutes great	er than the value of timer 13346.		
	The value of t	his timer is network	aependent.		
NOTE 5.	In NR N1 more	his timer is the sam	e as the value of timer 13512.	subclause 4 17	
NOTE 7	In NB-N1 mod	the timer value s	shall be calculated as described in	Subciause 4.17. S timer value which is	network dependent
NOTE 8	In WB-N1 mod	de if the LIF sunnor	rts CE mode B and operates in eith	er CE mode A or CE	mode R then the
10120.	timer value is	as described in this	table for the case of WB-N1/CF m	node (see subclause 4	4.19).
NOTE 9:	In WB-N1 mo	de. if the UE suppo	rts CE mode B. then the timer valu	e shall be calculated	by using an NAS
	timer value wh	nich value is networ	k dependent.		
NOTE 10:	If the AMF inc	ludes timer T3324 i	n the REGISTRATION ACCEPT n	nessage and if the UE	is not registered for
	emergency se	rvices, the value of	this timer is equal to the value of t	imer T3324.	-
NOTE 11:	The value of t	his timer needs to b	e large enough to allow a UE to co	omplete the configura	tion of one or more
	entries of the	"list of subscriber da	ata" and considering that configura	tion of SNPN subscri	ption parameters in
	PLMN via the	user plane or onbo	arding services in SNPN involves t	third party entities out	side of the operator's
	network.				
NOTE 12:	In satellite NG NR(GEO).	-RAN access, this	value shall be selected when satel	lite NG-RAN RAT type	e is NR(MEO) or

10.3 Timers of 5GS session management

Timers of 5GS session management are shown in table 10.3.1 and table 10.3.2.

NOTE: Timer T3396 is defined in 3GPP TS 24.008 [12].

Table 10.3.1: Timers of 5GS session management – UE side

TIMER	TIMER	STATE	CAUSE OF START	NORMAL STOP	ON
NUM.	VALUE				1 st , 2 nd , 3 rd , 4 th EXPIRY (NOTE 1)
T3580 NOTE 4 NOTE 5	16s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 21s NOTE 7	PDU SESSION ACTIVE PENDING	Transmission of PDU SESSION ESTABLISHMENT REQUEST message	PDU SESSION ESTABLISHMENT ACCEPT message received or PDU SESSION ESTABLISHMENT REJECT message received or PDU SESSION ESTABLISHMENT REQUEST message received in a DL NAS TRANSPORT message with 5GMM cause #22, #28, #65. #67, #69, #90, #91 or #92	Retransmission of PDU SESSION ESTABLISHMENT REQUEST message
T3581 NOTE 4 NOTE 5	16s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 21s NOTE 7	PDU SESSION MODIFICATION PENDING	Transmission of PDU SESSION MODIFICATION REQUEST message	PDU SESSION MODIFICATION COMMAND message with the same PTI is received or PDU SESSION MODIFICATION REJECT message received or PDU SESSION MODIFICATION REQUEST message received in a DL NAS TRANSPORT message with 5GMM cause #22, #28. #67, #69, or #90	Retransmission of PDU SESSION MODIFICATION REQUEST message
T3582 NOTE 4 NOTE 5	16s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 21s NOTE 7	PDU SESSION INACTIVE PENDING	Transmission of PDU SESSION RELEASE REQUEST message	PDU SESSION RELEASE COMMAND message with the same PTI is received or PDU SESSION RELEASE REJECT message received	Retransmission of PDU SESSION RELEASE REQUEST message
T3583	Default 1 min. NOTE 2	PDU SESSION ACTIVE	UE creates or updates a derived QoS rule	UE deletes the derived QoS rule (see subclause 6.2.5.1.4. 5)	On 1 st expiry: Deletion of the derived QoS rule

TIMER	TIMER	STATE	CAUSE OF START	NORMAL STOP	ON
NUM.	VALUE				THE
					FXPIRY (NOTE 1)
T3584	NOTE 3	PDU SESSION ACTIVE PENDING PDU SESSION MODIFICATION PENDING PDU SESSION ACTIVE or PDU SESSION INACTIVE PENDING	PDU SESSION ESTABLISHMENT REJECT, PDU SESSION MODIFICATION REJECT, or PDU SESSION RELEASE COMMAND received with 5GSM cause #67 and with a timer value for T3584 PDU SESSION ESTABLISHMENT REQUEST, or PDU SESSION MODIFICATION REQUEST received in a DL NAS TRANSPORT message with 5GMM cause #67 and with a timer value for T3584 (see subclause 5.4.5.3.3)	PDU SESSION RELEASE COMMAND message (see NOTE 6) or PDU SESSION MODIFICATION COMMAND message or PDU SESSION AUTHENTICATION COMMAND message or DEREGISTRATION REQUEST message with the re-registration type	EXPIRY (NOTE 1) None
				"re-registration	
T3585	NOTE 3	PDU SESSION ACTIVE PENDING PDU SESSION MODIFICATION PENDING PDU SESSION ACTIVE or PDU SESSION INACTIVE PENDING	PDU SESSION ESTABLISHMENT REJECT, PDU SESSION MODIFICATION REJECT, or PDU SESSION RELEASE COMMAND received with 5GSM cause #69 and with a timer value for T3585 PDU SESSION ESTABLISHMENT REQUEST, or PDU SESSION MODIFICATION REQUEST received in a DL NAS TRANSPORT message with 5GMM cause #69 and with a timer value for T3585(see subclause 5.4.5.3.3)	Tequired DU SESSION RELEASE COMMAND message (see NOTE 6) or PDU SESSION MODIFICATION COMMAND message or PDU SESSION AUTHENTICATION COMMAND message or DEREGISTRATION REQUEST message with the re-registration type "re-registration required"	None
Back-off			defined in 3GPP TS 24.008 [12]		
T3586 NOTE 4 NOTE 5	8s In WB- N1/CE mode, 16s For access via a satellite NG-RAN cell, 13s NOTE 7	PDU SESSION ACTIVE	REMOTE UE REPORT message sent	REMOTE UE REPORT RESPONSE message received	On the 1 st and 2 nd expiry, retransmission of REMOTE UE REPORT message On the 3 rd expiry, the procedure is aborted (see subclause 6.6.2.4).
T3587	NOTE 8	PDU SESSION ACTIVE	PDU SESSION MODIFICATION COMMAND message or PDU SESSION ESTABLISHMENT ACCEPT message received with Received MBS information that includes MBS decision set to "MBS join is rejected" and Rejection cause set to "multicast MBS session has not started or will not start soon" and an MBS back-off timer value	None	Initiating a request to join the mutlicast MBS session associated with the PDU session if still needed

		STATE	CAUSE OF START	NORMAL STOP			
NOW.	VALUE				1 st . 2 nd . 3 rd . 4 th		
					EXPIRY (NOTE 1)		
NOTE 1:	Typically, the procedures are aborted on the fifth expiry of the relevant timer. Exceptions are described in the						
	corresponding	g procedure descrip	tion.				
NOTE 2:	The network I	may provide the valu	ue of this timer applicable to the de	erived QoS rules of a	specific PDU session		
	as RQ timer v	alue in the PDU SE	SSION ESTABLISHMENT ACCE	PT message and PDL	J SESSION		
	MODIFICATIO	ON COMMAND me	ssage. The maximum value of the	timer is 30 min. If the	network indicates a		
	value greater	than the maximum	value, then the UE shall use the m	aximum value.			
NOTE 3:	The value of t	this timer is provided	d by the network.				
NOTE 4:	In NB-N1 mod	de, then the timer va	alue shall be calculated as describe	ed in subclause 4.18.			
NOTE 5:	In WB-N1 mo	de, if the UE suppor	rts CE mode B and operates in eitl	ner CE mode A or CE	mode B, then the		
	timer value is	as described in this	table for the case of WB-N1/CE n	node (see subclause 4	4.20).		
NOTE 6:	If the PDU SE	ESSION RELEASE	COMMAND message includes the	Back-off timer value	IE where the timer		
	value indicates neither zero nor deactivated and the 5GSM cause is not #39, the UE then starts the timer with						
	the value prov	vided in the Back-off	f timer value IE after stopping the o	existing timer (see sul	bclause 6.3.3.3).		
NOTE 7:	In satellite NG	G-RAN access, this v	value shall be selected when satel	lite NG-RAN RAT type	e is NR(MEO) or		
	NR(GEO).						
NOTE 8:	The value of t	this timer is provided	d by the network in the Received M	IBS container IE (see	subclause 6.3.2.3,		
	subclause 6.4	1.1.3 and subclause	9.11.4.31).				

- NOTE 1: The back-off timer is used to describe a logical model of the required UE behaviour. This model does not imply any specific implementation, e.g. as a timer of timestamp.
- NOTE 2: Reference to back-off timer in this section can either refer to use of timer T3396 or to use of a different packet system specific timer within the UE. Whether the UE uses T3396 as a back-off timer or it uses different packet system specific timers as back-off timers is left up to UE implementation.

TIMER	TIMER	STATE	CAUSE OF START	NORMAL STOP	ON	
NUM.	VALUE				THE 1 st , 2 nd , 3 rd , 4 th EXPIRY (NOTE 1)	
T3590 NOTE 3 NOTE 4	15s In WB- N1/CE mode, 23s For access via a satellite NG-RAN cell, 21s NOTE 5	PROCEDURE TRANSACTION PENDING	Transmission of PDU SESSION AUTHENTICATION COMMAND message	PDU SESSION AUTHENTICATION COMPLETE message received	Retransmission of PDU SESSION AUTHENTICATION COMMAND message	
T3591 NOTE 3 NOTE 4	16s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 22s NOTE 5	PDU SESSION MODIFICATION PENDING	Transmission of PDU SESSION MODIFICATION COMMAND message	PDU SESSION MODIFICATION COMPLETE message received or PDU SESSION MODIFICATION COMMAND REJECT message received	Retransmission of PDU SESSION MODIFICATION COMMAND message	
T3592 NOTE 3 NOTE 4	16s In WB- N1/CE mode, 24s For access via a satellite NG-RAN cell, 22s NOTE 5	PDU SESSION INACTIVE PENDING	Transmission of PDU SESSION RELEASE COMMAND message	PDU SESSION RELEASE COMPLETE message received or N1 SM delivery skipped indication received	Retransmission of PDU SESSION RELEASE COMMAND message	
T3593 NOTE 3 NOTE 4	Default 60s (NOTE 2)	PDU SESSION MODIFICATION PENDING	Reception of PDU SESSION MODIFICATION COMPLETE message for transmitted PDU SESSION MODIFICATION COMMAND message where the PDU SESSION MODIFICATION COMMAND message included 5GSM cause #39	PDU SESSION RELEASE REQUEST message received	Network-requested PDU session release procedure performed	
T3594 NOTE 3 NOTE 4	15s In WB- N1/CE mode, 23s For access via a satellite NG-RAN cell, 21s NOTE 5	PROCEDURE TRANSACTION PENDING	Transmission of SERVICE- LEVEL AUTHENTICATION COMMAND message	SERVICE-LEVEL AUTHENTICATION COMPLETE message received	Retransmission of SERVICE-LEVEL AUTHENTICATION COMMAND message	
NOTE 1:	Typically, the	procedures are abo	orted on the fifth expiry of the relev	ant timer. Exceptions	are described in the	
NOTE 2:	corresponding procedure description. If the PDU Session Address Lifetime value is sent to the UE in the PDU SESSION MODIFICATION COMMAND message then timer T3593 shall be started with the same value, otherwise it shall use a default value					
NOTE 3:	In NB-N1 mod	the timer value s	shall be calculated as described in	subclause 4.18.	mode B than the	
110124.1	timer value is	as described in this	table for the case of WB-N1/CE n	node (see subclause	4.20).	
NOTE 5: I	n satellite NG NR(GEO).	-RAN access, this v	value shall be selected when satel	lite NG-RAN RAT typ	e is NR(MEO) or	

Table 10.3.2: Timers of 5GS session management – SMF side

Annex A (informative): Cause values for 5GS mobility management

A.1 Causes related to UE identification

Cause #3 - Illegal UE

This 5GMM cause is sent to the UE when the network refuses service to the UE either because an identity of the UE is not acceptable to the network or because the UE does not pass the authentication check.

Cause #6 – Illegal ME

This 5GMM cause is sent to the UE if the ME used is not acceptable to the network, e.g. on the prohibited list.

Cause #9 – UE identity cannot be derived by the network.

This 5GMM cause is sent to the UE when the network cannot derive the UE's identity from the 5G-GUTI or 5G-S-TMSI because of e.g. no matching identity/context in the network, failure to validate the UE's identity due to integrity check failure of the received message.

Cause #10 - Implicitly de-registered

This 5GMM cause is sent to the UE either if the network has implicitly de-registered the UE, e.g. after the implicit de-registration timer has expired, or if the 5GMM context data related to the subscription does not exist in the AMF e.g. because of a AMF restart, or because of a registration request for mobility or registration update is routed to a new AMF.

A.2 Cause related to subscription options

Cause #5 - PEI not accepted

This cause is sent to the UE if the network does not accept an initial registration procedure for emergency services using a PEI.

Cause #7 - 5GS services not allowed

This 5GMM cause is sent to the UE when it is not allowed to operate 5GS services.

Cause #11 - PLMN not allowed

This 5GMM cause is sent to the UE if it requests service, or if the network initiates a de-registration request, in a PLMN where the UE, by subscription or due to operator determined barring, is not allowed to operate.

This 5GMM cause can also be sent to the UE when the disaster condition is no longer being applicable in the current location of the UE.

Cause #12 – Tracking area not allowed

This 5GMM cause is sent to the UE if it requests service, or if the network initiates a de-registration request, in a tracking area where the HPLMN or SNPN determines that the UE, by subscription, is not allowed to operate.

NOTE 1: If 5GMM cause #12 is sent to a roaming subscriber the subscriber is denied service even if other PLMNs are available on which registration was possible.

Cause #13 - Roaming not allowed in this tracking area

This 5GMM cause is sent to a UE which requests service, or if the network initiates a de-registration request, in a tracking area of a PLMN or SNPN which by subscription offers roaming to that UE but not in that tracking area.

This 5GMM cause can also be sent to the UE when the disaster condition is no longer being applicable in the current location of the UE.

NOTE 2: The network does not send 5GMM cause value #13 to the UE operating in SNPN access operation mode in this release of specification.

Cause #15 - No suitable cells in tracking area

This 5GMM cause is sent to the UE if it requests service, or if the network initiates a de-registration request, in a tracking area where the UE, by subscription, is not allowed to operate, but when it should find another allowed tracking area in the same PLMN or an equivalent PLMN or the same SNPN.

- NOTE 3: Cause #15 and cause #12 differ in the fact that cause #12 does not trigger the UE to search for another allowed tracking area on the same PLMN or SNPN.
- Cause #27 N1 mode not allowed

This 5GMM cause is sent to the UE if it requests service, or if the network initiates a de-registration request, in a PLMN or SNPN where the UE by subscription or operator policy, is not allowed to operate in N1 mode.

Cause #31 - Redirection to EPC required

This 5GMM cause is sent to the UE if it requests service in a PLMN where the UE by operator policy, is not allowed in 5GCN and redirection to EPC is required.

Cause #36 - IAB-node operation not authorized

This 5GMM cause is sent to the UE if a UE operating as an IAB-node requests service, or if the network initiates a de-registration procedure, in a PLMN or SNPN where the UE by subscription is not authorized for IAB operation.

Cause #72 - Non-3GPP access to 5GCN not allowed

This 5GMM cause is sent to the UE if it requests accessing 5GCN over non-3GPP access in a PLMN or SNPN, where the UE by subscription, is not allowed to access 5GCN over non-3GPP access.

- NOTE 3: The term "non-3GPP access" in an SNPN refers to the case where the UE is accessing SNPN services via a PLMN.
- Cause #74 Temporarily not authorized for this SNPN

This 5GMM cause is sent to the UE if it requests access, or if the network initiates a de-registration procedure, in a cell belonging to an SNPN for which the UE has no subscription to operate or for which the UE is not allowed to operate onboarding services.

Cause #75 - Permanently not authorized for this SNPN

This 5GMM cause is sent to the UE if it requests access, or if the network initiates a de-registration procedure, in a cell belonging to an SNPN with a globally-unique SNPN identity for which the UE either has no subscription to operate, the UE's subscription has expired or the UE is not allowed to operate onboarding services.

Cause #76 - Not authorized for this CAG or authorized for CAG cells only

This 5GMM cause is sent to the UE if the UE requests access or de-registration:

- i) in a CAG cell with a CAG-ID which is not included in the UE's "allowed CAG list" for the PLMN; or
- ii) in a non-CAG cell, wherein the UE is only allowed to access 5GS via CAG cells

Cause #77 - Wireline access area not allowed

This 5GMM cause is sent to the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) if the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) request accessing 5GCN over a wireline access network belonging to a wireline access area, where the 5G-RG or the W-AGF acting on behalf of the FN-CRG (or on behalf of the N5GC device) are not allowed by subscription to access the 5GCN over the wireline access.

Cause #79 - UAS services not allowed

This 5GMM cause is sent to the UE to indicate that the request of UAS services is not allowed.

Cause #80 - Disaster roaming for the determined PLMN with disaster condition not allowed

This 5GMM cause is sent by the network in a PLMN where the UE has requested registration for disaster roaming services for the determined PLMN with disaster condition, but the AMF determines that it does not support providing disaster roaming services to the UE for the determined PLMN with disaster condition as roaming agreement for disaster roaming services with HPLMN of the UE does not exist, or the determined PLMN with disaster condition is a forbidden PLMN of the UE.

A.3 Causes related to PLMN or SNPN specific network failures and congestion/authentication failures

Cause #20 - MAC failure

This 5GMM cause is sent to the network if the USIM detects that the MAC in the AUTHENTICATION REQUEST message is not fresh.

Cause #21 – Synch failure

This 5GMM cause is sent to the network if the USIM detects that the SQN in the AUTHENTICATION REQUEST message is out of range.

Cause #22 - Congestion

This 5GMM cause is sent to the UE because of congestion in the network (e.g. no channel, facility busy/congested etc.).

Cause #23 – UE security capabilities mismatch

This 5GMM cause is sent to the network if the UE detects that the UE security capability does not match the one sent back by the network.

Cause #24 - Security mode rejected, unspecified

This 5GMM cause is sent to the network if the security mode command is rejected by the UE for unspecified reasons.

Cause #26 - Non-5G authentication unacceptable

This 5GMM cause is sent to the network in N1 mode if the "separation bit" in the AMF field of AUTN is set to 0 in the AUTHENTICATION REQUEST message (see 3GPP TS 33.501 [24]).

Cause #28 - Restricted service area

This 5GMM cause is sent to the UE if it requests service in a tracking area of the 3GPP access or in an area of the wireline access, which is a part of the UE's non-allowed area or is not a part of the UE's allowed area.

Cause #43 - LADN not available

This 5GMM cause is sent to the UE if the user-plane resources of the PDU session are not established when the UE is located outside the LADN service area.

Cause #62 - No network slices available

This 5GMM cause is sent by the network if none of the requested network slice(s) in the registration request are allowed and there are no default network slice(s) configured in the network.

NOTE: Network does not send this cause in REGISTRATION REJECT message if the UE does not include a requested NSSAI in the REGISTRATION REQUEST message. In that case, if the UE is not registered for onboarding services in SNPN, the network uses other causes (e.g. #13, #15, etc.) based on the subscription.

Cause #65 - Maximum number of PDU sessions reached

This 5GMM cause is used by the network to indicate that the procedure requested by the UE was rejected as the network has reached the maximum number of simultaneously active PDU sessions for the UE.

Cause #67 - Insufficient resources for specific slice and DNN

This 5GMM cause is sent by the network to indicate that the requested service cannot be provided due to insufficient resources for specific slice and DNN.

Cause #69 - Insufficient resources for specific slice

This 5GMM cause is sent by the network to indicate that the requested service cannot be provided due to insufficient resources for specific slice.

Cause #71 - ngKSI already in use

This 5GMM cause is sent to the network in N1 mode if the ngKSI value received in the AUTHENTICATION REQUEST message is already associated with one of the 5G security contexts stored in the UE.

Cause #73 – Serving network not authorized

This 5GMM cause is sent to the UE if the UE initiates registration towards a serving network and the serving network fails to be authorized by the UE's home network.

Cause #78 -PLMN not allowed to operate at the present UE location

This 5GMM cause is sent to the UE to indicate that the PLMN is not allowed to operate at the present UE location.

NOTE: This cause is only applicable for satellite NG-RAN access.

Cause #90 - Payload was not forwarded

This 5GMM cause is sent by the network to indicate that the requested service cannot be provided because payload could not be forwarded by AMF.

Cause #91 - DNN not supported or not subscribed in the slice

This 5GMM cause is sent by the network to indicate that the requested service cannot be provided because payload could not be forwarded by AMF because the DNN is not supported or not subscribed in the slice selected by the network if the UE did not indicate a slice, or the DNN is not supported or not subscribed in the slice indicated by the UE.

Cause #92 - Insufficient user-plane resources for the PDU session

This 5GMM cause is sent by the network to indicate that the requested service cannot be provided due to insufficient user-plane resources for the PDU session.

Cause #93 - Onboarding services terminated

This 5GMM cause is sent by the network if the network initiates a de-registration procedure because the onboarding services are terminated.

A.4 Causes related to invalid messages

Cause #95 - Semantically incorrect message

This 5GMM cause is used to report receipt of a message with semantically incorrect contents.

Cause #96 - Invalid mandatory information

This cause 5GMM indicates that the equipment sending this 5GMM cause has received a message with a non-semantical mandatory IE error.

Cause #97 - Message type non-existent or not implemented

This 5GMM cause indicates that the equipment sending this 5GMM cause has received a message with a message type it does not recognize either because this is a message not defined, or defined but not implemented by the equipment sending this 5GMM cause.

Cause #98 - Message type not compatible with protocol state

This 5GMM cause indicates that the equipment sending this 5GMM cause has received a message not compatible with the protocol state.

Cause #99 - Information element non-existent or not implemented

This 5GMM cause indicates that the equipment sending this 5GMM cause has received a message which includes information elements not recognized because the information element identifier is not defined or it is defined but not implemented by the equipment sending the 5GMM cause. However, the information element is not required to be present in the message in order for the equipment sending the 5GMM cause to process the message.

Cause #100 - Conditional IE error

This 5GMM cause indicates that the equipment sending this cause has received a message with conditional IE errors.

Cause #101 – Message not compatible with protocol state

This 5GMM cause indicates that a message has been received which is incompatible with the protocol state.

Cause #111 - Protocol error, unspecified

This 5GMM cause is used to report a protocol error event only when no other 5GMM cause in the protocol error class applies.
Annex B (informative): Cause values for 5GS session management

B.1 Causes related to nature of request

Cause #8 - Operator Determined Barring

This 5GSM cause is used by the network to indicate that the requested service was rejected by the SMF due to Operator Determined Barring.

Cause #26 - Insufficient resources

This 5GSM cause is used by the UE or by the network to indicate that the requested service cannot be provided due to insufficient resources.

Cause #27 – Missing or unknown DNN

This 5GSM cause is used by the network to indicate that the requested service was rejected by the external DN because the DNN was not included although required or if the DNN could not be resolved.

Cause #28 – Unknown PDU session type

This 5GSM cause is used by the network to indicate that the requested service was rejected by the external DN because the requested PDU session type could not be recognised or is not allowed.

Cause #29 - User authentication or authorization failed

This 5GSM cause is used by the network to indicate that the requested service was rejected by the external DN due to a failed user authentication, revoked by the external DN, or rejected by 5GCN due to a failed user authentication or authorization.

Cause #31 - Request rejected, unspecified

This 5GSM cause is used by the network or by the UE to indicate that the requested service or operation or the request for a resource was rejected due to unspecified reasons.

Cause #32 - Service option not supported

This 5GSM cause is used by the network when the UE requests a service which is not supported by the PLMN.

Cause #33 - Requested service option not subscribed

This 5GSM cause is sent when the UE requests a service option for which it has no subscription.

Cause #35 – PTI already in use

This 5GSM cause is used by the network to indicate that the PTI included by the UE is already in use by another active UE requested procedure for this UE.

Cause #36 - Regular deactivation

This 5GSM cause is used to indicate a regular UE or network initiated release of PDU session resources.

Cause #37 - 5GS QoS not accepted

This 5GSM cause is used by the network if the new 5GS QoS that was indicated in the UE request cannot be accepted.

Cause #38 - Network failure

This 5GSM cause is used by the network to indicate that the requested service was rejected due to an error situation in the network.

Cause #39 – Reactivation requested

This 5GSM cause is used by the network to request a PDU session reactivation.

Cause #41 - Semantic error in the TFT operation

This 5GSM cause is used by the UE to indicate a semantic error in the TFT operation included in the request.

Cause #42 - Syntactical error in the TFT operation

This 5GSM cause is used by the UE to indicate a syntactical error in the TFT operation included in the request.

Cause #43 – Invalid PDU session identity

This 5GSM cause is used by the network or the UE to indicate that the PDU session identity value provided to it is not a valid value or the PDU session identified by the PDU session identity IE in the request or the command is not active.

Cause #44 – Semantic errors in packet filter(s)

This 5GSM cause is used by the network or the UE to indicate that the requested service was rejected due to one or more semantic errors in packet filter(s) of the QoS rule included in the request.

Cause #45 – Syntactical error in packet filter(s)

This 5GSM cause is used by the network or the UE to indicate that the requested service was rejected due to one or more syntactical errors in packet filter(s) of the QoS rule included in the request.

Cause #46 - Out of LADN service area

This 5GSM cause is used by the network to indicate the UE is out of the LADN service area.

Cause #47 - PTI mismatch

This 5GSM cause is used by the network or UE to indicate that the PTI provided to it does not match any PTI in use.

Cause #50 - PDU session type IPv4 only allowed

This 5GSM cause is used by the network to indicate that only PDU session type IPv4 is allowed for the requested IP connectivity.

Cause #51 - PDU session type IPv6 only allowed

This 5GSM cause is used by the network to indicate that only PDU session type IPv6 is allowed for the requested IP connectivity.

Cause #54 - PDU session does not exist

This 5GSM cause is used by the network at handover of a PDU session between non-3GPP access and 3GPP access, or at interworking of a PDN connection from non-3GPP access network connected to EPC or from E-UTRAN connected to EPC to a PDU session, to indicate that the network does not have any information about the requested PDU session.

Cause #57 - PDU session type IPv4v6 only allowed

This 5GSM cause is used by the network to indicate that only PDU session types IPv4, IPv6 or IPv4v6 are allowed for the requested IP connectivity.

Cause #58 - PDU session type Unstructured only allowed

This 5GSM cause is used by the network to indicate that only PDU session type Unstructured is allowed for the requested DN connectivity.

Cause #59 - Unsupported 5QI value

This 5GSM cause is used by the network if the 5QI indicated in the UE request cannot be supported.

Cause #61 - PDU session type Ethernet only allowed

This 5GSM cause is used by the network to indicate that only PDU session type Ethernet is allowed for the requested DN connectivity.

Cause #67 - Insufficient resources for specific slice and DNN

This 5GSM cause is by the network to indicate that the requested service cannot be provided due to insufficient resources for specific slice and DNN.

Cause #68 – Not supported SSC mode

This 5GSM cause is used by the network to indicate that the requested SSC mode is not supported.

Cause #69 - Insufficient resources for specific slice

This 5GSM cause is used by the network to indicate that the requested service cannot be provided due to insufficient resources for specific slice, maximum number of PDU sessions on a specific slice has been already reached, data rate on a specific slice has been exceeded, or UE-Slice-MBR has been exceeded.

Cause #70 - Missing or unknown DNN in a slice

This 5GSM cause is used by the network to indicate that the requested service was rejected by the external DN because the DNN was not included although required or if the DNN could not be resolved, in the slice.

Cause #81 - Invalid PTI value

This 5GSM cause is used by the network or UE to indicate that the PTI provided to it is invalid for the specific 5GSM message.

Cause #82 - Maximum data rate per UE for user-plane integrity protection is too low

This 5GSM cause is used by the network to indicate that the requested service cannot be provided because the maximum data rate per UE for user-plane integrity protection is too low.

Cause #83 - Semantic error in the QoS operation

This 5GSM cause is used by the network or the UE to indicate that the requested service was rejected due to a semantic error in the QoS operation included in the request.

Cause #84 - Syntactical error in the QoS operation

This 5GSM cause is used by the network or the UE to indicate that the requested service was rejected due to a syntactical error in the QoS operation included in the request.

Cause #85 - Invalid mapped EPS bearer identity

This 5GSM cause is used by the network or the UE to indicate that the mapped EPS bearer identity value provided to it is not a valid value or the mapped EPS bearer identified by the mapped EPS bearer identity does not exist.

Cause #86 - UAS services not allowed

This 5GSM cause is used by the network to indicate that the requested UAS services are not allowed.

B.2 Protocol errors (e.g., unknown message)

Cause #95 - Semantically incorrect message

This 5GSM cause is used to report receipt of a message with semantically incorrect contents.

Cause #96 - Invalid mandatory information

This 5GSM cause indicates that the equipment sending this 5GSM cause has received a message with a non-semantical mandatory IE error.

Cause #97 - Message type non-existent or not implemented

This 5GSM cause indicates that the equipment sending this 5GSM cause has received a message with a message type it does not recognize either because this is a message not defined, or defined but not implemented by the equipment sending this 5GSM cause.

Cause #98 – Message type not compatible with protocol state

This 5GSM cause indicates that the equipment sending this 5GSM cause has received a message not compatible with the protocol state.

Cause #99 - Information element non-existent or not implemented

This 5GSM cause indicates that the equipment sending this 5GSM cause has received a message which includes information elements not recognized because the information element identifier is not defined or it is defined but not implemented by the equipment sending the 5GSM cause. However, the information element is not required to be present in the message in order for the equipment sending the 5GSM cause to process the message.

Cause #100 - Conditional IE error

This 5GSM cause indicates that the equipment sending this cause has received a message with conditional IE errors.

Cause #101 – Message not compatible with protocol state

This 5GSM cause indicates that a message has been received which is incompatible with the protocol state.

Cause #111 - Protocol error, unspecified

This 5GSM cause is used to report a protocol error event only when no other 5GSM cause in the protocol error class applies.

Annex C (normative): Storage of 5GMM information

C.1 Storage of 5GMM information for UEs not operating in SNPN access operation mode

The following 5GMM parameters shall be stored on the USIM if the corresponding file is present:

a) 5G-GUTI;

- b) last visited registered TAI;
- c) 5GS update status;
- d) 5G NAS security context parameters from a full native 5G NAS security context (see 3GPP TS 33.501 [24]);
- e) K_{AUSF} and K_{SEAF} (see 3GPP TS 33.501 [24]);
- f) SOR counter (see subclause 9.11.3.51); and
- g) UE parameter update counter (see subclause 9.11.3.53A);

The UE may support multiple records of NAS security context storage for multiple registration (see 3GPP TS 31.102 [22]). If the UE supports multiple records of NAS security context storage for multiple registration, the first 5G security context of one access shall be stored in record 1 of the 5G NAS Security Context USIM file for that access and the second 5G security context of that access shall be stored in record 2 of the same file. The presence and format of corresponding files on the USIM is specified in 3GPP TS 31.102 [22].

If the corresponding file is not present on the USIM, these 5GMM parameters are stored in a non-volatile memory in the ME together with the SUPI from the USIM. These 5GMM parameters can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory; else the UE shall delete the 5GMM parameters.

The following 5GMM parameters shall be stored in a non-volatile memory in the ME together with the SUPI from the USIM:

- configured NSSAI(s);
- NSSRG information;
- NSSAI inclusion mode(s);
- MPS indicator;
- MCS indicator;
- operator-defined access category definitions;
- network-assigned UE radio capability IDs;
- "CAG information list", if the UE supports CAG;
- signalled URSP (see 3GPP TS 24.526 [19]);
- SOR-CMCI;
- one or more lists of type "list of PLMN(s) to be used in disaster condition", if the UE supports MINT;
- disaster roaming wait range, if the UE supports MINT;
- disaster return wait range, if the UE supports MINT;
- indication of whether disaster roaming is enabled in the UE; and

- indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN'.

The following 5GMM parameters should be stored in a non-volatile memory in the ME together with the SUPI from the USIM:

- allowed NSSAI(s).

Each configured NSSAI consists of S-NSSAI(s) stored together with a PLMN identity, if it is associated with a PLMN. The UE shall store the S-NSSAI(s) of the HPLMN. If the UE is in the VPLMN, the UE shall also store the configured NSSAI for the current PLMN and any necessary mapped S-NSSAI(s). The configured NSSAI(s) can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the configured NSSAI(s). A configured NSSAI may be associated with NSSRG information.

Each NSSAI inclusion mode is associated with a PLMN identity and access type. The NSSAI inclusion mode(s) can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the NSSAI inclusion mode(s).

The MPS indicator is stored together with a PLMN identity of the PLMN that provided it, and is valid in that RPLMN or equivalent PLMN. The MPS indicator can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME, else the UE shall delete the MPS indicator.

The MCS indicator is stored together with a PLMN identity of the PLMN that provided it, and is valid in that RPLMN or equivalent PLMN. The MCS indicator can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME, else the UE shall delete the MCS indicator.

Operator-defined access category definitions are stored together with a PLMN identity of the PLMN that provided them, and is valid in that PLMN or equivalent PLMN. The operator-defined access category definitions can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME, else the UE shall delete the operator-defined access category definitions. The maximum number of stored operator-defined access category definitions is UE implementation dependent.

Each network-assigned UE radio capability ID is stored together with a PLMN identity of the PLMN that provided it as well as a mapping to the corresponding UE radio configuration, and is valid in that PLMN. A network-assigned UE radio capability ID can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME, else the UE shall delete the network-assigned UE radio capability ID. The UE shall be able to store at least the last 16 received network-assigned UE radio capability IDs. There shall be only one network-assigned UE radio capability ID stored for a given combination of PLMN identity and UE radio configuration and any existing UE radio capability ID shall be deleted when a new UE radio capability ID is added for the same combination of PLMN identity and UE radio capability ID with a Version ID value different from the value included in the network-assigned UE radio capability ID(s) stored at the UE for the serving PLMN, the UE may delete these stored network-assigned UE radio capability ID(s).

The allowed NSSAI(s) can be stored in a non-volatile memory in the ME together with the SUPI from the USIM. Allowed NSSAI consists of S-NSSAI(s) stored together with a PLMN identity, if it is associated with a PLMN. If the allowed NSSAI is stored, then the UE shall store the S-NSSAI(s) of the HPLMN. If the UE is in the VPLMN, the UE shall also store the allowed NSSAI for the serving PLMN and any necessary mapping of the allowed NSSAI for the serving PLMN to the S-NSSAI(s) of the HPLMN. The allowed NSSAI(s) can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the allowed NSSAI(s).

If the UE is registered for emergency services, the UE shall not store the 5GMM parameters described in this annex on the USIM or in non-volatile memory. Instead the UE shall temporarily store these parameters locally in the ME and the UE shall delete these parameters when the UE is deregistered.

If the UE is configured for eCall only mode as specified in 3GPP TS 31.102 [22], the UE shall not store the 5GMM parameters described in this annex on the USIM or in non-volatile memory. Instead the UE shall temporarily store these parameters locally in the ME and the UE shall delete these parameters when the UE enters 5GMM-DEREGISTERED.eCALL-INACTIVE state, the UE is switched-off or the USIM is removed.

The "CAG information list" can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the "CAG information list".

The handling of the SOR-CMCI stored in the non-volatile memory in the ME is specified in 3GPP TS 23.122 [5].

Each "list of PLMN(s) to be used in disaster condition" is stored together with the PLMN identity of the PLMN that provided it. The stored lists of type "list of PLMN(s) to be used in disaster condition" can only be used if the SUPI from

the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the lists of type "list of PLMN(s) to be used in disaster condition". The UE shall store at least the "list of PLMN(s) to be used in disaster condition" provided by the HPLMN or EHPLMN. If the 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN' is set to "true", the UE should also store the "list of PLMN(s) to be used in disaster condition" provided by the VPLMN. The maximum number of stored lists of type "list of PLMN(s) to be used in disaster condition" provided by a PLMN other than the HPLMN or EHPLMN is UE implementation dependent.

The disaster roaming wait range can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the disaster roaming wait range.

The disaster return wait range can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the disaster return wait range.

The indication of whether disaster roaming is enabled in the UE can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the indication of whether disaster roaming is enabled in the UE.

The indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN' can only be used if the SUPI from the USIM matches the SUPI stored in the non-volatile memory of the ME; else the UE shall delete the indication of 'applicability of "lists of PLMN(s) to be used in disaster condition" provided by a VPLMN'.

C.2 Storage of 5GMM information for UEs operating in SNPN access operation mode

The 5GMM information for UEs operating in SNPN access operation mode and not registering or registered for the onboarding service in SNPN are stored according to the following conditions:

- if the UE does not support access to an SNPN using credentials from a credentials holder, the following 5GMM parameters shall be stored per subscribed SNPN in a non-volatile memory in the ME together with the subscriber identifier associated with the SNPN identity of the SNPN in the "list of subscriber data" configured in the ME (see 3GPP TS 23.122 [5]) or with the SUPI from the USIM if no subscriber identifier is configured in the entry of the "list of subscriber data" associated with the SNPN identity and the UE has a valid USIM;and
- if the UE supports access to an SNPN using credentials from a credentials holder, the following 5GMM parameters shall be stored in a non-volatile memory in the ME per:
 - i) the subscribed SNPN together with the subscriber identifier associated with the selected entry in the "list of subscriber data" configured in the ME (see 3GPP TS 23.122 [5]) or with the SUPI from the USIM if no subscriber identifier is configured in the selected entry of the "list of subscriber data" configured in the ME and the UE has a valid USIM; or
 - ii) the PLMN subscription together with the SUPI from the USIM which is associated with the PLMN subscription:
- a) 5G-GUTI;
- b) last visited registered TAI;
- c) 5GS update status;
- d) 5G NAS security context parameters from a full native 5G NAS security context (see 3GPP TS 33.501 [24]);
- e) K_{AUSF} and K_{SEAF} (see 3GPP TS 33.501 [24]);
- f) UE parameter update counter (see subclause 9.11.3.53A);
- g) configured NSSAI(s);
- g1)NSSRG information;
- h) NSSAI inclusion mode(s);
- i) MPS indicator;

- j) MCS indicator;
- k) operator-defined access category definitions;
- l) network-assigned UE radio capability IDs;
- m) zero or more instances of signalled URSP (see 3GPP TS 24.526 [19]), each associated with a non-subscribed SNPN, the subscribed SNPN or the HPLMN, which provided the URSP;
- n) optionally a non-subscribed SNPN signalled URSP handling indication indicating whether the UE is allowed to accept URSP signalled by non-subscribed SNPNs;
- o) permanently forbidden SNPNs list;
- p) temporarily forbidden SNPNs;
- q) SOR counter (see subclause 9.11.3.51); and
- r) SOR-CMCI.

The 5GMM information for UEs operating in SNPN access operation mode and registering or registered for the onboarding service in SNPN are stored as follows:

- a) 5G-GUTI;
- b) last visited registered TAI;
- c) 5GS update status;
- d) 5G NAS security context parameters from a full native 5G NAS security context (see 3GPP TS 33.501 [24]);
- e) K_{AUSF} and K_{SEAF} (see 3GPP TS 33.501 [24]);
- f) UE parameter update counter (see subclause 9.11.3.53A);
- g) network-assigned UE radio capability IDs;
- h) "permanently forbidden SNPNs" list for onboarding services; and
- i) "temporarily forbidden SNPNs" list for onboarding services.

The 5GMM information for UEs operating in SNPN access operation mode are stored according to the following conditions:

- if the UE does not support access to an SNPN using credentials from a credentials holder, the following 5GMM parameters should be stored per subscribed SNPN in a non-volatile memory in the ME together with the subscriber identifier associated with the SNPN identity of the SNPN in the "list of subscriber data" configured in the ME (see 3GPP TS 23.122 [5]) or with the SUPI from the USIM if no subscriber identifier is configured in the entry of the "list of subscriber data" associated with the SNPN identity and the UE has a valid USIM; and
- if the UE supports access to an SNPN using credentials from a credentials holder, the following 5GMM parameters should be stored in a non-volatile memory in the ME per:
 - i) the subscribed SNPN together with the subscriber identifier associated with the selected entry in the "list of subscriber data" configured in the ME (see 3GPP TS 23.122 [5]) or with the SUPI from the USIM if no subscriber identifier is configured in the selected entry of the "list of subscriber data" configured in the ME and the UE has a valid USIM; or
 - ii) the PLMN subscription together with the SUPI from the USIM which is associated with the PLMN subscription:
- a) allowed NSSAI(s).

If the 5GMM parameters are associated with the PLMN subscription, then the 5GMM parameters can only be used if the SUPI from the USIM which is associated with the selected PLMN subscription matches the SUPI stored in the non-volatile memory; else the UE shall delete the 5GMM parameters.

If the 5GMM parameters are associated with the subscribed SNPN of the entry in the "list of subscriber data", then the 5GMM parameters can only be used if the subscriber identifier of the selected entry of the "list of subscriber data" matches the subscriber identifier stored in the non-volatile memory or if the subscriber identifier from the USIM matches the subscriber identifier stored in the non volatile memory, no subscriber identifier is configured in the selected entry of the "list of subscriber data" configured in the ME and the UE has a valid USIM.

Each configured NSSAI consists of S-NSSAI(s) stored together with an SNPN identity, if it is associated with an SNPN. A configured NSSAI may be associated with NSSRG information.

Each NSSAI inclusion mode is associated with an SNPN identity and access type.

The MPS indicator is stored together with an SNPN identity of the SNPN that provided it, and is valid in that registered SNPN.

The MCS indicator is stored together with an SNPN identity of the SNPN that provided it, and is valid in that registered SNPN.

Operator-defined access category definitions are stored together with an SNPN identity of the SNPN that provided them, and are valid in that SNPN. The maximum number of stored operator-defined access category definitions is UE implementation dependent.

Each network-assigned UE radio capability ID is stored together with an SNPN identity of the SNPN that provided it as well as a mapping to the corresponding UE radio configuration, and is valid in that SNPN. The UE shall be able to store at least the last 16 received network-assigned UE radio capability IDs. There shall be only one network-assigned UE radio capability ID stored for a given combination of SNPN identity and UE radio configuration and any existing UE radio capability ID shall be deleted when a new UE radio capability ID is added for the same combination of SNPN identity and UE radio capability ID with a Version ID value different from the value included in the network-assigned UE radio capability ID(s) stored at the UE for the serving SNPN, the UE may delete these stored network-assigned UE radio capability ID(s).

The handling of the SOR-CMCI stored in the non-volatile memory in the ME is specified in 3GPP TS 23.122 [5].

The allowed NSSAI(s) can be stored in a non-volatile memory in the ME. Allowed NSSAI consists of S-NSSAI(s) stored together with an SNPN identity, if it is associated with an SNPN.

Annex D (normative): UE policy delivery service

D.1 General

D.1.1 Overview

The PCF may provide the UE with one or more UE policies using the network-requested UE policy management procedure. The UE provides the PCF with a list of one or more stored UE policy section identifiers (UPSIs), and the PCF provides each UE policy using one or more UE policy sections, each identified by a UPSI. The UPSI is composed of two parts:

- a) a PLMN ID part containing:
 - 1) the PLMN ID for the PLMN; or
 - 2) the PLMN ID part of the SNPN identity for the SNPN;

of the PCF which provides the UE policies; and

b) a UE policy section code (UPSC) containing a value assigned by the PCF.

The UE processes the UE policy sections, each identified by the UPSI, received from the PCF and informs the PCF of the result.

The UE provides the PCF with the UE policy related capabilities such as the UE's support for ANDSP and the UE's OS Id.

The UE can also request the PCF to provide V2XP as specified in 3GPP TS 24.587 [19B].

The UE can also request the PCF to provide ProSeP as specified in 3GPP TS 24.554 [19E].

D.1.2 Principles of PTI handling for UE policy delivery service procedures

When the PCF or the UE initiates a procedure, it shall include a valid PTI value in the message header of the command message or the request message. When the UE initiates a procedure, the UE shall use a PTI value in range between 01H and 77H. When the PCF initiates a procedure, the PCF shall use a PTI value in range between 80H and FEH.

When the PCF initiates a transaction related procedure (i.e. a procedure consisting of more than one message and the messages are related), it shall include a valid PTI value in the message header of the command message.

If a response message is sent as result of a received command or request message, the UE or the PCF shall include in the response message the PTI value received within the received command or request message (see examples in figure D.1.2.1, figure D.1.2.2 and figure D.1.2.3).

If a command message is sent as result of a received request message, the PCF shall include in the command message the PTI value received with the request message (see examples in figure D.1.2.3).



Figure D.1.2.2: UE-requested transaction related procedure rejected by the network



Figure D.1.2.3: UE-requested transaction related procedure triggering a network-requested transaction related procedure

NOTE: In earlier versions of this protocol, the UE can include in the response message a PTI value which is not the same as the one received within the command message, and therefore the PCF could not associate the response message from the UE to the command message sent.

D.2 Procedures

D.2.1 Network-requested UE policy management procedure

D.2.1.1 General

The purpose of the network-requested UE policy management procedure is to enable the network to:

- a) add one or more new UE policy sections to the UE;
- b) modify one or more UE policy sections stored at the UE; or
- c) delete one or more UE policy sections stored at the UE;

and optionally to enable the HPLMN or the subscribed SNPN to provide a UE accessing the subscribed SNPN or the HPLMN with a non-subscribed SNPN signalled URSP handling indication indicating whether the UE is allowed to accept URSP signalled by non-subscribed SNPNs.

D.2.1.2 Network-requested UE policy management procedure initiation

In order to initiate the network-requested UE policy management procedure, the PCF shall:

- a) if the network-requested UE policy management procedure is triggered by the UE-requested V2X policy provisioning procedure as specified in 3GPP TS 24.587 [19B] or the UE-requested ProSeP policy provisioning procedure as specified in 3GPP TS 24.554 [19E], then set the PTI IE to the PTI value of the received UE POLICY PROVISIONING REQUEST message of the UE-requested V2X policy provisioning procedure or the UE-requested ProSeP policy provisioning procedure, otherwise allocate a PTI value currently not used and set the PTI IE to the allocated PTI value;
- b) encode the information about the UE policy sections to be added, modified or deleted in a UE policy section management list IE as specified in subclause D.6.2 and include it in a MANAGE UE POLICY COMMAND message;
- c) if the PCF is a PCF of the HPLMN or the subscribed SNPN, optionally include the UE policy network classmark IE in a MANAGE UE POLICY COMMAND message and set the non-subscribed SNPN signalled URSP handling indication of the UE policy network classmark IE to "UE is not allowed to accept URSP signalled by non-subscribed SNPNs", or "UE is allowed to accept URSP signalled by non-subscribed SNPNs";
- d) send the MANAGE UE POLICY COMMAND message to the UE via the AMF as specified in 3GPP TS 23.502 [9]; and
- e) start timer T3501 (see example in figure D.2.1.2.1).
- NOTE: The PCF starts a different timer T3501 for each PTI value.



Figure D.2.1.2.1: Network-requested UE policy management procedure

Upon receipt of the MANAGE UE POLICY COMMAND message with a PTI value currently not used by a networkrequested UE policy management procedure, for each instruction included in the UE policy section management list IE, the UE shall:

- a) store the received UE policy section of the instruction, if the UE has no stored UE policy section associated with the same UPSI as the UPSI associated with the instruction;
- b) replace the stored UE policy section with the received UE policy section of the instruction, if the UE has a stored UE policy section associated with the same UPSI as the UPSI associated with the instruction; or
- c) delete the stored UE policy section, if the UE has a stored UE policy section associated with the same UPSI as the UPSI associated with the instruction and the UE policy section contents of the instruction is empty;

and if UE's RPLMN is the HPLMN or UE's RSNPN is the subscribed SNPN and the UE policy network classmark IE is included in the MANAGE UE POLICY COMMAND message, the UE shall delete the non-subscribed SNPN signalled URSP handling indication stored for the selected entry of "list of subscriber data" or the selected PLMN subscription, if any, and store the non-subscribed SNPN signalled URSP handling indication received in the UE policy network classmark IE, for the selected entry of "list of subscriber data" or the selected PLMN subscription.

The UE may continue storing a received UE policy section for a PLMN or SNPN when the UE registers in another PLMN or SNPN. If necessary, the UE may delete UE policy sections stored for a PLMN or SNPN other than the RPLMN and the HPLMN or the registered SNPN, before storing the new received UE policy sections.

When storing a UE policy sections received from an SNPN and the subscribed SNPN, the UE shall associate the NID of that SNPN with the UPSI of the stored UE policy section.

NOTE: The maximum number of UE policy sections for PLMNs or SNPNs other than the HPLMN and the RPLMN or the registered SNPN and the subscribed SNPN that the UE can store and how the UE selects the UE policy sections to be deleted are up to the UE implementation.

D.2.1.3 Network-requested UE policy management procedure accepted by the UE

If all instructions included in the UE policy section management list IE were executed successfully by the UE, the UE shall:

- a) create a MANAGE UE POLICY COMPLETE message including the PTI value received within the MANAGE UE POLICY COMMAND message; and
- b) transport the MANAGE UE POLICY COMPLETE message using the NAS transport procedure as specified in subclause 5.4.5.

Upon receipt of the MANAGE UE POLICY COMPLETE message, the PCF shall stop timer T3501. The PCF should ensure that the PTI value assigned to this procedure is not released immediately.

NOTE: The way to achieve this is implementation dependent. For example, the PCF can ensure that the PTI value assigned to this procedure is not released during the time equal to or greater than the default value of timer T3501.

D.2.1.4 Network-requested UE policy management procedure not accepted by the UE

If the UE could not execute all instructions included in the UE policy section management list IE successfully, the UE shall:

- a) set the PTI IE to the PTI value received within the MANAGE UE POLICY COMMAND message and encode the UPSI associated with the instructions which could not be executed successfully and the associated UE policy delivery service cause indicating the cause of the failure in a UE policy section management result IE as specified in subclause D.5.3 and include it in a MANAGE UE POLICY COMMAND REJECT message, and
- b) transport the MANAGE UE POLICY COMMAND REJECT message using the NAS transport procedure as specified in subclause 5.4.5.

Upon receipt of the MANAGE UE POLICY COMMAND REJECT message, the PCF shall stop timer T3501. Any instruction that was included in the UE policy section management list IE and whose associated UPSI is not included in a UE policy section management result IE of the received MANAGE UE POLICY COMMAND REJECT message is considered as successfully executed.

The PCF should ensure that the PTI value assigned to this procedure is not released immediately.

NOTE: The way to achieve this is implementation dependent. For example, the PCF can ensure that the PTI value assigned to this procedure is not released during the time equal to or greater than the default value of timer T3501.

Upon receipt of the notification from the AMF that the UE is not reachable, the PCF shall stop the T3501.

D.2.1.5 Abnormal cases on the network side

The following abnormal cases can be identified:

a) T3501 expired.

The PCF shall, on the first expiry of the timer T3501, retransmit the MANAGE UE POLICY COMMAND message and shall reset and start timer T3501. This retransmission is repeated four times, i.e. on the fifth expiry of timer T3501, the PCF shall abort the procedure and release the allocated PTI.

D.2.1.6 Abnormal cases in the UE

The following abnormal cases can be identified:

a) Receipt of an instruction associated with a UPSI which has a PLMN ID part that is not equal to the PLMN ID of the UE's HPLMN and the instruction contains a UE policy part with a UE policy part type set to "URSP" for a UE not operating in SNPN access operation mode, or receipt of an instruction associated with a UPSI which has a PLMN ID part that is not equal to the PLMN ID part of the selected SNPN and the instruction contains a UE policy part with a UE policy part type set to "URSP" for a UE operating in SNPN access operation mode.

The UE shall set the UE policy delivery service cause to #111 (Protocol error, unspecified) for the instruction in the UE policy section management result IE of the MANAGE UE POLICY COMMAND REJECT message.

b) Receipt of an instruction associated with a UPSI which has a PLMN ID part that is not equal to the PLMN ID of the UE's HPLMN or the UE's RPLMN and the instruction contains a UE policy part with a UE policy part type set to "ANDSP" for a UE not operating in SNPN access operation mode.

The UE shall set the UE policy delivery service cause to #111 (Protocol error, unspecified) for the instruction in the UE policy section management result IE of the MANAGE UE POLICY COMMAND REJECT message.

c) Transmission failure of the MANAGE UE POLICY COMPLETE message indication from lower layers.

The UE shall not diagnose an error and consider the network-initiated UE policy delivery procedure complete.

- NOTE 1: Considering the network-initiated UE policy delivery procedure complete as a result of this abnormal case does not cause the UE to revert the execution of the instructions included in the MANAGE UE POLICY COMMAND message.
- d) Transmission failure of the MANAGE UE POLICY COMMAND REJECT message indication from lower layers.

The UE shall not diagnose an error and consider the network-initiated UE policy delivery procedure complete.

- NOTE 2: Considering the network-initiated UE policy delivery procedure complete as a result of this abnormal case does not cause the UE to revert the execution of the instructions included in the MANAGE UE POLICY COMMAND message and successfully processed by the UE.
- e) Receipt of a MANAGE UE POLICY COMMAND message with a PTI set to the same value as the PTI of a previously received MANAGE UE POLICY COMMAND message.

The UE shall discard the message and retransmit the MANAGE UE POLICY COMMAND COMPLETE or MANAGE UE POLICY COMMAND REJECT message transmitted in response to the previously received MANAGE UE POLICY COMMAND message.

- NOTE 3: The way to achieve this is UE implementation dependent. For example, the UE can assume that on the fifth expiry of timer T3501, the PCF will abort the procedure and that the PTI value assigned to the procedure will be released.
- f) Receipt of an instruction associated with a UPSI which has a PLMN ID part that is equal to the PLMN ID part of the selected SNPN, the instruction contains a UE policy part with a UE policy part type set to "URSP", UE's RSNPN is a non-subscribed SNPN and:
 - the UE has a stored non-subscribed SNPN signalled URSP handling indication for the selected entry of "list of subscriber data" or the selected PLMN subscription indicating that the UE is not allowed to accept URSP signalled by non-subscribed SNPNs; or
 - 2) the UE does not have a stored non-subscribed SNPN signalled URSP handling indication for the selected entry of "list of subscriber data" or the selected PLMN subscription, and the non-subscribed SNPN signalled URSP handling indication preconfigured in the selected entry of "list of subscriber data" or the selected PLMN subscription indicates that the UE is not allowed to accept URSP signalled by non-subscribed SNPNs;

for a UE operating in SNPN access operation mode.

The UE shall set the UE policy delivery service cause to #111 (Protocol error, unspecified) for the instruction in the UE policy section management result IE of the MANAGE UE POLICY COMMAND REJECT message.

D.2.2 UE-initiated UE state indication procedure

D.2.2.1 General

The purpose of the UE-initiated UE state indication procedure is:

- a) to deliver the UPSI(s) of the UE policy section(s) which are:
 - identified by a UPSI with the PLMN ID part indicating the HPLMN or the selected PLMN, and stored in the UE, if any; or
 - identified by a UPSI with the PLMN ID part indicating the PLMN ID part of the SNPN identity of the selected SNPN and associated with the NID of the selected SNPN, and stored in the UE, if any;
- b) to indicate whether UE supports ANDSP; and
- c) to deliver the UE's one or more OS IDs;

to the PCF.

D.2.2.2 UE-initiated UE state indication procedure initiation

In order to initiate the UE-initiated UE state indication procedure, the UE shall create a UE STATE INDICATION message. The UE:

- a) shall allocate a PTI value currently not used and set the PTI IE to the allocated PTI value;
- b) if not operating in SNPN access operation mode, shall include the UPSI(s) of the UE policy section(s) which are identified by a UPSI with the PLMN ID part indicating the HPLMN or the selected PLMN available in the UE in the UPSI list IE;
- c) if operating in SNPN access operation mode, shall include UPSI(s) of the UE policy section(s) which are identified by a UPSI:
 - with the PLMN ID part indicating the MCC and MNC of the selected SNPN; and
 - associated with the NID of the selected SNPN;

available in the UE in the UPSI list IE;

- d) shall specify whether the UE supports ANDSP in the UE policy classmark IE; and
- e) may include the UE's one or more OS IDs in the UE OS Id IE.

The UE shall send the UE STATE INDICATION message (see example in figure D.2.2.2.1). The UE shall transport the created UE STATE INDICATION message using the registration procedure (see subclause 5.5.1).

UE

PCF

UE STATE INDICATION

Figure D.2.2.2.1: UE-initiated UE state indication procedure

D.2.2.3 UE-initiated UE state indication procedure accepted by the network

Upon receipt of the UE STATE INDICATION message, the PCF shall operate as described in 3GPP TS 23.502 [9] and 3GPP TS 29.525 [21].

D.2.2.4 Abnormal cases on the network side

Apart from the case described in subclause D.2.2.3, no abnormal cases have been identified.

D.3 UE policy re-assembly at the UE

When the UE needs to apply ANDSP as specified in 3GPP TS 24.502 [18], the UE shall consider all UE policy parts with ANDSP contents currently stored at the UE.

When the UE needs to apply URSP as specified in 3GPP TS 24.526 [19], the UE shall consider all UE policy parts with URSP contents currently stored at the UE.

When the UE needs to apply V2XP as specified in 3GPP TS 24.588 [19C], the UE shall consider all UE policy parts with V2XP contents currently stored at the UE.

When the UE needs to apply ProSeP as specified in 3GPP TS 24.555 [19F], the UE shall consider all UE policy parts with ProSeP contents currently stored at the UE.

D.4 Void

D.5 Message functional definition and contents

D.5.1 Manage UE policy command

D.5.1.1 Message definition

The MANAGE UE POLICY COMMAND message is sent by the PCF to the UE to request the UE to manage UE policy sections, see table D.5.1.1.1

Message type: MANAGE UE POLICY COMMAND

Significance: dual

Direction: network to UE

Table D.5.1.1.1: MANAGE UE POLICY COMMAND message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	PTI	Procedure transaction identity	М	V	1
		9.6			
	MANAGE UE POLICY	UE policy delivery service message	М	V	1
	COMMAND message identity	type			
		D.6.1			
	UE policy section management list	UE policy section management list	М	LV-E	11-65533
		D.6.2			
42	UE policy network classmark	UE policy network classmark	0	TLV	3-5
		D.6.7			
NOTE	: The total length of the MANAG	E UE POLICY COMMAND message co	ntent cannot e	xceed 655	35 octets
	(see Payload container content	s maximum length as specified in subcl	ause 9.11.3.3	9.1).	

D.5.1.2 UE policy network classmark

The UE policy network classmark is included when the PCF of a PLMN or an SNPN intends to provide the UE with information about the policy aspects of the network.

D.5.2 Manage UE policy complete

D.5.2.1 Message definition

The MANAGE UE POLICY COMPLETE message is sent by the UE to the PCF to report that all received instructions have been successfully executed at the UE, see table D.5.2.1.1

Message type: MANAGE UE POLICY COMPLETE

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length
	PTI	Procedure transaction identity	М	V	1
		9.6			
	MANAGE UE POLICY	UE policy delivery service message	М	V	1
	COMPLETE message identity	type			
		D.6.1			

Table D.5.2.1.1: MANAGE UE POLICY COMPLETE message content

D.5.3 Manage UE policy command reject

D.5.3.1 Message definition

The MANAGE UE POLICY COMMAND REJECT message is sent by the UE to the PCF to report that one or more instructions could not be successfully executed at the UE, see table D.5.3.1.1

Message type: MANAGE UE POLICY COMMAND REJECT

Significance: dual

Direction: UE to network

Table D.5.3.1.1: MANAGE UE POLICY COMMAND REJECT message content

IEI	Information Element	Type/Reference	Presence	Format	Length
	PTI	Procedure transaction identity	М	V	1
		9.6			
	MANAGE UE POLICY	UE policy delivery service message	М	V	1
	COMMAND REJECT message	type			
	identity	D.6.1.			
	UE policy section management	UE policy section management result	М	LV-E	11-65533
	result	D.6.3			
NOTE	: The total length of the MANAG	E UE POLICY COMMAND REJECT me	essage conten	t cannot ex	ceed
	65535 octets (see Payload con	tainer contents maximum length as spe	cified in subcla	ause 9.11.3	3.39.1).

D.5.4 UE state indication

D.5.4.1 Message definition

The UE STATE INDICATION message is sent by the UE to the PCF:

- a) to deliver the UPSI(s) of the UE policy section(s) stored in the UE;
- b) to indicate whether the UE supports ANDSP; and
- c) to deliver the UE's one or more OS IDs;

see table D.5.4.1.1.

Message type: UE STATE INDICATION

Significance: dual

Direction: UE to network

IEI	Information Element	Type/Reference	Presence	Format	Length		
	PTI	Procedure transaction identity 9.6	М	V	1		
	UE STATE INDICATION message identity	UE policy delivery service message type D.6.1	Μ	V	1		
	UPSI list	UPSI list D.6.4	М	LV-E	9-65531		
	UE policy classmark	UE policy classmark D.6.5	М	LV	2-4		
41	UE OS Id	OS ld D.6.6	0	TLV	18-242		
NOTE: The total length of the UE STATE INDICATION message content cannot exceed 65535 octets (see Payload container contents maximum length as specified in subclause 9.11.3.39.1).							

Table D.5.4.1.1: UE STATE INDICATION message content

D.6 Information elements coding

D.6.1 UE policy delivery service message type

Table D.6.1.1: UE	policy delivery	service mess	sage type
-------------------	-----------------	--------------	-----------

Bit	s							
8	7	6	5	4	3	2	1	
0	0	0	0	0	0	0	0	Reserved
0	0	0	0	0	0	0	1	MANAGE UE POLICY COMMAND message
0	0	0	0	0	0	1	0	MANAGE UE POLICY COMPLETE message
0	0	0	0	0	0	1	1	MANAGE UE POLICY COMMAND REJECT
								message
0	0	0	0	0	1	0	0	UE STATE INDICATION message
0	0	0	0	0	1	0	1	UE POLICY PROVISIONING REQUEST
								message (see NOTE)
0	0	0	0	0	1	1	0	UE POLICY PROVISIONING REJECT message
								(see NOTE)
All	oth	er v	alue	es a	ire r	ese	rved	
NC	DTE	E: Coding and usage of UE POLICY PROVISIONING REQUEST message an						of UE POLICY PROVISIONING REQUEST message and
			UE	PC	DLIC	Y F	PRÔV	ISIONING REJECT message are specified in
	3GPP TS 24.587 [19B].							

D.6.2 UE policy section management list

The purpose of the UE policy section management list information element is to transfer from the PCF to the UE a list of instructions to be performed at the UE for management of UE policy section stored at the UE.

The UE policy section management list information element is coded as shown in figure D.6.2.1, figure D.6.2.2, figure D.6.2.3, figure D.6.2.4, figure D.6.2.5, figure D.6.2.6, figure D.6.2.7 and table D.6.2.1.

The UE policy section management list information element has a minimum length of 12 octets and a maximum length of 65534 octets.



Figure D.6.2.1: UE policy section management list information element



Figure D.6.2.2: UE policy section management list contents

8	7	6	5	4	3	2	1	
								octet d
	Length of	r UE poi	icy sec	tion ma	nageme	ent subli	St	octet d+1
								octet d+2
	MCC	digit 2			MCC	digit 1		
								octet d+3
	MNC	digit 3			MCC	digit 3		
	MNC	digit 2			MNC	digit 1		octet d+4
								octet d+5
	UE polic	y sectio	n mana	gement	t sublist	content	ts	
								octet y

Figure D.6.2.3: UE policy section management sublist



Figure D.6.2.4: UE policy section management sublist contents



Figure D.6.2.5: Instruction



Figure D.6.2.6: UE policy section contents



Figure D.6.2.7: UE policy part

Table D.6.2.1: UE policy section management list information element

Value part of the UE policy section management list information element (octets 4 to z) The value part of the UE policy section management list information element consists of one or several UE policy section management sublists. UE policy section management sublist: Length of UE policy section management sublist (octets d to d+1) This field contains the binary encoding of the length of the UE policy section management sublist in units of octets. MCC, Mobile country code (octet d+2, and bits 4 to 1 of octet d+3) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A. MNC, Mobile network code (bits 8 to 5 of octet d+3, and octet d+4) The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111". UE policy section management sublist contents (octets d+5 to y) The UE policy section management sublist contents consist of one or several instructions. Instruction: Instruction contents length (octets d+5 to d+6) This field contains the binary encoding of the instruction contents length in units of octets. UPSC (octets d+7 to d+8) This field contains the binary encoding of the UPSC. The value of the UPSC is set by the PCF. UE policy section contents (octets d+9 to k) The UE policy section contents consist of one or several UE policy parts. UE policy part: UE policy part contents length (octets q to q+1) This field contains the binary encoding of the UE policy part contents length in units of octets. UE policy part type (bits 4 to 1 of octet q+2) Bits 4 3 21 0 0 0 0 Reserved 0 0 1 URSP 0 0 0 1 0 ANDSP 0 0 1 1 V2XP 1 0 0 ProSeP Ω All other values are reserved. Bits 8 to 5 of octet q+2 are spare and shall be coded as zero. UE policy part contents This field contains a UE policy part encoded as specified in 3GPP TS 24.526 [19] for the UE policy part type field set to "URSP" or "ANDSP", in 3GPP TS 24.588 [19C] for the UE policy part type field set to "V2XP" and in 3GPP TS 24.555 [19F] for the UE policy part type field set to "ProSeP" respectively.

D.6.3 UE policy section management result

The purpose of the UE policy section management result information element is to transfer from the UE to the PCF information about instructions for UE policy section management which the UE could not execute successfully.

The UE policy section management result information element is coded as shown in figure D.6.3.1, figure D.6.3.2, figure D.6.3.3, figure D.6.3.4, figure D.6.3.5 and table D.6.3.1.

The UE policy section management result information element has a minimum length of 12 octets and a maximum length of 65534 octets.



Figure D.6.3.1: UE policy section management result information element



Figure D.6.3.2: UE policy section management result contents

8	7	6	5	4	3	2	1	
		N	umber	of resul	ts			octet d
				1				
	мсс	digit 2			мсс	digit 1		octet d+1
								octet d+2
	MNC	digit 3			MCC	digit 3		
								octet d+3
	MNC	digit 2			MNC	digit 1		
								octet d+4
U	E policy	section	manag	ement s	subresu	It conte	ents	
			Ū					
								a statu
Ļ								ociet y

Figure D.6.3.3: UE policy section management subresult



Figure D.6.3.4: UE policy section management subresult contents



Figure D.6.3.5: Result

Table D.6.3.1: UE policy section management result information element

Value part of the UE policy section management result information element (octets 4 to z)
The value part of the UE policy section management result information element consists of one or several UE policy section management subresults.
UE policy section management subresult:
Number of results (octet d)
This field contains the binary encoding of number of results included in the UE policy section management subresult.
MCC, Mobile country code (octet d+1, and bits 4 to 1 of octet d+2)
The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.
MNC, Mobile network code (bits 8 to 5 of octet d+2, and octet d+3)
The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111".
UE policy section management subresult contents (octets d+4 to y)
The UE policy section management subresult contents consist of one or several results.
Result (octet f to f+4)
UPSC (octet f to f+1)
This field contains the binary encoding of the UPSC. The value of the UPSC is set by the PCF
Failed instruction order (octets f+2 to f+3)
This field contains the binary encoding of the order of the failed instruction in the UE policy section management sublist.
Cause (octet f+4) Bits 8 7 6 5 4 3 2 1 0 1 1 0 1 1 1 1 Protocol error, unspecified The receiving entity shall treat any other value as 0110 1111, "protocol error, unspecified".

D.6.4 UPSI list

The purpose of the UPSI list information element is to transfer from the UE to the PCF a list of UPSIs.

The UPSI list information element is coded as shown in figure D.6.4.1, figure D.6.4.2, and table D.6.4.1.

The UPSI list information element has a minimum length of 10 octets and a maximum length of 65532 octets.





8	7	6	5	4	3	2	1	
								octet d
		Len	gth of L	JPSI su	ublist			
			-					octet d+1
	MCC	digit 2			MCC	digit 1		octet d+2
	MNC	digit 3			MCC	digit 3		octet d+3
	MNC	digit 2			MNC	digit 1		octet d+4
								octet d+5
			UP	SC				
								octet d+6
								octet d+7*
			UP	SC				
								octet d+8*
								octet d+9*
								octet e*
								octet e+1*
			UP	SC				
								octet e+2*

Figure D.6.4.2: UPSI sublist

Table D.6.4.1: UPSI lis	t information element
-------------------------	-----------------------

MCC, Mobile country code (octet d+2, and bits 4 to 1 of octet d+3) The MCC field is coded as in ITU-T Recommendation E.212 [42], annex A.

MNC, Mobile network code (bits 8 to 5 of octet d+3, and octet d+4)

The coding of this field is the responsibility of each administration but BCD coding shall be used. The MNC shall consist of 2 or 3 digits. If a network operator decides to use only two digits in the MNC, MNC digit 3 shall be coded as "1111".

UPSC (octets d+5 to d+6)

This field contains the binary encoding of the UPSC. The value of the UPSC is set by the PCF.

D.6.5 UE policy classmark

The purpose of the UE policy classmark information element is to provide the network with information about the policy aspects of the UE.

The UE policy classmark information element is coded as shown in figure D.6.5.1 and table D.6.5.1.

The UE policy classmark is a type 4 information element with a minimum length of 3 octets and a maximum length of 5 octets.



Figure D.6.5.1: UE policy classmark information element



Support of ANDSP by the UE (SupportANDSP) (octet 3, bit 1)
Bit
1
0 ANDSP not supported by the UE
1 ANDSP supported by the UE
All other bits in octet 3 to 5 are spare and shall be coded as zero, if the respective octet is included
in the information element.

D.6.6 UE OS Id

The purpose of the UE OS Id information element is to provide the network with information about the OS of the UE.

The UE OS Id information element is coded as shown in figure D.6.6.1 and table D.6.6.1.

The UE OS Id is a type 4 information element with a minimum length of 18 octet and a maximum length of 242 octets.



Figure D.6.6.1: UE OS Id information element

Table	D.6.6.1:	UE OS	Id information	on element

OS Id:

The OS Id is coded as a sequence of a sixteen octet OS Id value field. The OS Id value field is defined as Universally Unique IDentifier (UUID) as specified in IETF RFC 4122 [35A].

D.6.7 UE policy network classmark

The purpose of the UE policy network classmark information element is to provide the UE with information about the policy aspects of the network.

The UE policy network classmark information element is coded as shown in figure D.6.7.1 and table D.6.7.1.

The UE policy network classmark is a type 4 information element with a minimum length of 3 octets and a maximum length of 5 octets.

929



Figure D.6.7.1: UE policy network classmark information element

Table D.6.7.1: UE policy network classmark information element

Non-subscribed SNPN signalled URSP handling indication (NSSUI) (octet 3, bit 1) (see NOTE) Bits

0 UE is allowed to accept URSP signalled by non-subscribed SNPNs

1 UE is not allowed to accept URSP signalled by non-subscribed SNPNs

All other bits in octet 3 to 5 are spare and shall be coded as zero, if the respective octet is included in the information element.

NOTE: Receiving entity shall ignore this bit, if received from the RPLMN which is not the HPLMN or from the RSNPN is not the subscribed SNPN.

D.7 Timers of UE policy delivery service

Timers of UE policy delivery service are shown in table D.7.1.

TIMER NUM.	TIMER VALUE	CAUSE OF START	NORMAL STOP	ON THE 1 st , 2 nd , 3 rd , 4 th EXPIRY			
T3501	NOTE 1	Transmission of MANAGE UE POLICY COMMAND	MANAGE UE POLICY COMMAND COMPLETE or MANAGE UE POLICY COMMAND REJECT message received	Retransmission of MANAGE UE POLICY COMMAND message			
NOTE 1: The value of this timer is network dependent.							

D.8 Handling of unknown, unforeseen, and erroneous UPDS data

D.8.1 General

The procedures specified in the subclause apply to those messages which pass the checks described in this subclause.

This subclause also specifies procedures for the handling of unknown, unforeseen, and erroneous UPDS data by the receiving entity. These procedures are called "error handling procedures", but in addition to providing recovery mechanisms for error situations they define a compatibility mechanism for future extensions of the UPDS.

Subclauses D.8.1 to D.8.8 shall be applied in order of precedence.

Detailed error handling procedures in the network are implementation dependent and may vary from PLMN to PLMN. However, when extensions of UPDS are developed, networks are assumed to have the error handling which is indicated in this subclause as mandatory ("shall") and that is indicated as strongly recommended ("should").

Also, the error handling of the network is only considered as mandatory or strongly recommended when certain thresholds for errors are not reached during a dedicated connection.

For definition of semantical and syntactical errors see 3GPP TS 24.007 [11], subclause 11.4.2.

D.8.2 Message too short or too long

D.8.2.1 Message too short

When a message is received that is too short to contain a complete message type information element, that message shall be ignored, c.f. 3GPP TS 24.007 [11].

D.8.2.2 Message too long

The maximum size of a UE policy delivery service message is 65535 octets.

D.8.3 Unknown or unforeseen procedure transaction identity

D.8.3.1 Procedure transaction identity

The following network procedures shall apply for handling an unknown, erroneous, or unforeseen PTI received in a UPDS message:

- a) In case the network receives a MANAGE UE POLICY COMPLETE message or MANAGE UE POLICY COMMAND REJECT message in which the PTI value is an assigned or unassigned value that does not match any PTI in use, the network shall ignore the UPDS message.
- b) In case the network receives a UPDS message in which the PTI value is a reserved value, the network shall ignore the UPDS message.

The following UE procedures shall apply for handling an unknown, erroneous, or unforeseen PTI received in a UPDS message:

a) In case the UE receives a UPDS message in which the PTI value is a reserved value, the UE shall ignore the UPDS message.

D.8.4 Unknown or unforeseen message type

If the UE or the network receives a UPDS message with message type not defined for the UPDS or not implemented by the receiver, it shall ignore the UPDS message.

NOTE: A message type not defined for the UPDS in the given direction is regarded by the receiver as a message type not defined for the UPDS, see 3GPP TS 24.007 [11].

If the UE receives a message not compatible with the UPDS state, the UE shall ignore the UPDS message.

If the network receives a message not compatible with the UPDS state, the network actions are implementation dependent.

D.8.5 Non-semantical mandatory information element errors

D.8.5.1 Common procedures

When on receipt of a message,

- a) an "imperative message part" error; or
- b) a "missing mandatory IE" error

is diagnosed or when a message containing:

- a) a syntactically incorrect mandatory IE;
- b) an IE unknown in the message, but encoded as "comprehension required" (see 3GPP TS 24.007 [11]); or
- c) an out of sequence IE encoded as "comprehension required" (see 3GPP TS 24.007 [11]) is received,

the UE shall ignore the UPDS message;

the network shall proceed as follows:

the network shall:

- 1) try to treat the message (the exact further actions are implementation dependent); or
- 2) ignore the message.

D.8.6 Unknown and unforeseen IEs in the non-imperative message part

D.8.6.1 IEIs unknown in the message

The UE shall ignore all IEs unknown in a message which are not encoded as "comprehension required" (see 3GPP TS 24.007 [11]).

The network shall take the same approach.

D.8.6.2 Out of sequence IEs

The UE shall ignore all out of sequence IEs in a message which are not encoded as "comprehension required" (see 3GPP TS 24.007 [11]).

The network should take the same approach.

D.8.6.3 Repeated IEs

If an information element with format T, TV, TLV, or TLV-E is repeated in a message in which repetition of the information element is not specified in subclause D.5, the UE shall handle only the contents of the information element appearing first and shall ignore all subsequent repetitions of the information element. When repetition of information elements is specified, the UE shall handle only the contents of specified repeated information elements. If the limit on repetition of information elements is exceeded, the UE shall handle the contents of information elements appearing first up to the limit of repetitions and shall ignore all subsequent repetitions of the information element.

The network should follow the same procedures.

D.8.7 Non-imperative message part errors

This category includes:

- a) syntactically incorrect optional IEs; and
- b) conditional IE errors.

D.8.7.1 Syntactically incorrect optional IEs

The UE shall treat all optional IEs that are syntactically incorrect in a message as not present in the message.

The network shall take the same approach.

D.8.7.2 Conditional IE errors

When upon receipt of a UPDS message the UE diagnoses a "missing conditional IE" error or an "unexpected conditional IE" error, or when it receives a UPDS message containing at least one syntactically incorrect conditional IE, the UE shall ignore the message.

When the network receives a message and diagnoses a "missing conditional IE" error or an "unexpected conditional IE" error or when it receives a message containing at least one syntactically incorrect conditional IE, the network shall either:

- a) try to treat the message (the exact further actions are implementation dependent); or
- b) ignore the message.

D.8.8 Messages with semantically incorrect contents

When a message with semantically incorrect contents is received, the UE shall perform the foreseen reactions of the procedural part of subclauses D.2. If, however no such reactions are specified, the UE shall ignore the message.

The network should follow the same procedure.

Annex E (informative): Void

Annex F (informative): Change history

Change history								
Date	Meeting	Tdoc	CR	Rev	Cat	Subject/Comment	New version	
2017-10	CT1#106	C1-174182				Draft skeleton provided by the rapporteur.	0.0.0	
2017-11	CT1#106					Implementing the following p-CRs agreed by CT1: (
2017-12	CT1#107					Implementing the following p-CRs agreed by CT1:	0.2.0	
						C1-175098, C1-175313. Corrections done by the rapporteur.		
2017-12	CT1 e- mail					Editorial corrections.	0.2.1	
0047.40	review					Do introduction of table in out-slower 0.0.004		
2017-12	mail					Re-Introduction of table in subclause 8.2.23.1	0.2.2	
0040.00	review					Inclumenting the following a CDs spread by CT4.	0.2.0	
2018-02	C11#108					Implementing the following p-CRs agreed by C11: C1-180663, C1-180224, C1-180046, C1-180437, C1-180438, C1-180448, C1-180307, C1-180211, C1-180316, C1-180221, C1-180451, C1-180339, C1-180361, C1-180148, C1-180415, C1-180451, C1-180453, C1-180455, C1-180459, C1-180482, C1-180483, C1-180484, C1-180619, C1-180620, C1-180623, C1-180624, C1-180627, C1-180628, C1-180664, C1-180665, C1-180668, C1-180672, C1-180673, C1-180679, C1-180680, C1-180684, C1-18077, C1-180721, C1-180725, C1-180736, C1-180737, C1-180738, C1-180739, C1-180740, C1-180741, C1-180750, C1-180751, C1-180013, C1-180311, C1-180312, C1-180197, C1-180313, C1-180283, C1-180037, C1-180041, C1-180464, C1-1806465, C1-180666, C1-180669, C1-180645, C1-180646, C1-180648, C1-180688, C1-180689, C1-180645, C1-180645, C1-180640, C1-180669, C1-180632, C1-180633, C1-180734, C1-180735, C1-180746, C1-180731, C1-180732, C1-18015, C1-180035, C1-180746, C1-180209, C1-180640, C1-180015, C1-180035, C1-180746, C1-180622, C1-180647, C1-180647, C1-180740, C1-180669, C1-180632, C1-180637, C1-1800400, C1-180640, C1-180669, C1-180630, C1-1800400, C1-180669, C1-1806421, C1-1806401, C1-180015, C1-1800400, C1-180664, C1-180669, C1-180647, C1-180015, C1-180747, C1-180746, C1-180704, C1-180740, C1-180162, C1-180190, C1-180664, C1-180605, C1-180666, C1-180641, C1-180644, C1-180664, C1-180605, C1-180666, C1-180641, C1-180644, C1-180646, C1-180605, C1-1806604, C1-180647, C1-180747, C1-180755, C1-180704, C1-180719, C1-180722, C1-180747, C1-180755, C1-180756	0.3.0	
2018-02	CT1 e-					Corrections done by the rapporteur. Resolution of collision among C1-180679. C1-180721 and C1-	0.3.1	
	mail review					180740. Resolution of collision among C1-180605, C1-180616 and C1- 180704. Re-implementation of parts of C1-180035, C1-180488, C1- 180605, C1-180606, C1-180729 and C1-180734 as some of the proposed changes were not implemented correctly in the previous version. Implementation of C1-180646 which was missed. Editorial corrections. Corrections done by the rapporteur.		
2018-03	CT1#109					Implementing the following p-CRs agreed by CT1: C1-181362, C1-181377, C1-181456, C1-181457, C1-181703, C1-181748, C1-181462, C1-181786, C1-181457, C1-181269, C1-181278, C1-181307, C1-181180, C1-181279, C1-181280, C1-181281, C1-181354, C1-181283, C1-181284, C1-181287, C1-181305, C1-181352, C1-181364, C1-181284, C1-181366, C1-181399, C1-181466, C1-181467, C1-181468, C1-181366, C1-181399, C1-181466, C1-181467, C1-181468, C1-181468, C1-181629, C1-181633, C1-181661, C1-181663, C1-181668, C1-181668, C1-181695, C1-181664, C1-181682, C1-181668, C1-181668, C1-181695, C1-181696, C1-181682, C1-181683, C1-181715, C1-181716, C1-181717, C1-181718, C1-181733, C1-181734, C1-181735, C1-181736, C1-181737, C1-181738, C1-181755, C1-181740, C1-181741, C1-181747, C1-181782, C1-181765, C1-181740, C1-181771, C1-181781, C1-181782, C1-181765, C1-181726, C1-181751, C1-181781, C1-181782, C1-181765, C1-181726, C1-181751, C1-181651, C1-181652, C1-181678, C1-181726, C1-181751, C1-181651, C1-181652, C1-181678, C1-181727, C1-181751, C1-181651, C1-181652, C1-181678, C1-181726, C1-181761, C1-181671, C1-181784, C1-181770, C1-181760, C1-181671, C1-181651, C1-181652, C1-181678, C1-181726, C1-181761, C1-181671, C1-181784, C1-181770, C1-181726, C1-181671, C1-181651, C1-181652, C1-181678, C1-181727, C1-181605, C1-181671, C1-181652, C1-181674, C1-181728, C1-181605, C1-181679, C1-181652, C1-181674, C1-181728, C1-181613, C1-181699, C1-181645, C1-181674, C1-181675, C1-1816177, C1-181679, C1-181600, C1-181700, C1-181618, C1-181613, C1-181615, C1-181600, C1-181700, C1-181728, C1-181613, C1-181679, C1-181700, C1-181731, C1-181732 Corrections done by the rapporteur.		
2018-03	CT1 e-			Re-implementation of C1-181168 and C1-181307.	0.4.1			
---------	-------------------------------------	-----------	--	--	-------			
	mail			Re-implementation of C1-181656 and C1-181606 so that C1-				
	review			181656 is implemented first.				
				Reventing to the old title.				
				Editorial corrections of some of the implemented p-CRS as well				
				Corrections done by the rapporteur				
2018-03	CT#79	CP-180101		Version 1.0.0 created for presentation to TSG CT#79 for	100			
_0.0 00	00	0		information.				
2018-05	CT1#110			Implementing the following p-CRs agreed by CT1:	1.1.0			
				C1-182219, C1-182493, C1-182496, C1-182202, C1-182497,				
				C1-182053, C1-182311, C1-182019, C1-182359, C1-182360,				
				C1-182361, C1-182358, C1-182305, C1-182306, C1-182354, C1-182455, C1-182455, C1-182455, C1-182450, C1-182404				
				C1-182600 C1-182601 C1-182605 C1-182606 C1-182607				
				C1-182608, C1-182609, C1-182610, C1-182614, C1-182615				
				C1-182621, C1-182662, C1-182664, C1-182665, C1-182708,				
				C1-182728, C1-182730, C1-182733, C1-182724, C1-182757,				
				C1-182759, C1-182760, C1-182768, C1-182772, C1-182775,				
				C1-182786, C1-182787, C1-182791, C1-182831, C1-182832,				
				C1-182833, C1-182834, C1-182835, C1-183836, C1-182838,				
				C1-182840, C1-182844, C1-182067, C1-182073, C1-182303,				
				C1-182647 C1-182648 C1-182650 C1-182651 C1-182657				
				C1-182659, C1-182660, C1-182741, C1-182742, C1-182761,				
				C1-182762, C1-182763, C1-182764, C1-182765, C1-182774,				
				C1-182789, C1-182789, C1-182815, C1-182845, C1-182797,				
				C1-182232, C1-182230, C1-182666, C1-182667, C1-182671,				
				C1-182673, C1-182677, C1-182800, C1-182824, C1-182710,				
				C1-182072, C1-182078, C1-182174, C1-182190, C1-182456,				
				C1-182636, C1-182637, C1-182638, C1-182639, C1-182726, C1-182729, C1-182747, C1-182749, C1-182766, C1-182767				
				C1-182841 C1-182847 C1-182043 C1-182057 C1-182260				
				C1-182044, C1-182617, C1-182618, C1-182619, C1-182620,				
				C1-182622, C1-182623, C1-182624, C1-182627, C1-182628,				
				C1-182629, C1-182802, C1-182808, C1-182345, C1-182461,				
				C1-182630				
0040.05	074.4			Corrections done by the rapporteur.	4 4 4			
2018-05	CT1 e-			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182407, C1-182067 and C1-	1.1.1			
2018-05	CT1 e- mail review			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some	1.1.1			
2018-05	CT1 e- mail review			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts.	1.1.1			
2018-05	CT1 e- mail review			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur.	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1:	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-180409, C1-180401, C1-180402, C1-180404, C1	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183207, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183275, C1-183245, C1-183277, C1-183277,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183243, C1-183146, C1-183297, C1-1832452, C1-183443, C1-183146, C1-183297, C1-183445, C1-183443, C1-183142, C1-183443, C1-183144, C1-183260, C1-183442, C1-183443, C1-183142, C1-183443, C1-183144, C1-183443, C1-183144, C1-183443, C1-183144, C1-183443, C1-183443, C1-183443, C1-183444, C1-	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223.	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182019, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183413, C1-183467, C1-183530, C1-183532,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182019, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183277, C1-183415, C1-183207, C1-183273, C1-183276, C1-1832277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183413, C1-183467, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183539, C1-183534, C1-18354, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183539, C1-183534, C1-183544, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183539, C1-183534, C1-183544, C1-183555, C1-183530, C1-183532, C1-183534, C1-183544, C1-183555, C1-183530, C1-183532, C1-183534, C1-183544, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183555, C1-183530, C1-183532, C1-183533, C1-183534, C1-183535, C1-183530, C1-183532, C1-183534, C1-183544, C1-183545, C1-1	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183413, C1-183467, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183715, C1-183716, C1-183710, C1-183718, C1-183720, C1-183714, C1-183720, C1 492720, C1 492744, C1 492744, C1-183720, C1 492720, C1 492744, C1 492744, C1 492727, C1 492744, C1 492744	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183145, C1-183207, C1-183415, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183205, C1-183142, C1-183151, C1-183154, C1-183260, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183534, C1-183467, C1-183530, C1-183532, C1-183721, C1-18374, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183749, C1-183741, C1-183744, C1-183745, C1-183748, C1-183749, C1-183741, C1-183744, C1-183745, C1-183748, C1-183749, C1-183740, C1-183744, C1-183745, C1-183748, C1-183749, C1-183740, C1-183744, C1-183745, C1-183748, C1-183749, C1-183741, C1-183744, C1-183745, C1-183748, C1-183749, C1-183740, C1-183744, C1-183745, C1-183748, C1-183749, C1-183745, C1-183744, C1-183745, C1-183748, C1-183749, C1-183740, C1-183744, C1-183745, C1-183748, C1-183749, C1-183744, C1-183745, C1-183749, C1-183749, C1-183744, C1-183745, C1-183748, C1-183749, C1-183744, C1-183745, C1-183748, C1-183749, C1-183744, C1-183745, C1-183748, C1-183749, C1-183744, C1-183745, C1-183748, C1-183749, C1-183744, C1-183745, C1-183744, C1-183	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183277, C1-183812, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183276, C1-183205, C1-183142, C1-183151, C1-183154, C1-183260, C1-183223, C1-183314, C1-183278, C1-183367, C1-183205, C1-183223, C1-183399, C1-183534, C1-183467, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183739, C1-183741, C1-183744, C1-183774, C1-183775, C1-183779, C1-1837750, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183775, C1-183779, C1-183780, C1-183778, C1-183779, C1-183778, C1-183780, C1-183778, C1-183779, C1-183778, C1-183781, C1-183778, C1-183779, C1-183778, C1-183781, C1-1	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183227, C1-183415, C1-183143, C1-183154, C1-183197, C1-183205, C1-183142, C1-183151, C1-183154, C1-183260, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183534, C1-183467, C1-183530, C1-183532, C1-183735, C1-183716, C1-183717, C1-183718, C1-183539, C1-183721, C1-183776, C1-183741, C1-183744, C1-183745, C1-183748, C1-183749, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183799, C1-183725, C1-183740, C1-183781, C1-183740, C1-183775, C1-183779, C1-183780, C1-183781, C1-183740, C1-183775, C1-183779, C1-183780, C1-183781, C1-183780, C1-183720, C1-183781, C1-183790, C1-183720, C1-183781, C1-183740, C1-183775, C1-183740, C1-183780, C1-183781, C1-183780, C1-183775, C1-183779, C1-183780, C1-183781, C1-183809, C1-183822, C1-183824, C1-183825, C1-183825, C1-183825, C1-183826, C1-183826, C1-183826, C1-183825, C1-183826,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183145, C1-183527, C1-183812, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183205, C1-183223, C1-183151, C1-183154, C1-183260, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183534, C1-183555, C1-183538, C1-183532, C1-183735, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183749, C1-183781, C1-183741, C1-183745, C1-183748, C1-183749, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183740, C1-183822, C1-183824, C1-183825, C1-183826, C1-183845, C1-183858, C1-183761, C1-183147, C1-183237,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183115, C1-183527, C1-183812, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183205, C1-183223, C1-183151, C1-183154, C1-183260, C1-183223, C1-183314, C1-183278, C1-183367, C1-183279, C1-183381, C1-183399, C1-183743, C1-183467, C1-183530, C1-183532, C1-183755, C1-183716, C1-183777, C1-183718, C1-183539, C1-183745, C1-183776, C1-183779, C1-183781, C1-183774, C1-183745, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183822, C1-183824, C1-183825, C1-183781, C1-183784, C1-183822, C1-183779, C1-183780, C1-183781, C1-183784, C1-183822, C1-183779, C1-183780, C1-183781, C1-183784, C1-183779, C1-183780, C1-183781, C1-183784, C1-183779, C1-183780, C1-183781, C1-183784, C1-183779, C1-183780, C1-183781, C1-183784, C1-183779, C1-183780, C1-183781, C1-1837809, C1-183822, C1-183824, C1-183825, C1-183826, C1-183845, C1-18353, C1-183778, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183780, C1-183781, C1-183845, C1-18353, C1-183780, C1-183781, C1-183845, C1-18353, C1-183780, C1-183781, C1-183845, C1-18353, C1-183378, C1-183877, C1-183329, C1-183353, C1-183378, C1-183877, C1-1833840, C1-183353, C1-183378, C1-183877, C1-1833929, C1-183353, C1-183378, C1-183877, C1-1834001, C1-183329, C1-183353, C1-183378, C1-183877, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001, C1-1834001,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183115, C1-183527, C1-183812, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183205, C1-183223, C1-183151, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183530, C1-183532, C1-183533, C1-18354, C1-183535, C1-183538, C1-183539, C1-183755, C1-18376, C1-183717, C1-183718, C1-183720, C1-183721, C1-18377, C1-183779, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183822, C1-183824, C1-183780, C1-183781, C1-183845, C1-18353, C1-183761, C1-183740, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183799, C1-183737, C1-183779, C1-183780, C1-183781, C1-183790, C1-183737, C1-183779, C1-183780, C1-183781, C1-183790, C1-183737, C1-183779, C1-183780, C1-183781, C1-183790, C1-183822, C1-183824, C1-183825, C1-183825, C1-183845, C1-183533, C1-1835761, C1-183781, C1-183408, C1-183533, C1-1835761, C1-183787, C1-183408, C1-183409, C1-183576, C1-183576, C1-183587, C1-183408, C1-183409, C1-183576, C1-183576, C1-183543, C1-183533, C1-1835761, C1-183577, C1-183540, C1-183533, C1-1835761, C1-1835761, C1-183408, C1-183409, C1-183576, C1-183576, C1-183540, C1-183533, C1-183576, C1-183543, C1-1835408, C1-183543, C1-183541, C1-183543, C1-1835408, C1-183543, C1-183541, C1-183543, C1-1835408, C1-183543, C1-183541, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543, C1-183543	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183205, C1-1832381, C1-183799, C1-183748, C1-183535, C1-183538, C1-183539, C1-183755, C1-183746, C1-183717, C1-183718, C1-183720, C1-183741, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183845, C1-18358, C1-1833761, C1-183787, C1-183237, C1-183329, C1-183353, C1-183378, C1-183387, C1-183237, C1-183408, C1-183353, C1-183378, C1-183377, C1-183780, C1-183408, C1-183726, C1-183776, C1-1837761, C1-183781, C1-183408, C1-183726, C1-183776, C1-183777, C1-183787, C1-183408, C1-183726, C1-183781, C1-183757, C1-183543, C1-183545, C1-183726, C1-183761, C1-183757, C1-183543, C1-183545, C1-183726, C1-183765, C1-183757, C1-183543, C1-183545, C1-183726, C1-183765, C1-183757, C1-183543, C1-183545, C1-183726, C1-183765, C1-183757, C1-183543, C1-183545, C1-183766, C1-183765, C1-183758, C1-183545, C1-183766, C1-183765, C1-183758, C1-183545, C1-183766, C1-183765, C1-183758, C1-183545, C1-183766, C1-183756, C1-183756, C1-183758, C1-183545, C1-183766, C1-183765, C1-1	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183205, C1-183381, C1-183799, C1-183748, C1-183555, C1-183538, C1-183539, C1-183755, C1-183746, C1-183717, C1-183718, C1-183720, C1-183745, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183845, C1-183858, C1-1833761, C1-183780, C1-183781, C1-183408, C1-183353, C1-183378, C1-183377, C1-183783, C1-183408, C1-183726, C1-183378, C1-183787, C1-183783, C1-183408, C1-183726, C1-183779, C1-183780, C1-183781, C1-183408, C1-183726, C1-183776, C1-183779, C1-183780, C1-183543, C1-183408, C1-183726, C1-183776, C1-183796, C1-183778, C1-183779, C1-183760, C1-183778, C1-183779, C1-183779, C1-183778, C1-183408, C1-183409, C1-183779, C1-183780, C1-183543, C1-183408, C1-183720, C1-183780, C1-183780, C1-183408, C1-183720, C1-183780, C1-183790, C1-183778, C1-183408, C1-183720, C1-183780, C1-183790, C1-183543, C1-183408, C1-183720, C1-183780, C1-183790, C1-183790, C1-183408, C1-183720, C1-183780, C1-183790, C1-183790, C1-183408, C1-183720, C1-183780, C1-183790, C1-183790, C1-183408, C1-183720, C1-183790, C1-183790, C1-183897, C1-183790, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183790, C1-183897, C1-183897, C1-183897, C1-183897, C1-183897, C1-1838	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183267, C1-183205, C1-183223, C1-183314, C1-183278, C1-183367, C1-183205, C1-183381, C1-183799, C1-183748, C1-18355, C1-183538, C1-183532, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183739, C1-183780, C1-183751, C1-183745, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183275, C1-183779, C1-183780, C1-183781, C1-18379, C1-183753, C1-183779, C1-183780, C1-183781, C1-183774, C1-18375, C1-183779, C1-183780, C1-183781, C1-183774, C1-18375, C1-183779, C1-183780, C1-183781, C1-183845, C1-18353, C1-183779, C1-183780, C1-183781, C1-183408, C1-183726, C1-183776, C1-183757, C1-183783, C1-183408, C1-183726, C1-183776, C1-183757, C1-183758, C1-183743, C1-183760, C1-183756, C1-183757, C1-183758, C1-183743, C1-183764, C1-183776, C1-183756, C1-183757, C1-183543, C1-183743, C1-183764, C1-183766, C1-183757, C1-183758, C1-183759, C1-183764, C1-183756, C1-183756, C1-183758, C1-183759, C1-183764, C1-183847, C1-183548, C1-183847, C1-183731, C1-183764, C1-183577, C1-183758, C1-183731, C1-183764, C1-183578, C1-183758, C1-183731, C1-183764, C1-183847, C1-183585, C1-183758, C1-183731, C1-183764,	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183267, C1-183205, C1-183223, C1-183314, C1-183154, C1-183267, C1-183205, C1-183381, C1-183399, C1-183413, C1-183467, C1-183205, C1-183381, C1-183730, C1-183748, C1-183747, C1-183538, C1-183539, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183749, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183755, C1-183761, C1-183780, C1-183781, C1-183409, C1-183533, C1-183538, C1-183542, C1-183545, C1-183455, C1-183533, C1-183761, C1-183780, C1-183781, C1-183408, C1-183499, C1-183761, C1-183757, C1-183543, C1-183408, C1-183499, C1-183561, C1-183757, C1-183543, C1-183740, C1-183726, C1-183760, C1-183757, C1-183758, C1-183741, C1-183764, C1-183766, C1-183756, C1-183757, C1-183741, C1-183744, C1-183576, C1-183757, C1-183543, C1-183408, C1-183499, C1-183561, C1-183542, C1-183543, C1-183741, C1-183744, C1-183543, C1-183543, C1-183545, C1-183764, C1-183576, C1-183756, C1-183758, C1-183731, C1-183784, C1-183578, C1-183756, C1-183758, C1-183731, C1-183784, C1-183578, C1-183585, C1-183798, C1-183741, C1-183784, C1-183578, C1-183585, C1-183583, C1-183761, C1-183764, C1-183578, C1-183563, C1-183798, C1-183661, C1-183247, C1-183578, C1-183563, C1-183798.	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183205, C1-183381, C1-183799, C1-183740, C1-183717, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183718, C1-183729, C1-183721, C1-183737, C1-183749, C1-183781, C1-183744, C1-183774, C1-183775, C1-183779, C1-183780, C1-183751, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183845, C1-183858, C1-183761, C1-183787, C1-183781, C1-183409, C1-183735, C1-183761, C1-183787, C1-183783, C1-183409, C1-183736, C1-183761, C1-183787, C1-183781, C1-183408, C1-183726, C1-183761, C1-183757, C1-183783, C1-183408, C1-183499, C1-183761, C1-183757, C1-183783, C1-183745, C1-183726, C1-183761, C1-183757, C1-183783, C1-183407, C1-183766, C1-183761, C1-183757, C1-183783, C1-183408, C1-183499, C1-183578, C1-183542, C1-183543, C1-183741, C1-183744, C1-183543, C1-183543, C1-183745, C1-183766, C1-183756, C1-183757, C1-183758, C1-183747, C1-183766, C1-183756, C1-183796, C1-183802, C1-183845, C1-183746, C1-183765, C1-183779, C1-183796, C1-183827, C1-183827, C1-183784, C1-183578, C1-183585, C1-183798, C1-183731, C1-183784, C1-183578, C1-183563, C1-183798, C1-183861, C1-183247, C1-183578, C1-183563, C1-183798, C1-18361, C1-183247, C1-183576, C1-183563, C1-183798, C1-183194, C1-183238, C1-183576, C1-183563, C1-183798, C1-183194, C1-183238, C1-183576, C1-183563, C1-183798, C1-183194, C1-183238, C1-183576, C1-183563, C1-183798, C1-18319	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183225, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183500, C1-183381, C1-183715, C1-183740, C1-183717, C1-183530, C1-183532, C1-183715, C1-183737, C1-183740, C1-183741, C1-183729, C1-183721, C1-183737, C1-183749, C1-183780, C1-183751, C1-183774, C1-183775, C1-183779, C1-183780, C1-183781, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183845, C1-183858, C1-183761, C1-183787, C1-1838401, C1-183409, C1-183726, C1-183776, C1-183757, C1-183781, C1-183408, C1-183858, C1-183761, C1-183757, C1-183781, C1-183408, C1-183726, C1-183776, C1-183757, C1-183781, C1-183407, C1-183766, C1-183776, C1-183757, C1-183781, C1-183408, C1-183858, C1-183761, C1-183757, C1-183781, C1-183407, C1-183726, C1-183750, C1-183757, C1-183781, C1-183408, C1-183499, C1-183761, C1-183757, C1-183783, C1-183845, C1-183726, C1-183761, C1-183757, C1-183543, C1-183845, C1-183766, C1-183757, C1-183798, C1-183827, C1-183784, C1-183545, C1-183798, C1-183845, C1-183784, C1-183545, C1-183798, C1-183799, C1-183766, C1-183795, C1-183798, C1-183827, C1-183784, C1-183578, C1-183583, C1-183798, C1-183827, C1-183784, C1-183578, C1-183583, C1-183798, C1-183827, C1-183247, C1-183578, C1-183563, C1-183798, C1-183799, C1-183798, C1-183578, C1-183563, C1-183798, C1-183799, C1-183798, C1-183562, C1-183563, C1-183798, C1-183799, C1-183798, C1-183579, C1-183763, C1-183798, C1-183799, C1-183798, C1-183579, C1-183763, C1-183	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183500, C1-183532, C1-183715, C1-183737, C1-183767, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183538, C1-183539, C1-183721, C1-183737, C1-183749, C1-183741, C1-183744, C1-183745, C1-183775, C1-183749, C1-183750, C1-183751, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183745, C1-183755, C1-183779, C1-183780, C1-183781, C1-183745, C1-183755, C1-183779, C1-183780, C1-183781, C1-183455, C1-183755, C1-183761, C1-183787, C1-183826, C1-183408, C1-183499, C1-183776, C1-183757, C1-183757, C1-183727, C1-183766, C1-183776, C1-183757, C1-183781, C1-183408, C1-183858, C1-183761, C1-183757, C1-183781, C1-183408, C1-183499, C1-183750, C1-183757, C1-183781, C1-183645, C1-183766, C1-183757, C1-183796, C1-183827, C1-183827, C1-183766, C1-183757, C1-183798, C1-183827, C1-183766, C1-183795, C1-183798, C1-183827, C1-183766, C1-183795, C1-183798, C1-183799, C1-183766, C1-183795, C1-183798, C1-183709, C1-183768, C1-183795, C1-183563, C1-183831, C1-183661, C1-183247, C1-183578, C1-183563, C1-183798, C1-183706, C1-183707, C1-183709, C1-183763, C1-183798, C1-183706, C1-183707, C1-183709, C1-183763, C1-183766, C1-183767, C1-183768, C1-183769, C1-183763, C1-183771, C1-183767, C1-183768, C1-183769, C1-183763, C1-183766, C1-183767, C1-183768, C1-183769, C1-1837763, C1-183766, C1-183767, C1-183768, C1-183769, C1-183763, C1-183	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183151, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183505, C1-183532, C1-183715, C1-183776, C1-183767, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183530, C1-183532, C1-183721, C1-183737, C1-183749, C1-183741, C1-183720, C1-183745, C1-183748, C1-183749, C1-183750, C1-183751, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183745, C1-183755, C1-183779, C1-183780, C1-183781, C1-183845, C1-183858, C1-183761, C1-183787, C1-183826, C1-183409, C1-183499, C1-183761, C1-183757, C1-183781, C1-183408, C1-183499, C1-183776, C1-183757, C1-183758, C1-183759, C1-183766, C1-183775, C1-183779, C1-183781, C1-183408, C1-183499, C1-183761, C1-183757, C1-183758, C1-183845, C1-183766, C1-183779, C1-183787, C1-183401, C1-183408, C1-183499, C1-183578, C1-183543, C1-183543, C1-183827, C1-183766, C1-183756, C1-183757, C1-183758, C1-183827, C1-183766, C1-183761, C1-183787, C1-183843, C1-183608, C1-183499, C1-183578, C1-183543, C1-183543, C1-18361, C1-183760, C1-183761, C1-183779, C1-183763, C1-183767, C1-183766, C1-183765, C1-183763, C1-183779, C1-183767, C1-183768, C1-183769, C1-183763, C1-183779, C1-183767, C1-183768, C1-183769, C1-183763, C1-183778, C1-183767, C1-183768, C1-183769, C1-183770, C1-183778, C1-183767, C1-183768, C1-183769, C1-183778, C1-183767, C1-183768, C1-183769, C1-183778, C1-183767, C1-183768, C1-183769, C1-183778, C1-183779, C1-183778, C1-183778, C1-1837	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183143, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183505, C1-183532, C1-183399, C1-183413, C1-183467, C1-183530, C1-183532, C1-183715, C1-183776, C1-183717, C1-183530, C1-183532, C1-183721, C1-183737, C1-183739, C1-183730, C1-183720, C1-183721, C1-183737, C1-183749, C1-183741, C1-183744, C1-183745, C1-183755, C1-183779, C1-183780, C1-183751, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183809, C1-183822, C1-183824, C1-183825, C1-183826, C1-183845, C1-183858, C1-183761, C1-183787, C1-183781, C1-183408, C1-183499, C1-183776, C1-183757, C1-183758, C1-183759, C1-183766, C1-183756, C1-183757, C1-183758, C1-183759, C1-183762, C1-183756, C1-183757, C1-183758, C1-183827, C1-183766, C1-183757, C1-183784, C1-183827, C1-183766, C1-183757, C1-183798, C1-183827, C1-183766, C1-183763, C1-183779, C1-183827, C1-183766, C1-183763, C1-183779, C1-183827, C1-183768, C1-183769, C1-183763, C1-183798, C1-183767, C1-183768, C1-183769, C1-183763, C1-183771, C1-183769, C1-183768, C1-183769, C1-183763, C1-183771, C1-183769, C1-183768, C1-183769, C1-183763, C1-183778, C1-183769, C1-183709, C1-183769, C1-183770, C1-183778, C1-183769, C1-183799, C1-183769, C1-183770, C1-1837	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-1832115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183146, C1-183197, C1-183260, C1-183142, C1-183143, C1-183154, C1-183255, C1-183205, C1-183223, C1-183314, C1-183278, C1-18367, C1-183505, C1-183532, C1-183399, C1-183716, C1-183717, C1-183530, C1-183532, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183739, C1-183741, C1-183744, C1-183745, C1-183748, C1-183749, C1-183750, C1-183751, C1-183774, C1-183755, C1-183779, C1-183780, C1-183781, C1-183809, C1-183822, C1-183824, C1-183780, C1-183781, C1-183845, C1-183858, C1-183761, C1-183147, C1-1832237, C1-183408, C1-183499, C1-1837761, C1-183747, C1-183543, C1-183408, C1-183499, C1-183776, C1-183757, C1-183758, C1-183759, C1-183766, C1-183775, C1-183779, C1-183781, C1-183408, C1-183499, C1-183578, C1-183542, C1-183543, C1-183455, C1-183766, C1-183761, C1-183747, C1-183543, C1-183408, C1-183499, C1-183578, C1-183542, C1-183543, C1-183408, C1-183499, C1-183578, C1-183543, C1-183543, C1-18361, C1-183766, C1-183765, C1-183779, C1-1837798, C1-183706, C1-183768, C1-183769, C1-183763, C1-183798, C1-183767, C1-183768, C1-183769, C1-183763, C1-183779, C1-183764, C1-183768, C1-183769, C1-1837763, C1-183778, C1-1837648, C1-183769, C1-183765, C1-1837763, C1-1837788, C1-1837648, C1-183769, C1-183769, C1-1837763, C1-1837788, C1-1834548, C1-183799, C1-183850, C1-	1.1.1			
2018-05	CT1 e- mail review CT1#111			Corrections done by the rapporteur. Re-implementation of C1-182768, C1-182841, C1-182841, C1-182619, C1-182665, C1-182497, C1-182067 and C1- 182078 to correct some editorials as well as adding some missing parts. Corrections done by the rapporteur. Implementing the following p-CRs agreed by CT1: C1-183268, C1-183109, C1-183281, C1-183517, C1-183518, C1-183519, C1-183791, C1-183115, C1-183527, C1-183812, C1-183813, C1-183141, C1-183148, C1-183406, C1-183070, C1-183207, C1-183273, C1-183276, C1-183277, C1-183415, C1-183143, C1-183154, C1-183197, C1-183260, C1-183142, C1-183143, C1-183154, C1-183276, C1-183205, C1-183281, C1-183399, C1-183278, C1-183367, C1-183205, C1-183281, C1-183399, C1-183413, C1-183467, C1-183530, C1-183532, C1-183533, C1-183534, C1-183535, C1-183538, C1-183539, C1-183715, C1-183716, C1-183717, C1-183718, C1-183720, C1-183721, C1-183737, C1-183739, C1-183781, C1-183744, C1-183749, C1-183775, C1-183761, C1-183750, C1-183751, C1-183774, C1-183775, C1-183761, C1-183780, C1-183774, C1-183809, C1-183822, C1-183824, C1-183780, C1-183781, C1-183845, C1-183726, C1-183761, C1-183147, C1-183743, C1-183845, C1-183726, C1-183776, C1-183757, C1-183758, C1-183845, C1-183726, C1-183776, C1-183757, C1-183758, C1-183845, C1-183726, C1-183756, C1-183757, C1-183758, C1-183845, C1-183726, C1-183756, C1-183757, C1-183758, C1-183845, C1-183744, C1-183576, C1-183757, C1-183758, C1-183845, C1-183762, C1-183756, C1-183757, C1-183758, C1-183769, C1-183762, C1-183756, C1-183757, C1-183758, C1-183761, C1-183247, C1-183562, C1-183763, C1-183758, C1-183761, C1-183247, C1-183562, C1-183763, C1-183758, C1-183767, C1-183762, C1-183766, C1-183763, C1-183771, C1-183761, C1-183247, C1-183562, C1-183763, C1-183783, C1-183763, C1-183707, C1-183769, C1-183770, C1-183778, C1-183763, C1-183709, C1-183769, C1-183770, C1-183771, C1-183768, C1-183799, C1-183769, C1-183770, C1-183771, C1-183458, C1-183499, C1-183560, C1-183770, C1-183771, C1-183458, C1-183510, C1-183805, C1-183761, C1-183832, C1-183	1.1.1			

2018-06	CT1 e-					Re-implementation of C1-183535, C1-183813, C1-183408,	1.2.1
	mail					U1-183766 and U1-183831.	
	leview					Editorial corrections of some of the implemented p-CRs.	
						Corrections done by the rapporteur.	
2018-06	CT#80	CP-181094				Version 2.0.0 created for presentation to TSG CT#80 for	2.0.0
						approval.	
2018-06	CT#80					Version 15.0.0 created after approval	15.0.0
2018-09	CT#81	CP-182139	0001		F	Replace unknown "registration update accept"	15.1.0
2018-09	CT#81	CP-182142	0003	2	F	Pass (Extended) Emergency Number List to upper layers	15.1.0
2018-09	CT#81	CP-182132	0004	1	F	Correcting access selection for SMS over NAS	15.1.0
2018-09	C1#81	CP-182138	0006	1	Г	Referring to the correct bits for SMS over INAS during the	15.1.0
2018-09	CT#81	CP-182140	0007	1	F	Setting and checking 5GS update status	15.1.0
2018-09	CT#81	CP-182132	0008	1	F	Clarifications on MICO indication	15.1.0
2018-09	CT#81	CP-182141	0009	1	F	Timer T3540 clarifications	15.1.0
2018-09	CT#81	CP-182137	0010	3	F	Network Slicing Subscription Change Indication	15.1.0
2018-09	CT#81	CP-182133	0012		F	Correction for PDU session context	15.1.0
2018-09	CT#81	CP-182133	0013	1	F	Correction for establishment of user-plane resources	15.1.0
2018-09	CT#81	CP-182133	0014	1	F	Correction for establishment cause	15.1.0
2018-09	CT#81	CP-182133	0016	4	F	Correction for maximum data rate per UE for integrity	15.1.0
2019.00	CTH04	CD 400407	0040	2	F		15 1 0
2018-09	CT#01	CP-18213/	0018	3 1	г F		15.1.0
2010-09	CT#81	CP-182132	0020	1	r F	Editor's notes in LIPDP	15.1.0
2018-09	CT#81	CP-182136	0023	2	F	Exchange of extended protocol configuration options	15.1.0
2018-09	CT#81	CP-182130	0024	1	F	5G QoS - restructuring QoS rules IF	15.1.0
2018-09	CT#81	CP-182133	0025	1	F	Correction for editor's note on further 5GSM causes and	15.1.0
						further minor issues	
2018-09	CT#81	CP-182132	0026		F	Correction and alignment of cause code values	15.1.0
2018-09	CT#81	CP-182141	0027	2	F	UAC information and establishment cause when uplink user	15.1.0
						data packet is to be sent for a PDU session with suspended	
0040.00	07/04	00.400404	0000	2	г	user-plane resources	45.4.0
2018-09	CT#81	CP-182134	0028	2	F	Corrections for operator-defined access categories	15.1.0
2016-09	01#01	CP-102131	0030	1	Г	(6 bits), AMF Set ID (10 bits), AMF POINTER	15.1.0
2018-09	CT#81	CP-182132	0031	1	F	Correcting message definition of message including FENI	15.1.0
	07/04			-	-		
2018-09	C1#81	CP-182141	0032	2	F	SMF knowledge that a UE is configured for high priority access	15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182141 CP-182131	0032	2 2	F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network	15.1.0 15.1.0
2018-09 2018-09 2018-09	CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130	0032 0035 0036	2 2 4	F F C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication	15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09	CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130	0032 0035 0036	2 2 4	F F C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure	15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131	0032 0035 0036 0037	2 2 4 2	F F C C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation	15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131	0032 0035 0036 0037	2 2 4 2	F F C C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO	15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131 CP-182130	0032 0035 0036 0037 0038	2 2 4 2 1	F F C C C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131 CP-182130 CP-182132	0032 0035 0036 0037 0038 0039	2 2 4 2 1 1	F C C C F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131 CP-182130 CP-182132 CP-182132	0032 0035 0036 0037 0038 0039 0040	2 2 4 2 1 1 1	F C C C F C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182131 CP-182130 CP-182132 CP-182132 CP-182139 CP-182139	0032 0035 0036 0037 0038 0039 0040 0041	2 2 4 2 1 1 1 1	F C C F C F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133	0032 0035 0036 0037 0038 0039 0040 0041 0042	2 2 4 2 1 1 1 1	F C C C F F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043	2 2 4 2 1 1 1 1 1 1	F C C C F F F B	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044	2 2 4 2 1 1 1 1 1 2	F F C C F F B B	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182133 CP-182137	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045	2 2 4 2 1 1 1 1 1 2 1 1 2 1	F F C C F F B B F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182135	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046	2 2 4 2 1 1 1 1 2 1 2 1 2	F F C C F F B B F F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182135 CP-182140	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0045	2 2 4 2 1 1 1 1 2 1 1 2 1 1 1	F F C C F F B B F F F F F F F F F F F F F F F F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode	15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182135 CP-182140 CP-182140	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0046	2 2 4 2 1 1 1 1 2 1 1 2 1 1 2 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F C C F B B F F F F F F F F F F F F F F F F F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response	15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182131 CP-182138	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049	2 2 4 2 1 1 1 1 2 1 2 1 2	F F C C F B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 RSA	15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182135 CP-182131 CP-182138 CP-182138	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049	2 2 4 2 1 1 1 1 2 1 1 2 2 2	F F C C F F B B F	SMF knowledge that a UE is configured for high priority access. Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA	15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182136 CP-182137 CP-182137 CP-182135 CP-182138 CP-182138 CP-182136 CP-182136 CP-182136 CP-182136	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055	2 2 4 2 1 1 1 1 2 1 1 2 1 2 2 2	F F C C F B B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPL paging	15.1.0 15.1.0
2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09 2018-09	CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182137 CP-182138 CP-182138 CP-182138 CP-182138	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0047 0048 0049 0055 0056	2 2 4 2 1 1 1 1 2 1 1 2 2 2 2 2	F F C C F B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182137 CP-182138 CP-182138 CP-182138 CP-182138 CP-182133 CP-182133	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0047 0048 0049 0055 0056 0058	2 2 4 2 1 1 1 1 2 1 1 2 2 2 2 1	F F C C F B B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182136 CP-182137 CP-182137 CP-182137 CP-182138 CP-182138 CP-182138 CP-182133 CP-182131 CP-182131 CP-182131 CP-182131	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0054 0055 0056 0058 0060	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3	F F C C F F B B F B	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182137 CP-182137 CP-182137 CP-182138 CP-182138 CP-182138 CP-182138 CP-182139 CP-182139 CP-182131 CP-182139	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055 0056 0055 0056 0058 0060 0064	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1	F F C C F F B B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182139 CP-182133 CP-182133 CP-182136 CP-182137 CP-182138 CP-182138 CP-182138 CP-182138 CP-182139 CP-182139 CP-182139 CP-182131 CP-182131 CP-182131 CP-182133	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055 0056 0055 0056 0058 0060 0064 0065	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1 4	F F C C F F B B F C	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182135 CP-182137 CP-182138 CP-182138 CP-182138 CP-182139 CP-182137 CP-182137 CP-182137 CP-182137	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0054 0055 0056 0058 0056 0058 0060	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1 4	F F C C F F B B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if	15.1.0 15.1.0
2018-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-09 2008-00 2008-0	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182137 CP-182138 CP-182138 CP-182138 CP-182139 CP-182139 CP-182137 CP-182137 CP-182137 CP-182137	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0054 0055 0056 0058 0056 0058 0060	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1 4 2	F F C C F F B B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182137 CP-182136 CP-182138 CP-182138 CP-182138 CP-182137 CP-182137 CP-182137 CP-182137 CP-182130	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055 0056 0055 0056 0058 0060 0069	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1 4 2	F F C C F F B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available Abnormal Cases in the UE for mobilty and periodic Daristica Landon and the Context and periodic Daristica Landon and the Context and periodic	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182137 CP-182136 CP-182138 CP-182138 CP-182139 CP-182137 CP-182137 CP-182137 CP-182130 CP-182130	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055 0056 0055 0056 0058 0060 0069 0069	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 1 3 1 4 2 1 1 2 1 2 2 2 2 2 1 3 1 4 2 2 2 2 2 2 2 2 2 1 3 1 4 2 2 2 2 2 2 2 2 2 2 2 2 2	F F C C F F B F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available Abnormal Cases in the UE for mobilty and periodic Registration Update Procedures Correction to 5GMM Substate	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182137 CP-182136 CP-182138 CP-182138 CP-182139 CP-182137 CP-182137 CP-182137 CP-182130 CP-182134 CP-182134 CP-182134 CP-182134	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0054 0055 0056 0058 0056 0058 0060 0069 0069 0070 0070	2 2 4 2 1 1 1 1 1 2 1 1 2 2 2 2 2 2 1 3 1 4 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 2 2 2 2 2 1 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 2 2 2 2 2 1 3 1 4 2 1 1 2 2 2 2 2 2 2 1 3 1 4 1 1 2 2 2 2 2 2 2 1 3 1 4 1 2 1 2 2 2 2 2 1 3 1 1 1 2 2 2 2 1 3 1 1 4 2 1 1 1 1 2 2 2 2 1 3 1 1 1 1 2 2 2 2 1 3 1 1 4 1 1 1 1 2 2 2 1 3 1 1 4 1 1 1 1 1 1 2 2 2 1 1 3 1 1 1 1 1 1 1 1 1 1 1 1 1	F F C C F F B B F D	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available Abnormal Cases in the UE for mobilty and periodic Registration Update Procedures Correction to 5GMM Substate Definition of emergency and non-emergency PDU sessions	15.1.0 15.1.0
2018-09 2018-09	CT#81 CT#81	CP-182141 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182133 CP-182137 CP-182138 CP-182138 CP-182139 CP-182137 CP-182137 CP-182137 CP-182130 CP-182134 CP-182134 CP-182134 CP-182134 CP-182134 CP-182134 CP-182134	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0046 0047 0048 0049 0055 0056 0055 0056 0058 0060 0064 0065 0066 0066 0069 0070 0071 0074	2 2 4 2 1 1 1 1 1 1 1 2 1 1 2 2 2 2 2 2 2 1 3 1 4 2 1 1 2 1 1 2 1 1 2 1 2 2 2 2 2 1 1 3 1 1 2 1 2 2 2 2 2 1 3 1 4 2 1 2 1 2 2 2 2 2 2 1 1 2 1 2 2 2 2 2 2 2 1 3 1 4 2 1 2 2 2 2 2 2 2 2 1 3 1 4 2 2 2 2 2 2 2 2 2 2 2 2 2	F F C C F F B B F F F F F F F F F F F F F F F F F F F D F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available Abnormal Cases in the UE for mobilty and periodic Registration Update Procedures Correction to 5GMM Substate Definition of emergency and non-emergency PDU sessions PDU session statlis	15.1.0 15.1.0
2018-09 2018-0	CT#81 CT#81	CP-182131 CP-182131 CP-182130 CP-182130 CP-182130 CP-182132 CP-182132 CP-182133 CP-182133 CP-182133 CP-182133 CP-182133 CP-182135 CP-182138 CP-182138 CP-182139 CP-182137 CP-182137 CP-182137 CP-182130 CP-182134 CP-182134 CP-182138 CP-182134 CP-182134	0032 0035 0036 0037 0038 0039 0040 0041 0042 0043 0044 0045 0044 0045 0046 0047 0048 0049 0055 0056 0056 0058 0056 0058 0060 0064 0065 0066 0066 0066 0067 0070 0071 0074 0074	2 2 4 2 1 1 1 1 1 1 1 2 1 1 2 2 2 2 2 2 2 2 1 3 1 4 2 1 3 1 4 2 3 3 1 1 2 3 3 1 2 3 3 3 3 3 3 3 3 3 3 3 3 3	F F C C F F B B F F F F F F F F F F F F F F F F F D F F F F	SMF knowledge that a UE is configured for high priority access Authentication for normal services not accepted by network Addition of ABBA in 5G based primary authentication procedure Alignment and correction of mapped security context creation at S1 to N1 mode HO Addition of NAS container IE for N1 mode HO Correction and update of S1 mode to N1 mode NAS transparent container Removal of MAC editor's note Removal of transparent container at N1 mode to S1 mode HO Correction of 5GS TAC LSB Handling of Emergency PDU sessions and null algorithms Request for Kamf re-derivation Mobility Registration when T3346 running DL NAS Transport message Single-registration mode Authentication Response Parameters for PDU session establishment due to change of SSC mode 3 or 2 PSA Equivalent PLMNs Remove the remaining instance of SUPI paging PDU session status Clarification on NAS level MM congestion Control SM cause for out of LADN service area Removal of redundant MICO statement LADN indication from UE at registration Mapping to configured NSSAI for HPLM shall be included if available Abnormal Cases in the UE for mobilty and periodic Registration Update Procedures Correction to 5GMM Substate Definition of emergency and non-emergency PDU sessions PDU session establish criteria for emergency PDU sessions PDU session establish criteria for emergency PDU sessions Service request allowed for PDU release outside LADN	15.1.0 15.1.0

2018-09	CT#81	CP-182136	0077	4	В	How to determine the maximum number of established PDU sessions	15.1.0
2018-09	CT#81	CP-182141	0079	2	F	UAC and setting of the Uplink data status IE	15.1.0
2018-09	CT#81	CP-182137	0080	1	С	Non-3GPP access to 5GCN not allowed	15.1.0
2018-09	CT#81	CP-182136	0082	2	В	Including S-NSSAI received in EPS in Requested NSSAI and in PDU session establishment request upon inter-system change from S1 mode to N1 mode	15.1.0
2018-09	CT#81	CP-182141	0083		В	UE configuration for AC 11-15 and MCS (access identity 2)	15.1.0
2018-09	CT#81	CP-182142	0084		В	UE configuration for NAS signalling low priority via OMA-DM or USIM not applicable in 5GS	15.1.0
2018-09	CT#81	CP-182137	0085	2	F	Miscellaneous corrections	15.1.0
2018-09	CT#81	CP-182132	0086	2	F	Clarifications on ATTEMPTING-REGISTRATION-UPDATE	15.1.0
2018-09	CT#81	CP-182141	0087	2	F	UE behaviour in substate ATTEMPTING-REGISTRATION- UPDATE	15.1.0
2018-09	CT#81	CP-182140	0090		F	Service area list IE	15.1.0
2018-09	CT#81	CP-182136	0091	1	F	Handling of forbidden tracking area list	15.1.0
2018-09	CT#81	CP-182134	0092	2	F	Corrections for authentication	15.1.0
2018-09	CT#81	CP-182141	0094	4	F	Trigger for mobility and periodic registration update	15.1.0
2018-09	CT#81	CP-182130	0095		F	Abnormal cases in the UE for network-initiated de-registration procedure	15.1.0
2018-09	CT#81	CP-182130	0096	4	F	Add attempt counter to Service Request procedure	15.1.0
2018-09	CT#81	CP-182131	0097	1	F	Authentication procedure during registration procedure for mobility and periodic registration update	15.1.0
2018-09	CT#81	CP-182138	0098	4	F	Registration procedure for mobility and periodic registration	15.1.0
2018-09	CT#81	CP-182138	0099	3	F	Release of the N1 NAS signalling connection	15.1.0
2018-09	CT#81	CP-182139	0100	2	F	Resetting of registration attempt counter	15.1.0
2018-09	CT#81	CP-182138	0101	4	F	On #27 N1 mode not allowed	15.1.0
2018-09	CT#81	CP-182130	0102	3	F	Adding EPLMN list related descriptions	15.1.0
2018-09	CT#81	CP-182138	0104	2	F	Provision of IWK N26 indication in registration update procedure	15.1.0
2018-09	CT#81	CP-182134	0105		F	Corrections for interworking with EPS	15.1.0
2018-09	CT#81	CP-182135	0106	4	F	Emergency Services Support indicator for non-3GPP access	15.1.0
2018-09	CT#81	CP-182219	0107	7	С	Network control for always-on PDU sessions	15.1.0
2018-09	CT#81	CP-182135	0108		F	Corrections on inconsistent descriptions for 5GSM and 5GMM	15.1.0
2018-09	CT#81	CP-182135	0110		F	Corrections on the timers of 5GMM	15.1.0
2018-09	CT#81	CP-182137	0112	1	F	No operation code for UE policy management	15.1.0
2018-09	CT#81	CP-182134	0113	1	F	Correction on UE behaviour for 5GSM congestion control	15.1.0
2018-09	CT#81	CP-182134 CP-182136	0114	5	F F	Including SD when Mapped configured SD is included in S-	15.1.0
2018-09	CT#81	CP-182142	0116	3	F	Updates to deleting a derived QoS rule in the UE	1510
2018-09	CT#81	CP-182138	0117	5	F	Provisioning of ANDSP for non-3GPP access	15.1.0
2018-09	CT#81	CP-182136	0118		D	Fix incorrect references	15.1.0
2018-09	CT#81	CP-182130	0120	2	F	Addition of ngKSI in DEREGISTRATION REQUEST	15.1.0
2018-09	CT#81	CP-182141	0123	2	F	Storing of MPS indicator in non-volatile memory of mobile	15.1.0
2018-09	CT#81	CP-182138	0124	4	С	NW slicing and delayed re-registration due to emergency services	15.1.0
2018-09	CT#81	CP-182131	0125	1	F	Addtion of cause values for service request reject	15.1.0
2018-09	CT#81	CP-182141	0126	3	F	UE actions when other causes received at SERVICE REJECT	15.1.0
2018-09	CT#81	CP-182137	0128		F	Missing general description on sub-clause 9.10	15.1.0
2018-09	CT#81	CP-182134	0130		F	Correction to general message format	15.1.0
2018-09	CT#81	CP-182138	0131	1	F	Plain 5GS NAS message	15.1.0
2018-09		CP-182134	0134	1	Г Б	Correction to the SCS identity time IF	15.1.0
2018-09	CT#01	CP-102134	0135	1	г F	Correction to the SCS network feature support IE	15.1.0
2018-09	CT#81	CP-182134	0137	1	F	Editorials and minor corrections	15.1.0
2018-09	CT#81	CP-182140	0138	1	F	Security procedures and handling after inter-system change	15.1.0
2018-09	CT#81	CP-182130	0140		F	5GMM aspects of NAS over non-3GPP access	15.1.0
2018-09	CT#81	CP-182139	0141	2	F	Resolution of editor's note on equivalent PLMN list	15.1.0
2018-09	CT#81	CP-182140	0142		F	Resolution of editor's notes on 5GMM sub-layer design	15.1.0
2018-09	CT#81	CP-182139	0143		F	Resolution of editor's note on UE behaviour in substate 5GMM-DEREGISTERED.ATTEMPTING-REGISTRATION	15.1.0
2018-09	CT#81	CP-182139	0144		F	Resolution of editor's note on other sub-states of state 5GMM- DEREGISTERED	15.1.0
2018-09	CT#81	CP-182139	0145	3	F	Resolution of editor's note on sub-states of state 5GMM- REGISTERED	15.1.0
2018-09	CT#81	CP-182139	0146	1	F	Resolution of editor's note on the key derivation function field	15.1.0
2018-09	CT#81	CP-182139	0147		F	Resolution of editor's note on security context coordination	15.1.0
2018-09	CT#81	CP-182139	0148		F	Removal of unnecesary editor's notes FFS	15.1.0
2018-09	CT#81	CP-182139	0151		F	Resolution of editor's note on handling of unknown, unforeseen, and erroneous protocol data	15.1.0

2018-09	CT#81	CP-182131	0153	2	С	AMF taking both EMC and EMC BS availability into account in	15.1.0
2018-09	CT#81	CP-182131	0154	1	F	Clarification on SM congestion control specific to PLMN	15 1 0
2018-09	CT#81	CP-182131	0155	1	F	Aligning T35cd bandling upon NW initiated SM request with	15.1.0
2010 00	01//01	01 102101	0100	-	-	T3396 and T35ef	10.1.0
2018-09	CT#81	CP-182132	0156	2	F	Clarification on stopping back-off timers upon reception of NW	15.1.0
2019.00	CT#04	CD 100101	0157	1	E	Initiated SM request	15 1 0
2016-09	C1#01	CP-162131	0157	1	Г	introduction of lower laver failure	15.1.0
2018-09	CT#81	CP-182137	0158	2	В	Introduction of 5GMM cause #15	15.1.0
2018-09	CT#81	CP-182141	0159	1	F	Timer for re-enabling N1 mode capability	15.1.0
2018-09	CT#81	CP-182137	0160		F	Lists of 5GS forbidden tracking areas	15.1.0
2018-09	CT#81	CP-182137	0162	1	F	Local release of a persistent PDU session	15.1.0
2018-09	CT#81	CP-182134	0163	1	F	Correction on retry of PDU session establishment procedure	15.1.0
2018-09	CT#81	CP-182134	0164		F	Correction to 5GSM/ESM coordination	15.1.0
2018-09	CT#81	CP-182133	0165		F	Correction on PDU address IE	15.1.0
2018-09	CT#81	CP-182130	0166	1	F	5GSM congestion control over AMF on PDU session	15.1.0
2018-09	CT#81	CP-182136	0167	3	F	Exception bandling in OoS operation	1510
2018-09	CT#81	CP-182134	0168	1	F	Correction on PTI mismatch	15.1.0
2018-09	CT#81	CP-182136	0169	1	B	Establishment of N1 NAS signalling connection due to change	15.1.0
						in the network slicing information	
2018-09	CT#81	CP-182138	0170	2	F	Release of N1 NAS signalling connection due to change in the	15.1.0
						network slicing information	
2018-09	CT#81	CP-182137	0171	3	F	Multiple S-NSSAIs in PDU session establishment	15.1.0
2018-09	CT#81	CP-182132	0173	3	F	Clarifications on UE 5GSM capabilities and procedures during	15.1.0
2018.00	CT#91	CD 192126	0174	1	D	Inter-working with EPS	15 1 0
2018-09	CT#81	CP-182130	0174	1	D	Interworking between E-LITRAN/EPC and N3IW/E/5GCN	15.1.0
2018-09	CT#81	CP-182130	0178	2	F	ITE re-registration when the AME cannot determine an Allowed	15.1.0
2010 00	01//01	01 102142	0170	-	-	NSSAI	10.1.0
2018-09	CT#81	CP-182137	0180	2	F	Local release of a PDU session due to 5GSM cause #43:	15.1.0
						Invalid PDU session identity	
2018-09	CT#81	CP-182132	0182	2	С	Common NAS security transparent container IE for intra-5G	15.1.0
2019.00	CT#04	CD 100106	0104	1	E	HO and S1 to N1 Inter-system HO	15 1 0
2016-09	CT#01	CP-102130	0164	1	Г	N1 mode to S1 mode	15.1.0
2018-09	CT#81	CP-182131	0185	2	F	Allowed NSSAI of a single-registration mode UE within a	15.1.0
						network with N26	
2018-09	CT#81	CP-182141	0187	1	F	SMF selection based on DNN for transfer a PDN connection	15.1.0
						from EPS to 5GS	
2018-09	CT#81	CP-182141	0188	1	F	UE behaviour for determination of the UE presence in LADN	15.1.0
2018-09	CT#81	CP-182133	0180	3	F	Correction on emergency PDLL session handling	15 1 0
2018-09	CT#81	CP-182137	0190	1	F	No EMM parameters handling for DR mode LIEs due to	15.1.0
2010 00	01//01	01 102101	0100	-	-	rejected service request	10.1.0
2018-09	CT#81	CP-182131	0192	3	F	Clarification on activation of UP resources of PDU session	15.1.0
2018-09	CT#81	CP-182130	0194	1	В	Access attempt barred for the UE-initiated NAS transport	15.1.0
						procedure	
2018-09	CT#81	CP-182142	0195	5	F	UE configured for EAB and access category 1	15.1.0
2018-09	CT#81	CP-182137	0197	1	F	No bearer for N1 NAS	15.1.0
2018-09	CT#81	CP-182133	0198	4	F E	Correction of S-NSSAI based congestion control	15.1.0
2018-09	CT#81	CP-182132	0200	1	F	Clarify abnormal cases in the LIE for independency of 5CMM	15.1.0
2010-09	01#01	01 102132	0200	-	*	procedures between accesses	10.1.0
2018-09	CT#81	CP-182141	0203	2	В	Storing Configured NSSAI when the PLMN is changed	15.1.0
2018-09	CT#81	CP-182136	0206		F	Incorrect statement for handling of security context at IWK	15.1.0
2018-09	CT#81	CP-182134	0207	2	F	Correction to SSC mode selection	15.1.0
2018-09	CT#81	CP-182135	0208	2	F	Corrections to terms and references	15.1.0
2018-09	CT#81	CP-182140	0209	1	С	Revision on AMF transport behaviour of 5GSM message	15.1.0
2018-09	CT#81	CP-182135	0210	2	F	Differences between NAS over 3GPP access and NAS over	15.1.0
2019.00	CT#04	CD 100100	0242	6	D	NON-JGPP ACCESS	15 1 0
2010-09	CT#81	CP-182130	0212	4	Б	Clarification on network-initiated de-registration procedure	15.1.0
2018-09	CT#81	CP-182131	0214	-	F	Correction of detach terminology	15.1.0
2018-09	CT#81	CP-182132	0215		F	Clarification on S-NSSAI based condestion control for PDU	15.1.0
						session modification procedure	
2018-09	CT#81	CP-182141	0216	4	В	SOR acknowledge message coding	15.1.0
2018-09	CT#81	CP-182141	0232	1	F	SMS over NAS re-transmission upon delivery failure on one	15.1.0
0010	07.02		005-		F	Access Type	45 4 5
2018-09	CT#81	CP-182134	0233	2	F	Corrections in EAP based primary authentication procedure	15.1.0
2018-09	CT#81	CP-182133	0234		F	Correction for multi-homed IPv6 PDI I session	1510
	21/01	2. 102100		1	1		

2018-09	CT#81	CP-182133	0235	1	F	Correction for transfer of a PDN connection from untrusted	15.1.0
2018-09	CT#81	CP-182133	0237	1	F	Correction for generation of QoS rules	1510
2018-09	CT#81	CP-182137	0239	1	F	Interworking for multi-homed IPv6 PDU session	15.1.0
2018-09	CT#81	CP-182131	0241	1	F	Clarification of N1 NAS signalling connection release in AMF on generic UE configuration update completion	15.1.0
2018-09	CT#81	CP-182139	0242	1	F	Requests for emergency services fallback from upper layers	15.1.0
2018-09	CT#81	CP-182135	0245		F	Corrections to the Identification and Registration procedures	15.1.0
2018-09	CT#81	CP-182132	0246		F	Correct abnormal procedures reference when handling CC	15.1.0
						#22 (Congestion)	
2018-09	CT#81	CP-182137	0247	1	F	Non-IP PDN connection type for S1 to N1 interworking	15.1.0
2018-09	CT#81	CP-182137	0248	1	F	Non-3GPP de-registration timer	15.1.0
2018-09	CT#81	CP-182141	0250	3	F	Substates for registration result	15.1.0
2018-09	CT#81	CP-182142	0251	1	F	Updating NS Configuration via registration procedure	15.1.0
2018-09		CP-182141	0254	4	B	SUCI encoding format and protection scheme	15.1.0
2010-09	01#01	CF-102132	0255	1	C	1	13.1.0
2018-09	CT#81	CP-182136	0256	1	В	Handling of error case when UE gets URSP from VPLMN	15.1.0
2018-09	CT#81	CP-182132	0258	2	F	Correction for abnormal cases in the UE of service request	15.1.0
2018-00	CT#81	CP-182140	0250	1	F	Setting of RPC establishment cause for operator-defined	15 1 0
2010-03	01#01	01-102140	0233	1	1	access categories	13.1.0
2018-09	CT#81	CP-182131	0262		F	Alignment with terminology "emergency PDU session"	15.1.0
					_	throughout TS 24.501	
2018-09	CT#81	CP-182141	0263	1	F	TAI removed from list of Servie area lists after reject from network	15.1.0
2018-09	CT#81	CP-182135	0264	1	F	EAP message IE mandatory in PDU SESSION	15.1.0
						AUTHENTICATION messages	
2018-09	CT#81	CP-182135	0269	1	F	Corrections related to the authentication procedure	15.1.0
2018-09	CT#81	CP-182140	0271	2	F	Security parameter carrying DL NAS COUNT during N1 to S1	15.1.0
2018-09	CT#81	CP-182130	0272	2	F	Adding procedures for updating local emergency numbers in	15.1.0
						other modes	
2018-09	CT#81	CP-182131	0275		F	Authentication response parameter IE to be of fixed length (24 501)	15.1.0
2018-09	CT#81	CP-182134	0276		F	Correction to the PDU Session ID value in Allowed PDU	15.1.0
						session status IE	
2018-09	CT#81	CP-182138	0277	2	F	Reactivation result indicating insufficient resources during	15.1.0
2018-09	CT#81	CP-182138	0278	1	С	Registration procedure triggered by a change of UE Radio	1510
2010 00	01//01	01 102100	0270	1	C	Capability	10.1.0
2018-09	CT#81	CP-182142	0280	2	F	Updates to RRC fallback indication while in 5GMM-	15.1.0
						CONNECTED mode, or while in 5GMM-CONNECTED mode	
2018.00	CT#91	CD 192129	0291	2	C	BDU Session Polesse due to Semantic or Syntactical Errors	15 1 0
2018-09	CT#81	CP-182138	0201	3	F	Removal of 5GSM cause from ePCO for PDU Session	15.1.0
2010 00	01//01	01 102100	0202	5	1	Release Complete	10.1.0
2018-09	CT#81	CP-182134	0283		F	Correction on PTI definition	15.1.0
2018-09	CT#81	CP-182140	0284	1	F	Resolving EN on fatal causes in 5GMM/5GSM state machine	15.1.0
2018-09	CT#81	CP-182130	0285	1	F	Uplink data handling for MT notification	15.1.0
2018-09	CT#81	CP-182136	0286	1	F	Fallback handling for RRC inactive	15.1.0
2018-09	CT#81	CP-182134	0287	1	F	Correction on PDU session modification procedure	15.1.0
2018-09	CT#81	CP-182140	0289	1	F	RRC establishment cause for EAB	15.1.0
2018-09	CT#81	CP-182138	0290	2	С	NW slicing and delayed registration due to emergency services reject PDU session request	15.1.0
2018-09	CT#81	CP-182134	0291	1	F	Correction to the UE security capability IE encoding	15.1.0
2018-09	CT#81	CP-182130	0292	3	F	Additions to UE configuration update completion clause	15.1.0
2018-09	CT#81	CP-182139	0293	2	F	Removal of local PDU session relase statement in UCU	15.1.0
						procedure	
2018-09	CT#81	CP-182140	0294		F	Resolution of editor's notes in D.3 and D.6.2	15.1.0
2018-09	CT#81	CP-182140	0295		F	Resolution of the editor's note on value of the non-3GPP de-	15.1.0
2018-00	CT#81	CP-182130	0206		F	Resolution of editor's note on the format of the authentication	1510
2010-03		01 102103	0200		1	parameters	10.1.0
2018-09	CT#81	CP-182140	0297	2	F	Resolution of editor's note on unknown or unforeseen PDU	15.1.0
2018-00	CT#81	CP-182130	0208		F	session identity Resolution of editor's note on other types of payload for the	15 1 0
2010-09	01#01	01-102139	0230		1	NAS transport procedure(s)	10.1.0
2018-09	CT#81	CP-182140	0300	2	F	Serving network name format for primary authentication	15.1.0
2018-09	CT#81	CP-182136	0301	1	В	Initial registration not accepted due to serving network not	15.1.0
						authorized	
2018-09	CT#81	CP-182130	0303	1	F	5GSM cause #xx –out of LADN service area	15.1.0
2018-09	CT#81	CP-182142	0304	1	F	UE policy delivery protocol in the scope	15.1.0

2018 00	CT#91	CD 192121	0205	1	F	AME pointer pointing one or more AMEs	1510
2010-03	01#01	00-102101	0303	1	г Г		15.1.0
2018-09	CT#81	CP-182135	0306		F	Corrections in the conditions for SMS via non-3GPP access	15.1.0
2018-09	CT#81	CP-182130	0308		F	5GMM STATUS message sent by the AMF when certain error	15.1.0
						conditions are detected upon receipt of 5GMM protocol data in	
						the AME	
2018-09	CT#81	CP-182137	0311	1	F	Interworking with E-UTRAN connected to EPC of a UE	15.1.0
						registered to 5GC via non-3GPP access	
2018 00	CT#91	CD 1921/2	0212	2	F	Lise of S NSSAL and session AMBP provided during the EPS	15 1 0
2010-09	01#01	GF-102142	0313	2	1.	Use of 3-NSSAT and session-AMDR provided during the LFS	13.1.0
						bearer context modification procedure	
2018-09	CT#81	CP-182136	0314	1	F	Handling of inter-access handover of a PDU session whose S-	15.1.0
						NSSAL is not allowed for the target access	
2019 00	CT#01	CD 102120	0210	1	F	ECSM sublever states for BDU session inactive	15 1 0
2010-09		CF-102130	0319	1	Г —		15.1.0
2018-09	CT#81	CP-182139	0321	1	F	Rename of T3584 and T3585	15.1.0
2018-09	CT#81	CP-182133	0322		F	Correction on acknowledgement handling of UE configuration	15.1.0
20.0000	00.	002.00	0022		-		
0040.00	07/04	00 400400	0000	1	F		45.4.0
2018-09	CT#81	CP-182133	0323	1	F	Correction on INVV initiated de-registration procedure	15.1.0
2018-09	CT#81	CP-182140	0326	1	F	Service area restrictions	15.1.0
2018-09	CT#81	CP-182135	0327	3	F	Disabling and re-enabling N1 mode capability	1510
2010 00	CT#04	OF 102100	0021	1	г Г		10.1.0
2016-09	CT#61	CP-162131	0320	1	F	Clarification on packet litters	15.1.0
2018-09	CT#81	CP-182133	0329	1	F	Correction on 5G-GUTI type encoding	15.1.0
2018-09	CT#81	CP-182133	0330	1	F	Correction on EAP-AKA' based primary authentication	15.1.0
2018 00	CT#91	CD_192122	0221		F	Correction on 5GSM state mapping when interworking	15 1 0
2010-09	01#01	000102133	0001	1	1		10.1.0
2018-09	CT#81	CP-182138	0333	1	F	Removal of Default EPS Bearer (DEB) indication	15.1.0
2018-09	CT#81	CP-182133	0334	1	F	Correction on interaction with upper layers	15.1.0
2018-00	CT#81	CP-182140	0337	2	F	Security Context	1510
2010-09	OT#01	CD 400405	00007	2	1 D	Emorranau coll in limited consists state	15.1.0
2018-09	CT#81	CP-182135	0339	3	В	Emergency call in limited service state	15.1.0
2018-09	CT#81	CP-182201	0340	3	С	Emergency call error handling	15.1.0
2018-09	CT#81	CP-182138	0341	2	F	PDU session status in notification response message	15.1.0
2018 00	CT#91	CD 192122	0242	1	Б	Clarifications on inclusion of SNSSAL in 5CSM request	15 1 0
2010-03	01#01	01-102102	0042	1	Г Г		15.1.0
2018-09	CT#81	CP-182133	0344		F	Correction for LADN	15.1.0
2018-09	CT#81	CP-182135	0345	1	F	Definition of user-plane resources	15.1.0
2018-09	CT#81	CP-182136	0346	1	F	Handling of PDU session(s) not supporting interworking with	15.1.0
20.0000	00.	002.00	00.0	-	-	EPS	
0040.00	OTHOL	00 400444	0050		D		45.4.0
2018-09	CT#81	CP-182141	0350		D	Terminology correction	15.1.0
2018-09	CT#81	CP-182131	0351	3	F	Clarification on MICO indication and LADN information via	15.1.0
						generic UE configuration update procedure	
2018 00	CT#91	CD 192122	0252	1	Б	Clarification on the temperany identity in the convice request	15 1 0
2010-09	01#01	GF-102132	0352	1	Г	Ciannealion on the temporary identity in the service request	15.1.0
						procedure	
2018-09	CT#81	CP-182130	0353		F	Abbreviation update for NITZ	15.1.0
2018-09	CT#81	CP-182135	0355		D	Duplicated subclause for registration request message	15.1.0
2018-00	CT#81	CP-182135	0356	1	D		15 1 0
2010 00	01#01	01-102100	0000	1	D		15.1.0
2018-09	C1#81	CP-182135	0357	2	В	DL and UL NAS Transport procedure updates for SOR	15.1.0
2018-09	CT#81	CP-182132	0358	1	F	Conditions to send registration complete message.	15.1.0
2018-09	CT#81	CP-182132	0359	1	F	Condition for starting timer T3540	15.1.0
2018-00	CT#81	CP-1821/0	0361	2	F	Security context mapping at 5GS to EPS HO	15 1 0
2010-03	01#01	00 102140	0001	2	т Г		15.1.0
2018-09	CT#81	CP-182134	0363	2	F	Correction to reset registration attempt counter	15.1.0
2018-09	CT#81	CP-182130	0365	3	F	Adding the reference to T3346	15.1.0
2018-09	CT#81	CP-182130	0367	1	F	Access attempt due to delivery of LPP message/transparent	1510
2010-03	01//01	02100	0007	1	-	container/LIE policy container	
	0=						
2018-12	CT#82	CP-183032	0103	6	F	Clarification on coordination between 5GMM and 5GSM	15.2.0
2018-12	CT#82	CP-183032	0121	10	F	Clarification to VoPS indicator	15.2.0
2018-12	CT#82	CP-183032	0274	4	F	Correct Extended Local Emergency Numbers List deletion	15.2.0
	01// OL	2. 10000Z		1	-	unon PI MN change	
0040.40	OT#00	00 400000	0045	4	г		45.0.0
2018-12	CT#82	CP-183032	0315	4	F	Operator-controlled inclusion of NSSAI in access stratum	15.2.0
						connection establishment	
2018-12	CT#82	CP-183032	0371	2	F	Correction for maximum data rate per UE for integrity	15.2.0
						protection for DRBs for PDLL sessions in non-3GPP access	
						which are transferable to 2000 access	
	a =			-			
2018-12	<u>CT#82</u>	CP-183032	0372	1	F	Correction for 3GPP PS data off and non-IP user data packets	15.2.0
2018-12	CT#82	CP-183032	0373	2	F	Corrections for determination of RRC establishment cause and	15.2.0
	-					establishment cause for non-3GPP access	-
2010 12	CT#02	CD 192022	0274		F	Corrections for MTU PCO parameters handling	15.2.0
2018-12		0P-103032	03/4		1		15.2.0
2018-12	CT#82	CP-183032	0376	2	F	Correction for SM PDU DN request container coding	15.2.0
2018-12	CT#82	CP-183032	0379	1	F	UAC: Correction for SMSoIP sent over DNN other that "IMS"	15.2.0
2018-12	CT#82	CP-183032	0380	2	F	UAC: Correction for operator-defined access categories	1520
2010-12	01#02	OD 100002	0000	-	<u>.</u> Г		10.2.0
2018-12	CT#82	CP-183032	0381	1	F	DRA parameters IE definition	15.2.0
2018-12	CT#82	CP-183032	0383	1	F	Removal of UE security capability from Intra N1 mode NAS	15.2.0
						container	
2018-12	CT#82	CP-183032	0385		F	Correct procedure for determining registered PLMN	1520
2010 12	CT#02	CD 100002	0200	1	- F		15.2.0
2018-12	01#82	GP-183032	0392	1	Г —	Deminuon or emergency registration	15.2.0
2018-12	CT#82	CP-183032	0393	1	F	Correction for EAP based primary authentication and	15.2.0
						AUTHENTICATION REJECT	
2010 12	CT#82	CP-183032	0304		F	Correction for LADN information encoding	15 2 0
2010-12		01 100002	000-		1		10.2.0

					-		
2018-12	CT#82	CP-183032	0395	3	В	Dynamic Routing indicator update description	15.2.0
2018-12	CT#82	CP-183032	0398	1	F	ABBA handling for 5G-AKA based authentication procedure	15.2.0
2018-12	CT#82	CP-183032	0399	2	F	ABBA handling when initiating FAP procedures	15.2.0
2018-12	CT#82	CP-183032	0401	3	F	Clarification on NAS message field format and manning	15.2.0
2010-12	CT#02	CF 103032	0401	3	Г		15.2.0
2018-12	CT#82	CP-183032	0402	1	F	Correction to nome country definition	15.2.0
2018-12	CT#82	CP-183032	0403	1	F	Signalling Default Configured NSSAI indication in the	15.2.0
						registration procedure	
2018-12	CT#82	CP-183032	0404	1	F	Rename "configured NSSAI not associated with a PLMN" to	15.2.0
						align to new stage 2 terminology	
2019 12	CT#02	CD 102022	0405	1	Б	"SMS subscribed indication" in CONFICURATION UDDATE	15 2 0
2010-12	01#02	CF-103032	0405	1	г		15.2.0
						COMMAND	
2018-12	CT#82	CP-183033	0406	1	F	S-NSSAI not allowed by AMF	15.2.0
2018-12	CT#82	CP-183033	0407		F	Continued need to align with terminology "emergency PDU	15.2.0
						session" throughout TS 24,501	
2018-12	CT#82	CP-183033	0408	1	F	Ambiguity in the use of the terms "no other parameter" and "no	1520
2010 12	01//02	01 100000	0400		1	parameters"	10.2.0
0040.40	07/00	00 400000	0.400		Г		45.0.0
2018-12	CT#82	CP-183033	0409		F	Change EMCW to EMCN3	15.2.0
2018-12	CT#82	CP-183033	0410	2	F	Miscellaneous wording, terminology and reference corrections	15.2.0
2018-12	CT#82	CP-183033	0411		F	Rejected NSSAI clarifications	15.2.0
2018-12	CT#82	CP-183033	0413	1	F	Rejected NSSAI sent in CONFIGURATION UPDATE	1520
2010 12	01//02	01 100000	0110		1	COMMAND = Alt 2	10.2.0
0040.40	OT#00	00 400000	0445		г		45.0.0
2018-12	CT#82	CP-183033	0415	-	F	UAC – meaning of the broadcast of categories a, b or c	15.2.0
2018-12	CT#82	CP-183033	0416	2	F	Clarification of "registration requested" in CONFIGURATION	15.2.0
						UPDATE COMMAND	
2018-12	CT#82	CP-183033	0417	5	F	Updates on steering of roaming handling and information	15.2.0
	002	0	• • • •	-	-	coding	
0040.40	OT#00	00 400000	0440	2	г	County	45.0.0
2018-12	CT#82	CP-183033	0419	3	F	Security context usage during NAS security mode control	15.2.0
						procedure	
2018-12	CT#82	CP-183033	0420	1	F	Corrections on GFBR parameter for QoS flow	15.2.0
2018-12	CT#82	CP-183033	0421	1	F	Correction to UE behavior when disabling N1 mode for non-	15.2.0
						3GPP access	
2019 12	CT#92	CD 192022	0422	1	F	Mannad OoS information validation (Solution 1)	15.2.0
2010-12	OT#02	CF-103033	0423	1	T D		15.2.0
2018-12	CT#82	CP-183033	0424	4	В	Protection of initial INAS messages – overall description	15.2.0
2018-12	CT#82	CP-183033	0425	4	В	Support for protection of initial NAS messages	15.2.0
2018-12	CT#82	CP-183033	0426	2	В	Update to 5GS registration type IE and introduction of a new	15.2.0
						5GS update type IE	
2018-12	CT#82	CP-183033	0427	1	F	S-NSSAL received in Notify payload during PDN connection	15 2 0
2010 12	01#02	01-100000	0721	1	1	establishment over ePDC/EPC	10.2.0
0040.40	OTUGO	0.0.400000	0.400		-		45.0.0
2018-12	C1#82	CP-183033	0428		F	No interworking to ePDG/EPC for Ethernet and unstructured	15.2.0
						PDU sessions	
2018-12	CT#82	CP-183033	0429		F	Addition of UE capability for LPP in 5GMM capability IE	15.2.0
2018-12	CT#82	CP-183033	0430	1	F	Correct superfluous test for N1 mode and S1 mode capability	15.2.0
2018-12	CT#82	CP-183033	0431	1	F	Change LIE policy classmark as mandatory IE	15.2.0
2010 12	OT#02	OF 100000	0400	1	Г		15.2.0
2018-12	CT#82	CP-183033	0432	2	F	Forward compatibility for UE security capability IE and a few	15.2.0
						other IEs	
2018-12	CT#82	CP-183033	0433	1	F	Introduction of Replayed S1 UE security capabilities	15.2.0
2018-12	CT#82	CP-183033	0434	1	F	Applicability of UAC for other cases of NAS message transport	15.2.0
2018-12	CT#82	CP-183034	0/35	5	F	Emergency registered	15.2.0
2010-12	01#02	CD 400004	0400	2	r E	Correction to domain actaction when far EDO/DAT E-III	15.2.0
2018-12	01#82	CP-183034	0436	5	Г	Correction to domain selection rules for EPS/RAT Fallback	15.2.0
2018-12	CT#82	CP-183034	0438		F	UL data status upon fallback indication from lower layers	15.2.0
2018-12	CT#82	CP-183034	0440	1	F	Misc. corrections to 24.501	15.2.0
2018-12	CT#82	CP-183034	0441	2	F	Clarification on handling of PDU session for LADN	15.2.0
2018.12	CT#92	CP_182024	0//2	2	F	Correction to determination method of LADN DNN	15.2.0
2010-12	OT#02	CD 100004	0442		r E	Correction to determination method of LADN DINN	15.2.0
2018-12	U1#82	CP-183034	0443	1	г	Correction to trigger or the mobility and periodic registration	15.2.0
						upaate procedure	
2018-12	CT#82	CP-183034	0444	1	F	Abnormal cases in the network side for Configuration update	15.2.0
						prcedure.	1
2018-12	CT#82	CP-183034	0445	1	F	Correction in emergency reg cause name	15.2.0
2019 12	CT#02	CP_192024	0446	-	F	UE behavior on NW/ initiated daragistration procedure with #22	15.2.0
2010-12	01#02	00 40000 1	0440	1	1' F		15.2.0
2018-12	CT#82	CP-183034	0447	1	F		15.2.0
2018-12	CT#82	CP-183034	0448	3	F	Resetting service request attempt counter upon receipt of the	15.2.0
						REGISTRATION ACCEPT message	
2018-12	CT#82	CP-183034	0450		F	Correction in determining sytactic errors for PDU session	15.2.0
	0.002	5	0.00		-	establishment accent	
2010 10	CT#00	CD 102024	0454		Б	T2540 started by the LIEse astting 50MM serves #07 MM	15.2.0
2018-12	01#82	CF-103034	0451		г	13040 statied by the UEOn getting 5GMIVI cause #27 – N1	15.2.0
			L			mode not allowed	L
2018-12	CT#82	CP-183034	0452	2	F	Clarification on UE identities used for registration	15.2.0
2018-12	CT#82	CP-183034	0454	1	F	Transition from 5GMM-CONNECTED mode with RRC inactive	15.2.0
						indication to 5GMM-IDLE mode triggered by radio capability	
1		1		1			1
						undate	
2010 40	07#00	CD 400004	0455	1	E	update	15.0.0

2018-12	CT#82	CP-183034	0458	1	F	Local PDU session release upon receipt of PDU Session Release Reject with 5GSM cause #34 (service option	15.2.0
						temporarily out of order) or #35 (PTI already in use)	
2018-12	CT#82	CP-183034	0460	3	F	Clarifications on the EAP based AKA procedure	15.2.0
2018-12	CT#82	CP-183034	0461	1	F	Applicability of the Service area list IE indicating all TAIs to equivalent PLMNs	15.2.0
2018-12	CT#82	CP-183034	0464	1	F	Clarification on PDU session transfer from non-3GPP to 3GPP access when 3GPP PS data off UE status is activated	15.2.0
2018-12	CT#82	CP-183034	0465	1	F	EAP-Identification in EAP-AKA' primary authentication and key	15.2.0
2018-12	CT#82	CP-183034	0467	1	F	Excluding mobility procedures from ODAC access control checks	15.2.0
2018-12	CT#82	CP-183034	0470	3	F	Alignment of 5G-GUTI assignment with SA3	15.2.0
2018-12	CT#82	CP-183034	0471	1	F	Network initiated de-registration in case of emergency PDU session	15.2.0
2018-12	CT#82	CP-183035	0472		F	Resolution of editor's note on always-on PDU session	15.2.0
2018-12	CT#82	CP-183035	0473		F	Addition of 5GSM cause values	15.2.0
2018-12	CT#82	CP-183035	0474		F	Resolution of editor's note on operator-defined access category	15.2.0
2018-12	CT#82	CP-183035	0476	6	С	Support for Traffic Segregation	15.2.0
2018-12	CT#82	CP-183035	0479		F	Correct reference to Mapped EPS bearer contexts	15.2.0
2018-12	CT#82	CP-183035	0480		F	Aborted UE-initiated NAS transport procedure for delivery of SMS/LPP message/UE policy container	15.2.0
2018-12	CT#82	CP-183035	0484	3	F	Clarification on back-off timer upon PLMN change	15.2.0
2018-12	CT#82	CP-183035	0485	2	F	Emergency call in limited service state	15.2.0
2018-12	CT#82	CP-183035	0490	2	F	Correction on Network slicing indication	15.2.0
2018-12	CT#82	CP-183035	0491	1	F	Correction on De-registration procedure	15.2.0
2018-12	CT#82	CP-183035	0492	1	F	Correction on dual-registration mode	15.2.0
2018-12	CT#82	CP-183035	0494	1	F E	Correction to Configured NSSAI for the HPLININ	15.2.0
2010-12	CT#82	CP-183035	0499	5	г F	Editorials and minor corrections	15.2.0
2018-12	CT#82	CP-183035	0500	3	F	Adding necessary term definitions	15.2.0
2018-12	CT#82	CP-183035	0502	1	F	Resolution of editor's notes on abnormal case handling when rejection with "Extended wait time" received from lower layers	15.2.0
2018-12	CT#82	CP-183035	0503	1	F	Resolution of editor's note on maximum length of the 5GS mobile identity IF	15.2.0
2018-12	CT#82	CP-183035	0505	1	F	Resolution of editor's note in sub-clause 4.8.3	15.2.0
2018-12	CT#82	CP-183035	0506	1	F	Correct unified access control applicability in 5GMM- REGISTERED.UPDATE-NEEDED	15.2.0
2018-12	CT#82	CP-183035	0507	3	F	Correct non-3GPP registration accept procedure when local emergency numbers are received from a different country	15.2.0
2018-12	CT#82	CP-183035	0509	1	F	Remove editor's note for MT LCS	15.2.0
2018-12	CT#82	CP-183035	0510	1	F	Reusing T3519 for Initial Registration procedure	15.2.0
2018-12	CT#82	CP-183035	0511	2	F	Abnormal cases for Registration procedure for mobility and periodic registration update	15.2.0
2018-12	CT#82	CP-183035	0516	1	F	Correction on handling of mandatory IE errors	15.2.0
2018-12	CT#82	CP-183035	0517	1	F	Correction on handling of invalid PSI	15.2.0
2018-12	CT#82	CP-183036	0518	2	F	Clarification on PLMN's maximum number of PDU sessions	15.2.0
2018-12	C1#82	CP-183036	0519	1	F	release procedures	15.2.0
2018-12	CT#82	CP-183036	0521		F	Correction on QoS rules IE	15.2.0
2018-12	CT#82	CP-183036	0522	2	F	Correction on QoS flow description IE	15.2.0
2018-12	CT#82 CT#82	CP-183036 CP-183036	0523 0525	1 2	F F	Correction of storage of operator-defined access categories Clarification on coordination between 5GMM and EMM for a	15.2.0 15.2.0
2018-12	CT#82	CP-183036	0526		F	The UE behavior in non-3GPP access and in state	15.2.0
2018-12	CT#82	CP-183036	0528	1	F		1520
2018-12	CT#82	CP-183036	0520	2	F	Apply service area restrictions in NAS procedures	15.2.0
2018-12	CT#82	CP-183036	0530	1	F	Abnormal cases in 5GSM procedures	15.2.0
2018-12	CT#82	CP-183036	0532		F	Clarifications on UE behaviour upon receiving RRC fallback	15.2.0
2018.12	СТ#82	CP-182026	0522	2	F	Corrections for interworking with EDS	15 2 0
2018-12	CT#82	CP-183036	0534	1	F	Release of the N1 NAS signalling connection upon Service	15.2.0
2018-12	CT#82	CP-183036	0535	1	F	QoS rules verification during PDU session establishment	15.2.0
2010 10	CT#00	CD 102020	0520	1	Б	Clarifications on UE and natwork state	15.2.0
2010-12	CT#82	CP-183036	0530	1	r F	T3517 handling and emergency services fallback	15.2.0
2018-12	CT#82	CP-183036	0540	2	F	Revisions on N1 NAS signaling connection establishment and	15.2.0
_0.0 .2	OTHOL	5	00-0	–	-	release	
2018-12	CT#82	CP-183036	0541	1	F	Disabling N1 mode capability for 3GPP access and impacts to PLMN selection	15.2.0

2018-12	CT#82	CP-183036	0543	1	F	UE's homogeneous support of emergency services fallback	15.2.0
2018-12	CT#82	CP-183036	0544		F	Context management for 3GPP access and non-3GPP access due to SR rejection	15.2.0
2018-12	CT#82	CP-183036	0545	2	F	Clarification on setting a service type of a SERVICE REQUEST message	15.2.0
2018-12	CT#82	CP-183036	0546	1	F	UL NAS TRNAPORT message pending due to network slicing information update	15.2.0
2018-12	CT#82	CP-183036	0547		F	Distinction in AMF-side abnormal cases for generic UE configuration update procedure with respect to 5G-GUTI update	15.2.0
2018-12	CT#82	CP-183036	0548	1	F	Handling of security contexts by a UE operating in dual- registration mode	15.2.0
2018-12	CT#82	CP-183036	0549	2	F	Establishment of secure exchange of NAS messages during inter-system change between N1 mode and S1 mode	15.2.0
2018-12	CT#82	CP-183037	0550	2	D	Terminology alignment regarding support for interworking without N26	15.2.0
2018-12	CT#82	CP-183037	0551	3	F	Correction on reporting change of 3GPP PS data off UE status	15.2.0
2018-12	CT#82	CP-183037	0553	1	F	Correction on Uplink data status handling	15.2.0
2018-12	CT#82	CP-183037	0557	2	F	Correction of LADN information for generic UE configuration	15.2.0
2018-12	CT#82	CP-183037	0559		F	Management of the registration attempt counter and the attach attempt counter during inter-system change	15.2.0
2018-12	CT#82	CP-183037	0560	1	F	No direct security protection to 5GSM	1520
2018-12	CT#82	CP-183037	0562	1	F	No S-NSSAL for emergency PDU sessions	15.2.0
2018-12	CT#82	CP-183037	0562	1	F	S-NSSAL selection for S-NSSAL based concestion control	15.2.0
2010-12	CT#02	CD_103037	0505	2	F	UE in registered state without registration complete	15.2.0
2010-12	CT#02	CP 103037	0500	2 1	T E	Apparmal Cases in the LIE for initial registration	15.2.0
2010-12	CT#62	CP-163037	0509	1	Г Г	Abhormal Cases in the OE for initial registration	15.2.0
2018-12	CT#82	CP-183037	0571	2	F	Addition of 24.501 overview	15.2.0
2018-12	CT#82	CP-183037	0572	2	F	Service area restrictions applicable for PLMNs in registration area	15.2.0
2018-12	CT#82	CP-183037	0573	1	F	Periodically erase of service area restriction list	15.2.0
2018-12	CT#82	CP-183037	0574	1	F	Completion of mobile identity IE definition in messages	15.2.0
2018-12	CT#82	CP-183037	0575		F	Correction of erroneously encoded IE	15.2.0
2018-12	CT#82	CP-183037	0576	2	F	Correction for local release	15.2.0
2018-12	CT#82	CP-183037	0577	3	F	UAC - providing access identities for barring checks of AS triggered access attempts	15.2.0
2018-12	CT#82	CP-183037	0578	2	F	Correction for indicating 3GPP PS data off status	15.2.0
2018-12	CT#82	CP-183037	0580	1	F	UPSIs in UE STATE INDICATION	1520
2018-12	CT#82	CP-183037	0582	1	F	Initial registration for emergency PDLI session	15.2.0
2018-12	CT#82	CP-183037	0583	2	F		15.2.0
2010-12	CT#02	CD 102027	0505	2	Г Г	Correction on Maximum number of supported packet filters	15.2.0
2010-12	CT#02	CF-103037	0504		Г Г	T2594 and T2595 undeto in Timor table	15.2.0
2010-12	CT#02	CF-103037	0565	1	Г	Absormal asso for T2550	15.2.0
2018-12	CT#82	CP-183037	0586	1	F	Abnormal case for 13550	15.2.0
2018-12	C1#82	CP-183038	0590	1	F	No UPF Resources at PDU Session Establishment	15.2.0
2018-12	CT#82	CP-183038	0591	2	F	Clarifications on NAS level mobility management congestion control	15.2.0
2018-12	CT#82	CP-183038	0592	1	F	SOR over control plane in non-3GPP access	15.2.0
2018-12	CT#82	CP-183038	0593	1	F	Handling of target CN type by NAS upon redirection to E-	15.2.0
2018-12	CT#82	CP-183038	0594	3	С	Procedure for UDM-triggered UE parameters update	1520
2018-12	CT#82	CP-183038	0595	1	F	Sending of 5GSM canability IE and Maximum number of	15.2.0
	001			-	-	supported packet filters IE in in PDU SESSION ESTABLISHMENT REQUEST with request type "existing PDU session"	
2018-12	CT#82	CP-183038	0597	3	F	Correction of reference	15.2.0
2018-12	CT#82	CP-183038	0598	2	С	Clarification on 5GSM cause_#46 out of LADN service area	15.2.0
2018-12	CT#82	CP-183038	0604	2	С	Clarification on congestion control upon intersystem change	15.2.0
2018-12	CT#82	CP-183038	0605		F	Editorial correction related to LADN	15.2.0
2018-12	CT#82	CP-183038	0606	1	F	Correcting the structure of LADN related description	15.2.0
2018-12	CT#82	CP-183038	0607	2	F	N3GPP de-registration timers handling at long MM back-off	15.2.0
2018-12	CT#82	CP-183038	0608	1	F	time Abnormal cases in the network side for Configuration update	15.2.0
2018-12	CT#82	CP-183038	0600	1	F	prcedure.	1520
2010 12	CT#02	CD 102020	0000	1	F	produre.	15.2.0
2018-12	01#82	07-183038	0010	1	r T	IPv4v6.	15.2.0
2018-12	CT#82	CP-183038	0611	1	F	Correction to QoS rules verification during PDU session establishment.	15.2.0
2018-12	CT#82	CP-183038	0612	1	F	QoS rules verification during PDU session modification.	15.2.0
2018-12	CT#82	CP-183038	0613	1	F	QoS flow descriptions IE to be set by SMF mandatory in PDU establishment accept for initial request.	15.2.0

2018-12	CT#82	CP-183038	0614	1	В	MCS Indicator and Access Identity 2	15.2.0
2018-12	CT#82	CP-183038	0615	3	F	SUCI encoding and support of NAI format	15.2.0
2018-12	CT#82	CP-183038	0617	1	F	UAC for simultaneous access attempt triggers	15.2.0
2018-12	CT#82	CP-183038	0618	1	F	Reset of registration attempt counter	15.2.0
2018-12	CT#82	CP-183038	0619	-	F	Reset of service request attempt counter	15.2.0
2018-12	CT#82	CP-183038	0622		F	CN paging bandling in RRC inactive	15.2.0
2018-12	CT#82	CP-183038	0623	2	F	Procedure collision handling in paging and notification	15.2.0
2018-12	CT#82	CP-183039	0624		F	Resolution of editor's note on different TAL for 4G and for 5G	1520
2010 12	CT#92	CP 182020	0625		F	Correction on 'E' bit for delating OoS flow description	15.2.0
2010-12	CT#02	CF-103039	0023		Г Б	EMM perometers bandling for EBC interworking	15.2.0
2010-12	CT#02	CF-103039	0020	1	Г	Evilvi parameters handling for EFC interworking	15.2.0
2018-12	C1#82	CP-183039	0627	1	Г	not allowed	15.2.0
2018-12	CT#82	CP-183039	0628	2	F	Correction to SUPI definition due to NAI format	15.2.0
2018-12	CT#82	CP-183039	0629		F	Correct secured packet procedures	15.2.0
2018-12	CT#82	CP-183039	0630	2	F	Abnormal cases for EAP-based AKA	15.2.0
2018-12	CT#82	CP-183039	0631	1	F	CONFIGURATION UPDATE COMMAND with no parameters other than registraion requested.	15.2.0
2018-12	CT#82	CP-183039	0633	1	F	Correction of the UE-initiated de-registration procedure	15.2.0
2018-12	CT#82	CP-183039t	0634	1	F	Definition of a 5GMM cause for DNN subscription check failure	15.2.0
2018-12	CT#82	CP-183039	0635	1	F	Stopping T3511 after transitioning to 5GMM-CONNECTED	15.2.0
					_	mode	
2018-12	CT#82	CP-183039	0636	2	F	Clarification on storage of UE policy sections of multiple PLMNs	15.2.0
2018-12	CT#82	CP-183039	0637	1	F	Corrections to UE handling of reject with cause #13	15.2.0
2018-12	CT#82	CP-183039	0639	1	F	Clarification on the Selected EPS NAS security algorithms	15.2.0
2018-12	CT#82	CP-183039	0641	3	F	Handling errors due to missing QoS flow descriptions parameters for GBR QoS flows	15.2.0
2018-12	CT#82	CP-183039	0642	2	F	Clarification on inclusion of Requested NSSAI during periodic	15.2.0
2018-12	CT#82	CP-183039	0643	1	F	The release of the existing N1 NAS signalling connection after	15.2.0
2018-12	CT#82	CP-183039	0646	1	F	Always-on PDLL sessions and associated access	1520
2018-12	CT#82	CP-183039	0647	1	F	Clarification on the applicability of NSSAL to the EPLMNs	15.2.0
2010-12	CT#02	CP 182020	0647	1	L.	Clarification on the applicability of NOSAL to the EPEMINS	15.2.0
2010-12	0T#02	OF-103039	0040	1	Г Г	the EPLMNs	15.2.0
2018-12	01#82	CP-183039	0649	1	F	QFI and QRI values in 5GSM messages	15.2.0
2018-12	CT#82	CP-183039	0650	2	F	Corrections on operation mode selection	15.2.0
2018-12	C1#82	CP-183039	0651	1	F	Clarification on handling of PDU sessions associated with 3GPP access and non-3GPP access for interworking	15.2.0
2018-12	CT#82	CP-183039	0652		F	Clarifications on Configuration Update procedure	15.2.0
2018-12	CT#82	CP-183039	0653		F	Correction on timer T3511	15.2.0
2018-12	CT#82	CP-183040	0654	2	F	Clarification on Notification procedure	15.2.0
2018-12	CT#82	CP-183040	0655	1	F	Clarification on handling of invalid LADN DNN in registration	15.2.0
2018-12	CT#82	CP-183040	0657	1	F	Correction for LADN information IF	15.2.0
2018-12	CT#82	CP-183040	0659		F	Editorial correction for the generic UE configuration update	15.2.0
2019 12	CT#00	CD 192040	0660	1	F	LIE behavior in ATTEMPTING RECISTRATION state	15.2.0
2010-12	CT#62	CP-163040	0000	1	Г	UE DENAVIOLINI ATTEMPTING-REGISTRATION State	15.2.0
2018-12	01#82	CP-183040	1000	1	Г Г	UE-AIVIE SEIECTEU PLIVIN ID MISMATCH IN INAUTIVE STATE	15.2.0
2010-12	CT#62	CP-163040	0662	1	Г Г		15.2.0
2018-12	C1#82	CP-183029	0663	1	F	acknowledgement	15.2.0
2018-12	CT#82	CP-183040	0664	2	F	Lower layer indication on the establishment/release of user plane resources	15.2.0
2018-12	CT#82	CP-183040	0665	3	F	Setting the Uplink data status and Service type IEs after receiving a fallback indication from the lower layers	15.2.0
2018-12	CT#82	CP-183040	0666	2	F	Indication of a reason for failure in reactivating a PDU session	15.2.0
2018-12	CT#82	CP-183040	0667	1	F	Adding the case that UE initiates the registration procedure for	15.2.0
2018-12	CT#82	CP-183040	0668	1	F	Handling of network rejection with 5GSM cause values #50,	15.2.0
2018-12	CT#82	CP-183040	0669	2	F	Mobility and periodic registration update triggered by	15.2.0
2018-12	CT#82	CP-183040	0673	1	F	UE STATE INDICATION message delivered in a	15.2.0
2018-12	CT#82	CP-183040	0674	1	F	Correction to operator-defined access category criteria type values	15.2.0
2018-12	CT#82	CP-183040	0677	1	F	Exceptions for UEs configured for high priority access in handling T3396, T3584, and T3585	15.2.0
2018-12	CT#82	CP-183040	0678		D	Removal of non-seamless non-3GPP offload from definitions	15.2.0
2018-12	CT#82	CP-183040	0679	4	F	UE identifier provided during an initial registration procedure	15.2.0
2018-12	CT#82	CP-183040	0680	1	F	Revisons on PDU Session Eestablishment procedure	15.2.0

2018-12	CT#82	CP-183040	0682	1	F	UE registered for emergency services upon authentication failure	15.2.0
2018-12	CT#82	CP-183040	0683		F	Update the trigger of mobility registration update initiation	15.2.0
2018-12	CT#82	CP-183040	0684	2	F	Clarification on T3346 for registration procedure	15.2.0
2018-12	CT#82	CP-183040	0686	2	F	Support sending multiple payloads via Payload container	15.2.0
2018-12	CT#82	CP-183040	0687	1	F	Correction on LADN information handling	15.2.0
2018-12	CT#82	CP-183040	0688	2	F	Mapping of a NOTIFICATION message to an access category	15.2.0
2018-12	CT#82	CP-183041	0690	2	F	Transmission of SUCI in DEREGISTRATION REQUEST	15.2.0
2018-12	CT#82	CP-183041	0691	-	F	Clarification on missing subclause	15.2.0
2010 12	CT#92	CP 192041	0602		F	Stan T2516 when authentication reject received	15.2.0
2010-12	CT#02	CP 103041	0602	1	г Б	Limited convice and no SUD states in 5CMM instance for non	15.2.0
2010-12	01#02	CF-103041	0093	1	Г	2CPD account	15.2.0
2010 12	OT#00	CD 102044	0005	1	E	Oos flow and managed EDS bearer contaxt	15.2.0
2016-12	CT#62	CP-163041	0695	1	F		15.2.0
2018-12	CT#82	CP-183041	0696	1	F	Corrections on 5GSM IEs	15.2.0
2018-12	C1#82	CP-183041	0697	1	F	Correction on 5GSM congestion control	15.2.0
2018-12	CT#82	CP-183041	0698		F	Correction on PTI value release	15.2.0
2018-12	CT#82	CP-183041	0699	1	F	Corrections on QoS rules IE	15.2.0
2018-12	CT#82	CP-183041	0700	1	F	UE handling for semantic error in the QoS operation	15.2.0
2018-12	CT#82	CP-183041	0701		F	Resolution of editor's note on abnormal case handling for the	15.2.0
						UE-initiated UE state indication procedure	
2018-12	CT#82	CP-183041	0702	1	F	Resolution of editor's note on abnormal case handling for the	15.2.0
						network-requested UE policy management procedure	
2018-12	CT#82	CP-183041	0703		F	Resolution of editor's note on the information the N3IWF	15.2.0
						maintains for a registered UE	
2018-12	CT#82	CP-183041	0709	2	F	Congestion control	15.2.0
2018-12	CT#82	CP-183041	0711	1	F	Correction to authentication abnormal cases	15.2.0
2018-12	CT#82	CP-183041	0712	1	F	Corrections for non-3GPP access idle mode	15.2.0
2018-12	CT#82	CP-183041	0715	1	F	Integrity protection maximum data rate for LIL and DI	15.2.0
2018-12	CT#82	CP-1830/1	0717	1	F	General section for limited service state over non-3GPP	15.2.0
2010-12	01#02	CF-103041	0/1/	1	г		15.2.0
2018 12	CT#92	CD 1920/1	0719		Б		15.2.0
2010-12	CT#02	CF-103041	0710		Г	Change of ICLucius from ZE to Z4 and from ZE to Z5	15.2.0
2019-01	CT#82	00.400004	0000	0	-	Change of IEI values from 7E to 74 and from 7F to 75	15.2.1
2019-03	CT#83	CP-190084	0382	9	F	Correct Extended Local Emergency Numbers List use	15.3.0
	07/00	05 (0000)		-	_		1.5.0.0
2019-03	CT#83	CP-190084	0638	3	F	Clarification for abnormal case handling of registration	15.3.0
						procedure after inter-system change from S1 mode to N1	
0040.00	OTHOD	05 400004	0700	-	-		45.0.0
2019-03	CT#83	CP-190084	0706	6	F	Resolution on the editor's note on abnormal cases in the UE	15.3.0
0040.00	OT	05 400004	0704	2	F	for the PDU EAP message reliable transport procedure	45.0.0
2019-03	CT#83	CP-190084	0721	2	F	Clarification on inclusion of the Uplink data status IE in the	15.3.0
						SERVICE REQUEST message after an RRC fallback	
	07/00	05 (0000)			-	Indication	
2019-03	CT#83	CP-190084	0724	4	F -	Corrections to Annex D	15.3.0
2019-03	CT#83	CP-190084	0725	2	F	Update reference for UE policy control service	15.3.0
2019-03	CT#83	CP-190084	0728	1	F	Completion of correction for local release	15.3.0
2019-03	CT#83	CP-190084	0729	2	F	UAC - access attempt matching criteria of operator-defined	15.3.0
						access categoryt	
2019-03	CT#83	CP-190084	0731	2	F	Correcting the name of ITU-T Recommendation E.212	15.3.0
2019-03	CT#83	CP-190084	0732	2	F	5GSM - request type not included in PDU SESSION	15.3.0
						ESTABLISHMENT REQUEST	
2019-03	CT#83	CP-190084	0734	2	F	AMF rejecting PDU session establishment when the DNN is	15.3.0
						not subscribed	
2019-03	CT#83	CP-190084	0735	1	F	Correction for acknowledgement of extended emergency	15.3.0
						number list	
2019-03	CT#83	CP-190084	0741	3	F	Mobility between 5GS over non-3GPP access and EPS over	15.3.0
						3GPP access	
2019-03	CT#83	CP-190084	0743	1	F	PDU session status for IWK without N26	15.3.0
2019-03	CT#83	CP-190084	0744	2	F	EPS GUTI provided to lower laver	15.3.0
2019-03	CT#83	CP-190084	0745	1	F	Length of 5G-S-TMSI	15.3.0
2019-03	CT#83	CP-190084	0746	3	F	S1 UE security canability	15.3.0
2010-02	CT#83	CP-100084	07/7	2	F	Clarification on creating new OoS flowst	15 3 0
2019-03	CT#00	CP 100004	0750	ے۔ 1	L L	Correction to bondling of #50 and #51	15.3.0
2019-03		CP 100005	0750	1	r E	DUI to nation modification for amorgans: DDU acceleration	15.3.0
2019-03	01#83	00-190085	0752	1	Г	PD0 Session modification for emergency PD0 sessions.	15.3.0
2019-03	CT#83	CP-190085	0753	1	F	Removal of unneessary text.	15.3.0
2019-03	CT#83	CP-190085	0754	1	F	QoS flow description to be added in PDU session modification	15.3.0
						command	
2019-03	CT#83	CP-190100	0758		F	Correction of the erroneous length of EAP messageIE	15.3.0
2019-03	CT#83	CP-190085	0759	2	F	Update of SUCI encoding	15.3.0
2019-03	CT#83	CP-190085	0762		F	Minor corrections to TS 24.501	15.3.0
2019-03	CT#83	CP-190085	0763	2	F	Cleanup on support of multiple payloads for NAS transport	15.3.0
			1	1.			1
2019-03	CT#83	CP-190085	0765	1	F	Correction on initial NAS message protection	15.3.0

2019-03	CT#83	CP-190085	0767	1	F	Addition of the 5GSM cause IE in the PDU SESSION	15.3.0
2019-03	CT#83	CP-190085	0768	2	D	Mipor corrections for interworking	1530
2019-03	CT#83	CP-190085	0769	1	F	Change of "a wildcard DNN" to "the wildcard DNN"	15.3.0
2019-03	CT#83	CP-190085	0771	2	F	Adding missing abnormal cases for initial registration (UE side)	15.3.0
2019-03	CT#83	CP-190085	0772	2	F	Clarification for the use of the default value for T3512	15.3.0
2019-03	CT#83	CP-190085	0773	3	F	Removal of UE security capabilities from the S1 mode to N1 mode NAS transparent container	15.3.0
2019-03	CT#83	CP-190085	0774	2	F	Clarification on NSSAI inclusion mode after an inter-system	15.3.0
2019-03	CT#83	CP-190085	0780	1	F	Corrections to LIE policy section management result	1530
2019-03	CT#83	CP-190085	0785	1	F	Removal of abnormal case handling for collision between initial	15.3.0
_0.0 00	01.000	0	0.00	•	-	registration and paging or notification	
2019-03	CT#83	CP-190085	0788		D	DNN as a common IE	15.3.0
2019-03	CT#83	CP-190085	0791	2	F	Generic UE configuration update procedure during registration procedure	15.3.0
2019-03	CT#83	CP-190085	0792		D	References for NAS signalling connection recovery and a fallback indication from the lower layers	15.3.0
2019-03	CT#83	CP-190086	0796	1	F	Corrections on fallback indication from lower layers	15.3.0
2019-03	CT#83	CP-190086	0798	2	F	Correction to the notification procedure	15.3.0
2019-03	CT#83	CP-190086	0801	1	D	Correction on sub-clause numbering	15.3.0
2019-03	CT#83	CP-190086	0802	3	F	Handling of QoS flow description without valid EPS bearer contextt	15.3.0
2019-03	CT#83	CP-190086	0803		F	Correction on the lengths of 5GSM procedures	15.3.0
2019-03	CT#83	CP-190086	0804	2	F	Corrections on Mapped EPS bearer context IE	15.3.0
2019-03	CT#83	CP-190086	0805	2	F	Handling of missing QoS flow description and QoS rule	15.3.0
2019-03	CT#83	CP-190086	0806	3	F	UE re-registration following UE parameters update	15.3.0
2019-03	CT#83	CP-190086	0807	1	F	Clarification on providing NSSAI to the lower layer	15.3.0
2019-03	CT#83	CP-190086	0808	1	F	Adjustment of relevent timers when T3346 is included in the 5GMM message	15.3.0
2019-03	CT#83	CP-190086	0809	1	F	Clarification on PDU Session Modification procedure	15.3.0
2019-03	CT#83	CP-190086	0810	1	F	Clarification on the behaviors of UE and SMF during the inter- system change	15.3.0
2019-03	CT#83	CP-190086	0812	2	F	Several corrections to messages and IEs	15.3.0
2019-03	CT#83	CP-190086	0814	2	F	Correction to IEI values	15.3.0
2019-03	CT#83	CP-190086	0815	2	F	Resolution on editor's notes on whether explicit start and stop indications for SMS over NAS is needed	15.3.0
2019-03	CT#83	CP-190086	0817		F	Correction to the definition of UE STATE INDICATION message content	15.3.0
2019-03	CT#83	CP-190086	0819	1	F	Reference corrections on UE's state change due to congestion control	15.3.0
2019-03	CT#83	CP-190086	0820	3	F	Correction to behaviour upon receipt of 5GMM reject cause for a UE in single-registration mode	15.3.0
2019-03	CT#83	CP-190086	0821	1	F	Informing the UE about the integrity protection failure at S1 to N1 mode change	15.3.0
2019-03	CT#83	CP-190086	0823	1	F	Handling of abnormal case when UE gets UE policies with incorrect PLMN ID	15.3.0
2019-03	CT#83	CP-190087	0824		F	Abnormal case handling for cause#72	15.3.0
2019-03	CT#83	CP-190087	0829	3	F	Correction on handling of invalid PSI	15.3.0
2019-03	CT#83	CP-190087	0831		F	Correction for missing 5Qls	15.3.0
2019-03	CT#83	CP-190174	0833	4	F	Possible criterion for the selection of the requested NSSAI	15.3.0
2019-03	CT#83	CP-190087	0834	2	F	Skip barring checks during fallback handling	15.3.0
2019-03	CT#83	CP-190087	0835	1	F	Correct missing Non-3GPP NW policies IE	15.3.0
2019-03	CT#83	CP-190087	0836	1	F	5G-GUTI provided to lower layer	15.3.0
2019-03	CT#83	CP-190087	0838	1	F	Handling of abnormal authentication errors	15.3.0
2019-03	CT#83	CP-190087	0839	1	F	Update the conditions on the AMF to provide the configured NSSAI to the UE	15.3.0
2019-03	CT#83	CP-190087	0844	1	F	Rejected NSSAI for current registration area	15.3.0
2019-03	CT#83	CP-190087	0845	1	F	Removal of Editor's note on home network public key and home network public key identifier update and removal of	15.3.0
2019-03	CT#83	CP-190087	0846	1	F	protection scheme identifier Update of PDU session authentication and authorization	15.3.0
2019-03	CT#83	CP-190087	0847	2	F	messages Update of error handling for Mapped EPS bearer contexts IE	15.3.0
2019-03	CT#83	CP-190087	0849	1	F	Correction for Cause code #7	15.3.0
2019-03	CT#83	CP-190087	0854	1	F	3GPP PS Data Off UE status change and congestion control in AMF - alternative 2	15.3.0
2019-03	CT#83	CP-190087	0855	1	F	Retransmission of PDU SESSION ESTABLISHMENT REQUEST	15.3.0
2019-03	CT#83	CP-190087	0856	1	F	Retransmission of PDU SESSION MODIFICATION REQUEST	15.3.0
2019-03	CT#83	CP-190087	0857	1	F	Retransmission of PDU SESSION RELEASE REQUEST	15.3.0
2019-03	CT#83	CP-190087	0859	1	F	Correction in EAP handling	15.3.0
2019-03	CT#83	CP-190087	0860		F	Issue in SNN	15.3.0

2019-03	CT#83	CP-190087	0862	2	F	Temporary identity allocation	15.3.0
2019-03	CT#83	CP-190088	0863		F	Initial maximum number of packet filters for associated PDU	15.3.0
						session	
2019-03	CT#83	CP-190088	0864	1	F	SUCI to be used for the registration for emergency services	15.3.0
2019-03	CT#83	CP-190088	0867		F	Correction to the inclusion of requested QoS rule and	15.3.0
						requested QoS flow descripstion in PDU session modification	
2010-03	CT#83	CP-100088	0868	1	F	Wrong message name	1530
2019-03	CT#03	CP 10008	0000	2	Г Б		15.3.0
2019-03	01#63	CP-190066	0070	2	Г Г		15.3.0
2019-03	C1#83	CP-190088	0873	1	F	procedure	15.3.0
2019-03	CT#83	CP-190088	0875	2	F	Upper layer request while T3540 is running	15.3.0
2019-03	CT#83	CP-190088	0876	1	F	Encoding of Routing indicator set to default value 0	15.3.0
2019-03	CT#83	CP-190088	0877	2	F	OS Id information element	15.3.0
2019-03	CT#83	CP-190088	0878	1	F	Clarification on deregistration procedure.	15.3.0
2019-03	CT#83	CP-190088	0880	1	D	Correction of the use of word "wants"	15.3.0
2019-03	CT#83	CP-190088	0883	1	F	Handling on errors of QoS flow description operations	15.3.0
2019-03	CT#83	CP-190088	0885	1	F	Addressing missing scenarios and providing other clarifications	1530
2010-00	01#00	01 100000	0000	1	1	related to fallback indication	10.0.0
2019-03	CT#83	CP-190088	0888	2	F	Handling when the UE indicated security capabilities are	15.3.0
2010-03	CT#83	CP-100088	0880	1	F	Correction in LIE-initiated de-registration procedure initiation	1530
2019-03	CT#03	CP 10008	0003	1	Г Б	Penerting QoS error when a delete or modify operation in	15.3.0
2019-03	01#03	CE-190088	0091	1	1,	reporting use error when a delete or modify operation IS	10.3.0
2040.00	07400	CD 400000	0000	1	E		45.0.0
2019-03	01#83	CP-190088	0892	1	F	Presence of the precedence and QFI fields in QoS rules	15.3.0
2019-03	CT#83	CP-190088	0894	2	F	Mandating UE sending registration complete for SOR	15.3.0
2019-03	CT#83	CP-190088	0895	1	F	Improvement on 5G-GUTI allocation after network triggered service request	15.3.0
2019-03	CT#83	CP-190088	0899	2	F	Clarification on the authorized QoS rule modification	15.3.0
2019-03	CT#83	CP-190089	0901	1	F	Clarification on congestion control upon intersystem change	15.3.0
2019-03	CT#83	CP-190089	0902	4	F	Default EPS bearer associates with the default OoS rule	1530
2010-03	CT#83	CP-100080	0002	2	F	NSSAL inclusion mode, EPI MNs, and registration area	15.3.0
2010 03	CT#93	CP 100080	0010	2	F	5CSM mossages for a DDL sossion associated with non	15.3.0
2019-03	01#03	CF-190069	0910	Z	Г	2CPD access exchanged via 2CPD access	15.5.0
2010.02	CT#02	CD 100090	0011	1	Б	3GPP access exchanged via 3GPP access	15 2 0
2019-03	01#03	CF-190069	0911	1	Г	inactive mode reselecting an EPLMN cell	15.5.0
2019-03	CT#83	CP-190089	0912	1	F	Change of N1 mode capability on UE mode change or on IMS	15.3.0
2010-03	CT#83	CP-100080	001/	1	F	Inclusion of a 5G-GUTI manned from the valid 4G-GUTI in the	1530
2019-03	01#05	01-190009	0314	1	1	REGISTRATION REQUEST message during an initial	15.5.0
				-	_	registration is not available in Rel-15	
2019-03	CT#83	CP-190209	0915	3	F	Correction to the REGISTRATION REQUEST msg when the Payload container IE is included	15.3.0
2019-03	CT#83	CP-190089	0917	1	F	Correction to the length of the IMEISV	15.3.0
2019-03	CT#83	CP-190089	0918	1	F	Correcton on handing of downlink signalling and data for non-	1530
2010 00	01#05	01-100000	0310	1	1	3GPP PDI I session	10.0.0
2019-03	CT#83	CP-190089	0919		F	Correction to the Payload container IF	1530
2010-03	CT#83	CP-100080	0010		F	Correction to several 5GMM IEs	15.3.0
2019-03	CT#03	CP 10009	0920	1	Г Б	Monned configured S NSSAI from the S NSSAI(a) of the	15.3.0
2019-03	01#03	CF-190069	0921	1	1,	HPLMN	15.5.0
2019-03	CT#83	CP-190089	0922		F	Use of stored NSSAI inclusion mode during initial registrationt	15.3.0
2019-03	CT#83	CP-190089	0923		F	Correction to the Operator-defined access category definitions	15.3.0
0040.05	07.000	00.465555	0007	-	-		45.0.0
2019-03	01#83	CP-190089	0925	1	F -	NON-DELIVERY OF POU SESSION ESTABLISHMENT ACCEPT	15.3.0
2019-03	CT#83	CP-190089	0926	1	F	Correction to Service area list IE Type of list "11"	15.3.0
2019-03	CT#83	CP-190089	0927	1	F	Setting of 5GSM cause value #36 regular deactivation	15.3.0
2019-03	CT#83	CP-190092	0929	3	F	Local release	15.3.0
2019-03	CT#83	CP-190089	0930	1	F	Adding the UE local configuration option in 24.501	15.3.0
2019-03	CT#83	CP-190089	0933		F	Handling of Replayed S1 UE security capabilities IE in Security	15.3.0
2019-03	CT#83	CP-190089	0934	2	F	Clarifications on use of PCF-provided PTI for UE policies	15.3.0
2019-03	CT#83	CP-190101	0720	1	F	Initiation of Service Request after reception of Notification over non-3GPP while the UE is in 5GMM-CONNECTED mode with PRC inactive indication in the 3GPP access	16.0.0
2019-03	CT#83	CP-190106	0730	4	В	SINE_5G: Back-off control and retry restriction mechanisms in 5GS	16.0.0
2010.02	CT#83	CP-100101	0722	1	F	Transfer of a PDU session among 5G-ANs and DNN	16.0.0
2010-03	CT#00	CP_100101	0720	5	F	Handling for OoS Flow status synchronization foilure	16.0.0
2019-03	CT#00	CP 100101	0139	5	L.	Clarification on rejected NSSAL for the DLMN	16.0.0
2019-03	OT#03	CF-190101	0700	J 1	Г Б		10.0.0
2019-03	U1#83	CP-190101	0782	1	r	4G-GUTI as additional guti in initial registration and UE holds	0.0.0

One-Bot CF-B01010 OP78 I I Dimension Dimension Dimension Display CF-B01010 CF-B010100 CF-C010000000000000000000000000000000000	2019-03	CT#83	CP-190101	0786	1	D	Consistent description on release of N1 NAS signalling	16.0.0
2019-03 CT#83 CP-190101 U789 I P Hending of indication to the SGM subjaye in case of SGM. 2019-03 CT#83 CP-190101 0787 I F Alignment for the notification procedure 16.0.0 2019-03 CT#83 CP-190101 0840 I New SGI to Chancer Farmework for Upinks Streaming 16.0.0 2019-03 CT#83 CP-190101 0840 I Alignment of terms of configured NSSAI and allowed NSSAI 16.0.0 2019-03 CT#83 CP-190101 0841 I Provision of the lower layers fills. 16.0.0 2019-03 CT#83 CP-190101 0841 I Provision of the lower layers fills. 16.0.0 2019-03 CT#83 CP-190101 0851 I Dudated value values fills. 16.0.0 2019-03 CT#83 CP-190101 0851 I Dudated value values fills relative values fills. 16.0.0 2019-03 CT#83 CP-190101 0851 I Dudated value values values fills. 16.0.0 2019-03 CT#83 <td>2010-03</td> <td>CT#83</td> <td>CP-100101</td> <td>0787</td> <td>1</td> <td>F</td> <td>Connection</td> <td>16.0.0</td>	2010-03	CT#83	CP-100101	0787	1	F	Connection	16.0.0
Inters Chillion Image and towarded to SMF due to service area restriction. 2019-03 CT483 CP-190101 0840 F Alignment for the notification procedure 16.0.0 2019-03 CT483 CP-190101 0840 F Update the conditions on the AMF to provide an allowed 16.0.0 2019-03 CT483 CP-190101 0841 F Alignment of the offault shared MSA1 and allowed NSA1 16.0.0 2019-03 CT483 CP-190101 0841 F Commotion to mellow configuration update procedure 16.0.0 2019-03 CT483 CP-190101 0851 F Update d'validation update procedure 16.0.0 2019-03 CT483 CP-190101 0866 F Update d'validation on allowed NSA1 storage in Non Volatile Memory 16.0.0 2019-03 CT483 CP-190101 0866 F Update divalidation and allowed NSA1 storage in Non Volatile Memory 16.0.0 2019-03 CT483 CP-190101 0866 F Carrier divalidation and allowed reduet procedure 16.0.0 2019-03 CT483 CP-190101 0866 </td <td>2019-03</td> <td>CT#83</td> <td>CP-190101</td> <td>0789</td> <td>4</td> <td>F</td> <td>Handling of indication to the 5GSM sublayer in case of 5GSM</td> <td>16.0.0</td>	2019-03	CT#83	CP-190101	0789	4	F	Handling of indication to the 5GSM sublayer in case of 5GSM	16.0.0
2019-03 CF433 CP-190101 0803 L New SQB for Enhanced Framework for Uplink Streaming 16.0.0 2019-03 CF433 CP-190101 0840 F Update the conditions on the AME to provide an allowed 16.0.0 2019-03 CF433 CP-190101 0841 I F Aligoment of tems of configured INSA I and allowed NSA I and allowee I and I allowee I and I allowee I and I allowe	2010 00	01#00		0100	•	-	message not forwarded to SMF due to service area restrictions	10.0.0
2019-03 CT#33 CP-190101 0840 F Update the conditions on the AMF to provide an allowed 16.0.0 2019-03 CT#33 CP-190101 0841 I Alignment of terms of conjuged NSSA and allowed NSSA 16.0.0 2019-03 CT#33 CP-190101 0841 I Provision of NSSA information to the lower layers 16.0.0 2019-03 CT#33 CP-190101 0842 I Provision of NSSA information to the lower layers 16.0.0 2019-03 CT#33 CP-190101 0851 I Current of terms operations to the lower layer services identities 1 and 2 16.0.0 2019-03 CT#33 CP-190101 0865 I F Update of validity conditions for access identities 1 and 2 16.0.0 2019-03 CT#33 CP-190101 0865 I F Ubdate of validity conditions for access identities 1 and 2 16.0.0 2019-03 CT#33 CP-190101 0865 I Correction on allowed NSSA storage in Non Volatile Memory 16.0.0 2019-03 CT#33 CP-190101 0867 I Correction on allowed NSSA storage	2019-03	CT#83	CP-190101	0797	3	F	Alignment for the notification procedure	16.0.0
2019-03 CT#83 CP-190101 0840 F Update the conditions on the AMF to provide an allowed 16.0.0 2019-03 CT#03 CP-190101 0841 F Alignment of terms of configured NSAI and allowed NSAI 16.0.0 2019-03 CT#03 CP-190101 0842 I F Correction for the UE configuration to pdate procedure 16.0.0 2019-03 CT#03 CP-190101 0851 I Correction for the UE configuration update procedure 16.0.0 2019-03 CT#03 CP-190101 0851 I Correction for the UE configuration update procedure 16.0.0 2019-03 CT#03 CP-190101 0866 I F UE state at lower layer failure on the WI side at initial 16.0.0 2019-03 CT#03 CP-190101 0866 I F UE state at lower layer failure on the WI side at initial 16.0.0 2019-03 CT#03 CP-190101 0866 F Correction of SMB payload container 16.0.0 2019-03 CT#03 CP-190101 0867 F Correction of SMB payload container 16.0.0 2019-03<	2019-03	CT#83	CP-190108	0830		В	New 5QIs for Enhanced Framework for Uplink Streaming	16.0.0
2019-03 CT#33 CP-190101 08411 F Night based upon the default S-NSSA(s) 2019-05 CT#33 CP-190101 08421 I Provision of NSSA information to the lower layers 16.0.0 2019-05 CT#33 CP-190101 08421 I Correction for the UIC engineering encodeure 16.0.0 2019-05 CT#33 CP-190101 08451 I Contention for the UIC engineering encodeure 16.0.0 2019-05 CT#33 CP-190101 08651 I Distribution on allowed MSX storage in Nov Valiel Memory 16.0.0 2019-05 CT#33 CP-190101 08661 F Correction on allowed MSX storage in Nov Valiel Memory 16.0.0 2019-05 CT#33 CP-190101 08671 F Correction on allowed MSX storage in Nov Valiel Memory 16.0.0 2019-05 CT#33 CP-190101 08672 F Correction on allowed MSX storage in Nov Valiel Memory 16.0.0 2019-05 CT#33 CP-190101 0867 F Correction on allowed MSK at instake in nov Valiel Markatin Stare instake interestin MSK at instake in nove Valiel	2019-03	CT#83	CP-190101	0840		F	Update the conditions on the AMF to provide an allowed	16.0.0
2019-03 CT-83 CP-190101 1942 I Provision of NS&AI information to the lower layers 16.0.0 2019-03 CT-83 CP-190101 1942 I Provision of NS&AI information to the lower layers 16.0.0 2019-03 CT-83 CP-190101 1944 I Correction for the UE configuration update procedure 16.0.0 2019-03 CT-83 CP-190101 1952 I Cubrate of validity conditions for access identifies 1 and 2 16.0.0 2019-03 CT-83 CP-190101 19652 I Clarification on allowed NSAI storage in Non Volatile Mamory 16.0.0 2019-03 CT-83 CP-190101 19661 F UE state tower layer failure on the NW side at initial 16.0.0 2019-03 CT-83 CP-190101 1987 I Correction of Wang preference 16.0.0 2019-03 CT-83 CP-190101 1987 I Correction of dwang preference 16.0.0 2019-03 CT-83 CP-190101 1987 I Correction of dwang preference 16.0.0 2019-04 CT-83	0040.00	OT "00	05 400404	0044	1	F	NSSAI based upon the default S-NSSAI(s)	10.0.0
All Bias Chinas China	2019-03	CT#83	CP-190101	0841	1	F	Alignment of terms of configured NSSAI and allowed NSSAI	16.0.0
201630 Cirres Cires Cires <td>2019-03</td> <td>CT#03</td> <td>CP-190101</td> <td>0042</td> <td>1</td> <td>Г Г</td> <td>Correction for the LIE configuration undate precedure</td> <td>16.0.0</td>	2019-03	CT#03	CP-190101	0042	1	Г Г	Correction for the LIE configuration undate precedure	16.0.0
2018-03 CT#83 CF-180101 0851 1 1 Update of validity conditions for access identities 1 and 2 16.0.0 2019-03 CT#83 CF-190101 0862 F SP procedure for emergency services failback when T3525 16.0.0 2019-03 CT#83 CF-190101 0866 I F Certrocodure for emergency services failback when T3525 16.0.0 2019-03 CT#83 CF-190101 0866 I F Contraction on allowed NSSAI storage in Non Volatile Memory 16.0.0 2019-03 CT#83 CF-190101 0896 F Contrection of wong reference 16.0.0 2019-03 CT#83 CF-190101 0896 F Correction on allowed inflationation 16.0.0 2019-03 CT#83 CF-190101 0896 F Correction on allowed signaling connection establishment 16.0.0 2019-03 CT#83 CF-190101 0894 F Correction on MIXOS signaling connection establishment 16.0.0 2019-03 CT#83 CF-190101 09924 F Correction on MIXOS signaling	2019-03	CT#83	CP-190101	0843	1	г F	Clarification to definition of "ongoing service"	16.0.0
2019-03 CT#83 CP-190101 0852 P SR procedure for emergency services failback when T3525 16.0.0 2019-03 CT#83 CP-190101 0865 P Clarification on allowed NISSAI storage in Non Volatile Memory 16.0.0 2019-03 CT#83 CP-190101 0866 P F Ustate at lower layer failure on the NW side at initial 16.0.0 2019-03 CT#83 CP-190101 0869 F Contrect of SMS apyload container 16.0.0 2019-03 CT#83 CP-190101 0896 F Contrect on SMS apyload container 16.0.0 2019-03 CT#83 CP-190101 0897 I Correction on short mapped EPS security context for EPC 16.0.0 2019-03 CT#83 CP-190101 0997 I Correction on short Maxue sets 16.0.0 2019-03 CT#83 CP-190101 0997 I Correction on short Maxue sets 16.0.0 2019-03 CT#83 CP-190101 0997 I Correction on SGMM acuse sets 16.0.0 2019-03 CT#8	2019-03	CT#83	CP-190101	0851	1	F	Update of validity conditions for access identities 1 and 2	16.0.0
Crites CP-190101 OB65 [2] F Clarification an allowed NSSAI storage in Non Volatile Memory [16.0.0] 2019-03 CT#83 CP-190101 0866 [1] F Clarification an allowed NSSAI storage in Non Volatile Memory [16.0.0] 2019-03 CT#83 CP-190101 0866 [1] F Content of SMS payload container 16.0.0 2019-03 CT#83 CP-190101 0881 [1] D Correction of Wong reference 16.0.0 2019-03 CT#83 CP-190101 0886 [1] C Correction on allowed NSSAI storage in Non Volatile Memory [16.0.0] 2019-03 CT#83 CP-190101 0886 [1] C Correction on allowed NSSAI storage in Non Volatile Memory [16.0.0] 2019-03 CT#83 CP-190101 0996 [1] F Correction on Allowed NSSAI storage in Non Volatile Memory [16.0.0] 2019-03 CT#83 CP-190101 0904 [2] F Correction on Allowing inconnection establishment [16.0.0] 2019-03 CT#83 CP-190101 0904 [2] F Correction on MANS signaling connection establishment [16.0.0] 2019-03 CT#83 CP-190101	2019-03	CT#83	CP-190101	0852	1	F	SR procedure for emergency services fallback when T3525	16.0.0
2019-03 CT#83 CP-190101 0866 1 F UE state at lower layer failure on the NW side at initial 16.0.0 2019-03 CT#83 CP-190101 0866 1 F UE state at lower layer failure on the NW side at initial 16.0.0 2019-03 CT#83 CP-190101 0868 1 F Correct mistake in case (i) in service request procedure 16.0.0 2019-03 CT#83 CP-190101 0886 2 F Clarifications related to failback indication 16.0.0 2019-03 CT#83 CP-190101 0887 F Correction on showing mapped EPS security cortext for EPC 16.0.0 2019-03 CT#83 CP-190101 0897 I F Correction on SOMM cause #55 16.0.0 2019-03 CT#83 CP-190101 0892 F Correction on SOMM cause #55 16.0.0 2019-03 CT#83 CP-191101 0924 F Correction on SOMM cause #55 16.0.0 2019-03 CT#83 CP-191137 0938 B Intervorking ditement PDU session ton VM						-	timer running	
2010-03 CT#83 CP-190101 0868 I F UE state at lower layer failure on the NW side at initial 16.0.0 2019-03 CT#83 CP-190101 0878 F Content of SMS payload container 16.0.0 2019-03 CT#83 CP-190101 0881 D Correction of wrong reference 16.0.0 2019-03 CT#83 CP-190101 0868 F Contrection on deriving mapped EPS security context for EPC 16.0.0 2019-03 CT#83 CP-190101 0896 F Correction on abnormal case handing at AMF for registration 16.0.0 2019-03 CT#83 CP-190101 0905 F Correction to N1X MS signaling connection establishment 16.0.0 2019-03 CT#83 CP-190101 0905 F Correction to N1X MS signaling connection establishment 16.0.0 2019-03 CT#83 CP-190110 0905 F Correction to N1X MS signaling connection establishment 16.0.0 2019-06 CT#84 CP-191137 0937 B Essentry Connestin to TX MS signaling contestin the Signaling to tX MS<	2019-03	CT#83	CP-190101	0865	2	F	Clarfication on allowed NSSAI storage in Non Volatile Memory	16.0.0
Open Cirr#3 CP-190101 0868 F Content of SMS payload container 16.0.0 2019-03 CT#83 CP-190101 0872 I F Correct mistake in case (i) in service request procedure 16.0.0 2019-03 CT#83 CP-190101 0868 2 F Clarifications related to fallback indication 16.0.0 2019-03 CT#83 CP-190101 0897 I F Correction on shoring mapped EPS security context for EPC 16.0.0 2019-03 CT#83 CP-190101 0987 I F Correction on SMCM cause #65 16.0.0 2019-03 CT#83 CP-190101 0924 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-191101 0924 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-191101 0932 F Correction on SGMM cause #65 16.1.0 2019-03 CT#84 CP-191137 0937 B Cerrection on SGM cause #65 16.1.0 2019-06 CT#84	2019-03	CT#83	CP-190101	0866	1	F	UE state at lower layer failure on the NW side at initial	16.0.0
2019-03 CT#83 CP-190101 0869 F Content of SMS payload container 16.0.0 2019-03 CT#83 CP-190101 0861 D Carrection of wrong reference 16.0.0 2019-03 CT#83 CP-190101 0868 F Clarifications related to fallback indication 16.0.0 2019-03 CT#83 CP-190101 0896 F Clarifications related to fallback indication 16.0.0 2019-03 CT#83 CP-190101 0996 F Clarification on MICO mode 16.0.0 2019-03 CT#83 CP-190101 0905 F Correction N 1N XNS signalling connection establishment 16.0.0 2019-03 CT#83 CP-1911147 0936 B Interworking of Enhomet PDU session to Ethernet PDN 16.1.0 2019-06 CT#84 CP-191137 0937 2 B PE1 for 5G-CRS and PN-CRG 16.1.0 2019-06 CT#84 CP-191137 0938 B Session-TMBR for PDU session to #CH488 16.0.2 2019-06 CT#84 CP-191132							registration	
2010-03 CT#83 CP-190101 0872 I I Correction of wong reference 16.0.0 2010-03 CT#83 CP-190101 0886 2 F Clarifications related to fallback indication 16.0.0 2019-03 CT#83 CP-190101 0896 F Correction on dwinying mapped EPS security context for EPC 16.0.0 2019-03 CT#83 CP-190101 0997 F Correction on shormal case handing at AMF for registration 16.0.0 2019-03 CT#83 CP-190101 0994 F Correction on SMM cause #65 16.0.0 2019-03 CT#83 CP-190101 0994 F Correction on SMM cause #65 16.0.1 2019-03 CT#83 CP-190113 0938 B Untervorking of Ehemet PDU session to Ethernet PDN 16.1.0 2019-06 CT#84 CP-191137 0933 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0944 F Handing of SO NAS security contexts 16.1.0 2019-06 C	2019-03	CT#83	CP-190101	0869		F	Content of SMS payload container	16.0.0
2019-03 CT#83 CP-190101 0881 D Constitution of wring reference 16.0.0 2019-03 CT#83 CP-190101 0886 F Correction on deriving mapped EPS security context for EPC 16.0.0 2019-03 CT#83 CP-190101 0897 I F Correction on abnormal case handing at AME for registration 16.0.0 2019-03 CT#83 CP-190101 0995 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-190101 0995 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-190101 0995 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-191137 0937 B De-implementation of CR468 16.0.2 2019-06 CT#84 CP-191137 0937 B PEI for 5G-CR6 and FN-CR6 16.1.0 2019-06 CT#84 CP-191137 0939 B Session-TM4R for PDU session tw-SGAN 16.1.0 2019-06 CT#84 CP-191132 0944	2019-03	CT#83	CP-190101	0872	1	F	Correct mistake in case (i) in service request procedure	16.0.0
2019-03 CT#83 CP-190101 0866 2 P Califications fielded Dialitack findcation 2019-03 CT#83 CP-190101 0866 2 P Correction on anomal case handing at AMF for registration 16.0.0 2019-03 CT#83 CP-190101 0897 1 P Correction on anomal case handing at AMF for registration 16.0.0 2019-03 CT#83 CP-190101 0904 P Clarification on MICO mode 16.0.0 2019-03 CT#83 CP-190101 0904 P Correction to N1 NAS signalling connection establishment 16.0.0 2019-03 CT#83 CP-190111 0934 F Correction to N1 NAS signalling connection establishment coses identities for establishment coses identities for establishment coses identities for establishment coses identities for establishment identicies of the properticies of anomal case intervorking of thermet PDU session tows for properity identicies of advation of RRC 16.1.0 2019-06 CT#84 CP-191132 0946 F Correction on the registration of deviation of RRC 16.1.0 2019-06 CT#84 CP-1	2019-03	CT#83	CP-190101	0881	2	D	Correction of wrong reference	16.0.0
Check Check <th< td=""><td>2019-03</td><td>CT#83</td><td>CP-190101</td><td>0000</td><td>2</td><td>Г F</td><td>Correction on deriving manned EPS security context for EPC</td><td>16.0.0</td></th<>	2019-03	CT#83	CP-190101	0000	2	Г F	Correction on deriving manned EPS security context for EPC	16.0.0
2019-03 CT#93 CP-190101 0997 I F Correction on abnormal case handing at AMF for registration 16.0.0 2019-03 CT#83 CP-190101 0904 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-190101 0924 F Correction on SGMM cause #65 16.0.0 2019-03 CT#83 CP-190101 0924 F Correction on SGMM cause #65 16.0.0 2019-06 CT#83 CP-19117 0937 C B De-implementation of CR468 16.0.2 2019-06 CT#84 CP-191137 0937 2 B PE1 for SG-CRG and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0938 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0944 F Handing of FDU session tpe 16.1.0 2019-06 CT#84 CP-191132 0946 F Clarifications on the validity of access identities for derivation of RRC 16.1.0 2019-06 CT#84	2019-03	01#05	CF-190101	0090		I.	interworking in connected mode	10.0.0
2019-03 CT#83 CP-190101 0904 2 r Clarification on MICO mode. 100.0000000000000000000000000000000000	2019-03	CT#83	CP-190101	0897	1	F	Correction on abnormal case handing at AMF for registration	16.0.0
2019-03 CT#93 CP-190101 0905 F Correction to SGMM cause #65 16.0 2019-03 CT#93 CP-190101 0924 F Correction to N1 NAS signalling connection establishment 16.0.1 2019-03 CT#83 C De-implementation of CR468 16.0.1 2019-06 CT#84 CP-191147 0936 B Interworking of Ethmert PDU session to Ethmert PDN 16.1.0 2019-06 CT#84 CP-191137 0937 2 B PE1 tor 50-CR6 and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0938 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of SQ NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of SQ NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F	2019-03	CT#83	CP-190101	0904	2	F	Clarification on MICO mode	16.0.0
2019-03 CT#83 CP-190101 0924 F Correction to N1 NAS signalling connection establishment 16.0.1 2019-03 CT#83 De-implementation of CR468 16.0.1 2019-06 CT#84 CP-191147 0936 B Interworking of Ethernert PDU session to Ethernet PDN 16.1.0 2019-06 CT#84 CP-191137 0937 B PEI for 5G-CR6 and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0938 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of SG NAS Security contexts 16.1.0 2019-06 CT#84 CP-191132 0948 F Handling of PDJ session type 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Correction on T339	2019-03	CT#83	CP-190101	0905		F	Correction on 5GMM cause #65	16.0.0
2019-03 CT#83 De-implementation of CR468 16.0.1 2019-06 CT#84 CP-191147 0936 B Interworking of Ethermet PDU session to Ethermet PDN 16.1.0 2019-06 CT#84 CP-191137 0937 2 B PEI for 5G-CRG and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0938 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0944 5 F Handling of SDAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0944 5 F Handling of SDAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0944 7 Precedence between access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0955 <	2019-03	CT#83	CP-190101	0924		F	Correction to N1 NAS signalling connection establishment	16.0.0
2019-06 CT#84 CP-191147 Og86 B Intervorking of Ethemet PDU session to Ethernet PDN 16.1.0 2019-06 CT#84 CP-191137 Og37 2 B PEI for 56-CRC and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 Og37 B PEI for 56-CRC and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 Og39 B Session-TMBR for PDU session in VP-GAN 16.1.0 2019-06 CT#84 CP-191132 Og46 F Handling of PDU session in VP-GAN 16.1.0 2019-06 CT#84 CP-191132 Og46 F Handling of PDU session in VP-GAN 16.1.0 2019-06 CT#84 CP-191132 Og46 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 Og45 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 Og85 F Correction on T3396 16.1.0 2019-06 CT#84 CP-191123 Og851	2019-03	CT#83					De-implementation of CR468	16.0.1
2019-06 CT#84 CP-191147 0936 B Interworking of Ethernet PD session to Ethernet PDN 16.1.0 2019-06 CT#84 CP-191137 0937 2 B PEI for 5G-CRG and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0939 B Session-TMBR for PDU session in W-5GAN 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of FDU session type 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of PDU session type 16.1.0 2019-06 CT#84 CP-191132 0948 I A Precedence between access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Correction on Ta399 16.1.0 2019-06 CT#84 CP-191122 0953 I A Correction on Ta390 16.1.0 2019-06 CT#84 CP-191122 095	2019-03	CT#83					Completion of de-implementation of CR468	16.0.2
2019-06 CT#84 CP-191137 0937 2 B PEI for 5G-CRG and FN-CRG 16.1.0 2019-06 CT#84 CP-191137 0939 B Usage of ACS Information PCO parameter 16.1.0 2019-06 CT#84 CP-191132 0939 B Session-TMBR for PDU session in W-SGAN 16.1.0 2019-06 CT#84 CP-191132 0946 I F Handling of 5G NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0946 I F Handling of 5G NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Correction on the descriptions of SGSM parameters and 16.1.0 2019-06 CT#84 CP-191122 0953 I A Correction on the descriptions of SGSM parameters and 16.1.0 2019-06 CT#84 CP-191122 0957 I A B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.	2019-06	CT#84	CP-191147	0936		В	Interworking of Ethernet PDU session to Ethernet PDN connection	16.1.0
2019-06 CT#84 CP-191137 0938 B Usage of ACS information PCO parameter 16.1.0 2019-06 CT#84 CP-191137 0938 B Session-TMBR for PDU session In W-SGAN 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of SO NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0944 F Handling of DDU session type 16.1.0 2019-06 CT#84 CP-191132 0948 2 A Precedence between access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access 16.1.0 2019-06 CT#84 CP-191132 0950 F Corrections to the length of the SOR transparent container and 16.1.0 2019-06 CT#84 CP-191132 0955 I F Correction on T3396 16.1.0 2019-06 CT#84 CP-191132 0955 I F Carection on Ta396 16.1.0 2019-06 CT#84 CP-191132<	2019-06	CT#84	CP-191137	0937	2	В	PEI for 5G-CRG and FN-CRG	16.1.0
2019-06 CT#84 CP-191132 0943 B Session-TMBR for PDU session in W-5GAN 16.1.0 2019-06 CT#84 CP-191132 0946 I F Handling of SQ NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0946 I F Handling of PDU session type 16.1.0 2019-06 CT#84 CP-191132 0948 2 A Precedence between access identities for derivation of RRC establishment cause for non-3GPP access 16.1.0 2019-06 CT#84 CP-191132 0950 F Clarifications on the length of the SOR transparent container and the 1.0 2019-06 CT#84 CP-191132 0955 I A Correction on the descriptions of 5GSM parameters and the 1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PD SDat Off status update 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PD SDat Off status update 16.1.0 2019-06 CT#84 CP-191123 0957 2 A <t< td=""><td>2019-06</td><td>CT#84</td><td>CP-191137</td><td>0938</td><td></td><td>B</td><td>Usage of ACS information PCO parameter</td><td>16.1.0</td></t<>	2019-06	CT#84	CP-191137	0938		B	Usage of ACS information PCO parameter	16.1.0
2019-06 CT#84 CP-191132 0944 5 F Handling of GC NAS security contexts 16.1.0 2019-06 CT#84 CP-191132 0948 1 F Handling of PDU session type 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0955 I F Carrection on T3396 2019-06 CT#84 CP-191123 0957 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 I A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191148 0963 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0	2019-06	CT#84	CP-191137	0939		В	Session-TMBR for PDU session in W-5GAN	16.1.0
2019-06 CT#84 CP-191132 0946 I F Handling of PDU session type 16.1.0 2019-06 CT#84 CP-191124 0948 2 A Precedence between access identities for derivation of RRC establishment cause and for derivation of establishment cause 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Corrections to the length of the SOR transparent container and the cause and for derivation of the SOR transparent container and the capsabilities 16.1.0 2019-06 CT#84 CP-191132 0957 1 A Correction on T3396 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191128 0957 1 A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191132 0957 2 A Handling of PS Data Off status update 16.1.0	2019-06	CT#84	CP-191132	0944	5	F	Handling of 5G NAS security contexts	16.1.0
2019-06 CT#84 CP-191124 0948 2 A Precedence between access identities for derivation of RRC is rom-3GPP access 16.1.0 2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Corrections to the length of the SOR transparent container and the 1.0. 2019-06 CT#84 CP-191122 0953 1 A Correction on T3396 2019-06 CT#84 CP-191132 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0953 1 A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191142 0953 3 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and the 1.0 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarificat	2019-06	CT#84	CP-191132	0946	1	F	Handling of PDU session type	16.1.0
Control Control <t< td=""><td>2019-06</td><td>CT#84</td><td>CP-191124</td><td>0948</td><td>2</td><td>A</td><td>Precedence between access identities for derivation of RRC</td><td>16.1.0</td></t<>	2019-06	CT#84	CP-191124	0948	2	A	Precedence between access identities for derivation of RRC	16.1.0
2019-06 CT#84 CP-191132 0949 F Clarifications on the validity of access identities 16.1.0 2019-06 CT#84 CP-191132 0950 F Corrections to the length of the SOR transparent container and the 16.1.0 2019-06 CT#84 CP-191122 0953 I A Correction on T3396 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 I A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN Information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0965 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 F Clarification on the checks on QoS rule operations - R16 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>for non-3GPP access</td> <td></td>							for non-3GPP access	
2019-06 CT#84 CP-191132 0950 F Corrections to the length of the SOR transparent container and 16.1.0 2019-06 CT#84 CP-191122 0953 I A Correction on T3396 16.1.0 2019-06 CT#84 CP-191132 0955 I F Correction on the descriptions of 5GSM parameters and 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 I A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0970 I B Small data rate control,	2019-06	CT#84	CP-191132	0949		F	Clarifications on the validity of access identities	16.1.0
UE parameters update transparent container 2019-06 CT#84 CP-191132 0953 1 A Correction on T3396 16.1.0 2019-06 CT#84 CP-191132 0957 1 F Correction on the descriptions of 5GSM parameters and capabilities 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 1 A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191132 0963 3 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0966 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 F Clarification on the creation of a single QOS flow during a PDU ission modification procedure 16.1.0 2019-06 CT#84 CP-191132 0968 F Clarification on the creation of a single QOS flow during a PDU ission modification procedure 16.1.0	2019-06	CT#84	CP-191132	0950		F	Corrections to the length of the SOR transparent container and	16.1.0
2019-06 CT#84 CP-191122 0953 I A Correction on T3396 16.1.0 2019-06 CT#84 CP-191132 0955 I F Correction on the descriptions of 5GSM parameters and capabilities 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 I A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191148 0963 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 I F Clarification on the creation of a single QoS flow during a PDU to 1.0 16.1.0 2019-06 CT#84 CP-191132 0968 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0970 I B <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>UE parameters update transparent container</td> <td></td>							UE parameters update transparent container	
2019-06 CT#84 CP-191132 0955 I F Correction on the descriptions of 5GSM parameters and capabilities 16.1.0 2019-06 CT#84 CP-191123 0957 2 A Handling of PS Data Off status update 16.1.0 2019-06 CT#84 CP-191124 0959 I A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191123 0963 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 I F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0970 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 B Small data rate	2019-06	CT#84	CP-191122	0953	1	А	Correction on T3396	16.1.0
Comparison Comparison <thcomparison< th=""> Comparison Comparis</thcomparison<>	2019-06	CT#84	CP-191132	0955	1	F	Correction on the descriptions of 5GSM parameters and	16.1.0
2019-06 C1#84 CP-191123 0957 2 A Handbing of PS Data On Status update 16.1.0 2019-06 CT#84 CP-191124 0959 1 A SR procedure for emergency services fallback when T3346 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0966 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0966 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0968 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0968 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0971 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activ	2010.00	CT#04	CD 101100	0057	2		capabilities	10 1 0
2019-06 C1#84 CP-191124 0939 1 A Ship proceeding for energy services railoack with 13540 10.1.0 2019-06 CT#84 CP-191148 0963 3 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191129 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0973	2019-00	CT#04	CP-191123	0957	1	A	Ranuling of PS Data Off Status update	16.1.0
2019-06 CT#84 CP-191148 0963 3 B Adding support for SNPNs (Stand-alone Non-Public Networks) 16.1.0 2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarification on the creation of a single QoS flow during a PDU session modification procedure 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84	2019-00	01#04	GF-191124	0939	1	л	timer running	10.1.0
2019-06 CT#84 CP-191132 0964 F Clarification on LADN information for the registered and equivalent PLMNs 16.1.0 2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 B Servic	2019-06	CT#84	CP-191148	0963	3	В	Adding support for SNPNs (Stand-alone Non-Public Networks)	16.1.0
2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the creation of a single QoS flow during a PDU 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0975 3	2019-06	CT#84	CP-191132	0964	1	F	Clarification on LADN information for the registered and	16.1.0
2019-06 CT#84 CP-191132 0965 3 F Clarification on the stop of T3540 16.1.0 2019-06 CT#84 CP-191132 0966 1 F Clarification on the creation of a single QoS flow during a PDU session modification procedure 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Serving PLMN rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129							equivalent PLMNs	
2019-06 CT#84 CP-191132 0966 I F Clarification on the creation of a single QoS flow during a PDU session modification procedure 16.1.0 2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 I B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 I B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 I B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 I B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-	2019-06	CT#84	CP-191132	0965	3	F	Clarification on the stop of T3540	16.1.0
2019-06 CT#84 CP-191132 0968 2 F Correction to the checks on QoS rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CloT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 1 B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3<	2019-06	CT#84	CP-191132	0966	1	F	Clarification on the creation of a single QoS flow during a PDU	16.1.0
2019-06 C1#84 CP-191132 0968 2 P Correction to the checks on Gos rule operations - R16 16.1.0 2019-06 CT#84 CP-191129 0969 1 B Introduction of extended DRX for 5G CIoT 16.1.0 2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Small data rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, activation with IE and indication flag 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 B	0040.00	07404	CD 404400	0000	2	Г	session modification procedure	10.1.0
2019-06 CT#84 CP-191129 0970 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Small data rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0971 1 B Small data rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 1 B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 1 B <td>2019-00</td> <td>CT#04</td> <td>CP-101132</td> <td>0060</td> <td>2</td> <td>r B</td> <td>Introduction of extended DPV for 5C Clot</td> <td>10.1.0</td>	2019-00	CT#04	CP-101132	0060	2	r B	Introduction of extended DPV for 5C Clot	10.1.0
2019-06 CT#84 CP-191129 0971 I B Small data rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0971 I B Small data rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 I B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0979 I <td< td=""><td>2019-00</td><td>CT#84</td><td>CP-191129</td><td>0909</td><td>1</td><td>B</td><td>Small data rate control, general description</td><td>16.1.0</td></td<>	2019-00	CT#84	CP-191129	0909	1	B	Small data rate control, general description	16.1.0
2019-06 CT#84 CP-191129 0972 4 B Serving PLMN rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, general description 16.1.0 2019-06 CT#84 CP-191129 0974 I B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 I B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, activation with IE and indication flag 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129	2019-06	CT#84	CP-191129	0971	1	B	Small data rate control, activation	16.1.0
2019-06 CT#84 CP-191129 0973 4 B Serving PLMN rate control, activation 16.1.0 2019-06 CT#84 CP-191129 0974 1 B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, activation with IE and indication 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0980 1 B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-191129	2019-06	CT#84	CP-191129	0972	4	B	Serving PLMN rate control, general description	16.1.0
2019-06 CT#84 CP-191129 0974 1 B Service Gap control in 5GS, general description 16.1.0 2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, activation with IE and indication 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0979 1 B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-191129 0980 1 B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-19	2019-06	CT#84	CP-191129	0973	4	В	Serving PLMN rate control, activation	16.1.0
2019-06 CT#84 CP-191129 0975 3 B Service Gap control in 5GS, activation with IE and indication flag 16.1.0 2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0980 1 B General Description for Restriction on use of enhanced coverage 16.1.0 2019-06 CT#84 CP-191129 0981 1 B Updates to Registration procedure for Restriction on use of 16.1.0	2019-06	CT#84	CP-191129	0974	1	В	Service Gap control in 5GS, general description	16.1.0
2019-06 CT#84 CP-191129 0977 3 B Service Gap control in 5GS, enforcement in UE 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in 5GS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 I B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0979 I B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0980 I B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-191129 0980 I B Updates to Registration procedure for Restriction on use of 16.1.0 2019-06 CT#84 CP-191129 0981 I B Updates to Registration procedure for Restriction on use of 16.1.0	2019-06	CT#84	CP-191129	0975	3	В	Service Gap control in 5GS, activation with IE and indication	16.1.0
2019-06 CT#84 CP-191129 0971 3 B Service Gap control in SGS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0978 3 B Service Gap control in SGS, enforcement in AMF 16.1.0 2019-06 CT#84 CP-191129 0979 1 B Service Gap control in 5GS, new time value via UCU 16.1.0 2019-06 CT#84 CP-191129 0980 1 B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-191129 0980 1 B General Description for Restriction on use of enhanced 16.1.0 2019-06 CT#84 CP-191129 0981 1 B Updates to Registration procedure for Restriction on use of 16.1.0	2010.06	CT#01	CP 101120	0077	2	D	IIdy Sorvice Cap control in 5CS, optorcoment in LIE	1610
2019-06CT#84CP-19112909791BService Gap control in SGS, endocement in AMP16.1.02019-06CT#84CP-19112909801BGeneral Description for Restriction on use of enhanced16.1.02019-06CT#84CP-19112909811BGeneral Description for Restriction on use of enhanced16.1.02019-06CT#84CP-19112909811BUpdates to Registration procedure for Restriction on use of16.1.0	2019-00	CT#94	CP-101129	09//	3	B	Service Gap control in 5GS, enforcement in AME	16.1.0
2019-06CT#84CP-19112909801BGeneral Description for Restriction on use of enhanced16.1.02019-06CT#84CP-19112909811BUpdates to Registration procedure for Restriction on use of16.1.0	2019-00	CT#84	CP-191129	0970	1	B	Service Gap control in 5GS, new time value via LICU	16.1.0
2019-06CT#84CP-19112909801BGeneral Description for Restriction on use of enhanced coverage16.1.02019-06CT#84CP-19112909811BUpdates to Registration procedure for Restriction on use of to heave descent procedure for Restriction on use of16.1.0	0	01//04	0. 101120		·	_	procedure	
2019-06 CT#84 CP-191129 0981 1 B Updates to Registration procedure for Restriction on use of 16.1.0	2019-06	CT#84	CP-191129	0980	1	В	General Description for Restriction on use of enhanced	16.1.0
	2019-06	CT#84	CP-191129	0981	1	В	Updates to Registration procedure for Restriction on use of	16.1.0

2019-06	CT#84	CP-191129	0982	2	В	Support for Reliable Data Service in 5GS	16.1.0
2019-06	CT#84	CP-191132	0983	1	F	Handling 5GMM cause #5 "PEI not accepted"	16.1.0
2019-06	CT#84	CP-191132	0984		F	Correcting UE state when disabling and re-enabling N1 mode	16.1.0
2019-06	CT#84	CP-191129	0985	2	F	Terminology definition for 5G_CIoT	16.1.0
2019-06	CT#84	CP-191129	0986	3	В	General introduction on CIoT 5GS optimizations	16.1.0
2019-06	CT#84	CP-191129	0987	5	В	CloT capability negotiation between UE and network	16.1.0
2019-06	CT#84	CP-191129	0988	3	В	CloT optimisations redirection betwee EPC and 5GC	16.1.0
2019-06	CT#84	CP-191129	0989	2	В	Inter-RAT mobility to and from NB-IoT in 5GS	16.1.0
2019-06	CT#84	CP-191129	0990	3	В	Congestion control for CP data transport in 5GS	16.1.0
2010-06	CT#84	CP-101122	0000	2	Δ	Correction on the 5GSM S-NSSAL congestion control	16.1.0
2010-00	CT#84	CP-101132	0002	2	F	Network slicing indication	16.1.0
2013-00	CT#04	CP 101122	0005	1	Λ Λ	Correction on the SMS over NAS	16.1.0
2019-00	CT#64	CP-191122	0995	1	A	Correction on the SMS over NAS	10.1.0
2019-06	CT#84	CP-191148	0997	2	В	Providing CAG ID to the lower layer	16.1.0
2019-06	CT#84	CP-191132	0998	5	в	UE behaviour upon receiving non-integrity protected NAS	16.1.0
2019-06	CT#84	CP-191217	1000	5	А	Multiple NAS connections and 5G NAS security context change handling	16.1.0
2019-06	CT#84	CP-191123	1002	1	А	EPS bearer synchronization when moving from EPC to 5GC	16.1.0
2019-06	CT#84	CP-191132	1018	6	F	UE-requested PDU session modification procedure and	16.1.0
						exemption indication	
2019-06	CT#84	CP-191129	1019	1	В	Active Time for MICO mode	16.1.0
2019-06	CT#84	CP-191138	1020	5	В	MA PDU request in UL NAS TRANSPORT message	16.1.0
2019-06	CT#84	CP-191138	1021	4	В	Update PDU session establishment procedure to support MA PDU session	16.1.0
2019-06	CT#84	CP-191132	1022		F	Correction to Mico mode activation in the UE	16.1.0
2019-06	CT#84	CP-191132	1023	4	F	Correction to rejected NSSAI deletion.	16.1.0
2019-06	CT#84	CP-101122	1025	·	A	Correction of timer table	16 1 0
2010-00	CT#84	CP-101122	1020	2	F	Coordination between GMM and 5GMM	16.1.0
2013-00	CT#04	CP 101122	1027	2	Г Б	Correction of requirements for the handling of access harring	16.1.0
2019-00	CT#04	CF-191132	1029	2	Г	Confection of requirements for the handling of access barring	10.1.0
2019-06	CT#84	CP-191129	1030	2	В	Strictly periodic registration timer indication for MICO mode	16.1.0
2019-06	CT#84	CP-191124	1037	2	A	Network initiated deregistration update for cause #3 and #6	16.1.0
2019-06	CT#84	CP-191132	1038		F	Correction of typos in octet numbering for AMF Set ID	16.1.0
2019-06	CT#84	CP-191132	1039		F	Stopping T3519 and deleting SUCI when receiving	16.1.0
						authentication reject	
2019-06	CT#84	CP-191122	1041	1	А	Correct UE parameters update data type	16.1.0
2019-06	CT#84	CP-191132	1043	2	F	Clarifications to the Routing indicator encoding	16.1.0
2019-06	CT#84	CP-191132	1045	1	F	Correction on Payload container type information element	16.1.0
2019-06	CT#84	CP-191129	1046	2	В	Introduction to UE selection of CN for 5G CIoT (for 24.501)	16.1.0
2019-06	CT#84	CP-191133	1047		F	Exception in suspension of 5GSM procedures in case of ongoing 5GMM procedures: Service request procedure initiated during connected mode	16.1.0
2019-06	CT#84	CP-191124	1050	1	А	PDU SESSION ESTABLISHMENT message sent via target access in case of handover of an existing PDU session between 3GPP access and non-3GPP access	16.1.0
2019-06	CT#84	CP-191133	1052	2	F	Retreival of an SMF ID during transfer of a PDN connection in EPS to 5GS	16.1.0
2019-06	CT#84	CP-191146	1053	3	F	Clarification of 5GSM cause values for which network may include a back-off timer or a re-attempt indicator	16.1.0
2019-06	CT#84	CP-191148	1054	4	В	Introduction of non-public network	16.1.0
2019-06	CT#84	CP-191148	1055	1	В	PLMN ID and NID provided to the lower lavers	16.1.0
2010-06	CT#84	CP-101122	1058	-	F	Extensible Authentication Protocol specified in IETE REC 2749	16 1 0
2010-00	CT#94	CP_101122	1050		• F	Corrections on security context terminologies	16.1.0
2010-00	CT#04	CP_101122	1000	2	F	Correction on LIE handling for notwork initiated do registration	16.1.0
2019-00	07#04	CF-191133	1000	2	Г Г	with #22	10.1.0
2019-06	C1#84	CP-191133	1061	1	F	13540 handling for re-registration triggered by UCU	16.1.0
2019-06	CT#84	CP-191133	1062	2	F	Correction on coding of "all other values are spare"	16.1.0
2019-06	CT#84	CP-191124	1064	2	А	Indication of resume failure from the lower layer	16.1.0
2019-06	CT#84	CP-191133	1066	1	F	Integrity protection failure	16.1.0
2019-06	CT#84	CP-191133	1067	1	F	Clarification related to dual registration mode	16.1.0
2019-06	CT#84	CP-191133	1070		F	Including the other cases in [S-NSSAI, DNN] combination back-off timer	16.1.0
2019-06	CT#84	CP-191123	1071	4	А	Correction to the Payload container IE	16.1.0
2019-06	CT#84	CP-191122	1073	3	А	Correction because of wrong implementation of CR0763 and CR0919	16.1.0
2019-06	CT#84	CP-191124	1075	2	A	Resolution of editor's notes on handling at emergency registration and emergency PDU sessions	16.1.0
2019-06	CT#84	CP-191124	1077	2	A	Resolution of editor's notes on handling at non-existing 5G NAS security context indicated by the UE when an emergency PDU session exists	16.1.0
2019-06	CT#84	CP-191137	1078	3	В	Introduction of general aspects of wireline access	16.1.0
2019-06	CT#84	CP-191137	1082	1	В	Introduction of references, definitions and abbreviations for	16.1.0
2010.00	CT#04	CD 101122	1000	1	F	5WWC	16.1.0
M-LID	VI#04	100-191133					

2019-06	CT#84	CP-191133	1090	1	F	Correction on PTI error handling of the network	16 1 0
2010 00	CT#04	CD 101122	1000	1	F		16.1.0
2019-00	01#04	CF-191133	1092	1	Г		10.1.0
2019-06	CT#84	CP-191122	1094	1	A	Correction on handling of mapped EPS bearer contexts IE	16.1.0
2019-06	CT#84	CP-191133	1096	2	F	Correction on associating default EPS bearer with the default	16.1.0
						QoS rule	
2019-06	CT#84	CP-191133	1098	1	F	Feature support after inter-system change from 5GS to EPS	16.1.0
2019-06	CT#84	CP-191133	1100	3	F	Correction on 5GSM IE handling when an error is detected	16 1 0
2010-00	CT#04	CD 101122	1100	1	T E	Security Mode Command precedure when S1 mode supported	16.1.0
2019-06	01#64	CP-191133	1105	1	F	Security Mode Command procedure when ST mode supported	10.1.0
2019-06	CT#84	CP-191133	1107	1	F	Correction of term "user preference" to "UE local configuration"	16.1.0
						for inter-system change without N26	
2019-06	CT#84	CP-191122	1109	1	А	Addition of missing codepoints for 5GSM causes	16.1.0
2019-06	CT#84	CP-101133	1114	2	F	Modification of the maximum number of supported packet	16 1 0
2010 00	01#04	01-101100	1114	2	1	filtore	10.1.0
	07/0				-		
2019-06	C1#84	CP-191133	1115	1	F	5GS-EPS interworking for Multi-homed IPv6 PDU Session not	16.1.0
						supported without N26	
2019-06	CT#84	CP-191133	1118	3	F	UE-requested PDU session modification for emergency PDU	16.1.0
						session	
2010-06	CT#84	CP-101133	1110		F	Clarification of emergency support indications	16 1 0
2010-00	CT#04	CD 101100	110	2	1	Assess sentral and indication that assess harring is applicable.	10.1.0
2019-06	CT#64	CP-191122	1121	3	А	Access control and indication that access barring is applicable	16.1.0
						for all access categories except categories '0' and '2'	
2019-06	CT#84	CP-191138	1122	2	В	Multiple Access PDU Session	16.1.0
2019-06	CT#84	CP-191122	1124	1	А	5GSM cause value #29 semantic extension	16.1.0
2019-06	CT#84	CP-191124	1128	4	А	NSSAI inclusion mode in ePI MN	16 1 0
2010 00	CT#04	CP 101124	1120	1	E	Interaction between active time for MICO mode and cDBV	16.1.0
2019-00	01#04	05-191129	1129	1	Г Ъ	Interaction between active time for which mode and eDRX	10.1.0
2019-06	CT#84	CP-191224	1130	4	В	User plane CI01 5GS optimization	16.1.0
2019-06	CT#84	CP-191129	1131	3	В	CP only indication	16.1.0
2019-06	CT#84	CP-191123	1133	1	А	Handling of PDU session modification while a back-off timer is	1610
2010 00	01#01	01 101120	1100	-			10.1.0
0040.00	07/04	00.404440	4404	1	D	Addie a suggest (an unified as a sector bis ONDNA (Or and	40.4.0
2019-06	C1#84	CP-191148	1134	1	в	Adding support for unified access control in SINPINS (Stand-	16.1.0
						alone Non-Public Networks)	
2019-06	CT#84	CP-191134	1135	1	D	The phrase "outside the scope of the present document" is not	16.1.0
						used consistently	
2019-06	CT#84	CP-101134	1137	3	F	Clarification regarding replayed LIE security capabilities	16 1 0
2010-00	01#04	00 101104	1107	1	1		10.1.0
2019-06	C1#84	CP-191124	1140	1	A	UE policy length mismatch	16.1.0
2019-06	CT#84	CP-191134	1142	2	F	Conditions for congestion control in AMF	16.1.0
2019-06	CT#84	CP-191134	1146	3	F	Correction of inconsistent requirements for the use of SUCI	16.1.0
2019-06	CT#84	CP-191134	1147	1	F	Disabling of N1 mode capability after emergency services	16.1.0
				-	-	fallback	
2010.06	CT#04	CD 101124	1140	1	Б	Clarification on disabling N1 mode conshility when there is	16 1 0
2019-06	CT#64	CP-191134	1149	1	Г	Clarification on disabiling NT mode capability when there is	16.1.0
						persistent PDU session	
2019-06	CT#84	CP-191123	1151	1	Α	Indicating PS data off status report for the UE in the Non-	16.1.0
						allowed Area	
2019-06	CT#84	CP-191134	1156	1	F	Clarification of "registration requested" with no other	16.1.0
						parameters	
2010.06	CT#94	CD 101124	1157		F	Undates to Notwork slicing indication	16 1 0
2019-00	CT#04	CD 404404	1107		г Г	Clarification on the length of the loss stream for initial NAC	10.1.0
2019-06	C1#84	CP-191134	1160		F	Clarification on the length of the key stream for initial NAS	16.1.0
						message protection	
2019-06	CT#84	CP-191207	1162	4	F	Applicability of the allowed NSSAI in an equivalent PLMN	16.1.0
						outside the UE's registration area	
2019-06	CT#84	CP-191123	1164	3	А	Handling of the ABBA parameter with a non-zero value and a	16.1.0
				-	-	length of more than 2 octets	
2010.00	CT#04	CD 101104	1165	1	Б	Clarifications to convice area restrictions	16 1 0
2019-00	01#04	00-191134	1105	1	1,		10.1.0
2019-06	CT#84	CP-191123	1168	1	А	IEI TOT THE NON-3GPP NW provided policies IE	16.1.0
<u>2019</u> -06	<u>CT#84</u>	CP-191136	1169		F	Reference to IEEE 802.3	16.1.0
2019-06	CT#84	CP-191123	1171	2	А	Correction to serving network name (SNN) reference	16.1.0
2019-06	CT#84	CP-191123	1173	1	А	IEI for the UE OS Id IE	16.1.0
2010 00	CT#04	CD 101123	1174	2	E	Correction for DDLL appaion modification with OFL short re-	16.1.0
2019-06		CF-191134	11/4	2	Г	Correction for PDU session modification with QFI change.	10.1.0
2019-06	CT#84	CP-191134	1175	2	F	Correction to PDU session release reject handling	16.1.0
2019-06	CT#84	CP-191134	1176	1	F	Correction to PDU session authentication result transport	16.1.0
						procedure.	
2019-06	CT#84	CP-191134	1177	1	F	Clarification for transfer of PDU session for LADN to FPS	16.1.0
2010.06	CT#04	CD 101124	1170	1	- E	Correction to Do registration and registration presedure	16 1 0
2019-00	01#04	66-191134	11/8	1	г		10.1.0
				ļ	L	collision	
2019-06	CT#84	CP-191134	1179		F	Clarification for 5GMM cause #3 and #6 in the SERVICE	16.1.0
						REJECT message	
2019-06	CT#84	CP-191134	1180	2	F	Added detailed description for substates INITIAL -	16.1.0
2010-00	01//04	0. 101104	1.100	-	-	REGISTRATION-NEEDED and LIDDATE-NEEDED	10.1.0
2010.00	07404		1400		D		10.1.0
2019-06		CF-191134	1182		ע		10.1.0
2019-06	CT#84	CP-191122	1185	1	А	Always-on PDU session	16.1.0
2019-06	CT#84	CP-191134	1186	1	F	S-NSSAI association for PDU session established in S1 mode	16.1.0
2019-06	CT#84	CP-191130	1187	1	В	5GMM capability for SRVCC from NG-RAN to LITRAN	16.1.0
2010.06	CT#94	CP_101120	1100	1	B	MS classmark 2 and supported codes	16 1 0
2019-00	01#04	00-191130	1100	1	D T		10.1.0
2019-06	UT#84	CP-191134	1191	2	Г	Handling of 5GSIVI parameters	16.1.0

2019-06	CT#84	CP-191134	1193	2	F	Handling of SM back-off timer	16.1.0
2019-06	CT#84	CP-191123	1195	1	А	DNN based congestion control for PDU session for LADN	16.1.0
2019-06	CT#84	CP-191135	1197	-	F	Terminology correction about PDU session type	16.1.0
2019-06	CT#84	CP-191135	1199	1	F	Definition of EMM-IDLE mode in TS 24 501	16.1.0
2010-06	CT#84	CP-101120	1200	1	F	No CloT 5GS ontimizations for non-3GPP access	16.1.0
2010-00	CT#84	CP-101124	1200	2	Δ	Network initiated EPS bearer synchronization when moving	16.1.0
2013-00	01#04	01-131124	1205	2	л	from EPC to 5GC	10.1.0
2010-06	CT#84	CP-101135	1204	1	F	Correction on follow-on request indicator	1610
2013-00	CT#04	CD 101125	1204	1	г Б	Correction on LE behaviour in ECMM RECISTERED DI MN	16.1.0
2019-06	CT#64	CP-191135	1205		Г		16.1.0
0040.00	OT#04	CD 404400	4000	1			40.4.0
2019-06	C1#84	CP-191123	1206	1	А	Deletion of the 5GSM cause IE in the PDU SESSION	16.1.0
0040.00	OT // 0.4	00 404405	1007		-		10.1.0
2019-06	CT#84	CP-191135	1207	1	F	Correction on the abnormal cases for registration procedure	16.1.0
2019-06	CT#84	CP-191129	1210		F	Alignment of the Abnormal cases for eDRX between 5GS and	16.1.0
						EPS	
2019-06	CT#84	CP-191135	1211	1	F	Collision handling	16.1.0
2019-06	CT#84	CP-191135	1212	2	F	Authenticate before deleting UE context	16.1.0
2019-06	CT#84	CP-191129	1213	3	В	Uplink NAS message transmission and CIoT data transfer	16.1.0
2019-06	CT#84	CP-191135	1214	1	F	Correction on T3346 and EPLMN handling	16.1.0
2019-06	CT#84	CP-191122	1219	2	А	Alignment of the 5G ciphering and integrity algorithm identifiers	16.1.0
2019-06	CT#84	CP-191135	1220	2	F	Conditions to apply the "null-scheme" to generate the SUCI	16.1.0
2019-06	CT#84	CP-191135	1221		F	Correction on the description of code point for 128-5G-EA3	16.1.0
2019-06	CT#84	CP-191129	1222	1	В	Service Gap control in 5GS. reject of UL NAS Transport	16.1.0
						message	
2019-06	CT#84	CP-191135	1227	2	F	Transmission failure of Registration Request during Initial Reg	16.1.0
2010 00	01//01		,	-	•	proc	10.110
2019-06	CT#84	CP-191135	1228	3	F	Handling of Radio Link failure during service Reg proc	16 1 0
2010-00	CT#84	CP-191148	1220	1	R	Abnormal case bandling for receipt of 5GMM cause value #11	16.1.0
2013-00	01#04	01-131140	1223	1	Б	from a PI MN	10.1.0
2010.06	CT#9/	CD 1011/9	1221	2	D	Packat filters based on N2IWE IP address and SPI for IPsec	16 1 0
2019-00	01#04	CF-191140	1231	2	Б		10.1.0
2010.00	CT#04	CD 101104	1000	1	٨	On flow for CID signalling ofter inter system shange	16.1.0
2019-00	CT#64	CP-191124	1230	1	A	Use dline of multiple OoC mule flow persentations included in ano	16.1.0
2019-06	C1#84	CP-191135	1239	1	F	Handling of multiple QoS rule/flow parameters included in one	16.1.0
0040.00	07/04	00 404405	10.10		-		10.1.0
2019-06	C1#84	CP-191135	1240	1	F	Error handling of optional IEs in a payload container entry of	16.1.0
						the Payload container IE	
0040.00	07/04	00 404405	10.14		-	A 1 1 1 (500M	40.4.0
2019-06	CT#84	CP-191135	1241		F	Add codings of 5GSM causes #41 and #42	16.1.0
2019-06 2019-06	CT#84 CT#84	CP-191135 CP-191124	1241 1245	1	F A	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy	16.1.0 16.1.0
2019-06 2019-06	CT#84 CT#84	CP-191135 CP-191124	1241 1245	1	F A	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE	16.1.0 16.1.0
2019-06 2019-06 2019-06	CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135	1241 1245 1246	1 3	F A F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network	16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06	CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135	1241 1245 1246	1 3	F A F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing	16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135	1241 1245 1246 1247	1 3	F A F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service	16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135	1241 1245 1246 1247	1 3	F A F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback	16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135	1241 1245 1246 1247 1248	3	F A F F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146	1241 1245 1246 1247 1248 1249	1 3 1	F A F F C	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123	1241 1245 1246 1247 1248 1249 1251	1 3 1 1 1	F A F F C A	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135	1241 1245 1246 1247 1248 1249 1251 1254	1 3 1 1 2	F A F F C A F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135	1241 1245 1246 1247 1247 1248 1249 1251 1254	1 3 1 1 2	F A F F C A F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135	1241 1245 1246 1247 1248 1249 1251 1254 1257	1 3 1 1 2	F A F F C A F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135 CP-191135	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264	1 3 1 1 2 1	F A F F C A F F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135 CP-191135	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264	1 3 1 1 2 1	F A F F C A F F F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073	1241 1245 1246 1247 1247 1247 1251 1251 1254 1257 1264 1056	1 3 1 1 2 1 5	F A F F C A F F F B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073	1241 1245 1246 1247 1247 1247 1251 1251 1254 1257 1264 1056	1 3 1 1 2 1 5	F A F F C A F F F B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057	1 3 1 1 2 1 5 8	F A F F C A F F F B B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189	1 3 1 1 2 1 5 8 3	F A F F C A F F F F B B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192073	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236	1 3 1 1 2 1 5 8 3 3	F A F F C A F F F B B F B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192073 CP-192073 CP-192073 CP-192073	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269	1 3 1 1 2 1 5 8 3 3 2	F A F F C A F F B B B B B B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192073 CP-192056 CP-192073 CP-192056	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270	1 3 1 1 2 1 5 8 3 3 2 1	F A F F C A F F B B B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192073 CP-192056	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270	1 3 1 1 2 1 5 8 3 3 2 1	F A F C A F F B B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192073 CP-192060 CP-192060	1241 1245 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270	1 3 1 1 2 1 5 8 3 3 2 1 1 2	F A F F C A F F B B F B B F B B B B B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAL pot allowed for MA PDU session establishment	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192060 CP-192060 CP-192060 CP-192060	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1264 1056 1269 1270 1271	1 3 1 1 2 1 5 8 3 3 2 1 2	F A F F C A F F B B F B F B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Lindetes to new stage 2 requirements of CAC information	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192060 CP-192060 CP-192073	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1264 1056 1269 1270 1271 1273	1 3 1 1 2 1 5 8 3 3 2 1 2 1 2	F A F F C A F B B F B F B F F B F B F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192060 CP-192060 CP-192060 CP-192073 CP-192060	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270 1271 1273	1 3 1 1 2 1 5 8 3 2 1 2 2 2	F A F F C A F F B B F B F B F B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192060 CP-192060 CP-192060 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270 1271 1273 1274	1 3 1 1 2 1 5 8 3 2 1 2 1 2 2 2 2 2 2 2 2 2	F A F F C A F B B B F B F F F F F F F F F F F F F F F F F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192056 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270 1271 1273 1274 1274	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2	F A F F C A F B B B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access S1	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192056 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1270 1271 1273 1274 1275	1 3 1 1 2 1 5 8 3 2 1 2	F A F F C A F B B F B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192056 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1257 1264 1056 1270 1270 1271 1273 1274 1276 1277 1278	1 3 1 1 2 1 5 8 3 2 1 2 2 1 2 3 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 <td< td=""><td>F A F F C A F B B B F B F B F B F B F B F B F B F B F B F B F B F F B F B F F F F B F B F F B F B F B F B F</td><td>Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt</td><td>16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0</td></td<>	F A F F C A F B B B F B F B F B F B F B F B F B F B F B F B F B F F B F B F F F F B F B F F B F B F B F B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192056 CP-192056 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1269 1270 1270 1271 1273 1274 1276 1277 1278 1279	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 2 2 1	F A F F C A F B B B F B F B C	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1057 1189 1236 1269 1270 1271 1273 1274 1276 1277 1278 1279 1280	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 1 2 2 2 1 2	F A F F C A F B B B F F B F B C B C B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG V2X capability and V2X PC5 capability	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191146 CP-191123 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192050 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1257 1264 1056 1269 1270 1271 1273 1274 1276 1277 1278 1279 1280 1281	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 2 2 1 2 2 2 1 2 1	F A F F C A F F B B F F B F F B C B C B C B B B C B B	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG V2X capability and V2X PC5 capability USPS extension for V2X policy	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192050 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192059 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053 CP-192053	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1269 1270 1271 1273 1274 1276 1277 1278 1277 1278 1279 1280 1281 1281 1282	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 2 1 2 2 1 2 1	F A F F C A F B B F F B F F B F B C B C B F F F F F F F F F F F F B B F	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG V2X capability and V2X PC5 capability USPS extension for V2X policy Incorrect security algorithm	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0
2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-06 2019-09	CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#84 CT#85	CP-191135 CP-191124 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-191135 CP-192073 CP-192073 CP-192056 CP-192050 CP-192059	1241 1245 1246 1247 1248 1249 1251 1254 1257 1264 1056 1257 1264 1056 1269 1270 1271 1273 1274 1276 1277 1278 1277 1278 1277 1278 1277 1280 1281 1282 1283	1 3 1 1 2 1 5 8 3 2 1 2 2 2 2 2 2 2 2 1 2 1 2 1 2 1 2 1	F A F F C A F B B F B F F B F F B F F F F F F F F B C B F <td< td=""><td>Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG V2X capability and V2X PC5 capability USPS extension for V2X policy Incorrect security algorithm Registration attempt counter correction</td><td>16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0</td></td<>	Add codings of 5GSM causes #41 and #42 Indication of syntactical or semantic errors related to SM policy association to UE Clean-up of general section for 5GMM aspects of network slicing Request from the upper layers to perform emergency service fallback Non-overlapping tracking areas 5GSM cause values #27, #50, #51 and #70 for SINE_5G Correction to handling of cause #72 Originating MMTEL voice due to upper layers request while T3346 is running Shared or valid 5G NAS security context EAP-success of EAP-TLS received in SECURITY MODE COMMAND Provisioning of an allowed CAG list and a CAG access only indication 5GMM cause value for CAG PDU session modication command not forwarded to 5G AN Storage for CAG information Procedure for Multiple Access upgrade of PDU Session Handling the non-current 5G NAS security context after inter- system change from N1 mode to S1 mode NSSAI not allowed for MA PDU session establishment Updates to new stage-2 requirements of CAG information structure Wireline access is a type of non-3GPP access Management of service area restrictions in wireline access IP address allocation for 5G-RG Security for W-AGF acting on behalf of an FN-RGt Alignment with stage-2 on PEI for 5G-RG and FN-RG V2X capability and V2X PC5 capability USPS extension for V2X policy Incorrect security algorithm Registration attempt counter correction	16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.1.0 16.2.0

2019-09	CT#85	CP-192052	1287	1	F	Clarification for UE selecting a suitable cell that supports CloT	16.2.0
2019-09	CT#85	CP-192056	1288	1	F	Core network type restriction determined by operator policy	16.2.0
2019-09	CT#85	CP-192056	1290	1	F	Disabling the N1 mode capability for 3GPP access in Idle	16.2.0
2019-09	CT#85	CP-192060	1293	1	С	PDU Session release for MA PDU Session	16.2.0
2019-09	CT#85	CP-192056	1295	1	F	Clarification on error check for QoS rules	16.2.0
2019-09	CT#85	CP-192060	1297	1	В	MA PDU session establishment reject due to unstructured PDU Session type	16.2.0
2019-09	CT#85	CP-192056	1298		F	Clarification for T3580 Stop condition	16.2.0
2019-09	CT#85	CP-192060	1300	1	В	MA PDU session modification for ATSSS parameters	16.2.0
2019-09	CT#85	CP-192056	1302	3	F	Staying in inactive upon resume failure with RRC staying in RRC INACTIVE	16.2.0
2019-09	CT#85	CP-192045	1306	1	А	Maintaining the UL and DL NAS COUNTs after a handover from 5GS to EPS	16.2.0
2019-09	CT#85	CP-192052	1308	2	BV	PDU sessions and QoS flows for NB-IoT UEs	16.2.0
2019-09	CT#85	CP-192052	1311	3	В	Idle mode optimizations for 5G Control plane CloT small data transfer t	16.2.0
2019-09	CT#85	CP-192056	1315	1	F	Performing registration update upon resume failure for reasons other than barring	16.2.0
2019-09	CT#85	CP-192062	1316	2	В	5GMM capability update for eNS	16.2.0
2019-09	CT#85	CP-192052	1318	1	В	Header compression for control plane user datat	16.2.0
2019-09	CT#85	CP-192052	1322	1	В	UE behavior when RRC connection resume failst	16.2.0
2019-09	CT#85	CP-192052	1323	1	F	Correction on service gap timer	16.2.0
2019-09	CT#85	CP-192052	1324	1	F	Correction on T3448 value IE	16.2.0
2019-09	CT#85	CP-192052	1325	1	F	Alleviation of SM concestion	16.2.0
2019-09	CT#85	CP-192056	1326		F	Correction wrt EPS attempt counter to be used for Single Registration Failure use cases	16.2.0
2019-09	CT#85	CP-192056	1328		F	Correction on terminology regarding EPS bearer contextst	16.2.0
2019-09	CT#85	CP-192056	1330	1	F	Correction to SM procedures for back off timer not forwarded	16.2.0
2019-09	CT#85	CP-192056	1332		F	from 5GMM case. Deletion of RAND and RES on receiving Service Accept	16.2.0
2010.00	CT#95	CP 102056	1222		F	message	16.2.0
2019-09	CT#05	CF-192030	1000	1	Г Е	running	16.2.0
2019-09	CT#05	CP-192032	1004	1	Г D	Addition of LCS indication in ECMM consulting and ECS NW	16.2.0
2019-09	CT#05	CF-192003	1330	1	D	capability	10.2.0
2019-09	CT#85	CP-192083	1336	1	В	information IE	16.2.0
2019-09	CT#85	CP-192056	1337		F	Minor miscellaneous corrections	16.2.0
2019-09	CT#85	CP-192056	1340		F	Clarify encoding of EPS bearer identity	16.2.0
2019-09	CT#85	CP-192071	1341		F	Keep equivalent PLMNs list for Deregistration Request message with 5GMM cause #7	16.2.0
2019-09	CT#85	CP-192056	1344		F	Correction to handling of operator-defined access category missing a standardized access category	16.2.0
2019-09	CT#85	CP-192056	1345	1	F	ODAC IEI correction	16.2.0
2019-09	CT#85	CP-192052	1346		F	Removal of eDRX support with RRC_INACTIVE for NB-IoT	16.2.0
2019-09	CT#85	CP-192057	1348	1	F	IMEI not required for non-3GPP only UEs	16.2.0
2019-09	CT#85	CP-192057	1351	1	F	Emergency services fallback from non-3GPP access or ePDG/EPC	16.2.0
2019-09	CT#85	CP-192057	1353	1	F	Re-ordering of text on the applicability of access identities	16.2.0
2019-09	CT#85	CP-192057	1354	1	F	Trigger for NAS procedure retry in case NAS is put back in RRC inactivet	16.2.0
2019-09	CT#85	CP-192069	1355		В	Adding general description of RACS	16.2.0
2019-09	CT#85	CP-192069	1356	1	В	Signalling of UE support for RACS and of UE radio capability ID	16.2.0
2019-09	CT#85	CP-192069	1357	2	В	UE radio capability ID assignment by the network	16.2.0
2019-09	CT#85	CP-192073	1358	1	В	Signalling of UE support for transfer of port management information containers, MAC address and DS-TT residence	16.2.0
2019-09	CT#85	CP-192073	1359	2	В	Adding support for transfer of Ethernet port management	16.2.0
2019-09	CT#85	CP-192073	1360	1	F	Removal of Editor's note on adding unified access control configuration to "list of subscriber data" for access to SNPNs	16.2.0
2019-09	CT#85	CP-192073	1361	2	F	Resolution of Editor's notes on abnormal case handling for	16.2.0
2019-09	CT#85	CP-192073	1363	1	F	Resolution of Editor's notes on the applicability of MPS, MCS and delay tolerant in SNPNs	16.2.0
2019-09	CT#85	CP-192073	1364	1	F	Update of requirements on UE to construct packet filters based on the N3IWF destination IP address and the SPI for the IPsec SA	16.2.0

2019-09	CT#85	CP-192083	1365	2	В	NAS transport of supplementary services messages for a	16.2.0
2019-09	CT#85	CP-192052	1366		F	Resolve Editor's note on support indication for Small Data	16.2.0
						Rate Control	
2019-09	CT#85	CP-192052	1367	1	В	Small data rate control parameters received in EPS	16.2.0
2019-09	CT#85	CP-192052	1369	1	F	Including EPS Preferred Network Behaviour to 5GCN	16.2.0
2019-09	CT#85	CP-192052	1370	1	F	General description on redirection of the UE by the core network	16.2.0
2019-09	CT#85	CP-192052	1371		F	No RRC inactive for NB-IoT	16.2.0
2019-09	CT#85	CP-192052	1372		F	Removal of Editor's Note for the T3348	16.2.0
2019-09	CT#85	CP-192057	1380		F	T3540 for 5GMM cause #31 or #73	16.2.0
2019-09	CT#85	CP-192052	1384	1	F	Service gap control timer and MICO mode	16.2.0
2019-09	CT#85	CP-192052	1385	2	F	Service gap control, stop of timer via configuration update command	16.2.0
2019-09	CT#85	CP-192052	1386		F	Service gap control, MO service request when connected and timer running	16.2.0
2019-09	CT#85	CP-192052	1387	2	С	Service gap control, follow-on request indicator at mobility update registration	16.2.0
2019-09	CT#85	CP-192052	1388	2	2	Service gap control, follow-on request indicator at initial registration	16.2.0
2019-09	CT#85	CP-192052	1391		F	MICO mode and stop of active timer in AMF when UE enters	16.2.0
2010-00	CT#85	CP-102062	1305	3	C	Registration reject due to no available allowed S-NSSA1(a)	16.2.0
2019-09	CT#85	CP-192002	1306	1	F	Service gap control MO SMS or LPP payload not allowed	16.2.0
2013-03	01#00	01 132032	1030	1	-	when connected	10.2.0
2019-09	CT#85	CP-192057	1398	1	F	Consistent use of PLMN ID for AKA	1620
2010-00	CT#85	CP-192057	1300	1	F	Service reject without integrity protection	16.2.0
2010-00	CT#85	CP-192052	1405		F	Strictly periodic handling due to emergency service	16.2.0
2010-00	CT#85	CP-102073	1406	2	F	Emergency call handling for a CAC only LIE	16.2.0
2019-09	CT#85	CP-102073	1400	2	Г F	Collision of deregistration and other NAS procedure	16.2.0
2019-09	CT#05	CP 102057	1400		Г D	Missing inactive form	16.2.0
2019-09	CT#00	CP-192037	1410	1	D E	Handling of an omorganou call in SNIPN accoss mode	16.2.0
2019-09	CT#05	CP 102057	1411	1	Г	Initial NAS manage protection	16.2.0
2019-09	CT#05	CP 102057	1412	1	Г	Stopping conditions for Timer T2565	16.2.0
2019-09	CT#05	CP 102062	1413	1	Г DV	Undate of existing subclause for network slice specific	16.2.0
2019-09	01#05	CF-192002	1421	4	Бv	authentication and authorizationt	10.2.0
2010-00	CT#85	CP-192060	1425	1	F	Undates to LIE-requested PDLL session procedures for	1620
2013 03	01#00	01-132000	1425	1	1	converting to MA PDU session	10.2.0
2019-09	CT#85	CP-192059	1427	2	F	Applicability of unified access control for wireline 5G access	1620
2010 00	01//00	01 102000		-	•	network	10.2.0
2019-09	CT#85	CP-192070	1434	2	F	Correction of handling of 5GSM causes #27 and #70 for	16.2.0
2010.00	CT#95	CP 102057	1/25	1	Б	SiNE_50	16.2.0
2019-09	CT#05	CP-192037	1435	1	Г	Confections to the disability and re-enabling of NT mode	16.2.0
2019-09	CT#05	CP-192057	1430	1	Г Г	Previous Tot SGPP and non-SGPP access types	10.2.0
2019-09	CT#05	CP-192037	1430	1	Г	Access control on MO L D	16.2.0
2019-09	CT#05	CP-192063	1440	1	Б		10.2.0
2019-09	CT#05	CP-192057	1442		Г	Correction of the OoS rule energian name	16.2.0
2019-09	CT#05	CP-192057	1440	2	Г D	Correction of the QOS fulle operation hame	10.2.0
2019-09	CT#05	CP-192073	1453	2	В	Handling of non-integrity protected messages in an SNPN	16.2.0
2019-09	01#85	CP-192073	1454	1	C E	Support of network slicing in an SNPN	16.2.0
2019-09	01#85	CP-192073	1456	1	F	Lists of forbidden networks in an SNPN	16.2.0
2019-09	01#85	CP-192073	1457	1	F		16.2.0
2019-09	CT#85	CP-192073	1458	1	F	5G-GUTI not globally unique in an SNPN	16.2.0
2019-09	CT#85	CP-192073	1460	2	С	Shared network broadcasting PLMN identity(ies) or SNPN identity(ies)	16.2.0
2019-09	CT#85	CP-192057	1461	2	F	Alignment on the implication description of type of list = "11" in service area list IE	16.2.0
2019-09	CT#85	CP-192073	1466	1	В	Support of time sensitive communication	16.2.0
2019-09	CT#85	CP-192073	1470	2	В	Port management information container: Delivery via the NAS protocol and coding	16.2.0
2019-09	CT#85	CP-192054	1474	2	F	Update the UE-requested PDU session establishment	16.2.0
						procedure regarding always-on PDU session for URLLC	
						service	
2019-09	CT#85	CP-192054	1476	2	F	Update the network-requested PDU session modification	16.2.0
						procedure regarding always-on PDU session for URLLC	
						service	
2019-09	CT#85	CP-192057	1482	1	F	Periodic update is allowed also in non-allowed area	16.2.0
2019-12	CT#86	CP-193094	1233	4	F	QoS operation upon activation of dedicated EPS bearer	16.3.0
2019-12	CT#86	CP-193100	1275	3	F	Management of forbidden area in wireline access	16.3.0
2019-12	CT#86	CP-193094	1284	2	F	Handling of unknown, unforeseen, and erroneous UPDS data	16.3.0
						in UE policy delivery service	
2019-12	CT#86	CP-193094	1327	4	F	Handing of 5GMM parameters during certain mobility	16.3.0
						registration failurest	

2019-12	CT#86	CP-193094	1339	2	F	Consistent back off timer handling for EPC interworking	16.3.0
2019-12	CT#86	CP-193094	1342	2	F	Local release when receiving REFRESH command for routing indicator in RRC inactive	16.3.0
2019-12	CT#86	CP-193094	1349	3	F	UAC check for services started in WLAN and being transferred to 3GPP access	16.3.0
2019-12	CT#86	CP-193094	1352	2	F	Addition of abnormal case handling for T3346 running in NAS transport procedure	16.3.0
2019-12	CT#86	CP-193094	1373	4	F	Maintenance of forbidden TA lists for non-integrity protected NAS reject	16.3.0
2019-12	CT#86	CP-193094	1374	2	F	EMM parameters handling for 5G ony causes	16.3.0
2019-12	CT#86	CP-193094	1375	5	F	Covering 5GMM cuase #31 for DoS attack	16.3.0
2019-12	CT#86	CP-193094	1376	3	F	UE checking the active EPS bearer ID for mapped QoS flows	16.3.0
2019-12	CT#86	CP-193094	1378	3	F	5G NAS security context for interworking	16.3.0
2019-12	CT#86	CP-193094	1397	2	F	Clarification on handling of MP-REG	16.3.0
2019-12	CT#86	CP-193094	1400	3	F	Correction on handling and coding of Mapped EPS bearer contexts	16.3.0
2019-12	CT#86	CP-193094	1401	4	F	QoS rule and QoS flow error handling	16.3.0
2019-12	CT#86	CP-193094	1402	2	F	QoS error operation during interworking	16.3.0
2019-12	CT#86	CP-193115	1403	1	F	Correction on the condition for handling reattempt for PDU session type related rejection	16.3.0
2019-12	CT#86	CP-193094	1404	1	F	Additional 5GS PDU session rejection cause values	16.3.0
2019-12	CT#86	CP-193094	1409	4	F	Emergency registered state handling.	16.3.0
2019-12	CT#86	CP-193094	1416	2	F	Correction for N1 signalling connection release	16.3.0
2019-12	CT#86	CP-193094	1418	3	F	Emergency PDU session establishment upon expiry of timer T3580	16.3.0
2019-12	CT#86	CP-193095	1426	1	F	Correction to the storage of 5GMM information; SOR counter and a UE parameter update counter	16.3.0
2019-12	CT#86	CP-193118	1431	8	В	Transmission of the UE CAG capability to the network	16.3.0
2019-12	CT#86	CP-193095	1432	3	F	Correction of 5GMM state for cause #27 "N1 mode not allowed"	16.3.0
2019-12	CT#86	CP-193095	1433	3	F	Correction and clarification of interworking with ePDG	16.3.0
2019-12	CT#86	CP-193095	1449	1	F	No requirement for network to store a back-off timer per UE	16.3.0
2019-12	CT#86	CP-193104	1450	5	В	Slice-specific authentication and authorization procedure	16.3.0
2019-12	CT#86	CP-193115	1452	3	В	Back-off control in case of routing failure	16.3.0
2019-12	CT#86	CP-193118	1469	3	С	PDU session used for TSC established as an always-on PDU	16.3.0
2019-12	CT#86	CP-103005	1480	7	F	Correction to delivery of manned S-NSSAI(s)	1630
2019-12	CT#86	CP-193095	1489	7	F	Correction of statement related to K'AME derivation during S1	16.3.0
2010 12	CT#96	CP 102005	1400	1	Г С	to N1 handover	16.2.0
2019-12	OT#00	OP 400005	1490	1	с Б	inter system change from S1 to N1 mode	10.3.0
2019-12	CT#86	CP-193095	1493	1	F	Removal of update status dependency for sub-state selection	16.3.0
2019-12	C1#86	CP-193089	1494	2	F	NAS signalling connection	16.3.0
2019-12	CT#86	CP-193088	1495	1	F	Apply UAC during resume of the N1 NAS signalling connection	16.3.0
2019-12	CT#86	CP-193095	1496	2	F	Receiving deregistration with cause #72 when registered for both 3GPP and Non-3GPP accesst	16.3.0
2019-12	CT#86	CP-193088	1497	2	F	Correcting DDX description	16.3.0
2019-12	CT#86	CP-193113	1499		F	Correct UE radio capability ID reference	16.3.0
2019-12	CT#86	CP-193101	1500	2	В	MA PDU session release	16.3.0
2019-12	CT#86	CP-193101	1501		F	Add the missing MA PDU request	16.3.0
2019-12	CT#86	CP-193101	1504	2	F	MA PDU session rejection due to operator policy and subscription policy	16.3.0
2019-12	CT#86	CP-193104	1505	6	С	Introduction of pending NSSAI for network slice-specific authentication and authorization	16.3.0
2019-12	CT#86	CP-193089	1506	1	F	Resolving Editor's Note for need of new EPD in CPSR message	16.3.0
2019-12	CT#86	CP-193089	1507	1	F	Resolving Editor's Note on whether CIoT small data container IE can be TV format in CPSR message	16.3.0
2019-12	CT#86	CP-193088	1509	2	F	Corrections of service gap controlt	16.3.0
2019-12	CT#86	CP-193118	1510	3	В	Primary authentication using EAP methods other than EAP- AKA' and EAP-TLS	16.3.0
2019-12	CT#86	CP-193118	1511	1	В	Serving network name in SNPN	16.3.0
2019-12	CT#86	CP-193118	1512	2	В	Extensions of EAP-TLS usage in primary authentication	16.3.0
2019-12	CT#86	CP-193118	1513		В	Extensions of EAP-AKA' usage in primary authentication	16.3.0
2019-12	CT#86	CP-193100	1514	1	F	Further alignment with stage-2 on PEI for 5G-RG and FN-RG	16.3.0
2019-12	CT#86	CP-193100	1515	1	F	Corrections for wireline access service area restrictions	16.3.0
2019-12	CT#86	CP-193088	1516	2	F	Corrections related to Service Gap timer	16.3.0
2019-12	CT#86	CP-193088	1517	1	F	5GS Control plane CloT data transfer for UE in connected mode	16.3.0
					E	NIAC measure container for Control plane contine require	1000

2019-12	CT#86	CP-193088	1519	4	F	Abnormal case handling for uplink NAS transport for non-	16.3.0
2019-12	CT#86	CP-193088	1520	2	F	Applicability of existing emergency PDU session request type	16.3.0
2019-12	CT#86	CP-193118	1523		F	Corrections on the Port management information container IE	16.3.0
2019-12	CT#86	CP-193104	1526	6	F	Deregistration due to the failed network Slice-Specific Authentication and Authorization	16.3.0
2019-12	CT#86	CP-193088	1527	1	F	Correction of the misuse of T3525	16.3.0
2019-12	CT#86	CP-193089	1529		F	Inclusion of the T3324 IE in REGISTRATION ACCEPT	16.3.0
2019-12	CT#86	CP-193088	1532	1	С	Control plane service request message and abnornal cases on	16.3.0
2010 12	CT#06	CD 102112	1505	1	D	the network side	16.2.0
2019-12	CT#86	CP-193113	1535	1	Б	RACS support at EPS to 5GS IWK	16.3.0
2019-12	CT#86	CP-193113	1541	1	Г В	Initial Small Data rate control parameters	16.3.0
2019-12	CT#86	CP-193104	1542	1	F	Reset of registration attempt counter at registration reject with cause #62	16.3.0
2019-12	CT#86	CP-193095	1543		F	Incorrect reference in Authentication subclause	16.3.0
2019-12	CT#86	CP-193095	1544		F	IE inclusion criteria style alignment	16.3.0
2019-12	CT#86	CP-193088	1545	1	С	CP CIoT header compression, UE initiated re-configuration	16.3.0
2019-12	CT#86	CP-193095	1546	1	F	Corrrection to the notification procedure and collision with UE initiated de-registration	16.3.0
2019-12	CT#86	CP-193095	1547	1	F	Request of IMEISV via the security mode control procedure	16.3.0
2019-12	CT#86	CP-193089	1548	1	F	Service gap control, simultaneous registration over N3GPP access in same PLMN	16.3.0
2019-12	CT#86	CP-193104	1549	3	F	Registration reject due to no allowed slices and NW slice specific authentication and authorization	16.3.0
2019-12	CT#86	CP-193118	1551	2	F	Clarification to forbidden TAI lists for SNPN	16.3.0
2019-12	CT#86	CP-193118	1552	1	F	Clarification to PLMN-SEARCH substate.	16.3.0
2019-12	CT#86	CP-193118	1553	1	F	Correction to SNPN enabled terminology.	16.3.0
2019-12	CT#86	CP-193118	1555	2	F	5GMM cause value #74 and requirements for non-integrity protected reject messages	16.3.0
2019-12	CT#86	CP-193118	1556	1	F	Discarding the REGISTRATION REJECT message with 5GMM cause #76 received without integrity protection	16.3.0
2019-12	CT#86	CP-193118	1558		F	Handling of the DEREGISTRATION REQUEST message with 5GMM cause value #74 or #75 in a PLMN	16.3.0
2019-12	CT#86	CP-193118	1559		F	Reset of the registration attempt counter upon receipt of NPN- based cause values	16.3.0
2019-12	CT#86	CP-193118	1560		F	Handling of T3510, T3517, and the service request attempt counter in an SNPN	16.3.0
2019-12	CT#86	CP-193118	1561	4	F	Missing requirement on UE policies for SNPN	16.3.0
2019-12	CT#86	CP-193088	1562	1	F	Corrections of RRC requirements specified in NAS specs	16.3.0
2019-12	CT#86	CP-193088	1563	5	F	Handling of maximum number of allowed active DRBs	16.3.0
2019-12	CT#86	CP-193118	1565		F	Correction to name of IE carrying residence time for TSN	16.3.0
2019-12	CT#86	CP-193095	1509	1	Г F	Correction to re-attempt indicator IE description	16.3.0
2019-12	CT#86	CP-193104	1571	2	F	5GMM Cause #62	16.3.0
2019-12	CT#86	CP-193095	1574	-	F	SMC message trigger	16.3.0
2019-12	CT#86	CP-193095	1575	1	F	Congestion control for UL NAS TRANSPORT message	16.3.0
2019-12	CT#86	CP-193088	1577	1	F	Applying small data rate control at inter-system change	16.3.0
2019-12	CT#86	CP-193118	1578	1	F	Counter events handling for SNPN	16.3.0
2019-12	CT#86	CP-193118	1579	1	F	5GMM cause handling for SNPN	16.3.0
2019-12	CT#86	CP-193101	1580	1	F	Provision of MA PDU session information during the UE- initiated NAS transport procedure initiation	16.3.0
2019-12	CT#86	CP-193095	1583		F	Octet alignment for 5G-GUTI in 5GS mobile identity IE	16.3.0
2019-12	CT#86	CP-193095	1584	1	F	Clarification on handling of 5G NAS security context	16.3.0
2019-12	C1#86	CP-193089	1585	1	С	Resolving Editor's notes on the Data Type field for the Control Plane Service Request message	16.3.0
2019-12	CT#86	CP-193095	1586	1	F	Registration upon change of UE radio capability during 5GMM- CONNECTED mode	16.3.0
2019-12	CT#86	CP-193095	1589		F	Handling of 5GSM cause #54 PDU session does not exist	16.3.0
2019-12	CT#86	CP-193096	1590	1	F	Correction to the indication upon receipt of 5GMM cause #91	16.3.0
2019-12	CT#86	CP-193096	1592	2	Г Б	Handling of errors in mapped EPS bearer contexts	16.3.0
2019-12	CT#86	CP-193090	1593	1	r R	S-NSSAI not available in the current PLMN" when non	1630
2013-12	01#00	01-193104	1550	4	Б	NSSAA subjected UE requesting the S-NSSAI subjects to NSSAA	10.5.0
2019-12	CT#86	CP-193104	1602	6	В	NSSAI storage impact with NSSAA	16.3.0
2019-12	CT#86	CP-193096	1603		F	EHPLMN and Dual registration	16.3.0
2019-12	CT#86	CP-193119	1604	7	F	CAG only UE and emergency procedure.	16.3.0
2019-12	CT#86	CP-193089	1605	3	F	SGC timer and handling during intersystem change	16.3.0
2019-12	CT#86	CP-193096	1608		F	Abbreviation of AUSF and UDM	16.3.0
2019-12	CT#86	CP-193106	1619	1	В	Release of NAS signalling connection for the UE authorized for V2X communication over PC5	16.3.0
2019-12	CT#86	CP-193113	1622	3	F	Identification procedure for RACS	16.3.0

2019-12	CT#86	CP-193088	1623	2	F	Control plane service request, correction regarding service	16.3.0
2010 12	CT#06	CD 102006	1605	2	Б	Dreadenee for approaction flow	16.2.0
2019-12	CT#00	CP-193096	1020	2	Г	Linified Access Central for IMC registration related signalling	10.3.0
2019-12	CT#00	CP-193096	1020	4	Г	Unified Access Control for IWS registration related signalling	16.3.0
2019-12	CT#00	CP-193096	1033	1	B	Failed and the second s	10.3.0
2019-12	C1#86	CP-193101	1636		D	A PDU session establisment request upgraded to	16.3.0
2019-12	CT#86	CP-193089	1640	1	С	Handling of MT paging for CP-CIoT	16.3.0
2019-12	CT#86	CP-193088	1647	2	B	5GS NAS extended timers for NB-N1 mode and WB-N1/CF	16.3.0
				-	_	mode devices	
2019-12	CT#86	CP-193089	1648	1	С	Serving PLMN rate control at PDU session modification	16.3.0
2019-12	CT#86	CP-193096	1649	1	F	Clarification on the Mapped EPS bearer context	16.3.0
2019-12	CT#86	CP-193096	1650	1	D	Editorial corrections to text related to the status of PDU	16.3.0
2010 12	01//00	01 100000	1000		D	sessions during SR procedure	10.0.0
2019-12	CT#86	CP-193089	1651	1	F	Short MAC and ngKSL in Control plane service request NAS	16.3.0
	0	0. 100000		-	-	message	
2019-12	CT#86	CP-193089	1657	2	В	Introduction of NSSAI efficient signalling for IoT devices	16.3.0
2019-12	CT#86	CP-193104	1658	1	С	Removal of Editor's note on conditions of accepting	16.3.0
				-	-	registration	
2019-12	CT#86	CP-193089	1660	2	F	UE behaviour when T3448 timer running	16.3.0
2019-12	CT#86	CP-193088	1661		F	AMF behaviour for mobility registration when SGC timer	16.3.0
						running	
2019-12	CT#86	CP-193096	1662		F	Clarification on the UE policy container	16.3.0
2019-12	CT#86	CP-193096	1663	1	F	DNN Replacement	16.3.0
2019-12	CT#86	CP-193096	1664	1	F	Faulty and missing reference	16.3.0
2019-12	CT#86	CP-193096	1665	1	F	Correction of handling of de-registration procedure in	16.3.0
				-	-	ATTEMPTING-REGISTRATION-UPDATE	
2019-12	CT#86	CP-193096	1666	1	F	Corrections and enhancements for T3540	16.3.0
2019-12	CT#86	CP-193084	1668	2	А	Handling of UE NAS Count during handover from N1 mode to	16.3.0
				_		S1 mode	
2019-12	CT#86	CP-193096	1669	2	F	Handling multiple QoS errors during a PDU session	16.3.0
						modification procedure – Option 1	
2019-12	CT#86	CP-193104	1674	1	F	No info on S-NSSAI subject to NSSAA in UE	16.3.0
2019-12	CT#86	CP-193169	1676	2	F	Equivalent SNPNs not supported for stand-alone non-public	16.3.0
						networks	
2019-12	CT#86	CP-193096	1680		F	5GMM state in non-3GPP access not impacting EMM state of	16.3.0
						single-registered UE	
2019-12	CT#86	CP-193096	1681	1	F	Registration attempt counter reset by single-registered UE	16.3.0
2019-12	CT#86	CP-193096	1682		F	Correction for 5GMM and inter-system change	16.3.0
2019-12	CT#86	CP-193096	1683		F	Correction for 5GSM and inter-system change with N26	16.3.0
2019-12	CT#86	CP-193097	1684	1	F	Clarification to forbidden PLMN list	16.3.0
2019-12	CT#86	CP-193100	1685		С	Access stratum connection and user-plane resources for	16.3.0
						trusted non-3GPP access and wireline access	
2019-12	CT#86	CP-193100	1686	1	В	Usage of PDU session identity for the PDU sessions	16.3.0
						requested by the TWIF	
2019-12	CT#86	CP-193100	1687		С	Removal of Session-TMBRt	16.3.0
2019-12	CT#86	CP-193100	1688		С	5G-RG and W-AGF acting on behalf of FN-RG performing UE	16.3.0
						requirements	
2019-12	CT#86	CP-193097	1690		F	Correction for 5GS network feature support IE	16.3.0
2019-12	CT#86	CP-193106	1692	1	В	UPDS updates enabling UE-requested V2X policy provisioning	16.3.0
	0					procedure	
2019-12	CT#86	CP-193119	1693	1	В	Exchange of port management capabilities during PDU	16.3.0
0010.15	OT "CC		400 1	1	a		40.0.0
2019-12	01#86	CP-193119	1694	1	C	Additional abnormal cases in SNPN	16.3.0
2019-12	CT#86	CP-193106	1697	1	В	5QI 86 introduction	16.3.0
2019-12	CT#86	CP-193097	1698	1	F	Deletion of UE radio capability in the network	16.3.0
2019-12	CT#86	CP-193115	1699	1	C	Retry restriction on non-3GPP access	16.3.0
2019-12	CT#86	CP-193115	1700	1	F	No retry restriction for 5GSM cause value #39	16.3.0
2019-12	CT#86	CP-193089	1702	3	C	Support of UE specific DRX for NB-IoT	16.3.0
2019-12	CT#86	CP-193090	1705	1	В	Transfer of Ciphering Key Information for Broadcast Location	16.3.0
			4=-		-	Assistance Data	10
2019-12	CT#86	CP-193089	1707		F	1 limer 13448	16.3.0
2019-12	CT#86	CP-193097	1709	1	F	Rejected NSSAI	16.3.0
2019-12	CT#86	CP-193090	1710	1	F	Sending location services data from 5GMM-IDLE mode using	16.3.0
					l	the Control Plane Service Request message	
2019-12	CT#86	CP-193088	1711	1	F	Correction of the format of CIoT small data container	16.3.0
2019-12	CT#86	CP-193089	1712	3	F	Removal of a Code-Point in Control Plane Service Type	16.3.0
2019-12	CT#86	CP-193097	1715	1	F	Correction to EPLMN list deletion for 5GMM cause #7	16.3.0
2019-12	CT#86	CP-193097	1716		F	Correction to UE OS ID encoding	16.3.0
2019-12	CT#86	CP-193119	1717	3	F	Removal of CAG suscription while emergency PDU session is	16.3.0
						established.	
2019-12	CT#86	CP-193119	1718	1	F	Detenition of CAG cell, CAG ID and CAG selection	16.3.0

2019-12	CT#86	CP-193119	1719	2	F	Handling of Service request message in a non-subscribed CAG cell	16.3.0
2019-12	CT#86	CP-193119	1720	1	F	Updation of LIMITED SERVICE state for CAG	16.3.0
2019-12	CT#86	CP-193116	1722	2	F	Handling of parameters stored in the ME memory	16.3.0
2019-12	CT#86	CP-193104	1723	2	F	Network slice authentication and emergency procedure	16.3.0
2019-12	CT#86	CP-193115	1728	-	F	Excluding 5GSM causes for congestion control from SINE	1630
2010-12	CT#86	CP-193089	1720	1	F	ngKSI for CONTROL PLANE SERVICE REQUEST message	16.3.0
2019-12	CT#86	CP-103007	1731	1	F	Inclusion of PDU session reactivation result error cause IE	16.3.0
2019-12	CT#00	CF-193097	1731	1	Г	INCload INCLS/ formate support	16.3.0
2019-12	CT#00	CF-193097	1732	1	Г		10.3.0
2019-12		CP-193097	1733	1	F	PEI format for non-3GPP access only UE	16.3.0
2019-12	C1#86	CP-193097	1/3/	1	F	Correction to the coding of EPS bearer identity	16.3.0
2019-12	CT#86	CP-193097	1740	1	F	UE handling upon receipt of 5GSM #46 out of LADN service area	16.3.0
2019-12	CT#86	CP-193088	1742	1	С	CIoT user data container in UL NAS transport message not routable	16.3.0
2019-12	CT#86	CP-193089	1744	1	С	Service gap control, supporting UE sends MO user data when connected when timer running	16.3.0
2019-12	CT#86	CP-193088	1749	2	F	Correction on the condition for including CP only indication	16.3.0
2019-12	CT#86	CP-193097	1752		F	T3540 in Service Accept Case	16.3.0
2019-12	CT#86	CP-193106	1753	1	С	Access control for UE triggered V2X policy provisioning procedure	16.3.0
2019-12	CT#86	CP-193119	1754	2	F	Abnormal cases for 5GMM cause values #74 and #75	16.3.0
2019-12	CT#86	CP-193110	1755	1	C	Rejected NSSAL in SNPNs	16.3.0
2010-12	CT#86	CP-103110	1756	2	C	Introduction of SNPN-specific attempt counter for pon-3GPP	16.3.0
2010 12	01#00	01-100110	1750	2	C	access and counter for "the entry for the current SNPN considered invalid for non-3GPP access" events	10.0.0
2019-12	CT#86	CP-193119	1758	1	F	Maintenance of forbidden TA lists for non-integrity protected NAS reject in an SNPN	16.3.0
2019-12	CT#86	CP-193097	1759	1	F	Handling of UAC for an MO IMS registration related signalling	16.3.0
2019-12	CT#86	CP-193097	1760	-	D	Correction of the definition of Network slicing information	1630
2010-12	CT#86	CP-103110	1766	2	F	Moving Appex E to TS 24 5vy	16.3.0
2019-12	CT#00	CP 102007	1767	1	Г Б	Follow on request addepoint value	16.3.0
2019-12	CT#00	CP-193097	1707	1	Г	No CAC appears control for amorganou convices	16.3.0
2019-12	CT#86	CP-193119	1770	1	F	No CAG access control for emergency services	16.3.0
2019-12	CT#86	CP-193119	1772	2	F	Coding of the CAG-ID	16.3.0
2019-12	CT#86	CP-193097	1773	1	D	limer order in timer tables	16.3.0
2019-12	CT#86	CP-193097	1774	1	F	Mobility registration accept with NSSAIs	16.3.0
2019-12	CT#86	CP-193097	1775	2	F	Correction to PLMN change with 5G-EA0	16.3.0
2019-12	CT#86	CP-193101	1778	1	D	Editorial on PDU session establisment request upgraded to MA PDU session	16.3.0
2020-03	CT#87e	CP-200117	1533	6	С	NW slice specific authentication and authorization failure and revocation	16.4.0
2020-03	CT#87e	CP-200130	1567	2	В	CAG information towards the lower layers for paging	16.4.0
2020-03	CT#87e	CP-200113	1641	5	В	PDU session handling for N5CW device	16.4.0
2020-03	CT#87e	CP-200107	1672	4	С	Handling of user-plane resources for NB-IoT UEs having at least two PDU sessions	16.4.0
2020-03	CT#87e	CP-200113	1689	2	С	Secondary authentication and W-AGF acting on behalf of FN- RG	16.4.0
2020-03	CT#87e	CP-200111	1781	3	F	Correcting unimplementable condition regarding N26	16.4.0
2020-03	CT#87e	CP-200111	1784	2	F	Maintain Selected EPS NAS security algorithms during N1	16.4.0
2020-03	CT#87e	CP-200111	1785		F	Correction for AUTHENTICATION REJECT handling	16.4.0
2020-03	CT#87e	CP-200111	1786	1	D	Editorial correction of an input parameter for 5G NAS message	16.4.0
2020-03	CT#87e	CP-200111	1789	1	F	Correction to sending of EPS NAS message container in	16.4.0
0000.00	OT#07-	00.000444	4700		D		40.4.0
2020-03		CP-200111	1793		D		16.4.0
2020-03	C1#87e	CP-200111	1794	2	F	Handling of unsupported SSC mode	16.4.0
2020-03	CT#87e	CP-200111	1797	1	F	Abnormal case for service request procedure	16.4.0
2020-03	CT#87e	CP-200111	1798	2	F	Mapped EPS bearer contexts deletion	16.4.0
2020-03	CT#87e	CP-200111	1799	1	F	Service Request for PS Data Off	16.4.0
2020-03	CT#87e	CP-200111	1800	1	F	Abnormal case for UL NAS TRANSPORT	16.4.0
2020-03	CT#87e	CP-200111	1804		F	Declare syntactical error when both MFBR uplink and MFBR downlink equal zero	16.4.0
2020-03	CT#87e	CP-200111	1805	1	D	Editorial correction to T3447 timer behavior	16.4.0
2020-03	CT#87e	CP-200111	1807	1	F	Handling multiple QoS errors during a PDU session establishment procedure	16.4.0
2020-03	CT#87e	CP-200111	1810	1	F	Inclusion of PDU session reactivation result error cause IF	1640
2020-03	CT#87e	CP-200111	1811	1	F	Correction on NAS transparent container for 5G-4G	16.4.0
2020 02	CT#97^	CP_200111	1210	2	F	Deletion of the rejected NSSAL for the current registration area	16.4.0
2020-03	CT#970	CP_200111	1012	2	r F	Trigger for stopping timer T2511	16.4.0
2020-03	CT#07-	CP 200111	1013		r E	Correction on T2502 for deastinated value	10.4.0
2020-03	01#0/6	06-200111	1014	1	1.	Conection on 13302 for deactivated value	10.4.0

2020-03	CT#87e	CP-200111	1815	1	F	5GMM cause #22 for resetting registration attempt counter	16.4.0
2020-03	CT#87e	CP-200111	1816		F	Consistent use of additional 5G security information IE	16.4.0
2020-03	CT#87e	CP-200111	1817		F	Correction on N26 interface indicator	16.4.0
2020-03	CT#87e	CP-200111	1818		F	Correction on reference of TS 36.304	16.4.0
2020-03	CT#87e	CP-200111	1819	1	F	Inclusion of 5GSM cause in PDU session modification request	1640
2020-03	CT#87e	CP-200111	1820	1	F	Inclusion of 5GSM cause in PDU session release request	16.4.0
2020-03	CT#87e	CP-200112	1821		F	PDU session establishment reject with 5GSM #29	16.4.0
2020-03	CT#076	CP 200112	1021	1	Г Г	Correction on OoS rule/OoS flow synchronization	16.4.0
2020-03	CT#07e	CP-200112	1022	1	Г	Correction on NAS COUNT handling for intro N1 handover	16.4.0
2020-03	CT#67e	CP-200112	1024	1	F	Correction on NAS COUNT handling for Intra-INT handover	16.4.0
2020-03	C1#87e	CP-200112	1825	1	F	Correction on Uplink data status IE coding	16.4.0
2020-03	C1#8/e	CP-200112	1826		F	Acknowledgement of UCU procedure	16.4.0
2020-03	CT#87e	CP-200112	1827	1	F	Update bullet index to include all NAS transport cases	16.4.0
2020-03	CT#87e	CP-200112	1828	1	D	Editorial correction on payload container	16.4.0
2020-03	CT#87e	CP-200112	1829	2	F	Corrections on UE-initiated NAS transport procedure initiation	16.4.0
2020-03	CT#87e	CP-200112	1834	2	D	Corrections in specifying reasons for errors	16.4.0
2020-03	CT#87e	CP-200112	1835	1	F	UE handling of invalid QoS flow description	16.4.0
2020-03	CT#87e	CP-200112	1836	2	F	S-NSSAI as a mandatory parameter to support interworking	16.4.0
				_	-	with 5GS	
2020-03	CT#87e	CP-200112	1837	1	F	Optional IF description for release assistance indication IF	1640
2020-03	CT#87e	CP-200112	1838	1	F	UE handling of multiple OoS errors in EPS	1640
2020 03	CT#970	CP 200112	1920	1	F	S NSSAL value acception with the BO timer applied for all	16.4.0
2020-03	C1#0/e	CF-200112	1039	1	г		10.4.0
2020.02	CT#07a	CD 200112	1010	1	Б	F LIVINS	16.4.0
2020-03	C1#87e	CP-200112	1840	1	Г		16.4.0
0000.00	0T/07	00.000440	10.10		F	a PDU SESSION MODIFICATION REQUEST message	40.40
2020-03	C1#87e	CP-200112	1842		F	Correction in handling of persistent PDU session during the	16.4.0
						mobility registration update	
2020-03	CT#87e	CP-200112	1843		D	NAS signalling spelling correction	16.4.0
2020-03	CT#87e	CP-200112	1845	2	F	Reject non-emergency PDU session request attempt while UE	16.4.0
						registered for emergency services in the network	
2020-03	CT#87e	CP-200112	1846		F	Correction to IEI values	16.4.0
2020-03	CT#87e	CP-200112	1847	1	F	Correction to 5GMM cause IE	16.4.0
2020-03	CT#87e	CP-200112	1848		F	Correction to UCU procedure abnormal cases on NW side for	16.4.0
						a new TAI list	
2020-03	CT#87e	CP-200112	1853	3	F	Service area restrictions, case missing for when UE is out of	16.4.0
2020 00	01.010	0. 2002		-	-	allowed tracking area list and RA	
2020-03	CT#87e	CP-200112	1854		F	Correction to the Manned NSSALIE	1640
2020-03	CT#87e	CP-200112	1858		D	Correction reference to 5CSM procedures	16.4.0
2020-03	CT#07e	CD 200112	1000	2	D E	ECCM conscients for MA DDL cossion	16.4.0
2020-03	CT#67e	CP-200114	1000	Z	Г	SGSM capabilities for MA PDU session	16.4.0
2020-03	CT#87e	CP-200114	1862	4	В	MA PDU session is not supported	16.4.0
2020-03	C1#8/e	CP-200117	1869	2	F	Cleanups on introduction of pending NSSAI	16.4.0
2020-03	CT#87e	CP-200113	1870		F	SUCI used by W-AGF acting on behalf of FN-RG	16.4.0
2020-03	CT#87e	CP-200113	1871		F	Resolving editor's note on W-AGF acting on behalf of FN-RG	16.4.0
						not using the "null integrity protection algorithm" 5G-IA0	
2020-03	CT#87e	CP-200113	1872		F	Resolving editor's note on service area restrictions in case of	16.4.0
						FN-BRG	
2020-03	CT#87e	CP-200113	1873		F	Resolving editor's note in forbidden wireline access area	16.4.0
2020-03	CT#87e	CP-200113	1874		F	Wireline 5G access network and wireline 5G access clean up	16.4.0
2020-03	CT#87e	CP-200113	1875	1	F	PEI clean up	16.4.0
2020-03	CT#87e	CP-200113	1876		F	Alignment for stop of enforcement of mobility restrictions in	16.4.0
2020 00	0010	5. 200110	1010		-	5G-RG and W-AGE acting on behalf of FN-CRG	10.4.0
2020-02	CT#87o	CP-200112	1877	1	F	Introduction of GCI and GU	1640
2020-03	CT#870	CP_200100	1879	1	F	Always-On PDU session and URU C	16/10
2020-03	CT#070	CP 200109	1070	1	r' F	CAG information list storage	16.4.0
2020-03	CT#07-	CP 200130	10/9	1	r E		10.4.0
2020-03		CP-200107	1881	2	Г Г	Abhormal case for cause #31	10.4.0
2020-03	CT#87e	CP-200130	1882		F	Removal of Editor's note on the use of the NOTIFICATION	16.4.0
						message in SNPNs	
2020-03	CT#87e	CP-200135	1884	3	В	Including CAG information list in REGISTRATION ACCEPT	16.4.0
						message	
2020-03	CT#87e	CP-200130	1885		F	Update of text on time synchronization	16.4.0
2020-03	CT#87e	CP-200125	1886		F	Removal of Editor's note on applicability of RACS to SNPNs	16.4.0
2020-03	CT#87e	CP-200125	1887		С	Finalizing the encoding of the UE radio capability ID	16.4.0
2020-03	CT#87e	CP-200125	1888	1	В	UE radio capability ID deletion upon Version ID change	16.4.0
2020-03	CT#87e	CP-200117	1889	2	В	Handling of S-NSSAIs in the pending NSSAI	16.4.0
2020-03	CT#87e	CP-200107	1891	-	F	Resolve Editor's Notes on NB-N1 mode extended NAS timers	16.4.0
2020 00	0010	5. 200107	1001		-	for CF	10.4.0
2020-02	CT#876	CP-200107	1802		F	Resolve Editor's Notes on W/B-N1 mode extended NAS timore	1640
2020-03		51-200107	1032		•	for CE	10.4.0
2020 02	CT#970	CP_200117	1202	1	F	Clarification on HPLMN S NSSAL	16.4.0
2020-03	OT#07	OF-200117	1093	1	r E		10.4.0
2020-03		CP-200114	1896	1	Г	IVIA PDU Session and one set of QoS parameters	10.4.0
2020-03	C1#8/e	CP-200117	1899	1	F F	Update to registration procedure due to eNS	16.4.0
	OT //						
2020-03	CT#87e	CP-200107	1900	1	F	Stop 13565 upon connection resumption	16.4.0

2020-03	CT#87e	CP-200114	1903		F	Minor Correction to ATSSS container IE desciption	16.4.0
2020-03	CT#87e	CP-200107	1907	3	В	Support for the signalling of the capability for receiving WUS assistance information	16.4.0
2020-03	CT#87e	CP-200113	1910	3	F	Enabling mobility with (emergency) sessions/connections between the (trusted) non-3GPP access network connected to	16.4.0
2020-03	CT#87e	CP-200117	1913	1	F	UE behaviour for other causes in the rejected NSSAI during deregistration procedure	16.4.0
2020-03	CT#87e	CP-200117	1914	1	F	Pending NSSAI update for the configured NSSAI in the CUC	16.4.0
2020-03	CT#87e	CP-200139	1015	2	F	Cleanup for NSSAA message and coding	1640
2020-03	CT#87e	CP-200117	1916	1	F	Rejected NSSAI during the initial registration procedure	16.4.0
2020-03	CT#87e	CP-200107	1917	1	F	ITE behaviour when T3447 running	16.4.0
2020-03	CT#87e	CP-200108	1918	1	C	PDLI session release	16.4.0
2020-03	CT#87e	CP-200113	1919	1	B	ACS information via DHCP	16.4.0
2020-03	CT#87e	CP-200117	1010	1	D	Name of the rejected NSSAI cause values	16.4.0
2020-03	CT#87e	CP-200125	1922	1	F	Clarification of the cause of start of T3550	16.4.0
2020-03	CT#87e	CP-200130	1923	1	F	Clarification of forbidden TAL lists for SNPN	16.4.0
2020-03	CT#87e	CP-200130	1924	1	F	Deletion of all CAG-IDs of a CAG cell 5GMM cause for #76	16.4.0
2020-03	CT#87e	CP-200130	1926		F	Clarification of the rejected NSSAL cause value	16.4.0
2020-03	CT#87e	CP-200130	1927		F	Removal of term CAG access control	16.4.0
2020-03	CT#87e	CP-200130	1928		F	Definition alignment for UF-DS-TT residence time	16.4.0
2020-03	CT#87e	CP-200107	1930		F	Ciphering and deciphering handling of CPSR message	16.4.0
2020-03	CT#87e	CP-200107	1932	1	C	Truncated 5G-S-TMSI over NAS	16.4.0
2020-03	CT#87e	CP-200107	1933	-	F	AMF behavior on stop T3448	16.4.0
2020-03	CT#87e	CP-200107	1934	1	F	Correction on SMS in payload container IE in CPSR message	16.4.0
2020-03	CT#87e	CP-200130	1935	1	F	Correction on 5GMM cause #74/#75 for no touching non-	16.4.0
				-	-	3GPP access	
2020-03	CT#87e	CP-200130	1937		F	Correction on term "non-3GPP access" used in SNPN	16.4.0
2020-03	CT#87e	CP-200130	1938		F	Reset the registration attempt counter for #76 in service reject	16.4.0
2020-03	CT#87e	CP-200117	1941	1	F	ENs resolution for revoked or failed NSSAA	16.4.0
2020-03	CT#87e	CP-200117	1942		D	Consistent name for NSSAA	16.4.0
2020-03	CT#87e	CP-200127	1943		F	No retry in 4G for PDU session type related 5GSM causes	16.4.0
2020-03	CT#87e	CP-200127	1944		F	Correction on UE retry restriction on EPLMN	16.4.0
2020-03	CT#87e	CP-200130	1945	1	F	Clarification on Public Network Integrated NPN in TS 24.501	16.4.0
2020-03	CT#87e	CP-200130	1946	1	F	UE receives CAG information in SNPN access mode	16.4.0
2020-03	CT#87e	CP-200130	1947	2	F	Establish PDU session to transfer port management	16.4.0
						information containers	
2020-03	CT#87e	CP-200114	1948	2	F	ATSSS Non-MPTCP traffic support	16.4.0
2020-03	CT#87e	CP-200130	1949	1	F	Correction for the wrongly implemented CR1693r1	16.4.0
2020-03	CT#87e	CP-200117	1953		F	NSSAA revocation function	16.4.0
2020-03	CT#87e	CP-200107	1956	1	F	Stopping of T3513 after connection resume for user plane CIoT 5GS optimization	16.4.0
2020-03	CT#87e	CP-200117	1958		F	Correction UE behaviour when the UE recives the pending NSSAI	16.4.0
2020-03	CT#87e	CP-200107	1961	1	F	Adding an editor's note for suspend indication due to user plane CloT 5GS optimization	16.4.0
2020-03	CT#87e	CP-200107	1966	1	С	Recovery from fallback for UEs using CP CloT optimization	16.4.0
2020-03	CT#87e	CP-200119	1968	1	В	Triggering service request procedure for V2X communication	16.4.0
						over PC5 interface	
2020-03	CT#87e	CP-200117	1971	1	С	Removal of the use of Service area list IE during NSSAAt	16.4.0
2020-03	CT#87e	CP-200117	1973		F	Additional triggers for deletion of pending S-NSSAI	16.4.0
2020-03	CT#87e	CP-200114	1976	1	В	Considering allowed NSSAI when establishing MA PDU session	16.4.0
2020-03	CT#87e	CP-200114	1977	1	В	UE Handling upon receipt of PDU session release command	16.4.0
2020-03	CT#87e	CP-200107	1978	1	F	Correction to UL CIoT user data container not routable or not	16.4.0
						allowed to be routed	
2020-03	CT#87e	CP-200107	1979	1	F	Single downlink data only indication and release of N1 NAS signalling connection	16.4.0
2020-03	CT#87e	CP-200107	1980	2	F	PDU session status with control plane service request	16.4.0
2020-03	CT#87e	CP-200107	1981	1	F	Service gap control, correction when to start service gap control timer in UE and NW	16.4.0
2020-03	CT#87e	CP-200107	1982	1	F	Clarification of control plane service request message options	16.4.0
2020-03	CT#87e	CP-200107	1983	1	С	UAC updates for NB-IoT to include "MO exception data"	16.4.0
2020-03	CT#87e	CP-200107	1984	1	В	Clarification on the use of exception data reporting	16.4.0
2020-03	CT#87e	CP-200286	1985	5	F	Update SNPN key differences	16.4.0
2020-03	CT#87e	CP-200109	1987	1	F	Setting the Always-on PDU session indication IE in the PDU SESSION ESTABLISHMENT ACCEPT message	16.4.0
2020-03	CT#87e	CP-200117	1991		F	AMF updates the UE NSSAI storage after network slice-	16.4.0
2020-03	CT#87e	CP-200117	1995		F	Clarification on the S-NSSAI not subject to NSSAA included in	16.4.0
						allowed NSSAI	

2020-03	CT#87e	CP-200117	1996	2	F	Subscribed S-NSSAI marked as default and NSSAA	16.4.0
2020-03	CT#87e	CP-200117	1997		F	Additional conditions to the presence in the subscribed S- NSSAIs	16.4.0
2020-03	CT#87e	CP-200130	1998	1	F	Triggering mobility registration update due to manual CAG selection	16.4.0
2020-03	CT#87e	CP-200117	2000	1	F	Emergency PDU session handling after NSSAA failure	16.4.0
2020-03	CT#87e	CP-200125	2002		F	UE behaviour upon receipt of a UE radio capability ID deletion indication	16.4.0
2020-03	CT#87e	CP-200125	2005	1	F	Additional condition to change UE radio capability ID during mobility registration update	16.4.0
2020-03	CT#87e	CP-200125	2006	1	F	UE radio capability information storage not needed for RACS	16.4.0
2020-03	CT#87e	CP-200097	2008	3	F	Handling of a UE with an emergency PDU session in terms of CAG	16.4.0
2020-03	CT#87e	CP-200130	2012	2	F	N1 mode capability disabling and re-enabling for SNPN	16.4.0
2020-03	CT#87e	CP-200130	2013	1	F	#31 not applicable in an SNPN	16.4.0
2020-03	CT#87e	CP-200130	2015	1	F	Validity of the USIM for an SNPN and for a specific access type	16.4.0
2020-03	CT#87e	CP-200130	2016		F	Handling of 5GMM cause values #62 in an SNPN	16.4.0
2020-03	CT#87e	CP-200130	2017	1	F	No mandate to support default configured NSSAI or network slicing indication	16.4.0
2020-03	CT#87e	CP-200130	2018	2	F	SNN coding	16.4.0
2020-03	CT#87e	CP-200130	2019	1	F	5GMM cause value #74 in an SNPN with a globally-unique SNPN identity	16.4.0
2020-03	CT#87e	CP-200113	2020	1	В	Registration of N5GC devices via wireline access	16.4.0
2020-03	CT#87e	CP-200113	2021	1	F	Correction on EUI-64 as PEI	16.4.0
2020-03	CT#87e	CP-200113	2022		F	Corrections on N5CW support	16.4.0
2020-03	CT#87e					Addition of IEI values, editorial corrections, implementation of missing CR1985	16.4.1
2020-06	CT#88e	CP-201102	0793	9	F	Inclusion of ATTACH REQUEST message in REGISTRATION REQUEST message during initial registration when 5G-GUTI mapped from 4G-GUTI is used	16.5.0
2020-06	CT#88e	CP-201102	1379	6	F	Corrections on the abnormal cases of registration procedure for initial registration	16.5.0
2020-06	CT#88e	CP-201102	1415	12	F	Handling of MCS data in various 5GMM states.	16.5.0
2020-06	CT#88e	CP-201096	1701	5	С	Enhancement on CPSR for CIoT CP data transport	16.5.0
2020-06	CT#88e	CP-201113	1734	2	F	S-NSSAL in rejected NSSAL for the failed or revoked NSSAA	16.5.0
2020-06	CT#88e	CP-201102	1782	5	F	not to be requested Correcting transfer of connections/sessions if there is an	16.5.0
				-		emergency call	
2020-06	CT#88e	CP-201102	1792	2	F	Restricting handling of cause #9 to the access on which it was received	16.5.0
2020-06	CT#88e	CP-201102	1795	3	F	Clarification on use of operator-defined access categories	16.5.0
2020-06	CT#88e	CP-201102	1803	3	F	Add handling for UE configured to use timer T3245 in 5GS via 3GPP access	16.5.0
2020-06	CT#88e	CP-201102	1823	4	F	Correction on UE behaviour for service area restriction	16.5.0
2020-06	CT#88e	CP-201102	1841	4	F	Paging with two valid 5G-GUTIs	16.5.0
2020-06	CT#88e	CP-201136	1880	2	С	CAG-ID not provided to lower layers during NAS signalling connection establishment	16.5.0
2020-06	CT#88e	CP-201113	1912	5	С	Deleting Editors note regarding indefinite wait at the UE for NSSAA completion	16.5.0
2020-06	CT#88e	CP-201102	1974	1	F	Dual-registration requirements for EHPLMNs	16.5.0
2020-06	CT#88e						16.5.0
2020-06	CT#88e	CP-201136	2010	2	С	Correction in UE behavior upon receipt of 5GMM cause value #74 or #75 via a non-integrity protected NAS message	16.5.0
2020-06	CT#88e	CP-201109	2027		В	EPS interworking of MA PDU session of 5G-RG when N26 is not supported	16.5.0
2020-06	CT#88e	CP-201108	2028	1	F	Secondary authentication and W-AGF acting on behalf of N5GC device	16.5.0
2020-06	CT#88e	CP-201109	2029	2	В	EPS interworking of MA PDU session of 5G-RG when N26 is supported	16.5.0
2020-06	CT#88e	CP-201096	2030	2	F	Indication of change in the use of enhanced coverage	16.5.0
2020-06	CT#88e	CP-201096	2031		F	Integrity protection data rate for UEs that don't support N3 data transfer	16.5.0
2020-06	CT#88e	CP-201096	2032		F	Addition of Control Plane Service Request in the abnormal cases for service request procedure	16.5.0
2020-06	CT#88e	CP-201132	2033		F	Correction of certain erroneous Information Element Identifiers	16.5.0
2020-06	CT#88e	CP-201096	2034	5	F	DRX parameters for NB-IoT	16.5.0
2020-06	CT#88e	CP-201096	2035	-	F	Correcting a wrong reference	16.5.0
2020-06	CT#88e	CP-201102	2037		F	Clarification on DL only match-all packet filer	16.5.0
2020-06	CT#880	CP-201112	2038	1	F	Clarification S-NSSAI status in AME for NSSAA	16.5.0
2020-00	CT#00c	CP 201113	2000	2	- F	Undate description on LIE indicate supporting NSSA	16.5.0
2020-00	01#008	06-201113	2039	2	1	opuate description on or indicate supporting NOOAA	10.5.0

2202.06 CT#988 CP-20113 2040 1 F Pending NSSA update for the configured NSSA in the UCU 16.5.0 2020.06 CT#988 CP-20113 2041 1 F Applicability of PS data of its MA PDU session 16.5.0 2020.06 CT#988 CP-20113 2044 1 F Mater inggers PDU session release 16.5.0 2020.06 CT#988 CP-201102 2047 F Correction of the handling of SGMM cause #27 16.5.0 2020.06 CT#988 CP-201102 2051 F Additional state#iss-states 16.5.0 2020.06 CT#988 CP-201102 2055 F ANDSP is not supported by SG-RG and WAGF 16.5.0 2020.06 CT#988 CP-201102 2056 F Clarkright description for NexeX-Sile S-RGFIne 16.5.0 2020.06 CT#988 CP-201102 2056 F Clarkright description for NexeX-Sile S-RGFIne 16.5.0 2020.06 CT#988 CP-20113 2058 3 C Adding a new shormal case on the network siles Paraces 16								
2020-66 CT4886 CP-201103 2043 I IF Applicability of PS dists of to MA PDU session If 6.5.0 2020-66 CT4886 CP-201113 2044 I F Missing condition for inclusion of "NISSA to be performer" If 6.5.0 2020-66 CT4886 CP-201102 2044 I F Correction of the handling of SGMM couse #27 If 6.5.0 2020-66 CT4886 CP-201102 2054 I F Slopping of T3486 differ receiving the NISSA command If 6.5.0 2020-66 CT4886 CP-201102 2051 I F Slopping of T3486 differ receiving the NISSA included in requested If 6.5.0 2020-66 CT4886 CP-201103 2065 I C Calification on the rejected SN-SSA included in requested If 6.5.0 2020-66 CT4886 CP-201113 2065 I ANDPS in a disarchine theory Silos SNA for Coarning scenarios If 6.5.0 2020-66 CT4886 CP-201113 2069 I F Definition of registered SNPN If 6.5.0 2020-66 CT4886	2020-06	CT#88e	CP-201113	2040	1	F	Pending NSSAI update for the configured NSSAI in the UCU	16.5.0
2020-06 CT#886 CP-201113 2043 I F Missing condition for inclusion of "NSSAA to be performed" 16.5.0 2020-06 CT#886 CP-20112 2047 F Correction of the handing of GMM cause w27 16.5.0 2020-06 CT#886 CP-20112 2051 F Correction of the handing of GMM cause w27 16.5.0 2020-06 CT#886 CP-20110 2051 I F Additional condition to start 754-0 16.5.0 2020-06 CT#886 CP-20110 2051 I F Additional condition to start 754-0 16.5.0 2020-06 CT#886 CP-20110 2056 I Additional condition to start 754-0 16.5.0 2020-06 CT#886 CP-20110 2056 I F Additional condition to start 754-0 16.5.0 2020-06 CT#886 CP-20110 2056 I F Additional condition to start 754-0 16.5.0 2020-06 CT#886 CP-20113 2059 I F Depolinin for Network Sile condin F 16.5.0	2020-06	CT#88e	CP-201109	2042	1	F	Applicability of PS data off to MA PDU session	16.5.0
Indicator Indicator Indicator Indicator 2020-06 CT4986 CP-20110 2044 I F AMF friggers PDU session release 16.5.0 2020-06 CT4986 CP-20110 2064 I F Correction of the handling of SGMM cause #27 16.5.0 2020-06 CT4986 CP-201102 2061 I F Specify UE behaviour for NOTFICATION message for 16.5.0 2020-06 CT4986 CP-201103 2063 I F Specify UE behaviour for NOTFICATION message for 16.5.0 2020-06 CT4986 CP-201103 2065 I AMDPS has to supported type. 16.5.0 2020-06 CT4986 CP-201113 2065 I AMDPS has to supported type. 16.5.0 2020-06 CT4986 CP-201113 2063 I I Definition of registered SNPN 16.5.0 2020-06 CT4986 CP-201136 2061 I I Definition of registered SNPN 16.5.0 2020-06 CT4986 CP-201136 20651 I	2020-06	CT#88e	CP-201113	2043	1	F	Missing condition for inclusion of "NSSAA to be performed"	16.5.0
20206 CT486 CP-201102 2047 I Correction of the handing of GGMA cause #27 16.5.0 20206 CT486 CP-201102 2050 2 FA Additional condition to start T3540 16.5.0 2020.06 CT4886 CP-201102 2050 2 FA Additional condition to start T3540 16.5.0 2020.06 CT4886 CP-201102 2051 1 F Specify UE baharoor for NOTIFICATION message for 16.5.0 2020.06 CT4886 CP-201103 2065 1 AMDSP is not supported by 50-RG and W-AGF 16.5.0 2020.06 CT4886 CP-201113 2059 3 C Updating regularenests of NSSAA for roaming scenarios 16.5.0 2020.06 CT4886 CP-201113 2059 3 C Updating regularenests of NSSAA for roaming scenarios 16.5.0 2020.06 CT4886 CP-201108 2061 I Derinition of registration proceed SINPN 16.5.0 2020.06 CT4886 CP-201108 2061 I Derintroamon fissic 16.5.0	2020.06	CT#00a	CD 201112	2044	4	E	indicator	16 5 0
202006 CT#886 CP-201113 2049 F Stopping of T3346 after receiving the NSSA Command 16.5.0 2020-06 CT#886 CP-201112 2060 2 F Additional concilion to start T3540 16.5.0 2020-06 CT#886 CP-201102 2061 F Additional concilion to start T3540 16.5.0 2020-06 CT#886 CP-201102 2065 I F Additional concilion to start T3540 16.5.0 2020-06 CT#886 CP-201108 2055 I Additional state/sub-states NSSAI in registration procedure. 16.5.0 2020-06 CT#886 CP-201113 2055 I Califying the description for Nelwork Side of CPSR 16.5.0 2020-06 CT#886 CP-201113 2050 I Clarifying the description for Nelwork Side of CPSR 16.5.0 2020-06 CT#886 CP-20113 2056 I Currention of SSC 16.5.0 2020-06 CT#886 CP-20108 2062 I Correction of SSC 16.5.0 2020-06	2020-06	CT#88e	CP-201113	2044	1	F	AMF triggers PDU session release	16.5.0
Chroso CP-20110 Carlo P Additional contains resuming in FLOR comman 16.5.0 2020-06 CT#88e CP-201102 2061 I F Additional contains resuming and FLOR Comman 16.5.0 2020-06 CT#88e CP-201103 2065 I F Clarification on the rejected S-NSAI included in requested 16.5.0 2020-06 CT#88e CP-201103 2055 F ANDSP is not supported by SG-RG and W-AGF 16.5.0 2020-06 CT#88e CP-201113 2059 3 C Updating requirements of NSSAA for rearring scenarios 16.5.0 2020-06 CT#88e CP-20106 2060 1 E Enregreency PDU session scabolished after WUS negotiation 16.5.0 2020-06 CT#88e CP-20106 2063 1 E Enregreency PDU session scabolished after WUS negotiation 16.5.0 2020-06 CT#88e CP-201136 2065 F tepportely and active and and active and	2020-00	CT#880	CP-201102	2047		Г F	Stopping of T3346 after receiving the NSSA Command	16.5.0
2020-06 CT#88e (P-201102) 2051 I F Additional condition to start T3540 16.5.0 2020-06 CT#88e (P-201102) 2051 I F Specify UE behaviour CN ONTFICATION message for additional starts/sub-states 16.5.0 2020-06 CT#88e (P-201103) 2055 F ADDITA 16.5.0 2020-06 CT#88e (P-201108) 2055 F ADDITA 16.5.0 2020-06 CT#88e (P-201108) 2055 F ADDITA 16.5.0 2020-06 CT#88e (P-201113) 2059 3 C Adding a new abnormal case on the network side for CPER 16.5.0 2020-06 CT#88e (P-201113) 2050 3 F Clarifying requirements of NSA for roaming scenarios 16.5.0 2020-06 CT#88e (P-201108) 2064 F update the counter for SNPN 16.5.0 2020-06 CT#88e (P-201104) 2067 F 500MI cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06	2020 00	01#000	01 201113	2045		1	message	10.0.0
2020-06 CT#88e CP-201102 2051 1 F Specify UE behaviour for NOTIFICATION mesage for additional state/substates 16.5.0 2020-06 CT#88e CP-201103 2063 1 F Clarification on the rejected S-NSSA included in requested 16.5.0 2020-06 CT#88e CP-201108 2055 F A NDSP is not supported by 5G-RG and W-AGF 16.5.0 2020-06 CT#88e CP-201113 2058 3 F Clarifying the description for Network side for CPSR 16.5.0 2020-06 CT#88e CP-20118 2060 1 F Definition or registered SNPN 16.5.0 2020-06 CT#88e CP-20108 2063 1 F Dergency PDU session established ater WUS negotation 16.5.0 2020-06 CT#88e CP-201138 2065 1 tupdate of the counter for SNPN 16.5.0 2020-06 CT#88e CP-201138 2065 1 totage of counter for UE in SNPN 16.5.0 2020-06 CT#88e CP-201138 2068 F SGMM cause value #74 in an SNPN wit	2020-06	CT#88e	CP-201102	2050	2	F	Additional condition to start T3540	16.5.0
2020-06 CT#886 CP-201113 2053 1 F Clarification on the rejected S-NSSAI included in requested 16.5.0 2020-06 CT#886 CP-201068 2055 F ANDSP in not supported by SG-RG and W-AGF 16.5.0 2020-06 CT#886 CP-20108 2058 3 C Adding a new abnormal case on the network side for CPSR 16.5.0 2020-06 CT#886 CP-20113 2058 3 C Updating requirements of NSAA for reaming scenarios 16.5.0 2020-06 CT#886 CP-20118 2066 1 F Definition of registered SNFN 16.5.0 2020-06 CT#886 CP-201138 2064 1 F Definition of registered SNFN 16.5.0 2020-06 CT#886 CP-201138 2065 F uterporally and permanently forbidden SNPNs lists per access 16.5.0 2020-06 CT#886 CP-201138 2067 1 F 6G.01T dSNPN 16.5.0 2020-06 CT#886 CP-201138 2067 1 F 6G.01T dSNPN <td>2020-06</td> <td>CT#88e</td> <td>CP-201102</td> <td>2051</td> <td>1</td> <td>F</td> <td>Specify UE behaviour for NOTIFICATION message for</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201102	2051	1	F	Specify UE behaviour for NOTIFICATION message for	16.5.0
Carbon Onesa CP-20108 2005 I NSSA in registration procedure. Minutate in Head All 2020-06 CT#886 CP-20108 2005 I Adding a new abnormal case on the network side for CPSR 16.5.0 2020-06 CT#886 CP-201113 2058 3 C Adding a new abnormal case on the network side for CPSR 16.5.0 2020-06 CT#886 CP-201131 2059 3 C Catrifying the description for Network Side for CPSR 16.5.0 2020-06 CT#886 CP-20108 2069 I P Definition of registrated SNPN 16.5.0 2020-06 CT#886 CP-20108 2063 I E Emergency PDU session established after VUS negotiation 16.5.0 2020-06 CT#886 CP-201136 2064 F update of the counter for SNPN 16.5.0 2020-06 CT#886 CP-201136 2066 I a storage of counter for SNPN 16.5.0 2020-06 CT#886 CP-201136 2069 I SGMM decase value #14 na SNPN with a globall	2020-06	CT#88e	CP-201113	2053	1	F	additional state/sub-states	1650
2202-06 CT#886 CP-20108 2055 F ANDSP is not supported by SC+G and W-AGF 16.5.0 2020-06 CT#886 CP-20113 2058 3 F Clarifying the description for Network Silee Specific 16.5.0 2020-06 CT#886 CP-20113 2059 3 C Updating requirements of NSSAA for roaming scenarios 16.5.0 2020-06 CT#886 CP-20138 2069 1 F Definition of registrated SNPN 16.5.0 2020-06 CT#886 CP-20138 2064 I Updating requirements of NSPN 16.5.0 2020-06 CT#886 CP-20138 2064 I Update of the counter for SNPN 16.5.0 2020-06 CT#886 CP-20138 2067 I F Storage of counters for UE in SNPN 16.5.0 2020-06 CT#886 CP-20138 2067 I F SGMM cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06 CT#886 CP-20113 2067 I F SGMM cause value #71 and SNPN with a globally-unique <td>2020-00</td> <td>C1#00e</td> <td>CF-201113</td> <td>2055</td> <td> </td> <td>1</td> <td>NSSAI in registration procedure.</td> <td>10.5.0</td>	2020-00	C1#00e	CF-201113	2055		1	NSSAI in registration procedure.	10.5.0
2202-06 C1#886 CP-20196 2065 3 C Adding a new abnormal case on the network slice Specific 16.5.0 2020-06 C1#886 CP-201113 2065 3 F Clerinying the description for Network Slice-Specific 16.5.0 2020-06 C1#886 CP-201136 2060 1 F Definition of registered SNPN 16.5.0 2020-06 C1#886 CP-201136 2064 F Definition of registered SNPN 16.5.0 2020-06 C1#886 CP-20136 2066 F temporally and permanently forbidder SNPN lists per access 16.5.0 2020-06 C1#886 CP-20136 2066 F storage of counters for UE in SNPN 16.5.0 2020-06 C1#886 CP-20136 2066 F SG GUT of SNPN 16.5.0 2020-06 C1#886 CP-20136 2069 F SGMM cause value #13 not supporting roaming for SNPN 16.5.0 2020-06 C1#886 CP-20113 2070 F Clarification or for GM and sulue #13 not supporting roaming for SNPN 16.5.0 <td>2020-06</td> <td>CT#88e</td> <td>CP-201108</td> <td>2055</td> <td></td> <td>F</td> <td>ANDSP is not supported by 5G-RG and W-AGF</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201108	2055		F	ANDSP is not supported by 5G-RG and W-AGF	16.5.0
2120-06 CT#886 CP-201113 2058 3 F Clarifying the description for Network Slice-Specific 16.5.0 2020-06 CT#886 CP-20113 2069 3 C Updating requirements of NSSAA for rearning scenarios 16.5.0 2020-06 CT#886 CP-20106 2062 F Correction of SGC 16.5.0 2020-06 CT#886 CP-201136 2064 F update of the counter for SNPN 16.5.0 2020-06 CT#886 CP-201136 2066 F temporatily and permanently forbiidden SNPNs lists per access 16.5.0 2020-06 CT#886 CP-201136 2066 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#886 CP-201136 2068 F SGMM cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06 CT#886 CP-201132 2071 C F SGMM cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06 CT#886 CP-201102 2071 C F SGMM cause value #74 in an SNPN with a globally-uniq	2020-06	CT#88e	CP-201096	2056	3	С	Adding a new abnormal case on the network side for CPSR	16.5.0
AutonOctation Revocation AutonOctation	2020-06	CT#88e	CP-201113	2058	3	F	Clarifying the description for Network Slice-Specific	16.5.0
2220-06 CT#886 CP-20113 2060 1 F Definition of registered SNPN 16.5.0 2202-06 CT#886 CP-20108 2060 F Correction of SGC 16.5.0 2202-06 CT#886 CP-201136 2064 F Update of the counter for SNPN 16.5.0 2202-06 CT#886 CP-201136 2064 F update of the counter for SNPN 16.5.0 2202-06 CT#886 CP-201136 2065 F terrogency PDU sesseion established after WUS negotiation 16.5.0 2202-06 CT#886 CP-201136 2066 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#886 CP-201136 2066 F SGMM cause value #13 not supporting roaming for SNPN 16.5.0 2020-06 CT#886 CP-201132 2071 Z F storage of counters for UE in PLMN 16.5.0 2020-06 CT#886 CP-201132 2074 L D Editorial corroctons 16.5.0 2020-06 CT#886 CP-201113	2020.06	CT#990	CP 201112	2050	2	C	Authorization Revocation	1650
CT#88e CP-201096 2062 F Correction of SGC Correction of SGC 2020-06 CT#88e CP-201086 2063 I F Emergency PDU session established after WUS negotiation 16.5.0 2020-06 CT#88e CP-201138 2066 F temporarily and permanently forbidden SNPNs lists per access 16.5.0 2020-06 CT#88e CP-201138 2066 3 F atorage of counter for SNPN 16.5.0 2020-06 CT#88e CP-201138 2066 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#88e CP-201138 2067 F SGMM cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06 CT#88e CP-201102 2072 F Clarification of the foure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2072 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2072 F Clarification of the figure of registration arcedure figistration and fiels.0 16.5.0	2020-00	CT#88e	CP-201136	2059	3	F	Definition of registered SNPN	16.5.0
2020-06 CT#886 CP-201036 2063 1 F Emergency PDU sesseion established after WUS negotiation 16.5.0 2020-06 CT#886 CP-201136 2064 F update of the counter for SNPN 16.5.0 2020-06 CT#886 CP-201136 2065 F temporarily and permanently forbidden SNPNs lists per access 16.5.0 2020-06 CT#886 CP-201136 2066 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#886 CP-201136 2068 F SGMM cause value #174 in an SNPN with a globally-unique 16.5.0 2020-06 CT#886 CP-201132 2071 F SGMM cause value #13 not supporting roaming for SNPN 16.5.0 2020-06 CT#886 CP-201122 2071 F storage of counters for UE in PLMN 16.5.0 2020-06 CT#886 CP-201102 2071 F Claiffication of the figure of registration procedure 16.5.0 2020-06 CT#886 CP-201112 2077 T C Cenestriction storaget counting period registration and the figure of	2020-06	CT#88e	CP-201096	2000	1	F	Correction of SGC	16.5.0
2020-06 CT#88e CP-201136 2005 F update of the counter for SNPN 16.5.0 2020-06 CT#88e CP-201136 2005 F temporanity and permanently forbidden SNPNs lists per access 16.5.0 2020-06 CT#88e CP-201136 2066 3 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F SGUMT of SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F SGMM cause value #14 in an SNPN with a globally-unique 16.5.0 2020-06 CT#88e CP-201132 2071 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2072 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2077 1 C Generic UE configuration update trigger for registration and tes.0 16.5.0 2020-06 CT#88e CP-20113 2075 F T3540 is not startred if the Registration Accept includes a neming NSAL <	2020-06	CT#88e	CP-201096	2063	1	F	Emergency PDL sesseion established after WLIS negotiation	16.5.0
2020-06 CT#88e CP-201136 2065 F temporanity and permanently forbidden SNPNs lists per access 16.5.0 2020-06 CT#88e CP-201136 2067 I F 5G GUTI of SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F SGMM cause value #71 in an SNPN with a globally-unique 16.5.0 2020-06 CT#88e CP-201136 2067 F SGMM cause value #71 in an SNPN with a globally-unique 16.5.0 2020-06 CT#88e CP-201136 2070 F Clarification of the cause of start of T3550 16.5.0 2020-06 CT#88e CP-201102 2071 F Starage of counters for UE in PLMN 16.5.0 2020-06 CT#88e CP-201102 2074 D Edianization corrections 16.5.0 2020-06 CT#88e CP-201102 2077 F Clarification charage 16.5.0 2020-06 CT#88e CP-201113 2075 F T3540 is not started if the Registration necedure 16.5.0 2020-06 CT#88e CP-201113 20771	2020-06	CT#88e	CP-201136	2000	1	F	update of the counter for SNPN	16.5.0
Type Type <th< td=""><td>2020-06</td><td>CT#88e</td><td>CP-201136</td><td>2065</td><td></td><td>F</td><td>temporarily and permanently forbidden SNPNs lists per access</td><td>16.5.0</td></th<>	2020-06	CT#88e	CP-201136	2065		F	temporarily and permanently forbidden SNPNs lists per access	16.5.0
2020-06 CT#88e CP-201136 2066 3 F storage of counters for UE in SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F SG QUTI of SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F SG MM cause value #74 in an SNPN with a globally-unique 16.5.0 2020-06 CT#88e CP-201136 2070 F Clarification of the cause of start of T3550 16.5.0 2020-06 CT#88e CP-201102 2071 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2071 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201113 2075 F T3540 is not started if the Registration Accept includes a pending NSSAI 16.5.0 2020-06 CT#88e CP-201113 2079 I F Clarification change 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>type</td> <td></td>							type	
2020-06 CT#88e CP-201136 2067 1 F 5G GUTI of SNPN 16.5.0 2020-06 CT#88e CP-201136 2068 F 5GMM cause value #14 in an SNPN with a globally-unique SNPN identity 16.5.0 2020-06 CT#88e CP-201102 2071 F SGMM cause value #13 not supporting reaming for SNPN 16.5.0 2020-06 CT#88e CP-201102 2071 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2074 D Editorial correction of the figure of registration Accept includes a pending NSSAI 2020-06 CT#88e CP-201172 2076 F T3540 is not started if the Registration Accept includes a pending NSSAI 2020-06 CT#88e CP-201172 2078 I F RAc S parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201172 2078 I F RAc S parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201132 2078 I F Rain	2020-06	CT#88e	CP-201136	2066	3	F	storage of counters for UE in SNPN	16.5.0
2020-06 CT#88e CP-201136 2068 F SGMM cause value #74 in an SNPN with a globally-unique SNPN identity 16.5.0 2020-06 CT#88e CP-201136 2069 2 F SGMM cause value #13 not supporting roaming for SNPN 16.5.0 2020-06 CT#88e CP-201102 2071 2 F Storage of counters for UE in PLMN 16.5.0 2020-06 CT#88e CP-201102 2072 F Clarification of the figure of registration procedure 16.5.0 2020-06 CT#88e CP-201102 2074 1 D Editorial corrections 16.5.0 2020-06 CT#88e CP-20113 2075 F T3840 is not started if the Registration Accept includes a 16.5.0 2020-06 CT#88e CP-20113 2079 1 F Clarify that NSSA4 can occur during periodic registration or field.0 16.5.0 2020-06 CT#88e CP-201113 2079 1 F Clarify that NSSA4 can occur during periodic registration or field.0 16.5.0 2020-06 CT#88e CP-201113 2080 <td>2020-06</td> <td>CT#88e</td> <td>CP-201136</td> <td>2067</td> <td>1</td> <td>F</td> <td>5G GUTI of SNPN</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201136	2067	1	F	5G GUTI of SNPN	16.5.0
2020-06 CT#88e CP-201136 2069 2 F SGMM cause value #13 not supporting roaming for SNPN 16.5.0 2020-06 CT#88e CP-201102 2071 2 F Storage of counters for UE in PLMN 15.5.0 2020-06 CT#88e CP-201102 2074 1 D Editorial corrections 16.5.0 2020-06 CT#88e CP-201102 2074 1 D Editorial corrections 16.5.0 2020-06 CT#88e CP-201102 2074 1 D Editorial corrections 16.5.0 2020-06 CT#88e CP-201096 2077 1 C Generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2079 1 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2080 F F ixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2081 F Adignment of UE actions of rejected NSSAI for the failed or revoked NSSAI <td>2020-06</td> <td>CT#88e</td> <td>CP-201136</td> <td>2068</td> <td></td> <td>F</td> <td>5GMM cause value #74 in an SNPN with a globally-unique</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201136	2068		F	5GMM cause value #74 in an SNPN with a globally-unique	16.5.0
Construction Construction<	2020-06	CT#88e	CP-201136	2069	2	F	5GMM cause value #13 not supporting roaming for SNPN	1650
Line Cite Cite <thcite< th=""> Cite Cite <thc< td=""><td>2020-00</td><td>CT#88e</td><td>CP-201136</td><td>2003</td><td>2</td><td>F</td><td>Clarification of the cause of start of T3550</td><td>16.5.0</td></thc<></thcite<>	2020-00	CT#88e	CP-201136	2003	2	F	Clarification of the cause of start of T3550	16.5.0
Classical Structure	2020-00	CT#88e	CP-201102	2070	2	F	storage of counters for LIF in PLMN	16.5.0
Correct Correction Construction Construction Construction Construction 2020-06 CT#88e CP-201102 2074 1 D Editorial corrections 16.5.0 2020-06 CT#88e CP-201102 2077 1 C Generic UE configuration Accept includes a period of the Registration and the Registration and the Registration and the Registration and the Registration change 2020-06 CT#88e CP-201127 2078 1 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2081 1 F Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2081 1 F Exception to initiate the service request procedure during the SA ddition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e <td>2020-06</td> <td>CT#88e</td> <td>CP-201102</td> <td>2072</td> <td>2</td> <td>F</td> <td>Clarification of the figure of registration procedure</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201102	2072	2	F	Clarification of the figure of registration procedure	16.5.0
Correct CP-201113 CP-201113 CP-201127 F T3540 is not started if the Registration Accept includes a pending NSSAI 16.5.0 2020-06 CT#886 CP-201127 2078 I F T3540 is not started if the Registration Accept includes a pending NSSAI 16.5.0 2020-06 CT#886 CP-201127 2078 I F RACS parameters in generic UE configuration procedure EC Restriction change 16.5.0 2020-06 CT#886 CP-201113 2080 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#886 CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#886 CP-201113 2081 F Alignment of UE actions of rejected NSSAI for the failed or revoked NSSAA 16.5.0 2020-06 CT#886 CP-201113 2089 1 F Exception to initiate the service request procedure during NSSAA when there is no allowed NSSAI 2020-06 CT#886 CP-201102 2093 1 F Add handling for parameters set to "value is not used" in 5GS 16.5.0 2020-06	2020-06	CT#88e	CP-201102	2074	1	D	Editorial corrections	16.5.0
2020-06 CT#88e CP-201096 2077 1 C gending NSSAI 2020-06 CT#88e CP-201127 2078 1 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2079 1 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2084 1 F Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2089 1 F Exception to initiate the service request procedure during the sale of the s	2020-06	CT#88e	CP-201113	2075		F	T3540 is not started if the Registration Accept includes a	16.5.0
2020-06 CT#88e CP-201096 2077 1 C Generic UE configuration update trigger for registration and EC Restriction change 16.5.0 2020-06 CT#88e CP-201112 2078 1 F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2084 1 F Alignment of UE actions of rejected NSSAI for the failed or revoked NSSAA 16.5.0 2020-06 CT#88e CP-201113 2081 1 F Exception to initiate the service request procedure during NSSAA when there is no allowed NSSAI 2020-06 CT#88e CP-201113 2081 F Addition of CAG information registration reject due to no available silces 16.5.0 2020-06 CT#88e CP-201102 2093 F Add handling for parameters set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2093 F Add handling for parameters set to "value is not used" in 5GS 16.5.0 2020-06 <							pending NSSAI	
CP-201127 2078 I F RACS parameters in generic UE configuration procedure 16.5.0 2020-06 CT#88e CP-201113 2079 1 F Clarify that NSSAA can occur during periodic registration or mobility updating for NB-N1 mode UEs 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2084 1 F Alignment of UE actions of rejected NSSAI for the failed or revoked NSSAA 2020-06 CT#88e CP-201113 2087 1 C Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2089 1 F Exception to initiate the service request procedure during NSSAA when there is no allowed NSSAI 2020-06 CT#88e CP-201113 2091 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameters included by AMF during inter-system change from S1 mode to N1 mode in SGMM-CONNECTED mode 16.5.0	2020-06	CT#88e	CP-201096	2077	1	С	Generic UE configuration update trigger for registration and	16.5.0
2020-06 CT#2012/2 2079 1 F CRCCD parameters in RSSA can occur during periodic registration or mobility updating for NB-N1 mode UEs 16.5.0 2020-06 CT#88e CP-201113 2080 F Fixing typo related to RSSA can occur during periodic registration or mobility updating for NB-N1 mode UEs 16.5.0 2020-06 CT#88e CP-201113 2084 1 F Alignment of UE actions of rejected NSSAI for the failed or revoked NSSAA 2020-06 CT#88e CP-20113 2087 1 C Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2089 1 F Exception to initiate the service request procedure during slices 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct parameters included by AMF during inter-system change fr	2020-06	CT#88e	CP-201127	2078	1	F	EC Restriction change	1650
Description Difference Difference <thdifferenc< th=""> Differenc Differenc</thdifferenc<>	2020-00	CT#88e	CP-201127	2070	1	F	Clarify that NSSAA can occur during periodic registration or	16.5.0
2020-06 CT#88e CP-201113 2080 F Fixing typo related to eNS 16.5.0 2020-06 CT#88e CP-201113 2084 1 F Alignment of UE actions of rejected NSSAI for the failed or revoked NSSAA 16.5.0 2020-06 CT#88e CP-201136 2087 1 C Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2089 1 F Exception to initiate the service request procedure during NSSAA when there is no allowed NSSAI 2020-06 CT#88e CP-201102 2093 1 F Addition of CAG information reject due to no available slices 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Correct UE behavior for receiving SGMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct UE behavior for receiving SGMM-CONNECTED mode 16.5.0 2020-06 CT#88e CP-201102 2096 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 <	2020 00	01//000	01 201110	2010	'	1	mobility updating for NB-N1 mode UEs	10.0.0
2020-06 CT#88e CP-201113 2084 1 F Alignment of UB actions of rejected NSSAI for the failed or revoked NSSAA 16.5.0 2020-06 CT#88e CP-201136 2087 1 C Addition of CAG information list in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2089 1 F Exception to initiate the service request procedure during NSSAA when there is no allowed NSSAI 16.5.0 2020-06 CT#88e CP-201113 2091 1 F Missing condition at registration reject due to no available slices 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2096 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 CT#88e CP-201102 2098 F Applicable URSP is not optional for a UE 16.5.0	2020-06	CT#88e	CP-201113	2080		F	Fixing typo related to eNS	16.5.0
Image: Construct of the construction of the	2020-06	CT#88e	CP-201113	2084	1	F	Alignment of UE actions of rejected NSSAI for the failed or	16.5.0
2020-06 CT#88e CP-20113 2081 I C Addition of CAS information fish in registration reject message. 16.5.0 2020-06 CT#88e CP-201113 2091 I F Exception to initiate the service request procedure during information reject fue to no available slices 16.5.0 2020-06 CT#88e CP-201102 2093 I F Addition of or parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2093 I F Add handing for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201102 2093 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2096 F Correct parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode 16.5.0 2020-06 CT#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201102 2109 2 F Inclusion of NSAI in AN Parameters for non-3GPP access	2020.06	CT#00a	CD 201126	2007	4	C	revoked NSSAA	10 5 0
2020-06 CT#88e CP-201113 2091 1 F Exception initiate instance soluced uses proceedure during instance in a soluced NSSAI 2020-06 CT#88e CP-201113 2091 1 F Missing condition at registration reject due to no available slices 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-20102 2094 1 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2096 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 CT#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06	2020-06	CT#88e	CP-201136	2087	1	C E	Addition of CAG information list in registration reject message.	16.5.0
2020-06 CT#88e CP-201113 2091 1 F Missing condition at registration reject due to no available slices 16.5.0 2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201096 2094 1 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode 16.5.0 2020-06 CT#88e CP-201102 2098 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 CT#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0	2020-06	C1#00e	CP-201113	2069	1	г	INSSAA when there is no allowed NSSAI	10.5.0
SlicesSlices2020-06CT#88eCP-20110220931FAdd handling for parameter set to "value is not used" in 5GS16.5.02020-06CT#88eCP-20109620941FCorrect UE behavior for receiving 5GMM cause #31 in 5GS16.5.02020-06CT#88eCP-2011022095FCorrect parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED16.5.02020-06CT#88eCP-2011022096FRemove invalid cases in error handling for QoS rule operation and TFT operation16.5.02020-06CT#88eCP-20110220982FApplicable URSP is not optional for a UE16.5.02020-06CT#88eCP-20110221011FAdditional QoS error handling related to mapped EBI16.5.02020-06CT#88eCP-20110221021FPS Data Off status report for non-3GPP access16.5.02020-06CT#88eCP-20110221031DUnify terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CloT16.5.02020-06CT#88eCP-20110221041FCorrecter Scale16.5.02020-06CT#88eCP-20110221063FAvoid repeated redirection for CloT16.5.0 <td>2020-06</td> <td>CT#88e</td> <td>CP-201113</td> <td>2091</td> <td>1</td> <td>F</td> <td>Missing condition at registration reject due to no available</td> <td>16.5.0</td>	2020-06	CT#88e	CP-201113	2091	1	F	Missing condition at registration reject due to no available	16.5.0
2020-06 CT#88e CP-201102 2093 1 F Add handling for parameter set to "value is not used" in 5GS 16.5.0 2020-06 CT#88e CP-201096 2094 1 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode 16.5.0 2020-06 CT#88e CP-201102 2096 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 CT#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0							slices	
2020-06 C1#88e CP-201096 2094 1 F Correct UE behavior for receiving 5GMM cause #31 in 5GS 16.5.0 2020-06 CT#88e CP-201102 2095 F Correct parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode 16.5.0 2020-06 CT#88e CP-201102 2096 F Remove invalid cases in error handling for QoS rule operation and TFT operation 16.5.0 2020-06 CT#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Avoid repeated redirection for CloT 16.5.0 2020-06	2020-06	CT#88e	CP-201102	2093	1	F	Add handling for parameter set to "value is not used" in 5GS	16.5.0
2020-06C1#88eCP-2011022095FCorrect parameters included by AMF during inter-system change from S1 mode to N1 mode in 5GMM-CONNECTED mode16.5.02020-06CT#88eCP-2011022096FRemove invalid cases in error handling for QoS rule operation and TFT operation16.5.02020-06CT#88eCP-20110220982FApplicable URSP is not optional for a UE16.5.02020-06CT#88eCP-20110221011FAdditional QoS error handling related to mapped EBI16.5.02020-06CT#88eCP-20110221021FPS Data Off status report for non-3GPP access16.5.02020-06CT#88eCP-20110221031DUnify terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CIoT16.5.02020-06CT#88eCP-20109621072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming scenerios16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.	2020-06	CT#88e	CP-201096	2094	1	F	Correct UE behavior for receiving 5GMM cause #31 in 5GS	16.5.0
ConstructionCP-2011022096FRemove invalid cases in error handling for QoS rule operation and TFT operation16.5.02020-06CT#88eCP-20110220982FApplicable URSP is not optional for a UE16.5.02020-06CT#88eCP-20110221092FInclusion of NSSAI in AN Parameters for non-3GPP access16.5.02020-06CT#88eCP-20110221011FAdditional QoS error handling related to mapped EBI16.5.02020-06CT#88eCP-20110221021FPS Data Off status report for non-3GPP access16.5.02020-06CT#88eCP-20110221031DUnify terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CIoT16.5.02020-06CT#88eCP-20110221072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming scenerios16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address allocation16.5.0	2020-06	CT#88e	CP-201102	2095		F	Correct parameters included by AMF during inter-system	16.5.0
2020-06CT#88eCP-2011022096FRemove invalid cases in error handling for QoS rule operation16.5.02020-06CT#88eCP-20110220982FApplicable URSP is not optional for a UE16.5.02020-06CT#88eCP-20114421002FInclusion of NSSAI in AN Parameters for non-3GPP access16.5.02020-06CT#88eCP-20110221011FAdditional QoS error handling related to mapped EBI16.5.02020-06CT#88eCP-20110221021FPS Data Off status report for non-3GPP access16.5.02020-06CT#88eCP-20110221031DUnify terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CIoT16.5.02020-06CT#88eCP-20110221072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming 16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.0							mode	
and TFT operation2020-06CT#88eCP-20110220982FApplicable URSP is not optional for a UE16.5.02020-06CT#88eCP-20114421002FInclusion of NSSAI in AN Parameters for non-3GPP access16.5.02020-06CT#88eCP-20110221011FAdditional QoS error handling related to mapped EBI16.5.02020-06CT#88eCP-20110221021FPS Data Off status report for non-3GPP access16.5.02020-06CT#88eCP-20110221031DUnify terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20110221063FAvoid repeated redirection for CIoT16.5.02020-06CT#88eCP-20109621072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.0	2020-06	CT#88e	CP-201102	2096		F	Remove invalid cases in error handling for QoS rule operation	16.5.0
2020-06 C I#88e CP-201102 2098 2 F Applicable URSP is not optional for a UE 16.5.0 2020-06 CT#88e CP-201144 2100 2 F Inclusion of NSSAI in AN Parameters for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Network triggered service request procedure over non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Network triggered service request procedure over non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2106 3 F Avoid repeated redirection for CloT 16.5.0 2020-06 CT#88e		0.5.1					and TFT operation	10 - 1
2020-06 C1#88e CP-201144 2100 2 F Inclusion of NSSAI in AN Parameters for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2101 1 F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Network triggered service request procedure over non-3GPP 16.5.0 2020-06 CT#88e CP-201096 2106 3 F Avoid repeated redirection for CloT 16.5.0 2020-06 CT#88e CP-201096 2107 2 F PDU session release due to CP only revocation 16.5.0 2020-06 CT#88e CP-201102 2107 2 F PDU session release due to CP only revocation 16.5.0 2020-06 CT#88e CP-201113	2020-06	CT#88e	CP-201102	2098	2	F	Applicable URSP is not optional for a UE	16.5.0
2020-06 C I#88e CP-201102 2101 I F Additional QoS error handling related to mapped EBI 16.5.0 2020-06 CT#88e CP-201102 2102 1 F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Network triggered service request procedure over non-3GPP 16.5.0 2020-06 CT#88e CP-201096 2106 3 F Avoid repeated redirection for CloT 16.5.0 2020-06 CT#88e CP-201096 2107 2 F PDU session release due to CP only revocation 16.5.0 2020-06 CT#88e CP-201113 2108 3 C Updating Rejected NSSAI IE for failed NSSAA case in roaming 16.5.0 2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0	2020-06	CT#88e	CP-201144	2100	2	F	Inclusion of NSSAI in AN Parameters for non-3GPP access	16.5.0
2020-06 CT#88e CP-201102 2102 I F PS Data Off status report for non-3GPP access 16.5.0 2020-06 CT#88e CP-201102 2103 1 D Unify terms network-initiated and network-requested 16.5.0 2020-06 CT#88e CP-201102 2104 1 F Network triggered service request procedure over non-3GPP 16.5.0 2020-06 CT#88e CP-201096 2106 3 F Avoid repeated redirection for CloT 16.5.0 2020-06 CT#88e CP-201096 2107 2 F PDU session release due to CP only revocation 16.5.0 2020-06 CT#88e CP-201113 2108 3 C Updating Rejected NSSAI IE for failed NSSAA case in roaming 16.5.0 2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0	2020-06	CT#88e	CP-201102	2101	1	F	Additional QoS error handling related to mapped EBI	16.5.0
2020-06CT#88eCP-20110221031DUnity terms network-initiated and network-requested16.5.02020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CloT16.5.02020-06CT#88eCP-20109621072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming scenerios16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address allocation16.5.0	2020-06	CT#88e	CP-201102	2102	1	F	PS Data Off status report for non-3GPP access	16.5.0
2020-06CT#88eCP-20110221041FNetwork triggered service request procedure over non-3GPP access16.5.02020-06CT#88eCP-20109621063FAvoid repeated redirection for CloT16.5.02020-06CT#88eCP-20109621072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming scenerios16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address allocation16.5.0	2020-06	C1#88e	CP-201102	2103	1	D	Unity terms network-initiated and network-requested	16.5.0
2020-06CT#88eCP-20109621063FAvoid repeated redirection for CloT16.5.02020-06CT#88eCP-20109621072FPDU session release due to CP only revocation16.5.02020-06CT#88eCP-20111321083CUpdating Rejected NSSAI IE for failed NSSAA case in roaming16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.02020-06CT#88eCP-2011022110FConsider PDU session type IE set by UE in IP address16.5.0	2020-06	C1#88e	CP-201102	2104	1	г	Network triggered service request procedure over non-3GPP	16.5.0
2020-06 CT#88e CP-201096 2107 2 F PDU session release due to CP only revocation 16.5.0 2020-06 CT#88e CP-201113 2108 3 C Updating Rejected NSSAI IE for failed NSSAA case in roaming 16.5.0 2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0	2020-06	CT#88e	CP-201096	2106	3	F	Avoid repeated redirection for CIoT	16.5.0
2020-06 CT#88e CP-201113 2108 3 C Updating Rejected NSSAI IE for failed NSSAA case in roaming 16.5.0 2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0	2020-06	CT#88e	CP-201096	2107	2	F	PDU session release due to CP only revocation	16.5.0
Scenerios Scenerios 2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0	2020-06	CT#88e	CP-201113	2108	3	С	Updating Rejected NSSAI IE for failed NSSAA case in roaming	16.5.0
2020-06 CT#88e CP-201102 2110 F Consider PDU session type IE set by UE in IP address 16.5.0					-		scenerios	-
	2020-06	CT#88e	CP-201102	2110		F	Consider PDU session type IE set by UE in IP address	16.5.0

2020-06	CT#88e	CP-201116	2111	1	F	T3540 for service request for V2X communications	16.5.0
2020-06	CT#88e	CP-201096	2112	1	F	Clarification on the UE behaviour when receiving T3448	16.5.0
2020-06	CT#88e	CP-201096	2113	2	F	Connection Resumption for Notification	16.5.0
2020-06	CT#88e	CP-201096	2114	1	F	CloT user or small data container in CPSR message not	16.5.0
2020 00	01//000	01 201000	2114	1		forwarded	10.0.0
2020-06	CT#88e	CP-201099	2115	2	F	Initial Registration after 5G-SRVCC	16.5.0
2020-06	CT#88e	CP-201103	2118		F	Fixing a reference in the service request procedure	16.5.0
2020-06	CT#88e	CP-201103	2120	1	F	Add MFBR as mandatory parameter in GBR QoS flow	16.5.0
2020-06	CT#88e	CP-201103	2121	2	F	Initial registration for initiating emergency PDU session	16.5.0
2020-06	CT#88e	CP-201103	2122	2	F	Missing QoS flow description parameters for GBR QoS flows	16.5.0
2020 00	01//000	01 201100	2122	5	1	in 5GSM and ESM coordination	10.0.0
2020-06	CT#88e	CP-201196	2128	4	F	Sending CAG information list	16.5.0
2020-06	CT#88e	CP-201096	2130	1	F	Correction on terminology for the Control plane CIoT 5GS	16.5.0
				-		optimization	
2020-06	CT#88e	CP-201096	2132	1	F	Handling of PDU session and PDN connection associated with	16.5.0
						Control plane only indication in case of N26 based	
2020.06	CT#990	CP 201126	2124	1	Б	Interworking procedures	1650
2020-00	C1#00e	CF-201130	2134	1	Г	including 5GMM cause #76	10.5.0
2020-06	CT#88e	CP-201114	2135	2	F	NSSAA in an SNPN	1650
2020-06	CT#88e	CP-201139	2140	2	F	Correction in the UE behaviour upon failure of the procedures	16.5.0
2020 00	01//000	01 201100	2140	5	1	initiated for ESFB	10.0.0
2020-06	CT#88e	CP-201103	2141		F	No emergency session transfer after ESFB	16.5.0
2020-06	CT#88e	CP-201103	2142	2	F	Indication that the emergency services fallback attempt failed	16.5.0
2020-06	CT#88e	CP-201113	2144	1	F	Handling of Pending S-NSSAI	16.5.0
2020-06	CT#88e	CP-201096	2145	1	F	QoS error checks for UEs in NB-N1 mode	16.5.0
2020-06	CT#88e	CP-201136	2149	2	F	NB-IoT not applicable for SNPN	16.5.0
2020-06	CT#88e	CP-201096	2150	1	F	Retransmission of a CPSR message after integrity check	16.5.0
2020 00	01//000	01 201000	2.00	1	-	failure at the AMF	10.0.0
2020-06	CT#88e	CP-201136	2152	1	F	Miscellaneous clean-up for SNPN	16.5.0
2020-06	CT#88e	CP-201136	2153	2	F	Service area restrictions in an SNPN	16.5.0
2020-06	CT#88e	CP-201103	2154	1	D	Corrections on MICO	16.5.0
2020-06	CT#88e	CP-201136	2156	3	F	5GSM back-off mechanisms in an SNPN	16.5.0
2020-06	CT#88e	CP-201136	2157	1	F	UE in the 5GMM-REGISTERED ATTEMPTING-	16.5.0
2020 00	01//000	201100	2.07	'		REGISTRATION-UPDATE substate operating in SNPN	10.0.0
						access mode	
2020-06	CT#88e	CP-201136	2158	2	F	Routing indicator update in an SNPN	16.5.0
2020-06	CT#88e	CP-201136	2159	1	F	3GPP PS data off in an SNPN	16.5.0
2020-06	CT#88e	CP-201103	2161	1	F	Correction to conditions for including the S-NSSAI(s) from	16.5.0
						default NSSAI in the requested NSSAI	
2020-06	CT#88e	CP-201096	2162		F	Corrections to CR#1907	16.5.0
2020-06	CT#88e	CP-201096	2164	1	В	Signalling of EPS APN rate control parameters during PDU	16.5.0
	07/00	05.00/000		-	5	session establishment	
2020-06	C1#88e	CP-201283	2165	3	В	Ethernet header compression for CP Clo1 – 5GMM aspects	16.5.0
2020-06	CT#88e	CP-201109	2169	1	F	Editorial fix in 9.11.4	16.5.0
2020-06	CT#88e	CP-201096	2173		F	Acknowledgement of truncated 5G-S-TMSI configuration	16.5.0
2020-06	CT#88e	CP-201096	2174		F	NAS-MAC calculation for RRC connection reestablishment for	16.5.0
2020.06	CT#992	CP-201007	2175		F	IND-IUT OF OPTIMISATION Removal of Editor's Note for CP congestion control	1650
2020-00	CT#000	CP 201097	21/0	4	r E	Correction on WUS accistonee	10.0.0
2020-06		CP-201097	21/0	1	Г Б		10.5.0
2020-06		CP-201136	21//	1	Г		16.5.0
2020-06		CP-201136	21/8		Г Г		10.5.0
2020-06	C1#88e	CP-201136	21/9		F	Correction on 5GMM #27 for CAG	16.5.0
2020-06	C1#88e	CP-201114	2180	2	F	Clarification on handling of pending NSSAI	16.5.0
2020-06	CT#88e	CP-201114	2181		F	I erm on rejected NSSAI for the failed or revoked NSSAA	16.5.0
2020-06	CT#88e	CP-201103	2182	1	F	Single-registration mode without N26	16.5.0
2020-06	CT#88e	CP-201103	2183	1	F	Handling of unallowed SSC mode	16.5.0
2020-06	CT#88e	CP-201103	2184	1	F	UAC exception for emergency	16.5.0
2020-06	CT#88e	CP-201103	2185	3	F	MRU after SR for ESFB aborted	16.5.0
2020-06	CT#88e	CP-201103	2187	1	F	Correction to criteria to enter 5GMM-REGISTERED.UPDATE-	16.5.0
0000 07	07/02		0.100		P	NEEDED substate after resumption failure	10 5 6
2020-06	C1#88e	CP-201103	2192		F	Correcting that 5G NAS integrity key is one of the input	16.5.0
2020.06	CT#990	CP-201102	2102	2	F	Parameters for integrity protection algorithm	1650
2020-00	CT#000	CP 201103	2100	2	r. E	Correction to Handling of #21	16.5.0
2020-00	CT#000	CP 201103	2194	4	T E	Correction to bondling of T2447 times	16.5.0
2020-00		CF-201097	2195	1	r D		10.5.0
2020-06	C1#886	CP-20113/	2196	1	ע	Correct theregistration	0.5.0

2020-06	CT#88e	CP-201103	2197		F	De-registration before initial registration for Emergency Services	16.5.0
2020-06	CT#88e	CP-201103	2198	2	F	Handling of allowed NSSAI when the RA includes the TAI belonging to EPI MN	16.5.0
2020-06	CT#88e	CP-201103	2200		F	Corrections on NSSAI storage	16.5.0
2020-06	CT#88e	CP-201103	2202	1	F	UE behaviour when the UE receives the rejected NSSAI for the current RA and does not have a valid RA	16.5.0
2020-06	CT#88e	CP-201109	2203	1	В	Handlings of MA PDU session when deregistration from an	16.5.0
2020-06	CT#88e	CP-201103	2204		F	Indicate support of ePCO length of two octets parameter when establishing the PDU session – Alt#2	16.5.0
2020-06	CT#88e	CP-201103	2205	2	F	Addition of 5GSM cause #59	16.5.0
2020-06	CT#88e	CP-201103	2214	1	F	Handling of mapped EPS bearer contexts	16.5.0
2020-06	CT#88e	CP-201103	2215	1	F	Integrity check interworking in 5GMM-CONNECTED mode	16.5.0
2020-06	CT#88e	CP-201103	2216	1	F	Correction on LADN DNN based congestion control	16.5.0
2020-06	CT#88e	CP-201108	2218	1	В	Primary authentication of an N5GC device	16.5.0
2020-06	CT#88e	CP-201108	2219	•	F	Stop of enforcement of wireline access service area	16.5.0
			_			restrictions and forbidden wireline access area	
2020-06	CT#88e						16.5.0
2020-06	CT#88e	CP-201104	2221		F	Incorrect set up of PDN type in inter-system change from S1	16.5.0
2020-06	CT#88a	CP-201100	2222		F	Session-AMBR and MA PDU session	1650
2020-00	CT#880	CP-201109	2222	1	F	Introduction of ATSSS	16.5.0
2020-00	CT#880	CP-201109	2223	1	Г	"MA PDU request" when the LIE has an MA PDU session	16.5.0
2020-00	C1#00E	CF-201109	2224		Г	established over one access and requests establishment of	10.5.0
						user plane resources over the other access	
2020-06	CT#88e	CP-201104	2227	1	F	Mobile Terminated Voice Gap for MPS	16.5.0
2020-06	CT#88e	CP-201114	2228	1	F	Adding AAA-S via NSSAAF to support NSSAA	16.5.0
2020-06	CT#88e	CP-201311	2229	2	F	Resolve EN for Ciphering Key data IE regarding positioning	16.5.0
2020-06	CT#88e	CP-201104	2230	1	F	SIBs MICO indication needs to be included without Network Slicing	16.5.0
						Subscription Change Indication in UCU.	
2020-06	CT#88e	CP-201104	2231	2	F	Conditions for use of S-NSSAIs after receiving Rejected NSSAI Conditions for use of S-NSSAIs after receiving Rejected NSSAI	16.5.0
2020-06	CT#88e	CP-201114	2232	2	F	Default S-NSSAI not subject to network slice-specific authentication and authorization	16.5.0
2020-06	CT#88e	CP-201114	2233	3	F	Performing network slice-specific re-authentication and re- authorisation	16.5.0
2020-06	CT#88e	CP-201114	2234	2	F	Storage of pending NSSAI	16.5.0
2020-06	CT#88e	CP-201104	2235	1	F	UE stopping back-off timer when receiving PDU SESSION	16.5.0
2020-06	CT#88e	CP-201114	2236	1	F	Pending NSSAL and equivalent PLMNs	1650
2020-06	CT#88e	CP-201132	2237	7	F	Correcting the incorrect mode of the LIF	16.5.0
2020-06	CT#88e	CP-201127	2241	1	F	Avoiding too frequent registration procedures due to signalling	16.5.0
2020-06	CT#88e	CP-201119	2242	1	F	Unified access control is not applicable to a UE operating as	16.5.0
2020-06	CT#88e	CP-201104	2247	1	F	Revert CR 0820t	1650
2020-00	CT#88e	CP-201104	2247	1	F	Change of E-LITRAN LIE Canability	16.5.0
2020-06	CT#88e	CP-201106	2250	1	F	Store the received S-NSSAI via ePDG in the configured	16.5.0
2020.06	CT#890	CP-201114	2251		F	Re-initiation of NSSAA - Reactive solution	1650
2020-00	CT#88e	CP-201104	2255	1	F	Clarification of the forbidden PLMN list used for non-3GPP	16.5.0
2020-06	CT#88e	CP-201104	2256		F	access Re-enabling the N1 mode capability upon request from the upper layers	16.5.0
2020-06	CT#88e	CP-201104	2257		F	Correction of re-enabling E-UTRA capability	16.5.0
2020-06	CT#88e	CP-201104	2258	1	F	Native 5G-GUTI in Additional GUTI IE	16.5.0
2020-06	CT#88e	CP-201104	2259	1	F	Correction of IEI	16.5.0
2020-06	CT#88e	CP-201097	2260	-	F	Maintenance of T3517	16.5.0
2020-06	CT#88e	CP-201137	2261		F	Operation of UE in SNPN access mode when timer T3247 expires	16.5.0
2020-06	CT#88e	CP-201137	2262		F	Reference correction for SNPN	16.5.0
2020-06	CT#88e	CP-201104	2264	1	D	Stop T3346 before sending NAS message	16.5.0
2020-06	CT#88e	CP-201097	2265	1	F	Condition under which the UE shall enter 5GMM-IDLE mode	16.5.0
2020-06	CT#88e	CP-201104	2267	1	F	Correction to handling of 5GSM timers in abnormal cases	16.5.0
2020-06	CT#88e	CP-201104	2269	1	F	Clarification for de-registration procedure initiation	16.5.0

2020-06	CT#88e	CP-201104	2270		F	Clarification in state transition of 5GMM-DEREGISTERED from another 5GMM state	16.5.0
2020-06	CT#88e	CP-201104	2271		F	Clarification of SMS over NAS supported bit in initial	16.5.0
2020-06	CT#88e	CP-201104	2272		F	Clarification on missing subclause in 5GMM-	16.5.0
2020-06	CT#88e	CP-201104	2273		F	Clarification on missing subclause in 5GMM- REGISTERED ATTEMPTING-REGISTRATION-UPDATE	16.5.0
2020-06	CT#88e	CP-201104	2274		F	Clarification regarding update status in NR RAT	16.5.0
2020-06	CT#88e	CP-201104	2275	1	F	Correction to paging timer stop in case of integrity check failure	16.5.0
2020-06	CT#88e	CP-201104	2277	1	D	Correction to spelling mistakes	16.5.0
2020-06	CT#88e	CP-201137	2278	2	F	UE shall use the GUTI assigned by the same SNPN during	16.5.0
2020.06	CT#990	CP 201127	2270	_	F	registration	1650
2020-00	CT#880	CP-201137	2219	1	F	Clarification on S-NSSAL deletion based on the rejected NSSAL	16.5.0
2020 00	01#000		2201	1	1	due to NSSAA in the roaming case	10.0.0
2020-06	CT#88e	CP-201114	2282	1	F	Correction on allowed NSSAI for UE not supporting NSSAA	16.5.0
2020-06	CT#88e	CP-201104	2283		В	Indicate 5GSM cause when initiating 5GSM procedure for error handling	16.5.0
2020-06	CT#88e	CP-201104	2284	1	F	PDU session release upon receipt of PDU session status IE	16.5.0
2020-06	CT#88e	CP-201130	2286	1	F	Correction to the handling for 5GSM #27	16.5.0
2020-06	CT#88e	CP-201104	2287		F	Stop back-off timer upon receipt of 5GSM #39	16.5.0
2020-06	CT#88e	CP-201104	2289		D	Removal of duplicate words	16.5.0
2020-06	CT#88e	CP-201105	2290	1	F	Correction to 5GMM-REGISTERED.NORMAL-SERVICE	16.5.0
2020-06	CT#88e	CP-201105	2292		F	Correction to handling of #3/#6/#7	16.5.0
2020-06	CT#88e	CP-201105	2294	1	F	Correction to 5GMM-DEREGISTERED.NORMAL-SERVICE	16.5.0
2020-06	CT#88e	CP-201105	2296		F	Correction to subclause in Requested NSSAI	16.5.0
2020-06	CT#88e	CP-201105	2297		F	Clarification in usage of SIM terminology in 5GS services	16.5.0
2020-06	CT#88e	CP-201105	2299	2	F	Support for continuity of emergency session upon registration failure	16.5.0
2020-06	CT#88e	CP-201137	2303		F	Updating Port management information container IE	16.5.0
2020-06	CT#88e	CP-201097	2304		F	De-registration request and CPSR collision case in the NW	16.5.0
2020-06	CT#88e	CP-201097	2305		F	Additional stop condition for timer T3580	16.5.0
2020-06	CT#88e	CP-201097	2306	1	F	Abonormal cases on UE side and the CPSR message	16.5.0
2020-06	CT#88e	CP-201097	2307	1	F	Service gap control: Alignment of NW and UE behaviour for timer T3447	16.5.0
2020-06	CT#88e	CP-201097	2308	1	F	Service gap control: Exceptions to start of timer T3447	16.5.0
2020-06	CT#88e	CP-201114	2309	1	F	UE behaviour when more than 16 S-NSSAIs received in pending NSSAI IE	16.5.0
2020-06	CT#88e	CP-201137	2313	1	F	Handling of CAG information list in REGISTRATION ACCEPT messages	16.5.0
2020-06	CT#88e	CP-201137	2314		F	Provision of CAG information list in SERVICE REJECT message.	16.5.0
2020-06	CT#88e	CP-201114	2315	1	F	Size of pending NSSAI in REGISTRATION ACCEPT message	16.5.0
2020-06	CT#88e	CP-201097	2316		F	Corrections for Enhanced Coverage in 5GS for CloT	16.5.0
2020-06	CT#88e	CP-201105	2317		F	Not including NSSAI for emergency session for interworking without N26 interface	16.5.0
2020-06	CT#88e	CP-201105	2318		F	AME not using 5GMM registration status in UF status IF	16.5.0
2020-06	CT#88e	CP-201097	2319	1	F	CPBO timer handling when AMF is congested for CP data	16.5.0
2020-06	CT#88e	CP-201108	2321	•	D	Removal of TMBR	16.5.0
2020-06	CT#88e	CP-201284	2323	2	B	Ethernet header compression for CP CIoT – 5GSM aspects	16.5.0
2020-06	CT#88e	CP-201105	2326	1	C	Associating S-NSSAI-based congestion backoff timers with S- NSSAI when S-NSSAI is provided during PDU session establishment	16.5.0
2020-06	CT#88e	CP-201097	2327	2	С	Updates to non-allowed area restrictions	16.5.0
2020-06	CT#88e	CP-201127	2328	1	F	Correction of RACS ID deletion via UCU	16.5.0
2020-06	CT#88e	CP-201114	2330	1	F	Correction related the pending NSSAI	16.5.0
2020-06	CT#88e	CP-201105	2332	1	F	Clarification on handling of rejected NSSAI for the current registration area	16.5.0
2020-06	CT#88e	CP-201114	2333	1	F	Clarification on S-NSSAI(s) in URSP(NSSP) be added into the request NSSAI	16.5.0
2020-06	CT#88e	CP-201097	2336	1	F	Multiple DRB support for UEs in NB-N1 mode	16.5.0
2020-06	CT#88e	CP-201097	2337	1	F	Establishment of UP resources for NB-IoT based on number of supported DRBs	16.5.0
2020-06	CT#88e	CP-201114	2339	1	F	A default S-NSSAI not subject to NSSAA	16.5.0
2020-06	CT#88e	CP-201137	2340	1	F	Clarification on the non-supported functions and procedures for SNPN	16.5.0

2020-06	CT#88e	CP-201105	2341		D	Correction on unclear texts regarding the CONFIGURATION UPDATE COMMAND message	16.5.0
2020-06	CT#88e	CP-201097	2342	1	F	IP header compression after inter-system change from S1 mode	16.5.0
2020-06	CT#88e	CP-201105	2344	1	F	Trigger SR over non-3GPP access after T3346 expiry.	16.5.0
2020-06	CT#88e	CP-201322	2345	2	В	Provisioning of DNS server security information to the UE	16.5.0
2020-06	CT#88e	CP-201097	2346	1	F	Adding a missing case on the UE side for CPSR	16.5.0
2020-06	CT#88e	CP-201105	2347		F	Condition for setting the Selected EPS NAS algorithm IE to NULL	16.5.0
2020-06	CT#88e	CP-201105	2348	2	F	Connected mode mobility from N1 mode to S1 mode and DL NAS COUNT handling	16.5.0
2020-06	CT#88e	CP-201105	2350	1	F	Correction in the AMF behaviour upon LADN information	16.5.0
2020-06	CT#88e	CP-201105	2352	1	F	Unify terminology for default S-NSSAIs and subscribed S- NSSAIs marked as default	16.5.0
2020-06	CT#88e	CP-201105	2356		F	Correction to Service Reject with cause #28	16.5.0
2020-06	CT#88e	CP-201105	2357	1	F	UE behaviour when receiving allowed NSSAI in CUC	16.5.0
2020-06	CT#88e	CP-201105	2358		F	Ciphering initial registration message with NULL algorithm	16.5.0
2020-06	CT#88e	CP-201105	2362		F	Clean up description of Cause #34 in TS 24.501	16.5.0
2020-06	CT#88e	CP-201105	2364	1	F	Semantic error check for duplicate QRI or QFI	16.5.0
2020-06	CT#88e	CP-201137	2366		D	Editorial change to SNPN	16.5.0
2020-06	CT#88e	CP-201137	2368	1	F	No CAG ID in de-registration request	16.5.0
2020-06	CT#88e	CP-201097	2369	1	F	No dedicated EPS bearer for interworking from WB-N1 to NB- S1 mode	16.5.0
2020-06	CT#88e	CP-201137	2371	1	F	Management for SNPN access mode per access type	16.5.0
2020-06	CT#88e	CP-201097	2372	2	F	Redirection of UE from N1 mode to S1 mode	16.5.0
2020-06	CT#88e	CP-201105	2375	1	F	NAS message transmission failure indication with delayed TAI change	16.5.0
2020-06	CT#88e	CP-201097	2376	1	F	PDU session transfer between 3GPP and non-3GPP when CP CloT 5GS optimization is being used	16.5.0
2020-06	CT#88e	CP-201097	2377	1	F	PDU session transfer between 3GPP and non-3GPP when UP CloT 5GS optimization is being used	16.5.0
2020-06	CT#88e	CP-201097	2379		F	Correction on CIoT small data container IE	16.5.0
2020-06	CT#88e	CP-201097	2380		F	Maximum length of Unstructured data via the control plane	16.5.0
2020-06	CT#88e	CP-201097	2381		F	Missing LCS/LPP container content in Payload container IE	16.5.0
2020-06	CT#88e	CP-201105	2382		F	Handling of multiple QoS flow descriptions	16.5.0
2020-06	CT#88e	CP-201105	2384	1	F	Clarification of NAS COUNT handling in 5G	16.5.0
2020-06	CT#88e	CP-201114	2386	1	F	Providing complete pending NSSAI for NSSAA	16.5.0
2020-06	CT#88e	CP-201137	2388	1	F	SNPN services via a PLMN over 3GPP access	16.5.0
2020-06	CT#88e	CP-201108	2390	1	В	N5GC NAS aspects	16.5.0
2020-06	CT#88e	CP-201108	2391	1	В	N5GC service area restrictions	16.5.0
2020-06	CT#88e	CP-201137	2394	2	F	Prevention of loop for 5GMM cause #62	16.5.0
2020-06	CT#88e	CP-201108	2398		F	N5CW device registration and IP assignment	16.5.0
2020-07	CT#88e					Editorial corrections by rapporteur and MCC. Addition of IEI values	16.5.1
2020-09	CT#89e	CP-202171	1970	2	F	Handling of LADN information when the UE is operating in SNPN access mode	16.6.0
2020-09	CT#89e	CP-202156	2086	5	F	S-NSSAIs always selected by AMF from allowed NSSAI	16.6.0
2020-09	CT#89e	CP-202166	2092	5	F	TA change during Authentication procedure in 5GMM- CONNECTED mode	16.6.0
2020-09	CT#89e	CP-202152	2220	2	F	IPv6 configuration for W-AGF acting on behalf of FN-RG	16.6.0
2020-09	CT#89e	CP-202156	2244	4	F	Disabling of N1 capabilities when all requested S-NSSAIs subjected to NSSAA are rejected due to failure of NSSAA or when no slice is available for UE	16.6.0
2020-09	CT#89e	CP-202171	2251	2	F	Alternative 2: UE behaviour regarding N1 mode capability upon T3247 expiry	16.6.0
2020-09	CT#89e	CP-202171	2252	1	В	Alternative 2: Handling of a UE not allowed to access SNPN services via a PLMN by subscription with 5GMM cause value	16.6.0
2020-09	CT#89e	CP-202150	2397	4	F	Transfer of PDN connection from untrusted non-3GPP access connected to EPC to 5GS	16.6.0
2020-09	CT#89e	CP-202146	2401		F	Correction to PDU session ID inclusion in UL and DL NAS transport	16.6.0
2020-09	CT#89e	CP-202156	2403	1	F	Clarification on the condition when the allowed NSSAI IE shall be included in the REGISTRATION ACCEPT message	16.6.0
2020-09	CT#89e	CP-202156	2405	1	F	Consistency of the term on rejected NSSAI for the failed or revoked NSSAA	16.6.0
2020-09	CT#89e	CP-202156	2409	1	F	Correction to clarify S-NSSAI(s) in allowed NSSAI doesn't require NSSAA	16.6.0
2020-09	CT#89e	CP-202156	2410	1	F	Clarification on the "NSSAA to be performed" indicator	16.6.0

2020-09	CT#89e	CP-202166	2411	1	F	Support of User Plane Integrity Protection for any data rates	16.6.0
2020-09	CT#89e	CP-202150	2418	1	F	URSP evaluation after rejection with the same URSP rule	16.6.0
2020-09	CT#89e	CP-202150	2419	1	F	Remove #43 in PDU session modification command not accepted by UE	16.6.0
2020-09	CT#89e	CP-202156	2420	1	F	NSSAA Slice handling for 1-to-many mapping in roaming scenario	16.6.0
2020-09	CT#89e	CP-202153	2422	1	F	Correcting partial implementation of CR#2029	16.6.0
2020-09	CT#89e	CP-202150	2423		F	Correcting partial implementation of CR#2221	16.6.0
2020-09	CT#89e	CP-202153	2424	1	F	"MA PDU request" when the 5G-RG performs inter-system change from S1 mode to N1 mode with an MA PDU session with a PDN connection as a user-plane resource	16.6.0
2020-09	CT#89e	CP-202152	2430		F	W-CP connection	16.6.0
2020-09	CT#89e	CP-202150	2433		F	SIM not applicable for 5GS cases	16.6.0
2020-09	CT#89e	CP-202150	2434	1	F	NAS MAC terminology	16.6.0
2020-09	CT#89e	CP-202150	2436	1	F	Concession bandling of initial registration for emergency	16.6.0
2020-09	CT#89e	CP-202150	2437	1	F	Corrections to the QoS parameter checks for "unstructured" data and for QoS flow deletion	16.6.0
2020-09	CT#89e	CP-202168	2441		F	Removal of Editor's note on inter PLMN mobility under same	16.6.0
2020-09	CT#89e	CP-202159	2442		F	Removal of Editor's note on UAC for IAB	16.6.0
2020-09	CT#89e	CP-202146	2443		F	Avoiding double barring for CPSR following NAS connection recovery from fallback	16.6.0
2020-09	CT#89e	CP-202146	2445		F	Correction to the 5GS network feature support IE	16.6.0
2020-09	CT#89e	CP-202171	2459		F	Correction of counters in an SNPN	16.6.0
2020-09	CT#89e	CP-202171	2460		F	Provisioning of a CAG information list in Service Request procedure	16.6.0
2020-09	CT#89e	CP-202156	2462	1	F	NSSAA during PDU session modification procedure	16.6.0
2020-09	CT#89e	CP-202153	2464	1	F	Clarification on the applicability of Allowed PDU session status IE to MA PDU	16.6.0
2020-09	CT#89e	CP-202153	2465	1	F	Correction on unnecessary restriction for modifying/upgrading a PDU session to an MA PDU session	16.6.0
2020-09	CT#89e	CP-202153	2466	1	F	Correction on PDU session status IE handling for MA PDU sessions	16.6.0
2020-09	CT#89e	CP-202153	2467	1	F	local release of an MA PDU session having user plane resources established on both 3GPP access and non-3GPP access	16.6.0
2020-09	CT#89e	CP-202150	2468	1	F	Clarification for SR attempt count reset	16.6.0
2020-09	CT#89e	CP-202150	2469	1	F	Handling for SR in 5U2 state	16.6.0
2020-09	CT#89e	CP-202156	2470	1	F	Clairification of Rejected NSSAI	16.6.0
2020-09	CT#89e	CP-202146	2473		F	CP data allowed in connected mode in Non-allowed area	16.6.0
2020-09	CT#89e	CP-202156	2474	1	F	Deleting Editors note regarding to network slice-specific re- authorization and re-authorization	16.6.0
2020-09	CT#89e	CP-202152	2476		F	IPv6 prefix not allocated	16.6.0
2020-09	CT#89e	CP-202171	2480		F	Minimum length of port management information container in SM messages	16.6.0
2020-09	CT#89e	CP-202150	2481	1	F	Mapped dedicated EPS bearer without default EPS bearer	16.6.0
2020-09	CT#89e	CP-202150	2482		F	Calculation of MAC in NAS transparent containers	16.6.0
2020-09	CT#89e	CP-202150	2483	1	F	Provisioning of DNS server security information to the UE- 25.401	16.6.0
2020-09	CT#89e	CP-202168	2484	1	F	Use existing NAS signalling connection to send mobility reg due to receipt of URC delete indication IE.	16.6.0
2020-09	CT#89e	CP-202156	2488	1	F	Clarification of Rejected NSSAI associated with 5GMM cause #62	16.6.0
2020-09	CT#89e	CP-202171	2491		F	CAG information list in Registration reject message.	16.6.0
2020-09	CT#89e	CP-202150	2492	1	F	CR#2299 clean up: continuity of emergency session upon registration failure	16.6.0
2020-09	CT#89e	CP-202156	2497	2	F	Correction on UE behavior for the rejected NSSAI for the failed or revoked NSSAA and the pending NSSAI when the Allowed NSSAI is received	16.6.0
2020-09	CT#89e	CP-202146	2499	1	F	Adding the handling of AMF for case k in the service request procedure	16.6.0
2020-09	CT#89e	CP-202146	2502	1	F	UE behavior when the timer T3447 is stopped	16.6.0
2020-09	CT#89e	CP-202171	2503		F	UE behavior on SNPN access mode when accessing to PLMN services via a SNPN	16.6.0
2020-09	CT#89e	CP-202150	2506	2	F	Mobility Registration for Inter-RAT movement	16.6.0
2020-09	CT#89e	CP-202171	2507	1	F	#76 cause handling in case of reception of Registration Reject in roaming scenarios	16.6.0
2020-09	CT#89e	CP-202150	2509	1	F	Corrections on the error check of QoS rules	16.6.0
2020-09	CT#89e	CP-202171	2512	1	F	Add definition of "allowed CAG list"	16.6.0

2020-09	CT#89e	CP-202032	2514	1	F	Paging not initiated for PDU session transfer from non-3GPP	16.6.0
2020-09	CT#89e	CP-202146	2515		F	LIE specific DRX value for NB-IoT	1660
2020-09	CT#89e	CP-202156	2521	1	F	Rejection of PDU session establishment associated with an S-	16.6.0
						NSSAI for which NSSAA is re-initiated	
2020-09	CT#89e	CP-202156	2525	1	F	Removal of the "failed or revoked NSSAA" definition	16.6.0
2020-09	C1#89e	CP-202171	2526		F	Finding a suitable cell in a PLMN where a UE is allowed to access a non-CAG cell	16.6.0
2020-09	CT#89e	CP-202171	2527	1	F	5GMM cause value #76 mapped to a different 5GMM cause	16.6.0
				•		value	
2020-09	CT#89e	CP-202171	2528	1	F	EAB not applicable for a UE operating in SNPN access mode	16.6.0
2020-09	CT#89e	CP-202171	2530	1	F	T3245 for a UE operating in SNPN access mode	16.6.0
2020-09	C1#89e	CP-202171	2531	1	F	Handling of back-off due to 5GSM cause value #27 "missing or unknown DNN" by a UE operating in SNPN access mode	16.6.0
2020-09	CT#89e	CP-202151	2534	1	F	Handing of QoS errors in ESM procedures	16.6.0
2020-09	CT#89e	CP-202151	2535		F	Delete unimplementable QoS operations in ESM procedure	16.6.0
2020-09	CT#89e	CP-202151	2536		F	Packet filter identifier setting when requesting new packet	16.6.0
	OT#00	00.000454	0507		-	filters	10.0.0
2020-09	C1#89e	CP-202151	2537	1	F	Update of the timers table for 5GS session management	16.6.0
2020-09	CT#89e	CP-202151	2539	1	F	Infinite De-registration attempt	16.6.0
2020-09	C1#09E	CF-202140	2540	1	Г	CPSR	10.0.0
2020-09	CT#89e	CP-202122	2552	4	F	Type of the N5GC indication information element	16.6.0
2020-09	CT#89e	CP-202156	2555		F	Clarification of conditions which the rejected NSSAI for the	16.6.0
0000.00	OT#00-	00.000454	0504		г	failed or revoked NSSAA is deleted	40.0.0
2020-09	CT#89e	CP-202151	2561	1	F	13525 clarification for UE configured with high priority access	16.6.0
2020-09	C1#89e	CP-202171	2002	1	Г	registration	16.6.0
2020-09	CT#89e	CP-202151	2564	1	F	Minimum length of "Plain 5GS NAS message"	16.6.0
2020-09	CT#89e	CP-202203	2568	2	F	Resolution of editor's notes on the handling of timers T3484	16.6.0
						and T3585 when the UE provided no S-NSSAI during PDU	
2020-09	CT#89e	CP-202151	2570	1	F	session establishment. Handling of timers T3484 and T3585 received with 5GSM	1660
2020-03	01#036	01-202131	2570	1	1	cause value #39	10.0.0
2020-09	CT#89e	CP-202151	2572	1	С	Allowed NSSAI assignment based on default subscribed	16.6.0
	07/00	05.0004.40	0575		-	NSSAI	10.0.0
2020-09	CT#89e	CP-202146	2575	4	F	Retry restriction for NB-IoT UEs due to out of tariff package	16.6.0
2020-09	CT#896	CP-202156	2577	1	F	Default subcribed S-NSSAIs for re-NSSAA or revoked NSSAA	16.6.0
2020-09	CT#89e	CP-202150	2570	1	Г F	Correction on OoS parameter "value is not used" in 5GS	16.6.0
2020-09	CT#89e	CP-202151	2587	1	F	Handling of T3520 in AUTH RE.	16.6.0
2020-09	CT#89e	CP-202151	2589	1	F	Correction that service reject is received not service request	16.6.0
2020-09	CT#89e	CP-202151	2590	•	F	Correction to implementation of CR2297	16.6.0
2020-09	CT#89e	CP-202151	2591		F	Correction to the implementation of CR0988	16.6.0
2020-09	CT#89e	CP-202146	2598	1	F	Fix of Timer T3448 encoding	16.6.0
2020-09	CT#89e	CP-202152	2602		F	Clarification on TWIF acting on behalf of N5CW device	16.6.0
2020-09	CT#89e	CP-202141	2606		F	AMF including CAG information list in rejection messages	16.6.0
2020-09	CT#89e	CP-202228	2608		F	Correction of the IEI of UE radio capability ID deletion	16.6.0
2020-00	CT#80c	CP-202172	22/12	2	F	Indication	17.0.0
2020-09	01#098	06-2021/3	2240	3	1.	Command	17.0.0
2020-09	CT#89e	CP-202173	2383	3	F	Single-registration mode without N26 for EPS NAS message	17.0.0
0000.00	07#00	00.000470	0404		E	container IE	47.0.0
2020-09	C1#89e	CP-2021/3	2404	1	г	clarification on the applicable access type for persistent PDU session	17.0.0
2020-09	CT#89e	CP-202173	2406	1	D	Clarification on protection of initial NAS messages	17.0.0
2020-09	CT#89e	CP-202173	2408	1	F	Fixing several typos and adding full form of abbreviation W-	17.0.0
						AGF	
2020-09	CT#89e	CP-202184	2416	1	D	Editorial changes – red text corrected to black text	17.0.0
2020-09	CT#89e	CP-202173	2425		D	Not capitalized 5GSM IE names	17.0.0
2020-09	CT#89e	CP-2021/3	2426		F	Incorrect IE names	17.0.0
2020-09		CP-202173	2421		Г	Overlapping requirements in 5.2.22	17.0.0
2020-09	CT#890	CP-202173	2429		r F	Minor corrections	17.0.0
2020-09	CT#800	CP-202173	2432		F	Dual-registration mode list correction	17.0.0
2020-09	CT#89e	CP-202173	2448	2	F	QoS error checks for unstructured PDU session type	17.0.0
2020-09	CT#89e	CP-202024	2452	2	F	The requirement of AMF to provide CAG information list for the	17.0.0
						current PLMN	
2020-09	CT#89e	CP-202173	2454	1	D	Abbreviations correction	17.0.0

2020-09	CT#89e	CP-202173	2456	1	F	Definition of Routing Indicator	17.0.0
2020-09	CT#89e	CP-202173	2457		F	Service Request procedure over non-3GPP access	17.0.0
2020-09	CT#89e	CP-202173	2458		D	Several editorial changes	17.0.0
2020-09	CT#89e	CP-202174	2471	1	F	Clarification of paging response	17.0.0
2020-09	CT#89e	CP-202174	2477		F	Misleading definition of 5G-IA and 5G-EA	17.0.0
2020-09	CT#89e	CP-202174	2479		D	Restructure the statement on establishment cause for non- 3GPP access	17.0.0
2020-09	CT#89e	CP-202174	2489	1	F	Correction to the octet number in 5GS network feature support	17.0.0
2020-09	CT#89e	CP-202174	2490	1	F	Correction to Configred NSSAI updation based on Rejected NSSAI.	17.0.0
2020-09	CT#89e	CP-202174	2505	1	F	Handling of 5GSM procedures when fallback is triggered	17.0.0
2020-09	CT#89e	CP-202174	2511		F	Include NAS message container in security mode complete	17.0.0
2020-09	CT#89e	CP-202113	2513	1	F	High priority access before pass the NSSAA	17.0.0
2020-09	CT#89e	CP-202174	2516	1	F	Exceptions in providing NSSAI to lower lavers	17.0.0
2020-09	CT#89e	CP-202033	2517	1	F	No VPLMN S-NSSAI change via the generic UE configuration	17.0.0
				•	-	update	
2020-09	CT#89e	CP-202174	2519	1	F	Correction in the session transfer	17.0.0
2020-09	CT#89e	CP-202174	2532		F	UE behaviour for service reject with #15	17.0.0
2020-09	CT#89e	CP-202174	2541		F	Mapped 5G security context deletion upon IDLE mode mobility from 5GS to EPS over N26 interface	17.0.0
2020-09	CT#89e	CP-202227	2547	2	F	Clarification to emergency registration procedure	17.0.0
2020-09	CT#89e	CP-202174	2573	2	С	Rejected NSSAI due to subscription	17.0.0
2020-09	CT#89e	CP-202174	2579	1	D	Deleting 5G NAS security context when 5G-EA0 used and PLMN changed	17.0.0
2020-09	CT#89e	CP-202184	2580	1	F	De-registration in ATTEMPTING-REGISTRATION-UPDATE	17.0.0
2020-09	CT#89e	CP-202174	2581	1	F	Correction on Payload container IE	17.0.0
2020-09	CT#89e	CP-202174	2583	1	F	EMM parameters handling for 5G only causes	17.0.0
2020-09	CT#89e	CP-202174	2584	2	F	Clarification on Operator-defined access category definitions	17.0.0
2020-09	CT#89e	CP-202174	2585	1	F	The suggestion on back-off timer for 5GSM #29	17.0.0
2020-09	CT#89e	CP-202174	2597	1	F	Fix of encoding errors in 5GS mobile identity IE	17.0.0
2020-09	CT#89e	CP-202249	2609		F	Avoid unnecessary signalling for CP only PDU sessions after	17.0.0
						inter-system change from S1 mode to N1 mode	
2020-12	CT#90e	CP-203169	2524	2	F	Clarification on HPLMN S-NSSAI	17.1.0
2020-12	CT#90e	CP-203169	2610	1	F	Clarification for CP only PDU session	17.1.0
2020-12	CT#90e	CP-203169	2611	1	F	Clarification for reflective QoS	17.1.0
2020-12	CT#90e	CP-203213	2615	1	А	Alignment of User Plane Integrity Protection description	17.1.0
2020-12	CT#90e	CP-203169	2617		F	Consistency of the term on rejection cause "S-NSSAI not	17.1.0
						available due to the failed or revoked network slice-specific authentication and authorization"	
2020-12	CT#90e	CP-203169	2618	1	F	Clarification on the condition when registration request is rejected for no network slices available	17.1.0
2020-12	CT#90e	CP-203169	2619		F	Correction on UE behaviour after receiving "Network slicing subscription changed" indication	17.1.0
2020-12	CT#90e	CP-203169	2620	1	F	Clarification on the S-NSSAI(s) included in the pending NSSAI	17.1.0
2020-12	CT#90e	CP-203169	2621	1	F	Consistency of NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message	17.1.0
2020-12	CT#90e	CP-203169	2623	1	F	AMF behavior upon receipt of NETWORK SLICE-SPECIFIC AUTHENTICATION COMPLETE message	17.1.0
2020-12	CT#90e	CP-203169	2624		F	Addition of used definitions and abbreviations	17.1.0
2020-12	CT#90e	CP-203169	2625		D	Editorial corrections in 24.501	17.1.0
2020-12	CT#90e	CP-203169	2626		F	Clarification on the 5GMM procedures which can be initiated by the UE in substate 5GMM-REGISTERED.ATTEMPTING- REGISTRATION-UPDATE	17.1.0
2020-12	CT#90e	CP-203169	2627		F	Removal of bullet irrelevant to tracking area concept	17.1.0
2020-12	CT#90e	CP-203169	2628	1	F	Merge of two bullets with the same handling for different Request type IE	17.1.0
2020-12	CT#90e	CP-203169	2631	1	D	Phrase that the abbreviation PCO represents	17.1.0
2020-12	CT#90e	CP-203169	2632	1	F	Protection of NAS IEs	17.1.0
2020-12	CT#90e	CP-203167	2635	2	A	QoS parameter handling for the PDU session transfer between 3GPP and non-3GPP access	17.1.0
2020-12	CT#90e	CP-203169	2637	1	F	Handling of QoS flow description without associated QoS rule	17.1.0
2020-12	CT#90e	CP-203169	2640	1	F	RFCs related to DHCPv6 are obsoleted by RFC 8415	17.1.0
2020-12	CT#90e	CP-203169	2642		F	Inclusion of requested NSSAI in the REGISTRATION REQUEST message	17.1.0
2020-12	CT#90e	CP-203169	2643		F	Clarification on the SPRTI bit of the MICO indication IE	17.1.0
2020-12	CT#90e	CP-203170	2644		F	UE behavior after receiving the rejected NSSAI with rejection cause "S-NSSAI not available in the current PLMN or SNPN"	17.1.0

2020-12	CT#90e	CP-203166	2645	1	А	5G-GUTI reallocation after resume from 5GMM-IDLE mode	17.1.0
2020-12	CT#90e	CP-203186	2647	1	А	Adding a missing "modification request" for the Request type	17.1.0
2020-12	CT#90e	CP-203218	2657	1	А	IE during NSSAA NAS signalling connection release upon CAG information	17.1.0
2020-12	CT#90e	CP-203170	2659		F	update via UCU Missing lower layer indications of barring and alleviation of	17.1.0
2020 12	CT#00-	CD 000470	2000	0	г Г	barring Latin data status IF in CDCD offer integrity shock follows	47.4.0
2020-12	CT#90e	CP-203170	2002	2	F		17.1.0
2020-12	C1#90e	CP-203170	2664	1	F	Missing Allowed PDU Session Status IE in CPSR	17.1.0
2020-12	CT#90e	CP-203179	2667	2	Α	Clarification on 2nd Leg PDU SESSION ESTABLISHMENT ACCEPT handling for MA PDU Sessions	17.1.0
2020-12	CT#90e	CP-203179	2669	2	А	Clarifications on Necessity of ATSSS Container IE	17.1.0
2020-12	CT#90e	CP-203170	2670	1	F	Prohibit UE from setting "Follow-on request pending" in the REGISTRATION REQUEST when UE is in non-allowed area	17.1.0
2020-12	CT#90e	CP-203170	2675		F	Clarifications on indicating subscribed MFBR/GFBR uplink/downlink	17.1.0
2020-12	CT#90e	CP-203170	2676	1	F	Update cases where whether ER-NSSAI IE is used	17.1.0
2020-12	CT#90e	CP-203170	2677	1	F	Extended rejected NSSAI storage	17.1.0
2020-12	CT#90e	CP-203170	2679	1	F	Update definition of Network slicing information	1710
2020 12	CT#000	CP 202196	2691	1 2	1	Excluding the S NSSAI(c) in the pending NSSAI from the	17.1.0
2020-12	CT#90e	CF-203100	2001	2	A .	requested NSSAI	17.1.0
2020-12	C1#90e	CP-203186	2683	1	A	NSSAA over 3GPP access and N3GPP access separately	17.1.0
2020-12	CT#90e	CP-203170	2687	2	F	Add some missing ESM causes on the network side	17.1.0
2020-12	CT#90e	CP-203166	2688	1	А	Timer value of active timer	17.1.0
2020-12	CT#90e	CP-203170	2689	1	F	Paging collision with 5GMM specific procedure or service request procedure	17.1.0
2020-12	CT#90e	CP-203211	2690	1	А	Correction to S-NSSAI based retry restriction	17.1.0
2020-12	CT#90e	CP-203167	2692	1	А	EN resolution on 5QI as criteria type for ODAC	17.1.0
2020-12	CT#90e	CP-203170	2694	•	F	Correction on slice based congestion control	1710
2020-12	CT#90e	CP-203170	2696	1	F	Rejected NSSAI bandling for 1-to-many manning in roaming	17.1.0
2020 12	01#300	01 200170	2000	1	1	scenario	17.1.0
2020-12	CT#90e	CP-203170	2697		F	Set T3517 to smaller value for emergency services fallback	17.1.0
2020-12	CT#90e	CP-203170	2700	1	F	Service request procedure and apportal cases in the LIE for	1710
2020 12	01//000	01 200170	2100	1	1	CPSR and emergency fallback	17.1.0
2020-12	CT#90e	CP-203170	2701	1	F	IEEE Std reference updates	17.1.0
2020-12	CT#90e	CP-203186	2703	1	А	Correct pending NSSAL bandling	1710
2020-12	CT#90e	CP-203170	2707	1	F	Correction to NAS transport procedure	1710
2020 12	CT#000	CP 202170	2709	1	г Г	Correction to FGMM cause #62 and allowed NSSAL	17.1.0
2020-12	CT#90e	CF-203170	2700	1	Г D	Conection to Solvin cause #02 and allowed NSSA	17.1.0
2020-12	CT#90e	CF-203170	2712	1	D	Correction in allowed NCCAL and nending NCCAL bandling	17.1.0
2020-12	C1#90e	CP-203186	2717	1	A	upon receipt of rejected NSSAI	17.1.0
2020-12	CT#90e	CP-203170	2719		F	Clarification the condition that the Extended NSSAI IE is included in the CONFIGURATION UPDATE COMMAND	17.1.0
						message	
2020-12	CT#90e	CP-203171	2722	1	F	Correction on inclusion criteria for IP header compression	17.1.0
	07/00	05.000/5/			_	configuration IE	
2020-12	C1#90e	CP-203171	2724	1	F	compression configuration IE	17.1.0
2020-12	CT#90e	CP-203171	2725	1	F	Correction on the rejected NSSAI in the registration reject message	17.1.0
2020-12	CT#90e	CP-203171	2727	1	F	SNPN access mode over 3GPP access when accessing SNPN services via a PLMN	17.1.0
2020-12	CT#90e	CP-203171	2729	3	F	Handling of pending NSSAI and allowed NSSAI during periodic registration update	17.1.0
2020-12	CT#90e	CP-203171	2730		F	Cell search in NG-RAN	17.1.0
2020-12	CT#90e	CP-203171	2731		F	Correction in the N1 mode capability handling	17.1.0
2020-12	CT#90e	CP-203171	2733	4	F	UE operation in case of routing failure	17.1.0
2020-12	CT#90e	CP-203171	2734	-	F	Paging a UE using eDRX	17.1.0
2020-12	CT#90e	CP-203171	2735	1	F	Correction in the AUSE operation in terms of checking the	17.1.0
2020 12	01#000	01 200111	2100	1	1	presence of the AT_RESULT_IND attribute in the EAP- response/AKA'-challence message	17.1.0
2020-12	CT#90e	CP-203218	2739	2	А	CAG information list in SR reject message	17.1.0
2020-12	CT#90e	CP-203171	2742	1	F	Provision CAG information list through de-registration	1710
2020-12	OT#00-	CD 202474	2740		r E	procedure	17.1.0
2020-12	C1#90e	CP-2031/1	2142	3	r r	INIODINITY Registration after back to coverage	17.1.0
2020-12	CT#90e	CP-2031/1	2/44		r T	Delete 5G NAS security context due to invalid key	17.1.0
2020-12	CT#90e	CP-203171	2745		F	Lack of ID for inter-system change from S1 mode to N1 mode	17.1.0
2020-12	CT#90e	CP-203166	2748	1	А	PDU session release in CP-SR	17.1.0

2020-12	CT#90e	CP-203218	2750	2	А	Update IEI of Port management information container	17.1.0
2020-12	CT#90e	CP-203171	2751		F	Correct location of ABO field	17.1.0
2020-12	CT#90e	CP-203171	2752		F	Correct reference of SM timer	17.1.0
2020-12	CT#90e	CP-203171	2754		F	Only CAG supported UE process CAG information list	17.1.0
2020-12	CT#90e	CP-203171	2757	1	F	Correction of EPS bearer context being activated	17.1.0
2020-12	CT#90e	CP-203171	2758	3	F	Clarification on LADN Information update	17.1.0
2020-12	CT#90e	CP-203171	2759	1	F	Absence of timer T3448	17.1.0
2020-12	CT#90e	CP-203186	2761	1	А	NSSAA for roaming UEs	17.1.0
2020-12	CT#90e	CP-203172	2764	1	D	IE length style in message definition	17.1.0
2020-12	CT#90e	CP-203172	2765		D	Minor corrections	17.1.0
2020-12	CT#90e	CP-203218	2771	1	А	Reception of CAG information list without serving PLMN's entry in roaming	17.1.0
2020-12	CT#90e	CP-203172	2772		F	N5CW device clean up	17.1.0
2020-12	CT#90e	CP-203172	2773		F	Correction in paging procedure	17.1.0
2020-12	CT#90e	CP-203172	2774	3	В	Usage of initial CAG information list	17.1.0
2020-12	CT#90e	CP-203218	2776	-	А	5GMM cause value #76 mapped to a different 5GMM cause	17.1.0
						value in network-initiated de-registration procedure	
2020-12	CT#90e	CP-203172	2779	4	F	Handling of radio link failure during NSSAA procedure	17.1.0
2020-12	CT#90e	CP-203172	2780	4	F	AMF behavior in case of NSSAA failure due to temporal	17.1.0
					_	network failure	
2020-12	CT#90e	CP-203172	2782	1	F	Addition of 5GSM cause #37	17.1.0
2020-12	CT#90e	CP-203172	2783	2	F	Handing of QoS flow description errors	17.1.0
2020-12	CT#90e	CP-203172	2785	1	F	Clarification on stopping back-off timers	17.1.0
2020-12	CT#90e	CP-203172	2786		F	Delete EBI in the QoS flow description when the	17.1.0
0000 40	OT#00+	00.000470	0707		г	corresponding mapped EPS bearer context is deleted	474.0
2020-12	C1#90e	CP-203172	2/8/		F	command	17.1.0
2020-12	CT#90e	CP-203178	2794	2	В	The impact on UE due to the introduction of Authentication and Key Management for Applications (AKMA)	17.1.0
2020-12	CT#90e	CP-203186	2796	2	А	PLMN selection condition upon reception Registration Reject with cause #62	17.1.0
2020-12	CT#90e	CP-203172	2807	1	F	Avoiding repeated inter-system re-directions	17.1.0
2020-12	CT#90e	CP-203172	2809	3	F	UE procedures when a request for emergency services	17.1.0
						fallback not accepted	
2020-12	CT#90e	CP-203215	2811	2	В	Notification to upper layer upper layer for MMTEL video call when T3346 or T3525 running	17.1.0
2020-12	CT#90e	CP-203172	2812		F	Correct UE behaviour for cause #31 in SR	17.1.0
2020-12	CT#90e	CP-203172	2813	1	F	Correction to T3502 for MRU	17.1.0
2020-12	CT#90e	CP-203172	2814	1	F	Deregistration before initial registration in SNPN selection	17.1.0
2020-12	CT#90e	CP-203172	2815	1	F	Clarification on description of triggering UE to enter 5GMM- DEREGISTERED state	17.1.0
2020-12	CT#90e	CP-203218	2817	2	А	Correction in 5GMM cause value #72	17.1.0
2020-12	CT#90e	CP-203166	2818		А	Rapporteur cleanup of editor's notes for 5G_CIoT	17.1.0
2020-12	CT#90e	CP-203172	2819	2	F	MO-SMS in non-3GPP access	17.1.0
2020-12	CT#90e	CP-203172	2824	1	F	Clarification on default configured NSSAI update will initiate a registration procedure by UE when "re-registration requested" indicated	17.1.0
2020-12	CT#90e	CP-203173	2826		F	Addition of used abbreviations	17.1.0
2020-12	CT#90e	CP-203173	2827	1	D	Editorial corrections in 24.501	17.1.0
2020-12	CT#90e	CP-203173	2828	1	F	Consistency of terms "5GMM-IDLE mode over non-3GPP access" and "5GMM-CONNECTED mode over non-3GPP access"	17.1.0
2020-12	CT#90e	CP-203173	2829	1	D	Consistent usage of acronym UE	17.1.0
2020-12	CT#90e	CP-203173	2832	1	F	Set the Follow-on request indicator to "Follow-on request pending"	17.1.0
2020-12	CT#90e	CP-203173	2833		F	requested NSSAI is Requested NSSAI IE or Requested	17.1.0
2020-12	CT#90e	CP-203173	2834		F	Mobility and periodic registration update when the UE receives	17.1.0
2020-12	CT#90e	CP-203173	2835		F	Condition when the UE shall include or not include the NAS	17.1.0
2020-12	CT#90e	CP-203173	2836		F	S-NSSAI(s) contained in the pending NSSAI	17.1.0
2020-12	CT#90e	CP-203173	2839	1	F	Default configured NSSAL storage after update by LIF	17.1.0
12	0	2. 200110		'	-	Parameters Update via UDM Control Plane Procedure	
2020-12	CT#90e	CP-203173	2841	1	F	Correcting the SERVICE ACCEPT message into SERVICE REQUEST message.	17.1.0
2020-12	CT#90e	CP-203173	2842		D	REGISTRATION ACCCEPT message	17.1.0
2020-12	CT#90e	CP-203173	2845	1	F	Correction on the condition of filling allowed NSSAI in registration accept message	17.1.0
2020-12	CT#90e	CP-203186	2847	2	А	Add a missing case for registration reject	17.1.0
---------	--------	-----------	------	---	---	---	--------
2020-12	CT#90e	CP-203173	2848		F	Limit the guidance only for UE not supporting ER-NSSAI	17.1.0
2020-12	CT#90e	CP-203173	2851	1	F	The handling of a CAG information list with no entry	17.1.0
2020-12	CT#90e	CP-203173	2853		F	Correction to incrementing the registration attempt counter during abnormal cases for Mobility and periodic registration update for initiating an emergency PDU session	17.1.0
2020-12	CT#90e	CP-203173	2854		F	Correction to the reference to service request abnormal cases	17.1.0
2020-12	CT#90e	CP-203173	2855		F	Clarify PDU session modification command reject due to QoS- related errors	17.1.0
2020-12	CT#90e	CP-203173	2858	1	F	Addition of missing requirements for storing KAUSF, KSEAF, SOR counter and UE parameter update counter	17.1.0
2020-12	CT#90e	CP-203173	2859	1	F	Correction of UE-requested PDU session modification	17.1.0
2020-12	CT#90e	CP-203205	2860	2	В	DNN setting in the 5GSM sublayer	17.1.0
2020-12	CT#90e	CP-203174	2861	1	F	Buffered T3512 handling in non allowed service area	17.1.0
2020-12	CT#90e	CP-203174	2863	1	F	Handling of Emergency Service Fallback procedure in NON- ALLOWED area	17.1.0
2020-12	CT#90e	CP-203174	2866	1	F	NAS signalling connection release when SAR list is received	17.1.0
2020-12	CT#90e	CP-203166	2868		А	5G-GUTI reallocation after MT service request but before connection suspend	17.1.0
2020-12	CT#90e	CP-203174	2869	1	F	Completion of service request procedure following CPSR for emergency fallback	17.1.0
2020-12	CT#90e	CP-203174	2870	1	F	Clarification on Selected EPS NAS security algorithms IE	17.1.0
2020-12	CT#90e	CP-203174	2872		F	SNPN access mode over 3GPP access when accessing PLMN services via a SNPN	17.1.0
2020-12	CT#90e	CP-203174	2876	1	F	Adding the definition of non-CAG cell	17.1.0
2020-12	CT#90e	CP-203174	2877	1	F	Protection of 5GSM message	17.1.0
2020-12	CT#90e	CP-203167	2879		А	Shared 5G NAS security context	17.1.0
2020-12	CT#90e	CP-203205	2882	1	F	Adding the abbreviations of PAP/CHAP in TS 24.501 and fixing a minor grammatical error in the NOTE on PAP/CHAP	17.1.0
2020-12	CT#90e	CP-203218	2884	1	А	AN Release triggered by CAG information Update	17.1.0
2020-12	CT#90e	CP-203174	2885	1	F	Miss local de-registration procedure before entering DEREGISTERED state	17.1.0
2020-12	CT#90e	CP-203270	2889	5	F	N1 mode disable when neither emergency services nor emergency services fallback works	17.1.0
2020-12	CT#90e	CP-203174	2890	1	F	Provide different UE IDs for trusted and untrusted non-3GPP access	17.1.0
2020-12	CT#90e	CP-203174	2895	1	F	Delay enabling N1 mode until NAS signalling connection or RR connection is released	17.1.0
2020-12	CT#90e	CP-203179	2900	1	A	Release MA PDU session when connecting to an ATSSS unsupported AMF	17.1.0
2020-12	C1#90e	CP-203174	2901		F	Correction of UE handlings on 5GSM cause #50 and #51	17.1.0
2020-12	CT#90e	CP-203174	2905	1	F	UE behavior when the UE receives the rejected NSSAI	17.1.0
2020-12	CT#90e	CP-203174	2907	1	F	S-NSSAI not available due to the failed or revoked NSSAA	17.1.0
2020-12	CT#90e	CP-203174	2908	1	F	PDU session ID in CPSR message	17.1.0
2020-12	CT#90e	CP-203174	2910	1	F	No available S-NSSAIs and emergency PDU session	17.1.0
2020-12	CT#90e	CP-203174	2917		F	Correction on handling for 5GMM #73 for DoS attack	17.1.0
2020-12	CT#90e	CP-203174	2918		F	Correction on MICO indication IE	17.1.0
2020-12	CT#90e	CP-203174	2922		F	Collision of error handling on QoS operations	17.1.0
2020-12	CT#90e	CP-203186	2924	1	А	Default S-NSSAI for which NSSAA has been successful, is included in allowed NSSAI in case of no eligible requested NSSAI	17.1.0
2020-12	CT#90e	CP-203179	2931	1	А	SM/MM coordination for MAPDUs	17.1.0
2020-12	CT#90e	CP-203179	2935	1	А	Clarification on release of MA PDU session over both accesses	17.1.0
2020-12	CT#90e	CP-203268	2921	3	F	UE reachability after NOTIFICATION RESPONSE	17.1.0
2020-12	CT#90e	CO-203253	2926	4	F	Checking ACK bit of the SOR container in the DL NAS TRANSPORT	17.1.0
2020-12	CT#90e	CP-203179	2937	1	А	MA PDU session modification rejection during change from S1 mode to N1 mode	17.1.0
2020-12	CT#90e	CP-203177	2939		А	Emergency registration not applicable for FN-RG	17.1.0
2020-12	CT#90e	CP-203174	2940	1	F	The selected PLMN for emergency services via trusted non- 3GPP access	17.1.0
2020-12	CT#90e	CP-203205	2941	1	F	Avoid including both PAP/CHAP and EAP identifiers in PDU session establishment request	17.1.0
2020-12	CT#90e	CP-203177	2945		А	Addition to the non 5G capable over WLAN (N5CW) device term	17.1.0
2020-12	CT#90e	CP-203274	2947	2	F	Delete previously allowed NSSAI upon receipt of "NSSAA to be performed"	17.1.0
2021-03	CT#91e	CP-210117	2549	2	F	UE behaviour in case of no allowed NSSAI is available	17.2.0

2021-03	CT#91e	CP-210117	2705	5	F	PDU session establishment request attempt during ongoing re-	17.2.0
2021-03	CT#91e	CP-210117	2864	1	F	I ocal release of PDL session due to Service Area Restriction	1720
2021-03	CT#91e	CP-210117	2865	3	F	Conflict of sub-state NON-ALLOWED-SERVICE with other	17.2.0
				Ŭ.		5GMM-REGISTERED sub-states	
2021-03	CT#91e	CP-210117	2898	7	F	Condition to stop timer T3540	17.2.0
2021-03	CT#91e	CP-210121	2948	3	F	Kaust change	17.2.0
2021-03	CT#91e	CP-210121	2949	1	F	Clarification on AKMA	17.2.0
2021-03	CT#91e	CP-210121	2952	3	F	Obtaining KAKMA and A-KID from NAS	17.2.0
2021-03	CT#91e	CP-210121	2953	1	F	Collision of AKMA and NAS AKA procedure handling	17.2.0
2021-03	CT#91e	CP-210121	2954	2	F	UE handling in case of no KAUSF available for AKMA	17.2.0
2021-03	CT#91e	CP-210117	2956	1	F	Suspention of 5GSM messages during SOR	17.2.0
2021-03	CT#91e	CP-210117	2958	1	F	Correction of Requested NSSAI nandling	17.2.0
2021-03	CI#91e	CP-210114	2960		A	contection of handling of CAG information from a PLININ equivalent to the HPI MN."	17.2.0
2021-03	CT#91e	CP-210117	2962		D	Minor corrections	17.2.0
2021-03	CT#91e	CP-210114	2964	1	А	Correction for SNPN access mode in non-3GPP access	17.2.0
2021-03	CT#91e	CP-210117	2965	1	F	NAS signalling connection release triggered by CAG	17.2.0
						information list without entry of current PLMN	-
2021-03	CT#91e	CP-210117	2967		F	Abnormal cases in the UE for PDU EAP result message	17.2.0
	07/04	0.0.4.4.4.4.5		-	_	transport procedure	
2021-03	CT#91e	CP-210095	2968	2	F	Handling of Kaust and Kseat created before EAP-success	17.2.0
2021-03	CT#91e	CP-210117	2969	1	F	PDU SESSION ESTABLISHMENT message	17.2.0
2021-03	CT#91e	CP-210249	2970	3	F	SOR transparent container coding	17.2.0
2021-03	CT#91e	CP-210107	2972	1	A	Fixing mis-implementation of CR2140	17.2.0
2021-03	CI#91e	CP-210117	2973	2	F	NB-N1 mode and max number of user planes resources	17.2.0
2021-03	CT#91e	CP-210117	2974		F	The handling of a CAG information list with no entry	1720
2021-03	CT#91e	CP-210117	2976	1	F	Correction for NB-N1 mode and maximum number of PDU	17.2.0
2021 00	01//010	01 210111	2010	1	1	sessions with active user plane resources	17.2.0
2021-03	CT#91e	CP-210117	2978	1	F	PEI for UE not supporting any 3GPP access technologies	17.2.0
2021-03	CT#91e	CP-210117	2979		F	Reference to UCU procedure is missing for a 5G-GUTI	17.2.0
0004.00	OT#04 a	00.040447	0000		F	reallocation variant	47.0.0
2021-03	C1#91e	CP-210117	2980	1	Г	revoked NSSAA when S-NSSAI rejected for the failed of	17.2.0
2021-03	CT#91e	CP-210169	2982	3	F	Update of CPSR procedure for low power location	17.2.0
						event reportingt	
2021-03	CT#91e	CP-210117	2983		F	UE-requested PDU session release with 5GSM cause #26	17.2.0
2021-03	CT#91e	CP-210117	2984	1	F	Clarify UE handling of receiving DL NAS TRANSPORT	17.2.0
2024 02	CT#01a	CD 010117	2005		F	message with 5GMM cause #28	17.0.0
2021-03	CT#91e	CP-210117	2900	1	Г	Clarify 5CSM pop congestion back off timer bandling for ro	17.2.0
2021-03	C1#916	CF-210116	2900	1	Г	registration required	17.2.0
2021-03	CT#91e	CP-210107	2989		А	Fix location of 5GSM congestion re-attempt indicator IE in	17.2.0
						PDU session establishment reject message and PDU session	
						modification reject message	
2021-03	CI#91e	CP-210118	2993	1	F	Correct description of #54 by taking into account its	17.2.0
2021-03	CT#91e	CP-210118	2995		F	Correction of Notification procedure	1720
2021-03	CT#91e	CP-210133	2000	1	F	Actions on T3247 expiry for other supported RATs	17.2.0
2021-03	CT#91e	CP-210118	2997	2	F	Timer related actions upon reception of ALITHENTICATION	17.2.0
	0010	2. 210110		<u>_</u>	-	REJECT	
2021-03	CT#91e	CP-210118	2998	1	F	State transition from 5GMM-CONNECTED mode with RRC	17.2.0
	0.5				-	inactive indication to LIMITED-SERVICE	
2021-03	CT#91e	CP-210118	3002	1	F	Clarifications to the handling of the stored pending NSSAI	17.2.0
2021-03	CT#91e	CP-210118	3004		F	Corrections for 5GS network feature support IE	17.2.0
2021-03	C1#91e	CP-210118	3005		F	UE behavior when received cause #62 in the REGISTRATION REJECT message	17.2.0
2021-03	CT#91e	CP-210118	3006		F	Consistency of the term on rejection cause "S-NSSAI not	17.2.0
						available due to the failed or revoked network slice-specific	-
						authentication and authorization"	
2021-03	CT#91e	CP-210118	3007	1	F	Inclusion of Extended rejected NSSAI IE	17.2.0
2021-03	CT#91e	CP-210118	3008	1	D	Editorial corrections on the first letter to be lowercase or	17.2.0
2021.02	CT#010	CP-210119	3000	1	F	uppercase	17 2 0
2021-03	CT#01c	CP_210110	3010		D 1	Fix several types	17.2.0
2021-03	CT#010	CP-210110	3010	2	F	Complement when and how the configured NSSAL rejected	17.2.0
2021-03		51 210110	0011	3	-	NSSAI and pending NSSAI may be changed	11.2.0
2021-03	CT#91e	CP-210118	3012		F	Deletion of the duplicated content about new allowed NSSAI	17.2.0
						storage	

2021-03	CT#91e	CP-210118	3013		F	Missing pending NSSAI and rejected NSSAI(s) for the failed or revoked NSSAA for no duplicated PLMN identities or SNPN identities	17.2.0
2021-03	CT#91e	CP-210118	3014		F	Add the native security context after changing to N1 mode in connected mode	17.2.0
2021-03	CT#91e	CP-210118	3016	1	F	Clarification of maintaining 5G-GUTI in an abnormal case	17.2.0
2021-03	CT#91e	CP-210118	3019		F	Delete previously allowed NSSAI upon receipt of "NSSAA to be performed" during initial registration	17.2.0
2021-03	CT#91e	CP-210118	3020		F	Cleanup of "NSSAA to be performed set to 1"	17.2.0
2021-03	CT#91e	CP-210119	3022		F	Remove the error case for mandatory IE of PDU SESSION MODIFICATION COMMAND message	17.2.0
2021-03	CT#91e	CP-210119	3024	1	F	T3540	17.2.0
2021-03	CT#91e	CP-210119	3025	2	F	Exception data in restricted service area for a UE in connected mode	17.2.0
2021-03	CT#91e	CP-210106	3027		А	T3575	17.2.0
2021-03	CT#91e	CP-210135	3028		D	Inclusive language review	17.2.0
2021-03	CT#91e	CP-210119	3029		F	Alignment of protection of NAS IEs	17.2.0
2021-03	CT#91e	CP-210119	3030	1	F	S-NSSAI association for 5GSM non-congestion control	17.2.0
2021-03	CT#91e	CP-210107	3033		А	Corrections to congestion control procedure	17.2.0
2021-03	CT#91e	CP-210114	3035	1	А	5GSM back-off mechanisms in PDU session release procedure for SNPN	17.2.0
2021-03	CT#91e	CP-210119	3036	1	F	Correction to the QoS operation error handlings in PDU session establishment procedure	17.2.0
2021-03	CT#91e	CP-210119	3037	1	F	Handling for collision of PDU session handover procedures	17.2.0
2021-03	CT#91e	CP-210119	3038		F	Mapped dedicated EPS bearer without default EPS bearer in the establishment procedure	17.2.0
2021-03	CT#91e	CP-210119	3041	1	F	AN Release triggered by CAG information list in Registration Accept message	17.2.0
2021-03	CT#91e	CP-210119	3042		F	Clarification on CAG-only UE behaviour for emergency PDU session	17.2.0
2021-03	CT#91e	CP-210119	3043	2	F	Clarification on EPS bearer identity handling	17.2.0
2021-03	CT#91e	CP-210119	3044	1	F	Clarification on the handling of QoS flow description without associated QoS rule	17.2.0
2021-03	CT#91e	CP-210119	3045	1	F	Correct a copy error	17.2.0
2021-03	CT#91e	CP-210133	3046		F	Correct the length of IE	17.2.0
2021-03	CT#91e	CP-210119	3048	3	F	Error check and handling for match-all packet filter	17.2.0
2021-03	CT#91e	CP-210119	3049	1	F	Handling of Rejected NSSAI in registration reject message without integrity protection	17.2.0
2021-03	CT#91e	CP-210119	3052		F	Unify terminology about the Authorized QoS rules IE	17.2.0
2021-03	CT#91e	CP-210119	3053		F	PLMN Search at Registered State	17.2.0
2021-03	CT#91e	CP-210119	3054		F	UE behaviour when rejected with #76 via a non-CAG cell	17.2.0
2021-03	CT#91e	CP-210119	3055	1	F	Deregister from emergency registered state as indicated	17.2.0
2021-03	CT#91e	CP-210119	3056		F	Disable N1 mode after change to S1 mode for emergency services	17.2.0
2021-03	CT#91e	CP-210120	3057		F	Clarification on NSSAI inclusion mode	17.2.0
2021-03	CT#91e	CP-210120	3058	1	F	Initiate SMC to provide Selected EPS NAS security algorithms	17.2.0
2021-03	CT#91e	CP-210120	3059	1	F	5GSM cause handling in UE-requested PDU session modification procedure	17.2.0
2021-03	CT#91e	CP-210107	3063	1	А	Local IP address in TFT negotiation in 5GS for 5G-4G interworking	17.2.0
2021-03	CT#91e	CP-210120	3066	1	F	Consistent ngKSI IE name	17.2.0
2021-03	CT#91e	CP-210120	3067	1	F	Correction on semantic errors in QoS operations	17.2.0
2021-03	CT#91e	CP-210120	3068		F	Semantic errors in QoS operations on EPS bearers vs. QoS rules	17.2.0
2021-03	CT#91e	CP-210120	3069		F	Syntactical errors on lack of mandatory parameters	17.2.0
2021-03	CT#91e	CP-210120	3070	1	F	Correction on UE retry restriction for 5GSM causes #50/#51/#57/#58/#61	17.2.0
2021-03	CT#91e	CP-210120	3071		F	Correction on UE retry restriction for 5GSM cause #68	17.2.0
2021-03	CT#91e	CP-210120	3073		F	Rejected NSSAI in registration accept for NSSAA	17.2.0
2021-03	CT#91e	CP-210120	3076	1	F	Prevention of loop scenario for 5GMM cause #62	17.2.0
2021-03	CT#91e	CP-210120	3080	2	F	Clarifications on PLMN and SNPN URSP storage - 24.501 part	17.2.0
2021-03	CT#91e	CP-210120	3082	3	F	Clarification on SNPN UE policy management procedure abnormal handling	17.2.0
2021-03	CT#91e	CP-210120	3087		D	Incorrect reference for NAS security algorithms	17.2.0
2021-03	CT#91e	CP-210120	3088		F	Clarification on NAS security context alignment on 3GPP access and non-3GPP access	17.2.0
2021-03	CT#91e	CP-210120	3089	1	F	Default configured NSSAI for PLMN	17.2.0
2021-03	CT#91e	CP-210120	3090		F	"No suitable cells in tracking area" not applicable to non-3GPP access	17.2.0

2021-03	CT#91e	CP-210120	3091	1	F	The UE behavior when the UE receives the allowed NSSAI	17.2.0
2021-03	CT#91e	CP-210120	3093	1	F	Correction on service area list IEs	17.2.0
2021-03	CT#91e	CP-210114	3097	•	A	T3245 of a UE operating in SNPN access operation mode	1720
2021 00	CT#010	01 210114	0007		11	Correction of misimplementation of CP 2001	17.2.0
2021-04	CT#02o	CD 211121	2106	1	٨	Clarification on MA BDU accords for LADN	17.2.1
2021-00	CT#92e	CP 211131	3190	1	A	Clarification on a PDU session for time synchronization: SSC	17.3.0
2021-00	01#926	CF-211130	3140		F	mode, always-on-ness	17.3.0
2021-06	CT#92e	CP-211136	24.44		-	DS-TT Ethernet port MAC address only sent when the PDU	17.3.0
0004 00	OT#00-	00.044400	3141			session type is Ethernet	47.0.0
2021-06	CT#92e	CP-211136	3142	1		UE-DS-TT residence time used for UE-DE TSC	17.3.0
2021-00	CT#92e	CP-211130	3139	4		Clarification on part management information container	17.3.0
2021-00	CT#920	CP-211130	3100	1	F		17.3.0
2021-00	01#926	CF-211130	3190	1	F	synchronization or TSC	17.3.0
2021-06	CT#92e	CP-211137	3144	1	C	Emergency services in an SNPN	1730
2021-06	CT#92e	CP-211137	0		•	SNN verification for SNPN supporting AAA-Server for primary	17.3.0
_000	0020	0. 20.	3137	1	В	authentication and authorization	
2021-06	CT#92e	CP-211137	3128	1	В	Lists of 5GS forbidden tracking areas	17.3.0
2021-06	CT#92e	CP-211137	3129	1	В	Forbidden SNPNs	17.3.0
2021-06	CT#92e	CP-211137				Enabling selection of an SNPN other than the subscribed	17.3.0
			3130	1	В	SNPN	
2021-06	CT#92e	CP-211137	3205		F	Emergency service fallback and SNPN	17.3.0
2021-06	CT#92e	CP-211137	3206		В	Inter-network mobility	17.3.0
2021-06	CT#92e	CP-211137			_	"List of subscriber data" handling for SNPN supporting AAA-	17.3.0
			3133	3	В	Server for primary authentication and authorization	
2021-06	CT#92e	CP-211137			<u>^</u>	Storage of 5GMM information for UEs in SNPN access	17.3.0
0004 00	OT #00 -	00.011107	3262	1	C	operation mode	47.0.0
2021-06	CT#92e	CP-211137	3259	1	C	The usage of the last visited registered TAI	17.3.0
2021-06	CT#92e	CP-211137	3260	1	В	Support of the default configured NSSAI in the SNPN	17.3.0
2021-06	CT#92e	CP-211137	3204	1	В	Onboarding in SNPN - slicing in initial registration	17.3.0
2021-00	CT#920	CP-211137	2220	1	D	Slice bandling in registration for SNPN onboarding	17.3.0
2021-00	CT#92e	CP-211137	3320	1	B	De-registration for SNPN onboarding registered LIE	17.3.0
2021-00	CT#92e	CP-211137	5521	1	D	DNN/S-NSSAI providing in PDL session establishment for	17.3.0
2021 00	01//020		3322	1	В	SNPN onboarding	11.0.0
2021-06	CT#92e	CP-211137				PVS information providing in PDU session establishment for	17.3.0
			3323	1	В	onboarding	
2021-06	CT#92e	CP-211137	3211	1	В	Onboarding in SNPN - mobility registration update	17.3.0
2021-06	CT#92e	CP-211138	3208	1	В	Support for MA PDU Session with 3GPP access in EPC	17.3.0
2021-06	CT#92e	CP-211138				Indication of UE supporting 3GPP access leg in EPC during	17.3.0
	07/00	0.5.044400	3194	1	F	MA PDU session establishment procedure	
2021-06	C1#92e	CP-211138	2240	4	р	Add target QoS flow capability for access performance	17.3.0
2021-06	CT#92e	CP-211139	3240	1	Б	Handling the paging cause in the LIE and the network for	1730
2021 00	01//020		3124	1	В	MUSIM in 5GS	11.0.0
2021-06	CT#92e	CP-211139	3119	5	В	Definitions and abbreviations for Multi-USIM in 5GS	17.3.0
2021-06	CT#92e	CP-211139	3166	1	В	Remove paging restriction via Registration	17.3.0
2021-06	CT#92e	CP-211139				Considering paging restrictions while paging the UE that is	17.3.0
			3224	1	В	MUSIM capable in 5GS	
2021-06	CT#92e	CP-211139				The MUSIM capable UE shall not initiate Service Request	17.3.0
			2225	1	D	procedure for Leaving the network if Emergency service is	
2021 06	CT#02c	CP_211120	5225	1	ט	Using Service Request procedure for removing paging	1730
2021-00	01#328	01-211139	3226	1	в	restrictions in 5GS for a Multi-USIM LIF	17.3.0
2021-06	CT#92e	CP-211139	3267	3	В	Updates to Registration procedure for MUSIM Leaving in 5GS	17.3.0
2021-06	CT#92e	CP-211139	5201	-	-	Handling of service request when responding to paging with	17.3.0
00	2		3170	2	С	voice service indication	
2021-06	CT#92e	CP-211139	3121	3	В	Multi-USIM UE support indications in 5GS	17.3.0
2021-06	CT#92e	CP-211140	3106	1	В	New cause value for rejected NSSAI	17.3.0
2021-06	CT#92e	CP-211140	3249	1	F	Correction of the definition of the Rejected NSSAI	17.3.0
2021-06	CT#92e	CP-211140	3111	3	В	Introducion of Network Slice Admission Control	17.3.0
2021-06	CT#92e	CP-211140				Clarificaiton on behaviors of the UE and the network	17.3.0
			3112	3	В	supporting Network Slice Admission Control	
2021-06	CT#92e	CP-211140	0.100			S-NSSAI rejected due to maximum number of UEs reached	17.3.0
2024 22	OT#00-	OD 044440	3123	4	В	and BO timer value	17.0.0
2021-06	C1#92e	CP-211140	2010	2	D	Iviaximum number of established PDU sessions already	17.3.0
2021.06	CT#020	CP-211144	3125	1	B	General section for ID 1149	1730
2021-00	CT#020	CP-211141	3218	1	B		17.3.0
2021-00	CT#92e	CP-211141	3244	1	B	UE configuration undate procedure undate for LILIAA	17.3.0
2021-06	CT#920	CP-211141	3138	3	B	Update on Registration procedure for UIIAA-MM	17.3.0
2021-06	CT#92e	CP-211142	3125	1	B	ProSe as a trigger for Service Request procedure	17.3.0
		· · · · · -					

2021-06	CT#92e	CP-211142			_	Network shall not release the RRC connection for ProSe	17.3.0
2021.06	CT#02a	CD 211142	3126	1	B	Services	17 2 0
2021-00	CT#920	CP-211142	2150	1	D	LIE ProSe appability pagetiction with ECC	17.3.0
2021-00	CT#920	CP-211142	3159	2	D	UE ProSe capability hegoliation with SGC	17.3.0
2021-00	CT#920	CP-211142	2160	3		Cerrostion of a mossage name	17.3.0
2021-00	CT#920	CP 211144	3100		Г	Povisions on the description of IEs in Service Request	17.3.0
2021-00	01#926	CF-211144	3174		F	message	17.5.0
2021-06	C1#92e	CP-211144	3181		F	sessions for MA PDU session	17.3.0
2021-06	CT#92e	CP-211144	3182		F	Clarification of MA PDU session handling after network	17.3.0
2021-06	CT#92e	CP-211144	3197		D	Correction on "security control mode procedure	17.3.0
2021-06	CT#92e	CP-211144	3198		F	Update cause of start T3540	17.3.0
2021-06	CT#92e	CP-211144	3212		F	Removal of editor's note on CAG information list in USIM	17.3.0
2021-06	CT#92e	CP-211144	3220		D	Minor corrections	17.3.0
2021-06	CT#92e	CP-211144	3234		F	Conditions for applying 5G-EA0 for the initial NAS message	17.3.0
2021-06	CT#92e	CP-211144				The UE enters the state 5GMM-SERVICE-REQUEST-	17.3.0
		-	3238		F	INITIATED after sending the SERVICE REQUEST message	
2021-06	CT#92e	CP-211144	3253		D	Editorial corrections in TS 24.501	17.3.0
2021-06	CT#92e	CP-211144				Abnormal cases in the PDU session authentication and	17.3.0
			3263		F	authorization procedure	
2021-06	CT#92e	CP-211144	3296		F	Use the latest UE security capability when selecting 5G security algorithms	17.3.0
2021-06	CT#92e	CP-211144	0200		•	null integrity protection algorithm used when UE has an	17.3.0
_000	0020	0	3297		F	emergency PDU session	
2021-06	CT#92e	CP-211144	3300		F	Clarification on the setting of packet filter identifier value	17.3.0
2021-06	CT#92e	CP-211144				Perform slice-independent services when no allowed NSSAI	17.3.0
			3051	2	F	available	
2021-06	CT#92e	CP-211144	3327		F	UL DRB setup collided with DL 5GSM message	17.3.0
2021-06	CT#92e	CP-211145	3325		F	Alignment on UE retry restriction for 5GSM causes	17.3.0
2021-06	CT#92e	CP-211145	3345		F	MNC digit 3 in the CAG information list IF	1730
2021-00	CT#92e	CP-211145	3102	1	F	T3447 value parameter in dICU procedure	17.3.0
2021-00	CT#92e	CP-211145	5152	1	1	CloT nw initiated re-negotiation of any header compression	17.3.0
2021 00	01#020	01 211110	3193	1	F	configuration	11.0.0
2021-06	CT#92e	CP-211145	3294	1	F	NSSAA and de-registration procedures collision	17.3.0
2021-06	CT#92e	CP-211145	3039	2	F	Handling of multiple SM Retry Timer values configured in a UE	17.3.0
2021-06	CT#92e	CP-211145	3344	1	F	Correction in the CAG only UE	17.3.0
2021-06	CT#92e	CP-211145	3278	1	F	Correction to resetting of the registration update counter	17.3.0
2021-06	CT#92e	CP-211145	3246	2	F	UE behavior when the UE receives the Allowed NSSAI	17.3.0
2021-06	CT#92e	CP-211145	3279	1	F	Correction to T3540 handling	17.3.0
2021-06	CT#92e	CP-211145	3280	1	F	Updation of stored pending NSSA for equivalent PLMN(s)	17.3.0
2021-06	CT#92e	CP-211145	3072	3	F	Excluding re-NSSAA for creating pending NSSAI	17.3.0
2021-06	CT#92e	CP-211145				Handling of collisions between UE-requested 5GSM	17.3.0
			2961	2	F	procedures and N1 NAS signalling connection release	
2021-06	CT#92e	CP-211145				Transmission failure handling for NETWORK SLICE-	17.3.0
			3329	1	F	SPECIFIC AUTHENTICATION COMPLETE	
2021-06	CT#92e	CP-211145	3233	1	F	LADN T3396 handling	17.3.0
2021-06	CT#92e	CP-211145	3272	2	F	Forbidding registration area when no slice available	17.3.0
2021-06	CT#92e	CP-211145	3236	2	F	Access barring for access categories '0' and '2' while timer RRC T302 is active	17.3.0
2021-06	CT#92e	CP-211146	3328		F	Correction on UE radio capability update	17.3.0
2021-06	CT#92e	CP-211146	3335		F	Relaxing requirement for NSSAA timing	17.3.0
2021-06	CT#92e	CP-211146	3346		F	AMF decision on the use of a 5G NAS security context	17.3.0
2021-06	CT#92e	CP-211146	3172	1	F	Retransmission timer starting for T3520 with emergency PDU session	17.3.0
2021-06	CT#92e	CP-211146	0475		_	Add EMM SR procedure for non-integrity protected reject	17.3.0
0004.00	OT//00		31/5	1	г	message	47.0.0
2021-06	C1#92e	CP-211146	2913	3	F	initiated in 5GMM-CONNECTED mode with RRC Inactive	17.3.0
2021-06	CT#92e	CP-211146		-		Clarification of Collision of PDU session establishment	17.3.0
			3185	1	F	procedure and network-requested PDU session release	
2021-06	CT#92e	CP-211146	0000		-	Disabling of N1 mode capability after failure in service request	17.3.0
0004.00	OT#22		3230	1	F	procedure triggered due to Emergency Service Fallback	47.0.0
2021-06	C1#92e	CP-211146	2240	1	E	Correcting the INCLES related to changes in some IEI values	17.3.0
2021 06	CT#022	CP-211146	JZ40	1		duiuss leteases	17 2 0
2021-00	01#320	01-211140	3291	1	F	procedure	17.3.0
2021-06	CT#92e	CP-211146	0000		-	UE handling of S-NSSAI when interworking with ePDG and	17.3.0
L		05 04440	3292	1			17.0.0
2021-06	CT#026	(P_{1111})	-10.11	12		Handling of 5(-1/1/1 cause #91	1730

2021-06	CT#92e	CP-211146	3318	1	F	DNN as an optional parameter when interworking with EPS	17.3.0
2021-06	CT#92e	CP-211146				AMF handling when none of the DNN's in LADN Indication IE	17.3.0
			3232	1	F	are part of subscribed DNN list	
2021-06	CT#92e	CP-211146	3239	1	F	ESFB handling in case of network authentication failure	17.3.0
2021-06	CT#92e	CP-211147	3270	1	F	RAT disable when re-attempts are not allowed	17.3.0
2021-06	CT#92e	CP-211147	3305	1	D	Remove duplicated text about semantic error handling	17.3.0
2021-06	CT#92e	CP-211147	3293	1	F	Correction to description of #54	17.3.0
2021-06	CT#92e	CP-211147	3295	1	F	Add a note to reference 24.173	17.3.0
2021-06	CT#92e	CP-211147	3299	1	F	Storage on counters and keys in 5G AKA	17.3.0
2021-06	CT#92e	CP-211147				UE does not delete 5G NAS security context in connected	17.3.0
			3301	1	F	mode	
2021-06	CT#92e	CP-211147				Clarification on CAG information list handling received in	17.3.0
			3302	1	F	HPLMN	
2021-06	CT#92e	CP-211147				Send REGISTRATION COMPLETE message only if the SOR	17.3.0
			3303	1	F	information is received	
2021-06	CT#92e	CP-211147	3187	1	F	Clarification on UE initialted MA PDU deactivation	17.3.0
2021-06	CT#92e	CP-211147	3347	1	F	5GMM procedure updating the default configured NSSAI	17.3.0
2021-06	CT#92e	CP-211147	3330	1	F	Correction to Ciphering key data IE	17.3.0
2021-06	CT#92e	CP-211147	3331	1	F	Registration attempt counter reset when in SNPN	17.3.0
2021-06	CT#92e	CP-211147	3264	1	F	Updating timer table for stopping timer T3540	17.3.0
2021-06	CT#92e	CP-211147				Updating timer talbe for stopping timer T3565 when receiving	17.3.0
			3261	1	F	CONTROL PLANE SERVICE REQUEST message	
2021-06	CT#92e	CP-211147	3265	1	F	The list of NSSAI(s)	17.3.0
2021-06	CT#92e	CP-211147	3221	1	F	Second SMC procedure after RINMR	17.3.0
2021-06	CT#92e	CP-211147	3247	2	F	Correction of Requested NSSAI handling	17.3.0
2021-06	CT#92e	CP-211148	3178		F	Non-3GPP access T3540 timer after service procedure	17.3.0
2021-06	CT#92e	CP-211148	3191	1	F	Non-3GPP access cannot use PLMN-SEARCH state	17.3.0
2021-06	CT#92e	CP-211148	3179	1	F	Non-3GPP access and 5GMM cause 76	17.3.0
2021-06	CT#92e	CP-211148	3050	5	F	MMTEL Voice and MMTEL Video in non-3GPP	17.3.0
2021-06	CT#92e	CP-211149	3145	3	В	Introduction of handling of Edge computing for 5GS	17.3.0
2021-06	CT#92e	CP-211149	2977	7	В	ECS address support indication and provisioning in ePCO	17.3.0
2021-06	CT#92e	CP-211150	3237		D	Various editorial corrections	17.3.0
2021-06	CT#92e	CP-211150	3243	1	F	Incorrect reference in subclause 4.5.4.2	17.3.0
2021-06	CT#92e	CP-211150	3336	1	В	Update Non-3GPP TAI to support N3SLICE	17.3.0
2021-06	CT#92e	CP-211151	3105	1	F	Support of SOR-CMCI	17.3.0
2021-06	CT#92e	CP-211151	3131	1	F	"ME support of SOR-CMCI" indicator	17.3.0
2021-06	CT#92e	CP-211151	3207	1	В	SOR-CMCI transport and usage	17.3.0
2021-06	CT#92e	CP-211152	3143	1	В	5QI for satellite access	17.3.0
2021-06	CT#92e	CP-211152	3101	2	В	New 5GMM cause for satellite access	17.3.0
2021-06	CT#92e	CP-211152	3219	2	В	UE's handling of the indication of country of UE location	17.3.0
2021-06	CT#92e	CP-211152	3348	1	В	PDU session establishment for NR satellite access	17.3.0
2021-06	CT#92e	CP-211152	3100	3	В	MCC list for 5GMM message	17.3.0
2021-06	CT#92e	CP-211153	3115		F	UE handling in case of no valid KAUSF for AKMA	17.3.0
2021-06	CT#92e	CP-211165	3103	4	В	Encoding of API-based DN-AA	17.3.0
2021-06	CT#92e	CP-211311	3258	2	В	Update of registration procedure for SNPN case	17.3.0
2021-06	CT#92e	CP-211312				S-NSSAI providing in UE-requested PDU session	17.3.0
						establishment procedure with "existing PDU session" request	
			2843	10	F	type	
2021-06	CT#92e	CP-211316	3202	3	В	Onboarding in SNPN - initial registration	17.3.0
2021-06	CT#92e	CP-211322				Updates to Service Request for MUSIM Leaving and Reject	17.3.0
			3268	4	В	Paging in 5GS	
2021-06	CT#92e	CP-211325	3215	3	F	Providing wildcard CAG-ID in the USIM	17.3.0
2021-06	CT#92e					Editorial correction in clause 5.3.19a	17.3.1
2021-09	CT#93e	CP-212120	3425	1	A	Enabling storing two 5G NAS security contexts	17.4.0
2021-09	CT#93e	CP-212239	3546	2	А	5G NAS Security Context handling for multiple registrations	17.4.0
2021-09	CT#93e	CP-212160	3521	2	В	Multiple round-trips of AA messages during UUAA-MM	17.4.0
2021-09	CT#93e	CP-212240	3581	-	F	Correction for incorrect CR implementation	17.4.0
2021-09	CT#93e	CP-212263	3250	6	В	C2 pairing authorization at PDU session establishment	17.4.0
2021-09	CT#93e	CP-212264	3251	6	В	C2 pairing authorization at PDU session modification	17.4.0
2021-09	CT#93e	CP-212125	3567	1	F	MINT: Added new registration type for disaster roaming.	17.4.0
2021-09	CT#93e	CP-212125	3512	1	В	Deregister for disaster inbound roaming services	17.4.0
2021-09	CT#93e	CP-212128	3463	-	F	UE-DS-TT residence time defined in 3GPP TS 23.501	17.4.0
2021-09	CT#93e	CP-212128	3464	-	F	Replacement of TS 24.519 with TS 24.539	17.4.0
2021-09	CT#93e	CP-212128				Cleanup limitation about Ethernet DS-TT port and Ethernet	17.4.0
			3538	1	F	type PDU session	
2021-09	CT#93e	CP-212128	3539	1	F	Supporting of TSCTSF	17.4.0
2021-09	CT#93e	CP-212128	3569	1	F	IEEE Std 1588-2019 reference update	17.4.0
2021-09	CT#93e	CP-212129	3558	-	F	Consistent terms on SNPN onboarding	17.4.0
2021-09	CT#93e	CP-212129	3557	1	В	Slice handling for SNPN onboarding	17.4.0
2021-09	CT#93e	CP-212129				NSSAAF : Network slice-specific and SNPN authentication	17.4.0
			3519	1	F	and authorization function	

2021-09	CT#93e	CP-212129	2507	1	E	Handling of AUTHENTICATION REJECT message in ON-	17.4.0
2021-09	CT#93e	CP-212129	5507	1	1	Association of NSSAL UE radio capability ID and back-off	17.4.0
	0	0. 2.2.20				timers for UE supporting access to an SNPN using credentials	
			3448	1	С	from a credentials holder	
2021-09	CT#93e	CP-212129	3446	1	С	No support for eCall over IMS in SNPNs	17.4.0
2021-09	CT#93e	CP-212129			-	UE identity when onboarding in SNPN for which the UE has	17.4.0
2021.00	CT#02o	CP 212120	3392	1	F	5G-GUTT Editor's note on enhanding SLICI derivation	1740
2021-09	CT#93e	CP-212129	2281	1	r	S-NSSAL not provided when registered for onboarding services	17.4.0
2021-05	01#000	01 212125	3390	1	F	in SNPN	17.4.0
2021-09	CT#93e	CP-212129	3389	-	F	Network identifier is not specified	17.4.0
2021-09	CT#93e	CP-212129	3387	1	В	Authentication handling	17.4.0
2021-09	CT#93e	CP-212129				NID of SNPN which assigned 5G-GUTI in mobility registration	17.4.0
			3384	1	F	update	
2021-09	CT#93e	CP-212129	3383	-	F	NID as cleartext IE	17.4.0
2021-09	CT#93e	CP-212129	3382	1	F	PVS PCO parameter providing	17.4.0
2021-09	C1#93e	CP-212129	3377	1	в	Clarification of UE status during registration procedure for	17.4.0
2021-09	CT#93e	CP-212129	3368	1	B	Service request not accepted by an ON-SNPN	1740
2021-09	CT#93e	CP-212130	0000		5	Rename the 5GSM capability of supporting access	17.4.0
_000	0	0. 2.2.00	3410	1	F	performance measurements per QoS flow	
2021-09	CT#93e	CP-212131	3489	2	F	Corrections in Service Request procedure	17.4.0
2021-09	CT#93e	CP-212131	3488	3	В	Reject RAN Paging using Service Request in RRC Inactive	17.4.0
2021-09	CT#93e	CP-212131	3487	-	F	Corrections to Paging restriction IE	17.4.0
2021-09	CT#93e	CP-212131	3476	1	В	NAS leaving to reject RAN paging	17.4.0
2021-09	CT#93e	CP-212131				Using Service Request procedure for removing paging	17.4.0
			2/20		D	Clot 5GS optimization	
2021-09	CT#03e	CP-212131	3439	-	D	Resolving the Editor's note related to Paging Rejection for	1740
2021-03	01#000	01 212101	3438	1	в	MUSIM UE in 5GS	17.4.0
2021-09	CT#93e	CP-212131	3421	2	F	Timer handling for MUSIM UEs (for 24,501)	17.4.0
2021-09	CT#93e	CP-212131	3372	-	D	Editorial corrections to CR#3170	17.4.0
2021-09	CT#93e	CP-212131	3360	1	В	MUSIM features are not applicable for non-3GPP access	17.4.0
2021-09	CT#93e	CP-212131	3354	1	В	T3540 for MUSIM	17.4.0
2021-09	CT#93e	CP-212131	3352	1	В	MUSIM and PEIs	17.4.0
2021-09	CT#93e	CP-212131	3350	1	В	NOTIFICATION RESPONSE message indicating failure	17.4.0
2021-09	CT#93e	CP-212132	3570	1	B	Clarification of the UCU procedure upon completion of NSSAA	17.4.0
2021-09	CT#93e	CP-212132	3561	1	F	Correction on AMF actions on NSAC	17.4.0
2021-09	CT#93e	CP-212132	3560	1	В	NSAC in de-registration procedure	17.4.0
2021-09	CT#93e	CP-212132	3535	1		Clarification on rejected NSSAL term	17.4.0
2021-09	CT#93e	CP-212132	5555	1	1	Stop associated back-off timer when remove S-NSSAI from	17.4.0
2021-05	01#000	01 212102	3534	-	F	rejected NSSAI for the maximum number of UEs reached	17.4.0
2021-09	CT#93e	CP-212132	3517	1	F	Removal of unnecessary ENs	17.4.0
2021-09	CT#93e	CP-212132				Clarify mobility management based on NSAC per access type	17.4.0
			3516	1	В	independently	
2021-09	CT#93e	CP-212142			_	Removal of CR3100r3 (C1-213895) (MCC list for 5GMM	17.4.0
0004.00	07#00-		3361	-	F	message)	47.4.0
2021-09	C1#93e	GP-212132	351F	1	в	Clarinication of mobility management based NSAC for roaming	17.4.0
2021-00	CT#030	CP-212132	3468	1	B	Lase	1740
2021-09	CT#93e	CP-212132	0.100	·	-	Clarification on network behavior when all S-NSSAIs included	17.4.0
00	2		3467	1	F	in the requested NSSAI are rejected	
2021-09	CT#93e	CP-212132	3417	1	F	Update the description of NSAC about SNPN	17.4.0
2021-09	CT#93e	CP-212132	3416	1	С	Resolution of an EN about pre-Rel-17 UE on NSAC	17.4.0
2021-09	CT#93e	CP-212132			L	The exception in Network Slice Admission Control for	17.4.0
0001	07:0-	00.010.00	3415	1	С	Emergency and Priority Services	47.10
2021-09	C1#93e	CP-212142	2540	1	E	Handling of a reject message including 5GMM cause value	17.4.0
2021.00	CT#020	CP-212142	3/1/	1	F	#ro without integrity protection	17/0
2021-09	CT#03e	CP-212142	5414	1	1	5GMM procedures for satellite access for reject cause on LIE	17.4.0
2021-03	0.7000	51 212142	3217	3	В	location - alternative handling	
2021-09	CT#93e	CP-212133	3566	-	F	Consistent term on USS communication	17.4.0
2021-09	CT#93e	CP-212133	3565	1	F	EN resolution on delivering UUAA-MM result via UCU	17.4.0
2021-09	CT#93e	CP-212133	3564	1	В	PDU session establishment reject for UUAA-SM	17.4.0
2021-09	CT#93e	CP-212133	3563	1	В	UAV registered as normal UE	17.4.0
2021-09	CT#93e	CP-212133	3523	1	F	Update of general section for ID_UAS	17.4.0
2021-09	CT#93e	CP-212133	3522	1	В	Multiple round-trip of AA messages during UUAA-SM	17.4.0
2021-09	CT#93e	CP-212133	2500	1	-	Clarification on UE behavior after Registration reject with UAV	17.4.0
2021.00	CT#02c	CP_212122	3/10	1	r B	DDLL session establishment request for LIAS convices	17/0
2021-09	CT#93e	CP-212133	3406	2	В	NW initiated de-registration upon failure of ongoing LILIAA-MM	17.4.0

2021-09	CT#93e	CP-212134	3486	-	F	Add the missing description on ProSe under avoiding double barring	17.4.0
2021-09	CT#93e	CP-212134	3430	1	F	Requesting ProSe resources as a trigger for Service Request	17.4.0
2021-09	CT#93e	CP-212134	3426	1	B	Introducing the Remote UE report procedure	17.4.0
2021-09	CT#93e	CP-212134				Resuming a connection due to ProSe	17.4.0
			3420	1	F	discovery/communication over PC5	
2021-09	CT#93e	CP-212136	3386	1	B	EAS rediscovery	17.4.0
2021-09	CT#93e	CP-212136	3385	1	В	(Re)configuring DNS server addresses	17.4.0
2021-09	C1#93e	CP-212139	3399	1	В	used for removing joined UE from MBS session(s)	17.4.0
2021-09	CT#93e	CP-212139	3395	1	В	Introducing the MBS join and leave procedures	17.4.0
2021-09	C1#93e	CP-212139	3394	1	В	PDU session modification procedure	17.4.0
2021-09	CT#93e	CP-212139	3370	2	В	Multicast join procedure - Alt.1	17.4.0
2021-09	CT#93e	CP-212139	3369	1	С	Paging with TMGI for multicast services	17.4.0
2021-09	CT#93e	CP-212140	3520	1	F	Correction on value of UE radio capability ID deletion	17.4.0
2021-09	CT#93e	CP-212140	3527	1	F	Sending P-CSCF address(es)	17.4.0
2021-09	CT#93e	CP-212140	3457	-	F	Incorrect reference in subclause 6.2.16	17.4.0
2021-09	CT#93e	CP-212140	3453	-	D	Editorial corrections	17.4.0
2021-09	CT#93e	CP-212140				Add handling of 5GMM cause #76 when UE does not have	17.4.0
			3423	-	F	any stored CAG information list	
2021-09	CT#93e	CP-212141	3525	-	D	Corrected S-NSSAI SST for SOR-CMCI	17.4.0
2021-09	CT#93e	CP-212141	3365	1	В	Alignments for the introduction of SOR-CMCI	17.4.0
2021-09	CT#93e	CP-212143	0450		-	Resolving the Editor's note in AKMA procedure related to	17.4.0
2021.00	CT#02o	CP 212152	3452	-	F	K_AUSE change after 5G AKA based primary authentication	17 4 0
2021-09	CT#93e	CP-212152	3403	-	U	Non-IP MTLL request in the PDLL session establishment	17.4.0
2021-03	01#336	01-212132	3482	-	F	procedure	17.4.0
2021-09	CT#93e	CP-212152	3478	-	F	Handling of multiple S-TAGs in the Ethernet header	17.4.0
2021-09	CT#93e	CP-212152	3477	-	F	Clarification of destination and source MAC addresses	17.4.0
2021-09	CT#93e	CP-212152	3475	-	F	Add a missing message to relax SM congestion control	17.4.0
2021-09	CT#93e	CP-212152	3470	-	D	Editorial corrections	17.4.0
2021-09	CT#93e	CP-212152	3460	-	F F	Clarification of NSAAA abnormal failure handling	17.4.0
2021-09	CT#93e	CP-212152	3458	-		Correction to CPSR handling in AMF	17.4.0
2021-09	CT#93e	CP-212152	3450	-	F	Clarification to collision of PDL sessions release procedures	17.4.0
2021-03	CT#93e	CP-212152	3432	-	D	Editorial corrections	17.4.0
2021-09	CT#93e	CP-212152	3422	_	F	Fix inconsistent QoS handling for network-requested PDU	17.4.0
2021-09	CT#93e	CP-212152	3411	-	F	AME provides the CAG information list	17.4.0
2021-09	CT#93e	CP-212152	3409	-	F	The incorrectly placed NOTE in QoS rule	17.4.0
2021-09	CT#93e	CP-212153	3542	-	F	5GSM state transition of MA PDU session	17.4.0
2021-09	CT#93e	CP-212153	3514	-	F	Asignment of IEI values	17.4.0
2021-09	CT#93e	CP-212153	3505	-	F	ANDSP not specified for a UE operating in SNPN access operation mode	17.4.0
2021-09	CT#93e	CP-212153				Processing Authentication Reject only if timer T3516 or T3520	17.4.0
			3502	1	F	is running	
2021-09	CT#93e	CP-212153	3495	1	F	UE changing from N1 mode to S1 mode	17.4.0
2021-09	CT#93e	CP-212153	3494	-	F	Deleting forbidden PLMNs list when UE is switched off	17.4.0
2021-09	CT#022	CP-212153	34/4	1	Г С	Aligh deregistration #62 with Initial registration reject	17.4.0
2021-09	CT#93e	CP-212153	3473	1	Г Г	Clarify on T3245 in each specific procedure	17.4.0
2021-09	CT#93e	CP-212153	3471	1	D	Remove duplicated MCC	17.4.0
2021-09	CT#93e	CP-212154	3568	1	Б Г	Handling redirection cause #31 for UE does not support S1	17.4.0
2021-09	CT#93e	CP-212154	3550	1	F	Rejected S-NSSAL update	1740
2021-09	CT#93e	CP-212154	3549	1	F	Correction on N5GC indication IE Format	17.4.0
2021-09	CT#93e	CP-212154		-		Not start T3540 if 5GMM cause is considered as abnormal	17.4.0
			3544	1	F	cases	
2021-09	CT#93e	CP-212154	3531	1	F	Include UE radio capability ID deletion indication IE and UE radio capability ID IE simultaneously	17.4.0
2021-09	CT#93e	CP-212154	3506	1	F	Handling two available native 5G-GUTIs during the registration procedure	17.4.0
2021-09	CT#93e	CP-212154	3504	1	F	Maximum number of S-NSSAIs in an NSSAI	17.4.0
2021-09	CT#93e	CP-212154	3481	1	F	Clarification of PF for Ethernet PDU session	17.4.0
2021-09	CT#93e	CP-212154	3427	1	F	Trigger PDU SESSION MODIFICATION for deletion of mapped EPS to ensure sync with network	17.4.0
2021-09	CT#93e	CP-212154	3408	1	F	Support MAC address range in packet filter	17.4.0
2021-09	CT#93e	CP-212154	3168	2	F	24.501 Redirect with MPS	17.4.0
2021-09	CT#93e	CP-212155	3555	1	F	Correction on UE parameters update data set type	17.4.0

2021	1-09	CT#93e	CP-212155	2554	1	F	Correction on indication of support of local address in TFT in	17.4.0
2021	1-09	CT#93e	CP-212155	3553	1	F	S1 mode Correction on LIE error handling on OoS operations	1740
2021	1-09	CT#93e	CP-212155	3503	2	F	SMC after Primary Authentication	17.4.0
2021	1-09	CT#93e	CP-212155	0000	-		Discarding the content of the container for SOR when the	17.4.0
				3493	1	F	security check about the container is failed.	
2021	1-09	CT#93e	CP-212155	2400	4	_	Simplification of description about rejected NSSAI and	17.4.0
2021	1.00	CT#02o	CD 212155	3469	1		correction of requested NSSAI handling	1740
2021	1-09	CT#93e	CP-212155	5401		1	Resuming the RRC connection upon requesting resources for	17.4.0
2021		01//000	01 212100	3449	1	F	V2X communication over PC5	
2021	1-09	CT#93e	CP-212155				Clarification that IP and Ethernet packets can be delivered	17.4.0
		07/00		3443	1	F	over Control Plane	
2021	1-09	CT#93e	CP-212155	3435	1		Uplink data status IE inclusion criteria clarification	17.4.0
2021	1-09	CT#93e	CP-212155	3434		Г	Correction on providing Selected EPS NAS security algorithms	17.4.0
2021	1 00	011/000	01 212100	3433	1	F	in SMC	17.4.0
2021	1-09	CT#93e	CP-212155	3359	1	D	Correction of MA PDU session network upgrade is allowed	17.4.0
2021	1-09	CT#93e	CP-212155	3326	3	F	Registered for emergency services due to CAG restrictions	17.4.0
2021	1-09	CT#93e	CP-212155			_	Clarification of reactivation requested in PDU SESSION	17.4.0
2021	1.00	CT#02o	CD 212156	3183	3	F	MODIFICATION COMMAND for an MA PDU session	17.4.0
2021	1-09	01#956	GF-212130	3480	-	F	access indicator	17.4.0
2021	1-09	CT#93e	CP-212156			-	Handling of N1 mode capability for non-3GPP access for voice	17.4.0
				3279	1	F	domain selection	
2021	1-09	CT#93e	CP-212156	2404		_	UE identity in NAS signalling connection establishment over	17.4.0
2021	1-09	CT#93e	CP-21215	3401	-	F	SMSoIP triggering mo-SMS establishment cause for non-	1740
2021		01//000	01 21210	3400	1	F	3GPP access	
2021	1-09	CT#93e					Editorial corrections and IEIs values added in tables 8.3.1.1.1,	17.4.1
							8.3.2.1.1, 8.3.7.1.1 and 8.3.9.1.1	
2021	1-12	CT#94e	CP-213211	3803	2	B	Paging Early Indication with Paging Subgrouping Assistance	17.5.0
2021	1-12	CT#94e	CP-213252	3786	2	В	Paging Subgrouping	17.5.0
2021	1-12	01#946	CF-213255	3780	2	F	ongoing 5G-MO-LR	17.5.0
2021	1-12	CT#94e	CP-213027				Signalling support for UPIP for UEs not supporting standalone	17.5.0
0004	1.40	OT#04a	00.010000	3445	1	A	NR connected to 5GCN	4750
2021	1-12 1-12	CT#94e	CP-213030	3701	1	В	Introduction of EPS-UPIP support indication in 5GC	17.5.0
2021	1-12	01#346	01-213030	3796	1	F	triagered by UE POLICY PROVISIONING REQUEST	17.5.0
2021	1-12	CT#94e	CP-213030	3800	-	D	Miscellaneous corrections	17.5.0
2021	1-12	CT#94e	CP-213030				Remove PLMN from the extension of the forbidden PLMNs list	17.5.0
		07/04	0.0.00000	3811	-	F	upon T3247 expiry	
2021	1-12	CT#94e	CP-213032	3688	1	В	Add requirements to support NR RedCap devices	17.5.0
2021	1-12	CT#94e	CP-213032	3695		F	From bandling for ORL and OFL set to zero by the network	17.5.0
2021	1-12	CT#94e	CP-213034	0000				
2021	1-12	CT#94e		3696	-	F	DNS server security information UE capability	17.5.0
		00.0	CP-213034	3696	-	F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of	17.5.0 17.5.0
2021		0010	CP-213034	3696 3700	-	F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message	17.5.0 17.5.0
2021	1-12	CT#94e	CP-213034 CP-213034	3696 3700 3707	- - 2	F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency	17.5.0 17.5.0 17.5.0
2004	1-12 1-12	CT#94e CT#94e	CP-213034 CP-213034 CP-213034	3696 3700 3707 3715	- 2 -	F F D	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections	17.5.0 17.5.0 17.5.0 17.5.0
2021	1-12 1-12 1-12	CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3725	- 2 - -	F F D D	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021	I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738	- 2 - - -	F F D D F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739	- 2 - - - - -	F F D D F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753	- 2 - - - - - -	F F D D F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753	- 2 - - - - - - -	F F D D F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3715 3731 3735 3738 3738 3739 3753 3755	- 2 - - - - - - -	F F D D F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3769	- 2 - - - - - - - - -	F F D D F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range 24 501	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774	- 2 - - - - - - - - - - - - - - - - - -	F F D F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range	17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753 3753 3755 3761 3768 3774 3806	- 2 - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806 3814 3806	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3755 3761 3768 3774 3806 3814 3814	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SoR in SNPN - procedures and coding Eprohidden lists when an entry of the list of subscriber data in	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806 3814 3584 3584 3393	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SoR in SNPN - procedures and coding Forbidden lists when an entry of the list of subscriber data is updated or UICC containing USIM is removed	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3755 3761 3765 3761 3768 3774 3806 3814 3884 3584 3393 3586	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SOR in SNPN - procedures and coding Forbidden lists when an entry of the list of subscriber data is updated or UICC containing USIM is removed NSSAI when registered for onboarding services in SNPN	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806 3814 3584 3584 3393 3586 3625	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SOR in SNPN - procedures and coding Forbidden lists when an entry of the list of subscriber data is updated or UICC containing USIM is removed NSSAI when registered for onboarding services in SNPN SMF selection for SNPN onboarding	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037 CP-213037 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806 3814 3584 3393 3586 3625 3625 3625	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SoR in SNPN - procedures and coding Forbidden lists when an entry of the list of subscriber data is updated or UICC containing USIM is removed NSSAI when registered for onboarding services in SNPN SMF selection for SNPN onboarding De-registration for onboarding registered UE	17.5.0 17.5.0
2021 2021 2021 2021 2021 2021 2021 2021	I-12 I-12 I-12 I-12 I-12 I-12 I-12 I-12	CT#94e CT#94e	CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213034 CP-213037 CP-213037 CP-213037 CP-213037 CP-213037 CP-213037 CP-213037 CP-213037	3696 3700 3707 3715 3731 3735 3738 3739 3753 3755 3761 3768 3774 3806 3814 3584 3584 3584 3586 3625 3632 3632 3632	- 2 - - - - - - - - - - - - - - - - - -	F F D D F F F F F F F F F F F F F	DNS server security information UE capability Correction for the protection of initial NAS messages in case of CPSR message Abort deregistration for emergency Minor corrections Add abbreviations of MPS and MCS Clarification of N1 mode capability UE handling upon receipt of 5GSM cause #33 Access type of the PDU session when re-activation failure Correction to item code UE re-initiate initial registration procedure if authentication procedure is failed Correction to 5GSM cause value list Resolution of an EN about CAG-ID range-24.501 Clarification on destination and source MAC address range Correction of PDU sessions release procedures Align mapping of SMS over IP & SMS over NAS during double barring CP SoR in SNPN - procedures and coding Forbidden lists when an entry of the list of subscriber data is updated or UICC containing USIM is removed NSSAI when registered for onboarding services in SNPN SMF selection for SNPN onboarding De-registration for onboarding registered UE AMF Onboarding Configuration Data as indication ragistered for SNPN onboarding	17.5.0 17.5.0

2021-12	CT#94e	CP-213037	3674		F	T3510 expiry for the UE registered for onboarding services in SNPN	17.5.0
2021-12	CT#94e	CP-213037	007 1			Kausf generation when AAA server of CH is EAP server of	17.5.0
-						EAP based primary authentication and key agreement	
			3675	4	F	procedure	
2021-12	CT#94e	CP-213037	3748	1	F	DNN and S-NSSAI used for onboarding in PLMN	17.5.0
2021-12	CT#94e	CP-213037	3750	1	F	SMF configuring to UE with PVS address	17.5.0
2021-12	CT#94e	CP-213037	3751	1	F	Secondary authentication/authorization by a DN-AAA server	17.5.0
2021-12	C1#94e	CP-213038	0770	4	F	Updating ATSSS parameter update with network-requested	17.5.0
2021-12	CT#0/o	CP-213030	3580	1	г В	PDU Session modification Paging timing collision control support in 5GS	1750
2021-12	CT#94e	CP-213039	3590	-	B	Reject RAN paging with optional paging restrictions	17.5.0
2021-12	CT#94e	CP-213039	3591	1	F	Clarification on removal of paging restrictions	17.5.0
2021-12	CT#94e	CP-213039	3601	1	C	Always-on PDU sessions for MUSIM UE	17.5.0
2021-12	CT#94e	CP-213039				Service request procedure due to MUSIM when no allowed	17.5.0
			3619	1	F	NSSAI is available	
2021-12	CT#94e	CP-213039			_	Service request procedure for NAS connection release when	17.5.0
	0 = 10	0.0.0000	3622	1	F	T3346 is running (for 24.501)	
2021-12	C1#94e	CP-213039	2000	4	-	NAS connection release from 5GMM-CONNECTED mode with	17.5.0
2021 12	CT#04a	CD 212020	3023	1	Г	RRC Indication	1750
2021-12	CT#94e	CP-213039	3667	-	D	5GS MUSIN SR transmission failule	17.5.0
2021-12	CT#94e	CP-213040	3402	-	B	MUSIM capability pegotiation in 5GCN	17.5.0
2021-12	CT#94e	CP-213040	3722	1	F	T3447 handling for MUSIM capable UF	17.5.0
2021-12	CT#94e	CP-213040	0	•		MRU procedure for allocation of 5G-GUTI when T3346 is	17.5.0
			3727	1	F	running	
2021-12	CT#94e	CP-213040	3766	1	F	Remove duplicated MUSIM definition	17.5.0
2021-12	CT#94e	CP-213040				Resolve EN on SR for rejecting RAN paging in 5GMM-	17.5.0
			3767	1	F	CONNECTED mode with RRC inactive	
2021-12	CT#94e	CP-213040	3773	1	C	Remove paging restrictions in case of service area restrictions	17.5.0
2021-12	CT#94e	CP-213040	3794	-	F	Only Paging for voice service	17.5.0
2021-12	C1#94e	CP-213040	3807	1	F	SGS	17.5.0
2021-12	CT#94e	CP-213040		•		Releasing NAS signalling connection and Paging restriction	17.5.0
-						during mobility registration in a TA outside the current	
			3808	1	F	Registration Area for MUSIM UE in 5GS	
2021-12	CT#94e	CP-213040			-	Network to accept or reject the paging restriction requested by	17.5.0
0004 40	CT#04a	00.0400.44	3809	1	В	MUSIM capable UE in 5GS	4750
2021-12	CT#94e	CP-213041	3349	4	Б	Add rejected page for max LE reached for #62	17.5.0
2021-12	CT#94e	CP-213041	3617	1	F	Clarification of the timer T3526	17.5.0
2021-12	CT#94e	CP-213041	5017	1	1	Removing the rejected S-NSSAI for the maximum number of	17.5.0
2021 12	01//010	01 210011	3618	1	F	UEs reached in case of IWK with EPC	11.0.0
2021-12	CT#94e	CP-213041	3620	1	F	Support of NSAC and interworking with EPC	17.5.0
2021-12	CT#94e	CP-213041	3621	1	F	Skip NSAC for existing PDU session request type	17.5.0
2021-12	CT#94e	CP-213041	3626	1	F	NSAC for legacy UEs	17.5.0
2021-12	CT#94e	CP-213041			_	Removal of S-NSSAI from rejected NSSAI for the maximum	17.5.0
0004 40	07/04-	00.0400.44	3627	1		number of UEs reached	47.5.0
2021-12	CT#94e	CP-213041	3640	2	F	AME handling of NSAC function for legacy UE.	17.5.0
2021-12	01#946	06-213041	3655	1	F	UEs reached with value 0 back-off timer	17.5.0
2021-12	CT#94e	CP-213041	3673	1	F	5GSM procedure when EAC is disabled	17.5.0
2021-12	CT#94e	CP-213041		1		Correction of the rejected NSSAI for the maximum number of	17.5.0
			3692	1	F	UEs reached handling	
2021-12	CT#94e	CP-213041	3698	-	F	Correction on SM based NSAC	17.5.0
2021-12	CT#94e	CP-213041			_	Clarification of registration procedure in which NSAC is	17.5.0
0001.17	OT "2 -		3724	-	F F	performed	47.5.0
2021-12	CT#94e	CP-213041	3/44	1		Clarification on SMF performing NSAC for an MA PDU session	17.5.0
2021-12	CT#946	CP-213041	3/03			Local limer for AIVIF to update rejected NSSAI	17.5.0
2021-12	CT#940	CP-213042	3628	- 2	n ⁻	Common IE for C2 authorization	17.5.0
2021-12	CT#94e	CP-213043	0020	<u> </u>	5	5GSM cause value of PDU session establishment reject for	17.5.0
	0.040	2. 2.00-0	3629	3	В	UAS services	
2021-12	CT#94e	CP-213043	3630	1	F	Missed CAA-Level UAV ID for C2 authorization	17.5.0
2021-12	CT#94e	CP-213043	3631	1	F	Miscellaneous corrections on Service-level-AA container IE	17.5.0
2021-12	CT#94e	CP-213043				Update the general part for Authentication and authorization of	17.5.0
			3637	1	С	UAV	
2021-12	CT#94e	CP-213043	3638	1	В	UUAA revocation for the case of UUAA-MM	17.5.0
2021-12	CT#94e	CP-213043	3641	<u>う</u>	Б	Using Service level AA container for CO with structure	17.5.0
2021-12	CT#946	CP-213043	3045	ა 1		Sing Service-level AA container for C2 authorization	17.5.U
2021-12	CT#940	CP-213043	3640	1	F	Restriction to non3app access	17.5.0
2021-12	CT#94e	CP-213043	3651	1	C	UUAA-SM procedure for re-authorization or re-authentication	17.5.0
					-		

2021-12	CT#94e	CP-213043	2052	4	-	Remove EN on payload differentiation of Service-level-AA	17.5.0
2021-12	CT#0/o	CP-2130/3	3658	3	F B	container SM request while LILIAA-SM is opgoing	1750
2021-12	CT#946	CP-213043	3770	3	B	Requirements related to LIAS subscription change	17.5.0
2021-12	CT#94e	CP-213043	3789	1	B	Collision between UUAA-MM and UE initiated deregistration	17.5.0
2021-12	CT#94e	CP-213043	0.00	•		Collision between UUAA-SM and UE requested PDU session	17.5.0
-			3790	1	В	release	
2021-12	CT#94e	CP-213043	3791	1	В	UUAA abnormal case	17.5.0
2021-12	CT#94e	CP-213045				Add the SMF shall provide the QoS flow description(s) for the	17.5.0
	0	0.0.00.40	3595	1	В	PDU sessions used for relaying	
2021-12	C1#94e	CP-213045	2040	4	-	NAS signalling recovery from fallback when the UE was only	17.5.0
2021-12	CT#0/o	CP-213045	3010	1	Г	Editorial corrections for the ProSe relay terminologies and	1750
2021-12	01#346	01-213043	3644	-	D	capabilities	17.5.0
2021-12	CT#94e	CP-213045			-	Triggering Service Request procedure due to lower layers	17.5.0
			3683	1	В	request for ProSe layer-2 UE-to-network relay	
2021-12	CT#94e	CP-213045				IPv6 prefix delegation via DHCPv6 for 5G ProSe layer-3 UE-	17.5.0
			3775	1	В	to-network relay	
2021-12	CT#94e	CP-213045	2776	1	-	The type of the port number in Remote UE context list	17.5.0
2021 12	CT#04o	CP 212045	3//0	1	Г	Moraing LIE triagoned V2X and ProSe policy provision	1750
2021-12	01#946	CF-213043	3782	1	F	procedure in UAC	17.5.0
2021-12	CT#94e	CP-213045	0.02	•		Correcting the reference of the spec in which the UE requests	17.5.0
-			3812	-	F	the PCF to provide ProSeP	
2021-12	CT#94e	CP-213047	3690	1	F	SNPN for NSSAI inclusion mode	17.5.0
2021-12	CT#94e	CP-213047	3725	1	F	Correction in mobility registration reject	17.5.0
2021-12	CT#94e	CP-213047	0700		-	Clarification on when a 5GSM procedure can be initiated for	17.5.0
0004 40	CT#04a	00.0400.47	3729	1	F	LADN DDU seesier medification often inter system shanne inter	4750
2021-12	C1#94e	CP-213047	3732	1	F	PDU session modification after inter-system change into a	17.5.0
2021-12	CT#94e	CP-213047	3733	1	F	The order of PDU sessions to be transferred to FPS	17.5.0
2021-12	CT#94e	CP-213047	0.00	•		Collision handling of UE-requested PDU session	17.5.0
						establishment procedure and network-requested PDU session	
			3734	1	F	modification procedure	
2021-12	CT#94e	CP-213047	0700		-	PDU session type required in PDU session establishment	17.5.0
2021 12	CT#04o	CP 212047	3/30	1	F	request	1750
2021-12	01#346	01-213047	3741	1	F	PDU session modification	17.5.0
2021-12	CT#94e	CP-213047	3756	1	F	Reattempting LADN DNN rejected with #46	17.5.0
2021-12	CT#94e	CP-213047	3757	1	F	UE behaviour on #29 related back-off timer	17.5.0
2021-12	CT#94e	CP-213047	3760	1	F	Add missing 5GSM cause values	17.5.0
2021-12	CT#94e	CP-213047	3762	1	F	Correction on unidentifiable example for syntactical error	17.5.0
2021-12	CT#94e	CP-213047	3772	1	F	Reservation of a bit in an entry of the CAG information list IE	17.5.0
2021-12	CT#94e	CP-213047	3805	1	F	Clarification for UE parameters update data handling	17.5.0
2021-12	CT#94e	CP-213047	3813	1	D	Reference correction - Editorial	17.5.0
2021-12	CT#94e	CP-213047	3023	-	D	Correction to condition to include the 5GS registration result IE	17.5.0
2021-12	01#040	01 210047	3831	-	F	in the REGISTRATION ACCEPT message	17.0.0
2021-12	CT#94e	CP-213047				Handling of the non-current 5G NAS security context at inter-	17.5.0
			3706	1	F	system change from N1 mode to S1 mode	
2021-12	CT#94e	CP-213048			-	The solution to the case the allowed CAG IDs of a PLMN	17.5.0
2024 42	OT#04-	CD 040040	3412	3		beyond the limit of one Entry-1	17 5 0
2021-12	CT#946	CP-213048	344/	∠ 1	F	Authentication failure when emergency service is ongoing	17.5.0
2021-12	CT#94e	CP-213046	5/10	1	1	Delete the PCO parameters after handover between 3GPP	17.5.0
2021012	0.000	0. 2.0040	3737	1	F	and non-3GPP access	
2021-12	CT#94e	CP-213048	3752	1	F	Acknowledgment for the security packet of SOR information	17.5.0
2021-12	CT#94e	CP-213048	3754	3	F	Clarification on semantic error about match-all packet filter	17.5.0
2021-12	CT#94e	CP-213048	3795	1	В	Derived QoS for UDP encapsulated IPsec packets	17.5.0
2021-12	CT#94e	CP-213048	3713	1	F	Storing Allowed NSSAI for EPLMNs	17.5.0
2021-12	CT#94e	CP-213048	3/14	1		Deleting rejected S-NSSAI	17.5.0
2021-12	CT#94e	CP-213048	3/12	<u>ک</u>	F	S-INSSAL WITH HOR-Standard Values	17.5.0
2021-12	CT#946	CP-213049	3654	1	B	Undate on FCS configuration information	17.5.0
2021-12	CT#94e	CP-213050	3681	1	F	ECS configuration information provisioning corrections	17.5.0
2021-12	CT#94e	CP-213053		-		Updating MBS service area for the MBS session that the UE	17.5.0
			3660	1	В	has joined	
2021-12	CT#94e	CP-213053				The MBS service area received in PDU SESSION	17.5.0
			2004		_	ESTABLISHMENT ACCEPT message can include both of	
2021 12	CT#04c	CP-212052	3662	-	Г R	IVIDS TALIIST and INK UGI IIST	1750
2021-12	CT#946	CP-213053	3663	1	F	Aligning the MBS procedures across different clauses	17.5.0
2021-12	CT#94e	CP-213053	3709	1	В	MBS operation in Requested MBS container IF	17.5.0

2021-12	CT#94e	CP-213053	3777	1	F	Correction to type of MBS session ID source specific IP multicast address	17.5.0
2021-12	CT#94e	CP-213053	3815	_	B	The impact of the De-registration procedure on the MBS	17.5.0
2021-12	CT#94e	CP-213053	2916	-	B	SMF to consider the UE as removed from the associated MBS	17.5.0
2021-12	CT#94e	CP-213053	3817	1	в	Removing the joined UE from MBS session due to becoming outside an undated MBS service area	17.5.0
2021-12	CT#94e	CP-213053	0010	1	5	Removing joined UE from MBS sessions at inter-system	17.5.0
0004 40	07/04-	00.040054	3818	-	В	change from N1 mode to S1 mode	47 5 0
2021-12	CT#94e	CP-213054	3670	1	В	Introduction of redundant PDU sessions	17.5.0
2021-12	CT#94e	CP-213054	3671	3	В	5GSM protocol update for redundant PDU sessions	17.5.0
2021-12	CT#94e	CP-213055	3437	6	В	Provisioning of parameters for disaster roaming in the UE	17.5.0
2021-12	C1#94e	CP-213055	3585	4	В	PLMN with disaster condition	17.5.0
2021-12	C1#94e	CP-213055			_	Addition of 5GS registration type for initial registration disaster	17.5.0
	07/0	0.0.00.00	3614	3	В	roaming.	1
2021-12	C1#94e	CP-213055	3642	-	F	Correction of implementation errors of CR3512 (C1-215139)	17.5.0
2021-12	CT#94e	CP-213055			_	Registration result IE indicate UE is registered for disaster	17.5.0
	07.004		3656	4	F	roaming service	
2021-12	C1#94e	CP-213055	3679	1	В	Introducing access identity 3 for disaster roamer	17.5.0
2021-12	CT#94e	CP-213055			_	Unnecessary signalling for providing selected EPS NAS	17.5.0
			3742	-	В	security algorithms to disaster roaming UEs	
2021-12	CT#94e	CP-213055	3820	1	В	Updating initial registration for disaster roaming services	17.5.0
2021-12	CT#94e	CP-213055	3825	1	В	Adding definition for registered for disaster roaming services	17.5.0
2021-12	CT#94e	CP-213056	3669	1	F	SOR-CMCI storage	17.5.0
2021-12	CT#94e	CP-213056	3702	-	В	Adding the SOR security check criterion to the SOR-CMCI	17.5.0
2021-12	CT#94e	CP-213056				UE performing deregistration procedure in 5GMM-	17.5.0
			3797	1	F	CONNECTED mode	
2021-12	CT#94e	CP-213057	3431	4	В	Validity of cause code #78	17.5.0
2021-12	CT#94e	CP-213057	3606	1	В	Alignment to KI#2 conclusions on EPLMN list	17.5.0
2021-12	CT#94e	CP-213057	3636	-	F	Access Technology Identifier satellite NG-RAN	17.5.0
2021-12	CT#94e	CP-213057	3833	1	С	Clarification of UE location verification in registration procedure	17.5.0
202203	CT#95e	CP-220022	3981	2	В	Failure case for 5G SRVCC	17.6.0
2022-03	CT#95e	CP-220147				Handling of forbidden TAI(s) within broadcast TACs in	17.6.0
			4131	-	В	registration procedure	
2022-03	CT#95e	CP-220213	3711	4	F	Paging restrictions with Connection Release in 5GS	17.6.0
2022-03	CT#95e	CP-220233				MUSIM capabilities exchange while Emergency service is	17.6.0
			3948	2	F	ongoing in 5GS	
2022-03	CT#95e	CP-220234				Adding the paging timing collision control in the definition of	17.6.0
			4077	2	F	the MUSIM UE in 5GS	
2022-03	CT#95e	CP-220250	3850	2	В	MUSIM support for SNPN access mode	17.6.0
2022-03	CT#95e	CP-220317	3868	2	В	UUAA-MM completion alignment	17.6.0
2022-03	CT#95e	CP-220318	3867	2	В	UUAA revocation alignment	17.6.0
2022-03	CT#95e	CP-220319	3765	5	В	UAS security information obtained during UUAA	17.6.0
2022-03	CT#95e	CP-220337	3916	2	F	General description on Multi-USIM UE in 5GS	17.6.0
2022-03	CT#95e	CP-220229	3984	1	A	RID for SNPN UEs	17.6.0
2022-03	CT#95e	CP-220229	3985	2	Δ	RID update for SNPN LIEs	17.6.0
2022-03	CT#95e	CP-220220	4002	1	Δ	NSSAA applicable for SNPN in Rel-17	17.6.0
2022-03	CT#050	CP-220223	38/0	- _	F	SUPLive of onboarding SUPL	17.6.0
2022-03	CT#95e	CP_220230	3049	1	F	Clarification on lists of forbidden SNDNs in an ON SNDN	17.0.0
2022-03	CT#958	CD 220230	2071	1	Г С	Handling of 5GMM cause values #2 and #6 in an ON SNDN	17.0.0
2022-03	CT#050	CP 220230	2072	1		Handling of 5CMM cause values #3 and #6 in an ON-SNPN	17.0.0
2022-03	CT#05-	CP 220230	3013	1		Handling of non-integrity protected reject researce with	17.0.0
2022-03	U1#956	GP-220236	2074		E	Francing of non-integrity protected reject messages with	17.0.0
2022.02	CT#05 c	CD 220220	30/4	<u> -</u>	1	ULE operation offer adding the ID of an ON SNDN in the	1760
2022-03	C1#95e	CP-220236	2017	4	-	UE operation after adding the ID of an ON-SNPN in the	17.6.0
0000.00	07/05		3917	1	F	permanentiy forbidden SiNPINS list	47.0.0
2022-03	U1#95e	GP-220236	2010	1	E	nationing of a non-integrity protected AUTHENTICATION	17.6.0
0000.00	OT#05 -		3910	1	Г Г	REJECT message received normal ON-SNPN	47.0.0
2022-03	01#956	CP-220236	3925	-		Contection in Scivily cause value #93 nandling	17.0.0
2022-03	01#956	CP-220236	3933	1	F		17.6.0
2022-03	CT#95e	CP-220237			-	I wo available native 5G-GUTIs during the registration	17.6.0
0000	OT "	00.000	3844	1		procedure not applicable in SNPN	47.0.0
2022-03	C1#95e	CP-220237	3845	1	F -	NID IE Inclusion condition	17.6.0
2022-03	CT#95e	CP-220237	3846	1	F	3GPP PS data off and UE in non-subscribed SNPN	17.6.0
2022-03	CT#95e	CP-220237	3847	1	В	Anonymous SUCI	17.6.0
2022-03	CT#95e	CP-220237	3905	1	F	NSAC for S-NSSAI used for onboarding services in SNPN	17.6.0
2022-03	CT#95e	CP-220237	3923	1	В	Onboarding indication over N11 in an ON-SNPN	17.6.0
2022-03	CT#95e	CP-220237	3966	1	F	Onboarding DNN/S-NSSAI	17.6.0
2022-03	CT#95e	CP-220237				Providing PVS information for obtaining credentials for NSSAA	17.6.0
						or PDU session authentication and authorization procedure in	
			3986	-	F	SNPN	

2022-03	CT#95e	CP-220237				Correction in the UE behavior upon receipt of a	17.6.0
			3924	1	F	value #75 from an ON-SNPN	
2022-03	CT#95e	CP-220238	3839	4	B	Enabling update of SOR-SNPN-SLin a PLMN	17.6.0
2022-03	CT#95e	CP-220238			_	Usage of indication to use MSK for derivation of KAUSF after	17.6.0
						success of primary authentication and key agreement	
			3843	2	В	procedure	
2022-03	CT#95e	CP-220238	3934	3	F	NSAC for SNPN onboarding	17.6.0
2022-03	CT#95e	CP-220238	3987	1	F	UE requesting PVS information	17.6.0
2022-03	CT#95e	CP-220238	2000	1	Р	Providing PVS addresses for obtaining SO-SNPN credentials	17.6.0
2022.02	CT#05o	CD 220229	3988	1	В	EAD TTLS with two phases of authentiation	1760
2022-03	CT#95e	CP-220238	4019	1	F	Congestion control for ophoarding in SNPN	17.0.0
2022-03	CT#95e	CP-220238	4013	1	1	Clarification of the LIE behavior for the pending NSSAI for the	17.0.0
2022 00	01//000	01 220200	4048	1	F	current SNPN	17.0.0
2022-03	CT#95e	CP-220238				Clarification of the UE behavior for the rejected NSSAI for the	17.6.0
			4049	1	F	failed or revoked NSSAA for the current SNPN	
2022-03	CT#95e	CP-220238				Missing cause value number for Onboarding services	17.6.0
			4064	1	F	terminated	
2022-03	CT#95e	CP-220239	2000	4	D	Local deactivation of UP resource for an MA PDU session with	17.6.0
2022.02	CT#05o	CD 220220	3860	1	В	PDN leg - 24501 Part	1760
2022-03	01#956	GF-220239	3861	1	F	reached	17.0.0
2022-03	CT#95e	CP-220239	3862	1	F	UE receives ATSSS not supported	17.6.0
2022-03	CT#95e	CP-220239				Abnormal handling for adding non-3GPP leg to an MA PDU	17.6.0
			3863	1	F	session already with PDN leg	
2022-03	CT#95e	CP-220239	3864	1	F	Clarification on QoS rules merge for MA PDU session	17.6.0
2022-03	CT#95e	CP-220240	3666	2	В	5GS MUSIM Paging restriction	17.6.0
2022-03	CT#95e	CP-220240	3851	-	F	Condition for removing the paging restriction	17.6.0
2022-03	CT#95e	CP-220240	3853	-	F	N1 NAS signalling connection release reformulation	17.6.0
2022-03	CT#95e	CP-220240	0054		-	Paging restriction in N1 NAS signalling connection release	17.6.0
0000.00	07405 -	CD 000040	3854	-		upon RAN paging rejection	47.0.0
2022-03	CT#95e	CP-220240	3857	1	F C	Correction of 13540 start scenarios for MRU procedure	17.6.0
2022-03	CT#95e	CP-220240	3913	1	F	Connection release for emergency service in 5GS	17.0.0
2022-03	CT#95e	CP-220240	5514	1	1	UE to release NAS signalling connection and indicate Paging	17.0.0
2022 00	01//000	01 220240				restriction during mobility Registration only if no emergency	17.0.0
			3949	-	F	service is ongoing in 5GS	
2022-03	CT#95e	CP-220240	3960	1	С	PTCC handling during emergency registration	17.6.0
2022-03	CT#95e	CP-220241	3912	2	F	The handling of paging cause support indicator in 5GS	17.6.0
2022-03	CT#95e	CP-220241	3937	1	F	Collision between UCU and SR	17.6.0
2022-03	CT#95e	CP-220241	3939	2	F	Clarification on Paging cause	17.6.0
2022-03	CT#95e	CP-220241	3982	-	F	Terminology clean up in 5GS	17.6.0
2022-03	CT#95e	CP-220241	4038	1	F	Paging restriction with connection release in non-allowed area	17.6.0
2022-03	CT#95e	CP-220241	4039	1	F	Correction of 13447 handling for Multi-USIM UE	17.6.0
2022-03	CT#95e	CP-220241	4040	-	Г	Conection of 15546 handling for Multi-OSIM DE	17.0.0
2022-03	01#956	CF-220241	4078	-	F	lin 5GS	17.0.0
2022-03	CT#95e	CP-220241	4081	1	F	Uplink data status IE not included when requesting release	17.6.0
2022-03	CT#95e	CP-220241		-		Clearing paging restrictions when no Allowed NSSAI is	17.6.0
			4051	1	F	available	
2022-03	CT#95e	CP-220241	4050	2	F	Clearing paging restrictions during lower layer failure in 5GS	17.6.0
2022-03	CT#95e	CP-220242	0.505		_	Rejected NSSAI for the maximum number of UE reached with	17.6.0
0000.00	OT #05		3536	2	F	attrerent PLMNs in RA	47.0.0
2022-03	CT#95e	CP-220242	2955	2	D	PUU session establishment reject for network slice data rate	17.6.0
2022-02	CT#050	CP-220242	3000	3	ט	NSAC for number of PDU sessions taking access type into	1760
2022-03	01#956	GF-220242	3875	1	F		17.0.0
2022-03	CT#95e	CP-220242	3876	1	F	Clarification on NSAC for emergency and priority services	17.6.0
2022-03	CT#95e	CP-220242	3877	1	F	NSAC for existing PDU session with inter access handover	17.6.0
2022-03	CT#95e	CP-220242	3878	1	F	S-NSSAIs in allowed NSSAI share common NSSRG value	17.6.0
2022-03	CT#95e	CP-220242	3879	2	F	Provide all subscribed S-NSSAIs in configured NSSAI	17.6.0
2022-03	CT#95e	CP-220242				NSAC during PDU session transfer with the Allowed PDU	17.6.0
	07	00.000	3881	1	F	session status IE	
2022-03	CT#95e	CP-220242	3886	1	F	5GSM message not forwarded in case of NSAC reject	17.6.0
2022-03	CT#95e	CP-220242	2000	1	-	Access type for rejected NSSAI for the maximum number of	17.6.0
2022.02	CT#05 -	CD 200040	3906	1	F	UES reached	17.0.0
2022-03	CT#950	CP-220242	2901	-		NOOAA TESULTIOLITIPACTED DY NOAU	17.0.0
2022-03	01#906	UF-220242	3926	2	F	restriction	17.0.0
2022-03	CT#95e	CP-220242	3929	1	F	NSAC applicable for SNPN	17.6.0
2022-03	CT#95e	CP-220242		1		Removal of several Editor's notes related to NSAC and	17.6.0
			0000	4			

2022.02	CT#050	CD 220242	1015	1	E	Coding of NSSBC information IE	1760
2022-03	CT#95e	GP-220242	4015	1	F		17.6.0
2022-03	CT#95e	CP-220242	4016	1	F	NSSRG feature applicable to SNPN	17.6.0
2022-03	CT#95e	CP-220242				5GSM message not forwarded in case of NSAC reject: AMF	17.6.0
			4111	1	F	oneration	
2022.02	074050	CD 220242	4440	•	- -	Correction on EAC mode to per alian lavel	1760
2022-03	01#956	GF-220242	4112	-	Г		17.0.0
2022-03	C1#95e	CP-220244				PDU session establishment with the DNN/S-NSSAI for UAS	17.6.0
						service from the UE whch has valid aerial subscription but	
			3792	5	В	UUAA-MM is failed abnormally	
2022.02	CT#050	CD 220244	0.02	J	-	Clarification of including Service level AA container in PDU	1760
2022-03	01#956	GF-220244	0044		-		17.0.0
			3841	1	F	SESSION MODIFICATION REQUEST message	
2022-03	CT#95e	CP-220244	3865	3	F	Adding missing UUAA-SM text	17.6.0
2022-03	CT#95e	CP-220244				Clarification on PDU session establishment for valid	1760
2022 00	011/000	0. 220211	3880	1	F	subscription of DNN/S-NSSAI	
	OTHOS	00.000044	0003	1	-		17.0.0
2022-03	CT#95e	CP-220244	3890	1	F	Generalizing 5GMM cause for UAS service not allowed.	17.6.0
2022-03	CT#95e	CP-220244				Clarification when valid UUAA result is available in the UE MM	17.6.0
			3891	1	F	context	
2022-03	CT#050	CP-220244	3802	_	F	Remove resolved ENs	1760
2022-03	01#956	OF-220244	3092	-	-		17.0.0
2022-03	C1#95e	CP-220244	3893	1	F	SA3 requirement for security protected UAS parameters.	17.6.0
2022-03	CT#95e	CP-220244	3945	1	F	Correction to the general part for UUAA	17.6.0
2022-03	CT#95e	CP-220244	3947	2	C	The handling of 5GMM#79	1760
2022 00	OT#000	CD 220244	4040	4	С Г		17.0.0
2022-03	C1#95e	CP-220244	4010	1	F	Modify service-level-AA parameters	17.6.0
2022-03	CT#95e	CP-220244	4011	1	F	Correction of procedure and text for UAS services	17.6.0
2022-03	CT#95e	CP-220244	4054	1	F	Clarification for revocation of C2 authorization	17.6.0
2022-03	CT#95e	CP-220244	4080	-	F	Undate IFL of Service-level-AA container	1760
2022-03	OT#356	00 220244	4440	4			47.0.0
2022-03	U1#95e	CP-220244	4113	1		Correction on description of C2 authorization parameters	17.6.0
2022-03	CT#95e	CP-220244	4114	-	F	Correction on service-level-AA response bit name	17.6.0
2022-03	CT#95e	CP-220245				Remote UE report not allowed when 5GSM BO timer is	17.6.0
2022 00	0.11000	J. 22027J	3883	1	F		
	07/07	00.000.00	3002	1	Г		
2022-03	C1#95e	CP-220245				Collision of network initiated PDU session release and remote	17.6.0
			3884	1	F	UE report procedures	
2022-03	CT#95e	CP-220245				Corrections for Service Request procedure when requesting	17.6.0
_0 00	011000	000	3956	_	F	5G ProSe resources is the trigger for the procedure	
	OTHOS	00.000045	3330	-	1		17.0.0
2022-03	CT#95e	CP-220245				Update to NAS security mode command during PC5 link	17.6.0
			3995	1	В	establishment	
2022-03	CT#95e	CP-220245				Correction of IEI values for the REMOTE UE REPORT	17.6.0
			1065	1	F		
0000.00	OTHOR	00.000.47	4003	1	1		47.0.0
2022-03	C1#95e	CP-220247			_	DINN as an optional parameter for emergency PDU session	17.6.0
			3967	-	F	when interworking with EPS	
2022-03	CT#95e	CP-220247				Applicability of NULL integrity protection algorithm in case of a	17.6.0
			3068	_	F	established emergency PDI I session	
0000.00	OT#05 -	00 000047	5500		•	Dischling of N4 mode in cost of #40 while Emergency call	47.0.0
2022-03	C1#95e	CP-220247			_	Disabiling of N1 mode in case of #10 while Emergency call	17.6.0
			3969	-	F	pending	
2022-03	CT#95e	CP-220247	3989	-	F	Lost text in 6.4.1.2	17.6.0
2022-03	CT#95e	CP-220247	4000	_	D	Two editorial corrections	1760
2022 00	OT#350	01 220247	4000		5		17.0.0
2022-03	C1#95e	CP-220247	4004	-	F	Clarification to payload container IE	17.6.0
2022-03	CT#95e	CP-220247	4006	-	F	Correction to 5GSM capability IE	17.6.0
2022-03	CT#95e	CP-220247	4021	-	F	Correct Re-attempt indicator IE for #39 in PDU session release	17.6.0
2022.02	CT#050	CP_220247	4024		F	Adding the missing implementation of C1 215154	1760
2022-03	01#308	01 -220247	4024		<u>'</u>		17.0.0
2022-03	CT#95e	CP-220247	4026	-	ט	Editorial correction of 5GS network support	17.6.0
2022-03	CT#95e	CP-220247				UE is allowed to use PCO IE after inter-system change from	17.6.0
			4027	-	F	N1 mode to S1 mode	
2022.02	CT#050	CP-220247				CloT user data container not forwarded due to conduction	1760
2022-03	01#308	01 -22024/	4007		-		17.0.0
L			4037	-		Control	
2022-03	CT#95e	CP-220247	4041		F	Correction for enabling use of MICO mode	17.6.0
2022-03	CT#95e	CP-220247	4042	-	F	Correction of eDRX handling in 5GS	17.6.0
2022.02	CT#050	CP_220247	4052		F	Clarification of Notification response message	1760
2022-03	01#908	07-220247	+003		-		17.0.0
2022-03	C1#95e	CP-220247	4061	-	F	Correction on UAC exception handling	17.6.0
2022-03	CT#95e	CP-220247	4063	-	F	Correction to the MBS back-off timer	17.6.0
2022-03	CT#95e	CP-220247	l			Correcting the terminology of the signalling between the LIF	17.6.0
2022 00	0.11000	J. 220271	4070	_	F	and the SME	
0000.05	OT "2-	00.0000	4079	-	-		47.0.0
2022-03	C1#95e	CP-220247	3997	1	F	Augnment of 5GSM state machine to procedural descriptions	17.6.0
2022-03	CT#95e	CP-220248	4023	1	F	Correction on #62	17.6.0
2022-03	CT#95e	CP-220248	4031	1	F	Handling of ESM non-congestion back-off timer	17.6.0
2022 00	CT#05-	CD 220240	4000	1	C	Correction to the DDU eccesion release are saling	1760
2022-03	01#956	CP-220248	4032	1	Г	Correction to the PDU session release procedure	17.0.0
2022-03	CT#95e	CP-220248			1	Starting T3540 only considers Service Request message but	17.6.0
			4034	1	F	not the CPSR Message	
2022-03	CT#95e	CP-220248	1		1	Correction on creating Oos rule in an ACTIVATE DEDICATED	17.6.0
2022-03	01#000	51 220240	1005		C	EDS READED CONTEVT DECHEST massage	
0000	07:0-	00.000	4000	I <u>-</u>		LES DEARER CONTEXT REQUEST MESSAge	47.0.0
2022-03	C1#95e	CP-220248	4087	-	F	Correction on number of standardized access category	17.6.0
2022-03	CT#95e	CP-220248				No modification operation permitted in ACTIVATE BEARER	17.6.0
_		_	4000	1	F	CONTEXT REOLIEST message	
			4093	-			
2022.02	CT#052	CD 220249	4093	-	і С	5CMM cause #31 not used instead of #76	1760
2022-03	CT#95e	CP-220248	4093	-	F	5GMM cause #31 not used instead of #76	17.6.0

2022-03	CT#95e	CP-220248	4107		E	Clarification on SM_RetryAtRATChange values configured in both ME and LISIM	17.6.0
2022-03	CT#95e	CP-220248	4107	-	Г	Correction on establishment of NAS signalling connection over	17.6.0
2022-03	CT#050	CP-220248	4116	-	F	non-3GPP access	1760
2022-03	01#356	01-220240	4118	-	F	operations	17.0.0
2022-03	CT#95e	CP-220249	4007	1	F	Duplicate 5GMM message type values	17.6.0
2022-03	CT#95e	CP-220249	4008	1	F	Indication of pending PDU session for NAS recovery	17.6.0
2022-03	CT#95e	CP-220249	4017	1	F	Correct Coding of PEIPS assistance information IE	17.6.0
2022-03	CT#95e	CP-220249	4026	4	F	Clarification on Allowed PDU session status IE included in	17.6.0
0000.00	07/05	00.000040	4030	1	Г Г	The head the set of a DDX is the AME	47.0.0
2022-03	CT#95e	CP-220249	4043	1	F	The handling of eDRX in the AMF	17.6.0
2022-03	CT#95e	CP-220249	4069	1	F -	Change in CloT optimizations preferred network behavior	17.6.0
2022-03	CT#95e	CP-220249	4080	1	F	Clarifications on the mapped 5G-GUTI terminology	17.6.0
2022-03	CT#95e	CP-220249	4083	1	D	Service request for redirecting CIoT UEs to EPC	17.6.0
2022-03	CT#95e	CP-220249	4086	1	F	Correction on syntactical error in QoS operation	17.6.0
2022-03	CT#95e	CP-220249	4090	1	F	Initiate PDU session modification procedure for emergency PDU session	17.6.0
2022-03	CT#95e	CP-220249	1000		•	Indicate change of PS data off LIE status outside of LADN	1760
2022 00	01//000	01 220240	4091	2	F	service area	17.0.0
2022-03	CT#95e	CP-220249	1001	-	•	Error bandling about OoS rule without corresponding OoS flow	1760
2022 05	01#330	01 220245	4092	1	F	description	17.0.0
2022-03	CT#95e	CP-220249	4094	1	F	Clarification on OoS flow handling	1760
2022 03	CT#950	CP-220240	4004	•	•	Protect emergency PDLL session when receiving #28 in the	17.6.0
2022-05	01#356	01-220243	1006	1	F	service reject message	17.0.0
2022.02	CT#050	CP 220240	4030	1	1	Correction in the AME operation upon initiating a LICLI for CAG	1760
2022-03	01#95e	GF-220249	4104	1	E C	information undate	17.0.0
2022.02	CT#050	CD 220240	4104	1	Г	Correction in the LIE operation upon receipt of a CAC	1760
2022-03	C1#95e	GF-220249	4105	1	E	information list during the initial registration	17.0.0
2022.02	CT#050	CD 220240	4105	1	Г	Authorized QoS flow provided by petwork	1760
2022-03	CT#95e	CF-220249	4117	1	Г	Authonized QOS now provided by hetwork	17.0.0
2022-03	C1#95e	GP-220249	1110	4	-		17.6.0
0000.00	OT#05 a	CD 000040	4119	1	Г Г	Inters	47.0.0
2022-03	CT#95e	CP-220249	4120	1	F	Correction on attempt counter reset	17.6.0
2022-03	C1#95e	CP-220249	1101	1	C	The solution to CAG IDs of a PLMN beyond the limit of one	17.6.0
0000.00	07/05	00.000040	4124	1	C	Entry-ie part	47.0.0
2022-03	C1#95e	CP-220249	4405	4	c	The solution to CAG IDs of a PLIVIN beyond the limit of one	17.6.0
0000.00	OT#05 a	00.000054	4125	1	C	Entry-Procedule part	47.0.0
2022-03	C1#95e	CP-220251	2006	4	-	Updates to 565 session management aspects over non-36PP	17.6.0
0000.00	OT#05 a	00 000050	3990	1	Г		47.0.0
2022-03	CT#95e	CP-220252	3040	1	D	EDC related PCO parameters usage	17.0.0
2022-03	CT#95e	CP-220252	3695	3	D	Spanar validity condition couling	17.0.0
2022-03	CT#95e	CP-220252	3957	3	В	Support of updating ECS configuration info	17.6.0
2022-03	CT#95e	CP-220256	3856	2	В	Update for multicast session release	17.6.0
2022-03	CT#95e	CP-220256	3900	2	В	UE MBS session local release at PDU session release	17.6.0
2022-03	C1#95e	CP-220256			-	Correction of the length field of the requested MBS container	17.6.0
0000.00	OT USE	00.00050	3903	1	F		47.0.0
2022-03	CT#95e	CP-220256	3920	1	В	UE handling of MBS back-off timer	17.6.0
2022-03	CT#95e	CP-220256	0004		-	UE Locally leaves the MBS session when the PDU session is	17.6.0
0000 05	07/07		3921	-	В		47.0.0
2022-03	C1#95e	CP-220256	3922	1	F	Associate the MBS service area with the TMGI	17.6.0
2022-03	CT#95e	CP-220256	3950	-	В	Using separate QoS flows dedicated for multicast	17.6.0
2022-03	CT#95e	CP-220256	3951	1	В	Introducing the security aspects for MBS	17.6.0
2022-03	CT#95e	CP-220256			_	Removing UE from MBS session when the UE moves outside	17.6.0
			3952	-	F	all the MBS service area(s) of that MBS session	
2022-03	CT#95e	CP-220256			_	Including the reason of removing a joined UE from an MBS	17.6.0
	0	00	3953	2	F	session by the network	
2022-03	CT#95e	CP-220256	3954	1	F	Correction to MBS service area	17.6.0
2022-03	CT#95e	CP-220256	3955	1	F	Correction for NR CGI list in the MBS service area	17.6.0
2022-03	CT#95e	CP-220256	3992	1	F	MCC and MNC coding in Received MBS container IE	17.6.0
2022-03	CT#95e	CP-220256				Remove UE from MBS session when the PDU session is	17.6.0
			4028	1	В	released implicitly	
2022-03	CT#95e	CP-220256				UE MBS session local leave when the 3GPP access UP	17.6.0
			4029	-	В	resources are released	
2022-03	CT#95e	CP-220256			_	UE locally leaves the MBS sessions locally when the PDU	17.6.0
			4035	1	F	session is released locally	
2022-03	CT#95e	CP-220256				Moving the impact of inter-system change from N1 mode to S1	17.6.0
			4072	-	D	mode on MBS to the correct clauses	
2022-03	CT#95e	CP-220256				Correcting the type of the Requested MBS container IE and	17.6.0
			4073	1	F	the Received MBS container IE to be type 6	
2022-03	CT#95e	CP-220256	4075	-	F	MBS inapplicability over non-3GPP access	17.6.0
2022-03	CT#95e	CP-220256				The impact of PDU session hand-over from 3GPP access to	17.6.0
			4076	-	F	non-3GPP access on MBS sessions	
						TAL configuration for non 2000 concer	1

2022-03	CT#95e	CP-220258	3919	1	В	PDU session associating with PDU session pair ID and RSN	17.6.0
2022-03	CT#95e	CP-220258	3928	1	F	Clarification in the NSAC for redundant PDU sessions	17.6.0
2022-03	CT#95e	CP-220258	3888	1	F	End of disaster condition during an emergency PDU session	1760
2022-03	CT#95e	CP-220258	3040	1	B	AME behaviors during the registration for disaster roaming	17.6.0
2022-03	CT#05e	CD 220230	4007	1		Correction to usage of disaster return weit renge	17.0.0
2022-03	CT#95e	CF-220200	4097	-	г г		17.0.0
2022-03	CT#95e	CP-220260	4098	-	F	Correction on disaster roaming information updating data	17.6.0
2022-03	C1#95e	CP-220260			_	Correction to disaster return wait range in Service reject	17.6.0
			4101	-	F	message.	
2022-03	CT#95e	CP-220260	4062	1	F	Correction of IEI value of the Disaster return wait range IE	17.6.0
2022-03	CT#95e	CP-220260	3970	1	В	PLMN with disaster condition IE as cleartext	17.6.0
2022-03	CT#95e	CP-220260	4067	1	F	Clarifications on wait timer	17.6.0
2022-03	CT#95e	CP-220260	4066	1	F	HPI MN control in roaming area	1760
2022 00	CT#050	CP 220260	2007	2	- -	Non supporting PI MN for disaster convice	17.6.0
2022-03	CT#95e	CF-220200	3007	2	1	Non supporting P LIVIN for ulsaster service	17.0.0
2022-03	C1#95e	CP-220261			_	Mismatch for MO SMS over NAS or MO SMSoIP service type	17.6.0
			3909	1	F	criterion between TS23.122 and TS24.501	
2022-03	CT#95e	CP-220261	3941	1	F	List indication not apply for secured packet	17.6.0
2022-03	CT#95e	CP-220261	3904	2	F	High Priority PLMN search due to SOR	17.6.0
2022-03	CT#95e	CP-220261	3908	1	F	De-registration handling due to Tsor-cm timer expiry	17.6.0
2022-03	CT#95e	CP-220261	4088	1	F	Clarification on list indication with secured packet	1760
2022-03	CT#050	CP-220262	3011		C	About the decision on eDBX parameters IE in AME	1760
2022-03	CT#95e	CF-220202	3944	-	C	About the decision on eDKA parameters IE In Alvin	17.0.0
2022-03	C1#956	CP-220264	2074		-	Remove Editor's Note in 1524.501 about RAN work on UPIP	17.6.0
			3974		F		
2022-03	CT#95e	CP-220265				Addition of NAS over Non-Terrestrial Network general clause	17.6.0
			3902		F	text	
2022-03	CT#95e	CP-220265				Applicability condition of the list of PLMNs not allowed to	17.6.0
			3975	1	F	operate at the present UE location	
2022-03	CT#95e	CP-220265				Storage and deletion of PI MNs not allowed to operate at the	17.6.0
	0	0. 220200	3998	1	F	present UF location list	
2022.02	CT#050	CD.220265	4120	1	C	Taking GNSS fix time into account in LIE MAS lover	1760
2022-03	CT#95e	CF-220205	4129	1	C		17.0.0
2022-03	C1#95e	CP-220265			-	Addition of extended NAS timers via a satellite NG-RAN cell -	17.6.0
			3901	3	В	Alternative A	
2022-03	CT#95e	CP-220265	4057	1	F	Update the contents of an entry in the PLMN List for #78	17.6.0
2022-03	CT#95e	CP-220265				Handling of PDU session modification not forwarded due to	17.6.0
			4009	1	F	#78	
2022-03	CT#95e	CP-220282	4123	1	В	Extension of SNN description for NSWO	17.6.0
2022-03	CT#95e	CP-220283		-		Paging Subgrouping undates in Registration and LIF	1760
2022 00	01#000	01 220200	3896	3	C	Configuration Undate procedure	17.0.0
2022.02	OT#05 a	CD 220202	4055	1		Desing subgroup during emergeney cell	17 6 0
2022-03	CT#95e	CP-220283	4055	1	F	Paging subgroup during emergency call	17.6.0
2022-03	C1#95e		<u> </u>		-	Editorial correction in clause 5.3.25	17.6.1
2022-06	CT#96	CP-221195	4275	1	A	SSC mode corrections	17.7.0
2022-06	CT#96	CP-221189	4139	4	F	Anonymous SUCI usage	17.7.0
2022-06	CT#96	CP-221226	4255	2	А	Indicating Supported SSC Mode(s) by the UE	17.7.0
2022-06	CT#96	CP-221280	4190	1	F	Correction on terminology and description for ID UAS	17.7.0
2022-06	07.000						
2022 00	C1#96	CP-221343	4327	4	F	LIE enter in substate NO-SUPI	1770
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	CT#96	CP-221343	4327	3	F	UE enter in substate NO-SUPI	17.7.0
2022-06	CT#96 CT#96	CP-221343 CP-221202	4327 4199	3 5	F B	UE enter in substate NO-SUPI Network slice AS group - General aspects	17.7.0 17.7.0
2022-06	CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202	4327	3	F B	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information	17.7.0 17.7.0 17.7.0
2022-06	CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202	4327 4199 4292	3 5 1	F B B	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding	17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202	4327 4199 4292 4295	3 5 1 1	F B B B	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability	17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202	4327 4199 4292 4295 4308	3 5 1 1 2	F B B B B	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203	4327 4199 4292 4295 4308 4166	3 5 1 1 2 2	F B B B B B B	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261	4 3 5 1 1 2 2 1	F B B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264	3 5 1 2 2 1	F B B B B F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264	3 5 1 2 2 1 - 1	F B B B B F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266	3 5 1 2 1 - 1 - 1	F B B B B F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267	3 5 1 2 1 - 1 1 1	F B B B B F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267	3 5 1 2 2 1 - 1 1	F B B B F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268	- 3 5 1 2 1 - 1 1 1	F B B B F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268	3 5 1 2 1 - 1 1 1 1 1	F B B B F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285	3 5 1 2 2 1 - 1 1 1 1 1 1 1	F B B B F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4268 4285 4285	3 5 1 2 2 1 - 1 1 1 1 1	F B B B B F F F F F F F F F D	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4285 4286 4337	- 3 5 1 2 1 - 1 1 1 1 1 1	F B B B B F F F F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4267 4268 4285 4285 4286 4337 4357	- 1 2 1 - 1 1 1 1 1 1 1	F B B B B F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368	- 3 5 1 2 1 - 1 1 1 1 1 1 1	F B B B F F F F F D F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368	- 3 5 1 2 1 - 1 1 1 1 1 1 1 1	F B B B F F F F F D F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368 4369	- 1 2 1 - 1 1 1 1 - 1 - - 1 - -	F B B B F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368 4369 4375	3 5 1 2 2 1 1 1 1 1 1 2 2 2 2 2 2 1 1 1 2	F B B B F F F F F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4285 4285 4285 4337 4368 4337 4369 4375 4421	- 1 2 1 2 1 1 1 1 1 2 1	F B B B F F F F F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4285 4286 4337 4368 4337 4369 4375 4421 4422	- 1 2 1 2 1	F B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SNPN Forbidden List Across Power Cycle	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4286 4337 4368 4369 4375 4421 4422 4257	- 1 2 1 2 1	F B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SNPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-220	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4266 4267 4268 4285 4286 4337 4368 4369 4375 4421 4422 4257	- 1 2 1 2 1	F B B B F F F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SNPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368 4369 4375 4421 4422 4257	- 1 2 1 2 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 2 1 1 2 1 1 2	F B B B B F F F F F F F F F F F F F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SNPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN Storage of 5GMM parameters mapping with SUPI from USIM for AKA based SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4286 4337 4368 4385 4389 4375 4421 4422 4257 4424	- 1 2 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 2 1 1 2 1 2 1 2	F B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SMPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN Storage of 5GMM parameters mapping with SUPI from USIM for AKA based SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4285 4285 4285 4337 4368 4337 4369 4375 4421 4422 4257 4424 4162	- 3 5 1 2 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 2 1 1 2 1 2 1 2 - 2 -	F B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SMPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN Storage of 5GMM parameters mapping with SUPI from USIM for AKA based SNPN	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96	CP-221343 CP-221202 CP-221202 CP-221202 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221203 CP-221204 CP-221204	4327 4199 4292 4295 4308 4166 4261 4264 4266 4267 4268 4285 4285 4286 4337 4368 4337 4368 4337 4369 4375 4421 4422 4257 4424	- 3 5 1 2 1 - 1 1 1 1 1 1 1 1 1 1 1 2 1 1 2 1 1 2 1 2 1 2 1 2 -	F B B B B F	UE enter in substate NO-SUPI Network slice AS group - General aspects Support NSAG - Procedure Message and NSAG information IE coding Support NSAG in 5GMM capability NSAG information storage Onboarding SNPN and secondary authentication support list of configuration data PVS address providing correction Editor's note in subclause 9.11.3.51 AMF onboarding configuration data clean up S-NSSAI when URSP rule triggering establishment of PDU session was signalled by non-subscribed SNPN UE parameter update data set for ME routing indicator update data Editorial change : onboarding indicator No NSSAI provided to lower layer for onboarding service Usage of the onboarding SUCI Correction for the note about the UE policy sections stored for PLMNs or SNPNs Mapped S-NSSAI applicable for SNPN SUPI handling in case of CH using AAA server Storage of SNPN Forbidden List Across Power Cycle Correction of non-3GPP access in SNPN Storage of 5GMM parameters mapping with SUPI from USIM for AKA based SNPN	17.7.0 17.7.0

202-86 CTF#86 CP-21206 H34 F 2030/10 2022-86 CT#96 CP-21206 H16 F H56 H2 H57 T 7.7.0 2022-86 CT#96 CP-22105 H167 F Uplink data status handling for NAS connection release 17.7.0 2022-86 CT#96 CP-22105 H169 F The handling of paging cause in SGS 17.7.0 2022-86 CT#96 CP-22105 H20 F The handling of the MUSIN UE 17.7.0 2022-86 CT#96 CP-22105 H20 F Referring to the correct terminology for the paging indication in T.7.0 2022-86 CT#96 CP-22105 H141 F Alignment for NSAC to mergancy and priority services 17.7.0 2022-86 CT#96 CP-22106 H181 F Delanit subscribed SNSAI not subject to NSAC NSAC<	2022-06	CT#96	CP-221204	4054		-	Clarification on PDU session establishment for MA PDU	17.7.0
CT222 CT222 CT2222 CT2222 CT22222 CT2292 CT2922	2022-06	CT#96	CP-221205	4354	-	F	session Completing terminology clean up in 5GS	1770
Construction 4166 I F 5GS 2022-06 CF280 CP-22105 4157 F Uplink data status handing for NAS connection release 17.7.0 2022-06 CF280 CP-22105 4150 F The handling of paging cause in SGS 17.7.0 2022-06 CF496 CP-22106 F P Intermine to the submit of the MUSM UE 17.7.0 2022-06 CF496 CP-22106 4220 F Referring to the consult terminology for the paging indication 17.7.0 2022-06 CF496 CP-22106 4427 F UF to the page mode of the submit o	2022-00	CT#96	CP-221205	4140	-	1	Uplink data status handling for removing paging restriction in	17.7.0
2022-66 CT#96 CP-221205 4159 1 Fit The handling of bright souce in SGS 17.7.0 2022-66 CT#96 CP-221205 4159 1 F PE1 handling of the MUSIM UE 17.7.0 2022-66 CT#96 CP-221205 4220 1 F De1 handling of the MUSIM UE in SGS 17.7.0 2022-66 CT#96 CP-221205 4221 F UE no longer a MUSIM UE in SGS 17.7.0 2022-66 CT#96 CP-221205 4374 2 F UE no longer a MUSIM UE in SGS 17.7.0 2022-66 CT#96 CP-221206 4421 F Clarification to MUSIM UE inspending in NB-N1 mode and WB 17.7.0 2022-66 CT#96 CP-221206 4142 F Dafatation on MASC tor SNPN onboarding 17.7.0 2022-66 CT#96 CP-221206 4181 F Clarification on update operading NSA14 UE reselves 17.7.0 2022-66 CT#96 CP-221206 4181 F Clarification on update operading NSA14 UE reselves 17.7.0 <	2022 00	01#00	01 221200	4156	1	F	5GS	
202260 CT#96 CP-221205 4160 1 F PEI handling of paging cause in SGS 17.7.0 202260 CT#96 CP-221205 4160 1 F PEI handling of the MUSIM UE in SGS 17.7.0 2022.06 CT#96 CP-221205 4221 1 F Referring to the correct terminology for the paging indication 17.7.0 2022-06 CT#96 CP-221205 4421 1 F Uter mode the mode	2022-06	CT#96	CP-221205	4157	-	F	Uplink data status handling for NAS connection release	17.7.0
2022-06 C1F#96 C-221205 d1F60 F PEI handling for the MUSIM UE 177.0 2022-06 C1F#96 C-221205 d221 F Admonratic cases for MUSIM UE in SQS maging indication 1 1 7.0 2022-06 C1F#96 C-221205 d221 F Edmonratic cases for MUSIM UE in SQS maging indication 1 1 7.0 2022-06 C1F#96 C-221205 d374 2 F UE no longer at MUSIM UE separating in NB-NT mode and WE 1 7.7.0 2022-06 C1F#96 C-221206 d1441 F Alignment for NSAC for SNPN nobcarding 17.7.0 2022-06 C1F#96 C-221206 d1481 F Clarification on update of pending NSAI HU E receives 17.7.0 2022-06 C1F#96 C-221206 d1481 F Clarification on update of pending NSAI HU E receives 17.7.0 2022-06 C1F#96 C-221206 d1481 F Clarification on update of pending NSAI HU E receives 17.7.0 2022-06 C1F#96 C-221206 d1481 F	2022-06	CT#96	CP-221205	4159	1	F	The handling of paging cause in 5GS	17.7.0
2122-06 C1#96 CP-221205 4239 1 F Diffection for sensing fine Follow-on request indication 17.7.0 2022-06 C1#96 CP-221205 4230 - F for wing to the control MUSIM block binding in 5GS 17.7.0 2022-06 C1#96 CP-221205 4247 - F for wing to the control MUSIM block binding in 5GS 17.7.0 2022-06 C1#96 CP-221205 4441 F Alignment for NSAC for emergency and priority services 17.7.0 2022-06 C1#96 CP-221206 41421 F Clarification on NSAC for SNN onboarding 17.7.0 2022-06 C1#96 CP-221206 41431 F Clarification on update of pending NSAI int SUBject to NSAC 17.7.0 2022-06 C1#96 CP-221206 41431 F Clarification on update of pending NSAI int SUBject to NSAC 17.7.0 2022-06 C1#96 CP-221206 41431 F Addition of the NSSAI int SUBject to NSAC 17.7.0 2022-06 C1#96 CP-221206 41431 F Examption SUSAI int	2022-06	CT#96	CP-221205	4160	1	F	PEI handling for the MUSIM UE	17.7.0
2022-06 CT#96 CP-221205 4374 P Referring to the correct terminology for the paging indication 17.7.0 2022-06 CT#96 CP-221205 4374 F UE no longer a MUSIM UE soperating in NB-N1 mode and WE 17.7.0 2022-06 CT#96 CP-221206 4427 F NI CE mode B 17.7.0 2022-06 CT#96 CP-221206 4142 F Clarification to MUSIM UE soperating in NB-N1 mode and WE 17.7.0 2022-06 CT#96 CP-221206 4143 F Clarification on update operating represents to NSAC tor SNPN orboarding 17.7.0 2022-06 CT#96 CP-221206 4145 F Clarification on update operating NSSAI to SNPC to NSAC 17.7.0 2022-06 CT#96 CP-221206 4200 2 F Triget to UB44 17.7.0 2022-06 CT#96 CP-221206 4200 2 F Triget to UB44 17.7.0 2022-06 CT#96 CP-221206 4201 F Triget to UB44 17.7.0 2022-06 CT#96	2022-06	C1#96	CP-221205	4220	1	F	Correction for setting the Follow-on request indicator in abnormal cases for MUSIM UE in 5GS	17.7.0
Chronols P For voice services for MUSIM handling in SGS For voice services for MUSIM handling in SGS 2022-06 CT#96 CP-221205 FOR	2022-06	CT#96	CP-221205	7225	1	1	Referring to the correct terminology for the paging indication	17.7.0
2022-06 CT#96 CP-21205 4374 P UE no longer a MUSIM UE 17.7.0 2022-06 CT#96 CP-21206 1444 F Alforment for NSAC for emergency and priority services 17.7.0 2022-06 CT#96 CP-21206 4182 I F Delament for NSAC for SNPN onboarding 17.7.0 2022-06 CT#96 CP-221206 4183 I F Delament subscribed 5-NSSAI not subject to NSAC 17.7.0 2022-06 CT#96 CP-221206 4185 - F Delamitation on update of pending NSAI I UE receives 17.7.0 2022-06 CT#96 CP-221206 4187 - F Trigeted NSAI NSAI SavaList NI UE receives 17.7.0 2022-06 CT#96 CP-221206 4200 P Trigeted NSAI NSAI I SavaList NI UE receives 17.7.0 2022-06 CT#96 CP-221206 4201 F Examption NI NSAI SavaList NI NI UE receives 17.7.0 2022-06 CT#96 CP-221206 4201 F Examption NI NSAI I SavaList NI NI UE rec				4230	-	F	for voice services for MUSIM handling in 5GS	
2022-06 CT#96 CP-221205 File Clarification to MUSIM UEs operating in NB-N1 mode and WB-17.7.0 2022-06 CT#96 CP-221206 4144 I F Alignment for NSAC for SMPA orbacing 17.7.0 2022-06 CT#96 CP-221206 4183 I F Clarification on SAC for SMPA orbacing 17.7.0 2022-06 CT#96 CP-221206 4183 I F Clarification on condition of registration registration 17.7.0 2022-06 CT#96 CP-221206 4185 - Clarification on condition of registration registration 17.7.0 2022-06 CT#96 CP-221206 4181 I F Information Information 17.7.0 2022-06 CT#96 CP-221206 4201 F Trager to update configured NSAI assocated with one or more 17.7.0 2022-06 CT#96 CP-221206 4241 F Exemptions for the network sile data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 F PDI second stactunatin with the number of MSAI	2022-06	CT#96	CP-221205	4374	2	F	UE no longer a MUSIM UE	17.7.0
2022-06 CT#96 CP-221206 4144 I F Alignment for NSAC for emergency and priority services 17.7.0 2022-06 CT#96 CP-221206 4182 I F Clarification on NSAC for SNPN onboarding 17.7.0 2022-06 CT#96 CP-221206 4181 I F Default subscribe 3-NSAS in ot subject to NSAC 17.7.0 2022-06 CT#96 CP-221206 4185 I Clarification on update opending NSAI and NSSAI associated with new or more 17.7.0 2022-06 CT#96 CP-221206 4200 2 F Trigeteld NSSAI Information 17.7.0 2022-06 CT#96 CP-221206 4201 F Trigeting to update configured NSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4201 F Exemptions for the network sile data rule limitation control 17.7.0 2022-06 CT#96 CP-221206 4221 F Eventorin NSSAI associated with the number of UEs associated 17.7.0 2022-06 CT#96 CP-221206 4231 <	2022-06	CT#96	CP-221205	4407	4	-	Clarification to MUSIM UEs operating in NB-N1 mode and WB-	17.7.0
202:06 CT#96 Ch-221208 HTF Clamination on ISAC for SNPM on boarding 177:0 202:06 CT#96 CP-221206 H181 F Default subscribed 5-MSAI not subscribed	2022-06	CT#96	CP-221206	4427 4144	1	F	Alignment for NSAC for emergency and priority services	1770
2022-06 CT896 CP-22120 4183 1 F Default subscribed S-NSSAI not subject to NSAC 17.7.0 2022-06 CT896 CP-221206 4184 F Chaffication on condition or nupdation rejection on update of pending NSSAI if UE receives 17.7.0 2022-06 CT896 CP-221206 4185 F rejected NSSAI 2022-06 CT896 CP-221206 4200 F Tringer to update configured NSSAI and NSSA for formation 17.7.0 2022-06 CT896 CP-221206 4200 F Tringer to update configured NSSAI and NSSA for formation 17.7.0 2022-06 CT896 CP-221206 4204 F Common NSSR0 values 17.7.0 2022-06 CT896 CP-221206 4242 I F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT896 CP-221206 4242 I F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT896 CP-221206 4333 F OUrection on NSAG faiformation 17.7.0	2022-00	CT#96	CP-221200	4182	1	F	Clarification on NSAC for SNPN onboarding	17.7.0
2022-06 CT#96 CP-221206 4184 I F Claffication on update of pending NSSAI if UE receives 17.7.0 2022-06 CT#96 CP-221206 4185 F Addition of the NSSRI information in the Network slicing 17.7.0 2022-06 CT#96 CP-221206 4187 1 F Addition of the NSSRI information in the Network slicing 17.7.0 2022-06 CT#96 CP-221206 4204 F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4204 F Ecomon NSSRG values 3 F Ean the Configured NSSAI and NSSRG information control 17.7.0 2022-06 CT#96 CP-221206 4241 F Evernptions for the network slice data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4331 F Correction on NSSRG information ore SGPP access and non-17.7.0 2022-06 CT#96 CP-221206 4333 F Correction on NSSRG information ore SGPP access and non-17.7.0 2022-06 CT#96 CP-221206 4331<	2022-06	CT#96	CP-221206	4183	1	F	Default subscribed S-NSSAI not subject to NSAC	17.7.0
2022-06 CT#96 CP-221206 1485 F rejected NSSAI 17.7.0 2022-06 CT#96 CP-221206 4187 1 F infication on update of pending NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4200 2 F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4200 2 F Tringer to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4201 F Exemptions for the network side data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 F With S-NSSAI reachs a certain threshold 17.7.0 2022-06 CT#96 CP-22120 4331 F Offerencion on SSRG information 17.7.0 2022-06 CT#96 CP-22120 4333 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-22120 4333 F Differential backoff timer in NSAC 17.7.0	2022-06	CT#96	CP-221206	4184	1	F	Clarification on condition of registration rejection	17.7.0
Line Line <thlin< th=""> <thline< th=""> Line Li</thline<></thlin<>	2022-06	CT#96	CP-221206				Clarification on update of pending NSSAI if UE receives	17.7.0
2122-06 C1#96 CP-221206 4187 I F information Information Information Information 2022-06 CT#96 CP-221206 4200 2 F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4204 - F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4201 F Exemptions for the network siles data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4201 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4331 F Ofference on NSSR information or 3GPP access and non- 17.7.0 2022-06 CT#96 CP-221206 4333 F Ofference on NSSR information or 3GPP access and non- 17.7.0 2022-06 CT#96 CP-221206 4331 F Differential backoft timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F co		OT "00	00.004000	4185	-	F	rejected NSSAI	1770
2022-06 CT#96 CP-221206 F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 F Trigger to update configured NSSAI and NSSRG information 17.7.0 2022-06 CT#96 CP-221206 F Common NSSRG values 17.7.0 2022-06 CT#96 CP-221206 4204 1 F Common NSSRG values 17.7.0 2022-06 CT#96 CP-221206 4204 1 F Exemptions for the network slice data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4224 1 F PUD sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4335 1 F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4381 - F NSSRG information value 17.7.0 2022-06 CT#96 CP-221206 4381 - F NSSRG information value 17.7.0 2022-06 CT#96 CP-221206 4381 - F NSSRG information value	2022-06	C1#96	CP-221206	1187	1	F	Addition of the NSSRG information in the Network slicing	17.7.0
2022-06 CT#96 CP-221206 4204 F Common NSSA is associated with one or more 17.7.0 2022-06 CT#96 CP-221206 4238 3 F Ite and the Configured NSSAI is 17.7.0 2022-06 CT#96 CP-221206 4240 1 F Exemptions for the network sice data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 F With S-NSSAI reaches a certain threshold 17.7.0 2022-06 CT#96 CP-221206 4241 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4333 F DUfferential backoff imer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4331 F Differential backoff imer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4331 F Extended rejected NSAC 17.7.0 2022-06 CT#96 CP-221206 4331 F Extended rejected NSAC 17.7.0 2022-06 CT#96 CP-221206	2022-06	CT#96	CP-221206	4200	2	F	Trigger to update configured NSSAI and NSSRG information	17.7.0
4204 F common NSSRG values 2022-06 CT#96 CP-221206 CP-221206 Canffication on the configured NSSAI IE 2022-06 CT#96 CP-221206 4240 I F Exemptions for the network sile data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4242 I F Exemptions for the network sile data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4242 I F DUI session reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4335 I F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4335 I F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4331 F Extended regicted NSSA11E mandator function NSA2 17.7.0 2022-06 CT#96 CP-221208 4331 F Extended regicted NSSA11E mandatory support 17.7.0 2022-06 CT#96 CP-221208	2022-06	CT#96	CP-221206	1200	-		The S-NSSAIs in an NSSAI associated with one or more	17.7.0
2022-06 CT#96 CP-221206 4238 F IE and the Configured NSSA II IE 2022-06 CT#96 CP-221206 4240 I F Exemptions for the network sile data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 I F Exemptions for the network sile data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4333 F Correcton on NSSR information orw 3GPP access and non- 17.7.0 2022-06 CT#96 CP-221206 4383 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4383 F Extended rejected NSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4183 F Extended rejected NSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4180 F Correction on LNAS TRANKA authorization 17.7.0 2022-0				4204	-	F	common NSSRG values	-
Classical F IE and the Configured NSSA IE F 2022-06 CT#96 CP-221206 4240 I F Exemptions for the network slice data rate limitation control 17.7.0 2022-06 CT#96 CP-221206 4241 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4334 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4365 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4361 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F Extended rejected NSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221206 4383 I F Extended rejected NSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4168 F Correction on LNAS TRMSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175	2022-06	CT#96	CP-221206		_	_	Clarification on the confliction between the NSSRG information	17.7.0
2122-06 C1#96 CP-221206 F Exemptions for the network side data are limitation control 17.7.0 022-06 CT#96 CP-221206 EAC mode is activated when the number of UEs associated 17.7.0 022-06 CT#96 CP-221206 EAC mode is activated when the number of UEs associated 17.7.0 022-06 CT#96 CP-221206 4334 F GPP access type 022-06 CT#96 CP-221206 4335 F Correcton on NSSRG information 17.7.0 022-06 CT#96 CP-221206 4381 F NSRG information value 17.7.0 022-06 CT#96 CP-221206 4381 F Information value 17.7.0 022-06 CT#96 CP-221206 4383 F Extended rejected NSSAI (E mandatory support 17.7.0 022-06 CT#96 CP-221208 4173 F Correction on procedures providing UUAA authorization 17.7.0 022-06 CT#96 CP-221208 4173 F Correction on UL NAS TRANSFER for UUAA muthorization 17.7.0		OT "00	00.004000	4238	3	F	IE and the Configured NSSAI IE	1770
2022-00 C1#90 CP-221206 4241 F Chindle is advance when be induced in the shold 17.7.0 2022-06 CT#96 CP-221206 1 F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 1 Manage NSSR information over 3GPP access and non- 17.7.0 2022-06 CT#96 CP-221206 4363 F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4363 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F Extender rejected NSSAI [s] mandatory support 17.7.0 2022-06 CT#96 CP-221208 Incorrect statement subscribed S-NSAI(s) marked as default 17.7.0 2022-06 CT#96 CP-221208 1413 F Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 1174 F Correction on DL NAS TRAINSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 1175	2022-06	CT#96	CP-221206	4240	1	F	Exemptions for the network slice data rate limitation control	17.7.0
2022-06 CT#96 CP-221206 4242 I F PDU sessions reactivation failure due to NSAC 17.7.0 2022-06 CT#96 CP-221206 4334 - F SGPP access type 2022-06 CT#96 CP-221206 4335 I F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4338 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F NSSRG information value 17.7.0 2022-06 CT#96 CP-221206 4382 F subject to NSAC 17.7.0 2022-06 CT#96 CP-221208 4382 F subject to NSAC 17.7.0 2022-06 CT#96 CP-221208 4174 F Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4175 I F Correction of LAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4178 I	2022-00	01#90	GF-221200	4241	1	F	with S-NSSAI reaches a certain threshold	17.7.0
2022-06 CT#96 CP-221206 4334 F Manage NSSRG information over 3GPP access and non- 3GPP access type 17.7.0 2022-06 CT#96 CP-221206 4335 1 F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4381 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F NSSRG information value 17.7.0 2022-06 CT#96 CP-221206 4383 1 F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UNAA authorization 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UNAA authorization 17.7.0 2022-06 CT#96 CP-221208 4171 F Correction on DUAA-MM handing at AMF 17.7.0 2022-06 CT#96 CP-221208 4171 F USS FQDN as service-level-AA container IE value 17.7.0 2022-0	2022-06	CT#96	CP-221206	4242	1	F	PDU sessions reactivation failure due to NSAC	17.7.0
1 4334 - F 3GPP access type 2022-06 CT#96 CP-221206 4363 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4363 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 - F NSSRG information value 17.7.0 2022-06 CT#96 CP-221208 Incorrect statement subscribed S-NSSAI(s) marked as default 17.7.0 2022-06 CT#96 CP-221208 F subject to NSAC 202-06 2022-06 CT#96 CP-221208 4168 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4174 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175 1 F UUSA-MM hailure delivery 17.7.0 2022-06 CT#96 CP-221208 4178 F Currection on UUAA-MM hailing at AMF 17.7.0 2022-06 CT#96 <td>2022-06</td> <td>CT#96</td> <td>CP-221206</td> <td></td> <td></td> <td></td> <td>Manage NSSRG information over 3GPP access and non-</td> <td>17.7.0</td>	2022-06	CT#96	CP-221206				Manage NSSRG information over 3GPP access and non-	17.7.0
2022-06 CT#96 CP-221206 4335 1 F Correction on NSSRG information 17.7.0 2022-06 CT#96 CP-221206 4381 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4382 F NSSRG information value 17.7.0 2022-06 CT#96 CP-221206 4382 F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4183 F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DLNAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on UUAA-MIN Faultine delivery 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on UUAA-MIN Faultine delivery 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on tuAA-MIN faultine delivery 17.7.0 2022-06 CT#96 CP-221208				4334	-	F	3GPP access type	
2022-06 CT#96 CP-221206 4363 F Differential backoff timer in NSAC 17.7.0 2022-06 CT#96 CP-221206 4381 F NSSR6 information value 17.7.0 2022-06 CT#96 CP-221206 4382 F NSSR6 information value 17.7.0 2022-06 CT#96 CP-221206 4383 I F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 Incorrect statement subscribed S-NSSAI(S) marked as default 17.7.0 2022-06 CT#96 CP-221208 4178 F Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4174 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 1477 F Correction on DL NAS TRANSFER for UUAA aprocedure 17.7.0 2022-06 CT#96 CP-221208 1471 F Correction on MuAAs traine datery as trained trained traine and traine at traine and traine at traine and traine at traine at traine at traine at traine at trained trained traine t	2022-06	CT#96	CP-221206	4335	1	F	Correcton on NSSRG information	17.7.0
2022-06 CT#96 CP-221206 4381 - P INSORG INDITIATION Value 17.7.0 2022-06 CT#96 CP-221206 4382 - F subject to NSAC 2022-06 CT#96 CP-221206 4383 1 F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 4168 - Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175 I F USS FQDN as envice-level-AA container IEI value 17.7.0 2022-06 CT#96 CP-221208 4175 I F USA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for 1D_UAS 17.7.0 2022-06 CT#96	2022-06	CT#96	CP-221206	4363			Differential backoff timer in NSAC	17.7.0
Loss OF 12 Loss 4382 F subject to NSAC 2022-06 CT#96 CP-221208 4383 I F Extended rejected NSAL IE mandators support 17.7.0 2022-06 CT#96 CP-221208 4383 I F Extended rejected NSAL IE mandators support 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4174 4 F Correction on DL NAS TRANSER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on UUAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4178 F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4191 F Retry restriction for 1D_UAS 17.7.0 2022-06 CT#96 CP-221208 4191 F Resolving editor's note for 1D_UAS 17.7.0 2022-06 CT#96 CP-221208 4191	2022-00	CT#90	CP-221200	4301	-	Г	Incorrect statement subscribed S-NSSAI(s) marked as default	17.7.0
2022-06 CT#96 CP-221206 4383 1 F Extended rejected NSSAI IE mandatory support 17.7.0 2022-06 CT#96 CP-221208 Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4168 F Correction to the Service-level-AA container IEI value 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175 1 F USS FODN as service-level-AA procedure 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on UUAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for SGM cause #86 17.7.0 2022-06 CT#96 CP-221208 4191 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Correction of the condition when the network initiates de- registration 17.7.0	2022 00	01//00	01 221200	4382	-	F	subject to NSAC	17.7.0
2022-06 CT#96 CP-221208 Correction of procedures providing UUAA authorization 17.7.0 2022-06 CT#96 CP-221208 4168 F Correction to the Service-level-AA container IEI value 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on UUAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4178 F Rever saticition for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4180 - F standalone IE 17.7.0 2022-06 CT#96 CP-221208 4192 F correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208	2022-06	CT#96	CP-221206	4383	1	F	Extended rejected NSSAI IE mandatory support	17.7.0
2022-06 CT#96 CP-221208 4178 F Correction to the Service-level-AA container IEI value 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction to the Service-level-AA container IEI value 17.7.0 2022-06 CT#96 CP-221208 4174 4 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175 1 F USS FQDN as service-level-AA server address 17.7.0 2022-06 CT#96 CP-221208 4178 1 F UUAA-MM failure delivery 17.7.0 2022-06 CT#96 CP-221208 4179 - Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 1191 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 191 2 F Correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 </td <td>2022-06</td> <td>CT#96</td> <td>CP-221208</td> <td></td> <td>_</td> <td>_</td> <td>Correction of procedures providing UUAA authorization</td> <td>17.7.0</td>	2022-06	CT#96	CP-221208		_	_	Correction of procedures providing UUAA authorization	17.7.0
2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4173 F Correction on DL NAS TRANSFER for UUAA procedure 17.7.0 2022-06 CT#96 CP-221208 4175 I F USS FQDN as service-level-AA procedure 17.7.0 2022-06 CT#96 CP-221208 4177 F Correction on UUAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for SGSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4180 - F standalone IE 17.7.0 2022-06 CT#96 CP-221208 4191 2 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4252 F Correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4252 F Correction of the definition of UE supporting UAS services 17.7.0	2022.06	CT#06	CD 221209	3866	5	F	payload	1770
2022:06 CT#96 CP-221208 4174 4 F Completion of service-level-AA procedure 17.7.0 2022:06 CT#96 CP-221208 4175 1 F USS FQDN as service-level-AA procedure 17.7.0 2022:06 CT#96 CP-221208 4177 - F Completion of Service-level-AA server address 17.7.0 2022:06 CT#96 CP-221208 4179 - F Correction on UJAA-MM handing at AMF 17.7.0 2022:06 CT#96 CP-221208 4179 - F Retry restriction for 5GSM cause #86 17.7.0 2022:06 CT#96 CP-221208 4179 - F standalone IE 2022:06 CT#96 CP-221208 4191 2 F Resolving editor's note for ID_UAS 17.7.0 2022:06 CT#96 CP-221208 4352 - F Correction of the definition of UE supporting UAS services 17.7.0 2022:06 CT#96 CP-221208 4367 1 F Editrification on service level AA procedure	2022-06	CT#96	CP-221208	4100	-	r E	Correction to the Service-level-AA container tel value	17.7.0
Correction CP-221208 4175 1 F Correction on UQA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4177 - F Correction on UQA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4178 1 F UUAA-MM failure delivery 17.7.0 2022-06 CT#96 CP-221208 4179 - F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4191 - F Retry restriction for to SGM cause #86 17.7.0 2022-06 CT#96 CP-221208 4191 - F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 <td>2022-00</td> <td>CT#96</td> <td>CP-221208</td> <td>4174</td> <td>4</td> <td>F</td> <td>Completion of service-level-AA procedure</td> <td>17.7.0</td>	2022-00	CT#96	CP-221208	4174	4	F	Completion of service-level-AA procedure	17.7.0
2022-06 CT#96 CP-221208 4177 - F Correction on UUAA-MM handling at AMF 17.7.0 2022-06 CT#96 CP-221208 4178 1 F UUAA-MM failure delivery 17.7.0 2022-06 CT#96 CP-221208 4179 - F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4180 - F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4180 - F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 1 F Correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 1	2022-06	CT#96	CP-221208	4175	1	F	USS FQDN as service-level-AA server address	17.7.0
2022-06 CT#96 CP-221208 4178 1 F UUAA-MM failure delivery 17.7.0 2022-06 CT#96 CP-221208 4179 F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4180 F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 4191 2 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction of the definition of UE supporting UAS services 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction on service level AA procedure for NI PDU 17.7.0 2022-06 CT#96 CP-221208 4405 <td>2022-06</td> <td>CT#96</td> <td>CP-221208</td> <td>4177</td> <td>-</td> <td>F</td> <td>Correction on UUAA-MM handling at AMF</td> <td>17.7.0</td>	2022-06	CT#96	CP-221208	4177	-	F	Correction on UUAA-MM handling at AMF	17.7.0
2022-06 CT#96 CP-221208 4179 - F Retry restriction for 5GSM cause #86 17.7.0 2022-06 CT#96 CP-221208 H180 - F standalone IE 17.7.0 2022-06 CT#96 CP-221208 H191 2 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 H192 1 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4192 1 F Correction of the condition when the network initiates de- 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the definition of UE supporting UAS services 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'compre	2022-06	CT#96	CP-221208	4178	1	F	UUAA-MM failure delivery	17.7.0
2022-06 CT#96 CP-221208 Parameters in Service-level-A4 container IE are not standalone IE 17.7.0 2022-06 CT#96 CP-221208 4180 - F standalone IE 17.7.0 2022-06 CT#96 CP-221208 4191 2 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the definition of UE supporting UAS services 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the ProSeP 17.7.0 2022-06 CT#	2022-06	CT#96	CP-221208	4179	-	F	Retry restriction for 5GSM cause #86	17.7.0
2022-06 CT#96 CP-221208 4191 2 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4192 1 F Resolving editor's note for ID_UAS 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 1 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 1 17.7.0 2022-06 CT#96 CP-221208 4387 1 F modification 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the	2022-06	CT#96	CP-221208	1100		E	Parameters in Service-level-AA container IE are not	17.7.0
2022-06 CT#96 CP-221208 Instruction Correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4252 - F Correction of the condition when the network initiates de- registration 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 1	2022-06	CT#96	CP-221208	4191	2	F	Resolving editor's note for ID_UAS	17.7.0
4192 I F registration 2022-06 CT#96 CP-221208 4252 - F Correction of the definition of UE supporting UAS services 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the ProSeP 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 F	2022-06	CT#96	CP-221208		-	•	Correction of the condition when the network initiates de-	17.7.0
2022-06 CT#96 CP-221208 4252 F Correction of the definition of UE supporting UAS services 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4387 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 </td <td></td> <td></td> <td></td> <td>4192</td> <td>1</td> <td>F</td> <td>registration</td> <td></td>				4192	1	F	registration	
2022-06 CT#96 CP-221208 4367 1 F Editorial correction 17.7.0 2022-06 CT#96 CP-221208 4387 1 F Clarification on service level AA procedure for NI PDU 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F He coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the ProSeP 17.7.0 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message	2022-06	CT#96	CP-221208	4252	-	F	Correction of the definition of UE supporting UAS services	17.7.0
2022-06 CT#96 CP-221208 4387 1 F Clarification on service level AA procedure for NI PDU 17.7.0 2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F I E coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4262 - F I E coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the ProSeP 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication and key agreement for 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 4219 3 B Authentication and k	2022-06	CT#96	CP-221208	4367	1	F	Editorial correction	17.7.0
2022-06 CT#96 CP-221208 4405 - F Handling of EMM parameters on getting #79 in SRM 17.7.0 2022-06 CT#96 CP-221209 4262 - F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 4316 1 F Interaction between 5GSM entity and upper layers with respect 17.7.0 2022-06 CT#96 CP-221209 4316 1 F to the ProSeP 17.7.0 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- 17.7.0	2022-06	CT#96	CP-221208	4387	1	F	Clarification on service level AA procedure for NI PDU	17.7.0
2022-06 CT#96 CP-221209 4262 F IE coded as 'comprehension required' 17.7.0 2022-06 CT#96 CP-221209 Interaction between 5GSM entity and upper layers with respect 17.7.0 2022-06 CT#96 CP-221209 4316 I F Interaction between 5GSM entity and upper layers with respect 17.7.0 2022-06 CT#96 CP-221209 4317 F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 I B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 I B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- 17.7.0 2022-06 CT#96 CP-221211 4280 F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 <td>2022-06</td> <td>CT#96</td> <td>CP-221208</td> <td>4405</td> <td>-</td> <td>F</td> <td>Handling of EMM parameters on detting #79 in SRM</td> <td>17.7.0</td>	2022-06	CT#96	CP-221208	4405	-	F	Handling of EMM parameters on detting #79 in SRM	17.7.0
2022-06 CT#96 CP-221209 Interaction between 5GSM entity and upper layers with respect 17.7.0 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06	2022-06	CT#96	CP-221209	4262	-	F	IE coded as 'comprehension required'	17.7.0
4316 1 F to the ProSeP 2022-06 CT#96 CP-221209 4317 - F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B UE-to-network relay 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211	2022-06	CT#96	CP-221209				Interaction between 5GSM entity and upper layers with respect	17.7.0
2022-06 C I #96 CP-221209 4317 F A few cleanups on 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4318 1 B The timer for authentication and key agreement for 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221201 4219 3 B Nuthentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4289		07.02	00.001	4316	1	F	to the ProSeP	47 - 4
2022-06 CT#96 CP-221209 4318 1 B Interferror autmentication and key agreement for 5G ProSe 17.7.0 2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	CT#96	CP-221209	4317	-	F	A rew cleanups on 5G ProSe	17.7.0
2022-06 CT#96 CP-221209 4360 - F Invalid PDU session identity in Remote UE Report message 17.7.0 2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221201 4219 3 B Authentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	U1#90	GP-221209	4318	1	в	The unier for authentication and key agreement for 5G ProSe	17.7.0
2022-06 CT#96 CP-221209 4365 1 B Secondary authentication via L3 relay 17.7.0 2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4219 3 B network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	CT#96	CP-221209	4360	-	F	Invalid PDU session identity in Remote UE Report message	17.7.0
2022-06 CT#96 CP-221210 Authentication and key agreement for 5G ProSe UE-to- network relay 17.7.0 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	CT#96	CP-221209	4365	1	В	Secondary authentication via L3 relay	17.7.0
4219 3 B network relay 2022-06 CT#96 CP-221211 4280 - F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	CT#96	CP-221210				Authentication and key agreement for 5G ProSe UE-to-	17.7.0
2022-06 C1#96 CP-221211 4280 F Correction on UE 5GMM state for 5GMM cause #76 17.7.0 2022-06 CT#96 CP-221211 4281 - F Correction on AMF handling on PDU session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	0000	07.02	00.001	4219	3	B	network relay	47 - 2
2022-06 CT#96 CP-221211 4281 F Correction on AME handling on PDD session release 17.7.0 2022-06 CT#96 CP-221211 4282 - F Correction on RRC resume indication at AMF 17.7.0 2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-06	CT#96	CP-221211	4280	-		Correction on UE 5GMM state for 5GMM cause #76	17.7.0
2022-06 CT#96 CP-221211 4289 - F Correction to PDU session type 17.7.0	2022-00	CT#90	CP-221211	4201	<u>-</u>	F		17.7.0
	2022-06	CT#96	CP-221211	4289	-	F	Correction to PDU session type	17.7.0

2022-06	CT#96	CP-221211	4299	-	F	Taking into account information from the NG-RAN when determining the Paging subgroup ID	17.7.0
2022-06	CT#96	CP-221211	4301	1	F	Conditions for an inactive UE to request the lower layers to transition to the connected mode	17.7.0
2022-06	CT#96	CP-221211	4301	-	F	Skipping access control checking for NAS signalling	17.7.0
2022-06	CT#96	CP-221211	4211		-	Condition of including new configured NSSAI in Registration	17.7.0
2022-06	CT#96	CP-221211	4312	-	F	Simplify enumeration of all kinds of rejected NSSAI	1770
2022-06	CT#96	CP-221211	4325	-	F	Removing the obsolete description of C1-211443	17.7.0
2022-00	CT#96	CP-221211	4330	_	F	Correction on 5GMM deregistration state for CC #62 and #79	17.7.0
2022-06	CT#96	CP-221211	1338	_	F	Correction on Extended rejected NSSALIE	17.7.0
2022-00	CT#06	CP-221211	4300	_	F	Coordination between 5GMM and EMM state	17.7.0
2022-00	CT#90	CP 221211	4341	-		Perform deregistration precedure in 5CMM_PECISTEPED	17.7.0
2022-00	C1#90	GF-221211	4343	-	F	state	17.7.0
2022-06	CT#96	CP-221211	4352	-	F	Clarification on CPSR procedure and minor correction	17.7.0
2022-06	CT#96	CP-221211				Correction on session-AMBR during the PDU session	17.7.0
			4353	-	F	establishment	
2022-06	CT#96	CP-221211	4356	-	F	Storing Allowed NSSAIs for EPLMNs during registration	17.7.0
2022-06	CT#96	CP-221211	4371	-	D	Wording correction for the UE policy classmark	17.7.0
2022-06	CT#96	CP-221211			_	N1 NAS signalling Connection maintenance for abnormal	17.7.0
2022 00	01//00	01 221211	4377	-	F	cases and PLMN selection	
2022-06	CT#96	CP-221211	_			Start T3540 when non-switch-off de-registration procedure	17.7.0
	0	0	4378	-	F	complete	
2022-06	CT#96	CP-221211	4379	-	F	Clarification for Semantic error in the mapped EPS bearer	17.7.0
2022-06	CT#96	CP-221211	107.0			Clarification of LIE initiated PDU procedure and NAS signalling	17.7.0
2022 00	01#00	01 221211	4390	_	F	connection release	17.7.0
2022-06	CT#96	CP-221211	4000		•	Correction of duplicated info in the Generic LIE Configuration	1770
2022 00	01//00	01 221211	4259	1	F		17.17.0
2022-06	CT#96	CP-221212	1200	•		Correction of the condition that the LIE removes the pending	1770
2022 00	01#00	01 221212	4251	1	F	NSSAI	17.7.0
2022-06	CT#96	CP-221212	4294	1	F	Clarification on the undate of allowed NSSAI	1770
2022-06	CT#96	CP-221212	4300	1	F	Note on the default configured NSSAI	17.7.0
2022-00	CT#90	CD 221212	4202	1			17.7.0
2022-00	CT#90	CF-221212	4302	1	Г		17.7.0
2022-06	CT#96	CP-221212	4320	1			17.7.0
2022-06	CT#96	CP-221212	4321	1	F	Scenarios to stop 13526	17.7.0
2022-06	CT#96	CP-221212	4322	1	F -	QoS error checks for unstructured PDU session type in PCO	17.7.0
2022-06	C1#96	CP-221212	4324	1	F	Missing state when disabling N1 mode for 3GPP access	17.7.0
2022-06	CT#96	CP-221212	1000		-	Perform eCall inactivity procedure in 5GMM-	17.7.0
0000.00	OT "00	00.004040	4329	1	F	REGISTERED.NON-ALLOWED-SERVICE substate	47.7.0
2022-06	CT#96	CP-221212	4004		-	No need to include Uplink data status IE in periodic registration	17.7.0
0000.00	OT // 00	00.004040	4331	1	F		4770
2022-06	CT#96	CP-221212	4344	1	F	Correction on trigger to initiate registration procedure	17.7.0
2022-06	CT#96	CP-221212	4347	1	F	Correction on the IE coding	17.7.0
2022-06	C1#96	00 00 10 10	1000		-		
2022-06		CP-221212	4366	2	F	UE delete NAS security context only when not be used	17.7.0
2022-06	CT#96	CP-221212 CP-221212	4366 4385	2 1	F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI	17.7.0 17.7.0
2022-06	CT#96 CT#96	CP-221212 CP-221212 CP-221212	4366 4385 4414	2 1 1	F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE	17.7.0 17.7.0 17.7.0
2022 00	CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414	2 1 1	F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging	17.7.0 17.7.0 17.7.0 17.7.0
2022 00	CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357	2 1 1 1	F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID	17.7.0 17.7.0 17.7.0 17.7.0
2022-00	CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357	2 1 1 1	F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06	CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357 4358	2 1 1 1 1	F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357 4358	2 1 1 1 1	F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357 4358 4359	2 1 1 1 1 1	F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212	4366 4385 4414 4357 4358 4359 4256	2 1 1 1 1 1 1 1	F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213	4366 4385 4414 4357 4358 4359 4256	2 1 1 1 1 1 1 1	F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284	2 1 1 1 1 1 1 1	F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309	2 1 1 1 1 1 1 1 1 1	F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313	2 1 1 1 1 1 1 1 1 1	F F F F F F F D	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323	2 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F D F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540	17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F D F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4348	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4348	2 1 1 1 1 1 1 1 1 1 1 1 1 2	F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4348 4350	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4345 4345 4345 4350 4351	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4345 4348 4350 4351	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information Indication for no 5GMM or 5GSM messages to the lower	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4326 4345 4345 4345 4348 4350 4351 4376	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information Indication for no 5GMM or 5GSM messages to the lower layers	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4326 4345 4345 4345 4348 4350 4351 4376	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information Indication for no 5GMM or 5GSM messages to the lower layers Handling of multiple TAGs in the Ethernet header for signalled	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4345 4348 4350 4351 4376 4380	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information Indication for no 5GMM or 5GSM messages to the lower layers Handling of multiple TAGs in the Ethernet header for signalled and derived QoS rules	17.7.0 17.7.0
2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06 2022-06	CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96 CT#96	CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221212 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213 CP-221213	4366 4385 4414 4357 4358 4359 4256 4284 4309 4313 4323 4326 4345 4345 4345 4345 4350 4351 4376 4380	2 1 1 1 1 1 1 1 1 1 1 1 1 1	F F F F F F F F F F F F F F	UE delete NAS security context only when not be used Use of definition default S-NSSAI Length information correction of 5GS update type IE Abnormal cases in Registration procedure for handling Paging subgroup ID Abnormal cases in Generic UE configuration update procedure for handling Paging subgroup ID Deleting PEIPS assistance information on Registration procedure failure Mismatch of the Legth Indicators between two similar IEs Correction on UE handling on extended local emergency numbers list via non-3GPP access Network slicing features applicable in SNPN Editorial corrections Correction on using T3540 Clarification on UE action for not forwarded 5GSM message Correction on access category about MO IMS registration related signalling Support MAC address range in packet filter The handling of establishing an emergency PDU session after WUS negotiation in 5GS Correction on the WUS assistance information Indication for no 5GMM or 5GSM messages to the lower layers Handling of multiple TAGs in the Ethernet header for signalled and derived QoS rules Correction to TFT check for PDU session establishment	17.7.0 17.7.0

| 2022-06 | CT#96 | CP-221213 | |

 | | Correction to TFT check for PDU session modification
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
|---|---|---|--
--
--
---|---|---|--|--|---|---|--|---|---
---|--|--|---|---|--
--
---|---|---|--|--|---|---|--|---
---|---|--|--|---|---|--|---|---|---|---|--|---|---|--
---|---|---|---|
| | | | 4389 | 1

 | F | procedure
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221213 | 4391 | 2

 | F | Clarification of Release of non-emergency PDU sessions
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221213 | 4392 | 1

 | F | Clarification of UE configuration parameter updates
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221213 | 40.00 |

 | - | Clarification of handlings of 5GMM cause #65 maximum
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 0000.00 | 07/00 | 00.004040 | 4393 | 1

 | | number of PDU sessions reached for SNPN
 | 4770 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221213 | 4394 | 1

 | | Storage of NSSAI
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221213 | 4395 | 1

 | D | Editorial corrections
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-00 | CT#90 | CP-221213 | 4399 |

 | D | NSSAL mapping during transfer of PDLL session from HPLMN
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-00 | 01#90 | GF-221213 | 3497 | 6

 | F | to VPI MN & VPI MN to HPI MN
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221214 | 4431 | 1

 | B | Support of provisioning FCS configuration info per FCSP
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4163 | 1

 | F | Correction on MBS service area indication
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | | -

 | - | Correction to the Requested MBS container and the Received
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | 4167 | -

 | F | MBS container IEI values
 | - | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | |

 | | Updating the MBS service area of MBS multicast session
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | 4225 | 1

 | F | using MBS Service Announcement
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4226 | 1

 | F | Corrections related to MBS multicast sessions
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | |

 | | Delivering multiple MBS service areas to the UE for Location
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | 4227 | -

 | F | dependent MBS service
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4228 | 1

 | F | Applicability of security protection for MBS session
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4235 | 1

 | F | Deregistration procedure impacts for MBS session
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4000 |

 | _ | MBS backoff timer in PDU SESSION ESTABLISHMENT
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 0000 07 | OT "CC | 00.001016 | 4236 | 1

 | В |
 | 4770 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4270 | 1

 | | Inviaximum number of associated MBS sessions
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4396 | 1

 | Г | Correction to IVIBS Service area indication
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221218 | 4397 | 1

 | D | Minor editorial
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | GP-221218 | 1400 | 1

 | E | Removing the EN related to the maximum lengths of the
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022.06 | CT#06 | CD 221210 | 4406 | 1

 | Г | Received MDS container IE
 | 1770 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-00 | C1#96 | CP-221210 | 4407 | 1

 | F | In the spec
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221219 | 4407 | 1

 | 1 | MS determined PLMN with disaster condition and
 | 1770 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022 00 | 01//00 | 01 221210 | 4141 | 1

 | F | broadcasting disaster related indication
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221219 | 4149 | 1

 | F | Cause code for MINT
 | 1770 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221219 | 1110 |

 | • | Resolution of Editor's note on handling of the indication of
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | |

 | | whether disaster roaming is enabled in the UE and the
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | |

 | | indication of 'applicability of lists of PLMN(s) to be used in
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | |

 | |
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| | | | 4170 | 3

 | F | disaster condition provided by a VPLMN'
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06 | CT#96 | CP-221219 | 4170
4209 | 3
1

 | F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06 | CT#96
CT#96 | CP-221219
CP-221219 | 4170
4209
4223 | 3
1
1

 | F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
 | 17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233 | 3
1
1
2

 | F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
 | 17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233 | 3
1
1
2

 | F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
 | 17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4233
4243 | 3
1
1
2
-

 | F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
 | 17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4243
4265 | 3
1
2
-
1

 | F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4233
4243
4265 | 3
1
2
-
1

 | F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4243
4265
4349 | 3
1
2
-
1
2

 | F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4265
4349 | 3
1
2
-
1
2

 | F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4265
4349
4361 | 3
1
2
-
1
2
-
-

 | F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4265
4349
4361
4384 | 3
1
2
-
1
2
-

 | F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
 | 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4384 | 3
1
2
-
1
2
-
1
-
1

 | F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413 | 3
1
2
-
1
2
-
1
-
1
-
1
-

 | F
F
F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
 | 17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0
17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-2212219
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340 | 3
1
2
-
1
2
-
1
-
1
-
1
1
1
1

 | F
F
F
F
F
F
F
F
F
F
F
F
D | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
 | 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362 | 3
1
2
-
1
2
-
1
-
1
-
1
1
1
1
1
1

 | F
F
F
F
F
F
F
F
F
F
F
D
B | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
 | 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409 | 3
1
2
-
1
2
-
1
-
1
1
1
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
F
D
B
D | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409 | 3
1
2
-
1
2
-
1
-
1
1
1
-
-
1
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
D
B
D | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410 | 3
1
2
-
1
2
-
1
-
1
1
1
-
1
1
1
-
1
1
1
-
1
1
1
-
1
1
1
-
-
1
1
-
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
D
B
D
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
 | 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588 | 3
1
2
-
1
2
-
1
-
1
1
1
-
1
1
-
1
1
-
1
1
1
-
1
1
1
1
1
-
-
1
1
-
-
-
1
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
D
B
D
F
B
B | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower lavers
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221223 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148 | 3
1
2
-
1
2
-
1
-
1
1
-
1
1
-
1
1
-
1
-
1
-
1
-
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
D
B
D
F
B
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148 | 3
1
2
-
1
2
-
1
-
1
1
-
1
1
-
1
1
-
1
-
1
-
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152 | 3
1
2
-
1
2
-
1
-
1
-
1
1
-
1
-
1
-
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165 | 3
1
2
-
1
2
-
1
-
1
-
1
1
-
1
1
-
1
1
-
-
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165
4194 | 3
1
2
-
1
2
-
1
-
1
-
1
1
-
1
1
-
1
-
1
-
1
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165
4194
4195 | 3 1 2 - 1 2 - 1 - 1 1 - 1 1 - 1 - 1 - 3 1 - 3 1 - 1

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
F
F
F
F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165
4194
4195
4272 | 3
1
2
-
1
2
-
1
-
1
-
1
1
-
1
1
-
1
-
1
-
1
-
-
1
-
-
1
-
-
-
1
-
-
-
-
-
-
-
-
-
-
-
-
-

 | F
F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
B
C
B
F
F
B
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165
4194
4195
4272
4305 | 3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 3 1 - 3 1 - 1 - - 1 - - - - - - - - - - - - - -

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
B
F
F
F
F
F
F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4165
4194
4195
4272
4305 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 - 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - <tr td=""> <!--</td--><td>F
F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
B
F
F
B
F
F
C</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU</td><td>17.7.0 17.7.0</td></tr> <tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4233
4243
4265
4349
4361
4384
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4195
4195
4272
4305</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1</td><td>F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
F
F
F
F
F
F
F
F
F
F
F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures</td><td>17.7.0 17.7.0</td></tr>
<tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
10-1</td><td>3 1 2 - 1 2 - 1 2 - 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - - 1 - 1 - - - - - - - <tr td=""> <!--</td--><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN</td><td>17.7.0 17.7.0</td></tr><tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372</td><td>3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location</td><td>17.7.0 17.7.0</td></tr><tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F
 F F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting #78</td><td>17.7.0 17.7.0</td></tr><tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<></td></tr></td></tr> | F
F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
B
F
F
B
F
F
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster
condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU | 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4195
4195
4272
4305 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
F
F
F
F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures | 17.7.0 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
10-1 | 3 1 2 - 1 2 - 1 2 - 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - - 1 - 1 - - - - - - - <tr td=""> <!--</td--><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN</td><td>17.7.0 17.7.0</td></tr>
<tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372</td><td>3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location</td><td>17.7.0 17.7.0</td></tr> <tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting #78</td><td>17.7.0 17.7.0</td></tr> <tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F F F F F F F F F F F F F F F F F
 F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<></td></tr> | F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN | 17.7.0 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372 | 3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 | F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT
support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location | 17.7.0 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 | F F F F F F F B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting #78 | 17.7.0 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1 | F F F F F F F B D F B C B F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<> | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not
transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729 | 17.7.0 17.7.0 |
| F
F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
B
F
F
B
F
F
C | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU | 17.7.0 17.7.0 | |

 | |
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4195
4195
4272
4305 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1

 | F
F
F
F
F
F
F
F
D
B
D
F
B
C
C
B
F
F
F
F
F
F
F
F
F
F
F
F
F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
 | 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
10-1 | 3 1 2 - 1 2 - 1 2 - 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - - 1 - 1 - - - - - - - <tr td=""> <!--</td--><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN</td><td>17.7.0 17.7.0</td></tr> <tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372</td><td>3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F F D B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location</td><td>17.7.0 17.7.0</td></tr>
<tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F</td><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting #78</td><td>17.7.0 17.7.0</td></tr> <tr><td>2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06</td><td>CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96</td><td>CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224</td><td>4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404</td><td>3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1</td><td>F F F F F F F B D F B C B F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<></td></tr>
 | F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
 | 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372 | 3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 | F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location | 17.7.0 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1

 | F F F F F F F B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting #78 | 17.7.0 | 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1 | F F F F F F F B D F B C B F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<> | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729 | 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0 17.7.0
 17.7.0 | | | | | | | | | | | | | | |
 | |
| F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN | 17.7.0 | |

 | |
 | | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4233
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372 | 3 1 2 - 1 2 - 1 - 1 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1

 | F F F F F F F F D B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction on the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
 | 17.7.0 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1

 | F F F F F F F B D F B C B F | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarification on DREI.
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the applicability of 5GMM cause value #78
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service intiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of EMM parameters on getting
#78 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |
| 2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06
2022-06 | CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96
CT#96 | CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221219
CP-221223
CP-221223
CP-221223
CP-221223
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224
CP-221224 | 4170
4209
4223
4243
4243
4265
4349
4361
4384
4411
4413
4340
4362
4409
4410
3588
4148
4152
4105
4194
4195
4272
4305
4306
4372
4306
4372
4401
4404 | 3 1 2 - 1 2 - 1 1 1 1 1 1 1 1 1 1 3 1 - 1 - 1 - 1 1 1 1 1 1 1

 | F F F F F F F B D F B C B F <td< td=""><td>disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release request not forwarded due to
#729</td><td>17.7.0 17.7.0</td></td<> | disaster condition provided by a VPLMN'
Trigger of UE-initiated de-registration procedure
Clarify that S1 mode is not supported for MINT
Correct on List of PLMNs to be used in disaster condition list
IEI
Editor's notes in subclause 5.4.4.1 and subclause 5.4.4.2
Emergency PDU session while the timer for disaster roaming
wait range is running
Non-emergency PDU sessions are not transferable to EPS
during disaster roaming
Storage of List of PLMNs to be used in disaster condition to
NVM
Resolution of editors note for registration type
Handling of EMM parameters on getting #80
Correction on CAG information list format
SDT support
Aligning the terminologies of signalling messages
Clarification for the encoding of MCC and MNC parameters in
TS 24.501
Multiple TACs from the lower layers
Removal of the indication of the country of UE location
Update the description on the lists of 5GS forbidden tracking
areas
Definition of last visited registered TAI for 5GSat
Correction in the applicability of 5GMM cause value #78
Correction in the note on GNSS fix time
Addition of lower bound IEs for #78, alt 2
Correction in the AMF operation to determine forbidden TAIs
Forbidden TAIs delivered to a UE during a successful MRU
and SR procedures
Clarification on emergency service initiation via the PLMN
which is not allowed to operate at the present UE location
Handling of EMM parameters on getting #78
Handling of PDU session release
request not forwarded due to
#729 | 17.7.0 | | | | | | | | | | | | |

 | | | | | | | | | | |
 | | | | | | | | | | | | | | | |
 |

2022-06	CT#96	CP-221224	3976	7	С	Forbidden TAI handling in case of multiple TACs	17.7.0
2022-06	CT#96	CP-221227	4433	1	В	Support of MC slicing configuration as part of UE local configuration	17.7.0
2022-06	CT#96	CP-221241	4231	_	F	The remote UE report procedure is initiated by a 5G ProSe	17.7.0
2022-06	CT#96	CP-221242	4201	1	E	UE-requested PDU session establishment procedure based on	17.7.0
2022-06	CT#96	CP-221245	4221	1	C	Considering eDRX parameter in the LISIM	1770
2022-00	CT#96	CP-221249	4150	1	0	UE required to not accent URSP signalled by non-subscribed	17.7.0
2022 00	01//00	01 221240	4135	1	В	SNPNs	17.17.0
2022-06	CT#96	CP-221249	4137	1	F	Editor's note in subclause 5.5.1.3.4	17.7.0
2022-06	CT#96	CP-221249			-	Access identities when UE accesses SNPN using PLMN	17.7.0
			4138	-	F	subscription	
2022-06	CT#96	CP-221249	4171	1	F	Signalling UE support for SOR-SNPN-SI in SOR ACK	17.7.0
2022-06	CT#96	CP-221249	4186	1	F	Support of mapped S-NSSAI in SNPN	17.7.0
2022-06	CT#96	CP-221249	4196	-	F	ON-SNPN: Correction in the operation of a UE entering the 5GMM-DEREGISTERED.PLMN-SEARCH state	17.7.0
2022-06	CT#96	CP-221249				NSSAA performed for a UE operating in SNPN access	17.7.0
			4202	-1	F	operation mode	
2022-06	CT#96	CP-221249	4203	2	F	Clarification of ProSe support in NPN	17.7.0
2022-06	CT#96	CP-221249	4208	1	F	URSP rules for SNPN	17.7.0
2022-06	CT#96	CP-221249	4213	-	F	Correction of definition given in TS 23.501 about GIN	17.7.0
2022-06	CT#96	CP-221256	4145	1	F	UCU for MPS	17.7.0
2022-06	CT#96	CP-221257				Access category assignment for an access attempt occurred	17.7.0
			3927	4	В	due to call pull	
2022-06	CT#96	CP-221258			_	Clarification on lists of 5GS forbidden tracking areas over non-	17.7.0
			4218	1	F	3GPP access	
2022-06	CT#96				_	Some IEI values assigned by the rapporteur of the TS	17.7.1
2022-09	CT#97e	CP-222137	4442	1	F	Correction for default UE credentials	17.8.0
2022-09	CT#97e	CP-222137	4444	-	F	PVS addresses for NSSAA not associated with DNN	17.8.0
2022-09	CT#97e	CP-222137	4445	1	F	Precedence between PVS addresses or PVS names	17.8.0
2022-09	CT#97e	CP-222137	4446	1	F	Alignment with SA3 on 5G AKA and EAP-AKA' based primary authentication and key agreement procedure used for onboarding services in SNPN	17.8.0
2022-09	CT#97e	CP-222137			-	Put the NOTE about network slice used for onboarding under	17.8.0
			4548	1	F	correponding bullet	
2022-09	CT#97e	CP-222137	4549	1	F	PVS information in SMF	17.8.0
2022-09	CT#97e	CP-222137	4557	-	F	HPLMN S-NSSAI in case of SNPN	17.8.0
2022-09	CT#97e	CP-222137	4620	1	F	Resolving EN on KSEAF derivation indicator in USIM	17.8.0
2022-09	CT#97e	CP-222137				Storage and access of 5GMM parameters mapping with SUPI	17.8.0
			4636	1	F	from USIM for AKA based SNPN	
2022-09	CT#97e	CP-222139	4538	1	F	The RAN paging handling for MUSIM UE in 5GS	17.8.0
2022-09	CT#97e	CP-222139				Harmonization for the paging restriction terminology for	17.8.0
			4613	-	F	MUSIM UE	
2022-09	CT#97e	CP-222139	4619	1	F	Rejection of paging correction	17.8.0
2022-09	CT#97e	CP-222140			_	Additional parameter with generic UE configuration update	17.8.0
			4495	1	F -	procedure	
2022-09	C1#97e	CP-222140	4499	1	F -	NSSRG and allowed NSSAI for the other access	17.8.0
2022-09	CT#97e	CP-222140	4525	1	F	Provide new NSSRG information to UE	17.8.0
2022-09	C1#97e	CP-222140	4566	2	F	Associate NSSRG values with HPLMN S-NSSAI	17.8.0
2022-09	C1#97e	CP-222140	4596	1	F	Correction on the rejected NSSAI due to maximum number of UEs reached	17.8.0
2022-09	C1#97e	CP-222143	4591	-	F -	Service-level-AA timer name correction	17.8.0
2022-09	CT#9/e	CP-222143	4598	1		Correction on Service-level-AA container IEI	17.8.0
2022-09	CT#9/e	CP-222144	4456	-		Add ProSeP request in UAC	17.8.0
2022-09	CT#9/e	CP-222144	4458	-			17.8.0
2022-09	CT#9/e	CP-222145	4448	1		Kernote UE IP INTO OF KEMUTE UE REPORT for IPv4	17.8.0
2022-09	CT#976	CP-222145	4455	1		Date not applied to L2 relay	17.8.0
2022-09	CT#976	CP-222145	4457	1		Clarification on the explanation for U2N relay	17.8.0
2022-09		CP-222145	4487	1	Г С	Correction on the expiry of 13586 timer	17.8.0
2022-09	CT#976	CP-222145	4314	1	r	Denviding LIDI MNUD together with DDUK ID is 04 bit states	17.0.0
2022-09	CT#97e	CP-222146	4553	1	F	format	17.8.0
2022-09	01#310	51 -222 140	4614	1	F	Remote UF report procedure	11.0.0
2022-00	CT#07≏	CP-222146	4615	1	F	Some corrections related to the Relay Key Request procedure	1780
2022-00	CT#97e	CP-222146		-	·	Introducing the 5GPRUK ID in the Relay key Request	17.8.0
-0-2 00	011010	51 222140	4616	1	В	procedure	
2022-09	CT#97e	CP-222146	4617	1	F	ProSe relay transaction identity as a type 3 IE	17.8.0
2022-09	CT#97e	CP-222146				Setting RRC establishment cause value when relav UE has its	17.8.0
			4639	2	F	own service	
2022-09	CT#97e	CP-222147				New 5QI values to support Advance Interactive Services (AIS)	17.8.0
	-		4501	1	С	in 5G	

2022-09	CT#97e	CP-222147	4542	1	F	Correction on serving PLMN rate control	1780
2022.00	CT#070	CP 222148	1192	2	C	Corroction to ECS Addross Provisioning	17.9.0
2022-09	CT#976	CF-222140	4402	2	-		17.0.0
2022-09	C1#97e	CP-222148	4492	-	F	Clarification on indicating the EDC support to network	17.8.0
2022-09	CT#97e	CP-222151				Handling of the MBS multicast session on local release of PDU	17.8.0
			4491	1	F	session	
2022-09	CT#97e	CP-222151	4568	1	F	Providing TMGI to lower layer for paging	17.8.0
2022-09	CT#97e	CP-222151				MBS session maintenance when releasing user-plane	1780
2022 00	01//070	01 222101	1592	1	F	resources of a MA RDU session	17.0.0
0000.00	07/07-	00.000454	4502	1	Г Г		47.0.0
2022-09	C1#97e	CP-222151	4583	1	F	MBS back-off timer for IP address	17.8.0
2022-09	CT#97e	CP-222151				Clarification on the determination of outside the MBS service	17.8.0
			4584	1	F	area	
2022-09	CT#97e	CP-222151	4585	1	F	PDU session status IE handling for MBS session	17.8.0
2022-00	CT#070	CP-222151	1586	-	F	MBS session maintenance after handover	17.8.0
2022-03	OT#976	00 000454	4500	-	- -		17.0.0
2022-09	C1#97e	CP-222151	4587	1	F	Request to join MBS session during establishment procedure	17.8.0
2022-09	CT#97e	CP-222151	4607	-	F	Delivering list of keys in MBS Security container	17.8.0
2022-09	CT#97e	CP-222151				Correction for the condition of including the Security container	17.8.0
			4609	-	F	in the Received MBS container IE	
2022.00	CT#070	CD 222151	4610	1		MBS Security keys undate to the LIE	17 9 0
2022-09	CT#976	CF-222131	4010	1	-		17.0.0
2022-09	C1#97e	CP-222151	4621	1	F	Correction to timers of multicast/broadcast services	17.8.0
2022-09	CT#97e	CP-222152	4440	-	F	Errors in PLMN ID IE	17.8.0
2022-09	CT#97e	CP-222155				Addition of the length value of the Negotiated eDRX	17.8.0
			4641	-	F	parameters IE	
2022-00	CT#070	CP-222156	1531	1	F	Add satellite E-LITRAN as an LIE supported access technology	1780
2022-03	CT#976	CD 222150	4004	-	•	Add satellite E-0 TRAN as all OE supported access technology	17.0.0
2022-09	C1#97e	CP-222158			_	Clarification that the NSAG information can not be sent with a	17.8.0
			4450	1	F	request to perform the registration procedure	
2022-09	CT#97e	CP-222158				Clarification that the NSAG information is sent over 3GPP	17.8.0
			4452	1	F	aceess only	
2022-09	CT#97e	CP-222158				Indicating the deletion or invalidation of the NSAG information	17.8.0
2022 00	01#010	01 222100	1171	1	F	to lower lavers	11.0.0
0000.00	07/07-	00.000450	4471	1	- -		47.0.0
2022-09	C1#97e	CP-222158	4483	1	F	Support NSAG for SNPN	17.8.0
2022-09	CT#97e	CP-222158	4484	1	F	NSAG priority	17.8.0
2022-09	CT#97e	CP-222158	4560	1	F	IEI assignment for the NSAG information IE	17.8.0
2022-09	CT#97e	CP-222159	4469	1	F	The consideration of avoiding unnecessary TAL change	1780
2022 00	CT#070	CP 222150	1100	1		Extended NAS timers based on satellite NG PAN PAT type	17.8.0
2022-09	01#976	CF-222159	4475	1	Г Г	Extended thas timers based on satellite the RAN RAT type	17.0.0
2022-09	C1#97e	CP-222159	4476	1	F	Alignment of terminology in current TAI definition	17.8.0
2022-09	CT#97e	CP-222159	4533	-	F	Suggest simplifying the selection for the current TAI	17.8.0
2022-09	CT#97e	CP-222159	4535	1	F	Update of conditions for deleting entries in # 78 list	17.8.0
2022-09	CT#97e	CP-222159	4556	1	F	Current TAI	1780
2022 00	CT#070	CP 222150	4624	2		Clarification on handling related to #78	17.8.0
2022-09	01#976	CF-222159	4024	2	Г —		17.6.0
2022-09	C1#97e	CP-222159	4625	2	F	Clarification on timer instance associated with the entry	17.8.0
2022-09	CT#97e	CP-222163	4498	1	F	MPS exemption in Attempting to reRegister	17.8.0
2022-09	CT#97e	CP-222163	4518	1	F	MPS exemption in Attempting to Register	17.8.0
2022-00	CT#070	CP-222168	1526	1	F	Condition of returning REGISTRATION COMPLETE by LIE	1780
2022-03	CT#976	CD 222100	4320		-		17.0.0
2022-12	CT#98e	CP-223121	4197	0	F	Handling of pending NSSALIN NSSRG proceduret	17.9.0
2022-12	C1#98e	CP-223136	4439	12	в	NAS operation for network slice-based random access	17.9.0
2022-12	CT#98e	CP-223122	4441	4	F	Editor's note in subclause 5.3.2	17.9.0
2022-12	CT#98e	CP-223122	4443	3	F	Editor's note in 6.4.1.2	17.9.0
2022-12	CT#080	CP-223116	4404	6	C	Manned S-NSSAI when LIF is roaming	17 9 0
2022-12	OT#308	01-223110	4494	4			17.00
2022-12	01#986	CP-223136	4051	<u>n</u>	Г _	Correction to the INSAG priority field reference	17.9.0
2022-12	CT#98e	CP-223116	4653	1	F	Correction to PDU session types	17.9.0
2022-12	CT#98e	CP-223116				Correction to the CONFIGURATION UPDATE COMMAND	17.9.0
			4657	2	F	message	
2022-12	CT#98e	CP-223136	4665	1	F	Correction to NSAG default area	17.9.0
2022 12	CT#092	CP_222122		<u> </u>		Correction to name of List of DLMNs offering disaster reaming	17.0.0
2022-12	C1#90e	01-223133	4070		-	correction to name or List or FLIVINS oriening disaster roaming	17.9.0
L			4670	1		Services	
2022-12	CT#98e	CP-223136	4678	1	F	Maximum and minimum length of NSAG information IE	17.9.0
2022-12	CT#98e	CP-223136	4680	1	F	Maximum number of TAI list restriction for NSAG	17.9.0
2022-12	CT#98e	CP-223136	4682	1	F	NSAG priority handling at the AMF	17.9.0
2022 12	CT#092	CP_222126	1691	1		Correction on NSAG information handling	17.0.0
2022-12	01#908	00-223130	4004		Г		17.9.0
2022-12	C1#98e	CP-223148	l	l .	-	Corrections on UE-initiated authentication and key agreement	17.9.0
			4703	1	F	procedure	
2022-12	CT#98e	CP-223121				Registration update triggered by NSSRG update in UCU	17.9.0
			4705	1	F	procedure	
2022.12	CT#090	CP-222126	4711	1		NSAG information provision over 3CPP access only	17 0 0
2022-12	OT#308	CD 000440	4/11			Initiation of outboation and have arrest and the f	17.3.0
2022-12	C1#986	68-223148	4-14		-	initiation of authentication and key agreement procedure for	17.9.0
			4718	1	F	5G ProSe U2N relay UE in NORMAL-SERVICE state	
2022-12	CT#98e	CP-223127	4720		F	Correction on USS FQDN	17.9.0
2022-12	CT#98e	CP-223127	4722	1	F	Clarification on authorization of UAV flight	17.9.0
2022-12	CT#080	CP-223116	4727	· ·	F	Correction on single-registration subclause	17 9 0
12022-12			17141	1	1	concertor on angle registration autorause	17.3.0
0000 40	CT#00	CD 000447	4700	A	L	Alignment of terminal and the second second ADO	1700
2022-12	CT#98e	CP-223117	4733	1	F	Alignment of terminology on multicast MBS session	17.9.0

2022-12	CT#98e	CP-223117	4737		F	Handling of the MBS multicast sessions when the PDU	17.9.0
2022-12	CT#98e	CP-223148	4740			Rename 5GPRUK ID and 5GPRUK in CP based solution and	17.9.0
0000.40	07/00	05 000 107	4742	1	- F	rename PRUK and PRUK ID in UP based solution	17.0.0
2022-12	CT#98e	CP-223127	4750	1		Correction on unused value of payload type	17.9.0
2022-12	CT#96e	CP-223116	4700	1	г г	Addition of DE requested 13512 value at MICO	17.9.0
2022-12	CT#98e	CP-223117	4767	1	F	MBS address information type in the received MBS information	17.9.0
2022-12	CT#98e	CP-223148	4775	1		Including TCP/UDP port ranges in REMOTE DE REPORT	17.9.0
2022-12	CT#98e	CP-223121	4780		Г	Remove redundant content about NSSRG information	17.9.0
2022-12	C1#98e	CP-223148	1707	1	F	Mandatory Inclusion of 5GPRUK ID in the RELAY KEY	17.9.0
2022 12	CT#090	CD 222149	4707	1		Correcting the reference for the DDTI	17.0.0
2022-12	CT#98e	CP 223140	4709	1		Indicate to lower layer to delete stored TMCI	17.9.0
2022-12	CT#90e	CP-223117	4794	I		Clarification on providing SOP CMCL in SNPN access	17.9.0
2022-12	01#906	GF-223122	4800	2	F	operation mode	17.9.0
2022-12	CT#98e	CP-223135	4803	1	F	Paging rejection in RRC inactive	1790
2022-12	CT#98e	CP-223116	4828		F	Correction of implementation error of CR4124	17.0.0
2022-12	CT#98e	CP-223148	4830		F	Correction of implementation error of CR4615	17.0.0
2022-12	CT#08o	CP-223133	4030		1	Allowed access attempts while timer precluding registration is	17.9.0
2022-12	01#306	01-220100	4832		F	running in 24 501	17.3.0
2022-12	CT#98e	CP-223116	4002			Lost scope of AME requirements for allowed NSSAL in NR-N1	1790
2022 12	01#000	01 220110	4843	1	F	mode	17.0.0
2022-12	CT#98e	CP-223115	4853	3	F	AME behaviour on Forbidden TAIs list IEs	17.9.0
2022-12	CT#98e	CP-223115	4855	1	F	UE behaviour on Forbidden TAIs list IEs	17.9.0
2022-12	CT#98e	CP-223116				Registration procedure triggered by a change of UF	17.9.0
LOLL IL	01#000	01 220110	4863	1	F	Requested T3512	11.0.0
2022-12	CT#98e	CP-223116				Correction on conditions for using SPI for UE derived QoS	17.9.0
			4871	1	F	rules	
2022-12	CT#98e	CP-223136	4881	1	F	Maximum length of NSAG information IE	17.9.0
2022-12	CT#98e	CP-223136	4883	2	F	TAI lists restriction for NSAG information	17.9.0
2022-12	CT#98e	CP-223127				Procedure type for service-level authentication and	17.9.0
-			4885		F	authorization procedure	
2022-12	CT#98e	CP-223143				Modify network handling of PDU sessions for emergency	17.9.0
			4888	1	F	request in abnormal case	
2022-12	CT#98e	CP-223127				Clarification on payload and payload type for UUAA and C2	17.9.0
			4903	2	F	authorization	
2022-12	CT#98e	CP-223115				Correction in the forbidden TAI lists in NAS messages over	17.9.0
			4919	3	F	satellite access	
2022-12	CT#98e	CP-223117	4942	2	F	Multicast MBS session join or leave for local multicast service	17.9.0
2022-12	CT#98e	CP-223121	4950	1	F	Maximum length of NSSRG information IE	17.9.0
2022-12	CT#98e	CP-223115	4970		F	Correction of IEI values	17.9.0
2022-12	CT#98e	CP-223278	4972	3	F	Notification of change of aerial service availability to the UE	17.9.0
2023-03	CT#99	CP-230214	5032	-	F	Fix encoding of 5QI 87	17.10.0
2023-03	CT#99	CP-230214				Correction of incomplete implementation of CR4380 (C1-	17.10.0
			5089	-	F	224129)	
2023-03	CT#99	CP-230232			_	UE and AMF comply with NSSRG restriction across different	17.10.0
			5077	1	F	access types over the same PLMN and different PLMNs	
2023-03	CT#99	CP-230271	5124	1	F	Correction for SNN related to 5G NSWO	17.10.0
2023-03	CT#99	CP-230307			_	Clarification on NSAG information validity when TAI list is	17.10.0
	07/00	05.000044	5072	1	-	absent	
2023-03	CT#99	CP-230214	4992	1	F	5GMM IEI Duplicated in Registration Accept	17.10.0
2023-03	C1#99	CP-230312	E407	4	-	Impact on network and relay UE behaviour when CP-PRUK is	17.10.0
0000.00	07/00	00.00000	5127		Г	not round	47 40 0
2023-03	C1#99	CP-230236	E104	2	г	Rejecting PDU session for C2 communication when UAS	17.10.0
2022.02	CT#00	CD 220207	5154	3 2	Г Г	Service is not allowed	17 10 0
2023-03	CT#99	CP-230307	5056	2	г г	Undete MDS indicator in CLIC measure D47	17.10.0
2023-03	CT#99	CF-230214	5100	3 2	г г	Opuale MPS indicator in COC message RT7	17.10.0
2023-03	CT#99	CP-230307	5163	3		Correction of misimplementation of CR 4663, 4005, and 4676	17.10.0
2023-03	CT#99	CF-230224	5054	-	F	Correction of Implementation errors and misimplementation	17.10.0
2023-03	CT#100	CD 221226				Contection of formalling errors and misimplementation	17.10.1
2023-00	C1#100	CF-231230	5/11	1	_	container alternative 2	17.11.0
2023.06	CT#100	CP-231206	5/02	1	Δ	Correction to the CloT small data containor IE	17 11 0
2020-00	CT#100	CP_221200	5/06	1	Δ	Correction to the MA PDU session information IE	17 11 0
2020-00	CT#100	CP_221246	5362	1		Correction to the Service-level A server address IE	17 11 0
2020-00	CT#100	$C1_{2}^{2}$	5160		F	Correction to the Service-level-AA server address IE	17 11 0
2023-00	CT#100	CP_231227	5460	-	F	ATSSS capabilities of MA PDI Lession	17 11 0
2020-00	CT#100	CD_221215	5409	1	Г	Sending the registration complete message after receiving the	17 11 0
2020-00	01#100	01-201210	5435	4	F	NSAG information	17.11.0
2023-06	CT#100		0.00	т		Clashes resolved	17 11 1
2023-00	CT#101	CP-232231	5540	_	F	Correction to the Service-level-AA service status indication IF	17 12 0
2023-00	CT#101	CP-232230	5535	1	F	Correction to the clause on the ECS address IF	17 12 0
2023-00	CT#101	CP-232200	5626	2	Δ	Handling of un-authorized IAR LIFs	17 12 0
	00	<u> </u>	0020	-			

ETSI TS 124 501 V17.16.0 (2024-10)

2023-12	CT#102	CP-233140				Allowed NSSAI includes S-NSSAI(s) not contained in	17.13.0
			5767	1	F	requested NSSAI	
2023-12	CT#102	CP-233140	5769	2	F	Update configured NSSAI if slice subscription changed	17.13.0
2023-12	CT#102	CP-233140	5914		F	Add missing posSibType to ciphering key data Rel-17	17.13.0
2023-12	CT#102	CP-233140	5869	1	F	Correction of definition and handling of HPLMN S-NSSAIs	17.13.0
2023-12	CT#102	CP-233179	5874	2	F	Correction on NSAG priority	17.13.0
2024-03	CT#103	CP-240109	6060	1	F	Correction on the minimum length of the SLAC	17.14.0
2024-06	CT#104	CP-241197	6328	1	F	Custom throttling to temporary failed 5GSM procedure	17.15.0
2024-09	CT#105	CP-24216	6329	-	F	Adding inclusion criteria and correcting information element	17.16.0

	Document history						
V1761	July 2022	Publication					
V17.7.1	July 2022	Dublication					
V1/./.1	July 2022						
V17.8.0	October 2022	Publication					
V17.9.0	January 2023	Publication					
V17.10.1	April 2023	Publication					
V17.11.0	July 2023	Publication (withdrawn)					
V17.11.1	July 2023	Publication					
V17.12.0	October 2023	Publication					
V17.13.0	January 2024	Publication					
V17.14.0	April 2024	Publication					
V17.15.0	July 2024	Publication					
V17.16.0	October 2024	Publication					

History